Niranjan Bhattacharya Phillip G. Stubblefield *Editors*

Human Fetal Growth and Development

First and Second Trimesters



Human Fetal Growth and Development

Niranjan Bhattacharya Phillip G. Stubblefield Editors

Human Fetal Growth and Development

First and Second Trimesters



Editors Niranjan Bhattacharya Calcutta School of Tropical Medicine Kolkata India

Phillip G. Stubblefield Boston University Jamaica Plain Massachusetts USA

ISBN 978-3-319-14873-1 ISBN 978-3-319-14874-8 (eBook) DOI 10.1007/978-3-319-14874-8

Library of Congress Control Number: 2016938227

© Springer International Publishing Switzerland 2016

This work is subject to copyright. All rights are reserved by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

The publisher, the authors and the editors are safe to assume that the advice and information in this book are believed to be true and accurate at the date of publication. Neither the publisher nor the authors or the editors give a warranty, express or implied, with respect to the material contained herein or for any errors or omissions that may have been made.

Printed on acid-free paper

This Springer imprint is published by Springer Nature The registered company is Springer International Publishing AG Switzerland The book is dedicated to two scientists working on two sides of the globe. They are both no more but their work lives on. They were not connected to each other, had never met each other and worked in very different fields within the broad parameters of fetal growth, abortion, and continuation of pregnancy; yet they are connected by their honest and sincere dedication to scientific research.

Prof. A E Beer of the University of Michigan, USA, is the father of the "Beer Baby." He was one of the best known Reproductive Immunologist in the world during his lifetime. From 1977 till his death he had regular correspondence with Dr. Niranjan Bhattacharya, then a young scientist, a correspondence that spanned thousands of questions raised by Dr. Bhattacharya and answered patiently and in great depth by Dr. Beer. Once Dr. Bhattacharya went to meet him in 1986 at Ann Abor and after a talk of over two hours felt that Dr. Beer was not his usual self. He queried him about this, and Dr. Beer's answer reflects the man he was and his dedication to science: "I am sorry but my brother is now undergoing corrective operation for the dissecting aneurism of the aorta, but I had given you an appointment…"

Prof. Chameli Ganguly was a Physiologist-turned-Biochemist. She started her career as a Junior Research Fellow under Prof. Kanailal Mukherjee, a legendary biochemist who worked in the Institute of Post Graduate Medical Research, Calcutta, India. She was married to science. She considered any manipulation of scientific data as a rape of science and would often repeat an experiment 20 times with immense patience if the need arose. She is instrumental in supporting this book from the conceptual phase because she believed that it was important to globally disseminate the findings of her and others' work in this very challenging field of fetal growth and development, especially to understand why the fetus survives or why it is sometimes expelled.

The editors dedicate this book to Dr. Beer and Dr. Ganguly. May they motivate many more to follow in their footsteps.

> Prof. Niranjan Bhattacharya, Kolkata, India Prof. Phillip G. Stubblefield, Boston, USA

Preface

Omh Ayamaramva sybhya bhabatu (Sanskrit) Let this effort be for the welfare of all.

I met a little Elfman once, Down where the lilies blow, I asked him why he was so small, and why he did not grow. He slightly frowned and with his eyes He looked at me through and through. "I'm quite as big for me", he said "As you are big for you".

J.K. Bangs

In *Scientific Foundations of Pediatrics* edited by J.A. Devis and J. Dobbing, William Heinemann Medical books Ltd., 1981 (2nd edition).

Poets are visionaries. They can see beyond themselves. He has expressed exactly what I want to say. The growth and development of a human fetus within the mother's womb is conditioned by the genetic makeup and the micro and macro environment provided by the mother.

Back in 1964, Prof. K L Mukherjee undertook a project for the anthropometry of human fetuses. Much information is available about the growth and development of human babies after parturition but because of the inaccessibility of human fetuses in intrauterine life not many studies are available on anthropometry of fetuses. In 1964, when the medical terminations of pregnancy clinics were started, Prof. K L Mukherjee had the opportunity of studying the anthropometry of human fetuses at different periods of gestation.

At that time, ultrasonography was not available in all hospitals. The studies also include the structure and some functional aspects of individual organs. Multiparous mothers undergoing the operation of hysterotomy and ligation at the same time at the Medical Termination of Pregnancy clinic were the subjects of the present investigation. The project was explained to the mothers who signed the consent form for the use of the fetuses. The project was duly cleared through the Ethical Committee of the Institute. The usual modus operandi was as follows:

Anesthetized mothers preoperatively received an injection of ergometrine. An incision was made above symphysis publes. The uteri were cut through usual steps and whole fetuses including the intact amniotic sacs were taken out with the help of fingers and placed on an ice bucket. The fetuses were brought to the laboratory and dissections were started within 45 min. The fetuses were weighed to the nearest milligram. The crown heel and crown rump length were measured to the nearest millimeter with a tape measure. The biparital diameter at the level of the eyebrow were measured with a thread and the length was calculated from the length of the thread.

A total of 1396 fetuses were studied during the period 1964–2007. Some details of anthropometric measurements are available in Langman's Medical Embryology and Morsman's Human Embryology. Baslinsky's book on embryology was a source of many informations. The reasons which prompted us to publish the results were the large number of fetuses which were available to us.

Besides anthropometry, attempts were made to study the histological structure and some of the characteristic biochemical functions of individual organs of the fetuses. The following organs were specially studied: liver, lungs, brain, intestine, kidneys, gonads, thymus, adrenal cortex and skeletal muscle. Since our limitations for the hysterotomy extended up to 20 weeks only, most of our studies were limited to that period of intrauterine life. The gestation periods were calculated from the last menstrual period. Only a few fetuses of older periods were available for studies; all of them were stillborns.

The mothers were all normally nourished women and did not have any sign of under nutrition. It is presumed that all the fetuses were fully nourished for their periods of gestation. None of the mothers had anomalies of placenta. The idea behind the study is to understand the growth of the individual fetus along with its development and maturation of all the organs.

Growth implies both cell multiplication and accumulation of both intraand extracellular material. It seldom involves just cell multiplication and matrix secretion, but simultaneously it causes cell differentiation, pattern formation, and changes in form. In case of normal development, a great range of functions is simultaneously orchestrated to produce a harmonious pattern which is characteristic of normal development. The growth and maturation of an organ at the cellular level comprises of four stages, viz., (1) proliferation, (2) migration, (3) differentiation, and (4) death.

From fertilization, the embryogenesis period continues up to the tenth week of gestation. The fetal period begins at the end of the tenth week of gestation (eighth week of development). Since the precursors of all the major organs are created by this time, the fetal period is described both by organ and by a list of changes by weeks of gestational age. Because the precursors of the organs are formed, the fetus is also not as sensitive to damage from environmental exposure as the embryo. Instead, toxic exposures often cause physiological abnormalities or minor congenital malformation.

There is much variation in the growth of the fetus. When fetal size is less than expected, that condition is known as intrauterine growth restriction (IUGR), also called fetal growth restriction (FGR). Factors affecting fetal growth can be *maternal*, *placental*, or *fetal*. Maternal factors include maternal weight, body mass index, nutritional state, emotional stress, toxin exposure (including tobacco, alcohol, heroin, and other drugs which can also harm the fetus in other ways), and uterine blood flow. Placental factors include size, microstructure (densities and architecture), umbilical blood flow, transporters and binding proteins, nutrient utilization, and nutrient production. Fetal factors include the fetus genome, nutrient production, and hormone output. Also, female fetuses tend to weigh less than males at full term.

Fetal growth is often classified as follows: small for gestational age (SGA), appropriate for gestational age (AGA), and large for gestational age (LGA). SGA can result in low birth weight, although premature birth can also result in low birth weight. Low birth weight increases risk for perinatal mortality (death shortly after birth), asphyxia, hypothermia, polycythemia, hypocalcemia, immune dysfunction, neurologic abnormalities, and other long-term health problems. SGA may be associated with growth delay, or it may instead be associated with absolute stunting of growth.

Kolkata, India Kolkata, India Jamaica Plain, MA, USA Kanailal Mukherjee Niranjan Bhattacharya Phillip G. Stubblefield

Introduction

Research studies on the growth and development of the human fetus, an area of study that has important implications for the development of a human being, and possible diseases that the adult human may face later in life due to problems suffered in utero, are few. This is understandable because while studies on animal models are easier to conduct with fewer ethical restrictions, research on human fetuses carries with it inherent difficulties and complications. However, with the development of new technologies, imaging and other ways of monitoring growth have become easier and safer. Further, medical termination of pregnancy for different purposes and reasons are permitted in several countries, providing the opportunity for study and research after obtaining informed consent from all concerned. The present book is an attempt to bring together knowledge in this very significant field from all over the world under the covers of one book.

The present project, which now sees fruition in a published form, actually started a long time back as an experimental effort to study the growth and development of the human fetus in utero after parturition by a group of researchers, senior and junior, in Kolkata (then known as Calcutta) under the leadership of a senior professor, Dr. K L Mukherjee. As mentioned, due to the inaccessibility of human fetuses in intrauterine life, not many studies are available on the anthropometry of fetuses. In 1964, medical termination of pregnancy clinics were started by Prof. K L Mukherjee at the Institute of Post Graduate Medical Education and Research (IPGMER), a premier research institution situated in the SSKM Hospital, Calcutta. The setting up of the clinic in SSKM Hospital provided the opportunity for studying of anthropometry of the human fetus at different periods of gestation on the aborted fetuses. The Medical Termination of Pregnancy Act was passed in 1971; this specifies, among other things, that any pregnant woman, within the prescribed weeks, can legally request her pregnancy to be terminated in selected government hospitals, equipped for the purpose, without legal harassment, for the purpose of family planning. The study of fetuses became much easier after the passage of this Act, since mothers admitted for hysterotomy and ligation, if they gave informed consent and the Ethical Committee gave clearance, were taken as subjects and their fetuses were collected from the operation theater of government hospitals only, for study and research.

And the ball started rolling. On the 50th year of our continued research, we decided to publish our findings through an international book project on the growth and development of the human fetus up to the second trimester.

The person who had initiated the idea of publishing this book, which would also cover research on the subject done all over the world, Prof. Chameli Ganguly, a sincere researcher who had been part of the team since the beginning, recently passed away. We expedited the entire process, convinced that the fruits of our research should not be lost but shared with future generations working in this and related fields, particularly since there is so little in-depth work in this very important area with its manifold implications. The global giant in medical publication, Springer International, was convinced of the importance and hence this book.

When we began our study in the mid-1960s, the criteria for prematurity was (1) babies born before term, or (2) weighing less than 2500 g at birth. However, the situation has changed over the last 50 years of development of medical science.

Currently the world's smallest premature baby to have survived from birth until hospital discharge was born 15 weeks before its due date (born: September 4, 2004). At birth, the baby weighed just 260 g, or 8.6 oz. She was 9.8 in. long. Her mother suffered from severe pregnancy-specific gestosis. This tiny little girl became clinically normal at 7 years, though she had to undergo laser eye surgery for retinopathy of prematurity (ROP).

This is not an isolated case. The second smallest baby in the USA and the world's third smallest baby to survive long enough to leave the hospital was born at 24 weeks because her mother suffered from similar pregnancy-specific hypertensive disorder. She weighed just 9 ½ oz at birth. Supplemental oxygen was given to her at home to treat bronchopulmonary dysplasia (BPD), and she had to have surgery to repair a PDA and laser eye surgery for ROP. Her brain was free from any bleeding.

Another interesting case is the birth of a boy in Germany, who was delivered in June 2009 at 25 weeks gestation. He weighed 275 g, just over 9 ½ oz. He wears glasses and needed physical therapy during his first year of life, but is otherwise a normal, healthy toddler. Another girl baby was born in 1989, who became the world's sixth smallest surviving preemie. She weighed just 280 g at birth (9.9 oz) at 26.6 weeks gestation. She is only 4 ft 7 in. tall, wears glasses, and has asthma; she has no other long-term effects of her premature birth.

The record for the world's most longest surviving (till now) premature baby belongs to James Elgin Gill, a Canadian male born at just 21 weeks 5 days in 1988. James beat all of the odds, and in 2006 was a healthy teenager heading off to college. Another American baby girl was born at 21 weeks 6 days in October of 2006 via in vitro fertilization; her gestational age can be pinpointed exactly, an impossibility for most infants. Although she needed oxygen at hospital discharge, was anemic, and has mild osteopenia, she is otherwise a normal, healthy little girl. The babies mentioned above show no major developmental delays. However, the outcomes for micropreemies are not always as good.

The present book started with an attempted understanding of a basic issue, what is fetal growth?

This issue has been examined from various angles, starting with discussions on the anthropometric measurements of infants born at various gestational periods which are an indicator of gestational age (Chap. 1), the use of developmental immunology to understand the immunological system of a fetus and explain why a newborn is vulnerable to many infections (Chap. 3), debates regarding the vaccination of the unborn (Chap. 2) and tracing the actual growth of a fetus from the zygote stage through the second semester (Chap. 5), to the search for a safe, effective, mid-trimester abortifacent, which involved the use of antigens to disrupt the growth of the fetus and which has revealed many hitherto unknown secrets about the growth and development of the human fetus up to 20 weeks (Chap. 4).

The book goes on to particulars from this basic issue and delves into the complexities of growth, structural and functional development, carbohydrate metabolism and urea biosynthesis in the human fetus, development of the fetal brain and the role of lipid metabolism in human fetal growth. Chapter 6 suggests that growth of individual organs in a fetus is conditioned by the genetic potential and the environment provided by the mother and a third component, the fetus itself. The importance of maternal circulation in the supply of glucose to the developing fetus is highlighted in Chap. 7, and Chap. 8 studies the appearance of mucopolysaccharides, water, and electrolytes in developing human fetal organs, namely, the liver, lungs, and brain, and finds that in all three the water content is much higher in fetal life than in adult life and that there is slow decrease in the water content with the progression of development. Chapter 9 examines the biosynthesis of urea in the fetal liver up to the second semester: one of the various findings was that very little urea was synthesized by the fetal liver, especially in the earlier period of gestation; the rate increased with progress of gestation but in comparison to the situation in an adult, very little urea was synthesized even in the later periods of the gestation. It was further found that of the two possibilities of utilization of ornithine besides the urea cycle, the proline biosynthesis and polyamine biosynthesis, the former pathway was found to be absent in fetal livers. Chapter 10 raises and tries to answer an interesting question: can the fetal and adultogenic layer be distinguished by any other characteristics? Since the functional characteristics of the two kinds of cells also differ, can we identify any protein or other component present in the fetal layer of cells which is absent in the adult layer? The researchers, from their experiments, found that a set of antigens were found in adult extracts which were not present in fetal extracts. The researchers inferred that either the adult saline extract does not have some common fetal antigens or the differentiation of the adult adrenal gland is associated with the appearance of a host of specific antigens. Chapter 11 traces the growth of the human fetal brain up to 28 weeks of gestation and finds that there is a straight-line relationship between the body weight and brain weight. The weight of the brain of fetus is around 14 % of the body weight throughout the gestation period. The experiments also suggested that in the case of brain, some acid soluble nucleotides are very importantly and significantly involved in myelin synthesis and there may be higher metabolic activity and turnover at earlier periods. Chapter 11 discusses lipid metabolism in human fetal development. The authors note that the fetus requires a substantial amount of lipids throughout its development; the lack of lipids including cholesterol (CHO) affects growth disorders. The two lipoproteins

(LPs) involved in supporting the placental CHO need are low density lipoprotein (LDL) and high density lipoprotein (HDL). Fetal steroid precursors of estrogens regulate the uptake of maternal LPs to promote the placental progesterone synthesis. Both estrogen and progesterone are key determinants in pregnancy maintenance and fetal growth; this is a basic role of fetal and maternal LPs.

The next seven chapters study various implications of genes on fetal development up to the second trimester. Chapter 13 is a study of gene regulatory networks and epigenetic modifications in cell fate decisions during early embryonic development. Cell differentiation takes place as the fertilized totipotent cell progresses along the developmental pathway. The authors discuss gene regulatory networks and epigenetic modifications in early development as regulation of expression of genes is the key to understanding how cell differentiation occurs in early developmental stages. The author also discusses master regulators involved in differentiation of the trophectoderm and the inner cell mass, which are incompletely known and the chromatin modifiers involved in these steps. Further, he also describes the master regulators that control the cell fate decisions that give rise to the epiblast and the primitive endoderm from the inner cell. Chapter 14 is a different kind of chapter; it explores the possibilities of finding out when an embryo reaches "personhood." This has major implications for fetal stem cell research because the generation of human embryonic stem cells sometimes requires the destruction of early human embryos, and ethical questions are raised about stem cell research. Chapter 16, by the same author, is another theoretical exploratory chapter on the embryological basis of virgin birth. Chapter 15, by the same author, discusses dependence of fetal hairs and sebaceous glands on the fetal adrenal cortex and the possible control from epidermal Merkel cells and adrenal medulla. The role of the fetal adrenal cortex is little understood. This chapter reviews the literature and attempts to synthesize the current understanding of the developmental and functional biology of the fetal adrenal cortex. Chapter 7 discusses aneuploidy in human preimplantation embryos. The author notes that the number of abnormal zygotes cannot simply be calculated on the basis of the aneuploidy rates of gametes but may result from abnormal fertilization. Furthermore mitotic abnormalities during cleavage division give rise to mosaic chromosome abnormalities of the embryos. Chapter 17 throws new light on the potential impact of maternal milk consumption during pregnancy on mTORC1-driven fetal growth. The author notes that milk is a highly specialized, complex signaling system developed by mammalian evolution to promote mTORC1-driven postnatal growth but milk is not designed by nature to promote prenatal fetal growth. Continued consumption of cow's milk and dairy products during adolescence, adulthood, and pregnancy is an evolutionarily novel behavior that may have longterm adverse effects on human health. The authors of Chap. 19 present their findings on genetic disease specific human embryonic cell lines. They have detected affected embryos, which were used for derivation of the genetic disease specific hESC lines, which presently contain 87 hESC lines obtained from embryos with single-gene and chromosomal disorders. Further, screening of 137 normal hESC lines for polymorphism in the chemokine receptor 5

(CCR5) CMKBR5 gene resulted in detection of 12 hESC lines with CCR5del32 allele, including one with two copies of the gene, conferring resistance to HIV.

The next section of the book discusses the impact of stem cells on the growth and development of the human fetus. Stem cells are considered the origins of all organisms. In fact, all organisms are formed and developed from a single cell – the zygote, which is the product of oocyte fertilization by the sperm. Through the process of cell division, the totipotent stem cell produces all tissue types within an organism. By means of self-renewal, stem cells are maintained during embryonic development right through to adulthood. The basics of role of stem cells in the development of the human fetus are discussed in Chap. 20. The next chapter discusses how and possibly why a human fetus survives in utero without being rejected by the mother (Chap. 21). This involves the mother's immune system. The authors suggest that during a successful pregnancy, the immunity that may cause rejection of the baby and placenta is shut off in the mother's uterus. Throughout the rest of the mother's body, her immune system is fully functional, allowing her to deal in a proper way with any infections that may come her way. In cases of recurrent miscarriage, this immune camouflage and protection of the embryo is perhaps not initiated in such a way that the uterine local and systemic immune responses of the mother are modulated in favor of a decrease in cell mediated immunity and increase in humoral immunity.

The next few chapters go on to discuss the development of different systems like the immune system, the endocrine system, the hepatic system in the human fetus and its impact on growth and maturation. Chapter 22 focuses on glucose metabolism in the fetus and its relationship with fetal insulin. The regulation of glucose, the main energy source of a fetus, is kept fairly constant by the maintenance of maternal glucose concentration by increasing rates of maternal glucose production and development of relative maternal glucose intolerance and insulin resistance, and secondly, the transfer of maternal glucose to the fetus by the placenta, which is buffered by placental glucose utilization. A third regulating factor is the production of insulin by the developing fetal pancreas, which enhances glucose utilization among the insulin-sensitive tissues (skeletal muscle, liver, heart, adipose tissue) that increase in mass and meet the glucose need during late gestation. Chapter 23 notes that systemic maturation is essential for the survival of a neonate, and focuses on the maturation of the endocrine system which is necessary to generate hormones to support normal development. A bouquet of transition factors and epigenetic events act hand in hand with autocrine, paracrine and endocrine network of hormones and growth factors leading to the evolution of the fetal endocrine system with the hypothalamus-pituitary system providing the pivotal controlling template. However, most of the evidence in this chapter, the author mentions, comes from animal studies. The next chapter (Chap. 24) discusses development of the human immune system in the first and second trimesters. The immune paradox lies in the fact that the semiallogenic embryo (or allogenic embryo in surrogate mothers) is not immunologically rejected. She notes that the fetal immune system can tolerate antigens (including its mother and its own organs) better than to eliminate

antigens from its environment. Trophoblast-derived thymic stromal lymphopoietin (TSLP) instructs decidual CD11c+ dendritic cells (dDCs) with increased costimulatory molecules, MHC class II, and Th2/3-type, but not Th1-type, cytokines. TSLP-activated dDCs cause proliferation and differentiation of decidual CD4+ CD252 T cells into CD4+ CD25+ FOXP3+ regulatory T cells (Tregs) through TGF-b1. Decidual CD4+ CD25+ FOXP3+ Tregs promote invasiveness and HLA-G expression of trophoblasts, resulting in preferential production of Th2 cytokines and reduced cytotoxicity in decidual CD56brightCD162 NK cells.

The next few chapters focus on fetal hepatic development (Chaps. 25 and 26), fetal cardiovascular development (Chaps. 27 and 28), fetal neurological development and its implications on diseases on adult life (Chaps. 29 and 30), fetal nephrological development (Chap. 31), and fetal hematological development (Chap. 32). The proper development of the fetal liver, heart, and all other organs is vital to later well being. The liver is indispensable for maintenance of normal homeostasis. The development of the liver is a complex process involving an intricate web of various transcription factors which interact among each other and are regulated by extracellular signaling (Chap. 25). The liver is also the abode for a variety of stem cells – both for the bone marrow derived, which are pluripotent and home to the liver, as well as the hepatic progenitor cells (HPC), which are more differentiated, oval cells - playing roles in hepatocyte regeneration and response to injury (Chap. 26). Understanding the process of development of the liver may help in the creation of new cell therapies: growing tissues ex vivo for use in transplantation or for coaxing cells in vivo to acquire characteristics that can restore function in disease states. So far as the heart is concerned, abnormal embryonic development of the vascular system causes a variety of vascular anomalies. Vasculogenesis is genetically determined with a specific sequence of formation, selective regression and remodeling that must occur in the correct sequence to produce the typical mature vascular system. Investigation into the genes and signaling pathways that guide this complex process are crucial to understanding how vascular anomalies develop and may ultimately provide clues for therapeutic intervention or treatment of these conditions (Chap. 27). However, modern imaging technologies provide hope for structural cardiac defects (Chap. 28). Once the structural cardiac defects are identified by an obstetric sonologist, the precise details of CV defects are studied by a fetal echocardiographer. In selected subsets of fetuses, further details may be gathered with other imaging modalities. Finally, a detailed treatment policy can be formulated well before the baby is born. Fetal growth and development also has an impact on adult neurodegeneration. Chronic placental insufficiency may result in fetal hypoxemia which leads to synaptic dysfunction that triggers neurodegeneration in neonates. Maternal hormonal disturbances also affect the fetus adversely (Chap. 29). However, there are few human studies showing the effect of in utero exposure on neurodegeneration. The next chapter has collected data on low birth weight babies and observes that for over 50 years, preterm outcome studies have reported higher rates of cerebral palsy, intellectual disability, and sensory impairment in preterm survivors as compared with peers born at term. Chapter 31 discusses the development of kidneys in fetuses up to the second trimester, and claims that this organ is the "most important organ evaluated in nature." Chapter 32 examines fetal hematological development, noting the various stages of development and observing that like primitive erythropoiesis in the yolk sac, definitive erythropoiesis in the fetal liver is essential for continued survival of the embryo.

The next three chapters evaluate the pharmalogical implications of fetal development and the liabilities of treating the fetus as a patient. Chapter 33 discusses twin-twin transfusion syndrome and perinatal mortality. However, overall rates of perinatal survival have enhanced as a consequence of a scope of treatment modalities, including amnioreduction, septostomy, and laser ablation. Amnioreduction offers great results in right on time stage sickness, with at least one fetus surviving in more than 85 % of cases and two surviving in 66.7 % of cases with stage I or stage II disease. Chapter 34 also notes that previously treatment was restricted to the neonatal period, but with recent advancements in the basic science of medicine and surgery, there is a high possibility to detect and treat any anomaly of the fetus prenatally. Chapter 35 focuses on structural and functional perspectives on the placental barrier and its role in fetal development in the first two trimesters. The authors note that the placenta plays a significant role in providing a micro niche which supports the fetal growth through transfer of nutrients and immunological properties which helps in imparting protective immunological properties to the fetus. It also acts as a major fetomaternal barrier against pathogens from the maternal system. It also undertakes the secretion of different hormones, cytokines, growth factors, and other bioactive products essential for the fetus.

Chapters 36 and 37 explain surgery on the unborn. This is a relatively new field and is still developing. Surgeries are being undertaken for several anomalies like congenital diaphragmatic hernia, thoracic malformations, twin-twin transfusion syndrome, airway obstruction, aortic valve stenosis, and myelomeningocele. The first chapter reviews prenatal process and a variety of operating approaches for fetal surgery, while the second discusses an experimental therapy involving maternal laparotomy and hysterotomy, delivering the fetus and performing the necessary fetal surgery, after which the fetus is returned to the uterus and the pregnancy allowed to continue. The author observes that at subsequent delivery, there was a lack of scarring and contracture associated with the surgery. While this makes the procedure attractive, there are risks involved in interrupting a healthy and viable pregnancy. He notes that perhaps more relevant, at this stage, is investigating the biology of fetal wound healing to discover the mechanisms of scarless healing.

Chapter 38 focuses on the relation between intrauterine growth retardation (IUGR) and medical complications in adult life. Gross IUGR can lead to hypertension, increased risk of metabolic like dyslipidemia, diabetes mellitus, and cardiovascular disorders, renal impairments in some cases, improper brain and neurofunctional development and, in some rare cases, onset of auto-immune diseases also like rheumatoid arthritis and systemic lupus erythematosus.

The following six chapters are rather unique and provide very different perspectives on human fetal growth. There are many streams of alternative medicine, and the fact that they have survived over hundreds of years and sometimes millennia means that they cannot be ignored and that modern medicine should incorporate scientific truth from wherever it can be found. Chapters 39 and 42 provide perspectives on fetal growth and development from the point of view of traditional Chinese medicine, which also has potential for the tapeutic remedies for threatened miscarriages. Chapter 40 gives an indication on how tribal people in India can predict the sex of a baby by measuring the circumference of the belly of a pregnant woman during the second trimester and observing fetal movement. Tribal medicine also prescribes nutrients and discourages some kinds of food and drinks during pregnancy for the proper development of a fetus. The next chapter (Chap. 41) discusses applications of Unani medicine in pregnancy, though it is generally considered unscientific in the West. Homeopathy (Chap. 43), although the invention of a German scientist, is also discarded by the West as a non-viable science, though it has helped many mothers through difficult pregnancies. Chapter 44 evaluates the potential of Ayurvedic medicine in the first two trimesters of pregnancy.

Finally, the book also contains some articles which cannot be labeled in the categories mentioned above. Chapter 45 delves into the problems of prematurity. It describes the short-term and long-term complications of preterm birth in terms of fetal development as well as injury to fragile organ systems during the perinatal and neonatal periods. Although some randomized, controlled trials demonstrate the safety and effectiveness of a few treatments for neonates, scientific neonatal intensive care unit (NICU) treatments and interventions have not been adequately investigated. Therefore, the author recommends more rigorous studies of new therapies and medications before they are adopted. The next chapter (Chap. 46) also discusses the problems of extreme prematurity and notes that preterm birth is the most common cause of death among infants worldwide, and that the chances of survival without long-term difficulties is low. Chapter 47 focuses on the idea of sequencing the naturally aborted fetuses and its potential biological implications. The author believes that such a resource can easily be generated in India and such a database can provide answers to many crucial yet unsolved problems related to premature delivery to rejection of the fetus. This is also likely to answer several developmental issues that decide the progress from successful fertilization and fetal development to a fully mature baby. The final chapter (Chap. 48) pertains to the ethics of the use of aborted human tissues for research and therapeutic purposes. The book would not be complete without this chapter since ethics is an intrinsic part of medical research.

The editors gratefully acknowledge all the contributors to this volume. They have not been individually named here; instead, the editors have tried to highlight the multiple themes that have been discussed in the book. But it is essential at this point to say that contributors from all over the world – the Chinese University of Hong Kong; Yale University School of Medicine; the University of Science, Vietnam National University; Reproductive Genetic Institute, Chicago; University of Osnabrück, Germany; Maastricht University Medical Center, the Netherlands; Gulhane Military Medical Academy, Ankara, Turkey; Department of Health Science and Pediatrics, Turin

University, Italy – participated in this book project. The book also has a large number of contributors from premier Indian medical research institutions and other scientific institutions. The book, while beginning with the basic work done by Prof. Kanailal Mukherjee and his team, encompasses other research and theoretical perspectives and speculations from researchers across India. The editors have tried to make the book as comprehensive as possible, particularly because published works in the area of human fetal growth, development, and maturation are few and far between. Yet this is an area of growing importance, given the emergence of stem cell biology and regenerative medicine as an increasingly significant discipline because of the new potentials of cure and therapy that it promises. However, without understanding the growth and development of the human fetus, especially in its early stages, it will not be possible to understand how stem cells can help in regeneration of damaged tissues. The editors are therefore thankful to all the contributors for sharing their knowledge in this book. However, needless to say, the editors bear no responsibility for the views of the contributors, which are the products of their study and research and, therefore, their own.

Prof. Niranjan Bhattacharya and Prof. Phillip G. Stubblefield



To all scientists – present, past and future – whose sincere dedication to science and the search for truth has made science into a religion

Dr. Niranjan Bhattacharya, Kolkata, India Prof. Phillip G. Stubblefield, Boston, USA

Acknowledgment

We would like to express our gratitude to the many people who saw us through this book; to all those who provided support, talked things over, read, wrote, offered comments, allowed me to quote their remarks and assisted in the editing and design.

A book of this nature involves the cooperation of many: the contributors, publisher, as well as patients, researchers, and others who have helped the medical scientists with their work. Our thanks go out to all of them although it is not possible to name everyone. However, there are some who need special mention; without them, the book may never have been published.

More than 60 professors and senior researchers participated in this international collaboration. The editors are particularly grateful to Weston Grant and Victoria John, Senior Editors, Springer-Verlag London Limited, also Julia Megginson and Ron Jaworsky, only to name a few of the Springer's dedicated team, for their keen interest, advice, support, and guidance.

We gratefully acknowledge advise and involvement of leading experts on the science of fetal growth and development, including its clinical applications in day-to-day practice, who eagerly and actively participated and shared their state-of-art knowledge in the field. To name the leaders of the group let us start with Prof. K L Mukherjee of Institute of Post Graduate Medical Education and Research, Calcutta, India, along his disciple Prof. Chameli Ganguly, who persuaded the editors to accept the challenge of editing this multidisciplinary book. Other important participants include Prof. Siddhartha Roy, Department of Biophysics, Bose Institute, P1/12 C.I.T. Scheme VIIM, Kolkata 700054, India; Prof. M. Kemal Irmak High Council of Science, Gulhane Military Medical Academy, Ankara, Turkey; Prof. Joep Geraedts, Department of Genetics and Cell Biology, Maastricht University Medical Center, the Netherlands; Prof. Dr. med. Bodo C. Melnik, Department of Dermatology, Environmental Medicine and Health Theory, University of Osnabrück, Sedanstrasse 115, D-40090 Osnabrück, Germany; Prof. A Kuliev and his colleagues O Verlinsky, S Rechitsky, from Reproductive Genetic Institute, Chicago, IL, USA; Prof. Phuc Van Pham from Laboratory of Stem Cell Research and Application, University of Science, Vietnam National University, Ho Chi Minh city, Viet Nam; Prof. Alan Dardik, MD, PhD and his colleagues from the Vascular Biology and Therapeutics Program and the Department of Surgery, Yale University School of Medicine, New Haven, Connecticut, USA, and the Veterans Affairs, Connecticut Healthcare Systems, West Haven, Connecticut, USA; Prof. Bhabotosh Biswas MS, MCh

(Cardiothoracic Surgery), Vice Chancellor, West Bengal University of Health Sciences, Calcutta, India; Prof. Andrew Burd, MD, FRCS, Chinese University of Hong Kong leaving aside a group of brilliant Professors from SSKM Hospital like Prof. Gopal Dhali, Prof. Abhijeet Chaudhuri, Prof. Pradeep Saha, and Prof. Rajen Pandey, who also contributed with responsible articles. Other very distinguished participation for this international book project included Prof. Sanjukta Bhattacharya of Jadavpur University and Prof. Samir K. Brahmachari of the Academy of Scientific and Innovative Research, New Delhi, India, and CSIR – Institute of Genomics and Integrative Biology, New Delhi, India

The editors also gratefully acknowledge the contributions of all the participants of this edited book volume. These authors took precious time from their busy schedules in order to help us to complete the book in time.

The editors are also grateful to their wives for keeping their homes peaceful and creative and for maintaining a true academic and creative ambiance for research work (Prof. Sanjukta Bhattacharya for Prof. Niranjan Bhattacharya, and Linda Stubblefield, MSW, for Prof. Phillip Stubblefield). We thank them for their encouragement, understanding, and forbearance. Given their own interest in research in their respective fields, it is no surprise that their affection for the book is no less than that of ours.

We were also encouraged and facilitated in our work with creative criticism, comments, suggestions, and guidance from members of our fraternity, students, social activists, and patients, without whose keen interest, advice, and support it would have been difficult to proceed further in this new and vastly unknown field of modern regenerative medicine. May God bless them all for their goodwill and support. Last and not least: We beg forgiveness of all those who have been with me over the course of the years and whose names I have failed to mention. Prof. Niranjan Bhattacharya and Prof. Phillip Stubblefield.

Contents

Part I Introduction

1	Fetal Growth 3 Niranjan Bhattacharya and Phillip G. Stubblefield
2	Vaccination of the Unborn: A Perspective11Niranjan Bhattacharya and Sanjukta Bhattacharya
3	Understanding the Growth of the Fetus in Utero from the Immunologists Angle
4	Intraamniotic Antigen and Disruption of Human Fetal Growth:A Study from 1978–2002 with Subsequent Follow Up19Niranjan Bhattacharya
5	Fetal Growth and Developmentin the First Two Trimesters49Aditi Aikat, Tarun Kumar Roy, and Niranjan Bhattacharya
Par	t II Structural and Functional Fetal Development up to Second Trimester
6	
	Anthropometric Measurement of the Human Fetus
7	Chameli Ganguly [†] , Bimal Samanta, Gitanjaly Guha Thakurata [†] , Nemaichand Chandra, Sukla Ghosh, K.L. Mukherjee [†] ,
7 8	Chameli Ganguly [†] , Bimal Samanta, Gitanjaly Guha Thakurata [†] , Nemaichand Chandra, Sukla Ghosh, K.L. Mukherjee [†] , and Niranjan Bhattacharya Human Fetus: Carbohydrate Metabolism

10	Mechanism of Rejection of Human Fetal Adrenal Cortex	135
	Chameli Ganguly [†] , Bimal Samanta, Gitanjaly Guha Thakurata [†] , Chaitali Bhattacharya, Reba Bhattacharya [†] , K.L. Mukherjee [†] , and Niranjan Bhattacharya	
11	Development of the Human Fetal Brain Bimal Samanta, Chameli Ganguly [†] , Gitanjaly Guha Thakurata [†] , K.L. Mukherjee [†] , and Niranjan Bhattacharya	161
12	Lipid Metabolism in the Human Fetus Development Ornella Guardamagna and Paola Cagliero	183
Par	t III Gene and Human Fetal Development up to Second Trime	ster
13	Gene Regulatory Networks and Epigenetic Modifications in Cell Fate Decisions During the Early Embryonic Development Siddhartha Roy	199
14	Personal Human Life Begins with the Formation of Adult Type Hippocampus at 13th Week of Development M. Kemal Irmak	207
15	Dependence of Fetal Hair and Sebaceous Glands on Fetal Adrenal Cortex and Possible Control from Epidermal Merkel Cells and Adrenal Medulla M. Kemal Irmak	215
16	Theoretical Postulation of the Embryological Basis of the Virgin Birth and Role of Embryonic Stem Cells Localized Out of the Embryo M. Kemal Irmak	223
17	Aneuploidy in Human Preimplantation Embryos	231
18	The Potential Impact of Maternal Milk Consumption During Pregnancy on mTORC1-Driven Fetal Growth Bodo C. Melnik	237
19	Genetic Disease Specific Human Embryonic Stem Cell Lines Anver Kuliev, O. Verlinsky, and S. Rechitsky	259
Par	t IV Impact of Stem Cell on Growth and Development of Human Fetus	
20	Stem Cells in Growth and Developmentof the Human Fetus.Phuc Van Pham	267

21	The Mystery of the Survival of the Human Fetus in Utero Niranjan Bhattacharya, Sanjukta Bhattacharya, and Phillip G. Stubblefield	277
Par	t V Fetal Endocrine Development up to Second Trimester	
22	Glucose Metabolism in Foetus and Its Relationship with Foetal Insulin Prabir Kumar Kundu	285
23	Growth and Maturation of the Human Fetal Endocrine System Up to Twenty Four Weeks of Gestation Subhankar Chowdhury	291
Par	t VI Fetal Immune Development: Up to Second Trimester	
24	Development of Human Immune System in First and Second Trimester Swapna Chaudhuri	305
Par	t VII Fetal Hepatic Development: Up to Second Trimester	
25	Growth and Development of Human Liver and Its Future Implications Gopal Krishna Dhali and Gurubasava Lakamaji	319
26	Embryonic Development of Liverand Its Future ImplicationsAbhijeet Chaudhuri	331
Par	t VIII Fetal Cardiovascular Development: Up to Second Trimester	
27	Embryological Basis for Vascular Anomalies Trenton R. Foster, Jason A. Chin, Kirstyn E. Brownson, Hualong Bai, and Alan Dardik	349
28	Cardio Vascular Developmental Abnormality of the Human Fetus Appearing Within Second Trimester: Detection and Treatment Bhabotosh Biswas	359
Par	t IX Fetal Neurological Development: Up to Second Trimester and Its Implication in Adult Neurodegenerative Diseases	
29	Implications of Foetal Development and Neurodegenerationin Adult Life.Biman Kanti Ray	369

Part	t X Fetal Neuropsychiatric Development: Up to Second Trimes	ter
30	Very Low Birth Weight Babies and Their MentalHealth OutcomePradeep K. Saha	379
Part	XI Fetal Nephrological Development: Up to Second Trimester	r
31	Development of Renal Tissues in First Twenty Weeks Rajendra Pandey	391
Part	XII Fetal Haematological Development	
32	Development of the Haemopoiesis System before the Second Trimester of Pregnancy Mainuddin Naskar and Niranjan Bhattacharya	399
Part	XIII Pharmacological Implication of Fetal Development: Up to Second Trimester	
33	Liabilities of the Fetus as a Patient Mursheed Ali and Subhas Chakraborty	407
34	Fetus as a Patient During the First and Second Trimestersof Growth and DevelopmentPriyodarshi Sengupta, Mainuddin Naskar, Raj Gupta,Nandita Bose, Sushanta Banerjee, and Niranjan Bhattacharya	415
35	Structural and Functional Developmental Perspectives of the Placental Barrier and Its Role in the Fetal Development During the First and Second Trimester Priyodarshi Sengupta, Biplabendu Talukdar, Indranil Roy, Santanu Tripathi, Nandita Bose, Sushanta Banerjee, and Niranjan Bhattacharya	441
Part	XIV Surgical Implications of Fetal Development: Up to Second Trimester	
36	Surgery of the Unborn/Fetal Surgery	459
37	Reflections on Reality: Towards Scarless Healing: Déjà Vu Andrew Burd	467
Part	XV Fetal Growth and Adult Life Implications	
38	Implications of Gross IUGR in Adult Life with Respectto Some Major DiseasesPriyodarshi Sengupta and Niranjan Bhattacharya	495

Par	t XVI Understanding Fetal Growth from the Perspective of Alternative Medicine	
39	Growth and Development of the Human Fetus Up to the Second Trimester: A Perspective of Traditional Chinese Medicine Mursheed Ali and Niranjan Bhattacharya	507
40	Perspectives of Tribal Medicine on the First6 Months of PregnancyPriyodarshi Sengupta, Dibyendu Bandhopadhyay,and Niranjan Bhattacharya	513
41	General Perspectives and Applications of Unani Medicine in Pregnancy Up to the 2nd Trimester Priyodarshi Sengupta, Anamika Ishani Upadhyay, Abdus Salam Ansari, Akash Bhattacharya, Nandita Bose, Sushanta Banerjee, and Niranjan Bhattacharya	517
42	Applications of Traditional Chinese Medicine Conceptsin the Safety and Development of the Fetus Duringthe 1st and 2nd Trimesters of PregnancyPriyodarshi Sengupta, Akash Bhattacharya,and Niranjan Bhattacharya	525
43	General Perspectives of Homeopathic Medicine During the First and Second Trimesters of Pregnancy Priyodarshi Sengupta, Nutan Gavhane, Nandita Bose, Sushanta Banerjee, and Niranjan Bhattacharya	533
44	Ayurvedic Perspective of Pregnancy and Fetal Development During the First and Second Trimester Priyodarshi Sengupta, Madhav Rayate, Abhishek Kumar, Anamika Kumari Prasad, Nandita Bose, Sushanta Banerjee, and Niranjan Bhattacharya	541
Par	t XVII Miscellaneous	
45	Problems of Prematurity Annie Abraham and C.S. Rejiya	553
46	Problems of Extreme Prematurity Prasanta Choudhuri, Dhritidipa Chaudhuri, and Niranjan Bhattacharya	561
47	Sequencing of Naturally Aborted Human Foetuses: A Resource for New Knowledge Samir K. Brahmachari	571

Part XVIII Ethics and Human Fetal Tissue Research

48	Ethics Pertaining to the Use of Aborted Human Tissues	
	for Research and Therapeutic Purposes	575
	Priyadarshi Sengupta, Niranjan Bhattacharya,	
	Sanjukta Bhattacharya, and Phillip G. Stubblefield	
Ind	ex	583

Contributors

Annie Abraham, PhD Department of Biochemistry, University of Kerala, Thiruvananthapuram, Kerala, India

Aditi Aikat, MD Regenerative Medicine, Calcutta School of Tropical Medicine, Kolkata, India

Mursheed Ali, MSc Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

Abdus Salam Ansari, BUMS Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

Hualong Bai, MD Vascular Biology and Therapeutics Program, Department of Surgery, Yale University School of Medicine, New Haven, CT, USA

Veterans Affairs Connecticut Healthcare Systems, West Haven, CT, USA

Dibyendu Bandhopadhyay, MSc Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

Sushanta Banerjee, MD Director, Medical Education, Govt. of West Bengal, Kolkata, India

Formerly, Professor and Head, Department of Pharmacology, RG Kar Medical College, Kolkata, India

Akash Bhattacharya, MTech Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

Chaitali Bhattacharya, MSc, PhD (Cal) UGC Fellow, Institute of Child Health, Kolkata, India

Niranjan Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India

Reba Bhattacharya[†], MSc, PhD (Cal) Formerly, Research Fellow, Department of Biochemistry, Institute of Postgraduate Medical Education and Research, Kolkata, India Sanjukta Bhattacharya, PhD Professor, Department of International Relations, Jadavpur University, Kolkata, West Bengal, India

Bhabotosh Biswas, MS, MCh (Cardiothoracic Surgery) Vice-Chancellor, West Bengal University of Health Sciences, Kolkata, West Bengal, India

Formerly, Professor, Cardio-Cascular Surgery, RG Kar Hospital, Kolkata, West Bengal, India

Nandita Bose, MD Director, School of Tropical Medicine, Kolkata, India

Formerly, Professor, Department of Pathology, IPGMER, SSKM Hospital, Kolkata, India

Samir K. Brahmachari, PhD Former, Director-General, Council of Scientific and Industrial Research, Govt of India, New Delhi, India

Kirstyn E. Brownson, MD Vascular Biology and Therapeutics Program, Department of Surgery, Yale University School of Medicine, New Haven, CT, USA

Veterans Affairs Connecticut Healthcare Systems, West Haven, CT, USA

Andrew Burd, MD, FRCS Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

Paola Cagliero, PhD Department of Health Science and Pediatrics, Turin University, Torino, Italy

Nemaichand Chandra, MSc, PhD (Cal) Department of Biochemistry, All India Institute of Medical Sciences, Patna, India

Abhijeet Chaudhuri, MD, DM Department of Hepatology and Transplantation, School of Digestive and Liver Diseases, Institute of Post Graduate Medical Education & Research and SSKM Hospital, Kolkata, India

Dhritidipa Chaudhuri, MBBS, Dip National Board Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

Swapna Chaudhuri, PhD Emeritus Professor, Department of Physiology, ICMR, New Delhi, India

Jason A. Chin, MD Vascular Biology and Therapeutics Program, Department of Surgery, Yale University School of Medicine, New Haven, CT, USA

Veterans Affairs Connecticut Healthcare Systems, West Haven, CT, USA

Prasanta Choudhuri, MBBS, M.Sc (Thalassemia) Department of Regenerative Medicine and Translational Science, Kolkata School of Tropical Medicine, Kolkata, India

Subhankar Chowdhury, DTM&H, MD, DM, MRCP, FRCP Department of Endocrinology and Metabolism, Institute of Post Graduate Medical Education and Research and SSKM Hospital, Kolkata, West Bengal, India Alan Dardik, MD, PhD Vascular Biology and Therapeutics Program, Department of Surgery, Yale University School of Medicine, New Haven, CT, USA

Veterans Affairs Connecticut Healthcare Systems, West Haven, CT, USA

Gopal Krishna Dhali, MD, DM Department of Gastroenterology, IPGMER and SSKM Hospital, Kolkata, India

School of Digestive and Liver Diseases, Institute of Post Graduate Medical Education & Research, Kolkata, India

Trenton R. Foster, MD Vascular Biology and Therapeutics Program, Department of Surgery, Yale University School of Medicine, New Haven, CT, USA

Veterans Affairs Connecticut Healthcare Systems, West Haven, CT, USA

Chameli Ganguly[†], **MSc**, **PhD** Former Biochemist Central Calcutta Society for Advancement of Human Development and Research, Kolkata, India

Nutan Gavhane, BHMS Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

Joseph P.M. Geraedts Department of Genetics and Cell Biology, Maastricht University Medical Center, Maastricht, AZ, The Netherlands

Sukla Ghosh, MSc, PhD (Cal) Department of Zoology, Ballygunge Science College, Calcutta University, Kolkata, India

Ornella Guardamagna, MD Department of Health Science and Pediatrics, Turin University, Torino, Italy

Raj Gupta, MSc Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

M. Kemal Irmak Gulhane Military Medical Academy, High Council of Science, Ankara, Turkey

Anver Kuliev Reproductive Genetic Innovations, Chicago, IL, USA

Abhishek Kumar, BAMS Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

Prabir K. Kundu, MD, MD Department of Endocrine, Metabolic and Nutrition Diseases, Calcutta School of Tropical Medicine, Kolkata, India

Calcutta School of Tropical Medicine, Kolkata, India

Bodo C. Melnik Department of Dermatology, Environmental Medicine and Health Theory, University of Osnabrück, Osnabrück, Germany

K.L. Mukherjee[†], **MB**, **PhD** (**Cal**), **PhD** (**Wisconsin**) Former Head of the Department of Biochemistry, Institute of Post Graduate Medical Education and Research, Kolkata, West Bengal, India

Subhas Chakraborty Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

Mainuddin Naskar, MBBS Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

Rajendra Pandey, MD, DM Department of Nephrology, Institute of Post Graduate Medical Education & Research, Kolkata, India

Anamika Kumari Prasad, BAMS Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

Biman Kanti Ray Department of Neurology, RG Kar Medical College, Kolkata, India

Madhav Rayate, BAMS Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

S. Rechitsky Reproductive Genetic Innovations, Chicago, IL, USA

C.S. Rejiya, PhD Department of Biochemistry, Sree Ayyappa College (TDB), Eramallikkara, Kerala, India

Indranil Roy, MSc Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

Siddhartha Roy, PhD Department of Biophysics, Bose Institute, Kolkata, India

Tarun Kumar Roy, MBBS, DMRD Regenerative Medicine, Calcutta School of Tropical Medicine, Kolkata, India

Pradeep K. Saha, MD Department of Psychiatry, IPGME&R, Kolkata, India

Institute of Psychiatry - A Centre of Excellence, Kolkata, India

Bimal Samanta, MSc, PhD (Cal) Central Calcutta Society for Advancement of Human Development and Research, Kolkata, India

Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

Priyodarshi Sengupta, MPhil Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

Phillip G. Stubblefield, MD Emeritus Professor of Obstetrics and Gynecology at Boston University School of Medicine, Boston University, Jamaica Plain, MA, USA

Biplabendu Talukdar, MD Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

Gitanjaly Guha Thakurata[†], MSc, PhD (Cal) Formerly, Department of Biochemistry, National Medical College and Hospital, Kolkata, India

Santanu Tripathi, MD Professor, Department of Clinical and Experimental Pharmacology, School of Tropical Medicine, Kolkata, India

Anamika Ishani Upadhyay, BUMS, M.Phil Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

Phuc Van Pham, PhD Laboratory of Stem Cell Research and Application, Ho Chi Minh City University of Science, Vietnam National University, Ho Chi Minh City, Vietnam

O. Verlinsky Reproductive Genetic Innovations, Chicago, IL, USA

[†]Author was deceased at the time of publication.
Part I

Introduction

Fetal Growth

Niranjan Bhattacharya and Phillip G. Stubblefield

Introduction

Prenatal studies of commonly used research animals do not provide good models for fetal growth patterns. Studies of human intrauterine growth usually are based on anthropometric measurements of infants born at various gestational periods. Gestation is usually estimated in weeks from the first day of the last menstrual period before conception. Thus, conceptual age and the period of actual embryonal and fetal growth may be 1–3 weeks less than the estimated gestational age. Numerous centile charts have been constructed which relate fetal measurements to gestational age [1, 2]. Studies of birth weight in relation to gestational age have been carried out in large population samples usually on a retrospective basis. They have emphasized that maternal height, weight gain in mid-pregnancy, parity, age, ethnic group,

e-man: sanjuktaniranjan@gman.co

P.G. Stubblefield, MD

genetic factors, socio-economic factors, nutrition before and during pregnancy and smoking during pregnancy, all may influence fetal growth.

Fetal growth like post-natal infant growth undoubtedly is influenced by nutrient availability for mother and fetus and nutritional state of the mother before and during pregnancy. The nutritional state of the mother when she was herself a fetus may also be a significant regulator of the growth of her fetus [3].

The maximal velocity of fetal growth appears to be between 32 and 38 weeks of gestation. During that interval the birth weight virtually doubles. By 39 weeks the peak growth is reached. Normal full term gestation is considered to be between 37 and 41 weeks. If born before 37 weeks, the infant is considered premature. If born after 41 weeks it is considered post-mature. Mean birth weight for babies are smaller in most developing countries than in Europe, the United States of America and Canada. This may be due to a shorter period of high velocity of growth, perhaps related to maternal nutrition prenatally and during pregnancy.

Physical Growth and Its Measurements

Although growth is a highly complex process, it takes place in a completely ordered fashion in the biological system. The intrauterine period of mammalian growth can be divided into three

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India e-mail: sanjuktaniranjan@gmail.com

Emeritus Professor of Obstetrics and Gynecology at Boston University School of Medicine, Boston University, Jamaica Plain, MA, USA e-mail: phillip.stubblefield@gmail.com

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development: First and Second Trimesters*, DOI 10.1007/978-3-319-14874-8_1

distinct phases, in each of which the mode of nutrition is different [4]. In the first blastocyst phase, nutrition is derived from the uterine secretions. In the second or embryonic phase, nutrition is derived from the active erosive agency of the fetal trophoblast, while in the third fetal phase, trophoblastic activity ceases and nutrition is derived by diffusion from the maternal blood stream.

EM Widdowson [5] described the changes in the body composition of the fetus during growth. The initial division of the fertilized ovum is not accompanied with an increase in its size, and this may be true for a few further divisions, so that each cell becomes progressively smaller and there is no increase in total size. By the time the blastocyst has become implanted in the wall of the uterus, about the ninth day after fertilization, synthesis of proteins from the amino acids, and water and inorganic substances that reach it, begin to enlarge the cells before they divide. There is an alteration too in the timing of the divisions. The first few occur almost simultaneously, but after a few days, cell division becomes staggered so that only a few cells are dividing at the same time.

The immature organism is characterized by the large percentage of extracellular fluid within it. When the ovum is fertilized and for the first few generations of divisions, the new organism is entirely cellular, so there must come a time in early development when extracellular fluid becomes a part of it. There is an upper limit to the size of a cell that it can attain. This is partly because as it grows, its volume increases with the cube of the radius but its surface area expands with the square. The volume determines its biochemical activity, but the materials necessary for this activity must pass in through the surface membrane. Further, as the cell increases in size the ratio between the cytoplasm and the nucleus increases, and there is again an upper limit to the amount of cytoplasm that the nucleus can control.

In the first 2 or 3 weeks of human development inside the uterus, growth takes place entirely by cell division. Differentiation soon begins and as the organs and tissues appear, each develops its own characteristic kind of material and its own contribution to the weight of the organism as a whole. With differentiation comes a slowing down of cell division and as this occurs the average size of the cells begins to increase. This is accompanied by a progressive increase in the proportion of organs occupied by cells and a decrease in the percentage of extracellular fluid. This process of decreasing of extracellular fluid goes on during fetal and early postnatal life [6].

As an animal grows, its body proportions and composition change. These alterations are brought about by a differential rate occurring in the different parts and tissues of the body [7]. The order in which the various parts and tissues develop is much the same in all species, for it is based on the relative importance of the functions of the parts or tissues for the survival of the animal.

The changes in the form of embryo as it develops, repeat in general, the evolution of the species [8]. In horses, cattle and sheep, whose young are born at an advanced stage of development and have to follow their mother, the maximum proportions of leg length occur at birth, while in rabbits, which are born at a less advanced stage of development, this does not occur until later. In general, the head and the legs form a high proportion of the body weight at birth. With development, the body first lengthens and later deepens. The growth gradient starts at extremities and passes inward to the loin, while the lower parts of the ribs are the latest to develop.

When growth is limited by a low level of nutrition, the earlier developing parts and tissues have priority of supply [9].

The growth of the body and the changes in its composition during development may be considered in a number of different ways. Growth and development begin at the moment of conception and are continuous from then on until the epiphyses fuse and growth in height ceases, and the processes concerned with chemical development come to an end. Data on growth and composition of the body before term at 40 weeks gestation are of necessity cross-sectional. Measurement of growth in size after births are mainly longitudinal, but measurement of body composition at all ages has not been longitudinal [6]. Before describing early growth it is necessary to consider briefly the measurements of growth which are available in clinical practice, i.e., measurement of physical growth.

Weight

Weight is still the most widely used single clinical measurement of growth in intrauterine and postnatal life. Weight can be regarded as the sum of body fat and lean body tissue so that weight gain represents the sum of increments in different body components including muscles, skeleton, adipose tissue and water. It is therefore a non specific measurement of growth. In order to provide a more detailed assessment, other measurements are needed and crown-heel length, head circumference, biparietal diameter, etc. are now being increasingly used. The velocity of the growth of the body in weight and length is more rapid during the last 3 months of gestation and the first 4 or 5 months after birth than at any other age [6].

After birth the internal organs – heart, kidneys, lungs and liver – tend to grow parallel within the body, and they form roughly the same percentage of the body weight in all ages. Data based on P Gruenwald and HN Mainh [10] and E Boyd [11] show that before term, most organs contribute a little more to the weight of the body than they do during childhood and adolescence, but the spleen stays constant at 0.3 % of the body weight from 26 weeks gestation to 13 years.

Length

The accurate measurement of crown-heel provides the best clinical measurement of skeletal growth since, unlike weight gain, it is not influenced by the accumulation of water and fat [12].

Head Circumference

Indirect measurements have been used to follow the normal growth of the human brain. The most common of these is cranial circumference. Some correlations have been made between increase in cranial circumference and the cellular growth of the brain. Approximate formulae have been worked out relating head circumference to brain weight, protein and DNA content during the first year of life [13].

Nucleic Acids

In the rat, DNA synthesis and hence, cell division cease in the brain 20 days after birth [14]. Net protein synthesis continues to about 99 days of age and lipids are synthesized for varying periods between these extremes. Thus, the total number of cells is reached before the total protoplasmic mass of the brain is attained. There is a phase during which already existing cells must be enlarging in order to account for the over-all enlargement of the brain. Actually, careful examination of all non-regenerating organs reveals three distinct phases of growth. The first stage of growth is characterized by a proportional increase in weight, protein and DNA content. The number of cells is increasing whereas the ratios of the size of the individual cells are not changing. Simple hyperplasia in this case is occurring. This phase ends as the rate of net DNA synthesis begins to slow down and weight and protein content continue to increase at the same rate giving rise to a transitional phase of hyperplasia and concomitant hypertrophy which lasts until net DNA synthesis stops. After this point, all growth is by hypertrophy. Finally, when weight stabilizes and net protein synthesis stops, growth is finished. The ultimate packaging of the protoplasm into individual cells depends on the rate of DNA synthesis. The time during the growing period when cells are actively proliferating in any organ is presumably under genetic control and these times are different in different species. For example, in the rat, cell division in the brain is over by 21 days postnatally. In the guinea pig there is very little cell division in the brain after birth [15]. In contrast, in the human, cell division in the brain continues until around the end of the first year of life [16]. The enzyme DNA polymerase is essential for the synthesis of DNA. It has been shown that the activity of this enzyme mimics the rate of DNA synthesis in the chick brain [17] and in the rat brain. In the rat, the maximum rate of DNA synthesis is achieved at around 10 days of age. Peak enzyme activity is attained at 6-10 days of age and the activity is highest in those regions where cell division is most rapid [18].

EM Widdowson et al. in 1972 [19] examined cellular growth in the human fetal liver as judged by measurements of DNA and protein. They found that total DNA increased rapidly from around 15 weeks until term. The protein DNA ratio however was virtually unchanged from 15 to 25 weeks, suggesting that during the early fetal stages, the cell number is increasing more rapidly than the cell size. Also, some lipid components (diphosphoinositide, sphingomyelin, cerebrosides etc.) are absent in the early stages of individual development and appear only in the last stage of fetal development or in the first weeks of postnatal life [20].

Proteins

Development and metabolism of macromolecules are closely linked phenomena, since growth is necessarily the result of protein and nucleic acid synthesis. The fact that all the information that is characteristic of a living human being is stored in one fertilized cell, not only with regard to cell divisions involving billions of cells, and their structural and functional maturations but also to the time sequence of development of these characteristic events make us wonder with awe. There has been much progress in our scientific understanding of development, and much of this knowledge concerns the mechanism of protein synthesis in response to the transcription and translation of the genetic message. However, there are many gaps in knowledge about the exact way how and when the genetic message begins to be transcribed. The rate of net protein synthesis may be taken as an index of growth, since the increase in the mass of the cell may result from an increase in either the number of organelles within the cells or an increase in the soluble cytoplasm and both these types of increases involve synthesis of proteins. Protein synthesis maybe of two types: one is non-specific protein synthesis and the other is specific protein synthesis. Non-specific protein synthesis involves total protein synthesis during development, whereas specific protein synthesis involves the appearance or increase in the synthesis of different enzymes or special proteins characteristic of a particular kind of disease.

In the century and a half or more of the history of research into the proteins of nerve tissue, it is possible to distinguish two principal periods, the boundary between which is formed by the fifth decade of the twentieth century. In that decade radioactive methods, especially the use of amino acids labeled with radioactive isotopes, began to be used extensively for the study of the metabolic conversion of proteins in the nervous system.

The first period, which began in 1811 and which can be called the "pre-isotope" period, lasted about 140 years and ended with the discovery of the chemical topography of proteins in the tissues of the nervous system. For several decades in the nineteenth century, the attention of scientists was concentrated chiefly on the study of the lipid and carbohydrate composition of nerve tissue. As such, protein received far less attention. The main reason behind this was that the histochemical methods existing at that time and the biochemical methods that followed soon after were mainly for studying lipid and carbohydrate tissue components. Furthermore, for a long time the important role of proteins in the activity of the nervous system was not recognized. Studies revealed the complexity of the protein composition and the existence of protein complex with nucleic acids, lipids, carbohydrates and other substances in the nervous tissue. During the second or "isotope" period, although shorter in duration than the first, considerable progress has been made in studying the role of proteins in the functions of the nerve tissue. Because of the high sensitivity and high specificity of the isotopic indicator method, it is now possible to detect and to determine quantitatively, conversions of matter of such small magnitude in the living organism as were previously beyond the limit of methods earlier. Another unique feature of the isotope method is that the transformations of biologically important substances, especially proteins, on an extremely small scale can be detected in the intact organism without any significant disturbance of the functions of the organs of the experimental animal. In the case of dynamic equilibrium between the two processes, as is observed in the tissues of adult animals, the total protein content does not appreciably change, and for that reason, the methods available previously could not provide information about the rates of protein synthesis or breakdown which can be obtained by the isotope technique. In developing animals, unlike the adult, there is a steady increase in the total content of proteins, nucleic acids, carbohydrates, lipids and other tissue components which can easily be determined by ordinary chemical methods. Even in this case, however, the use of labeled precursors has certain advantages, for, in conjunction with other modern methods it can be used to determine not only a change in the content of the proteins, carbohydrates and other substances in the tissue, but also the intensity and rate of the metabolic conversion of these substances, their individual fractions and individual components at all stages of ontogenic development. So by exploiting the radio isotope technique in the protein synthesis of developing organs, the differential rates of development in individual organs during fetal life, can be estimated.

All functions of the cell and all physiological phenomenon involve conversions of proteins. The elucidation of the biochemical basis of the various physiological functions of any cell, including the most highly organized and specialized nerve cell, is therefore impossible without the knowledge of the composition of proteins, their physiochemical properties and biological properties and the principles governing their conversions during cell function. Without minimizing the importance of other macro-molecular substances of the cell, it could be said that proteins do occupy a central part of biochemistry. The multiple nature of its functions are fascinating. All physiological functions and the biochemical processes on which they are based take place with the active participation of proteins.[21]. Some proteins perform enzymatic functions in the body, others have hormonal, protective, transport, structural and other functions [22–24].

Considerable progress has been made in the study of protein metabolism in animal tissues. The biosynthesis of tissue proteins takes place throughout development. In the developing organism this process provides chiefly for differentiation and growth of the tissues through the formation of their protein spectra, while in the later stages of ontongeny it functions chiefly for the renewal of the tissue proteins. Several evidences bearing on the morphogenetic changes in proteins in the course of differentiation of the various tissues during embryogenesis and the early stages of post-embryonic development have been summarized [25]. Regulation of protein composition, distribution and metabolism play a crucial role not only during the physiological functioning of the nervous system but also during cellular differentiation and during pathological changes. Before the widespread introduction of isotopes into biochemical research, age changes in protein metabolism were chiefly measured by determinations of the content of nitrogen and total proteins in organs, by changes in the qualitative composition and physiochemical properties of proteins and their complexes with other biologically important compounds and by the quantitative and the qualitative composition of the end products of dissemination. The first and fundamental indicator of these changes was the protein content in the brain at various stages of ontogeny. Evidences in relevant literature [26, 27] show that the protein content, calculated per weight of brain tissue, increases with age, although in old age its concentration falls [28]. During fetal life life there is a decrease in protein content (percent in relative population of solids) in the pig with the progress of gestation.

The process of protein synthesis has been extensively studied in the sea urchin [29]. In this primitive organism, protein synthesis is accelerated immediately after fertilization with an increased concentration of DNA, and the first signs of biochemical differentiation are apparent at the initiation of gastrulation [30]. Similar differentiation has also been observed at this stage of development in the amphibian embryo [31].

Substantial age and other functionally determined changes have been discovered in the chemical compositions [32], the structural organization [33] and the functional activity [34] of the cell organelles. These differences are concerned chiefly with the number, size and shape of the organelles [35] and also the distribution of protein [32], nucleic acids and free nucleotides and lipids [33].

Specific Protein Synthesis

Development is characterized by growth as well as by structural and functional differentiation. In the developing tissues and organs, specific enzyme patterns are expressed enabling the growing organisms to cope with the demands of fetal life after separation from the mother. Hence profound alterations in the enzymatic equipment of a growing and differentiating organ may be expected to occur.

Conclusion

The timing of the successive developmental processes is one of the most fundamental aspects of ontogenesis. The early phases of development depend primarily on the genetic program. However, the sequential gene expression is apparently under the control of metabolic influences as the dominant though perhaps not exclusive mechanism. For many decades, scientists have collected descriptive information about the morphological aspects of development, but the study of enzymatic development and differentiation and especially its regulation is comparatively recent. Enzymatic difference of any given tissue is the process whereby it acquires its characteristic quantitative enzyme pattern. This process involves both positive changes, the appearance or increase in synthesis of enzymes de novo, and negative changes, the diminution in the amount of other enzymes. The fundamental problem underlying this process is that of gene expression.

The carefully programmed normal formation of enzymes continues to a rather late stage in mammalian development, namely to the late fetal stage and early post natal period. It is only when enzymatic differentiation has been completed that any organ or organism reaches full physiological maturity and functions. The lack of complete enzymatic development forms the biochemical basis for the concept of 'functional immaturity'. Many clinical problems associated with infants of low birth weight, less matured infants and infants of diabetic mothers are those of functional immaturity or alterations in the normal course of enzymatic development. Knowledge and understanding of the normal process of enzymatic differentiation and its regulation, especially in the human fetus and neonate are of the greatest importance in the care of the newborn with problems of functional immaturity.

References

- Lubchenko LO, Hansman C, Dressler M, Boyd E. Intrauterine growth as estimated from live born birth weight data at 24 to 38 weeks of ghestation. Pediatrics. 1963;32:739–800.
- Tanner JM, Thomson AM. Standards for birth weight at gestation periods from 32 to 42 weeks, allowing for maternal height and weight. Arch Dis Child. 1970;45:566.
- 3. Barker, DJP. The malnourished baby and infant Relationship with Type 2 diabetes. Br Med Bull 2001;60(1):69–88. doi: 10.1093/bmb/60.1.69
- Hammond J. Growth in size and body proportions in farm animals. In: Zarrow MX, editor. Growth in living systems. New York: Basic Book; 1961. p. 321–34.
- Widdowson EM. Changes in body composition during growth. In: Davis JA, Dobbing J, editors. Scientific foundations of paediatrics. London: William Heinemann Medical Books Ltd.; 1981. p. 330–42.
- Widdowson EM. Growth and body composition in childhood. In: Carraza BFR, Grace M, Nichols BL, Senteree J, editors. Clinical nutrition of the young child. New York: Nestle Nutrition SA and RAVEN Press; 1985. p. 1–14.
- Palson H. Conformation and body composition. In: Hammond J, editor. Progress in the physiology of farm animals, vol. 2. London: Butterworth's Scientific Publications; 1955.
- Hammond J. Farm animals. 2nd ed. New York: St Martin's; 1952.
- 9. Hammond J. Physiological factors affecting birth weight. Proc Natl Acad Sci U S A. 1944;2:8.
- Gruenwald P, Minh HN. Evaluation of body and organ weights in perinatal pathology. 1. Normal standards derived from autopsies. Am J Clin Pathol. 1960;34:247–53.
- Boyde E. Growth, including reproduction and morphological development. In: Alton PL, Dittmer DS, editors. Biological handbooks. Washington, DC: Federation of American Societies for Experimental Biology; 1962. p. 346–8.
- Babson SC, Bramhall JL. Diet and growth in the premature infant: the effect of different dietary intakes of ash electrolyte and protein on weight gain and linear growth. J Pediatr. 1969;74:890.

- Winick M, Rosso P. Head circumference and cellular growth of the brain in normal and marasmic children. J Pediatr. 1969;74:774–8.
- Fish I, Winick M. Cellular growth in various regions of the developing rat brain. Pediatr Res. 1969;3: 407–12.
- Mandel P, Rein H, Harth-Edel S, Mandel R. Distribution and metabolism of ribonucleic acid in the vertebrate central nervous system. In: Richter D, editor. Comparative neurochemistry. New York: Pergamon Press; 1964.
- Winick M. Changes in nucleic acid and protein content of the human brain during growth. Pediatr Res. 1968;2:352–5.
- Margolis FL. DNA and DNA polymerase activity in chicken brain regions during ontogeny. J Neurochem. 1969;16:447–56.
- Brasel JA, Ehrenkranz RA, Winick M. DNA polymerase activity in rat brain during ontogeny. Dev Biol. 1970;23:424–32.
- Widdowson EM, Crabb DE, Millner RDG. Cellular development of some human organs before birth. Arch Dis Child. 1972;47:655.
- de Almeida DF, Pearse AGE. Comparative histochemistry of lipids in relation to myelination in rabbit brain. J Neurochem. 1958;3:132–8.
- Liao SF, Wang T, Regmi M, Lysine nutrition in swine and the related monogastric animals: muscle protein biosynthesis and beyond. SpringerPlus 2015;4:147. doi:10.1186/s40064-015-0927-5
- 22. Haurowitz F. Chemistry and function of proteins. New York: Academic; 1963.
- 23. Dixon M, Webb EC. Enzymes. London: Longmans; 1964.
- Schweet R, Owen R. In Current problems in biochemistry (Russian translation)). IL, Moscow; 1961. p. 324–43.

- Brachet J. Biochemical embryology (Russian translation) IL, Moscow; 1961.
- Bennett EL, Rosenzweig MR, Krech D, Ohlander A, Morimoto H. Cholinesterase activity and protein content of the rat brain. J Neurochem. 1961;6:210–8.
- Koch WJ, Koch ML. Contributions to the chemical differentiation of the CNS III. The chemical differentiation of the brain of the albino rat during growth. J Biol Chem. 1913;15:423–48.
- Himwich WA, Himwich HE. Brain composition during the whole life span. Geriatrics. 1957;12:19–27.
- Hultin T, Bergstrand A. Incorporation of C14 L-Leucine into protein by cell-free systems for sea urchin embryos at different stages of development. Dev Biol. 1960;2:61–75.
- Immers J. Comparative study of the localization of incorporated C14 labelled amino acid and sulphate 35 in the sea urchin ovary, egg and embryo. Exp Cell Res. 1961;24:356.
- Loutrop S, Werdinius H. Metabolic phases during amphibian embryogenesis. J Exp Zool. 1957;135:203.
- Dahl DR, Samson FE. Metabolism of rat brain mitochondria during postnatal development. Am J Physiol. 1959;196:470–2.
- Weinbach EC, Garbus J. Oxidative phosphorylation in mitochondria from aged rats. J Biol Chem. 1959; 234:412–7.
- 34. Jordan WK, March R. Partition of adenosine triphosphate in intracellular fractions of mature and immature rat cerebrum. J Histochem Cytochem. 1956;4:301–11. http://jhc.sagepub.com/content/4/4/301.long
- 35. Murthy MRV. Protein synthesis in growing rat tissues. I. Effect of various metabolites and inhibitors on phenyl alanine incorporation by brain and liver ribosome's. II. Polyribosome concentration of brain and liver as a function of age. Biochem Biophys Acta. 1966;119(3):586–98.

Vaccination of the Unborn: A Perspective

Niranjan Bhattacharya and Sanjukta Bhattacharya

Introduction

Shortly after the Salk and Sabin polio vaccines had demonstrated the transformative benefits of childhood vaccination but long before the ill-informed controversy over the measles-mumps-rubella vaccine became a concern for refusal of vaccination, the Vaccination Assistance Act of 1962 established a U.S. vaccination program against polio, diphtheria, tetanus, and pertussis. With that effort launched and growing attention directed at imminent vaccination campaigns against influenza, measles, and rubella, a leadership group was formed by the US Government. That group, the Advisory Committee on Immunization Practices (ACIP), marks its 50th anniversary this year (2015) [1]. Vaccine refusal not only increases the individual's risk of disease but also increases the risk for the whole community. As a result of substantial gains in reducing vaccine-preventable diseases, the memory of several infectious diseases has faded from public

e-mail: sanjuktaniranjan@gmail.com

consciousness and the risk-benefit calculus seems to have shifted in favor of the perceived risks of vaccination in some parents' minds [2].

Theoretical Prospect/Potentialities of Vaccinating the Unborn

Infections are an important cause of morbidity and mortality worldwide and it is young infants who often suffer from a disproportionately great incidence of infection and subsequent mortality. Currently, vaccine strategies to prevent infections in infants generally direct initiation of immunization at around 2-4 months and protective levels of antibodies may not appear until 6-7 months of age [3]. This strategy fails to prevent important infections of the newborn in the first month. Experience with hepatitis B immunization of the newborn (both passive and active) shows that it can safely prevent neonatal infection and long term sequelae [4]. Tetanus in neonates has been dramatically reduced in many parts of the world in which pregnant women are vaccinated with tetanus toxoid [5].

Advantages and Disadvantages of Immunization

Strategy I (maternal immunization): Advantages: (a) a term baby has antibody at birth, and (b) there is active immunity in the mother. Disadvantages:

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India

S. Bhattacharya, PhD

Professor, Department of International Relations, Jadavpur University, Kolkata, West Bengal, India

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_2

(a) decreased passive antibody in premature infants, (b) passive immunity in the newborn baby wanes progressively within 6 months.

Strategy II (Neonatal Immunization): Advantages: (a) enhanced compliances at birth with mother's support; (b) the beginning of active immunity of the neonate. Disadvantages: (a) response to vaccination may be weak, and (b) development of protective immunity may take some more weeks.

Strategy III: Prospects of intra amniotic or intra fetal injection after 20 weeks; on the basis of the experimentation carried out by us (discussed in Chapter 4), there is abortion but no fetal death was reported in the growing fetus above 18 weeks.

Ours is possibly the first global experience with intrafetal vaccination attempts before 20 weeks of gestation. On the basis of our experiences we are strongly against vaccination of the unborn through the intra amniotic or intra fetal route, because of its potential adverse effects as seen in the histology of a non-aborted anencephaly fetus at 28 weeks and non-aborted fetus at 14 weeks leaving aside the problem of poor antibody response (depending on gestational maturity, time of exposure and the optimum dosage of antigen). The toxic/non specific reaction could be due to reduction in antibody formation and other forms of immuno-incoordination in the immature fetus.

While theoretically it may appear fascinating to vaccinate the grossly premature high risk newborn though the intra amniotic or intra fetal route and thus protect the premature newborn against possible infections, our practical experience goes strongly against such interventions at least not before 20 weeks.

Role of Antigenic Stimulation on Fetal Intra Uterine Growth: Implications of Increase in Liver Weight/Spleenic Weight/Thymic Weight and Rise of DNA in the Fetus Before the Process of Abortion

Antigenic (tetanus toxoid) stimulation through the intra amniotic route showed a definite increase in fetal liver weight, thymus weight, spleenic weight and also an increase in cell numbers when compared to the control (blank antigen) challenged fetus. There was also some increase in essential enzymes concerned with glutamine synthesis of the human fetal liver glutamyl dehydrogenase and gamma glutamyl transferase. Seeing such changes within the wider physiological content, attention has been increasingly drawn to the interactions between the immunological and the neuroendocrine systems. Immunological cells have the receptors which enable them to receive signals from a wide range of hormones, corticosteroids, insulin, growth hormones, testosterone, estradiol, B adrenergic agents, acetylcholine, endorphin and encephalins, but by and large, glucocorticoids and androgens depress immune response whereas estrogen, growth hormone, thyroxine and insulin do the opposite [6]. These hormones work through different cytokines/growth factors and their binding protein or proteases. Hence the precise mechanism of how the antigenic stress at a critical period of growth in vitro possibly disrupts the fine immune-neuroendocrine coordination of the receptor growth factor cytokine mediated action at the cellular level, would be an interesting field of research for future researchers in this area.

Theoretical Problem of Graft versus Host Reaction and Autoimmunity in the Aborted and Non Aborted Fetus

The rise in spleenic weight and liver weight along with the necrosis of the liver, denudation of intestinal epithelium and skin lesion constitute the cardinal features of GVH disease [7]; hence its presence strongly suggests the possibility of graft versus host type of injury effect with antigenic challenge through the intra amniotic route due to a disruption in the thymic schooling of the T lymphocyte during the human fetal ontogeny of the immune system. Spleenic rise in weight from 12 weeks onwards and pre-spleenic, i.e., before 11 weeks, liver weight rise, which continued even after the appearance of the spleen, up to 20 weeks of the study of a human gestation with intra amniotic antigenic challenge as well as study of the histology of the skin/liver, suggests the possibility of graft vs host type of reaction effect. This hypothesis regarding graft vs host reaction may not hold true for the early weeks of gestation of a fetus, that is, before 16 weeks, because antigens cause direct stimulation of the fetal system, resulting in congestion, hemorrhage, varying degrees of loss of architecture in the early weeks and mononuclear invasion and thrombosis in all the organs and in the placenta with varying degrees of villitis in later weeks (17 weeks or more). We believe that the early acute reactions in the embryo in the pre-immune status of the fetus (16 weeks) are hitherto unnamed inflammations or perhaps even an autoimmune reaction in case there is participation of the premature immune system, while reactions in the 17 or more weeks fetus may have a component of graft vs host reaction as may be perceived from the placenta with villitis, thrombosis, mononuclear invasion etc. However, further specialized immunological study, which will differentiate the fetal leucocyte from the maternal leucocyte, can confirm or reject our hypothesis that antigenic intra amniotic stimulation during the early weeks of gestation can cause an autoimmune type of acute reaction due to poor schooling at the thymus. The fetal lymphocytes would be able to distinguish properly or at least partially, its own property (antigen) from the foreign property (antigen) and the war would be at the fetomaternal border (in the placenta). Conscription of the immature fetal leucocytes due to antigenic stress or war like situations in the fetus, can provoke the much more powerful (superpower) maternal immune systems' sanctions which can lead to premature birth (independent immature systems statehood for the fetus) or death and abortion depending on the gestational maturity of the fetus.

Advantage of Hypoimmune Fetal System in Biomedical Research

What have we learned from our study of antigenic stimulation of the growing human fetus in the utero for the last 40 years? We have learned that the growing human fetus (8–20 weeks) in utero can react to an antigenic challenge; for humoral immunity (tetanus toxoid) and cellular immunity (BCG). However, in case of tetanus toxoid stimulation the effect widely varies, as there is wide variation in the induction-abortion interval. Why is this so? Is it the lower expression of receptors, or neoantigens in the pre-immune or the hypoimmune fetal system which make it resistant to antigenic stimulation? The answer is possibly in the affirmative. This could explain the delayed response even with a very high dosage of antigenic stimulation (2–4 cc tetanus toxoid) for a tiny fetus weighing 8–150 g in the uterus, when the adult dosage is 1 cc.

This hypoimmune fetal tissue has an edge over the adult tissue in situations requiring transplant of fetal tissue in the adult system. For example, in case of stem cell transplantation (umbilical cord blood stem cell) in adults or fetal thymus transplantation (unpublished observation), there is survival of the fetal tissue without any graft versus host reaction in the adult system [8–11].

References

- Schwartz JL, Mahmoud A. A half-century of prevention – the advisory committee on immunization practices. N Engl J Med. 2014;371:1953–6. doi:10.1056/ NEJMp1410049.
- Omer SB, Salmon DA, Orenstein WA, deHart P, Halsey N. Vaccine refusal, mandatory immunization, and the risks of vaccine-preventable diseases. Engl J Med. 2009;360:1981–8. doi:10.1056/NEJMsa0806477.
- Fischer GW, Offoloni MG, Mond JJ. Prospects of vaccination during pregnancy and the newborn period. Clin Perinatol. 1997;24(1):231–49.
- Chanock R, Kapikian A, Mills J, et al. Influence of immunological factors in respiratory syncytial virus disease. Arch Environ Health. 1970;21:347–55.
- Baltazar JC, Sarol JN. Prenatal tetanus immunization and other practices associated with neonatal tetanus. Southeast Asian J Trop Med Public Health. 1994;25(1):132–8.
- Roitt IM. Roitt's essential immunology. London: Blackwell Science Ltd; 1997. p. 216.
- Kay HEM Fetal thymus transplants in man. In: Proceedings of the Ciba Foundation Symposium on ontogeny of acquired immunity, London 23–25th Nov 1971, Published by Elsevier, Excerpts Medica, North Holland; 1972. p. 257.

- Wegman TC, Lin H, Guelbert L, Mossman TH. Bi-directional cytokine interactions in the Maternofetal relationship. Successful Allopregnancy is a Th1 phenomenon. Immunol Today. 1993;14:353–5.
- Ribbing SL, Hoversland RC, Beaman KD. T cell suppressor factors play an integral role in preventing fetal rejection. J Reprod Immunol. 1988; 14:83–95.
- Rubesa G, Beaman KD, Lucin P, Beer AE, Rukavina D. Expression of TJ6 protein in the human first trimester decidual lymphocytes. Reg Immunol. 1994;6:331–3.
- Nicholas TC, Kang JA, Angkachatchi V, Beer AE, Beaman KD. Lymphocyte expression of the pregnancy associated protein TJ6. Cell Immunol. 1994;155:219–29.

Understanding the Growth of the Fetus in Utero from the Immunologists' Angle

Niranjan Bhattacharya and Phillip G. Stubblefield

Development and Growth of the Human Immune System in Utero

Developmental immunology can be defined as the study of how adaptive host defense blood cells within an individual sequentially respond to repetitive environmental challenges in a way that promotes the health and survival of an individual. According to classical principles an individual becomes immune or protected from re-infection in response to an antigenic encounter during an initial infection. Mature immunological competence is ultimately achieved through cumulative adaptive changes stimulated by exposure to a large repertoire of foreign antigenic material. Since the in utero fetal environment is sequestered from frequent

P.G. Stubblefield, MD Emeritus Professor of Obstetrics and Gynecology at Boston University School of Medicine, Boston University, Jamaica Plain, MA, USA e-mail: phillip.stubblefield@gmail.com encounters with micro-organisms, the host defense system of the human newborn is inexperienced. This partially accounts for why the human newborn is vulnerable to human microbial attacks during the first 6 weeks of life. Furthermore although many components of the immune system of the fetus are present early in gestation some are immature and do not become fully functional (compared with the activity of immune defenses of adult subjects) until sometime after birth. Despite these limitations fetal host defense systems are capable of active engagement and an immune response does occur when the fetus is infected or immunologically challenged in utero.

The cells involved in the human immune system are derived from the stem cells originating in the yolk sac. In the human system, fetal liver and bone marrow takes the responsibility of the yolk sac at 5 weeks gestation. Natural Killer (NK) cells and T and B lymphocyte precursors are detectable in the fetal liver at 6 weeks and 7-8 weeks gestation. The fetal thymus is colonized by T cell precursors at 8-9 weeks and pre B cells are found in the bone marrow at 13 weeks, as evidenced due to the attachment of cluster of differentiation (CD) molecules in each of the sub types. Mature T and B lymphocytes are detectable by the onset of second trimester pregnancy. Some complement components can be detected from 6 to 14 weeks gestation although at a much lower level than in the adult.

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India e-mail: sanjuktaniranjan@gmail.com

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development: First and Second Trimesters*, DOI 10.1007/978-3-319-14874-8_3

Natural Immunity (Cellular Component of Macrophage Monocyte Natural Killer Cell System)

In the fetus polymorphonuclear (PMN) leucocytes are first identified at the fetal yolk sac stage of hematopoiesis [1]. Mature PMN's are not identifiable in the fetal liver or bone marrow until approximately 14 weeks of gestation [2]. By 22–23 weeks of gestation the circulatory PMN count has increased but still it is only 2 % of the circulatory PMN concentration measured in the cord blood of term gestational newborns. This low PMN count is compensated with high hematopoietic progenitor cells. This may be due to shifting of stem cells from the liver to the bone marrow.

Before the liver becomes the major site of hematopoietic activity, macrophages constitute 70 % of the blood cells present in the liver. Circulating monocytes do not appear in the fetal blood before 20 weeks of gestation. However, at 30 weeks gestation, monocyte concentration reaches 3–7 % of the circulating formed blood cells. As a whole, the influx of mononuclear phagocytes to the site of inflammation is delayed and attenuated in the new born. Similarly, in the case of NK cells whose number is normal in the neonatal period, the surface membrane expression of certain antigens is altered in comparison to adult NK cells. This may explain the diminished cytotoxicity in vitro and vulnerability of the newborn to viral and parasitic infection.

Natural Immunity (Humoral Component)

Opsonin activity is reduced in the new born at term and more so in the prematurely born at test concentration of plasma and serum above 10 %. This deficiency may be in part related to the lower complement or immunoglobulin (IgM and IgG) concentration in the new born [3].

Complement proteins are synthesized early in gestation [4]. Synthesis of C4, C2, C3 and C5 can be confirmed between 8 and 14 weeks of gestation. Evidence derived by several methods confirms that there is no transplacental passage of complement components. Components of the alternate pathway activity are more deficient in concentration than classical pathway activity in both term and pre-term infants.

Fibronectins are a class of multifunctional high molecular weight glycoproteins that serve to facilitate cell to cell to sub stratum adhesion and thus play an important role in directing cell migration, proliferation and differentiation. Thus fibronectins are essential for certain aspects of embryological development of the fetus and for hemostasis, hematopoiesis, inflammation and wound healing. In many pathological conditions like sepsis, fibrosis, etc the normal structure, physiology and functions of fibronectin are altered in a way that contributes to the underlying tissue or organ dysfunction. Circulatory plasma fibronection concentrations are reduced in fetal cord blood and in the term infant [5]. Plasma fibronectins are further reduced in respiratory distress syndrome, birth depression, sepsis, and intra uterine growth retardation. Fibronectin biosynthesis by macrophages in vitro is decreased in the neonatal period [6]. Fibronectin improves leucocyte function in vitro and promotes neutrophilic adhesion migration, chemotaxis and also assists killing of opsonized bacteria, yeasts etc.

Other Humoral Factors of Natural Immunity

C reactive protein (CRP) is synthesized by the fetus and the newborn and is one of the most important acute phase proteins. It helps to activate the classical complement pathway by beading with the C1q. CRP binding with the bacteria helps in the opsonization and rapid clearance of the bacteria through the neutrophil, monocyte or macrophage phagocytosis.

Lactoferrin is a positively charged iron binding glycoprotein present in the granules of the neutrophils. This helps in neutrophilic reactive oxygen intermediate production (OH), chemotaxis, endothelial adhesion, aggregation etc. Neonatal cord blood neutrophils are profoundly deficient in lactoferrin [7].

Acquired Immunity-Cell Mediated and Antibody Mediated Responses of T Lymphocytes

In the human, the thymus develops embryologically as an outgrowth from the third and fourth pharangeal pouches between the sixth and seventh week of gestation. Lymphocytes destined to become T cells appear among epithelial cells during the ninth and tenth weeks [8]. However it is not until the tenth week of gestation that the cortex and the medulla of the thymus region begin to demarcate and not until the 12th weeks of gestation that Hassall's corpuscles appear. The undifferentiated cells that first enter the thymus at the seventh week of gestation do not possess either CD4 or CD8 antigen but do express the T cell markers [9]. Within the thymus, maturation of the T cells is accompanied by the sequential appearance of surface phenotype markers CD7 followed by CD1, CD2, CD5 and somewhat later, CD3. As gestation progresses most cells leaving the thymus express either the CD4 or CD8 surface antigen. Those cells that lack both the antigens retain the stem cell function and possess a receptor (CD25 Tac antigen) for IL2 which plays an essential role in promoting simultaneous appearance of CD4 or CD 8 antigens. At this stage, the transcribed T cell alpha chain precedes that of the Beta chain. At the 12th week of the intra-uterine of life in the fetus, CD3 positive cells can be identified at the peripheral blood, which increase progressively with the progress of gestation and represent more than 50 % of T lymphocytes by the 22nd week of gestation. These CD3 positive cells also express either the CD4 or the CD8 antigen. By the 13th week of gestation CD3 positive cells appear in the fetal liver or spleen and by the end of the second trimester represent more than 50 % of the T lymphocytes in those organs. The T cell helper/suppressor ratio (CD4/CD8 ratio) in the cord blood is approximately 1.7:1 (versus the adult ratio of 2:1). In the peripheral blood 20 % of the T cells express thymocyte antigens compared with fewer than 1 % expression in the adult [10].

Lymphokines Production and Role in Cell Mediated Immunity

Functional responses involving the T lymphocytes and T cell derived lymphokines can be demonstrated relatively early in gestation. By the 12th week of gestation, lymphocytes obtained from the human thymus respond both to mitogens and to foreign histocompatiblity antigens in mixed lymphocyte culture [11]. In addition fetal cells stimulated with allo-antigens exhibit normal antigen specific cytotoxicity. In contrast, the phenotype appearance and proportion of circulatory cells is diminished and the production of some lymphokines are reduced in the neonate. The most significant defect appears to be a deficiency of memory T cells, because expansion of the memory T cell production is dependent on the exposure to foreign antigens. This T cell inexperience may be partially responsible for the diminished production of gamma interferon in the neonate [12].

B lymphocyte and Antibody Production: B cell maturation occurs in two stages. In the first stage undifferentiated stem cells mature into cells identifiable as B lymphocytes. This is an antigen independent phase that takes place in the fetal liver and the bone marrow in humans [13].

The second stage of lymphoid differentiation is antigen dependent and during this phase B lymphocytes are transformed into plasma cells. The first recognizable cell in the B cell lineage is the pre B cell. This cell can be detected in the fetal liver in the seventh to eighth week of gestation. The presence of heavy chain IgM [14] could be seen at this stage. As gestation progresses, pre-B cell can be detected in the fetal bone marrow. It is at the pre-B cell stage of development that clonal diversity is generated. Intact immunoglobulin genes are formed by the re-arrangement of gene segments comprising each heavy and light chain family. Pre-B cells also give rise to immature B lymphocyte which expresses only surface IgM and complement receptors [15]. These cells could be detected in the fetal liver at 8-9 weeks of gestation. However, clonal anergy could be noticed in case of exposure to selfantigen. Cells that express other immunoglobulin IgA or IgG can be demonstrated by the 12th week of gestation. At a somewhat later state (the mature B cell stage), cells express membrane bound IgG or IgA in association with the membrane IgM or IgD. By the 12th week of gestation, the normal fetus has levels of circulating B lymphocytes that are equal or higher than the adult. Fetal B lymphocytes can be demonstrated in highest proportion in the spleen (30 %), blood (35 %) and lymph nodes (13 %).

Antibody Production in the Fetus and Neonates

The fetus acquires the ability to produce serum immunoglobulins early in gestations [16]. In vitro studies have demonstrated the ability of fetal cells to produce antibody (IgM) by the eighth week gestation. IgG synthesis appears slightly later and IgA synthesis begins at about 30 weeks of gestation. However due to the sterile environment of the uterus the inability of the fetus to respond to certain kinds of antibody, T cell suppression and B cell differentiation, results in a very little antibody level before the time of birth [17]. At the time of birth most of the circulating antibodies in the fetal system are IgG antibodies passed through the placenta. Low level of fetal IgM (less than 10 % of the adult level) are demonstrable at term gestation and reach the adult level by 1-2 years of age. Adult level of IgG is attained at 4-6 years and adult IgA level is attained at puberty.

Placental Transfer of Antibody

Although B lymphocytes are present in the fetus, by the end of the first trimester there is very little active fetal immunoglobulin production. Without exposure to an antigen in the sterile environment of growth and differentiation of the fetal system serum immunoglobulin, levels are extremely low until 20-22 weeks of gestation, at which time an accelerated active transport of IgG takes place across the placenta [18]. Only the maternal IgG is transported due to the presence of specific placental receptors for heavy chain IgG molecules. This active transport of IgG causes a rise of IgG concentration in the fetus 5-10 % higher than the mother. Elevated levels of IgM or IgA in the cord blood demonstrates that the infant has been exposed to an antigen in utero and has synthesized antibody itself. Congenital infection with syphilis and rubella characteristically produces elevation of the cord blood IgM concentration. Elevated levels of both IgM and IgA also maybe found if there is transplacental materno-fetal hemorrhage.

References

- Christensen RD. Neutrophil kinetics in the fetus and neonate. Am J Pediatr Hematol Oncol. 1989;11(2):215.
- Ohis RK, et al. Neutrophil poolsizes and granulocyte colony stimulating factor production in human mid trimester fetuses. Pediatr Res. 1995;37:806.
- Baker CJ, et al. Role of antibody to native type III. Polyusaccharide of group B streptococcus in infant infection. Pediatrics. 1981;68:544.
- Kohler P. Maturation of the human complement septem I, onset time and sites of fetal C1q, C4, C3 and C5 synthesis. J Clin Invest. 1973;52:671.
- Yoder MC, et al. Plasma fibronectin in healthy infants, respiratory distress syndrome and perinatal asphyxia. J Pediatr. 1983;102:777.
- Gerdes JS et al. Decreased fibronectin biosynthesis by human cord blood mononuclear phagocytes in vitro-S. Leukocytic Biol. 1984;35:91.
- Hill HR. Biochemical structural and functional abnormalities of polymorpho nuclear leukocytes in the neonate. Pediatr Res. 1987;22:375.
- Haynes BF, et al. Early human T cell development; analysis of the human thymus at the time of initial entry of hematopoietic stem cells into the fetal thymic micro environment. J Exp Med. 1988;168:1061.
- Haynes BF, et al. Early events in human T cell ontogeny: phenotypic characterization and immunohistological localization of T cell precursors in early human fetal tissues. J Exp Med. 1988;168:1061.
- Campbell AC, et al. Lymphocyte subpopulation in the blood of newborn infants. Clin Exp Immunol. 1974;18:469.
- Leikin S, et al. Differences in transformation of adult and newborn lymphocytes stimulated by antigen, antibody and antigen antibody complexes. Cell Immunol. 1971;I:468.
- Nogoyama S, et al. Diminished expression of CD40 ligand by activated neonatal T cells. J Clin Invest. 1995;95:66.
- Kamps WA, et al. Micro-environmental studies of pre-B and B cell development in human and mouse fetus. J Immunol. 1982;129:526.
- Owen JJT, et al. In vitro generation of B lymphocytes in the mouse fetal liver; a mammalian bursa equivalent. Nature. 1974;249:361.
- Cooper MD. Pre-B cells normal and abnormal development. J Clin Immunol. 1981;1:81.
- 16. Ballow M, et al. Development of immune system in very low birth weight (less than 1500 gm) premature infants: concentration of plasma immunoglobin and the pattern of infections. Pediatr Res. 1986;20:899.
- Hayward AR. Development of lymphocyte responses and interaction in human fetus and newborn. Immunol Rev. 1981;57:39.
- Yeung CY, et al. Serum gamma globulin levels in normal, premature, post mature and small for date newborn babies. Lancet. 1968;I:1167.

Intraamniotic Antigen and Disruption of Human Fetal Growth: A Study from 1978–2002 with Subsequent Follow Up

4

Niranjan Bhattacharya

The search for a safe, effective, mid-trimester abortifacent (1978-2002) in which a group of researchers were engaged in Calcutta, revealed many hitherto unknown secrets about the growth and development of the human fetus up to 20 weeks. A major finding was that while expulsion by intra amniotic tetanus toxoid or auto absorption in case of BCG is the effect, the cause is the disequilibrium of the coordination of the growth and maturation of the developing fetus. That the fetus, even at a very early stage of growth, can react to an antigen, even in a sterile environment leading to embryopathic changes at the teratogen phase (up to 9 weeks) or the pre-immune phase (10-16 weeks), in a non-specific manner resembling an acute non-specific infection, or the hypo-immune phase (from 17 weeks onwards), is a new finding. These finding may have serious implications for future medical research. The major thrust areas in this context are discussed below.

Intra Amniotic Antigens as Abortifaecients

This article will showcase our experience of 688 cases of abortion after single intra amniotic injection of 2 cc tetanus toxoid (Table 4.1). It was observed that if the period of gestation is randomized the overall success rate of abortion after a single injection is 92.80 % in the present study. When the cut off period was 7 days after intra amniotic 2 cc injection, the abortion rate was 52.8 % (363 cases) which cumulatively became 72.4 % (498 cases) if the cut off period was extended for another week. On the 21st day cut off schedule, the rate was 86.2 % of abortion (593 cases), which figure eventually shot up to 92.8 % (638 cases) when the cut off period was 30 days. In sum, the experience showed that with 2 cc single intra amniotic injection of tetanus toxoid, there was 94.74 % abortion in the 8-11 weeks group and 94.05 % in the 12-15 weeks group, if the cut off period was 1 month. However, the induction-abortion interval varied widely. With progression of gestation 16-19 weeks or more, there was a gradual increase in the inductionabortion interval as is apparent from the mean induction-abortion interval. The success rate (if there is a cut off period of 1 month) of a single dose of intra amniotic injection of 2 cc schedule progressively comes down from 94.74 % in the 8–11 weeks group to 88.61 % in the 16–19 weeks

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India e-mail: sanjuktaniranjan@gmail.com

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_4

No. of cases (in %)	Weeks of gestation	Type of antigen used	Induction abortion interval range (mean ± SD)	Success rate (in %)	Condition of abortus (living)	Dead
19	8–11 weeks	Tetanus toxoid 2 cc single intra-amniotic injection schedule	48 hours– 11 days hour mean with $SD=96 h \pm 24.6 h$	94.74	_	19
319	12–15 weeks	Tetanus toxoid 2 cc single intra-amniotic injection schedule	56 h-13 days 4 h mean with $SD = 118 \pm 32.4$ h	94.05	_	319
316	16–19 weeks	Tetanus toxoid 2 cc single intra-amniotic injection schedule	76 h-26 days 4 h mean with $SD=14 h\pm 48.6 h$	88.61	36 fetuses all above 16 weeks	280 dead fetuses
34	20 weeks and above	Tetanus toxoid 2 cc single intra-amniotic injection schedule	116 h–28 days 4 h mean with $SD = 186 \pm 36.4$ h	73.67	All fetuses living	_

Table 4.1 Intra amniotic tetanus toxoid single injection 2 cc series cut off period for success rate in 30 days

group and eventually to 73.67 % in the 20 weeks or more group. But after randomization of gestation from 8 to 20 weeks the overall success rate, after single intra amniotic tetanus toxoid 2 cc injection, was 92.80 % in the present study.

It appears from Table 4.2 that with a multiple injection schedule of 2 cc intra amniotic tetanus toxoid weekly among the non-aborters there were two cases of failure on the 21st day, i.e., 8.33 % failure (even with multiple injection). It may be mentioned here that in the multiple injection schedule, certain associated problems can be found. Apart from inconvenience to the patients, there is a distinct possibility of fetomaternal exchange. If the fetomaternal exchange is 1 % with the USG amniocentesis protocol, it eventually becomes 3 % with the third prick/attempt on the non-aborters. In case of clinically guided amniocentesis (on the side of the fetal limbs or behind the fetal back in a bigger fetus) the cumulative percentage of fetomaternal exchange would be around 10 % or even slightly more. Under these circumstances, ascertaining the specific effects of tetanus toxoid would be difficult, leaving aside the problem of immune paralysis, because the effective immune response of an antigen requires a synchronization of a number of different physiological and immunological events. The immune system is controlled and

regulated by the genetic background. The specific responses of all antigens are determined by the immune response (Ir) gene and the immune suppressor gene (Is) with the development of its peripheral systems and sub systems including control of protein synthesis, located at the major histocompatibilty complex. Another problem in the adult or mature system is the nutritional status of the individual. Further, we are not certain about the receptor expression in the different afferent immune arch of the growing fetus. Seen from a phylogenetic angle, the embryonal and fetal growth in vitro shows a wide species variation so far as the afferent immune arch is concerned. Summing up, it appears that to generate an ideal immune response, an antigen (vaccine) should have a few general requirements like the (a) activation of antigen presenting cells to initiate the antigen processing, (b) the activation of both B and T cells to provide memory cells, (c) the generation of response to a variety of epitopes, (d) the persistence of the antigen so that the immune system remains stimulated for a long time. And most importantly, (e) safety.

All ante-natal mothers globally receive routine intra-muscular vaccination of tetanus toxoid. In this study, the objective was to observe the impact of tetanus toxoid which is an antigen, when it is injected intra-amniotically or

		•	,	-		····· · · · · · · · · · ·			
				1st injection to		2nd injection to		3rd injection to	
			1st injection to wait till day	wait till day	2nd injection to	wait till day	3rd injection to wait till	wait till day	
			wait till day 1–7 1–7	1–7	wait till day 8-14	8-14	wait till day 15–21	15-21	
	No. of		Success within		Success within		Success within		
Age	cases	Weeks of gestation	7 days	Failure	14 days	Failure	21 days	Failure	Comments
18-30	24	Range 8–20 weeks	11 (45.8 %)	13 patients	11 + 9 = 19	5 patients	11 + 8 + 3 = 22	2 patients	
mean age in years with SD 28.2±1.4		gestation mean with SD= 16.6±1.4 week			79.1 % cumulative success rate after 2nd injection within 14 days		91.6% cumulative success rate after 3rd injection within 21 days		

Table 4.2 The effect of intra amniotic multiple 2 cc injection 7 days apart on the growing human fetus up to 20 weeks

intra-fetally in an immature or a growing immune system on the growth, maturation and development of the fetus. It may be mentioned here that medical termination of pregnancy is legally permissible in India up to 20 weeks for the purpose of family planning. A number of government hospitals provide this service free of cost. As such, many mothers come to these hospitals for termination of pregnancy, which is done on a regular basis along with suggestions of other ways of prevention of pregnancy in future. Standardized procedures are followed like prostaglandin injection/vaginal application/oral application apart from hypertonic solutions like urea, manitol, saline etc. If any problems occur or there is failure of abortion or anaphylactoid reaction, patients are given the option of intra-amniotic antigen experimentation. Tetanus toxoid is the commonly used antigen because of its safety profile, easy availability and low cost.

Clostridium tetani is the causative agent of tetanus which is one of the important species of the genus clostridium, which comprises of spore forming gram positive anaerobic bacilli. Two products liberated by C. tetani are the classical neurotoxin (tetanospasmin) and haemolysin (tetanolysin). The haemolysin is heat liable and inactivated by oxygen. All the symptoms in tetanus are attributable to an extremely toxic neurotoxin. Tetanus toxin is produced in vitro in amounts up to 5–10 % of the bacterial weight. The purified neurotoxin is a simple protein containing no carbohydrate and with a molecular weight of 150,000 daltons. The toxin exists in two stages: the toxin monomer and a dimer of about twice the molecular weight. The dimer is non-toxic but antigenic. Treatment of this toxin with formaldehyde results in polymerization of toxin and results in the formation of toxoid which is nontoxic but antigenic. In the detoxification procedure, a standard hypertoxigenic strain (Harvard strain) of C.tetani is used. In case of an adult/ baby, the number of Lf of toxoid should not exceed 25 Lf for primary immunization. The immunization of all women in the child bearing age in tetanus endemic areas has been recommended by the World Health Organization

(WHO). There is no specific time period of pregnancy when women should be immunized against tetanus. In those countries where the risk of tetanus neonatorum is low, usually the immunization is deferred during pregnancy. It has been seen that immunization during the fifth and eighth months of pregnancy results in the formation of antibodies in the infant and also enhances the response of these infants to subsequent immunization. This phenomenon has been termed as transplacental immunization.

Though extremely rare, certain adverse effects of tetanus vaccination have been reported viz., local reactions like swelling, redness and pain up to 10 days at the site of injection; then there are systemic reactions like pyrexia, myalgia, malaise, acute anaphylaxis, peripheral neuropathy, elevated IGE level and elevated Anti A and Anti B antibodies. Very occasionally a reaction has been seen in individuals who had high levels of circulatory anti toxin and in whom bolstering was attempted. Tetanus toxoid has been reported to increase Anti A and Anti B antibodies owing to the traces of blood group antigen in the vaccine. Immunization of patients (pregnant) with the aim of preventing neonatal infection may therefore increase the risk of hemolytic disease of the newborn [1].

With this backdrop of information in mind, let us now explore our results. The problem is what should be the ideal dosage for embryonal and fetal immunization through the intra amniotic route. As we do not have a set of rules to follow and there is prevailing confusion over the degree of fetal and amniotic fluid exchange of secretions and excretions and the subsequent dilutions and exchanges with the maternal compartment, the problem is how much antigen injected through the intra amniotic route can cause stimulation of the fetal system (a) non-specifically and (b) specifically, with the progressive maturation and the gradual development of the fetal immune system. This is a most important gap in the existing state of knowledge.

Tables 4.3, 4.4, 4.5, 4.6, and 4.7 demonstrate the effects of intra-amniotic tetanus toxoid injection, varying from $\frac{1}{2}$ to 4 cc (single dose).

Mean age in years with SD	Mean parity with SD	No. of cases	Weeks of gestation range	Antigen tetanus toxoid	Induction abortion interval range	Comments cut off period 14th day
28.6±4.4	2.8±0.6	20 patients	9–19 weeks. Mean gestation with SD 13.8±1.2	¹ ∕2cc tetanus toxoid (single injection)	48 h–13 days 2 h	12 patients did not abort on the 7th day, 6 patients did not abort on the 14th day

Table 4.3 Intra amniotic tetanus toxoid with changes in dosage (single ½cc age parity injection) schedule

Table 4.4 Intra amniotic tetanus toxoid 1 cc series (single injection)

Mean age in years with SD	Mean parity with SD	No. of cases	Weeks of gestation range	Antigen tetanus toxoid	Induction abortion interval range	Comments cut off period 14th day
32.6±1.4	3.4±1.2	14 patients	8–19 weeks. Mean with SD 14.4±2.4	1 cc tetanus toxoid (single injection)	72 h–11 days 14 h	10 patients did not abort on the 7th day, 4 patients did not abort on the 14th day

Table 4.5 The effect of intra amniotic 1.5 cc tetanus toxoid single dose on the growing human fetus

Mean age in years with SD	Mean parity with SD	No. of cases	Weeks of gestation range	Antigen tetanus toxoid	Induction abortion Interval range	Comments cut off period 14th day
33.8±1.8	2.8±0.6	16 patients	9–19 weeks. Mean with SD 12.6±1.2	1.5 cc tetanus toxoid (single injection)	68 h–11 days 12 h	6 patients did not abort on the 7th day, 4 patients did not abort on the 14th day

Table 4.6 The effect of 3 cc single intra amniotic injection of tetanus toxoid on the growing human fetus in the below table

Mean age in years with SD	Mean parity with SD	No. of cases	Weeks of gestation range	Antigen tetanus toxoid	Induction abortion interval range	Comments cut off period 14th day
34.2±2.8	3.6±1.2	10	8–18 weeks. Mean with SD 14.6±2.8 weeks	3 cc tetanus toxoid (single) injection intra-amniotically	112 h–12 days 16 h	6 patients did not abort on the 7th day, 3 patients did not abort on the 14th day

If we analyze the gestation randomized results of Tables 4.3, 4.4, 4.5, 4.6, and 4.7, we find that the abortion failure rate on the seventh day with $\frac{1}{2}$ cc tetanus toxoid is 60 % (n=20); with 1 cc the failure rate on the seventh day is 71.42 % (n=14)

and with 1 $\frac{1}{2}$ cc tetanus toxoid, the failure rate on the seventh day is 37.5 % (n=16) and with 2 cc tetanus toxoid the failure on the seventh day is 47.2 %, and again with 3 cc tetanus toxoid dosage the failure rate is 60 % (n=10) on the seventh

Table 4.7	The effect of single intra-amniotic 4 cc tetanus toxoid on the growing human fetus as shown in the table
below	

Mean age in years with SD	Mean parity with SD	No. of cases	Weeks of gestation range	Antigen tetanus toxoid	Induction abortion interval range	Comments cut off period 14th day
29.6±2.8	2.8±0.6	12 cases	8–19 weeks. Mean with SD 14.6±2.4	4 cc single injection intra-amniotically	110 h–13 days 12 h	8 patients did not abort on the 7th day, 4 patients did not abort on the 14th day

day; again with 4 cc tetanus toxoid dosage the failure rate on the seventh day is 66.6 % (n=12). Now if we analyze the cumulative failure on the 14th day we will get the result of failure of abortion with 1/2 cc of tetanus toxoid on the 14th day single injection to be 30 % (Table 4.3). With 1 cc tetanus toxoid intra amniotic injection the cumulative failure rate on the 14th day is 42.85 %. With 1 1/2 cc of tetanus toxoid intra amniotic regime of single injection the cumulative failure on the 14th day is 25 %. However, with 2 cc of tetanus toxoid injection the cumulative failure rate is 27.6 %. With 3 cc intra amniotic tetanus toxoid regime, the cumulative failure of abortion on the 14th day is 30 % and lastly, in the 4 cc single intra amniotic antigen injection cases, the cumulative failure of abortion on the 14th day is 33.3 %. In all such cases from Tables 4.3, 4.4, 4.5, 4.6, and 4.7, the gestation period has been randomized (8.9-19 weeks from LMP). If the results are compared from 1/2 to 4 cc of single dose of tetanus toxoid with the multiple injection schedule of Table 4.2, it appears that there is a marginal superiority (n=24) with multiple injection of Table 4.2 where cumulative success rate is 79.1 % (n=24) on the 14th day as against 72.4 % with 2 cc single injection. However, this inference is too simplistic, because then we have to ignore the cumulative possibility of fetomaternal hemorrhage as a sequel to each injection and the immaturity of the memory cells of the fetus.

During discussions with Prof. Arnold Klopper of the Royal College of Gynaecologists and Obstreticians at the time of his visit to our laboratory on 18th January 1980, Prof. Klopper suggested that the unpredictability of fetal stimulation

can be overcome with intra-fetal injection under ultrasound control [4]. As a matter of fact this event appears to be the first attempted vaccination of the human fetus in utero. Here, too, there are certain technical problems, e.g., fetal movement, and we presume there may be even some regurgitation of the antigen to the amniotic cavity depending on the site of injection, needle bore, fetal movements etc. Similarly though we always tried to target the fetal back muscles for intrafetal injection, inadvertent injury to fetal vital structures like liver, lung and surrounding structures cannot be ignored.

The following Tables depict the results of dosage variation in the injection schedule (Table 4.8).

Though the series is small (n=12) with single injection schedule, 50 % of those injected aborted within the seventh day and cumulative 83.33 % abortions were noted on the 14th day, which is possibly the highest performance, even better than with the 2 cc schedule 72.4 % success and multiple injection schedule 79.1 % success rate on the 14th day. However, here too, interpretation on the basis of simple statistics would be too simplistic because we have to ignore then the effect of injury to the fetal vital structure with the antigen injection process and immaturity of the fetal immune system.

In one experiment, we wanted to see whether the addition of a fetotoxic or immunomodulatory substance like Vitamin A in the dosage of 300,000 IU if injected with tetanus toxoid can change the scenario. Here again we found that in 70 % cases there was failure of abortion on the seventh day and a cumulative 40 % abortion failure was noted on the 14th day (Table 4.9).

Mean age in years with SD	Mean parity with SD	No. of cases	Weeks of gestation range	Antigen tetanus toxoid	Induction abortion interval range	Comments cut off period 14th day
26.64±2.6	3.2±1.2	12 patients.	12–18 weeks. Mean gestation with SD 12.6±1.4	Single intra fetal injection of 2 cc under USG'	68 h–11 days 14 h	6 patients did not abort on the 14th day, 2 patients did not abort on the 14th day

Table 4.8 The effect of direct 2 cc intra-fetal injection under ultrasound guidance on the growing human fetus up to 18 weeks

Table 4.9 The effect of an immunomodulation and fetotoxic substance, vitamin A and 2 cc of tetanus toxoid on the growing human fetus up to 18 weeks

Mean age in years with SD	Mean parity with SD	No. of cases	Weeks of gestation range	Antigen tetanus toxoid	Induction abortion interval range	Comments cut off period 14th day
24.6±2.6	2.6±0.4	10	8–18 weeks. Mean with SD 14.6±2.4	Single intra amniotic injection of 2 cc tetanus toxoid and 300,000 IU of Vitamin A	48 h–13 days 12 h	4 patients did not abort on the 14th day, 7 patients did not abort on the 7th day

Intra-amniotic tetanus toxoid 2 cc+300,000 IU of Vitamin A

Table 4.10 Intra amniotic tetanus toxoid 2 cc and oral levamisole (150 mg daily) to mother for 7 days uninterrupted

Mean age in years with SD	Mean parity with SD	No. of cases	Weeks of gestation range	Antigen tetanus toxoid	Induction abortion interval range	Comments cut off period 14th day
28.6±1.4	2.9±1.9	12	8–18 weeks mean with SD 13.8±0.8	Tetanus toxoid 2 cc with levamisole orally to mother	49 h–11 days 14 h.	7 patients did not abort on 7th day. 3 patients did not abort on the 14 th day

Table 4.11 The effect of maternal non-immunization and subsequent intra amniotic instillation of 2 cc tetanus toxoid on the growing human fetus up to 20 weeks

Mean age in					Induction	
years with	Mean parity	No. of	Weeks of gestation	Antigen tetanus	abortion interval	Comments cut off
SD	with SD	cases	range	toxoid	range	period 14th day
34.2±2.8	3.6±1.2	16	8–19 weeks mean with SD 14.2 ± 1.4 weeks	2 ml of single fixed dose of intra-amniotic injection	51 h–8 days 12 h	6 patients did not abort on 7th day. 4 patients did not abort within 14th day

Intra amniotic tetanus toxoid without maternal immunization of tetanus toxoid

In order to understand whether maternal T cell stimulation or modulation of T cell function pharmacologically can make the induction of abortion interval more predictable, we utilized oral levamisole (150 mg OD \times 7 days). Here 58.33 % cases failed to abort on the seventh day and 25 % cases failed to abort on the 14th day (Table 4.10).

Mean age in years with SD	Mean parity with SD	No. of cases	Weeks of gestation range	Antigen tetanus toxoid	Induction abortion interval range	Comments cut off period 14th day
33.4±1.4	3.6±1.2	14	8–19 weeks mean gestation with SD 13.6±1.4 weeks	2 cc of intra amniotic tetanus toxoid and oral cimetidine 400 mg BID to mother	52 h–13 days 12 h	8 patients did not abort on 7th day. 6 patients did not abort within 14th day

Table 4.12 Intra amniotic tetanus toxoid 2 cc along with oral cimetidine 400 mg BD to mother for 7 days uninterrupted

Table 4.13 The effect of intra amniotic 1 cc glutamate BCG on the growing human fetus up to 15 weeks

Mean age in years with SD	Mean parity with SD	No. of cases	Weeks of gestation range	Antigen tetanus toxoid	Induction abortion interval range	Comments cut off period 14th day
33.3±1.4	3.4±1.2	26	11–15 weeks mean gestation with SD 13.2±3 weeks	1 cc of glutamate BCG	No abortion varying degree of dissolution vis-à-vis auto absorption of the fetus on the 14th day	Mothers who were Mantoux test negative were enrolled. Mantoux test positive and ELISA Tb IgA, IgG and IgM (+) cases were discarded from study design

The Table 4.11 shows mothers who were not immunized with tetanus toxoid. We wanted to see the role of maternal immunization on the induction-abortion interval of the growing human fetus before 20 weeks, when challenged by tetanus toxoid through the intra-amniotic route. Here, we find 62.5 % cases aborted within 7 days and 75 % cases of abortion within 14 days. Excepting for these 15 subjects in the above Table, all other mothers were properly immunized with tetanus toxoid.

In Table 4.12, the results of mothers who were adviced to take cimetidine (H2 receptor antagonist) 400 mg BD dosage along with intra amniotic tetanus toxoid 2 cc injected are shown. Here also we found that eight cases did not abort on the seventh day (57.14%) and six cases did not abort on the 14th day (42.85 % failure).

In Table 4.13, we have shown our experience with intra amniotic instillation of BCG. All commercial BCG vaccinations owe their parentage to the culture developed by Calmette and Guerin about 60 years ago. Subsequently daughter strains were named differently such as Montreal, Connaught, Glaxo and so on. The standard dose of BCG vaccine is 0.1 mg in 0.1 ml volume injected intradermally. The protective efficacy of BCG vaccination has been the subject of considerable controversy. There have been nine major internationally controlled trials with BCG vaccination in various parts of the world during the past 60 years. While highly significant protection (70-80%) has been obtained in several trials conducted in Europe, Canada and the USA this has been offset by a report of an almost total lack of protection (0-37 %) in Puerto Rico and Chengleput (South India) [1]. There are many explanations of which the most plausible is that the immune reactivity of the population had been conditioned by prior exposure to the mycobacterium in the environment. Contra indication for inclusion in the above Table for BCG trial is Mantoux test positivity. (Later, as per Ethical Committee suggestions, ELISA Tb IgG, IgM,

Mean age in years with SD	Mean parity with SD	No. of Cases	Weeks of gestation range	Antigen tetanus toxoid	Induction abortion Interval range	Comments
25±2.4 years	2.8±6	14	10 to 16 weeks. Mean gestation with SD 13.4 ± 1.4	0.5 cc of dyphteria and 0.5 cc antigen commonly known as double antigen	88 h–12 days 12 h	6 patients aborted within 7 days. 10 patients cumulatively aborted within 14 days

Table 4.14 The effect of intra amniotic double antigen challenge on the growing human fetus up to 20 weeks

IgA studies were made mandatory before inclusion in this trial).

As we have stated earlier, due to the prevailing unpredictability of the actual amount of antigens entering the fetal circulation due to several factors, we used 1 cc of glutamate BCGH intra amniotically in all the cases (Table 4.13) irrespective of gestational period, i.e., 11–15 weeks in this study. To our utter surprise there was no abortion. However there was progressive diminution of the size of the uterus. When we did hysterotomy on the 14th day after the injection, the uterus was found to be practically free of the conceptus. We repeated the experiment again and again, but the result remained the same on the 14th day of the instillation of BCG. Twenty-six such cases were experimented with and the gestation period varied from 11 to 15 weeks. However, when analyzing the data, the question arises why pregnancies above 15 weeks were not included. The answer to this question is the restriction suddenly imposed by the Institutional Ethical Committee on BCG experimentation on the plea that persistence of placenta or any other fetal tissue in the maternal system may have the possibility of inducing hydatidiform mole or choriocarcinomatous changes to mothers in the long run.

In the animal system, auto absorption is seen in the conceptus frequently, especially in cases of multiple pregnancies. In human pregnancy we have frequently noted a vanishing twin or a part of the gestational triplet or quadruplet. There, placental vascular factors have been primarily or secondarily attributed as the cause. However, there are published reports of successful pregnancy after selective termination of multiple pregnancy with KCL or subsequent normal pregnancy after treatment of ectopic pregnancy with methotrexate. Hence, there should not be undue fear about this novel approach of fetal dissolution which greatly minimizes the maternal inconvenience of expulsion of the fetus.

In Table 4.14, the effect of 1 cc of double antigen containing dyptheria toxoid 0.5 cc and pertussis 0.5 cc intra amniotically in 14 cases of pregnancy with gestation varying from 10 to 16 weeks was observed. Here 42.85 % failure was noted on the seventh day and cumulative 28.51 % cases of failure on the 14th day.

The pathogenesis of diptheria is attributed to the release of a potent exdotoxin by the bacteria with a molecular weight of 55,350 dalton and with the help of trypsin it can be fragmented into two dissimilar fragments called A and B. Fragment A has a molecular weight of 21,150 daltons and if a single molecule gains entry into a cell it can catalyze ADP ribosylation of Elongation Factor 2 (EF2) using its nicotinamide adenine dinucleotide as a substrate. Ribosylation of EF2, which is essential for protein synthesis, leads to inhibition of proteins synthesis and death of cell which clinically manifests as necrotic lesion of dyptheria. When the dyptheria toxin is converted into toxoid, the immunogenicity is retained but the virulence is lost; however there may be rare reactions in the form of Type I (immediate type) hypersensitivity and delayed hypersensitivity [1].

The causative agents of pertussis are members of the genus Bordetella: they are small gram negative coccobacilli which are obligatory respiratory tract pathogens of warm blooded animals including birds. These are unable to survive outside their hosts. Four species are recognizable

Mean age in years with SD	Mean parity with SD	No. of cases	Weeks of gestation range	Antigen tetanus toxoid	Induction abortion interval range	Comments
28.2±2.4 years	2.8±6	14	11–18 weeks. Mean gestation with SD 14.6±1.2	2 cc 20 % BSA antigen	77 h–12 days 4 h	8 patients did not abort on the 7th day. 4 patients did not abort on the 14th day

Table 4.15 The effect of intra amniotic challenge of 2 cc 20 % bovine serum albumin on the growing human fetus up to 18 weeks

Table 4.16 The effect of maternal blood injection intra amniotically on the growing human fetus up to 14 weeks

Mean age in years with SD	Mean parity with SD	No. of cases	Weeks of gestation range	Antigen schedule	Induction abortion interval range	Comments
24.4±2.2 years	2.8±6	10	8–14 weeks. Mean gestation with SD 13.4±1.2	10 cc of maternal whole blood.	72 h–12 days 6 h	6 patients did not abort on the 7th day. 3 patients did not abort on the 14th day

and all of them excepting B. avium are pathogenic for human beings and can cause pertussis. B. pertussis is the most common agent for whooping cough. The pertussis toxin is synonymous with earlier known terms like histamine sensitizing factor, lymphositosis promoting factor, haemagglutinin, islet activating protein and pertussiger. It has a molecular weight of 117,000 daltons and consists of six polypeptides and two units, viz., a unit which is enzymatically active and the remaining five polypeptides cause the B unit which is needed to cross the membrane, eventually entering the cell and increasing the cAMP concentration and disturbing the normal cell function [1].

In the experiment relating to Table 4.15 a very simple antigen, namely 20 % BSA 2 cc, was injected intra amniotically in 14 cases with gestation varying from 11 to 18 weeks. Here 57.14 % of the cases did not abort on the seventh day and cumulatively 28.57 % cases did not abort on the 14th day.

We were extremely puzzled with the result. Initially we thought that the abortion is antigen specific, i.e., tetanus toxoid specific. Then in the course of our experimentations we concentrated on making the effect of induction to abortion predictable, i.e., abortion predictable within a fixed time. We did not achieve this goal, even after multiple injections or intrafetal injection of reducing the induction-abortion interval. Therefore we changed the antigen to BCG and saw a different effect altogether, i.e., fetal dissolution auto absorption. Due to ethical committee restrictions we used other antigens like double antigen (dyptheria and pertussis) without any gross variation in the result. However, even with BSA, there was abortion, thus making the effect of tetanus toxoid non specific in nature.

Average fetal blood is 110 cc/kg which is approximately 50 % higher than the adult blood volume (75 cc/kg). Approximately 30 % of this circulating fetal blood resides in the umbilical cord and placenta. Thus, after excluding this blood the fetus has a net blood volume of approximately 80 cc/kg which is similar to the weight of normalized blood volume in lean adult humans.

If we look at Table 4.16, here too there is randomization of the weeks of gestation from the 8th to the 14th week. Here 10 cc maternal blood was injected into the amniotic fluid in ten cases. On the seventh day there was abortion in 40 % cases

					Induction	
Mean age in years with SD	Mean parity with SD	No. of cases	Weeks of gestation range	Antigen schedule	abortion interval range	Comments cut off period 14th day
24.4±2.2 years	3.1±1.6	16	9–15 weeks. Mean gestation with SD 12.6±0.8 weeks	Buffy coat 10 cc of maternal blood	77 h–11 days 2 h	11 patients did not abort on the 7th day. 8 patients did not abort on the 14th day

 Table 4.17
 Intra amniotic leucocytes of 10 cc maternal whole blood

Table 4.18 The effect of 5 cc freshly collected allogeneic amniotic fluid injection on the growing human fetus up to 16 weeks

Mean age in years with SD	Mean parity with SD	No. of cases	Weeks of gestation range	Antigen schedule	Induction abortion interval range	Comments cut off period 14th day
26.4±2.8 years	3.3±1.2	12	12–16 weeks. Mean gestation with SD 14.0±1.8 weeks	5 CC of allogeneic amniotic fluid of the allogeneic mother	120 h–9 days 4 h	5 patients did not abort on the 7th day. 7 patients did abort on the 14th day

 Table 4.19
 Intra amniotic steroidin (Bacterial polysaccharide)

Mean age in years with SD	Mean parity with SD	No. of cases	Weeks of gestation range	Antigen schedule	Induction abortion interval range	Comments cut off period 14th day
25.6±1.4 years	3.4±1.2	10 cases	9–18 weeks. Mean gestation with SD 13.6 ± 2.4 weeks	Bacterial antigen (polysaccharide 2 cc)	56 h–11 days 8 h	6 patients did not abort on the 7th day. 4 patients did abort on the 14th day

The effect of 2 cc of bacterial polysaccharide injection on the growing human fetus up to 18 weeks

and on the 14th day 70 % cases aborted. Strangely, it appears that very little maternal blood (10 cc), can induce abortion in 70 % of the cases. This raises concern regarding certain serious questions of the critical amount of fetomaternal exchange in normal pregnancy and also regarding the problems of surgery on the unborn in the early weeks from the point of view of inadvertent fetomaternal exchange and the triggering off of abortion.

The effect of intra amniotic injection of buffy coated leucocyte collected from 10 cc maternal blood on the growing human fetus up to 15 weeks is shown in the Table 4.17.

If we analyze the above Table we see that with gestation varying from 9 to 15 weeks, injection of

freshly prepared buffy coat of 10 cc of maternal blood, once, intra amniotically can trigger abortion (gestation randomized) in 31.25 % cases on the seventh day and cumulatively 50 % within 14 days.

In the experiment recorded in Table 4.18, we injected 5 cc of allogeneic amniotic fluid (single injection) to 12 cases carrying pregnancy from 12 to 16 weeks. Here too, seven cases (58.83 %) aborted within 14 days cumulatively and five cases (41.66 %) aborted within the seventh day.

In Table 4.19, we have used bacterial polysaccharide antigen 2 cc (steroidin) intra amniotically once in ten cases carrying gestations from 9 to 18 weeks; however, with stimulation of polysaccharide bacterial antigens, 60 % cases did

Mean age in years with SD	Mean parity with SD	No. of cases	Weeks of gestation range	Antigen schedule	Induction abortion interval range	Comments cut off period 14th day
23.8±4.4 years	2.8±4	106	10–20 weeks. Mean gestation with SD 14.8±1.8 weeks	2 cc of normal saline with thiomersol and aluminium phosphate	No abortion within 1 month in 102 patients. 4 patients which aborted on the 18th hour after injection and 42 h injection, 116 h and 22 h	All 4 patients had hype pyrexia and incomplete abortion. Amniotic fluid bacterial culture suggested haemolytic streptococcus and staph. Aureus in 3 cases, cervical culture positive for E.Coli in one case

Table 4.20 To examine the effect of preservative and adsorbent in identical dosage in 2 cc blank (normal saline) on the growing human fetus

Intra amniotic 2 cc of normal saline (with aluminum phosphate 0.01% and thiomersol 200 microgram) in the strength of tetanus toxoid

not abort on the seventh day and 40 % cumulative failure was noted on the 14th day.

As with other experiments discussed earlier, whenever there was slight/overt hemorrhagic tap during amniocentesis, the case was discarded from this study. In four cases there were incomplete abortions due to inadvertent infection and amniotic fluid culture was positive for haemolytic streptococcus, staph aureus etc. in three cases and cervical swab culture for E.Coli was positive in one case.

In Table 4.20, we have included 114 control cases where we brought specially prepared antigen blank ampoules from Chowgule e Hind, India Ltd, containing 2 cc of normal saline with thiomersol 0.01 % and aluminium phosphate 1 mg/ml. We waited after intra amniotic injection for 1 month to see if the injection prick itself or fetomaternal micro-exchange or maternal-fetus exchange or inadvertent trauma induced more exchange, i.e., fetomaternal transfusion or maternal fetal exchange, can cause abortion; and whether adjuvants like aluminium phosphate and preservatives like thiomersol can cause abortion. Eight cases were discarded from the study due to hemorrhagic tap and out of 106 cases there was abortion in 4 cases (3.77%). We are not precisely in a position to say whether fetomaternal exchange through the vent of placental or other fetal tissue at the site of amniocentesis or contamination of infection inadvertently or hematogenous infection from preexisting maternal source or the adsorbent, aluminium phosphate itself, either singularly or in combination with preservative thiomersol, triggered the abortion process. The contributory role of materno-fetal transfusion is also not precisely known. And the last factor is that most of the control cases were injected in the year 1977–1984 when we did not have USG support to guide the amniocentesis.

In this connection, the reasons for *not using viral antigens*, may be mentioned. The institutional Ethical Committee suggested that the technical possibility/potentiality that attenuated virus may revert back to virulence in the hypoimmune or semi-immune or protected environment of the fetus. At least in case of rubella, the vaccine virus is capable of crossing the placenta and has been isolated from aborted fetal tissue from sites like the eye, kidney, bone marrow and the decidua. Whether the vaccine virus can damage the fetal tissue, as wild virus does in the case of adult tissue, is yet to be investigated.

In the experiment recorded in Table 4.21, 15 cases were included who opted for a second consecutive termination with tetanus toxoid. Here we find 86.66 % success rate on the 14th day. If we compare this with the first abortion scenario, the induction abortion interval does not show any gross variation.

Mean age in years with SD	Mean parity with SD	No. of cases	Weeks of gestation range	Antigen schedule	Induction abortion interval range	Comments cut off period 14th day
29.6±4.2 years	3.1±0.8	15	13–16 weeks. Mean 13.6±1.4 weeks	2 cc of single fixed intra-amniotic injection	47.8 h–13 days 2 h (success rate is 86.66 %)	6 patients did not abort within 7 days and 2 patients did not abort within 14 days

Table 4.21 The impact of 2 cc tetanus toxoid on consecutive second pregnancy termination attempt through intraamniotic route

Analysis of Table 4.22 shows that all the antigens used in the experiment in a fixed single injection schedule showed certain symptoms like aches and pain all over the body with or without mild pyrexia. This pyrexia mostly subsided after abortion, but very rarely it continued, viz., in tetanus toxoid single injection schedule, 112 cases (16.56 %) had aches and pain all over the body preceding abortion; and in 63 cases (9.15 %) the temperature was more than 99 °F, while in 1.01 % cases (7 cases) the temperature continued even after abortion. In control cases with normal saline, and adsorbant and preservative the incidence of pyrexia was 3.77 % (4 cases out of 106 cases). In the control cases where pyrexia continued even after abortion the amniotic fluid culture was found to be positive for pathogenic bacteria in three cases and cervical culture was positive in one case. In case of other antigens like sterodin, 20 % BSA, pertussis and dyptheria toxoid injection, the incidence of aches and pain all over the body varied like (a) 60 %, (b) 28.57 % and (c) 21.42 % respectively. With (d) whole maternal blood injections, (e) allogenic amniotic fluid injection, and (f) buffy coat leucocyte intra amniotic injection, the incidence of aches and pain varied, i.e., 40 %, 25 % and 31.25 % respectively. The overall incidence of aches and pain, if we randomize, was around 16.74 %. Now the question may be asked why there were aches and pains all over the body preceding abortion. In this connection certain trends may be noted.

From the onset of aches and pain all over the body the abortion takes place within 6–36 h. (mean 16.4 ± 4.6 h SD) and the abortion is inevitable. (As per WHO Prostaglandin Task Force

Data Form, we gave a standard instruction to the on duty nursing staff not to give aspirin or other NSAIDs drug unless the pain was unbearable, for symptomatic relief, because of the theoretical possibility of delayed abortion and incomplete abortion). While the aetiology of aches and pains and temperature is not known to us in most of the cases as yet, the analogy lies in case of aseptic inflammation of myocardial infarction, where we find temperature due to release of cellular damaged products in the circulation like lysosomal enzymes etc.

A second question may be raised as to why aches and pains were present only in 16.74 % in all antigens. The answer is not straightforward. In the present series, most patients who volunteered for antigenic termination came from the lower socioeconomic strata, who were financially unable to afford prostaglandin or other methods. It has been found that such patients have a wide variation in the subjective tolerance of pain and ability to report its incidence to the attending nurse. We have included those cases where the pain was severe, i.e., unbearable, necessitating analgesic support like pethidine/morphine/pentazocin etc. In case of the BCG group this aches and pain syndrome necessitating request for analgesic support is noted in 79.92 %. Whether absorption of the fetal tissue, like the lysosomal content, is the cause of this generalized body ache is an important unresolved question for us. In this connection, during our limited unpublicized experience, two cases of selective fetal killing by (KCL) injection in the fetal heart (through USG) may be cited which also caused aches and pain to the mother within 48 h of fetal death.

Antigen used	No. of cases	Mean gestation	Aches and pain all over the body preceding abortion	Pyrexia	Pyrexia subsided after abortion	Pyrexia continued after abortion	Rupture of the membranes
2 cc 20 % BSA	14	14.6±1.2	4 (28.57 %)	4	3	1	1
1 cc Dyptheria with pertussi	14	13.4±1.4	3 (21.42 %)	3	2	Nil	1
10 cc of maternal blood	10	13.4±1.2	4 (40 %)	2	2	Nil	Nil
Buffy coat of 10 cc maternal blood	16	14.4±2.6	5 (31.25 %)	3	2	1	1
1 cc of Glutamate BCG single injection	26	13.2±0.8	Instead of abort the fetus (76.92) degree)				auto absorption of on (varying
Single injection 2 cc tetanus toxoid	688	13.6±2.4	114 (16.56 %)	63	56	7	68* 70 fetuses were living at birth; all of them were 18 weeks or more. Not single fetuses aborted before 16 weeks and were living at birth
2 cc of normal saline with adjuvant and preservative	106	14.2±1.2					re suggested staph complete abortion
2 cc of bacterial antigen Sterodin	10	13.6±2.4	6 (60 %)	4	4	X	X** (One case had persistent bleeding after abortion necessitating 4 bottles of fresh whole blood transfusion and antibiotic and I.V. electrolyte solutions as a life saving gesture
			1				
5 CC amniotic fluid of the allogenic mother	12	14.2±1.2	3 (25 %)	x	X	X	1

Table 4.22 Showing the clinical events after intra amniotic antigenic challenge

Post Abortion Maternal Effect

Table 4.23 suggests post-abortion mild hemorrhage, i.e., 1–5 days of bleeding (85.5 %): postabortion moderate hemorrhage-6 to 10 days of bleeding (14.1 %); post-abortion severe hemorrhage -11 days or more of bleeding (0.4 %). Some blood loss during abortion is inevitable and this increases with the length of gestation. For the clinician there are two aspects of hemorrhage, i.e., an excess rate of blood loss or loss of an abnormally large volume of blood.

In all the cases the assessment is based on the subjective and objective coordination by the

No. of cases	Weeks of gestation	Mild hemorrhage	Moderate hemorrhage	Severe hemorrhage
32 Cases	8–11 weeks	21 cases	10 cases	1 case
336 cases	12-15 weeks	246 cases	88 cases	2 cases
186 cases	16–19 weeks	138 cases	48 8 cases	Nil
82 cases	20 weeks	78 cases	48 cases	Nil
82 cases	20 weeks and above	78 cases	3 cases	1 case

Table 4.23 The distribution of cases by the amount of post-abortion bleeding after intra amniotic tetanus toxoid termination

Table 4.24 The distribution of cases and histological correlationship of grade-II bleeding (5 days or more) after intra amniotic tetanus toxoid termination

Histopathology report after	Retained product of	Fragments of deciduas		
curettage	conception	only	Blood clots only	Others
48 cases	16	14	14	4

non-duty nursing staff and the resident doctor. In all cases, whenever the bleeding continued for more than 5 days, i.e., moderate bleeding, methergin (methyl ergometrin) support was started, and if inspite of methyl ergometrin support there was continuation of bleeding, evacuation of the uterus and a broad spectrum antibiotic (prophylactically) was advised (Ampicillin and Cloxacillin). In Table 4.23, out of 153 cases where the bleeding was more than mild, ergometrin support and clinical examination reduced the number of actual cases going for dilatation and evacuation (D & E) to 48 cases only. Clinical examination followed by ultrasound diagnostic support was utilized as and when necessary (Table 4.24).

The Table shows that out of 48 cases, 62.5 % of the cases showed some degree of conception material either in the form of retained product (16 cases, 33.3 %) or fragments of deciduas (29.16 %) at the time of histopathological examination.

Early resumption of normal menstruation visà-vis restoration of fertility, i.e., ovulation and reproductive function, is a serious concern for mothers who opted for abortion. The guilt and tension about abortion is largely abated with restoration of normal menstrual flow. For the clinician, the concern is the prevention of early complications, viz., hemorrhage, infection and uterine injury and not missing an ectopic pregnancy, leaving aside the problem of incomplete evacuation and infertility as late complications.

Interval		
between		
abortion and		
the onset of		
first menstrual		Gestation in weeks
cycle	No. of cases	(randomized)
21-30 days	146	In 76.2 % of the cases
31-40 days	136	menstruation returned
41-50 days	28	within 40 days after termination with
51-60 days	46	tetanus toxoid
61-above		
	Total cases	
	356 cases	
	followed up	

 Table 4.25
 Post-abortion maternal effect

Distribution of cases on the basis of the interval between abortion and resumption of 1st menstrual cycle onset, after 2 cc fixed dosage intra amniotic tetanus toxoid termination protocol

The following few Tables show the follow-up study of abortion with (single fixed dose 2 cc) tetanus toxoid, with references to the menstruation of the patient. Here, too, all the cases which were terminated with tetanus toxoid did not turn up for follow-up study and advice inspite of our request and assurance of free treatment facility.

Therefore in the above Table 4.25, it appears that 76.2 % of the cases resumed their menstruation within 40 days.

Table 4.26 indicates that 58.1 % of the patients did not have any change in their period (when compared with the pre-abortion status) and

Women followed up	196	(Cycles followed up: 784)
No change in menstruation	114 cases	=58.1 %
Regular to irregular cycles	16 cases	=8.8 %
Irregular to regular cycles	66 cases	=33.1 %

Table 4.26 Post-abortion maternal effect

Behavior of menstrual cycle for the first 4 cycles after termination with intraamniotic tetanus toxoid single fixed dosage. (2 cc)

Table 4.27 Post-abortion maternal effect

Women followed up	176	(Cycles followed up: 528)
No change in cycle	146 cases	=82.9 %
Regular to irregular cycles	8 cases	=4.54 %
Irregular to regular cycles	22 cases	=12.56 %

Showing changes in their menstrual cycle, i.e., regularity for subsequent cycles, viz. 5–7th cycle after abortion with intra amniotic tetanus toxoid single fixed dose of 2 cc

33.1 % cases showed improvement, i.e., became regular from irregular bleeding (before abortion).

In Table 4.27, we have noted the fifth to seventh cycles after abortion and compared these with pre-abortional menstrual cycles. 82.9 % of our patients did not perceive any gross change in their cycle.

Table 4.28 shows the duration of menstrual flow for the first four cycles. Here we have noted on day 1, variation or no change in 49.4 % cases, and 2 days variations (decrease) were found in 33.77 % cases. However, 16.43 % of the cases suggested 2 days variation (increase) in their menstrual flow.

Table 4.29 depicts that when analyzing the subjective amount of menstrual flow, 59.1 % cases showed no change in the menstrual bleeding after abortion and 40.7 % cases showed marginal changes, i.e., slight increase or decrease of the menstrual amount.

Table 4.30 shows a history of dysmenorrhoea in 29 cases (before abortion). Out of 196 cases who reported for follow up clinic, in 11 cases, the dysmenorrhoea continued after abortion. Three

Table 4.28 Post abortion maternal effect: showing the changes in duration of menstrual flow for the first 4 cycles after intra-amniotic 2 cc tetanus toxoid termination

Women followed up	196	(Cycles followed up: 528)
No change in or 1 day variation (increase or decrease)	97 cases	=49.4 %
Duration of period increased by 2 days or more	33 cases	=16.43 %
Duration of period decreased by 2 days or more	66 cases	=33.77 %

Table 4.29 Post abortion maternal effect

Women followed up	196	(Cycles followed up: 784)
No change in amount of periodic bleeding (increase or decrease)	116 cases	59.1 %
Increase in the amount of periodic bleeding	42 cases	21.4 %
Decrease in the amount of periodic bleeding	38 cases	19.3 %

Single fixed dose (2 cc) abortion with intra amniotic tetanus toxoid showing changes in amount of menstrual flow in the first 4 cycles

Table 4.30 Post abortion maternal effect

No. of cases followed up=196 (No. of cycles followed up=784

History of dysmenorrhoea before abortion	29 cases
Continuation dysmenorrhoea after abortion	11 cases
Development of dysmenorrhoea after abortion (no previous history)	3 cases

Showing incidence of post-abortional dysmenorrhoea in cases of intra amniotic tetanus toxoid termination with fixed single dose schedule. (2 cc)

cases who did not have a past history suggestive of dysmenorrhoea developed dysmenorrhoea after abortion with tetanus toxoid.

From Table 4.31, it appears that 32 cases, i.e. 16.32 % of the cases developed oligomenorrhoea to amenorrhoea after abortion with tetanus toxoid of which 2.04 % had amenorrhoea for 3 months follow up.

No. of cases followed up:	' 96	
No of cycles followed up	588	
Incidence of oligomenorrhoea	28	14.28 %
Incidence of amenorrhoea	4	2.04 %

 Table 4.31
 Post abortion maternal effect

Showing the incidence of amenorrhoea to oligomenorrhoea (subjective assessment in comparison with the prepregnancy bleeding) after termination with single fixed intra amniotic dosage (2 cc) injection protocol with tetanus toxoid, follow up for 3 months

Table 4.32 Post-abortion maternal effects

No. of patients studies=88, (Sequential-12th, 14th, 16th, 18th day folliculometry)

Ovulation present	51	57.95 %
Ovulation absent	15	17.04 %
Ovulation doubtful	22	25 %

Single fixed (tetanus toxoid) intra amniotic injection schedule and post-abortion follow-up showing detection of ovulation by sequential folliculometry on the first menstrual cycle

Table 4.33 Post-abortion maternal effects

)

Single fixed (tetanus toxoid) intra amniotic injection schedule and post-abortion follow-up showing folliculometry on the first 4 menstrual cycles (up to 7 months)

The above two Tables (4.32 and 4.33) show resumption of ovulation with tetanus toxoid termination (fixed dose schedule) as per the report of sequential USG examination from the 12th to the 18th day after period. Ovulation is noted in 57.95 % cases out of 88 cases who reported for this study in the first month of termination and 88.8 % cases had folliculometry suggesting ovulation within the first four cycles (up to 7 months).

Future Pregnancy Potential

The vast majorities who have a safe induced abortion experience suffer no long term effects on their general or reproductive health. The

Table 4.34 Post-abortion maternal effects. Future pregnancies follow up termination of a fetus with intra amniotic tetanus toxoid

No. of cases studies = 146 (po termination of pregnancy)	ost tetanus toxoid
a. IUGR	6 (4.10 %)
b. Gestosis (PET only)	14 (9.58 %)
c. Gestosis+IUGR ^a	4 (2.73 %)
d. Ectopic pregnancy	Nil
e. Placenta previa	Nil
f. Miscarriage	1
g. Pre-term labour	1

IUGR Intrauterine growth retardation, PET Pre-eclampsia toxemia

^aVery poor nutritional background and poor compliance with antenatal advice

exceptions are a small number of women who have severe, immediate or delayed complications like secondary infertility, ectopic pregnancy, miscarriage and a fear of adverse effects on the subsequent pregnancies like low birth weight, pre-term labour, fetal abnormalities, placenta previa, psychological effects and fear of cancer.

Table 4.34 shows a follow up of 146 cases with the age of the mother randomized. The incidence of idiopathic IUGR without any obvious reason is 4.10%, i.e., six cases. The incidence of specific conditions like edema (E) proteinuria (P), hypertension (H) syndrome specific to pregnancy (EPH Gestosis or toxemia of pregnancy) is 9.58 % of the total cases. Again another 2.73 % cases had IUGR with pre-eclamptic toxemia. Here too, any specific conclusion would be too simplistic because of age, parity, nutritional status randomization.

If we analyze Table 4.35, 26 cases were randomized who were enrolled for auto immune study. We have noted that 11.53 % cases had marginally positive anti cardiolipin antibody titre both IgG and IgM (positive) and 7.69 % cases had a positive anti nuclear antibody. Another 7.69 % cases had a positive report for lupus anti coagulant and 3.84 % cases had a positive anti double strand DNA antibody. However, except in one case where the complement reactive protein continued to be positive even after 3 months, the other auto immune parameters became negative. On closer scrutiny of past history and through

Auto immune study	No. of cases	Normal range	Positive result	Percentage
Complement reactive protein (CRP)	26	Less than 6 mg/ml	1	3.84 %
Anti nuclear antibody	26	1:40-1:80 above significant	2	7.69 %
Anticardiolipin antibody (IgG and IgM)	26	Anticardiolipin antibody. IgG up to 15 GPL units/ml		
IgM up to 12:5 MPL units/ml	2	11.53 %		
Lupus anti coagulant	26	Negative	2	7.69 %
Anti double strand DNA antibody	26	1:10 and above is significant	1	3.84 %

 Table 4.35
 Immuno-endocrino-hematological impact of intra amniotic antigenic challenge to the mother with tetanus toxoid

Post-abortion maternal effects: Follow-up immunological study after 1st pregnancy termination with tetanus toxoid showing residual reaction of the mother to this new method of abortion

Table 4.36 Auto immune profile of mother during her consecutive second pregnancy termination with intra amniotic tetanus toxoid showing long term reactions of the mother in case of 2nd pregnancy termination (consecutive with intra amniotic antigenic challenge tetanus toxoid)

Auto-immune study	No. of cases	Range (normal)	Result (positive)
Complement reactive protein (CRP)	12	Less than 6 mg/ml	2 (16.66 %)
Anti nuclear antibody	12	1:40–1:80 above significant	1* (8.33 %). Report became negative on 3 months follow up
Anticardiolipin antibody (IgG and IgM)	12	IgG up to 15 GPL unit/ml. IgM up to 12.5 MPL units/ml	1* (8.33 %). Report became negative on 3 months follow up
Lupus anti coagulant	12	Negative	1* (8.33 %). Report became negative on 3 months follow up
Anti DNA antibody (double strand)	12	1:10 and above is significant'	1* (8.33 %). Report became negative on 3 months follow up

clinical examination of the same patient, she was found to be suffering from rheumatic arthritis and her ASO titre was found to be high.

From the above experiment pertaining to Table 4.36, it appears after a second consecutive termination with tetanus toxoid, there was no residual unusual autoimmune activity in the mother; 8.33–16.66 % of the cases which showed a positive titre became negative at 3 months follow up.

If Table 4.36 is analyzed, 8.33 % had a positive anti nuclear antibody, anti phospholipid antibody (anti cardiolipin antibody), lupus anti coagulant, anti- DNA antibody (double strand) which became negative at the follow-up study in the third month. However, 16.66 % of the cases showed a positive complement reactive protein which also became negative in the 3 months follow up study. In the experiment relating to Table 4.37, we wished to examine whether sharing of DR/DQ triggers early abortion, i.e., within 72 h of first trimester cases only. Though the series was small (N=10), we did not find any correlation between the parents' HLA DR/DQ studies and early abortion. Hence, accidental or chance sharing of HLA DR/DQ is possibly not a cause of early abortion. Rather, the intra amniotic tetanus challenge may be the possible trigger for the abortion process.

In Table 4.38 we find maceration of the fetus more in the below 17 weeks gestation group, viz., 48 cases showed maceration (58.53 %) and 9 cases did not show any clinical maceration (10.97 %); here again the male fetuses were 66 % in the macerated group and 22.22 % in the nonmacerated group. In cases of fetuses above 17 weeks, 8 % cases were found to have maceration with 4 % each from each sex. **Table 4.37** Paternaland maternal HLADR/DQ study of the 1sttrimester early abortions(within 72 hours afterintra amniotic singledosage, fixed tetanustoxoid termination)

		HLA type	class II			
		DR	Locus	DQ	Locus	Others
Case I	Father	15 (27)	-	1	-	-
	Mother			2	-	DR 53
Case II	Father	3	_	1	_	DR 52
	Mother	7	17 (3)	-	-	DR 52
Case III	Father	15 (2)	17 (3)	1	2	DR 53
	Mother	7	_	3	-	DR 52
Case IV	Father	15 (2)	17 (3)	1	-	DR 53
	Mother	7	_	2	-	DR 52
Case V	Father	15 (2)	-	2	-	DR 53
	Mother	7	_	1	-	-
Case VI	Father	3	-	1	-	DR 52
	Mother	7	17 (3)	-	-	DR 53
Case VII	Father	1 (03)	-	1	-	DR 52
	Mother	7	_	4	2	DR 52/53
Case VIII	Father	11 (5)	-	2	-	DR 52
	Mother	7	17 (3)	3	-	-
Case IX	Father	1	15 (2)	3	-	-
	Mother	7	15 (2)	1	-	DR 53
Case X	Father	14 (6)				
	Mother					

However in the non-macerated group above 17 weeks, 56 % cases were found to be female fetuses and 36 % were male fetuses.

In Table 4.39, from the macroscopical study of the abortus, it appears that the fetus, if it is above 17 weeks or more, the degree of maceration abruptly decreases. Whether this is the result of the appearance of the HLA system in the growing fetus or due to some other factor like maturation and growth of the organs, is a problem to be solved in the future. The other theoretical possibility is the delay in expulsion of the fetus after its death. Is this the result of the immaturity of the fetal system or the inability of the uterus to expel the dead fetus? This is also a problem to be solved in the future. There is also the third possibility of non-fetal participation, due to death, in the labour initiation process; however, the most important clue may be that fetal death is not immediately followed by placental non-function, delayed expulsion and the reverse, i.e., placental dysfunction followed by expulsion or death of the fetus, in case of early expulsion of the fetus. In case of pre- HLA

(before 17 weeks), the fetal death may precede in most cases of abortion and with progressive maturity of the fetus, placental dysfunction may precede the expulsion/death of the fetus. If the expulsion is earlier, the fetus may be living, depending on the state of gestational maturity of the fetal organs.

What Is Our Ultimate Learning from This Search for a Safe, Effective, Cheap Midtrimester Abortifaecient

Combating the menace of unsafe or criminal abortion in the developing world in particular, by the intra amniotic injection of 2 cc tetanus toxoid to mothers who want an abortion could be a good and rational suggestion for our unfortunately poor patients who want a safe abortion.

Adequate popularization of this simple, safe, cheap and easily procurable, effective method of abortion [1-24] can combat the criminal abortion menace. In the developing countries in particular,

ion	
stat	
ge	
s of	
term	
п.	
tuses	
l fe	
llec	
expe	
e of	
tat	
the s	
and	
toxoid)	
tetanus	
schedule	
dosage	
e fixed	
gle 1	
Sing	
4.38	
ā	
a	

Table 4.3	8 Single fixed do:	Table 4.38 Single fixed dosage schedule (tetanus toxoid) and the state of expelled fetuses in terms of gestation	nus toxoid) and the	e state of expelled f	fetuses in terms o	f gestation			
No. of	Gestation up to 1	No. of Gestation up to 16 weeks (10.99 %)			17 weeks and ab	17 weeks and above macerated 8 $\%$			
cases	Macerated (58.53 %)	1 %)	Non macerated		Macerated		Non-macerated		Comments
	Male	Female	Male	Female	Male	Female	Male	Female	
82	32 (58.53 %) 16 (58.53 %)	16 (58.53 %)	2 (10.97 %)	7 (10.97 %)	1 (2.43 %)	1 (2.43 %)	9 (28.04 %)	14 (28.04 %)	 14 (28.04 %) Living fetus at birth 17 weeks and above (non macerated) 23 cases; before 17 weeks group: 6 cases were living at the time of abortion
No. of cases	Weeks of gestation	Mean induction abortion interval with SD	No. of macerations	Maceration of the fetus	Living fetus or dead fetus at abortion				
--------------	--------------------	--	--------------------	-------------------------	---				
9	8-11	96±24.6 h SD	X	9	29 living				
42	12–15	118±32.4 h SD	18	39	fetuses were				
20	16–19	144±48.6 h SD	11	2	seen at birth. 53				
11	20 above	186±36.4 h SD			 dead fetuses were seen at birth 				

Table 4.39 Single fixed dosage schedule (tetanus toxoid) induced maceration and the induction – abortion interval co-relationship

local health departments can take the initiative by organizing camps and workshops and encouraging the participation of social workers, nurses, midwives, village level quack doctors or barefoot doctors, pharmacists or even criminal abortionists agreeable to accept change, from any sphere of life in urban, semi-urban and rural areas. Inadvertent intramuscular (intra-uterine musculature) injection of the mother will sensitize and boost the maternal existing antibody against tetanus. If the mother in such a case decides to continue the pregnancy after 18 weeks the mother would be protected against neonatal tetanus with a subsequent injection after the first month. Global mortality from tetanus is 1 million of which 880,000 deaths are recorded for neonatal tetanus [1-25].

Reaction of the Pre-immune or Hypo-immune Fetus (up to 20 weeks)

Earlier investigations reported the implications of congenital infections in case of maternal syphilis [26] and rubella in the human system [27–32], rubella infection and blue tongue virus infections in lambs and lymphocytic choriomeningites virus in mice [33]. They suggested that the developing fetus of many mammalian species is able to mount a highly efficient immunological response to the agents responsible for congenital infections.

The present research is the first documented work (Medline search verification, 24 October 2014) on intra-amniotic direct antigenic challenge to the human fetus showing the reactions of the developing pre-immune and hypo-immune fetus; however, there are certain obvious differences with the congenital infection scenario.

In case of congenital infections, the placental barrier is formidable and unless this is broken, access to the fetal compartment or fetal organs is not possible. The sequence of events are maternal severe infection, involvement of placenta and loss of placental function – in all these, the fetus is involved. The damage to the fetal system is due to the resultant action of toxins/chemicals liberated by the living or dead/offending organisms and their interaction with the growing fetal system as well as the primitive non-specific to specific developing immunological defense against the offending organisms.

In case of our experimentation with different antigenic challenges, the placenta is never damaged primarily. It is only damaged secondarily to fetal damage or death. Hence the changes we find are primarily the result of fetal changes due to the interaction of the fetal system with the antigenic stress. On the basis of macroscopic and microscopic studies, we found similar hitherto unknown fetal reactions to an antigenic invasion of the fetal system: to 20 % bovine serum albumin 2 cc, 10 cc maternal whole blood, collected from leucocytes or buffy coated layer collected from 10 cc maternal blood, allogeneic amniotic fluid 5 cc of other mothers, bacterial antigen like sterodin, double antigen, tetanus toxoid, etc.

However, the reactions are much more with glutamate BCG (up to 15 weeks). Here fetal death and fetal dissolution was observed of varying degrees without abortion, a hitherto unknown phenomenon (vide Medline search covering the last 40 years).

If we analyze the sensitivity of the growing human embryo to fetus in the light of our experiments with intraamniotic challenge, we can observe the following:

- (a) Intra amniotic tetanus toxoid antigen challenge in the teratogenic phase (up to 9 weeks) to the pre-immune phase of human fetal growth (10–16 weeks) leads to massive hemorrhage; also congestion of all the viscera and changes in the growing architecture of the fetal organs, eventually leading to death and abortion of the fetus.
- (b) Tetanus toxoid challenge at the hypo-immune phase (17 weeks onwards) of human fetal growth leads to mono cellular cell invasion, hemorrhage, thrombosis in all the organs including the placenta where there are additional features of villites and deep subchorionic fibrosis, eventually leading to abortion in a less dramatic way with mostly living fetuses.

Hence, we can say that even in the sterile environment of the amniotic fluid, the developing fetal system can also react to any antigenic challenge; however, the reactions are dependent on type/route/virulence/dosage of the antigenic assault. It is believed that this at least will partially explain the apparent benign presence of treponema palladium over the very early fetus [26]. On the basis of research conducted over the last three to four decades it is felt that a little more presence of treponema antigenic load in a very young fetus will abruptly change the entire scenario from benign presence to a cruel presence triggering death cum deformity and abortion of the fetus depending upon the stage of fetal growth from the teratogen to the pre-immune phase or hypoimmune phase of the fetus due to the crossing of the critical mark of fetal susceptibility cum tolerance of the antigen load. This may result in an exaggerated and hypersensitive type of acute response in the early weeks (up to 16 weeks) and a chronic response in the later weeks (above 17 weeks) as seen with different antigenic challenges which is evident clinically

in the case of the aborted fetus, depending on the gestational age of the fetus.

The above were the research findings regarding the development of different parameters for growth and differentiation of the human fetus from 8 to 20 weeks time.

It is well known that the function of the immune system is to protect the individual from invasion of foreign antigens by distinguishing the self from the non self. A normal immune response relies on the careful coordination of a complex network of specialized cells, organs and biological factors necessary for the recognition and subsequent elimination of foreign antigens. An exaggerated immune response can result in the hypersensitivity to foreign antigens with resultant tissue injury and the expression of a variety of clinical syndromes like Type 1 (IgE mediated) immediate hypersensitivity reaction, Type II (IgG, IgM, complement activation) antibody mediated cytotoxicity reaction, Type III (antigen antibody complex and complement activation) immune complex reaction and Type IV (lymphocyte) delayed hypersensitivity reaction. Two other mechanism are also proposed: Type V and Type VI, of which, Types I-III are humoral immune responses and Type IV response results from initial sensitization and subsequent antigenic challenge. This kind of antigen elimination through the cellular or humoral process is integrally linked to the inflammatory response in which the cellular messengers (cytokines) and antibodies trigger vaso active inflammatory mediators. This inflammation has got a positive role, i.e., to promote efficient elimination of foreign antigens and to prevent uncontrolled lymphocytic activation and antibody production and also negative and deleterious effects. In the case of the growing innocent embryo or the hypo immune human fetus, all the factors responsible for inflammatory response and its coordination are also developing at that stage of gestational maturity. Hence, poor coordination may lead to the negative effect of inflammation due to inappropriate activation in a growing fetal system thus causing dysregulation and perpetuation of the inflammatory process which leads to tissue damage and organ dysfunction and the death of the fetus with massive hemorrhage and congestion of all the viscera (up to 16 weeks).

From 17 weeks onwards, due to the progression of the ontogeny of the human immune system, the reactions are less dramatic, typified by the characteristic changes of mononuclear invasion, less hemorrhage and congestion and damage to the architecture of the growing human fetus. Researchers from immunology to endocrinology and from gynaecology to biochemistry should unite together and sit together to standardize the reactions because these have immense implications for fetal organ transplants and surgery on the unborn, apart from the problem of fetal tissue transplant in the adult system. In this connection, umbilical cord blood stem cell transplants are extremely encouraging because of its hypo-immunogenicity and no graft versus host reaction when the umbilical cord blood stem cells are transplanted into the adult system.

Implications of This Type of Research in the Field of Fetal Surgery and in Understanding the Cause behind Premature Labour in Cases of Surgery of the Unborn

Powerful imaging and sampling techniques have stripped the veil of mystery from the once secretive fetus. Although most fetal defects are best managed after birth, a few predictable life threatening developmental consequences have been successfully corrected in utero; however pre-term labour remains a significant risk to both the mother and the fetus [34].

Our experience with maternal 10 cc blood injection tion or leucocytes of 10 cc maternal blood injection through the intra-amniotic route and even a nonspecific antigenic challenge like BSA 20 % 2 cc shows that these can trigger abortion in 71.4 % of cases within 14 days. Hence, we believe that the most important problems in surgery of the unborn is feto -maternal exchange vis a vis sensitization which can initiate premature labour, whereas materno-fetal exchange can cause death, deformity, abortion as well as premature labour depending on the antigenic load, period of exposure and gestational maturity of the fetus at that point of time.

We also wanted to remind all investigators in the field of the various theories in the pathogenesis of vascular disrupting syndromes following the commonly practiced chorionic villous sampling (CVS) for detection of congenital abnormality with the initiation of a vascular incident after CVS with thrombosis at the sampling site, may result in (a) atrophoblast embolism and embolisation of fetal vessels, resulting in hypoperfusion of fetal circulation followed by voluminous fetal maternal transfusion or release of vaso active substances [35–38], (b) damage of the extra amniotic tissue, especially the fetal membranes with subsequent limb malformation due to amniotic band of oligohydramnios [39–41] or (c) entrapment of limbs to the extra coelomic space [42, 43]. However, the studies referred to above emphasized on the non-immunological link of a vascular disrupting syndrome. Maternal sensitization due to inadvertent fetomaternal hemorrhage which can be followed by materno-fetal hemorrhage as a compensatory exchange transfusion has been noted by AH Lipson and WS Webster [44]. On the basis of our experience the two hit phenomenon of an immunological model as suggested by AH Lipson and WS Webster [44] in case of fetomaternal surgical operations on the unborn as a triggering factor for premature labour. Hence, it is suggested to practice extreme caution and incorporation of micro surgical procedures to minimize the risk of fetomaternal vascular disruption for mothers undergoing surgery on the unborn fetus, resulting in sensitization of the fetus vis-à-vis the mother or vice versa and its implications on the mature immunological system (the mother's) and the immature immunological system (the fetal) and its overall impact on the initiation of abortion or premature labour, depending on gestational maturity.

Future of Hematopoietic Stem Cell Transplant in Utero

A variety of congenital defects, potentially curable by hematopoietic stem cell transplantation are now feasible in early gestational anomalies like immunodeficiencies, haemoglobinopathies and storage diseases. However, there are problems related to donor availability, graft vs host disease, graft rejection in case of adult bone marrow.

The alternate sources of hematopoietic stem cells are either the liver from an aborted fetus or the neonatal umbilical cord blood. In the case of stem cells collected from the liver of a preimmune less than 15–16 weeks fetus, there is no problem of graft versus host disease; hence T cell depletion is not necessary [45] there is the added advantage of homing as they are primed to travel to the waiting fetal marrow and the fetal cells then can be reprogrammed to a lifetime of selfrenewal. Investigators already transplanted hematopoietic stem cells in a fetal sheep and monkeys [46, 47] and demonstrated long lasting hematopoietic chimerism of all cell lines without rejection, host versus graft disease or need for marrow ablation or immunosuppression. Umbilical cord blood stem cells may also be an attractive alternative source of hematopoietic stem cells for adult patients. The observation that is most intriguing is the low incidence of acute or chronic graft versus sot disease [48], regardless of the HLA background or the pregnancy induced maternal tolerance or the blunted proliferation response to neo antigens and T cell cytotoxicity of the umbilical cord blood stem cells [49] is responsible for less graft versus host response in case of pediatric or adult transplantation is a matter under intense scrutiny by the scientific community.

On the basis of human ethically permitted volunteer research on the disruption efficacy of intra amniotic antigenic challenge on the pre-immune fetus up to 15–16 weeks with 2 cc bovine serum albumin 20 %, WBC of 10 cc maternal whole blood, maternal 10 cc whole blood, 5 cc allogeneic amniotic fluid of other allogeneic mothers, it is noted by us that there is abortion of the fetus in varying percentages within 14 days. We also confirmed to the entire scientific community working on the problem of surgery on the unborn that caution should be exercised and that antigenic challenge even to the pre-immune fetus can cause hitherto unnamed fetal reactions, leading to death and abortion of the conceptus. Extreme caution



Fig. 4.1 Shows a non macerated fetus of 18 weeks gestation. Softening of the viscera is noted. There is deep subchorionic fibrosis or whitish appearance of the placenta



Fig. 4.2 Shows a fetus and placenta at 19 weeks. Here the placenta showed whitish, deep sub-chronic fibrosis. Here, the abortion took place on the 11th day after intra amniotic instillation of tetanus toxoid

should be exercised at least in the case of hematopoietic stem cell transplantation in utero to combat and cure genetic diseases because of the negative impact of inflammation type responses in the growing phase of the embryo or the fetus [50–55] (Figs. 4.1, 4.2, 4.3, 4.4, 4.5, 4.6, 4.7, 4.8, 4.9, and 4.10).

Conclusion

What has not been done yet but essentially needs to be done in the future to find a solution to the many questions raised regarding the issue of antigenic challenge to the growing human fetus in utero is listed below:

(a) If we consider pregnancy as a balance of Th1/Th2 response [55], sequential study



Fig. 4.3 (**a**–**c**) This is a unique representative photograph showing a fetus (18 weeks) expelled after intra amniotic instillation of (2 cc) tetanus toxoid on the seventh day. The important aspect of the photograph is the fact that the entire fetus, placental membrane and the amniotic fluid inside the amniotic sac came out simultaneously, justifying the

abortion as complete. Our coagulation studies (vide text) had proved that there is no fall in platelet, fibrinogen and a change of fibrinogen degradation products with this method of abortion. In this connection it is worth mentioning that the price for 2 cc tetanus toxoid is Rs. 4 (Four Indian Rupees only which is equivalent to Five American cents)



Fig. 4.4 The migration inhibition of fetal thymocytes noted in the presence of tetanus toxoid antigen, when tetanus toxoid injection (three injections at 1 month interval) was given intra muscularly to the mother (18 weeks pregnancy)



Fig. 4.6 Depicts the effect of intra amniotic antigen (tetanus toxoid) on migration of fetal thymocytes (17 weeks) on the 14th day



Fig. 4.5 Migration inhibition of fetal thymocytes noted in the presence of tetanus toxoid antigen, after 14 days of intra amniotic tetanus toxoid injection (18 weeks)

of the Th1 mediator (tumor necrotic factor, gamma interferon etc.) and Th2 mediators after intra amniotic challenge, would provide an exact understanding of the cellular action involved in abortion.

(b) A study of TJ6 protein of the fetal thymus after antigenic challenge as well as maternal blood lymphocytes sequential TJ protein binding [56–58] estimation will help us to understand the fetomaternal impact of intra amniotic antigen challenge.



Fig. 4.7 Depicts the effect of intra amniotic antigen (tetanus toxoid) on migration of fetal thymocytes (17 weeks) on the seventh day

(c) Sequential fetal immunophenotypes after the antigenic challenge to the growing fetus help to answer many unanswered questions related to this research work, particularly in the area related to fetal growth and maturation and other specific responses of the fetus with antigenic challenge.



Fig. 4.8 Depicts the effect of intra amniotic antigen (tetanus toxoid) on migration of fetal thymocytes (17 weeks) of fetal thymocytes (17 weeks) on the 21st day

(d) Sequential specific immunoperoxidase staining of the fetal and maternal leucocytes after intra amniotic antigenic challenge and its interactions will boost the existing field of knowledge in the field of fetomaternal interactions after intra amniotic antigenic challenge in pregnancy.

Acknowledgement I gratefully acknowledge the guidance of the Ethical Committee headed by late Prof KP Sengupta, PhD, Director, IPGMER, SSKM, Calcutta, India. I am also grateful to Dr. MK Chhetri, Director of Health Services (retired) for his guidance. I sincerely acknowledge and thank all my patients for their wholehearted objective support and trust in our research which entails the search for a safe, cheap and effective abortifaecient.



Fig. 4.9 (**a**, **b**) Tetanus toxoid injected I.M in the above two pictures to the mother does not show migration inhibition of fetal thymocytes. We have seen in the preview of the present study that antigen (tetanus toxoid) stimulation of the mother causes liberation of MIF factors, but maternal sensitivity with antigen has got no effect on the fetal system and vice versa, i.e., fetal antigen sensitivity has got no impact on maternal system up to 20 weeks of human fetal growth. We can suggest that the impact of antigen possibly depends on gestation, maturity, exposure and amount of antigen in a developing immune system



Fig. 4.10 Immunoflourescence study was done to reconfirm positive cases only and document autoimmune reactions, i.e., DS (DNA), and of ANA, of a mother undergoing abortion with intra amniotic antigen, though routinely all cases were done by ELISA (Enzyme Linked Immunosorbent assay) for semi-quantitative detection of auto antibodies

References

- Bhattacharya N, Choudhuri N, Banerjee S, Mukherjee KL. Intraamniotic tetanus toxoid as an abortifacient. Indian J Med Res. 1979;70:435–9.
- Bhattacharya N. Intra-amniotic instillation of tetanus toxoid-a safe cheap effective abortifacient in the light of our experiences with different intra amniotic instillation of antigens for alteration of pregnancy immune tolerance: a study from 1978–1996, from the Proceeding of the 9th Congress of the Federation of the Asia and Oceania Perinatal Societies, Singapore, 10–14th November 1996. Tamby Raja RL, Ho NK, editors. Relevance and excellence in perinatal care. Monduzzi Editore, International Proceedings Division (Bologna). p. 193–200.
- Bhattacharya N. Study of aborted fetus after intra amniotic instillation of tetanus toxoid. Monduzzi Editore, Bologna, Italy, 1996. p. 187–92.
- Bhattacharya N. Dissolution of the fetus: a new experience with intra amniotic BCG instillation. Monduzzi Editore, Bologna, Italy, 1996. p. 201–6.

- Bhattacharya N. Intra amniotic antigen as abortifacient. Letter to the Editor, Clinical and experimental obstetrics and gynecology, Vol XXIII, No. 4. (Ed., A. Onnis); Canada, 1996. p. 272–5.
- Bhattacharya N, et al. Intra-amniotic tetanus toxoid as a safe abortifacient. 9th World Congress of Obstetrics and Gynaecology. Tokyo, Sept 1979 (Vide book of abstracts).
- Bhattacharya N, et al. Intra amniotic antigen as abortifacient. 10th world Congress of Fertility and Sterility, Madrid, July 1980 (Vide book of abstracts).
- Bhattacharya N. et al. Intra amniotic tetanus toxoid as abortifacient. 10th World Congress of Obstetrics and Gynaecology, Madrid, July 1980. (vide book of abstracts).
- Bhattacharya N, et al. Follow up studies with intraamniotic tetanus toxoid as abortifacient. Asia Oceania Congress of Obstetrics and Gynaecology, Melbourne, Aug 1981 (Vide book of abstracts).
- Bhattacharya N, et al. Tetanus toxoid with Vitamin A. An effective intra-amniotic abortifacient. Xth World Congress of Obstetrics and Gynaecology, Sept 1982, San Francisco.

- 11. Bhattacharya N, et al. Experience with intra amniotic antigen as abortifacient. Vth Annual Meeting of the American Society of the Immunology of Reproduction, May 1984, Durham (Vide book of abstracts).
- Bhattacharya N, et al. Intra-amniotic tetanus toxoid with Vitamin A as abortifacient. XVI World Congress of Pathology, Vienna, Sept 1986 (Vide book of abstracts).
- Bhattacharya N, et al. Experience with intra amniotic antigen for altering pregnancy immunotolerance. XVI World Congress of Pathology, Vienna, Sept 1986 (Vide book of abstracts).
- 14. Bhattacharya N, et al. Alteration of pregnancy induced immunotolerance by intra amniotic antigen IIIrd International Congress of Reproduction Immunology, July 1986, Canada. (Vide book of abstracts).
- Bhattacharya N, et al. Study with intra amniotic antigen in pregnancy. 4th International Congress of Reproduction Immunology, Kiel, FRG, 1989. (Vide book of abstracts).
- Bhattacharya N, et al. Intra amniotic tetanus toxoid as abortifacient. XIVth World Congress of Obstetrics and Gynaecology, Montreal, Sept 1994. (Vide book of abstracts).
- Bhattacharya N, et al. Experience with intra amniotic antigen in 2nd trimester pregnancy. XIVth World Congress of Fertility and Sterility, Montpellier, France, 1995. (Vide book of abstracts).
- Bhattacharya N, et al. Disruption of fetal growth with intra-amniotic antigen. Asia Oceania Congress of Obstetrics and Gynaecology, Bali, 1995 (Vide book of abstracts).
- Bhattacharya N, et al. Experience with intra amniotic antigen in pregnancy. 52nd Annual meeting of the Society of Obstetrics and Gynaecology, Canada, June 1996 (Vide book of abstracts).
- Bhattacharya N, et al. Experience with intra amniotic tetanus toxoid in pregnancy. 11th European Association of Obstetrics and Gynaecology, Budapest, June 1996. (Vide book of abstracts).
- Bhattacharya N, et al. Sequence of events after intra amniotic instillation of tetanus toxoid. International Congress on Gestosis, Romania, May 1996 (Vide book of abstracts).
- Bhattacharya N. Study of Intra amniotic tetanus toxoid as abortifacient. 15th World Congress of Obstetrics and Gynaecology, Copenhagen, Aug 1997 (Vide book of abstracts).
- 23. Bhattacharya N. Intra-amniotic antigen instillation for alteration of pregnancy induced immunotolerance. Annual Congress of the Federation of Obstetrics and Gynecological Society of India, Dec 1997, Siliguri, West Bengal (Vide book of abstracts).
- 24. Bhattacharya N. Intra amniotic tetanus toxoid in the alteration of pregnancy immunotolerance. South Asia Association for Regional Cooperation (SAARC) Congress at Dhaka on Women's Reproductive Health in February 1998. (Vide book of abstracts).

- Bhatia R, Ichhpujani RL. Immune response to vaccine. In: Rajesh B, Ichhpujani RL, editors. Immunization against infectious disease. New Delhi: Jaypee Brothers, Medical Publishers (P) Ltd.; 1994. p. 57.
- Silverstein AM. Congenital syphilis and the timing of immunogenesis in the human fetus. Nature (London). 1962;194:196–7.
- Singer DB, Rudolph AJ, Rosenberg HS, Rawis WE, Boniuk M. J Pediatr. Pathology of the congenital rubella syndrome. 1967;71:665–75.
- 28. Tonduri G. Embryopathien. Berlin: Springer; 1962.
- 29. Tonduri G, Smith DW. J Pediatr. Fetal rubella pathology. 1966;68:867–79.
- Rawls WE, Melnick JL. Rubella virus carrier cultures derived from congenitally infected infants. J Exp Med. 1966;123:795–816.
- Hardy JB, Mc Cracken GH, Gilkeson MR, Sevev JL. Adverse fetal outcome following maternal rubella after the first trimester of pregnancy. JAMA. 1969; 207(13):2414–20.
- 32. Osburn BI. The relation of fetal age to the character of lesions in fetal lambs infected with brucella ovis. Pathol Veternary. 1968;5:395–406.
- 33. Silverstein AM. Immunological maturation of the fetus: modulation of the pathogenesis of congenital infectious diseases. In: Ontogeny of acquired immunity-proceeding of Ciba Foundation Symposium, Elsevier-Excerpta Medica, North Hollan (23–25th Nov 1971); 1972. p. 14–25.
- Harrison MR. Fetal surgery. Am J Obstet Gynaecol. 1996;174(4):1255.
- Quintero RA, Romero R, Mahoney MJ, Vecchilo M, Hobbins JC. Fetal haemmorhagic lesions after chorionic villous sampling. Lancet. 1992;339:193.
- Burton BK, Schulz CJ, Burd LI. Limb anomalies associated with chorionic villus sampling. Obstet Gynaecol. 1992;79:726–30.
- Rodeck CH. Fetal development after chorionic villus sampling. Lancet. 1993;341:468–9.
- Firth HV, Boyd PA, Chamberlain P, Mc Kenzie IZ, Lindenbaum RH, Huson SM. Severe limb abnormalities after chorionic villus sampling at 56–66 days gestation. Lancet. 1991;337(8744):762–3.
- Planteydt HT, Van de Vooren MJ, Verweij H. Amniotic band and malformations in a child etc. Lancet. 1986;ii:756–7.
- Christians GC, Van Baarlen ML, Huber J, Leschot NJ. Fetal limb constriction: a possible complication of CVS. Prenat Diagn. 1989;9:67–71.
- Smidt-Jensen S, Philip J, Zachary JM, Fowler SE, Norgaard-Pedersen B. Implication of maternal serum AFP elevation. Prenat Diagn. 1994;14:35–45.
- Shepard TJ, Kapur RP, Fantel AG. Limb reduction defects and chorionic villus sampling. Lancet. 1991;337:1092.
- 43. Ghirardine G, Camurri L. Exocoelomic space limb reduction and CVS. Lancet. 1991;338:695.
- 44. Lipson AH, Webster WS. Transverse limb deficiency, Oromandibular limb hypogenesis sequences. Am J Med Genet. 1993;47:1141–3.

- 45. Harrison MR. Fetal surgery. Am J Obstet Gynaecol. 1996;1744(4):1262.
- 46. Flake AW, Harrison MR, Zanjani ED. In utero stem cell transplantation. Exp Hematol. 1991;19:1061–4.
- 47. Flake AW, Harrison MR, Adzic NS, Zanjani ED. Transplantation of fetal hematopoietic stem cells in utero: the creation of hematopoietic chimeras. Science. 1986;233:776–8.
- 48. Thomas ED. Marrow transplantation for malignant diseases. J Clin Oncol. 1983;1:517.
- Tafuri A, Alferink J, Moller P, et al. T cell awareness of paternal allo-antigens during pregnancy. Science. 1995;270:630–3.
- Fischer GW, Offoloni MG, Mond JJ. Prospects of vaccination during pregnancy and the newborn period. Clin Perinatol. 1997;24(1):231–49.
- Chanock R, Kapikian A, Mills J. Influence of immunological factors in respiratory syncytial virus disease. Arch Environ Health. 1970;21:347–55.
- Baltazar JC, Sarol JN. Prenatal tetanus immunization and other practices associated with neonatal tetanus. Southeast Asian J Trop Med Public Health. 1994; 25(1):132–8.

- Roitt IM. Roitt's essential immunology. London: Blackwell Science Ltd; 1997. p. 216.
- 54. Kay HEM. Fetal thymus transplants in man. In: Proceedings of the Ciba Foundation symposium on ontogeny of acquired immunity, London 23–25th Nov 1971, Published by Elsevier, Excerpts Medica, North Holland; 1972. p. 257.
- 55. Wegman TC, Lin H, Guelbert L, Mossman TH. Bi-directional cytokine interactions in the Maternofetal relationship. Successful allopregnancy is a Th1 phenomenon. Immunol Today. 1993;14: 353–5.
- Ribbing SL, Hoversland RC, Beaman KD. T cell suppressor factors play an integral role in preventing fetal rejection. J Reprod Immunol. 1988;14:83–95.
- Rubesa G, Beaman KD, Lucin P, Beer AE, Rukavina D. Expression of TJ6 protein in the human first trimester decidual lymphocytes. Reg Immunol. 1994;6:331–3.
- Nicholas TC, Kang JA, Angkachatchi V, Beer AE, Beaman KD. Lymphocyte expression of the pregnancy associated protein TJ6. Cell Immunol. 1994; 155:219–29.

Fetal Growth and Development in the First Two Trimesters

5

Aditi Aikat, Tarun Kumar Roy, and Niranjan Bhattacharya

The story of human life begins much beyond birth...

As soon as an oocyte is fertilized by a sperm, a journey ensues from the totipotent zygote undergoing cell division, cellular migration, differentiation, apoptosis (programmed cell death), growth and cell re-arrangement that involves temporal and spatial control of gene expression guided by epigenetic parameters, to evolve into an organised multi-cellular human being.

The word fetus (plural fetuses) is from the Latin fetus, "the bearing or hatching of young, a bringing forth," from Latin base *fe- "to generate, bear," also "to suck, suckle" [1]. The fetal period extends from ninth week of gestation, the first 8 weeks being the embryonic period, during which the embryo gradually transforms into a fetus with the primordial characteristics of all major systems having formed and bearing a recognizable human appearance. The development in the fetal period is primarily

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India e-mail: sanjuktaniranjan@gmail.com concerned with rapid body growth and differentiation of tissues, organs and systems, while the rest of the body takes over growth rate of the fetal head deviating from the embryonic period where there is a head start regarding growth rate [2].

First Two Trimesters

The first two trimesters, which are each a period of 3 months are again unique with all the major systems getting developed in the first trimester itself and the second one seeing the fetus gain sufficient growth size-wise with detailed visualization possible through USG [3]. The fetus during this period exhibits the unique characteristic of healing without scar, having a critical balance between Matrix metallo proteinases (MMPs) and Tissue Inhibitors of MetalloProteinases (TIMPs) [4], demonstrating lack of fibroplasia, increased Hyaluronidases and increased expression of home box genes like HOX B13 and PAX 2 genes [5]. Beyond these two trimesters, the fetal cells slowly acquire adult characteristics.

Perspectives on Fetal Growth and Development across Culture and Time

Fetal growth and development have been always an enigma that people over the ages have tried to

A. Aikat, MD • T.K. Roy, MBBS, DMRD Regenerative Medicine, Calcutta School of Tropical Medicine, Kolkata, India

[©] Springer International Publishing Switzerland 2016 N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development: First and Second Trimesters*, DOI 10.1007/978-3-319-14874-8_5

unravel, with each having his own perspectives and answers.

The Egyptian texts show Akhnaton (Amenophis IV) praising the Sun God Aton as the creator of germ in a woman, maker of seed in men and giver of life. The ancient Egyptians professed that the soul entered the child at birth through the placenta.

A Hindu Scripture penned in Sanskrit embodying the development of the fetus has been traced to 1416 BC, and is known as the Garbha Upanishada.

Even Greek scholars made many contributions to this with books authored by the Father of Medicine, Hippocrates of Cos (Circa 460–377 BC) featuring the first recorded embryologic studies. Aristotle of Stagira (circa 384–322 BC), a Greek philosopher and scientist promoted the idea that the embryo developed from a formless mass, which he described as a "less fully concocted seed with a nutritive soul and all bodily parts" [6].

A Roman scientist, Claudius Galen (circa 130– 201) described the development and nutrition of fetus in his book, *On the Formation of Fetus*.

The Jewish physician Samuel El Yehudi (second century AD) mentioned six stages of development.

Views in the medieval period hardly deviated from the theory of Aristotle. But with the dawn of the Renaissance, we find Leonardo da Vinci (1452–1519) making accurate sketches of pregnant uteri containing fetus (Fig. 5.1) and introducing the quantitative approach to embryology [7]. The revolution began with the publication of William Harvey's (1578–1657) book, *De Generatione Animalium*. He is said to be greatly influenced by one of his Professors Fabricius (1537–1619), the author of *De Formato Fetus* [8].

In 1759, Casper Friedrich Wolff proposed the layer concept, which formed the basis of the theory of Epigenesis [9].

These were followed by Genetic perspectives of fetal development with James Watson and Francis Crick deciphering the molecular structure of DNA in 1953 [10].

Rapid advances in Molecular Biology next started addressing issues related to temporal and spatial regulation of gene expression and the pro-



Fig. 5.1 Author's representation of Leonardo da Vinci's drawing made in the fifteenth century AD showing a fetus in utero

cess of cellular commitment towards formation of various parts [11].

The Anatomical and Physiological Perspective of Fetal Development in the First Two Trimesters

For our convenience of understanding, the anatomical and physiological development of the fetus is discussed in periods of 4–5 weeks.

Nine to Twelve Weeks

At the beginning of the ninth week, the head constitutes approximately half the CRL of the fetus. Subsequently, growth in body length accelerates rapidly so that by the end of 12 weeks, the CRL has more than doubled. Although growth of the head slows down considerably by this time, it is still disproportionately large compared to the rest of the body [12].

At 9 weeks, the face is broad, the eyes are widely separated, the ears are low set, and the eyelids are fused. By the end of 12 weeks, primary ossification centers appear in the skeleton, especially in the cranium (skull) and long bones [13]. The legs are short and the thighs are relatively small during the early ninth week. The upper limbs have almost reached their final relative lengths by the end of 12 weeks, but the lower limbs are still not so well developed and are slightly shorter than their final relative lengths.

The external genitalia of males and females are hard to distinguish until the end of the ninth week. It is by the 12th week that the mature fetal form is established. Intestinal coils are clearly visible in the proximal end of the umbilical cord until the middle of the tenth week. The intestines have returned to the abdomen by the 11th week.

At 9 weeks, the liver is the major site of erythropoiesis (formation of red blood cells) [14]. By the end of 12 weeks, the site of erythropoiesis shifts to the spleen. Urine formation begins between the 9th and 12th weeks, and urine is discharged through the urethra into the amniotic fluid. The fetus reabsorbs some amniotic fluid after swallowing it. It is through the placental membrane, that the fetal waste products are transferred to the maternal circulation (Fig. 5.2).

Thirteen to Sixteen Weeks

Rapid growth is observed during this period. By 16 weeks, the head is relatively small compared with that of the 12-week fetus and the lower limbs have lengthened. Limb movements, which first occur at the end of the embryonic period, become coordinated by the 14th week but are too slight to be felt by the mother [15].

Limb movements are visible during ultrasound examinations. Ossification of the fetal skeleton is active during this period, and the bones are clearly visible on ultrasound images by the beginning of the 16th week. Slow eye movements occur at 14 weeks. Scalp hair patterning is also determined during this period [16].

By 16 weeks, the ovaries are differentiated and contain primordial ovarian follicles that contain oogonia.

By 12–14 weeks, the genitalia of the fetuses can be recognized. The eyes face anteriorly rather than anterolaterally by 16 weeks. Besides, the external ears are close to their definitive position on the sides of the head.

Seventeen to Twenty Four Weeks

Growth slows down during this period, but the fetus still increases its CRL by approximately 50 mm [17]. Fetal movements—quickening—are commonly felt by the mother. The skin is now covered with a greasy, cheese-like material—*vernix caseosa*. It consists of a mixture of dead epidermal cells and a fatty substance (secretion) from the fetal sebaceous glands.

At 20 weeks, eyebrows and head hair are visible. The fetuses are usually completely covered with fine downy hair called **lanugo**, that helps to hold the *vernix caseosa* on the skin [18]. By 18 weeks, the fetal uterus is formed. The canalization of the vagina begins by this time with many primordial ovarian follicles containing oogonia being visible. By 20 weeks, the testes have begun to descend, but they are still located on the posterior abdominal wall, as are the ovaries in female fetuses.

Systemic Development

Respiratory System

- In the first two trimesters, lung development remains restricted to two of the four stages of maturation: pseudoglandular (6–16 weeks), and canalicular (16–26 weeks).
- In the beginning of the pseudoglandular period, the lungs only contain exocrine gland and when the fetus reaches 16 weeks, most of the important elements of the lung have been created except the part which involves gas exchange. Any fetus born during this period will not survive because no respiration is able to take place. By 20–22 weeks, type II pneumocytes begin to secrete pulmonary surfactant. Deficiency of surfactant results in respiratory distress syndrome (RDS) or hyaline membrane disease (HMD).
- The canalicular period refers to the period when there is growth of the lungs tree. The distal part of the lungs mature later than the proximal part. Owing to this the canalicular period overlaps with the pseudoglandular



Fig. 5.2 (a) and (b) Sonographical images of fetuses of 9 weeks and 11 weeks of age, (c) and (c (i)) showing sonographical images of fetus of 13 weeks. (**d**–**f**) sonographi-

cal images of fetuses of 17, 21 weeks and 24th week of development

period. During this period, the lung tissue is highly vascularized and the lumen of the bronchi and terminal bronchioles are enlarged. Towards the end of the canalicular period, approximately between 24 and 26 weeks of gestation, it is possible for respiration to occur after the passages have completed development up to the primordial alveolar sacs and the lung tissues is also well vascularised [19].

Cardio-Vascular System

The cardiovascular system is the first major system to function in the embryo. The primordial heart and vascular system appear in the middle of the third week. This precocious cardiac development occurs because the rapidly growing embryo can no longer satisfy its nutritional and oxygen requirements by diffusion alone. Consequently, there is a need for an efficient method of acquiring oxygen and nutrients from the maternal blood and disposing of carbon dioxide and waste products. The cardiovascular system is derived mainly from:

- Splanchnic mesoderm, which forms the primordium of the heart
- Paraxial and lateral mesoderm near the otic placodes
- Pharyngeal mesoderm
- Neural crest cells from the region between the otic vesicles and the caudal limits of the third pair of somites.
- Endocardium formed from the Splanchnic mesoderm migrate and localize between the developing Myocardium above and the endoderm below. N- Cadherin may also be implicated in the cell sorting. The catenins are linked with these cadherin molecules and involved in the wnT signaling pathway in the development of the heart tube [20].

The fetal circulatory system operates very differently from that of the newborn, being organized in parallel circles [21]. Blood carrying the highest oxygen saturation goes to the fetal heart, the brain, the upper extremities while the other parts of the fetal body receive blood with lower oxygen saturation. Also fetal blood pO2 is much lower than maternal blood pO2 and hence fetal environment have been equated with Mount Everest in utero [22]. Despite the low pO2 in the fetal blood, the fetus is not exposed to hypoxia due to its adaptive responses and the hemoglobin concentration of fetal blood is about 50 % greater than that of the mother and interestingly the fetal hemoglobin can carry more oxygen at a low PCO2 than it can at high PCO2 and are alkaline in nature. Capacity of fetal blood to combine with oxygen increases, while maternal blood has diminished capacity to do so, thus enforcing the Double Bohr effect [23].

The first line of supervision over the blood circulation of the fetus are mediated by the carotid chemoreceptors, with the second line being hormones, Antidiuretic hormone (ADH), Angiotensin II, catecholamines and cortisol [24, 25].

Gastrointestinal System

The primitive gut that has already developed in the embryonic phase, starts peristaltic activities in the large intestine by 10 weeks and by 11 weeks in the small intestine [26]. Swallowing amniotic fluid also reflects fetal CNS maturity and contributes to somatic growth, development and maturation of the fetal Gastrointestinal System providing 10-15 % of nitrogen requirement in the normal fetus. Upper gastrointestinal tract obstruction in human fetuses are associated significantly with fetal growth restriction and amniotic fluid volume [27, 28]. Besides, fetal swallowing is influenced by hypoxia, hypotension and plasma osmololality changes as well [27]. The main feeding regulatory factors, neuropeptide Y (NPY) and Leptin are secreted in human fetuses as early as 16–18 weeks respectively [29, 30]. Leptin may be implicated in the development of the fetal gastrointestinal tract as well; this is suggested by the presence of Ob-Rb (functional receptor of Leptin) in the mucosa [31].

It is observed that by 13 weeks the intestine starts to absorb glucose and water [26]. Enzyme

activities increase after 14 weeks of gestation with the first traces of gastric acidity appearing in a 4 month old fetus [26].

Urinary System

From the beginning of the fourth week of the embryonic period the excretory organs starts to form pro-nephros, mesonephros and metanephros, with ultimately metanephros forming the definitive kidney. The nephrons are derived from the metanephric blastema [32]. The first nephrons appear in the kidney medulla, around 20-22 weeks of gestation. Failure in the maturation of the primitive kidneys can lead to abnormal development of the genital system, adrenal glands and lungs [33]. Studies have revealed that retinoids, metabolites of vitamin A, have a significant impact on the number of nephrons and this happens to be dose- dependent and are commonly associated with cigarette smoking, alcoholism and unbalanced diet. Ironically the congenital nephron deficit could be a missing link in understanding the etiopathogenesis of essential hypertension [34].

The fetus usually produces hypotonic urine and begins producing the anti-Diuretic hormone (ADH) by the 11th week of gestation. Aquaporin, special water channels appear by 12–15th week [35]. However the fetal kidney does not play a significant role in the acid base balance of the body [33]. It starts excreting urine from the third month of pregnancy [36].

Central Nervous System (CNS)

CNS develops from the embryonic ectoderm. After the early embryonic development characterized by its immobilization, the earliest neural activity originates from spinal motoneurons [37]. From early fetal period the first signs of supraspinal control on motor activity begins to be evident [38]. By the ninth week, the brainstem, the diencephalon and cerebral hemispheres are formed [39]. Facial movements, which are controlled by V and VII cranial nerve, appear around 10–11 weeks of gestation [38]. After the 12th week, the fetal movement become more variable in speed and amplitude [40]. While the midbrain maturation begins as late as the middle of the second trimester, that controls the eye movements [41]. The second trimester of pregnancy is characterized by organization of fetal movement patterns and increase in complexity of movements, with rest activity cycles becoming recognizable [42]. It is unknown whether and when the fetus begins to understand pain. Functional Thalamo-cortical connections are required for this awareness. It is supposedly formed between 22 and 26 weeks, and beyond this the fetus can probably feel pain since it is able to perceive the stimuli [43].

Genetic and Epigenetic Perspectives of Fetal Development

Understanding the epigenetic regulation of fetal development requires the integration of many fields of scientific inquiry including phylogeny, the evolution of placental anatomy, comparative genomics, and epigenomics. Pregnancies vary enormously in terms of timing of embryo implantation, placental anatomy, uterine shape, rates of fetal growth and development, neonatal body mass, number of offspring per pregnancy, and gestation length.

A growing body of evidence supports the notion that epigenetic changes such as DNA methylation and histone modifications, both involving chromatin remodeling, contribute to fetal metabolic programming. Research has revealed that impaired intrauterine growth and adult metabolic and cardiovascular disorders. including coronary heart disease, type 2 diabetes, and insulin (INS) resistance, are strongly associated. It was David Barker and coworkers, who followed-up a cohort of 499 men and women born in Preston (Lancashire, UK) during 1935-1943, and found correlation that adults having highest blood pressures were those who had been small babies with large placentas. Barker and Hales also explained that fetuses having an impaired supply of nutrients try to adapt by changing their physiology and metabolism, and altering the sensitivity of tissues; this leads to an abnormal structures and functions in adult life. This led to the formulation of the "thrifty phenotype hypothesis".

A dynamic cross- talk between the fetal and maternal environment and modulation of gene expression was strongly hypothesised. Epigenetic modifications such as DNA methylation and covalent posttranslational histone modifications, which mediate phenomena such as genomic imprinting and chromatin remodeling, eventually emerged as a plausible molecular explanation of fetal metabolic programming [44].

Factors Influencing Fetal Growth and Development

The fetus requires various substrates for growth and production of energy and several other factors that govern fetal morphogenesis in varying proportion to the growth and development, the principal determinants being the fetal genotype and in utero environment.

Fetal Parameters

- Genetic factors contribute to fetal growth and development. Studies involving knock out genes or gene over-expression have revealed that both maternal and paternal influences are present during fetal development and are passed on to the developing fetus by spermatozoa or oogonia by a mechanism called imprinting [45]. It has been observed gynogenetic zygotes (two maternal genome copies) lead to under-developed extra-embryonic tissues but well developed embryos. Insulin like growth factor I (IGF I) and IGF II are two protein products of gene that specifically regulate the development of trophoblast cells, which form the placenta. Studies have shown that paternal IGF-II allele in mice model produce 60 % less fetal growth.
- Nutritional factors that govern fetal growth and development, has the placenta providing a major role. During the first two trimesters the

placenta is functional as a nutritive, respiratory, excretory and endocrine organ and from the very inception it is responsible for the diffusion of oxygen and nutrients from the maternal blood to the embryonic blood and passing the metabolic wastes in the reverse direction [46]. With increasing months of pregnancy the placental growth and thinning of its membrane diffusion layers, the placental diffusion rate gets a surge but exchange of oxygen is more related to the blood flow than the placental thickness [47]. The main ingredient of fetal diet is carbohydrate. The fetus has low capacity for gluconeogenesis as the necessary enzymes are inactive due to low fetal arterial pO2. The fetal insulin determines the glucose utilization by the fetus. Hormonal regulation of fetal growth differs from the post-natal regulation. ACTH and glucocorticoids stimulate the storage of glucose as glycogen [47]. Fetal growth hormone also has a small role in stimulating fetal growth as there is lack of functional GH receptors. But fetal insulin significantly stimulates fetal growth. It is known that pancreatic agenesis leads to fetal growth restriction whereas fetal hyperinsulinemea leads to fetal mass overgrowth [48]. IGFs are present in human fetal tissue extract after 12 weeks of gestation [49]. Its level in fetal and cord circulation directly correlates with fetal length and mass [2].

Maternal Parameters

- Parity, maternal age and multiple pregnancy are also factors influencing fetal growth and development. Primiparous mothers have been found to bear smaller babies than multiparous mothers. Individuals of multiple births usually weigh considerably less than infants resulting from a single pregnancy. It is evident that the total metabolic requirements of two or more fetuses exceed the nutritional supply available from the placenta during the third trimester [50].
- Maternal Nutrition: There is growing evidence that maternal nutritional status can alter

the epigenetic state (stable alterations of gene expression through DNA methylation and histone modifications) of the fetal genome. This may provide a molecular mechanism for the impact of maternal nutrition on both fetal programming and genomic imprinting. Promoting optimal nutrition will not only ensure optimal fetal development, but will also reduce the risk of chronic diseases in adults [51].

- Cigarette smoking is highly detrimental to fetal development as the mother inhales nicotine, lead, arsenic and carbon monoxide. These cause placental insufficiency. Smoking during pregnancy can cause low-birth weight, preterm delivery, and infant death. Smoking during pregnancy is estimated to account for 20–30 % of low-birth weight babies, up to 14 % of preterm deliveries, and about 10 % of all infant deaths according to the American Lung Association. Smoking is a wellestablished cause of IUGR (Intra uterine growth retardation) [52].
- Alcohol and Illicit Drugs: Fetal alcohol syndrome (FAS) or fetal alcohol syndrome is a pattern of physical and mental defects that can develop in a fetus in association with high levels of alcohol consumption during pregnancy. Alcohol crosses the placental barrier and can stunt fetal growth or weight, create distinctive facial stigmata, damage neurons and brain structures, which can result in intellectual disability and other psychological or behavioral problems, and also cause other physical damage [2, 53-55]. The main effect of FAS is permanent central nervous system damage, especially to the brain. Developing brain cells and structures can be malformed or have development interrupted by prenatal alcohol exposure; this can create an array of primary cognitive and functional disabilities (including poor memory, attention deficits, impulsive behavior, and poor cause-effect reasoning) as well as secondary disabilities.
- Impaired utero-placental and Feto-placental Blood Flow: Maternal placental circulation may be reduced by conditions that decrease uterine blood flow (e.g., small chorionic ves-

sels, severe maternal hypotension, and renal disease). Chronic reduction of uterine blood flow can cause fetal starvation resulting in IUGR. Placental dysfunction or defects can also cause IUGR. The net effect of these placental abnormalities is a reduction of the total area for exchange of nutrients between the fetal and maternal blood streams. It is very difficult to separate the effect of these placental changes from the effect of reduced maternal blood flow to the placenta. In some instances of chronic maternal disease, the maternal vascular changes in the uterus are primary and the placental defects are secondary [2].

Microchimerism

During the period of fetal development an interphenomena takes called esting place Microchimerism. It has only recently become apparent that naturally-acquired microchimerism is common in humans. The placenta was previously thought to barricade the mother and fetus thus preventing maternal rejection of the fetus. It is now known that during pregnancy some cells traffic from mother to fetus and from fetus to mother. Surprisingly, some of the mother's cells can be found in her adult offspring and some cells from the fetus can be found in the mother decades later. Microchimerism (Mc) refers to harboring a small number of cells or DNA from a genetically different individual [56].

Cytokine Response to Prenatal Infections during Fetal Development

Bacteria and viruses are potent activators of proinflammatory cytokines, such as tumor necrosis factor (TNF)- α , interleukin (IL)-1 β and IL-6, which are key effectors of the peripheral immune response [57] and the main mediators of brain orchestrated sickness responses following infection and inflammation [58]. In the periphery these cytokines act in a cascade-

like fashion, which generally involves the sequential induction of TNF- α , IL-1 β and IL-6 [59]. Although the cytokine-dependent inflammatory response is essential in resolving infection, abnormal changes in TNF- α , IL-1 β and IL-6 levels during gestation, as a result of infection, may affect fetal brain development in utero [60]. Indeed, while cytokines have a key role in normal brain development [61], they are implicated in the death and dysfunction of neurons following injury or chronic disease in the adult brain [62], and are elevated in neurodevelopmental insults, such as hypoxia, also associated with an increased risk of schizophrenia [63]. Thus, cytokines are logical candidates to mediate the effects of maternal infection on the fetus; however, there have been contradictory reports of both increased and decreased expression of these mediators in the fetal brain following exposure of pregnant rats to bacterial LPS [64, 65]. Therefore, it remains unclear whether maternal LPS exposure during pregnancy results in altered cytokine expression in the fetal brain. Downstream consequences of systemic cytokine induction during pregnancy include changes in circulating levels of glucocorticoid [66, 67] and thyroid hormones [68, 69]. Maternal glucocorticoid [70, 71] and thyroid [72] hormones can cross the blood placental barrier and are essential for normal fetal neurodevelopment [73, 74]. However, in humans, exposure of the fetus to excess glucocorticoids can have detrimental effects on fetal brain development [73, 75].

Bacterial infection of the developing fetal lungs too begins with an inflammatory cascade resulting in cytokine injury and oxidative stress. For some pathogens like P. Falciparum, the mechanisms involve oxidative stress and apoptosis to disrupt placental and fetal growth. An in utero infection may also affect the long-term health of the infant; in many cases, a viral infection in utero increases the risk of developing type 1 diabetes in childhood. Understanding the varied mechanisms employed by these pathogens may enable therapies to attenuate changes in fetal development, decrease preterm birth, and improve survival.

Fetal Allograft

The feto-placental unit may be regarded as an allograft with respect to the mother, yet the maternal immune system does not reject it, giving rise to a major biologic enigma in nature. For nearly 50 years, the subject has been dominated by Medawar's 'fetal allograft analogy' hypothesis, in which the fetus is assumed to be a semi-allogeneic conceptus that evaded rejection [76].

The syncytiotrophoblast of the chorionic villi, although exposed to maternal immune cells within the blood sinusoids, lack major histocompatibility (MHC) antigens and hence no rejection responses are evoked. However, extravillous trophoblast (EVT) cells, which invade the uterine decidua and its vasculature (spiral arteries), express class I MHC antigens. These antigens include HLA-G, which, being nonpolymorphic (class Ib), is poorly recognizable by T lymphocytes as an alloantigen, as well as HLA-C, which, being polymorphic (class Ia), is recognizable by T cells. In addition to averting T cells, EVT cells must also shield themselves from potential attack by natural killer (NK) lymphocytes and injury inflicted by activation of complement [77].

The strategic location of HLA G in the placenta is believed to evade T-cell recognition owing to its nonpolymorphic nature, and recognition by the "killer-inhibitory receptors" on NK cells, thus turning off their killer function. Arguments against this hypothesis have been suggested by the identification of healthy individuals showing biallelic loss of HLA-G1, which goes on to sustain the thought that HLA-G is not essential for fetoplacental survival. Besides human EVT cells were found to be vulnerable to NK cellmediated killing; and it was unable to explain the reason behind non rejection of HLA-C, a polymorphic antigen, also expressed by EVT cells.

A probable explanation could be that both HLA-G and HLA-C were shown to have the unique ability to resist human cytomegalovirusmediated MHC class I degradation, with the probability of having selective location of these two antigens at the fetomaternal interface which may help to withstand viral assault.

- Immunoprotection is provided locally by certain immunosuppressor molecules, such as prostaglandin E₂, transforming growth factor (TGF)-β and interleukin-10. Decidua-derived prostaglandin E₂ was shown to block activation of maternal T cells as well as NK cells in situ. Indeed, immunoregulatory function of decidual cells is consistent with their genealogy. It was shown that uterine endometrial stromal cells, which differentiate into decidual cells during pregnancy, are derived from progenitor (stem) cells that migrate from hemopoietic organs such as the fetal liver and the bone marrow during ontogeny.
- Transient tolerance of maternal T-cell to the fetal MHC antigens may be a backup mechanism for placental immunoprotection. A similar B-cell tolerance has also been observed.
- A trafficking of activated maternal leukocytes into the placenta or the fetus is prevented by deletion of these cells triggered by apoptosisinducing ligands present on the trophoblast.
- Based on genetic manipulation in mice, it was shown that the presence of complement regulatory proteins (Crry in the mouse, membrane cofactor protein or CD46 in the human) [78], which can block activation of the third component of complement (C3) in the complement cascade, protects the placenta from complement-mediated destruction, which may happen otherwise because of residual C3 activation remaining after defending against pathogens. Crry gene knockout mice died in utero because of complementmediated placental damage, which could be averted by additional knockout of the C3 gene.
- Experiments in mice revealed that the presence of the enzyme indoleamine 2,3-deoxygenase in trophoblastic cells was critical for immunoprotection of the allogeneic conceptus by suppressing T cell-driven local inflammatory responses including complement activation [79].

Another hypothesis is put forward to account for the non-rejection of the fetal allograft, based on shared surface-repellent molecules (S.R.M.s) which prevent close apposition of maternal and fetal immunocompetent cells. This hypothesis is shown to be compatible with the major theoretical requirements of the immune system. Thus it provides possible explanations for the self/notself recognition process, tolerance to organ transplants, the development of autoimmune disease, and high and low zone tolerance. The recognition of self is important with reference to cellular antigens, but lesser less so for humoral antigens [80].

The survival of the fetal allograft also depends on a fine well coordinated systemic and uterine support by the reproductive hormones Progesterone (PR). Estrogen (ER), Human chorionic gonadotropin (HCG), prolactin (PRL), and others. These hormones and their cytokine nexus are essential for maintaining the pregnancy specific immunomodulation, including its up and down regulations.

Diagnostic Parameters

Ultrasonography

Ultrasonographical examination of the fetus became integrated into the prenatal care soon after its introduction in the 1950s, because of its wide availability, low cost, and lack of known adverse effects. The chorionic sac and its contents may be visualized by ultrasonography during the embryonic and fetal periods. The benefits of antenatal ultrasound examination can proceed from assessment in the first trimester. Determination of risk assessment of aneuploidy in singleton and multiple gestations, fetal viability, accurate gestational age, multiple gestations and chorioicity and amnionicity and detection of fetal anomalies have been described with first trimester ultrasonography [81].

The benefit burden calculus of routine antenatal ultrasonography support its use and fulfillment of ethical principles of beneficence and respect for patient's autonomy [82].

Ultrasound scans give accurate measurements of the biparietal diameter of the fetal cranium (skull), from which close estimates of fetal age and length can be made. Despite difficulty of accurate diagnosis, detection rate of fatal skeletal abnormality with ultrasonography is as high as 94–96 %. With the advancement in technology of three-dimensional ultrasonography, abnormality in face, relative proportion of limbs, hands and feet are observed with more accuracy [83].

Diagnostic Amniocentesis

Amniocentesis is a diagnostic test that may be recommended following an abnormal triple test result. This is a common invasive prenatal diagnostic procedure, usually performed between 15 and 18 weeks gestation. Amniotic fluid is sampled by inserting a 22-gauge needle through the mother's anterior abdominal and uterine walls into the amniotic cavity by piercing the chorion and amnion. Because there is relatively little amniotic fluid before the 14th week, amniocentesis is difficult to perform before this time. The amniotic fluid volume is approximately 200 ml, and 15–20 ml can be safely withdrawn [84]. Amniocentesis detects chromosome abnormalities, neural tube defects and genetic disorders. Down syndrome or Trisomy 21 is the most common chromosome abnormality. Genetic disorders include disorders like cystic fibrosis. The most common neural tube defect is spina bifida. Amniocentesis is occasionally used late in pregnancy to assess fetal lung maturity [84, 85].

Alpha Feto Protein Assay

Alpha-fetoprotein (AFP) is a glycoprotein that is synthesized in the fetal liver, umbilical vesicle, and gut. AFP is found in high concentration in fetal serum, peaking 14 weeks after the LNMP. Small amounts of AFP normally enter the amniotic fluid. This protein is produced early in gestation by the fetal yolk sac and then later in the liver and gastrointestinal tract. The true function of AFP is unknown. We do know that this protein's level increases and decreases during certain weeks of pregnancy which is why accurate pregnancy dating is crucial.

The AFP test is measuring high and low levels of alpha-fetoprotein. The results are combined with the mother's age and ethnicity in order to assess probabilities of potential genetic disorders. Amniotic fluid AFP concentration is measured by immunoassay, and, when used with ultrasonographic scanning, approximately 99 % of fetuses with these severe defects can be diagnosed prenatally [85].

Characterization of First Trimester Fetal Erythroblasts for Non-invasive Prenatal Diagnosis

Isolation of fetal nucleated red blood cells (NRBC) from maternal blood would allow noninvasive prenatal diagnosis of chromosomal and monogenic disorders without the inherent risks of invasive procedures [86].

Components of this technique include enrichment, identification and molecular genetic diag-One study showed that primitive nosis. erythroblasts were the predominant cell type until 12 weeks gestation, after which time their numbers declined steeply; 100 % were ϵ -globinpositive versus <0.06 % definitive erythroblasts. Buoyant densities of first trimester fetal erythroblasts ranged from 1.077 to 1.130 g/ml, and optimal recoveries were obtained with Percoll 1118. Although primitive erythroblasts carried a negative surface charge and were resistant to NH₄Cl lysis, these properties had only a limited role in fetal cell enrichment. Immunophenotyping showed that primitive, like definitive, erythroblasts were GPA+, CD47+, CD45- and CD35-, whereas CD71 expression was weak/undetectable on primitive erythroblasts but strongly positive on 100 % of definitive erythroblasts; primitive erythroblasts were also CD36- whereas definitive erythroblasts were CD36+. Using chromosomal fluorescence in-situ hybridization (cFISH) or polymerase chain reaction (PCR) [87].

Spectrophotometric Studies

Examination of amniotic fluid by this method may be used for assessing the degree of erythroblastosis fetalis, also called hemolytic disease of the newborn. This disease results from destruction of fetal red blood cells by maternal antibodies. In 1961, Liley demonstrated the prognostic value of amniotic fluid spectrophotometry to identify these infants and then showed that intrauterine transfusions could prevent fetal deaths [88].

Chorionic Villus Sampling

Chorionic *villous* sampling (CVS) is a form of prenatal diagnosis to determine chromosomal or genetic disorders in the fetus. It entails sampling of the chorionic villus (placental tissue) and testing it for chromosomal abnormalities, usually with FISH or PCR. CVS usually takes place at 10–12 weeks' gestation, earlier than amniocentesis or percutaneous umbilical cord blood sampling. It is the preferred technique before 15 weeks [89, 90]. CVS was performed for the first time by Italian biologist Giuseppe Simoni, scientific director of Biocell Center, in 1983.

Conclusion

The fetus in its struggle for survival in the first two trimesters braves the environmental mileau that it is exposed to and adapts and modifies to realize its developmental potential determined by genetic and epigenetic factors. These two trimesters provide the fetus with unique protective abilities as the fetal cellular characteristics are quite divergent from its postnatal characteristics, being many times more sensitive, with the ability to heal without scar formation and fetal stem cells having much higher proliferative potential as well as differentiation capacity as the fetal tissue extracted in this period are able to regenerate diseased and damaged tissues with minimal GVHD. The fetal growth and development in the first two trimesters masterminds the health outcome of the adult and are hence of paramount importance.

References

- Harper D. Online etymology dictionary. 2001. http:// etymonline.com/. Retrieved 16 Apr 2015.
- Moore KL, Persuad TVN. The developing human: clinically oriented embryology; Ninth week to Birth: Fetal Period. 9th ed, Philadelphia: Elsevier Saunders; 2011. p. 93.

- Dunstan GR. The embryo, from Aristotle to Alton. Hist Today. 1988; 38(4). Accessed from www.historytoday.com/gr-dunstan. 17 Apr 2015.
- Visse R, Nagase H. Matrix metalloproteinases and tissue inhibitors of metalloproteinases: structure, function, and biochemistry. Accessed from http://circres. ahajournals.org/content/92/8/827.full.
- Garg HG, Longaker MT, editors. Scarless wound healing. Accessed from http://www.scribd.com/ doc/36358407/Wound-Healing#.
- Hilary G. Lkeonardo da Vinci's embryological annotations. Accessed from https://embryo.asu.edu/handle/10776/2031. 17 Apr 2015.
- An analysis of the De generationeAnimalium of William Harvey by Arther William Meyer. Accessed from http://www.jstor.org/stable/226279. 17 Apr 2015.
- Horder TJ, Witkowski JA, Wylie CC, editors. A history of embryology. Cambridge: Cambridge University Press; 1986. p. 113–46.
- Medina J. Womb with a view: the research genetics of the 21st century. Accessed from http://www.counterbalance.org/scisuff-print. 16 Apr 2015.
- Moore KL, Persuad TVN. The developing human: clinically oriented embryology; Introduction to the Developing Human. 9th ed, Philadelphia: Elsevier Saunders; 2011. p. 10.
- Barton PJR, Boheler KR, Brand NJ, Thomas PS. Basic concepts in molecular biology; Molecular biology of cardiac development and growth. 1995. p. 1–24. Accessed from http://www.Link.springer/chapter/10. 18 Apr 2015.
- Langman's medical embryology. Accessed from http://www.slideshare.net/mejar1/langmans-medicalembryology. 19 Apr 2015.
- Lee S, Kim T, Lee H, Park J, Chung S, Jeon D. Length measurement of fetal long bone and fetal anomaly detection. Webmed Cent Obstet Gynaecol. 2013;4(5), WMC004236. doi:10.9754/journal. wmc.2013.004236.
- Handin RI, Lux SE, Stossel TP. editors. Blood: principles and practice of hematology, vol 1. Accessed from: edited by https://books.google.co.in/books/ isbn=0781719933.
- Moore KL, Persaud TVN, Torchia MG. Before we are born: essentials of embryology and birth defects. Accessed from https://books.google.co.in/ books?isbn=1437701051.
- Moore KL, Persuad TVN. The developing human: clinically oriented embryology; Ninth week to Birth: Fetal Period. 9th ed, Philadelphia: Elsevier Saunders; 2011.
- Falkner F, Tanner JM. Human growth: a comprehensive treatise volume 1 developmental biology. Accessed from https://books.google.co.in/ books?isbn=1461321018.
- Pregnancy and the development of the child. Accessed from http://www.ewtn.com/library/PROLIFE/pregnancy.HTM.
- Embryology of the respiratory system. Accessed from http://embryology4genius.weebly.com/maturationof-the-lungs.html.

- Cardiac development; Cell biology. Margaret L. Kirby Professor of Pediatrics, and Biology Duke University Medical Center. Accessed from https://books.google. co.in/books?isbn=019972086X.
- Berne RM, Levy MN. Fijiologija. 3rd ed. Zagreb: Medicinska naklada; 1996. p. 489–91. 831–63, 879– 907, 908–48.
- 22. Bancroft J. Researches in prenatal life. Oxford: Blackwell; 1946.
- Hall GI. Medicinska Fijiologija, 11. Izdanje. Zagreb:Medicinska naklada; 2006. p. 1042–52, 1027–41, 918–30.
- Jones CT, Robinson RO. Plasma catecholamines in fetal and adult sheep. J Physiol. 1975;248:15–33.
- 25. Green LR, Mc Garrigle HHG, Bennet L, et al. The effect of acute hypoxemia on plasma angiotensin II in intact and carotid sinus denervated fetal sheep. J Physiol. 1994;470(P):81P.
- Cunnighum FG, Mc Donald PC, Gant NF, et al. Williams obstetrics. 20th ed. Stanford: Appleton and Lange; 1997.
- 27. Diamant NE. Development of esophageal function. Am Rev Respir Dis. 1985;131:S 29–32.
- El Haddad MA, desai M, Gayle D, et al. In uero development of fetal thirst and appetite potential for programming. J Soc Gynaecol Investig. 2004;11(3):123–30.
- Kawamura k, Takebayashi S. The development of nor-adrenaline, acetylcholinesterase, neuropeptide Y VIP containing nerves in human cerebral arteries. Neurosci Lett. 1994;175(1–2):1–4.
- Jacquet D, Leger J, Levy Merchal C, et al. Ontogeny of leptin in human fetuses and newborns; effects of intrauterine growth retardation on serum leptin concentration. J Clin Endicrinol Metab. 1998;83(4):1243–6.
- Aparacio T, Kermorgrant S, Darmoul D, et al. Leptin and Ob- Rb receptor isoforms in the human digestive tract during fetal development. J Clin Endicrinol Metab. 2005;90(11):6177–84.
- Trivedi VN, Hay P, Hay JC. Normal embryonic and fetal development. Oxford: Blackwell Publishing; 2007. p. 19–35.
- Kleinman LI. The kidney. In: Stave U, editor. Perinatal physiology. New York: Plenum Medical Book Company; 1978. p. 589–616.
- Gilbert T, Merlet-Benichou C. Retinoids and nephron mass control. Pediatr Nephrol. 1994;8(2):175–80.
- Battaglia FC, Meschia IG. An introduction to fetal physiology. Orlando: Academic; 1986. p. 154–67.
- McCance RA, Widdowson EM. Renal function before birth. Studies in perinatal physiology. Bath: Pitman Press; 1980. p. 94–103.
- Okado N, Kakimi S, Kojima T. Synaptogenesis in the cervical cord of human embryo; sequence of synapse formation in a spinal reflex pathway. J Comp Neurol. 1979;184(3):491–518.
- Joseph R. Fetal brain and cognitive development. Dev Rev. 1999;20:81–98.
- Pomeroy SL, Volpe JJ. Development of nervous system. In: Pollin RA, Fox WW, editors. Fetal and neonatal physiology. Philadelphia: WB Saunders Company; 1992. p. 1491–509.

- Kurjak A, Azumendi G, Vecek N, et al. Fetal hand movement and facial expression in normal pregnancy studied by 4D sonography. J Perinat Med. 2003;31(6):496–508.
- Awoust J, Levi S. Neurological maturation of the human fetus. Ultrasound Med Biol. 1983;9(2);Suppl 2:583–7.
- Yan F, Dai SY, Akther N, et al. 4 D sonographical assessment of fetal facial expression. Int J Gynaecol Obstet. 2006;94(2):108–13.
- Lowery CL, Hardman MP, Manning N, et al. Neurodevelopmental changes of fetal pain. Semin Perinatol. 2007;31(5):275–82.
- 44. Sookoian S, Fernández Gianotti T, Burgueño AL, Pirola CJ. Fetal metabolic programming and epigenetic modifications: a systems biology approach. Pediatr Res. 2013;73:531–42. Accessed from http://www. nature.com/pr/journal/v73/n4-2/full/pr20132a.html.
- 45. Timothy RH, Limesand SW, Hay WW. Factors influencing fetal growth. 2001. Accessed from http:// www.Neoreviews.aappublications.org/content/2/6/ e119.extract. 18 Apr 2015.
- Marieb EN. Human anatomy and physiology. 5th ed. San Francisco: Benjamin Cummings; 2000. p. 1118–48.
- Jhonson MH, Everitt BL. Essential reproduction. 5th ed. Oxford: Blackwell Science; 2000. p. 203–22.
- Lemons JA, Ridenour R, Orsini EN. Congenital absence of pancreas and intrauterine growth retardation. Pediatrics. 1979;64(2):255–7.
- 49. Ross MG, Ervin MG, Novak D. Fetal physiology. In: Gabbe SG, Niebyl JR, Simpson JL, editors. Obstetrics normal and problem pregnancies. Philadelphia: Churchill Livingstone Elsevier; 2007. p. 26–54.
- Wu G, Bazer FW, Cudd TA, Meininger CJ, Spencer TE. Maternal nutrition and fetal development. The American Society for Nutritional Sciences. J Nutr. 2004;134(9):2169–72.
- 51. American pregnancy association. Promoting pregnancy wellness. Smoking during pregnancy. Accessed from http://americanpregnancy.org/ pregnancy-health/smoking-during-pregnancy/.
- Ulleland CN. The offspring of alcoholic mothers. Ann N Y Acad Sci. 1972;197:167–9. doi:10.1111/ j.1749-6632.1972.tb28142.x. PMID 4504588.
- Lemoine P, Harousseau H, Borteyru JB, Menuet JC. Les enfants de parents alcooliques. Anomalies observées, à propos de 127 cas. Quest Med. 1968;21:476–82. PMID 12657907.
- Streissguth A. Fetal alcohol syndrome: a guide for families and communities. Baltimore: Brookes Publishing; 1997. ISBN 1-55766-283-5.
- 55. Ethen MK, Ramadhani TA, Scheuerle AE, et al. Alcohol consumption by women before and during pregnancy. Matern Child Health J. 2008;13(2):274–85. doi:10.1007/s10995-008-0328-2.
- 56. Fred Hutchinson cancer research center. Microchimerism: Retained cells from transfer during pregnancy. A new paradigm for human health and disease? Accessed from http://www.microchimerism. org/what-microchimerism-does.html. 18 Apr 2015.

- Borish LC, Steinke JW. 2. Cytokines and chemokines. J Allergy Clin Immunol. 2003;111(2 Suppl):S460–75. | Article | PubMed | ChemPort |.
- Dantzer R, Wollman EE. Relationships between the brain and the immune system. J Soc Biol. 2003;197:81–8. | PubMed |.
- Luheshi GN. Cytokines and fever. Mechanisms and sites of action. Ann N Y Acad Sci. 1998;856:83–9. | PubMed | ISI | ChemPort |.
- Dammann O, Leviton A. Infection remote from the brain, neonatal white matter damage, and cerebral palsy in the preterm infant. Semin Pediatr Neurol. 1998;5:190–201. | PubMed | ChemPort |.
- Zhao B, Schwartz JP. Involvement of cytokines in normal CNS development and neurological diseases: recent progress and perspectives. J Neurosci Res. 1998;52:7–16. | Article | PubMed | ISI | ChemPort |.
- Allan SM, Rothwell NJ. Inflammation in central nervous system injury. Philos Trans R Soc Lond B Biol Sci. 2003;358:1669–77. | Article | PubMed | ISI | ChemPort |.
- Marx CE, Jarskog LF, Lauder JM, Lieberman JA, Gilmore JH. Cytokine effects on cortical neuron MAP-2 immunoreactivity: implications for schizophrenia. Biol Psychiatry. 2001;50:743–9. | Article | PubMed | ISI | ChemPort |.
- 64. Cai Z, Pan ZL, Pang Y, Evans OB, Rhodes PG. Cytokine induction in fetal rat brains and brain injury in neonatal rats after maternal lipopolysaccharide administration. Pediatr Res. 2000;47:64–72. | PubMed | ISI | ChemPort |.
- 65. Urakubo A, Jarskog LF, Lieberman JA, Gilmore JH. Prenatal exposure to maternal infection alters cytokine expression in the placenta, amniotic fluid, and fetal brain. Schizophr Res. 2001;47:27–36. | Article | PubMed | ISI | ChemPort |.
- Beishuizen A, Thijs LG. Endotoxin and the hypothalamo-pituitary-adrenal (HPA) axis. J Endotoxin Res. 2003;9:3–24. | Article | PubMed | ISI | ChemPort |.
- Turnbull AV, Rivier CL. Regulation of the hypothalamic-pituitary-adrenal axis by cytokines: actions and mechanisms of action. Physiol Rev. 1999;9:1–71.
 PubMed | ISI | ChemPort |.
- Dubuis JM, Dayer JM, Siegrist-Kaiser CA, Burger AG. Human recombinant interleukin-1 beta decreases plasma thyroid hormone and thyroid stimulating hormone levels in rats. Endocrinology. 1988;123:2175– 81. | PubMed | ISI | ChemPort |.
- Wartofsky L, Burman KD. Alterations in thyroid function in patients with systemic illness: the 'euthyroid sick syndrome'. Endocr Rev. 1982;3:164–217. | PubMed | ISI | ChemPort |.
- Dupouy JP, Coffigny H, Magre S. Maternal and foetal corticosterone levels during late pregnancy in rats. J Endocrinol. 1975;65:347–52. | PubMed | ISI | ChemPort |.
- Zarrow MX, Philpott JE, Denenberg VH. Passage of 14C-4-corticosterone from the rat mother to the foetus and neonate. Nature. 1970;226:1058–9. | Article | PubMed | ISI | ChemPort |.

- Porterfield SP, Hendrich CE. The thyroidectomized pregnant rat – an animal model to study fetal effects of maternal hypothyroidism. Adv Exp Med Biol. 1991;299:107–32. | PubMed | ChemPort |.
- Matthews SG. Antenatal glucocorticoids and programming of the developing CNS. Pediatr Res. 2000;47:291–300. | PubMed | ISI | ChemPort |.
- 74. Sampson D, Pickard MR, Sinha AK, Evans IM, Leonard AJ, Ekins RP. Maternal thyroid status regulates the expression of neuronal and astrocytic cytoskeletal proteins in the fetal brain. J Endocrinol. 2000;167:439–45. | Article | PubMed | ISI | ChemPort |.
- Koenig JI, Kirkpatrick B, Lee P. Glucocorticoid hormones and early brain development in schizophrenia. Neuropsychopharmacology. 2002;27:309–18. | Article | PubMed | ISI | ChemPort |.
- Mor G, editor. Immunology of pregnancy. Springer Landes Biosciences, Georgetown, Texas, USA. 2006. p. 1–6.
- Sargent I, Sacks G, Redman C. Immunology of implantation and pregnancy. Accessed from http:// www.endocrine-abstracts.org/ea/0004/ea0004s19. htm. Published: 2002-11-01. ISBN: 978-0-387-30612-4 (Print) 978-0-387-34944-2 (Online).
- Mead R, Hinchliffe SJ, Morgan BP. Molecular cloning, expression and characterization of the rat analogue of human membrane cofactor protein (MCP/ CD46). Immunology. 1999;98(1):137–43. Accessed from: www.ncbi.nlm.nih.gov/pubmed/10469244.
- Mor G, editor. Immunology of pregnancy. Springer Landes Biosciences, Georgetown, Texas, USA. 2006. p. 101–8.
- Finn R. Survival of the genetically incompatible fetal allograft. Lancet. 1975;1(7911):835–8. Accessed from http://www.ncbi.nlm.nih.gov/pubmed/48059. on 18 Apr 2015. ISBN: 978-0-387-30612-4 (Print) 978-0-387-34944-2 (Online).
- Sepulveda W, Odibo A, Sebire NJ, et al. The lambda sign at 10 to 14 weeks of gestation as a predictor of chorionicity in twin pregnancies. Ultrasound Obstet Gynecol. 1996;7(6):421–3.
- Kurjak A, Chervenak FA, editors. Donald school text book of ultrasound in obstetrics & gynaecology. 3rd ed. New Delhi: Jaypee Brothers Medical Publishers; 2011.
- Lee S, Kim T, Lee H, Park J, Chung S, Jeon D. Length measurement of fetal long bone and fetal anomaly detection. Obstet Gynecol. 2013;4(5):WMC004236. doi:10.9754/journal.wmc.2013.004236.
- 84. Moore KL, Persuad TVN, Torchia MG. The developing human: clinically oriented embryology; Ninth week to birth: fetal period. 9th ed, Philadelphia: Elsevier Saunders; 2011. p. 102.
- 85. Prenatal diagnosis: amniocentesis and CVS. Accessed from http://familydoctor.org/familydoctor/ en/pregnancy-newborns/fetal-health/prenataldiagnosis-amniocentesis-and-cvs.html.
- 86. Accessed from http://americanpregnancy.org/ prenatal-testing/amniocentesis/.

- 87. Bianchi DW, Simpson JL, Jackson LG, Elias S, Holzgreve W, Evans MI, Dukes KA, Sullivan LM, Klinger KW, Bischoff FZ, et al. Fetal gender and aneuploidy detection using fetal cells in maternal blood: analysis of NIFTY I data. Prenat Diagn. 2002;22:609–15.
- Choolani M, O'Donoghue K, Talbert D, Kumar S, Roberts I, Letsky E, Bennett PR, Fisk NM. Characterization of first trimester fetal erythroblasts for non-invasive prenatal diagnosis. Mol Hum Reprod. 2003;9(4):227–35. doi:10.1093/ molehr/gag027.
- Liley AW. The use of amniocentesis and fetal transfusion in erythroblastosis fetalis. Pediatrics. 1965;35:836. Cited in chapter 58. Alloimmune hemolytic disease of the newborn; Williams hematology. Accessed from https://medtextfree.wordpress.com/chapter-58-alloimmune-hemolytic-disease-of-the-newborn.
- Alfirevic Z, von Dadelszen P. Alfirevic Z. editor. Instruments for chorionic villus sampling for prenatal diagnosis. Cochrane Database Syst Rev. 2003;(1): CD000114. doi:10.1002/14651858.CD000114. PMID 12535386.

Part II

Structural and Functional Fetal Development up to Second Trimester

Anthropometric Measurement of the Human Fetus

Chameli Ganguly[†], Bimal Samanta, Gitanjaly Guha Thakurata[†], Nemaichand Chandra, Sukla Ghosh, K.L. Mukherjee[†], and Niranjan Bhattacharya

Anthropometric Measurement of Human Fetuses

The present chapter reports the findings of a study conducted at the Institute of Post Graduate Medical Education and Research, Calcutta, by a group of researchers led by Prof KL Mukherjee, who headed the Department of Biochemistry. The study was conducted from 1977 to 1987. The foetuses were collected from SSKM Hospital, Calcutta, with the informed consent of mothers

Former Biochemist Central Calcutta Society for Advancement of Human Development and Research, Kolkata 700040, India

B. Samanta, MSc, PhD (Cal) Central Calcutta Society for Advancement of Human Development and Research, Kolkata, India

Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

G.G. Thakurata, MSc, PhD (Cal) Formerly, Department of Biochemistry, National Medical College and Hospital, Kolkata 700040, India

N. Chandra, MSc, PhD (Cal) Department of Biochemistry, All India Institute of Medical Sciences, Patna, India

S. Ghosh, MSc, PhD (Cal) Department of Zoology, Ballygunge Science College, Calcutta University, Kolkata, India who had undergone medical termination of pregnancy, which is legal under India's MTP law under certain circumstances. For the sake of convenience the fetuses have been divided into six groups: – A, B, C, D, E and F according to a difference in the gestation period of 4 weeks. The gestational age was ascertained form the date of the last menstrual period.

Table 6.1 shows the grouping and anthropometric measurement of human fetuses in this series.

- Group A contained 90 fetuses of 9–12 week of gestation. Their weights varied form 1–14 g, the C.R. lengths form 22–67 mm, C.H. lengths form 30–96 mm and H.C. form 34–82 mm.
- Group B contained 337 fetuses of 13–16 weeks of gestation. Their weights varied form 15–105 g, the C.R. lengths from 62 to 113 mm, C.H. lengths form 72 to 185 mm, and H.C. lengths form 58–145 mm.

K.L. Mukherjee, MB, PhD (Cal), PhD (Wisconsin) Former Head of the Department of Biochemistry, Institute of Post Graduate Medical Education and Research, Kolkata, West Bengal, India

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India e-mail: sanjuktaniranjan@gmail.com

© Springer International Publishing Switzerland 2016 N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development: First and Second Trimesters*, DOI 10.1007/978-3-319-14874-8_6

[†]Author was deceased at the time of publication.

C. Ganguly, MSc, PhD

Group	Gestation period (wks)	Range of body weight (gms)	No. of cases	C.R. (mm)	C.H. (mm)	H.C. (mm)
А	(9–12)	(1–14)	90	22-67	30–96	34-82
В	(13–16)	(15–105)	337	62–113	72–185	58–145
С	(17–20)	(106–310)	425	107-228	107-330	130–248
D	(21–24)	(311–640)	531	140-220	169–372	160-235
Е	(25–28)	(641–1,071)	10	193–247	303-336	198–270
F	(29–32)	(1,072–1,520)	2	250	400	280
G	(33–36)	(1,521–2,196)	1	280	418	316

 Table 6.1 Grouping of fetuses and anthropometric measurements

CR Crown Rump

CH Crown Heel

HC Head Circumference

- Group C contained 425 fetuses of 17–20 weeks of gestation. Their weights varied form 106– 310 g, the C.R. lengths from 107 to 228 mm, C.H. lengths form 117–330 mm, and H.C. lengths form 130–248 mm.
- Group D contained 531 fetuses of 21–24 weeks of gestation. Their weights varied form 331–640 g, the C.R. lengths from 140 to 220 mm, C.H. lengths form 169–372 mm, and H.C. lengths form 160–235 mm; some still born fetuses of longer gestation periods were available for these measurements. They include E,F and G.
- Group E contained ten fetuses of 25–28 weeks of gestation. Their weights varied form 641– 1,071 g, the C.R. lengths from 193 to 247 mm, C.H. lengths form 303–336 mm, and H.C. lengths form 198–270 mm.
- Group F and G contained only three, whose gestation period were 29–32 and 33–36 weeks, C.R.,
 C.H. and H.C. lengths in group F, 250 mm, 400 mm and 280 mm respectively. In group G they were 280, 418 and 316 mm respectively.

Since much of the body weight consists of the skeleton, it is apparent that the overall anthropometric measurements bear curvilinear relationship to the body weight (Fig. 6.1).

Fetuses of group E to group G were all still born fetuses.

Table 6.2 and Fig. 6.2 shows the relation between the body weight and the corresponding crown rump lengths. There is no straight line relationship between the length and the body weight of fetuses. The slope of the figure indicates that the rate of length (mm) change per unit mass (gm) decreases with time. Figure 6.1 shows that the body weight increases with progress of gestation and rate of increase of body weight rises with progress of gestation.

The question here is how crown-rump length (CR), crown-heel length (CH), head circumference (HC) grow in relation to body weight of human fetuses at different periods of gestation; this issue has been the subject of great concern to bio-statisticians over decades. Studies on the variation of these vital biological parameters during human fetal growth are based mainly on fitting of linear curves [1–7]. In our analysis (Figs. 6.3, 6.4, and 6.5) we have fitted three different curves for the purpose of analysis of trends in growth of CR, CH and HC corresponding to different body weights of human fetuses from 9 to 28 weeks of gestation.

We related the various anthropometric parameters to the fetal body weight. The body weight itself is a function of growth in respect to gestational age. There will be variations in the body weights of fetuses of a given gestational age. Thus a fetus of 16 weeks of gestation may weigh from 90 to 110 g. This amount of variation applies to all our anthropometric measurements.

There is an intrinsic growth rate of a normal fetus with respect to time in a given species. We have no idea about the factor controlling this rate of growth. We cover up our ignorance by saying that the rate of growth of a particular human fetus depends upon its intrinsic growth potential, inherent in its genomic complex which is modified by the environment, provided locally by the mother's uterine complex. It we assume that the local environment of a normal mother is optimal the individual growth of the fetus is a function of the genetic potential. Furthermore, the fetus is a



Table 6.2 Grouping of fetuses and weight length relationship

Group	No. of fetuses	Week	Weight (gm)	Length (mm) (crown rump)	Length/weight
А	15	8-12	1.5-14.5	25–53	4.6
В	25	13–16	15.6-103	62–109	1.1
С	36	17–20	115-295	110–157	0.90
D	21	21–24	330-660	163–211	0.37
Е	14	25–28	715–1,025	203–252	0.20
F	3	29–32	1,055-1,650	230–277	0.17

composite of the growth of the organs it is constituted of. Growth of the individual organs is conditioned again by the genetic potential and the environment provided by the mother and a third component, the fetus itself. With these limitations in mind, we approach the data on the relationship of the various anthropometric measurements to fetal growth (which is the body weight in our case). Such relationships have been determined by a number of workers including ours [8-14]. Older data were based on the Carnegle Institute Collection. Streeter et.al. [15] and Schults et.al.



Fig. 6.3 Trend in

Fig. 6.4 Trend in

of foetuses (9-28

in relation to

weeks)





70



[16, 17] studied the dimensions in fetuses obtained information from spontaneous abortions or pathological pregnancies, on mainly formal and fixed specimens.

nancies, on mainly formal and fixed specimens. Figure 6.6 shows the volume of fetuses bears a relationship to the body weight which almost passes through the origin of X and Y axis.

Anthropometry of Individual Organs

The growth of an organ from inception to a definitive functional stage is an integrated function of the whole organism. The related information of such growth is inbuilt in the nucleic acids of the cells of the organ. Such growth is probably independent of the development of the organism as a whole, except in so far as it is a subject to hormonal regulation. However, when it starts functioning, the functioning may be important for the development of the whole organism and thus will be related to the overall growth. As noted before organogenesis and throughout fetal life there is a characteristic growth rate of each individual organ of the fetus. In most organs the process is continued after birth.

Group	Period of gestation (weeks)	Number of cases	Body weight mean ± S.D (gm)	Liver weight mean ± S.D (gm)	Liver weight per 100 g body weight (gm)
А	9–12	74	9±3	0.41 ± 0.20	4.60
В	13–16	208	50±25	2.07 ± 1.04	4.25
С	17–20	192	187±56	7.70 ± 2.36	4.13
D	21–24	68	420±80	16.70±3.35	3.95
Е	25–28	27	738±116	30.10±4.38	4.10

 Table 6.3
 Relationship between the body weight and the weight of liver of human fetuses





As the fetus grows during gestation, its organs also grow in progression to the body weight. Regulation of the growth process is complex and not completely understood, while the genetic potentiality of the different endocrine organs related to growth and various unknown parameter mediate the growth regulatory process. It is important to determine some of these parameters.

Growth of the Liver

The growth in mass of the human fetal liver was compared to the overall body weight of the fetus. The liver at 8–12 weeks of gestation is relatively a bigger organ with respect to body weight, forming 4.5–5.5 % of body weight. It protrudes through the abdominal wall. Thereafter from 13 to 32 weeks of gestation it forms 3.4-4.0 % of body weight. As a result the liver weight forms a more or less constant proportion of body weight. Results are shown in Table 6.3 and Fig. 6.7. They show that the liver weight is directly proportional to body weight.

Growth of the Lung

The lung is a respiratory organ, but it is not so in fetal life, although the fetal lungs are known to expand and contract in the last stages of development. Therefore, whatever growth pattern is observed in the fetal lung, it will be an intrinsic pattern of the organ.

The total weight of the right and left lungs and both lungs together are determined at different periods of gestation and expressed as a function of the total body weight. Results are shown in Table 6.4 and Fig. 6.8. There was a variation in individual fetuses; the weight of right and left lung and total weight of the lungs increased with the total weight of the fetus (Fig. 6.8). When the weight of the lung was expressed as gm/Kg of body weight (Table 6.4) and plotted against the body weight, we get somewhat irregular plot as shown in Fig. 6.9. Initially, that is, upto a body weight 350 g, there was a rather sharp decline in the rate of growth of the lungs. The decline was maintained upto a body weight of 850 g after which the rate of growth was uniformly proportional to the total body weight.

Growth of the Brain

The brain, for our studies, included the central nervous system upto the medulla at the level of the second cervical vertebra. The whole mass of tissue was dissected out, the cerebrospinal fluid was drained by decantation and the weight of the brain was determined. The smallest fetus we could examine weighed 1.5 g and thus belonged to a gestation period of 8.5 weeks. The brain at this time had already assumed the appearance of a human brain; i.e., the primary divisions and flexures had already occurred previously and the prosen, messen, and rhomben cephalon had already given rise to the constituent derivatives of the brain like rhinencephalon, corporastraita, cerebral cortex, thalamus and epi and hypothalamus, col-

Gestational age	Body wt. (gm)	Number of fetuses	Lung wt. $(M \pm SD)$ gm/Kg of body wt.		
(weeks)			Right lung	Left lung	Total lung
12–16	14–105	10	19±2.5	16±2.1	35 ± 1.0
16–20	105-310	9	16±1.6	13±1.6	30±3.4
20–24	310-640	18	15±2.0	12±2.1	27±2.3
24–28	640-1,080	11	12±1.2	10±0.8	22±2.0
28-32	1,080-1,650	10	12±2	10±1.4	22±2.8

Table 6.4 Weight of fetal lung in different gestation period









 Table 6.5
 Brain weight g/Kg. of body weight

Body wt.	No. of samples	Mean ± S.D
0–100	26	172±28
100-200	14	165 ± 26
200-300	12	160±17
300-400	5	153±23
400–500	3	165±15
500-600	2	130±
600–700	3	157±12
700-800	3	191±14
800–900	2	160±
900-1,000	3	172±16
1,000-2,000	5	157±15

liculi, tegmentum, crurcerebri, cerebellum, pons and medulla, albeit to a relatively less differentiated extent. From this time onwards the growth of the brain was proportional to the weight of the fetus. The brain weight per Kg of body weight is shown in Table 6.5 and Fig. 6.10. It formed 16-18 % of the body weight. There were occasional fetuses whose brains weighed 20 % of the body weight and some brains were 12-13 % of the body weight. But by and large, the brains weighed 16-18 % of the body weight. Figure 6.11 shows that brain weight is directly proportional to the body weight.

Why the variations of weight are related to the large capacity of the brain is not known. Thus, why the brain of Stephen Hawking may belong to the first groups of 16–18 % of the body weight, while the brains of Tom, Dick and Harry may belong to the second group of 12–13 % of body weight, is not known.

Growth of the Kidney and the Adrenal Gland

Table 6.6 and Fig. 6.12 shows the total weights of the adrenal glands and the kidneys in 12 human fetuses arranged according to body weight. In early periods of gestation, the adrenal glands out weighed even the metanephric kidneys. It was comparatively a bigger organ. After the tenth week of fetal life, the kidneys began to grow in mass at a faster rate than the adrenal glands so that the adrenals and the kidneys weighed more or less the same upto the 12th week. Thereafter, the kidneys weighed more than the adrenals. But even so, the adrenal gland is a comparatively bigger organ in the fetus than in an adult.

Table 6.7 shows the adrenal weights of 90 human fetuses of different gestation periods. Figure 6.12 shows the results graphically where the weights of the adrenal glands are plotted against gestation age. There is a progressive



increase in the weights of the adrenal glands as the fetus increase in weight. The rate of increase, however, varies at different gestational periods, and is not uniform all throughout. Figure 6.13 shows the weights of adrenal glands expressed as mg. per gm of body weight against gestational age. There is a progressive decrease in the relative weight of the adrenal glands as the fetus advances in age. The rate of decreases is not uniform and greatest after birth.

Growth of the Human Fetal Testes

The growth and development of the testes in human fetuses are not uniform throughout gestation. The right and left testes were dissected individually, weighed and expressed as mg tissue weight. Results are shown in Table 6.8. The right testes were, as a rule, 4-10 % heavier than the left with few exceptions. There was a gradual

increase in the weight of the testes (combined weights) as the gestation increased. The individual weights of the combined testes in mg were plotted against the weight of the fetus in gms. Results are shown in Fig. 6.14. The growth was not uniform; initially there was a proportional growth which soon flattened to increase with different slopes again and again throughout the whole length of the gestation. This is in marked contrast with the growth of an organ in mg per 100 g of body weight and results are plotted in a graphic form, (Fig. 6.15). There is a steep decline in the early gestation periods from about 200 mg per 100 g of body weight to about 60 mg per 100 g when the fetus weighted from 1.5 g to about 20 g. Later on in gestation, the decline is gradual. We could obtain data only upto a body weight of about 1,560 g when the testes weighed about 20 mg per 100 g of body weight. The decline was not uniformly maintained throughout the gestation period.





Table 6.6 Relationship between the total weight adrenalglands and kidneys with the increase of body weight

Body	Approx. gestation period	Total weight of kidneys	Total weight of adrenal glands
weight (gm)	(weeks)	(gm)	(gm)
6.578	10	0.0145	0.0266
7.0702	10	0.0235	0.0320
11.1834	11	0.0391	0.0264
13.2	11.5	0.0666	0.0511
14	12	0.1141	0.1010
14.4755	12	0.0711	0.0535
102	16	0.6011	0.2731
280	19.4	2.3626	0.9658
457	21.6	2.8350	1.3713
735	24.6	4.6049	2.3364
1,010	28	8.609	2.4395
1,650	32	13.350	4.1457

Growth of the Human Fetal Ovaries

The ovaries were dissected individually from each side and immediately weighed. Results are shown in Table 6.9 and Fig. 6.16. There was a progressive increase of the weight of the ovary with increase in body weight. In most instances fetuses of identical body weights had almost identical ovarian weights. But occasionally, within each group there were a few fetuses, whose ovarian weights did not have the expected ovarian weights.

The individual weights of the combined ovaries in mg were plotted against the weight of the fetuses in gms. Results are shown in Fig. 6.17. Like the testes, here also the growth was not uniform.


Fig. 6.12 Relation of weight of adrenal glands to gestational age

1	1	
Gestational period (weeks)	No. of cases	Total weight of the adrenal
Below 12 weeks	7	0.0451 ± 0.0277
	/	
12–16 weeks	23	0.1623 ± 0.0097
16-20 weeks	24	0.7341 ± 0.28
20-24 weeks	18	1.3008 ± 0.23
24-28 weeks	16	2.136±0.3
28-32 weeks	2	3.4471
New born ^a	1	4.5621
2 days	2	4.748
18 days ^b	1	3.7116
7 months	1	3.4211
3 years ^c	1	4.7264
4 and 1/2 years ^d	1	3.8876
Adult		8.0

Table 6.7	Weight	of	the	adrenal	glands	at	different
periods of h	uman fet	tal li	ife a	nd upto a	ıdult		

Results are expressed in mean and standard deviation for the number of fetuses indicated in parenthesis ^aPremature and 12 h of age

^bDied of Paralytic ileus

^cVitamin deficiency Patient

^dDied of Encephalitis

The weight of the organ in mg. is expressed per 100 g of body weight and results are plotted in a graphic form (Fig. 6.17). There is a steep decline in the early gestation period. The decline is not uniformly maintained throughout the gestation period.

The Growth in Mass of the Thymus

Detectable thymus was observed earliest in about 8 weeks of gestation when the fetus weighed about 1 g. In weeks earlier than 8, the thymus cannot be identified as an organ even under a magnifying glass. The thymic weight was expressed as mg/100 g of body weight. Table 6.10 shows some of these anthropometric measurements. Group A included 39 fetuses varying from a body weight of 1.3–14.7 g. In many of these fetuses, especially of in the smaller ones, thymic tissue could not be dissected. In all fetuses weighing more than 5 g,



Fig. 6.13 Weight of adrenal glands mg per gm body weight/gestational age

Gestation period	No. of	Bodyweight	Weight of te	stes (mg) mea	n ± S.D	Tissue weight in mg/100 g of
(weeks)	sample	(gm)	Right	Left	Total	body weight mean ± S.D
9–12	11	5-11	4.5 ± 2.0	4.0 ± 2.0	8.5 ± 3.0	133.16±36.05
3–16	24	22-80	12.0 ± 4.0	9.0 ± 4.0	21.0±8.0	50.75±17.35
17–20	31	130–250	39.0±12.0	36.0 ± 14.0	75.0±23.0	40.28±2.89
21–24	24	350-450	70.0 ± 9.0	64.0±11.0	134 ± 20	30.42±2.60
25–28	13	700–1,000	93±11	87±9	180±19	21.70±1.42
29–32	4	1,100-1,400	132±12	0.113 ± 20	245±31	20.06±0.46
33–36	1	1553.00	161	157	318	20.47

Table 6.8 Growth of the testes in developing human fetus

thymic tissue is dissectable. The organ formed 52 mg/100 g of body weight in this group. Group B included 28 fetuses of body weights varying from 15 to 100 g. The thymic weight expressed in mg per 100 g of body weight was 77 mg. This was higher than the thymic weight

of fetuses of Group A. The relative growth of thymus was thus more in this period than the previous. Group C included 39 fetuses of body weights varying form 100–300 g. The relative rate of growth of the thymus in this period was still higher than that of fetuses of group B, in as



Fig. 6.14 Weight of testes plotted against weight of fetus



Fig. 6.15 Relationship of fetal testes/body weight

much as it formed 136 mg per 100 g of body weight in contrast to 77 in group B. This increase in the relative growth of the thymus in proportion to body weight was maintained throughout the period of gestation. We could study foetuses upto 28 weeks; it was highest in the

Group and gestation	No. of	Body weight	Weight of ovary in (mg) mean +/- S.D			Tissue weight in mg/100 g
period (wks)	sample tested	in gms	Right	Left	Total	of body weight mean ± S.D
9–12	13	5-11	3.0 ± 1.5	3.0 ± 1.5	6.0 ± 3.0	100.30 ± 43.76
13–16	38	22-80	15.5 ± 8.0	13.6 ± 6.0	29±13	49.79±8.61
17–20	38	130-250	38 ± 8.0	34 ± 7.0	72±15	40.97 ± 6.83
21–24	26	350-450	81±9.0	72 ± 9.0	154 ± 15	35.74±4.39
25–28	10	700-1,000	95 ± 6.0	105 ± 35	0.200 ± 30	23.97 ± 2.77
29–32	4	1,100-1,400	110 ± 7.0	123±12	234±8	17.64±1.48
33–36	1	1553.00	152	149	298	19.10

Table 6.9 Growth of the ovaries in developing human fetus



Fig. 6.16 Relationship of fetal ovary and fetal body weight

fetuses of large body weight. This is in contrast to many other organs like the brain and liver which constitute a more or less constant proportion of body weight except in the case of very early fetuses. Since the weight of the thymuses is expressed per 100 g of body weight, it can be inferred that with progress of gestation there is an absolute as well as a relative increase in the mass of the thymus. However one thing must be mentioned in this context. There were occasional fetuses in all the groups in whom the thymic weight was not as high as would be expected from the body weight. In almost all cases we could predict the weight of the liver or



Fig. 6.17 Individual weights of combined ovaries in mg plotted against the weight of the fetuses in gms

Table 6.10	Growth of the thymus in developing human
fetus	

Gestational period (weeks)	Body wt. of fetuses (gms)	Thymic wt. in mg/100 g body weight
8-12 (39)	1.3–14.7	52.2±8.39
12-16 (28)	15-100	76.6±5.52
16-20 (39)	100-300	135.8±7.66
20-24 (24)	300–580	218.6±16.53
24-28 (19)	580-1,080	362.3±24.23

The results are expressed as mean and S.E.M. of the number of fetuses indicated in parenthesis

the brain or the kidney from the body weight but we could not do so in the case of thymuses.

Figure 6.18 shows the relationship between the body weight of the fetus (in gm) and the thymic weight (in mg.). It is apparent that there is no straight line relationship between the body weight and the thymic weight. The rate of growth varies at different periods of gestation.



Fig. 6.18 Relationship between thymic weight and body weight of fetus

References

- Lubchenko LO, Hansman C, Dressler M, Boyd E. Intrauterine growth as estimated form liveborn birth weight data at 24 to 38 weeks of gestation. Pediatrics. 1963;32:793–800.
- Lubchenko LO, Hansman C, Boyd E. Intrauterine growth in length and head circumference as estimated form live births at gestational ages form 26–42 weeks. Pediatrics. 1966;37:403–8.
- Usher R, McLean F. Intrauterine growth of live born Caucasian infants at sea level; standards obtained form measurements in 7 dimensions of infants born between 25 and 44 weeks of gestation. J Pediatr. 1969;74:901–10.
- 4. Berger GS, Edelman DA, Kerenyi TD. Fetal crownrump length and biparietal diameter in the second

trimester of pregnancy. Am J Obstet Gynecol. 1975;122:9–12.

- Birkbeck JA, Billewicz WZ, Thomson AM. Human fetal measurements between 50 and 150 days of gestation in relation to crown-heel length. Ann Hum Biol. 1975;2:173–8.
- Drumm JE, Clinch J, Mackenzie G. The ultrasonic measurement of fetal crown-rump length as a method assessing gestational age. Br J Obstet Gynaecol. 1976; 83:417–21.
- Bhatia BD, Tyagi NK. Birth weight; relationship with other fetal anthropometric parameters. Indian Pediatr. 1984;21:833–8.
- Campbell S. The prediction of fetal maturity by ultrasonic measurement of the biparietal diameter. J Obset Gynaecol Br Commonw. 1969;76:603–9.
- 9. Parekh UC, Pherwani A, Udani PM, Mukherjee S. Brain weight and head circumference in fetus, infant

and children of different nutritional and socio economic groups. Indian Pediatr. 1970;7:347–58.

- Prema LN, Nagaswamy S, Raju VB. Fetal growth as assessed by anthropometric measurements. Indian Pediatr. 1974;11:803–10.
- Brenner WE, Edelman DA, Hendricks CH. A standard of fetal growth for the United States of America. Am J Obstet Gynecol. 1976;126:555–64.
- Hern WM. Correlation of fetal age and measurements between 10 and 26 weeks of gestation. Obstet Gynaecol. 1984;63:26.
- 13. Mukherjee B, Mitra SC, Gunasegaran JP. Fetal crown-rump length and body weight at different

gestational periods. Indian J Med Res. 1986;83: 495–500.

- Sood M, Hingorani V, Kashyap N, Kumar S, Berry M, Bhargava S. Ultrasonic measurement of fetal parameters in normal pregnancy and in intrauterine growth relation. Indian J Med Res. 1988;87:453.
- Streeter GL. Weight, sitting height, foot length and menstrual age of the human embryo. Contrib Embryol (Carnegie Inst Wash). 1920;11:143–70.
- Schults AH. Fetal growth in man. Am J Phys Anthropol. 1923;6:389–99.
- 17. Schults AH. Fetal growth in man and other primates. Q Rev Boil. 1926;1:465–521.

Human Fetus: Carbohydrate Metabolism

Chameli Ganguly[†], Gitanjaly Guha Thakurata[†], Sukla Ghosh, K.L. Mukherjee[†], and Niranjan Bhattacharya

Introduction

Glucose is the unit of carbohydrate metabolism. All the three classes of carbohydrates mono, di and polysaccharides ultimately are converted to glucose and then metabolized. Some sugars join the common metabolic pathway of anaerobic glycosis. Except under special circumstances, for e.g., in galactosemic children there is no dearth of myelin on delivery. Therefore, the baby must have utilized galactose form the maternal circulation. In adult life, when the supply of glucose is not forthcoming like in the postprandial period, stored glycogen is broken down to glucose. Synthesis of glycogen is a function of the period when glucose is being absorbed form the gut. The state of glycogen synthesis and the breakdown becomes important in fetal life. In fetal

[†]Author was deceased at the time of publication

C. Ganguly, MSc, PhD Former Biochemist Central Calcutta Society for Advancement of Human Development and Research, Kolkata 700040, India

G.G. Thakurata, MSc, PhD (Cal) Formerly, Department of Biochemistry, National Medical College and Hospital, Kolkata 700040, India

S. Ghosh, MSc, PhD (Cal) Department of Zoology, Ballygunge Science College, Calcutta University, Kolkata 700040, India

K.L. Mukherjee, MB, PhD (Cal), PhD (Wisconsin) Former Head of the Department of Biochemistry, Institute of Post Graduate Medical Education and Research, Kolkata, West Bengal 700015, India life, however, the fetus is constantly supplied with glucose from maternal circulation. There is no need for glycogen to be synthesized and broken down in order to supply glucose to the circulation.

The history of glycogen synthesis and break down forms a very interesting part of the effort of many scientists.

In the inimitable word of Haldane, "Bernard was the first to formulate the extremely fruitful idea that the blood of a living animal in an internal medium is kept remarkably constant as regards its physicochemical condition by the co-ordinated influence upon it of the various organs of the body." The term internal medium, the milieu interieur, has been associated with the name of Claude Bernard [1], a pupil of the leading French physiologist Francois Magendie. Although Charles Robin [2] might have used the term earlier than Claude Bernard in his book, "Treatise of Anatomical and Physiological Chemistry, Normal and Pathologic," Bernard contemplated and discussed the idea of the internal environment for over 20 years and gradually added new facts of

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India e-mail: sanjuktaniranjan@gmail.com meaning to the idea of the milieu interieur. Almost at the same time P. Fluger [3], Frederick [4] and Riched [5] contributed immensely to the idea that internal environment is maintained by virtue of dynamic self-regulating mechanism, stability and constancy in the face of changing environment. Claude Bernard [1] said, "it is the fixity of the milieu interieur which is the condition of free and independent life; all the vital mechanisms, however varied they may be, have only one object, that of preserving constant the condition of life in the internal environment." According to him, the conditions which must be maintained constant in the fluid matrix of the body in order to favour freedom from external limitation are water, temperature and nutrient (including sugar). The concept of homeostasis was further elaborated by Canon [6]. The storage and release of carbohydrate as a means of keeping the blood sugar at a constant level under changing environmental conditions [7] was realized. Hansen [8] pointed out that there were normal oscillations in blood sugar occurring within a narrow range. Such insight and intuition in the 1920s were a real marvel when we realize that the methods of laboratory investigations were crude and rather imprecise in those days. That in the milieu interieur the blood sugar was maintained at a reasonably constant level was stressed by many authors ever since Otto Folin introduced the method of blood sugar estimation as a clinical laboratory technique. Thus it was found that after a meal rich in carbohydrate the bold sugar started to rise upto approximately 30-40 mg% above the fasting level at one and half hours and returned to the fasting level at two and half hours and remained at the level for a considerable period of time. During this period it was gradually realized that the human body was an organized system and an organized system required expenditure of energy in order to prevent the organization form spontaneous natural disorganization. It was found out that the energy was chemical in nature and was ultimately derived from the hydrolysis of nucleoside triphosphates. The synthesis of nucleoside triphosphates occurs mainly in the respiratory chain during operation of the tricarboxylic acie cycle. The tricarboxilic acid is formed by conjugation of oxaloacetic acid with acetyl coA. The Ac coA is derived from the decarboxlation of pyruvic acid.

The pyruvic acid is produced by anaerobic glycolysis of glucose. There are some organs of the body which are obligatorily dependent on a supply of glucose. Glucose must, therefore, be supplied to these organs constantly. The supply line is through the blood. The blood must therefore carry a more or less constant amount of glucose at all times.

This concept of homeostatic regulation of internal environment in the face of constantly changing external environment was forcefully advanced by Claude Bernard in the later half of the nineteenth century, as mentioned earlier. Subsequently it was gradually realized that all the chemical reactions in our body occurred at constant body temperature and required presence of catalysts. These biocatalysts were found to be the different enzymes having definite specificities with regard to the substrates they catalyze. These enzymes regulate the level of metabolites found in an ever changing system, wherein the metabolites enter and leave the system in a finely and delicately controlled manner. Organic reactions are reversible. Since the enzyme catalyzed reactions are organic in nature, it is to be expected that they may occur in both directions depending upon the energy consideration in the system. Moreover, most of the bodily reactions are in a flux, i.e., sequential, the products of one reaction are substrates for other reactions and thus the flow may virtually be unidirectional under a particular set of conditions. Hence in order to maintain the level of a metabolite at a particular concentration it is required that constant rate of input into the system is maintained as also the same rate of output out of the system.

When we come to consider the maintenance of a more or less constant level of blood sugar, we are confronted with a number of changing external conditions which tries to offset the maintenance of a constant level.

Firstly, we should be aware of the fact that blood sugar is never constant. For example, after a meal containing carbohydrates the blood sugar concentration rise to a level between 30 and 40 mg% higher than the fasting level at about an hour after the meal and return to the fasting level between two and two and a half hours after the meal. So there are periods in which blood sugar concertration is higher than the fasting level. However, the organisms tend to adopt to the high blood sugar by distributing it between the various fluids and metabolizing it either by catabolic reactions or by anabolic reactions. Conversely if food is withheld from our system, substances are broken down and mobilized in order to produce glucose and thus keep the blood sugar from falling. Therefore, by the statement "blood sugar concentration is kept constant", it is implied that blood sugar is kept within a certain level which varies depending upon various external conditions playing on the system.

In the human body the brain is dependent on the energy provided by glucose oxidation. Therefore the brain must always be supplied with glucose. In postnatal life, in the postprandial period glucose is added to the system by Glycogenolysis and Gluconeougenesis.

The human fetus is supplied with glucose from the maternal circulation. There is no necessity for Glycogenolysis and Gloconeogenesis.

Therefore it was considered important to study the glycogen content of the liver and muscle in different gestation period. This was part of the experiments done on the human fetus, mentioned in Chap. 6, by our group of researchers led by Prof. K.L. Mukherjee, at SSKM Hospital, Calcutta, from 1977 onwards.

Blood Sugar Concentration in Fetal and Maternal Blood

The mothers from whom the fetuses were obtained had opted for Medical Termination of Pregnancy which is legal under India's MTP Act, 1971, under specific circumstances. Due permission was taken from the hospital ethical committee and informed consent was taken from the mothers. Fetal blood was drawn from the heart within 10 min after the fetuses were taken out by hysterotomy. A sample of venous blood was obtained from the mother while she was under anesthesia, from the arm, opposite to the one, through which she was being infused with either saline or 5 % dextrose in saline. Blood sugar was estimated by the glucose oxidase method; in some cases the blood sugar was estimated by Folin and Wu's method in addition to the glucose oxidase method. Mothers were grouped in four groups of /4 week period beginning from

Table 7.1	Blood sugar concentration of fetal a	nd mater-
nal blood (glucose oxidase method)	

			Mean ± SD	Mean ± SD
	Gest		blood sugar	blood sugar
	period	No. of	in fetus	in mother
Group	(wks)	samples	(mg/dl)	(mg/dl)
А	13–16	8	80.6±10.72	59.2 ± 11.09
В	17–20	10	87.8 ± 16.4	77.9 ± 15.7
С	21–24	6	114 ± 26.5	106 ± 20.9
D	25–28	1	35.0	62.0

Table 7.2 Relationship between the high fetal blood sugar and the different glycemic condition of the mother

	Fetal blood sugar higher that maternal (% of total)		
	Higher	Lower	
Normogycemic mother	16	84	
Potentially diabetic mother	48	52	
Frankly diabetic mother	35	65	

13 weeks of gestation. Table 7.1 show the results obtained by glucose oxidase method on the pairs tested in different groups. The mean fetal blood sugar was higher than the mean maternal level but the difference was not statistically significant. In approximately 80 % cases the fetal blood sugar was lower than the maternal but in 20 %, the fetal sugar level was higher than the maternal.

Since in all published literature on the relationship between fetal and maternal blood sugar concentration, the fetal blood was shown to contain sugar at a lesser concentration than the maternal one [9, 10], we tried to investigate the cause of the high fetal sugar found to be present in some cases. The glycemic condition of the mother, whether normoglycemic, potentially diabetic or frankly diabetic had any relationship with the high fetal blood sugar is shown in Table 7.2. The potentially diabetic mother refers to those mothers whose hyperglycemia became normoglycemia after the termination of pregnancy. Of the normoglycemic mothers, 16 % showed high blood sugar level in the fetuses. In potentially diabetic mother about a half showed higher blood sugar level in the fetuses than the mothers. In frankly diabetic mother about a third showed higher blood sugar concentration in the fetuses.

Normoglycemia:	Fasting blood sugar 4.5–5.5 m Mol/l and post prandial blood sugar less than 6.5 m Mol/l
Potentially diabetic:	Blood sugar higher than 6.5 M Mol/l during pregnancy but below 5.5 m Mol/l after puerperium
Frankly diabetic:	The diabetic state persisting after puerperium

Glycogen Content in the Human Fetal Liver

One of the important parameters which maintains blood sugar level in the adult is liver glycogen. Glycogenolysis provides glucose-6-Phosphate (through glucose-1-P) which undergoes hydrolysis to glucose. The glycogen contents of the liver of those human fetuses were extracted within 15 min after removal from the uterus. The fetal liver glycogen was found to be reasonably stable for upto 24 h after the fetus was taken out. Portions of fetal livers were cut at 0.25, 1, 12 and 24 h. after their removal from the uterus and their glycogen contents were estimated (Table 7.3). Results agreed within reasonable limits (the later

Table 7.3 Glycogen stability in human fetal liver(gm/100 g)

Time	Temperature	Temperature of storage				
Hours	20 °C	−5 °C				
0.25	1.66	1.89				
1	1.59	1.77				
2	1.90	1.59				
12	1.30	1.87				
24	1.55	1.62				

values were more than 80 % of the 0 h values). Three methods of extraction were tried: hot 33 % KOH, hot water and cold TCA. Hot 33 % KOH yielded the highest glycogen content. Therefore, all subsequent investigations were carried out on hot alkali extracted liver. Glycogen was precipitated with alcohol upto 60 %. Two washings with 60 % ethanol were enough to isolate the glycogen. Final color reaction was carried out with anthrone reagent. Results were multiplied by 1.1 to compensate for the hydrolysis to glucose equivalent. Results are shown in Table 7.4. The data were statistically analysed with respect to significant differences between the liver glycogen contents of fetuses at different stages of gestation. At 9–12 weeks the liver glycogen was very little; especially, most fetal livers of 9 weeks hardly contained any glycogen at all.

The liver glycogen contents gradually increased from a mean value of less than 0.1 g% at 9–12 weeks around 1.8 g% between 21 and 24 weeks. The pairs were matched with regard to the mean of similar glycogen contents in order to compare the values statistically. The increase of liver glycogen contents between the fetuses of different gestation groups was highly significant.

Table 7.5 shows that the glycogen content of skeletal muscles was again lowest at 9-12 week (0.4 %). But from 13 to 24 weeks the muscle glycogen content remained more or less steady at around 1 %. As a result, the difference in skeletal muscle glycogen content between fetuses at different groups was only significant between the earliest gestation groups and between no other groups. If we compare the glycogen contents of the liver and muscles in the corresponding gestation periods, we find in Table 7.6 that skeletal muscles had higher glycogen contents than the liver at

 Table 7.4
 Glycogen content human fetal liver at different gestation period

Group	Weeks	Number of samples	Mean gm%	S.D.	P value	Remarks
А	9–12	14	0.08	0.09	< 0.001	Highly significant
В	13–16	32	0.57	0.54	< 0.001	Highly significant
С	17–20	32	0.93	0.36	< 0.001	Highly significant
D	21–24	19	1.83	0.70	0.001>P<0.005	Highly significant

		Number of				
Group	Weeks	samples	Mean gm%	S.D.	P value	Remarks
А	9-12	10	0.40	0.52		
В	13–16	10	1.07	0.29	< 0.001	Highly significant
С	17–20	10	0.96	0.36	< 0.500	N.S.
D	21–24	10	1.12	0.34	P >0.4 <0.2	N.S.

 Table 7.5
 Glycogen content of human fetal muscle at different gestation period

Table 7.6 Inorganic phosphorus content of human fetal liver at different gestation period

Age in week	Number of sample	Body weight (gms)	Liver weight (gms)	Total Pi (mg/100 g)
9–12	2	1–14	57.41 (mean)	0.47 (mean)
13-16	15	15-110	65.83 ± 4.82	3.12 ± 0.62
17-20	14	110-300	66.04 ± 4.89	7.18 ± 0.63
21-24	9	300-600	114±6.7	16.00±1.02
25–28	2	600– 1,000	199.2(mean)	-
37–40	2	1,533	1,133.2(mean)	-
Adult	2	_	53.7	_

9–12 week. Thereafter the values were similar and later on the liver glycogen contents were higher than the muscles.

Inorganic Phosphorus Content in the Human Fetal Liver

Phosphorus is a very important constituent of the organ both as regards its own metabolism as well in relation to the handling of glucose. Unlike the adult whose dietary intake contains abundance of phosphorus, the intrauterine fetus depends entirely on the maternal supply of inorganic phosphorus through its placental circulation. It was considered, therefore of interest to study the magnitude of phosphorus content in fetal organs at different periods of gestation. The human fetal liver was studied, and the results are shown in Table 7.6. The results are expressed as mg of inorganic phosphorus per 100 g of liver weight. From 9 to 21 weeks the total inorganic phosphorus contents were higher. Total liver inorganic phosphorus content were plotted against the body weight of the fetuses. There was an almost straight line relationship between the two parameters upto a body weight of 300 g; thereafter the slope was steeper. The inorganic phosphorus content of the mitochondrial fractions was usually lower than that of the cytosolic fraction. Two adult livers could be obtained during the period of study. The inorganic phosphorus content was lower than that of the fetal liver.

Phosphorylase Activity in the Human Fetal Liver

Glycogen is broken down to glucose-1-p by the activity of the enzyme phosphorylase. Since Keq of the enzyme is close to 1, the activity of the enzyme is usually measured by the synthetic reaction, i.e., incorporation of glucose into the growing glycogen chain and consequent liberation of inorganic phosphate form glucose-1-p in $8,000 \times g$ supernatant. Phosphorylase activity of the liver of a few fetuses was measured according to the method of Stalmans and Hers [11]. Results are shown in Table 7.7; the activity at 9-12 weeks was about a tenth of the adult liver activity of the adult livers. At 13-16 and 17-20 weeks the activity is about a fifth of the adult liver activity. According to Hers [12] the phosphorylase activity of adult human liver as obtained from liver biopsy, is 27.8 µ Mole pi/g tissue/min (mean value). Phophorylase activity of the skeletal muscle was estimated form crude homogenate of the thigh muscle (quadriceps Femoris). Results are shown in Table 7.8.

The enzyme activity of the fetal muscle was much lower than that of adult muscle reported by Hers [13]. In fact adult muscle phosphorylase activity was about four times higher than that of the adult liver. In comparison, however the

Gestation period (wks)	Number of specimen	Enzyme activity µ Mole pi/g/ tissue/min	Specimen activity µ Mole pi/g/ protein/min
9–12	2	2.52	13.39
13–16	7	5.60 ± 0.46	21.94±4.37
17–20	6	5.58±0.68	16.21±3.6
21–24	1	4.09	12.4
Adult ^a	4	27.8	-

Table 7.7 Phosphorylase activity of the human fetal liver at different period of gestation

^aAccording to Hers [12]

Table 7.8 Phosphorylase activity of human fetal muscle at different period of gestation

Period of gestation (wks)	Body weight (gm)	Phosphorylase activity µ Mole pi/g T/min
10	4.3	1.70
14	35.2	0.58
18	187.0	1.75
24	549.0	4.6
Adult ^a	_	100

^aAccording to Hers [12]

phosphorylase activity of the fetal muscle was lower than that of the fetal liver. The reported value of adult muscle phosphorylase by Hers [13] is 100 μ Mole Pi/g tissue/min. In other words the fetal muscle had very little phosphorylase activity.

Glucose: 6 Phosphatase Activity of the Human Fetal Liver

The hydrolysis of liver glucose -6-phosphate was carried out according to the method of Swanson [14]. The liver homogenate was carried out in maleate buffer (0.1 M, PH 6.5) at a concentration of around 50 mg per ml of liver homogenate. It was centrifuged at $600 \times g$ and the supernatant was used as the enzyme source. It was soon realized that the activity of the enzyme which was observed in human fetal liver of earlier gestation period might be an artifact, possibly due to non specific phosphatases present in the homogenate supernatant. Accordingly and as suggested by Hers [13], the nonspecific phosphatase was inhibited by prior incubation of the enzyme at pH

Table 7.9 Non specific activity of glucose -6 – pase in the human fetal liver

	Mg pi/g/ tissue/15 min	
	(according to the	Mg pi/gm/tissue/15 min
Gestation	method of	(after prior incubation
period (wks)	Swanson)	at Ph 5)
8	0.24	0
20	5.76	1.44

5.0 in an acetate buffer and then the enzyme activity was determined. An example may perhaps, be cited in one of the cases where enzyme activity was found to be present by the original method of Swanson but it was shown to be absent when the nonspecific phosphatase activity was subtracted. Table 7.9 shows the result. In an 8 week old human fetus some measurable activity was present when it was estimated according to the original method of Swanson. But prior incubation at pH 5.0 inhibited the not specific phosphatase and no G-6-Pase activity was found in the liver at this period of gestation.

The method of glucose-6-phosphatase activity is based on liberation of inorganic phosphorus from glucose-6-phosphate, according to the method of Swanson [14]. Results are shown in Table 7.10. In the fetus of 9–12 weeks no glucose 6-pase activity was detected. Both activities, per gm tissue and specific activities of the enzyme increased progressively.

Earliest evidence of glucose-6-pase activity was demonstrated at 14.5 weeks of gestation. From then on, the activity increased progressively, but even at 24 weeks of gestation the

Gestation period (wks)	No. of samples	Activity μMole pi/g T/min	Sp. activity µMole pi/g protein/min
9–12	14	Nil	Nil
13–16	8	0.248 ± 0.017	2.53±0.14
17–20	6	0.764 ± 0.49	5.22 ± 0.34
21–24	6	1.269 ± 0.08	6.488 ± 0.359
Adult	4	2.5	9.6

Table 7.10 Glucose – 6 – pase activity of activity of human fetal liver at different period of gestation

enzyme activity was lower than that found in adult livers.

Since glucose-6-pase activity was not found in livers of human fetuses in the earlier weeks of gestation, it was considered advisable to study if the placenta could serve the purpose of the liver in this respect. Accordingly the whole placenta was homogenized in a waring blender in the cold and aliquots of placental homogenate were centrifuged at $500 \times g$; the supernatant was used as the enzyme source. Glucose-6-pase was estimated by the method of Swanson and about 5 mg tissue equivalent was used in the assay. Results are shown in Table 7.11. In the placenta of earlier gestation period, the specific activity of glucose-6-pase was much higher than that of later gestation period.

Some properties of the fetal liver glucose-6pase activity was studied in the microsomal pellet $(105,000 \times \text{g pellet})$. The pellets were resuspended in cacodylate buffer (0.15 M, Ph 6.5) and enzyme activity was determined by the method of Segal and Washko [15]. Result are shown in Table 7.12. The specific activity of the microsomal pellet was about twice that of the 5,000 × g supernatant. The 105,000 × g supernatant had no activity.

The 105,000 × g pellet was used to measure the activity of the enzyme by the above method with and without the presence of certain compounds and results are shown in Tables 7.13. The end product measured was glucose by the method of Nelson and Somogyi [16]. The full system showed an activity of 6.1 μ Mol glucose liverated/ mg protein per min. Inorganic phosphate at 100 Mol concentration produced a complete inhibition. Similarly ATP at 0.2 Mol concentration produced a complete inhibition.

Table 7.11 Glucose -6 – pase activity of the human placenta at different period of gestation

Gestation (wks)	Fetal body Wt. (g)	Placental g-6-pase activity µMole pi/g protein/min
14.5	48.2	67.8
19	219	2.14
19	237	4.98

Table 7.12 Glucose – 6 – pase activity of the human fetal liver in different subcellar fraction

Gestation period (wks)	Fraction	Glucose – 6 – pase activity µMole pi/g protein/min
21	5,000 × g sup	4.44
_	105,000 × sup	Nil
-	105,000 × pellet	7.87

Table 7.13 Effect of inorganic phosphate and ATP on the activity of glucose -6 -pase of human fetal liver

Addition of	Concentration (µ M)	Sp. activity of G – 6 – pase µ M g/mg pro/min
Inorganic phosphate	25	6.1
_	50	2.0
_	100	0
ATP	0.20	0
_	0.50	0
-	1.00	0

Fructose 1,6: Diphosphatase in Human Fetal Liver

Fructose 1,6 diphosphatase is one of the key enzymes of gluconeogenesis and highest activity is present in gluconeogenetic tissues, e.g., liver [17]. The activity of the enzyme is subjected to modifiers and regulators [18].

Since one of the objectives is consideration of the factors producing gluconeogienesis, the activity of the enzyme was measured in the livers of human fetuses of different gestation periods. The method used was a modification of Lahiri et al. [19]. The activity was measured at pH 7.5 and pH 9.5; the results are shown in Table 7.14. The activity as measured at pH 7.5 was always less than the activity at pH 9.5 at all stages of gestation. The developing

-			
Gestation period (wks)	Number of sample	F – D pase activity at pH 7.5 and pH 9.5 μ Mole pi/ pro h	
13–16	4	0.20	0.38
Do	-	0.37	0.86
Do	-	0.57	0.60
Do	-	0.17	0.25
17-20	2	0.49	0.60
Do	-	0.33	0.43
21–24	2	0.17	0.20
Do	-	0.19	0.49

Table 7.14F - D pase activity of human fetal liver atdifferent periods of gestation of 10 K fraction

human fetus receives glucose continuously through the placental circulation *vide* Biswas et al. [20]. However, apart from the supply of glucose from the mother, demonstration of FD Phase activity in liver within 8–32 weeks of gestation evokes gluconeogenic possibilities, [21]. Participation of the gluconeogenic enzyme during development for either fetal or neonatal energy homeostasis has been documented in several mammals [22]. Nevertheless, a clear insight regarding the fetal gluconeogenic capacity in mammals, particularly in human, is yet to be elaborated.

Human fetal liver fructose 1,6 – Diphosphatase has been purified about 200-fold to homogeneity at pH 7.5. The purification procedure for the enzyme was applicable throughout human fetal gestation. The purified enzymes recorded an approximate Km of 1.603×10^{-5} M for FDP and 1,6-biphospahte at pH 7.5. The enzyme could hydrolyze sedoheptulose - 1,7 - biphosphatase also. The molecular weight of the native enzyme as determined by gel filtration was approximately 146,000. The enzyme was Mg²⁺ depended, which could be replaced by Mn²⁺. Ca²⁺ was absolutely inhibitory. The Monovalent cations had no effect except Li1+ which inhibited the enzyme activity at pH 7.5. Histidine, citrate and EDTA were found to be stimulatory at pH 7.5, with optimal stimulation at 1 mM, 2 mM and 0.04 mM respectively. Zn²⁺ exhibited both stimulation and inhibition of the human fetal liver FDPhase. However, optimum stimulatory concentration of histidine partially recovered the enzyme activity at pH 8.5 from Zn^{2+} inhibition. 5⁻ – AMP inhibited the enzyme activity at pH 7.5 and inhibition was temperature dependent. The inhibition of the neutral FDPhase activity by 5° – AMP was potentiated by fructose -2-6 – biphosphate. The fructose -2-6 – biphosphate inhibition of the enzyme in absence of 5'AMP was found both at pH 7.5 and pH 8.5 but not at pH 9.5.

Discussion

A number of studies have been carried out on blood sugar concentration of fetal and maternal blood in different species [9, 10, 23-25] and in man. Most of these studies has reported lower sugar concentration in fetal blood than in the maternal. This is logical since transport of sugar across the placental circulation has been thought to be facilitated by diffusion type involving a carrier [26, 27]: and the process does not involve expense of energy. The carrier mediated transport is through GAP junction. Subsequent to the binding of the ligand to the receptor the pore is widened to transport the ligand. Since it does not involve waste of energy, the transport must be down a concentration gradient. Therefore, in order that glucose be transported to fetal circulation the concentration in the maternal circulation must be higher than in the fetal. It is rational, therefore that most studies found this to be so. The greater the difference in the maternofetal concentration the higher the transport. There were few studies on the blood sugar concentration in the human uterine artery and umbilical vein simultaneously because of obvious ethelical consideration. Furthermore it is difficult to measure the uterine and umbilical blood flows in order to calculate the rate of transport of sugar across the placental barrier. Such studies have been carried out in lambs [28]. By means of antipyrine steadystate diffusion technique, simultaneous umbilical and uterine blood flows were measured. Fetal oxygen consumption was found to be relatively constant among species whereas the metabolic rates were found to depend on the body size [29]. Glucose/oxygen quotients across the umbilical circulation during the last third gestation period in lambs was found to be 0.5 which was interpretated to mean that only 50 % of oxygen consumption in fetal lamb could be accounted for by umbilical glucose uptake.

Blood sugar concentration in the human fetus is different in different parts of the fetal circulation [30]. Thus, in the umbilical artery the fetal sugar concentration is almost always lower than the maternal sugar. However, in blood drawn from the heart the concentration is occasionally higher; this may be due to a number of reasons. Blood flowing through the liver may acquire some sugar by glycogenolysis or gluconeogenesis. Furthermore, some amount of delay was inevitable between hysterotomy and drawing of fetal blood. During this interval, the fetus is totally anoxemic, although the heart was still found to pump blood when the fetal thorax was opened. What effect the anoxemia has on glucose homeostasis it is difficult to say. On the one hand, tissues will try to derive energy by increased glycolysis. On the other hand, the lactic acid may be utilized to produce glucose by gluconeogenesis. The ultimate effect on blood sugar concentration may be provided by the relative rate of glycogenolysis and gluconeogenesis.

The percentage of fetal blood sugar higher than the mother, was more in diabetic state either overt or gestational. In maternal hyperglycemic condition, excess glucose in transferred to the fetus which develops hyperinsulinemia [31]. Elevated insulin level in the fetal plasma may lead to increased glycogen synthetase activity [32]. However the response is slow and not immediate. Some fetuses may not have developed the insulin receptor fully and fetuses may vary in their time sequence of development of the insulin receptors. Maternal hyperglycemia is more amenable to be regulated by insulin response than fetal. Thus in potentially diabetic mothers, more instances of higher fetal glycemia than maternal may be found. Glucose homeostasis in fetuses of diabetic mothers is therefore. more unstable than in nondiabetic mothers. Diabetes is associated with increased production of ketone bodies.

If the ketone bodies pass to the fetal circulation they may be preferentially utilized by the fetus so that the glucose remains underutilized. The underutilization of glucose especially if associated with some gluconeogenesis may explain the higher glucose concentration in some fetuses than in their mothers.

Most of these studies are one point studies, i.e., blood sugar concentrations have been determined once only, maternal sugar, either when she was on an empty stomach or during the operation of hysterotomy, and the fetal, during the dissection. Such studies leave much to be desired. Although we can imagine organismal homeostasis to be operative, the homeostasis would take place over a time period and not instantaneously especially in the mother during stress or in the developing fetuses. In the meanwhile a number of environmental alterations are occurring which may have an effect on glucose homeostasis. One of these alterations is the operation of hysterotomy and attendant infusions of saline or other fluids during the operation. Ordinarily, an infusion of either saline or dextrose at around 2 ml per min does not produce an alteration of blood sugar concentration in a normal adult person. But the conditions may not be called normal in pregnant mothers undergoing the operation of hysterotomy. The relationship between fetal and maternal blood sugar in mothers being infused with either saline or 5 % dextrose was more or less the same. We may, perhaps, imagine that maternal homeostasis under these circumstances led to no relative alteration in the maternal sugar concentration to have corresponding effect on the sugar concentration in the fetus.

Serum inorganic phosphate concentration of mothers on an empty stomach and during the operation of hysterotomy were found to be more or less the same, although the blood sugar concentrations during the operation was higher than the concentrations on an empty stomach. A rise of blood sugar after a glucose tolerance test is attended with fall of serum inorganic phosphorus. It is attributed to the entry of sugar into the cells, consequent to the stimulation of insulin release. The sugar would form the different phosphorylated compounds thus utilizing the serum inorganic phosphorus. The rise of blood sugar during the operation of hysterotomy is perhaps produced by glycogenolysis. The utilization of phosphorus during the phosphorolysis is perhaps

compensated by the glucose-6-phosphatase activity. The net result under these circumstances would be no change in the serum inorganic phosphorus. Furthermore, the increased blood sugar level during the operation of hysterotomy may not be attended with increased utilization of sugar in the cells because of the stress of anesthesia and the operation. The serum inorganic phosphorus would, therefore, remain unchanged in the mother. Fetal serum inorganic phosphorus was almost always higher than the maternal. The transport of inorganic phosphorous should be receptor mediated or coupled to some other metabolic transport, from the maternal to the fetal side: as otherwise a reverse transport should occur since the fetal serum inorganic phosphorus is at a higher concentration than the maternal. Utilization of inorganic phosphorus is mainly in the formation of the various organic phosphates including the high energy phosphates and in the deposition of inorganic phosphates in the skeletal tissues. We can assume both these processes to be very active in the growing fetus. So, other process remaining unaltered, the serum inorganic phosphorus in the fetus should be lower than the maternal. An adult organism gets rid of the excess inorganic phosphorus through renal excretion. Fetus is a closed system within the mother. It cannot excrete inorganic phosphorus into the outside environment. It only can utilize the placental circulation to get rid of the excess inorganic phosphorus. Perhaps, this process is not very efficient and there is an accumulation of serum inorganic phosphorus in the fetus. The serum inorganic phosphorus is also high in the newborn and young children. In them also, renal excretion may not have matured enough to excrete excess inorganic phosphorus. Liver glycogen content of human fetuses were studied by Szendi [33], Villee [23, 34], Gennser et al. [35] and Capkova and Jirasek [36]. Our results vary somewhat from the previously published results. Capkova and Jirasek found about 0.34 % of glycogen at 9 weeks gestation, whereas we found a much lower amount (less than 0.1 %) at this period. The reason is unclear the mother, in our case, came from a lower socioeconomic status but did not differ very much in nutritional status from other mothers in respect to body weight,

height and skin fold thickness. Furthermore, the studies of Capkova and Jirasek stressed on two spurts in respect of glycogen contents at around 14 and 17 weeks. We however, found liver glycogen content to rise throughout the period of gestation upto 22 weeks. Our results are in agreement with those of Villee [23] who also found a proportionate increase in liver glycogen content from 9 to 22 weeks gestation. In contrast to the liver, the muscles maintained a more or less steady concentration of glycogen, except during the early gesta-Capkova and Jirasek tion period. [36] differentiated three types of glycogen in human fetal organs. In one type, glycogen content is high at first and decreases with progress of gestation. In the second type, glycogen content increases with gestation and in the third type, the glycogen content does not show any change during the course of gestation. Heart muscle and kidneys belonged to the third type. We found that skeletal muscle glycogen belongs to the third type. The fetal liver glycogen belongs to the second type, i.e., it increases with increase of gestation. Under steady concentration of glucose as it is obtained in the human fetus, why should the liver and muscle behave in a different way with respect to glycogen content? The liver is an active metabolic organ even in the fetus, i.e., it systhesizes compounds which are necessary for fetal homeostasis; the fetal muscle, on the other hand, is more concerned with its own development rather than in the development of any other organ. The glycogen content is perhaps a reflection of the different fundamental roles played by the two organs in fetal life.

Normal glycogen possessing the usual branched structure in the presence of iodine has maximum absorption at higher wave lengths (460 mu) which shifts to lower values in limit dextrin (390 mu). Furthermore, amylopectin in the presence of iodine has maximum absorption at around 520 mu. The thrice precipated human fetal liver glycogen in the presence of iodine had higher absorption at 460 mu than at 520 mu; this shows that the liver glycogen had a branched structure and is not like that of amylopectin. Thus the glycogen synthesized in the fetal liver must have the branching enzyme creating a large num-

ber of nonreducing ends which are the sites of action of both glycogen phosphorylase and systhetase. In this way both synthesis and breakdown of glycogen is enhanced.

Fetal liver glycogen was found to be unusually stable. Whereas adult liver glycogen was degraded rapidly at room temperature, fetal liver glycogen was degraded less than 8 % only at room temperature for 24 h. Perhaps lysosomal activation in fetal liver was not profound as in adult liver. Lysomal enzymes were found to be present in fetal liver but the enzyme activity was much lower than in adult liver. The glycogen content of the fetal liver can, therefore, be considered representative of actual in vivo glycogen content.

There was no correlation between fetal blood sugar and fetal liver glycogen content. One of the ways in which glucose homeostasis is obtained in the adult is by glycogenesis. Thus the rise of blood sugar following a meal stimulates insulin release from the pancreas leading to increased glucose penetration in the muscle and stimulation of glycogen synthesis both in the muscles and the liver. In the fetus, however the glucose concentration is most probably uniformly held constant: in other words there is no fluctuation in the concentration. So presumably there is no equivalent post absorptive hyperglycemia and no stimulation of insulin release. Thus glycogen synthesis in the fetus is not regulated by sugar concentration. Rapidly dividing cells, e.g., in fetal life, may be associated with low cyclic AMP, which in turn may favour the co- valent modification of glycogen synthetase, which synthesizes glycogen at a rate independent of blood sugar concentration and at its own characteristic rate.

In adult the serum Pi is regulated by the excretion of phosphorus in the urine. The absorption is regulated by 1:25 dihydroxy cholecalciferol and the excretion by parathyroid hormone. In the human fetus the excretion of phosphorus through the renal tubules cannot be regulated. The supply of phosphorus is not discontinous as in adults but occurs uniformly through the placental circulation. Under these modified circumstances the serum Pi content of the human fetus from 9 to 20 weeks of gestation is about three times that of the adult. One of the reasons must be that the human fetus passes very little urine and therefore, cannot excrete much phosphorus. The only avenue left is backflow through the placental circulation. The backflow is probably prevented by the gap junctions possessing vectorial transport, i.e., from mother to fetus only and not in the reverse direction. The serum Pi is related to sugar metabolism in various ways. Thus the element is necessary for formation of the various hexose phosphates and other intermediates inside the cell, once glucose enters it. If there is significant glycogenolysis, Pi is utilized to produce glucose-1-p. There are various other uses of Pi. One of the main uses is deposition of calcium phosphate in the developing and growing skeletal system as hydroxyl apatite, the chief salt being tricalcium phosphate. The maintenance of high fetal serum Pi indicates vectorial properties of Pi transport across the placenta. The relative independence of the level of serum Pi on the face of the widely different blood sugar values in the fetus indicates that the two levels are independently regulated.

Formation of glucose from glycogen or lactic acid or glucogenic amino acids involves the ultimate enzyme G-6-Pase.In connection with gluconeogenesis in human fetuses, therefore, the enzyme has received considerable attention. Incubation of human fetal liver from therapeutic abortion with labeled substrates showed that the liver can produce glucose from pyruvate and glycerol in vitro [34]. The low G-6-Pase found in our series may be commensurate with high G-6-P dehydrogenase in the hexose monophosphate shunt pathway. Thus more of the pentose phosphates may be produced and the ribonucleotide diphosphate reduced to the deoxyribonucleotide diphosphate by the NADPH generated in shunt pathway. Thus the tissues with high mitotic rates may be assured of a supply of the required intermediates by the operation started by the low G-6-Pase pathway. Furthermore, the presence of the G-6-Pase in vitro may not necessarily be synonymous with operation of gluconeogenesis as the translocase system for the G-6-Pase may be deficient in the human fetus at least in the early period of gestation. The enzyme was inactivated by both ATP and Pi in accordance with the usual properties of the enzyme.

Studies on the activity of the human fetal liver fructose 1:6 phosphates have been very limited. The gluconeogenic enzyme, of which this enzyme forms one of the key enzymes, has been studied in human, sheep, guineapig and baboon fetuses [24, 37]. A detailed study of these enzymes in the human fetus obviously could not be undertaken: most of such studies were carried out on subhuman primates [22, 38] and these studies furthermore, were confined to estimation of the enzyme activity in cell extracts from liver or kidney.

Glycogen content of the liver and muscles are broken down to glucose-1-phosphate by the enzyme glycogen phosphorylase. Fetuses of different species have different behaviours with regard to synthesis and breakdown of glycogen. They also behave differently with regard to phosphorylase activity. Thus in rats the activity of the enzyme begins to rise abruptly from about 3 days prior to birth; the rate of increase continues for another couple of days postpartum and then begins its decline to reach the adult activity [39]. The equilibrium constant of phosphorylase activity is close to 1. Under normal conditions prevailing in vivo, the product of glycogen and inorganic phosphate concentration far exceeds that of glucose-1-p, so that in vivo the enzyme exhibits Glycogenolysis only. The inorganic phosphate concentration in human fetuses is almost three times that of the adult. Hence, the enzyme activity is far in the direction of breakdown than synthesis.

The enzyme activity in human fetal liver from 13 to 20 weeks of gestation was more or less unchanged. We did not do estimation of active and inactive phosphorylase activities of fetal liver as was done in the case of rat [40]. We cannot say how much of the phosphorylase activities we measured were due to inactive or active enzymes, although it is possible that during manipulation after the fetuses were taken out, much of the otherwise inactive enzymes would have been converted to the active one [11].

Considering the very small amount of phosphorylase activities present in the human fetal liver upto 20 weeks of gestation, it may be concluded that the derivative activity is very limited in human fetal liver upto this age.

Summary and Conclusion

- 1. The relationship between human fetal and maternal blood sugar concentrations was studied in 25 normal fetuses of different gestation period and their corresponding mothers. In 80 % of these pairs the sugar content from fetal heart blood was lower that the maternal but, in 20 % it was higher than the maternal.
- 2. In the case of those fetuses whose blood sugar was higher than the maternal about half the mothers were pre-diabetic.
- 3. There was no significant relationship between the infusions the mothers were receiving at the time of operation and the blood sugar content of the fetuses.
- 4. The serum inorganic phosphorus contents of the fetal blood were higher (more than double) than that of the mothers. But these had no relationship with the relative glycemic condition of the fetomaternal pairs.
- 5. The glycogen contents of the fetal livers were analyzed. At 9–12 weeks the glycogen contents were very low; thereafter the values increased gradually form less than 0.1 g% at 9–12 weeks to around 1.8 g% at 21–24 weeks.
- 6. The fetal muscle glycogen contents were very low at 9–12 weeks and thereafter stayed at around 1 %.
- The fetal liver glycogen had normal structure as judged by the iodine color and was unusually stable even at room temperature upto 24 h.
- There was no relationship between fetal liver glycogen content and relative glycemic condition of the fetuses in the fetomaternal pairs.
- The inorganic phosphorus contents of the liver were between 57 and 66 mg per 100 g tissue at 9–20 weeks and 114 mg per cent at 21–24 weeks.
- 10. Glucose-6-pase activity of human fetal liver was undetectable below 12 weeks of gestation. It was demonstrable thereafter and increased progressively but even at 24 weeks the activity was frequently less than that of the adult activity. Some experiments were undertaken to standardize the measurement

of activity of human fetal livers and the buffer systems, substrate concentration, homogenate aliquot inactivation at acid PH etc., were standardized.

- 11. The human placenta was found to possess Glucose 6-Pase activity at a higher level in earlier gestation period than in later.
- 12. The activity of fructose 1:6 biphosphatase in human fetal liver of different gestation periods was studied. The activity at pH 7.5 was found to be less than activity at pH 9.5.Measurable activity of the enzyme was detected from 11 weeks of gestation onwards. Some properties of the human fetal liver enzyme was studied like cytosolic location, heat treatment, and inhibition and activation by various organic compounds. Some attempts at purification of the enzyme were also undertaken.
- 13. The phosphorylase activity of human fetal livers of different gestation periods was studied. The enzyme activity throughout the gestation period was found to be very low.
- 14. The phosphorylase activity of human fetal muscles of different gestation periods was determined and compared to that of adult human muscles. The activity of the fetal muscles was found to be much lower than that of the adult muscles. In adult muscles the phosphorylase activity was higher than that of the liver but, in fetal muscles the reverse was found to be true.
- 15. These investigations tended to show that components are available in human fetal organs for gluconeogenesis but, it is doubtful if the process plays any significant role in the fetal system in vivo.

References

- 1. Bernard C. Les Phenomenes de la vie Paris. 1878. Paris, Bailliere. p. 113, 121.
- Robin Charles P. Treatise of anatomical and physiological chemistry, normal and patholigic. Publisher, unknown; 1853. p. 14.
- 3. PFluger E. The teleologic mechanism of living nature. Pflugers Arch. 1877;15:57–103.

- Fredericq L. The influence of the environment on the composition of blood of aquatic animals. Arch Zool Exper Gen. 1885;3:34–8.
- Richet C. Functions of defense. Dictinnaire Physioligie. 1900;4:121.
- Canon WB. Organization & physiological homeostatis. Physiol Rev. 1929;1:399–431.
- Campos FA, Cannon WB, Lundin, Walker TT. Some conditions affecting the capacity for prolonged muscular work. Am J physiol. 1929;87:680.
- Hansen KM. Oscillations in blood sugar of fasting normal persons. Acta Med Scand 4 (Suppl. 1):27–58, 1923.
- Shelley HJ, Neligan GA. Neonatal hypoglycemia. Br Med Bull. 1966;22:37.
- Shelley HJ, Bassett JM, Milner RDG. Control of carbohydrate metabolism in the fetus and the newborn. Br Med Bull. 1975;31:37.
- Stalmans W, Dewulf H, Hue L, Hers HG. The sequential inactivation of glycogen in liver after administration of glucose to mice and rats. Eur J Biochem. 1974;41:127.
- Hers HG. Glycogen Storage disease. In: Levine R, Luft R editors. Advances in metabolic disorders, Academic Press, New York, USA. vol 1. 1975. p. 36.
- Hers HG. Glycogen Storage disease. In: Levine R, Luft R editors. Advances in metabolic disorders, Academic Press, New York, USA. vol 1. 1964. p. 36.
- Swanson MA. Phosphatases in liver. I. glucose-6phosphatase. J Biol Chem. 1950;184:647–59.
- Washko ME, Segal HL. Studies of liver G-6-pase, solubilization and properties of the enzyme form normal and diabetic rats. J Biochem. 1959;234:1937.
- 16. Somogyi M. J Biochem. 1952;195:19.
- Clark MG, Lardy HA. Regulation of carbohydrate metabolism. In: Whelan WJ, editor, MTP international review of science. Biochemistry. Series one: Biochemistry of carbohydrates. vol 5, Butterworth, London, 1975 p. 239–41.
- Pontremoli S. Structure and function of liver fructose – I, 6-Diphosphatase. Biochem J. 1972;130:1p.
- Majumder AL, Eisenberg Jr F. Proc Natl Acad Sci (USA). 1977;74:3222.
- Biswas R, Majumder AL. Fructose-1, 6-bisphosphatase I. Purification, properties and proteolytic modification of the fish liver enzyme. Ind. J. Biochem. Biophys. 1985;22:293–9. http://bic.boseinst.ernet.in/ dpb/ALMFPUB.htm
- Biswas T, Majumder AL, Thakurta GG, Mukherjee KL. J Biosci. 1985;4:167–73.
- 22. Sheerwood WG, Robinson BH, Mayes S, Freire E, Oei J, Dibattista D. Biol Neonate. 1980;37:67.
- 23. Vilee CA. The metabolism of human placenta in vitro. J Biol Chem. 1953;205:113–23.
- Ballard FJ, Oliver IT. Carbohydrate metabolism in liver from fetal & neonatal sheep. Bio Chem J. 1965; 95:191–200.
- Girard JR, Ferre P, Gilbert M, Kervran A, Assan R, Marliss EB. Fetal metabolic response to meternal fasting sugar in the rat. Am J Physiol. 1977;232:456–63.

- Widdas WF. The inability of diffusion to account for placental glucose transfer & consideration of the kinetics of a possible carrier transfer. J Physiol (Lond). 1952;118:23–9.
- Chinard FP, Danesino V, Hartmann WL, Hugget AS, Paul W, Reynolds SRM. The transmission of hexoses across the placenta in the human and the rhesus monkey (Macaca mulatta). J Physiol. 1956;132(2): 289–303. http://www.ncbi.nlm.nih.gov/pmc/articles/ PMC1363496/
- Battagalia FC, Meschia G. Foetal & placental metabolism: their interrelationship and impact upon maternal metabolism. Proc Nutr Soc. 1981;40:99–113.
- 29. Battagalia FC, Meschia G. Principal substrates of fetal metabolism. Physiol Rev. 1978;58:499–527.
- Ghosh S, Guhathakurta G, Mukherjee KL. Glucose homeostasis in human fetuses. Ind. J. Pediatr. 1986;53:281–9. http://www.suklaghosh.com/pdfs/ glucose_homeostasis.pdf
- 31. Adam PAJ, Kalhanm SC, Savin SM. Fuel metabolism in the infant of the diabetic mother: Attenuated mobilization of alternate fuels. Symposium on diabetes and other endocrine disorders during pregnancy and in the new born. Natl. Foundation March of Dimes, New York, A.R. Liss, Inc., publ; 1976, 51–67. http:// geology.case.edu/documents/SavinCV_February-2011_001.pdf

- Sparks JW, Lynch A, Chez RA, Glinsman WH. Glycogen regulation in the isolated perfused near term monkey liver. Ped Res. 1976;10:51–6.
- Szendi B. Intrauterine function der lunge and leber des fetus Arch. Gynak. 1936;162:27–41.
- Villee CA. The intermediary metabolism o human fetal tissues. Cold Spr Harb Symp Quant Biol. 1954; 19:186–99.
- Gennser G, Landquist I, Nilson E. Glycogenolytic Activity in the Liver of the Human Foetus. Biol Neonate. 1971;19:1–23.
- Capkova A, Jirasek JE. Glycogen reserves in organs of human foetuses in the first half of pregnancy. Biol Neonate. 1968;13:129–42.
- 37. Levitsky LL, Paton JB, Fisher DE, Delannoy CW. Blood levels of gluconeogenic precursors & renal gluconeogenesis in the fasting baboon infant. Ped Res. 1976;10:412.
- Robinson BH, Sherwood WG, Mayes S, Freire E, Oei J, Doe Battista D. Neonate. 1980;37:60.
- Burch HB, Lowry, Kuhlmans AM, Skerjance J, Diamont EJ, Lowry SH, Van Dippe J. Changes in patterns of carbohydrate metabolism in the developing rat liver. J Biol Chem. 1963;234:2267.
- Watts C, Gain. Glycogen metabolism in the liver of the developing rats. Biochem J. 1976;160:263–70.

Mucopolysaccharides, Water and Electrolytes of Human Fetal Organs

8

Chameli Ganguly[†], Gitanjaly Guha Thakurata[†], K.L. Mukherjee[†], and Niranjan Bhattacharya

Introduction

Structure of Mucopolysaccharide

A group of long chin heteroglycans or heteropolysaccharides consisting of N-acetylhexosamine, hexuronic acid or hexose as repeating units, occurring in animal tissues, is termed "Mucopolysaccharide" (MPS). Some of these polysaccharides are also polyelectrolytes, behaving as polyanionic substances due to the presence of free carboxylic (--COO⁻) and sulphate (--SO₃⁼) groups. Such acidic

C. Ganguly, MSc, PhD Former Biochemist Central Calcutta Society for Advancement of Human Development and Research, Kolkata 700040, India

G.G. Thakurata, MSc, PhD (Cal) Formerly, Department of Biochemistry, National Medical College and Hospital, Kolkata 700040, India

K.L. Mukherjee, MB, PhD (Cal), PhD (Wisconsin) Former Head of the Department of Biochemistry, Institute of Post Graduate Medical Education and Research, Kolkata, West Bengal 700015, India

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India e-mail: sanjuktaniranjan@gmail.com mucopolysaccharides (AMPS) mainly occur in connective tissues and universally contain acetylated hexosamine. Jeanloz [1] termed these MPS as Glycosaminoglycans (GAG) (Fig. 8.1).

The elucidation of the basic structure and some characteristic properties of at least seven different nitrogen containing heteropolysaccharide-protein complexes in the ground substance of connective tissue was established (Fig. 8.2).

Mucoproteins are in Stacey's terminology, carbohydrate protein complexes with a relatively high protein or peptide content. Ovomucoid, serum mucoprotein, pituitary hormones, and submaxillary gland mucin belong to this group.

The high proportion of carbohydrate (80 % of the total wt. in blood group substance) had led some authors [2, 3] to classify these substances together with hyaluronateprotein complex and Chondroitin Sulphate (CS)-protein complex, as MPS. The macromolecular structure of this protein carbohydrate complex has been studied by various groups of workers, e.g., Tsiganos and Muir [4] and Robinson [5] etc., and termed as proteoglycan.

Buddecke et al. [6] and Rosenberg et al. [7] and many other workers observed that the proteoglycan molecule is polydisperse in nature.

Tsiganos and Muir [4] have reported that proteoglycans may be aggregated with one another by means of glycoprotein linkage (Figs. 8.3 and 8.4).

A tiny part of a proteoglycan aggregate form cartilage matrix. The "core" proteins of each proteoglycan are attached to the hyaluronic acid backbone at intervals of 30–60 sugar dimmers (uronic acid plus amino

© Springer International Publishing Switzerland 2016

[†]Authors was deceased at the time of publication.

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_8





Fig. 8.2 Structure of mucopolysaccharides



structure



sugar) along the backbone. Linking proteins are found at the attachment sites. The GAG molecules (keratin sulfate and chondroitin sulfate) are attached to the core proteins; a given core protein molecule (molecular weight up to 2×10^5) may have 100 sidechains of chondroitin sulfate (molecular weight up to 2 X 104) and 30 to 60 sidechains of keratin sulfate (molecular weight up to 4 or 8 X 10³), and so will be much larger and more complex than the drawing indicates. Since a hyaluronic acid backbone may vary between 4 and 0.4 microns in length, since the length of core proteins is highly variable, and since the length of GAG molecules on the core proteins is also variable and large, the total volume occupied by an aggregate is very great. And, of course, its potential variability in detailed structure is enormous [7, 8].

MPS or GAG is classified as neutral and acidic form. Neutral MPS contains no carboxglic (--COO⁻) or Sulfate (--SO₃⁼). These are present in blood group





Fig. 8.6 Structure of connective tissue

substance. Acid MPS (AMPS) are nonsulfated and sulfated. Example of nonsulfated AMPS is hyaluronic Acid (Hy). and chondroitin, main sulphated AMPS are chondrotin sulfates (C-S-,A,B,C,D,E). Heparin and Heparan Sulphate (Fig. 8.5).

Matrix

Acidmucopolysaccaride (AMPS) in different connective tissues present mostly in intersticial space, which is known as Matrix (Fig. 8.6). The matrix consists of ground substance, fibres and impregnating materials. The Matrix contains AMPS, inorganic ions, water, organic compound in transit from the blood to the cell and from the cell to the blood and some macromolecules which have escaped from the Arteriolar end of capillaries. All the compounds and the ions are in some sort of association with the AMPS either directly or indirectly. Thus cations formed electrostatic bond with the polyanaion as also the water as Hydronium ions (H_3 O⁺). The salts which are transported consist of the number of cations and anions. The other compound for e.g., Nutrients, intermediate compounds, waste products and probably the salts too are held in interstices of the gel formed by the union of the hydronium ion with the anion of the AMPS.

It may be assumed that a large quantity of extra cellular fluid is maintained by a large amount of polyelectrolyte, i.e., the AMPS molecule. These AMPS secondarily enhance the transport of cations which are indispensible for cellular activities mainly for synthetic process.

Function of Mucopolysaccharide

The presence of free carboxylic ($-Coo^{-}$) and sulpahte ($-SO_3^{=}$) group on each repeating unit of AMPS molecule confers a high negative (-) charge on it; such a highly charged polyelectrolyte, linear in configuration, interacts with water (H_3O^+) and cations (+), such as Na⁺, Ca⁺⁺etc. This property of interaction between oppositely charged ions in AMPS is responsible for some of their biological functions. It seems likely that specific functions result from both the quantitative localization of AMPS as well as the specific chemical and physicochemical properties of the individual substances. It is clear from Grossman's finding that even very closely related AMPS may exhibit markedly different properties. The effect

Fig. 8.5 Classification of

mucopolysaccharide

of various hormones (relaxin, cortisone, testosterone) on certain tissues causes polymerization of the ground substance. Depolymerization of mucoprotein decreases the binding capacity of AMPS for cations. The binding of metals by monomers is less strong than by polymerized acids.

The AMPS fulfill diverse functions in the animal body. According to Dorfman et al. [9] the functions are as follows:

- 1. Supporting medium of any organ.
- 2. Water metabolism in connective tissue.
- 3. Control of electrolyte in connective tissue.
- 4. Organization of inorganic crystal or calcification.
- Anticoagulant function of mucopolysaccharides.
- 6. Defensive function of mucopolysaccharides.
- 7. Wound healing.
- 8. Lubrication function.
- 9. Cleaning activity.
- 10. Maintance of stable transparent medium of the eye.
- 11. Interaction of MPS with collagen drug metabolite.

Among the above functions of AMPS we are interested in only water and electrolyte metabolism in fetuses and make correlation with AMPS concentration.

Total body water in an adult person constitutes 60 % of body weight; the water is distributed in the following way: 30-40 % is intracellular and 20-25 % is extracellular. In a new born child, however, the total body water is 78 % of body weight, 30-40 % is intracellular and nearly the same amount is extracellular. In fetuses around 6 months, the total body water is about 82 % of body weight; the intracellular water is again around 30-40 % and the extracellular water is either the same as or even more than the intracellular. Fetuses about 10 weeks of age contain 85 % of water where extracellular water is perhaps more than the intracellular. Extracellular water, therefore is highest in early fetal life and decreases as it grows older. The decrease of extracellular water relative to intracellular continues after birth and is most evident in the first few months of life. The tissues in the early periods of gestation are in a highly anabolic state, which of course, require a large amount of nutrient substances, which are supplied continuously through the blood stream. The excess extracellular water probably allows solution of large amount of such nutrients and enables the tissue cells to be supplied with the required compounds from the blood stream through the connective tissue spaces.

The negative charges of AMPS maintain an accumulation of water (H_3O^+) in the interstitial tissue space. Hyaluronate as a hydrated gel has an important function to bind water in the skin. As ageing progresses, there is a relative decrease of Hy and an increase of CS in skin [10]. According to Dorfman and Schiller [11], it is not easy to explain why tissue hydration is specifically connected only with Hy but not with sulphated MPS. Both fibroblast cells and mast cells play an important role in the homeostatic mechanism maintaining water in intercellular space.

The structural organization of water in the protein has its role in biological functions such as ionic cell selectivity [12]. Tropocollagen molecule holds much of CT water [13].

Electrolytes Control in the Connective Tissue

The ground substance of the CT contains all the transit compounds other than water between parenchymal cells and circulation. Among the transit compounds diffusible cations (Na⁺, K⁺, Ca⁺⁺, Mg⁺⁺) are mostly related to one or other of the ground substance colloid compounds, the negatively charged immobile aggregate of AMPS-protein complexes.

In tissues, the most common cations of extracelluar space are the sodium ions so that the polysaccharides occur in tissues almost entirely as sodium salts. K⁺, Ca⁺⁺ and Mg⁺⁺ ions are also associated with polysaccharides.

The object of our work:

- To determine the total water and extracellular water of the organs and electrolyte (Na⁺,K⁺) content in tissues and Serum at different gestation period.
- To estimate the total AMPS content of certain human fetal organs in different gestation periods.

- 3. To fractionate the AMPS into the three usual fractions, i.e., hyaluronate, chondrotin sulphates and heparin in these organs.
- 4. To undertake some studies on the composition of these classes of AMPS form the organs.
- To determine the major electrolyte concentrations of the organs and to correlate the quantity and quality of the AMPS to the water and the major electrolyte concentrations in these organs.
- To measure the activity of the enzyme glucosamine-6-phosphate synthetase (Gm-6-p is the major precursor of AMPS) in different gestation periods.

Methods of Investigation

Fetuses were obtained by hysterotomy after obtaining required consent of mothers (from 1977) admitted for Medical Termination of Pregnancy (MTP). The fetuses were dissected at once in most of the cases. The organs were dissected in toto. After the determination of total weights of the organs aliquots were taken for respective measurements. The following organs were selected:- liver, lung and brain. The following estimations were carried out in aliquots of the above organs.

1. Total water and extracellular water and electrolyte content:

Total water content was determined by weighing about 300 mg. of tissue in a preweighed tube and keeping it at 105 °C for 18 h. The difference in weight was taken to be due to evaporation of water.

Extracellular water of tissues was determined from Na-thiocyanate space by the method of Eder [14]. Mothers were injected 10 ml. of Na-thiocyanate solution, 5.0 % (in sterile form) 4–6 h prior to operation. Nathiocyanate concentration of fetal serum and tissues were determined.

After the evaporation of water the tissue was ashed in a muffle furnace at 500 ± 25 °C for 4 h; after the furnace was cooled the tubes were taken out and when they came to room

temperature, they were reweighed. The difference was taken to represent the ash content. The ash was dissolved in 5 ml. of 6-N HCL. The solution was centrifuged. Sodium and potassium concentrations were determined in appropriately diluted supernatant by flame photometry.

2. Acid Mucopolysaccharides (AMPS) in fetal tissues:

Extraction of AMPS and fractionation were carried out according to Singh and Bachhawat [15], with slight modification. Bachhawat's method was based on the method of Dorfman [16] which was itself a modified method of Scott [17].

Aliquotes of the tissues were dried with acetone and made lipid free with 2:1 Chloroform and Methanol mixture. AMPS was extracted from the defatted dry tissues with 0.1M acetate buffer, pH 5.0. Extracted AMPS was made protein free by papain digestion.

Fractionation of AMPS

Extracted AMPS was fractionated by the method of Schiller et al. [16]. Extracted AMPS is precipitated with Cetyl Pyridinium Bromide (CPB). The CPB-AMPS complex was collected along with celite. From this complex AMPS were fractionated to fractions I, II and III by different NaCl Concentration as 0.4M, 1.2M and 2.1M respectively. AMPS was precipitated by Alcohol, centrifuged and dissolved in a small volume of water. It was ready for constitutional analysis.

Hexuronic acid content was determined by Dische's carbozole method [18] as modified by Bitter and Muir [19] and further slight modification by Singh [15]. The Uronic acid content was multiplied by 2.4 to get an approximate value of total AMPS content.

Hexosamine content was estimated from various AMPS fractions. The fractions were hydrolyzed with concentrated HCl (to make the final conc. 0.6 N) at 100 °C. in the sealed tubes for 4 h. The hydrolysates were dried in a vaccum desiccators. The dried samples were dissolved in a

small volume of water and made ready for Hexosamine (Hm.) analysis. Determination of Hm is based on the Elson-Morgan [20] and Morgan-Elson [21] methods for free Hm and N-Ac-Hm respectively. Judoseig and Benmanan [22] used a method to determine Galm and total Hm content by measuring the color intensity in cold and hot reactions.

Sulphate estimation from total AMPS was estimated by $BaCl_2$ – gelatin method of Dodgson and Price [23].

Glucosamine-6-p Synthetase (E.C. 2.6.1.16)

It is the primary reaction in the Biosynthesis of AMPS. Assay of this Enzyme was done by the method of Ghosh et al. [24]. Liver is the main source of this enzyme. Fetal liver homagenate was incubated with substrate G-6-P and Glutamine in K-Phosphate buffer (0.2M, pH7.6). Glm-6-p was estimated by the method of Morgon and Elson [21]. Enzyme activity was expressed as μ mole/mg tissue/h and Specific activity as μ mole/mg protein/h. Soluble protein was estimated by the method of Lowry et al. [25].

Result

For the sake of convenience the fetuses have been divided into six groups -A, B, C, D, E and F according to a difference in the gestation period of 4 weeks. Table 8.1 shows the grouping of the 115 fetuses investigated in the present series of these fetuses; 15 belonged to group A, 26 to B, 36 to C, 21 to D, 14 to E and 3 to F. Three organs, Liver, Lungs and Brain, were selected for the study of AMPS; water electrolytes and the Key enzyme of AMPS Synthesis Glucozamine-6-p synthetase was also studied in the above three organs.

Liver

AMPS Concentration in Human Fetal Liver Tissues

The dehydrated defatted tissues were used for the determination of AMPS. Table 8.2 shows the

Table 8.1 Grouping of fetuses according to body weight and gestational age (week)

Group	No. of fetuses	Week	Weight g
А	15	8-12	1.5-14.5
В	26	13–16	15.6–108
С	36	17–20	115-295
D	21	21–24	330-660
Е	14	25–28	715–1025
F	3	29–32	1,055–1,650

Table 8.2 Total mucopolysaccharieds of human fetal liver of different gestation period

Group	Age weeks	No. of fetuses	Total uronic acid mg/g. Dry defated tissue
А	8-12	-	ND
В	13–16	7	0.74 ± 0.16
С	17–20	10	0.65 ± 0.07
D	21–24	10	0.56 ± 0.07
Е	25–28	5	0.55 ± 0.09
F	29–32	-	ND
Postmortem	0–8 days	6	0.35 ± 0.02

total AMPS concentration in fetal tissues in different gestational periods. The result is expressed as Uronic acid per g of dry lipid free tissues. The uronic acid content of fetal liver varied from 0.5 to 0.7 mg/g, in postnatal life the amount was around 0.3 mg/g.

Mucopolysaccharides in human fetal liver and their fractions (Tables 8.3 and 8.4). In group A, the livers were too small to be analyzed. In group B, we could determine the total amount of UA in fraction 1. In groups B and C the total amount of fraction I was found to be the same; the hexosamine present was mostly glucosamine but the content of sulphate was almost equal to that of the UA. Thus this fraction contained the nonsulfated hyaluronic acid, some amount of heparitin sulfate and a small amount of CS. In groups D and E, total AMPS slightly decreased and this fraction also contained predominantly hyaluronic acid but considerable amount of chrondroitin and heparatin sulfate was found to be eluted in this fraction. In post mortem specimens of babies of 0-8 days ages, the total amount of fraction I was found to be much less than in the fetal specimens but the nature of the

				Uronic acid	Mg/g dry det	fatted tissue Fractions
Group	Age (wks.)	No. of cases	Total	Ι	II	III
А	8-12	0	ND	ND	ND	ND
В	13–16	3	0.74 ± 0.16	0.34 ± 0.06	0.25 ± 0.03	0.08 ± 0.04
С	17–20	12	0.65 ± 0.07	0.40 ± 0.02	0.21 ± 0.02	0.04 ± 0.02
D	21–24	10	0.56 ± 0.07	0.30 ± 0.05	0.24 ± 0.04	0.03 ± 0.02
Е	25–28	5	0.55 ± 0.09	0.38 ± 0.08	0.20 ± 0.07	0.07 ± 0.01
F	28–32	0	ND	ND	ND	ND
P.M	0–8 days	7	0.15	0.15	0.04	0.35 ± 0.02

 Table 8.3
 Mucopolysaccharides in human fetal liver and their fractions

Table 8.4 Composition of AMPS of fraction –I

Group	Age	No. of cases	Uronic acid mg/g dry defatted tissue	Hexosamine*	Glucosamine*	Galactosamine*	Sulfate*
А	8-12	0	ND	ND	ND	ND	ND
В	13–16	3	0.34 ± 0.06	ND	ND	ND	ND
С	17–20	12	0.40 ± 0.02	1.30	1.27	0.03	0.9
D	21–24	10	0.30 ± 0.05	1.25	1.02	0.23	1.0
Е	25–28	5	0.38 ± 0.08	0.90	0.75	0.15	0.8
F	29-32	-	ND	ND	ND	ND	ND
P.m	0-8 days	7	0.15	0.75	0.60	0.15	1.37

*Expressed as molar ratio of Uronic Acid

Table 8.5 Composition of AMPS of fraction -II

Group	Age	No. of cases	Uronic acid mg/g dry defatted tissue	Hexosamine*	Glucosamine*	Galactosamine*	Sulfate*
А	8-12	0	-	ND	ND	ND	ND
В	13–16	3	0.26 ± 0.08	ND	ND	ND	ND
С	17–20	12	0.21 ± 0.02	0.97	0.13	0.84	0.87
D	21-24	10	0.24 ± 0.04	1.20	0.13	1.07	1.5
Е	25-28	5	0.19 ± 0.07	0.86	0.38	0.48	0.98
F	29-32	-	-	ND	ND	ND	ND
P.m	0–8 days	7	0.16±0.02	0.70	0.20	0.50	0.98

*Expressed as molar ratio of Uronic Acid

material eluted in this fraction was observed to be a mixture of HY and CS and heparitin sulfate.

Table 8.5 shows that in groups A and B, the fraction II was insufficient in amount for determination of various constituent. In group C, the amount of fraction II as determined by the total hexuronic acid content was higher than in the livers of babies dying after 0–8 days of life. When the fraction was, however analyzed for the glucosamine and galactosamine contents, most of the aminosugar was found to be galactosamine rather than glucosamine, which

should be the preponderant aminosugar in this fraction.

In Table 8.6, Fraction III shows the content of heparin was low as compared to Fractions I and II but by and large the heparin content in fetal life was slightly greater in group B and E than in postmortem liver in postnatal life. This fraction also cannot be considered to consist of heparin only, as a considerable amount of galm was present although heparin contains Glm rather than Glam. The sulfate content was characteristic of heparin in groups D and E but low in group C and post-mortem.

			Uronic acid mg/g				
Group	Age	No. of cases	dry defatted tissue	Hexosamine*	Glucosamine*	Galactosamine*	Sulfate*
А	8-12	0	-	ND	ND	ND	ND
В	12–16	3	0.08 ± 0.04	ND	ND	ND	ND
С	16–20	9	0.04 ± 0.02	0.88	0.44	0.44	1.0
D	20–24	8	0.02 ± 0.02	0.90	0.77	0.13	3.0
Е	24–28	7	0.07 ± 0.02	0.71	0.68	0.03	3.1
F	28–32	-	-	ND	ND	ND	ND
P.m	0–8 days	6	0.04 ± 0.02	1.2	1.1	0.1	1.0

Table 8.6 Composition of AMPS of fraction –III

*Expressed as molar ratio of Uronic Acid

Table 8.7 Total water, ash, sodium and potassium contents of human fetal liver. (Values are expressed in wet tissue)

Group	Gest, period weeks	Water g%	Ash g%	K+ mEq/kg	Na⁺ mEq/kg
А	8-12(7)	80.7±1.2	1.62 ± 0.93	41±5	94±10
В	13-16(10)	79.4±0.5	1.90 ± 0.13	43±4	71±7
С	17-20(20)	78.5±0.3	1.63 ± 0.02	51±2	67±8
D	21-24(11)	78.3±0.6	1.68 ± 0.15	48±5	60 ± 10
E	25-28(8)	77.6±0.4	1.31 ± 0.18	44 ± 4	66 ± 10
	0-8 days(4)	78.2±1.3	1.57±0.19	44±3	62±3
	8 (adults)	69.0±1.6	1.42 ± 0.07	57±3	63±12

Water and Electrolyte Content of Human Fetal Liver (Including Different Compartments of the Tissue)

Water content of human fetal livers at different periods of gestation is shown in Table 8.7. It varied from 80.5 % of the organ weight at 8–12 weeks (Gr. A) to 77.6 % at 25–28 weeks (Gr. E). The decrease was progressive except in the last two groups where the water content was almost unchanged. We could obtain livers of four children who died from prematurity and other causes; the water content in their livers was also like that of a fetus at 21–28 weeks (Gr. F), namely around 78 %. The water content of adult livers obtained by surgical biopsy was around 69 %. Water in the body was compartmentalized mainly into two compartments, the extra and intracellular compartments.

We studied the water and electrolyte in different compartments of tissues. Many compounds as per e.g., inuline, sucrose, NaSCN, radioactive Na⁺, Cl, SO₄⁼ etc. have been used for the determination of extracellular space. We selected NaSCN to determine extra cellular space. We injected 10 ml sodium thiocyanate solution (5 %) to the

Table 8.8 Thiocyate concentrations in body fluids (mg/dl)

Maternal serum	(9)	4.7 ± 0.91
Fetal serum	(9)	2.7±0.21
Amniotic fluid	(3)	0.48, 0.22, 0.94
Fetal bladder fluid	(2)	0.81, 0.48
Fetal C.S.F.	(1)	0.44

mother 4 h before the fetus was taken out. The fetus was put as usual, on ice at the operation theater, and immediately brought to the laboratory where it was dissected. Fetal blood was obtained by cardiac puncture into a heparinized syringe and the blood was immediately centrifuged to get the plasma. Plasma thiocyanate concentration was measured by the method described by Eder [14]. The tissues were homogenized in water (1 ml per g T) and then the method was followed exactly as described for the plasma.

Table 8.8 shows the concentration of thiocyanate in maternal serum was higher than the concentration in fetal serum. The amiotic fluid had lower thiocyanate concentration than fetal serum; the fetal bladder fluid similarly had lower concentration as also did the cerebrospinal fluid. Since throcyanate was found to be permeable across the placenta and since we assumed that fetal circulation was greater than the adult and fetal urinary excretion slower than the adult, we presumed that thiocyanate had assumed equilibrium in the fetus by 4 h. We, therefore, took thiocyanate space concentration in fetal organs to represent extracellular space.

The extracellular space per 100 g of tissue was calculated. According to the following formula – Extracellular space = $x/y \times 100 \times 1.1$, where x = Thiocyanate concentration of tissue in mg% and y = thiocyanate, concentration of plasma in mg% and 1.1 is a constant for the Donnan Equilibrium factor. The total water per 100 g tissue was calculated by drying the tissues to constant weight at 105 °C (usually for 18 h). The intracellular water was taken to be the difference of total organ water and extracellular space. Results are shown

Table 8.9 Water content and distribution of human fetal liver

Body weight	Approx. gest. period	Total water g%	Extracellular g%	Intracellular g%
95	16	80	53	27
101	16	79	51	28
107	16	79	51	28
120	17	80	53	27
250	20	81	45	36
330	23	79	46	33
375	24	80	46	34
402	25	79	46	33
877	27	78	43	35

in Table 8.9. The thiocyanate space varied from 53 ml per 100 g of liver at 16 weeks of gestation to 43 ml at 27 weeks. As the total water did not vary much during the different gestation period, the corresponding intracellular water (i.e., total water – thiocyanate space) increased from 27 ml per 100 g at 16 weeks to 35 ml at 27 weeks.

The metabolism of water, electrolytes and MPS is closely interrelated. The matrix containing the polyanionic AMPS is supposed to attract the hydronium inos (H_3O^+) and the common cations like sodium (Na^+) and potassium (K^+) ions.

The more charged it will be, the more it attracts the cations which again, in their turn, will bring the anions like the chloride and bicarbonate around them. We therefore, determined the sodium and potassium concentrations in the ashed materials of different organs including the liver. Results on the liver are shown in Table 8.7. Potassium concentration in mEq/kg of body weight varied from 41 to 51; virtually there was no change at different periods of gestation. Sodium content, of the liver, however, in earlier periods of gestation, e.g., in group A and B was higher than in later periods.

Correlation of Water, Electrolytes and AMPS Concentration

Fetal tissues contain a greater quantity of water than adult tissues. This extra water is chiefly extracellular in location. The thiocyanate space in adult liver varied from 30 % to 35 % whereas in fetal liver it was from 43 % to 53 %.

We measured only the sodium and potassium contents of the fetal and maternal plasma; the

Table 8.10 Sodium and potassium concentrations in human fetal liver (in different compartments). Valued are expressed in wet tissue

Gr	CNS spray	Intracellular water	Plasma Na⁺	Plasma K⁺	Extracellular liver Na ⁺	Extracellular liver K ⁺	Intracellular liver Na ⁺	Intracellular liver K ⁺
_	Ml/kg	Ml/kg	mEq/L	mEq/L	mEq/kg	mEq/kg	mEq/kg	mEq/kg
С	520	275	140	11.6	79.1	6.6	0.0	44.4
D	465	320	138	10.1	70.6	5.2	0.4	42.8
Е	460	325	135	12.5	68.4	6.3	0.6	37.7
Adults	320	370	142	4.8	49.9	2.0	7.1	55.0

Intracellular Water = Total water - CNS space

Extracellular Na = Plasma Na X 1.1 (Donnan Eq) X CNS Space

Extracellular K = plasma K X 1.1 (Donnan Eq) X CNS Space

Intracellular Na = Total sodium - Extracellular sodium

Intracellular K = Total Potassium - Extracellular Potassium

		Total	Uronic acid mg/g dry defatted tissue fraction				
Gest. Pd	No. of cases		Ι	II	III		
8–12	0	ND	ND	ND	ND		
13–16	4	4.59 ± 0.49	3.26 ± 0.35	1.21±0.16	0.13±0.04		
17–20	7	2.94 ± 0.24	1.79±0.12	0.96 ± 0.07	0.19 ± 0.09		
21–24	3	2.42 ± 0.28	1.67 ± 0.47	0.65 ± 0.21	1.12±0.019		
25-28	5	2.21±0.11	1.26 ± 0.08	0.85±0.13	0.09 ± 0.02		
28–32	0	ND	ND	ND	ND		
0–8 days	6	0.70 ± 0.14	0.23 ± 0.06	0.39 ± 0.09	0.08 ± 0.02		

 Table 8.11
 Mucopolysaccharides of the developing human lung and their Fractionations

total cations (Na⁺, k⁺) in fetal plasma was 151.6 mEq/L in the serum and 146.8 mEq/L in the mother (Table 8.7). The difference was statistically significant (p<0.02). A greater difference which may have some role to play in the amount of the extracellular fluid in the tissues space in the fetal life lies in the hyaluronate and total AMPS contents of the fetal organs.

The thiocyanate concentration in fetal plasma was less than the thiocyanate concentration in maternal plasma. The elaboration of fetal urine is negligible and the thiocyanate concentration in bladder fluid was also much lower than in the plasma. We should have expected the thiocyanate concentration in fetal plasma to be higher than the maternal plasma if the compound was continually being filtered through the placenta. Perhaps, only the maternal thiocyanate which was bound to the plasma proteins actually filtered through the placenta and the free fetal thiocyanate might pass in the reverse direction. An equilibrium may probably be attained which ultimately resulted in fetal thiocyanate concentration being lower than the maternal.

Table 8.10 notes that the electrolyte concentration in the different compartments of the fetus was apparently quite distinct from those of an adult. Sodium contents of the extracellular fluid in fetal liver were greater than in adult liver. Since the total thiocyanate space was also bigger in the fetal liver, the extracellular space in fetal liver contained as much as twice the amount of sodium per kg of organ weight as in adult liver. The potassium concentrations as well as contents in the total thiocyanate space of the fetal livers were higher than those of the adult livers. Potassium concentration per litre of intracellular fluid was found to be 152 mEq in adult liver, whereas in fetal livers it progressively decreased form 161 mEq/L in fetuses of 17–20 weeks gestation to 116 mEq/L in 25–28 weeks of gestation.

Lung

Total AMPS of Lung and Its Fractionations

Total AMPS content of lung was estimated as total uronic acid content in the CPB precipitated material of the proteolyzed extract and expressed as mg per g of dry defatted tissue (Table 8.11). The lung is a soft tissue as compared with the liver. The process of extraction of AMPS from lung tissue was exactly the same as from the liver.

It is apparent from the Table 8.11 that the total AMPS content of the fetal lung was higher in fetuses of early gestation period than in those of the later periods. When compared to the liver (Table 8.11), the AMPS content of the lung was four to six times higher in fetuses of corresponding gestational age. With growth of the lung the amount of AMPS per g of tissue gradually and progressively decreased until at 29–32 weeks, the amount of AMPS was about a half of what was found in fetuses of 12–16 weeks. However, in all these ages the uronic acid content of the lung was about four to six times that of the liver.

Different Fractions of Total AMPS of Human Fetal Lung and Its Composition

Aliquots of each fraction were tested for uronic acid, total hexosamine, galactosamine, glucosamine and sulphate Results were expressed as molar ratio of uronic acid.

Group	Week age	No. of cases	Total Ua mg/g of dry defatted tissue	Hexosamine ^a	Glucosamine ^a	Galactosamine ^a	Sulphate ^a
А	8-12	-	ND	ND	ND	ND	ND
В	13–16	4	3.26 ± 0.04	ND	ND	ND	ND
С	17–20	7	1.79±0.12	0.94	0.90	0.04	0.17
D	21–24	3	1.66 ± 0.47	0.79	0.78	0.01	0.20
Е	25-28	5	1.26 ± 0.08	0.81	0.65	0.16	0.29
F	29-32	-	ND	ND	ND	ND	ND
P.M.	0–8 days	6	0.23 ± 0.66	1.2	0.90	0.13	0.55

Table 8.12 Composition of different fractions of AMPS. Composition of mucopolysaccharides of human fetal lung (fraction I)

ND Not Done

^aExpressed as molar ration of Uronic acid

		No. of	Total Ua mg/g of				
Group	Week age	cases	dry defatted tissue	Hexosamine ^a	Glucosamine ^a	Galactosamine ^a	Sulphate ^a
А	8-12	-	ND	ND	ND	ND	ND
В	13–16	4	1.22±0.16	ND	ND	ND	ND
С	17-20	7	0.96 ± 0.07	0.87	0.20	0.67	0.73
D	21–24	3	0.64 ± 0.21	0.82	0.20	0.62	0.87
E	25–28	5	0.86±0.13	0.73	0.17	0.56	0.86
F	29-32	-	ND	ND	ND	ND	ND
P.M.	0–8 days	6	0.39 ± 0.09	0.74	0.03	0.71	1.01

Table 8.13 Composition of mucopolysaccharides of human fetal lung (fraction II)

ND Not Done

^aExpressed as molar ration of Uronic acid

		No. Of	Total Ua mg/g of				
Group	Week age	cases	dry defatted tissue	Hexosamine ^a	Glucosamine ^a	Galactosamine ^a	Sulphatea
А	8-12	-	ND	ND	ND	ND	ND
В	13–16	4	0.13 ± 0.04	ND	ND	ND	ND
С	17-20	7	0.19 ± 0.09	0.67	ND	ND	0.65
D	21-24	3	0.12 ± 0.02	0.62	0.42	0.20	2.1
Е	25-28	5	0.09 ± 0.02	0.67	0.41	0.46	2.5
F	29-32	-	ND	ND	ND	ND	ND
P.M.	0–8 days	6	0.082 ± 0.02	1.1	1.0	0.1	0.8

 Table 8.14
 Composition of mucopolysaccharides of human fetal lung (fraction III)

ND Not Done

^aExpressed as molar ration of Uronic acid

Table 8.12 shows that fraction I predominantly consisted of hyaluronic acid. The hexosamine content was similar to the uronic acid content. Galm was virtually absent; almost all of the hexosamine was constituted of glucosamine, indicating all are Hyaluronic acid. However, there was not considerable amount of SO_4 in this fraction.

Table 8.13 shows that fraction II predominantly consisted of chondroitin sulphate. Of the total hex-

osamine, galactosamine constituted about twothird and glucosamine one-third; sulphate contents were present in almost 1:1 ratio to the uronic acid.

Table 8.14 shows that fraction III mainly consisted of heparin and as it contained glucosamine with N-sulphation and variable sulfation of four and six hydroxy groups of glucosamine, fraction III contributed only 3–4 % of the total AMPS. Throughout the period of gestation the

Period of gest. weeks	Water g%	Ash g%	K⁺ mEq/kg	Na⁺ mEq/kg
9–12(5)	91.2±0.8	1.80 ± 0.11	31±6.1	130±12
13-16(15)	89.0±0.3	1.50±0.10	30±2.5	128±18
17-20(20)	88.0±0.3	1.30±0.10	33±3.7	104±6
21-24(12)	86.6±0.3	1.25 ± 0.07	34±3.2	90±16
25-28(9)	85.5 ± 0.8	1.05±0.19	32±1.5	97±8
29-32(4)	82	1.2	25	110
0-8 days(5)	83	1.2	25	100
(8)	70±1.5	1.6±0.08	55±1.3	60±11

Table 8.15 Sodium and potassium concentrations of human fetal lungs (walues are expressed in wet weight)

Table 8.16a Extra and intracellular water in lung

Gr.	Gest. per weeks	Water ml/kg	CNS space ml/kg	Intracellular water ml/kg
С	17-20(4)	880	570	380
D	21-24(4)	860	480	380
Е	25-28(4)	850	452	398
Adult	(2)	690	325	365

 Table 8.16b
 Sodium and potassium concentrations in different compartments

Gest. per weeks	Plasma Na⁺ mEq/L	Plasma K⁺ mEq/L	Extracellular lung Na ⁺ mEq/kg	Extracellular lung K ⁺ mEq/kg	Intracellular lung Na ⁺ mEq/kg	Intracellular lung K+ mEq/kg
17-20(4)	137	13.7	85.9	8.5	18.1	22.5
21-24(4)	141	12.9	74.4	6.8	15.6	24.3
25-28(4)	145	12.0	72.1	5.9	24.9	25.3
(2)	133	3.8	47.5	1.5	12.5	53.5

content remained more or less the same. The hexosamine to uronic acid ratio was close to 1. The sulphate to uronic acid ratio of 2–2.5 is in conformity with the higher sulfation of heparin; but in early age of fetuses and in postmortem materials, the sulfate content was not that high.

Water and Electrolyte Contents of Lung

Water, ash, total potassium and total sodium contents of human fetal lung as found are shown in Tables 8.15, 8.16a, and 8.16b. Among human fetal organs, lungs were found to contain water in quantities more than any other organ except the brain. The water content gradually diminished, the ash content was more or less the same. The total potassium contents did not change very much with increasing age of the fetus and were lower than in the adult lungs. The total sodium contents were, however, very high at the early period of gestation and came down somewhat as the age of the fetus increased but even at 32 weeks were considerably higher than in the adult lung. The thiocyanate space in a fetus of earlier gestation period was very higher and came down as the age of the fetus increased but the intracellular space was more or less constant. In the human adult, the cation of the extracellular space was sodium. In the intracellular fluid of the fetus, however, sodium and potassium was calculated to be almost equal in amount. In the adult lung, the major intracellular cations was the potassium ion but some sodium was calculated to be present in the intracellular fluid also. Compared to the fetal lung, the adult lung contained much less mucopolysaccharides and extracellular sodium and potassium.

Brain

Mucopolysaccharides of the Human Fetal Brain at Different Gestation Periods

The amount of AMPS did not change with progress of gestation (Table 8.17). This is in contrast

			Uronic acid mg	Uronic acid mg/g defatted tissues			
Age	No of cases	Total	Ι	II	III		
8-12(3)	3	1.88 ± 0.20	0.80 ± 0.20	0.77 ± 0.20	0.31 ± 0.04		
13-16(5)	3	2.78±0.10	1.75±0.12	0.83±0.14	0.20±0.02		
17-20(6)	6	2.39 ± 0.11	1.4±0.26	0.76 ± 0.06	0.23 ± 0.12		
21-24(3)	3	2.38 ± 0.06	1.53 ± 0.30	0.74 ± 0.04	0.11±0.02		
25-28(6)	4	2.5 ± 0.02	1.62 ± 0.05	0.80 ± 0.09	0.12 ± 0.02		
P.m(3)	3	0.72 ± 0.06	0.17±0.04	0.47±0.10	0.08±0.03		

Table 8.17 AMPS in fetal brain and their fractions

Table 8.18 Composition of mucopolysaccharides of Human Fetal Brain(Fraction I)

Age (Wk)	Uronic acid mg/g of dry defatted tissue	Hexosamine ^a	Glucosamine ^a	Galactosamine ^a	Sulphate ^a
8-12(3)	0.80 ± 0.20	-	-	-	-
13-16(5)	1.75 ± 0.12	1.01	0.95	0.05	0.23
17-20(6)	1.4±0.26	1.01	0.90	0.11	0.53
21-24(3)	1.53 ± 0.30	1.05	1.00	0.05	0.61
25-28(6)	1.62±0.05	1.1	1.00	0.10	0.50
P.m(3)	0.17±0.04	1.0	0.92	0.08	1.05

^aExpressed as molar ration of uronic acid. (Mean and standard deviation)

 Table 8.19
 Composition of mucopolysaccharides of human fetal brain (fraction II)

Age (Wk)	Uronic acid mg/g of dry defatted tissue	Hexosamine ^a	Glucosamine ^a	Galactosamine ^a	Sulphate ^a
8-12(3)	0.77 ± 0.20	-	_	-	_
13-16(5)	0.83 ± 0.14	1.5	-	-	-
17-20(6)	0.76±0.06	1.1	0.30	0.80	1.65
21-24(3)	0.74 ± 0.04	0.90	0.35	0.55	1.12
25-28(6)	0.80 ± 0.09	0.85	0.31	0.54	1.0
P.m(3)	0.47 ± 0.10	0.97	0.07	0.9	1.1

^aExpressed as molar ration of uronic acid. (Mean and standard deviation)

to the situation in the lung where the total AMPS underwent progressive decrease with increase in gestation period; in this respect the results were similar to the findings in the fetal liver although the amount of AMPS in the fetal liver was lower than that of the brain. In postmortem materials on neonatal brain the amount of AMPS was found to be in lesser amount than in fetal brains. When the AMPS of the fetal brain were fractionated, it was again found that fraction I was in a light preponderance, Tables 8.18, 8.19, and 8.20, show that hyluronate constituted a slightly greater fraction of the polysaccharides than the other two fractions. Table 8.18 shows that the hexosamine was

mostly glucosamine but again there was considerable quantity of sulfates in this fraction. The high hyaluronate was a characteristic of fetal brain since postmortem specimens had a much lower hyaluronate and a relatively higher chondroitin sulphate content (Tables 8.18 and 8.19). Fraction II (Table 8.19) contained the chondroitin sulfates and heparitin sulphate; the predominant sugar was galactosamine although some glucosamine was found to be present in this fraction indicating the presence of heparitin sulfate. Fraction III (Table 8.20) constituted a smaller fraction of the AMPS as in the other organs. The molar ration of sulphate to uronic acid was more

Age (Wk)	Uronic acid mg/g of dry defatted tissue	Hexosamine ^a	Glucosamine ^a	Galactosamine ^a	Sulphate ^a
8-12(3)	0.31 ± 0.04	-	-	-	-
13-16(5)	0.20 ± 0.02	0.6	-	-	1.4
17-20(6)	0.23±0.12	0.6	-	-	1.4
21-24(3)	0.11±0.02	0.85	0.71	0.14	1.5
25-28(6)	0.12±0.02	0.89	0.34	0.55	1.2
P.m(3)	0.08 ± 0.03	0.3	_	-	0.8

Table 8.20 Composition of mucopolysaccharides of human fetal brain (fraction III)

^aExpressed as molar ration of uronic acid. (Mean and standard deviation)

Gest. period(wk) Water g% Ash % K+ mEq/kg Na⁺ mEq/kg 9-12(7) 91 ± 0.5 1.46 ± 0.06 40 ± 4 89 ± 3 13-16(12) 90 ± 0.9 1.26 ± 0.17 41 ± 2 98 ± 5 17 - 20(21) 90 ± 0.2 1.22 ± 0.06 48 ± 2 100 ± 2 21 - 24(10) 89.5 ± 0.2 1.19 ± 0.20 39 ± 3 78 ± 6 25-28(9) 88.2 ± 0.2 0.99 ± 0.06 36 ± 2 76 ± 10 0-8days(4)81.7 1.25 58 ± 2 70 ± 5 (4) 81.5 1.15 64 ± 3 72 ± 4

Table 8.21 Water, sodium and potassium contents in human fetal brain (per 100 g wet weight)

 Table 8.22
 Sodium and potassium concentration in different fluid compartments of human fetal brain

		CNS space					
Ges. Wk	Water ml/kg	ml/kg	ICW ml/kg	EC Na ⁺ mEq/kg	EC K ⁺ mEq/kg	IC Na ⁺ mEq/kg	IC K ⁺ mEq/kg
17-20(4)	902	460	442	69.0	6.6	31.0	41.4
21-24(4)	905	430	475	66.7	6.7	11.7	32.3
25-28(4)	910	460	450	71.3	5.9	4.7	30.1

No adult brain could be analyzed

than 1 in fraction III as it should be (Tables 8.17, 8.18, 8.19, and 8.20.

Sodium and Potassium Concentrations in the Human Fetal Brain

The water content of the brain varied form 91 % at 9–12 weeks to 86 % at 32 weeks. The potassium content of whole human fetal brain was between 36 and 48 mEq/kg.; in adult brains the potassium content was much higher (64 mEq/kg). The sodium content of the fetal brain varied from 76 to 100 mEq/kg. The sodium content of adult brain was found to be around 72 mEq/kg. The difference between the fetal and adult brain was very great indeed with regard to the sodium content (Table 8.21).

The thiocyanate space was determined in four fetuses each of groups C, D and E. Results are shown in Table 8.22. The total water and thiocyanate space did not change very much in the three groups. The extracellular sodium and potassium did not change. But there was a great reduction in the intracellular sodium content. In fetuses of earlier gestation period, i.e., 17–20 weeks, a large amount of intracellular sodium was present (Table 8.22).

Glucosamine-6-Phosphate Synthetase (EC 2.6.1.16) in Liver, Brain and Lung

Table 8.23 shows the activity of Gm-6-P synthetase of human fetal liver at different gestation periods upto 28 weeks. Results are compared with the activity of adult human liver obtained during cholecystectomy. The specific activity of fetal liver varied from 63 to 74 units (uM/g protein/h) at different gestation periods. These activities were 2.5–2.8 times that of human adult
		µM product/g tis	μM product/g tissue/h			
Age in weeks	No. cases	Liver	Lung	Brain		
8–12	6	5.58 ± 0.75	ND	0.26 ± 0.01		
13–16	7	3.10±0.39	0.27 ± 0.07	0.27 ± 0.02		
17–20	11	2.89 ± 0.24	0.23 ± 0.02	0.17±0.01		
21–24	6	2.90 ± 0.27	0.17 ± 0.02	0.17 ± 0.02		
25–28	ND	ND	ND	ND		
29–32	ND	ND	ND	ND		
35–55 years	5	1.72 ± 0.24	0.50 ± 0.10	ND		

 Table 8.23
 Glucosamine-6-phosphate synthetase in liver, brain and lung. (Enzyme activity in tissue)

Table 8.24 Glucosamine-6-phosphate synthetase in liver, brain and lung. (Specific activity of the enzyme)

		µM product/g p	μM product/g protein/h			
Age in weeks.	No. cases	Liver	Lung	Brain		
8–12	6	65.0 ± 4.0	ND	10.5 ± 1.6		
13–16	7	68.0±7.3	12.0±2.3	11.5±2.5		
17–20	11	73.9±3.4	11.5±1.5	10.5±0.7		
21–24	6	68.0±5.1	8.8±1.1	14.0±0.7		
25–28	ND	ND	ND	ND		
29–32	ND	ND	ND	ND		
35-55 years	5	27.0±5.1	9.4±2.0	ND		

liver. The difference between the activity of fetal liver and adult was statistically significant (p < 0.001). Ghosh et al. [24] showed the activity in 18,000 g supernatant in rat liver to be 20.2 μ M/h/g protein which is also similar to human adult value. Spiro [26] from specific activity curve showed that the liver is the primary site for the synthesis of serum glucosamine. Phelp [27] purified and studied the kinetic property of rat liver enzyme.

The activity of the fetal lung was one-fifth to one-eighth of the activity of the corresponding fetal livers and the adult lung obtained in pneumonectomy specimen away from the site of lesions had an activity which was about a third of the adult liver. Enzyme activity is measured as the formation of product (Glucosamine-6-p) as μ M/g tissue/h and Specific activity as μ M/g protein/h (Tables 8.23 and 8.24).

Glucosamine-6-phosphate synthetase activity of the human fetal cerebral cortex is shown in Table 8.23, from 8 weeks of gestation to 24 weeks the activity varied from 10.5 to 14 units/g protein/h. The activity was compared to that of the lungs at corresponding gestation. Values are more or less similar. The enzyme activity of liver was six times higher than that of brain.

The effect of various hormones on certain tissues (relaxin, cortisone, testosterone) causes polymerization of the ground substance. It also increases the proportion of the soluble fraction of colloids compensated by the decrease of bound electrolytes arising from the increased binding strength of water. Depolymerisation of the mucroproteins decreases the binding capacity of MPS for cations. The binding of metal by monomers is less strong than by polymerized acids.

Discussion

MPS and Relationship with Water and Electrolytes

The amount of hexuronic acid in the extracted, partially purified MPS material is, therefore, a measure of the total MPS of an organ. In embryos the mesenchyme is principally cellular but in fetuses the matrix is quantitatively greater than the actual cellular element. The higher uronic acid in fetal organs of early gestation period may be due to a relatively greater content of mesenchymal cells in a given volume of tissue. It must however, be cautioned that these cells are relatively undifferentiated and less ready to extrude the proteoglycans in the interstitial space.

Liver

Mucopolysaccharide

Fraction I, comprising Hy should not contain Glam and sulfate. Although glam content was proportionately low, there was quite a large amount of sulfate. It appeared, therefore, that some amount of Cs and Hs was present in this fraction. There was a diminution in the amount of fraction I, at later periods of gestation and in immediate postnatal life. Fraction II was also not a pure class of compounds in as much as in this fraction a large quantity of glucosamine was found. The sulfate content more or less corresponded with an equimolar quantity of hexosamine. The contamination might therefore, be either due to heparitin sulfate or even to heparin. The quantity was almost half of that found in Fraction I. Fraction III constituted less than 10 % of the total UA content and the fraction also contained a mixture of a variety of compounds. If fraction III constituted entirely of heparin, it should have a hexuronic acid ratio to sulfate more than 2. However in the earlier period of gestation and in adult livers this ratio was less than 2. It is only in the later periods of second trimester (21-28 weeks) that the ratio corresponded to that of heparin. During the progress of gestation there was a decrease in the total UA content of the liver; the decrease was mostly in the Hy fraction. We do not know the contribution of the individual proteoglycans to the overall function of the MPS. The decrease in Hy fraction with presumably a greater stability of the matrix specially in extrauterine life would go against the indispensability of Hy to form a stable structure with link proteins for the other polysaccharides.

Water and Electrolytes

The main difference in the sodium and potassium concentrations of the serum in fetal and adult sera

consisted in the higher serum potassium content in fetal life. Whereas adult serum potassium concentrations were 4.8 mEq/L ±0.5, the corresponding concentrations in fetal sera were 10.8 mEq/L ± 1.3 . Since there was no difference in the sodium contents of fetal and adult sera, the total cation concentration and the total osmolarity of fetal sera were higher than those of adult sera. This higher osmolarity of fetal water probably has something to do with the floatation characteristic of the fetus in the amiotic fluid. The density of a few smaller fetuses was found to be 1.103 by water displacement method, whereas that of the amniotic fluid was around 1.117. The maintenance of this density was a function of the osmotic tension of the body fluids of the fetus. A teleological explanation of the higher serum potassium in fetal life may also be given. A growing organ actively synthesizing protein and depositing the protein in the tissues needs potassium. The higher serum potassium level in fetal life ensures a liberal supply. Furthermore, potassium concentration in the intracellular fluid of the fetal liver was lower than that of the adult liver. Perhaps, the sodium pump in the tissue is not as effective in fetal liver as in adult liver so that potassium leaks out into the extracellular fluid.

The thiocyanate space in fetal liver was greater than in adult liver. The extracellular fluid was therefore present in greater quantity in fetal life than in adult liver. In fact, the higher total water content of fetal liver may be ascribed entirely to the greater quantity of extracellular fluid. The intracellular fluid was not increased in fetuses. This huge amount of extracellular fluid is not present as free water but must be in a soluble form in combination with a macromolecule. The macromolecule is not presumably a protein but a polysaccharide, a mucoploysaccharide to be exact. Total mucopolysaccharide contents was higher in earlier gestation period than in later. The decrease was still more manifest in neonatal livers on postmortem examination. The total water content of fetal liver was lower than that of the adult liver. However, the decrease in total mucopolysaccharides from fetal to neonatal liver was more than the decrease in total water from fetal to adult life. It might therefore, be assumed

that amount the various functions of the mucopolysaccharides, only one function is related to its binding property with water. There are other important functions of this class of compounds, which are perhaps, more important in fetal life. The decrease of mucopolysaccharides during progress of gestation was almost entirely ascribable to a decrease in hyaluronates. Since the binding of hyaluronate to water is a well-known phenomenon, the relation between the decrease of hyaluronate content and water is more than coincidental.

The presence of a higher amount of water in the extracellular space necessitates retention of salts to maintain osmotic equilibrium. The total cation content of the extracellular fluid per kg of organ weight in fetal liver amounted to 75 mEq whereas in the adult liver it was only 52 mEq. The increase of extracellular water in fetal liver as compared to adult liver was between 50 % and 70 %. Therefore, the total cation content in extracellular fluid of fetal liver was increased relatively as well as absolutely. The total mucopolysaccharide contents of fetal liver in earlier gestation periods was about double that of neonatal livers. Hence the mucopolysaccharides must play an important role in the greater binding of not only water but of cations as well.

Lung

Mucopolysaccharide

Total MPS of the fetal lung were in all instances higher than those of the liver at the corresponding periods of gestation. It might signify a greater extracellular material in the fetal lung than in the fetal liver. However, as already mentioned, all of the acetone extracted, papain digested material of the fetal lung went into solution in the dilute alkali; but a fraction of the corresponding part of the fetal liver did not. It might be interpreted to mean that the fetal liver contained more differentiated fibrous elements than the lung counterpart. With progress of gestation there was decrease in total MPS per gm. of dry lipid free tissue in the lung as in the liver. These findings are in contrast with the observations of Horwiz and Crystal [28] on rabbit lung. They observed the GAG content

to be relatively constant around 3.4 uM/g of dry weight in fetal lungs and to increase to around 6 uM/g in adult lungs. In human fetuses on the other hand the GAG content were found to be much higher (around 24 uM/g dry weight in the early period of gestation) than in rabbit lung and during growth of the lung there was marked decrease in the MPS content. The adult human lung was found to have a much smaller MPS content (around 3.5 uM/g. dry weight). The decrease might signify that the amount of matrix becomes less as the organ becomes more compact or the more differentiated a tissue becomes the less amount of extracellular space it holds. The decrease of the amount of uronic acid with the onset of extrauterine respiration is real and probably results from compression of the surrounding lung parenchyma by the expanding alveoli and absorption of intra alveolar and perialveolar material. The decrease of MPS with increase in gestation period may also be related to the changing populations of cell types found in the rat lung in the immediate neonatal period (Kaufman, Burris [29] and Wiebel, 1975). When the MPS was fractionated into Fraction 1, containing predominantly hyaluronic acid it was found that there was a pronounced reduction in this fraction with progress of gestation and especially after both, i.e., after extrauterine respiration has occurred. Fraction 2 containing mainly CS was also found to be highest at 14-16 weeks of gestation but the decrease with increasing period of gestation was not spectacular like Fraction 1. Again after respiration had set in, fraction 2 was also considerably reduced. Fraction 3 containing mainly heparin constituted 3-4 % of the total mucopolysaccharides in the fetal lung; in the neonatal lung this fraction was about 10 % of the total mucopolysaccharides. The heparin content of human fetal lung was therefore lower than that of the neonatal lung.

Water and Electrolytes

Among human fetal organs, lung was found to contain more water than any other organ except the brain. Thus at 9–12 weeks of gestation the water content of the lung was 91 %, while that of the liver was 80.5 %. The MPS content of the

fetal lung was also very high as compared to the liver. Thus at 13-16 weeks of gestation the total uronic acid per g of dry defatted lung was 4.6 mg, while that of the liver was only 0.74 mg. It may be that the greater amount of water in the fetal lung of earlier gestation period could be accounted for by its association as hydronium ions to the polyanions of the AMPS especially with Hy in extracellular fluid. The higher water content of the fetal lung at earlier gestation period could not be due to lesser cellularity at this period as the amount of DNA per gm. of tissue was found to be highest (unpublished doctoral dissertation of 1st author, Dr. Chameli Ganguly, 1978, Calcutta University, available at Calcutta University) in the earliest gestation period studied. As in the fetal liver the thiocyanate space was highest at the earliest gestation period and gradually come down. The intracellular water did not change much with increase in gestation period. Hence, the higher water content could be ascribed to the extracellular fluid only. The high amount of water in the extracellular space naturally would necessitate movement of salts in order to keep proper osmotic equilibrium. The total of sodium and potassium ions in the tissue, as a whole, underwent a progressive diminution from about 160 mEq at 9-12 weeks of gestation to 130 mEq at 25-28 weeks. In adult lungs, however, total sodium content was lower and total potassium higher than in fetal lungs. The distribution of these ions in the extra and intracellular fluid was also different in the fetal and adult lungs. The total sodium and potassium ions in the extracellular fluid of the fetal lung was much higher than that of the adult lung; the potassium ions in the intra-cellular fluid was correspondingly lower in the fetal lung than in the adult lung. Although the major cation of the extracellular space in fetal lung was calculated to consist of sodium, surprisingly in the intacellular fluid, a little less than half of the major cation was found to be sodium was calculated to be present in the intracellular fluid. The reason for this is unclear the estimation could not be wrong. We checked and rechecked in all possible ways. We sent some dried specimens abroad. Their values were similar with our.

Brain

Mucopolysaccharide

The MPS of the rat brain during development have been studied by Margolis et al. [30] and by others. We found that the total MPS were around 2 mg uronic acid per g of dry defatted tissue in human fetal brain at 8-28 weeks of gestation. The total MPS did not change significantly from 9 to 28 weeks of gestation. The water content in these brains, however, decreased progressively from 91 % at 9-12 weeks to 88.2 % at 25–28 weeks. Thus the water content could not be related to the total MPS content. The nonvariability of the total MPS at 9-28 weeks of gestation is also reflected in the hyaluronate (Fraction I) and chondroitin sulfate (Fraction II) contents; they also did not register much change. In neonatal postmortem brains both total MPS and hyaluronate contents (Fraction I) were much lower than in the fetal brain. Fraction II (chondroitin sulphate) was relatively predominant in the postmortem specimens. There was a reduction in Fraction III as gestation period increased. In previous studies on developing rat brain, a high amount of hyaluronic acid in early gestation was thought to be responsible for greater retention of water [30]; since the polymeric chains of hyaluronic acid were shown to have significant role in hydration and solvent transfer in tissues [31]. However, in none of the studies significant attention has been paid to Fraction III, containing heparin. Our findings that Fraction III in fetal brain underwent a decrease with progress of gestation may have something to do with the interstices of the brain mass permitting easier migration of neuronal cells, at earlier periods of gestation. The degree of polymerization of the polysaccharide units in Fraction III may determine the electrostatic holding of H₃O ions in the interstices of the neuronal and possibly glial cells too.

Water and Electrolytes

Brain is the most watery of all fetal organs; at very early periods of gestation, around 9 weeks, brain and lungs have water contents of around 90 % of the wet weight. This huge amount of water in the brain is largely extracellular as judged by the thiocyanate space. Our finding of brain thiocyanate space in fetal brain of 46 % is not very disproportionate. Tissue spaces available to thiocyanate are higher than inulin spaces. Intracellular water which is calculated from fetal total water minus the extracellular water came to around 46 %. Our observations are on the whole cerebral matter; we are, therefore, far from understanding the water content of any particular kind of cells of the brain.

The electrolyte concentrations of the human fetal brain shows a higher water, ash and sodium content at earlier periods than at later periods of gestation, in fetal brain potassium contents were lower than sodium; whereas in adult brain the contents were around the same. Intracellular fluid of the fetal brain sodium and potassium was found to be present in almost equal concentrations at earlier periods of gestation. In adult cat cerebral cortex [32], intracellular sodium could be calculated to be only a small amount of the intracellular potassium contents.

Glucosamine-6-Phosphate Synthetase

In fetal life, the glucosamine-6-phospahate synthetase activity of the liver was more than double that of adult liver. This enzyme starts the reaction sequence necessary for the synthesis of the AMPS and is the rate limiting step. The high activity in fetal life is consistent with the AMPS content in this period of life. The activity of their enzyme in the liver is many times higher than the activity in several other growing organs. The high quantity of Glm-6-P synthesized. Glucosamine-6-phosphate synthetase activity of the fetal lung was only about a fifth of the activity of the liver. Whereas there was no reduction of the liver enzyme activity with progress of gestation. The AMPS content of the fetal lung. Admittedly, one of the chief functions of the Glm-6-P synthetase in to provide the glucosamine for synthesis of the AMPS. It is also true for adult liver that the Glm-6-P synthesized in the liver undergo dephosphorylation to 9 Glm and the Glm is utilized elsewhere. Glm-6-P synthetase activity of brain and also of the lung was lower than that of the liver. When expressed per gm. of tissue the activity of the liver was between 12 and 22 times that of the brain; when expressed per gm. of protein the hepatic activity was between five and seven times that of the brain. We did not have any adult brain to analyse for the enzyme activity. So we had to be content with determination of the enzyme content in fetal brain upto 24 weeks of gestation.

Conclusions

An organ consists of a supporting framework which is permeated by the characteristic cells of the organ and vascular network. A growing organ of the fetus is at first characterized by condensation of the mesodermal and endo-or ectodermal cells, which divides and differentiates into the tissues and cells of the particular organ. It has been envisaged that, at first, there is proliferation and differentiation of mesenchymal cells, along with the characteristic cells of the organ. The function of the mesenchymal cells is to provide a suitable matrix formed by the mucopolysaccharides, collagen fibers and the fibroblast and mast cells. This complex matrix is permeated three dimensionally by the characteristic cells of the organ. This matrix constituting the interstitial material undergoes variation in composition as the organ grows, divides, differentiates, and becomes mature. The object of the enquiry has been to study some of the properties of this matrix in three human fetal organs, namely the liver, lungs and brain. In all three organs the total water content was higher in fetal life than in adults. Water was present at the highest concentration at the earliest gestation period that could be studied. It gradually decreased with the progress of gestation. The thiocyanate space was found to be much higher in fetal organs than in adult organs; the intracellular water in fetal and adult organs was more or less the same. Thus the higher water content of fetal organs could be entirely accounted by the extracellular compartment. The acidic mucopolysaccharides of the liver was present in almost twice the concentration found in neonatal livers and the extracellular fluid in fetal life was also about twice that in adult liver. Thus there was a positive correlation between the acidic mucopolysaccharides and the

extracellular fluid. Perhaps the water molecules were bound in the interstices of the polyanions as hydronium ions (H_3O^+) . The decrease of acidic mucopolysaccharides with progress of gestation was found to be due to a decrease of hyaluronic acid which was more concerned with water binding. The total extracellular sodium and potassium contents in fetal liver was higher than in adult livers and intracellular potassium was lower. The higher extracellular water in fetal livers attracted the hydronium ions and the sodium and potassium ions in order to maintain proper osmotic tension. These cations, in turn, attracted anions in order to preserve electrochemical neutrality. The total osmotic tension in fetal organs was found to be somewhat higher than in adult organs, which finding may have something to do with the buoyant density of the fetus in the amniotic fluid. Total mucopolysaccharides of the fetal lung was much higher than in fetal liver, so also was the total water content. There was a progressive decrease in the water and hyaluronate contents of the lung as gestation advanced. As in the fetal liver, the fetal lung had an increased thiocyonate space, where the acidic mucopolysaccharides attracted hydronium ions and other counter ions like sodium and potassium. A considerable quantity of intracellular sodium was calculated to be present in the lung for which no explanation could be given. Intracellular potassium content was lower in fetal lung than in adult lung. In fetal brain the total acidic mucopolysaccharide content was higher than in fetal liver. There was no measurable decrease of total acidic mucopolysaccharides and hyaluronates from 9 to 28 weeks of gestation and the water content also did not change much during this period. In neonatal and adult life the brain had lower acid mucopolysaccharide, hyaluronate and water contents. In fetal brains the extracellular and intracellular water were found to be equal in amounts. Sodium was present mostly in the extracellular compartment. The intracellular potassium content in fetal brains was lower than reported adult values. During the biosynthesis of mucopolysaccharides, the primary enzymatic reaction is amination of fructose-6-phosphate brought about by the enzyme glucosamine-6-phosphate synthetase. Upto the year 1978, there has been no report on the activity of this enzyme in human fetal organs. An assay system has been developed to measure the activity of this enzyme. Some properties of the enzyme have been studied and its purification has been partially successful. The activity of the enzyme was highest in the fetal liver, about five to seven times that of the fetal lung or brain. The fetal liver had also 2.5 times the activity of the adult liver. It appears, therefore, that the glucosamine-6-phospahte synthesized in the liver undergoes dephorphorylation to glucosamine which is made available to other tissues for the necessary biosynthesis of the mucopolysaccharides and other compounds.

Summary

- (A) The acidic mucopolysaccharides, especially the hyaluronate content of human fetal organs bore a positive correlation to the water content; probably the polyanions became associated with the hydromium ions.
- (B) The anionic macromolecules formed complexes not only with hydronium ions but with other cations like sodium and potassium which in turn attracted other anions like chlorides, bicarbonates etc.
- (C) The human fetal organs had higher osmotic tension in their body fluids than human adult organs, which may have something to do with the buyant density of the fetus in the amniotic fluid.
- (D) The glucosamine-6-P synthetase, the primary rate limiting enzyme in the biosynthesis of mucopolysaccharides was present in a very high concentration in the fetal livers.

References

- Jeanloz RW. Chemical structure of the polysaccharides of connective tissue (mucopolysaccharides). Bull Soc Chim Biol. Paris. 1960;42:1829–31.
- Stacey M. The chemistry of mucopolysaacharides and mucoproteins. Adv Carbohydr Chem. 1946;2:161.
- 3. Stacey M, Barker SA. Carbohydrates of living tissues. London: van. Nostrand; 1962.

- Tsiganos TCP, Muir H. In: Vogell HG (ed.). Connective tissue and ageing, Amsterdam: Excerpta Medica; 1973;1:132.
- Hopwood JJ, Robinson HC. The structure and composition of cartilage keratin sulphate. Biochem J. 1974;141:517.
- 6. Buedecke E, Kroz W, Tittor W. Hoppe-Seyler's. Z Physiol Chem. 1967;384:651.
- Rosenberg L, Hellman AK, Kleinschmit WJ. Macromolecular models of protein-polysaccharide from bovine nasal cartilage based on electron microscope studies. London: London Academic Press, Biol Chem. 1970;245:4123.
- Hascall VC, Riolo REJ. Characteristics of the proteinkeratan sulfate core and of keratan sulfate prepared from bovine nasal cartilage proteoglycan. J Biol Chem. 1972;247:4529.
- Grossman BJ, Dorfman A. In vitro comparison of the antithrombic action of heparin and chondroitinsulfuric acid-B. Pediatrics. 1957;20:506.
- Leoewi G, Mdyer K. The acid mucopolysaccharides of embryonic skin. Biochim Biophys Acta. 1958;27:453.
- Dorfman A, Schiller S. Biological structure and function, London: Academic; 1961;1. p. 328.
- Stryer L. Energy transfer in proteins and polypeptides. Rad Res. 1960; (Suppl. 2):432–51.
- Spiers CH. Das Wasseraufnahmevermogen von Kollagen und Leder. J Soc Leather Trades' Chem. 1952;36:20.
- Eder HA. Determination of the thiocyanate space. In: Visscher MB (ed.). Methods in medical research, Chicago: Yearbook Publishers, 1951;4:48–53.
- Singh M, Bachhawat BK. Isolation and characterization of glysoaminoglycans in human brain of different age groups. J Neurochem. 1968;15:249–58.
- Schiller S, Slover GA, Dorfman AJ. A method for the separation of acid mucopolysaccharides: its application to the isolation of heparin from the skin of rats. Biol Chem. 1961;236:983.
- Scott JE. In: Glick D (ed.). Methods on biochemical analysis. New York: Inter Science Publishers; 1960;8:145.
- Dische Z. A new specific color reaction of hexuronic acids. J Biol Chem. 1947;167:189.

- Bitter T, Muir HM. A modified uronic acid carbazole reaction. Anal Biochem. 1962;4:330–4.
- Elson LA, Morgan WT. A colorimetric method for the determination of glucosamine and chondrosamine. Biochem J. 1933;27(6):1824–8.
- Morgan WT, Elson LA. A colorimetric method for the determination of *N*-acetylglucosamine and *N*-acetylchrondrosamine. Biochem J. 1934;28:988.
- Ludowieg J, Bewrmaman JD. A method for analysis of amino sugars: specificity and mechanism of the reaction. Carbohydr Res. 1968;8:185–92.
- Dodgson KS, Price RG. A note on the determination of the ester sulphate content of sulphated polysaccharides. Biochem J. 1962;84:106.
- Ghosh S, Blumenthal HJ, Davidson E, Roseman S. Glucosamine metabolism. V. Enzymatic synthesis of glucosamine 6-phosphate. J Biol Chem. 1960;235:1265–73.
- Lowery OH, Rosebrough NJ, Farr AL, Randall RJ. Protein measurement with the Folin phenol reagent. J Biol Chem. 1951;193:265.
- 26. Spiro RG. Studies on the biosynthesis of glucosamine in the intact rat. J Biol Chem. 1959;234:742–8.
- Winterburn PJ, Phelps CF. Purification and some kinetic properties of rat liver glucosamine synthetase. Biochem J. 1971;121(4):701–9.
- Harwitz AL, Crystal RC. Content and synthesis of glycosaminoglycans in the developing lung. J Clin Invest. 1975;56(5):1312–8.
- 29. Kauffman SL, Burris PH, Weibal ER. The post-natal growth of the rat lung. II. Autoradiography. Anat Rec. 1974;180:63–76.
- Margolis RU, Margolis RK, Chang LB, Preti C. Glycosaminoglycans of brain during development. Biochem. 1975;14(1):85–8.
- Laurent TC, Barany EH, Carlsson B, Tidare E. Determination of hyaluronic acid in the microgram range. Anal Biochem. 1969;31:133–45.
- Bourke RS, Tower DB. Fluid compartmentation and electrolytes of cat cerebral cortex in vitro–II sodium, potassium and chloride of mature cerebral cortex. J Neurochem. 1966;13(11):1099–117.

Urea Biosynthesis in the Human Fetal Liver

Chameli Ganguly[†], Bimal Samanta, Gitanjaly Guha Thakurata[†], Chaitali Bhattacharya, K.L. Mukherjee[†], and Niranjan Bhattacharya

Introduction

There are three types of animals: (1) Ammoniotelic, (2) Uricotelic, and (3) ureotelic. All mammals are Ureotelic.

Ammonia which is one of the main substrates producing urea is toxic to the system. There are a number of ways by which ammonia is produced in the system of mammals.

Kerbs [1] demonstrated that the mammalian liver and kidney catalyzed the oxidative deamination of both D- and L- amino acids. It was shown that L- amino acid oxidases do not have the activity to account for the catabolism of all L- amino acids. Except L- glutamic dehydrogenase, there is very few known L- amino acid oxidative system in mammals. But transamination reaction by transaminases are very active in mammals. On this basis, the mechanism of the L- amino acid oxidation producing ammonia is said to be by involvement by transamination of L- amino acid with α - ketoglutarate producing L- glutamate which is then oxidatively deaminated by glutamate dehydrogenase to form ammonia and α - keto glutarate. The following are the routes by which ammonia can be produced in the mammalian system.

G.G. Thakurata, MSc, PhD (Cal) Formerly, Department of Biochemistry, National Medical College and Hospital, Kolkata, India

C. Bhattacharya, MSc, PhD (Cal) UGC Fellow, Institute of Child Health, Kolkata 700017, India

K.L. Mukherjee, MB, PhD (Cal), PhD (Wisconsin) Former Head of the Department of Biochemistry, Institute of Post Graduate Medical Education and Research, West Bengal, Kolkata, India N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India e-mail: sanjuktaniranjan@gmail.com

[†]Author was deceased at the time of publication

C. Ganguly, MSc, PhD • B. Samanta, MSC, PhD (Cal) Former Biochemist, Central Calcutta Society for Advancement of Human Development and Research, Kolkata, India



The enzyme glutamate dehydrogenase is a mitochondrial enzyme of widespread distribution. In animal tissues, the activity of the enzyme is high in liver, kidney, brain but low in heart.

That the liver is the site of urea formation in ureotalic organisms was proved by Bollman et al. [2] in 1924.

Sir Hans Krebs and Hanseleit [3], in 1932, demonstrated an entirely novel mechanism of urea synthesis by the tissue slice technique.

Measurement of the rate of the production of urea in the presence of ammonia and other substrates showed that, among a large number of substances tested, two aminoacids, ornithine and citrulline, can under certain conditions, accelerate the rate of formation of urea from ammonia.

Krebs [4] was led to regard the function of ornithine as that of an intermediate in a series of reactions which would bring about its regeneration and arginine filled a required niche in the concept. On the basis of this hypothesis and the effect of citrulline on liver slices to produce urea from ammonia, it was assumed that citrulline may also be an intermediate in the conversion of ornithine into arginine according to the scheme:

Ammonia and carbon dioxide react with ornithine to form citrulline (Step I). Citrulline then condenses with a second molecule of ammonia to form arginine (Step II). The cycle is completed by the action of arginase (Step III). Under suitable conditions a small amount of ornithine will thus catalyse the formation of a large amount of urea provided the supplies of ammonia and carbon dioxide are adequate. The reaction of citrulline to arginine could occur in two distinct step – condensation and simultaneous breakdown of the condensation product.

The condensation reaction is said to take place by the condensation of aspartic acid and citrulline in the presence of ATP and Mg^{2+} .



This intermediary condensation product was named as argininosuccinic acid [5]. The enzyme responsible for the formation of arginino succinic acid was named as arginino-succinic acid synthetase.

The enzyme that claves argininosuccinic acid to arginine and fumaric acid is known as argininosuccinase or L-argininosuccinate arginine lyase.

> $Mq^{2}+$ $NH_4+ HCO_3+ 2ATP ----- NH_4-C-O-PO_3 + 2ADP + Pi$ N-acetyl glutamate Arginase (EC 3.5.3.1)

Arginase (L- arginineurohydrolase) is one of the earliest enzyme known. It catalyzes the

NH₂

(CH₂)₃

CH NH₃+

COO-Arginine

The enzyme is present in a high concentration in the livers of all ureotelic animals [7].

The location of the enzyme is in the cytoplasm. Co²⁺ and Mn²⁺ are the activators of the enzyme.

Biosynthesis of Urea in Mammals during Development

The evolutionary adaptation of nitrogen excretion was first expressed in an elegant form by Balwin [8, 9]. The nitrogenous excretory products may be varied by metabolic adaptations, with evolutionary branching. Needham [10, 11] suggested that during development, the emoryo excretes ammonia first and eventually develops a mechanism for nitrogen excretion characteristics of the species, recapitulating its evolution. This was supported by Plentin [12] in 1957 who showed that 50-fold

hydrolysis of arginine to ornithine and urea as shown by the equation.

Among mammals, the enzymes of the urea cycle were extensively studied in rats. Kennan and Cohen [15] and Rahis and Suihkonen [16] showed that the activity of the urea cycle enzymes was very low in fetal rat liver but it showed a rapid postnatal increase. The overall capacity of liver slice to synthesize urea was almost absent from fetal rats and guinea pigs and increased rapidly to adult capacity after birth [17]. Kennan and

biosynthesis would take place in preparation for

the nonaquatic environment after birth [15].

ter was carbamyl phosphate synthetase. The formation of carbamyl phosphate was then caralyzed by carbamyl phosphate synthetase of liver according to the following reaction [6].

The name of the enzyme proposed for the lat-



It

Cohen [15] also showed that the enzymes of the urea cycle were present in significant amounts in the liver of the youngest pig embryo studied (28 days). The species difference, which is quite marked in these two groups, viz., rat and pig, was also discussed by Kennan and Cohen [15] in relation to fetal membranes of the two animals and the maturation of the fetus and new born [15]. They (1959) also differentiated the two animals with regard to the maturity of the animals at birth, the development of the mesonephric kidney and the thickness of the placental membrane.

Miller and Chen [18] showed that the activities of the enzymes of the urea cycle are very low in the fetal liver of rats and they increased rapidly after birth; the rate limiting enzyme was again found to be the condensing enzyme of arginine synthetase system, i.e., argininosuccinic acid synthetase. Illnerova and Kubat [19] also investigated whether there were some factors which could be responsible for the sudden increase of the urea cycle enzymes after the 14th day of its gestation and also for the postnatal increase of the enzymes.

Slemons and Morriss [20] studied the maternal and fetal serum concentrations of non-protein nitrogen and urea and showed that there was no difference in concentrations in these two compartments in human beings and rats. The direct measurements of the production of urea was made by Manderscheid [21] and Kennan and Cohen [15] by using the tissue slice techniques of Krebs and Henseleit [3] and they observed that urea could be produced in liver slices from human fetal liver at 3–4 months of gestation. In humans, urea can cross the thin fetal membrane system to the mother efficiently, for there is no allantoic vesicles in the human being [15].

From the proceeding brief review, it appears that although urea biosynthesis in human fetuses has been studied by various observers, there is need for further study in order to evaluate the role of the fetal liver in making urea and also to use this parameter as an index of biochemical differentiation of the fetal liver. Our group of researchers led by Prof. K.L. Mukherjee, studied the human fetal liver at various stages of gestation at SSKM Hospital, Calcutta, from 1977 onwards, as mentioned in Chapter 6. The fetuses were obtained from consenting mothers undergoing hysterotomy under the MTP Act or for other reasons. Due ethical consent was obtained from the then institutional ethical committee. The object of our study was:

- 1. Measurement of urea and ammonia nitrogen concentrations in the bladder fluid of fetuses.
- 2. Measurement of the activities of ornithine transcarbamylase, argininosuccinate synthetase, argininosuccinase and of arginase in the livers of these fetuses.
- 3. Measurement of the concentration of ornithine in fetal livers.
- 4. Some studies on the metabolism of ornithine in fetal livers.

Ammonia Nitrogen and Urea Nitrogen Concentrations of the Bladder Fluid and Plasma

All the fetuses studied in the present series contained some fluid in the bladder. Amount of the fluid varied from fetus to fetus and cannot be taken as an indication of maturity of the fetus. Although it was generally true that the amount of fluid obtained in fetuses of earlier gestation period was smaller in quantity than in the more matured fetuses, fetuses of similar weights had widely varying amounts of bladder fluid. Ammonia nitrogen and urea nitrogen concentrations of the bladder fluid were determined in 23 fetuses; Results are shown in Table 9.1. The ammonia nitrogen concentrations varied from 0.23 to 1.25 mg per 100 ml. There was no correlation between body weight and ammonia nitrogen concentrations of the bladder fluid. The same thing was observed with urea-nitrogen concentrations of the bladder fluid. It varied from 8.5 to 24.4 mg per 100 ml. Here also, there was no correlation between body weight and urea nitrogen concentrations. Table 9.2 showed that concentrations of ammonia nitrogen and urea nitrogen did not change with gestation.

The ammonia nitrogen concentrations of the plasma of seven human fetuses and their mother's were determined along with the urea nitrogen concentrations. Results are shown in Table 9.2. In this table is also shown the composition of the urine passed on the first day after birth and the

composition of the bladder fluid. It appears from the data that the urea content of the bladder fluid was more or less the same as that of the fetal and maternal plasma. The ammonia nitrogen concentration of the bladder fluid was somewhat higher than that of the fetal plasma which was more or less the same as that of the maternal plasma.

Table 9.1 Concentration of ammonia-nitrogen and ureanitrogen with respect to different gestational age and corresponding body weights

Gestational age (weeks)	Body weights (grams)	Ammonia- nitrogen	Urea- nitrogen
9–12	6-15(0)	x	x
12–15	15-65(0)	x	X
15-18	65-255(6)	0.42+0.064	15.0±1.12
18-21	255-377(4)	0.6 ± 0.064	11.1±1.79
21–24	377-630(5)	0.61 ± 0.165	17.9 ± 2.74
24–27	630-855(4)	0.36 ± 0.054	11.0±0.22
27-30	855-1,490(4)	0.57+0.079	16.1+2.26

Table 9.2 Composition of bladder fluid, newborn urine, fetal and maternal plasma at 20–28 weeks of gestation

	Bladder fluid	New born urine	Fetal plasma	Maternal plasma
Urea	35	378	29	32
Ammonia	0.42	-	0.045	0.04
Sugar	37	Nil	45	93
Sodium	145	359	127	132
Potassium	4.7	12.7	3.7	4.9
Chloride	99	279	105	93

Enzyme Activity

Ornithine Transcarbamylase during Development

Activities of ornithine transcarbamylase in the livers of the fetuses are shown in Table 9.3. Activities were low in the fetuses of earlier gestation periods. There appear to be no period in the life of the fetus when the increase of this enzyme activity was abrupt and more than at other periods. However, the number of samples examined was rather small and hence no definitive statement can be made. The functional differentiation of the liver with respect to this enzyme, appears to take place throughout gestation. Even when the baby weight is about 1.5 kg. and is certainly viable, the ornithine transcarbamylase activity is less than half of the adult value and the specific activity is only a fifth.

Activities of Argininosuccinic Acid Synthetase during Development

This method of measurement of argininosuccinic acid synthetase activity involves the determination of citrulline concentration before and after incubation. The difference found was never too high.

The activity of this enzyme followed the pattern for ornithine transcarbamylase rather closely. The activity of argininosuccinic acid synthase per gram wet weight of liver was lowest in

 Table 9.3
 Activities of ornithine transcarbamylase in developing human fetus

		Activity	Specific activity	Sp. activity (fetus)
Gestational ages (weeks)	Body weights (grams)	Range	Range	Sp. activity (adult)
9–12	6-15 (3)	85.4-100.2	1.1–1.5	1/32
12–15	15-65(3)	104.0-120.0	1.5-1.8	1/26
15–18	65-255(5)	150.5-278.0	1.7–5.2	1/13
18–21	255-377(6)	329.5-387.0	4.8-5.5	1/8
21–24	377-630(4)	229.0-472.5	3.52-6.47	1/8
24–27	630-855(3)	457.5-480.3	7.05-7.19	1/6
27–30	855-1,490(4)	674.1–749.6	8.2–9.2	1/5
Adult 29 years				1
Adult 55 years				1

N.B.: * Activity is expressed as micromole product per gram wet weight

Specific activity is expressed as micromole product per milligram protein

the smallest fetuses and increased with increasing gestational age (Table 9.4). The activity of the enzyme per gram of wet weight of liver was 13.09 in fetuses of 12–15 weeks of gestation and increased with progress of gestation at the highest gestational age, viz., during 27–30 weeks of gestation to 81.97. The specific activity, i.e., the activity per milligram of protein was also low ranging from 0.165 in the earliest fetuses studied to 1.17 in the 27–30 weeks of gestation. We could get only two livers from adult persons who were 29 and 55 years old. The activity found was 90 and 143 units respectively pre gm of liver. The specific activity however was rather close in the two subjects.

Activities of Argininosuccinase during Development

The method of estimation of argininosuccinase activity involves the determination of arginine by the conversion of latter to urea by added arginase to the system.

The results of determination of argininosuccinase activity in the livers of fetuses of different gestation periods are shown in the Table 9.5. The enzyme activity could not be carried out in the very small fetuses. From 12 to 15 weeks of gestation the enzyme activity steadily increased to 24 weeks. The increase was real in as much as the

Table 9.4 Activities of argininosuccinic acid synthetase in developing human fetuses

		Specific	Sp. activity (fetus)
Gestational ages (weeks)	Activity range	activity range	Sp. activity (adult)
9–12	Not done	Not done	Not done
12–15	10.56-17.2	0.13-0.22	1/30
15–18	13.4–34.0	0.19-0.495	1/14
18–21	31.7-45.7	0.46-0.64	1/9
21–24	30.1-39.5	0.41-0.58	1/10
24–27	38.0-68.04	0.5-1.036	1/7
27-30	70.4-89.5	0.99–1.4	1⁄4
Adult 29 years	90.0	4.74	1
Adult 55 years	142.8	5.30	1

N.B.: * Activity is expressed as micromole product per gram wet weight

Specific activity is expressed as micromole product per milligram protein

specific activity was also found to increase. At about 30 weeks of gestation the enzyme activity per gram of wet weight of the liver was almost the same as that found in adults but the specific activity was half that of adults.

Activities of Arginase in Developing Human Fetuses

Both the enzyme activity per gram of fresh tissue as well as the specific activity were more or less the same throughout the gestation periods investigated namely 9–30 weeks, except perhaps in the very small fetuses, where the activity found was somewhat low. These values are shown in Table 9.6. As compared to the adult values, the enzyme activity of the fetal livers was almost half of that in the adult.

The Probable Rate in Which Urea Can Be Produced

Five enzymes are consecutively required for the production of urea from ammonia, the product of one reaction serving as the substrate for the next. We will consider the last four enzymes in this sequence. The rate of ornithine transcarbamylase at 15 weeks of gestation is 1/32nd of the rate in the adult. We, can, therefore, assume that the

Table 9.5 Activities of argininosuccinase in developing human fetuses

		Specific	Sp. activity (fetus)
Gestational ages (weeks)	Activity range	activity range	Sp. activity (adult)
9–12	Not done	Not done	Not done
12–15	14.1-20.8	0.2-0.29	1/9
15-18	13.5-32.2	0.29-0.52	1/6
18-21	32.0-49.8	0.52-0.71	1⁄4
21–24	38.4–64.6	0.5-1.01	1/13
24–27	62.2–77.4	0.93-1.03	1/2
27-30	62.3-88.0	0.69-1.1	1/2
Adult 29 years	88.3	2.21	1
Adult 55 years	55.0	2.40	1

N.B.: * Activity is expressed as micromole product per gram wet weight

Specific activity is expressed as micromole product per milligram protein

product is formed at 1/32nd of the adult rate. The rate of the next enzyme, the arginosuccinic acid synthetase is also 1/30th of the adult rate. The problem now is whether the product of the first reaction is sufficient to saturate the second enzyme to form ES complex? Since the second enzyme activity was found reduced to also the same extent, the reduction may be assumed to involve either the amount of the enzyme or the presence of an inhibitor. In the former event, the reduced amount of the product of the first reaction may or may not be sufficient to saturate the second enzyme. In the latter event the rate of the overall reaction occurring as a result of the

Table 9.6 Activities of arginase in developing human fetuses

Gestational		Specific	Sp. activity (fetus)
ages	Activity $\times 10^3$	activity	Sp. activity
(weeks)	range	range	(adult)
9-12	6.93-10.02	40.8-70.6	Not done
12-15	8.734–18.76	67-13-	1/3
		120.0	
15-18	11.5–16.8	80.8–98.8	1⁄2
18-21	10.32–19.7	86.0-126.0	1⁄2
21-24	12.337-18.816	92.2-120.0	1⁄2
24–27	12.264–25.872	91.5-134.0	1⁄2
27-30	11.2–25.5	85.0-118.0	1/2
Adult	48.72	193.3	1
29 years			
Adult	50.75	193.5	1
55 years			

N.B.: * Activity is expressed as micromole product per gram. Wet weight

Specific activity is expressed as micromole product per milligram protein

combined action of the first and second enzymes will be found to be successively affected. From many considerations, it appear that the overall rate of the reaction is successively affected in the sequence of reactions.

Since the reactions are mutually dependent the laws of conditional probability was used to calculate the final overall rate of urea production from ornithine and carbamyl phosphate to urea (Table 9.7). To take an example, the probability of a molecule of carbamyl phosphate forming citrulline is 1/32nd of the adult rate; the probability of citrulline forming argininosuccinic acid is 1/30th of the adult rate. Eventually, it can be argued that the probability of a molecule of carbamyl phosphate going on to urea is 1/14,598 of the adult rate in a fetus of 12–15 weeks to 1/104th of the adult rate in a fetus of 27–30 weeks.

Concentration of Ornithine in Human Fetal Liver

Since arginase activity of the fetal livers was comparatively higher that the other enzymes of the urea cycle and since arginine was presumably available to the fetus through the placental circulation, it appeared reasonable to study the metabolism of the products of arginase activity. The urea being a highly diffusible substance was probably eliminated by the maternal kidneys. The level of free ornithine in the liver was, therefore, studied in fetuses of different gestation periods.

	Ratio of specific ad	Ratio of specific activities (fetus/adult)				
Gestational age (weeks)	Ornithine transcarbamylase	Argininosuccinic acid synthetase	Argininosuccinase	Arginase	Probability of the rate of urea production fetus/adult	
9–12	1/32	X	X	1/3	X	
12-15	1/26	1/30	1/9	1/2	1/14,598	
15-18	1/13	1/14	1/6	1/2	1⁄2,364	
18-21	1/8	1/9	1⁄4	1/2	1/656	
21-24	1/8	1/10	1/3	1/2	1/552	
24–27	1/61	1/7	1/2	1/2	1/200	
27-30	1/5	1/4	1/2	1/2	1/104	

Table 9.7 Probability of the rate of urea production in human fetuses

We estimated the ornithine content by employing a partially purified ornithine transcarbamylase and estimating the citrulline produced.

The concentrations of free ornithine in human fetal livers are shown in Table 9.8. It varied from 0.22 to 0.58 μ mol/g of wet weight of liver. There was no correlation between the concentration of ornithine and gestational age. In most of the cases, the concentration of ornithine was found to vary from 0.2 to 0.3 μ mol/g wet weight of liver. It appears from these data that the concentration of ornithine remained almost constant throughout the gestation of ornithine in the adult. The concentration of ornithine in fetal livers was between two and four times than that of reported values for adult livers [22].

Since the free ornithine content of fetal livers was found higher than the reported values of adult livers, it was thought worthwhile to enquire into the contents of polyamines in fetal livers since they are synthesized from ornithine.

Table 9.8 Content of ornithinine in human fetal liver

Sl. No.	Body weight (grams)	Gestation period (weeks)	Micromole of ornithine/gram wet weight of liver
1.	175	17	0.28
2.	295	20	0.22
3.	330	21	0.52
4.	420	22	0.27
5.	590	24	0.58
6.	660	25	0.246
7.	1,010	28	0.2
8.	1,450	30	0.32

Concentration of Polyamines in Human Fetal Liver

The polyamines were extracted from the livers and separated by high voltage electrophoresis. Results are shown in Table 9.9.

The concentrations of putrescine and spermine expressed in micromoles and millimicromoles per gram of wet weight of liver in the Table. Concentration of putrescine increased from 0.34 µmol in the fetuses of 15–18 weeks of gestation to 3.08 µmol in the gestation period of 21-24 weeks. The concentration of this polyamine decreased to 1.2 µmol per gram of wet weight of liver in the higher gestation periods. The increase of putrescine concentration from 12 to 24 weeks and its subsequent decrease could not be fully ascribed to the higher water content of the liver in the fetuses of earlier gestation periods. When the concentrations were expressed per 100 g of dry weight, then also the same increase and the subsequent decrease was demonstrable.

The concentration of spermine on the other hand steadily increased from the earlier gestation period to the late gestation periods. The concentration of spermine was expressed in millimicromoles per gram of wet weight of liver were as those of putrescine were expressed in micromoles, indicating the higher concentration of the latter polyamine in fetal livers. Spermine content increased from 9 millimicromoles per g to 113.7 millimicromoles per g with the increase of gestational age from 15 to 30 weeks. The steady increase in the spermine with increase of gesta-

 Table 9.9
 Concentration of polyamines in human fetal liver

	Putrescine		Spermine	Spermine		Molar ratio	
Gestational ages (weeks)	Micromoles per gram. Wet weight	Micromomes per 100 g dry weight	Millimicromoles per gram. Wet weight	Micromoles per 100 g. Dry weight		ne wet weight/ e dry weight	
9–12	X	Х	X	X	Х	X	
12–15	X	Х	X	X	X	X	
15-18 (5)	0.34 ± 0.11	200	9±0.81	5.3	37.7	37.7	
18-21 (4)	1.84 ± 0.8	920	50.01±5.1	25.0	36.7	36.8	
21-24 (6)	3.08±0.6	1,400	66.6±12.2	30.0	46.2	46.6	
24–27 (6)	1.127 ± 0.73	427	85.91±47.5	35.7	13.1	11.9	
27-30 (7)	1.2±0.19	500	113.7±51.0	47.1	10.5	10.6	
3 years	1.217	420	103.0	35.5	11.8	11.8	

tion was also apparent even when the data were expressed per 100 g of dry weight. The ratio of putrescine to spermine which was high at first, gradually came down to a steady proportion from 24 weeks of gestation onward.

In Table 9.10, the concentrations of putrescine and spermine were expressed per milligram of DNA in the human fetal liver and the results were compared with the concentrations of RNA per milligram of DNA in the same fetuses. Even then, the results are reminiscent of the earlier mode of expression, i.e., per milligram of tissue weight. The putrescine level increased from 0.009 μ mol at 15 weeks to 0.37 μ mol at 30 weeks. In the 3 year old child, the level was 0.32 μ mol.

Spermine concentrations per milligram DNA steadily increased from $0.25 \,\mu$ mol at 15 weeks to 11.7 millimicromoles at 30 weeks. In the 3 year old child the level was 27.5 millimicromole.

RNA concentrations per milligram of DNA was more or less the same from 15 to 30 weeks of gestation. The concentration was higher in the liver of the 3 year old child.

Activities of Ornithine Amino Transferse (EC.2.6.1.13) during Development of Human Fetuses

The activity of this enzyme was studied in a small number of fetuses, upto 24 weeks of gestation by the method described by Jenkins and Tsai [23]. No activity was found in any of the fetuses studied.

Discussion

In the present investigation, the composition of the bladder fluid shows that urea concentration of the bladder fluid was more or less the same as that of the fetal and maternal plasma. The fluid that is filtered at the glomerulus naturally contains an ultrafiltrate of plasma and, therefore, the glomerular fluid contains urea in the same concentration as it is present in the fetal plasma. The fluid is reabsorbed in the proximal tubules in an iso-osmotic concentration so that when the fluid reaches the descending limb of the loop of Henle the osmolar concentration is the same as that of the glomerular filtrate as it formed. The urea concentration is also the same as that of the plasma. Later on, when the water gets reabsorbed in the loop and the distal tubules the urea concentration of the urine also increases proportionally. Since in the fetal bladder fluids, urea concentration was the same as that of the fetal plasma it can be presumed that there was no reabsorption of water in the distal tubules. Whether there was reabsorption in the proximal tubules or not it is difficult to say, because in either case the osmolar and urea concentrations of the fluid will be the same as that of the ultrafiltrate.

Table 9.10 Comparison of concentrations between polyamines and ribonucleic acid per milligram of deoxyrihonucleic acid during development of human fetal liver

Gestational ages (weeks)	Putrescine	Spermine	Molar ratio putrescine/spermine	Ribonucleic acid
9–12	X	X	Х	Х
12–15	X	X	X	Х
15–18 (5)	0.0094 ± 0.004	0.25 ± 0.1	37.6	0.369 ± 0.107
18–21 (4)	0.1505 ± 0.06	4.56±1.23	33.0	0.321 ± 0.073
21-24 (6)	0.3705 ± 0.01	6.88 ± 1.75	53.8	0.349 ± 0.082
24–27 (6)	0.128 ± 0.088	9.16±4.8	14.0	0.335 ± 0.071
27-30 (7)	0.16 ± 0.085	11.7±4.7	13.7	0.385 ± 0.088
3 years	0.3245	2,746	11.8	0.54

Results are expressed as means ± S.E.M. to the number of fetuses indicated in parenthesis. Concentrations of putrescine and spermine are expressed in micromoles and millimicromoles respectively per milligram of DNA. RNA is expressed in milligram per milligram of DNA.

The ammonia concentrations of the bladder fluid, although higher than that of the fetal plasma, was nowhere around the concentrations found in new born or adult urine. Urine is a freely diffusible substance; in fact it has been used to determine the total body water concentration. The exchange of urea across the placenta is, therefore, complete. The fetal plasma, in fact contains urea in the same concentration as that of the maternal plasma. If any urea is added to the fetal circulation in the course of fetal metabolism it would eventually find its way to the maternal circulation and be dealt with by the mother. The urea concentration of the fetal plasma, would not, therefore, increase as a result of addition by the fetus and cannot be used as critical of urea production by the fetus, even though fetus has no direct access to the environment to discharge its urinary contents. The fetal bladder fluid closely corresponds to an ultrafiltrate of fetal plasma in as much as it contains more or less the same concentrations of sugar, sodium, potassium and chloride. Sabrazes and Fauquet [24] showed that urines formed in utero had urea but the concentration found was extremely low and less than 50 mg per 100 ml. Barlow and McCance [25] also showed that fetal urines contained total nonprotein nitrogen to the extent of 41-61 mg per 100 ml.

The total amount of the fluid in the fetal bladder probably increased with gestation but the concentration of urea nitrogen and ammonia nitrogen remained constant throughout the gestation. Rubatelli and Formentin [26] as already described showed that the concentrations of various non-protein nitrogeneous compounds in the umbilical vein and artery and antecubital vein of the human fetus were similar to that of a normal person. Urea nitrogen and ammonia nitrogen excreted in the urine per day are 10-15 g and 400-1,000 mg respectively. A normal adult excretes an average of 1,500 ml of urine per day. So if the concentrations be expressed in 100 ml of urine, the concentrations of urea nitrogen and ammonia nitrogen are 700-1,000 mg and 26-67 mg respectively. The concentrations of urea and ammonia nitrogen found in the bladder fluids of these fetuses were far

from the concentrations found in the urines elaborated after birth. The fluid in the urinary bladder of such fetuses can, therefore, not be properly be called urine, although the ultrafiltration must have occurred through the glomerular filter. No selective reabsorption in the tubule had taken place. It is of some comparative interest to note that in the marine teleost fish which do not synthesize urea, the functional adult kidney has been lost. Birds and saurian reptiles, which are uricotelic, have a marked reduction in the number of glomeruli and of course, lack of a functioning urea cycle.

Thus there would appear to be some relationship between the ability to synthesize urea and the degree of kidney development. Kennan and Cohen [15] thought that the placental mammals apparently began to synthesize urea when the embryonic glomerual kidney had developed and the time of the appearance of urea cycle enzymes might in turn be correlated with the thickness of the fetal membrane system, the presence and size of the allantoic vesicle and the relative maturity of the new born animal.

For the conversion of urea from ammonia, five enzymes, viz., carbamyl phosphate synthetase, ornithine transcarbamylase, argininosuccinic acid synthetase, argininosuccinase and arginase are required in sequence. The product of one enzyme serves as the substrate for the next. Of these five enzymes, the first enzyme carbamyl phosphate synthetase was not measured in this study. Raiha and Suihkonen [27] measured the activity of this enzyme and found it to increase with increasing gestational age, thus producing carbamyl phosphate. There are two major reactions in which carbamyl phosphate participates, namely pyrimidine biosynthesis and urea formation. The synthesis of carbamyl phosphate brings the product to a branch point and hence is subjected to autoregulation.

It is not surprising to find that there seems to be a reciprocal relationship between the activities of the two enzymes, aspartate and ornithine transcarbamylase, both utilizing carbamyl phosphate as their substrates. The former one is necessary to produce pyramidine precursors and the latter enzyme produces citrulline in the formation of urea. During regeneration of the liver the activity of aspartate transcarbamylase increased while that of ornithine transcarbamylase decreased [28]. Charbonneau et al. [29] also found that during embryonic life in the rat, the activities of all the enzymes of the urea cycle were very low, but aspartate transcarbamylase activity was relatively high. It was claimed that the mother synthesized and eliminated urea, sparing the embryo of these functions. The embryo would, however, have to synthesize nucleic acids for growth and hence pyramiding biosynthesis from carbamyl phosphate would be obligatory. Kretchmer et al. [30] showed that the *de novo* pathway for the synthesis of pyrimidines was at least as active in the fetal liver of the developing animals like rat, rabbit etc. as it was in the adult and that there was no relation between this pathway and that of urea biosynthesis in liver. Hager and Jones [31] suggested from their experiment that the glutamine dependent carbamyl phosphate synthetase provided carbamyl phosphate for the pyrimidine pathway. Rapidly growing normal tissues (fetal liver, spleen, testis) have much more carbamyl phosphate synthetase II activity that other tissues (lung, uterus, brain, heart and muscle) [32]. In the regenerating liver and also in fetal rat liver ornithine transcarbamylase was found to be low [28]. When the regeneration was complete or after the rat was born, the aspartate transcarbamylase activity relatively decreased and the activity of ornithine transcarbamylase correspondingly increased thus channeling the carbamyl phosphate along one or the other route according to the metabolic requirement characteristic of the particular condition. The urea cycle starts from the carbamyl phosphate synthetase I activity: actually speaking the characteristic reaction starts from ornithine transcarbamylase. As is so often the case in a chain reaction like in the biosynthesis of urea, the first step is usually the rate limiting reaction. In this cycle, therefore, the ornithine transcarbamylase may be considered to play a rate limiting role in the sequence of reactions ultimately ending in biosynthesis of urea.

The activity of ornithine transcarbamylase in human fetal livers was found to be very low in fetuses of early gestation period. It is very easy to find a teleological explanation for this phenomenon. Since the fetus is a rapidly growing organism it needs to synthesize the nucleic acid pyrimidines and, therefore, all the carbamyl phosphate is channelized to pyrimidine biosynthesis.

The growth of the human fetal liver mass with respect to time is more or less uniform [33]. For example the liver represents 44 mg per g of body weight at 12 weeks and 38 mg per g at 32 weeks. The slight decrease that occurs may be accounted for by a relative decrease of water. Even when the baby is 6 months old it still weighs 46 mg per g of body weight. Even though the mass of the liver increases proportionately to the mass of the body weight, there is a definite decrease of relative amounts of DNA in the liver [34]. For example, the DNA represents 0.87 g % in the fetuses at 14 weeks and 0.50 g % in the fetuses at 28 weeks. Thus with continuing growth of body weight, there is relative decrease of DNA and hence the number of cells per unit mass of the liver. The cells, therefore, accumulate matter probably at the expense of water.

One of the ways of accumulation of matter will be to increase the protein content and hence presumably of the enzyme amount. To take the instance of ornithine transcarbamylase during the growth of the human fetus, can we assume that the relatively undifferentiated liver cell of early gestation period produces little enzyme whereas the amount of the enzyme increases with differentiation. The DNA of the liver cell containing the operator gene of ornithine transcarbamylase in the early period of gestation lies unmasked only in 3 % of the liver parenchymal cells so that they can be transcribed and translated; in the other 97 %, the gene is blocked. The process of differentiation, therefore, can be interpreted as unmasking of the gene for a particular enzyme. However, even at this level we cannot get away from speaking about vague mystery things, of whose nature we do not know. The liver performs its functions remarkably well in fetuses very early in life. For example, it produces plasma proteins of the fetus and synthesizes all kinds of intermediates of biosynthesis for the overall economy of the fetus even at 12 weeks of gestation. Why is it that such a liver performs plasma protein synthesis so effectively but synthesizes ornithine transcarbamylase very poorly? Is there a hierarchy of functions that are to be eventually performed by the liver. We still do not know answers to these questions.

The next enzyme in the sequence, the argininosuccinic acid synthetase behaves in the same way as ornithine transcarbamylase. If we represent the activity found in adults as one, the respective relative activity of ornithine transcarbamylase and argininosuccinate synthetase bears a very close parallelism at each period of gestation. It is possible, therefore, that the operator gene for the two enzymes may be either closely linked or the same in human fetuses. An alternative explanation would be, that unmasking of the DNA for the synthesis of mRNA for both the enzymes occurs by the same substance which may be either an initiator or a derepressor of an inhibitor.

The argininosuccinate synthetase activity is normally about a hundredth of the ornithine transcarbamylase. Hence it is presumed synthetase is the rate limiting enzyme in the urea cycle. In the present experiments both ASA synthetase and ASA-ase activities are about a tenth of the ornithine transcarbamylase; so it is possible that the overall reaction of arginine synthesis will be about a hundredth of the rate of the ornithine transcarbamylase.

The next enzyme in the sequence in urea biosynthesis is argininosuccinase which splits argininosuccinic acid into arginine and fumaric acid.

Relatively speaking, the activity of this enzyme as compared to the activity in the adult was not as low as the other enzymes of the urea cycle except arginse.

When an organ like the human fetal liver functionally differentiates, can we justifiably say that such a functional differentiation occurs in almost all the cells or in a small number of cells comprising a tissue? That there is a specific pattern in the rate of increase of each of these individual enzymes might be interpreted to mean that a certain percentage of cells in assuming the function and the rate of increase is a function of the individual enzyme concerned. Thus the rate of increase of argininosuccinase is different from the rate of increase of the other enzymes of the urea cycle.

The next enzyme in sequence in urea biosynthesis is arginase, which splits arginine into urea and ornithine. Among the enzymes of the urea cycle, the activity of this enzyme was found to be relatively higher than any other enzyme of the urea cycle, when the activity in the fetus was compared to that in an adult. Furthermore, the activity was relatively constant throughout the gestation period studied. That the activity was found to be relatively high even at an early period of gestation could be interpreted to mean that the body uses the products of arginase activity, i.e., ornithine or urea.

The activities of these enzymes in the liver of adult human beings or in the livers of fetuses are more or less in the ranges described by previous workers like Cohen and Brown [35]. It is hard to explain the low value in the fetuses of the different gestation periods.

If the probability of the rate of production of urea by the fetus be compared to that of an adult, it appears that the fetus can produce urea no doubt, but the rate as compared to the adult is negligible. Thus the little amount of urea that is formed probably goes to the mother's circulation to be eliminated in her urine. It is probable that the fetal kidneys do not perform the excretion of urea, as the bladder fluid did not contain any extra urea in any of these fetuses.

Ornithine occurs at a point of intersection of a number of metabolic pathways. It is one of the products of arginase activity on arginine. It can be decarboxylated to yield putrescine. By transamination it can form γ -glitamic semialdehyde. After decarboxylation to putrescine it can form the oligoamines like spermiding and spermine. From glutamic semialdehyde, proline can be formed by reduction.

The discovery that polyamines stimulate growth of microorganisms, and in animal and plant tissues, is of particular interest, and the interaction between polyamines and nucleic acids already studied in cell free extracts probably provides the mechanism for this stimulation [36].

It appears from the Table 9.9 that the concentration of putrescine increases upto a certain gestation period and then decreases whereas the concentration of spermine increases with the gestational age of the fetus. The sudden drop of putrescine content is also evident from the molar ration of putrescine to spermine, which also increases upto a certain gestation and then falls down. During early fetal life putrescine may be the main requirement because it regulates the synthesis of other polyamines.

In human fetuses the polyamine content is not high in liver cells and therefore cannot be said to perform a stabilizing function on DNA, nor can it have much effect of RNA synthesis either. The control of polyamine synthesis in human fetal livers is unlike that in embryonic chick, toad, rad or sea urchin, where polyamine biosynthesis is definitely increased in embryonic life.

Summary

- Normal human fetuses of different gestation periods were collected on ice after hysterotomy and the different enzymes of the urea cycle were estimated in the liver. Mother's blood, fetal blood, amniotic fluid and bladder fluids were collected and ammonia nitrogen and urea nitrogen concentrations were measured in these fluids.
- The urea concentrations of the bladder fluids of these fetuses were very low as compared to the urea concentrations of the adult urines. The values were more or less similar to the values found in the maternal blood and fetal blood.
- 3. Thus the fluid in the urinary bladder of such fetuses could not properly be called urine although the ultrafiltration must have occurred through the glomerular filter. No selective reabsorption in the tubule had taken place.
- 4. Of the four enzymes of the urea cycle that were studied, ornithine transcarbamylase had the lowest activity along with argininosuccinic acid synthetase. Their activities increased with progression of gestation.
- Activity of argininosuccinase was higher than the previous two enzymes, when they were compared to the values found in the adult. It increased with gestation and attained

a half of the adult activity at later periods of gestation.

- 6. Activity of arginase, was found to be the highest in the cycle. The activity of this enzyme in contrast with other enzymes of the urea cycle did not progressively increase with progression of gestation.
- 7. In the series of reactions ultimately producing urea, since the activity of all these enzymes were found reduced, it appeared that in vivo very little urea was synthesized by the fetal liver, especially in the earlier periods of gestation; the rate increased with progress of gestation but in comparison to the situation in an adult, very little urea was synthesized even in the later periods of gestation. Some urea may be produced by the comparatively higher arginase activity from agrinine, made available to the fetus through the maternal circulations.
- 8. The very low ornithine transcarbamylase and argininosuccinic acid synthetase activities and comparatively higher arginase activity of the fetal liver might give rise to increased ornithine concentration in fetal liver. The ornithine concentrations were in fact found to be two to four times higher in the fetal liver than in adult livers. Since ornithine does not occur in any body proteins, the ornithine must be metabolized through some other alternate metabolic pathway.
- 9. Of the two possibilities of utilization of ornithine besides the urea cycle, the proline biosynthesis and polyamine biosynthesis, the former pathway was found to be absent in fetal livers.
- 10. Some polyamine concentrations were measured in these livers and were found to increase from a very low value at earlier periods of gestation to about a half to a third of the value found after birth. The concentration of putrescine was found to increase upto a certain gestational age of the fetus and then it decreased. Perhaps, the ornithine had to accumulate to some critical value before it could induce the ornithine decarboxylase and putrescine also had to rise to a respectable concentrations before it could stimulate spremidine and spermine biosynthesis.

11. When the concentrations of polyamines were compared with RNA content by expressing the values per milligram of DNA, it was found that RNA concentration per milligram of DNA was more or less the same but the spermine concentration increased with progression of gestation. The ration of putrescine to spermine at first increased and then decreased indicating the formation of spermine in later periods of gestation.

References

- 1. Kerbs HA. The metabolism of amino acids in the animal body. Z Physiol Chem. 1933;217:191–227.
- Bollman JL, Mann FC, Magath TB. Studies on the physiology of the liver. Am J Physiol. 1924;69:371–92.
- Kerbs HA, Henseleit H. Untersuchungen uber die harnstoffbildung im tierkorper (Studies on urea formation in mammals). *Hoppe-Seylers* Z Physiol Chem. 1932;210:33–6.
- Kerbs HA. Urea synthesis. In: Sumher, Mryback, editors. The enzymes, vol. II, Pt. 2. New York: Academic Press. pp. 866–85.
- Walker JB. Argininosuccinic acid form chorella. Proc Natl Acad Sci U S A. 1952;38:561–6.
- Anderson PM, Meister A. Evidence for asn activated form of carbon dioxide in the reaction catalyzed by Escherichia coli carbamyl phosphate synthetase. Biochemistry. 1965;4:2803–8.
- Greenberg DM. Arginase. In: Boyer L, Myrback, editors. The enzymes, vol. 4. 2nd ed. New York: Academic; 1960. p. 257–67.
- Baldwin E. An introduction to comp. Biochemistry. 1st ed. London: Cambridge Univ. Press; 1937.
- 9. Baldwin E. Dynamic aspects of biochemistry. 2nd ed. London: Cambridge Univ. Press; 1952.
- Needham J. Chemical embryology. New York: Macmillan; 1931.
- 11. Needham J. Biochemistry and morphogenesis. London: Cambridge Univ. Press; 1942.
- Plentl AA. The origin of amnioktic fluid. In: Ville CA. editor. Gestation: Transactions of the Fourth Conference. New York: Josiah Macy Jr. Foundation; 1957. pp. 71–114.
- Brown Jr GW, Cohen PP. Biosynthesis of urea metamorphing tadpoles. In: McElory WD, Glass B, editors. The chemical basis of development. Baltimore: Jhons Hopkins Press; 1958. p. 495–513.
- Brown Jr GW, Cohen PP. Comparative biochemistry of urea synthesis. J Biol Chem. 1959;234:1769–74.
- Kennan AL, Cohen PP. Biochemical studies of the developing mammalian fetus. Dev Biol. 1959;1:511–25.
- Raiha NCR, Suihkonen J. Development of the enzymes of urea biosynthesis in rat and human liver. J Padiatr. 1966;69:934–5.

- Raiha NCR, Kretchmer N. Urea; biosynthesis during development of the mammal. J Padiatr. 1965;67:950–1.
- Miller AL, Chen P. Development of urea cycle enzyme activity in the liver-of fetal and neonatal rats. Enzymol Biol Clin. 1970;11:497–503.
- Illnerova H, Kubat M. Factors possibly inducing the rise in activity of urea cycle enzymes in rat liver during development. Physiol Bohemoslov. 1968;17(1):77–80.
- Slemons JM, Morriss WH. The non-protein nitrogen and urea in the maternal and fetal bold at the time of the birth. Bull Johns Hopkins Hosp. 1916;27:343–50.
- 21. Manderscheid H. Uber die harnstoffbildung bei den wirbeltieren. Biochem Z. 1933;263:2456–249.
- 22. Ryan WL, Carver MJ. Free amino acids of human foetal and adult liver. Nature. 1966;212:292–3.
- Jenkins WT, Tsai H. Ornithine aminotransferase (pig kidney). In: Tabor J, Tabor CW, editors. Methods in enzymology, vol XVII, Pt. A. New York/London: Academic press. pp. 281–5.
- Sabrazes, Fauquet. Proprietés hematolytiques de la pre-mière urine du nouveau-né. CR Soc Biol Paris. 1901;53:272.
- Barlow A, McCance RA. The nitrogen partition in new born infants' urine. Arch Dis Child. 1948;23:225–30.
- Ruabtelli FF, Formentin PA. Ammonia nitrogen, urea, uric acid blood levels in the mother and in both umbilieal vessels at delivery. Biol Neonatorum. 1968;13(3–4):147–54.
- Raiha NCR, Suihkonen J. Development of urea synthesizing enzymes in human liver. Acta Padiatr Scand. 1968;57:121–4.
- Sallach HJ, Fahien LA. Nitrogen metabolism of amino acids. In: Greenberg DM, editor. Metabolic pathways, vol. III. 3rd ed. New York: Academic; 1969. p. 1–94.
- 29. Charbonneau R, Roberge A, Berlinguet L. Variation with age of the enzymes of the urea cycle and aspartate transcarbamylase in rat liver. Can J Biochem. 1967;45:1427–32.
- Kretchmer N, Hurwitz R, Raiha NCR. Urea and pyrimidine metabolism during development. Biol Neonatourm. 1966;9(1–6):187–93; Discussion 93–96.
- Hager SE, Jones ME. A glutamine dependent enzyme for the synthesis of carbamyl phosphate for pyrimidine biosynthesis in fetal rat liver. J Biol Chem. 1967;242:5674–80.
- Yip MCM, Knox WE. Glutamine dependent carbamyl phosphate synthetase. J Biol Chem. 1970;245:2199–204.
- Dutta G, Mukherjee KL. Unpublished observations. 1974.
- Mitra, Mukherjee KL. Unpublished observations. 1974.
- Cohen PP, Brown Jr GW. Ammonia metabolism and urea biosynthesis. In: Florkin, Mapson, editors. Comparative biochemistry, vol. III. New York: Academic; 1960. p. 161–244.
- Russel DH, Potyraj JJ. Spermine synthesis in the uterus of the ovariectomized rat in response to oestradiol-170. Biochem J. 1972;128:1109–15.

Mechanism of Rejection of Human Fetal Adrenal Cortex

10

Chameli Ganguly[†], Bimal Samanta, Gitanjaly Guha Thakurata[†], Chaitali Bhattacharya, Reba Bhattacharya[†], K.L. Mukherjee[†], and Niranjan Bhattacharya

Introduction

In the course of development from early recognizable fetal stage to about 2 years after birth, the adrenal gland in man and some other primates undergo an interesting sequence of changes, which has been described as physiological involution. The so called involution occurs in what is known as the fetal type of cortical cells. The recognition of fetal adrenal cortex as distinct from the adult cortex is ascribed to Elliot and Armour in 1911 [1]. They found that the adrenal cortex was composed of a narrow rim of cells which constituted the periphery of the cortex and from which the adult cortex was derived. This peripheral zone was named the "Permanent Zone" and the area between the permanent zone and the

C. Ganguly, MSc, PhD

B. Samanta, MSc, PhD (Cal) Biochemist, Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

G.G. Thakurata, MSc, PhD (Cal) Formerly, Department of Biochemistry, National Medical College and Hospital, Kolkata 700040, India

C. Bhattacharya, MSc, PhD (Cal) UGC Fellow, Institute of Child Health, Kolkata 700017, India medulla was occupied by cells which were called the so called "Fetal Cortical Cells". This fetal zone degenerated after birth and was subsequently replaced by regeneration from the peripheral rim of cells.

The human adrenal shows some interesting variations in size and weight at different ages of life. In the fetus for example, it increases steadily in size and is very large in proportion to the size of the fetus. This remarkably large size of the fetal adrenal gland is mainly due to the presence of a well developed fetal zone which involutes after birth.

The organ, therefore, in fetal life is about 20 times heavier than in the adult. At midterm the fetal zone constitutes 85 % of the total adrenal cortex. At birth it is 84 % and 82 % for the still

K.L. Mukherjee, MB, PhD (Cal), PhD (Wisconsin) Former Head of the Department of Biochemistry, Institute of Post Graduate Medical Education and Research, Kolkata, West Bengal 700015, India

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India e-mail: sanjuktaniranjan@gmail.com

[†]Author was deceased at the time of publication

Former Biochemist Central Calcutta Society for Advancement of Human Development and Research, Kolkata 700040, India

R. Bhattacharya, MSc, PhD (Cal) Formerly, Research Fellow, Department of Biochemistry, Institute of Postgraduate Medical Education and Research, Kolkata 700020, India

[©] Springer International Publishing Switzerland 2016 N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development: First and Second Trimesters*, DOI 10.1007/978-3-319-14874-8_10

born and live born infants, respectively. Two weeks later, the fetal zone is 58 %; 3-12 months later it is only 13 % and seen as a narrow rim [2]. Swinyard [3] and Tahka [4] also found that the permanent zone about the time of birth, occupied 15-20 % of the whole adrenal cortex.

Immediately following birth, there is a spectacular decline in weight; an abrupt 50 % reduction in their mass occurs. The weight of adrenals decreasing after birth appear to be due to the fact that the fetal cortex undergoes a process of degeneration and is not completely replaced until puberty. At the end of the first year, in the human adrenal gland, the fetal cortex has completely disappeared and has been replaced by typical normal cortex but the gland still continues to grow until the individual reaches the age of 20 or 21. Since such a great part of the fetal adrenal cortex persists only during the fetal period, it seems reasonable to assume that the growth and differentiation of this fetal zone are related to some specific functions during pregnancy.

The histological structure of the human fetal adrenal gland differs from that of the adult adrenal gland. The fetal adrenal gland is composed of an outer or true cortical part consisting of a narrow zone of small deeply staining cells arranged in ovoid or acinar groups and an inner portion made up of large polyhedral cells which are frequently fat containing, take a pale acidophilic stain and have no definite arrangement. In the most central region, a few deeply staining medullary cells may be seen.

Keene and Hewer [5] confirmed the derivation of the adrenal cortical cells from the coelomic epithelium, but showed that the fetal and adult type cortical cells arose separately. In the 10 mm embryo, the cells forming the fetal adrenal cortex were already migrating into the mesenchyme, whereas those that were to form the adult type cortex first appeared in the 12 mm stage as a cap overlying the previously differentiated fetal cortex; the extension of this cap to envelop the fetal cortex followed, simultaneously with invasion of the gland with chromaffin cells. Uotila [6] also studied the development of these zones in man and found that fetal cortical cells were first observed in an 8 mm embryo (about 21 days old), whereas the first definite adult-type cortex was observed at 14 mm (about six and half weeks). The primary inner zone of fetal cortex, is present as a definite structure by the fourth fetal month [7] and continues with no morphological change up to the time of birth. Hyperemia is a feature of the fetal cortex at all times and is maximal at the time of birth [8]. The true cortex is relatively richer in nuclei, poorer in cytoplasm and less vascular. The cells are arranged in arcades beneath the fibrous capsule of the gland and resemble those of the adult glomerular zone. These arcades are continuous centrally with columns of cells which are similar to those of the adult fascicular zone, but much shorter. Stoner et al. [9] claimed that differentiation of the adult cortex into zona glomerulosa and fasciculata first appeared between the second and fourth week of fetal life. The adult type reticular zone is absent at birth according to most observers, but develops postnatally [10].

The fetal cortex undergoes a number of changes at birth. The fetal zone degenerates, while the adult zone proliferates and starts to differentiate into three adult zones.

We are proposing a mechanism for the involution of the fetal adrenal cortex. The fetal cortical cells may have surface receptors for placental progesterone or estrogens. The hormones need to be continually replaced. As long as the hormones are bound to the cell surface, the T cells cannot recognize that they are foreign cells; the recognition site is masked. After the placental hormones are withdrawn the recognize them as foreign cells and produce cytotoxic factors which eventually lead to the death of the fetal cortical cells.

We studied the mode of involution of human fetal adrenal cortex in post-uterine life. From the preceding brief review, it appears that the process is due to either a withdrawal of the placental hormones or appearance of a rejection phenomenon of an auto-immune nature. It is proposed in this investigation to study the latter aspect, namely, the non-recognition of a fetal autograft in postnatal life. Therefore, the present investigations are designed to study whether the involution of the fetal adrenal cortex is similar to a graft rejection phenomenon or not. This work was conducted at the Institute of Post Graduate Medical Education and Research, Calcutta, from 1977 onwards.

The study includes the following aspects-

- 1. Histological observations on the adrenal glands of human fetuses of 7–36 weeks of gestation and on that of infants and adults.
- 2. The developmental pattern of soluble proteins of adrenal glands in human fetuses.
- 3. The possibility of detecting humoral antibodies in the serum of infants against extracts of fetal adrenal cortex.
- 4. The nature of antigen and antibody reaction in the rejection phenomenon.

Collections and Samples – Normal human fetuses, obtained by hystecotomy, after obtaining due consent as per the norms of the institutional Ethical Committee of the time, were received directly on ice. The adrenals were taken out by dissecting the abdomen of the fetuses. Adrenals were also dissected from children and adults who died of infectious diseases in which adrenals were not affected, and also of accident victims. Blood was collected from the fetuses, infants aged between 6 months to 1 year and adults. Rabbits were used for immunization.

Histological Study

Although the principal object was to study the involution of fetal cortical cells, we studied the development of the adrenal gland as a whole, during the development of the fetus as a part of our study. At a very early stage of development, at about 3 weeks of gestation the presumptive fetal adrenal cortex first appears as a condensation of celomic epithelium at the root of the dorsal mesentery. The islands rapidly proliferate; however all the cells look alike and there is no evidence of any differentiation at this stage. Thereafter, according to Keene and Hewer [5] and Uotila [6], there appear a rim of presumptive permanent fetal zone, which develops separately and then forms a cap around the fetal zone to form ultimately the definitive fetal cortex together. According to others like Gruenwald [11] and Velican [12], the

same primitive fetal cortex differentiates into the adult and fetal zones. Bachmann [13] and Lanman [14] have reviewed these morphological aspects.

At About 8 Weeks of Gestation

Figures 10.1 and 10.2 show the histological studies of adrenal gland of the fetus weighed at 1.5 g and was about 8 weeks gestation; it was found that there was a layer of presumptive capsule forming cells, 1 or 2 cells thick, followed by a layer of presumptive adult cortex, 7–8 cells thick, followed by the fetal type of cells 40–45 cells



Fig. 10.1 Histological picture of the adrenal gland of a fetus of 8 weeks: the capsular layer, the presumptive adult layer and the fetal layer $\times 100$



Fig. 10.2 Histological picture of the adrenal gland of a fetus of 8 weeks showing the fetal layer and the islands of blood forming tissue $\times 400$

thick, followed by the 7–8 cells thick adultogenic, and 1–2 cells thick capsulogenic cells. At a number of places in the fetal cortical layer as well as in the adult cortical layer there were islands of blood forming cells without any attempt at capillary formation. The adultogenic layer consists of cells with small and large nuclei and rather smaller cytoplasm as compared to the fetal layer. The fetal layer consists of cells which are larger than the adultogenic layer; the cytoplasm is more in amount and stains more intensely with eosin than the cells of the adultogenic layer. The blood forming cells stain the most with eosin.

At About 9 Weeks of Gestation

The histology of the adrenal gland of a fetus weighing 6.5 g and of about 9 weeks of gestation is shown in Figs. 10.3 and 10.4. From within inwards and then outwards the following layers can be distinguished, i.e., the capsulogenic layer is two to three cells thick and are is more elongated; at places the capsular fibrus connective tissue has been laid down. The adultogenic layer has increased in thickness, i.e., 13-14 cells thick and the fetal layer has also increased in thickness. The blood islands are more in number; some are not enclosed in capillaries. A small number of cells in the blood islands are mature erythrocytes, i.e., non-nucleated cells in contrast to that of the previous fetus (Fig. 10.1) where all the cells were nucleated.

At About 12 Weeks of Gestation

The histology of the adrenal gland of a fetus weighing 14.5 g, at about 12 weeks of gestation is shown in Figs. 10.5 and 10.6. The haematoxylineosin (HE) staining shows that the capsule is now well formed. The adultogenic layer consist of cells which are smaller than the fetal cortical cells; the adultogenic layer is very much less vascularized than the fetal layer, which is well vascularized; the fetal layer of cells contains blood in between two rows of cells; the blood contains non-nucleated erythrocytes. Occasionally, within the gland, a very large cell is seen, which for want



Fig. 10.3 Histological picture of the adrenal gland of a fetus weighing 6.5 g and of 9 weeks: two layers of capsular layer, a small adultogenic zone and a large fetal zone; small islands of blood forming cells are seen, \times 100



Fig. 10.4 Histological picture of the adrenal gland of a fetus weighing 6.5 g and of 9 weeks of gestation showing fetal zone containg a blood forming island which has few mature erythrocytes, a few erythroblasts and a large number of myeloid cells; $\times 400$

of terms, I call a "Titan Cell". Its length and diameter are at least three times that of the fetal cortical cells, shown in Fig. 10.7. The capsule contains amylase resistant PAS positive material.



Fig. 10.5 Histological picture of the adrenal gland of a fetus weighing 14.5 g; note the large number of blood forming islands, $\times 100$



Fig. 10.6 High power view of the blood forming island of the fetal adrenal gland (14.5 g); note the erythrocytes in the sinusoids between rows of fetal cells. The blood forming islands are most developed in this period; $\times 400$

At About 13 Weeks of Gestation

The histology of the adrenal gland of a fetus weighing 25.4 g and belonging to 13 weeks gestation is shown in Figs. 10.8 and 10.9. The capsule is well formed, contains three layers of cells. The adultogenic layer is of the same type as shown in the previous fetus. The fetal cells are big; towards the center of the fetal cells, the spaces between the cords of cells are occupied by sinusoids containing blood. The blood islands referred to previously are also there; they predominantly appear to be of the nature of a lymphoid follicle, naked intensely staining nuclei without much ctyoplasm; it is hard to be definite about the character of these cells.



Fig. 10.7 A 'Titan cell' in the adrenal cortex; note the large size of the cell in comparison to others; $\times 1000$



Fig. 10.8 Histological picture of the adrenal gland of a fetus weighing 25.4 g; note the indentation produced by a transmigrating medulla into the adrenal cortex pushing the adultogenic layer in front; $\times 100$



Fig. 10.9 Histological picture of the adrenal gland of a fetus weighing 25.4 g showing a 'Titan cell'; × 100

The most striking thing about the histology of this fetus was the demonstrability of the infiltration of medullary chords from the outside through the capsule, through the adultogenic layer, and through the fetal layer towards the center of the gland. It appears that the medullary cells also form sort of islands which approach the capsule of the cortex and either the islands split the capsule by secreting hyaluronidase or the capsule sort of envelops the medullary islands, which then assume chord like structure and start to transmigrate.

At About 17 Weeks of Gestation

The histology of a fetus weighing 135 g and of 17 weeks gestation is shown in Figs. 10.10, 10.11, 10.12, and 10.13. The capsule is better formed than in the previous adrenal gland of 13 weeks. At places the connective tissue is more prominent and at places the cells are more prominent. What absorbs one's attention

is the transmigration of the adrenal medulla. It appears as if the capsule splits into two layers to receive medullary tissue in its midst or,



Fig. 10.11 Histological picture of the adrenal gland of a fetus weighing 135 g showing the details of the transmigration of the medulla; the rounded mass lengthens into strands and layers of medullary tissue going right through the adultogenic layer and the fetal cortex towards the centre of the gland; $\times 100$



Fig. 10.10 Histological picture of the adrenal gland of a fetus weighing 135 g; the medulla is a rounded mass and is in the process of transmigrating into the cortex; the capsule splits to receive the medulla and the adultogenic layer is pushed inwards; $\times 100$



Fig. 10.12 Histological picture of the adrenal gland of 135 g in another area showing the ingrowth of a capillary in the wake of migrating medulla; × 100



Fig. 10.13 A high power view of the capillary shown in Fig. 10.12

when the latter approaches the capsule, the capsule layer envelops and spreads a layer over it. The medullary tissue then produces an indentation on the inner layer of the capsulogenic mass, where the capsulogenic cells show signs of granulation (in PAS stained material). The adultogenic layer is bodily pushed aside by the transmigrating tissue (the original medullary tissue as it approaches the adrenal gland, begging to be included in the gland, has a capsule of its own. But as it penetrates the capsule and as it pushes the adultogenic layer, the capsule becomes disintegrated and the naked cells transmigrate). The capillaries follow in the wake of the path laid down by the adrenal medulla to go towards the center. At places they go by themselves without the help of the medullary tissue. At some places the actual capillaries have been cut in longitudinal section passing through the capsule and adultogenic layer into the fetal layer. As in the other glands, the adultogenic layer contains a few Titan cells. The fetal layer consists of large polyhedral cells, almost looking at you through microscope and begging you to unravel their mystery. The histology at this stage is a real beauty to see and to appreciate. The sinusoids are there containing blood. The islands of cells looking like lymphoid follicles are also there.

At About 19 Weeks of Gestation

The histological picture of the adrenal gland of a fetus of about 19 weeks gestation and weighing 240 g is shown in Fig. 10.14. The histological picture is a continuation of the process seen in the previous fetus as shown in Figs. 10.10 and 10.11. The capsule is at places permeated by blood vessels and the medulla, going to be incorporated in the substances of the adrenal gland. The adultogenic and fetal layers show the same general characteristics as shown in the illustrations of the previous fetus. The medullary tissue which has already gone in, forms a generally disorganized array of cells in the centre of the gland, and surrounded by fetal cortical cells.



Fig. 10.14 Histological picture of the adrenal gland of a fetus of 19 weeks gestation, weighing 240 g: note the well formed capsules; the adultogenic layer is the same as in the earlier fetuses; the fetal cortex is well formed; the distinct sinusoidal lining contains erythrocytes, especially towards the centre of the gland; \times 100



Fig. 10.15 Histological picture of the adrenal gland of a fetus weighing 635 g showing increased activity in the adultogenic layer and sinusoids in the adultogenic layer; × 100

At About 24 Weeks of Gestation

The histological picture of the adrenal gland of a fetus of about 24 weeks of gestation and weighing 635 g is shown in Fig. 10.15. The capsule, in some areas, consists of a few layers of connective tissue enclosing a large amount of loose areolar tissue. Included in the aredolar tissue are numerous blood vessels, arteries, veins and capillaries cut in cross sections. The incursion of medullary tissue is almost complete, i.e., in only one place in the section was an island of medullary tissue seen lying just outside the capsule.

The adultogenic layer is proliferating, and sinusoids are developing in this layer now. There are attempts at differentiation of this layer into a zona glomerulosa and zona fascicuata. The fetal cells are still good looking polyhedral cells containing sinusoids in between the chords. The medullary tissue is still a disorganized mass lying strewn in chord like structures within the fetal layer of cortical cells.

At About 28 Weeks of Gestation

The histology of the adrenal gland of a fetus weighing 1010 g and belonging to a gestation period of about 28 weeks is shown in Figs. 10.16 and 10.17. The capsule consists predominantly of fibrous tissue and contains in its midst a number of vascular channels, arteries, veins and capillaries. The adultogenic layer consists of incompletely formed zona glomerulosa and attempted formation of zona fasciculate. Strands of fibrous tissue dip down from the capsule separating the adultogenic layers into sort of lobules. The fetal cells still look pretty, healthy and well vascularized: the sinusoids are full of blood. Some wide vascular channels traverse the substance of the gland at places. The medulla could not be identified.

At About 32 Weeks of Gestation

Histology of the adrenal gland of a fetus weighing 1650 g and of about 32 weeks of gestation is shown in Fig. 10.18. The capsule is now fully differentiated. The adultogenic layer is similar to the layer in the earlier fetus shown in Figs. 10.12 and 10.13. The biggest change that has occurred is in the fetal layer. The cells no longer look that pretty; they have assumed a ground glass appearance, the nuclei have started to be pyknotic. Near the centre of the gland small lakes are forming in which amidst a large number of capillaries the fetal cells have started to degenerate. The degeneration is more severe in the centre than towards the peripaery.



Fig. 10.16 Histological picture of the adrenal gland of a fetus, 28 weeks of gestation, weighing 1010 g showing increased activity in the adultogenic layer and necrosis of the fetal zone towards the centre of the gland; \times 100



Fig. 10.17 Histological picture of the adrenal gland shown in Fig. 10.16, showing strands of fibres dipping down from the capsule towards the centre of the gland; $\times 400$

At About 2 Days of Age

The histology of the adrenal gland of a child weighing 2500 g and who died of respiratory distress on the second day after birth is shown in



Fig. 10.18 Histological picture of the adrenal gland of a fetus weighing 1650 g showing vacuoles inside the fetal cortex; $\times 100$



Fig. 10.19 Histological picture of the adrenal gland of a child of 2 years of age: note the tremendous proliferation of the adultogenic layer and the hyaline regeneration of the fetal layer; $\times 100$

Fig. 10.19. The adrenal gland is still of the fetal type. The fetal layer is preparing to be degenerated and involuted. At places, especially towards the center of the gland there are a number of open spaces, possibly made by degeneration and disappearance of groups of fetal cells. The fetal cells which are remaining look sick.

At About 7 Days of Age

The histology of the adrenal gland of a premature child weighing 1100 g and who died 7 days after birth of cerebral hemorrhage (found in post mortem) showed that the gland was still a fetal gland. The adultogenic layer merged imperceptibly into the fetal layer which showed all signs of degeneration. Small islands were present where all the cells had autolyzed and were being removed. At places groups of cells had disappeared leaving tiny open spaces. But the whole process takes place so smoothly and efficiently that one cannot but wonder at the engineering feats of mother nature.

Fractionation of Soluble Proteins in the Adrenal Glands of Fetuses

When suprarenal glands of fetuses of different gestation periods were homogenized in buffered saline and centrifuged at $2000 \times g$ for 15 min, the supernatants were found to contain proteins at a lower concentration in the fetuses of earlier gestation periods than in the fetuses of later gestation periods. One hundred microgram protein equivalent of these supernatants were separated by polyacrylamide gel electrophoresis and stained with amidoschwartz. From 10 to 15 different kinds of protein could be identified in the extracts of these fetuses; the pattern changed form one period of gestation to another; the change was quantitative as well as qualitative; i.e., new proteins appear, some old proteins disappear, others increase in intensity and some decrease. Their relationships to the functional differentiation of the gland as a whole are too complex to understand at this state of our knowledge but a start has been made. It is probable that some of these changes are associated with distinct functions like appearance of the different hydroxylases or in the later fetuses of the lysosmal enzyme. This change in the pattern of the soluble proteins continues through early postnatal life to adulthood.

Figure 10.20 shows the electrophoretic pattern of the soluble proteins of the adrenal cortex in fetuses weighing from 7 to 17 g, having gestation periods of 9–12 weeks. The line drawing of the pattern of one of these fetuses is shown in Fig. 10.21. The pattern was identical in these fetuses. Altogether eleven bands could be distinctly identified; they have been numbered form above downwards. The electrophoresis was run in such a way that the tubes were taken off just as the



Fig. 10.20 The electrophoretic patterns of soluble proteins of the adrenal cortex in fetuses weighing from 7 g to 14 g, gestation periods 9 to 12 weeks

SOLUBLE PROTEIN PATTERN OF FETAL ADRENAL GLAND



Fig. 10.21 Line drawing of soluble protein patterns of a fetus weighing 7.07 g, having gestation period of about 9 weeks approximately

bromophenol blue dye band reached the lowermost point of the gel, so that the total length of the gel corresponded to the actual length of the column. The position of each band was then measured with a slide caliper and the distance of the band from the junction of the large and small pore gels divided by the total length of the gel was expressed as Rfs. Identical proteins in these bands were assumed to have the same Rfs. Figures 10.20 and 10.21 show that band No. 6 stained the most, followed by band 1 and 9 in decreasing order.

Figure 10.22 shows the electrophoretic pattern of soluble proteins of the adrenal cortex in fetuses weighing from 24 to 48 g, having gestation periods of 12–13 weeks. Altogether 15 bands could be observed. If we compare the pattern with that of the earlier stage, a profound change in pattern



Fig. 10.22 The electrophoretic patterns of soluble proteins of the adrenal cortex in fetuses weighing from 24 to 48 g, gestation period 12–13 weeks; the pattern is compared to the fetus in Fig. 10.21

SOLUBLE PROTEIN PATTERN OF FETAL ADRENAL GLAND



Fig. 10.23 Line drawing of soluble protein patterns of a fetus weighing 24.3 g, gestation period 12.6 weeks approximately

is certainly evident, i.e., 4 extra protein bands were present and the intensity of staining in band I increased and band 6 decreased. The pattern is compared to that of a the fetus shown in Fig. 10.23.

It can be speculated that one of these extra proteins can be the 21 hydroxylase which converts progesterone to deoxycorticosterone. The other three bands could be the other hydroxylases or some special proteins derived from the differentiating capsule or from the transmigrating medulla.

Figure 10.24 shows the electropolytic pattern of the soluble protein of the adrenal cortices in



Fig. 10.24 The electrophoretic patterns of soluble proteins of the adrenal cortex in fetuses weighing 135–280 g, gestation period between 16 to 19 weeks

SOLUBLE PROTEIN PATTERN OF FETAL ADRENAL GLAND



Fig. 10.25 Line drawing of soluble protein patterns of a fetus weighing 177 g, gestation period approximately 17.2 weeks

fetuses weighing from 135 to 280 g, having gestation period of 16–19 weeks. The line drawings of the same are given in Figs. 10.25, 10.26, and 10.27. The pattern shown in the fetus is more or less the same as the fetus shown in Fig. 10.22. But the pattern changes in the next fetus differing from it by only a few grams. In the pattern of soluble proteins, only 10 bands are discernible in contrast to 15 bands in the fetuses shown earlier. Four bands stain strongly 1, 5, 6 and 9, in contrast to 1, 6 and 9 in the fetus of later gestational age.

The banding patterns in the next three fetuses studied as exemplified in Fig. 10.28, shows more or less uniformity but the pattern again showed some change in fetuses that belonged to 20–24 weeks of gestation.

SOLUBLE PROTEIN PATTERN OF FETAL ADRENAL GLAND



Fig. 10.26 Line drawing of soluble protein patterns of a fetus weighing 199 g, gestation period approximately 17.4 weeks

SOLUBLE PROTEIN PATTERN OF FETAL ADRENAL GLAND



Fig. 10.27 Line drawing of soluble protein patterns of a fetus weighing 280 g, gestation period approximately 19 weeks



Fig. 10.28 The electrophoretic patterns of soluble proteins of the adrenal cortex in fetuses weighing 330 to 590 g, gestation period 20 to 24 weeks



Fig. 10.29 The electrophoretic patterns of soluble proteins of the adrenal cortex in fetuses weighing 530 to 1010 g, gestation period 24 to 28 weeks

The banding patterns again changed in fetuses of 24 to 28 weeks gestation (Fig. 10.29).

The soluble protein pattern in postnatal life is exemplified in Figs. 10.30 and 10.31. The first column in the figure is from a baby who lived for 2 days and died of respiratory distress syndrome and the second column is from an adult person who died of a street accident. The number of bands in the two columns are the same but the individual staining intensity differed in the two cases. Altogether 24 bands were discernible in both of them. The most prominent difference between the two columns was with reference to bands 22 and 23. Whereas in the 2 day old baby the protein stained quite deeply, the staining was not pronounced in the case of the adult.

Fetal Specific Adrenal Protein

The change in the protein pattern of the developing fetal adrenal gland and comparing it with that of the adult adrenal gland suggested that the fetal gland probably contained some components which were not present in the adult gland and vice versa. So attempts were made to identify such proteins by immunological methods. The



Fig. 10.30 The electrophoretic patterns of soluble proteins of the adrenal gland in post-natal life

SOLUBLE PROTEIN PATTERN OF ADULT ADRENAL GLAND



Fig. 10.31 Line drawing of soluble protein patterns of adult adrenal glands

fetal and adult cortical extracts were injected into rabbits in order to raise antibodies against the proteins present in the respective extracts.

Production of Rabbit Antibodies against Human Fetal and Adult Adrenal Cortical Extracts

Antisera were prepared by immunizing rabbits against (a) the soluble components, i.e., phosphate



Fig. 10.32 The reaction pattern of antiserum to the phosphate buffered saline extract of fetal adrenal cortex (central well) with the phosphate buffered saline extract of the fetal adrenal gland (peripheral well) in Ouchterlony system



Fig. 10.33 The reaction pattern of antiserum to the phosphate buffered saline extract of fetal adrenal cortex (central well) with the phosphate buffered saline extract of the fetal adrenal gland (peripheral wells No. 2) and adult (peripheral wells No. 1) adrenal gland

buffered saline extracts of adrenal glands of both the fetus and adult and (b) membrane components, i.e., 1 M NaCl extracts of adrenal glands of both the fetus and adult, from which the soluble components were pre-extracted.

The extracts were injected along with equal volumes of Freund's adjuvants in order to have the maximum number of specific antibodies formed against the maximum number of proteins present in the extract. Not all proteins are equally immunogenic and fetal proteins are perhaps less immunogenic than adult proteins. Since a protein has to possess a structure, different from that of the corresponding protein of the host in order to behave as an antigen, many proteins of the human adrenal extract may possess structure identical with that of the rabbit and therefore will not behave as antigens. Even with these limitations in view the different antisera were compared with respect to their reactions with antigen chiefly by Ouchterlony technique [15].

Rabbit Antibodies to Fetal Adrenal Cortex (Phosphate Buffered Saline)

In the central well was placed the rabbit antibodies to the phosphate buffered saline extract of fetal adrenal cortex and in the peripheral wells, the phosphate buffered saline extract of the fetal adrenal gland, which was used to immunize the rabbits. Results are shown in Fig. 10.32. About five or six bands were discernible in 48 h as a result of the interaction, about two thirds of the distance away from the rim of the central hole.

When such antibodies were reacted alternately with phosphate buffered saline (PBS) extract of the fetal and adult adrenal glands, the reaction pattern was quite different in the two cases, as shown in Fig. 10.33. The reaction was read at the end of 24 h. The peripheral holes marked 1 contained adult PBS extract and those marked 2 contained fetal PBS extract. The reaction pattern with fetal PBS extract, of course, was similar to Fig. 10.32. About five bands could be observed, which, however, were closer to the central well than to the peripheral well, because the reaction was read at 24 h. The reaction pattern observed with adult PBS extract as shown in peripheral wells marked 1 was entirely different form that in wells numbered 2. Firstly, the reaction was much fainter; secondly, a number of bands were discernible which were identical with bands of wells numbered 2 but the bands travelled more towards the peripheral hole than the bands in wells numbered 2; thirdly a faint reaction could be observed with another set of antigens in the adult PBS extract.



Fig. 10.34 The reaction pattern of antiserum to 1M NaCl extract of fetal adrenal gland (central well) with 1M NaCl extract of fetal adrenal glands (peripheral wells)



Fig. 10.35 The reaction pattern of antiserum to 1M NaCl extract of fetal adrenal gland (central well) with 1M NaCl extract of fetal and adult adrenal cortices; peripheral well No.1 contained fetal 1M NaCl extract and peripheral well No.2 contained adult 1M NaCl extract

Antibodies to Fetal 1M Extract of Adrenals

Rabbit antibodies to 1M NaCl extracts of fetal adrenal glands were placed in the central well and were allowed to react with 1M NaCl extract of the fetal adrenal gland which was used to immunize the rabbits. The pattern of reactions is shown in Fig. 10.34. Two groups of antigens were discernible, one group having smaller molecular weights on the average than the other. The group having the higher molecular weights was present in a greater concentration than the group having the smaller molecular weights.

When such antibodies were reacted alternately with 1M NaCl extracts of fetal and adult adrenal cortices, the reaction pattern was again different (Fig. 10.35). Peripheral wells marked 1 contained 1M NaCl extract of the fetal adrenal gland and those marked 2 had 1M NaCl extract of the adult adrenal gland. The reaction in the holes marked 1 showed identical pattern as shown in Fig. 10.34, i.e., two sets of proteins were discernible; but in the holes marked 2, only one set of proteins was discernible with showed reactions of identity with the group of antigens of fetal 1M extracts having the higher molecular weights. No proteins were found in the adult 1M adrenal extract corresponding to the second set of proteins of the fetal 1M adrenal extract.



Fig. 10.36 The reaction pattern of antiserum to PBS extracts of adult adrenal gland (central well) with same PBS extracts of adrenal gland (peripheral wells)

Antibodies to PBS Extracts of Adult Adrenal Glands

Rabbit antibodies to PBS extracts of adult adrenal glands were placed in the central hole and were reacted with the same PBS extracts of adrenal glands which were used as the immunogen. Results are shown in Fig. 10.36. Two broad groups of protein were discernible in the reaction pattern, separated from each other by a clear zone. These two groups of protein had again differing molecular weights, one group having a group average molecular weight more than the other.

When such antibodies were reacted alternately with PBS extracts of fetal and adult adrenal



Fig. 10.37 Same antiserum reacted alternately with PBS extracts of fetal (Hole No. 1) and adult (Hole No. 2) adrenal glands

glands, results are shown in Fig. 10.37. Wells marked 1 had extract of fetal glands and two contained extracts of adult glands. The reaction with the fetal extracts was much fainter than the reaction with adult extracts. Only one group of proteins were discernible, this group giving reactions of identity with the group in the adult cortical extract, the inner group.

The reaction with the adult PBS extract, however, revealed two seats of bands as in Fig. 10.36; the external set was specific for the adult PBS extract in the sense that no corresponding bands could be found in the fetal PBS extract. The inner set of bands, as stated previously showed reactions of identity with the fetal PBS extract; however it stained more intensely than the corresponding bands of the fetal PBS extract. It appeared as if the inner set with the adult PBS extract contained more bands than with the fetal PBS extract but since the corners were well rounded it appears that these sets of bands were identical.

Antibodies to 1M NaCl Extracts of Adult Adrenal Glands

Rabbit antibodies to 1M NaCl extracts of adult adrenal glands were placed in the central well and the corresponding antigen was placed in the peripheral wells. Results are shown in Fig. 10.38. Two sets of bands were discernible one external and the other internal, differing in average molecular weights with a gap in between.



Fig. 10.38 The reaction pattern of antiserum to 1M NaCl extracts of adult gland (central well) with corresponding antigen (peripheral wells)



Fig. 10.39 Same antiserum reacted with 1M NaCl extracts of fetal (Hole No. 1) and adult (Hole No. 2) adrenal gland

When such antibodies were reacted with 1M NaCl extracts of fetal (hole 1) and adult (hole 2) adrenal glands, the reactions were as shown in Fig. 10.39. Two sets of bands were observed in each case and they were identical.

Antibodies to PBS Extracts of Fetal Adrenal Glands

When rabbit antibodies to PBS extracts of fetal adrenal glands were placed in the central hole and reacted with 1M NaCl extracts of either fetal (hole 1) or adult (hole 2) adrenal glands, results are shown in Fig. 10.40. Almost no precipitating bands were detectable in either case. This shows that PBS extracted antigens were different from


Fig. 10.40 The reaction pattern of antiserum to PBS extracts of fetal adrenal gland (central well) with 1M NaCl extracts of fetal (hole 1) and adult (hole 2) adrenal glands



Fig. 10.42 The reaction pattern of antiserum to PBS extracts of adult adrenal gland (central well) with 1M NaCl extracts of fetal adrenal gland (peripheral wells)



Fig. 10.41 The reaction pattern of antiserum to PBS extracts of adult adrenal gland (central well) with 1M NaCl extracts of adult adrenal gland (peripheral holes)

the 1M NaCl extracted antigens (compare Fig. 10.40 with Fig. 10.33.)

Antibodies To PBS Extract of Adult Adrenal Gland

Rabbit antibodies to PBS extract of adult adrenal glands were placed in the central well and reacted with 1M NaCl extract of adult adrenal glands in peripheral wells. The reaction is shown in Fig. 10.41. Two groups of antigens were discernible. It appears, therefore, that PBS extracted antigens in the case of adult glands had common antigens with the 1M NaCl extracted antigens. When such antibodies were reacted with 1M NaCl extracts of fetal glands, results were



Fig. 10.43 The reaction pattern of antiserum to 1M NaCl extracts of adult adrenal glands (central hole) with PBS extracts of adult (hole 1) and fetal adrenal (hole 2) adrenal glands. Near Hole 1, arrow indicates the antigen antibody precipitin band

obtained as in Fig. 10.42. The results are identical with the results of Fig. 10.41.

Antibodies to One M Nacl Extracts of Adult Adrenal Glands

When rabbit antibodies to 1M NaCl extracts of adult adrenal glands were extracted with PBS extracts of adult (hole 1) and fetal (hole 2) adrenal glands results are shown in Fig. 10.43. With adult PBS extracts (hole 1), three sets of bands were discernible. The photographic reproduction is poor. But in the actual plate two sets of antigen antibody reactions were seen very close to the antigen hole (shown by the arrow); these were



Fig. 10.44 The reaction pattern of antiserum to 1M NaCl extracts of fetal adrenal gland (central hole) with 1M NaCl extracts of fetal (peripheral well no. 1) and adult (peripheral well no. 2) adrenal cortices

not present near hole, 2 i.e., with fetal PBS extracts. The third set of bands was near the central hole and consisted of a broad band, not much well differentiated. Near hole 2, i.e., with fetal PBS extracts, two sets of bands were observed; the inner set was identical with the innermost bands of the adult PBS extracts but the second set of bands was characteristic of fetal PBS extracts.

Characterization of Fetal 1M Antigens

Figure 10.44 shows the reaction of rabbit antibodies against fetal 1M antigens with fetal (hole 1) and adult (hole 2) 1M antigens. As noted before, the reaction with fetal 1M extract showed two sets of bands, one of which was absent in the reaction with corresponding adult antigen in hole 2. One set of antigens was, therefore, common to fetal and adult 1M NaCl extracts but one set of antigens were characteristic of fetal 1M NaCl extracts. It was, therefore, considered worthwhile to study if such fetal specific antibodies could be characterized. Accordingly 1 ml of the rabbit antibodies was incubated with 0.1 ml of the adult 1M NaCl extract at 4 °C for 18 h after which the mixture was centrifuged; the supernatant was again treated with 0.1 ml of the adult 1M NaCl extract for 18 h at 4 °C and recentrifuged; the process was repeated for two more times; at the end of which time the antibodies did not show any reaction when reacted with 1M NaCl extract of adult adrenal glands as shown in Fig. 10.45.



Fig. 10.45 The reaction pattern of the supernatant (central well) from the precipitated antibody with the 1M NaCl extract of adult adrenal gland (peripheral wells)



Fig. 10.46 The reaction pattern of the same supernatant (central well) with 1M NaCl extracts of fetal adrenal gland (peripheral wells)

The central well contained the supernatant from the precipitated rabbit antibody and the peripheral wells contained the 1M NaCl extract of adult adrenal glands. Although the rabbit antibodies by such a treatment did not react with 1M NaCl extracts of adult adrenal gland they still showed reaction with 1M NaCl extracts of fetal adrenal glands. Figure 10.46 shows the reaction pattern when such a supernatant was placed in the central well and the 1M NaCl extracts of fetal adrenal glands in the peripheral wells. It shows that 1M NaCl extracts of fetal adrenal glands contained some immunogenic components which were probably not present in the extracts from adult adrenal glands.

Presence of Autoantibodies in Postnatal Life against Fetal Adrenal Antigens

Ouchterlony System

If the fetal cortical cells possessed a soluble antigen, which was not present in the adult type of cortical cells, it may be possible that such antigen might give rise to humoral antibodies in postnatal life. It was presumed, of course, that in antenatal life, the synthesis of such antibodies was inhibited, possibly by placental hormones.

An infant, especially in the first few months of life, is active in the involution of fetal cortical cells. If the involution is due to complement mediated antigen antibody reaction, the serum of an infant is most likely to show the presence of such antibodies. The serum of a number of infants aged between 4 and 12 months was placed in the central well of an Ouchterlony system and fetal cortical extracts both with phosphate buffered saline and with 1M NaCl solution were placed in the peripheral wells. Even in a 1 week old infant, no precipitin band was found to develop. Figure 10.47 shows one such experiment. No antigen antibody precipitating band was demonstrable.

Quantitative Immunoprecipitation Test

The presence of certain specific components in the fetal adrenal extract in contradistinction to adult adrenal extract invited some studies on the humoral and cell mediated immune response of infant to the fetal cortical cells. No frank autoantibodies could be demonstrated in the serum of infants who were active in the involution of the fetal cortical cells, to the fetal adrenal extracts. In vitro study of quantitative Immuno Precipitation test was done by using infants serum at various dilution and either fetal cortical extract or a child's cortical extract. Results are shown in Table 10.1. When infant serum as different dilutions was mixed with fetal cortical extracts, measureable precipitate could be obtained up to a serum dilution of 1 in 10. Under identical conditions almost no precipitate was noted when the same sera were mixed with adrenal extract of children. Whether the precipitate obtained with the fetal cortical was real or not is a problem for



Fig. 10.47 The reaction pattern of infant serums (central well) with fetal cortical extracts both with phosphate buffered saline (peripheral hole no. 1) and with 1M NaCl solution (peripheral hole no. 2) in Ouchterlony system

 Table 10.1
 Precipitin reaction infant serum with adrenal extracts

Infant serum dilution	mean difference i.e.			Child extracts/protein mean difference i.e. ppt. (µg)		
Control	14.8	11		41.44	42	
1/5	34.60	47	36	61.45	53	11
1/10	28.46	37	26	33.16	24	0
1/20	7.11	9	0	44.31	37	0
1/40	16.26	21	0	40.40	40	0
1/50	13.14	13	0	38.44	41	0

discussion. As can be seen from the table, the protein content in the control tube with fetal cortical extract was less than that in the control tube with child's adrenal extract. It may be that the proteins of the fetal cortical extract were more soluble than those of the child's adrenal extract and did not precipitate so much as a result of incubation at 4 °C for 96 h. But the increase of precipitate when infant's serum was added to the fetal cortical extract was real and consistently found in four consecutive experiments, of which one is shown in Table 10.1.

Fifty μ l of saline extracts of the adrenal cortices from the fetus and a 1 year old child, containing 100 μ g protein was mixed with 50 μ l of the appropriate dilutions of the infant serum. The mixture was incubated at 4 °C for 96 h, and centrifuged; the precipitates were once washed with cold saline; protein was then estimated directly in

the tube. From the protein contents of the experimental tubes, that of the control tube were subtracted and results are expressed as the difference from the control.

Passive Hemagglutination Test

When sheep erythrocytes were either tanned or trypsinized and then coated with fetal and adult cortical PBS extract, it was found that the erythrocytes to which the fetal cortical extract was adsorbed, could be agglutinated by infant serum up to a dilution of 1 in 20, but not by fetal serum or adult serum. No hemagglutination was observed with erythrocytes coated with adult adrenal PBS extract. Results are shown in Table 10.2, against fetal serum, infant serum and adult serum dilutions the results were consistent in six experiments, one of which is reported in Table 10.2.

The results of passive hemagglutination test with both the fetal and adult adrenal extracts against different dilutions of infant, fetal and adult sera are given.

Passive Cutaneous Anaphyl Axis Test

We next tested the assumption that the infant in whom the involution of fetal adrenal cells is taking place, may manifest some cell mediated immune response to the fetal cortical cell, provided some immunological mechanism is at the basis of the involution. Accordingly fetal and adult adrenal extract (both saline and 1M NaCl extracts) were made up to the same protein concentration of 40 µg per 10 µl and then sterilized by passing through Millipore filter. Such extracts were sterilized by passing through millipore filter. Such extracts, however, when injected intradermelly into a number of infants failed to any measurable produce induration. An Arthustype reaction also could not be demonstrated with such adrenal extracts. The infants were injected with Even's blue 15 min prior to the intradermal injection of the adrenal extracts. No bluish discoloration could be observed at the site of injection (Fig. 10.48).

Migration Inhibition Factor of Sensitized Lymphocytes and Macrophages

Leucocytes were prepared form the blood of a child aged 10 months and who was tuberculin positive; they were cultured in Lucite Kennedy Chambers in the presence and absence of fetal cortical extracts and tuberculin. Results are shown in Fig. 10.49. In chamber Nos. 1 and 2, the leucocytes were cultured without any antigen. The area of leucocyte migration was measured by plainmetry and found to be 2.8 and 3.2 mm² respectively. In Chamber Nos. 3 and 4, the medium contained 6 µg of protein equivalent of fetal cortical extract per ml of medium, there was complete inhibition of migration. Since the child was tuberculin positive, Chamber Nos. 5 and 6 contained medium to which was added 1 µl of PPD. In this case too, there was complete inhibition.

Table 10.2	Passive hemagglutination test					
Reaction with fetal serum						

Reaction with fetal seru	ım							
Infant serum dilution	1/5	1/10	1/20	1/40	1/80	1/160	1/320	Control
Fetal Ext.	+	+	+	_	-	_	_	-
Child Ext.	-	-	-	-	-	-	-	-
Reaction with fetal seru	ım							
Fetal serum dilution	1/5	1/10	1/20	1/40	1/80	1/160	1/320	Control
Fetal Ext.	-	_	-	_	-	_	_	-
Child Ext.	-	_	-	-	-	-	-	-
Reaction with adult service	um							
Adult serum dilution	1/10	1/20	1/40	1/80	1/160	1/320	Control	
Fetal Ext.	_	_	-	-	_	-	-	
Child Ext.	_	_	_	_	_	_	_	



Fig. 10.48 Passive outaneous anaphylaxis reaction in infants, 48 hours after intradermal injection of extracts; 2 ml of 2 % Evan's blue solution (50 gm/kg) was injected intradermally prior to the administration of the extract: (a) PBS extracts of both the fetal and adult adrenal glands were injected intradermally on the left and right forearms

When leucocytes were prepared from an adult blood in the same way and then cultured in the presence and absence of adult and fetal cortical extracts, the migration observed is also depicted in Fig. 10.49. Whereas in the control Chamber Nos. 7 and 8, the area of leucocyte migration amounted to 3.0 and 2.5 mm² respectively, there was complete inhibition of migration in the presence of fetal cortical extract (Chamber Nos. 10 and 11) and tuberculin (Chamber No. 12) in the presence of adult cortical extract, however, the leucocyte migration was not completely inhibited (Chamber No. 9). The area of migration was found to be 1.6 mm by plainmetry.

There was complete inhibition of infant leucocyte migration with as little as $6 \mu g$ of protein equivalent form fetal cortical extract. Thus the leucocytes of infants may be presumed to be sen-



of the infant; (b) 1M NaCl extracts of both fetal and adult adrenal glands were injected intradermally on the left and right forearms of the infant. A control site injected with saline is also marked on both. No reaction can be seen in either case

sitized to the fetal cortical extract. What part it plays in the death of these cells and their involution we cannot say.

Conclusion

We have studied histologically the developmental pattern of human fetal adrenal cortex from early gestation period towards the end of gestation period (3–32 weeks of gestation), infants, children, adult. Histologically the human fetal adrenal cortex was different from adult adrenal cortex. The two kinds of cells are probably distinct to their derivation too in fetal adrenal cortex. On the periphery of glands deep stained, there are a few layers of cells called "Permanent zone" or "Adult zone" and the majority of the gland is



Fig. 10.49 Inhibition of sensitized lymphocytes migration. 1–6: lymphocytes from a 10 month old child who was tuberculin positive exhibiting sensitization to fetal adrenal cortical extract (1) and (2) without any antigen; the area of migration is 2.8 and 3.2 sq. mm; complete inhibition of migration has been exhibited in chamber 3. 4, the medium containing fetal cortical extract, and in chamber (5) and (6), the medium containing PPD. &-12:

occupied by the large polyhedral cells which are frequently fat containing and have no definite arrangement; this portion is called "Fetal zone". In the most central region of the gland a few deeply staining medullary cells may be seen. The involution of the fetal cortical cells, which starts in the last couple of weeks of gestation and continues through postnatal life is associated with death of the fetal cells. In adult adrenal gland no fetal zone is found and adult zone proliferates into three zones, i.e., Zona glomerulosa, Zona fasciculate and Zona reticilaris.

Besides their distinctive histological characters, can the fetal and adultogenic layer be distinguished by any other characteristics? The soluble proteins of the fetal adrenal glands of different

containing lymphocytes from an adult tuberculin-positive person; 7 and 8 – without any antigen: the area of migration is 3.0 and 2.5 sq. mm; (9): the area containing adult cortical extract: the area of migration is 1.6 sq. mm; complete inhibition of migration has been exhibited in (10) and (11) – the medium contained fetal cortical extract; (12) contained PPD

gestation period was studied by Polyacrylamide gel electrophoresis method and stained by amidoschwartz from 10 to 15 different kinds of protein could be identified in the extracts of the fetuses; [19] the pattern change from one period of gestation to another, Some old proteins disappear, new proteins appear, other increase in intensity and some decrease. The majority of the workers believe that the principal steroidogenic cells are the fetal, while some specialized function like formation of pregnenolone and 11- β - OH-androsteredion by the whole adrenal gland is carried on [16].

Besides their distinctive histological characters, can the fetal and adultogenic layer be distinguished by any other characteristics? Since the functional characteristics of the two kinds of cells also differ, can we identify any protein or other component present in the fetal layer of cells which is absent in the adult layer? Accordingly, we prepared saline and 1M NaCl extracts of fetal and adult adrenal glands. The saline extracts of course contain besides cytosol, the mitochondria, endoplasmic reticula, golgi bodies and other particulate matter. The 1M NaCl extracts contain nuceloprotiens, glycoprotenis, membraneous semisoluble proteins and the like.

These two kinds of extracts from the fetal and adult adrenal glands were injected into rabbits with Freund's adjuvants in order to raise antibodies against antigens present in the different extracts. These antibodies were, then, reacted against these extracts *in vitro* in Ouchterlony systems. When antibodies against saline extracts of fetal adrenal glands were reacted with saline extracts of fetal and adult adrenal glands, the reaction pattern showed that in addition to common antigens, the adult saline extract contained some antigens which were not present in the fetal saline extract.

When antibodies to 1M NaCl extracts of fetal glands were reacted with 1M NaCl extracts of fetal and adult adrenal glands, some components in the fetal and adult extracts reacted in common, the reaction pattern with the common antigens were reactions of identity. There was a separate set of antigens in the 1M NaCl extracts of fetal glands; these antigens were not present in the adult adrenal gland. The findings of some fetus specific component in the 1M NaCl extracts of fetal glands, therefore, assumes some importance.

When rabbit antibodies were raised against saline extracts of adult adrenal glands and such antisera were tested with saline extracts of adult and fetal adrenal glands, the antigen antibody reactions were more intense with adult extract than with fetal extract. Furthermore, a set of antigens were found in adult extracts which were not present in fetal extract. It can be inferred that either the adult saline extract does not have some common fetal antigens or the differentiation of the adult adrenal gland is associated with the appearance of a host of specific antigens. In the case of antibodies to 1M NaCl extracts of adult adrenal glands, however, the immunogenic components in the corresponding adult and fetal extracts were similar.

The antibodies to saline extracts of the fetal gland did not react with 1M NaCl extracts of either fetal or adult glands. So the components extracted with saline and 1M NaCl were antigenically distinct and specific.

The antibodies to saline extracts of the adult gland reacted with 1M NaCl extracts of the adult as well as fetal glands. In the case of adult glands, therefore, the saline extracted components had common antigens with 1M NaCl extracted antigens.

Rabbit antibodies against 1M NaCl extracts of the fetal glands showed one set of precipitating bands which was common to 1M NaCl extracts of both fetal and adult adrenal glands and one set of bands which was present with corresponding fetal extract only. The antibodies to the immunogens, which were common to 1M NaCl extracts of adult and fetal glands could be precipitated with 1M NaCl extracts of adult glands. Such serum no longer reacted with 1M NaCl extract of adult glands but still reacted with 1M NaCl extracts of fetal glands. The fetal glands, therefore, had perhaps, located on the cell membrane, some immunogenic components, which were unique to the fetal cells and not present in either the adultogenic layer of the fetal gland or the differentiated cells of the adult gland.

Can this unique fetal component on the membrane of the fetal cortical cell serve as a heterologous antigen, which the immunologic apparatus of the older fetus and infant recognize as a foreign component and so it proceeds to reject the cells containing the foreign components? So it was considered worthwhile to investigate whether the new born or infant serum contains some humoral antibodies to soluble extract of the fetal adrenal gland, or whether the lymphocytes or macrophages of the infant can produce cell mediated immune response against fetal adrenal component.

When untreated sera of new-borns and infants were reacted with either saline or 1M NaCl extracts of fetal adrenal glands in an Ouchterlony system, no precipitating band could be demonstrated. Even when γ -globulins were precipitated with salts and such γ -globulins were dialyzed and concentrated and then use the system, no humeral antibody could be detected.

The immunoprecipitin test as well as the passive hemagglutination test with coated tanned erythrocytes show the presence of some sort of humoral antibodies to the fetal cortical extract in infant serum, although the reaction in either case was not pronounced one. As a matter of fact, it was not expected that the reaction would be a very pronounced reaction considering that the involution of the fetal cortex takes place over a matter of months in a slow and gradual way, rather than like the involution that occurs in the uterus after parturition which can be regarded as an explosive event.

The humoral reaction, however, cannot by itself account for the ordered progression of death and involution of the fetal cortical cells. It was considered worthwhile to test of the presence of cell mediated immune response of infants to fetal cortical extract. However, when both kinds of extracts of fetal and adult adrenal glands were injected intradermally into the forearm of 5, 6 month old infants, no induration could be detected 2, 4, 24 and 48 h after the injection. No Arthus type of reaction was also demonstrable in as much as there was no discoloration at the site of intradermal injection when such infants were preinjected with Evan's blue (T 1824).

If we presume that the fetal membrane component of fetal cortical cells is unique to fetal cells, and that this component is recognized by the involuting individual as foreign to the organism, we can expect the immunocompetent cells to react to it. Such reaction may take various forms, such as migration inhibition factor, skin reactive factor, lymphotoxin, inhibitor of DNA synthesis, chemotactic factor, blastogenic factors etc. [17].

The leucocytes will be sensitized to the fetal component and show disapproval on coming in contact with fetal adrenal extract. One of the ways in which disapproval may be manifested is by secreting some migration inhibitory factor. Accordingly, we cultured leucocytes from infants and adults in capillary tubes and tested for the migration of leucocytes in the Kennedy chambers. There was complete inhibition of infant leucocyte migration with as little as $6 \mu g$ of protein equivalent from fetal cortical extract. Thus the leucocytes of infants may be presumed to be sensitized to the fetal cortical extract. What part it plays in the death of these cells and their involution we cannot say.

From our studies on human fetal adrenal cortex, we have to explain why does not a fetus of earlier gestation period recognize this component as foreign or else become tolerant to it. Our explanation is, as long as the placenta secretes enough estrogens and gonadotrophins, the hormones, probably have a high affinity for this fetal adrenal component and prevents recognition of the fetal cortical cells.

The steroid hormones are known to have high affinity for certain proteins and thus may change the structure of repressor (which is most probably a protein) allowing the operon to proceed with the synthesis of mRNA [18] for let us say a vital protein for the fetal cortical cell. When the placental hormones become inadequate, no such binding can occur and, therefore, the operon becomes inoperative and this vital protein is no longer synthesized thus leading to the death of the fetal cortical cell.

It might also be that this fetal cortical component when combined with placental hormones cannot be recognized as foreign cells by the immunologic apparatus of the fetus. When the placental hormones decrease in amount, there is no more binding, and the protein assumes a conformation which can be recognized as foreign component by the individual, who therefore, proceeds to deal with the foreign component by both humoral and cell mediated immune responses. The process is necessarily slow and takes place over a period of months instead of days. The fetal adrenal cells probably become subject to apoptosis and removed from the system.

It would have been ideal if we could have isolated the fetal component and studied its binding to the placental hormones. But it was not possible to do so.

Summary

- The human fetal adrenal gland is characterized by a smaller peripheral rim of adultogenic cells and a much larger layer of what are termed fetal cortical cells. During the greater period of gestation the fetal cells, which occupy the central four fifths of the gland appear to dominate the morphology of the gland. Starting at the last couple of weeks before parturition the fetal cells go through a process of involution by which the whole fetal layer of cells dies out and by and by the adultogenic cells proliferate to form the ultimate mature adrenal gland.
- 2. The histology of the adrenal gland was studied from about 7 weeks of gestation to a few days after birth. The gland arises from a mesothelial condensation on each side. The two zones of the human fetal adrenal glands can be distinguished by the structure of the cells and also by staining capacity. Beginning at about 12-16 weeks, the medullary tissue which descends from migratory cells that leave the primitive ganglia of the celiac plexus of the autonomic nervous system, approach the capsule, probably split it, produce an indentation in the adultogenic layer, push it, and gradually become incorporated into the substance. Originally when they were outside the gland they probably were rounded masses, but when they travel inside the adrenal gland, they transform themselves into chords of cells. At the end of the gestation period the pretty polyhedral, preserved cells of fetal zones begin to look sickly, the cytoplasm degenerates and undergo dissolution and gradually disappear. But the adult zone proliferates to three zones i.e. zona glomerulosa, zona fasciculota, zona retucularis.
- 3. The soluble proteins of the fetal adrenal glands of different gestation period was studied by Polyacrylamide gel electrophoresis method and stained by amidosehwartz from 10 to 15 different kinds of protein could be identified in the extracts of the fetuses [19];

the pattern change from one period of gestation to another; some old proteins disappear, new proteins appeared, other increase in intensity and some decrease.

- 4. To find out the fetal specific protein which is present only in fetal gland but not in adult adrenal gland and vice versa. Rabbits were immunized against the buffered saline extract of fetal and adult adrenal gland and 1M NaCl extract of fetal and adult adrenal glands. The antigen antibody reactions were studied in ouchterlony system where antisera are given in central hole and specific and non specific antigen on the peripheral holes. When antibodies against saline extracts of fetal adrenal glands were reacted with saline extracts of fetal and adult adrenal glands, the reaction pattern showed that in addition to the common antigens, the adult saline extracts contained some antigens which were not present in the fetal saline extract. When antibodies against 1M NaCl extracts (membrane fraction) of the fetal adrenal gland was reacted with 1M NaCl extracts of fetal and adult adrenal glands, it was found that 1M NaCl extract of the fetal gland contained some proteins not present in the corresponding extract of the adult gland. When the fetal extract was treated with 1M NaCl extract of the adult gland, the adult components were precipited. Such treated rabbit antisera no longer reacted with 1M NaCl extract of adult adrenal glands but still reacted with 1M NaCl extract of fetal adrenal glands. The fetal glands, therefore, contained some specific components probably on their membranes which were not present in the adult adrenal gland.
- 5. By immunoprecipitation tests with infant serum, however, a higher quantity of protein was found to precipitate (serum dilution 1 in 10) with fetal adrenal extract than with adrenal extract of children, which did not contain any fetal component.
- 6. When sheep erythrocytes were either coated with tannic acid or trypsinized and then coated with saline extract of fetal adrenal gland, they were found to be agglutinated by infant's

serum up to a dilution of 1 in 20, but not by fetal serum. When such erythrocytes were coated with saline extracts of children adrenal glands no agglutination was found to occur with infant serum. The hemagllutination observed with infant's serum could not be detected with adult serum.

- 7. The infant in whom the involution of fetal adrenal cells is taking place, may manifest some cell mediated immune response to the fetal cortical cell, provided some immunological mechanism is at the basis of the involution. However, when both kinds of extracts of fetal and adult adrenal glands were injected intradermally into the forearm of 5, 6 month old infants, no induration could be detected at 2, 4, 24 and 48 h after the injection. No Arthus type of reaction was also demonstrable in as much as there was no discoloration at the site of intradermal injection when such infants were preinjected with Evan's blue (T 1824).
- 8. When leucocytes were prepared form the infant blood and cultured in the Kennedy chambers in the presence and absence of fetal cortical extracts in the medium, the fetal cortical extract was found to inhibit completely the migration of leucocytes from the capillaries.
- 9. The involution of the fetal cortical cell was, therefore, thought to be due to recognition of some fetal specific component as a sort of foreign proteins and consequent immunologic response of the infant. The failure of recognition during intrauterine life was presumed to result from altered configuration of the same protein, possibly by being attached to placental hormones.

References

1. Elliot T, Armour R. The development of the cortex in the human suprarenal gland and its condition in hemicephaly. J Pathol Bacteriol. 1911;15:481–9.

- Bech K, Tygstrup I, Nerup J. The involution of the fetal adrenal cortex. Acta Pathol Microbiol Scand. 1969;76:391–400.
- 3. Swinyard CA. Growth of the human suprarenal glands. Anat Rec. 1943;87:141.
- Tahka H. On the weight and structure of the adrenal glands and the factors affecting them, in children of 0–2 years. Acta Paediatr. 1951;40(81):1–95.
- Keene MFL, Hewer EE. Observations on the development of the human suprarenal gland. J Anat Physiol. 1927;61:302.
- Uotila UU. The early embryological development of the fetal and permanent adrenal cortex in man. Anat Rec. 1940;76:183–95.
- Maximow AA. A text book of histology. Philadelphia: W.B. Saunders Company. Completed and edited by William Bloom; 1930.
- Benner MC. Studies on the involution of the fetal corntex of the adrenal gland. Am J Pathol. 1940;16:787.
- Stoner HB, Whiteby HJ, Emery JL. The effect of systemic disease on the adrenal cortex of the child. J Pathol Bacteriol. 1953;66:171–83.
- Blackman SS. Concerning the function and origin of the reticular zone of the adrenal cortex. Bull Jonhs Hopkins Hosp. 1946;78:180.
- Gruenwald P. Embryonic and postnatal development of the adrenal cortex, particularly the zona glomerulosa and accessory nodules. Anat Rec. 1946;95:391–422.
- 12. Velican C. La zone transitoire de la cortico-surrenale humaine. Arch Anat Micr (Paris). 1948;37.
- Bachmann R. Die Nebenniere. In Handbuch der mikroskopischen Antomie des Menschen. Ed. Bargman W., Vol. 6. Blutgeefass-und Lymphgefassapparat. Innersekretorische. Part 5, 1954, p. 1–952. Berlin: Springer-Verlag.
- Lanman JT. The fetal zone of the adrenal gland. Medicine. 1953;32:389–430.
- Ouchterlony O. Diffusion in gel methods for immunological analysis. Progress in Aergy, 5:1–78;6:30 (1962), (S. Karger, Basel); 1958.
- Bloch E, Benirschke K. Steroidogenic capacity of fetal adrenal *in vitro*. In: Currie AR, Symington T, Grant JK, editors. The human adrenal cortex. The Williams and Wilkins Company. Baltimore. p. 580–89.
- Bloom BR. In: Bloom BR, Glade PR, editors. In vitro methods in cellmediated immunity. New York: Academic; 1971. p. 4.
- Tomkins GM, Maxwell ES. Some aspects of steroid hormone action. Ann Rev Biochem. 1963;32:677.
- Ornstein L. Disc electrophoresis I. Background and theory. Ann N Y Acad Sci. 1964;121(2):321–49.

Development of the Human Fetal Brain

11

Bimal Samanta, Chameli Ganguly[†], Gitanjaly Guha Thakurata[†], K.L. Mukherjee[†], and Niranjan Bhattacharya

Introduction

Growth is a very important feature of development of a multicellular organism. It involves both cell multiplication [1] and making more intra and extracellular material. Both types of increases

B. Samanta, MSc, PhD (Cal) (🖂)

Central Calcutta Society for Advancement of Human Development and Research, Kolkata, India

Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

C. Ganguly, MSc, PhD Former Biochemist Central Calcutta Society for Advancement of Human Development and Research, Kolkata 700040, India

G.G. Thakurata, MSc, PhD (Cal) Formerly, Department of Biochemistry, National Medical College and Hospital, Kolkata 700040, India

K.L. Mukherjee, MB, PhD(Cal), PhD (Wisconsin) Former Head of the Department of Biochemistry, Institute of Post Graduate Medical Education and Research, Kolkata 700015, West Bengal, India

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India

e-mail: sanjuktaniranjan@gmail.com

involve the synthesis of DNA, RNA, Proteins, Lipids and other constituent materials of the cells. From the point of view of development, an important feature of growth is that it occurs to different extents in different parts of the developing embryo and different patterns of growth can continue for long periods [2].

The growth and maturation of an organ at the cellular level comprises four stages: (1) Proliferation, (2) Migration, (3) Differentiation and (4) Death. This sequence of events is observed in all parts of the brain; however, the timing differs from one cell type to another and from region to region and even within the brain there are remarkably complex differences in the rate of growth and differentiation between the various regions [3, 4]. This results in an extremely complex pattern of development when the brain is regarded as an organ.

The mechanisms involved in cell differentiation are almost certainly of a biochemical nature. Cells which have differentiated along a specific pathway are characterized among other things, by the spectrum of proteins they contain.

Proliferation

The formation of the brain, for example, in the rat, begins around the seventh gestational day, at about the time when implantation occurs in the uterus. Proliferation of cells occurs from the ventricular zone of the primitive brain regions around the 11th

© Springer International Publishing Switzerland 2016 N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development: First and Second Trimesters*, DOI 10.1007/978-3-319-14874-8_11

[†]Author was deceased at the time of publication.

day and proceeds very rapidly [5] producing an enormous number of cells in a few days. Malnutrition during early pregnancy can reduce the number of cells formed, as measured by the DNA content of the brain [6, 7]. Human fetal nutritional requirements for normal growth are inferred either from comparisons between maternal diet during pregnancy and the size or condition of the baby at birth or from animal experiments [8, 9]. Protein (essential amino acid) or glucose deprivation results in an underdeveloped fetal brain [10]. It is generally assumed that a nutritionally excellent diet consumed by the mother during pregnancy is good for the fetus, while a poor diet will embarrass fetal nutrition and growth [11, 12]. Studies in Guatemala among rural American women of uniformly low socioeconomic state indicate that supplemental protein and energy intake during pregnancy are associated with increased birth weights and decreased morbidity and mortality among newborns [13].

Migration

After an unknown number of mitoses of stem cells, each newly formed cell of the central nervous system migrates out of the "proliferative compartment" [5] and form layers of homogeneous cells:

- Different cell types separate from each other and cluster together.
- 2. Within the homogeneous cell population the cells align in parallel to each other. They show a basal-apical orientation.

The capacity of self-orientation is strictly limited to a short period during embryonic development.

Differentiation

During or following migration a second type of differentiation takes place: the formation of cellspecific structures. Neurons grow axons and dendrites and form synaptic interconnections. Glial cells differentiate into astroglia and oligodendroglia. The oligodendrocytes develop and differentiate their plasma membrane into myelin sheaths.

Cell formation and migration are accompanied by death of a certain number of newly formed cells [14]. Studied on the whole, the brain has shown that its rate of development is not linear but that at a certain moment suddenly increases dramatically. This period called "brain growth spurt period" [15] is characterized by the sudden formation of a great number of new cells (hyperplasic phase) which rapidly grow. The "growth spurt period" is associated with a rapidly increasing metabolic activity and the appearance of new functions. More detailed studies have shown that the time and duration of the "growth spurt period" differ not only from species to species [7, 16], but also from region to region and from cell type to cell type in the same species. This results in an extremely complex pattern of development for the whole brain of a given species.

Modification of hormone levels and of cellular environment appear to be important factors in CNS epigenesis. Malnutrition during development is one of the factors which appears to permanently impair CNS functioning particularly at the behavioral level and to provoke in humans permanent intellectual deficit.

The brain, the most complex organ of the body, takes long to mature. In some respects the brain is changing until the day we die; presumably every new memory that is formed depends on a modification of the functional organization of some part of the brain. However, the most rapid period of growth and anatomical change is during fetal and early postnatal life.

Dobbing and Sands [7] studied the growth and chemical development of the human brain. As in other parts of the body there is a fall in the percentage of water from about 92 % at 20–22 weeks gestation to 77 % in the adult. There is a simultaneous rise in the percentage of lipids in the form of myelin. Dobbing and Sands used cholesterol as an index of myelination since its accumulation bears a constant relation to cerebroside. There is a rapid rise in the concentration of cholesterol in the forebrain up to 4 years of age, followed by a small, gradual rise to adult values.

There are two main types of cells in the brain, those of the neurons and those of the glia. Dobbing and Sands were able to identify two phases of rapid cell multiplication in the forebrain. The first lasted from 10 to 18 weeks gestation and correspondent to the multiplication of neuroblasts; this process ceased at about 18 weeks gestation when the neuroblasts are differentiated into neurons. At about this time glial multiplication began, and all further increase in number of cells was due to an increase in number of glial cells. In the later half of pregnancy there is rapid growth of the brain, of which one component is hyperplasia of the glial cells [17]. Over the same period the dendrites of the neurons which are laid down in the first half of pregnancy are also rapidly developing in size and complexity. The resulting change in brain weight is referred to as the brain growth spurt [9] and this continues well into the second postnatal year.

Growth in Numbers

The total number of cells in any organ can be measured by determining the total organ DNA content and dividing it by a constant for the particular species which represents the DNA content per diploid cell in that species [18]. Total brain DNA content represents the total number of brain cells and total cerebellar or cerebral DNA represents the total number of cells in each of those regions.

Once the number of cells is determined, the average weight per cell (total organ DNA: DNA content per cell) [19], protein content per cell can be determined simply by analyzing for each of these components and dividing by the number of cells. This can be expressed chemically as a weight/DNA, Protein/DNA ratio which may be a useful indication of cell size [20].

Anthropometric Measurement of the Human Fetal Brain

According to clinical studies done by our group at SSKM Hospital, Kolkata (mentioned in Chapter 6), from 1979 onwards on the measurement of

growth and development of the human fetus including the fetal brain, there was a straight line relationship between the body weight and the weight of the brain (Fig. 11.1).

The weight of the whole brain up to the medulla oblongata formed 13.96-14.5 % of the body weight from 9 to 28 weeks of gestation (Table 11.1).



Fig. 11.1 Relationship between the bodyweight and brainweight of human fetuses

Table 11.1 Relationship between the body weight and the weight of brain of human fetus

Period of gestation		Body weight	Brain weight	Brain weight per 100 g
(weeks)	No. of cases	$(Mean \pm S.D.) (gm)$	$(Mean \pm S.D.) (gm)$	Body weight (gm)
9–12	74	8.73±3.16	1.24 ± 0.50	14.45
13–16	208	48.94±25.15	6.92±3.59	14.14
17–20	192	186.62±56.03	26.80±8.15	14.24
21–24	68	419.88±79.68	57.84±13.10	13.96
25–28	27	737.48±117.56	106.54 ± 20.00	14.47

	No. of cases	DNA content				
Period of gestation			Midbrain (mg/g			
(weeks)		Cerebellum (mg/g tissue)	tissue)	Cerebrum (mg/g tissue)		
9–12	25	6.69 ± 0.26	3.55 ± 0.25	2.97 ± 0.27		
13–16	38	5.51±0.34*	2.85±0.58***	2.38±0.24*		
17–20	46	3.49±0.21**	2.25±0.46***	1.54±0.31*		
21–24	23	2.69±0.24***	1.44±0.33***	1.33±0.39***		
25–28	01	2.01	1.30	1.22		

 Table 11.2
 DNA content in developing human fetal cerebrum, cerebellum and midbrain

Data are mean \pm SEM

Statistics: P value, *P<.005; **P<.001; ***P>.05 (not significant), as compared with the preceding period of gestation

Table 11.3 Cell number of the developing human fetal cerebrum, cerebellum and midbrain

Period of gestation	No. of cases	Cell number				
(weeks)		Cerebrum	Cerebellum	Midbrain		
9–12	25	$(1.24 \pm 0.37) \times 10^9$	$(0.10 \pm 0.03) \times 10^9$	$(0.24 \pm 0.08) \times 10^9$		
13–16	38	$(5.28 \pm 1.97) \times 10^9$	$(0.23 \pm 0.09) \times 10^9$	$(0.41 \pm 0.21) \times 10^9$		
17–20	46	$(16.95 \pm 5.63) \times 10^9$	$(0.45 \pm 0.12) \times 10^9$	$(0.48 \pm 0.15) \times 10^9$		
21–24	23	$(27.64 \pm 3.44) \times 10^9$	$(0.87 \pm 0.28) \times 10^9$	$(1.28 \pm 0.37) \times 10^9$		
25–28	01	31.96×10 ⁹	1.42×10^{9}	2.08×10^{9}		

Data are mean ± S.D.

Biochemical Changes in the Human Fetal Brain

DNA and RNA Content

The brain was divided into three regions, (A) The two cerebral hemispheres, (B) The cerebellum and (C) The mid and hind brain up to the lower end of the medulla oblongata at the level of the first cervical vertebra. In the course of gestation there was a progressive decrease in the DNA content per gm of the nervous tissues from 9 to 28 weeks of gestation. In the cerebrum the rate of decrease was more or less uniform which is shown in the Table 11.2. In the cerebrum and midbrain the rate of decrease was more or less the same up to 24 weeks; but the rate of decrease slowed down to 10 % from 24 to 28 weeks.

The total number of cells in the brain was calculated from the DNA content and the total weight, taking 6.2 pg as the DNA content per cell which is shown in Table 11.3. There was about 30-fold increase in the number of cells from 9 to 28 weeks of gestation in the cerebrum, 15-fold in the cerebellum and eightfold in the midbrain.

The total number of cells per mg of tissue was higher at earlier weeks of gestation in all areas of the brain. Thus at 9–12 weeks of gestation the number of cells per mg of the cerebral tissue was about 10⁹ and at 13–16 weeks 0.8×10^9 , at 17–20 week 0.5×10^9 . It is apparent, therefore, that during growth and differentiation there was accumulations of cellular and extracellular materials. There was a decrease of RNA per mg of tissue form 9–24 weeks in the cerebellum and midbrain than in the cerebrum which is shown in Table 11.4.

The increase in the weight of the tissue per unit area is, therefore, not due to accumulation of RNA, grossly speaking. There might, however, be a different micro-environmental change not detectable by total amount of RNA during the progress of gestation.

Protein Content

Total protein content of the human fetal brain at various period of gestation was studied. The protein content in mg per gm of wet weight (Table 11.5a) and the protein content in mg per gm dry weight (Table 11.5b) of cerebrum, cerebellum and midbrain decreased significantly from 13 to 16 weeks of gestation. It was also noted that the

Period of gestation	No. of cases	RNA content				
(weeks)		Cerebrum mg/g tissue	Cerebellum mg/g tissue	Midbrain mg/g tissue		
9–12	25	3.243 ± 0.27	1.825 ± 0.28	1.579±0.19		
13–16	38	3.178*±0.20	$1.650*\pm0.23$	1.380**±0.09		
17–20	46	2.316***±0.32	1.360**±0.07	1.060***±0.11		
21–24	23	2.073*±0.21	1.66***±0.11	$0.980*\pm0.09$		
25–28	01	1.98	1.41	1.30		

Table 11.4 RNA content in developing human fetal cerebrum, cerebellum and midbrain

Data are mean \pm SEM

Statistics: P value, *P>0.05; (not significant) **P<0.005; ***P<0.001, as compared with the preceding period of gestation

Table 11.5a Protein content in human fetal cerebrum, cerebellum and midbrain

	No. of cases	Protein content		
Period of gestation		Cerebrum mg/g wet	Cerebellum mg/g wet	Midbrain mg/g wet
(weeks)		tissue	tissue	tissue
9–12	11	63.81 ± 1.98	61.90±0.95	60.13 ± 1.71
13–16	19	60.70±1.10**	59.21±0.68*	$58.89 \pm 0.55*$
17–20	16	58.11±1.66**	57.30±0.72*	56.91±0.26*
21–24	9	57.50±2.03*	56.42±0.79*	55.01±0.95**

Data are mean ± SEM

 Table 11.5b
 Protein content in human fetal cerebrum, cerebellum and midbrain

	No. of cases	Protein content				
Period of gestation		Cerebrum mg/g dry	Midbrain mg/g dry			
(weeks)		tissue	tissue	tissue		
9–12	11	709 ± 32.7	687 ± 19.9	668 ± 38.5		
13–16	19	$607 \pm 30.6*$	592±14.7*	589±13.8*		
17–20	16	581±25.2****	573±10.6*	$569 \pm 10.9^*$		
21–24	9	575±25.9***	564 ± 6.8	550±18.9**		

Data are mean ± SEM

Statistics: P value, *P<0.001; **P<0.005; ***P>0.05, ****<0.01; (not significant) as compared with the preceding period of gestation

protein content of the above regions were nearly the same at the corresponding period of gestation.

¹⁴C-leucine incorporation was studied in the human fetal brain at different gestational periods. It was found in Table 11.6, that 16,000 g with supernatant incorporation in the cerebrum and midbrain the rate of incorporation increased from 13 to 20 week of gestation and then declined up to 24 weeks of gestation to values even lower than those at 13–16 weeks of gestation in the respective tissues.

Incorporation of 14 C-leucine into proteins of mitochondrial fractions (16,000×g pellet) of human fetal brain tissue appeared to be highest in cerebrum followed by midbrain at earlier periods of gestation (13–20 week). The rate of incorporation into the

cerebrum protein remained almost unchanged throughout the period of gestation studied. The incorporation rate in the midbrain increased gradually with progress of gestation. Here also a twofold increase was found at 21–24 weeks of gestation.

P³² Incorporation in DNA, RNA and Phospholipid

We studied P^{32} incorporation into phospholipid fraction (Table 11.7) and in Acid soluble fraction, Nucleotide, DNA and RNA (Table 11.8). We also studied P^{32} incorporation into total Nucleic Acid (Table 11.9). All the studies were done in fetal

	No. of cases	Incorporation (c.p.m/mg protein/10 ⁴ counts)					
Period of gestation		Supernatant		Pellet			
(weeks)		Cerebrum 120 min	Midbrain 120 min	Cerebrum 120 min	Midbrain 120 min		
13–16	4	5.06 ± 0.24	3.12±0.32	5.04 ± 0.35	1.53 ± 0.26		
17–20	5	6.25 ^b ±0.41	$3.42^{d} \pm 0.24$	5.17°±0.27	1.95°±0.22		
21–24	5	$4.72^{b} \pm 0.44$	$3.01^{d} \pm 0.16$	5.72°±0.38	2.84 ^b ±0.20		

Table 11.6 In vitro incorporation of L-(14 C-U)-leucine into TCA precipitable proteins from 16,000 × g supernatant and of pellet of human fetal brain tissues

Each value represents the Mean ± SEM for 4-5 determinations

 $_{b}^{a}$ < 0.001; $_{p}^{b}$ < 0.005; $_{p}^{c}$ < 0.01 and $_{p}^{c}$ < 0.05; $_{p}^{d}$ < 0.05 (not significant), compared with the preceding period of gestation

No. of cases Incorporation (c.p.m/g tissue/100 counts) Period of gestation Cerebrum Cerebellum Midbrain (weeks) 9-12 8 10.05 ± 2.32 N.D. 22.11 ± 2.71 13-16 10 19.10 ± 2.05 18.12 ± 2.83 15.50 ± 3.15 17 - 207 14.51 ± 3.05 9.42 ± 2.91 13.69 ± 2.85 21 - 246 13.33 ± 2.19 5.91 ± 2.75 12.17 ± 3.06 Adult 2 N.D. N.D. N.D.

 Table 11.7
 P³² incorporation into lipid fractions of different human fetal brain

The results are expressed as mean \pm S.D.

N.D. not done

Table 11.8 P³² incorporation into nucleic acids of different human fetal organs at different period of gestation

Period of gestation	EE	Incorporation (c.p	Incorporation (c.p.m/g tissue/100 counts incubated ×10 ⁻²)				
(weeks)		Cerebrum	Cerebellum	Midbrain			
13–16	10	10.5 ± 2.19	8.83 ± 2.10	6.52 ± 1.84			
17–20	7	5.79 ± 1.84	3.69 ± 1.13	3.10 ± 1.66			
21–24	6	4.91±1.06	3.60 ± 1.56	2.90±1.07			
Adult	2	N.D.	N.D.	N.D.			

The results are expressed as mean \pm S.D.

N.D. not done

Table 11.9 P-incorporation into acid soluble nucleotides of different human fetal organs at different period of gestation

Period of gestation	No. of cases	Incorporation (c.p.m/g tissue/100 counts incubated)				
(weeks)		Cerebrum	Cerebellum	Midbrain		
9–12	8	5.31 ± 1.82	N.D	6.51 ± 1.51		
13–16	10	4.32 ± 1.30	2.10±0.4	5.13 ± 1.80		
17–20	7	4.57 ± 1.20	3.52 ± 1.0	4.79 ± 1.03		
21–24	6	5.98 ± 2.15	3.81±1.31	6.83 ± 1.62		
Adult	2	N.D.	N.D.	N.D.		

The results are expressed as mean \pm S.D.

N.D. not done

Cerebrum, Cerebellum and Midbrain of human fetuses at different periods of gestation. The results showed that incorporation of P³² into total Nucleic acid was less than into the lipid fraction. P³² incorporation into nucleic acid were highest in the earlier period of gestation, i.e., 13–16 weeks, than in the later periods.

Synthesis in the Human

Glutamine synthetase catalyzes a number of reactions [21], among which there are two important reactions. The first reaction catalyzes glutamine formation from glutamate, ammonia and ATP and the second causes the formation of gamma-glutamyl hydroxamate from glutamate, hydroxyl amine and ATP. No detectable amount of glutamine synthetase activity was found in the human fetal brain using the first reaction. The enzyme activities of the human fetal cerebrum, cerebellum and midbrain were studied. In all the fetal organs that were studied, namely cerebrum, cerebellum and midbrain, the enzyme activity of the second reaction showed two to threefold increase from 13 to 28 weeks of gestation. The increase was progressive throughout the period of gestation.

Histological Studies of the Human Fetal Central Nervous System

Only some aspects of the histology of the central nervous system in developing human fetuses are illustrated. Figure 11.2 shows the disposition of the neural tube in a human fetus of below 7 weeks of gestation at a very low magnification. The total field of photomicrograph was obtained by joining the photomicrographs of five different parts carefully. The neuroblasts arise from the ependyma of the neural canal. Even at this stage there is some differentiation of the neuroblasts pushing towards the periphery (Fig. 11.2). The

ventricular zone is a densely stained layer of cells with evidences of a layer of cells migrating towards the periphery. The intermediate zone contains fibres and sparsely situated cells, while the marginal layer contains fibres only. There is a layer of nerve fibres surrounding the neural tube (Fig. 11.3). The fetus has 22–24 somites. At some places the growing neurons seem to innervate growing muscular tissues within the somite (Fig. 11.4). The transition between the deeply staining packed neuroblasts and the differentiating sparsely disposed neural cells is rather abrupt (Fig. 11.7). Figure 11.8 shows the general appearance of the central nervous system at the cranial end. The cells are less differentiated than the cells at the level of the spinal canal (Figs. 11.2, 11.3, 11.4, 11.5, 11.6, 11.7, and 11.8). The ventricles are lined by almost a single layer of epithelial cells from which (perhaps?) the generally undifferentiated cells arise and spread (Fig. 11.9). Incidentally, the picture shows a beautiful reproduction of the developing eye (Figs. 11.10, 11.11, and 11.12). Figure 11.13 shows the development around a capillary with nucleated cells (1.725 g fetus) and Fig. 11.14 shows the histology of a part of the cerebrum of a fetus weighing 2.3 g. It illustrates the rapid growth of the invaginations of the cephalic ends of the neural



Fig. 11.2 Photomicrograph of a sagital section in a human fetus of below 7 weeks of gestation showing the disposition of the neural tube at a very low magnification



Fig. 11.4 Layer of nerve fibers surrounding the neural tube



Fig. 11.5 Somites with innervations growing neurons to muscular tissue

Fig. 11.3 High power field of neuroblasts showing the ventricular zone, intermediate zone and the marginal layer of the human fetus below 7 weeks of gestation

Fig. 11.6 Somites containing centrally located blood vessels



Fig. 11.7 High power field of a blood vessel containing nucleated erythrocytes within the somites

Fig. 11.8 High power field of neuroblasts

Fig. 11.9 Low power view of CNS at the cranial end of 100 mg fetus







Fig. 11.10 High power view of the CNS of 100 mg fetus showing relatively uniform looking undifferentiated cells

Fig. 11.11 Low power view of a developing eye (below 7 weeks)

Fig. 11.12 High power view of developing eye (below 7 weeks)





Fig. 11.13 High power field showing development around a capillary with nucleated cells (1.725 g fetus) at the upper part

Fig. 11.14 Low power view of cerebrum of a 2.3 g fetus showing development of layers

tube with mostly undifferentiated neuroblasts, whereas, in other places (Fig. 11.15) the same brain is consolidating the neuronal growth to be differentiated further, later on, in life. Figure 11.16 shows the histology of the part of the cerebrum of fetus weighing 36 g (about 14 weeks of gestation). It shows extensive growth of capillaries within the developing neuronal cells. Figure 11.17 shows the histology of the cerebrum of a fetus weighing 95.5 g (about 16 weeks of gestation).

It shows the development of a sulcus in the brain. At about this time there is extensive capillary growth within the developing cerebrum. Figure 11.18 gives a view of the cere-

Fig. 11.15 Low power view of cerebrum of 2.3 g fetus in another place showing the neuronal migration

brum-subgerminal layer showing at least five capillaries (167 g fetus). Figure 11.19 gives a view of the sub-cortical region showing fibers and a capillary (224 g fetus), and Figure 11.20 gives a view of an area of the midbrain of a 12 g fetus showing peripherally the cells circumscribed by fibrous layer (perhaps tract) and smaller islands of neurons. Figure 11.21 shows the histology of cerebellum of a fetus weighing 90 g (about 16 weeks of gestation). The external granular layer is in the proliferative stage. Figure 11.22 shows the histological appearance of cerebellum of a fetus weighing 150 g (about 17 weeks of gestation). There are only two layers, an outer granular and an inner molecular.



Fig. 11.16 High power view of extensive capillary network in the cerebrum (36 g fetus)

Fig. 11.17 High power view of cerebrum showing development of sulcus (95.5 g of fetus)



Fig. 11.18 High power view of cerebrumsubgerminal layer showing at least five capillaries (167 g fetus)

Fig. 11.19 High power view of sub-cortical region showing fibers and a capillary (224 g fetus)

Fig. 11.20 High power view of an area of the midbrain of a 12 g fetus showing peripherally the cells circumscribed by fibrous layer (perhaps tract) and smaller islands of neurons







Fig. 11.21 High power view of mid brain of 90 g of fetus showing peripheral layer of neuronal cells and central layer of more fibers

Fig. 11.22 High power view of mid brain of 150 g fetus showing two capillaries in the central part of the fiber rich area

There is no evidence of the single layer of the purkinje cell.

Discussion

Anthropometric Measurement

The rate of growth of a particular human fetus depends upon its intrinsic growth potential, inherent in its genomic complex which is modified by the environment, provided locally by the mother's uterine complex. Individual growth of the fetus is a function of the genetic potential.

Nucleic Acid Contents

DNA contents per gram wet weight of human fetal cerebrum, cerebellum and midbrain decreased with progress of gestation. Such a decrease in the brain was also described by Subba Rao and Janardana Sarma [22]. This can be explained by the fact that the growth of the brain entails not only of cellular growth but of the neuronal processes as well. In the earlier periods of gestation the cellular element was more but as more and more processes develop in the course of differentiation, cells per unit mass of the brain naturally will decrease. It was noted that the rate of cell multiplication was more in the earlier period of gestation. However, this rate varies in different parts of the brain. For example, in the midbrain which consists predominatly of white matter, the rate of increase is slow. The rate of growth in the cerebellum does not decrease with progress of gestation as much as in the cerebrum. This has also been the experience of other workers, although Winick et al. [23] think that the rate of cell growth in the cerebrum and cerebellum is similar.

Enzymes: Glutamine Synthetase

Glutamine Synthetase catalyzes not only synthesis of glutamine from glutamate, NH₃ and ATP, but also has three other enzymatic activities one of which is formation of gammaglutamyl hydroxamate and ammonia from glutamine and hydroxyl amine. In human fetal organs, including liver, cerebrum, midbrain and cerebellum synthesis of glutamine could not be detected from glutamate, ammonia and ATP.

Nor was it possible to detect any activity of forming gamma-glutamyl hydroxamate and ADP from glutamate, NH₂OH and ATP. The third reaction, i.e., formation of gamma-glutamyl hydroxamate and NH₃ from glutamine and NH₂OH could be demonstrated in these human fetal organs.

Glutamine synthesis has been speculated to be a means of detoxication of ammonia locally produced [24–26]. Most of the ammonia, in this process, comes from the dehydrogenation of glutamic acid. Glutamic acid is an alternate metabolic fuel for the brain, apart from glucose and ketone bodies. The glutamine produced in the brain may diffuse into the plasma and serve as the building blocks of protein, and also as a means of bicarbonate formation in the kidney tubules in conditions of acidosis.

In Vitro ³²P: Incorporation

Incorporation of ³²P into acid soluble nucleotides of liver decreases in course of gestation. The higher incorporation at earlier periods of gestation indicates faster metabolic turnover in general. The nucleotides are utilized for biosynthesis of various organic compounds. At earlier periods of gestation cell division takes precedence over other metabolic activities. Thus the neucleoside triphosphates are more utilized for the biosynthesis of the nucleic acids. In our own experiments the total DNA contents increased about six times from a gestation period of 9-12 to 13-16 weeks, whereas they increased less than two times only from a gestation period of 21-24 to 25-28 weeks. In later periods of gestation, other metabolic activities involving biosynthesis, which all require nucleoside triphospahte, increase.

However the rate of cell division decreases thus accounting for the overall decrease of formation of acid soluble nucleotides in fetal livers of later gestation periods. However, it is not possible to account for similar rates of incorporation of ³²P into acid soluble nucleotides of the brain. The total

Period of gestation (weeks)	No. of cases	Specific Activity (µmol/h/g protein)		
		Cerebrum	Cerebellum	Midbrain
9–12	12	200.41 ± 30.32	N.D.	N.D.
13–16	33	708.43±40.11**	761.33±68.52	652.42±46.15
17–20	27	870.91±74.72*	1,204.73±106.10**	1,300.81±114.61**
21–24	18	911.35±83.20***	1,422.54±117.76****	1,853.22±132.11**
25–28	07	1,312.46±111.20**	1,508.97±140.64***	2,018.16±139.95****
	01	N.D.	N.D.	N.D.

Table 11.10 Gamma-glutamyl transferase activity in human fetal brain

Data are mean ± SEM

Statistics: P value, *P<0.005; **P<0.001; ***P<0.01, ****P>0.05; (not significant) as compared with the preceding period of gestation

DNA, for example, in the cerebrum increases about four times from 9-12 to 13-16 weeks of gestation (Table 11.10), whereas it increases only 1.5 times from 21-24 to 25-28 weeks. But the rate of incorporation of ^{32}P into acid soluble nucleotides is more or less the same in the above gestation periods. Perhaps in the case of brain, acid soluble nucleotides are very importantly and significantly involved in myelin synthesis thus overshadowing the effect of rate of synthesis of DNA.

In the fetal brain, there was again no gross difference in the ³²P incorporation into the lipids of the cerebrum and midbrain. In the cerebellum, however, the rate of incorporation diminished. The growth of the cerebellum differs in different species. In man there is a contention that the cerebellum continues to grow even after the cerebrum has stopped growing. Cell division is accompanied by membrane synthesis in the two daughter cells, which continues to grow after ceasation of mitosis. The issue is whether the decrease in incorporation of ³²P into phospholipids signifies rapid membrane growth.

In Vitro ¹⁴C-Leucine Incorporation

During the growth of the human fetus in uterus, the brain increased in mass in proportion to the body weight similar to the liver, from 13 to 28 weeks of gestation. There are other human fetal organs like the adrenal cortex or femur which do not grow in proportion to the total body weight (our unpublished results). This difference in response to the growth potential may be related to the hormonal characteristics of the different organs, in terms of receptors for the various hormones, directly or indirectly related to growth, and also in terms of the pattern of response to the receptor bound hormones.

It may only be surmised that human fetal brain and liver from 13 to 28 weeks of gestation grow uniformly and unfold the receptors and respond to the receptor bound hormones in a uniform way. The protein content of the human fetal liver increased during progress of gestation. The water content decreased from 80% at 9 weeks to 77.5% at 24 weeks. The increase of protein content in the human fetal liver might have been due to increased rate of protein synthesis commensurate with assumption of specific functions in the economy of the developing fetus.

It may be stated that the slight reduction of protein content per unit dry weight of the tissue in brain regions of human fetuses is not due to reduction of protein synthesis during the period of gestation studied. Further, it may be suggested that the slight reduction of protein content per dry weight of the brain regions accompanied by an almost unaltered rate of protein synthesis is possibly due to the rapid accumulation of lipid components in the respective brain regions, indicating the comparatively high rate of lipid synthesis in the brain during the fetal life studied.

Incorporation of U¹⁴C- Glucose In Vitro

Radiolabelled glucose was incorporated into radioactive compounds in brain tissues in vitro. The incorporation into amino acids were characteristic of adult brain tissues in contrast to the fetal brains. Very little incorporation of counts from glucose into either the lipid fractions or into nucleic acids could be demonstrated.

Histological Studies of the Human Fetal Central Nervous System

A detailed histological study of the human fetal central nervous system is beyond the purview of the present investigations. Only some points are being presented.

Development of the Eye

We have a histological study of a 35 mm human embryo. The photomicrograph shows that the inner and outer layer of optic cups seem to encircle the lens completely whereas conventionally the nervous layer is illustrated to fold back in front of the lens demarcating the anterior chamber and developing cornea. In the present study, the sclera and presumptive cornea form one complete sheet of tissue. We, therefore, present two possibilities:

- 1. The plane of the section of the fetus is responsible for the presented picture. But then there should have been at least traces of the folding back of the nervous tissues.
- 2. The development of the anterior chamber and cornea is a later phenomenon. However we have no evidence of the anterior chamber and of the cornea as yet. The cellular nature of the developing lens is well illustrated in the present histology of the eye. The small conjunctival sac is lined by epithelial cells.

The developing neural tube at this stage of fetal life contains a more or less uniform type of tissue, with a ventricular, subventricular, intermediate and marginal zone. This is true both in the spinal region as well as in the cephalic parts of the tube (Figs. 11.2 and 11.4). The neural tube is a closed system at this stage and the histology shows more differentiative neural tissues.

Development of the Somites

The other interesting histology is that of the somites. These paraxial mesodermal condensations are the forerunners of the dermotome, myotome and sclerotome. We can count about 25 somites in this particular fetus. At this stage the myocele has been obliterated by the proliferation of cells. The epitheloid cells have changed their character into mesenchymal tissue. The capillary has complete endothelial wall and within the capillary almost all cells are nucleated erythrocytes consistent with their (?) yolk sac origin rather than of hepatic origin. The picture is consistent with the existence of a nutrient vessel within each individual somite for its further development and differentiation. This description I could not find in any text book of human embryology.

Cortical Histogenesis

The earliest histology of the cerebral cortex is from a fetus of about 8 weeks' gestation. The capillaries contain nucleated erythrocytes. The cellular development is similar to that found everywhere in the central nervous system – neuroepithelial cells from the ventricles dividing and migrating to the pial surface. It is difficult to trace this migration by usual hematoxylin eosin staining. Except for a layer of cells lining the ventricles, which stain deeply, the rest of the cells form almost a continuum bridging the two layers of the neuroepithelium. The blood supply in the developing cerebrum is plentiful. The vessels dip into the cortical matter from the surface.

Development of the Midbrain

The general plan of histogenesis in the midbrain is basically similar to other areas of the brain. The blood vessels are numerous. A distinguishing feature are the centres of neuronal congregations deep within the white matter, which are perhaps in connection with specific centre of the midbrain.

Development of the Cerebellum

The histological development of the human cerebellum has been studied only in certain aspects. The development in fetal life has been well described by Uzman [27] and Langman [28]. The development in fetal life of human cerebellum consists of an external granular layer, an internal granular layer and a layer of ventricular neuroepithelial layer. The rapid proliferation of the external granular layer is well illustrated in Fig. 11.23. The zone consists of a broad band of cells more or less uniform in character, whose fibres from the white matter internal to the zones. The ventricular neuroepihelial cells migrate towards the surface, forming an internal granular layer, internal to the

Fig. 11.23 High power view of mid brain of 347 g of fetus showing perhaps a cluster of neurons for either a cranial nerve centre or a group of neurons for a centre like the respiratory.(?) white matter formed by the proliferated external granular layer of cells (Figs. 11.24 and 11.25). Further development of the cerebellum occurs at periods of gestation later than 20 weeks and continues postnatally (Fig. 11.26).

Summary and Conclusions

- 1. There is a straight line relationship between the body weight and brain weight. Weight of the brain of the fetus is around 14 % of the body weight throughout the gestation period.
- 2. The total nucleic acid contents (DNA and RNA) in the human fetal brain increased with



Fig. 11.24 Low power view of cerebellum of a 90 g fetus showing neuronal cells originating from two sides but distinctive characteristics in the two lines of origins

Fig. 11.25 High power view of cerebellum of a 150 g fetus showing cell migrating



Fig. 11.26 Low power view of cerebellum of a 150 g fetus showing cell migration



the progress of gestation along with the increase of cell number of the organ. The rate of increase in total DNA content in early periods of gestation was much higher than at later periods. The DNA content per gm tissue decreased with progress of gestation. The rate of decrease in the cerebrum was more or less uniform while that in the cerebellum and midbrain was more or less the same up to 24 weeks and then further slowed down.

 The number of cells increased 30-fold in the cerebrum, 15-fold in the cerebellum and eightfold in the midbrain form 9–12 weeks to 25–28 weeks of gestation. In the earlier stages of gestation, the growth of the brain entails cellular growth as well as neuronal processes. In earlier periods of gestation the cellular element was more but in the course of development due to more and more processes, cells per unit mass of the brain decreased.

- 4. The total protein content per gm wet wt of the human fetal brain was found to decrease with the increase of gestation.
- No glutamine synthetase activity was detectable in the human fetal brain. Gammaglutamyl transferase activity increased progressively with gestation periods in the cerebrum,

cerebellum and midbrain. Glutamine synthesis is not a prominent feature of human fetal organs. The mammalian fetus is constantly supplied with glucose through the maternal circulation. So the alternate metabolic fuel, i.e., glutamic acid oxidation is perhaps not needed. A high level of glutaminase activity was found in the cerebrum and midbrain at 9–12 weeks of gestation. The neural tissues in earlier periods of gestation consist mostly of neurons rather than glial cells which may be relatively deficient in glutaminase activity.

- 6. In vitro incorporation of ³²P into the acid soluble nucleotides of human fetal liver slices decreased with progress of gestation. The rate of incorporation of ³²P into acid soluble nucleotides in the brain was more or less the same throughout the gestation studied. This signifies that perhaps in the case of the brain, the acid soluble nucleotides are very importantly and significantly involved in myelin synthesis. ³²P incorporation into lipids was much less than in acid soluble nucleotides. More incorporation in the earlier periods of gestation than in the later periods might signify higher metabolic activity and turnover at earlier periods.
- 7. The rate of ¹⁴C- leucine incorporation into protein gradually increased up to 20 weeks and then declined. The tissue in brain regions of human fetuses at the later gestation periods accompanied by an almost unaltered rate of protein synthesis is possibly due to rapid accumulation of lipid components indicating a comparative high rate of lipid synthesis in the brain during the fetal life studied.
- The amount of incorporation of ¹⁴C- glutamate was more than that of ¹⁴C- leucine at each gestation period. This suggests that glutamate has probably a special role to play in brain metabolism.
- The incorporation of U-¹⁴C-glucose into human fetal brain slices suggested that the fetal brain can utilize glucose for energy metabolism.
- 10. From the seventh week to the 28th week, the total weight of the brain forms almost a constant percent of the body weight. Certain

aspects of the development of the eye were was included in the histological studies. The histological picture closely followed the pattern described by standard text books of embryology except in certain minor points. During early fetal life each somite has a centrally located blood vessel which supplies presumably nutrients for the further development of the somites. The histological development of the cerebrum, cerebellum and midbrain followed closely the description in standard text books of embryology.

References

- Stroker MGP. The multiplication of cells. Postgrad Med J. 1978;54 Suppl 1:5–11.
- 2. Wolpert L. The development of the pattern of growth. Postgrad Med J. 1978;54 Suppl 1:15–21.
- Davies DP. Physical growth from fetus to early childhood. In: Davis JA, Dobbing J, editors. Paediatrics. 2nd ed. London: Scientific Foundations, William Heinemann Medical Books Ltd; 1981. p. 303–30.
- Fish I, Winick M. Cellular growth in various regions of the developing rat brain. Pediatr Res. 1969;3:407–12.
- 5. Jacobson M. Developmental neurobiology. New York: Plenum Press; 1978. p. 27–55.
- Winick M. Nutrition and nerve cell growth. Fed Proc. 1970;29:1510.
- Dobbing J, Sands J. Quantitative growth and development of human brain. Arch Dis Child. 1973;48:757–67.
- Davison AN, Dobbing J. Malnutrition as a vulnerable periods in brain development. Br Med Bull. 1966;22:40–4.
- Guthrie HA, Brown ML. Effect of severe undernutrition in early life on growth, brain size and composition in adult rats. J Nutr. 1968;94:419–26.
- Zamenhof S, Van Marthens E. Study of factors influencing prenatal brain development. Mol Cell Biochem. 1974;4:157–66.
- 11. McCanace RA, Widdowson EM. The chemical structure of the body. Q J Exp Physiol. 1956;41:1.
- Urrusti j, Yoshida P, Velasco L, Frenks S, Rosado A, Sosa A, Morales M, Yoshida T, Metcoff J. Human fetal growth retardation. Clinical features of sample with intrauterine growth retardation. Pediatrics. 1972;50:547–58.
- Lechtig A, Habicht JP, Delgado H, Klein RE, Yarbrough C, Martorell R. Effect of food supplementation during pregnancy on birth-weight. Pediatrics. 1975;56:508.
- 14. Prestige MC. Differentiation, degeneration and the role of the periphery: Quantitative considerations. In:

Schmitt FO, editor. The neurosciences: second study program. New York: The Rockefeller University Press; 1970. p. 73–82.

- Dobbing J. Lipids, malnutrition and developing brain, a CIBA foundation symposium. Amsterdam: Elsevier-Excerpta Medua; 1972. p. 9–29.
- Dobbing J. The developing brain; a plea for more critical interspecies extrapolation. Nutr Rep Int. 1973;7:401.
- Dobbing J, Smart JL. Vulnerability of developing brain and behavior. Br Med Bull. 1974;30:164.
- Enesco M, Leblond CP. Increase in cell number as a factor in the growth of the organs and tissues of the young male rat. J Embryol Exp Morphol. 1962;10:530–6.
- 19. mirsky AE, Ris H. Variable and constant components of chromosomes. Nature. 1949;163:666–9.
- Epstein CJ. Cell size, nuclear content and the development polyploidy in mammalian liver. Proc Natl Acad Sci (U S A). 1967;52:327.
- Meister A. Glutamine synthesis. In: Meister A, editor. Biochemistry of the amino acids, vol – I, New York, London: Academic Press Inc.; 1965. p. 439–592.

- 22. Subba Rao K, Janardana Sarma MK. Growth and development in different regions of human fetal brain: changes in wet weight, moisture content and nucleic acids. Ind J Med Res. 1976;64:144–53.
- Winick M, Rosso P, Walterlow J. Cellular growth of cerebrum, cerebellum and midbrain stem in normal and marasmic children. J Exp Neurol. 1970;26:393–400.
- Kerbs HA. Metabolism of amino acids. The synthesis of glutamine from glutamic acid and ammonia and the enzymic hydrolysis of glutamine in animal tissues. Biochem J. 1935;29:1951–69.
- Malherbe HW. Significance of glutamic acid for the metabolism of nervous tissue. Physiol Rev. 1950;30:549–69.
- Strecker HJ. In: Richter D, editor. Metabolism of the nervous system. New York: Pergamon Press; 1957. p. xiv–l–599.
- Lahut Uzman L. The histogenesis of the mouse cerebellum as studied by its tritiated thymidine uptake. J Comp Neurol. 1960;114:137–159. doi: 10.1002/cne.901140204.
- Langman J. Medical Embryology. Baltimore: Williams and Wilkins; 1975. p. 109.

Lipid Metabolism in the Human Fetus Development

12

Ornella Guardamagna and Paola Cagliero

Abbreviations

ABCA1	ATP-binding cassette A1
ABCG1	ATP-binding cassette G1
ApoA-1	Apoprotein A-1
ApoA-4	Apoprotein A-4
ApoE	Apoprotein E
CE	Cholesteryl ester
CETP	Colesteryl ester transfer protein
СНО	Cholesterol
HDL	High density lipoprotein
FA	Fatty acid
FFA	Free fatty acid
LCAT	Lecithin:cholesterol acyl transferase
LDL	Low density lipoprotein
LP	Lipoprotein
LRP-1	LDL-receptor related protein 1
LRP-2	LDL-receptor related protein 2
NPC1L1	Niemann-Pick C1-like1
PTP	Phospholipid transfer protein
SPC-X/2	Sterol carrier protein X and 2
SR-B1	Scavenger receptor class B
TG	Triglyceride
VLDL	Very low density lipoprotein

e-mail: ornella.guardamagna@unito.it

Introduction

Lipids including cholesterol (CHO) and fatty acids (FAs) are main constituents of human body cells and actors in physiological functions thus representing a critical requirement for the embryonic and fetal development.

CHO and FAs attain multiples functions: CHO is a cellular membrane constituent, a steroid hormone, bile acids and oxysterol precursor and it is essential for activation of various signalling pathway [1, 2]. CHO plays an important role before implantation, as a precursor of progesterone synthesis, and helps in maintaining the early pregnancy [3–7]. When the embryo is implanted in the uterine wall, CHO is determinant for the embryogenesis and morphogenesis and patterning of the central nervous system [8]. As well FAs and triglycerides (TGs) are cellular membrane constituents, represent an energy source and take part in neuronal and visual development [9].

The embrio and fetus do not come in direct contact with the maternal circulation thus are dependent upon tissues surrounding them to receive the nutritional support. These tissues are represented by the yolk sac and trophoblasts, early in the first trimester, then the placenta since the end of the first trimester and the second trimester [5]. The placenta is an hemochorial villous organ with multiple functions: oxygen and CO_2 exchange, nutrient absorbption and immune

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_12

O. Guardamagna, MD (⊠) • P. Cagliero, PhD Department of Health Science and Pediatrics, Turin University, Piazza Polonia, 94, Torino 10126, Italy

[©] Springer International Publishing Switzerland 2016

barrier. It represents a bridge connecting mother and fetus through the maternal-placental (uteroplacental) blood circulation and the fetalplacental (fetoplacental) blood circulation. The functional unit of the placenta is the chorionic villus which contains syncytiotrophoblast/cytotrophoblast, villous stroma and fetal vascular endothelium, layers that separate the maternal blood from the fetal circulation [10]. The yolk sac and the placenta provide an adequate nutrient supply by transporting a wide variety of maternal molecules to the embrio and fetus, including lipids, so promoting the intrauterine development. The transfer of some nutrients is regulated by the placenta itself through specific enzymes, receptors and transport proteins; others nutrients are directly metabolized by the placenta.

During gestation metabolic changes intervene with a shift from carbohydrates to lipids for maternal energy production in order to make nutrients available for the fetus [11]. Glucose is the main substrate that crosses the placenta but other factors may also contribute to the fetal growth. The fetus requires a substantial amount of lipids throughout its development, the lack of CHO affecting growth disorders [12]. To satisfy these needs maternal physiological hyperlipidemia is manifest in pregnancy; CHO, TGs and FAs concentrations increase in both maternal plasma and erythrocytes thus allowing the fetus to rapidly receive and store fat, which exceeds by far that of any other nutrient [13]. Maternal plasma CHO may increase through the 12th week

> Fetal LP-CHO

of gestation while TGs reach the 150–300 % of increase in the third trimester of pregnancy [6, 14, 15]. The two lipoproteins (LPs) classes involved in supporting the placental CHO need are low density lipoprotein (LDL) and high density lipoprotein (HDL) [16–18]. A supply of CHO requirement as a precursor for the production of steroid hormones in the placenta is further critical [19]. Fetal steroid precursors of estrogens regulate the uptake of maternal LPs to promote the placental progesterone are thus key determinants in pregnancy maintenance and fetal growth so being evident the basic role of fetal and maternal LPs [20, 21] (Fig. 12.1).

Lipids Synthesis and Transport

The fetus has two potential sources of CHO and FAs that include the endogenous and exogenous metabolic supply. The endogenous pathway is represented by lipids synthesized by the embrio and fetus themself, the exogenous pathway concerns lipids provided by the maternal and placental circulation. Fetal CHO and FAs are thus either taken up from the maternal circulation or synthesized "*de novo*".

In humans very low density LP (VLDL) carries CHO and TG from the liver where originates to peripheral cells, LDL carries mainly CHO. HDL represents the reverse CHO pathway deputed to carry out free CHO from

> Maternal LP-CHO



Fig. 12.1 Feto-maternal CHO intake and steroid hormonal synthesis regulation

peripheral cells, to promote CHO esterification to cholesteryl ester (CE), to provide exchange of CE and TG between circulating LPs and to bring up CE to the liver (Fig. 12.2). The delivery of lipids to the fetus is made available by the physiological maternal LPs increase which allow VLDL, LDL, HDL to be taken up by the placenta [6] (Fig. 12.3). Lipid metabolism undergoes particular changes during pregnancy, despite the fact that the placenta is practically impermeable to TGs, except for FAs. Through the early two third of pregnancy the mother



Fig. 12.2 Lipoprotein metabolism in unpregnant women



Fig. 12.3 Mother to fetus CHO and lipid exchange

accumulates fat stores thus providing nutrients sources to the fetus in the third part of pregnancy by the placenta transfer [22].

Fetal Cholesterol

Most of CHO required for the fetal growth is "de novo" synthesized by the fetus itself, thus making him autonomous from maternal or placental cholesterol supply [23]. The fetal CHO synthesis amount has been demonstrated to be higher than in adults and the endogenous production of CHO has been confirmed a mandatory need for fetal development [24]. The requirement of CHO is particularly high in brain and liver tissues, the synthesis by the liver being postulated to support other tissues requirement as demonstrated in animals [25]. Sterols markers of the CHO synthesis which include lanosterol, dihydrolanosterol, and lathosterol increase strongly since the 19th week of gestation while low till that period [26].

A maternal contribution to fetal CHO levels is anyway sustained by evidences supporting the postulate that up to 20 % of the sterol used by the fetus in the first trimester origins from maternal CHO and that an even greater percentage could be derived from the placenta with higher CHO concentrations [5]. More recently it was demonstrated that this figures grow up to around 22–40 % in the last trimester of pregnancy with a peak in the 2nd trimester [27] (Fig. 12.3).

Data, demonstrating that fetuses affected by CHO synthesis defect show at birth or later detectable CHO in tissues and plasma, support the hypothesis of a maternal supply to the fetal CHO pool [28, 29]. Furthermore comparing the LDL levels flux through the umbilical artery, which transports blood to the fetus via the placenta and that of the cord vein, which translates blood in the opposite direction (via the placenta to the fetus), it appears that LDL concentrations are higher in the umbilical artery [30]. CHO maternal LPs uptake can be also influenced by maternal, but not paternal, apolipoproteinE (APOE) phenotype thus adding a subject to exogenous CHO transfer to the fetus from maternal blood [31]. A further marker for maternalfetal CHO transport consists of beta-sitosterol levels detectable in the amniotic fluid [26]. The maternal CHO availability in early pregnancy seems relevant to placental and embryonic development in humans [32–37] thus on the basis of the above demonstrations it has been hypothesized that the fetus can acquire maternal CHO whenever this hypothesis it is not definitively accepted and needs further confirmations.

CHO Transport

Maternal CHO has to cross the barriers between maternal and fetal tissues: the yolk sac in early pregnancy and, from approximately the fourth gestational week, the placenta (Fig. 12.4). Since the 10–12 weeks of gestational age the placental syncytiotrophoblast layer plug maternal blood on their apical side and fetal microvessels at its basolateral side. In-vitro studies suggest that the yolk sac is able to transfer externally derived CHO by receptor-independent processes such as aqueous diffusion [5, 38]. Most evidences come from studies conducted in animals that strongly suggest a transport over the yolk sac membrane during pregnancy [24].

Humans studies indicate that maternal CHO markedly contributes to the fetal CHO pool at early stages of gestation, both resulting significantly correlated [39]. The placental layer overcome by lipids could be more difficult. CHO uptake by syncytiotrophoblast cells is the result of receptor mediated as well as receptorindependent processes [24]. This step is allowed by means of different mechanisms, already demonstrated in animals, but still questionable in humans. These involve:

(a) LPs receptor mediated mechanisms involving LDL and VLDL receptors. LPs bound to the receptor undergo endocytosis to lysosomes/endosomes where CHO esters are degraded. The free CHO is transported across the cells via sterol carrier proteins, as Niemann-Pick C1-Like 1 (NPC1L1) and sterol carrier protein X and 2. The receptor is then recycled to the membrane [40].



Fig. 12.4 Placental uptake and efflux of maternal LPs

- (b) Receptors that transfer the CHO across the plasma membrane without internalization of the receptor. These include:
 - 1. The scavenger receptor class B (SR-B1) that transports CHO mainly from HDL and with less affinity from LDL;
 - LDL receptor-related protein 1 (LRP-1) that binds apoE-containing particles such as VLDL. An increase in maternal blood CHO was demonstrated to reduce LDL receptor protein in trophoblasts and this result was considered a regulatory effect of maternal CHO on these receptor expression [41].
 - LDL receptor-related protein 2 (LRP-2), also named megalin, binds LDL and apoB, while cubilin binds HDL, apoE and apoA-1.

The efflux of CHO across cellular membranes is promoted by HDL through the ATP-binding cassette (ABC) transporters ABCG1, but contrasting data are available about ABCA1 [27, 42]. Moreover the efflux can be sustained by aqueous diffusion or apoE and phospholipid complex [24].

The placenta contains several LP receptors supporting its ability in taking up CHO from maternal blood but this subject is still uncertain as are the procedures that need to be explained [30].

Fetal Fatty Acids and Triglycerides

FAs can be synthesized "*de novo*" by the fetus, including some satured FAs and monounsatured
FAs originating from glucose. This appears particularly in advanced gestation when there is a gradual shift from embryonic to fetal lipids maternally derived [43]. The relevance of FAs in the fetal development is suggested by their maternal circulation increase during gestation [44, 45] and were shown to correlate with fetal lipid con-

Fetal Fatty Acids and Triglycerides Transport

centrations and fetal growth [46–48].

The energy provided by FAs is available from the maternal circulation in two different forms: FAs in their free form bound to albumin, or TGs transported as LPs (Fig. 12.4).

FAs are hydrolized by lipase activity that produces free fatty acids (FFAs). FFAs can also be directly uptaken from the maternal circulation then enter the placenta and trophoblasts cells through passive diffusion or by membrane carrier proteins to reach the fetus [13, 49–51]. When taken up by trophoblast cells FFAs are either transported transcellulary to the basolateral membrane, via an undefined mechanism, or utilized by the placenta itself for energy or as membrane substrates [6, 52].

Whenever LPs cross the placenta with difficulty [53], TGs could anyway be available to the fetus [6]. Maternal TGs carried by plasma LPs can be taken up intact by the placenta or undergo enzymatic lipase activity as shown in the placenta. Two lipases achieve FFAs from maternal circulating LPs containing TGs: the lipoprotein lipase (LPL) and the endothelial lipase. LPL shows the main relevant TGs lipase activity, is abundant in the human placenta and ensures that TGs are hydrolyzed into FFAs; the endothelial lipase is a phospholipase with little triacylglycerol lipase activity. FAs are then re-esterified to synthesize glycerolipids thus providing an energy reservoir in the placenta [52]. Glycerolipids undergo a further hydrolysis to allow FAs to be released into the fetal blood bounded to the alfafetoprotein to be rapidly transported to the fetal liver.

A relatively high placental transfer of lipids is related to the fetal requirement of some essential FAs to satisfy their increased request during gestation, as the fetus cannot synthesize them by itself. These are recognized as linoleic acid (LA), α -linolenic acid (ALA) and long chain polyunsatured fatty acids (LC-PUFA), in particular docosahexaeoic acid (DHA). LC-PUFA are mainly transported associated with plasma TG-rich lipoproteins as TGs rather than as FFAs, as demonstrated in humans [54]. DHA is particularly enriched in maternal plasma phospholipids (especially phosphatidylcholine) while ALA is present in TGs but not in phospholipids [55]. The correlation between maternal and late gestation fetal levels supports the mother to fetus transport of essential fatty acids with origins from the maternal diet and metabolism. An adequate availability of LC-PUFA to the fetus is clearly needed to preserve the normal fetal growth and this mechanism is preserved by the development of maternal hyperlipidaemia through pregnancy [6].

Fetal Lipoproteins and Molecular Mechanisms

Fetal LP levels in plasma were quantified during normal human gestation showing marked fluctuations (Table 12.1). The latter should be referred to the gestational age influence on fetal CHO levels with a strong inverse correlation. Early in the gestation Johnson [56] found that total plasma CHO levels were high thus suggesting a rapid biosynthesis rate of lipoprotein containing CHO occurring in the fetal liver (Table 12.2). As well CHO levels resulted significantly and directly correlated with maternal concentrations in fetuses younger than 6 months [6].

Table 12.1 Change in LP levels and gestational age

Week of gestation	CHO (mg/dl)	LDL CHO (mg/dl)	HDL CHO (mg/dl)
31–32	68.0 ± 7.0	44.0±5.0	24.0 ± 2.0
33–34	73.0 ± 7.0	49.0 ± 6.0	24.0 ± 2.0
35-36	65.0 ± 7.0	35.0 ± 3.0	22.0 ± 4.0
37–38	64.0 ± 4.0	37.0 ± 3.0	23.0 ± 2.0
39–40	56.0 ± 2.0	30.0 ± 2.0	22.0 ± 1.0
41-42	53.0 ± 3.0	28.0 ± 2.0	22.0 ± 1.0

Human fetal adrenal glands make use of LPs containing CHO, and in particular LDL, as a substrate for steroid hormones production. Further LDL levels are inversely related to plasma concentrations of the major fetal adrenal secreted hormone: the dehydroepiandrosterone sulfate. The early CHO decrease in plasma fetal levels is concurrent with the adrenal gland size increase as happens through the 12–20 weeks of the gestation so providing a putative cause [57].

Some major variations characterize fetal and maternal LPs (Fig. 12.5). LDL and VLDL are poorly represented in fetal blood [58] while HDL represents the main lipoprotein class in cord blood [59, 60]. The fetal transport of CHO by HDL accounts at least the 50 % of the whole fetal pool,

Table 12.2 Fetal cholesterol levels through the gestational period

Week of gestation	CHO (mg/dl)
10–16 (n=68)	85.4 ± 30.7
16.5-20 (n=19)	39.9 ± 21.0
26.5–32 (n=17)	67.8 ± 5.8
32.5–36 (n=16)	58.8±13.6
36.5–40 (n=44)	51.4±11.5

while LDL is the mainly transporter in the mother [61]. Fetal HDL differs from adult ones for physical-chemical properties. First the HDL₂ sub-fraction is mainly represented in fetal blood whereas HDL₃ sub-fraction is more prevalent in adults [62]. Second the fetal HDL apoprotein composition shows high apoE contents [62] and apoA-4 enrichment. ApoE is a relevant player in LP metabolism as interacts with cell surface receptors [63] and is mainly carried by HDL than TG rich LPs [61] so it is likely that the main role of apoE is to participate to the HDL metabolism. For instance apoE may facilitate the uptake of HDL by the fetal liver. ApoA-4 shows great structural similarities with apoA-1, which does not efflux CHO from the trophoblast [51], thus being postulated apoA-1 and apoA-4 to be exchangeable in the fetus [64]. The fetal HDL apoproteins profile gives an explanation for changes in their functions including an increased atheroprotective effect in the feto-placental vasculature. This effect could be explained by the role of apoA-4 and apoE in promoting lecithin:cholesterol acyl transferase (LCAT) activation [65]. To the antiatherogenic effect contributes the lower fetal colesteryl ester



Fig. 12.5 LPs, apoprotein and enzyme activity variation in fetal plasma compared to maternal plasma

transfer protein (CETP) activity accounting for larger HDL subfraction [66] and the phospholipid transfer protein that increases particle's efflux capacity [67]. Fetuses exhibit major modifications of their HDL proteome in addition to the quantitative decrease of HDL-TGs and increase of HDL CHO levels [65].

FA transfer proteins expressed by the placenta include FA transport proteins (FATP1-4, FATP6), FA translocase protein (FAT/CD36), and plasma membrane fatty-acid-binding protein (FABPpm) [68, 69] which are devoted to secrete into the fetal blood the maternally derived FFAs [70]. FABPpm shows a selective action addressed to LC-PUFA transport to the fetus whenever in smaller quantity with respect to TGs.

Fetal genome and in particular APOE and LPL genes were demonstrated to modulate the maternal LP phenotype when particular polymorphisms change related are considered. In the presence of APOE2 isoforms the fetal liver uptake of HDL may be reduced [71]. Following metabolic changes include the depletion of intrahepatic CHO content and the reduction of LDL CHO concentration as occurs in postnatal life. Otherwise fetal genetic polymorphisms produce lipid phenotypes that are contrary to those observed in adults as the case of APOC3 (APOC3*S2 lowers LDL in fetus while increase LDL in adults). Furthermore the effects of fetal polymorphisms are strongly modulated by maternal polymorphisms suggesting that LP effects of these polymorphisms may differ before and after birth [33]. These data confirm a fetal contribution to maternal LP metabolism through the pregnancy [72]. Furthermore maternal genetic polymorphisms seems to impact on fetal LP levels, independently of maternal LPs and of fetal genome.

Fetal Consequences of Maternal and Fetal Disorders

The altered feto-maternal LP metabolism and exchanges exit in different degree disorders ranging from inability to the embrio implantation, leading to heavy malformations and abortion or
 Table 12.3
 Fetal outcome related to fetal and/or maternal lipoprotein metabolism disorders

Maternal disorders	Fetal consequences	
Familial	High LDL CHO	
hypercholesterolemia	Preterm birth	
Pre-eclampsia	Altered placental transfer of lipids	
	High TGs	
Gestational diabetes mellitus	LGA	
	High FFAs	
	Low TGs	
Diabetes	Malformation and macrosomia	
	High VLDL and LDL	
Fetal disorder		
IUGR	Low CHO, LDL CHO, HDL CHO	
	High TGs	
SLOS	Central nervous system anomalies	
	Deficient CHO	

minor disorders which impact on the future life. Disorders related are both of maternal or fetal origin (Table 12.3): the former include the Familial Hypercholesterolemia (FH), Pre-eclampsia and Diabetes; the latter the Intrauterine Growth Restriction (IUGR) and defects of CHO biosynthetic pathway.

Maternal Disorders

Familial Hypercholesterolemia

Pregnant women affected by FH show higher CHO levels [73] than non-FH pregnant women whenever both show a similar percentage rise [15]. As well LDL CHO levels in the cord blood of FH newborns are higher if compared with non-FH newborn ones. On the contrary cord blood TGs and HDL CHO levels were similar to those of controls [74]. FH women gave birth to normal weight infants so indicating the good nutritional status of mothers and no impact on the fetal growth [75]. Furthermore epidemiological data indicate that maternal hyperlipidemia in pregnant FH women is associated with a more procoagulant profile and changes in fetal-uteroplacental circulation, suggesting the fetus being at risk of preterm birth [76]. A study on over 2000 individuals showed that FH patients born from an FH mother present with higher LDL CHO levels than those who inherited FH from their father [77]. This observation open the discussion toward the atherogenic LPs profile of FH pregnant women, on deleterious effects on their offspring considering the risk of CVD later in life, and on the need of a treatment while on pregnancy of FH woman.

Pre-eclampsia

Pre-eclampsia (PE) is a multiple system disorder that affects the mother and can adversely influence the feto-placental unit. PE is associated with placental dysfunction, oxidative stress [78], endothelial cell activation [79], and it is a cause of maternal and fetal morbidity. PE affected women demonstrate marked dyslipidaemia, hypertension and an increased systemic inflammatory condition potentially triggered by widespread endothelial dysfunction.

PE pregnancy is characterized by a proatherogenic lipid profile with increased TGs levels, HDL reduction and increased small dense LDL particles [80]. This altered LP metabolism is involved in the pathogenesis playing a role in the development of the disorder. The placental vascular bed of PE pregnant women shows acute atherosis and atherosclerotic placental lesions characterized by the accumulation of foam cells and perivascular cell infiltration. These abnormalities reduce the placental perfusion and placental/fetal hypoxia may develop [78]. Fetal lipid metabolism can be affected due to an altered placental lipid transfer but contrasting data are described. Rodie et al. [81] observed that CHO levels were higher in the umbilical cord blood from pre-eclamptic pregnant women (with respect to controls) while HDL values were unchanged, supposing that placental transport mechanisms could be upregulated. These results were not confirmed by Catarino et al. [50] who showed lower CHO and HDL values in the umbilical cord blood of PE pregnant women. This hypothesis is sustained by the evidence that LP receptor expression is decreased in the placenta of women with PE [82]. Moreover higher values of fetal TGs were observed in the umbilical cord blood associated with a significant increase in maternal blood TGs as a compensatory way to face the uteroplacental hypoperfusion [83].

Diabetes

Gestational diabetes mellitus (GDM) and maternal diabetes mellitus type 1 or 2 are characterized by high incidence of fetal macrosomia and neonates that are large for gestational age (LGA) [84]. Poor metabolic control of diabetes early in pregnancy is associated with an increased risk of fetal malformations [85]. Poorly controlled diabetes in the second half of pregnancy could exit in neonatal macrosomia. Fetus from mother under poor metabolic control shows an LP cord blood level increase when compared with fetuses from non-diabetic mothers. This case frequently occurs in diabetic mothers.

Maternal hypertriglyceridaemia and hyperglycaemia are a consequence of the augmented insulin resistant condition, these changes enhancing the substrate availability to the fetus. In GDM pregnancies, maternal lipids correlate with fetal lipids and fetal growth; the increase of mother LP concentrations could have effects on lipids transferred to the fetus by their intensified passage [85]. This process enhances the risk for oxidative stress and lipid peroxidation [86]. Also in pregnancies under well-controlled GDM, both maternal TGs and FFAs levels have been shown to correlate positively with neonatal weight and fat mass, indicating that maternal hyperlipidaemia in GDM actively enhances the availability of lipids to the fetus, contributing to his fat depot accumulation. Maternal FFAs and TGs levels predict also LGA birth weight and these values are linked with those measured in cord blood serum [48].

Fetal Disorders

Intrauterine Growth Restriction (IUGR)

Small for gestational age (SGA) newborns could be divided into two groups, depending on the causes of low birth weight. SGA neonates include those who are genetically small and IUGR. In this latter pathologic condition the fetus does not reach its genetically growth potential, with a growth velocity reduction and related fetal disorders. The current hypothesis about pathogenesis include the insufficient trophoblast development that may lead to atheroclerotic placental lesions. Thus IUGR has similar placental pathology as PE [87].

Authors evidenced lower maternal LDL and CHO concentrations in pregnancies complicated by IUGR but few data are available about fetal LPs in IUGR condition. Pecks et al. [88] showed a significant decrease of HDL and LDL CHO levels in the umbilical cord blood of IUGR pregnant women. Furthermore authors underlined the current increase of oxLDL/LDL ratio which was negatively correlated to HDL concentrations. On the contrary TG levels were significantly increased.

IUGR fetuses show a proatherogenic profile. Based on Baker's hypothesis, change in lipid concentrations may represent one of the pathogenic links between low birth weight for gestational age and subsequent cardiovascular events in adulthood. This suggests a metabolic programming in intrauterine environment resulting from placental insufficiency [88].

Defects in the Cholesterol Biosynthetic Pathway

Seven known disorders involving enzyme defects in post-squalene cholesterol biosynthesis have been identified: desmosterolosis, X-linked dominant chondrodysplasia punctata, CHILD syndrome, lathosterolosis, hydrops-ectopic calcification-moth-eaten skeletal dysplasia, Antley– Bixler syndrome, Smith–Lemli–Opitz syndrome. The most common is the Smith-Leimli-Opitz syndrome (SLOS), while the other six syndrome are extremely rare and often lethal.

SLOS is a congenital multiple anomaly/intellectual disability syndrome caused by a deficiency of CHO synthesis resulting from an inherited deficiency of 7-dehydrocholesterol (7DHC)reductase, encoded by DHCR7 gene. The enzyme catalyses the last step of CHO biosynthesis, the conversion of 7DHC to CHO. As a result deficient CHO levels are produced while the precursor 7DHC and derivatives accumulate both during embryonic development and after birth. Tissues (especially brain) deprived of CHO, or because of sterol precursors and derivatives deposit, develop abnormally and function poorly. Substitution of 7DHC for CHO alters the lipid raft stability, protein compositions and decreases membrane bending rigidity. The precocious altered sterol composition in SLOS affects the physical and chemical properties as well as the function of cellular membranes. These changes are causative of signal transduction [89]. IUGR is the most frequent ultrasound finding, detected in 67-100 % of affected fetuses [90, 91]. SLOS affected newborns have a distinctive appearance with specific facial dysmorphism and suffer from multiple congenital anomalies including cleft palate, congenital heart disease, genitourinary abnormalities, and malformed limbs. They often manifest mentally retardation with significant central nervous system anomalies [92, 93].

Conclusion

Lipoprotein metabolism in human fetus is uncompletely understood but it is finally clear that a strict relationship between mother and fetus LP phenotypes effect pregnancy outcome and mother and fetus well-being. This correlation is related to the genetic fetal and maternal background and LP polymorphisms associated, besides environmental conditions.

Main key points concern the relevance of fetus CHO pool and FAs that are critical to the growth rate, the fetus as auxotrophic human being for CHO and lipid synthesis and the LP transport as feasible from maternal blood through the placenta. CHO transport in the fetus is mainly supported by HDL subclass and among FAs LC-PUFA uptake by the placenta is preferential. The transport is made available by transporters or by acqueous diffusion including both CHO and FAs. Considering the CHO efflux from the placenta to the fetus this is mainly provided by HDL or by simple diffusion while mechanisms regarding FFAs efflux are not finally established. Further studies aimed to ascertain actually unrecognized physio-pathological mechanisms are requested whenever ethical issues should not be neglected.

References

- Fielding CJ, Fielding PE. Membrane cholesterol and the regulation of signal transduction. Biochem Soc Trans. 2004;32:65–9.
- Cooper MK, Wassif CA, Krakowiak PA, et al. A defective response to Hedgehog signaling in disorders of cholesterol biosynthesis. Nat Genet. 2003;33: 508–13.
- Baardman ME, Kerstjens-Frederikse WS, Berger RMF, et al. The role of maternal-fetal cholesterol transport in early fetal life: current insights. Biol Reprod. 2013;88:1–9.
- 4. Murphy CR. The plasma membrane transformation of uterine epithelial cells during pregnancy. J Reprod Fertil Suppl. 2000;55:23–8.
- Woollett LA. Maternal cholesterol in fetal development: transport of cholesterol from the maternal to the fetal circulation. Am J Clin Nutr. 2005;82:1155–61.
- Herrera E, Amusquivar E, Lopez-Soldado I, Ortega H. Maternal lipid metabolism and placental lipid transfer. Horm Res. 2006;65:59–64.
- Weissgerber TL, Wolfe LA. Physiological adaptation in early human pregnancy: adaptation to balance maternal-fetal demands. Appl Physiol Nutr Metab. 2006;31:1–11.
- Bertrand N, Dahmane N. Sonic Hedgehog signaling in forebrain development and its interactions with pathways that modify its effects. Trends Cell Biol. 2006;16:597–605.
- Innis SM. Dietary omega 3 fatty acids and the developing brain. Brain Res. 2008;1237:35–43.
- Wang Y, Zhao S. Vascular biology of the placenta. San Rafael: Morgan & Claypool Life Sciences; 2010.
- Di Cianni CG, Miccoli R, Volpe L, et al. Intermediate metabolism in normal pregnancy and in gestational diabetes. Diabetes Metab Res Rev. 2003;19(4):259–70.
- Tint GS, Salen G, Batta AK, et al. Correlation of severity and outcome with plasma sterol levels in variants of the Smith-Lemli-Opitz syndrome. J Pediatr. 1995;127:82–7.
- Gil-Sánchez A, Koletzko B, Larqué E. Current understanding of placental fatty acid transport. Curr Opin Clin Nutr Metab Care. 2012;15:265–72.
- Martin U, Davies C, Hayavi S, et al. Is normal pregnancy atherogenic? Clin Sci. 1999;96:421–5.
- Amundsen AL, Khoury J, Iversen PO, et al. Marked changes in plasma lipids and lipoproteins during pregnancy in women with familial hypercholesterolemia. Atherosclerosis. 2006;189:451–7.
- Tuckey RC. Progesterone synthesis by the human placenta. Placenta. 2005;26:273–81.
- Henson MC, Shi W, Greene SJ, Reggio BC. Effects of pregnant human, nonpregnant human, and fetal bovine sera on human chorionic gonadotropin, estradiol, and progesterone release by cultured human trophoblast cells. Endocrinology. 1996;137:2067–74.
- Knopp RH, Warth MR, Charles D, et al. Lipoprotein metabolism in pregnancy, fat transport to the fetus, and the effects of diabetes. Biol Neonate. 1986;50:297–317.

- Saarelainen H, Laitinen T, Raitakari OT, et al. Pregnancy-related hyperlipidemia and endothelial function in healthy women. Circ J. 2006;70:768–77.
- Pepe GJ, Albrecht E. Actions of placental and fetal adrenal steroid hormones in primate pregnancy. Endocr Rev. 1995;16:608–49.
- Desoye G, Schwenditisch MO, Pfeiffer KP, Zechner R, Kostner GM. Correlation of hormones with lipid and lipoprotein levels during normal pregnancy and postpartum. J Clin Endocrinol Metab. 1987;64:704–12.
- Herrera E, Lasunción MA, Gomez-Coronado D, et al. Role of lipoprotein lipase activity on lipoprotein metabolism and the fate of circulating triglycerides in pregnancy. Am J Obstet Gynecol. 1988;158:1575–83.
- Jurevics HA, Kidwai FZ, Morell P. Sources of cholesterol during development of the rat fetus and fetal organs. J Lipid Res. 1997;38:723–33.
- Avis HJ, Hutten BA, Twickler MT, et al. Pregnancy in women suffering from familial hypercholesterolemia: a harmful period for both mother and newborn? Curr Opin Lipidol. 2009;20:484–90.
- Haave NC, Innis SM. Cholesterol synthesis and accretion within various tissues of the fetal and neonatal rat. Metabolism. 2001;50:12–8.
- Baardman ME, Erwich JJHM, Berger RMF, et al. The origin of fetal sterols in second-trimester amniotic fluid: endogenous synthesis or maternal-fetal transport? Am J Obstet Gynecol. 2012;207:19–25.
- Larque E, Ruiz-Palacios M, Koletzko B. Placental regulation of fetal nutrient supply. Curr Opin Clin Nutr Metab Care. 2013;16:292–7.
- Linck LM, Hayflick SJ, Lin DS, et al. Fetal demise with Smith–Lemli–Opitz syndrome confirmed by tissue sterol analysis and the absence of measurable 7-dehydrocholesterol delta(7)-reductase activity in chorionic villi. Prenat Diagn. 2000;20:238–40.
- Nowaczyk MJM, Farrell SA, Sirkin WL, et al. Smith– Lemli–Opitz (RHS)syndrome: holoprosencephaly and homozygous IVS8-1G C genotype. Am J Med Genet. 2001;103:75–80.
- Parker Jr CR, Deahl T, Drewry P, Hankins G. Analysis of the potential for transfer of lipoprotein-cholesterol across the human placenta. Early Hum Dev. 1983;8:289–95.
- Witsch-Baumgartner M, Gruber M, Kraft HG, et al. Maternal apo E genotype is a modifier of the Smith– Lemli–Opitz syndrome. J Med Genet. 2004;41:577–84.
- 32. Wadsack C, Hammer A, Levak-Frank S, et al. Selective cholesteryl ester uptake from high density lipoprotein by human first trimester and term villous trophoblast cells. Placenta. 2003;24:131–43.
- 33. Descamps OS, Bruniaux M, Guilmot PF, et al. Lipoprotein concentrations in newborns are associated with allelic variations in their mothers. Atherosclerosis. 2004;172:287–98.
- Madsen EM, Lindegaard ML, Andersen CB, et al. Human placenta secretes apolipoprotein B-100-containing lipoproteins. J Biol Chem. 2004;279:55271–6.
- 35. Wittmaack FM, Gafvels ME, Bronner M, et al. Localization and regulation of the human very low

density lipoprotein/apolipoprotein-E receptor: trophoblast expression predicts a role for the receptor in placental lipid transport. Endocrinology. 1995;136:340–8.

- 36. Rindler MJ, Traber MG, Esterman AL, et al. Synthesis and secretion of apolipoprotein E by human placenta and choriocarcinoma cell lines. Placenta. 1991;12:615–24.
- Lopez D, McLean MP. Estrogen regulation of the scavenger receptor class B gene: anti-atherogenic or steroidogenic, is there a priority? Mol Cell Endocrinol. 2006;247:22–33.
- Woollett LA. The origins and roles of cholesterol and fatty acids in the fetus. Curr Opin Lipidol. 2001;12:305–12.
- 39. Napoli C, D'Armiento FP, Mancini FP, et al. Fatty streak formation occurs in human fetal aortas and is greatly enhanced by maternal hypercholesterolemia. Intimal accumulation of low density lipoprotein and its oxidation precede monocyte recruitment into early atherosclerotic lesions. J Clin Invest. 1997;100:2680–90.
- Woollett LA. Review: transport of maternal cholesterol to the fetal circulation. Placenta. 2011;32 Suppl 2:S218–21.
- Ethier-Chiasson M, Duchesne A, Forest JC, et al. Influence of maternal lipid profile on placental protein expression of LDLr and SR-BI. Biochem Biophys Res Commun. 2007;359:8–14.
- 42. Lindegaard ML, Wassif CA, Vaisman B, et al. Characterization of placental cholesterol transport: ABCA1 is a potential target for in utero therapy of Smith–Lemli–Opitz syndrome. Hum Mol Genet. 2008;17:3806–13.
- 43. Van Aerde JE, Feldman M, Clandinin MT. Accretion of lipid in the fetus and newborn. In: Polin RA, Fox WW, editors. Fetal and neonatal physiology. 2nd ed. Philadelphia: W. B. Saunders Co; 1998. p. 458–77.
- 44. Burt RL, Leake NH, Pulliam RP. Regulation of plasma NEFA in pregnancy and the puerperium. Preliminary observations. Obstet Gynecol. 1961;17:215–21.
- Herrera E, Ortega H, Alvino G, et al. Relationship between plasma fatty acid profile and antioxidant vitamins during normal pregnancy. Eur J Clin Nutr. 2004;58:1231–8.
- 46. Kitajima M, Oka S, Yasuhi I, et al. Maternal serum triglyceride at 24–32 weeks' gestation and newborn weight in nondiabetic women with positive diabetic screens. Obstet Gynecol. 2001;97:776–80.
- Nolan CJ, Riley SF, Sheedy MT, et al. Maternal serum triglyceride, glucose tolerance, and neonatal birth weight ratio in pregnancy. Diabetes Care. 1995;18:1550–6.
- Schaefer-Graf UM, Graf K, Kulbacka I, et al. Maternal lipids as strong determinants of fetal environment and growth in pregnancies with gestational diabetes mellitus. Diabetes Care. 2008;31:1858–63.
- 49. Haggarty P. Fatty acid supply to the human fetus. Annu Rev Nutr. 2010;30:237–55.
- Catarino C, Rebelo I, Belo L, et al. Fetal lipoprotein changes in pre-eclampsia. Acta Obstet Gynecol Scand. 2008;87:628–34.

- Schmid K, Davidson W, Myatt L, Woollett A. Transport of cholesterol across a BeWo cell monolayer: implications for net transport of sterol from maternal to fetal circulation. J Lipid Res. 2003;44:1909–18.
- Coleman RA, Haynes EB. Synthesis and release of fatty acids by human trophoblast cells in culture. J Lipid Res. 1987;28:1335–41.
- Herrera E, Lasunción MA. Maternal–fetal transfer of lipid metabolites. In: Polin RA, Fox WW, Abman SH, editors. Fetal and neonatal physiology. 3rd ed. Philadelphia: W.B. Saunders Co; 2004. p. 375–88.
- Herrera E. Lipid metabolism in pregnancy and its consequences in the fetus and newborn. Endocrine. 2002;19:43–55.
- 55. Otto SJ, Houwelingen AC, Antal M, Manninen A, et al. Maternal and neonatal essential fatty acid status in phospholipids: an international comparative study. Eur J Clin Nutr. 1997;51:232–42.
- 56. Johnsonh Jr J, Simpsonb ER, Carr P, et al. The levels of plasma cholesterol in the human fetus throughout gestation. Pediatr Res. 1982;16:682–3.
- Carr BR, Porter JC, Masnald PC, et al. Metabolism of low-density lipoprotein by human fetal adrenal tissue. Endocrinology. 1980;107:1034–340.
- Dolphin PJ, Breckenridge WC, Dolphin MA, Tan MH. The lipoproteins of human umbilical cord blood apolipoprotein and lipid levels. Atherosclerosis. 1984;51:109–22.
- Averna MR, Barbagallo CM, Di Paola G, et al. Lipids, lipoproteins and apolipoproteins AI, AII, B, CII, CIII and E in newborns. Biol Neonate. 1991;60: 187–92.
- Parker Jr CR, Carr BR, Simpson ER, MacDonald PC. Decline in the concentration of low-density lipoprotein-cholesterol in human fetal plasma near term. Metabolism. 1983;32:919–23.
- Nagasaka H, Chiba H, Kikuta H, et al. Unique character and metabolism of high density lipoprotein (HDL) in fetus. Atherosclerosis. 2002;161:215–23.
- Augsten M, Hackl H, Ebner B, et al. Fetal HDL/ apoE: a novel regulator of gene expression in human placental endothelial cells. Physiol Genomics. 2011;43:1255–62.
- 63. Herz J, Hamann U, Rogne S, et al. Surface location and high affinity for calcium of a 500-kd liver membrane protein closely related to the LDL-receptor suggest a physiological role as lipoprotein receptor. EMBO J. 1998;7:4119–27.
- 64. Sreckovic I, Birner-Gruenberger R, Obrist B, et al. Distinct composition of human fetal HDL attenuates its anti-oxidative capacity. Biochim Biophys Acta. 2013;1831:737–46.
- 65. Zhao Y, Thorngate FE, Weisgraber KH, et al. Apolipoprotein E is the major physiological activator of lecithin-cholesterol acyltransferase (LCAT) on apolipoprotein B lipoproteins. Biochemistry. 2005;44:1013–25.
- 66. Schaefer EJ, Asztalos BF. Increasing high-density lipoprotein cholesterol, inhibition of cholesteryl ester transfer protein, and heart disease risk reduction. Am J Cardiol. 2007;100:25–31.

- 67. Scholler M, Wadsack C, Metso J, et al. Phospholipid transfer protein is differentially expressed in human arterial and venous placental endothelial cells and enhances cholesterol efflux to fetal HDL. J Clin Endocrinol Metab. 2012;97:2466–74.
- Duttaroy AK. Transport of fatty acids across the human placenta: a review. Prog Lipid Res. 2009;48:52–61.
- 69. Haggarty P. Placental regulation of fatty acid delivery and its effect on fetal growth-a review. Placenta. 2002;23:S28–38.
- Hanebutt FL, Demmeimair H, Schiessl B, et al. Longchain polyunsaturated fatty acid (LC-PUFA) transfer across the placenta. Clin Nutr. 2008;27:685–93.
- Mahley RW, Huang Y. Apolipoprotein E: from atherosclerosis to Alzheimer's disease and beyond. Curr Opin Lipidol. 1999;10:207–17.
- 72. Descamps OS, Bruniaux M, Guilmot PF, et al. Lipoprotein metabolism of pregnant women is associated with both their genetic polymorphisms and those of their newborn children. J Lipid Res. 2005;46:2405–14.
- Potter JM, Nestel PJ. The hyperlipidemia of pregnancy in normal and complicated pregnancies. Am J Obstet Gynecol. 1979;133:165–70.
- Vuorio AF, Miettinen TA, Turtola H, et al. Cholesterol metabolism in normal and heterozygous familial hypercholesterolemic newborns. J Lab Clin Med. 2002;140:35–42.
- Toleikyte I, Retterstøl K, Leren TP, et al. Pregnancy outcomes in familial hypercholesterolemia: a registrybased study. Circulation. 2011;124:1606–14.
- 76. Khoury J, Amundsen AL, Tonstad S, et al. Evidence for impaired physiological decrease in the uteroplacental vascular resistance in pregnant women with familial hypercholesterolemia. Acta Obstet Gynecol Scand. 2008;29:1–5.
- 77. van der Graaf A, Vissers MN, Gaudet D, et al. The dyslipidemia of mothers with familial hypercholesterolemia deteriorates lipid levels in their adult offspring. Boston: Oral presentation at the International Atherosclerosis Society Conference; 2010.
- Serdar Z, Gur E, Colakodullary' M, et al. Lipid and protein oxidation and antioxidant function in women with mild and severe preeclampsia. Arch Gynecol Obstet. 2003;268:19–25.
- Var A, Kuscu N, Koyuncu F, et al. Atherogenic profile in preeclampsia. Arch Gynecol Obstet. 2003;268:45–7.

- Belo L, Caslake M, Gaffney D, et al. Changes in LDL size and HDL concentration in normal and preeclamptic pregnancies. Atherosclerosis. 2002;162:425–32.
- Rodie V, Caslake M, Stewart F, et al. Fetal cord plasma lipoprotein status in uncomplicated human pregnancies complicated. Atherosclerosis. 2004;176:181–7.
- 82. Murata M, Kodama H, Goto K, et al. Decreased very-low-density lipoprotein and low-density lipoprotein receptor messenger ribonucleic acid expression in placentas from preeclamptic pregnancies. Am J Obstet Gynecol. 1996;175:1551–6.
- Tabano S, Alvino G, Antonazzo P. Placental LPL gene expression is increased in severe intrauterine growthrestricted pregnancies. Pediatr Res. 2006;59:250–3.
- DeRuiter MC, Alkemade FE, Gittenberger-de Groot AC, et al. Maternal transmission of risk for atherosclerosis. Curr Opin Lipidol. 2008;19:333–7.
- Herrera E, Ortega-Senovilla H. Disturbances in lipid metabolism in diabetic pregnancy – are these the cause of the problem? Best Pract Res Clin Endocrinol Metab. 2010;24:515–25.
- Herrera E, Ortega-Senovilla H. Lipid metabolism during pregnancy and its implications for fetal growth. Curr Pharm Biotechnol. 2014;15:24–31.
- Sattar N, Greer IA, Galloway PJ, et al. Lipid and lipoprotein concentrations in pregnancies complicated by intrauterine growth restriction. J Clin Endocrinol Metab. 1999;84:128–30.
- Pecks U, Brieger M, Schiessl B, et al. Maternal and fetal cord blood lipids in intrauterine growth restriction. J Perinat Med. 2012;40:287–96.
- Porter FD. Smith–Lemli–Opitz syndrome: pathogenesis, diagnosis and management. Eur J Hum Genet. 2008;16:535–54.
- Goldenberg A, Wolf A, Chevy F, et al. Antenatal manifestations of Smith-Lemli-Opitz (RSH) syndrome: a retrospective survey of 30 cases. Am J Med Genet A. 2004;124:423–6.
- Quelin C, Loget P, Verloes A, et al. Phenotypic spectrum of fetal Smith-Lemli-Opitz syndrome. Eur J Med Genet. 2012;55:81–90.
- Irons M, Elias ER, Salen G, et al. Defective cholesterol biosynthesis in Smith-Lemli-Opitz syndrome. Lancet. 1993;341:1414.
- Nowaczyk MJ, Irons MB. Smith-Lemli-Opitz syndrome: phenotype, natural history, and epidemiology. Am J Med Genet C: Semin Med Genet. 2012;15:250–62.

Part III

Gene and Human Fetal Development up to Second Trimester

13

Gene Regulatory Networks and Epigenetic Modifications in Cell Fate Decisions During the Early Embryonic Development

Siddhartha Roy

Introduction

The fertilized zygote is totipotent, capable of differentiating into all the cell types present in an adult body as well as extra-embryonic tissues required for embryo development. The fertilized totipotent single cell divides, giving rise to the multi-cell morula and later the blastocyst. By the morula stage, the first differentiation events occur, giving rise to the trophectoderm and the Inner Cell Mass (ICM) [1]. Trophectoderm later gives rise to the trophoblasts, while the ICM develops into the embryo. The inner cell mass has a more restricted potency than the zygote in that it cannot give rise to trophoblast cells. Thus, these cells are referred to as pluripotent [2]. By the late blastocyst stage the inner cells mass further differentiates into the epiblast and the primitive endoderm layers (Fig. 13.1) [1]. Further down the embryonic development road, the epiblast gives rise to the germ layers, three layers of further differentiated cells, the ectoderm, the

Department of Biophysics, Bose Institute, P1/12 C.I.T. Scheme VIIM, Kolkata 700054, India e-mail: sidroykolkata@gmail.com mesoderm and the endoderm that ultimately give rise to all the cell types in the adult animal. Differentiated cells that originate from the zygote do not normally cross the lineage boundaries. Thus, they become committed to their lineage and restricted in potency.

How this continued restriction of potency occurs is not fully clear at the molecular level. However, recent path-breaking research is pointing toward primacy of gene regulation in this process [3, 4]. It has been known for a long time that the DNA sequence of the genome is identical in the totipotent cell and all the differentiated nucleated cells. Thus, different stages of cell differentiation differ only with respect to expression of genes. First conceptual model of cell differentiation was proposed by Waddington [5]. In this model, the pluripotent cells sit on top of a mountainous landscape, which describes some sort of downward gradient of potency. As the cell differentiates, the cells on the top of the mountain roll down and settle into various local minima; rolling down being equivalent to loss of potency and differentiation (Fig. 13.2). Although this model had an iconic influence in this field, it is largely metaphoric without giving any hint to the actual underlying mechanisms. In this article, we will explore how recent elucidation of regulation of gene expression and epigenetic regulation is throwing light on molecular mechanisms of cell differentiation at the early stages of development.

S. Roy, PhD

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_13



Fig. 13.1 Morphological and lineage specification steps that occur during early mouse embryonic development (The figure is reproduced from Krupinski et al. under Creative Common License [54])



Gene Regulatory Networks

Pioneering work by Gurdon and others on nuclear transfer indicated that differentiated cell nucleus contains all the information to revert back to a pluripotent state [6]. Much later, Yamanaka and co-workers created induced pluripotent cells from differentiated cells by expressing a small number of relevant transcription factors [7]. Thus, it is now well established that it is not the sequence of genes per se, but its expression that is the key to inducing a particular differentiated cellular state. A particular pattern of gene expression results in a particular cell type and this pattern of expression may be heritable over cell generations. Thus, in order to understand how cells differentiate, we delve into the present level of understanding of eukaryotic gene regulation.

Broadly, gene expression is primarily regulated at two different levels. At one level, genes are regulated by proteins that bind in a DNA sequence specific manner, called transcription factors [4]. The activity of a transcription factor dictates the state of expression of a gene that it regulates and in combination with other transcription factors determine the overall pattern of gene expression. This type of regulation is not heritable across cell generations. The second type of regulation occurs through modifications of DNA itself and its associated proteins and is one of the major mechanisms of the epigenetic regulation. The latter is often heritable across cell generations, giving a sense of quasi-permanency. The two different modes of gene regulation are not mutually exclusive and often deeply interrelated. Binding of transcription factors often lead to recruitment of chromatin modification machineries and consequently to changes in status of epigenetic modifications. Similarly, changes in chromatin modification may lead to

changes in accessibility of transcription factor binding sites. Thus, the two modes may be considered complementary and as two sides of the same coin.

Transcription factors rarely work alone in eukaryotes. Unlike in prokaryotes, eukaryotic gene regulation occurs through combinatorial binding of several transcription factors in the upstream regulatory regions of genes [8]. It is becoming apparent that one of the fundamental reasons for this phenomenon may lie in the fact that eukaryotic transcription factors by themselves lack sufficient DNA sequence specificity for the target site. The transcription factors have to operate and specifically recognize a small piece of DNA sequence in the backdrop of more than a billion non-target basepairs in the genome. This would require DNA sequence discriminatory ability of a very high order which is not reached and perhaps is not attainable by an isolated transcription factor [9]. Eukaryotes reach this level of discrimination by combinatorial binding of various transcription factors in the same gene regulatory region. Thus, we usually observe multiple transcription factors cooperating with each other, forming a network of gene regulatory proteins. Together with the genes they regulate and other involved players, they form the gene regulatory network [10]. Gene regulatory networks have the general characteristics of networks in general and may have different topologies. It is difficult to discuss the types and properties of gene regulatory networks in detail here and the reader is referred to some excellent literature on this [11-13]. However, we will focus on one relevant property which we think is crucial for the understanding of cell differentiation.

Gene regulatory networks are critical for understanding molecular nature of cell differentiation and development. External signals in the form of cytokines, positional information etc., are transduced through signaling pathways to gene regulatory networks. The gene regulatory networks are the final executors of cell fate decisions and within them one or a limited number of transcription factors, the master regulators, play a dominant role [14, 15]. These cell fate determining gene regulatory networks are required to exist in steady-states (changing concentrations will also change the occupancy of promoters and hence the gene expression) so as to dictate a particular pattern of gene expression. Existence of steady-states is dependent upon the nature of the network architecture [16, 17]. Many biological networks have architecture that fall into the class that contains a positive circuit [18, 19]. Having positive circuits is the necessary condition to have multiple steady-states, that is, possessing the property of multi-stationarity [20]. For gene regulatory networks that determine the cell fate decisions, access to multiple steady-states leads to the possibility of expressing multiple patterns of gene expression. Steady-states are important as in these states the concentrations of the transcription factors remain more or less invariant, leading to a stable pattern of gene expression. Expression of different combination of genes is possible if the target sequences have different affinities and change of activity of the transcription factor(s) will lead to different fractional occupancies and consequently expression of different genes. Such different classes of target sequences and their biological consequence have been described for transcription factors such as p53 [21]. Thus, external signaling leads to changes of transcription factor concentrations in the nucleus which leads to a different steadystate of one or more gene regulatory networks, resulting in different transcription factor occupancy and changed pattern of gene expression (Fig. 13.3). Clearly, understanding the gene regulatory networks that execute the cell fate decisions are of prime importance for understanding cell differentiation.

Chromatin Modifications and Cell Differentiation

As pointed out before, change in gene expression pattern as a consequence of changes in gene regulatory networks (i.e. resulting from changes in transcription factor occupancy) does not lead to heritable gene expression patterns per se. Resistance to change of gene expression patterns in response to sudden alterations in signaling



Fig. 13.3 A schematic figure of how external signals are transmitted to transcription factors via signal transduction pathways. The signal causes activities of relevant transcription factors to increase in the nucleus causing change in promoter occupancy of transcription factors of various genes. The *oval* shaped objects are the transcription factors, the *arrows* represent signal transduction pathways and the *round* objects represent the signaling ligands occupying respective receptors

pattern or stochastic fluctuations and its heritability across cell generations are important factors for maintaining the quasi-irreversible nature of lineage committed or differentiated cells. Such stability and heritability is offered by epigenetic chromatin modifications, which are thus essential components of lineage commitment and cell differentiation [22, 23].

Chromatins undergo a large number of modifications including that of the DNA and associated proteins [24]. Among the modifications, DNA methylations and histone acetylation and methylations are most well studied. DNA methylations are used to silence genes quasipermanently [25]. Histone modifications are generally used as softer marks, allowing reversions under certain conditions. Apart from their direct effects on transcription, the chromatin modifications also are important players in changing the way the chromatins are condensed, leading to further regulations of gene expression. Two major regions of chromatins which show variations in condensation properties can be readily identified: The condensed heterochromatin, containing mostly transcriptionally silent genes and the less condensed euchromatin and containing regions rich in genes that undergo active transcription. Some heterochromatin regions are fixed in all cell types, whereas some others vary from cell type to cell type. Thus, accessibility of genes to transcription factors and machineries could be restricted in the condensed heterochromatin segments, adding another layer of regulation [26].

Upon fertilization and formation of the zygote the genome undergoes de-methylation and by the morula stage it has reached the minimum. Re-methylation then starts. Beyond the initial developmental stage, further differentiation is accompanied by rewriting of epigenetic marks, suggesting that this form of regulation of gene expression plays a vital role in cell differentiation in the early developmental stages [27]. The precise mechanism of how epigenetic marks affect cell differentiation is still not well understood. Similarly, the relationship between the cell fate determining gene regulatory networks and the epigenetic regulation is not quite clear. In a previous article, we have attempted to connect these two modes of gene regulation and suggested that epigenetic marks restrict accessibility to certain steady-states of cell fate determining networks, thereby channeling cells to a particular lineage committed state [28]. Thus, a fuller understanding of cell fate determining gene regulatory networks, epigenetic gene regulation and their interrelationship is vital for elucidation of molecular mechanisms underlying early development.

Embryonic stem cells are derived from the inner cell mass of an embryo. Embryonic stem cells have been instrumental in developing many of our ideas about the pluripotent state and its lineage commitment. Their chromatin is believed to be in a hyper-dynamic state with very few regions in hetero-chromatinized state [29, 30]. Probably, as a consequence they are able to switch between different gene expression programs and they appear to do so in vitro, stochastically [31, 32].

Transcription Factors Involved in Differentiation of the Zygote

As mentioned above, the first segregation and differentiation event occurs in the morula stage. The trophectoderm and inner cell mass lineages segregate, with former giving rise to trophoblasts and the latter forming the epiblast and the primitive endoderm layers. The inner cell mass is pluripotent and it appears that the state of pluripotency is initiated and maintained largely by three transcription factors, Nanog, Oct4 and Sox2 [33]. It appears that Nanog also plays a central role in the differentiation of ICM [34]. Nanog belongs to the homeodomain class of transcription factors and has been used in induction of pluripotency from the differentiated cells. Oct4 and Sox2 are also important components of pluripotency maintenance [35]. Oct4 belongs to the POU family of transcription factors, whereas Sox2 has the HMG box binding domain. Oct4 and Sox2 bind to many promoters of pluripotency genes together with cooperativity.

The first differentiation step from zygote to trophectoderm and inner cell mass is dictated by a limited number of transcription factors [36], of which Oct4 and Cdx2 (dominant determinants for ICM and TE lineages, respectively) were thought to be the master regulators of cell fate; Oct4 priming the ICM lineage and Cdx2 priming the trophectoderm lineage [37-39]. They were reported to inhibit each other's function reciprocally, thus maintaining a balance [38, 40-42]. Appropriate signals may increase activity of one or the other, tilting the balance in favor of a particular cell fate. These could be the crucial components of the master regulatory switch regulating the earliest cell fate decision. However, some recent studies have raised doubts about the primacy of these transcription factors in the cell fate determination and other transcription factors may be involved. A very recent study has suggested that Arid3a could play important roles for determining the cell fate to trophectoderm lineages. Thus, it appears that a network of a number of transcription factors are involved in priming the cell fate decisions at this stage. A full elucidation of the network and its properties may be a very important goal in the near future.

Epigenetic modifications also play crucial role in lineage commitment of trophectoderm and inner cell mass [43]. Most notable among them is the histone methyl transferase, SETDB1 [44, 45]. In mouse, SETDB1 expression occurs in the inner cell mass which is initiated at the blastocyst stage. It methylates H3K9 and thus is a repressor of gene expression. It interacts with Oct4 and recruited by it to the appropriate sites. The targeted genes are likely to be involved in lineage commitment process of trophectoderm, such as Cdx2. The multiprotein complex NuRD [46] is also an important component of maintaining inner cell mass identity and prevents lineage commitment to trophectoderm. Embryonic stem cells containing defective NuRD shows inappropriate expression of trophectoderm specific genes such as Elf5 [47]. NuRD deficient embryonic stem cells can be easily converted to trophoblasts cells, suggesting that it is part of a barrier that prevent inappropriate conversion to trophectoderm lineage. Many other histone modifiers and chromatin remodelers are involved in the survival of inner cell mass, including Tip60 and MOF [48]. Clearly, the histone modifiers and chromatin remodeling proteins play an essential role in the early development.

Differentiation of Inner Cell Mass into Epiblast and Primitive Endoderm

The inner cell mass at the blastocyst stage differentiates into the epiblast and the primitive endoderm layers. Just prior to implantation, the inner cell mass starts segregating into two populations: The progenitor of the epiblast cells and the progenitors of the primitive endoderm cells. Recent studies suggest that two transcription factors play lineage determining roles, Nanog and Gata6. These two transcription factors seem to express in a mutually exclusive manner in the inner cell mass cells, with the cell expressing Nanog becoming the epiblast and cells expressing Gata6 proceeding to primitive endoderm. Although, these two transcription factors appear to be the master regulators in the cell fate determination at the inner cell mass level, there are other transcription factors involved in the gene regulatory network [49, 50]. It appears that Gata6 progressively activates Sox17, Gata4 and Sox7 [51,

52]. The segregation and differentiation of primitive endoderm and epiblast is coupled through the Fgf4 and Fgf receptor pathways [53].

Discussion and Conclusion

A major component of early development is lineage commitment of totipotent zygote to more differentiated cell types. We have described some of the underlying principles of this lineage commitment. Although, the principles are not fully elucidated, it is clear that the gene regulatory networks play the master regulatory role. Although they initiate the process of lineage commitment upon receiving the appropriate signals, the final execution depends on many other players acting downstream. Most important among these players are the enzymes that modify the chromatin components, particularly histones and DNA, which results in change of conformation of the chromatin. These changes result in a more permanent state, defining a differentiated cell type. All the players in either the regulatory networks or the chromatin modifiers are not fully known. However, we have described some players that are involved in fate decisions of the more differentiated cell types that emerge in the early development and expect that in the near future a full catalog of players involved will be known. It should then be possible to fully construct the gene regulatory networks, which should allow us to lay a fuller molecular and systems level understanding of the early development process.

Acknowledgement We acknowledge JC Bose National Fellowship, Department of Science and Technology, Govt. of India.

References

- Artus J, Chazaud C. A close look at the mammalian blastocyst: epiblast and primitive endoderm formation. Cell Mol Life Sci. 2014;71(17):3327–38.
- Roper S, Hemberger M. Defining pathways that enforce cell lineage specification in early development and stem cells. Cell Cycle. 2009;8:1515–25.
- Marikawa Y, Alarcón VB. Establishment of trophectoderm and inner cell mass lineages in the mouse embryo. Mol Reprod Dev. 2009;76:1019–32.

- Hobert O. Gene regulation by transcription factors and microRNAs. Science. 2008;319:1785–6.
- Waddington CH, Kacser H. The strategy of genes. London: George Allen and Unwin; 1957.
- Gurdon JB. Adult frogs derived from the nuclei of single somatic cells. Dev Biol. 1962;4:256–73.
- Takahashi K, Yamanaka S. Induction of pluripotent stem cells from mouse embryonic and adult fibroblast cultures by defined factors. Cell. 2006;126:663–76.
- Zinzen RP, Girardot C, Gagneur J, Braun M, Furlong EE. Combinatorial binding predicts spatio-temporal cis-regulatory activity. Nature. 2009;462:65–70.
- Slutsky M, Mirny LA. Kinetics of protein-DNA interaction: facilitated target location in sequencedependent potential. Biophys J. 2004;87:4021–35.
- Alon U. An introduction to systems biology. Boca Raton: Chapman & Hall/CRC; 2007.
- Junker BH, Schreiber F. Analysis of biological networks. Hoboken: Wiley-Blackwell Online Library; 2008.
- Karlebach G, Shamir R. Modelling and analysis of gene regulatory networks. Nat Rev Mol Cell Biol. 2008;9:770–80.
- Lewis TG. Network science: theory and applications. Hoboken: Wiley-Blackwell; 2011.
- Yang J, Mani SA, Donaher JL, Ramaswamy S, Itzykson RA, Come C, et al. Twist, a master regulator of morphogenesis, plays an essential role in tumor metastasis. Cell. 2004;117:927–39.
- Honda K, Yanai H, Negishi H, Asagiri M, Sato M, Mizutani T, et al. IRF-7 is the master regulator of type-I interferon-dependent immune responses. Nature. 2005;434:772–7.
- Richard A, Comet J-P. Necessary conditions for multistationarity in discrete dynamical systems. Discret Appl Math. 2007;155:2403–13.
- Thomas R, Kaufman M. Multistationarity, the basis of cell differentiation and memory. I. Structural conditions of multistationarity and other nontrivial behavior. Chaos. 2001;11:170–9.
- Conradi C, Flockerzi D, Raisch J. Multistationarity in the activation of a MAPK: parametrizing the relevant region in parameter space. Math Biosci. 2008;211:105–31.
- Xiong W, Ferrell JE. A positive-feedback-based bistable 'memory module' that governs a cell fate decision. Nature. 2003;426:460–5.
- Thomas R, Kaufman M. Multistationarity, the basis of cell differentiation and memory. II. Logical analysis of regulatory networks in terms of feedback circuits. Chaos: Interdiscip J Nonlinear Sci. 2001;11:180–95.
- Weinberg RL, Veprintsev DB, Bycroft M, Fersht AR. Comparative binding of p53 to its promoter and DNA recognition elements. J Mol Biol. 2005;348:589–96.
- 22. Guo G, Huss M, Tong GQ, Wang C, Li Sun L, Clarke ND, et al. Resolution of cell fate decisions revealed by single-cell gene expression analysis from zygote to blastocyst. Dev Cell. 2010;18:675–85.
- Delcuve GP, Rastegar M, Davie JR. Epigenetic control. J Cell Physiol. 2009;219:243–50.
- Kouzarides T. Chromatin modifications and their function. Cell. 2007;128:693–705.

- Smith ZD, Meissner A. DNA methylation: roles in mammalian development. Nat Rev Genet. 2013;14:204–20.
- Grewal SI, Elgin SC. Heterochromatin: new possibilities for the inheritance of structure. Curr Opin Genet Dev. 2002;12:178–87.
- 27. Smith ZD, Chan MM, Mikkelsen TS, Gu H, Gnirke A, Regev A, et al. A unique regulatory phase of DNA methylation in the early mammalian embryo. Nature. 2012;484:339–44.
- Roy S, Kundu TK. Gene regulatory networks and epigenetic modifications in cell differentiation. IUBMB Life. 2014;66:100–9.
- Meshorer E, Yellajoshula D, George E, Scambler PJ, Brown DT, Misteli T. Hyperdynamic plasticity of chromatin proteins in pluripotent embryonic stem cells. Dev Cell. 2006;10:105–16.
- Park S-H, Park SH, Kook M-C, Kim E-Y, Park S, Lim JH. Ultrastructure of human embryonic stem cells and spontaneous and retinoic acid-induced differentiating cells. Ultrastruct Pathol. 2004;28: 229–38.
- Gupta PB, Fillmore CM, Jiang G, Shapira SD, Tao K, Kuperwasser C, et al. Stochastic state transitions give rise to phenotypic equilibrium in populations of cancer cells. Cell. 2011;146:633–44.
- Macfarlan TS, Gifford WD, Driscoll S, Lettieri K, Rowe HM, Bonanomi D, et al. Embryonic stem cell potency fluctuates with endogenous retrovirus activity. Nature. 2012;487:57–63.
- Silva J, Nichols J, Theunissen TW, Guo G, van Oosten AL, Barrandon O, et al. Nanog is the gateway to the pluripotent ground state. Cell. 2009;138:722–37.
- 34. Sun LT, Yamaguchi S, Hirano K, Ichisaka T, Kuroda T, Tada T. Nanog co-regulated by Nodal/Smad2 and Oct4 is required for pluripotency in developing mouse epiblast. Dev Biol. 2014;392:182–92.
- 35. Loh Y-H, Wu Q, Chew J-L, Vega VB, Zhang W, Chen X, et al. The Oct4 and Nanog transcription network regulates pluripotency in mouse embryonic stem cells. Nat Genet. 2006;38:431–40.
- 36. Jedrusik A, Parfitt D-E, Guo G, Skamagki M, Grabarek JB, Johnson MH, et al. Role of Cdx2 and cell polarity in cell allocation and specification of trophectoderm and inner cell mass in the mouse embryo. Genes Dev. 2008;22:2692–706.
- Nichols J, Zevnik B, Anastassiadis K, Niwa H, Klewe-Nebenius D, Chambers I, et al. Formation of pluripotent stem cells in the mammalian embryo depends on the POU transcription factor Oct4. Cell. 1998;95:379–91.
- Niwa H, Miyazaki J-i, Smith AG. Quantitative expression of Oct-3/4 defines differentiation, dedifferentiation or self-renewal of ES cells. Nat Genet. 2000;24:372–6.
- 39. Palmieri SL, Peter W, Hess H, Schöler HR. Oct-4 transcription factor is differentially expressed in the mouse embryo during establishment of the first two extraembryonic cell lineages involved in implantation. Dev Biol. 1994;166:259–67.
- Dietrich J-E, Hiiragi T. Stochastic patterning in the mouse pre-implantation embryo. Development. 2007; 134:4219–31.

- 41. Niwa H, Toyooka Y, Shimosato D, Strumpf D, Takahashi K, Yagi R, et al. Interaction between Oct3/4 and Cdx2 determines trophectoderm differentiation. Cell. 2005;123:917–29.
- 42. Strumpf D, Mao C-A, Yamanaka Y, Ralston A, Chawengsaksophak K, Beck F, et al. Cdx2 is required for correct cell fate specification and differentiation of trophectoderm in the mouse blastocyst. Development. 2005;132:2093–102.
- Ng RK, Dean W, Dawson C, Lucifero D, Madeja Z, Reik W, et al. Epigenetic restriction of embryonic cell lineage fate by methylation of Elf5. Nat Cell Biol. 2008;10:1280–90.
- 44. Cho S, Park JS, Kwon S, Kang Y-K. Dynamics of Setdb1 expression in early mouse development. Gene Expr Patterns. 2012;12:213–8.
- 45. Bilodeau S, Kagey MH, Frampton GM, Rahl PB, Young RA. SetDB1 contributes to repression of genes encoding developmental regulators and maintenance of ES cell state. Genes Dev. 2009;23:2484–9.
- Ahringer J. NuRD and SIN3: histone deacetylase complexes in development. Trends Genet. 2000;16:351–6.
- 47. Latos PA, Helliwell C, Mosaku O, Dudzinska DA, Stubbs B, Berdasco M, et al. NuRD-dependent DNA methylation prevents ES cells from accessing a trophectoderm fate. Biol Open. 2012;1:341–2. doi:10.1242/ bio.2012513.
- Hu Y, Fisher JB, Koprowski S, McAllister D, Kim MS, Lough J. Homozygous disruption of the Tip60 gene causes early embryonic lethality. Dev Dyn. 2009;238:2912–21.
- Plusa B, Piliszek A, Frankenberg S, Artus J, Hadjantonakis A-K. Distinct sequential cell behaviours direct primitive endoderm formation in the mouse blastocyst. Development. 2008;135:3081–91.
- Chazaud C, Yamanaka Y, Pawson T, Rossant J. Early lineage segregation between epiblast and primitive endoderm in mouse blastocysts through the Grb2-MAPK pathway. Dev Cell. 2006;10:615–24.
- 51. Niakan KK, Ji H, Maehr R, Vokes SA, Rodolfa KT, Sherwood RI, et al. Sox17 promotes differentiation in mouse embryonic stem cells by directly regulating extraembryonic gene expression and indirectly antagonizing self-renewal. Genes Dev. 2010;24:312–26.
- 52. Takash W, Cañizares J, Bonneaud N, Poulat F, Mattéi M-G, Jay P, et al. SOX7 transcription factor: sequence, chromosomal localisation, expression, transactivation and interference with Wnt signalling. Nucleic Acids Res. 2001;29:4274–83.
- 53. Kuijk EW, van Tol LT, Van de Velde H, Wubbolts R, Welling M, Geijsen N, et al. The roles of FGF and MAP kinase signaling in the segregation of the epiblast and hypoblast cell lineages in bovine and human embryos. Development. 2012;139:871–82.
- Krupinski P, Chickarmane V, Peterson C. Simulating the mammalian blastocyst – molecular and mechanical interactions pattern the embryo. PLoS Comput Biol. 2011;7:e1001128.
- Mitchell K. The genetics of brain wiring: from molecule to mind. PLoS Biol. 2007;5:e113.

Personal Human Life Begins with the Formation of Adult Type Hippocampus at 13th Week of Development

M. Kemal Irmak

When Does Personal Human Life Begin?

One of the most controversial topics in modern bioethics, science, and philosophy is the beginning of personal human life (personhood). Many philosophers and scientists have argued about the definition of personhood and the time for the beginning of a personal human life, however an acceptable explanation has not been provided yet. The consequences of this discussion are vitally important, as they may help to articulate more adequate arguments on some bioethical issues, like the definition of the moral status of embryos, abortion and embryo research. The purpose of the present chapter is therefore to propose when, in the course of normal development, a personal human life begins.

The availability of embryonic stem cells may open novel avenues for medical treatment of otherwise incurable diseases [1]. Embryonic stem cells are able to make any cell except placental cells and are also immortal [2]. However, the generation of human embryonic stem cells sometimes requires the destruction of early human embryos. This raises the same ethical questions and conflicts that are often heard when the ethics

M.K. Irmak

Gulhane Military Medical Academy, High Council of Science, Ankara, Turkey e-mail: mkirmak@gata.edu.tr of abortion are discussed. Some people regard an embryo to be a full human person; it has all of the rights of any citizen - including the right to life; thus any procedure that injures or kills an embryo is murder of a human. Most people however regard the beginning of individual human life to occur much later in pregnancy; thus, killing a recently fertilized embryo is not murder of a human person [3]. What is the status of the embryo at several days old? Is the embryo alive? Yes, from its beginning the embryo is cellular and it is alive, no one questions this. But, is the embryo human? If we could catch the embryo before it reaches a stage of being judged human, we would take the embryonic stem cells without any concern. Seeing that monozygotic twinning can occur as late as day 14 after conception and such identical twinning will produce two individuals with different lives, this could be proposed as a pre-embryonic stage where the single individual person is not yet fixed [4]. Today, embryo research is allowed up to 14 days in United Kingdom after which the splitting and the forming of twins is no longer possible [4].

Most religious traditions hold that what makes one a person is the possession of a soul [4, 5]. When the body meets with the soul, it comes to be a human person, with all the attendant rights, especially his basic right to life. However, the exact gestational age at which ensoulment is believed to happen is debatable. Common views include ensoulment happening

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_14 14

[©] Springer International Publishing Switzerland 2016

at the moment of conception, at the formation of brain and at the time of viability [5, 6]. In all religious groups abortion has always been regarded as sinful. However, for many centuries the termination of a pregnancy at an early stage carried lesser penalties than one later. This was related to the view that the human soul does not enter the embryo until 40 days or so after conception [5, 7]. Most religious groups thus, make a distinction between the moral status of the unformed and formed embryo, and think of the human person in the full sense, coming only with a delayed ensoulment [6].

Brain Cortex and Personhood

What makes human beings human and when does personal human life begin? From a medical point of view the function of the brain is fundamentally linked to being human. The brain controls almost all functions of the body and determines its psychological makeup [6]. Legal definition of death is based on the absence of brain activity, even though all the rest of the body is viable. Children born with an encephaly can never live a personal human life. Indeed, a living body without a brain would be a mass of cells without personhood. A conglomeration of cells in the early phase of pregnancy can hardly be characterized as a human person. Therefore personhood cannot occur until it has something that counts as a brain. The human identity, personality, and worth is associated with the functioning of the brain, so only when the brain is developed can there be any talk about an unborn human being [8].

In elaborating on this theory, Goldenring [9] defended the 8-week mark as the starting point for personal human life, indicating in his view the point at which there is integration of the brain as a whole. Kushner [10] argues that the initiation of brain activity is the most reasonable time at which to fix the start of life. Similar arguments are put forward by Shea [11] who recognizes that a new human life comes into being when the newly developing body organs and systems begin to function as a whole under the direction of a

functioning brain at 8 weeks of gestation. Emphasis is placed on a point during development when brain tissue begins to function. Brain death is the cessation of function of a brain; brain birth is the very gradual acquisition of function of a developing neural system. Is this developing neural system a brain? At some point, it must become a brain, but at what point? In particular, can we legitimately call it a brain at around 8 weeks' gestation?

The fetal brain develops very gradually over time from a comparatively simple to a more complex structure that comprises a number of functionally differentiated neurological components, including the critical cortex. Brain cortex is the outer layer of cerebral hemispheres and it is responsible for sensory perception, movement, language, thinking, memory and consciousness [12]. Brain cortex is also necessary for the personhood [6, 13]. The development of different parts of brain begins at different stages in fetal development [14, 15]. The structure of the adult cortex is highly complex and specialized and a great deal of change must occur during fetal development before an adult-like cortical structure is recognizable. However, the cerebral cortex is an extremely rudimentary structure at 8 weeks of gestation on which so much emphasis is placed [15]. At this point, we must remember the existence of two different types of cerebral cortex and two different stages of cortex development. The bulk of the cerebral cortex is a six-layered structure called neocortex (young cortex). The remaining of the cortex around the hilus of the cerebral hemisphere is known as allocortex (other cortex) [12]. Allocortex has a relatively elementary structure with three basic layers and is composed of the olfactory cortex and hippocampal formation [12].

The neocortex undergoes a long period of differentiation and maturation and neocortical 'life' does not begin while the neocortex is getting organized [15]. Neocortical birth could be located after the 24th weeks of gestation when cortical organization in the fetus begins to resemble that of adult type [15]. During the early fetal period and midgestation, the individual neocortical layers are not easily identifiable since there is a continuous influx of new neurons into the cortical plate and this plate does not contain synapses until 22–24 weeks of gestation. The six-layered adult laminar pattern gradually appears after 24 weeks of gestation owing to the progressive differentiation of the cortical plate [16–18]. Thus, neocortical birth could be located at the occurrence of the first minimal level of structural organization of neocortex which starts after the 24th week of gestation [19].

Hippocampal Development

The term allocortex is applied to the part of the brain that consists of a rim of cortical tissue around the hilus of the cerebral hemisphere [12]. It is the first part of the cortex to initiate differentiation from the eighth week onward and it forms a continuous, almost circular strip on the medial and inferior aspects of the hemisphere before 13th week of development [20–22]. There are two allocortical formations in the mammalian brain: the hippocampal formation and the olfactory cortex [12]. At first, olfactory cortex occupies a large part of the basal aspect of the brain. Later in development, the olfactory cortex becomes restricted to a small part of the brain and hippocampus becomes the main structure of the adult allocortex [23, 24].

Hippocampal formation is a large C-shaped allocortical structure in the medial aspects of the temporal lobe [12]. It begins to develop by 8-week gestation and one of the earliest sulci identified by fetal MRI is the hippocampal sulcus [25]. In the ninth week, the hippocampal primordial can be identified on the medial aspect of the developing cerebral hemisphere [26]. The hippocampal formation of adult type may be recognized at MRI of 13th week of development [26]. Most pyramidal cells are generated by the 13th week [27] and the subsequent period is characterized by an increase in the volume of the hippocampal formation. As most of the pyramidal cells have already formed and are currently undergoing the process of differentiation, the cellular components that increase in number thereafter are mainly of glial origin [26]. Hippocampal connections also develop prior to neocortical pathways, and reciprocal entorhinal-hippocampal projections are among the first cortico-cortical connections to be established in the human brain [28].

Hippocampal Birth at 13th Week of Development

The most striking feature of the early human fetal brain is the differentiation of hippocampal formation, and the intense expression of functional proteins in these areas [29–32]. As seen in the previous section, hippocampal cortex reaches its final structure at an earlier stage than the neocortex, and it is already developed [20-22, 25, 26, 32–34] at 13th week of development. From now on, the macroscopic aspect of the hippocampus remains unchanged until birth, and its morphology in the early fetal period, neonatal and pediatric periods is quite similar [20, 26]. Thus, three-layered adult laminar pattern of hippocampus appears around 13th weeks [21, 22] and hippocampal birth could be located at the occurrence of the first minimal level of structural organization of hippocampus at the 13th week of development.

Hippocampal formation plays a key role in memory and emotion. It acts as a comparator of novel and familiar stimuli and in the initiation or inhibition of behavioral strategies as appropriate to the situation. It therefore becomes linked to emotions such as anxiety and to setting appropriate motor responses [29]. Elaboration of hippocampal functions is also associated with the development of self-awareness, consciousness of the self [9], and hippocampus seems to be essential for the preservation of identity [12]. Thus, hippocampus can be regarded as the seat of emotion, memory and personhood.

We propose here a hippocampal birth theory suggesting that a personal human life or personhood cannot begin until the development of the hippocampus. Hippocampus is reliably present at about 13th weeks' gestation, and formation of hippocampus would indicate that the second level of life has commenced, and therefore could be taken to herald the beginning of a personal human life. We can say that the informational capacity of the zygote and early embryo is not sufficient to direct the development of anything personal, and is not sufficient to constitute a genetically stable subject as a human being. At the 13th week, fetus has distinctive human characteristics and possesses the primordia of all the internal and external organs and parts. An adult type hippocampus also emerges at about 13 weeks' gestation, and the fetus could be considered a personal human being starting then. Hippocampal birth, the emergence of a mature hippocampus is therefore proposed as a reasonable time to demarcate the beginning of personal human life (personhood). Prior to 13 weeks, existence is limited to cells, organs, and organ systems which have the potential for integration into a full human organism. Standardization and widespread acceptance of a definition of "hippocampal birth" would potentially clarify many medical and legal questions regarding elective abortion and embryo research.

There must be a point during development following fertilization at which it is reasonable to claim that the organism changes from nonperson to person. When does this special point in development occur? Seifert treats the human body as a composite of a biological organism and an intellective soul [35]. If we accept that the soul essentially has the capacity for personhood, it seems that the soul does not begin to exist until there occurs an appropriate seat for the soul in the fetal brain. This position would seem to require that the immortal soul only be infused into a fetus with sufficient cortical development. We suggest therefore that the soul is added to the already-existing physical body, when the hippocampus is formed at 13th week [36] even in a pregnancy occurred by self-fertilization [37, 38]. Prior to this point, there exists a cluster of biological cells which has the potential to develop as one human organism [39]. This alternative is compatible not only with the facts of modern medicine but also with the traditional understanding of ensoulment defended by many theologians who argue that the peculiarly human soul is not incarnated until there is appropriately an organized matter [1].

Soul Must Have a Material Component

Mental functions are powers that the soul has exercised by means of the physical entity called the brain [40]. A question is raised: "How is it possible for an immaterial substance to interact with a physical body?" [40]. No experimental data can be sufficient to bring us to the recognition of a soul, but there must be a substance as the basis of personal identity, for without spaceoccupying substance, there would be no way to account for the soul's ability to interact with the body [41, 42]. It was suggested that the soul substance consists of cosmological dark matter [43]. The dark matter is a universal connecting medium, filling all space to the furthest limits, penetrating the interstices of the atoms without a break in its continuity. So completely does it fill space that it is sometimes identified with space itself, and universe is built up in this fluid and move through a sea of it [43]. Astrophysical observations indicate that dark matter constitutes most of the mass in our universe, but its nature remains unknown [44]. It is called dark matter since it neither emits nor absorbs light. The existence of dark matter is inferred by its gravitational effects on ordinary matter and radiation [45, 46]. With the conception that the dark matter is the primary form of all substance, that all other forms of matter are merely differentiations of it, then it seems that soul substance which is in this life linked organically with the body can be identical with the dark matter. The soul is likely to work into man's physical body directly via that dark matter [43]. Evidence of the existence of dark matter has been found in large high-energy particle accelerators at CERN and Fermilab (Illinois) [47].

Vomeronasal System as a Point of Entry for the Soul Together with Dark Matter

We thought that while the soul has a material component (dark matter), there must be an open window to the brain for the entrance of the soul with dark matter. In this respect, vomeronasal organ (VNO) which is found in the nasal cavity and which has connections with the brain only between the 12th and 14th weeks of human development – a period including the time of ensoulment at the 13th week – seems to be the most appropriate window through which the soul and dark matter can enter the brain [48]. Actually, VNO is said to be the place in the body where the nervous system is closest to the external world [49]. Sensory cells of the vomeronasal system (VNS) are located within the sensory epithelium of the VNO, bipolar in shape with a single dendrite and an axon originating from its soma. The dendrite reaches the surface of the lumen of the VNO to form a dendritic terminal that bears microvilli [50]. The axon leaves the sensory epithelium, forming the vomeronasal nerve with neighboring axons, traveling toward the brain, and terminate on dendrites of second-order neurons in the accessory olfactory bulb [49, 50]. Axons of the second-order neurons of the VNS make close connections with the amygdala and hippocampus [49–51], the seat of emotion, memory and personality; shortly the seat of the soul.

The vomeronasal organ is a fluid-filled, tubular structure located at the base of the nasal septum that opens into the nasal cavity via a duct at its anterior end [52]. It is a chemoreceptive structure with direct axonal connections to the accessory olfactory bulb in many terrestrial vertebrates [53]. Pheromones presumably bind to the vomeronasal organ and exert behavioral or physiologic responses, thereby allowing chemical communication between animals of the same species [54]. The effects of pheromones are thought to be mediated by signals from the accessory olfactory bulbs to the amygdala and hypothalamus [52]. The vomeronasal system is well developed and functional in adult animals [55], while human VNO becomes rudimental before birth [56]. VNO in the human embryo contains bipolar cells similar to the developing vomeronasal sensory neurons of other species, but the structure becomes more simplified later in development [57-63], having no obvious way of communication with the brain. In humans the VNO, including the vomeronasal nerve and associated

ganglion cells, is first recognizable at eighth week of development [64]. VNO is well developed during the 12-14th weeks of development [48], but VNO loses receptor cells and becomes a ciliated, pseudostratified epithelium after 14th week of age [65, 66]. Moreover, the vomeronasal nerve connecting the vomeronasal organ with the accessory olfactory bulb (AOB) degenerates between week 14 and 28 [67, 68] leaving the function of the human VNO unclear. The AOB which is a primary brain center for the VNS [51, 62], is present in human embryos and certain stages of fetuses, but becomes degenerated and it is not identifiable after 7 months [69]. These observations support the view that vomeronasal system functions mainly during the intrauterine period in humans, especially during the period of ensoulment [36, 70]. Thus the development of the vomeronasal structures seems to be limited to a restricted time frame in humans, when they play a role for the ensoulment [59]. It was proposed therefore that the human VNO has functions mainly during fetal development when the VNO, along with the vomeronasal nerve, contributes to the transfer of the soul and its dark matter to their proper sites in the brain [71].

Conclusion

Above considerations make it appear likely that the hippocampus is the primary center harboring the soul and the human life (personhood) begins at the 13th week of development with a delayed ensoulment. The soul may also have a component of dark matter; and they both enter the brain through the window of vomeronasal organ which is functional and has connections with the brain only during the time of ensoulment. Therefore lawgivers, philosophers, scientists and whoever related may consider the beginning of personal human life in their decisions and procedures as the 13th week of development. Before this period, embryo must be regarded as a cell cluster which is to be respected but not accorded absolute protection. The hippocampal birth theory suggests that an abortion before 13 weeks' gestation kills potential human life, whereas abortion at a later point terminates

actual human life (a person). This theory also says that society should not require funerals such a religious burial until the conception has reached 13 weeks of gestation. The hippocampal birth theory further offers a clear choice point for the ethics of embryo research: stem cell research with spare embryos produced during infertility treatment, or embryos formed specifically for research or therapeutic purposes (even by cloning), is ethically acceptable up to the point of the formation of hippocampus.

References

- 1. Glannon W. Tracing the soul: medical decisions at the margins of life. Christ Bioeth. 2000;6:49–69.
- Kurjak A, Carrera JM, McCullough LB, Chervenak FA. Scientific and religious controversies about the beginning of human life: the relevance of the ethical concept of the fetus as a patient. J Perinat Med. 2007;35:376–83.
- Leist M, Bremer S, Brundin P, et al. The biological and ethical basis of the use of human embryonic stem cells for in vitro test systems or cell therapy. ALTEX. 2008;25:163–90.
- Eberl JT. The beginning of personhood: a thomistic biological analysis. Bioethics. 2000;14:134–57.
- Waite L, Nindl G. Human embryonic stem cell research: an ethical controversy in the US & Germany. Biomed Sci Instrum. 2003;9:567–72.
- Smith A. Embryonic stem cells. In: Marshak DR et al., editors. Stem cell biology. New York: Cold Spring Harbor Laboratory Press; 2001. p. 2–19.
- Heinemann T, Honnefelder L. Principles of ethical decision making regarding embryonic stem cell research in Germany. Bioethics. 2002;16:530–43.
- 8. Jones DG. Brain birth and personal identity. J Med Ethics. 1989;15:173–8.
- Goldenring JM. The brain-life theory: towards a consistent biological definition of humanness. J Med Ethics. 1985;11:198–204.
- Kushner T. Having a life versus being alive. J Med Ethics. 1984;10:5–8.
- 11. Shea MC. Embryonic life and human life. J Med Ethics. 1985;11:205–9.
- Hendelman WJ. Atlas of functional neuroanatomy. 2nd ed. New York: Taylor & Francis Group; 2006. p. 1–202.
- Himma KE. A dualist analysis of abortion: personhood and the concept of self qua experiential subject. J Med Ethics. 2005;31:48–55.
- Bergstrom RM, Bergstrom L. Prenatal development of stretch reflex functions and brainstem activity in the human. Ann Chir Gynaecol Fenn. 1963;52(Suppl):1–21S.

- Burgess JA, Tawia SA. When did you first begin to feel it? Locating the beginning of consciousness. Bioethics. 1996;10:1–26.
- Kostovic I. Structural and histochemical reorganization of the human prefrontal cortex during perinatal and postnatal life. Prog Brain Res. 1990;85:223–39.
- Mrzljak L, Uylings HB, Kostovic I, Van Eden CG. Prenatal development of neurons in the human prefrontal cortex: I. A qualitative Golgi study. J Comp Neurol. 1988;271:355–86.
- Mrzljak L, Uylings HB, Van Eden CG, Judas M. Neuronal development in human prefrontal cortex in prenatal and postnatal stages. Prog Brain Res. 1990;85:185–222.
- Chan WY, Lorke DE, Tiu SC, Yew DT. Proliferation and apoptosis in the developing human neocortex. Anat Rec. 2002;267:261–76.
- Huang H. Delineating neural structures of developmental human brains with diffusion tensor imaging. Sci World J. 2010;10:135–44.
- Kostovic I, Vasung L. Insights from in vitro fetal magnetic resonance imaging of cerebral development. Semin Perinatol. 2009;33:220–33.
- Rados M, Judas M, Kostovic I. In vitro MRI of brain development. Eur J Radiol. 2006;57:187–98.
- Donkelaar HJ, Lammens M, Hori A. Clinical neuroembryology, development and developmental disorders of the human central nervous system. Heidelberg: Springer; 2006. p. 429–46.
- Lohman AH, Lammers HJ. On the structure and fibre connections of the olfactory centres in mammals. Prog Brain Res. 1967;23:65–82.
- 25. Glenn OA. Normal development of the fetal brain by MRI. Semin Perinatol. 2009;33:208–19.
- Prayer D, Kasprian G, Krampl E, Ulm B, Witzani L, Prayer L, Brugger PC. MRI of normal fetal brain development. Eur J Radiol. 2006;57:199–216.
- Seress L, Abraham H, Tornoczky T, Kosztolanyi G. Cell formation in the human hippocampal formation from mid-gestation to the late postnatal period. Neuroscience. 2001;105:831–43.
- Hevner RF, Kinney HC. Reciprocal entorhinalhippocampal connections established by human fetal midgestation. J Comp Neurol. 1996;372:384–94.
- Heimer L, Van Hoesen GW, Trimble M, Zahm DS. Anatomy of neuropsychiatry. London: Elsevier; 2008. p. 69–99.
- Ulfig N. Calcium-binding proteins in the human developing brain. Adv Anat Embryol Cell Biol. 2002;165(III–IX):1–92.
- Ulfig N, Setzer M, Bohl J. Ontogeny of the human amygdala. Ann N Y Acad Sci. 2003;985:22–33.
- 32. Wang X, Dow-Edwards D, Keller E, Hurd YL. Preferential limbic expression of the cannabinoid receptor mRNA in the human fetal brain. Neuroscience. 2003;118:681–94.
- Herlenius E, Lagercrantz H. Development of neurotransmitter systems during critical periods. Exp Neurol. 2004;190 Suppl 1:S8–21.

- 34. Yew DT, Chan WY. Early appearance of acetylcholinergic, serotoninergic, and peptidergic neurons and fibers in the developing human central nervous system. Microsc Res Tech. 1999;45:389–400.
- Seifert J. Leib und Seele. Salzburg: Universitatsverlag Anton Pustet; 1973.
- Irmak MK. Beginning of individual human life at 13th week of development. J Exp Integr Med. 2011;1:235–9.
- Irmak MK. Self-fertilization in human: having a male embryo without a father. Med Hypotheses. 2010;75:448–51.
- Irmak MK. Embryological basis of the virgin birth of Jesus. J Exp Integr Med. 2014;4:143–6.
- 39. Qur'an; 2-233 and 46-15: Statements indicating the beginning of personhood (ensoulment) at the end of 3th lunar month (13th week) of development.
- Beckwith FJ. Of souls, selves, and cerebrums: a reply to Himma. J Med Ethics. 2005;31:56–60.
- 41. Murphy N. Whatever happened to the soul? Theological perspectives on neuroscience and the self. Ann N Y Acad Sci. 2003;1001:51–64.
- MacDougall D. Hypothesis concerning soul substance, together with experimental evidence of the existence of such substance. Am Med. 1907;2:240–3.
- 43. Gonzalez de Posada F. Reflections on the ether. An R Acad Nac Med (Madr). 2001;118:43–72.
- 44. Bai Y, Carena M, Lykken J. Dilaton-assisted dark matter. Phys Rev Lett. 2009;103:261803.
- 45. Sadoulet B. Particle dark matter in the universe: at the brink of discovery? Science. 2007;315:61–3.
- Ostriker JP, Steinhardt P. New light on dark matter. Science. 2003;300:1909–13.
- Stapnes S. Detector challenges at the LHC. Nature. 2007;448:290–6.
- Moore KL, Persaud TVN. The developing human, clinically oriented embryology. 7th ed. Philadelphia: Saunders; 2003. p. 185.
- Halpern M. The organization and function of the vomeronasal system. Annu Rev Neurosci. 1987;10:325–62.
- Meredith M. Sensory processing in the main and accessory olfactory system: comparisons and contrasts. J Steroid Biochem Mol Biol. 1991;39:601–14.
- Gottfried JA, Deichmann R, Winston JS, Dolan RJ. Functional heterogeneity in human olfactory cortex: an event-related functional magnetic resonance imaging study. J Neurosci. 2002;22:10819–28.
- Kandel ER, Schwartz JH, Jessel TM, editors. Principles of neural science. 4th ed. New York: McGraw-Hill; 2000.
- Zbar RI, Zbar LI, Dudley C, Trott SA, Rohrich RJ, Moss RL. A classification schema for the vomeronasal organ in humans. Plast Reconstr Surg. 2000;105:1284–8.
- Meredith M. Human vomeronasal organ function. A critical review of best and worst cases. Chem Senses. 2001;26:433–45.

- 55. Salazar I, Lombardero M, Aleman N, Sanchez Quinteiro P. Development of the vomeronasal receptor epithelium and the accessory olfactory bulb in sheep. Microsc Res Tech. 2003;61:438–47.
- Knecht M, Witt M, Abolmaali N, Huttenbrink KB, Hummel T. The human vomeronasal organ. Nervenarzt. 2003;74:858–62.
- Boehm N, Gasser B. Sensory receptor-like cells in the human fetal vomeronasal organ. Neuroreport. 1993;4:867–70.
- Bhatnagar KP, Smith TD. The human vomeronasal organ. III. Postnatal development from infancy to the ninth decade. J Anat. 2001;199:289–302.
- Trotier D, Eloit C, Wassef M, Talmain G, Bensimon JL, Doving KB, Ferrand J. The vomeronasal cavity in adult humans. Chem Senses. 2000;25:369–80.
- Moran DT, Jafek BW, Rowley JC. The vomeronasal (Jacobson's) organ in man: ultrastructure and frequency of occurrence. J Steroid Biochem Mol Biol. 1991;39:545–52.
- Stensaas LJ, Lavker RM, Monti-Bloch L, Grosser BI, Berliner DL. Ultrastructure of the human vomeronasal organ. J Steroid Biochem Mol Biol. 1991;39:553–60.
- Meisami E, Bhatnagar KP. Structure and diversity in mammalian accessory olfactory bulb. Microsc Res Tech. 1998;43:476–99.
- Chuah MI, Zeng DR. Olfactory marker protein is present in olfactory receptor cells of human fetuses. Neuroscience. 1987;23:363–70.
- 64. Kreutzer EW, Jafek BW. The vomeronasal organ of Jacobson in the human embryo and fetus. Otolaryngol Head Neck Surg. 1980;88:119–23.
- Smith TD, Bhatnagar KP. The human vomeronasal organ. Part II: prenatal development. J Anat. 2000;197:421–36.
- 66. Witt M, Georgiewa B, Knecht M, Hummel T. On the chemosensory nature of the vomeronasal epithelium in adult humans. Histochem Cell Biol. 2002;117:493–509.
- Kjaer I, Hansen BF. The human vomeronasal organ: prenatal developmental stages and distribution of luteinizing hormone-releasing hormone. Eur J Oral Sci. 1996;104:34–40.
- Nakashima T, Kimmelman CP, Snow JB. Vomeronasal organs and nerves of Jacobson in the human fetus. Acta Otolaryngol (Stockh). 1985;99:266–71.
- Humphrey T. The development of the olfactory and the accessory olfactory formation in human embryos and fetuses. J Comp Neurol. 1940;73:431–68.
- Takami S. Recent progress in the neurobiology of the vomeronasal organ. Microsc Res Tech. 2002;58:228–50.
- Irmak MK. Cosmological dark matter and ensoulment. J Exp Integr Med. 2013;3:343–6.

Dependence of Fetal Hair and Sebaceous Glands on Fetal Adrenal Cortex and Possible Control from Epidermal Merkel Cells and Adrenal Medulla

M. Kemal Irmak

Introduction

We still do not fully understand the physiological and biological role of the fetal adrenal cortex. The purpose of the present chapter is therefore to review the literature and synthesize the current understanding of the developmental and functional biology of the fetal adrenal cortex. First, a role will be attributed to the fetal adrenal cortex in the regulation of fetal development of hairs and sebaceous glands and their common product vernix caseosa. The literature will also be discussed concerning the role of Merkel cells and adrenal medulla on the regulation of fetal adrenocortical function. As the fetal zone is principally an androgen-producing adrenal cortical zone, understanding of its regulation may provide insights into the regulation of adrenal androgen production in general.

M.K. Irmak

Gulhane Military Medical Academy, High Council of Science, Ankara, Turkey e-mail: mkirmak@gata.edu.tr

Development of Fetal Adrenal Cortex

Human fetal adrenal development is characterized by rapid growth, high steroidogenic activity, and a distinct morphology, including a cortical compartment known as the fetal zone. This fetal zone is a unique adrenal cortical compartment that exists only during fetal life in humans and higher primates [1-3]. Rapid growth of the human fetal adrenal cortex begins at approximately the eighth week of gestation and continues to term [4, 5]. The growth is almost entirely due to enlargement of the fetal zone and, as a consequence, the gland achieves a relative size 10- to 20-fold that of the adult adrenal by 30 weeks [1]. During midgestation, the fetal zone occupies 80-90 % of the cortical volume and produces 100-200 mg/day of the dehydroepiandrosterone sulfate (DHEA-S), which is quantitatively the principal steroid product of the fetal adrenal gland throughout gestation [6, 7]. Morphological and biochemical analyses indicate that the human fetal adrenal cortex has steroidogenic capabilities early in gestation at about 6-8 weeks [7, 8]. Estriol, as an indicative of fetal adrenal steroidogenic activity [8] can first be detected in the maternal circulation at the eighth week of gestation, indicating that DHEA-S is being produced by the fetus at this stage. At around the 12th week of gestation, estriol concentrations in the maternal circulation rapidly

© Springer International Publishing Switzerland 2016 N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development:*

First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_15

15

increase approximately 100-fold [9]. This increase coincides with the initiation of fetal zone hypertrophy [10]. These observations indicate that the human fetal adrenal cortex produces DHEA-S beginning at around 8–10 weeks of gestation in sufficient quantities to affect increases in maternal estriol levels. Production of DHEA-S by the fetal adrenal cortex continues for the remainder of pregnancy and during the second and third trimesters, it increases considerably such that by term the human fetal adrenal produces around 200 mg/day of DHEA-S [8].

However, the fetal adrenal, in the presence of ACTH, prolactin, and growth hormones, involutes after birth [11, 12]. Associated with these changes is a 50 % decline in the absolute weight of the adrenal during the first few postnatal weeks which is reflected by a sharp decrease in DHEA-S concentrations [12], but at the end of the third year, a new zone (the reticular zone) develops from the outermost layer of the fetal cortex [13].

Dependence of Fetal Hair and Sebaceous Glands on Adrenal Androgens

The physiological role of the fetal adrenal cortex during intrauterine life is not well understood. Early in gestation (8–10 weeks), the glands appear to be capable of DHEA-S synthesis. This event seems to be related with the differentiation of hairs and sebaceous glands. Hairs begin to develop early in the fetal period between 9 and 12 weeks [14]. The first hair that appear – **lanugo** – are fine, soft, and lightly pigmented. Lanugo is plentiful by 17–20 weeks, and covers the skin of the fetus densely and helps to hold the vernix caseosa on the skin. Lanugo is shed between the first and the fourth month postpartum, and is replaced by coarser vellus hair [15]. Sebaceous glands develop around the 13th to 15th gestational week as buds from the sides of developing hair follicles [16, 17] and form an oily secretion - sebum - that passes to the surface of the skin, where it mixes with desquamated epidermal cells to form vernix caseosa. Vernix caseosa is a greasy substance that covers the fetal skin and it protects the developing skin from constant exposure to amniotic fluid [18]. Human sebaceous glands attain quite a large size by the time of birth, but then shrink to comparatively small structures until the onset of adrenarche. Androgens are important in determining the type and distribution of hairs over the human body [19]. Androgens are also a prerequisite for sebaceous gland development [20] and they increase sebum secretion. Therefore, not only do androgens alter the type of hair present, but they will increase the oiliness of skin and hair. It seems that the human fetal adrenal cortex produces enough DHEA-S beginning at around 8-10 weeks of gestation to influence the growth of fetal hairs and sebaceous glands. While the fetal zone atrophies soon after birth, fetal hairs are shed and sebaceous glands shrink to small structures in concordance with the rapid decrease in adrenal androgen levels. These postnatal changes in the hair and sebaceous glands strongly support the view about the effects of fetal adrenal cortex on the development of fetal hair and sebaceous glands.

Regulation of Adrenal Androgen Secretion

After the neonatal period, until approximately 6 years of age, levels of adrenal androgens in normal children are at their lifetime minimum [21]. The zona reticularis, not perceptible in children under 6, later recapitulates the secretory pattern of the fetal zone, forming DHEA-S [22-25]. This increase takes place in girls and boys between 6 and 8 years of age, approximately 2 years before the onset of gonadal maturation [23, 26] and termed adrenarche [27]. Adrenarche is accompanied clinically by pubarche, the appearance of axillary and pubic hair [28]. After a peak of adrenal androgen production at age 20–25 [22], DHEA-S, particularly, begins a steep, continuous decline [29], reaching about 20 % of peak levels in those older than age 70. The mechanism responsible for adrenarche and the control of fetal adrenal androgen secretion have both been the subject of considerable investigation, but

remain a matter of controversy [30, 31]. Elucidation of the mechanisms has been hampered by a paucity of appropriate animal models. None of the common laboratory animals has a comparable fetal zone and the event of adrenarche occurs only in humans and higher primate species that have a long childhood preceding the advent of puberty [32-34]. It is well known that the adrenal androgens emanate chiefly from the zona reticularis of the adrenal cortex [35, 36], and hormones, such as ACTH, prolactin, gonadotropins, and estrogens, do not appear to affect adrenal androgen secretion and do not cause the adrenarche [26, 37]. Moreover, during aging in healthy people, serum levels of aldosterone, cortisol and corticosterone undergo relatively little change [29, 38, 39]; in contrast, serum concentrations and excretion of adrenal androgens decline markedly in both sexes [40, 41]. Considering the fact that there is a discrepancy in ACTH levels in plasma and androgen release in the time of adrenarche and in several other clinical situations [26, 35, 38, 42-44], adrenal androgen production seems to be regulated by a different mechanism other than ACTH.

Evidences for a Regulatory Paracrine Effect of Medullary Cells on Cortical Cells

Previous considerations support the hypothesis of a distinct adrenal androgen-stimulating mechanism which might be dependent on intra-adrenal factors that control growth and differentiation of the zona reticularis. In this respect, it was suggested that during adrenarche, it is the adrenal medulla which might influence the adrenal androgen secretion [45]. This suggestion is consistent with numerous lines of evidence which indicate that adrenal medulla exerts a paracrine control on the secretory activity of the cortex by releasing catecholamines and several regulatory peptides [46–58], suggesting that adrenomedullary function might be linked to the androgen secretion. Moreover, the dose-dependent inhibition of adrenomedullary catecholamine secretion in response to adrenal androgens [59, 60] and the presence of β_2 -receptors on cortical cells [61] suggest interdependence of epinephrine and DHEA secretion in the adrenal gland. Several previous findings which show the anatomical proximity and close intermingling of chromaffin and cortical cells in the adrenal [62–68], the disintegration of medullary capsule of the adrenal at adrenarche [69] and the innervation of the adrenal cortex by nerve fibers originating in the medulla [70] also suggest that sympathoadrenal hormones may play a role in the complex developmental process of adrenarche [45]. Therefore, adrenomedullary function appears to be tightly linked to adrenal cortex, explaining the variations in the adrenomedullary hormones with adrenarche [45]. Studies undertaken so far suggest a local intraadrenal peptidergic regulatory concept, and above considerations support the hypothesis that medullary cells may also be involved in the regulation of fetal cortical cell activity in a paracrine manner.

Merkel Cells and Adrenal Medulla Might Play a Role in the Fetal Adrenal Androgen Secretion

Fetal adrenal cortex seems to affect the development of fetal hair and sebaceous glands, but the mechanism that regulates fetal adrenal androgen production is a key unanswered problem in human adrenal biology. It was suggested that Merkel cells and adrenal medulla might play a role in the induction and control of fetal adrenal androgen secretion [71, 72].

The adrenal gland originates from two distinct primordial in mammals: the cortex derived from mesoderm, and the medulla derived from the ectodermal neural crest [1, 73]. What is the biological meaning of this spatial integration of two embryologically distinct organs? What are the interrelations between the two secretory cell lines? It was suggested that this fusion may be an important achievement that facilitates the adaptation to environmental changes, thus enabling better coordination of the functions of the mammals [62]. Environmental factors seem to affect the adrenal physiology. For example, adaptation of higher animals to cold is maintained by the participation of the sympatho-adrenal system. The adaptation to cold develops successfully owing to the marked compensatory hypersecretion of epinephrine by the adrenal glands [74]. Hyperproduction of cathecolamines leads to heat production and combines with an increased reaction of the organism to epinephrine and norepinephrine [75]. That is why, in adrenalectomized animals, adaptation to cold proved impossible, and the animals died after duration of exposure to cold that was well tolerated with an intact sympatho-adrenal system [75]. Exposure to amniotic fluid during gestation is also another environmental factor from which the fetus should be protected. We can suggest therefore that longterm exposure to amniotic fluid may affect adrenal gland activity by medullary cells known to be involved in environmental adaptation.

Merkel cells also arise from stem cells of neural crest origin that migrated during the sixth embryonic week in human skin [76]. Merkel cells with a density of 1,700 per mm² showed cytoplasmic processes directed towards the basal lamina and extending between neighboring keratinocytes. However, these cells generally disappear at the end of gestation and innervation is necessary for the survival of Merkel cells in postnatal life [77, 78]. It has been demonstrated that Merkel cells in the developing skin transiently expressed mRNA for the serotonin transporter indicating an intact and active 5-hydroxytryptamine (5-HT) uptake system [79]. But, Merkel cells themselves are not able to synthesize 5-HT and the source of 5-HT as a substrate for the transporter in Merkel cells is the amniotic fluid, which contains 5-HT transported from the maternal circulation [80]. By expressing the 5-HT transporter, Merkel cells might have a transient role in sensing the 5-HT levels in amniotic fluid [79] and playing a trophic role for the development of fetal adrenal cortex through their endocrine secretions. Receiving or sensing the extracellular 5-HT levels may therefore allow the Merkel cells to modulate the adrenocortical function via adrenal medulla and thus to protect fetal skin from harmful effects of amniotic fluid during gestation by forming vernix caseosa. The postnatal decrease in 5-HT transporter

mRNA and 5-HT uptake might reflect this physiological phenomen [79].

Conclusion

The biological meaning of the spatial integration of cortical and medullary cells seems to be a functional coordination of both epithelial cell lines as has been suggested for the interaction between parafollicular and follicular cells of the thyroid gland [81, 82]. Functional interdependence between the medullary and cortical cells explains the biological significance of their spatial integration. The colocalization of cathecholamine-secreting and steroid hormone-producing cells under a common capsule [62, 67, 68, 83-87] may, via their paracrine interactions, coordinate adaptations to stress and the environment. In this respect, fetal skin seems to be protected from an environmental stressor (contact exposure to amniotic fluid) by vernix caseosa - a product of fetal adrenal cortex, and this stimulus seems to be transferred to the adrenal gland via epidermal Merkel cells. We therefore conclude that both Merkel cells and adrenomedullary cells may facilitate the adaptation of the cortical cells to environmental changes, enabling more effective coordinated functions of the body.

References

- Jirasek J. Human fetal endocrines. London: Martinus Nijhoff; 1980. p. 69–82.
- Hornsby PJ. The regulation of adrenocortical function by control of growth and structure. In: Anderson DC, Winter JSD, editors. Adrenal cortex. London: Butterworths; 1985. p. 1–31.
- Winter JSD. The adrenal cortex in the fetus and neonate. In: Anderson DC, Winter JSD, editors. Adrenal cortex. London: Butterworths; 1985. p. 32–56.
- Hatano O, Takakusu A, Nomura M, Morohashi KI. Identical origin of adrenal cortex and gonad revealed by expression profiles of Ad4BP/SF-1. Genes Cells. 1996;1:663–71.
- Lanman JT. Fetal zone of the adrenal gland; its developmental course, comparative anatomy, and possible physiological functions. Medicine. 1953;32:389–430.
- Mesiano S, Jaffe RB. Regulation of growth and differentiated function in the human fetal adrenal. In: Saez JM, Brownie AC, Capponi A, Chambaz EM, Mantero F, editors. Cellular and molecular biology of

the adrenal cortex. Paris: INSERM/Libbey Eurotext; 1992. p. 235–45.

- McNutt NS, Jones AL. Observations on the ultrastructure of cytodifferentiation in the human fetal adrenal cortex. Lab Invest. 1970;22:513–27.
- Siiteri P, MacDonald P. The utilization of circulating dehydroepiandrosterone sulfate for estrogen synthesis during human pregnancy. Steroids. 1963;2:713–16.
- Yen SSC. Endocrine-metabolic adaptations in pregnancy. In: Yen SSC, Jaffe RB, editors. Reproductive endocrinology. Philadelphia: W.B. Saunders; 1991. p. 936–81.
- Baker BL, Jaffe RB. The genesis of cell types in the adenohypophysis of the human fetus as observed with immunocytochemistry. Am J Anat. 1975;143:137–61.
- De Peretti E, Forest M. Unconjugated DHA plasma levels in normal subjects from birth to adolescence in human: the use of a sensitive radioimmunoassay. J Clin Endocrinol Metab. 1976;43:982–91.
- Mesiano S, Jaffe RB. Developmental and functional biology of the primate fetal adrenal cortex. Endocr Rev. 1997;18:378–403.
- Sadler TW. Langman's medical embryology. Baltimore: Williams & Wilkins; 1984. p. 365–6.
- Muller M, Jasmin JR, Monteil RA, Loubiere R. Embryology of the hair follicle. Early Hum Dev. 1991;26:159–66.
- Whiting DA, Howsden EL. Color atlas of differential diagnosis of hair loss. Cedar Grove: Canfield Publishing; 1996. p. 16.
- Downing DT, Stewart ME, Strauss JJ. Biology of sebaceous glands. In: Fitzpatrick TB, Eisen AZ, Wolff K, Freedberg IM, Austen KF, editors. Dermatology in general medicine, vol. 1. 3rd ed. New York: McGraw-Hill Book Co; 1987. p. 185–90.
- Montagna W. Embryology and anatomy of the cutaneous adnexa. J Cutan Pathol. 1984;11:350–1.
- Moore KL, Persaud TVN. The developing human, clinically oriented embryology. 7th ed. Philadelphia: Saunders; 2003. p. 304–489.
- Azziz R, Carmina E, Sawaya ME. Idiopathic hirsutism. Endocr Rev. 2000;21:347–62.
- Rosenfield RL, Deplewski D. Role of androgens in the developmental biology of the pilosebaceous unit. Am J Med. 1995;98:80S–8.
- Miller WL, Tyrell JB. The adrenal cortex. In: Felig P, Baxter JD, Frohman LA, editors. Endocrinology and metabolism. 3rd ed. New York: McGraw-Hill; 1995. p. 555–711.
- Meikle AW, Daynes RA, Araneo BA. Adrenal androgen secretion and biologic effects. Endocrinol Metab Clin N Am. 1991;20:381–400.
- Babalola AA, Ellis G. Serum DHEAS in a normal pediatric population. Clin Biochem. 1985;18:184–9.
- 24. Pepe GJ, Albrecht ED. Regulation of the primate fetal adrenal cortex. Endocr Rev. 1990;11:151–76.
- Hopper BR, Yen SS. Circulating concentrations of dehydroepiandrosterone and dehydroepiandrosterone sulfate during puberty. J Clin Endocrinol Metab. 1975;40:458–61.

- 26. Apter D, Pakkerinen A, Hammond GL, Vihko R. Adrenocortical function and puberty, serum ACTH, cortisol and dehydroepiandrosterone in girls and boys. Acta Paediatr Scand. 1979;69:599–606.
- 27. Baulieu EE, Corpechot C, Dray F, et al. An adrenal secreted androgen: dehydroisoandrosterone sulfate: its metabolism and a tentative generalization on the metabolism of other steroid conjugates in man. Recent Prog Horm Res. 1965;21:411–500.
- Ferriman D, Gallwey JD. Clinical assessment of body hair growth in women. J Clin Endocrinol Metab. 1961;21:1440–7.
- Hornsby PJ. The biosynthesis of DHEA by the adrenal cortex and its age-related decline. In: Watson RW, editor. DHEA: health promotion and aging. Harwood: Academic; 1998. p. 1–13.
- Lee PA, Kowarski A, Migeon CJ, Blizzard RM. Lack of correlation between gonadotropin and adrenal androgen levels in agonadal children. J Clin Endocrinol Metab. 1975;40:664–9.
- Parker LN. Control of adrenal androgen secretion. Endocrinol Metab Clin N Am. 1991;20:401–21.
- Cutler Jr GB, Glenn M, Bush M, Hodgen GD, Graham CE, Loriaux DL. Adrenarche: a survey of rodents, domestic animals, and primates. Endocrinology. 1978;103:2112–18.
- Cutler GB, Loriaux DL. Adrenarche and its relationship to the onset of puberty. Fed Proc. 1980;39:2384–92.
- Smail PJ, Faiman C, Hobson WC, Fuller GB, Winter JSD. Further studies on adrenarche in nonhuman primates. Endocrinology. 1982;111:844–8.
- Parker LN, Odell W. Control of adrenal androgen secretion. Endocr Rev. 1980;1:392–410.
- 36. Endoh A, Kristiansen SB, Carson PR, Buster JE, Hornsby PJ. The zona reticularis is the site of biosynthesis of DHEA and DHEAS in the adult human adrenal cortex. J Clin Endocrinol Metab. 1996;81:3558–65.
- Yen SCS, Jaffe RB. Reproductive endocrinology: physiology, pathophysiology and clinical management. 2nd ed. Philadelphia: WB Saunders Company; 1986. p. 340–2.
- Parker LN, Gral T, Perrigo V, Skowsky R. Decreased adrenal androgen sensitivity to ACTH during aging. Metab Clin Exp. 1981;30:601–4.
- Orentreich N, Brind JL, Rizer RL, Vogelman JH. Age changes and sex difference in serum dehydroepiandrosterone sulfate concentrations throughout adulthood. J Clin Endocrinol Metab. 1984;59:551–5.
- Parker L, Lifrak E, Ramadan M, et al. Aging and the human adrenal zona reticularis. Arch Androl. 1983;10:17–20.
- Thomas G, Frenoy N, Legrain S, Sebag-Lanoe R, Baulieu EE, Debuire B. Serum dehydroepiandrosterone sulfate levels as an individual marker. J Clin Endocrinol Metab. 1994;79:1273–6.
- Yamaji T, Ishibashi M, Sekihara H, et al. Serum DHAS in Cushing's syndrome. J Clin Endocrinol Metab. 1984;59:1164–8.

- 43. Albertson B, Hobson W, Barnett B, et al. Dissociation of cortisol and adrenal androgen secretion in the hypophysectomized, ACTH-replaced chimpanzee. J Clin Endocrinol Metab. 1984;59:13–8.
- 44. Hauffa BP, Kaplan SL, Grumbach MM. Dissociation between plasma adrenal androgens and cortisol in Cushing's disease and ectopic ACTHproducing tumour: relation to adrenarche. Lancet. 1984;1:1373–6.
- 45. Weise M, Eisenhofer G, Merke DP. Pubertal and gender-related changes in the sympathoadrenal system in healthy children. J Clin Endocrinol Metab. 2002;87:5038–43.
- 46. Mazzocchi G, Musajo F, Neri G, Gottardo G, Nussdorfer GG. Adrenomedullin stimulates steroid secretion by the isolated perfused rat adrenal gland *in situ*: comparison with calcitonin gene-related peptide effects. Peptides. 1996;17:853–7.
- 47. Mesiano S, Katz SL, Lee JY, Jaffe RB. Insulin-like growth factors augment steroid production and expression of steroidogenic enzymes in human fetal adrenal cortical cells: implications for adrenal androgen regulation. J Clin Endocrinol Metab. 1997;82:1390–6.
- Bornstein SR, Ehrhart-Bornstein M, Scherbaum WA, Pfeiffer EF. Effects of splanchnic nerve stimulation on the adrenal cortex may be mediated by chromaffin cells in a paracrine manner. Endocrinology. 1990;127:900–6.
- Ehrhart-Bornstein M, Hinson JP, Bornstein SR, Scherbaum WA, Vinson GP. Intraadrenal interactions in the regulation of adrenocortical steroidogenesis. Endocr Rev. 1998;19:101–43.
- Ehrhart-Bornstein M, Bornstein SR, Güse-Behling H, et al. Sympathoadrenal regulation of adrenal androstenedione release. Neuroendocrinology. 1994;59:406–12.
- Ehrhart-Bornstein M, Bornstein SR, Trzeciak WH, et al. Adrenaline stimulates cholesterol side chain cleavage cytochrome P450 mRNA accumulation in bovine adrenocortical cells. J Endocrinol. 1991;131:R5–8.
- Güse-Behling H, Ehrhart-Bornstein M, Bornstein SR, Waterman MR, Scherbaum WA, Adler G. Regulation of adrenal steroidogenesis by adrenaline: expression of cytochrome P450 genes. J Endocrinol. 1992;135:229–37.
- Schultzberg M, Lundberg JM, et al. Enkephalin-like immunoreactivity in gland cells and nerve terminals of the adrenal medulla. Neuroscience. 1978;3:1169–86.
- Toth IE, Hinson JP. Neuropeptides in the adrenal gland: distribution, localization of receptors, and effects on steroid hormone synthesis. Endocr Res. 1995;21:39–51.
- 55. Bernet F, Bernard J, Laborie C, Montel V, Maubert E, Dupouy JP. Neuropeptide Y (NPY)- and vasoactive intestinal peptide (VIP)-induced aldosterone secretion by rat capsule/glomerulosa zone could be mediated by catecholamine via β1 adrenergic receptors. Neurosci Lett. 1994;166:109–12.

- Pelto-Huikko M. Immunocytochemical localization of neuropepides in the adrenal medulla. J Electron Microsc Tech. 1989;12:364–79.
- Haidan A, Bornstein SR, Glasow A, Uhlmann K, Lübke C, Ehrhart-Bornstein M. Basal steroidogenic activity of adrenocortical cells is increased tenfold by co-culture with chromaffin cells. Endocrinology. 1998;139:772–80.
- Nussdorfer GG. Paracrine control of adrenal cortical function by medullary chromaffin cells. Pharmacol Rev. 1996;48:495–530.
- Liu PS, Lin MK, Hsieh HL. Dehydroepiandrosterone sulfate inhibition of catecholamine secretion from bovine adrenal chromaffin cells. Neurosci Lett. 1996;204:181–4.
- Dar DE, Zinder O. Short term effect of steroids on catecholamine secretion from bovine adrenal medulla chromaffin cells. Neuropharmacology. 1997;36:1783–8.
- 61. Shima S, Komoriyama K, Hirai M, Kouyama H. Studies on cyclic nucleotides in the adrenal gland. XI. Adrenergic regulation of adenylate cyclase activity in the adrenal cortex. Endocrinology. 1984;114: 325–9.
- Irmak MK. Strain differences in rabbit adrenocortical structure and morphological evidence of corticomedullary communication. Acta Physiol Hung. 2000;87:167–72.
- 63. Bornstein SR, Gonzalez-Hernandez JA, Ehrhart-Bornstein M, Adler G, Scherbaum WA. Intimate contact of chromaffin and cortical cells within the human adrenal gland forms the cellular basis for important intraadrenal interactions. J Clin Endocrinol Metab. 1994;78:225–32.
- Fortak W, Kmiec B. About occurrence of the chromophilic cells in the adrenal cortex of white rats. Endokrynol Pol. 1968;19:117–28.
- Palacios G, Lafraga M. Chromaffin cells in the glomerular zone of adult rat adrenal cortex. Cell Tissue Res. 1975;164:275–8.
- 66. Gallo-Payet N, Pothier P, Isler H. On the presence of chromaffin cells in the adrenal cortex: their possible role in adrenocortical function. Biochem Cell Biol. 1987;65:588–92.
- Bornstein SR, Ehrhart-Bornstein M, Usadel H, Böckmann M, Scherbaum WA. Morphological evidence for a close interaction of chromaffin cells with cortical cells within the adrenal gland. Cell Tissue Res. 1991;265:1–9.
- Bornstein SR, Ehrhart-Bornstein M. Ultrastructural evidence for a paracrine regulation of the rat adrenal cortex mediated by the local release of catecholamines from chromaffin cells. Endocrinology. 1992;131:3126–8.
- Dhom G. The prepubertal and pubertal growth of the adrenal (adrenarche). Beitr Pathol. 1973;150:357–77.
- Hinson JP. Paracrine control of adrenocortical function: a new role for the medulla? J Endocrinol. 1990;124:7–9.

- Irmak MK. Multifunctional Merkel cells: their roles in electromagnetic reception, finger-print formation, Reiki, epigenetic inheritance and hair form. Med Hypotheses. 2010;75:162–8.
- Irmak MK, Oztas E, Vural H. Dependence of fetal hairs and sebaceous glands on fetal adrenal cortex and possible control from adrenal medulla. Med Hypotheses. 2004;62:486–92.
- Orth DN, Kavacs WJ, Rowan DeBold C. The adrenal cortex. In: Wilson JD, Foster DW, editors. William's textbook of endocrinology. Philadelphia: Saunders; 1992. p. 489–619.
- Leduc J. Catecholamine production and release in exposure and acclimation to cold. *Acta Physiol Sc* and Suppl. 1961;183:1–101.
- Johnson GE, Schonbaum E, Sellers EA. Cold exposure: pharmacologic investigation of the compensatory mechanisms in the maintenance of normothermia. Fed Proc. 1966;25:1216–19.
- Lucarz A, Brand G. Current considerations about Merkel cells. Eur J Cell Biol. 2007;86:243–51.
- Kinkelin I, Stucky CL, Koltzenburg M. Postnatal loss of Merkel cells, but not of slowly adapting mechanoreceptors in mice lacking the neurotrophin receptor p75. Eur J Neurosci. 1999;11:3963–9.
- Moll IM, Roessler M, Brandner JM, Eispert AC, Houdek P, Moll R. Human Merkel cells: aspects of cell biology, distribution and functions. Eur J Cell Biol. 2005;84:259–71.
- Hansson SR, Hoffman BJ. Transient expression of a functional serotonin transporter in Merkel cells during

late gestation and early postnatal rat development. Exp Brain Res. 2000;130:401–9.

- Yavarone MS, Shuey DL, Sadler TW, Lauder JM. Serotonin uptake in the ectoplacental cone and placenta of the mouse. Placenta. 1993;14:149–61.
- Irmak MK, Özcan O. Human diversity, environmental adaptation and neural crest. Med Hypotheses. 1997;48:407–10.
- Irmak MK, Kirici Y. Time to reevaluate the therapeutic use of calcitonin and biological role attributable to parafollicular (C) cells. Med Hypotheses. 2004;62:486–92.
- Belloni AS, Andreis PG, Meneghelli V. Adrenomedullin and calcitonin gene-related peptide (CGRP) interact with a common receptor of the CGRP1 subtype in the human adrenal zona glomerulosa. Endocr Res. 1999;25:29–34.
- Cunningham LA, Holzwarth MA. Vasoactive intestinal peptide stimulates adrenal aldosterone and corticosterone secretion. Endocrinology. 1988;122:1090–7.
- Kapas S, Hinson JP. Actions of adrenomedullin on the rat adrenal cortex. Endocr Res. 1996;22:861–5.
- Ottenweller JE, Meier AH. Adrenal innervation may be an extrapituitary mechanism able to regulate adrenocortical rhythmicity in rats. Endocrinology. 1982;111:1334–8.
- Weinkove C, Anderson DC. Interactions between the adrenal cortex and medulla. In: Anderson DC, Winter JSD, editors. Adrenal cortex. London: Butterworth Co; 1985. p. 208–34.

16

Theoretical Postulation of the Embryological Basis of the Virgin Birth and Role of Embryonic Stem Cells Localized Out of the Embryo

M. Kemal Irmak

Introduction

Human sex is determined at fertilization by the sex chromosome of the sperm. While X-bearing sperms form XX females, those containing a Y chromosome give rise to XY males. Y chromosome triggers the undifferentiated gonad into the path of testicular development; in its absence, the indifferent gonad develops into an ovary [1]. The differentiation of the male reproductive tract and external genitalia depends on hormones secreted by the fetal testis, whereas the fetal ovary is not required for the development of the female tract, which occurs simply in the absence of testicular secretions [2]. Primordia of female and male genital ducts are present in the early embryo of both sexes [1], and the normal development of the internal genital ducts is determined by the presence or absence of testicular tissue. In males, testicular secretions induce the development of male ducts while they lead to the regression of the female ducts [1, 2]. In females, the lack of testicular secretions allow the development of the female genital tract, while the male ducts regress due to a lack of testosterone [1].

Gulhane Military Medical Academy, High Council of Science, Ankara, Turkey e-mail: mkirmak@gata.edu.tr

A Human May Have Functional Male and Female Gonads

In some persons, both ovarian and testicular tissues are present, either in the same (ovotestis) or in opposite gonads [2, 3], and they are called true hermaphrodites.

Ovulation and spermatogenesis can be observed in the gonads of these subjects [4-7]. It was reported that 35 % of true hermaphrodites have bilateral ovotestes, 38 % have one ovotestis and a contralateral ovary or testis, and 27 % have one ovary and a contralateral testis [4]. An ovotestis is a bipolar structure showing ovarian tissue in an upper pole and the testicular tissue in the lower pole [5]. Efferent ducts and vas deferens are never found together next to an ovotestis which can be explained by lower amount of testosterone [6]. While the ovarian compartment of an ovotestis shows evidence of ovulation at puberty in about 50 % of cases, spermatogenesis has never been observed in the testicular portion of an ovotestis [6, 7]. Spermatogenesis was observed in only solitary testes found in true hermaphroditism [7]. About 10 % of true hermaphrodites are derived from more than one zygote and are chimeras (chi 46,XX/46,XY) [3, 7]. In chimerism, two oocytes are fertilized by two spermatozoa, then fusion of two different zygotes into a single embryo takes place [8]. The external appearance of these subjects is extremely variable, they may be normal male and normal

N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development: First and Second Trimesters*, DOI 10.1007/978-3-319-14874-8_16

M.K. Irmak

[©] Springer International Publishing Switzerland 2016

female, or ambiguous [9]. Many such cases have been reported in the literature and healthy off-springs have been observed in a few men and women [9-16].

A Left Ovary and a Right Abdominal Testis May Develop in a Human

In chimeric subjects, ovary is localized frequently on the left side of the body, while testis is found more common on the right [4]. During development of a chimera, XX cells tend to gather on the left side of the embryo while XY cells on the right [2, 17–19] which may explain the preferred locations of the gonads (left ovary and right testis). In human true hermaphrodites, regression of the male and female genital duct systems is different because of the presence of both types of the gonads. Embryonic testis has its effect mainly by local action on the adjacent ducts [20-22]. Therefore, testicular tissue may not influence the growth of the female duct on the contralateral side [23], but at least 15 % of testicular tissue is enough to inhibit the growth of the ipsilateral female duct. Moreover, the uterine and oviductal portions of the female duct differ in their distance and sensitivity to testicular secretions [21]. Thus the regression of the female duct system is variable in human true hermaphrodites [5]. The uterus was present throughout its length and normal in 10 % of cases, and present but hypoplastic in 46 %. Only one uterine horn was present in 10 % and only the uterine body was present in 14 % of subjects [5]. On the other hand, the growth of the male duct is not correlated to the amount of testicular tissue in the ipsilateral gonad [23]. One explanation for the absence of the male ducts at the right side is that the maximal response may only be triggered by a critical higher hormone level.

Transperitoneal Migration of Gamets to Contralateral Oviduct

During ovulation, estrogens increase the motility of the uterine tubes which results in a negative pressure in the oviducts [24–28]. This produces a

vacuum effect which has made several pregnancies possible in subjects lacking an ipsilateral ovary by allowing the transperitoneal migration of oocyte from the contralateral gonad [29–34]. In a case of ectopic pregnancy in the left oviduct, sperms gain access to the left oviduct after entering the peritoneal cavity via the right oviduct [35]. These suggest that ipsilateral oviduct is not always essential for fertilization and negative pressure in the oviducts is much enough to pickup the gametes on the contralateral side.

Pregnancies in Chimera

Many pregnancies were reported in chimera of 46,XX/46,XY type while a few has fathered a child [4, 36-40]. It is interesting to note that all pregnancies resulted in male fetuses [36, 41–43]. The probability of having many consecutive male newborns is unlikely to be related to chance alone. Our observations and several other studies indicate that at fertilization, warmer oocytes prefer Y-bearing sperms to give rise to a male embryo while cooler oocytes prefer X-bearing sperms to give rise to a female embryo [44–47]. Since local temperature is higher at the left ovary than the right one because of the differences in the speed of venous drainage (at the left side, slower drainage to the renal vein with a right angle; and at the right side, faster drainage to the vena cava inferior with an oblique angle result in different gonadal temperatures at the two sides of the abdomen [48]), the oocytes in those women with unilateral left ovary seem to prefer Y-bearing sperms resulting in pregnancies with male embryo. Cooler temperature at the right side also seems to allow the testis to be functional at the right side of the abdominal cavity.

Self-Fertilization

Self-fertilization was reported in many flowering plants, in a kind of fish and in a case of rabbit [49–52]. They have both eggs and sperms in their body and at fertilization, one sperm cell fuses with oocyte to form an embryo. The mangrove

killifish which can reproduce by self-fertilization [51] has bilobed gonad consisting mainly of ovarian tissue with small amount of testicular tissue that simultaneously produces eggs and sperm [52]. It is interesting to note that a true hermaphrodite rabbit with two functional ovaries and two testes was housed in isolation and became pregnant and delivered seven healthy young of both sexes [50]. Is self-fertilization possible for a human? The answer is yes, but it seems to occur once in the history.

Scenario for Mary to Give Birth to Jesus without a Father

Virgin Mary is a chimera of 46,XX/46,XY type resulting from the fusion of two embryos of different sex types and both ovarian and testicular tissues develop in Her body as seen in a beautiful plant [53]. In another word, She develops from the fusion of dizygotic male and female twins and She had both reproductive cell types (oocytes and sperms) in Her body. Since XX cells tend to gather on the left side while XY cells on the right, an ovary develops on the left side with a patent oviduct and a testis develops at the right side of the abdominal cavity with no duct [54]. Her intact female reproductive tract on the left side indicates that contralateral testicular secretion is not adequate to induce complete duct regression during development on the left. Female genital duct regression on the right side is however mediated by the secretions from the ipsilateral testis, but testosterone secreted from right testis does not prevent the regression of the male duct due to a lack of response to it or inadequate hormone levels at the site. Therefore, no duct is present next to the testis at the right side, and tubules of the testis have an open access to the abdominal cavity allowing the sperms to be picked-up by the contralateral oviduct. Both gonads are functional and produce oocyte and spermatozoa respectively after puberty. At the time of ovulation, estrogens increase the motility of the oviduct on the left side which results in a negative pressure in the tube and oocyte and sperms are picked-up into the tube with the help of this vacuum effect,

taking both gametes into the fertilization site in the oviduct. Because of the higher temperature of the oocyte derived from the left ovary, fertilization occurs with a Y-bearing sperm to give rise to a XY male embryo which becomes miraculously Jesus [55].

Embryonic Stem Cells Localized out of the Embryo

In mammals, all cells produced by the first few divisions of the fertilized egg are totipotent - that is, they all have the potential to give rise to an embryo with the placenta [56]. Each cell (blastomere) is equipotent up to the eight-cell stage (Fig. 16.1a), after which there is a gradual restriction of the developmental potency of the cells [57]. The differentiation of the inner cell mass and the trophoblast lineage is the first restriction event in embryogenesis. The inside blastomeres are made up of a population of pluripotent cells and are called inner cell mass which give rise to the embryo, whereas the outside cells give rise to the trophoblast (Fig. 16.1b), which is the future placenta [58]. After implantation takes place, the inner cell mass develops to a simple epithelium called epiblast (Fig. 16.1c). Epiblast cells are still pluripotent after the implantation [59].

While all somatic cells are destined for death once they start differentiation, germ cells are the only cells which can give rise to successive generations. Primordial germ cells, the earliest recognizable precursors of germ cells are recognizable at a surprisingly early stage in development [60]. They arise from a pluripotent population of cells in the epiblast (Fig. 16.1c) [61] and then take up temporary residence in extraembryonic tissues before returning to the body of the embryo proper [62]. Primordial germ cells actively migrate into the proximal allantois, and directly into the adjacent embryonic endoderm [63]. They spend the early stages of development within these extraembryonic tissues as a small cluster of cells [64]. The allantois arises early in the third week as an endodermal outgrowth from the dorsocaudal part of the yolk sac into the connecting stalk. The main function of



Fig. 16.1 Developmental fate of a group of embryonic stem cells in the early developmental stages: (a) All cells up to the eight-cell stage after the first few divisions of the fertilized egg are totipotent – that is, all cells (blastomeres) have the potential to give rise to an embryo with the placenta. (b) The differentiation of the inner cell mass which gives rise to the embryo and the trophoblast lineage

which is the future placenta is the first restriction event in embryogenesis. (c) Primordial germ cells arise from a pluripotent population of cells in the epiblast. (d) Pluripotent primordial germ cells that enter the allantois remain in the allantois and do not return to the embryo whereas other primordial germ cells migrate dorsocranially in the embryo to reach the developing gonads

the allantois is to fuse with the chorion and vascularize, thereby forming the umbilical component of the placenta [62]. Evidence suggests that bone morphogenetic protein 4 secreted by cells lying outside the embryo proper is required for the formation of both primordial germ cells and allantois [65]. These cells divide and mature, turning off the expression of a number of somatic cell genes and turning on the expression of genes involved in maintaining the special character of the germ cells and allantois [56].

Primordial germ cells are found at the base of the allantois and in the adjacent endoderm of yolk sac in the early developmental stages [66]. All of these cells do not migrate into the embryo, only a specific subpopulation does. The population of primordial germ cells that enter the allantois contribute only to the allantois, and do not return to the embryo [62, 67, 68], whereas other primordial germ cells migrate dorsocranially in the embryo to reach the developing gonads (Fig. 16.1d) [69]. The involution of allantois results in the formation of the urachus and medial umbilical ligament localized between the urinary bladder and the navel [60]. Allantoid remains and primordial germ cells may still be present inside the ligament all the way to adulthood; which can develop into cysts, diverticula and neoplasms [70].

The base of the allantois which contains the future germ line is the only allantoic region to exhibit pluripotency, colonizing several derivatives of all three primary germ layers [68]. These cells retain an unrestricted developmental potential and can give rise to all the tissues and cell types in the body, including germ cells [56]. Thus,

in a suitable environment, they can produce any of the cell types of the body, although not the extraembryonic cells that go on to form structures such as the placenta; for this reason, they are said to be pluripotent, rather than totipotent [57]. Cells with this property are called embryonic stem cells [56]. By not migrating into the embryo and staying in the allantois arrests the normal program of change of their character with time and so enables these stem cells to live without differentiating [56]. It seems that in the early developmental stages, a group of pluripotent embryonic stem cells each of which can give rise to an embryo are set aside out of the embryo for a possible use, may be a rebirth, in the future [71].

References

- Moore KL, Persaud TVN. The developing human, clinically oriented embryology. 7th ed. Philadelphia: Saunders; 2003. p. 16.
- Inoue H, Nomura M, Yanase T, Ichino I, Goto K, Ikuyama S, Takayanagi R, Nawata H. A rare case of 46, XX true hermaphroditism with hidden mosaicism with sex-determining region Y chromosome-bearing cells in the gonads. Intern Med. 1998;37:467–71.
- Behrman RE, Kliegman RM, Jenson HB. Nelson textbook of pediatrics. 17th ed. Philadelphia: Saunders; 2004. p. 1945.
- Krob G, Braun A, Kuhnle U. True hermaphrodite: geographical distribution, clinical findings, chromosomes and gonadal histology. Eur J Pediatr. 1994;153:2–10.
- 5. van Niekerk WA, Retief AE. The gonads of human true hermaphrodites. Hum Genet. 1981;58:117–22.
- Fischer HW, Lischer CE, Byars LT. True hermaphroditism. Ann Surg. 1952;136:864–73.
- Aaronson IA. True hermaphroditism: a review of 41 cases with observations on testicular histology and function. Br J Urol. 1985;57:775–9.
- Strachan T, Read AP. Human molecular genetics. 3rd ed. New York: Taylor & Francis; 2004. p. 110.
- Malan V, Vekemans M, Turleau C. Chimera and other fertilization errors. Clin Genet. 2006;70:363–73.
- Schultz BA, Roberts S, Rodgers A, Ataya K. Pregnancy in true hermaphrodites and all male offspring to date. Obstet Gynecol. 2009;113:534–6.
- van Niekerk WA. True hermaphroditism: an analytic review with a report of 3 new cases. Am J Obstet Gynecol. 1976;126:890–907.
- Haqq CM, Donahoe PK. Regulation of sexual dimorphism in mammals. Physiol Rev. 1998;78:1–33.
- Minowada S, Fukutani K, Hara M, Shinohara M, Kamioka J, Isurugi K, Niijima T. Childbirth in true hermaphrodites. Eur Urol. 1984;10:414–5.

- Uehara S, Nata M, Nagae M, Sagisaka K, Okamura K, Yajima A. Molecular biologic analyses of tetragametic chimerism in a true hermaphrodite with 46, XX/46, XY. Fertil Steril. 1995;63:189–92.
- Green AJ, Barton DE, Jenks P, Pearson J, Yates JR. Chimaerism shown by cytogenetics and DNA polymorphism analysis. J Med Genet. 1994;31:816–7.
- Salas-Cortes L, Jaubert F, Nihoul-Fekete C, Brauner R, Rosemblatt M, Fellous M. SRY protein is expressed in ovotestis and streak gonads from human sex-reversal. Cytogenet Cell Genet. 2000;91:212–6.
- Hadjiathanasiou CG, Brauner R, Lortat-Jacob S, Nivot S, Jaubert F, Fellous M, Nihoul-Fekete C, Rappaport R. True hermaphroditism: genetic variants and clinical management. J Pediatr. 1994;125:738–44.
- Giltay JC, Brunt T, Beemer FA, Wit JM, van Amstel HK, Pearson PL, Wijmenga C. Polymorphic detection of a parthenogenetic maternal and double paternal contribution to a 46, XX/46, XY hermaphrodite. Am J Hum Genet. 1998;62:937–40.
- Hawkins JR, Taylor A, Berta P, Levilliers J, Van der Auwera B, Goodfellow PN. Mutational analysis of SRY: nonsense and missense mutations in XY sex reversal. Hum Genet. 1992;88:471–4.
- Koopman P, Gubbay J, Vivian N, Goodfellow P, Lovell-Badge R. Male development of chromosomally female mice transgenic for Sry. Nature. 1991;351:117–21.
- Meyers-Wallen VN, Donahoe PK, Manganaro T, Patterson DF. Müllerian inhibiting substance in sexreversed dogs. Biol Reprod. 1987;37:1015–22.
- Sommer MM, Meyers-Wallen VN. XX true hermaphroditism in a dog. J Am Vet Med Assoc. 1991;198:435–8.
- Andersen CY, Byskov AG, Grinsted J. Growth pattern of the sex ducts in foetal mouse hermaphrodites. J Embryol Exp Morphol. 1983;73:59–68.
- McComb PF, Coppo ME. The transperitoneal migration of ova in the rabbit. Acta Eur Fertil. 1986;17:5–7.
- Pettersson A, Larsson B, Einarsson E. The effect of unilateral ovariectomy on intraluminal pressure in the porcine oviductal isthmus. Zentralbl Veterinarmed A. 1991;38:481–4.
- Mwanza AM, Englund P, Kindahl H, Lundeheim N, Einarsson S. Effects of post-ovulatory food deprivation on the hormonal profiles, activity of the oviduct and ova transport in sows. Anim Reprod Sci. 2000;59:185–99.
- Pettersson A, Einarsson S, Kindahl H. Intraluminal pressure variations in the isthmus of the porcine oviduct after intrauterine insemination with saline, oestrogen solution or boar seminal plasma. Acta Vet Scand. 1993;34:109–16.
- Mwanza AM, Englund P, Pettersson A, Einarsson S. Oviductal isthmic motility patterns as monitored by polyview in unrestrained sows around ovulation. Anim Reprod Sci. 2000;62:309–20.
- Ombelet W, Deblaere K, Grieten M, Verswijvel G, Nijs M, Hinoul P, de Jonge E. Intrauterine pregnancy following transperitoneal oocyte and/or sperm

migration in a woman with an ectopic (undescended) ovary. Reprod Biomed Online. 2003;7:110–3.

- Motta T, Marchini M, Fadin M, D'Alberton A, Candiani GB. Successive transperitoneal migration of ova in a woman with extensive pelvic adhesions. Fertil Steril. 1993;59:1311–2.
- Ben-Nun I, Fejgin M, Gruber A, Ben-Aderet N. Transperitoneal ovum migration in women with unilateral congenital ovarian absence. Acta Obstet Gynecol Scand. 1988;67:665–7.
- Wheeler JM, Dodson MG. Transperitoneal migration of the ovum. A case report. J Reprod Med. 1985;30:895–8.
- 33. Gabriel B, Fischer DC, Sergius G. Unruptured pregnancy in a non-communicating heterotopic right fallopian tube associated with left unicornuate uterus: evidence for transperitoneal sperm and oocyte migration. Acta Obstet Gynecol Scand. 2002;81:91–2.
- Kamrava MM, Seibel MM, Thompson IE, Berger MJ, McArdle CR. Intrauterine pregnancy following transperitoneal migration of the ovum. Obstet Gynecol. 1982;60:391–3.
- Ansari AH, Miller ES. Sperm transmigration as a cause of ectopic pregnancy. Arch Androl. 1994;32:1–4.
- Williamson HO, Phansey SA, Mathur RS. True hermaphroditism with term vaginal delivery and a review. Am J Obstet Gynecol. 1981;141:262–5.
- Tanaka Y, Fujiwara K, Yamauchi H, Mikami Y, Kohno I. Pregnancy in a woman with a Y chromosome after removal of an ovarian dysgerminoma. Gynecol Oncol. 2000;79:519–21.
- Talerman A, Verp MS, Senekjian E, Gilewski T, Vogelzang N. True hermaphrodite with bilateral ovotestis, bilateral gonadoblastomas and dysgerminomas, 46, XX/46, XY karyotype, and a successful pregnancy. Cancer. 1990;66:2668–72.
- 39. Verp MS, Harrison H, Ober C, Oliveri D, Amarose AP, Lindgren V, Talerman A. Chimerism as the etiology of a 46XX/46XY fertile true hermaphrodite. Fertil Steril. 1992;57:346–9.
- Tiltman AJ, Sweerts M. Multiparity in a covert true hermaphrodite. Obstet Gynecol. 1982;60:752–4.
- Mayou BJ, Armon P, Lindenbaum RH. Pregnancy and childbirth in a true hermaphrodite following reconstructive surgery. Br J Obstet Gynaecol. 1978;85:314–6.
- 42. Tegenkamp TR, Brazzel JW, Tegenkamp I, Labidi F. Pregnancy without benefit of reconstructive surgery in a bisexually active true hermaphrodite. Am J Obstet Gynecol. 1979;135:427–8.
- Schoenhaus SA, Lentz SE, Saber P, Munro MG, Kivnick S. Pregnancy in a hermaphrodite with a male-predominant mosaic karyotype. Fertil Steril. 2008;90:E7–10.
- Robert KA, Thompson MB. Viviparity and temperature-dependent sex determination. Sex Dev. 2010;4:119–28.
- Pieau C, Dorizzi M. Oestrogens and temperaturedependent sex determination in reptiles: all is in the gonads. J Endocrinol. 2004;181:367–77.

- McLachlan JC, Storey H. Hot male: can sex in humans be modified by temperature? J Theor Biol. 2003;222:71–2.
- Hunter RH. Temperature gradients in female reproductive tissues. Reprod Biomed Online. 2012;24:377–80.
- Standring S. Abdomen and pelvis. In: Gray's anatomy. 39th edn. Philadelphia: Elsevier, Churchill Livingstone; 2005, p. 66–109.
- Alberts B, Johnson A, Lewis J, Raff M, Roberts K, Walter P. In: Gibbs S, editor. Molecular biology of the cell. New York: Taylor & Francis; 2002. p. 1243–6.
- Frankenhuis MT, Smith-Buijs CM, de Boer LE, Kloosterboer JW. A case of combined hermaphroditism and autofertilisation in a domestic rabbit. Vet Rec. 1990;126:598–9.
- Kanamori A, Yamamura A, Koshiba S, Lee JS, Orlando EF, Hori H. Methyltestosterone efficiently induces male development in the self-fertilizing hermaphrodite fish, Kryptolebias marmoratus. Genesis. 2006;44:495–503.
- 52. Mackiewicz M, Tatarenkov A, Taylor DS, Turner BJ, Avise JC. Extensive outcrossing and androdioecy in a vertebrate species that otherwise reproduces as a selffertilizing hermaphrodite. Proc Natl Acad Sci U S A. 2006;103:9924–8.
- 53. Qur'an. 3–37: development of Mary as a beautiful plant; 66–12: Mary is called "Him".
- Irmak MK. Self-fertilization in human: having a male embryo without a father. Med Hypotheses. 2010;75:448–51.
- Irmak MK. Embryological basis of the virgin birth of Jesus. J Exp Integr Med. 2014;4:143–6.
- Alberts B, Johnson A, Lewis J, Raff M, Roberts K, Walter P. Sexual reproduction: meiosis, germ cells, and fertilization. In: Gibbs S, editor. Molecular biology of the cell. New York: Taylor & Francis Group; 2007. p. 1282–304.
- 57. Matsui Y. Developmental fates of the mouse germ cell line. Int J Dev Biol. 1998;42:1037–42.
- Hogan BL, Blessing M, Winnier GE, Suzuki N, Jones CM. Growth factors in development: the role of TGF-beta related polypeptide signalling molecules in embryogenesis. Dev Suppl. 1994;120:53–60.
- 59. Gardner RL, Rossant J. Investigation of the fate of 4–5 day post-coitum mouse inner cell mass cells by blastocyst injection. J Embryol Exp Morphol. 1979;52:141–52.
- Carlson BM. Patten's foundations of embryology. New York: McGraw-Hill; 1996. p. 75.
- Lawson KA, Hage WJ. Clonal analysis of the origin of primordial germ cells in the mouse. Ciba Found Symp. 1994;182:68–84.
- Anderson R, Copeland TK, Schöler H, Heasman J, Wylie C. The onset of germ cell migration in the mouse embryo. Mech Dev. 2000;91:61–8.
- Standring S. Implantation, placentation, pregnancy and parturition. In: Gray's anatomy. Philadelphia: Churchill Livingstone; 2008. p. 1250–1355.
- 64. Lawson KA, Dunn NR, Roelen BA, Zeinstra LM, Davis AM, Wright CV, Korving JP, Hogan BL. Bmp4 is
required for the generation of primordial germ cells in the mouse embryo. Genes Dev. 1999;13:424–36.

- 65. Fujiwara T, Dunn NR, Hogan BL. Bone morphogenetic protein 4 in the extraembryonic mesoderm is required for allantois development and the localization and survival of primordial germ cells in the mouse. Proc Natl Acad Sci U S A. 2001;98:13739–44.
- 66. Inman KE, Downs KM. The murine allantois: emerging paradigms in development of the mammalian umbilical cord and its relation to the fetus. Genesis. 2007;45:237–58.
- 67. Anderson R, Fässler R, Georges-Labouesse E, Hynes RO, Bader BL, Kreidberg JA, Schaible K, Heasman J, Wylie C. Mouse primordial germ cells lacking beta1 integrins enter the germline but fail

to migrate normally to the gonads. Development. 1999;126:1655-64.

- Downs KM, Harmann C. Developmental potency of the murine allantois. Development. 1997;124:2769–80.
- Gomperts M, Garcia-Castro M, Wylie C, Heasman J. Interactions between primordial germ cells play a role in their migration in mouse embryos. Development. 1994;120:135–41.
- Romero-Rojas AE, Messa-Botero OA, Melo-Uribe MA, Díaz-Pérez JA, Chinchilla-Olaya SI. Primary yolk sac tumor of the urachus. Int J Surg Pathol. 2011;19:658–61.
- Irmak MK. Why are some of embryonic stem cells localized out of the embryo? J Exp Integr Med. 2012;2:373–5.

Aneuploidy in Human Preimplantation Embryos

Joseph P.M. Geraedts

Introduction

Even when conditions are optimal, the chance of a clinically recognized pregnancy occurring in a given menstrual cycle is much less than half. Besides failure of conception pre-clinical pregnancy loss is a major cause for the low fecundity observed in humans. Chromosome abnormalities are the predominant cause of this clinical problem, both in natural conceptions as well as after assisted reproduction [21]. About 42 % of all oocytes exposed to natural fertilization survive until after implantation, and only 31 % will lead to a live birth (Fig. 17.1).

The number of abnormal zygotes cannot simply be calculated on the basis of the aneuploidy rates of gametes but may result from abnormal fertilisation. Furthermore mitotic abnormalities during cleavage division give rise to mosaic chromosome abnormalities of the embryos.

Department of Genetics and Cell Biology, Maastricht University Medical Center, P.O. Box 5800, Maastricht, AZ 6202, The Netherlands e-mail: joep.geraedts@mumc.nl

Lessons from Cytogenetics

The frequency of chromosome abnormalities in oocytes, zygotes and early stages of embryo development have been studied since the early 1980s of last century. After the introduction of fertilisation in the IVF laboratory as a means of infertility treatment it has become possible to study human gametes and the early stages of embryonic development. Due to legal restrictions in most countries it is not allowed to create embryos for study purposes and the superfluous embryos which originate within the context of infertility treatment do not reflect the normal situation. However, it is possible to obtain normal oocytes from couples with male factor infertility and visa versa. Furthermore, the introduction of molecular methods has allowed more unselected studies to be carried out. Fluorescence In Situ Hybridization (FISH) has first provided the opportunity to analyse the chromosomal content of the earliest conception products [28]. It allows the detection of aneuploidy in single cells. However, only a limited number of chromosomes can be studied using this method. Furthermore, the estimation of hypohaploidy (loss of chromosomes) is more problematic using FISH.

Therefore the more recent introduction of techniques that allow the simultaneous study of all chromosomes have contributed most. Metaphase comparative genomic hybridisation (mCGH) was

J.P.M. Geraedts

[©] Springer International Publishing Switzerland 2016 N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development: First and Second Trimesters*, DOI 10.1007/978-3-319-14874-8_17



originally developed for cytogenetic analysis of solid tumors [17]. At the beginning of this century it has been introduced for the study of preimplantation embryos [37, 38]. The principle of this method is that control and study DNA are simultaneously hybridised onto a normal metaphase. On the basis of a shift in the ratio of two colours quantitative fluorescence microscopy reveals chromosomes (in case of aneuploidy) and chromosomal regions (in case of unbalanced structural rearrangements) which are lost or amplified. Based on the same principle as metaphase CGH, array CGH differs in that genomic clones from selected regions of the genome are spotted on a slide, replacing normal 'control' metaphase cells as the target DNA [19]. Since then various other technologies have been developed that allow copy number analysis of all 23 pairs of chromosomes, 22 autosomes, and the sex chromosomes, or "24-chromosome" copy number analysis in single or small numbers of cells [13].

Aneuploidy in Oocytes

Since the first report on the study of meiosis in human oocytes [39], it has become known that aneuploidy frequently arises during female meiosis and that it is closely associated with advancing maternal age. The relationship between increasing maternal age and trisomy is well recognized [16]. The risk of trisomy in a clinically recognized pregnancy rises from about 2-3 % for women in their 20s to an astounding 30 % or more for women in their 40s [16].

Published reports of full karyotype analysis on the first polar body, confirmed by the results of MII oocyte analysis, indicate that there is a low aneuploidy rate (3 % in one study) in oocytes of young fertile women [8]. To shed further light on the nature of aneuploidy in human oocytes, CGH was also used to provide a detailed cytogenetic analysis of first and second polar bodies biopsied from the fertilized oocytes of women with an average age of more than 40 years [7, 9]. The total oocyte abnormality rate was 70 % and MII anomalies predominated over MI (50 % aneuploidy rate versus 40.3 %).

In a large-scale series of over 20,000 oocytes, aneuploidies were detected in polar bodies using FISH for chromosomes 13, 16, 18, 21 and 22. Almost every second oocyte (46.8 %) was found to be abnormal, with a predominance of errors involving chromatid loss, predicting predominance of trisomies (53 %) over monosomies (26 %) in the resulting embryos [18]. Of the detected anomalies in oocytes, 40 % were complex.

Therefore it is clear that both female meiotic divisions and all chromosomes can be, and frequently are, affected by errors in their segregation. In general, oocyte chromosome abnormalities increase dramatically in women over the age of 40 years. This steep increase in the presence of chromosome errors may reflect adverse intrinsic (oocyte) or extrinsic (ovarian or hormonal) effects related to advancing maternal age. However, it is noteworthy that the vast majority of existing data on maternally derived aneuploidy come from oocytes generated via ovarian stimulation, and it remains possible that oocytes from natural ovulatory cycles may exhibit some differences [7, 9].

There seems to be no effect of maternal age on triploidy and tetraploidy. These abnormalities originate at fertilisation and during preimplantation development respectively. This confirms that maternal age is a factor, which only influences meiosis.

The cause of non-disjunction in oocytes of older women is largely unknown. The classic textbook mechanism for aneuploidy arising in meiosis is non-disjunction of whole chromosomes or sister chromatids in the first (meiosis I) and second meiotic divisions (meiosis II), respectively [10]. Premature centromere division during meiosis-I has been suggested as an alternative mechanism for trisomy formation [1, 2]. This could be studied for the first time at a large scale in a series of women having aneuploidy testing by array CGH mainly because of advanced maternal age (mean age 40 years). Nearly three quarters (72 %) had one or more aneuploidies in either one or both polar bodies, which was confirmed in almost all corresponding fertilised oocytes [12]. By examining the copy number changes resulting from these meiotic errors, in the three products of female meiosis, that is, both polar bodies and the corresponding zygote, it was possible to distinguish non-disjunction from premature predivision of sister chromatids as the cause of aneuploidy in the corresponding zygotes. The results of this analysis demonstrated that almost all meiosis I errors for all chromosomes are caused by premature predivision of sister chromatids. As would be expected by the random segregation of single chromatids at meiosis II to either the PB2 or the zygote, however, a significant proportion of meiosis I errors did not result in aneuploidy in the zygote. Overall, therefore, there were more meiosis II-derived maternal aneuploidies in the zygote. Furthermore, over

half of the zygotes examined had multiple aneuploidies [14].

Aneuploidy in Spermatozoa

Using FISH the majority of chromosomes have been studied in spermatozoa. However specific data are lacking for chromosomes 5, 10, 11, 14, 17 and 19. The majority of authors report on chromosomes 13, 18 and 21 and the sex chromosomes, thus, the estimates of aneuploidy for these chromosomes are the most reliable. It has been found that disomy frequencies differ among chromosomes. For autosomes, disomy is in the order of 0.1 % but may range from 0.03 (chromosome 8) to 0.47 (chromosome 22). The great majority of authors find that chromosome 21 (0.17%) has a significantly elevated frequency of disomy, and that the sex chromosomes have the highest frequency (0.27 %), although these findings are not universal. The lower estimate of total aneuploidy in FISH studies is 4.5 % (2× disomy of 2.26 %) [33]. The variations observed between studies can be attributed to technical factors such as decondensation and denaturation of the DNA in the specimen and distortion of the morphology after pretreatment. Besides technical factors explaining a proportion of the variation observed, it should not be forgotten that differences might be observed between patients. It has been shown that the percentage of numerical abnormalities is significantly increased in a group of patients with abnormal semen characteristics [27, 35]. Finally it has been observed that sperm of non-obstructive azoospermic men had a higher incidence of chromosome abnormalities, of which sex chromosome abnormality was the most predominant [24]. A higher incidence of numerical chromosome abnormalities in sperm has also been detected in ICSI zygotes [20]. This is confirmed by the limited available data on ICSI foetal karyotypes which reveal that, in comparison with a general neonatal population, there is: (i) a slight but significant increase in de-novo sex chromosomal aneuploidy (0.6 % instead of 0.2 %) and structural autosomal abnormalities (0.4 % instead of 0.07 %) [34].

Abnormalities in Preimplantation Embryos

The reported rate of abnormal fertilisation observed during IVF varies from 2 % to about 9 %. About 20 h after insemination, the zygotes are studied for the presence of pronuclei. The two main categories of abnormalities are parthenogenetic activation (one pronucleus) and triploidy (three pronuclei). Tripronuclear zygotes result in most cases from dispermy [29].

Cytogenetic studies of preimplantation embryos carried out using classical techniques have reported rates of abnormalities which varied from 23 % to 90 % [21].

The development of FISH technology first enabled arrested human embryos to be studied. These studies revealed mostly mosaic chromosome complements [6, 15]. Also in dividing preimplantation embryos chromosomal mosaicism has been shown to be a normal feature of both morphologically abnormal [4, 25, 26] as well as normal embryos. Chromosomal mosaicism might be one of the reasons why the application of preimplantation aneuploidy screening to ensure transfer of euploid and improvements of live-birth rates from IVF has not worked using FISH [11]. The first studies applying the (metaphase) comparative genome hybridisation technique reported an identical result of 3 out of 12 preimplantation embryos consisting of only normal cells [36, 37]. In 2013 Mertzanidou et al. [22] used aCGH to study all blastomeres from 14 good-quality human preimplantation embryos from <35 years old) IVF patients. Only four of these were uniformly diploid, while the remaining ten were mosaic. None of the embryos had the same aneuploidy pattern in all cells. Of the 70 analysed blastomeres, 55.7 % were diploid and 44.3 % had chromosomal abnormalities, while 29 % of the abnormal cells carried structural aberrations.

It has been reported that during preimplantation development, a high prevalence of mosaicism persists to the blastocyst stage [3, 30]. However, the percentage of abnormal cells per embryo was 16 %, which is less than that of cleavage stage embryos. At the blastocyst stage aneuploidies were found to be significantly less common among embryos of optimal morphological quality, while such abnormalities were overrepresented amongst embryos considered to be of poor morphology. However, many blastocysts affected by forms of aneuploidy with the greatest capacity to produce clinical pregnancies, such as trisomy 21, were indistinguishable from euploid embryos [9]. Even though there is a strong selection against chromosomally abnormal embryos, extended culture to day 5 or 6 cannot be used as a reliable tool to select against clinically relevant chromosome abnormalities such as trisomies [31].

Already during preimplantation development the potential of human embryos depends on their chromosomal constitution. The majority of all normal embryos reach the blastocyst stage, while this is the case in only a minority of chromosomally abnormal embryos [31]. Since trisomies and triploidies are much more viable than monosomies and more complex aneuploidies, human miscarriage in the first trimester mainly results from embryos with extra chromosomes [32]. Recently a meta-analysis was performed to obtain a more precise evaluation of the risk of embryonic chromosomal abnormalities in firsttrimester miscarriage after ART in which a total of 15 studies were included. No statistical difference was found in risk of chromosomally abnorcompared mal miscarriage with natural conception and the different types of ART utilized, whereas the risk of foetal aneuploidy significantly increased with maternal age ≥ 35 [23].

Conclusion

To a large extend, the low fecundity of humans can be explained by preimplantation embryonic loss as a result of the occurrence of chromosomal abnormalities in the human conceptus. They can originate during gamete formation, at fertilization and during various stages of early embryo development. As a result of the latter, the majority of cleavage stage embryos created by IVF are chromosomal mosaics. The contribution of abnormalities via sperm is much less than via the oocyte. Aneuploidy of meiotic origin in older women mainly results from premature centromere division. In young women the vast majority of chromosomally abnormal embryos are caused by post-zygotic mitotic errors.

References

- Angell RR. Predivision in human oocytes at meiosis I: a mechanism for trisomy formation in man. Hum Genet. 1991;86:383–7.
- Angell RR. First-meiotic division nondisjunction in human oocytes. Am J Hum Genet. 1997;61:23–32.
- Bielanska M, Tan SL, Ao A. Chromosomal mosaicism throughout human preimplantation development in vitro: incidence, type, and relevance to embryo outcome. Hum Reprod. 2002;17:413–9.
- Bongso A, Ng SC, Lim J, Fong CY, Ratnam S. Preimplantation genetics: chromosomes of fragmented human embryos. Fertil Steril. 1991;56:66–70.
- Chard T. 11 Frequency of implantation and early pregnancy loss in natural cycles. Baillieres Clin Obstet Gynaecol. 1991;5(1):179–89.
- Coonen E, Harper JC, Ramaekers FC, Delhanty JD, Hopman AH, Geraedts JP, Handyside AH. Presence of chromosomal mosaicism in abnormal preimplantation embryos detected by fluorescence in situ hybridisation. Hum Genet. 1994;94:609–15.
- Fragouli E, Alfarawati S, Goodall NN, Sánchez-García JF, Colls P, Wells D. The cytogenetics of polar bodies: insights into female meiosis and the diagnosis of aneuploidy. Mol Hum Reprod. 2011;17:286–95.
- Fragouli E, Alfarawati S, Spath K, Wells D. Morphological and cytogenetic assessment of cleavage and blastocyst stage embryos. Mol Hum Reprod. 2014;20:117–26.
- Fragouli E, Wells D, Delhanty JDA. Chromosome abnormalities in the human oocyte. Cytogenet Genome Res. 2011;133:107–18.
- Gardner RJM, Sutherland GR. Chromosome abnormalities and genetic counseling. New York: Oxford University Press; 2004.
- Geraedts J, Collins J, Gianaroli L, Goossens V, Handyside A, Harper J, Montag M, Repping S, Schmutzler A. What next for preimplantation genetic screening? A polar body approach! Hum Reprod. 2010;25:575–7.
- 12. Geraedts J, Montag M, Magli MC, Repping S, Handyside A, Staessen C, Harper J, Schmutzler A, Collins J, Goossens V, van der Ven H, Vesela K, Gianaroli L. Polar body array CGH for prediction of the status of the corresponding oocyte. Part I: clinical results. Hum Reprod. 2011;26:3173–80.
- Handyside AH. 24-chromosome copy number analysis: a comparison of available technologies. Fertil Steril. 2013;100:595–602.
- Handyside AH, Montag M, Magli MC, Repping S, Harper J, Schmutzler A, Vesela K, Gianaroli L, Geraedts J. Multiple meiotic errors caused by predivi-

sion of chromatids in women of advanced maternal age undergoing in vitro fertilization. Eur J Hum Genet. 2012;20:742–7.

- 15. Harper JC, Coonen E, Ramaekers FC, Delhanty JD, Handyside AH, Winston RM, Hopman AH. Identification of the sex of human preimplantation embryos in two hours using an improved spreading method and fluorescent in-situ hybridization (FISH) using directly labelled probes. Hum Reprod. 1994;9: 721–4.
- Hassold T, Hunt P. Maternal age and chromosomally abnormal pregnancies: what we know and what we wish we knew. Curr Opin Pediatr. 2009;21:703–8.
- Kallioniemi A, Kallioniemi OP, Sudar D, Rutovitz D, Gray JW, Waldman F, Pinkel D. Comparative genomic hybridization for molecular cytogenetic analysis of solid tumors. Science. 1992;258:818–21.
- Kuliev A, Zlatopolsky Z, Kirillova I, Spivakova J, Cieslak Janzen J. Meiosis errors in over 20,000 oocytes studied in the practice of preimplantation aneuploidy testing. Reprod Biomed Online. 2011;22:2–8.
- Le Caignec C, Spits C, Sermon K, De Rycke M, Thienpont B, Debrock S, Staessen C, Moreau Y, Fryns JP, Van Steirteghem A, Liebaers I, Vermeesch JR. Single-cell chromosomal imbalances detection by array CGH. Nucleic Acids Res. 2006;34:e68.
- Macas E, Imthurn B, Keller PJ. Increased incidence of numerical chromosome abnormalities in spermatozoa injected into human oocytes by ICSI. Hum Reprod. 2001;16:115–20.
- Macklon NS, Geraedts JP, Fauser BC. Conception to ongoing pregnancy: the 'black box' of early pregnancy loss. Hum Reprod Update. 2002;8:333–43.
- 22. Mertzanidou A, Wilton L, Cheng J, Spits C, Vanneste E, Moreau Y, Vermeesch JR, Sermon K. Microarray analysis reveals abnormal chromosomal complements in over 70% of 14 normally developing human embryos. Hum Reprod. 2013;28:256–64.
- Qin JZ, Pang LH, Li MQ, Xu J, Zhou X. Risk of chromosomal abnormalities in early spontaneous abortion after assisted reproductive technology: a metaanalysis. PLoS One. 2013;8:e75953.
- Palermo GD, Colombero LT, Hariprashad JJ, Schlegel PN, Rosenwaks Z. Chromosome analysis of epididymal and testicular sperm in azoospermic patients undergoing ICSI. Hum Reprod. 2002;17:570–5.
- Pellestor F, Dufour MC, Arnal F, Humeau C. Direct assessment of the rate of chromosomal abnormalities in grade IV human embryos produced by in-vitro fertilization procedure. Hum Reprod. 1994;9:293–302.
- Pellestor F, Girardet A, Andreo B, Arnal F, Humeau C. Relationship between morphology and chromosomal constitution in human preimplantation embryo. Mol Reprod Dev. 1994;39:141–6.
- 27. Pfeffer J, Pang MG, Hoegerman SF, Osgood CJ, Stacey MW, Mayer J, Oehninger S, Kearns WG. Aneuploidy frequencies in semen fractions from ten oligoasthenoteratozoospermic patients donating sperm for intracytoplasmic sperm injection. Fertil Steril. 1999;72:472–8.

- Pieters MH, Geraedts JP, Meyer H, Dumoulin JC, Evers JL, Jongbloed RJ, Nederlof PM, van der Flier S. Human gametes and zygotes studied by nonradioactive in situ hybridization. Cytogenet Cell Genet. 1990;53:15–9.
- Plachot M, Mandelbaum J, Junca AM, de Grouchy J, Salat-Baroux J, Cohen J. Cytogenetic analysis and developmental capacity of normal and abnormal embryos after IVF. Hum Reprod. 1989;4(Suppl):99–103.
- Ruangvutilert P, Delhanty J, Serhal P, Simopoulou M, Rodeck C, Harper J. Fish analysis on day 5 postinsemination of human arrested and blastocyst stage embryos. Prenat Diagn. 2000;20:552–60.
- Sandalinas M, Sadowy S, Alikani M, Calderon G, Cohen J, Munné S. Developmental ability of chromosomally abnormal embryos to develop to the blastocyst stage. Hum Reprod. 2001;16:1954–8.
- Simpson JL. Causes of fetal wastage. Clin Obstet Gynecol. 2007;50:10–30.
- Templado C, Vidal F, Estop A. Aneuploidy in human spermatozoa. Cytogenet Genome Res. 2011;133:91–9.
- Van Steirteghem A, Bonduelle M, Devroey P, Liebaers I. Follow-up of children born after ICSI. Hum Reprod Update. 2002;8:111–6.

- 35. Vegetti W, Van Assche E, Frias A, Verheyen G, Bianchi MM, Bonduelle M, Liebaers I, Van Steirteghem. Correlation between semen parameters and sperm aneuploidy rates investigated by fluorescence in-situ hybridization in infertile men. Hum Reprod. 2000;15:351–65.
- Voullaire L, Slater H, Williamson R, Wilton L. Chromosome analysis of blastomeres from human embryos by using comparative genomic hybridization. Hum Genet. 2000;106:210–7.
- Wells D, Delhanty JDA. Comprehensive chromosomal analysis of human preimplantation embryos using whole genome amplification and single cell comparative genome hybridization. Mol Hum Reprod. 2000; 6:1055–62.
- Wilton L, Williamson R, McBain J, Edgar D, Voullaire L. Birth of a healthy infant after preimplantation confirmation of euploidy by comparative genomic hybridization. N Engl J Med. 2001;345:1537–41.
- Yuncken C. Meiosis in the human female. Cytogenetics. 1968;7:234–8.

The Potential Impact of Maternal Milk Consumption During Pregnancy on mTORC1-Driven Fetal Growth

GHR

GIP

RAS

Growth hormone receptor

Glucose-dependent

oncogene homolog

gene homolog

V-HA-RAS rat sarcoma viral onco-

$\mathbf{18}$

insulinotropic

Bodo C. Melnik

Abbreviations

4E-BP-1	Eukaryotic translation initiation fac-		polypeptide
	tor 4E-binding protein 1	GIPR	GIP receptor
AAs	Amino acids	GLP-1	Glucagon-like peptide-1
AGA	Appropriate for gestational age	GRB2	Growth factor receptor-bound protein 2
AKT	V-AKT murine thymoma viral onco-	HPL	Human placental lactogen
	gene homolog (protein kinase B)	IGF-1	Insulin-like growth factor-1
AMP	Adenosine monophosphate	IGF1R	Insulin-like growth factor-1 receptor
AMPK	AMP-activated protein kinase	IR	Insulin receptor
BCAA	Branched-chain amino acid	IRS	Insulin receptor substrate
BM	Basal membrane	JAK	Janus-activated kinase
BMI	Body mass index	LAT	L-type amino acid transporter
EIF4A	Eukaryotic translation initiation fac-	Leu	Leucine
	tor 4A	LGA	Large for gestational age
EIF4E	Eukaryotic translation initiation fac-	MEK	Mitogen-activated protein kinase
	tor 4E		kinase 1
ERK	Mitogen-activated protein kinase	MiR	Micro-ribonucleic acid
	kinase 4	mTORC1	Mechanistic (mammalian) target of
FFA	Free fatty acid		rapamycin complex 1
FGF21	Fibroblast growth factor 21	MVM	Microvillous plasma membrane
GDH	Glutamate dehydrogenase	PDCD4	Programmed cell death 4
GDM	Gestational diabetes mellitus	PGC-1a	Peroxisome proliferator-activated
GF	Growth factors		receptor-γ coactivator-1α
GH	Growth hormone	PGH	Placental growth hormone
		PI3K	Phosphoinositide-3 kinase
		PRLR	Prolactin receptor
B.C. Melnik		PTEN	Phosphatase and tensin homolog
Department of Dermatology,		RAF	V-RAF-1 murine leukemia viral

Environmental Medicine and Health Theory,

University of Osnabrück, Sedanstrasse 115, Osnabrück 40090, Germany

e-mail: melnik@t-online.de

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_18

[©] Springer International Publishing Switzerland 2016

RHEB	RAS homolog enriched in brain
S6K1	Ribosomal protein S6 kinase 70-kD
	kinase 1
SOCS	Suppressor of cytokine signaling
SPRY2	Sprouty2
Trp	Tryptophan
TSC2	Tuberin

Introduction

Continued consumption of cow's milk and dairy products during adolescence, adulthood and pregnancy is an evolutionarily novel behavior that may have long-term adverse effects on human health [1]. Milk is a highly specialized, complex signaling system developed by mammalian evolution to promote mTORC1-driven postnatal growth [2]. Milk is not designed by nature to promote prenatal fetal growth. The consumption of milk of another mammal during pregnancy is an exceptional behavior of Neolithic humans and is not observed in any other mammal on the planet. Milk is the only and sufficient nutrient system that allows adequate postnatal growth and anabolism of the newborn milk recipient. On the molecular level, cell growth, cell proliferation, nucleotide-, protein- and lipid synthesis, and other anabolic metabolic processes, and inhibition of autophagy are all mediated by the nutrient-sensitive kinase mechanistic target of rapamycin complex 1 (mTORC1) [3–12].

After a short introduction into the major regulatory pathways activating mTORC1, mTORC1activating nutritional and hormonal signals induced by persistent cow's milk and dairy protein consumption will be presented. After that, the effect or increased insulin/IGF-1 and amino acid signaling induced by cow's milk consumption on placental growth and placental nutrient transfer will be dissected. Finally, recent epidemiological evidence will be presented that confirms the association between increased maternal weight gain during pregnancy, increased placental weight, enhanced fetal growth and increased birth weight with milk consumption during pregnancy. In contrast to cow's milk consumption, inhibition of placental mTORC1 and insulin/ IGF-1 signaling resulted in downregulation of placental nutrient transporters and link maternal undernutrition to mTORC1-mediated restriction of fetal growth [13].

mTORC1: The Nutrient-Sensitive Kinase Orchestrating Cell Growth

The nutrient- and growth factor-sensitive kinase mTORC1 is the cell's central regulator of growth and proliferation [3–12]. mTORC1 activates nucleotide, protein and lipids synthesis under conditions of nutrient availability. mTORC1 is activated at the lysosomal membrane. Two major signaling pathways are of pivotal importance for mTORC1 activation: (1) the growth factor-PI3K-AKT pathway that activates the GTPase RHEB and (2) the translocation of mTORC1 to the lysosomal site of activation via amino acid-mediated activation of the small GTPase RAGA and RAGB.

mTOR is a multi-domain protein of approximately 300 kDa exhibiting a serine/threonine protein kinase domain at its C-terminus related to phosphoinositol-3-kinases (PI3Ks). In mammalian cells two functionally different mTOR complexes exist: mTORC1 and mTORC2, respectively [3–7]. Among other functional proteins, mTORC1 contains the important partner protein RAPTOR, which interacts with substrates mTORC1-mediated for phosphorylation. mTORC1 controls the G₁/S transition and G₂/M progression of the cell cycle [14]. In contrast to mTORC2, which contains the partner protein RICTOR, only mTORC1 plays a special role in sensing cellular nutrients such as glucose, amino acids, palmitic acid, energy (ATP) levels, and cell stress (ROS) all important for the regulation of cell growth and proliferation.

Insulin/IGF-1 Signaling Activates mTORC1

Growth factor signals such as insulin and IGF-1 are integrated by the tuberous sclerosus protein TSC1 (hamartin) and TSC2 (tuberin) that regulate

RHEB (RAS-homolog enriched in brain), one essential activator of mTORC1 [15–20]. In its GTP-bound form, RHEB directly activates mTORC1. The RHEB-specific GTPase-activating protein (GAP) is the TSC2 protein, which functions as a heterotrimer with its binding partners TSC1 and TBC1D7 [20]. Growth factor signaling via TSC2 phosphorylation reduces the inhibitory function of the TSC1/TSC2/TBC1D7 complex towards RHEB which results in activation of RHEB and finally of mTORC1.

Amino Acid-Mediated Activation of mTORC1

Amino acids, especially leucine, glutamine and arginine, play a most important role for the activation of mTORC1 [5–11]. Amino acids activate mTORC1 even in the absence of insulin but not vice versa [9, 21, 22]. The activation of mTORC1 depends on two major pathways: (1) by the upstream activation of RHEB by signals derived from growth factors and (2) by the amino aciddependent translocation of inactive mTORC1 to active RHEB localized in lysosome compartments [23–25]. Insulin and IGF-1 signaling, via activated AKT phosphorylate TSC2 and thereby suppress the inhibitory function of the TSC1/ TSC2/TBC1D7 complex towards RHEB. The TSC complex associates with the lysosome in a RHEB-dependent manner, and its dissociation in response to insulin requires AKT-mediated TSC2 phosphorylation. These recent findings provide a unifying mechanism by which independent pathways affecting the spatial recruitment of mTORC1 and the TSC complex to RHEB at the lysosomal surface serve to integrate diverse growth signals [26]. The inhibition of either TSC1 or TSC2 leads to activation of RHEB and ultimately of mTORC1 [19, 26, 27].

Amino acid uptake into the cell is crucial for mTORC1 signaling. Nicklin et al. [28] suggested that cellular export of glutamine is required for cellular leucine uptake and subsequent leucinemediated mTORC1 activation. Intracellular glutamine is required for preloading the SLC7A5/ SLC3A2 bidirectional amino acid transporter, which drives the efflux of glutamine and influx of leucine for leucine-mediated mTORC1 activation [28, 29]. Remarkably, in response to amino acid depletion, mTORC1 activity is rapidly abolished [30], and amino acid starvation even impairs binding of mTORC1 to RHEB [31].

Amino acids play a pivotal role in the translocation of inactive mTORC1 to lysosomal compartments enriched in activated RHEB [23, 24]. The spatial regulation of inactive mTORC1 by amino acids is mediated by an active RAG heterodimer and is of crucial importance for amino acid sensing and activation of mTORC1 [32]. The pentameric Ragulator complex acts as a scaffold for the RAG GTPases and mTORC1 at the lysosomal membrane. According to the recent opinion, RHEB and RAGs come together at the lysosome to activate mTORC1 [26, 33] (Fig. 18.1).

Amino acid accumulation in the lysosomal lumen generates an activating signal that is transmitted in a vacuolar H+-ATPase (v-ATPase)dependent fashion to activate the guanine nucleotide exchange factor (GEF) activity of Ragulator towards RAGA. Upon RAGA-GTP loading, mTORC1 is recruited to the lysosomal surface where it interacts with RHEB and becomes activated [32]. Thus, mTORC1 integrates insulin, IGF-1, energy-derived signals to RHEB but in parallel requires sufficient amino acid signals for maximal mTORC1 activity [34]. Recent evidence has been provided that proton-assisted amino acid transporters (PATs) localized on late endosomes and lysosomes (LEL) interact with RAGs and are required for mTORC1 activation [35]. PAT1 (SLC36A1) is expressed at the luminal surface of the small intestine and is commonly found in lysosomes of many cell types [36]. The v-ATPase interacts with the activated RAG/Ragulator complex to control amino acid-dependent mTORC1 activation, which is regulated by the rapid accumulation of extracellular amino acids in LELs [37, 38]. Thus, in response to amino acids these molecules form a signaling complex that has been called the 'nutrisome' [35, 37]. Cycling of protons through this nutrisomal engine induces conformational changes that may activate mTORC1. Importantly, signaling from the insulin receptor



Fig. 18.1 Synopsis of milk-modified mTORC1-signaling of the human syncytiotrophoblast. Leucine (*Leu*) and glutamine (*Gln*) derived from whey protein hydrolysis increase postprandial serum insulin levels. Milk protein consumption increases serum levels of IGF-1. Leucine, insulin and IGF-1 stimulate mTORC1 of the syncytiotrophoblast and placenta growth in general. Milk consumption is associated with placental growth. Placental weight is related to increased serum levels of human placental lactogen (*HPL*). HPL via STAT5/SOCS signaling as well as increased maternal mTORC1/S6K1 signaling induce insulin resistance increasing maternal serum levels of insulin and

and subsequent activation of the PI3K/AKT/ RHEB cascade promotes shuttling of PATs from the cell surface to LEL membranes, hence increasing PAT-dependent mTORC1 activation and cell growth [35, 37]. In addition, the accumulation of amino acids in the LEL lumen presumably involves transport into intracellular endosomal compartments via currently unknown amino acid transporters (AATs) or potentially via endocytosis. Cytoplasmic leucine, which is brought into cells via the heterodimeric amino acid transporter CD98 [39], has been shown to play a key role in activating mTORC1 in some cultured cells and may be important in this process. Influx of leucine or other amino acids into the LEL system may ultimately allow the amino acid substrates of PAT1 to accumulate in the LELs through amino acid exchange mechanisms, leading to PAT1mediated activation of the nutrisome [35].

glucose. Increased trophoblast mTORC1 activity and HPL stimulate placental expression of FGF21 that upregulates GLUT1 expression. Thus milk consumption promotes exaggerated glucose transfer to the fetus. As trophoblast mTORC1 activity controls the expression of L-type amino acid transporters (*LAT*) (*dotted line*), milk not only provides abundant leucine but also stimulates trophoblast-mediated leucine transfer to the fetus. Both glucose and leucine are pivotal stimuli required for mTORC1-driven fetal growth. Milk, a signaling system designed for postnatal growth, apparently interferes with prenatal growth trajectories when "abused" by humans during pregnancy

Leucyl-tRNA synthetase (LeuRS) acts as a cytosolic amino acid sensor [40, 41]. LeuRS also plays a critical role in amino acid-induced mTORC1 activation by sensing intracellular leucine concentration and initiating mTORC1 activation by binding to and activating RAG GTPase [40, 41]. LeuRS acts as a GTPase-activating protein (GAP) for RAGD, enhancing the GTP-bound form of RAGA/B crucial for amino acid-mediated mTORC1 activation [41].

The Glutaminolysis Pathway Activates mTORC1

Duran et al. [42] suggested that leucine stimulates mTORC1 indirectly through its effects on glutaminolysis. Glutamine in combination with leucine increased GTP charging of exogeneously expressed RAGB, promoting mTORC1 activation by enhancing glutaminolysis and the production of α -ketoglutarate (α -KG) [42]. In contrast, the eIF2 α (eukaryotic initiation factor 2 α) kinase GCN2 (general amino acid control-nondepressible 2) senses the absence of one or more amino acids by virtue of direct binding to uncharged cognate tRNAs [43].

Palmitate Activates mTORC1

It has recently been demonstrated that the saturated C16:0 fatty acid palmitate activates mTORC1 [44]. Palmitate activated mTORC1 by enhancing recruitment of mTOR onto lysosomal membranes [44]. Notably, the Ragulator complex component p18 is anchored to the lysosomal membrane via palmitoylation, however palmitate did not alter localization of either p18 or RAGC proteins [44]. Thus, the mechanism of palmitate-induced mTORC1 activation is yet unknown.

Glucose Activates mTORC1

Glucose is the major carbohydrate and energy source of the cell that via glycolysis is converted into pyruvate. The first step of glycolysis is the phosphorylation of glucose by hexokinase-II to form glucose 6-phosphate. The free energy released in the process of glycolysis and subsequent downstream oxidation of citrate in the tricarboxylic acid cycle is used to form the high energy compound adenosine triphosphate (ATP). High cellular ATP levels result in low levels of adenosine monophosphate (AMP). In states of starvation or restriction of glucose increased levels of AMP activate the energy-sensing enzyme AMP-activated protein kinase (AMPK). Liver kinase B1 (LKB1) and AMPK are critical regulators of mTORC1 [45, 46]. The serine/threonine kinase LKB1 represents the major kinase phosphorylating the AMPK activation loop (α-subunit of AMPK) under conditions of energy stress [47, 48]. Thus, AMPK plays a key role in energydependent regulation of mTORC1. AMPK is activated during energy-deficient conditions,

when AMP levels rise. AMPK phosphorylates TSC2 and RAPTOR, thereby suppressing mTORC1 activity [15]. Activation of AMPK inhibits the mTORC1 pathway, activates the autophagy initiating kinase ULK-1 [49, 50]. In response to glucose deprivation, hexokinase-II, which catalyzes the first step of glycolysis, binds to mTORC1 through its TOS motif, decreasing mTORC1 activity [51, 52]. Thus, there exists a multilayer regulation of glucose-mediated mTORC1 activation [53].

Cow's Milk Consumption Increases mTORC1 Signaling

As outlined above, mTORC1 is activated by four major pathways: (1) growth factor signaling via insulin and IGF-1, (2) amino acid signaling predominantly via leucine, (3) glutaminolysismediated signaling, and (4) saturated fatty acid-derived signals primarily via palmitate. It will be shown that persistent cow's milk consumption in humans stimulates all four signaling pathways activating mTORC1 of the cell's of the milk consumer (Fig. 18.1).

Milk Induces Hyperinsulinemia

Milk is a materno-neonatal vector system that stimulates insulin production of pancreatic β -cells of the milk recipient, physiologically of the species-specific newborn mammal until weaning [2]. The *insulinemic index* of whole cow's milk (148 ± 14) and skim milk (140 ± 13) is much higher than the glycemic indices of whole milk (42 ± 5) and skim milk (37 ± 9) , respectively [54, 55]. Fast hydrolysis and immediate intestinal absorption of insulinotropic amino acids of the whey protein fraction of cow's milk [56, 57] raises insulin levels of much higher magnitudes than intestinal digestion of structural proteins such as beef (insulinemic index: 51). A recent study in healthy humans confirmed that maximal plasma amino acid increase occurs within the first hour after whey protein ingestion [58]. Whey proteins, the animal proteins that contain the highest amounts insulinotropic branched-chain

amino acids (BCAAs), are the proteins that exhibit fastest hydrolysis [59]. The major insulinotropic protein fraction of cow's milk is the whey protein fraction in comparison to the casein fraction that preferentially elevates serum IGF-1 levels [60]. Furthermore, whey-derived leucine and other whey-derived amino acids stimulate the incretin secretion of enteroendocrine K- and L-cells [61, 62]. Milk is a unique carrier of essential BCAAs. Whey protein contains the highest amount of leucine (14 %), followed by total milk proteins (10 % leucine), and casein (10 %) compared to egg protein (8.5 %), meat protein (8.0 %), soy protein (8.0 %), and wheat protein (7.0 %), respectively [63]. Enteroendocrine Kand L- cells sense whey protein-derived amino acids, especially leucine, and respond by release of the incretins glucose-dependent insulinotropic polypeptide (GIP) and glucagon-like peptide-1 (GLP-1) [64–66] (Fig. 18.1). In the neonatal breastfeeding period glucose-dependent insulinotropic polypeptide (GIP) signals not primarily in response to glucose but predominantly in response to whey proteins, thus operates as a whey-dependent insulinotropic polypeptide ("WIP").

Apart from the incretin induction of whey proteins, whey-derived amino acids, predominantly leucine and glutamine, exert direct inulinotropic effects on pancreatic β -cells [67–69]. It is thus fully comprehensible, that milk protein consumption in comparison to meat protein intake results in hyperinsulinemia as demonstrated in healthy 8-year old boys differentially fed with equal amounts of either milk or meat protein [70].

In obesity, an unfavorable condition promoting fetal overgrowth, plasma leucine levels are significantly elevated [71]. It has been demonstrated in obese children that additional supply of leucine resulted in excessive hyperinsulinemia [72]. Thus, there appears to be a synergism of hyperinsulinemia mediated by milk-derived leucine intake and obesity-associated hyperleucinemia. Both metabolic deviations may be unfavorable during pregnancy and may adversely affect placental insulin-dependent nutrient transfer to the fetus as discussed later.

Milk Increases Serum IGF-1 Levels

A meta-analysis confirmed that continued milk consumption increases serum levels of the growth hormone insulin-like growth factor-1 (IGF-1) [73]. The European Prospective Investigation into Cancer and Nutrition [74] confirmed a positive relationship between milk intake in 2,109 European women with increased IGF-1 serum levels. An increase in serum IGF-1 concentration of more than 20 % has also been observed in prepubertal children not used to milk consumption after a 4-week intake of 710 mL of whole milk daily [75]. A recent study including 193 overweight adolescents aged 12-15 years drank either 1 L/day of skimmed milk, whey, casein or water for 12 weeks. All milk-based-drinks contained 35 g milk protein/L. IGF-1 significantly increased with skimmed milk and tended to increase with casein compared to the pre-test control group [76]. Casein in comparison to whey protein has been shown to differentially enhance hepatic IGF-1 synthesis [58] (Fig. 18.1). After the widespread implementation of household cooling technology in the 1950s per capita cheese consumption, the major dairy source of casein, increased in Germany by the factor 5 from 5 kg in the early 1950s to nearly 24 kg in 2013 [77]. Taken together, milk consumption has a profound IGF-1-enhancing effect on prepubertal, pubertal and adult humans, thus increases serum IGF-1 levels of young adult women in the period of reproduction.

Milk-Derived Amino Acids Activate mTORC1

Milk proteins in comparison to all other animal proteins provide highest amounts of essential BCAAs, especially highest amounts of leucine [64]. Milk protein (8.09 g glutamine/100 g) in comparison to beef protein (4.75 g glutamine/100 g) provides 70 % more glutamine than beef [78]. Glutamine is an important activator of mTORC1 signaling via (1) its function as a gate-keeper for the uptake of the mTORC1-activating amino acid leucine [28] and (2) via its availability

as a precursor of the glutaminolysis pathway that activates mTORC1 [42]. Furthermore, leucine is an allosteric activator of glutamate dehydrogenase (GDH), the key regulating enzyme of the glutaminolysis pathway [79]. On the other hand, GDH contributes to leucine sensing in the regulation of autophagy by activating mTORC1 [80]. Remarkably, the combination of glutamine and leucine, which maximizes the flux through GDH, is most effective in phosphorylation of S6K in β -cells [81]. Thus, leucine and glutamine, two predominant amino acids of cow's milk, interact at most critical steps of amino acid signaling resulting in the upregulation of mTORC1.

Milk-Derived Palmitate Activates mTORC1

Bovine milk contains about 3.5-5 % total lipid, existing as emulsified globules $2-4 \mu m$ in diameter and coated with a membrane derived from the secreting cell. In homogenized milk, the coat is mostly casein. About 98 % or more of the lipid is triacylglycerol, which is found in the globule [82]. The major fatty acid of the total fatty acid composition of bovine milk lipids is palmitate (C16:0) with 32.3 wt.% [82, 83]. Palmitate like BCAAs has recently been demonstrated to enhance the lysosomal translocation of mTORC1, a most critical step for the activation of mTORC1 [44].

In obesity and states of insulin resistance, in addition to increased serum levels of essential BCAAs [84–86], palmitate serum levels are significantly increased [87–89]. Thus, milk and dairy consumption in obese women enhances already elevated amino acid- and fatty acid-mediated signaling promoting the activation mTORC1.

Taken together, milk provides the ideal "cocktail of mammalian evolution" for the activation of mTORC1 by increasing insulin/IGF-1 signaling [90, 91], by providing essential BCAAs and abundant glutamine [64, 78], and by transferring palmitate, the fatty acid activator of mTORC1 [44, 82]. In fact, the activation of mTORC1 via measurement of increased phosphorylation of the mTORC1 target S6K1 in liver and white adipose tissue of mice fed on cow's milk has been demonstrated [92]. It has been reported that milk consumption increased body mass index and body weight in children [93, 94], adolescents [95], and adults [96]. Thus, the question arises whether milk protein intake in pregnancy may increase placental weight and upregulate mTORC1-mediated growth and nutrient transfer of the syncytiotrophoblast to the fetus. Furthermore, it will discussed whether the milk-induced increase of placenta mass increases the amount and signaling of placental hormones such as human placental growth hormone (PGH) and human placental lactogen (HPL).

Milk-Derived Glucose Activates mTORC1

Lactose is a disaccharide sugar derived from galactose and glucose and makes up around 5 % of milk by weight. Lactose is hydrolyzed by the enzyme lactase encoded by the LCT gene, which is physiologically expressed along the brush border membrane of differentiated enterocytes lining the villi of the small intestine during the period of milk feeding [97]. Since lactase's only function is the digestion of lactose in milk, in most mammalian species the activity of lactase is dramatically reduced after weaning [98]. Lactase persistence has resulted from mutations of the LCT gene that allowed persistent milk consumption in European and African populations [99, 100]. However, the "original" genetic design of mammalian biology is exclusive and species-specific milk consumption restricted to the postnatal growth phase until weaning. Lactase persistent pregnant women who may consume 1 L cow's milk are thus able to metabolize approximately 25 g of glucose, a substantial amount that may further activate milkmediated mTORC1 signaling.

Milk Consumption in Pregnancy Increases Placental, Fetal and Birth Weight

As milk promotes mTORC1 signaling [2] and has been demonstrated to increase linear growth in children [101] as well as BMI in children, adolescents and adults [93-96], it is most conceivable that increased milk consumption may influence placental growth as well. Data from 50,117 mother-infant pairs of the *Danish National Birth Cohort* collected from 1996 to 2002 showed a placental weight increase across the whole range of milk intake [102]. A linear increment of placental weight from 13.3 g (0–1 glass of milk/day) to 26.4 g (>6 glasses of milk/day) (p<0.001) has been reported.

An increase in placental weight may not only increase the amount and capacity of nutrient transporters for nutrient transfer to the fetus but may also increase the amount of placentaderived growth hormones that may decrease maternal insulin sensitivity thereby raising maternal blood glucose levels promoting fetal growth and birth weight. The Generation RStudy, a population-based prospective cohort study from fetal life until young adulthood in Rotterdam investigated 3,405 mothers during pregnancy. Maternal milk consumption of >3 glasses (450 mL of milk) per day was associated with greater fetal weight gain in the third trimester of pregnancy, which led to an 88 g higher birth weight than that with milk consumption of 0–1 glass per day [103]. Intriguingly, maternal protein intake, but not fat or carbohydrate intake, from dairy products was associated with higher birth weight [103]. This association was limited to milk, whereas protein intake from nondairy food or cheese was not associated with birth weight. Compared with the lowest reference category of milk consumption (0-1 glasses/ day), maternal milk intakes of >1-2 glasses/day, 2-3 glasses/day, and >3 glasses/day was associated with increased fetal weight gain. Fetal weight gain has been estimated by morphometric ultrasound examinations during pregnancy according to the procedure of Hadlock et al. [104]. Milk-mediated differences in fetal weight gain appeared from 20th week onward, but became most evident in the last part of the third trimester [103]. There are other international studies, which report an increase of birth weight in relation to milk consumption during pregnancy (Table 18.1). A retrospective cohort in Sweden reported a birth weight increase of 75 g and 134 g in the offspring of mothers consuming >200 mL and 1 L milk daily, respectively [105]. A prospective study in India reported that the frequency of milk consumption at 18th week of gestation was positively associated with placental weight, birth weight, birth length, and head circumference [106]. According to a prospective study in Canada, maternal daily consumption of an additional 1 cup (237 mL) of milk was associated with a 41 g increase in offspring birth weight [107]. A prospective Australian study in 557 mothers reported that protein intake from dairy products was associated with a higher birth weight of the offspring [108]. In a randomized controlled trial of 72 adolescent pregnant mothers, 25 mothers were counseled to consume >4 servings of dairy products a day, which resulted in a 240 g higher birth weight in this group compared to the control group [109]. According to a recent systematic literature review, the majority of studies reported positive associations between milk and/or dairy consumption and birth weightrelated outcomes [110].

Taken together, accumulating evidence underlines that milk consumption during pregnancy increases placental weight, fetal growth and birth weight as well as birth size (Table 18.1). Noteworthy, significant and dose-dependent increase in fetal growth and birth weight is associated with milk protein intake during pregnancy, whereas no effect was observed with cheese protein intake [102, 103]. Future studies should elucidate the differential contribution of the whey protein and the casein fraction of milk on the growth promoting effects of the placenta and fetal growth. In comparison to caseins, which clot in the stomach, whey proteins are fast proteins that are easily hydrolyzed in the intestine and deliver immediate "bursts" of BCAAs [57-59]. Whey-derived insulinotropic BCAA pulses apparently induce postprandial hyperinsulinemia [60, 64–69]. Notably, whey-induced BCAA and insulin pulses may evoke exaggerated effects on syncytiotrophoblast mTORC1 signaling.

Effect of milk intake	Outcome	Studies [references]
Pre-pregnancy body weight	Increase	Randomized intervention study, Denmark [95]
Gestational weight gain	Increase	Observational study, Iceland [192]
Placenta weight	Increase	Danish National Birth Cohort, Denmark [102]
		Pune Maternal Nutrition Study, India [106]
Fetal weight	Increase	Generation R Study, Netherlands [103]
Birth weight	Increase	Generation R Study, Netherlands [103]
		Observational study, Sweden [105]
		Pune Maternal Nutrition Study, India [106]
		Prospective observational study, Australia [108]
		Randomized controlled trial, USA [109]
		Systematic literature review, Norway [110]

 Table 18.1 Impact of milk consumption during pregnancy on maternal-fetal weight parameters

Milk Consumption Placental Weight Gain and Growth Hormone Signaling

Milk-induced weight gain of the placenta may modify the synthesis and amount of placental growth hormones. Changes in fetal growth associated with differential gene expression of PGH and HPL (also called chorionic somatomammotropin, CSH) may be modulated by maternal nutrinonal status [111]. CSH1, the predominant transcript of HPL is increased in placentas of large for gestational age (LGA) pregnancies [111]. Notably, maternal blood levels of HPL are correlated with placenta weight [112–114] and fetal weight [114– 117]. Moreover, a link between fetal growth velocities in the second half of the pregnancy and changes in HPL in maternal serum has been demonstrated [118]. In LGA newborns the expression of *CSH1-1*, *CSH2-1*, and *CSHL1-4* mRNA transcripts in placenta was significantly increased compared with appropriate for gestational age (AGA) newborns [119]. Women with LGA newborns had an increased BMI before pregnancy (25 kg/m²), an increased gestational weight gain (19 kg), and placental weight (777.6 g) compared to AGA newborns associated with a normal maternal BMI before pregnancy of 22.4 kg/m², a gestational weight gain of 15.8 kg, and a placental weight of 650 g, respectively [119].

Recent evidence indicates that maternal serum PGH is positively associated with fetal growth in the first half of pregnancy [120]. It is not known whether PGH correlates with placental weight. However, PGH acts to augment the supply of fatty acids and glucose to the placenta by the induction of insulin resistance [121–123]. Several studies suggest an important role for both PGH and HPL in the control of fetal growth [124–126].

Milk- and Placental Growth Hormone-Induced Insulin Resistance

The somatogenic and lactogenic hormones of the placenta and maternal pituitary gland integrate the metabolic adaptations of pregnancy with the demands of fetal and neonatal development. Dysregulation of PGH and/or HPL in pathologic conditions of pregnancy may adversely impact fetal growth and postnatal metabolic function [127]. In addition to promoting growth of maternal tissues, PGH induces maternal insulin resistance and thereby facilitates the mobilization of maternal nutrients for fetal growth. HPL and prolactin increase maternal food intake by induction of central leptin resistance and promote maternal β -cell expansion and insulin production [127]. Maternal insulin resistance increases serum glucose levels and thus the glucose supply to the placenta.

Milk is the animal protein source providing the highest amounts of essential BCAAs [64],

which have recently been linked with the development of insulin resistance [84-87]. It has been shown in children, who in comparison to adults exhibit increased pituitary GH levels, that milk consumption increases serum insulin, GH and IGF-1 levels and induces insulin resistance [70, 75]. GH and PGH bind and activate the maternal GH receptor (GHR), whereas HPL activates the maternal prolactin receptor (PRLR) [128]. Both receptor pathways via activation of signal transducer and activator of transcription 5 (STAT5) induce the expression of suppressor of cytokine signaling proteins (SOCS) [128]. GH induces the expression of SOCS2 and SOCS3, whereas HPL induces SOCS1 and SOCS2 [120]. SOCS1, SOCS3, SOCS6 and SOCS7 are negative regulators of insulin signaling [129–131]. SOCS proteins inhibit insulin receptor signaling by binding to the insulin receptor (IR), thereby blocking access of signaling intermediates and inhibiting insulin receptor tyrosine kinase activity, leading to a reduction of IR-directed phosphorylation of insulin receptor substrate-1 (IRS-1) and its downstream events, and by targeting IRS-1 and IRS-2 for proteasomal degradation [129, 130]. PGH and HPL signaling via SOCS expression thus induce SOCS-mediated insulin resistance [129-132]. In pregnancy, PGH- and HPL-mediated maternal insulin resistance is necessary for adequate glucose supply to the fetus. However, uncontrolled SOCS-mediated insulin resistance will result in gestational diabetes mellitus (GDM).

Moreover, the whey component of milk increases GIP signaling [64–66]. GIP receptors are not only expressed by pancreatic β -cells and GIP may not only signal via the entero-insular axis stimulating insulin secretion but also enhances GH secretion of the somatotroph cells of the pituitary, which express the GIP-receptor (GIPR) [133]. GIPR activation elevates cAMP, which drives GH-promoter activity [133]. As GHR signaling controls the expression SOCS2 and SOCS3, consumption of whey-containing milk products may overstimulate SOCS-induced maternal insulin resistance and may contribute to increased glucose flux to the fetus overstimulating fetal growth. Thus, extensive milk consump-

tion during pregnancy may increase the risk for GDM.

In milk consuming pregnant women, placental growth hormone signaling overlaps with milk signaling, which apparently via mTORC1activation and S6K1-mediated phosphorylation of IRS-1 and SOCS-mediated degradation of IRS-1 reduce insulin signaling of the pregnant mother. BCAAs by increasing mTORC1-S6K1 signaling act as positive signals for maintenance of protein stores, while inhibiting other actions of insulin at multiple levels [134]. In amino acidinfused humans, over-activation of mTORC1-S6K1 pathway increased inhibitory IRS-1 phosphorylation at Ser312, Ser636/639 and Ser1101 resulting in insulin resistance of skeletal muscle [135–137]. Thus, there is substantial evidence that inappropriate activation of mTORC1-S6K1 signaling by amino acids induces insulin resistance, the fundamental metabolic deviation leading to type 2 diabetes mellitus and GDM [135–138]. Whey proteins in contrast to meat proteins provide fast hydrolysable BCAAs mimicking a BCAA infusion that promotes insulin secretion and insulin resistance, major intrinsic mechanisms of milk signaling [2].

Both, milk-mediated mTORC1-induced insulin resistance and HPL-mediated insulin resistance may synergistically overstimulate the physiological insulin resistance of pregnancy. Furthermore, milk signaling and PGH signaling increase maternal hepatic IGF-1 synthesis [73-75, 126]. Chellakooty et al. [126] observed a positive association between the change in PGH and the change in IGF-1 levels throughout gestation. The change in IGF-1 throughout gestation was significantly positively associated with placental weight at birth [126]. Thus, the IGF-1increasing effect of milk consumption may result in an additional increase of placental weight, which in fact has been confirmed epidemiologically [102].

In mammals the growth and insulin resistance promoting system milk is intended to induce and maintain postnatal growth and is never combined with the insulin resistance promoting placental growth hormone system of pregnancy except for milk-consuming pregnant women.

Milk Consumption and FGF21-Mediated Over-Expression of GLUT1

Cow's milk consumption during pregnancy increases placental weight [102]. Notably, placental weight is associated with increased maternal serum levels of HPL [112-114]. Lactogens, especially HPL, have been demonstrated to activate the Janus-activated-kinase-2 (JAK2)/signal transducer and activator of transcription-5 (STAT5) signaling pathway [139–141]. Increased circulating serum levels of the recently discovered fibroblast growth factor 21 (FGF21) have been related with insulin resistance, type 2 diabetes, obesity and metabolic syndrome [142, 143]. In comparison to control subjects, plasma FGF21 levels were significantly higher in women with GDM [144]. Dekker Nitert et al. [145] demonstrated that FGF21 is expressed in the placenta. FGF21 mRNA expression is increased in women with GDM [145]. Notably, the FGF21 promoter contains three putative STAT5-binding sites [146]. It has been shown that GH increased STAT5 binding to the FGF21 promoter and enhanced FGF21 expression in the liver [146]. Increased FGF21 production has been observed in late pregnancy in the mouse [147]. In mice, GH-induced hepatic FGF21 production is stimulated by free fatty acid (FFA)-mediated activation of PPAR α [148]. In neonatal mice, milk intake via FFA-mediated stimulation of PPARa enhanced hepatic FGF21 expression [149]. Furthermore, recent evidence indicates that hepatic mTORC1 signaling regulates the expression of FGF21. Ectopic activation of hepatic mTORC1 in liver-specific Tsc1 knockout mice resulted in peroxisome proliferator-activated receptor- γ coactivator-1 α (PGC-1 α)-dependent expression of FGF21 [150]. Intriguingly, overexpression of FGF21 in 3T3-L1 adipocytes upregulated glucose uptake rates and increased the mRNA expression of glucose transporter 1 (GLUT1) [151].

GLUT1 is believed to be the primary glucose transporter isoform in the human placenta and increases its expression over gestation [152]. GLUT1 has been localized to both the maternal facing microvillous plasma membrane (MVM) with threefold higher expression as compared to the basal plasma membrane (BM) [153]. In maternal diabetes, the expression of GLUT1 in the BM has been reported to increase [154, 155]. Notably, increased syncytiotrophoblast BM expression of GLUT1 has been associated with increased birth weight of large babies of non-diabetic mothers [156].

Thus, milk-activated mTORC1 signaling via mTORC1-mediated overexpression of FGF21 and enhanced HPL/STAT5-mediated expression of placental FGF21 may overstimulate the expression of basal membrane GLUT1, which increases glucose flux to the fetal capillaries, thus promotes fetal weight gain. Taken together, there is good reason to assume that milk consumption during pregnancy enhances PGH/HPL-STAT5 signaling, milk-derived FFA-PPARα-signaling, and milk-driven mTORC1 signaling that may overstimulate placental FGF21-mediated GLUT1-expression enhancing glucose transport to the fetus.

Potential Role of Milk miR-21 on Fetal Growth

Exosomal microRNA-21 (miR-21) is a consistent component of cow's milk [157]. miR-21 appears to play a key role in mTORC1 signaling. Critical targets of miR-21 are mRNAs of important tumor suppressor proteins involved in upstream and downstream suppression of mTORC1 signaling, i.e., PTEN [158–161], Sprouty1 and Sprouty2 [162–164], PDCD4 [165–167]. Furthermore, miR-21 has been shown to induce the cell cycle promoter cyclin D1 in an mTORC1-dependent manner [168]. Supposed that milk-derived miR-21 reaches distant cells of the milk recipient, PTEN suppression could insulin/IGF-1/PI3K/AKT increase signaling, which further augments mTORC1 activation. MiR-21-induced inhibition of Sprouty1 and 2 would amplify RAS-RAF-MEK-ERK signaling, which additionally suppresses TSC2 and thus raises mTORC1 activity. Furthermore, miR-21 could stimulate the initiation of translation by

repression of PDCD4, which is a suppressor of translation initiation that inhibits the RNA helicase eIF4A [169]. Both, 4E-BP-1 and PDCD4 are crucial regulatory inhibitors of translation initiation and thus of protein synthesis. Activation of the mTORC1 pathway and its substrate kinase S6K1 results in subsequent phosphorylation of 4E-BP-1 and PDCD4 that promote eIF4E-eIF4G complex assembly and stimulate mRNA translation [168]. MiR-21-mediated suppression of PDCD4 expression may further amply translation initiation, a reasonable regulatory step of milk signaling to promote postnatal growth. Intriguingly, Jiang et al. [170] recently reported on aberrant upregulation of miR-21 in placental tissue of macrosomia. Among others the target genes of miR-21 were involved in JAK/STATand the mTORC1 signaling pathways [170]. Future research should clarify whether milkderived miR-21 may contribute to the development of mTORC1-driven fetal macrosomia.

The Role of mTORC1 in Placental Nutrient Transfer

The nutrient-sensitive kinase mTORC1 regulates cell metabolism and growth in response to altered nutrient levels and growth factor signaling [4–11]. During gestation the placenta integrates and regulates nutrient transfer to the fetus required for fetal growth [171, 172]. The syncytiotrophoblast represents the transporting epithelium and the primary endocrine cell of the human placenta, which plays the most critical role in determining fetal growth (Fig. 18.1). Trophoblast cell metabolism and associated signaling influence fetal nutrient availability. Trophoblast nutrient sensors may have a unique role in regulating fetal growth. Recent evidence underlines a pivotal role of mTORC1 in placental nutrient sensing [173]. mTOR is highly expressed in the syncytiotrophoblast of the human placenta [174]. It has been shown in cultured primary human trophoblast cells that mTORC1 is regulated by glucose, amino acid- and growth factor signaling [175]. Placental insulin/IGF-1 signaling and fetal levels of oxygen, glucose and amino acids are altered in pregnancy complications such

as intrauterine growth restriction, and all these factors are well-established upstream regulators of mTORC1 [4–11]. Furthermore, mTORC1 is a positive regulator of placental system A and system L amino acid transporters, suggesting that trophoblast mTORC1 modulates amino acid transfer across the placenta [173]. Syncytiotrophoblast mTORC1 activation increases cell surface density of amino acid transporters in the trophoblast and thus links maternal nutrient availability and growth factor signaling to fetal growth by modulating mTORC1-mediated flux of amino acids across the placenta finally stimulating amino acid-mediated mTORC1 activation of fetal cells for growth [173]. In this regard, cow's milk consumption during pregnancy may enhance the magnitude of trophoblast mTORC1 signaling. Milk consumption increases maternal blood levels of insulin and IGF-1 and increases the amount of essential BCAAs, glutamine and palmitate, all important nutrient and growth factor signals activating mTORC1 [44, 53–57, 60, 66, 73–75].

Placental mTORC1 signaling is compromised in pregnancy complications associated with altered fetal growth. It has ben shown that mTORC1 in the human placenta is downregulated in restricted fetal growth [174]. Kavitha et al. [13] recently demonstrated in pregnant baboons that maternal nutrient restriction downregulated placental mTOR, insulin/IGF-1 signaling and nutrient transporters.

In contrast, activation of placental mTORC1 signaling has been observed in association with maternal obesity [176]. In female Albino Wistar rats maternal overweight induced by a diet with high content of saturated fat activates placental mTOR and eIF2 α signaling and increases fetal growth [177]. Obesity is associated with elevated circulating levels free palmitate [178] and hyperleucinemia, hyperinsulinemia and insulin resistance [71, 84–86]. Thus, obesity exhibits a metabolic environment with enhanced insulin, BCAA, glucose and palmitate signals that overstimulate trophoblast mTORC1 activity. In obese women giving birth to LGA newborns the activity of placental insulin/IGF-1 and mTORC1 signaling was positively correlated with birth weight [176].

Milk consumption during pregnancy thus exaggerates obesity-mediated mTORC1-activating conditions of the trophoblast by increasing insulin/ IGF-1 signaling and providing abundant amounts of BCAAs and glutamine, all converging in the activation of placental mTORC1. This mechanistic view clearly supports the adverse impact of milk consumption during pregnancy in the promotion of placental and fetal overgrowth as confirmed by epidemiological data [101, 102].

Milk Consumption Increases Maternal Weight

Rapid weight gain in early prenatal life, increased birth weight and accelerated weight gain during early postnatal life exert adverse effects of nutritional and hormone-dependent programming [179–182] and are associated with increased risk of obesity, diabetes [183–185], asthma [186], hypertension, cardiovascular diseases [181, 187], and cancer [188, 189]. Epidemiological studies in human and animal experiments showed that nutrition during fetal and neonatal life promote diseases of civilization in adulthood. The overall effects of the maternal environment on the placenta are the product of its exposures throughout gestation, the "placental exposome" [190]. Milkderived BCAAs, insulin, IGF-1 and mTORC1 signaling most likely result in significant adverse alterations of the placental exposome.

Maternal overweight and obesity during pregnancy are associated with higher birth weights and more body fat of the offspring [191]. Arnberg et al. [95] investigated 203 overweight adolescents with a BMI of 25.4 ± 2.3 kg/m² (mean \pm SD), who received an additional daily amount of 35 g milk protein either as 1 L/d of skim milk, whey, or casein, or water as a control for 12 weeks. BMI-for-age Z-score was greater at 12 weeks in the skim milk, whey, and casein groups compared with baseline and with the water and pretest control groups. Thus, milk consumption increased body weight of adolescents. Intriguingly, pregnant women gaining excessive weight gain in comparison to women gaining optimal weight gain reported a twofold intake of dairy products of about 200 g/day [192]. Of all dairy products, dairy protein and dairy fat, the strongest predictor of increased maternal weight gain during the last trimester of pregnancy was milk [192]. Thus, milk consumption during pregnancy contributes significantly to increased maternal weight. Babies of obese women are often large at birth [193–197]. Jansson et al. [176] recently demonstrated that the activity of placental insulin/IGF-1 and mTORC1 signaling was positively correlated to birth weight. Milk signaling during pregnancy by high cow's milk intake appears to exaggerate imbalanced metabolic and hormonal signaling of obesity and diabetes during pregnancy.

Conclusions and Outlook

Milk, is not just food but represents a specific postnatal programming system for postnatal growth, a powerful nutritional cocktail of mammalian evolution that drives the adequate growth axis of the newborn infant [2]. Cow's milk supports the growth of Bos taurus, a species that doubles birth weight four times faster than human infants [198]. The persistent intake of bovine milk, a secretory product promoting speciesspecific postnatal growth, by humans during pregnancy is an absolute biological exception not observed in any other mammalian species. No adult pregnant mammal will be exposed to the signaling system of another species' milk. Cow's milk consumption during pregnancy promotes weight gain in the pre-pregnancy period [95], during pregnancy, especially in the third trimenon [192], increases placental weight [102, 106], increases fetal growth especially in the third trimenon [103, 192] and consecutively increases birth weight of the offspring [105, 106, 108–110] (Table 18.1). There is accumulating evidence that milk consumption during childhood and adolescence increases BMI and body weight [92–96], thus may result in unfavorable pre-pregnancy conditions that adversely affect weight gain during pregnancy [195]. Milk consumption during pregnancy accelerates both placental and fetal growth [102, 190]. There is good reason to assume that cow's milk-driven mTORC1 signaling during pregnancy exaggerates placental mTORC1 signaling. The milk-induced increase of placental weight may be associated with increased serum levels of HPL, which via induction of SOCS1/SOCS2 signaling promote maternal insulin resistance. Milk-driven maternal mTORC1 signaling via activation of S6K1 may further impair maternal insulin-IRS-1 signaling thereby enhancing the materno-fetal glucose gradient, which leads to overstimulating fetal growth. Moreover, milk-mediated overstimulation of mTORC1 signaling may directly upregulate syncytiotrophoblast amino acid transporter expression [173–176], thus increasing the transfer of amino acids to the fetus (Fig. 18.1). High glucose and amino acids (BCAAs) levels finally promote mTORC1 signaling of fetal cells resulting in fetal overgrowth and macrosomia.

Milk, the mTORC1 promoting "starter kit" of postnatal life represents an underestimated risk factor with significant impact on the placental exposome. It is thus of critical concern that gynecologists frequently recommend milk and dairy protein consumption during pregnancy as a rich source of calcium "to improve" fetal bone mineralization. Milk unfortunately not only delivers abundant calcium but most critically transfers a signaling system that increases the magnitude of mTORC1 signaling of the milk recipient and her fetus. Overstimulated mTORC1-mediated metabolic programming during the perinatal period has negative effects on lifelong metabolic and immunological programming [179-184, 197-200], which are intimately involved in the pathogenesis of mTORC1-driven age-related diseases of civilization [37, 201, 202]. Modern concepts of mTOR biology lead to the conclusion that only maternal, appropriate placental and fetal mTORC1 signaling guarantees healthy metabolic programming. Accelerated mTORC1 signaling as a consecutive reaction pattern of intrauterine growth restriction as well as over-stimulated perinatal mTORC1 signaling by overnutrition, especially milk consumption may adversely affect health and life span.

Milk is nature's design for postnatal mammalian growth but not for fetal growth. In terms of

anthropological biology, the introduction of milk consumption since the Neolithic revolution [203], promoted by refrigerator mass production since the 1920s in industrialized societies, is a fundamental change of human behavior bearing the risk of longterm adverse effects on human health [1]. Cow's milk consumption during pregnancy and cow's milk protein-based artificial infant formula feeding deviate perinatal axes of mTORC1 signaling and accelerate perinatal growth. In 1934, McCay and Crowell [204] from Cornell University provided translational evidence that slow growth favors longevity. Milk consumption during pregnancy disturbs a most critical period of perinatal growth, which may result in irreversible changes in the developmental trajectory [205]. Contemporary urban women of the "latte macchiato/cappuccino generation" are already exposed to increased intake of milk. These pregnant women are further encouraged by their obstetricians and gynecologists to increase their milk and dairy consumption with the intention to provide sufficient amounts of calcium [206]. However, it is of most critical concern that these recommendations stress milk signaling during pregnancy, a most sensitive period of lifelong mTORC1-driven metabolic programming. Milk, a signaling system for postnatal growth, should not interfere with prenatal growth and should not be recommended for pregnant women. Milk consumption during pregnancy appears to be a yet underestimated health hazard for the human fetus.

References

- Wiley AS. Cow milk consumption, insulin-like growth factor-I, and human biology: a life history approach. Am J Hum Biol. 2012;24(2):130–8.
- Melnik BC, John SM, Schmitz G. Milk is not just food but most likely a genetic transfection system activating mTORC1 for postnantal growth. Nutr J. 2013;12:103.
- Foster KG, Fingar DC. Mammalian target of rapamycin (mTOR): conducting the cellular signaling symphony. J Biol Chem. 2010;285(19):14071–7.
- Inoki K, Ouyang H, Li Y, Guan KL. Signaling by target of rapamycin proteins in cell growth control. Microbiol Mol Biol Rev. 2005;69(1):79–100.
- 5. Avruch J, Long X, Ortiz-Vega S, Rapley J, Papageorgiou A, Dai N. Amino acid regulation of

TOR complex 1. Am J Physiol Endocrinol Metab. 2009;296(4):E592–602.

- Sengupta S, Peterson T, Sabatini DM. Regulation of the mTOR complex 1 pathway by nutrients, growth factors, and stress. Mol Cell. 2010;40(2):310–22.
- Laplante M, Sabatini DM. mTOR signaling. Cold Spring Harb Perspect Biol. 2012;4(2):1–4.
- Kim J, Guan KL. Amino acid signaling in TOR activation. Ann Rev Biochem. 2011;80:1001–32.
- Kim S, Buel GR, Blenis J. Nutrient regulation of the mTOR complex 1 signaling pathway. Mol Cells. 2013;35(6):463–73.
- Jewell JL, Guan KL. Nutrient signaling to mTOR and cell growth. Trends Biochem Sci. 2013;38(5): 233–42.
- Efeyan A, Sabatini DM. Nutrients and growth factors in mTORC1 activation. Biochem Soc Trans. 2013;41(4):902–5.
- Laplante M, Sabatini DM. Regulation of mTORC1 and its impact on gene expression at a glance. J Cell Sci. 2013;126(Pt 8):1713–9.
- Kavitha JV, Rosario FJ, Nijland MJ, McDonald TJ, Wu G, Kanai Y, Powell TL, Nathanielsz PW, Jansson T. Down-regulation of placental mTOR, insulin/ IGF-I signaling, and nutrient transporters in response to maternal nutrient restriction in the baboon. FASEB J. 2014;28(3):1294–305.
- Wang X, Proud CG. Nutrient control of mTORC1, a cell-cycle regulator. Trends Cell Biol. 2009;19(6): 260–7.
- Gwinn DM, Shackelford DB, Egan D, Mihaylova MM, Mery A, Vasquez DS, Turk BE, Shaw RJ. AMPK phosphorylation of raptor mediates a metabolic checkpoint. Mol Cell. 2008;30(2):214–26.
- Inoki K, Li Y, Zhu T, Wu J, Guan KL. TSC2 is phosphorylated and inhibited by Akt and suppresses mTOR signalling. Nat Cell Biol. 2002;4(9): 648–57.
- Inoki K, Zhu T, Guan KL. TSC2 mediates cellular energy response to control cell growth and survival. Cell. 2003;115(5):577–90.
- Manning BD, Tee AR, Logsdon MN, Blenis J, Cantley LC. Identification of the tuberous sclerosis complex-2 tumor suppressor gene product tuberin as a target of the phosphoinositide-3-kinase/akt pathway. Mol Cell. 2002;10(1):151–62.
- Tee AR, Fingar DC, Manning BD, Kwiatkowski DJ, Cantley LC, Blenis J. Tuberous sclerosis complex-1 and -2 gene products function together to inhibit mammalian target of rapamycin (mTOR)-mediated downstream signalling. Proc Natl Acad Sci U S A. 2002;99(21):13571–6.
- Dibble CC, Elis W, Menon S, Qin W, Klekota J, Asara JM, Finan PM, Kwiatkowski DJ, Murphy LO, Manning BD. TBC1D7 is a third subunit of the TSC1-TSC2 complex upstream of mTORC1. Mol Cell. 2012;47(4):535–46.
- Dodd KM, Tee AR. Leucine and mTORC1: a complex relationship. Am J Physiol Endocrinol Metabol. 2012;302(11):E1329–42.

- Thedieck K, Hall MN. Translational control by amino acids and energy. Handbook of cell signaling. Three-volume Set. 2nd ed. Waltham Massachchusetts USA: Elsevier, Academic Press; chap 274, 2010. p. 2285–93.
- Sancak Y, Peterson TR, Shaul YD, Lindquist RA, Thoreen CC, Bar-Peled L, Sabatini DM. The Rag GTPases bind raptor and mediate amino acid signaling to mTORC1. Science. 2008;320(5882):1496–501.
- 24. Sancak Y, Bar-Peled L, Zoncu R, Markhard AL, Nada S, Sabatini DM. Ragulator-Rag complex targets mTORC1 to the lysosomal surface and is necessary for its activation by amino acids. Cell. 2010; 141(2):290–303.
- Goberdhan DC. Intracellular amino acid sensing and mTORC1-regulated growth: new ways to block an old target? Curr Opin Investig Drugs. 2010;11(12): 1360–7.
- Menon S, Dibble CC, Talbott G, Hoxhaj G, Valvezan AJ, Takahashi H, Cantley LC, Manning BD. Spatial control of the TSC complex integrates insulin and nutrient regulation of mTORC1 at the lysosome. Cell. 2014;156(4):771–85.
- 27. Manning BD, Cantley LC. United at last: the tuberous sclerosis complex gene products connect the phosphoinositide 3-kinase/Akt pathway to mammalian target of rapamycin (mTOR) signalling. Biochem Soc Trans. 2003;31(Pt 3):573–8.
- Nicklin P, Bergman P, Zhang B, Triantafellow E, Wang H, Nyfeler B, Yang H, Hild M, Kung C, Wilson C, Myer VE, MacKeigan JP, Porter JA, Wang YK, Cantley LC, Finan PM, Murphy LO. Bidirectional transport of amino acids regulates mTOR and autophagy. Cell. 2009;136(3):521–34.
- Cohen A, Hall MN. An amino acid shuffle activates mTORC1. Cell. 2009;136(3):399–400.
- Hara K, Yonezawa K, Weng QP, Kozlowski MT, Belham C, Avruch C. Amino acid sufficiency and mTOR regulate p70 S6 kinase and eIF-4EBP1 through a common effector mechanism. J Biol Chem. 1998;273(23):14484–94.
- Long X, Ortiz-Vega S, Lin Y, Avruch J. Rheb binding to mammalian target of rapamycin (mTOR) is regulated by amino acid sufficiency. J Biol Chem. 2005;280(25):23433–6.
- Bar-Peled L, Schweitzer LD, Zoncu R, Sabatini DM. Ragulator is a GEF for the Rag GTPases that signal amino acid levels to mTORC1. Cell. 2012;150(6): 1196–208.
- Groenewoud MJ, Zwartkruis FJ. Rheb and Rags come together at the lysosome to activate mTORC1. Biochem Soc Trans. 2013;41(4):951–5.
- 34. Nobukuni T, Joaquin M, Roccio M, Dann SG, Kim SY, Gulati P, Byfield MP, Backer JM, Natt F, Bos JL, Zwartkruis FJ, Thomas G. Amino acids mediate mTOR/raptor signaling through activation of class 3 phosphatidylinositol 3OH-kinase. Proc Natl Acad Sci U S A. 2005;102(40):14238–43.
- Ögmundsdóttir MH, Heublein S, Kazi S, Reynolds B, Visvalingam SM, Shaw MK, Goberdhan DC. Proton-

assisted amino acid transporter PAT1 complexes with Rag GTPases and activates TORC1 on late endosomal and lysosomal membranes. PLoS ONE. 2012;7(5), e36616.

- Thwaites DT, Anderson CM. The SLC36 family of proton-coupled amino acid transporters and their potential role in drug transport. Br J Pharmacol. 2011;164(7):1802–16.
- Zoncu R, Efeyan A, Sabatini DM. mTOR: from growth signal integration to cancer, diabetes and ageing. Nat Rev Mol Cell Biol. 2011;12(1):21–35.
- Zoncu R, Bar-Peled L, Efeyan A, Wang S, Sancak Y, Sabatini DM. mTORC1 senses lysosomal amino acids through an inside-out mechanism that requires the vacuolar H+-ATPase. Science. 2011;334(6056):678–83.
- Reynolds B, Laynes R, Ögmundsdóttir MH, Boyd CA, Goberdhan DC. Amino acid transporters and nutrient-sensing mechanisms: new targets for treating insulin-linked disorders? Biochem Soc Trans. 2007;35(Pt 5):1215–7.
- Bonfils G, Jaquenoud M, Bontron S, Ostrowicz C, Ungermann C, De Virgili C. Leucyl-tRNA synthase controls TORC1 via the EGO complex. Mol Cell. 2012;46(1):105–10.
- Han JM, Jeong SJ, Park MC, Kim G, Kwon NH, Kim HK, Ha SH, Ryu SH, Kim S. Leucyl-tRNA synthase is an intracellular leucine sensor for the mTORC1signaling pathway. Cell. 2012;149(2):410–24.
- Duran RV, Oppliger W, Robitaille AM, Heiserich L, Skendaj R, Gottlieb E, Hall MN. Glutaminolysis activates Rag-mTORC1 signaling. Mol Cell. 2012; 47(3):349–58.
- 43. Gallinetti J, Harputlugil E, Mitchell JR. Amino acid sensing in dietary-restriction-mediated longevity: roles of signal-transducing kinases GCN2 and TOR. Biochem J. 2013;449(1):1–10.
- 44. Yasuda M, Tanaka Y, Kume S, Morita Y, Chin-Kanasaki M, Araki H, Isshiki K, Araki S, Koya D, Haneda M, Kashiwagi A, Maegawa H, Uzu T. Fatty acids are novel nutrient factors to regulate mTORC1 lysosomal localization and apoptosis in podocytes. Biochim Biophys Acta. 2014;1842(7):1097–108.
- Shaw RJ. LKB1 and AMPK control of mTOR signalling and growth. Acta Physiol (Oxf). 2009;196(1): 65–80.
- Xu J, Ji J, Yan XH. Cross-talk between AMPK and mTOR in regulating energy balance. Crit Rev Food Sci Nutr. 2012;52(5):373–81.
- 47. Shaw RJ, Kosmatka M, Bardeesy N, Hurley RL, Witters LA, DePinho RA, Cantley LC. The tumor suppressor LKB1 kinase directly activates AMP-activated kinase and regulates apoptosis in response to energy stress. Proc Natl Acad Sci U S A. 2004;101(10):3329–35.
- Hardie DG. AMP-activated/SFN1 protein kinases: conserved guardians of cellular energy. Nat Rev Mol Cell Biol. 2007;8(10):774–85.
- Egan D, Kim J, Shaw RJ, Guan KL. The autophagy initiating kinase ULK1 is regulated via opposing phosphorylation by AMPK and mTOR. Autophagy. 2011;7(6):643–4.

- 50. Shang L, Chen S, Du F, Li S, Zhao L, Wang X. Nutrient starvation elicits an acute autophagic response mediated by ULK1 dephosphorylation and its subsequent dissociation from AMPK. Proc Natl Acad Sci U S A. 2011;108(12):4788–93.
- Roberts DJ, Tan-Sah VP, Ding EY, Smith JM, Miyamoto S. Hexokinase-II positively regulates glucose starvation-induced autophagy through TORC1 inhibition. Mol Cell. 2014;53(4):521–33.
- Kundu M. Too sweet for autophagy: hexokinase inhibition of mTORC1 activates autophagy. Mol Cell. 2014;53(4):517–8.
- Melnik BC, Schmitz G. Metformin: an inhibitor of mTORC1 signaling. J Endocrinol Diabetes Obes. 2014;2(2):1029.
- Holt S, Brand Miller J, Petocz P. An insulin index of foods: the insulin demand generated by 1000-kK portions of common foods. Am J Clin Nutr. 1997;66(5):1264–76.
- Hoyt G, Hickey MS, Cordain L. Dissociation of the glycaemic and insulinaemic responses to whole and skimmed milk. Br J Nutr. 2005;93(2):175–7.
- Boirie Y, Dangin M, Gachon P, Vasson MP, Maubois JL, Beaufrère B. Slow and fast dietary proteins differently modulate postprandial protein accretion. Proc Natl Acad Sci U S A. 1997;94(26):14930–5.
- He T, Giuseppin ML. Slow and fast dietary proteins differentially modulate postprandial metabolism. Int J Food Sci Nutr. 2014;65(3):386–90.
- 58. Boutrou R, Gaudichon C, Dupont D, Jarden J, Airinei G, Marsset-Baglieri A, Benamouzig R, Tomé C, Leonil J. Sequential releases of milk protein-derived bioactive peptides in the jejunum in healthy humans. Am J Clin Nutr. 2013;97(6):1414–23.
- 59. Mahé S, Roos N, Benamouzig R, Davin L, Luengo C, Gagnon L, Gaussergès N, Rautureau J, Tomé D. Gastrojejunal kinetics and the digestion of [15N]betalactoglobulin and casein in humans: the influence of the nature and quantity of the protein. Am J Clin Nutr. 1996;63(4):546–52.
- Hoppe C, Mølgaard C, Dalum C, Vaag A, Michaelsen KF. Differential effects of casein versus whey on fasting plasma levels of insulin, IGF-1 and IGF-1/ IGFBP-3: results from a randomized 7-day supplementation study in prepubertal boys. Eur J Clin Nutr. 2009;63(9):1076–83.
- Thomas FB, Sinar D, Mazzaferri EL, Cataland S, Mekhjian HS, Caldwell JH, Fromkes JJ. Selective release of gastric inhibitory polypeptide by intraduodenal amino acid perfusion in man. Gastroenterology. 1978;74(6):1261–5.
- 62. Chen Q, Reimer RA. Dairy protein and leucine alter GLP-1 release and mRNA of genes involved in intestinal lipid metabolism in vitro. Nutrition. 2009;25(3):340–9.
- 63. Millward DJ, Layman DK, Tomé D, Schaafsma G. Protein quality assessment: impact of expanding understanding of protein and amino acid needs for optimal health. Am J Clin Nutr. 2008;87(5): 1576S–81.

- 64. Nilsson M, Stenberg M, Frid AH, Holst JJ, Björck IM. Glycemia and insulinemia in healthy subjects after lactose-equivalent meals of milk and other food proteins: the role of plasma amino acids and incretins. Am J Clin Nutr. 2004;80(5):1246–53.
- 65. Nilsson M, Holst JJ, Björck IM. Metabolic effects of amino acid mixtures and whey protein in healthy subjects: studies using glucose-equivalent drinks. Am J Clin Nutr. 2007;85(4):996–1004.
- 66. Salehi A, Gunnerud U, Muhammed SJ, Ostman E, Holst JJ, Björck I, Rorsman P. The insulinogenic effects of whey protein is partially mediated by a direct effect of amino acids and GIP on β-cells. Nutr Metab (Lond). 2012;9(1):48.
- McDaniel ML, Marshall CA, Pappan KL, Kwon G. Metabolic and autocrine regulation of the mammalian target of rapamycin by pancreatic β-cells. Diabetes. 2002;51(10):2877–85.
- Yang J, Chi Y, Burkhardt BR, Guan Y, Wolf BA. Leucine metabolism in regulation of insulin secretion from pancreatic beta cells. Nutr Rev. 2010;68(5): 270–9.
- 69. Le Bacquer O, Queniat G, Gmyr V, Kerr-Conte J, Lefebvre B, Pattou F. mTORC1 and mTORC2 regulate insulin secretion through Akt in INS-1 cells. J Endocrinol. 2013;216(1):21–9.
- Hoppe C, Mølgaard C, Vaag A, Barkholt V, Michaelsen KF. High intakes of milk, but not meat, increases s-insulin and insulin resistance in 8-yearold boys. Eur J Clin Nutr. 2005;59(3):393–8.
- 71. She P, Van Horn C, Reid T, Hutson SM, Cooney RN, Lynch CJ. Obesity-related elevations in plasma leucine are associated with alterations in enzymes involved in branched-chain amino acid metabolism. Am J Physiol Endocrinol Metab. 2007;293(6):E1552–63.
- Loridan L, Sadeghi-Nejad A, Senior B. Hypersecretion of insulin after the administration of L-leucine to obese children. J Pediatr. 1971;78(1): 53–8.
- Qin LQ, He K, Xu JY. Milk consumption and circulating insulin-like growth factor-I level: a systematic literature review. Int J Food Sci Nutr. 2009;60 Suppl 7:330–40.
- 74. Norat T, Dossus L, Rinaldi S, Overvad K, Grønbaek H, Tjønneland A, Olsen A, Clavel-Chapelon F, Boutron-Ruault MC, Boeing H, Lahmann PH, Linseisen J, Nagel G, Trichopoulou A, Trichopoulos D, Kalapothaki V, Sieri S, Palli D, Panico S, Tumino R, Sacerdote C, Bueno-de-Mesquita HB, Peeters PH, van Gils CH, Agudo A, Amiano P, Ardanoz E, Martinez C, Quirós R, Tormo MJ, Bingham S, Key TJ, Allen NE, Ferrari P, Slimani N, Riboli E, Kaaks R. Diet, serum insulin-like growth factor-I and IGF-binding protein-3 in European women. Eur J Clin Nutr. 2007;61(1):91–8.
- Rich-Edwards JW, Ganmaa D, Pollak MN, Nakamoto EK, Kleinman K, Tserendolgor U, Willet WC, Frazier AL. Milk consumption and the prepubertal somatotropic axis. Nutr J. 2007;6:28.
- Larnkjær A, Arnberg K, Michaelsen KF, Jensen SM, Mølgaard C. Effect of milk proteins on linear growth

and IGF variables in overweight adolescents. Growth Horm IGF Res. 2014;24(2–3):54–9.

- Melnik BC. Leucine signaling in the pathogenesis of type 2 diabetes and obesity. World J Diabetes. 2012;3(3):38–53.
- Lenders CM, Liu S, Wilmore DW, Sampson L, Dougherty LW, Spiegelman D, Willett WC. Evaluation of a novel food composition database that includes glutamine and other amino acids derived from gene sequencing data. Eur J Clin Nutr. 2009;63(12):1433–9.
- Li M, Li C, Allen A, Stanley CA, Smith TJ. The structure and allosteric regulation of mammalian glutamate dehydrogenase. Arch Biochem Biophys. 2012;519(2):69–80.
- Lorin S, Tol MJ, Bauvy C, Strijland A, Poüs C, Verhoeven AJ, Codogno P, Meijer AJ. Glutamate dehydrogenase contributes to leucine sensing in the regulation of autophagy. Autophagy. 2013;9(6):850–60.
- Xu G, Kwon G, Cruz WS, Marshall CA, McDaniel ML. Metabolic regulation by leucine of translation initiation through the mTOR-signaling pathway by pancreatic beta-cells. Diabetes. 2001;50(2):353–60.
- Jensen RG, Ferris AM, Lammi-Keefe CJ. The composition of milk fat. J Dairy Sci. 1991;74(9):3228–43.
- Bitman J, Wood DL. Changes in milk fat phospholipids during lactation. J Dairy Sci. 1990;73(5):1208–16.
- 84. Newgard CB, An J, Bain J, Muehlbauer MJ, Stevens RD, Lien LF, Haqq AM, Shah SH, Arlotto M, Slentz CA, Rochon J, Gallup D, Ilkayeva O, Wenner BR, Yancy Jr WS, Eisenson H, Musante G, Surwit RS, Millington DS, Butler MD, Svetkey LP. A branchedchain amino acid-related metabolic signature that differentiates obese and lean humans and contributes to insulin resistance. Cell Metab. 2009;9(4):311–26.
- Morris C, O'Grada C, Ryan M, Roche HM, Gibney MJ, Gibney ER, Brennan L. The relationship between BMI and metabolomic profiles: a focus on amino acids. Proc Nutr Soc. 2012;71(4):634–8.
- 86. McCormack SE, Shaham O, McCarthy MA, Deik AA, Wang TJ, Gerszten RE, Clish CB, Mootha VK, Grinspoon SK, Fleischman A. Circulating branchedchain amino acid concentrations are associated with obesity and future insulin resistance in children and adolescents. Pediatr Obes. 2013;8(1):52–61.
- Zhou YP, Grill VE. Long-term exposure of rat pancreatic islets to fatty acids inhibits glucose-induced insulin secretion and biosynthesis through a glucose fatty acid cycle. J Clin Invest. 1994;93(2):870–6.
- 88. Dresner A, Laurent D, Marcucci M, Griffin ME, Dufour S, Cline GW, Slezak LA, Andersen DK, Hundal RS, Rothman DL, Petersen KF, Shulman GI. Effects of free fatty acids on glucose transport and IRS-1-associated phosphatidylinositol 3-kinase activity. J Clin Invest. 1999;103(2):253–9.
- Unger RH, Clark GO, Scherer PE, Orci L. Lipid homeostasis, lipotoxicity and the metabolic syndrome. Biochim Biophys Acta. 2010;1801(3):209–14.
- Melnik BC, John SM, Schmitz G. Over-stimulation of insulin/IGF-1 signaling by Western diet may

promote diseases of civilization: lessons learnt from Laron syndrome. Nutr Metab (Lond). 2011;8:41.

- Melnik BC, John SM, Schmitz G. Adipogenic and insulin resistance-promoting effects of milk consumption. Mol Nutr Food Res. 2014;58(6):1166–7.
- 92. Yamin HB, Barnea M, Genzer Y, Chapnik N, Froy O. Long-term commercial cow's milk consumption and its effects on metabolic parameters associated with obesity in young mice. Mol Nutr Food Res. 2014;58(5):1061–8.
- Wiley AS. Dairy and milk consumption and child growth: is BMI involved? An analysis of NHANES 1999–2004. Am J Hum Biol. 2010;22(4):517–25.
- Berkey CS, Rocket HRH, Willet WC, Colditz GA. Milk, dairy fat, dietary calcium, and weight gain. Arch Pediatr Adolesc Med. 2005;159(6):543–50.
- 95. Arnberg K, Mølgaard C, Michaelsen KF, Jensen SM, Trolle E, Larnkjær A. Skim milk, whey, and casein increase body weight and whey and casein increase plasma C-peptide concentration in overweight adolescents. J Nutr. 2012;142(12):2083–90.
- 96. Barr SI, McCarron DA, Heaney RP, Dawson-Hughes B, Berga SL, Stern JS, Oparil S. Effects of increased consumption of fluid milk on energy and nutrient intake, body weight, and cardiovascular risk factors in healthy older adults. Am J Diet Assoc. 2000;100(7):810–7.
- Skovbjerg H, Sjöström H, Norén O. Purification and characterization oft he amphiphilic lactase/phlorizin hydrolyase from human samm intestine. Eur J Biochem. 1981;114(3):653–61.
- Swallow DM. Genetics of lactase persistence and lactose intolerance. Ann Rec Genet. 2003;37:197–219.
- Ingram CJ, Mulcare CA, Itan Y, Thomas MG, Swallow DM. Lactose digestion and the evolutionary genetics of lactase persistence. Hum Genet. 2009;124(6):579–91.
- 100. Ingram CJ, Elamin MF, Mulcare CA, Weale ME, Tarekegn A, Raga TO, Bekele E, Elamin FM, Thomas MG, Bradman N, Swallow DM. A novel polymorphism associated with lactose tolerance in Africa: multiple causes for lactase persistence? Hum Genet. 2007;120(6):779–88.
- 101. Hoppe C, Mølgaard C, Michaelsen KF. Cow's milk and linear growth in industrialized and developing countries. Ann Rev Nutr. 2006;26:131–73.
- 102. Olsen SF, Halldorsson TI, Willett WC, Knudsen VK, Gillman MW, Mikkelsen TB, Olsen J, NUTRIX Consortium. Milk consumption during pregnancy is associated with increased infant size at birth: prospective cohort study. Am J Clin Nutr. 2007;86(4): 1104–10.
- 103. Heppe DH, van Dam RM, Willemsen SP, den Breeijen H, Raat H, Hofman A, Steegers EA, Jaddoe VW. Maternal milk consumption, fetal growth, and the risks of neonatal complications: the Generation R Study. Am J Clin Nutr. 2011;94(2):501–9.
- 104. Hadlock FP, Harrist RB, Sharman RS, Deter RL, Park SK. Estimation of fetal weight with the use of head, body, and femur measurements – a prospective study. Am J Obstet Gynecol. 1985;151(3):333–7.

- 105. Ludvigsson JF, Ludvigsson J. Milk consumption during pregnancy and infant birthweight. Acta Paediatr. 2004;93(11):1474–8.
- 106. Rao S, Yajnik CS, Kanade A, Fall CH, Margetts BM, Jackson AA, Shier R, Joshi S, Rege S, Lubree H, Desai B. Intake of micronutrient-rich foods in rural Indian mothers is associated with the size of their babies at birth: Pune Maternal Nutrition Study. J Nutr. 2001;131(4):1217–24.
- 107. Mannion CA, Gray-Donald K, Koski KG. Association of low intake of milk and vitamin D during pregnancy with decreased birth weight. CMAJ. 2006;174(9): 1273–7.
- Moore VM, Davies MJ, Willson KJ, Worsley A, Robinson JS. Dietary composition of pregnant women is related to size of the baby at birth. J Nutr. 2004;134:1820–6.
- 109. Chan GM, McElligott K, McNaught T, Gill G. Effects of dietary calcium intervention on adolescent mothers and newborns: a randomized controlled trial. Obstet Gynecol. 2006;108(3 Pt 1):565–71.
- 110. Brantsæter AL, Olafsdottir AS, Forsum E, Olsen SF, Thorsdottir I. Does milk and dairy consumption during pregnancy influence fetal growth and infant birthweight? A systematic literature review. Food Nutr Res. 2012;56:20050.
- Freemark M. Placental hormones and the control of fetal growth. J Clin Endocrinol Metab. 2010;95(5): 2054–7.
- 112. Cramer DW, Beck P, Makowski EL. Correlation of gestational age with maternal human chorionic somatomammotropin and maternal and fetal growth hormone plasma concentrations during labor. Am J Obstet Gynecol. 1971;109(4):649–55.
- Lindberg BS, Nilsson BA. Human placental lactogen (HPL) levels in abnormal pregnancies. J Obstet Gynaecol Br Commonw. 1973;80(12):1046–53.
- 114. Sciarra JJ, Sherwood LM, Varma AA, Lundberg WB. Human placental lactogen (HPL) and placental weight. Am J Obstet Gynecol. 1968;101(3):413–6.
- Boyce A, Schwartz D, Hubert C, Cedard L, Dreyfus J. Smoking, human placental lactogen and birth weight. Br J Obstet Gynaecol. 1975;82(12):964–7.
- 116. Letchworth AT, Boardman RJ, Bristow C, Landon J, Chard T. A rapid semi-automated method for the measurement of human chorionic sommatomammotrophin. The normal range in the third trimester and its relation to fetal weight. J Obstet Gynaecol Br Commonw. 1971;78(6):542–8.
- 117. Lindberg BS, Nilsson BA. Variations in maternal plasma levels of human placental lactogen (HPL) in normal pregnancy and labour. J Obstet Gynaecol Br Commonw. 1973;80(7):619–26.
- Henleigh PA, Cheatum SG, Spellacy WN. Oxytocinase and human placental lactogen for the prediction of intrauterine growth retardation. Am J Obstet Gynaecol. 1977;129(6):675–8.
- 119. Männik J, Vaas P, Rull K, Teesalu P, Laan M. Differential placental expression profile of human Growth Hormone/Chorionic Somatomammotropin

genes in pregnancies with pre-eclampsia and gestational diabetes mellitus. Mol Cell Endocrinol. 2012;355(1):180–7.

- 120. Pedersen NG, Juul A, Chrisitansen M, Wojdemann KR, Tabor A. Maternal serum placental growth hormone, but not human placental lactogen or insulin growth factor-1, is positively associated with fetal growth in the first half of pregnancy. Ultrasound Obstet Gynecol. 2010;36(5):534–41.
- 121. Barbour LA, Shao J, Qiao L, Puluwa LK, Jensen DR, Bartke A, Garrity M, Draznin B, Friedman JE. Human placental growth hormone causes severe insulin resistance in transgenic mice. Am J Obstet Gynecol. 2002;186(3):512–7.
- 122. Goodman HM, Tai LR, Ray J, Cooke NE, Liebhaber SA. Human growth hormone variant produces insulin-like and lipolytic responses in rat adipose tissue. Endocrinology. 1991;129(4):1779–83.
- 123. Barbour LA, Shao J, Qiao L, Leitner W, Anderson M, Freidman JE, Draznin B. Human placental growth hormone increases expression of the p85 regulatory unit of phosphtidylinositol 3-kinase and triggers severe insulin resistance in skeletal muscle. Endocrinology. 2004;145(3):1144–50.
- 124. Freemark M. Ontogenesis of prolactin receptors in the human fetus: roles on fetal development. Biochem Soc Trans. 2001;29(Pt 2):38–41.
- 125. Fleenor D, Oden J, Kelly PA, Mohan S, Alliouachene S, Pende M, Wentz S, Kerr J, Freemark M. Roles of the lactogens and somatogens in perinatal and post-natal metabolism and growth: studies of a novel mouse model combining lactogen resistance and growth hormone deficiency. Endocrinology. 2005; 146(1):103–12.
- 126. Chellakooty M, Vangsgaard K, Larsen T, Scheike T, Falck-Larsen J, Legarth J, Andersson AM, Main KM, Skakkebaek NE, Juul A. A longitudinal study of intrauterine growth and the placental growth hormone (GH)-insulin-like growth factor 1 axis in maternal corculation: association between placental GH and fetal growth. J Clin Endocrinol Metab. 2004;89(1):384–91.
- 127. Newbern D, Freemark M. Placental hormones and the control of maternal metabolism and fetal growth. Curr Opin Endocrinol Diabetes Obes. 2011;18(6): 409–16.
- Hennighausen L, Robibson GW. Interpretation of cytokine signalling through the transcription factors STAT5A and STAT5B. Genes Dev. 2008;22(6): 711–21.
- Howard JK, Flier JS. Attenuation of leptin and insulin signaling by SOCS proteins. Trends Endocrinol Metab. 2006;17(9):365–71.
- Lebrun P, Van Obberghen E. SOCS proteins causing trouble in insulin action. Acta Physiol (Oxf). 2008;192(1):29–36.
- 131. Lebrun P, Cognard E, Gontard P, Bellon-Paul R, Filloux C, Berthault MF, Magnan C, Ruberte J, Luppo M, Pujol A, Pachera N, Herchuelz A, Bosch F, Van Obberghen E. The suppressor of cytokine sig-

nalling 2 (SOCS2) is a key repressor of insulin secretion. Diabetologia. 2010;53(9):1935–46.

- 132. Yang Z, Hulver M, McMillan RP, Cai L, Kershaw EE, Yu L, Xue B, Shi H. Regulation of insulin and leptin signaling by muscle suppressor of cytokine signaling 3 (SOCS3). PLoS ONE. 2012;7(10), e47493.
- 133. Occhi G, Losa M, Albiger N, Trivellin G, Regazzo D, Scanarini M, Monteserin-Garcia JL, Fröhlich B, Ferasin S, Terreni MR, Fassina A, Vitiello L, Stalla G, Mantero F, Scaroni C. The glucose-dependent insulinotropic polypeptide receptor is overexpressed amongst GNAS1 mutation-negative somatotropinomas and drives growth hormone (GH)-promoter activity in GH3 cells. J Neuroendocrinol. 2011;23(7): 641–9.
- 134. Patti ME, Brambilla E, Luzi L, Landaker EJ, Kahn CR. Bidirectional modulation of insulin action by amino acids. J Clin Invest. 1998;101(7):1519–29.
- 135. Krebs M, Krssak M, Bernroider E, Anderwald C, Brehm A, Meyerspeer M, Nowotny P, Roth E, Waldhäussl W, Roden M. Mechanism of amino acidinduced skeletal muscle insulin resistance in humans. Diabetes. 2002;51(3):599–605.
- 136. Tremblay F, Krebs M, Dombrowski L, Brehm A, Bernroider E, Roth E, Nowotny P, Walshäusl W, Marette A, Roden M. Overactivation of S6 kinase 1 as a cause of human insulin resistance during increased amino acid availability. Diabetes. 2005; 54(9):2674–84.
- 137. Tremblay F, Brulé S, Um SH, Masuda K, Roden M, Sun XJ, Krebs M, Polakiewicz RD, Thomas G, Marette A. Identification of Ser-1101 as a target of S6K1 in nutrient- and obesity-induced insulin resistance. Proc Natl Acad Sci U S A. 2007;104(35):14056–61.
- Shah OJ, Wang Z, Hunter T. Inappropriate activation of the TSC/Rheb/mTOR/S6K cassette induces IRS1/2 depletion, insulin resistance, and cell survival deficiencies. Curr Biol. 2004;14(18):1650–6.
- 139. Cao J, Gowri PM, Ganguly TC, Wood M, Hyde JF, Talamates F, Vore M. PRL, placental lactogen, and GH induce NA(+)/taurocholate-cotransporting polypeptide gene expression by activating signal transducer and activator of transcription-5 in liver cells. Endocrinology. 2001;142(10):4212–22.
- 140. Kondegowda NG, Mozar A, Chin C, Otero A, Garcia-Ocana A, Vasavada RC. Lactogens protect rodent and human beta cells agianst glucolipotoxicity-induced cell death through Janus kinase-2 (JAK2)/signal transducer and activator of transcription-5 (STAT5) signalling. Diabetologica. 2012;55(6):1721–32.
- 141. Fujinaka Y, Takane K, Yamashita H, Vasavada RC. Lactogens promote beta cell survival through JAK2/STAT5 activation and Bcl-XL upregulation. J Biol Chem. 2007;282(42):30707–17.
- 142. Iglesias P, Selgas R, Romero S, Díez JJ. Biological role, clinical significance, and therapeutic possibilities of the recently discovered metabolic hormone fibroblastic growth factor 21. Eur J Endocrinol. 2012;167(3):301–9.

- 143. Zhang X, Yeung DC, Karpisek M, Stejskal D, Zhou ZG, Liu F, Wong RL, Chow WS, Tso AW, Lam KS, Xu A. Serum FGF21 levels are increased in obesity and are independently associated with the metabolic syndrome in humans. Diabetes. 2008;57(5):1246–53.
- 144. Tan BK, Sivakumar K, Bari MF, Vatish M, Randeva HS. Lower cerebrospinal fluid/plasma fibroblast growth factor 21 (FGF21) ratios and placental FGF21 production in gestational diabetes. PLoS ONE. 2013;8(6), e65254.
- 145. Dekker Nitert M, Barrett HL, Kubala MH, Scholz Romero K, Denny KJ, Woodruff TM, McIntyre HD, Callaway LK. Increased placental expression of fibroblast growth factor 21 in gestational diabetes mellitus. J Clin Endocrinol Metab. 2014;99(4):E591–8.
- 146. Yu J, Zhao L, Wang A, Eleswarapu S, Ge X, Chen D, Jiang H. Growth hormone stimulates transcription of the fibroblast growth factor 21 gene in the liver through the signal transducer and activator of transcription 5. Endocrinology. 2012;153(2):750–8.
- 147. Cui Y, Giesy SL, Hassan M, Davis K, Zhao S, Boisclair YR. Hepatic FGF21 production is increased in late pregnancy in the mouse. Am J Physiol Regul Integr Comp Physiol. 2014;307:R290–8.
- 148. Chen W, Hoo RL, Konihi M, Itoh N, Lee P, Ye H, Lam KS, Xu A. Growth hormone induced hepatic production of fibroblast growth factor 21 through a mechanism dependent on lipolysis in adipocytes. J Biol Chem. 2011;286(40):34559–66.
- 149. Hondares E, Rosell M, Gonzalez FJ, Giralt M, Iglesias R, Villaroya F. Hepatic FGF21 expression is induced at birth via PPARα in response to milk intake and contributes to thermogenic activation of neonatal brown fat. Cell Metab. 2010;11(3):206–12.
- 150. Cornu M, Oppliger W, Albert V, Robitaille AM, Trapani F, Quagliata L, Fuhrer T, Sauer U, Terracciano L, Hall MN. Hepatic mTORC1 controls locomotor activity, body temperature and lipid metabolism through FGF21. Proc Natl Acad Sci U S A. 2014;111(32):11592–9.
- 151. Li K, Li L, Yang M, Liu H, Boden G, Yang G. The effects of fibroblast growth factor-21 knockdown and over-expression on its signaling pathway and glucose-lipid metabolism in vitro. Mol Cell Endocrinol. 2012;348(1):21–6.
- 152. Ericsson A, Hamark B, Powell TL, Jansson T. Glucose transporter isoform 4 is expressed in the syncytiotrophoblast of first trimester human placenta. Hum Reprod. 2005;20(2):521–30.
- 153. Jansson T, Wennergren M, Illsley NP. Glucose transporter protein expression in human placenta throughout gestation and in intrauterine growth retardation. J Clin Endocrinol Metab. 1993;77(6):1554–62.
- 154. Jansson T, Wennergren M, Powell TL. Placental glucose transport and GLUT 1 expression in insulindependent diabetes. Am J Obstet Gynecol. 1999; 180(1 Pt 1):163–8.
- 155. Gaither K, Quraishi AN, Illsley NP. Diabetes alters the expression and activity of the human placental GLUT1 glucose transporter. J Clin Endocrinol Metab. 1999;84(2):695–701.

- 156. Acosta O, Ramirez VI, Lager S, Gaccioli F, Dudley DJ, Powell TL, Jansson T. Increased glucose and placental GLUT-1 in large babies of obese non-diabetic mothers. Am J Obstet Gynecol. 2015;212(2): 227.e1–7.
- 157. Chen X, Gao C, Li H, Huang L, Sun Q, Dong Y, Tian C, Gao S, Dong H, Guan D, Hu X, Zhao S, Li L, Zhu L, Yan Q, Zhang J, Zen K, Zhang CY. Identification and characterization of microRNAs in raw milk during different periods of lactation, commercial fluid, and powdered milk products. Cell Res. 2010;20(10):1128–37.
- 158. Meng F, Henson R, Wehbe-Janek H, Ghoshal K, Jacob ST, Patel T. MicroRNA-21 regulates expression of the PTEN tumor suppressor gene in human hepatocellular cancer. Gastroenterology. 2007;133(2): 647–58.
- 159. Han M, Liu M, Wang Y, Chen X, Xu J, Sun Y, Zhao L, Qu H, Fan Y, Wu C. Antagonism of miR-21 reverses epithelial-mesenchymal transition and cancer stem cell phenotype through AKT/ERK1/2 incactivation by targeting PTEN. PLoS ONE. 2012;7(6), e39520.
- 160. Dey N, Das F, Mariappan MM, Mandal CC, Ghosh-Choudhury N, Kasinath BS, Choudhury GG. MicroRNA-21 orchestrates high glucose-induced signals to TOR complex 1, resulting in renal cell pathology in diabetes. J Biol Chem. 2011;286(29): 25586–603.
- 161. Dey N, Ghosh-Choudhury N, Kasinath BS, Choudhury GG. TGFβ-stimulated microRNA-21 utilizes PTEN to orchestrate AKT/mTORC1 signaling for mesangial cell hypertrophy and matrix expansion. PLoS ONE. 2012;7(8), e42316.
- 162. Sayed D, Rane S, Lypowy J, He M, Chen IY, Vashistha H, Yan L, Malhotra A, Vatner D, Abdellatif M. MicroRNA-21 targets Sprouty2 and promotes cellular outgrowths. Mol Biol Cell. 2008;19(8): 3272–82.
- 163. Dariminpourain M, Wang S, Ittmann M, Kwabi-Addo B. Transcriptional and post-transcriptional regulation of Sprouty1, a receptor tyrosine kinase inhibitor in prostate cancer. Prostate Cancer Prostatic Dis. 2011;14(4):279–85.
- 164. Frey MR, Carraro G, Batra RK, Polk DB, Warburton D. Sprouty keeps bowel kinases regular in colon cancer, while miR-21 targets Sprouty. Cancer Biol Ther. 2011;11(1):122–4.
- 165. Asangani IA, Rasheed SA, Nikolova DA, Leupold JH, Colburn NH, Post S, Allgayer H. MicroRNA-21 (miR-21) post-transcriptionally downregulates tumor suppressor Pdcd4 and stimulates invasion, intravasation and metastasis in colorectal cancer. Oncogene. 2008;27(15):2128–36.
- 166. Lu Z, Liu M, Stribinskis V, Klinge CM, Ramos KS, Colburn NH, Li Y. MicroRNA-21 promotes cell transformation by targeting the programmed cell death 4 gene. Oncogene. 2008;27(31):4373–9.
- 167. Carayol N, Katsoulidis E, Sassano A, Altman JK, Druker BJ, Platanias LC. Suppression of programmed cell death 4 (PDCD4) protein expression by BCR-ABL- regulated engagement of the mTOR/

p70 S6 kinase pathway. J Biol Chem. 2008; 28(13):8601–10.

- 168. Ng R, Song G, Roll GR, Frandsen NM, Willenbring H. A microRNA-21 surge facilitates rapid cyclin D1 translation and cell cycle progression in mouse liver regeneration. J Clin Invest. 2012;122(3): 1097–108.
- 169. Dennis MD, Jefferson LS, Kimball SR. Role of p70S6K1-mediated phosphorylation of eIF4B and PDCD4 proteins in the regulation of protein synthesis. J Biol Chem. 2012;287(51):42890–9.
- 170. Jiang H, Wu W, Zhang M, Li J, Peng Y, Miao TT, Zhu H, Xu G. Aberrant upregulation of miR-21 in placental tissues of macrosomia. J Perinatol. 2014;34(9):658–63.
- Lager S, Oowell TL. Regulation of nutrient transport across the placenta. J Pregnancy. 2012;2012:179827.
- 172. Larqué E, Riuz-Palacios M, Koletzko B. Placental regulation of fetal nutrient supply. Curr Opin Clin Nutr Metab Care. 2013;16(3):292–7.
- 173. Jansson T, Aye IL, Goberdhan DC. The emerging role of mTORC1 signaling in placental nutrientsensing. Placenta. 2012;33 Suppl 2:e23–9.
- 174. Roos S, Jansson N, Palmberg I, Säljö K, Powell TL, Jansson T. Mammalian target of rapamycin in the human placenta regulates leucine transport and is down-regulated in restricted growth. J Physiol. 2007;582(Pt 1):449–59.
- 175. Roos S, Lagerlöf O, Wennergren M, Powell TL, Jansson T. Regulation of amino acid transporters by glucose and growth factors in cultured primary human trophoblast cells is mediated by mTOR signaling. Am J Physiol Cell Physiol. 2009;297(3):C723–31.
- 176. Jansson N, Rosario FJ, Gaccioli F, Lager S, Jones HN, Roos S, Jansson T, Powell TL. Activation of placental mTOR signaling and amino acid transporters in obese women giving birth to large babies. J Clin Endocrinol Metab. 2013;98(1):105–13.
- 177. Gaccioli F, White V, Capobianco E, Powell TL, Jawerbaum A, Jansson T. Maternal overweight induced by a diet with high content of saturated fat activates placental mTOR and eIF2alpha signaling and increases fetal growth in rats. Biol Reprod. 2013;89(4):96.
- 178. Lu Y, Qian L, Zhang Q, Chen B, Gui L, Huang D, Chen G, Chen L. Palmitate induces apoptosis in mouse aortic endothelial cells and endothelial dysfunction in mice fed high-calorie and high-cholesterol diets. Life Sci. 2013;92(24–26):1165–73.
- 179. Guilloteau P, Zabielski R, Hammon HM, Metges CC. Adverse effects of nutritional programming during prenatal and early postnatal life, some aspects of regulation and potential prevention and treatments. J Physiol Pharmacol. 2009;60 Suppl 3:17–35.
- Plagemann A. Perinatal nutrition and hormonedependent programming of food intake. Horm Res. 2006;65 Suppl 3:83–9.
- Melnik BC. Milk the promoter of chronic Western diseases. Med Hypotheses. 2009;72(6):631–9.
- Melnik BC. Excessive leucine-mTORC1-signalling of cow milk-based infant formula: the missing link

to understand early childhood obesity. J Obes. 2012;2012:197653.

- 183. Boney CM, Verma A, Tucker R, Vohr BR. Metabolic syndrome in childhood: association with birth weight, maternal obesity, and gestational diabetes mellitus. Pediatrics. 2005;115(3):e290–6.
- 184. Sørensen HT, Sabroe S, Rothman KJ, Gillman M, Fischer P, Sørensen TI. Relation between weight and length at birth and body mass index in young adulthood: cohort study. BMJ. 1997;315(7116):1137.
- 185. Leunissen RW, Stijnen T, Hokken-Koelega AC. Influence of birth size on body composition in early adulthood: the programming factors for growth and metabolism (PROGRAM)-study. Clin Endocrinol (Oxf). 2009;70(2):245–51.
- Brüske I, Flexeder C, Heinrich J. Body mass index and the incidence of asthma in children. Curr Opin Allergy Clin Immunol. 2014;14(2):155–60.
- 187. Skilton MR, Siitonen N, Würtz P, Viikari JS, Juonala M, Seppälä I, Laitinen T, Lehtimäki T, Taittonen L, Kähönen M, Celermajer DS, Raitakari OT. High birth weight is associated with obesity and increased carotid wall thickness in young adults: the cardio-vascular risk in young Finns study. Arterioscler Thromb Vasc Biol. 2014;34(5):1064–8.
- 188. Bukowski R, Chlebowski RT, Thune I, Furberg AS, Hankins GD, Malone FD, D'Alton ME. Birth weight, breast cancer and the potential mediating hormonal environment. PLoS ONE. 2012;7(7), e40199.
- 189. Spracklen CN, Wallace RB, Sealy-Jefferson S, Robinson JG, Freudenheim JL, Wellons MF, Saftlas AF, Snetselaar LG, Manson JE, Hou L, Qi L, Chlebowski RT, Ryckman KK. Birth weight and subsequent risk of cancer. Cancer Epidemiol. 2014; 38(5):538–43.
- 190. Lewis RM, Demmelmair H, Gaillard R, Godfrey KM, Hauguel-de Mouzon S, Huppertz B, Larque E, Saffery R, Symonds ME, Desoye G. The placental exposome: placental determinants of fetal adiposity and postnatal body composition. Ann Nutr Metab. 2013;63(3):208–15.
- 191. Symonds ME, Mendez MA, Meltzer HM, Koletzko B, Godfrey K, Forsyth S, van der Beek EM. Early life nutritional programming of obesity: mother-child cohort studies. Ann Nutr Metab. 2013;62(2):137–45.
- 192. Olafsdottir AS, Skuladottir GV, Thorsdottir I, Hauksson A, Steingrimsdottir L. Maternal diet in early and late pregnancy in relation to weight gain. Int J Obes (Lond). 2006;30(3):492–9.
- 193. Sebire NJ, Jolly M, Harris JP, Wadsworth J, Joffe M, Beard RW, Regan L, Robinson S. Maternal obesity and pregnancy outcome: a study of 287,213 pregnancies in London. Int J Obes Relat Metab Disord. 2001;25(8):1175–82.
- 194. Jolly MC, Sebire NJ, Harris JP, Regan L, Robinson S. Risk factors for macrosomia and its clinical consequences: a study of 350,311 pregnancies. Eur J Obstet Gynecol Reprod Biol. 2003;111(1):9–14.
- 195. Baeten JM, Bukusi EA, Lambe M. Pregnancy complications and outcomes among overweight and

obese nulliparous women. Am J Public Health. 2001;91(3):436-40.

- 196. Ehrenberg HM, Mercer BM, Catalano PM. The influence of obesity and diabetes on the prevalence of macrosomia. Am J Obstet Gynecol. 2004;191(3): 964–8.
- 197. Crane JM, White J, Murphy P, Burrage L, Hutchens D. The effect of gestational weight gain by body mass index on maternal and neonatal outcomes. J Obstet Gynaecol Can. 2009;31(1):28–35.
- 198. Bounous G, Kongshavn PA, Taveroff A, Gold P. Evolutionary traits in human milk proteins. Med Hypotheses. 1988;27(2):133–40.
- 199. Melnik BC. Formula feeding promotes adipogenic, diabetogenic, hypertonic and allergic mTORC1programming. In: Preedy VR, Watson RR, Zibadi S, editors. Handbook of dietary and nutritional aspects of bottle feeding, Human Health Handbooks no 8. Wageningen: Wageningen Academic Publihers; 2014. p. 545–68, chapt 34.

- Melnik BC. The potential mechanistic link between allergy and obesity development and infant formula feeding. Allergy Asthma Clin Immunol. 2014;10(1):37.
- 201. Johnson SC, Rabinovitch PS, Kaeberlein M. mTOR is a key modulator of ageing and age-related disease. Nature. 2013;493(7432):338–45.
- 202. Xu S, Cai Y, Wei Y. mTOR Signaling from cellular senescence to organismal aging. Aging Dis. 2014; 5(4):263–73.
- 203. Curry A. Archaeology: the milk revolution. Nature. 2013;500(7460):20–2.
- McCay, Crowell MF. Prolonging the life span. Sci Mon. 1934;39(5):405–14.
- 205. Vickers MH. Developmental programming of the metabolic syndrome – critical windows for intervention. World J Diabetes. 2011;2(9):137–48.
- 206. The American College of Obstetricians and Gynecologists. FAQ1. Nutrition during pregnancy. 2013; http://www.acog.org/~/media/For%20Patients/ faq001.pdf?dmc=1&ts= 20140823T1014147121.

Genetic Disease Specific Human Embryonic Stem Cell Lines

19

Anver Kuliev, O. Verlinsky, and S. Rechitsky

Introduction

Preimplantation Genetic Diagnosis (PGD) provides unlimited source for obtaining human embryonic stem cell (hESC) lines. As PGD involves the pre-selection of the genetic disease free embryos for transfer back to uterus the affected embryos were either discarded, or used for confirmation of diagnosis. These affected embryos otherwise provided the valuable source for the derivation of the genetic disease derivation of hESC lines with genetic and chromosomal disorders, which will be described in this paper. There were also attempts to obtain hESC lines from a single cell, removed from the 8-cell stage embryo at the time of embryo biopsy for PGD purposes, so these lines could be used for transplantation purposes even before the baby is born and can be a potential donor of HLA matched stem cells for the affected sibling [1, 2]. Despite feasibility of this approach, it does not seem practical, as removing additional material from the embryos at this stage may compromise the viability of the embryo.

Feasibility of establishing hESC lines from the inner cell mass of the blastocyst was first demonstrated in 1998 [3], and presently hESC lines for research purposes are provided by the NIH repository of hESC lines. This registry initially contained 78 lines, of which only 11 have met NIH scientific criteria, including the presence of L-alkaline phosphatase (TRA-2-39), Oct-4, high molecular weight glycoproteins (antibodies TRA-1-60, TRA-1-81), stage specific embryonic antigens (SSEA-3, SSEA-4), euploid karyotype and teratoma formation in SCID mice [4]. The list of hESC lines in the NIH registry is currently being extended, following Executive Order 13505, entitled "Removing Barriers to Responsible Research Involving Human Stem Cells." The Order was issued by President Obama on March 9, 2009, lifting the ban regarding the use of Federal funds for hESC research. Although the NIH registry has been significantly expanded, it contains only a few hESC lines with different genetic abnormalities I (see below).

The main source for obtaining hESC lines in our collection was the blastocyst stage embryo inner cell mass (ICM), isolated by immunosurgery and placed on a feeder layer, although other sources, such as morula was also attempted but appeared impractical [5–8]. The established hESC lines were tested for alkaline phosphatase, stage specific antigens SSEA-3 and SSEA-4, high molecular weight glycoproteins or tumour rejection antigens, TRA-1-60 and Oct-4. They were maintained in vitro from 10 to 15 passages before freezing in sufficient amounts, with no

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_19

A. Kuliev (⊠) • O. Verlinsky • S. Rechitsky Reproductive Genetic Innovations, 2910 MacArthur Blvd., 2825 N Halsted St, Chicago, IL, USA e-mail: anverkuliev@hotmail.com

[©] Springer International Publishing Switzerland 2016

observation of differences in the efficiency of obtaining hESC lines depending on the source.

Our collection of the hESC lines with genetic and chromosomal disorders, described in this paper, contains 87 lines, including 14 with chromosomal abnormalities and 73 with monogenic conditions, which are frozen and are available at different passages.

Chromosomal Disease Specific Human Embryonic Stem Cell Lines

The list of the chromosomal disease specific hESC lines is presented in Table 19.1. These are however only those few lines that were established, while the majority of attempts failed due to a poor outcome of hESC lines from the embryos with autosomal aneuploidies. Also, because of selective disadvantage of abnormal cells in culture, some of the hESC lines attempted to derive from chromosomally abnormal embryos ended up having a normal karyotype, phenomenon observed also reported previously [9–11]. On the other hand, it was reported that there is a risk of de novo chromosomal abnormalities in the process of the propagation and maintenance of hESC lines [12]. One of such incidental chromosomal abnormalities was detected in the hESC line, distributed from NIH registry to more than

 Table 19.1
 Chromosomal disease-specific hESC lines (total 14 lines)

	Stem cell karyotype
1	69,XXY
2	47,XXY
3	46,XX, der(4) t(4;13)
4	47,XX,+13
5	47,XX,+14
6	47,XXY
7	47,XXX
8	47,XY + 12
9	47,XY + der(21)t(2,21)
10	46,X + MAR
11	46,XX (T14;17)
12	47,XX + 21
13	46,XX iso (17q)
14	47,XX + 12

100 research laboratories around the world. This hESC line was obtained from the Wisconsinbased stem cell registry WiCell, and have had a normal female karyotype, which was stable though the establishment and maintenance for several months, with the ability to differentiate with the formation of neural and beating cardiac muscle cells. Karyotyping changes, involving the gain of the chromosome 17q, were observed in three independent hESC lines on the five independent occasions, together with the occasional gain of chromosome 12, which was suggested to be attributable to a selective advantage of these

cells to the propagation of undifferentiated cells.

This phenomenon created a concern over the use

of hESC lines for hESC -based therapies, because cytogenetic changes may be only the part of

genetic abnormalities acquired in the process of

the establishment, maintenance and differentiation of hESC lines. In a special multicenter study of 127 hESC lines from 39 laboratories worldwide for genetic changes occurring during culture, it was demonstrated that although these lines remained cytogenetically normal, a progressive tendency to acquire karyotypic changes on prolonged culture were also observed, commonly affecting chromosomes 1, 12, 17 and 20 [13]. Analysis for single nucleotide polymorphisms (SNP) showed that there were also genomic structural variants, which appeared sporadically but no common variants related to culture conditions were observed on chromosomes 1, 12 and 17. However, overlapping structural variants were acquired on chromosome 20p11.21 during culture of multiple cell lines, suggesting the anti-apoptotic gene, BCL2L1, as the most likely driver of culture adaptation in this chromosomal region. This makes initial genetic testing of the embryos used as a source of ESC lines one of the basic requirements.

A few hESC lines with chromosomal disorders were reported previously, including lines with trisomy 13, triploidy and mosaicism ([14]; NIH Registry 2011). As presented in Table 19.1, our collection of hESC lines contains 14 hESC lines obtained from the embryos with chromosomal disorders, including 4 lines with translocations, 1 trisomy 14, 1 triploidy, 1 trisomy 13, 2 trisomy 12, 1 trisomy 21, and 4 with an euploidy of sex chromosomes (1 with 45, X+mar, 1 with 47, XXX and 2 with 47, XXY, one of which was derived from the same embryo that was the source of the hESC line with EmeryDreifuss (carrier) type muscular dystrophy (see below).

However, as mentioned, this is only a small proportion of those embryos from which hESC lines were obtained, as the majority of the attempts failed. In fact, not all embryos are able even to reach the blastocyst stage, to allow attempting the obtain hESC lines, and this also may depend on the origin of aneuploidy, that affects the derivation success rates. We have previously demonstrated that the plating efficiency differed significantly depending on the origin of aneuploidy, with very low plating efficiency, when the embryos with prezygotic aneuploidy were the source, such as those originating from the zygotes with MI or MII errors [15]. Karyotypically normal hESC lines may be also produced from the embryos with chromosomal abnormalities [9–11], but undetected mosaicism could not be excluded in these cases, as the diagnosis was obtained at the cleavage stage that is characterized by an extremely high mosaicism rate, so selective advantage of normal cell in such cases may have taken place.

Genetic Disease Specific Human Embryonic Stem Cell Lines

Tables 19.2, 19.3, and 19.4 present our collection of the monogenetic disease specific hESC lines, representing the world's largest repository with 73 hESC lines with genetic disorders [16–19]. These lines provide an unlimited source for understanding the mechanisms of the phenotypic realization of genetic defects and for the development of the approaches for their possible treatment. A few hESC lines with genetic disorders are available in NIH registry, and were reported also by other laboratories, including cystic fibrosis, Charcot-Marie-Tooth disease, Duchenne Muscular Distrophy, congenital nephrotic syndrome, spinal muscular atrophy, and Marfan syndrome ([20]; NIH Stem Cell Registry). **Table 19.2** Autosimal recessive disease-specific hESClines: total 24 lines

Hemoglobin – alpha locus; HBA, affected $(/)$
Hemoglobin – beta locus; HBB, affected (cd39/ IVS1-110)
Hemoglobin – beta locus; HBB, affected (cd8 + G/619del)
Hemoglobin – beta locus; HBB, affected (HbS/HbS – sickle cell anemia)
Hemoglobin – beta locus; HBB, affected (IVS1-5/Cd8 + G)
Hemoglobin – beta locus; HBB, affected (IVSI-6/ IVSI-6)
Hemoglobin – beta locus; HBB, affected (Unknown/ IVSII-1)
Hemoglobin – beta locus; HBB, affected (Unknown/ IVSII-1)
Hemoglobin – beta locus; HBB, carrier (N/IVS 1-1)
Hemoglobin – beta locus; HBB, carrier (N/ IVS1-110)
Cystic fibrosis; affected (Δ F508/1717–1 G > A)
Cystic fibrosis; affected (Δ F508/1717–1 G > A)
Cystic fibrosis; affected ($\Delta F508/\Delta F508$)
Cystic fibrosis; affected ($\Delta F508/\Delta F508$)
Cystic fibrosis; affected ($\Delta F508/\Delta F508$)
Cystic fibrosis; affected ($\Delta F508/\Delta F508$)
Cystic fibrosis; affected (N1303K/ΔF508)
Cystic fibrosis; affected (W1282X/R117C)
Fanconi anemia, complementation group A; FANCA, carrier of 14 bp deletion, spinal muscular atrophy, type I; SMA1, affected, exon 7 deletion $(n=2)$
Sandhoff disease, affected (n=3)

The success rate of derivation of the genetic disease specific hESC lines was 20–25 %, not different from the success rate observed for hESC lines obtained from normal embryos. As seen from Table 19.2, our collection of hESC lines with genetic disorders contains 24 hESC lines derived from the embryos with autosomal recessive disorders, including 10 with beta-globin mutations (including thalassaemia and with sickle cell disease), 1 Fanconi anaemia, complementation group A, 8 cystic fibrosis, 2 spinal muscular atrophy, 3 Sandhoff disease, and 1 spinal muscular atrophy.

Fourteen hESC lines were obtained from the embryos with X-linked disorders (Table 19.3), including 1 adrenoreukodystrophy, 2 fragile site mental retardation (one affected male and one carrier female), 2 ocular albinism, 1 Becker, 4

 Table 19.3
 X-linked disease-specific hESC lines: total 14 lines

Albinism, ocular, type I; OA1, (c.251del C), affected male (1 line)

Albinism, ocular, type I; OA1, (N/c.251del C), carrier (1 line)

Adrenoleukodystrophy; ALD, (1801 del AG) affected male (1 line)

Muscular dystrophy, Becker type; BMD, affected male (1 line)

Muscular dystrophy, Duchenne type; DMD, affected (2 lines)

Muscular dystrophy, Duchenne type; DMD, carrier (2 lines)

Emery-Dreifuss muscular dystrophy, X-linked; EDMD, affected male (3 lines)

Emery-Dreifuss muscular dystrophy, X-linked; EDMD, carrier (1 line)

Fragile site mental retardation 1 affected male, expansion (1 line)

Fragile site mental retardation 1, carrier female (1 line)

 Table 19.4
 Dominant disease specific hESC lines: total

 35 lines

Breast cancer, familial (brca2); affected (n/ivs7 gt del)	
(1 line)	

Breast cancer, familial (brca2); affected (n/ivs7 gt del) and multiple endocrine

Neoplasia, type I; men1 affected (n/3036 4 bp del) (1 line)

Huntington disease; HD, affected, expansion (7 lines)

Marfan syndrome; MFS, affected, g7712a/n (1 line)

Dystrophia myotonica 1, affected, expansion (2 lines)

Neurofibromatosis, type I; NF, affected, (7 lines) Torsion dystonia 1, autosomal dominant; DYT1, affected, exon 7 gag deletion (3 lines)

Treacher Collins-Franceschetti syndrome; TCOF, affected (nt. 4374 ins. a/n) (3 lines)

Tuberous sclerosis type 1, affected (2 lines)

Popliteal pterygium syndrome; PPS, affected (1 line)

Facioscapulohumeral muscular dystrophy 1a;

FSHMD1A, affected (7 lines)

Duchenne (two affected and two carriers), and 4 Emery-Dreifuss (one affected and one carrier) type muscular dystrophy.

Thirty-five hESC lines were derived from the embryos with autosomal dominant conditions (Table 19.4), including 7 with neurofibromatosis type 1, 1 Marfan syndrome, 3 torsion dystonia, 2 tuberoses sclerosis, 1 pterigium syndrome, 7 fascioscapular muscular dystrophy 1A, 2 muscular dystrophy, 3 Treacher Collins-Franceschetti syndrome, 1 familial breast cancer BRCA-2, 1 familial breast cancer BRCA-2 together with type I multiple endocrine neoplasia and 7 Huntington's disease.

In addition, 12 hESC lines with polymorphism in the chemokine receptor 5 (CCR5) CMKBR5 gene, which in homozygous status confers resistance to HIV [21, 22], was detected by screening of 137 genetically normal hESC lines from our collection. One of them was established from the blastocyst deriving from partenogenetic embryo (46, XX), with two copies of all maternally derived genes, as demonstrated by polymorphic markers for X-chromosome, and chromosomes 3, 6, 11, 13, 18 and 21 [23]. Although this polymorphism was linked to HIV resistance long time ago, providing an immense promise of being able to treat HIV positive patients [21, 22], finding acceptable donor matches homozygous for the CCR5del32 presented a real challenge. So with the establishment of larger repositories of hESC lines, a search for finding HLA match for HIV patients may also be performed. At this time, the hESC lines with this polymorphism, and especially the one that has two copies of CCR5del32 allele, may provide an unlimited source for the research into transplantation treatment of this devastating disease.

Conclusion

Although initially the major goal of the establishment of hESC lines was the development of the cell-replacement therapies, it is presently obvious that hESC lines will have an important role in the studies of mechanisms of genetic disorders through generating the sources of normal and genetically abnormal cells and tissues. The ability to obtain hESC lines with specific genetic disorders, that could produce unlimited quantities of the disease tissue where the disease has a genetic basis, makes it realistic to undertake research on the primary disturbances of the cellular processes in the genetically abnormal cells and to identify the molecular mechanisms that might be blocked to prevent the disease progression. Therefore, there is obvious need for establishment of hESC lines originating from embryos with genetic and chromosomal abnormalities, to provide the basis for understanding of the mechanisms of phenotype realization of genetic defects and for the development of new approaches for their possible treatment.

Our repository has a large collection of hESC lines, which provides a unique opportunity to screen available hESC lines for polymorphisms associated with susceptibility and/or resistance to diseases in humans, as demonstrated by the establishment of the first such line containing CCR5-32bp deletion, conferring resistance to HIV. So this approach may be productive for finding hESC lines with rare mutations which may prove invaluable to the future stem cell therapy of severe disorders for which there is no available treatment. This may have particular potential for research into the mechanisms of predisposition or resistance to common disorders, the results of which could lead to new treatment regiments to those common conditions for which there is no available treatment.

References

- Ilic D, Giritharan G, Zdravkovic T, et al. Derivation of hESC lines from biopsied blastomeres on human feeders with minimal exposure to xenomaterials. Stem Cell Dev. 2009;18:1343–50.
- Yang G, Mai Q, Li T, Zhou C. Derivation of hESC lines from single blastomeres of low quality embryos by direct plating. J Assist Reprod Genet. 2013;30: 953–61.
- Tompson JA, Itskovitz-Eldor J, Shapiro SS, et al. Embryonic stem cell lines derived from human blastocysts. Science. 1998;282:1145–7.
- National Institutes of Health guidelines for research using human pluripotent stem cells. NIH stem cells information archives. 25 Aug 2000. www.nih.gov/ news/stemcell/stemcellguidelines.htm.
- Soukoyan MA, Vatolin SY, Golubitsa AN, et al. Embryonic stem cells derived from morulae, inner cell mass and blastocyst of mink: comparison of their pluripotencies. Mol Reprod Dev. 1993;36:148–58.
- Stice SL, Strelchenko NS, Keefer CL, Matthews L. Pluripotent bovine embryonic stem cell lines direct embryonic development following nuclear transfer. Biol Reprod. 1996;54:100–10.

- Shamblott MJ, Axelman J, Littlefield JM, et al. Human embryonic germ cell derivatives express a broad range of developmentally distinct markers and proliferate extensively in vitro. Proc Natl Acad Sci U S A. 2001;98:113–8.
- Strelchenko N, Verlinsky O, Kukharenko V, Verlinsky Y. Morula derived human embryonic stem cells. Reprod BioMed Online. 2004;9:623–9.
- Peura T, Boswan A, Chami O, et al. Karyotypically normal and abnormal hESC lines derived from PGDanalyzed embryos. Cloning Stem Cells. 2008;10:203–16.
- Sous-Toby E, Gerecht-Nir S, Amit M, et al. Derivation of a diploid hESC line from mononuclear zygote. Hum Reprod. 2004;19:670–5.
- Chen X, Luo Y, Fan Y, et al. Triploid and diploid embryonic stem cell lines derived from tripronuclear human zygotes. Assist Reprod Genet. 2012;29:713–21.
- Draper JD, Smith K, Gokhale P, et al. Recurrent gain of chromosomes 17q and 12 in cultures human embryonic stem cells. Nat Biotechnol. 2004;22:53–4.
- The International Stem Cell Initiative. Screening ethnically diverse hESC lines identifies a chromosome 20 minimal amplicon conferring growth advantage. Nat Biotechnol. 2011;29:1132–46.
- Heins N, Englund MCO, Sjoblom C, et al. Derivation, characterization, and differentiation of human embryonic stem cells. Stem Cells. 2004;22:367–76.
- Verlinsky Y, Strelchenko N, Kukharenko V, et al. Impact of meiotic and mitotic non-disjunction on generation of hESC lines. Reprod BioMed Online. 2009;18:120–6.
- Verlinsky Y, Strelchenko N, Kukharenko V, et al. Hman embryonic stem cell lines with genetic disorders. Reprod BioMed Online. 2005;10:105–10.
- 17. Verlinsky Y, Kuliev A. Practical preimplantation genetic diagnosis. London: Springer; 2006.
- Verlinsky Y, Strelchenko N, Kukharenko V, et al. Repository of human embryonic stem cell lines and development of individual specific lines using stembrid technology. Reprod BioMed Online. 2006;13:547–50.
- Verlinsky Y, Strelchenko N, Kukharenko V, et al. Isolation of human embryonic stem cells from various stages of the human embryo. In: Lakshmipathy, editor. Emerging technology platforms for stem cells. Hoboken: Wiley; 2009. p. 19–27.
- Pickering S, Minger S, Patel M, et al. Generation of a hESC line encoding the cystic fibrosis mutation Delta F-508 using preimplantation genetic diagnosis. Reprod BioMed Online. 2005;10:390–7.
- Liu R, Paxton WA, Choe S, Ceradini D, Martin SR, Horuk R, MacDonald ME, Stuhlmann H, Koup RA, Landau NR. Homozygous defect in HIV-1 coreceptor accounts for resistance of some multiply-exposed individuals to HIV-1 infection. Cell. 1996;86:367–77.
- Hütter G, Nowak D, Mossner M, et al. Long-term control of HIV by CCR5 Delta32/Delta32 stem-cell transplantation. N Engl J Med. 2009;360:692–8.
- Pomerantseva E, Kukharenko V, Goodman A, Vrlinsky O, Rechitsky S, Kuliev A. Human embryonic stem cell lines with CCR5-del32 allele conferring resistance to HIV. Stem Cell Discov. 2011;1:67–70.

Part IV

Impact of Stem Cell on Growth and Development of Human Fetus

Stem Cells in Growth and Development of the Human Fetus

20

Phuc Van Pham

Stem Cells

Stem cells are considered the origins of all organisms. In fact, all organisms are formed and developed from a single cell – the zygote, which is the product of oocyte fertilization by the sperm. This cell is referred to as a totipotent stem cell. Through the process of cell division, the totipotent stem cell produces all tissue types within an organism.

Stem cells have two important characteristics; the ability to undergo cell differentiation and self-renewal [5, 24]. Differentiation is a process by which a stem cell can become a functional cell. For example, hematopoietic stem cells (HSCs) can differentiate into red blood cells [8]. HSCs cannot transfer oxygen unlike their red blood cell decedents whose primary role is to carry oxygen. Self-renewal is a complex biological system which enables stem cells to maintain as well as increase their number. Stem cells perform self-renewal via asymmetric or symmetric division to ensure that at least one stem cell is produced from the cell division process.

By means of self-renewal, stem cells are maintained during embryonic development right through to adulthood. However, there are significant differences in stem cell properties between these stages. Therefore, stem cells are divided into at least two groups, including embryonic stem cells (ECSs) derived from the blastocyst and adult stem cells derived from the fetus or adult [5]. The main difference between them is that ESCs exhibit pluripotency, while adult stem cells only exhibit multipotency. Differences in the potential of these stem cells are summarized in Table 20.1.

In 2006, Yamanaka et al. (2006) showed that fibroblasts can be reprogrammed into pluripotent stem cells, and called these cells induced pluripotent stem cells (iPSCs) [49]. From this study, almost all mature cells or adult stem cells can be reprogrammed into iPSCs. iPSCs exhibit almost all ESC properties, including expression of pluripotent genes, such as Oct-3/4 and Nanog, multiple lineage differentiation, and successful production of chimeric mice. These cells offer much promise within biology and medicine. With respect to biology, studies of these cells have confirmed that all cells in the human body originate from a single cell, and that the offspring cells are different because of differences in their gene expression profiles. With respect to medicine, iPSCs open up a whole new therapy approach for the treatment of many chronic diseases [3, 44].

Another type of stem cell of particular interest in the field of cancer research is the cancer

P. Van Pham, PhD

Laboratory of Stem Cell Research and Application, Ho Chi Minh City University of Science, Vietnam National University, Ho Chi Minh City, Vietnam e-mail: pvphuc@hcmuns.edu.vn

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_20
Potential	Number of differentiated cells	For example	Types of differentiated cells
Totipotent	All	Zygote (or Blastomere)	All cells in human body and extra- embryonic parts
Pluripotent	All, exclude trophoblast	All cells from inner cell mass	All cells from three germinal layers
Multipotent	Many	Hematopoietic stem cells	Cardiac cells; muscle cells; osteoblasts; hepatic cells; all kinds of blood cells
Oligopotent	Some	Myeloid	Five kinds of blood cells: red blood cells; monocytes; macrophages; eosinophils; neutrophils
Quadripotent	Four	Mesenchymal progenitor cells	Chondrocytes; adipocytes; glial cells; osteoblasts
Tripotent	Three	Glial progenitor cells	Two kinds of astrocytes; and oligo-dendrocytes
Bipotent	Two	Mouse liver derived precursor cells	Lymphocyte B and macrophage
Unipotent	One	Mast cell precursor	Mast cells

Table 20.1 The differences in potential of stem cells

stem cell. Cancer stem cells are defined as the original cells giving rise to malignant tumors [6, 47]. To date, the origin of cancer stem cells is hypothesized differently; with the most accepted origin of the cancer stem cell being a mutated stem cell. When normal stem cells become mutated, especially with respect to their self-renewal-related genes, they can proliferate uncontrollably and thus form tumors [51].

In general, stem cells play an extremely important role in the formation, growth and development of all organisms. In this chapter, the role of stem cells in the growth and development of the human fetus is demonstrated.

Early Embryo Development and the Origin of Stem Cells

To date, there has been much debate about the existence of stem cells in the early embryo. Some authors suggest that in fact no stem cell exists in the early embryo and that at this stage there are complex changes in the development of embryonic cells, such as multiplication, commitment, differentiation, and death.

For example, the unfertilized oocyte may be regarded as a pluripotent cell, which can be activated to develop into the embryo in a parthenogenic manner [46]. Although ESCs can be successfully isolated from these embryos, results in primates and mice have shown that they cannot grow and develop into whole organisms [17, 26, 52]. This means oocytes cannot be regarded as totipotent cells, while zygotes are totipotent stem cells because they can develop into whole organisms. Results from embryo splitting technology show that only the zygote and blastomere at the eightcell stage embryo are in fact totipotent [38, 43].

In the next stage of embryonic development, differentiation of totipotent stem cells starts to form at least two cell types within the blastocyst, including trophoblasts and the inner cell mass. Cells in the inner cell mass maintain the stemness with pluripotency and are so-called ESCs [2, 48]. Conversely, trophoblasts are differentiated cells that form the extraembryonic tissues, such as the placenta. This is the first commitment event in mammalian development. When ESCs loose the capacity to produce trophoblasts, they become pluripotent not totipotent stem cells.

The next commitment event relates to the formation of the primitive endoderm, the precursor of the yolk sac. The yolk sac can be considered a primitive form of the placenta, which plays the vital role in the transport of nutrients from the mother to fetus.

The inner cell mass comprising ESCs continues to develop and form an epithelium, known as the epiblast. The epiblast continues to maintain pluripotency with the ability to differentiate into multiple tissues, as well as form teratomas when transplanted into ectopic sites. In fact, ESCs can also be isolated from the epiblast [10, 12, 25]. However, there is a difference between the inner cell mass-derived ESCs, which can integrate into a host blastocyst to form a chimera, and epiblastderived ESCs, which loose this potential. In the next stage of embryonic development, the epiblast progresses to the gastrulation stage, with the formation of three germinal layers. The cells in these three germinal layers become more specific, thus loosing pluripotency. In fact, during the process of gastrulation, the intensities of Wnt, nodal and BMP signaling determine cell fate in the epiblast. Development from gastrulation to the fetus is a complex process that results in the formation of all specific tissues and organs. It is believed that gastrulation-specific stem cells are produced by self-renewal of the epiblast.

Roles of Tissue Specific Stem Cells in Morphogenesis: Breast Model

The mammary gland is a specific organ in the human body that continues to undergo morphogenesis postnatally. Therefore, mammary gland morphogenesis has been studied in-depth. In fact, the mammary gland changes with respect to both tissue structures as well as cell population during puberty, pregnancy and menopause. To date, many studies have confirmed that all these processes are driven by mammary stem cells. Early evidence has shown that normal human mammary epithelium is developed from mammary stem cells, a process first observed through studies of the X-chromosome inactivation pattern. In 1996, Tsai et al. analyzed cells from normal mammary epithelium and showed that all cells had the same pattern of X-chromosome inactivation [50]. This means that all cells in the same mammary gland originate from a single cell.

In the mouse model, one single mammary stem cell can regenerate the entire mammary gland *in vivo* [35, 36]. In humans, xenotransplantation using human mammary stem cells injected into the

mammary fat pad or under the renal capsule of immune-deficient mice *in vivo*, results in outgrowths equivalent in size to one human mammary lobule.

Further investigations have enabled normal human mammary stem cells to be characterized based on specific marker expression and differentiation potential. As a stem cell, human mammary stem cells exhibit prolonged self-renewal; differentiate along both luminal and myoepithelial lineages, undergo branching morphogenesis in three dimensional culture, and generate outgrowths in xenotransplantation experiments. From the mammary cell population, mammary stem cells can be enriched by surface markers including CD49f^{high}EpCAM^{low} [9], CD73⁺CD90⁻ [40], CD10⁺ [23], and CD49f⁺DLL1⁺DNER⁺ [34]. Mammary stem cells can also be enriched based on expression of high expression of aldehyde dehydrogenase (ALDH) [13]; or based on mammosphere culture [34].

Human mammary development starts from 4 to 6 months of embryonic development [18]. At this time point, the primary mammary gland contains a central and peripheral basal cell population that produces two different cell layers [22]. From 21 to 25 weeks, epithelial buds form from the primary bud. In some cases, branched ducts highly structured together with regular lobules form similar to those observed in adults [31, 41] (Fig. 20.1).

Neural Stem Cells: Generating the Brain

Formation of the CNS starts with neural plate formation, followed by neural tube formation, which comprises one layer of neuro-epithelial cells. After the neural tube matures, the cellular architecture becomes stratified and neural stem cells are found in the ventricular layer, while the post-mitotic cells migrate toward the brain surface [1]. Cortical development is the result of symmetric cell division of neural stem cells. Asymmetric cell division produces two distinct cell types: one stem cell and one immature neuron or intermediate progenitor cell (IPC) (also called a basic progenitor). Immature



Fig. 20.1 Stem cells and mammary development. Mammary gland is formed by mammary stem cells. These cells can differentiate into two kinds of mammary gland included myoepithelial cells and luminal cells

neurons migrate away from the ventricular zone and become mature neurons of the cortical plate, whereas the intermediate progenitors reside in the subventricular zone, where they continue to divide and constitute an important reservoir for new neurons throughout neurogenesis. The intermediate progenitors are also able to divide symmetrically, generating two progenitors or two neurons [16, 30].

Neuroepithelial cells are gradually replaced by radial glia [27]. These radial glial cells express several astroglial properties and markers specific to the glial lineage, such as RC2 [29]. Radial glia can divide asymmetrically and serve as progenitors of neurons and glia, as well as constitute a scaffold onto which neurons migrate in the developing brain. Furthermore, it is now appreciated that they are in fact stem cells and similar to neuroepithelial cells. However, radial glial cells have a more restricted potential than neuroepithelial cells. In vivo evidence of neuroepithelial cell tripotency has been demonstrated by retroviral trace labeling, whereas the majority of labeled radial glia are found to give rise to a single cell type, for example a neuron, astrocyte, or oligodendrocyte [15, 28] (Fig. 20.2).

Hematopoietic Stem Cells and Their Roles in Blood Formation

In humans, HSCs originate from the mesoderm/ hemangioblast. From day 21, hemangioblasts mature into pre-HSCs and HSCs first appear in the AGM (aorta-gonad-mesonephros) region from day 28 [14]. They continue to colonize the developing fetal liver, therefore the fetal liver becomes the main source of HSCs until birth, at which point bone marrow hematopoiesis is established. However, there are two regions related to HSC production including the AGM and yolk sac.

A long-standing hypothesis posits that hematopoietic cells originate from the hemangioblast – a mesodermal precursor cell that gives rise to blood and endothelial cells. These precursors develop between days 2.5 and 4.0 of embryoid body (EB) differentiation and form blast-like colonies (referred to as blast colony forming cells, BL-CFCs). These precursors give rise to primitive and definitive hematopoietic, endothelial and vascular smooth muscle cells (Fig. 20.3).

BL-CFCs have been isolated from brachyury^{+/} Flk1^{+/}Kdr⁺ (Kdr: kinase insert domain protein receptor) cells, indicating that they are a specialized subset of the mesoderm [11]. Their subsequent commitment into the hematopoietic fate is driven and marked by expression of Scl/ Tal1 (T-cell acute lymphocytic leukemia 1) [7, 37]. More importantly, Flk1⁺brachyury⁺ hemangioblasts can also be detected in the gastrulating mouse embryo [19]. However, surprisingly few of the BL-CFCs that display both hematopoietic and vascular potential are found in the yolk sac.

Hemangioblasts are suggested to be precursors of yolk sac hematopoietic cells. This implies



Fig. 20.2 Neural stem cells (NSCs) in development and in the adult. Neuroepithelial cells in early development divide symmetrically to generate more neuroepithelial cells. Some neuroepithelial cells likely generate early neurons. As the developing brain epithelium thickens, neuroepithelial cells elongate and convert into radial glial cells. Radial glial cells divide asymmetrically to generate neurons directly or indirectly through neural stem cells. Oligodendrocytes are also derived from radial glial cells through neural stem cells that generate oligodendrocytes

Fig. 20.3 Establishment of definitive hematopoietic stem cell (HSC) pools in human. Hematopoietic development starts as specification of primitive streak mesoderm into hematopoietic and vascular fates. Pre-HSCs at yolk sac, AGM or placenta would go to a maturation process to produce HSCs in yolk sac, AGM, placenta, fetal liver and bone marrow. Subsequently, fetal HSCs expand rapidly, after which a steady state is established in which HSCs reside in a relatively quiescent state in the bone marrow



that the first stages of hematopoietic development take place before the cells migrate into the yolk sac, prior to the formation of blood islands, which consist of developing primitive red cells and endothelial cells adjacent to the visceral endoderm [33].

Migration of hematopoietic precursors from the primitive streak to the yolk sac depends on vascular endothelial growth factor signaling through Flk1 [45]. Further studies have shown that specification of hematopoietic fate in the yolk sac also depends on the visceral endoderm and on signals from Ihh (Indian hedgehog) [4] and Bmp4 (bone morphogenetic protein 4) [42], demonstrating the importance of the microenvironment for hematopoietic commitment.

Errors of Stem Cell Self-Renewal, Differentiation and Diseases

Fetal Alcohol Syndrome and Neural Stem Cells

Fetal alcohol syndrome (FAS) is considered the result of the effect of alcohol on neural development of the fetus, which interferes with differentiation and the formation of neurons and glial cells from neural stem cells. FAS usually occurs in children born to mothers who have abused alcohol during pregnancy. Some FAS defects include pre- and post-natal growth deficiencies, minor facial abnormalities, and damage to the developing central nervous system (CNS). Many FAS-related defects occur in the CNS, with gross morphological abnormalities of the brain and when compared to normal children, an overall decrease in white matter, particularly in the cerebrum or forebrain. Certain areas of the developing CNS are particularly susceptible to alcohol-induced birth defects, including the ocular system, corpus callosum, basal ganglia, and cerebellum. From 5 to 6 weeks after fertilization, alcohol also impacts the corpus callosums of fetuses. The corpus callosum is a dense band of white matter that separates the right and left hemispheres of the brain and is responsible for coordinating communication between hemispheres.

Corpus callosum defects can lead to poor bimanual motor coordination or motor-visual coordination, and issues with faculties like abstract thought and decision making.

Alcohol is also considered as an apoptotic agent that causes damage to neural stem cells or neural progenitor cells, like radial glia, which give rise to neurons and supporting glial cells in the CNS. Radial glia not only provide a physical scaffolding but also chemically guide neurons as they form. Alcohol can impact the development and migration of these radial glia progenitor pools, thereby resulting in fewer neurons and glia being produced. In particular, neurons as well as glia with morphological abnormalities can be produced. These effects reduce cell volumes and cause structural abnormalities, which can in turn impact the CNS.

Myelodysplastic Syndrome (MDS): A Stem Cell Disorder

Myelodysplastic syndrome (MDS) comprises a group of diverse bone marrow disorders in which the bone marrow does not produce enough healthy blood cells. MDS is often referred to as a "bone marrow failure disorder". MDS is primarily a disease of the elderly (most patients are older than 65 years), but MDS can affect younger patients too. There are three or four phases of MDS, including progression to acute leukemia, and the loss of proliferation or differentiation capability, leading to progressive cytopenias. Cytopenias are low blood cell counts, and are a hallmark feature of MDS responsible for some of the symptoms that MDS patients experience, including infection, anemia, spontaneous bleeding, or easy bruising. In addition to reduced numbers of blood cells, the mature blood cells circulating in the blood system may not function properly because of dysplasia. Failure of the bone marrow to produce mature healthy cells is a gradual process, and therefore MDS is not necessarily a terminal disease.

To date, a number of studies have confirmed that MDS is related to defects in HSCs. These defects give rise to secondary stem cell abnormalities, for example involving stem cell niche interactions or immune function. The MDS patient HSC pool has been shown to undergo increased apoptosis as well as decreased proliferation and survival. HSCs up-regulate a number of interferon (IFN)-inducible genes. Depletion of early HSCs is seen in MDS, suggestive of defective selfrenewal. This means there are some errors in the genes regulating the self-renewal, differentiation and quiescence of HSCs in MDS patients. The stem cell niche is a micro-environment providing protection to HSCs, especially in keeping HSCs quiescent and protecting them from oxidative stress. Therefore, defects in signaling between stem cells and their niche also cause depletion of HSCs.

Auto-immune Diseases as Stem Cell Disorders

Using animal models, a number of studies have found that autoimmune diseases are actually stem cell disorders. Interestingly, HSC transplantation can be used to treat these diseases; however transplantation of abnormal HSCs from autoimmune disease mice to normal mice may trigger autoimmune disease in the recipients. Some researchers have successfully identified and cultured both normal and abnormal HSCs and identified some of the differences between them.

In an early study, Morton and Siegel transplanted bone-marrow cells from lupus-prone NZB mice into lethally irradiated non-susceptible strains and induced lupus syndrome [32]. In another study using the same method, bone marrow transplantation transferred the autoimmune diseases systemic lupus erythematosus (SLE), experimental autoimmune encephalomyelitis (EAE), adjuvant arthritis, antiphospholipid syndrome and type 1 diabetes, into the recipients [20].

It is interesting that bone marrow transplantation, also called HSC transplantation, is reported to be the best way to treat many autoimmune diseases. For example, in a report of more than 195 methods used to prevent or delay type 1 diabetes in non-obese diabetic (NOD) mice, HSC transplantation was highlighted as the most effective method [39]. Allo-transplantation of HSCs also was considered as a suitable and effective therapy in treating SLE [21].

Conclusions

Origins of stem cells in human fetal development and growth are unclear. Stem cells exist in the early embryo, including the blastocyst, and in the adult, as confirmed during gastrulation and fetal development stages. Furthermore, adult stem cells can be produced from ESCs. Theoretically, all adult stem cells originate from ESCs via the process of selfrenewal and differentiation. However, to date, gaps remain between the science of stem cells and that of human development. Some scientists believe that adult stem cells are formed early during fetal formation, however different terminologies exist between development and stem cell biologists regarding these cell types. It is therefore essential to unify these different terms in order to clarify stem cell roles and formation. Despite this, stem cells clearly exhibit important roles during fetal development and growth. A greater number of stem cell disorders are currently being discovered and the effects of stem cell damage on normal fetal development are gradually being elucidated.

References

- Altmann CR, Brivanlou AH. Neural patterning in the vertebrate embryo. Int Rev Cytol. 2001;203:447–82.
- Amit M. Sources and derivation of human embryonic stem cells. Methods Mol Biol (Clifton, NJ). 2013;997:3–11.
- Andrews PW, Cavagnaro J, Deans R, Feigal E, Horowitz E, Keating A, Rao M, Turner M, Wilmut I, Yamanaka S. Harmonizing standards for producing clinical-grade therapies from pluripotent stem cells. Nat Biotechnol. 2014;32:724–6.
- Baron M. Induction of embryonic hematopoietic and endothelial stem/progenitor cells by hedgehog-mediated signals. Differ Res Biol Divers. 2001;68:175–85.
- Bongso A, Eng HL. Stem cells: their definition, classification and sources. In: Bongso A, Lee EH, editors. Stem cells: from benchtop to bedside. Singapore: World Scientific; 2005.

- Chen SY, Huang YC, Liu SP, Tsai FJ, Shyu WC, Lin SZ. An overview of concepts for cancer stem cells. Cell Transplant. 2011;20:113–20.
- Chung YS, Zhang WJ, Arentson E, Kingsley PD, Palis J, Choi K. Lineage analysis of the hemangioblast as defined by FLK1 and SCL expression. Development (Cambridge, England). 2002;129: 5511–20.
- Eaves CJ. Hematopoietic stem cells: concepts, definitions, and the new reality. Blood. 2015;125:2605–13.
- Eirew P, Stingl J, Raouf A, Turashvili G, Aparicio S, Emerman JT, Eaves CJ. A method for quantifying normal human mammary epithelial stem cells with in vivo regenerative ability. Nat Med. 2008;14:1384–9.
- Factor DC, Najm FJ, Tesar PJ. Generation and characterization of epiblast stem cells from blastocyst-stage mouse embryos. Methods Mol Biol (Clifton, NJ). 2013;1074:1–13.
- Fehling HJ, Lacaud G, Kubo A, Kennedy M, Robertson S, Keller G, Kouskoff V. Tracking mesoderm induction and its specification to the hemangioblast during embryonic stem cell differentiation. Development (Cambridge, England). 2003;130:4217–27.
- Gillich A, Bao S, Surani MA. Reversion of mouse postimplantation epiblast stem cells to a naive pluripotent state by modulation of signalling pathways. Methods Mol Biol (Clifton, NJ). 2013;1074:15–29.
- Ginestier C, Hur MH, Charafe-Jauffret E, Monville F, Dutcher J, Brown M, Jacquemier J, Viens P, Kleer CG, Liu S, et al. ALDH1 is a marker of normal and malignant human mammary stem cells and a predictor of poor clinical outcome. Cell Stem Cell. 2007;1: 555–67.
- Godin I, Cumano A. The hare and the tortoise: an embryonic haematopoietic race. Nat Rev Immunol. 2002;2:593–604.
- Grove EA, Williams BP, Li DQ, Hajihosseini M, Friedrich A, Price J. Multiple restricted lineages in the embryonic rat cerebral cortex. Development (Cambridge, England). 1993;117:553–61.
- Haubensak W, Attardo A, Denk W, Huttner WB. Neurons arise in the basal neuroepithelium of the early mammalian telencephalon: a major site of neurogenesis. Proc Natl Acad Sci U S A. 2004;101:3196–201.
- Hirabayashi M, Goto T, Tamura C, Sanbo M, Hara H, Kato-Itoh M, Sato H, Kobayashi T, Nakauchi H, Hochi S. Derivation of embryonic stem cell lines from parthenogenetically developing rat blastocysts. Stem Cells Dev. 2014;23:107–14.
- Howard BA, Gusterson BA. Human breast development. J Mammary Gland Biol Neoplasia. 2000;5:119–37.
- Huber TL, Kouskoff V, Fehling HJ, Palis J, Keller G. Haemangioblast commitment is initiated in the primitive streak of the mouse embryo. Nature. 2004;432:625–30.
- 20. Ikehara S, Kawamura M, Takao F, Inaba M, Yasumizu R, Than S, Hisha H, Sugiura K, Koide Y, Yoshida TO, et al. Organ-specific and systemic autoimmune diseases originate from defects in hematopoietic stem cells. Proc Natl Acad Sci U S A. 1990;87:8341–4.

- 21. Ikehara S, Yasumizu R, Inaba M, Izui S, Hayakawa K, Sekita K, Toki J, Sugiura K, Iwai H, Nakamura T, et al. Long-term observations of autoimmune-prone mice treated for autoimmune disease by allogeneic bone marrow transplantation. Proc Natl Acad Sci U S A. 1989;86:3306–10.
- Jolicoeur F, Gaboury LA, Oligny LL. Basal cells of second trimester fetal breasts: immunohistochemical study of myoepithelial precursors. Pediatr Dev Pathol: Off J Soc Pediatr Pathol Paediatr Pathol Soc. 2003;6:398–413.
- 23. Keller PJ, Arendt LM, Skibinski A, Logvinenko T, Klebba I, Dong S, Smith AE, Prat A, Perou CM, Gilmore H, et al. Defining the cellular precursors to human breast cancer. Proc Natl Acad Sci U S A. 2012;109:2772–7.
- Lajtha LG. Stem cell concepts. Differ Res Biol Divers. 1979;14:23–34.
- Li W, Ding S. Converting mouse epiblast stem cells into mouse embryonic stem cells by using small molecules. Methods Mol Biol (Clifton, NJ). 2013;1074:31–7.
- Mai Q, Yu Y, Li T, Wang L, Chen MJ, Huang SZ, Zhou C, Zhou Q. Derivation of human embryonic stem cell lines from parthenogenetic blastocysts. Cell Res. 2007;17:1008–19.
- Malatesta P, Hartfuss E, Gotz M. Isolation of radial glial cells by fluorescent-activated cell sorting reveals a neuronal lineage. Development (Cambridge, England). 2000;127:5253–63.
- McCarthy M, Turnbull DH, Walsh CA, Fishell G. Telencephalic neural progenitors appear to be restricted to regional and glial fates before the onset of neurogenesis. J Neurosci: Off J Soc Neurosci. 2001;21:6772–81.
- 29. Misson JP, Edwards MA, Yamamoto M, Caviness Jr VS. Identification of radial glial cells within the developing murine central nervous system: studies based upon a new immunohistochemical marker. Brain Res Dev Brain Res. 1988;44:95–108.
- Miyata T, Kawaguchi A, Saito K, Kawano M, Muto T, Ogawa M. Asymmetric production of surfacedividing and non-surface-dividing cortical progenitor cells. Development (Cambridge, England). 2004;131: 3133–45.
- Monaghan P, Perusinghe NP, Cowen P, Gusterson BA. Peripubertal human breast development. Anat Rec. 1990;226:501–8.
- 32. Morton JI, Siegel BV. Transplantation of autoimmune potential. I. Development of antinuclear antibodies in H-2 histocompatible recipients of bone marrow from New Zealand Black mice. Proc Natl Acad Sci U S A. 1974;71:2162–5.
- Palis J, Yoder MC. Yolk-sac hematopoiesis: the first blood cells of mouse and man. Exp Hematol. 2001; 29:927–36.
- 34. Pece S, Tosoni D, Confalonieri S, Mazzarol G, Vecchi M, Ronzoni S, Bernard L, Viale G, Pelicci PG, Di Fiore PP. Biological and molecular heterogeneity of breast cancers correlates with their cancer stem cell content. Cell. 2010;140:62–73.

- 35. Prater MD, Petit V, Alasdair Russell I, Giraddi RR, Shehata M, Menon S, Schulte R, Kalajzic I, Rath N, Olson MF, et al. Mammary stem cells have myoepithelial cell properties. Nat Cell Biol. 2014;16:942–50, 941–7.
- Rios AC, Fu NY, Lindeman GJ, Visvader JE. In situ identification of bipotent stem cells in the mammary gland. Nature. 2014;506:322–7.
- Robertson SM, Kennedy M, Shannon JM, Keller G. A transitional stage in the commitment of mesoderm to hematopoiesis requiring the transcription factor SCL/ tal-1. Development (Cambridge, England). 2000;127: 2447–59.
- Robl JM. Development and application of technology for large scale cloning of cattle. Theriogenology. 1999;51:499–508.
- Roep BO, Atkinson M. Animal models have little to teach us about type 1 diabetes: 1. In support of this proposal. Diabetologia. 2004;47:1650–6.
- 40. Roy S, Gascard P, Dumont N, Zhao J, Pan D, Petrie S, Margeta M, Tlsty TD. Rare somatic cells from human breast tissue exhibit extensive lineage plasticity. Proc Natl Acad Sci U S A. 2013;110:4598–603.
- Rudland PS. Histochemical organization and cellular composition of ductal buds in developing human breast: evidence of cytochemical intermediates between epithelial and myoepithelial cells. J Histochem Cytochem: Off J Histochem Soc. 1991; 39:1471–84.
- 42. Sadlon TJ, Lewis ID, D'Andrea RJ. BMP4: its role in development of the hematopoietic system and potential as a hematopoietic growth factor. Stem Cells (Dayt, Ohio). 2004;22:457–74.
- Schramm RD, Paprocki AM. Strategies for the production of genetically identical monkeys by embryo splitting. Reprod Biol Endocrinol: RB&E. 2004;2:38.

- Seki T, Fukuda K. Methods of induced pluripotent stem cells for clinical application. World J Stem Cells. 2015;7:116–25.
- 45. Shalaby F, Ho J, Stanford WL, Fischer KD, Schuh AC, Schwartz L, Bernstein A, Rossant J. A requirement for Flk1 in primitive and definitive hematopoiesis and vasculogenesis. Cell. 1997;89:981–90.
- 46. Shaw JM, Trounson AO. Parthenogenetic activation of unfertilized mouse oocytes by exposure to 1,2-propanediol is influenced by temperature, oocyte age, and cumulus removal. Gamete Res. 1989;24:269–79.
- Shipitsin M, Polyak K. The cancer stem cell hypothesis: in search of definitions, markers, and relevance. Lab Invest; J Tech Methods Pathol. 2008;88:459–63.
- 48. Sviridova-Chailakhyan TA, Tzoy NG, Panchenko MM, Akatov VS, Chailakhyan LM. An efficient method for isolation of inner cell masses from the mouse blastocysts for culturing embryonic stem cells. Dokl Biol Sci: Proc Acad Sci USSR, Biol Sci Sect/ Translated Russ. 2008;423:469–72.
- Takahashi K, Yamanaka S. Induction of pluripotent stem cells from mouse embryonic and adult fibroblast cultures by defined factors. Cell. 2006;126:663–76.
- Tsai YC, Lu Y, Nichols PW, Zlotnikov G, Jones PA, Smith HS. Contiguous patches of normal human mammary epithelium derived from a single stem cell: implications for breast carcinogenesis. Cancer Res. 1996;56:402–4.
- Valent P, Bonnet D, De Maria R, Lapidot T, Copland M, Melo JV, Chomienne C, Ishikawa F, Schuringa JJ, Stassi G, et al. Cancer stem cell definitions and terminology: the devil is in the details. Nat Rev Cancer. 2012;12:767–75.
- 52. Yang H, Liu Z, Ma Y, Zhong C, Yin Q, Zhou C, Shi L, Cai Y, Zhao H, Wang H, et al. Generation of haploid embryonic stem cells from Macaca fascicularis monkey parthenotes. Cell Res. 2013;23:1187–200.

The Mystery of the Survival of the Human Fetus in Utero

21

Niranjan Bhattacharya, Sanjukta Bhattacharya, and Phillip G. Stubblefield

Introduction

The study of the body's immunity and immune defense system has enabled physicians to understand how in a successful pregnancy, the baby is allowed to grow inside the mother's body without being rejected. All healthy individuals have an immune system that will reject germs, viruses and donated organs, e.g., kidneys, hearts and livers, with the immune rejection reaction. However, during a successful pregnancy, this outside interference with the mother's immune system in not activated.

During a successful pregnancy, the immunity that may cause rejection of the baby and placenta is shut off in the mother's uterus. Throughout the rest of the mother's body, her immune system is fully functional, allowing her to deal in a proper

S. Bhattacharya, PhD

Professor, Department of International Relations, Jadavpur University, Kolkata, West Bengal, India

P.G. Stubblefield, MD

way with any infections that may come her way. Unfortunately many couples with recurrent spontaneous miscarriages find themselves in a situation where, following conception, this immune camouflage and protection of the embryo is not initiated in such a way that the uterine local and systemic immune responses of the mother are modulated in favour of a decrease in cell mediated immunity and increase in humoral immunity [1].

Discussion

Scientists have understood for years how to manipulate and interfere with the immune system and its functioning and the prevention of Rh sensitization during pregnancy, hemolytic disease of the newborn being a clear example. In this case, the injection of readymade Rh immunoglobulin (Rh antibody) in to the Rh negative mother carrying an Rh positive baby prevents her own immune system from recognizing and rejecting the baby's Rh positive cells. Briefly, the motherto-be receives signals (molecular messages) from the conceptus. Many of these messages are hormonal (endocrine) in nature; however, others are direct genetic messages that the father contributed to the child. Some of these involve the tissue type (HLA antigens) that the father gives to the child. If the parents are appropriately "mis-matched", so far as their tissue types are concerned, the mother's system recognizes that the baby's cells

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn),

MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India e-mail: sanjuktaniranjan@gmail.com

Emeritus Professor of Obstetrics and Gynecology at Boston University School of Medicine, Boston University, Jamaica Plain, MA, USA

[©] Springer International Publishing Switzerland 2016 N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development:*

First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_21

(the trophoblasts) that will form the placenta are "foreign". The lymphocytes of her immune system that have congregated in her uterus communicate with the molecular messages from the baby's cells and begin to protect the baby from immune rejection. The mother's immune system forms an antibody that can be measured as early as 5 weeks of pregnancy. This antibody attaches to the molecules on the cells of the baby that induced them. Here, they camouflage the baby's cells from the mother's immune killer cells that can destroy. In addition, the attachment of this specific antibody to the baby's cells brings

with it a signal that makes them grow and divide [2]. Without this "growth" signal, the baby and the placenta regress and die. If the maternal antibody which leads to the protection of the baby is low the baby is not

protection of the baby is low, the baby is not effectively shielded and protected. The cells of the placenta are not stimulated to grow, because there is no antibody produced by the mother. The outcome over the next few days or weeks will be the rejection of the fetus. By immunological testing and HLA tissue typing, these couples look more like brother and sister than unrelated husband and wife. In sum, the message from the father communicated to the mother via the tiny conceptus is not heard by the mother. Immunoprotection is not induced and the pregnancy fails repeatedly [3].

The problems of "lucky" or "unlucky" match can be addressed together through tissue typing and identifying the HLA antigens of both. In addition, the couple's ability to respond appropriately to each other can be tested through a laboratory assay called the mixed lymphocyte culture reaction.

Another problem that interferes with a subsequent pregnancy success operates through an alteration of the blood clotting mechanisms. Each pregnancy which the mother loses, is associated with an approximately 20 % increased chance that she will make an autoimmune response to fatty molecules called phospholipids that are integral parts of the baby's cells. There is a family of phospholipids: the one best known and most tested for is cardiolipin—the heart fat. Cardiolipin or one of its 'brothers' and "sisters" in the mother, may activate and speed up her blood clotting in the vicinity of early pregnancy and this results in its loss. Conclusions in pregnancy may occur later in gestation such as intrauterine growth retardation, toxemia and even intrauterine fetal demise. Half of these will be premature and the other half will have features of growth retardation, the remainder of the babies will possibly die during pregnancy due to abnormalities of the anti-phospholipid antibody and a demonstrated clotting abnormality (i.e., the lupus anti coagulant) [4].

Other causes of recurrent spontaneous abortion include hormonal problems, abnormal development or diseases of the uterus or cervix, poor sperm quality, infections, chromosome abnormalities and chronic diseases.

Every person's white blood cell type (tissue type, HLA type) consists of ten numbers, half of which comes from each parent. These white blood cell numbers are molecules (antennae) that are present on the surface of the person's white blood cells, and they serve important functions in recognizing foreign antigens (germs and viruses) that enter our body. This recognition even results in an immunity response that eventually leads to death of the germ or virus and the formation of an antibody against the germ or virus that prevents recurrence of the disease.

A representative white blood cell type is as follows: A1.2, B7.12, C4.2, DR 3.8 and DQ 4.1. One number at each locus (A, B, C, DR and DQ) comes from the mother and the other from the father.

Couples with secondary abortions (one living child and then spontaneous miscarriages) are a distinct group. Sixty percent of the first-born children of secondary couples are DQ 0501/501 (formally called DQ 4.1 homozygous). This means that they are purebred with regard to this part of their white blood cell type. Both mother and father contributed the same 4.1 gene. An analysis of several hundred live born children who were born following immunological treatment of their mothers revealed no live born DQ 0501/0501 babies. Women whose first babies are DQ 0501/0501 are at high risk of making an autoimmunity response that involves the production of an antibody (immunity) in their body that attacks the glue (phospholipids) that are important in building the placenta of the baby. They are also prone to increase the numbers of natural killer cells in their blood from 4 % to over 20-30 % [3].

Further studies in the laboratory of Prof. Alan Beer in 1995 showed that this type of natural killer cells in women with recurrent miscarriage can also damage the early cells that will eventually make the placenta. (Beer AE. Personal communication on the problem of abortion with intra-amniotic antigen on 09.05.1997.)

Now, thanks to a better understanding because of contributions from AE Beerr, AM Bahar et al., JYH Kwak et al. [5–6] on immunological factors that protect or destroy the developing embryo and fetus, researchers using simple new therapies are reporting very high rates of successful pregnancies among thousands of women who have experienced previously unexplained recurrent miscarriages [7–10].

The treatments are designed to counter one or two types of immunological problems that can result in destruction of the placenta, embryo or fetus by antibodies produced by the pregnant woman. One approach is designed to block antibodies against the fetus by the mother's own tissues. In the language of Prf. Beer, 'Blocking antibodies create a disguise and the fetus becomes a wolf in sheep's clothing as far as the mother's immune system is concerned' [3, 11, 12].

Now, with a better understanding of the pathophysiology of pregnancy we can see that highly specific, protecting systems operate during pregnancies which do not interfere with any of the mother's immune responses. Immunoglobins (IgG) are available commercially. If a pregnant woman were to receive a kidney from an unmatched donor, she would reject it at any other time. Similarly, her body can continue to fight off outside infectious agents [13–16] but not the pregnancy due to the presence of this blocking antibody in the maternal system.

Since half the genetic information in the fetus comes from a "foreigner", namely the father, without blocking antibodies acting as an immunological fortress, the mother's body would launch an attack against it. But nature does not rely on the mother alone to protect the fetus. As indicated by another set of studies, the substance that stimulates the production of blocking antibodies comes from the father, carried in by the sperm that produces the pregnancy [17].

Dr. JYH Kwak et al. [18] and Prof AE Beer et al. [18] have however, indicated that sometimes the tissue of the father and the mother are too much alike immunologically and the woman's body fails to respond adequately to the father's feeble signal [19] to produce blocking antibodies. This results in outright rejection of the pregnancy tissue or failure of the fetus to thrive.

Normally, the signal to produce blocking antibodies comes from the paternal contribution to the embryo. The antibodies can be measured as early as 5 weeks of pregnancy. But if the maternal and paternal tissue types are too similar the signal is not received and the pregnant woman does not produce the protective antibody [20].

Allopregnancy Is a Th2-Like Phenomenon

According to one study, CBA×DBA/2 mice are particularly susceptible to low doses of Th1 mediators (TNF, gamma IFN), which act in synergy to induce high fetal loss. In addition, CBA×DBA/2 placentae and deciduae produce few, if any, Th2 cytokines such as IL-4, IL-3 and IL-10. Alloimmunisation, which is known to prevent resorptions in this abortion-susceptible combination and to prevent TNF- or LPS-induced abortions/resorptions, enhances the placental production of IL-4 and IL-10 in CBA×DBA/2 matings. Furthermore, rIl-10 given alone by intraperitoneal injection completely reverses the high incidence of fetal resorptions, in control to anti IL-10. Antigamma IFN and pentoxifylline (an anti-TFN agent), which partially reduce resorption, act in synergy for optimal fetal protection. Injection of recombinant bovine trophoblast protein (r.o TP) corrects the high rate of fetal resorption in CBA×DBA/2 mice and is correlated with increased placental IL-4 and IL-10 production. These results show that Th2 play a more important role than Th1 and suggest that a simple immune-endocrine network is involved in establishing the characteristic Th1/Th2 balance observed during pregnancy [21].

The results of many studies indicates that non-specific killer cells and inflammatory, cytostatic and cytotoxic lymphokines play a crucial role in immunologically mediated fetal death in both naturally abortion-prone CBA×DBA/2 and B10×B10. A mice have a higher resorptions rate, which increases with age and the prevalence of "bad father" [22]. A correlation has been shown between these high resorptions rates and accumulation of asialo GM1+cells at the implantation sites of the embryos that are to be resorbed [23]. Activation of these cells by poly-IC or gamma-IFN lead to abortion [1, 24, 25].

Role of TJ6 Protein during Pregnancy

TJ6 is a novel immune suppressor protein which can induce apoptosis.

Early studies demonstrated that mice treated with antibody to the TJ6 protein early in pregnancy resulted in ablation of those pregnancies. The gene for TJ6 was subsequently cloned in both man and mice and were found to be nearly identical. Antibodies and DNA probes prepared from one species are reactive with the other. Expression appears to be highly regulated and in part (1) Cell activation and (2) Steroid hormone. Other factors such as mitogens and cytokines appear to modulate these effects.

Flow cytometric analysis of TJ6 on peripheral blood lymphocytes of pregnant women revealed that TJ6 was expressed on CD19 positive B cells. Specifically, TJ6 is expressed as a 45 kD membrane form on the surface of these cells. While the initial protein was shown to be approximately 70 kD in length, two additional forms have been identified. Subsequent studies show that the TJ6 is post-translationally modified to produce a 45 kD membrane form and an approximately 18 kD soluble form. Some preliminary studies have shown that the membrane form is involved in programmed cell death of the cells expressing it. The soluble form appears to be involved in antiproliferative effects of anti-CD3 and allogeneic stimulated cells [26].

Reproductive Immunophenotype

If we study in detail the reproductive phenotype which is essential for its manipulative role in the continuation of pregnancy, it is seen that there are 30 different types of white blood cells (Lymphocytes). However, eight of them are most important and can be assessed by flow cytometry. Disorder of these eight cell types may predict a future pregnancy loss. These tests were developed by Professor Alan Beer. (Personal communication and suggestions of Prof. Alan E. Beer on the problem of abortions with intra amniotic antigens on 09/05/1997.)

- A. CD3 (Pan T cells) range between 63 % and 86 %. They are low when the immune system is suppressed. They are high when the immune system is overactive.
- B. CD4 (T Helper cells) range 31–53 %. They are destroyed by HIV virus. If they are low the aetiology of the deficiency must be studied.
- C. CD5 (T cytotoxic suppressors) range 17–35 %. They coordinate the strong or weak immune reaction by coordinating Pan T and helper T cells.
- D. CD19 (B cells) range 3–8 %. These lymphocytes are plasma cells that produce antibody 19 m in the blood, then IgG in the lymph system and lastly, the IgA protects the organ (organ immunity). CD19 is high in cases of women with an immune related cause for infertility.
- E. CD56+CD16+Natural killer cells, range 3–12 %. These are natural killer cells produced in the bone marrow and produce tumor necrotic factor. If level varies above 18 %, intravenous immunoglobin may help the situation, otherwise may result in reproductive failure.
- F. CD56+Natural killer cells, range 3–12 %. They lack CD16 molecules or rather, known as CD56+CD16 natural killer cells, which could be identified in decidua and may also

produce large quantity of tumor necrotic factor in the decidua and can kill placental cells or the fetal cells. A level of 18 % or above predicts poor pregnancy outcome and may necessitate I.V. Immunoglobin G therapy which can decrease the killing potential of the natural killer (NK) cells.

- G. CD3 IL-2 R cells, normal range 0–5 %. They are high in autoimmune disease in case of rejection of kidney or bone graft. If the level is above 10 % I.V. immunoglobin may prevent its activation.
- H. CD19+CD5+ (B1 cells), range 0–3 %. When activated they produce antibody against hormone, hormone receptor and neurotransmitters and may also justify autoimmune condition or rejection of an organ, e.g., bone marrow.

Conclusion

Why a pregnancy survives, remains an eternal mystery. There are many concepts and working hypotheses, that try to explain why the fetus survives in utero. Many factors, notably complement's role, chromosomal role of the parents, steroid and immunosuppressive role, manipulation by immunophenotypes, novel immunosuppressor's role etc have been highlighted in many research papers over the last 50 years. Yet, none of these explanations can comprehensively explain the survival of the human fetus.

References

- Wagman TG, Lin H, Guilbert J, Mosmann TR. Bidirectimal cytokine interactions in the maternal fetal relationship: is successful pregnancy a TH2 phenomenon. Immunol Today. 1993;14:353–5.
- Gilman Sachs A, Luo SP, Beer AE, Beaman KD. Identification of antilymphocytic antibodies by flow cytometric analysis of microlymphocytotoxicity in women with recurrent spontaneous abortion with paternal leukocytes. J Clin Lab Immunol. 1989;30:53–9.
- Beer AE, Kwak JYH, Ruiz JE: Immunophenotypic profiles of peripheral blood lymphocytes in women with recurrent pregnancy losses and in infertile

women with multiple failed in vitro fertilization cycles. Am J Reprod Imm. 1996;35:376–382.

- 4. Beer AE, Joanne YH, Kwak KJ, Gilman Sachs A, Beaman KD. New Horizons in the evaluation and treatment of recurrent pregnancy loss. In: Hunt SJ, editor. Immunobiology of reproduction. New York: Springer Verlag; 1994.
- Bahar AM, Kwak JYH, Beer AE, Kim LJ, Nelson LA, Beaman KD, Gilman-Sachs A. Antibodies to phospholipids and unclear antigens in no pregnant women with unexplained spontaneous recurrent abortions. Reprod Immunol. 1993;24:213–22.
- Kwak KJ, Ober C, Barini R, Beer AE. Maternal autoimmune abnormalities and fetal HLA-DQ AI alleles in women with recurrent spontaneous abortions (abstract). 39th Annual meeting of the Society for Gynecological Investigation: Location: San Antonio, Texas; 1992.
- Barbui T, Radici E, Cortelazzo S, Rosste Colli M, et al. Antiphospholipid antibody in early repeated abortion. A case controlled study. Fertil Steril. 1998;50:589–92.
- Beer AE, Kwak JYH. What is the evidence of immunologic pregnancy loss? Lymphocytic immunization, the supportive view. In: Chaout G, Mowbry J, editors. Molecular and cellular biology of maternal and fetal relationship. London: Inserm; 1990. p. 285–92.
- Blank M, Cohen J, Toder V, Schoenfel Y. Indication of antiphospholipid syndrome in naïve mice with mouse lupus monoclonal and human polyclonal anticordiolipid antibody. Proc Natl Acad Sci USA. 1991;88:3069–77.
- Coulam CB. Epidemiology of recurrent spontaneous abortion. Am J Reprod Immunol. 1991;26:23.
- Kwak JYH, Beaman KD, Gilman Sachs A, Ruiz JE, Schweitz D, Beer AE. Upregulated expression of CD 56 +CD 56+/16+ and CD19+ cells in peripheral lymphocytes in women with recurrent pregnancy losses. Am J Reprod Immunol. 1995;34:93.
- Coulam CB, Goodman C, Rousseu RG, Thomson EJ, Beaman KD. Systemic CD56+ cells can predict pregnancy outcome. Am J Reprod Immunol. 1995;33:40.
- Trinchier G. Biology of natural killer cells. Adv Immunol. 1989;47:187–376.
- Johnson PM. Immunobiological characterization of the trophoblast decidual interface in human pregnancy. In: Hunt JS, editor. Immunology of reproduction. Boston: Springer; 1993. p. 3–13.
- Brikly C, Loke YW. Early human decidual cells exhibit NK activity against the K562 cell line but not against first trimester trophoblast. Cell Immunol. 1989;118:337.
- Kwak JYH, Gilman Sachs A, Beaman KD, Beer AE. Immunogenetic profiles in women with primary recurrent spontaneous abortions (RSA) and secondary RSA. Am J Reprod Immunol. 1991;25:50–1.
- Kwak JYH, Gilman Sachs A, Beaman KD, Beer AE. Autoantibody in women with primary recurrent

spontaneous abortion of unknown aetiology. J Reprod Immunol. 1992;22:15–31.

- 18. Kwak JYH, Barine R, Gilman Sachs A, Beaman KD, Beer AE. Down regulation of maternal antiphospholipid antibody during early pregnancy and pregnancy outcome (abstract). Society of Gynaecological Investigations, 40th meeting: Toronto, Canada; 1993. p. 2, 107.
- Kovatis S, Main EK, Librach C, Stubblefiedl M, Fisher SJ, Demar R. A class I antigen HLA G expresses in human trophoblast. Science. 1990;248:220–3.
- Beer AE. Immunology of reproduction. In: May S, editor. Immunological diseases. 4th ed. Boston: Little Brown Company; 1988. p. 329–60. Ch-15.
- 21. Chaouat G, Clark DA, Weegman TG. Genetic aspects of the CBA/J ×DBA/2j and B10 ×B10A model of murine spontaneous abortions and prevention by levcoctic immunization. In: Allen WR, et al. editors. Early pregnancy loss mechanism and treatment. 18th

RCOG Study Group RCOG: RCOG Press, London; 1988. p. 89–105.

- Gendron R, Baines M. Infiltrating decidual natural killer cells are associated in the spontaneous abortion in mice. Cell Immunol. 1988;113:261.
- De Fougerolles R, Baines M. Modulation of natural killer activity influences resorptions rates in CBA ×DBA 12 matings. J Reprod Immunol. 1988;II:147–53.
- Kinsky R, Delage G, Rosin M, Thanh MN, Hoffman M, Chaouat G. A murine model of NK cell mediated resorptions. Am J Immunol. 1990;23:73.
- Lin H, Mossman TR, Guikbert L, Tontipoppipat S, Wegmann TG. Synthesis of T helper-TH2, cytokines at the maternal fetal interphase. J Immunol. 1993;151:4562.
- 26. Rubesa G, Beaman KD, Lucin P, Beer AE and Rukavina D: Expression of TJ6 protein in the human first trimester decidual lymphocytes. Regional Immunology. 1994;6:331–333.

Part V

Fetal Endocrine Development up to Second Trimester

Glucose Metabolism in Foetus and Its Relationship with Foetal Insulin

22

Prabir Kumar Kundu

Introduction

Glucose is the principal blood sugar of the human foetus and its main energy source. Not surprisingly, therefore, its supply to these tissues is regulated by a relatively complex set of mechanisms that tend to keep its metabolism relatively constant [1]. The first point in this regulation is the maintenance of maternal glucose concentration by increasing rates of maternal glucose production and development of relative maternal glucose intolerance and insulin resistance. The second point is the transfer of maternal glucose to the fetus by the placenta, which is buffered by placental glucose utilization. The third point is the production of insulin by the developing foetal pancreas, which enhances glucose utilization among the insulin-sensitive tissues (skeletal muscle, liver, heart, adipose tissue) that increase in mass and thus glucose need during late gestation [2, 3]. Glucose uptake into foetal tissues is regulated by glucose transporters that increase or decrease in response to both acute and chronic changes in foetal glucose concentration and conditions of intrauterine growth restriction (IUGR). At the same time, signal transduction protein regulators of amino acid synthesis into protein are downregulated, emphasizing that IUGR presents a mixed phenotype, with increased propensity to take up energy substrates, such as glucose, and diminished capacity for protein synthesis and growth [4].

Knowledge and understanding of the foetal glucose understanding is necessary to understand the nutritional requirements for normal growth and development, to provide rationale treatment for acute foetal distress and to provide prenatal care of a diabetic patient. Foetal glucose metabolism depends on additive effects of foetal plasma glucose and insulin. Glucose-stimulated insulin secretion increases over gestation, is down-regulated by constant hyperglycemia, but enhanced by pulsatile hyperglycemia. Insulin production is diminished in fetuses with intrauterine growth restriction (IUGR) by inhibition of pancreatic cell replication, but not by mechanisms that regulate insulin production or secretion, while the opposite occurs with hypoglycemia alone, despite its common occurrence in IUGR. Chronic hyperglycemia down-regulates glucose tolerance and insulin sensitivity with decreased expression of skeletal muscle and hepatic Glut 1 and 4 glucose transporters, while chronic hypoglycemia up-regulates these transporters. The opposite occurs for signal transduction proteins that regulate amino acid synthesis into protein. Such adaptations might underlie childhood and adult metabolic disorders of insulin resistance, obesity, and diabetes mellitus.

First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_22

P.K. Kundu, MD(Medicine), MD(Tropical Medicine) Department of Endocrine, Metabolic and Nutrition Diseases, Calcutta School of Tropical Medicine, Kolkata, India

Calcutta School of Tropical Medicine, Kolkata, India e-mail: drpkkund080255@gmail.com

[©] Springer International Publishing Switzerland 2016 N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development:*

Glucose Homeostasis Regulation in the Foetus

The foetus is constantly supplied with glucose that passes through the placenta. Glucose, aminoacids and lactate are the main energetic substrates during the foetal development. Glucose is the source of approximately half of the total energy necessary for the growth and development of the foetus. It passes through the placenta by diffusion and the amount that is being used up by the foetus depends on the mother's blood glucose level as well as on the transplacental glucose gradient. Foetal blood glucose level comprises approximately 70-80 % of the glucose concentration in the mother's venous blood serum. The foetus, besides taking the glucose from the maternal blood, utilizes also quite high amounts of amino acids and in this way the mother is being deprived of gluconeogenesis precursors. For this reason pregnant women develop hypoglycaemia faster than before pregnancy. Foetal metabolism is set on anabolic processes. This is the reason for building up glycogen, which is stored in the liver during late pregnancy and is being used as an easily accessible source of glucose in the first hours after birth. Enzymatic systems necessary for gluconeogenesis and glycogenolysis are present in the foetal liver, but remain inactive and are activated only in case of extreme maternal starvation. The foetal liver contains approximately three times more glycogen than the liver of an adult and during birth this amount constitutes the energetic reserve for the newborn. Fat oxidation plays during foetal live a less important role than the oxidation of aminoacids and glucose, so the production of ketones is low. Insulin, similarly as in the other periods of life, is the main anabolic hormone of the foetus and can be detected in the foetal pancreas already in early development stages. Endocrine pancreatic cells develop between the further investigations. It seems however, that to protect the anabolic processes - glycogen and fat tissue formation - a high insulin/glucagon ratio is being maintained.

These observations underline the importance of the appropriate functional development of the endocrine part of the pancreas for the foetal development. Investigations concerning the amount of receptors and their hormonal affinity revealed, that these both parameters are decreased for glucagon receptors in contrary to insulin receptors, which number and insulin affinity are significantly higher than in an adult pancreas. This phenomenon also indicates a major predominance of anabolic processes in this part of life.

Insulin does not pass through the placenta and therefore the foetus has to be independent from the mother. Insulin secretion in the foetus is regulated by the concentration of glucose as well as the concentration of amino acids, but the β cell of the foetal pancreas becomes with time more and more susceptible to glucose stimulation. A particularly intensive increase of β cell mass takes place in the last trimester of pregnancy. This development influences numerous other processes in the future life and may play a deciding role in the predisposition towards different diseases in adulthood. Young adults with a very low birth weight have higher indexes of insulin resistance and glucose intolerance and higher blood pressure than those born at term [6]. During this time the development of the exocrine part of the pancreas takes place and the islets begin to be formed (these structures are fully developed in the 22 gestation week). In comparison to adults, foetal β cells are characterized by decreased response to blood glucose and amino acids levels alterations in vivo as well as in vitro.

Foetal Endocrine Pancreas

The foetal pancreas appears during the fourth week of foetal life. The alpha cells, which contain glucagon, and the beta cells, which contain somatostatin, develop before the beta cells differentiate; however, insulin can be recognized in the developing pancreas before beta cell differentiation is apparent. Human pancreatic insulin and glucagon concentrations increase with advancing foetal age, and are higher than concentrations found in the adult pancreas. In vivo studies of umbilical cord blood obtained at delivery and foetal scalp blood samples obtained at term show that foetal insulin secretion is low and tends to be relatively unresponsive to acute changes in glucose. In contrast, foetal insulin secretion, in vitro, is responsive to amino acids

and glucagon as early as 14 weeks' gestation. In maternal diabetes mellitus, foetal islet cells undergo hypertrophy such that the rate of insulin secretion increases [9, 10].

- Week 7 to 20 pancreatic hormones secretion increases, small amount maternal insulin
- Week 10 glucagon (alpha) differentiate first, somatostatin (delta), insulin (beta) cells differentiate, insulin secretion begins
- Week 15 glucagon detectable in foetal plasma

Foetal Insulin Secretion

The foetal pancreas develops in the late first to early second trimester, producing measurable insulin concentrations by mid-gestation. There is a gradual increase in basal insulin concentration and glucose- and arginine-induced insulin secretion towards term in foetal sheep [2].

Regulation of Insulin Secretion by Glucose Concentration

Foetal insulin secretion responds variably to changes in glucose concentration that are dependent on the absolute change in glucose concentration and its duration, magnitude and pattern [6]. For example, glucose stimulated- and basal insulin secretion in foetal sheep in late gestation are down-regulated in the presence of chronic, sustained, marked hyperglycaemia [5, 6] but pulsatile hyperglycaemia increases foetal insulin secretion, as occurs in human fetuses and neonates [6, 12]. Interestingly, hypoglycaemia also decreases basal and glucose-induced insulin secretion [16, 17], evidence of how intrauterine growth restriction, which characteristically among human clinical cases and animal models involves foetal hypoglycaemia, decreases foetal pancreatic development and insulin secretion capacity. Amino acids also regulate insulin secretion, but for the most part, acute and very

high physiological to pharmacological changes in a given amino acid's concentration are necessary to elicit a significant change in foetal insulin concentrations [11].

Changes in Foetal Insulin Secretion with Intrauterine Growth Restriction

Human foetuses with severe intrauterine growth restriction (IUGR) have less pancreatic endocrine tissue and exhibit β -cell dysfunction [22, 27]. Such defects, if permanent, might limit β -cell function in later life and contribute to the increased incidence of non-insulin-dependent diabetes mellitus in individuals who were subject to growth restriction in utero [13, 14]. Similar observations have been made experimentally. For example, in the maternal hyperthermia-induced placental insufficiency model of IUGR in sheep, plasma insulin concentrations in the IUGR fetuses are 64 % lower at baseline and 77 % lower after glucose-stimulated insulin secretion (GSIS) [18]. Modelling of changes in plasma insulin concentration in response to glucose and arginine stimulation revealed deficits in the insulin secretion rate in the IUGR fetuses. When tested in vitro, pancreatic islets from the IUGR foetuses secreted insulin in response to increasing glucose and potassium chloride (KCl) concentrations, but the mass of insulin released per IUGR islet was lower than control islets from normally grown fetuses, due to an 82 % reduction in their insulin content [15]. A deficiency in islet glucose oxidation, but not total utilization rate, was found at maximal stimulatory glucose concentrations (11 mmol l-1). Interestingly, however, these same studies showed that insulin release as a fraction of total insulin content was greater in the IUGR islets. Thus pancreatic islets from placental insufficiency-induced IUGR foetuses have impaired β -cell stimulus–secretion coupling as a result of reduced glucose-stimulated glucose oxidation rates, insulin biosynthesis and insulin content, despite increased fractional rates of insulin release that results from a greater proportion of releasable insulin in the presence of lower insulin stores [16, 18].

Changes in Foetal Insulin Secretion with Hypoglycaemia

A common biochemical characteristic of fetuses with IUGR is relative hypoglycaemia; it is reasonable therefore to consider whether foetal pancreatic adaptations to hypoglycaemia alone, without other general nutrient deficiencies associated with IUGR (reduced amino acid and oxygen delivery, for example), would, by itself, limit the capacity of foetal pancreatic β -cells to produce and/or secrete insulin. Preliminary studies measured glucose-stimulated insulin secretion (GSIS) in normal control foetal sheep, foetal sheep made hypoglycaemic by maternal insulin infusion for 2 weeks in late gestation, and in a similar hypoglycaemic group of fetuses after a 5-day euglycaemic recovery period [16]. Hypoglycaemia significantly decreased plasma insulin concentrations in the hypoglycaemic (0.13}0.01 ng ml-1) and recovery foetuses (0.11 }0.01 ng ml-1); insulin concentrations returned to euglycaemic control values $(0.30\}0.01$ ng ml-1) in recovery fetuses (0.29}0.04 ng ml-1) during their euglycaemic recovery period. Mean steady state plasma insulin concentration during the GSIS study was reduced in hypoglycaemic fetuses (0.40}0.07 versus 0.92}0.10 ng ml-1 in controls); there was some recovery [19, 20].

Other Aspects of Glucose and Amino Acid Metabolism in IUGR

Up-regulation of mechanisms regulating glucose utilization. Many different models of IUGR [29] have shown that when the fetus is deprived of glucose, either selectively such as with maternal hypoglycaemia or generally as with placental insufficiency, foetal weight-specific glucose utilization rate is not very much different from normal rates [30]. These conditions both produce foetal hypoglycaemia, which increases the maternal-foetal glucose concentration gradient and thus the driving force for glucose transport across the placenta, thereby compensating in part for the reduction in glucose supply. For the fetus to accomplish this, there must be an increase in the foetal tissue capacity for glucose uptake and/or utilization, which could come about by increased concentrations and/or activity and/or plasma membrane localization of glucose transporters, increased insulin signal transduction and thereby effectiveness to promote Glut 4 (and perhaps Glut 1) translocation to the cell membrane, or mechanisms of insulin metabolism into oxidative and/or non-oxidative pathways. In both foetal sheep and foetal rats with IUGR, Glut 1 and Glut 4 concentrations in myocardium, adipose tissue and skeletal muscle do not decrease with sustained hypoglycaemia [8, 25, 26], perhaps a positive adaptation to maintain glucose utilization despite hypoglycaemia.

According to other preliminary data (M. A. Anderson, J. Friedman, W. W. Hay Jr, unpublished results, 2001) in placental insufficiency-induced IUGR, foetal sheep have shown increased levels of insulin receptor protein and decreased levels of the insulin signal transduction inhibitors, P85 protein subunit of phosphatidylinositol 3 kinase (PI3K), which negatively regulates the effect of insulin to promote Glut 4 translocation from inactive intracellular storage pools to active sites in the cell membrane where it enhances glucose uptake across the cell membrane, and glycogen synthase kinase (GSK), which negatively regulates glycogen synthase and the synthesis of glucose into glycogen [23, 24]. Therefore, foetal tissues in these chronically IUGR foetuses had adapted to their hypoglycaemic environment by developing mechanisms to promote glucose uptake and utilization, possibly via enhanced insulin action.

Other Changes in Foetal Metabolism in Response to Glucose (and Thus Insulin) Deficiency

A number of mechanisms have been noted, including glucose production from glycogen breakdown and the substitution of amino acids for glucose oxidation with acute glucose deprivation. With chronic glucose deprivation, however, foetal gluconeogenesis develops [21] and protein breakdown diminishes, allowing conservation of amino acids in protein structure while growth remains limited by the decrease in amino acid transfer to the fetus by the placenta [28, 7]. If such adaptations are linked to the decreased insulin signal transduction protein concentrations found in skeletal muscle and liver in IUGR fetuses, it is possible that chronic nutrient deprivation leading to growth restriction might portend limited capacity for protein growth in later life.

Conclusion

The fetus has considerable capacity to adapt metabolically to acute and chronic changes in glucose supply by relatively common and understandable mechanisms. Current review dwells on the molecular and physiological changes in glucose utilization capacity which could be reversed by selective re-introduction of nutrient (glucose or amino acids, or both) and hormonal (insulin or IGF-1 or both) supplies, directly by infusion into the fetus or indirectly by infusion into the mother.

References

- Aldoretta PW, Carver TD, Hay Jr WW. Ovine uteroplacental glucose and oxygen metabolism in relation to chronic changes in maternal and fetal glucose concentrations. Placenta. 1994;15:753–64.
- Aldoretta PW, Carver TD, Hay Jr WW. Maturation of glucose-stimulated insulin secretion in fetal sheep. Biol Neonate. 1998;73:375–86.
- Aldoretta PW, Hay Jr WW. Chronic hyperglycemia induces insulin resistance and glucose intolerance in fetal sheep. Pediatr Res. 2001;49:307A, abstract 1758.
- Anderson MS, Thamotharan M, Kao D, Devaskar SU, Qiao L, Friedman JE, Hay Jr WW. Effects of acute hyperinsulinemia on insulin signal transduction and glucose transporters in ovine fetal skeletal muscle. Am J Physiol Regul Integr Comp Physiol. 2005;288:R473–81.
- Carver TD, Anderson SM, Aldoretta PA, Esler AL, Hay Jr WW. Glucose suppression of insulin secretion in chronically hyperglycemic fetal sheep. Pediatr Res. 1995;38:754–62.
- Carver TD, Anderson SM, Aldoretta PW, Hay Jr WW. Effect of low-level basal plus marked 'pulsatile' hyperglycemia on insulin secretion in fetal sheep. Am J Physiol. 1996;271:E865–71.
- Carver TD, Quick AA, Teng CC, Pike AW, Fennessey PV, Hay Jr WW. Leucine metabolism in chronically hypoglycemic hypoinsulinemic growth-restricted fetal sheep. Am J Physiol. 1997;272:E107–17.

- Das UG, Schroeder RE, Hay Jr WW, Devaskar SU. Time-dependent and tissue-specific effects of circulating glucose on fetal ovine glucose transporters. Am J Physiol. 1999;276:R809–17.
- Davidson LS, Hay Jr WW. Effect of hyperinsulinemia on amino acid utilization independent of glucose metabolism in the ovine fetus. Pediatr Res. 2004;55: 382A (abstract).
- Digiacomo JE, Hay Jr WW. Fetal glucose metabolism and oxygen consumption during sustained hypoglycemia. Metabolism. 1990;39:193–202.
- Gresores A, Anderson S, Hood D, Zerbe GO, Hay Jr WW. Separate and joint effects of arginine and glucose on ovine fetal insulin secretion. Am J Physiol. 1997;272:E68–73.
- Hay Jr WW. Glucose metabolism in the fetal-placental unit. In: Cowett RM, editor. Principles of perinatal-neonatal metabolism. New York: Springer; 1998. p. 337–68.
- Hay Jr WW. Nutrition and development of the fetus: carbohydrate and lipid metabolism. In: Walker WA, Watkins JB, Duggan CP, editors. Nutrition in pediatrics (basic science and clinical applications). Ontario: BC Decker Inc; 2003. p. 449–70.
- Hay Jr WW, Anderson MS. Fuel homeostasis in the fetus and neonate. In: Jameson JL, Degroot LJ, editors. Endocrinology. Philadelphia: W.B. Saunders Company; 2005. p. 3387–406.
- Hay Jr WW, Digiacomo JE, Meznarich HK, Hirst K, Zerbe G. Effects of glucose and insulin on fetal glucose oxidation and oxygen consumption. Am J Physiol. 1989;256:E704–13.
- Limesand SW, Hay Jr WW. Adaptation of ovine fetal pancreatic insulin secretion to chronic hypoglycaemia and euglycaemic correction. J Physiol. 2003;547:95–105.
- Limesand SW, Jensen J, Hutton JC, Hay Jr WW. Diminished beta-cell replication contributes to reduced beta-cell mass in fetal sheep with intrauterine growth restriction. Am J Physiol Regul Integr Comp Physiol. 2005;288:R1297–305.
- Limesand SW, Rozance PJ, Zerbe GO, Hutton JC, Hay Jr WW. Attenuated insulin release and storage in fetal sheep pancreatic islets with intrauterine growth restriction. Endocrinology. 2006;147:1488–97.
- Limesand SW, Smith D, Rozance PJ, Hay Jr WW. Enhanced insulin sensitivity in fetal sheep with placental insufficiency and intrauterine growth restriction (IUGR). 3rd international congress of the developmental origins of health and disease, Toronto, Ontario, Canada, November 16–20, 2005. Pediatr Res. 2005;58:1053, abstract p. 1–049.
- McGowan JE, Aldoretta PW, Hay Jr WW. Contribution of fructose and lactate produced in placenta to calculation of fetal glucose oxidation rate. Am J Physiol. 1995;269:E834–9.
- Narkewicz MR, Carver TD, Hay Jr WW. Induction of cytosolic phosphoenolpyruvate carboxykinase in the ovine fetal liver by chronic fetal hypoglycemia and hypoinsulinemia. Pediatr Res. 1993;33:493–6.
- Nicolini U, Hubinont C, Santolaya J, Fisk NM, Rodeck CH. Effects of fetal intravenous glucose chal-

lenge in normal and growth retarded fetuses. Horm Metab Res. 1990;22:426–30.

- Rozance PJ, Limesand SW, Wyckoff MH, Hay Jr WW. Correlation between insulin secretion and glucose metabolism in ovine fetal pancratic islets. Pediatric Res. 2003;53:394A, abstract 2233.
- 24. Rozance PJ, Limesand SW, Wyckoff MH, Hay WW Jr. Diminished insulin secretion is not associated with changes in glucose utilization or oxidation in pancreatic islets obtained from hypoglycemic fetal sheep. Endocrine Societies 85th annual meeting, Philadelphia; 2003. Abstract p. 1–410.
- 25. Sadiq HF, deMello DE, Devaskar SU. The effect of intrauterine growth restriction upon fetal and postnatal hepatic glucose transporter and glucokinase proteins. Pediatr Res. 1998;43:91–100
- 26. Sadiq HF, Das UG, Tracy TF, Devaskar SU. Intrauterine growth restriction differentially regulates peri-

natal brain and skeletal muscle glucose transporters. Brain Res. 1999;823:96–103

- Van Assche FA, De Prins F, Aerts L, Verjans M. The endocrine pancreas in small-for-dates infants. Br J Obstet Gynaecol. 1977;84:751–753
- Van Veen LC, Teng C, Hay WW, Jr, Meschia G, Battaglia FC. Leucine disposal and oxidation rates in the fetal lamb. Metabolism. 1987;36:48–53
- 29. Wallace JM, Bourke DA, Aitken RP, Leitch N, Hay WW., Jr Blood flows and nutrient uptakes in growthrestricted pregnancies induced by overnourishing adolescent sheep. Am J Physiol Regul Integr Comp Physiol. 2002;282:R1027–R1036
- Wallace JM, Aitken RP, Milne JS. Nutritionallymediated placental growth restriction in the growing adolescent: consequences for the fetus. Biol Reprod. 2005 DOI 10.1095/biolreprod.104.030965

Growth and Maturation of the Human Fetal Endocrine System Up to Twenty Four Weeks of Gestation

23

Subhankar Chowdhury

The fetus is nourished by a continuous supply of carbohydrates, amino acids, essential fatty acids, calcium, and oxygen in a warm environment, protected from outside influences by the fluid surrounding the fetus (amniotic fluid) which acts as a hydrostatic brace. However, at the time of birth, the fetus must have grown and developed, and its physiological systems must have sufficiently matured to survive a given level of parental care. In terms of fetal development, "maturation" refers to changes that facilitate survival following the transition from intra- to extrauterine existence. The human fetus must undergo endocrine maturation to generate hormones to support normal development and whose abnormal development/function can impact various systems with effects ranging from functional or structural defects to even death. A bouquet of transition factors and epigenetic events act hand in hand with autocrine, paracrine, and endocrine network of hormones and growth factors leading to the evolution of the fetal endocrine system with the hypothalamus-pituitary system providing the pivotal controlling template. Endocrine maturation commences early in gestation and continues till term and even thereafter to prepare

e-mail: subhankar.chowdhury@gmail.com

the fetus for the external world. Most of the evidence available in this regard comes from animal studies with actual human data being available only in certain steps of this process. This chapter will look at the endocrine maturation of the human fetus up to 20 weeks of gestation and will also dwell upon its aberrations and subsequent implications.

Pituitary Gland

The anterior and intermediate lobes of the pituitary gland are derived from oral ectoderm, whereas the posterior pituitary is derived from neural ectoderm [1].

Anterior Pituitary

Anterior pituitary develops in four distinct stages beginning at around 4–6 weeks of gestation [2, 3].

- Stage 1 (Pituitary placode). The anterior neural ridge is displaced ventrally to form the oral epithelium which forms the roof of the oral cavity. Onset of pituitary organogenesis coincides with a thickening (pituitary placode) in the roof of the oral ectoderm.
- <u>Stage 2 (Rudimentary Rathke's pouch).</u> Invagination of the oral ectoderm forms a rudimentary pouch, and evagination of the

S. Chowdhury, DTM&H, MD, DM, MRCP, FRCP (⊠) Department of Endocrinology and Metabolism, Institute of Post Graduate Medical Education and Research and SSKM Hospital, Kolkata, West Bengal 700020, India

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_23

ventral diencephalon forms the posterior pituitary. The pituitary placode makes contact with the floor of the ventral diencephalon. Apposition between the rudimentary Rathke's pouch and neural ectoderm of the diencephalon is critical to normal development of the anterior pituitary.

- Stage 3 (Definitive Rathke's pouch). The rudimentary Rathke's pouch deepens and folds on itself until it closes and forms a definitive pouch (incomplete closure may persist as Rathke's cleft). The pituitary stalk is formed by evagination of the posterior part of the presumptive diencephalon.
- Stage 4 (Adult gland). The Rathke's pouch is completely separated from the oral cavity. The final formation of a complicated secretory organ containing five different cell types occurs under the influence of several signaling and transcription factors over space and time. Remnants of pituitary tissue may persist in the nasopharyngeal midline and may rarely give rise to functional ectopic hormone-secreting tumors in the nasopharynx.

Critical neuroectodermal signals for initiating dorsal pituitary morphogenesis include infundibular bone morphogenetic protein 4 (BMP4), which is required for initial pouch invagination and also fibroblast growth factor 8 (FGF8), Wnt5 and Wnt4 [4, 5]. BMP4 and FGF8 are present only in the diencephalon which underlines the importance of apposition of the rudimentary Rathke's pouch and diencephalic neural ectoderm [6]. Subsequent ventral development is determined by expression of factors like BMP2 and sonic hedgehog protein (Shh). Early cell differentiation requires intracellular Rpx (HESX1) and Ptx (PITX) expression. Rathke's pouch expresses transcription factors like Lhx3, Lhx4, and IsI-1 which are required for progenitor cell survival and proliferation. Ptx1 is expressed in the oral ectoderm, and subsequently in all pituitary cell types, particularly those arising ventrally while Ptx2 mutation causes Rieger's syndrome, characterized by defective eye, tooth, umbilical cord, and pituitary development. Ptx is an universal pituitary regulator and activates transcription of α GSU, POMC, LH β (Ptx1), and GH (Ptx2). Lhx3 determines GH-, PRL-, and TSH-cell differentiation. PROP1 acts as a prerequisite for POU1F1, which activates GH, PRL, TSH, and growth hormone- releasing hormone (GHRH) receptor transcription. TSH and gonadotropinexpressing cells share α GSU expression under developmental control of GATA2 and FOXL2. Pituitary cell type diversification is mediated by binary Wnt/β-catenin signaling which causes induction of PROP1 and suppression of HESX1. PROP1 expression is particularly important as it development determines subsequent of POU1F1-dependent (GH, PRL, TSH, and GHRH receptor) and gonadotroph cell lineages [7]. Signal-dependent coactivating factors also cooperate with POU1F1 to determine specific hormone expression (Estrogen induces PRL TSH while TEF induces expression). Steroidogenic factor (SF1) and DAX1 (encoded by NROB1) determine subsequent gonadotroph development [8, 9]. Corticotroph cell commitment occurs earliest during fetal development and requires TPIT protein (encoded by TBX19) for POMC expression. Table 23.1 gives an outline of the syndromes arising from mutation in these transcription factors [6].

Corticotroph cells are morphologically identifiable at 6 weeks and immunoreactive ACTH is detectable by 7 weeks. Somatotroph cells are subsequently evident with abundant immunoreactive cytoplasmic GH expression at 8 weeks. Nine- to sixteen-week-old human fetal pituitary cells in culture show a predominant response to GHRH and a limited effect of somatostatin, suggesting that the inhibitory action of somatostatin develops later in gestation [10]. At 12 weeks, differentiated thyrotrophs and gonadotrophs express TSH, LH and FSH, respectively. Immunoreactive PRL is detectable in mixed mammosomatotrophs along with GH in the early half of gestation and pituitary PRL content increases progressively from 12 to 15 weeks. PRL receptors are present in most fetal tissues during first trimester and are implicated in fetal

Gene	Phenotype		
HESX1	Variable: septo-optic dysplasia, combined pituitary hormone deficiency, isolated growth hormone deficiency (IGHD) with ectopic posterior pituitary (EPP)		
	Anterior pituitary hypoplastic or absent		
	Posterior pituitary ectopic or eutopic		
	Frequency of mutations: <1 %		
OTX2	Anophthalmia, anterior pituitary hypoplasia (APH), EPP, absent infundibulum		
	Frequency of mutations: 2–3 % of anophthalmia/microphthalmia cases		
SOX2	Hypogonadotropic hypogonadism; APH, abnormal hippocampi, bilateral anophthalmia/ microphthalmia, abnormal corpus callosum, learning difficulties, esophageal atresia, sensorineural hearing loss, hypothalamic hamartoma		
	Frequency of mutations: 3.4 %		
SOX3	IGHD and mental retardation, hypopituitarism; APH, infundibular hypoplasia, EPP, midline abnormalities		
	Frequency of mutations: 6 % (duplications), 1.5 % (mutations)		
GLI2	Holoprosencephaly, hypopituitarism, craniofacial abnormalities, polydactyly, single nares, single central incisor, partial agenesis of corpus callosum		
	Frequency of mutations: 1.5 %		
LHX3	GH, TSH, gonadotropin deficiency with pituitary hypoplasia		
	ACTH insufficiency variable		
	Short, rigid cervical spine		
	Variable sensorineural hearing loss		
	Frequency of mutations: 1.3 %		
LHX4	GH, TSH, cortisol deficiency, persistent craniopharyngeal canal and abnormal cerebellar tonsils; APH, ectopic/eutopic posterior pituitary, absent infundibulum		
	Frequency of mutations: 1.2 %		
PROP1	GH, TSH, PRL, and gonadotropin deficiency		
	Evolving ACTH deficiency		
	Enlarged pituitary with later involution		
	Frequency of mutations: 1.1 % sporadic cases, 29.5 % familial cases		
POU1F1	Variable anterior pituitary hypoplasia with GH, TSH, and PRL deficiencies		
	Frequency of mutations: 3.8 % sporadic cases, 18 % familial cases		

 Table 23.1
 Genetic mutations and alterations in pituitary development

From Dattani et al. [6]. With permission from Elsevier

growth, skeletal maturation and adipose tissue maturation [11, 12].

The hypothalamic cell condensations which are the precursors of the hypothalamic nuclei, and the interconnecting fiber tracts are demonstrable by 15–18 weeks of gestation [3]. Concentrations of dopamine, TRH, GnRH, and somatostatin are significant in hypothalamic tissue by 10–14 weeks of gestation. Capillaries develop within the proliferating anterior pituitary mesenchymal tissue around Rathke's pouch by 8 weeks of gestation, and intact hypothalamic-pituitary portal vessels are present by 12–17 weeks but the maturation process extends almost till end gestation.

Intermediate Lobe of the Pituitary

The intermediate lobe is prominent in early gestation and is virtually absent in the adult human pituitary [13]. The major secretory products of the intermediate lobe are α -MSH and β -endorphin but the role of these intermediate lobe peptides in the fetus remain obscure [14].

Posterior Pituitary

The fetal neurohypophysis is well developed by 10–12 weeks of gestation and contains both arginine vasopressin (AVP) and oxytocin (OT) [15]. In addition, arginine vasotocin (AVT) is present in the fetal pituitary and pineal gland but its actual role is unknown [16]. AVP functions in the fetus mainly as a stress-responsive hormone responding to stress like hypoxia while AVP response to osmolar stimuli and its antidiuretic effects are prominent during the last trimester of gestation [15].

Adrenal System

Understanding adrenal embryogenesis is mostly derived from studies in transgenic mice and humans with various forms of adrenal hypoplasia. The primordia of the adrenal glands can be recognized just above the bilaterally developing mesonephros by 3–4 weeks of gestation [17, 18]. The adrenal cortex is derived from a thickening of the intermediate mesoderm at 4-5 weeks of gestation while the adrenal medulla originates from the ectoderm. This region is known as the gonadal ridge and gives rise to the steroidogenic cells of both the adrenal gland and the gonads. Early differentiation of the adrenogonadal primordium from the urogenital ridge requires signaling cascades and transcription factors GLI3, SALL1, FOXD2, WT1, PBX1, WNT4, and the regulator of telomerase activity, ACD. Separation of the adrenogonadal primordium and formation of the adrenal primordium depend on the actions of transcription factors SF1, DAX1, WNT4, and CITED2. Among these, SF1 and DAX1 play a particularly important role in early adrenal development [19]. DAX1 probably acts as a repressor of SF1 transcription and regulates adrenal progenitor cell development and maturation. Adrenal tissue precursors migrate retroperitoneally to the upper pole of the mesonephros and are infiltrated at 7-8 weeks of gestation by sympathetic cells derived from the neural crest that subsequently form the adrenal medulla. The adrenocortical primordium develops at approximately 8 weeks of gestation and can differentiates into two distinct layers, the inner fetal zone (FZ) and the outer definitive zone (DZ) which depends on the temporal expression of transcription factors including Pref-1/ZOG, inner zone antigen, and SF1 [20]. Encapsulation of the adrenal gland occurs after 8 weeks of gestation and results in the formation of a distinct organ just above the developing kidney.

The fetal adrenal has three functional zones – an outer definitive zone capable of producing both glucocorticoids and mineralocorticoids, a transitional zone with enzymes for cortisol production, and a much larger inner fetal zone capable of producing significant amounts of C19 androgens such as dehydroepiandrosterone (DHEA) and dehydroepiandrosterone sulphate (DHEAS), which are then converted to estrogens by the placenta. The large eosinophilic cells of the fetal zone are well differentiated by 9-12 weeks of gestation and are capable of active steroidogenesis. Quantitative differences in the relative activities of the five steroidogenic apoenzymes are found between cells derived from the fetal versus the definitive zones due to regulated steroidogenic gene transcription. The fetal zone has relatively high steroid sulfotransferase activity which together with low 3β-HSD results in the predominant production of DHEA, DHEAS, pregnenolone sulfate, several $\Delta 5 \ 3\beta$ hydroxysteroids, while only limited amounts of $\Delta 5$ 3-ketosteroids, including cortisol and aldosterone are produced by the fetal adrenals [17, 18]. This programming is designed to provide DHEA substrate for placental estrone and estradiol production. The fetal zone contains more LDL binding sites and manifests a greater rate of de novo cholesterol synthesis than does the definitive zone, in keeping with its greater steroidogenic activity. Glucocorticoids are synthesized in the first trimester due to transient expression of type 2 3β-hydroxysteroid dehydrogenase (HSD3B2) using pregnenolone as the precursor instead of denovo cholesterol synthesis [21]. Fetal pituitary ACTH stimulates steroid production by activating StAR and increasing delivery of substrate cholesterol to P450scc and is the major stimulus to adrenal function as evidenced by the reduction

of placental lactogen production by 90 % and the involution of the adrenal gland after 15 weeks in the anencephalic fetus [17, 21]. Angiotensin II inhibits 3β -HSD activity and promotes DHEA production in the fetal zone in early gestation.

Glucocorticoid receptors (GR) are present at midgestation in most tissues, including placenta, lung, brain, liver, and gut and play an important role in fetal development. Mice lacking GR function manifest enlarged and disorganized adrenal cortices, adrenal medullary atrophy, lung hypoplasia, and defective gluconeogenesis [22]. Fetal cortisol is converted to inactive cortisone through 11BHSD2 in fetal tissues and this metabolism protects the anabolic milieu of the fetus, as cortisol can retard both placental and fetal growth [23]. Mineralocorticoid receptors (MR) are present from 12 to 16 weeks in fetal kidney, skin, hair follicles, trachea and bronchioles, esophagus, colon, small intestine, and pancreatic exocrine ducts but their role remains unclear with aldosterone secretion remaining low in midgestation and being unresponsive to secretagogues [24].

Adrenal insufficiency may be secondary to ACTH deficiency or may be due to primary adrenal failure. ACTH deficiency may be isolated (TPIT mutation) or a part of multiple pituitary hormone deficiency (discussed earlier in Table 23.1). Recessive mutations in the T-box factor TPIT (TBX19) have been identified in patients with severe, early-onset isolated ACTH deficiency with profound hypoglycemia, prolonged jaundice, and sudden neonatal death [25]. TPIT is required for the specification, maturation, and maintenance of both precorticotrope and premelanotrope populations and for the suppression of gonadotrophs. It is also required to activate the expression of POMC in conjunction with the transcription factor PTX1. SF1 gene mutation causes adrenal and gonadal agenesis, gonadotropin deficiency, and absence of the hypothalamic ventromedial nucleus [26]. Mutations in SF1 have been associated with gonadal dysgenesis in 46XY individuals without any adrenal involvement and have also been associated with primary ovarian failure.

Inactivating DAX1 gene mutations are associated with adrenal hypoplasia and gonadotropin deficiency [26]. In humans, inactivating *StAR* mutations cause adrenal hypoplasia and adrenal hormone insufficiency, and normal to varying degree of ambiguous genitalia in XY males [27]. Various forms of congenital adrenal hyperplasia may be associated with variable adrenal failure (e.g., mutations in CYP11A1, StAR, HSD3B2I, CYP17, CYP21, CYP11B1) with varying degrees of genital ambiguity [28, 29].

Autonomic Nervous System

The paired adrenal glands are well developed by 10-12 weeks of gestation. There is progressive growth and maturation of the adrenal medulla with increasing gestational age and histological maturation is complete only at 1 year of postnatal life [6]. In the adrenal glands, the progenitor cells derived from common neuroectodermal stem cells differentiate into neuroendocrine cells, expressing tyrosine hydroxylase and dopamine β -hydroxylase in response to a series of transcription factors including PHOX2B, MASH1, PHOX2A, and dHAND [30, 31]. The PHOX2B gene plays a central role in the development of most relays of the autonomic nervous system and its mutation has been associated with the congenital hypoventilation syndrome, Hirschsprung's disease, and a predisposition to neuroblastoma [32]. Catecholamines are present in the chromaffin tissue by 10-15 weeks of gestation and the adrenal medulla can respond directly to asphyxia, long before splanchnic innervation develops, by secreting norepinephrine. Catecholamines are critical for fetal cardiovascular function and fetal survival and its absence leads to midgestation fetal death in the majority [33]. Catecholamines are also the major stress hormone in the fetus and is discharged in large amounts into the circulation in response to fetal hypoxia [34]. Also, the defense against fetal hypoxia involves action of catecholamine on the cardiac α -receptors which predominate early in gestation and gradually decline in number as β -adrenergic receptors increase.

Thyroid

The pharyngeal floor thickens in the midline (median anlage) and acts as the precursor of the T4-producing follicular cells while paired caudal extensions of the fourth pharyngobranchial pouches (lateral anlagen) give rise to the parafollicular calcitonin-secreting cells (C cells) [35, 36]. These structures are identifiable by 16-17 days of gestation, and by 24 days the median anlage develops a thin, flask-like diverticulum extending from the buccal floor to the fourth branchial arch. The primitive structure connecting the primordium to the pharyngeal floor elongates into the thyroglossal duct. At 24-32 days, this median anlage is a bilobed structure, and by 50 days of gestation, the median and lateral anlagen fuse with each other. Dissolution and fragmentation of the thyroglossal duct usually occurs by 37-50 days and a small dimple is left at the point of origin of the thyroglossal duct at the junction of the middle and posterior thirds of the tongue which is known as the foramen caecum. Cells of the lower portion of the thyroglossal duct differentiate into thyroid tissue, forming the pyramidal lobe of the gland [37]. Meanwhile, the thyroid gland migrates caudally from the pharyngeal floor, through the anterior midline of the neck, to its definitive location in the anterior neck. At 51 days, the gland exhibits its definitive external form, with an isthmus connecting the two lateral lobes, and it reaches its final position below the thyroid cartilage by the seventh week of embryonic life. An ectopic thyroid and persistent thyroglossal duct or cyst may occur as a consequence of abnormalities of thyroid descent. Rarely, thyroid tissue may develop from remnants of the thyroglossal duct near the base of the tongue forming a lingual thyroid tissue which may be the sole functioning thyroid present [37]. This has clinical importance as its inadvertent removal may lead to iatrogenic hypothyroidism.

A Tbx1-Fgf8 pathway in the pharyngeal mesoderm plays a key role in early thyroid migration and development. Several developmental genes are involved in thyroid and para-

thyroid gland embryogenesis. These include the genes for thyroid transcription factors HEX, TTF1 (Titf1/Nkx2.1), FOXE1 (Titf2/Foxe1), NKX2-5, and PAX8 [38–40]. HEX gene mutation is associated with thyroid agenesis or severe hypoplasia. TTF2 mutation results in thyroid dysgenesis and cleft palate while TTF1 defect leads to pulmonary hypoplasia, parafollicular C-cell aplasia and thyroid agenesis. Inactivating Pax8 mutations lead to thyroid hypoplasia and renal anomalies. TTF1/NKX2.1 and PAX8 also play a role in the survival of thyroid cell precursors and regulation of thyroid specific gene expression, while FOXE1/TTF2 is essential for cellular migration. The SHH gene may have an important role in later stages of thyroidogenesis in the symmetric bilobation of the gland and it also suppresses the ectopic expression of thyroid follicular cells. However, mutations in these genes account for fewer than 10 % of patients with familial thyroid dysgenesis and congenital hypothyroidism and most cases occur sporadically and are of unknown aetiology.

Thyroglobulin (Tg) starts forming in the future follicular cells from as early as the 29th day of gestation, while the capacities to concentrate iodide and synthesize thyroxine (T4) are delayed until about the 11th week [41]. Radioactive iodine given inadvertently to the mother before this period is not actually harmful while later administration may ablate the fetal thyroid. Early growth and development of the thyroid is not dependent on TSH as pituitary TSH secretion is not apparent until the 14th week. Thyroxine-binding globulin (TBG), the major thyroid hormone-binding protein in plasma, is detectable by the tenth gestational week and increases in concentration progressively till term. The terminal differentiation of thyroid follicular cells as evidenced by expression of the genes encoding the TSH receptor (TSHR), the sodium-iodide symporter (NIS), thyroglobulin (Tg), and thyroperoxidase (TPO) and the formation of follicles - occurs in the normal embryo only after migration is complete [42]. Genes encoding Tg, TPO, and pendrin are expressed as early as 7 weeks' gestational age.

NIS expression appears last from 11th gestational week onwards suggesting a key role of NIS in the onset of thyroid function while Tg is detectable in unpolarized thyrocytes before follicle formation [43].

Classic thyroid hormone actions are mediated via functional thyroid hormone nuclear receptors- TR α 1, TR α 2, TR β 1, and TR β 2. TR α 1 and TR β 1 isoforms and receptor binding are present by 8–10 weeks of gestation in the fetal brain with TR α transcription and receptor occupancy increasing eight- to tenfold by 16–18 weeks. Liver, heart, and lung receptor binding can be identified by 13–18 weeks of gestation [36].

In early gestation, placental transfer of T4 is the only source of thyroid hormone and is critical for fetal neurodevelopment playing an important role in neurogenesis and neural cell migration, neuronal differentiation, dendritic and axonal growth, synaptogenesis, myelination, and neurotransmitter enzyme synthesis [44]. Most of the thyroid hormone in the fetal compartment is inactivated to sulfated and deiodinated analogues until the perinatal period to maintain a low T3 metabolic state, facilitating fetal growth and programmed tissue maturation [45]. In congenital hypothyroidism, cord T4 concentrations are 25-50 % of normal due to an increased net transplacental flux of maternal thyroid hormone. The transplacental passage of thyroid hormone along with alteration in brain deiodinase activity plays a crucial role in minimizing the adverse effects of fetal hypothyroidism and helps explain the near-normal outcome of hypothyroid fetuses (with adequate postnatal levothyroxine supplementation). On the other hand, in the presence of both maternal and fetal hypothyroidism (due to potent TSH receptorblocking antibodies, maternal and fetal POU1F1 deficiency, and severe iodine deficiency), there is severe neurocognitive impairment despite early and adequate commencement of thyroid replacement. Significantly, the presence of maternal hypothyroxinemia or inadequately controlled hypothyroidism results in significant and irreversible neurocognitive deficit in the offspring [46, 47].

Parathyroid Gland

Inferior and Superior parathyroid glands develop from the third and fourth pharyngeal pouches respectively in a co-ordinated fashion with thyroid gland formation [35]. The parathyroid anlage of the third pouch is carried caudally with the migrating thyroid anlage and ends up at the lower poles of the thyroid lobes as the inferior parathyroid glands. The fourth pouch encounters the thyroid anlage later and comes to rest at the upper poles of the thyroid lobes as the superior parathyroid glands. HOX15 plays a pivotal role in the gene cascade programming normal thyroid-parathyroid gland development and its silencing results in parathyroid gland aplasia. Genes like SOX3, GCM2, GATA3, CRKL, and TBX1 are also involved in the development of the parathyroid gland [47, 48]. CRKL and TBX1 mutations are associated with the DiGeorge syndrome and GATA3 mutation gives rise to a DiGeorge-like syndrome while GCM2 mutation leads to isolated hypoparathyroidism. Transplacental calcium transfer maintains high fetal calcium levels and occurs across the syncytiotrophoblast, which contains a calcium-binding protein that buffers intracellular calcium ions as they are transported across the syncytial cell to the basement membrane [49]. A placental ATP-dependent calcium pump transports the calcium to the fetal circulation and is stimulated by a midmolecule portion of PTHrP (via an unidentified receptor) secreted by the fetal parathyroid gland and by the placenta, via a paracrine effect [50]. Absence of fetal parathyroids due to any cause results in a low fetal plasma calcium concentration and a loss of the placental calcium gradient. PTH and PTHrP, through the PTH/PTHrP receptor, probably modulates fetal skeletal calcium flux, calcium excretion through the fetal kidney, stimulates fetal renal 1,25(OH)2D production and reabsorption of calcium from amniotic fluid. PTHrP has a major role in fetal bone development and metabolism as well as fetal calcium homeostasis and its absence results in neonatal death due to severe and widespread bony defects. Chronic stimulation by fetal hypercalcemia results in high blood levels of calcitonin which inhibits bone resorption and promotes bone mineral anabolism [51].

Gonadal System

Two anlagens give rise to the gonads - the primordial germ cells of the yolk sac wall and the somatic, stromal cells that migrate from the primitive mesonephros [52, 53]. Germ cells begin their migration from the yolk sac by 4-5 weeks of gestation when the gonadal ridge appears as a derivative of the mesonephros. The germ cells get included in the developing gonadal ridge during the sixth week. The primitive gonad is composed of a surface epithelium, primitive gonadal cords continuous with the epithelium, and a dense cellular mass referred to as the gonadal blastema. Primordial cells that fail to migrate normally may account for the location of extragonadal germ cell cancers in men. The fetal testis and ovary are indistinguishable until 6 weeks when the testicular cords made up of Sertoli cells appear under the influence of the Y chromosome. The SRY (sex-determining region of the Y chromosome) gene on the pseudoautosomal region of the Y chromosome encodes a transcription factor that increases the expression of SRY-box 9 (SOX9), which directs formation of Sertoli cells and testis differentiation [54]. The primitive cords lose their connections with the epithelium, primitive Sertoli cells and spermatogonia become visible within the cords, and the epithelium differentiates to form the tunica albuginea. SRY gene expression is activated by steroidogenic factor 1 (SF1) and the binding protein GATA4 [55]. SF1 also results in SRY-independent SOX9 expression, AMH gene expression and gonadotropin production. SOX9 drives expression of other genes like fibroblast growth factor 9 (FGF9) and AMH that are necessary for differentiation of testis and genes like WNT4 and NROB1 which repress ovary formation. WT1 is expressed in the genital ridge and its expression mediates the transition of mesenchyme to epithelium. In XX individuals, SOX9 is repressed by factors like β -catenin leading to the development of the follicular cells and ovaries. Leydig cells begin to form at 8 weeks of gestation from the undifferentiated interstitium and make up about 50 % of the cell mass at 14 weeks [56]. The differentiation, proliferation, and organization of the gonads

occur initially under the influence of maternal human chorionic gonadotropin (hCG) and later LH and FSH from the fetal pituitary gland. Testosterone production from fetal Leydig cells increases progressively between 10 and 20 weeks and induces development of the male internal genitalia while conversion of testosterone to 5α -dihydrotestosterone (5α -DHT) in the urogenital tract leads to the formation of the prostate and male external genitalia [57]. Androgen receptors appear in the mesenchyme of urogenital structures at 8 weeks of gestation, followed by appearance of the receptors in the epithelium during development at 9-12 weeks. Male phenotypic development is complete by around 15 weeks of gestation. The developing testis is attached to the diaphragm by the craniosuspensory ligament and anchored to the inguinal region by a caudal ligament known as the gubernaculum. The initial transabdominal phase of testicular descent occurs between 10 and 23 weeks of gestation and depends on two processes: (1) testosterone induced regression of the craniosuspensory ligament (2) thickening of the gubernaculum, which is controlled by INSL3 produced by Leydig cells and its cognate receptor, relaxin family peptide receptor 2 (RXFP2, also known as leucine-rich repeat-containing G protein-coupled receptor 8 [LGR8] or G protein-coupled receptor affecting testis descent [GREAT]) [58]. AMH secretion from the fetal Sertoli cells causes regression of the Müllerian ducts preventing the formation of uterus and fallopian tubes while female genitalia develop in the absence of testosterone. AMH gene expression is activated by the SRY and SF1 genes and also has autocrine and paracrine effects on fetal testicular steroidogenic function. AMH mutation results in persistent Mullerian duct syndrome in the XY fetus [59].

Loss-of function mutations of SRY or SOX9 produce XY sex reversal, whereas gain-offunction mutations or duplications produce XX sex reversal [53]. SOX9 mutations may also cause camptomelic dysplasia. SF1 mutations may lead to 46XY sex reversal with gonadal dysgenesis, with or without adrenal failure and may also result in vanishing testes syndrome. WT1 mutations produce several syndromes associated with abnormal testicular embryogenesis (WAGR syndrome [Wilms' tumor, aniridia, genitourinary anomalies, and mental retardation], Denys-Drash syndrome, and Frasier syndrome) and renal abnormalities such as Wilms' tumor or glomerulosclerosis. WNT4 gain-of-function mutations result in XX sex reversal, whereas NROB1 duplication is associated with XY sex reversal.

In females, in the absence of SRY, the gonadal blastema differentiates into interstitium and medullary cords containing the primitive germ cells (oogonia). By 11-12 weeks of gestation, clusters of dividing oogonia are surrounded by cord cells within the cortex while the medulla at this time consists largely of connective tissue [60]. At 12 weeks of gestation, primitive granulosa cells begin to replicate and many of the large oogonia in the deepest layers of the cortex enter their first meiotic division. Primordial follicles are observed initially at about 18 weeks and the number increases rapidly thereafter although the number of oocytes progressively declines (from a peak of three to six million at 5 months' gestation) [61]. Interstitial cells capable of steroid production are present after 12 weeks although steroids are practically not produced by the developing ovary [21]. RSPO1 mutation may cause 46XX sex reversal as it belongs to the R-spondin family of proteins that is implicated in Wnt/β-catenin signaling, and the gene encoding the protein may determine ovary formation [62]. Estrogen receptors have been characterized in the 16- to 23-week human fetus. The ER β message is commonly present, particularly in testis, ovary, spleen, thymus, adrenal, brain, kidney, and skin. The ER α message is prominent in the uterus, with relatively low levels in most other tissues. However, the clinical significance of ERs remains uncertain as ER knockout laboratory animals are normal at birth [63].

Endocrine Pancreas

Pancreatic budding occurs from the gut tube and differentiation of exocrine and endocrine cell lineages occur from endodermal tissue under the influence of a number of genes and transcription

factors [6, 64]. SOX9 and HNF3B are required for early foregut formation and pancreas specification while PDX1, HLXB9, and ISL1 play a role early in pancreatic development. HES1 and neurogenin 3 (NGN3) take part in development of the islets of Langerhans. Pancreatic agenesis may result from PDX1 mutation causing neonatal diabetes mellitus. The human fetal pancreas is identifiable by 4 weeks of gestation, and alpha and beta cells can be recognized by 8–9 weeks. Insulin, glucagon, somatostatin, and pancreatic polypeptide are measurable by 8-10 weeks of gestation [65]. Alpha cells outnumber the beta cells in the early fetal pancreas and reach peak at midgestation, while beta cells increase in the latter half of gestation and by 20 weeks, endocrine cells are dispersed throughout the exocrine tissues. The fetal beta cell begins to function from around the 14th week but to a very limited extent such that the secretion of insulin into the bloodstream by the fetal pancreas is low. Development of the pancreas is independent of the pituitary as the endocrine pancreas develops normally in an encephalic fetuses; however, beta cell hyperplasia and hypertrophy to hyperglycemia exposure is lacking which may be due to deficiency of GH-IGF-1, which plays a permissive role otherwise [66]. Pancreatic glucagon content is high in midgestation but its secretion is blunted in the fetus. The blunted capacity for fetal insulin and glucagon secretion may be on account of the relatively stable fetal serum glucose concentrations maintained by transplacental maternal glucose transfer, lack of enteric incretin signal to the pancreas from feeding and deficient cAMP in the fetal pancreatic islet due to deficient production or rapid destruction. In early gestation, hepatic metabolism and substrate utilization are insulin-independent and modulated in an autoregulatory fashion by glucose, while glycogen storage in the fetus is modulated by fetal glucocorticoids and probably by placental lactogen [67].

The above knowledge and ongoing research has set the stage for fetal endocrine disease diagnosis, therapy for fetal endocrine and metabolic disorders and diagnosis and management of neonatal endocrine dysfunction. Moreover, management strategies for premature infants and infants and children with fetal growth restriction can be better devised with an understanding of endocrine maturation. We are on the doorstep of an era of direct management of the intrauterine environment both medically and surgically for which an elaborate understanding of fetal development is of prime importance. Intrauterine diagnosis and treatment of fetal adrenal and thyroid disorders are already in vogue and increasing availability of synthetic hormones and growth factor agonists and antagonists facilitate direct fetal endocrine therapy [68, 69]. In early gestation, the fetus is a favorable recipient of stem cells and this technology may be applicable to therapy for selected endocrine and metabolic diseases [70]. Cutting edge research has increased our experience with fetal gene therapy in animals and in the near future, similar therapy may be a reality in humans too.

References

- Kelberman D, Rizzoti K, Lovell-Badge R, et al. Genetic regulation of pituitary gland development in human and mouse. Endocr Rev. 2009;30:790–829.
- Dasen JS, Rosenfeld MG. Signaling and transcriptional mechanisms in pituitary development. Annu Rev Neurosci. 2001;24:327–55.
- Grumbach MM, Gluckman PD. The human fetal hypothalamus and pituitary gland: the maturation of neuroendocrine mechanisms controlling secretion of fetal pituitary growth hormone, prolactin, gonadotropins, adrenocorticotropin-related peptides, and thyrotropin. In: Tulchinsky D, Little AB, editors. Maternal fetal endocrinology. 2nd ed. Philadelphia: WB Saunders; 1994. p. 193–261.
- Scully KM, Rosenfeld MG. Pituitary development: regulatory codes in mammalian organogenesis. Science. 2002;295:2231–5.
- Zhu X, Wang J, Ju BG, et al. Signaling and epigenetic regulation of pituitary development. Curr Opin Cell Biol. 2007;19:605–11.
- Dattani MT, Hindmarsh PC, Fisher DA. Endocrinology of fetal development. In: Melmed S, Polonsky KS, Larsen PR, Krokenberg HM, editors. Williams textbook of endocrinology. Philadelphia: Elsevier Saunders; 2011. p. 833–67.
- Pulichino A-M, Vallette-Kasic S, Drouin J. Transcriptional regulation of pituitary gland development: binary choices for cell differentiation. Curr Opin Endocrinol Metab. 2004;11:13–7.

- Muscatelli F, Strom TM, Walker AP, et al. Mutations in the DAX-1 gene give rise to both X-linked adrenal hypoplasia congenita and hypogonadotropic hypogonadism. Nature. 1994;372:672–6.
- Tabarin A, Achermann JC, Recan D, et al. A novel mutation in DAX1 causes delayed-onset adrenal insufficiency and incomplete hypogonadotropic hypogonadism. J Clin Invest. 2000;105:321–8.
- Goodyear CG, Sellen JM, Fuks M, et al. Regulation of growth hormone secretion from human fetal pituitaries, interactions between growth hormone releasing factor and somatostatin. Reprod Nutr Dev. 1987;27:461–70.
- 11. Clément-Lacroix P, Ormandy C, Lepescheux L, Ammann P, Damotte D, Goffin V, et al. Osteoblasts are a new target for prolactin: analysis of bone formation in prolactin receptor knockout mice. Endocrinology. 1999;140:96–105.
- Gluckman PD, Pinal CS. Growth hormone and prolactin. In: Polin RA, Fox WW, Abman SH, editors. Fetal and neonatal physiology. 3rd ed. Philadelphia: WB Saunders; 2004. p. 1891–5.
- Visser M, Swaab DF. Life span changes in the presence of alpha melanocyte- stimulating-hormonecontaining cells in the human pituitary. J Dev Physiol. 1979;1:161–78.
- Facchinetti F, Storchi AR, Petraglia F, et al. Ontogeny of pituitary β-endorphin and related peptides in the human embryo and fetus. Am J Obstet Gynecol. 1987;156:735–9.
- Leake RD. The fetal-maternal neurohypophysial system. In: Polin RA, Fox WW, editors. Fetal and neonatal physiology. 2nd ed. Philadelphia: WB Saunders; 1998. p. 2442–6.
- Ervin MG, Leake RD, Ross MG, et al. Arginine vasotocin in ovine maternal and fetal blood, fetal urine, and amniotic fluid. J Clin Invest. 1985;75:1696–701.
- Winter JSD. Fetal and neonatal adrenocortical physiology. In: Polin RA, Fox WW, Abman SH, editors. Fetal and neonatal physiology. Philadelphia: WB Saunders; 2004. p. 1915–25.
- Geller DH, Miller WL. Molecular development of the adrenal gland. In: Pescovitz OH, Eugster EA, editors. Pediatric endocrinology. Philadelphia: Lippincott Williams & Wilkins; 2004. p. 548–67.
- Ikeda Y, Swain A, Weber TH, et al. Steroidogenic factor 1 and DAX-1 localize in multiple cell lineages: potential links in endocrine development. Mol Endocrinol. 1996;10:1261–72.
- Okamoto M, Takemori H. Differentiation and zonation of the adrenal cortex. Curr Opin Endocrinol Diabetes. 2000;7:122–7.
- Mesiano S, Jaffe RB. Neuroendocrine-metabolic regulation of pregnancy. In: Strauss III JF, Barbieri RL, editors. Reproductive endocrinology. 5th ed. Philadelphia: WB Saunders; 2004. p. 327–66.
- McKenna NT, Moore DD. Nuclear receptors: structure, function and cofactors. In: DeGroot LJ, Jameson JL, editors. Endocrinology. 5th ed. Philadelphia: Elsevier Saunders; 2006. p. 277–87.

- Johnson JW, Mitzner W, Beck JC, et al. Long term effects of betamethasone in fetal development. Am J Obstet Gynecol. 1981;141:1053–64.
- 24. Hirasawa G, Sasano H, Suzuki T, et al. 11β-hydroxysteroid dehydrogenase type 2 and mineralocorticoid receptor in human fetal development. J Clin Endocrinol Metab. 1999;84:1453–8.
- Pulichino AM, Vallette-Kasic S, Couture C, et al. Human and mouse TPIT gene mutations cause early onset pituitary ACTH deficiency. Genes Dev. 2003;17:711–6.
- Hammer GD, Parker KL, Schimmer BP. Minireview: transcriptional regulation of adrenocortical development. Endocrinology. 2005;146:1018–24.
- 27. Lin L, Philibert P, Ferraz-de-Souza B, et al. Heterozygous missense mutations in steroidogenic factor 1 (SF1/Ad4BP, NR5A1) are associated with 46, XY disorders of sex development with normal adrenal function. J Clin Endocrinol Metab. 2007;92:991–9.
- Lourenco D, Brauner R, Lin L, et al. Mutations in NR5A1 associated with ovarian insufficiency. N Engl J Med. 2009;360:1200–10.
- Miller WL. Steroid hormone biosynthesis and actions in the maternofeto- placental unit. Clin Perinatol. 1998;25:799–817.
- Padbury JF. Functional maturation of the adrenal medulla and peripheral sympathetic nervous system. Baillieres Clin Endocrinol Metab. 1989;33:689–705.
- Huber K, Karch N, Ernsberger U, et al. The role of PHOX2B in chromaffin cell development. Dev Biol. 2005;279:501–8.
- Gaultier C, Trang H, Dauger S, et al. Pediatric disorders with autoimmune dysfunction: what role for PHOX2B? Pediatr Res. 2005;58:1–6.
- Thomas SA, Matsumoto AM, Palmiter RD. Noradrenaline is essential for mouse fetal development. Nature. 1995;374:643–6.
- 34. Slotkin TA, Seidler FJ. Adrenomedullary catecholamine release in the fetus and newborn: secretory mechanisms and their role in stress and survival. J Dev Physiol. 1988;10:1–16.
- Santisteban P. Development and anatomy of the hypothalamicpituitary axis. In: Braverman LE, Utiger RD, editors. The thyroid. 9th ed. Philadelphia: JB Lippincott; 2005. p. 8–25.
- 36. Brown RS, Huang SA, Fisher DA. The maturation of thyroid function in the perinatal period and during childhood. In: Braverman LE, Utiger RD, editors. The thyroid. 9th ed. Philadelphia: JB Lippincott; 2005. p. 1013–28.
- 37. Salvatore D, Davies TF, Schlumberger MJ, Hay ID, Larsen PR. Thyroid physiology and diagnostic evaluation of patients with thyroid disorders. In: Melmed S, Polonsky KS, Larsen PR, Krokenberg HM, editors. Williams textbook of endocrinology. Philadelphia: Elsevier Saunders; 2011. p. 327–61.
- Parlato R, Rosica A, Rodriguez-Mallon A, et al. An integrated regulator network controlling survival and migration in thyroid organogenesis. Dev Biol. 2004;276:464–75.

- De Felice M, Di Lauro R. Thyroid development and its disorders: genetics and molecular mechanisms. Endocr Rev. 2004;25:722–46.
- Trueba SS, Auge J, Mattei G, et al. PAX8, TITF1 and FOXE1 gene expression patterns during human development: new insights into human thyroid development and thyroid dysgenesis-associated malformations. J Clin Endocrinol Metab. 2005;90:455–62.
- Burrow GN, Fisher DA, Larsen PR. Mechanisms of disease: maternal and fetal thyroid function. N Engl J Med. 1994;331:1072–8.
- Macchia P. Recent advances in understanding the molecular basis of primary congenital hypothyroidism. Mol Med Today. 2000;6:36–42.
- Szinnai G, Lacroix L, Carre A, et al. Sodium/iodide symporter (NIS) gene expression is the limiting step for the onset of thyroid function in the human fetus. J Clin Endocrinol Metab. 2007;92:70–6.
- 44. Morreale de Escobar G, Obregon MJ, Escobar del Rey F. Role of thyroid hormone during early brain development. Eur J Endocrinol. 2004;151:U25–37.
- 45. Kerry R, Hume R, Kaptein E, et al. Sulfation of thyroid hormone and dopamine during human development: ontogeny of phenol sulfotransferases and arylsulfatase in liver, lung, and brain. J Clin Endocrinol Metab. 2001;86:2734–42.
- 46. Pop VJ, Brouwers EP, Vader HL, et al. Maternal hypothyroxinaemia during early pregnancy and subsequent child development: a 3-year follow-up study. Clin Endocrinol (Oxf). 2003;59:282–8.
- Hochberg Z'e, Tiosano D. Disorders of mineral metabolism. In: Pescovitz OH, Eugster EA, editors. Pediatric endocrinology. Philadelphia: Lippincott Williams & Wilkins; 2004. p. 614–40.
- Bowl MR, Nesbit MA, Harding B, et al. An interstitial deletion/insertion involving chromosomes 2p25.3 and Xq27.1 near SOX3 causes X-linked recessive hypoparathyroidism. J Clin Invest. 2005;115:2822–31.
- Kovaks CS, Kronenberg HM. Maternal-fetal calcium and bone metabolism during pregnancy, puerperium and lactation. Endocr Rev. 1997;18:832–72.
- 50. Kovacs CS, Lanske B, Hunzelman JL, et al. Parathyroid hormonerelated peptide (PTHrP) regulates fetal-placental calcium transport through a receptor distinct from the PTH/PTHrP receptor. Proc Natl Acad Sci U S A. 1996;93:15233–8.
- Prada JA. Calcium-regulating hormones. In: Polin RA, Fox WW, editors. Fetal and neonatal physiology. 2nd ed. Philadelphia: WB Saunders; 1998. p. 2287–96.
- Erickson RP, Blecher SR. Genetics of sex determination and differentiation. In: Polin RA, Fox WW, Abman SH, editors. Fetal and neonatal physiology. 3rd ed. Philadelphia: WB Saunders; 2004. p. 1935–41.
- Lee MM. Molecular genetic control of sex differentiation. In: Pescovitz OH, Eugster EA, editors. Pediatric endocrinology. Philadelphia: Lippincott Williams & Wilkins; 2004. p. 231–42.

- Biason-Lauber A. Control of sex development. Best Pract Res Clin Endocrinol Metab. 2010;24:163–86.
- Sekido R, Lovell-Badge R. Sex determination and SRY: down to a wink and a nudge? Trends Genet. 2009;25:19–29.
- Aslan AR, Kogan BA, Gondos B. Testicular development. In: Polin RA, Fox WW, Abmans SH, editors. Fetal and neonatal physiology. 3rd ed. Philadelphia: WB Saunders; 2004. p. 1950–5.
- Tapanainen J, Kellokumpu-Lehtinen P, Pelliniemi L, et al. Age-related changes in endogenous steroids of human fetal testis during early and midpregnancy. J Clin Endocrinol Metab. 1981;52:98–102.
- Hughes IA, Acerini CL. Factors controlling testis descent. Eur J Endocrinol. 2008;159 suppl 1:S75–82.
- Josso N, Belville C, Dicard JY. Mutations in AMH and its receptors. Endocrinology. 2003;13:247–51.
- Byskov AG, Westergaard LG. Differentiation of the ovary. In: Polin RA, Fox WW, Abman SH, editors. Fetal and neonatal physiology. 3rd ed. Philadelphia: WB Saunders; 2004. p. 1941–9.
- Fulton N, da Silva SJM, Bayne RAL, et al. Germ cell proliferation and apoptosis in the developing human ovary. J Clin Endocrinol Metab. 2005;90:4664–70.
- Parma P, Radi O, Vidal V, et al. R-spondin 1 is essential in sex determination, skin differentiation and malignancy. Nat Genet. 2006;38:1304–9.

- 63. Couse JF, Korach KS. Estrogen receptor null mice: what have we learned and where will they lead us? Endocr Rev. 1999;20:358–417.
- Habener JF, Kemp DM, Thomas MJ. Mini review: transcriptional regulation in pancreatic development. Endocrinology. 2005;146:1025–34.
- Edlund H. Pancreatic organogenesis: developmental mechanisms and implications for therapy. Nat Rev Genet. 2002;3:524–32.
- 66. Formby B, Ullrich A, Coussens L, et al. Growth hormone stimulates insulin gene expression in cultured human fetal pancreatic islets. J Clin Endocrinol Metab. 1988;66:1075–9.
- Sperling MA. Carbohydrate metabolism: insulin and glucagons. In: Tulchinsky D, Little AB, editors. Maternal-fetal endocrinology. 2nd ed. Philadelphia: WB Saunders; 1994. p. 380–400.
- Fisher DA. Fetal thyroid function: diagnosis and management of fetal thyroid disorders. Clin Obstet Gynecol. 1997;40:16–31.
- 69. New MI, Carlton A, Obeid J, et al. Update: prenatal diagnosis for congenital adrenal hyperplasia in 595 pregnancies. Endocrinologist. 2003;13:233–9.
- Flake AW, Zanjani ED. In utero hematopoietic stem cell transplantation. JAMA. 1997;278:932–7.

Part VI

Fetal Immune Development: Up to Second Trimester

Development of Human Immune System in First and Second Trimester

24

Swapna Chaudhuri

Introduction

The immune paradox lies in the fact that the semi-allogenic embryo (or allogenic embryo in surrogate mothers) is not immunologically rejected [95]. A pregnancy may be spontaneously aborted in 10-20 % of cases without any apparent reason [2, 60]. There are multiple sets of immunological needs for the fetus and newborn, like protection against infection and bypassing harmful inflammatory immune responses that can lead to pre-term delivery, and also the transition from a sterile intra-uterine environment to external world full of unknown antigens. These immunological needs shape neonatal innate immune system dampening the production of pro-inflammatory cytokines rendering newborns at risk of infection and impairing responses to many vaccines [63]. Ontogeny of human fetal immune system in mid-term fetus is derived from a completely different set of stem cells than the adult immune system. Fetal immune system can tolerate antigens (including its mother and its own organs) better than to eliminate antigens from its environment. Midterm fetal immune system has about three times more regulatory T cells than newborns or adults, which render tolerance

S. Chaudhuri

Emeritus Professor, Department of Physiology, ICMR, New Delhi, India e-mail: swapna.chaudhuri@gmail.com [75]. Treg blocked fetal T-cell responses to maternal cells.

Trophoblast-derived thymic stromal lymphopoietin (TSLP) instructs decidual CD11c+ dendritic cells (dDCs) with increased costimulatory molecules; MHC class II; and Th2/3-type, but not Th1-type, cytokines. TSLP-activated dDCs cause proliferation and differentiation of decidual CD4+ CD252 T cells into CD4+ CD25+ FOXP3+ regulatory T cells (Tregs) through TGF-b1 [26]. Decidual CD4+ CD25+ FOXP3+ Tregs promote invasiveness and HLA-G expression of trophoblasts, resulting in preferential production of Th2 cytokines and reduced cytotoxicity in decidual CD56brightCD16 NK cells [26].

It is of primary need to strengthen all the factors to protect the mother and the neonate since, pregnancy represents the most important period for the conservation of the species. To protect the mother against the environment and preventing damage to the fetus the immune system plays an important role. The maternal immune system during pregnancy is characterized by a reinforced network of recognition, communication, trafficking and repair; to maintain the well-being of the mother and the fetus it is able to raise the alarm, if necessary. On the other side, the fetus provides a developing active immune system that modifies the way the mother responds to the environment. Therefore, it is appropriate to refer to pregnancy as a unique immune condition that is modulated,

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_24

but not suppressed. This unique behaviour explains why pregnant women respond differently to the presence of microorganisms or its products [76].

There are three threats to the embryo: (i) lysis or immobilization by anti-embryonic antibodies, (ii) attack by non-MHC-restricted cells, and (iii) ongoing maternal immune response directed towards the implantation site. Acceptance of the fetal "allograft" by the maternal immune system is one of the mysteries of the immune system. The active expression of T-cell immunity against HLA antigens expressed on fetal tissues, results in placental detachment and fetal loss. When reactivity is less intense, it may result in preeclampsia and premature delivery. In this regard T-cell responses are Th1-polarized and are dominated by IFN- γ , which is highly toxic to the placenta [114].

Absence of MHC antigens of human preimplantation embryo prevents direct attacks by MHC-restricted T cells. Maternal APC presenting paternal or embryo-derived peptides triggers CD4+ and CD8+ MHC effector T cells yielding vulnerability to the human embryo [17]. During the first trimester, NK cells, dendritic cells and macrophages infiltrate the decidua and accumulate around the invading trophoblast cells [4, 97]. uNK cells are critical for trophoblast invasion in the uterus and uDC are necessary for decidual formation and may affect the angiogenic response by inhibiting blood vessel maturation [7].

A series of overlapping control mechanisms selectively downregulate Th1 immunity at the fetomaternal interface and within the fetal microenvironment itself operate at the level of the placenta. Expression of FasL on fetal cells interact with Fas on activated T cells as a potential means of elimination of T cells by apoptosis [39, 40]. CD95 receptor mediates cell death signal for apoptosis. CD95 ligand has a crucial role in the homeostasis of haematopoietic cell populations in adults. Compared to adult peripheral blood lymphocytes the low level of CD95 ligand expression of cord blood mononuclear cells enjoy some immune privilege [25]. Local production of T-cell suppressive tryptophan metabolites via indoleamine 2,3-dioxygenase, which is expressed in syncytiotrophoblasts and macrophages [79]. Immunosuppressive molecules in the placenta, such as progesterone [88, 104, 105], prostaglandin E2 [1, 44] and early pregnancy factor [78] helps to prevent immunological rejection of an embryo. Moreover, Th2-trophic and/or Th1-suppressive factors are produced by the placenta in high levels including IL-4 and IL-10 [93]. Progesterone ensures that any environmental antigens/allergens that pass via the maternal circulation across to the developing fetus will elicit fetal immune system Th2 cytokines [89]. Trophoblasts express nonclassical MHC molecules HLA-G, HLA-C, HLA-E, and HLA-F [109]. HLA-G binds to the killing inhibitory receptor on NK cells and protects trophoblasts from NK cell-mediated attack. HLA-G may also protect the fetus from an allogeneic T cell response.

The commensal microflora of the gastrointestinal tract imparts signals from the microbial environment to stimulate postnatal maturation of the immune function in mammals. Infections, in the gastrointestinal and respiratory tracts, may also contribute to this process [46]. The principal focus of this late-stage maturation process is upregulation of Th1 functions, which, are dampened during fetal life. Absence of adequate microbial stimulation during infancy results in preponderance of Th2 phenotype, resulting in blunted expression of Th1 immunity at peripheral challenge sites [49], at mucosal surfaces a failure of the immune deviation mechanisms that normally regulate induction of Th2 responses [103], and excessive class switching of immature B cells toward IgE commitment [27]. The molecular signalling between the microbial environment and the immune system through the TOLL receptors [51, 116], as well as the high-affinity receptor for bacterial lipopolysaccharide (CD14), are found to be central in the process.

Development of the Fetal Immune System

At about gestational 6 weeks thymus arises from the third branchial arch and cortex arises from ectodermal layer and the medulla arises from the
endoderm. Over the next 2–3 weeks lymphoid cells migrate, initially from the yolk sac and fetal liver, and then from the bone marrow to colonize the fetal thymus [20, 46].

The initial stage of human fetal haemopoiesis occurs in the yolk sac mesoderm and the extraembryonic mesenchymal tissue. At 3-4 weeks of gestation pluripotent erythroid and granulomacrophage progenitors appear in the yolk sac of human embryos. At 5–6 weeks of gestation these primitive cells can then be detected in the circulation, and from 4 weeks of gestation they migrate to the liver which becomes the major site of haemopoiesis. Liver size increases significantly with rise in the number of nucleated cells from the 5–10 weeks. Although these early progenitors keeps on proliferating very little differentiation occurs although a discrete granulocyte/macrophage population emerges at this time. The thymus and spleen are seeded from the liver and stem cells are detectable in the bone marrow at 11–12 weeks of gestation [73]. Hepatic haematopoiesis declines in the third trimester of pregnancy but ceases soon after birth.

Fetal blood collected by fetoscopy at 12–19 weeks of gestation when cultured yields high levels of both erythroid and granulocytic/ monocytic progenitor cells – monocytes comprising 42–68 %, neutrophils 27–41 %, and eosinophils 5–30 % [65]. Although the number of granulocyte progenitors in the circulation is high at this time, granulocytes are not formed in significant amount in the foetuses until after birth. Neutrophils appear last in the blood during fetal life [32].

Developmental Origin of Hematopoietic Stem Cells (HSCs)

The first wave of hematopoiesis occurs in the yolk sac in the first month [87] and yields initially macrophages and mature red blood cells that remain nucleated before the circulation of blood begins [85]. Mature and more primitive hematopoietic cell including the first lymphoid cells generate subsequently [11, 50]. Between weeks 7 and 19 of gestation [12] a significant

number of committed and multi-potent CD34+ progenitors with a capacity for expansion circulate in the fetal blood. Later in midgestation, cells that display the self-sustaining properties of HSCs arise predominantly from hemogenic cells with endothelial features lining both dorsal aorta and blood vessels in the placenta [28]. These HSCs then translocate to the fetal liver and spleen with rapid amplification. Just before birth they migrate again to developing niches in the bone marrow where HSCs remain concentrated throughout adult life, although some may also recirculate [77]. The HSCs migration during embryogenesis in the bone marrow microenvironment is controlled by adhesion receptors including CD34, CD44, selectins, and integrins [48], $\alpha 4\beta 1$ (VLA-4) and its ligand VCAM-1 are involved in the homing of hematopoietic progenitors to spleen and bone marrow [86]. In the developing fetal liver, the lymphoid-deficient α -HSCs constitute only less than 5 % of all HSCs and less than 10 % of the HSCs with durable selfrenewal ability. However, these are much more prevalent (~3× more) amongst the initial HSC population that can be detected in fetal bone marrow (BM) just before birth [5].

In normal pregnancy, the rate of fetal mononuclear-cell division rate decreases with gestational age from 1.8 % at 18 weeks to 1 % at 40 weeks [107]. The rate of mononuclear-cell division is elevated in early pregnancy and in chromosomally abnormal fetuses, probably as a consequence of higher number of circulating haemopoietic precursors. Hematopoietic and progenitor stem cells is found in both the fetal blood and liver [83], but the fetal liver acts as a richer source of primitive hematopoietic progenitor cells than the fetal blood. The fetal red blood cells, white blood cells, and platelet counts all increase with gestation, from weeks 8 to 17, reflecting hematopoietic development. During gestation, normoblasts decrease dramatically. The number of circulating and hepatic T lymphocytes increases before 13 week of gestation, reflecting thymic maturation. The fetal liver contains fewer T lymphocytes than the fetal blood (2.5 % vs. 18.6 %) and more CD34+ hematopoietic stem and progenitor cells (17.5 % vs. 4.3 %). At an early

gestational age (21–22 weeks) fetal blood has a higher concentration of primitive hematopoietic progenitor cells (CD34+/CD38– cells) than umbilical cord blood at full term (39–40 weeks) [82]. The cord blood cytokine-receptor network, consisting of IL-1, IL-2, IL-12, IFN- γ and TNF- α is biased towards anti-inflammatory activity in fetuses [41]. Endothelial progenitor cells derive mainly from the monocyte/macrophagecontaining CD34- mononuclear cells and only in part from the hematopoietic stem-cell-containing CD34+ mononuclear cells [92].

Cytokine Profile of Fetal Pre-immunocytes

Increased IL-2 γ receptor expression in cordblood mononuclear can mediate the prevention of neonatal infection [94]. The mononuclear cells from neonatal blood produce less IL-10 than adults while the primary cells of origin and the regulatory mechanisms may differ in adults [57]. Enhanced IL-6 and decreased IFN γ production by the cord-blood mononuclear cells are detectable hallmarks of newborns population with the high-risk allergy population [64]. The fetal peripheral blood mononuclear cells exhibit mitogenic responses to allergenic stimuli during gestation [54].

Neonatal monocytes widely differ from adult monocytes in cytokine-expression profile [43]. Following exposure to lipopolysaccharide fetal monocytes produce more IL-8 and less TNF- α . In neonatal sepsis caused by Streptococcus agalactiae, a major cause of severe infection in newborns and pregnant females, different cytokine expression patterns (IL-6, IL-1 beta, and IL-12p40) have been found in the cord-blood mononuclear cells [6] compared to *Escherichia coli* lipopolysaccharide, where TNFa, IL-6, and IL-8 appear almost simultaneously. The human monocyte response to S. agalactiae results in the production of TNF-a, and delayed appearance of IL-6, and IL-8 [61]. The lymphocyte response to S. agalactiae is manifested by IFN γ and IL-12 secretion, which the E. coli lipopolysaccharide fails to produce. This suggests an important role for TNFa, IFNy, and IL-12 in S. agalactiae pathogenesis and/or immunity. Up-regulation of macrophage migration inhibitory factor (MIF) in the intervillous blood is found in placental malaria in pregnant women. MIF may play a role in immune responses to malaria during pregnancy by virtue of its ability to activate macrophages and to overcome the immunosuppressive effect of glucocorticoids [16]. A consistent pattern of MIF expression is found in the syncytiotrophoblasts, extravillous trophoblasts, intervillous blood mononuclear cells and amniotic epithelial cells [15]. Significantly higher levels of MIF expression is exhibited in the amniotic epithelial and intervillous blood mononuclear cells from infected placentas compared to that in uninfected placentas.

Development of Macrophages and Dendritic Cells

At 4–6 weeks of age there are found two populations of cells with a dendritic/macrophage structure in the yolk sac and mesenchyme and also prehaematopoietic liver at 5 weeks of gestation. The major population of yolk sac macrophages is MHC class II-negative, and there is a minor population that is MHC class II-positive [52]. MHC class II-negative cells appear in the thymic cortex, in the marginal zones of lymph nodes, in the splenic red pulp, and in the midst of erythopoietic activity in the bone marrow. A few MHC class II-positive cells are seen in the liver at 7–8 weeks of gestation, the lymph nodes at 11-13 weeks of gestation, and the T-cell areas of the developing thymic medulla by 16 weeks of gestation, whereas thymic epithelium expresses class II at 8–9 weeks [52].

MHC class II-positive cells also are present in the skin, gastrointestinal tract, and hepatic systems. The number of hepatic sinusoidal macrophages (Kupffer cells) is low in early gestation (17 weeks) but increases to nearly adult levels in the neonatal period. By 6 weeks of intrauterine development the blood flow to the liver passing through the left umbilical vein comes directly from the placenta, providing a rich nutrient supply to these cells [19]. By 6–7 weeks of gestation HLA-DR+ Langerhans cells are detectable in the skin. Langerhans cells migrate into the epidermis during the first trimester and resemble the adult phenotype by the second trimester [33]. There are MHC class II-positive cells in unidentified cell type in the lamina propria of the fetal gut as early as 11 weeks of gestation [67]. Cord-blood dendritic cells express relatively poor accessory function [47]. Umbilical cord-blood dendritic cells have lower levels of ICAM-1 and MHC classes I and II than peripheral blood dendritic cells from adults.

T Cell Development

At 8-9 weeks of gestation CD7+ T-cell precursors from the fetal liver seed the thymus, 60 % of which are cytoplasmic CD2+ and only 4 % are cytoplasmic CD3+. From 9.5 weeks until birth, TCR β + cells increase to form over 90 % of the CD7+ population [13]. The prothymocytes by interacting with the stroma, proliferate, and differentiate with expression of the first T cellspecific surface molecules like CD2 and later, CD4 and CD8 [3]. At 12 weeks of gestation delineation of the thymic cortical and medullary regions occur, following which Hassall's corpuscles appear [8]. The most immature thymocytes are found in the subcapsular cortical region, and cells move into the deeper layers as they mature [9].

The early prothymocytes do not express CD3, the T cell receptor (TCR), CD4, or CD8 and are often referred to as "triple-negative thymocytes" [70]. The progeny continue to divide and rearrange their TCR genes, and since these cells express both CD4 and CD8, they are now called "double-positive" cells [58, 70]. They undergo positive selection by self-major histocompatibility complex (MHC) restriction. More than 95 % (nearly 50 million) of the cells die each day during this stage [70].

From 18 to 24 weeks of gestation, though the mesenteric lymph nodes have very few B cells

or monocytes but a high percentage of CD45RA+ T cells, the fetal spleen at this time is invaded by equal numbers of T cells, B cells, and monocytes/macrophages [112]. Lymph-node and thymus T cells at these gestational ages express activation marker CD69. By 18 weeks of gestation T cells from fetal spleen have adult levels of CD3, CD4, CD8, CD2 and CD11a making the spleen fully immunocompetent, with sufficient accessory cells to ensure T-cell activation. But at this stage the mesenteric lymph nodes are deficient in accessory cells numerically or functionally.

Between 6 and 9 weeks of gestation prior to being detectable in the thymus rearranged TCR δ genes are first seen in the liver and primitive gut [71]. Though early in development the thymic and gut γ/δ T-cell repertoires overlap but they diverge and become non overlapping during the second trimester [117]. The γ/δ T-cell population in the fetal liver is distinct from the thymus. At 20–22 weeks of gestation, 63 % of CD3+ cells are TCR α/β and 32 % are TCR γ/δ in the liver, which is considered to be a site of γ/δ T-cell development. Mature $\alpha\beta$ T cells can be found in the periphery of the human fetus as early as 10–12 gestational weeks [29].

Neonatal human T cells proliferate in response to different antigens, like allergens [89, 106, 111, 113], auto antigens [118], and parasite antigens [31, 81]. Mounting of an antigen-specific response of neonatal T and B cells is indicated by presence of antigen-specific IgE in umbilical cord blood [56].

Treg Development

In the fetal thymus, double-positive cells initiate expression of CD25, GITR, CTLA4 and CD122 during their transition from the CD27– to the CD27+ stage. Moreover, CD4 + CD25+ fetal thymocytes already have the potential to suppress proliferation of CD25- cells. After leaving the thymus, FoxP3 + CD4 + CD25+ Treg enter the fetal lymph nodes and spleen, where they acquire a primed/memory phenotype [21].

Development of B Cells

As early as 8 weeks of gestation Pro- (CD24+/ surface IgM-negative) and pre-B cells (cytoplasmic IgM+/surface IgM-negative) can be detected in the fetal liver and omentum. The percentage of pre-B cells decreases during weeks 13-23 in the omentum though they are similar to fetal liver over 8-12 weeks gestation [101]. B-cell development in the omentum is transitory and become detectable in the spleen at 13-23 weeks of gestation. At 15 weeks of gestation CD5+ B cells can be found in the human peritoneal cavity and pleural cavity [80]. In the fetal circulation percentage of CD5+ B cells (B-1 B cells) is higher than the adult, and declines with increasing gestational age [10, 110]. CD5+ B cells are largely T-independent and produce polyreactive antibodies which have a role in the primary immune response. This immune mechanism form a part in the first line of defence, which is a necessary function in the newborn.

The pre-B cell, is a large lymphoid cell containing scanty intracytoplasmic μ heavy chains and lack surface immunoglobulins (sIg) and intracytoplasmic light chains [34, 42, 90]. There were twice as many pre-B cells as B cells in the livers of 7–14 week old fetuses [34, 55]. Although both small and large pre-B cells were seen in the liver and BM at all times of gestation investigated, the small pre-B cells by far outnumbered the large ones in both organs [59]. At 8 weeks of gestation, liver pre-B cells express the cytoplasmic μ chain, and by 10–12 weeks, surface IgM is expressed on liver B cells. In the livers of 8 week old fetuses, more than 90 % of the total pre-B plus B cell population consisted of pre-B cells; the relative number of liver pre-B cells gradually decreased with increasing gestational age and, after the 14th week, B cells outnumbered pre-B cells [35].

From 13 weeks of gestation surface IgD is detectable and CD24 expression precedes μ -chain expression and is retained throughout differentiation into adulthood. Liver B cells are negative for CD21 and CD22 though they express CD20 [10]. B cells strongly express IgM+ which are detectable in the lymph nodes from 16 to 17 weeks and spleen at 16–21 weeks [10]. From 17 weeks of gestation primary nodules develop around the follicular dendritic cells of the lymph nodes. At 16–20 weeks of gestation B cells are abundant in the bone marrow. B cells in the spleen are diffusely distributed at 22 weeks, and then form primary nodules around 24 weeks; this is later than seen in lymph nodes. B cells positive for CD19, CD20, CD21, CD22, HLA-DR, IgM, and IgD emerge into the peripheral circulation at 12 weeks of gestation [110].

Immunoglobulin Production

At 10 weeks of gestation early IgG and IgM synthesis occurs primarily in the spleen in large amounts, and is maximal at 17–18 weeks of gestation. Between 5.5 and 22 weeks, serum IgG level slowly increase with greater increase in 26 weeks, and finally a very significant increase at birth [37]. Throughout gestation IgG crosses the placenta, the transfer rate rising markedly from 20 weeks, and is maximal from 32 weeks of gestation [69, 84]. At 11 weeks of gestation IgE synthesis was observed in fetal liver and lung, and by 21 weeks in the spleen [74].

Neonates have very low serum IgM, IgA and IgE levels, and the IgG present is essentially of maternal origin. Surface IgM and CD79 (signal transducer for membrane Ig and necessary for all IgM functions) are elevated on cord-blood B cells compared to the adult. But similar levels of CD19, 21, 22, and 81 are expressed in cord and adult B cells, although CD32 is lower on cord B cells [66].

Newborns of mothers vaccinated with tetanus toxoid during pregnancy have specific antibody of the IgM class in their serum [36]. Umbilical cord mononuclear cells collected at birth at full term exhibit antigen-specific reactivity to allergens, including those of house-dust mite and cow's milk [12, 81, 83, 107]; parasite antigens such as those of *Plasmodium* spp. [115] and *Schistosoma* spp. [92]; and autoantigens, including myelin basic protein [41].

Mucosal Immunity

IgA and IgM are important in the first line of defence. Occasional IgM- and IgA-producing cells were observed in the fetal parotid gland (20–40 weeks), but cells producing D, G, or E isotypes were not seen [108]. The IgA1 subclass mostly J-chain-positive predominates. Amylase, lysozyme, and lactoferrin were detectable early fetal life.

Duodenal expression of secretory component, classes I and II is seen and IgA-, IgM-, and IgG-producing cells are detectable from 24 to 32 weeks of gestation. A population of MHC class II-positive cells exists in the lamina propria from 11 weeks of gestation [53], and T cells are found at this site from 12 to 14 weeks of gestation [30]. From the second postnatal week, intense expression of epithelial HLA-DR, secretory component, and IgA is seen, reflecting modulation by environmental factors [108].

Mucosal surfaces have an important role in the development of allergic responses and disease immune responses. Prior to birth both the skin and gastrointestinal tract are relatively immunologically mature, whereas the airways show little evidence of population by haematopoietic cells; an influx is seen during the first week postnatally [102]. Allergic disease is first manifest in the gut and skin while airways inflammation appear later in infancy/childhood. This may be due to developmental delay in the airways.

Development of Eosinophils

Eosinophil granulopoiesis occurs in the fetal liver, at 5 weeks in the hepatic laminae [100], and after 20 weeks of gestation, they appear in the portal areas. There are very few studies on either the phenotype or function of fetal and neonatal eosinophils. Interestingly, fetal eosinophils (23–34 weeks of gestation) have adult levels of L-selectin, while newborns have less L-selectin on their eosinophils than those of the adult [98, 99].

Mother–Placenta–Fetus: Relation to Infection and Immunity

Neonates with placental infections have very high circulating levels of inflammatory cytokines, such as IL-1, IL-6, IL-8 and TNF- α , which has been termed as fetal inflammatory response syndrome (FIRS) [22, 68, 91]. These cytokines have been shown to affect the CNS and the circulatory system [14, 24, 68]. Fetal morphologic abnormalities were found in animal models leading to ventriculomegaly and hemorrhages, which were due to fetal pro-inflammatory cytokines such as II-1, TNF α , MCP-1, MIP1- β and INF- γ . The presence of FIRS increases the future risk for autism, schizophrenia, neurosensorial deficits and psychosis induced in the neonatal period. The presence of FIRS increases the future risk for autism, schizophrenia, neurosensorial deficits and psychosis induced in the neonatal period [38, 72, 96]. An inflammatory response in the placenta, altering the cytokine balance in the fetus, may affect the normal development of the fetal immune system leading to anomalous responses during childhood or later in life. A significant impact on later vaccine responses may result from antenatal infections. Surviving infants with placental malaria may suffer adverse neurodevelopmental stages and may have abnormal responses to a later parasitic infection [62]. In all these cases the parasite did not reach the placenta, but the inflammatory process in the placenta affected the normal fetal development [23].

References

- Abe N, Katamura K, Shintaku N, Fukui T, Kiyomasu T, Iio J, Ueno H, Tai G, Mayumi M, Furusho K. Prostaglandin E2 and IL-4 provide naive CD4+ T cells with distinct inhibitory signals for the priming of IFN-gamma production. Cell Immunol. 1997;181(1): 86–92.
- Adolfsson A, Larsson PG. Cumulative incidence of previous spontaneous abortion in Sweden in 1983–2003: a register study. Acta Obstet Gynecol Scand. 2006;85:741–7.
- Anderson G, Moore NC, Owen JJ, Jenkinson EJ. Cellular interactions in thymocyte development. Annu Rev Immunol. 1996;14:73–99.

- Ashkar AA, Di Santo JP, Croy BA. Interferon gamma contributes to initiation of uterine vascular modification, decidual integrity, and uterine natural killer cell maturation during normal murine pregnancy. J Exp Med. 2000;192:259–70.
- Benz C, Copley MR, Kent DG, et al. Hematopoietic stem cell subtypes expand differentially during development and display distinct lymphopoietic programs. Cell Stem Cell. 2012;10:273–83.
- Berner R, Csorba J, Brandis M. Different cytokine expression in cord blood mononuclear cells after stimulation with neonatal sepsis or colonizing strains of streptococcus agalactiae. Pediatr Res. 2001;49: 691–7.
- Birnberg T, Plaks V, Berkutzki T, Mor G, Neeman M, Dekel N, Jung S. Dendritic cells are crucial for decidual development during embryo implantation. Am J Reprod Immunol. 2007;57:342.
- Bodey B, Kaiser HE. Development of Hassall's bodies of the thymus in humans and other vertebrates (especially mammals) under physiological and pathological conditions: immunocytochemical, electronmicroscopic and in vitro observations. In Vivo. 1997;11:61–85.
- Bodey B, Bodey Jr B, Siegel SE, Kaiser HE. Novel insights into the function of the thymic Hassall's bodies. In Vivo. 2000;14:407–18.
- Bofill M, Janossy G, Janossa M, et al. Human B cell development. II. Subpopulations in the human fetus. J Immunol. 1985;134:1531–8.
- Boiers C, Carrelha J, Lutteropp M, et al. Lymphomyeloid contribution of an immune-restricted progenitor emerging prior to definitive hematopoietic stem cells. Cell Stem Cell. 2013;13:535–48.
- Campagnoli C, Fisk N, Overton T, et al. Circulating hematopoietic progenitor cells in first trimester fetal blood. Blood. 2000;95:1967–72.
- Campana D, Janossy G, Coustan-Smith E, et al. The expression of T cell receptor-associated proteins during T cell ontogeny in man. J Immunol. 1989; 142:57–66.
- 14. Cardenas I, Aldo P, Koga K, Means R, Lang SH, Mor G. Subclinical viral infection in pregnancy lead to inflammatory process at the placenta with nonlethal fetal damage. Am J Reprod Immunol. 2009; 61:397.
- Chaisavaneeyakorn S, Lucchi N, Abramowsky C, et al. Immunohistological characterization of macrophage migration inhibitory factor expression in Plasmodium falciparum-infected placentas. Infect Immun. 2005;73:3287–93.
- Chaisavaneeyakorn S, Moore JM, Othoro C, et al. Immunity to placental malaria. IV. Placental malaria is associated with up-regulation of macrophage migration inhibitory factor in intervillous blood. J Infect Dis. 2002;186:1371–5.
- Chaouat G. The immunological status of the preimplantation embryo. In: Kurpisz M, Fernandez N, editors. Immunology of human reproduction. Oxford: Bios Scientific Publ; 1995. p. 285.

- Clerici M, DePalma L, Roilides E, Baker R, Shearer GM. Analysis of T helper and antigen-presenting cell functions in cord blood and peripheral blood leukocytes from healthy children of different ages. J Clin Invest. 1993;91:2829–36.
- Cope EM, Dilly SA. Kupffer cell numbers during human development. Clin Exp Immunol. 1990; 81(3):485–8.
- Crompton T, Outram SV, Hager-Theodorides AL. Sonic hedgehog signalling in T-cell development and activation. Nat Rev Immunol. 2007;7: 726–35.
- Cupedo T, Nagasawa M, Weijer K, Blom B, Spits H. Development and activation of regulatory T cells in the human fetus. Eur J Immunol. 2005;35: 383–90.
- 22. Davies JK, Shikes RH, Sze CI, et al. Histologic inflammation in the maternal and fetal compartments in a rabbit model of acute intra-amniotic infection. Am J Obstet Gynecol. 2000;183:1088–93.
- Desai M, Ter Kuile FO, Nosten F, et al. Epidemiology and burden of malaria in pregnancy. Lancet Infect Dis. 2007;7:93–104.
- Deverman BE, Patterson PH. Cytokines and CNS development. Neuron. 2009;64:61–78.
- Drenou B, Choqueux C, El GA, et al. Characterisation of the roles of CD95 and CD95 ligand in cord blood. Bone Marrow Transplant. 1998;22 Suppl 1: S44–7.
- 26. Du MR, Guo PF, Piao HL, Wang SC, Sun C, Jin LP, Tao Y, Li YH, Zhang D, Zhu R, Fu Q, Li DJ. Embryonic trophoblasts induce decidual regulatory T cell differentiation and maternal-fetal tolerance through thymic stromal lymphopoietin instructing dendritic cells. J Immunol. 2014;192(4):1502–11.
- Durkin HG, Bazin H, Waksman BH. Origin and fate of IgE-bearing lymphocytes. I. Peyer's patches as differentiation site of cells. Simultaneously bearing IgA and IgE. J Exp Med. 1981;154:640–8.
- Dzierzak E, Speck NA. Of lineage and legacy: the development of mammalian hematopoietic stem cells. Nat Immunol. 2008;9:129–36.
- Krow-Lucal ER, McCune JM. Distinct functional programs in fetal T and myeloid lineages. Front Immunol. 2014. http://dx.doi.org/10.3389/ fimmu.2014.00314.
- Fahey JV, Humphrey SL, Stern JE, Wira CR. Secretory component production by polarized epithelial cells from the human female reproductive tract. Immunol Invest. 1998;27:167–80.
- Fievet N, Ringwald P, Bickii J, et al. Malaria cellular immune responses in neonates from Cameroon. Parasite Immunol. 1996;18:483–90.
- Forestier F, Daffos F, Catherine N, Renard M, Andreux JP. Developmental hematopoiesis in normal human fetal blood. Blood. 1991;77(11):2360–3.
- 33. Foster CA, Holbrook KA, Farr AG. Ontogeny of langerhans cells in human embryonic and fetal skin: expression of HLA-DR and OKT-6 determinants. J Invest Dermatol. 1986;86:240–3.

- 34. Gathings WE, Lawton AR, Cooper MD. Immunofluorescent studies of the development of pre-B cells, B lymphocytes and immunoglobulin isotype diversity in humans. Eur J Immunol. 1977; 7:804–10.
- Asma GE, Langlois van den Bergh R, Vossen JM. Development of pre-B and B lymphocytes in the human fetus. Clin Exp Immunol. 1984;56:407–14.
- Gill III TJ, Repetti CF, Metlay LA, et al. Transplacental immunization of the human fetus to tetanus by immunization of the mother. J Clin Invest. 1983;72:987–96.
- 37. Gitlin D, Biasucci A. Development of gamma G, gamma A, gamma M, beta IC-beta IA, C 1 esterase inhibitor, ceruloplasmin, transferrin, hemopexin, haptoglobin, fibrinogen, plasminogen, alpha 1-antitrypsin, orosomucoid, beta-lipoprotein, alpha 2-macroglobulin, and prealbumin in the human conceptus. J Clin Invest. 1969;48:1433–46.
- Golan HM, Lev V, Hallak M, Sorokin Y, Huleihel M. Specific neurodevelopmental damage in mice offspring following maternal inflammation during pregnancy. Neuropharmacology. 2005;48:903–17.
- Guller S, LaChapelle L. The role of placental fas ligand in maintaining immune privilege at maternal-fetal interfaces. Semin Reprod Endocrinol. 1999;17:39–44.
- 40. Hammer A, Blaschitz A, Daxbock C, Walcher W, Dohr G. Fas and fas-ligand are expressed in the uteroplacental unit of first-trimester pregnancy. Am J Reprod Immunol. 1999;41:41–51.
- Han P, Hodge G. Intracellular cytokine production and cytokine receptor interaction of cord mononuclear cells: relevance to cord blood transplantation. Br J Haematol. 1999;107:450–7.
- 42. Hayward AR, Simons MA, Lawton AR, Mage RG, Cooper MD. Pre-B and B cells in rabbits. Ontogeny and allelic exclusion of kappa light chain genes. J Exp Med. 1978;148:1367–77.
- Hebra A, Strange P, Egbert JM, et al. Intracellular cytokine production by fetal and adult monocytes. J Pediatr Surg. 2001;36:1321–6.
- 44. Hilkens CM, Vermeulen H, van Neerven RJ, et al. Differential modulation of T helper type 1 (Th1) and T helper type 2 (Th2) cytokine secretion by prostaglandin E2 critically depends on interleukin-2. Eur J Immunol. 1995;25:59–63.
- Hofman FM, Danilovs J, Husmann L, Taylor CR. Ontogeny of B cell markers in the human fetal liver. J Immunol. 1984;133:1197–201.
- Holt PG, Jones CA. The development of the immune system during pregnancy and early life. Allergy. 2000;55:688–97. http://onlinelibrary.wiley.com/ doi/10.1111/all.2000.55.issue-8/issuetoc
- Hunt DWC, Huppertz HI, Jiang HJ, Petty RE. Studies of human cord blood dendritic cells: evidence for functional immaturity. Blood. 1994;12:4333–43.
- Hynes RO. Integrins: versatility, modulation, and signaling in cell adhesion. Cell. 1992;69:11–25.
- 49. Inagaki H, Suzuki T, Nomoto K, Yoshikai Y. Increased susceptibility to primary infection with listeria monocytogenes in germfree mice may be due to lack

of accumulation of L-selectin + CD44+ T cells in sites of inflammation. Infect Immun. 1996;64:3280–7.

- Inlay MA, Serwold T, Mosley A, et al. Identification of multipotent progenitors that emerge prior to hematopoietic stem cells in embryonic development. Stem Cell Reports. 2014;2:457–72.
- Janeway Jr CA. The immune system evolved to discriminate infectious nonself from noninfectious self. Immunol Today. 1992;13:11–6.
- Janossy G, Bofill M, Poulter LW, et al. Separate ontogeny of two macrophage-like accessory cell populations in the human fetus. J Immunol. 1986; 136:4354–61.
- Johansen FE, Braathen R, Brandtzaeg P. Role of J chain in secretory immunoglobulin formation. Scand J Immunol. 2000;52:240–8.
- Jones AC, Miles EA, Warner JO, et al. Fetal peripheral blood mononuclear cell proliferative responses to mitogenic and allergenic stimuli during gestation. Pediatr Allergy Immunol. 1996;7:109–16.
- Kamps WA, Cooper MD. Microenvironmental studies of pre-B and B cell development in human and mouse fetuses. J Immunol. 1982;129:526–31.
- King CL, Malhotra I, Mungai P, et al. B cell sensitization to helminthic infection develops in utero in humans. J Immunol. 1998;160:3578–84.
- Kotiranta-Ainamo A, Rautonen J, Rautonen N. Interleukin-10 production by cord blood mononuclear cells. Pediatr Res. 1997;41:110–3.
- Kraft DL, Weissman IL, Waller EK. Differentiation of CD3-4-8- human fetal thymocytes in vivo: characterization of a CD3-4 + 8- intermediate. J Exp Med. 1993;178:265–77.
- Kubagawa H, Gathings WE, Levitt D, Kearney JF, Cooper MD. Immunoglobulin isotype expression of normal pre-B cells as determined by immunofluorescence. J Clin Immunol. 1982;2:264–9.
- Kutteh WH. Recurrent pregnancy loss. Obstet Gynecol Clin North Am. 2014;41:xi–xiii.
- Kwak DJ, Augustine NH, Borges WG, et al. Intracellular and extracellular cytokine production by human mixed mononuclear cells in response to group B streptococci. Infect Immun. 2000;68:320–7.
- Labeaud AD, Malhotra I, King MJ, King CL, King CH. Do antenatal parasite infections devalue childhood vaccination? PLoS Negl Trop Dis. 2009;3:e442.
- Levy O. Innate immunity of the newborn: basic mechanisms and clinical correlates. Nat Rev Immunol. 2007;7:379–90.
- 64. Liao SY, Liao TN, Chiang BL, et al. Decreased production of IFN gamma and increased production of IL-6 by cord blood mononuclear cells of newborns with a high risk of allergy. Clin Exp Allergy. 1996;26:397–405.
- Linch DC, Knott LJ, Rodeck CH, Huehns ER. Studies of circulating hemopoietic progenitor cells in human fetal blood. Blood. 1982;59:976–9.
- Macardle PJ, Weedon H, Fusco M, et al. The antigen receptor complex on cord B lymphocytes. Immunology. 1997;90:376–82.

- MacDonald TT, Weinel A, Spencer J. HLA-DR expression in human fetal intestinal epithelium. Gut. 1988;29:1342–8.
- Madsen-Bouterse SA, Romero R, Tarca AL, et al. The transcriptome of the fetal inflammatory response syndrome. Am J Reprod Immunol. 2010;63: 73–92.
- Malek A, Sager R, Kuhn P, Nicolaides KH, Schneider H. Evolution of maternofetal transport of immunoglobulins during human pregnancy. Am J Reprod Immunol. 1996;36:248–55.
- Mathieson BJ, Fowlkes BJ. Cell surface antigen expression on thymocytes: development and phenotypic differentiation of intrathymic subsets. Immunol Rev. 1984;82:141–73.
- McVay LD, Jaswal SS, Kennedy C, Hayday A, Carding SR. The generation of human gammadelta T cell repertoires during fetal development. J Immunol. 1998;160:5851–60.
- Meyer U, Feldon J, Yee BK. A review of the fetal brain cytokine imbalance hypothesis of schizophrenia. Schizophr Bull. 2009;35:959–72.
- Migliaccio G, Migliaccio AR, Petti S, et al. Human embryonic hemopoiesis. Kinetics of progenitors and precursors underlying the yolk sac—liver transition. J Clin Invest. 1986;78:51–60.
- Miller DL, Hiravonen T, Gitlin D. Synthesis of IgE by the human conceptus. J Allergy Clin Immunol. 1973;52:182–8.
- Mold JE, Venkatasubrahmanyam S, Burt TD, et al. Fetal and adult hematopoietic stem cells give rise to distinct T cell lineages in humans. Science. 2010; 330:1695–9.
- Mor G, Cardenas I. The immune system in pregnancy: a unique complexity. Am J Reprod Immunol. 2010;63:425–33.
- Morrison SJ, Scadden DT. The bone marrow niche for haematopoietic stem cells. Nature. 2014;505: 327–34.
- Morton H. Early pregnancy factor: an extracellular chaperonin 10 homologue. Immunol Cell Biol. 1998;76:483–96.
- Munn DH, Zhou M, Attwood JT, et al. Prevention of allogeneic fetal rejection by tryptophan catabolism. Science. 1998;281:1191–3.
- Namikawa R, Mizuno T, Matsuoka H, et al. Ontogenic development of T and B cells and nonlymphoid cells in the white pulp of human spleen. Immunology. 1986;57:61–9.
- Novato-Silva E, Gazzinelli G, Colley DG. Immune responses during human schistosomiasis mansoni. XVIII. Immunologic status of pregnant women and their neonates. Scand J Immunol. 1992;35:429–37.
- Opie TM, Shields LE, Andrews RG. Cell-surface antigen expression in early and term gestation fetal hematopoietic progenitor cells. Stem Cells. 1998; 16:343–8.
- Pahal GS, Jauniaux E, Kinnon C, Thrasher AJ, Rodeck CH. Normal development of human fetal hematopoiesis between eight and seventeen weeks'

gestation. Am J Obstet Gynecol. 2000;183: 1029–34.

- Palfi M, Selbing A. Placental transport of maternal immunoglobulin G. Am J Reprod Immunol. 1998;39:24–6.
- Palis J, Robertson S, Kennedy M, Wall C, Keller G. Development of erythroid and myeloid progenitors in the yolk sac and embryo proper of the mouse. Development. 1999;126:5073–84.
- 86. Papayannopoulou T, Craddock C, Nakamoto B, Priestley GV, Wolf NS. The VLA4/VCAM-1 adhesion pathway defines contrasting mechanisms of lodgement of transplanted murine hemopoietic progenitors between bone marrow and spleen. Proc Natl Acad Sci U S A. 1995;92:9647–51.
- Peault B, Tavian M. Hematopoietic stem cell emergence in the human embryo and fetus. Ann N Y Acad Sci. 2003;996:132–40.
- Piccinni MP, Giudizi MG, Biagiotti R, et al. Progesterone favors the development of human T helper cells producing Th2-type cytokines and promotes both IL-4 production and membrane CD30 expression in established Th1 cell clones. J Immunol. 1995;155:128–33.
- Prescott SL, Macaubas C, Holt BJ, et al. Transplacental priming of the human immune system to environmental allergens: universal skewing of initial T cell responses toward the Th2 cytokine profile. J Immunol. 1998;160:4730–7.
- Raff MC, Megson M, Owen JJ, Cooper MD. Early production of intracellular IgM by B-lymphocyte precursors in mouse. Nature. 1976;259:224–6.
- Romero R, Gotsch F, Pineles B, Kusanovic JP. Inflammation in pregnancy: its roles in reproductive physiology, obstetrical complications, and fetal injury. Nutr Rev. 2007;65:S194–202.
- Rookmaaker MB, Vergeer M, van Zonneveld AJ, Rabelink TJ, Verhaar MC. Endothelial progenitor cells: mainly derived from the monocyte/ macrophage-containing. Circulation. 2003;108:e150.
- Roth I, Corry DB, Locksley RM, et al. Human placental cytotrophoblasts produce the immunosuppressive cytokine interleukin 10. J Exp Med. 1996;184:539–48.
- 94. Saito S, Morii T, Umekage H, Makita K, Nishikawa K, Narita N, Ichijo M, Morikawa H, Ishii N, Nakamura M, Sugamura K. Expression of the interleukin-2 receptor gamma chain on cord blood mononuclear cells. Blood. 1996;87:3344–50.
- Sargent IL, Borzychowski AM, Redman CW. Immunoregulation in normal pregnancy and preeclampsia: an overview. Reprod Biomed Online. 2007;14(Spec No 1):111–7.
- 96. Shi L, Smith SE, Malkova N, et al. Activation of the maternal immune system alters cerebellar development in the offspring. Brain Behav Immun. 2009;23:116–23.
- Shimada S, Nishida R, Takeda M, et al. Natural killer, natural killer T, helper and cytotoxic T cells in the decidua from sporadic miscarriage. Am J Reprod Immunol. 2006;56:193–200.

- Smith JB, Tabsh KM. Fetal neutrophils and eosinophils express normal levels of L-selectin. Pediatr Res. 1993;34:253–7.
- Smith JB, Kunjummen RD, Kishimoto TK, Anderson DC. Expression and regulation of L-selectin on eosinophils from human adults and neonates. Pediatr Res. 1992;32:465–71.
- 100. Sohn DS, Kim KY, Lee WB, Kim DC. Eosinophilic granulopoiesis in human fetal liver. Anat Rec. 1993;235:453–60.
- 101. Solvason N, Kearney JF. The human fetal omentum: a site of B cell generation. J Exp Med. 1992; 175:397–404.
- 102. Stoltenberg L, Thrane PS, Rognum TO. Development of immune response markers in the trachea in the fetal period and the first year of life. Pediatr Allergy Immunol. 1993;4:13–9.
- 103. Sudo N, Sawamura S, Tanaka K, et al. The requirement of intestinal bacterial flora for the development of an IgE production system fully susceptible to oral tolerance induction. J Immunol. 1997;159: 1739–45.
- Szekeres-Bartho J, Wegmann TG. A progesteronedependent immunomodulatory protein alters the Th1/ Th2 balance. J Reprod Immunol. 1996;31:81–95.
- 105. Szekeres-Bartho J, Faust Z, Varga P, Szereday L, Kelemen K. The immunological pregnancy protective effect of progesterone is manifested via controlling cytokine production. Am J Reprod Immunol. 1996;35:348–51.
- 106. Szepfalusi Z, Nentwich I, Gerstmayr M, et al. Prenatal allergen contact with milk proteins. Clin Exp Allergy. 1997;27:28–35.
- 107. Thilaganathan B, Stagiannis K, Meher-Homji NJ, Plachouras N, Nicolaides KH. Fetal blood mononuclear cell division in normal and pathological pregnancies. Fetal Diagn Ther. 1994;9:79–83.
- 108. Thrane PS, Halstensen TS, Rognum TO, Brandtzaeg P. Expression of HLA class I and II (DR, DP and DQ) determinants in fetal and postnatal salivary glands. Scand J Immunol. 1991;34:539–48.

- 109. Tilburgs T, Scherjon SA, Claas FH. Major histocompatibility complex (MHC)-mediated immune regulation of decidual leukocytes at the fetal-maternal interface. J Reprod Immunol. 2010;85:58–62.
- 110. Tucci A, Mouzaki A, James H, Bonnefoy JY, Zubler RH. Are cord blood B cells functionally mature? Clin Exp Immunol. 1991;84:389–94.
- 111. Van Duren-Schmidt K, Pichler J, Ebner C, et al. Prenatal contact with inhalant allergens. Pediatr Res. 1997;41:128–31.
- 112. Von HP, Sarin S, Krowka JF. Deficiency in T cell responses of human fetal lymph node cells: a lack of accessory cells. Immunol Cell Biol. 1995;73: 353–61.
- 113. Warner JA, Miles EA, Jones AC, et al. Is deficiency of interferon gamma production by allergen triggered cord blood cells a predictor of atopic eczema? Clin Exp Allergy. 1994;24:423–30.
- 114. Wegmann TG, Lin H, Guilbert L, Mosmann TR. Bidirectional cytokine interactions in the maternal-fetal relationship: is successful pregnancy a TH2 phenomenon? Immunol Today. 1993;14:353–6.
- 115. Wiener E, Mawas F, Dellow RA, Singh I, Rodeck CH. A major role of class I Fc gamma receptors in immunoglobulin G anti-D-mediated red blood cell destruction by fetal mononuclear phagocytes. Obstet Gynecol. 1995;86:157–62.
- Wright SD. Toll, a new piece in the puzzle of innate immunity. J Exp Med. 1999;189:605–9.
- 117. Wucherpfennig KW, Liao YJ, Prendergast M, et al. Human fetal liver gamma/delta T cells predominantly use unusual rearrangements of the T cell receptor delta and gamma loci expressed on both CD4 + CD8- and CD4-CD8- gamma/delta T cells. J Exp Med. 1993;177:425–32.
- 118. Yu M, Fredrikson S, Link J, Link H. High numbers of autoantigen-reactive mononuclear cells expressing interferon-gamma (IFN-gamma), IL-4 and transforming growth factor-beta (TGF-beta) are present in cord blood. Clin Exp Immunol. 1995;101: 190–6.

Part VII

Fetal Hepatic Development: Up to Second Trimester

Growth and Development of Human Liver and Its Future Implications

25

Gopal Krishna Dhali and Gurubasava Lakamaji

Anatomy of Liver

Liver is the largest gland in the body comprising $\sim 5 \%$ of body mass in mammals. It performs important metabolic, endocrine and exocrine functions. Human liver normally weighs 1.44–1.66 kg and is located in right upper quadrant of abdomen. Two major types of cells populate the liver lobes; parenchymal and non-parenchymal cells. Eighty percent of the liver volume is occupied by parenchymal cells i.e. hepatocytes. Non-parenchymal cells constitute 40 % of the total number of liver cells but only 6.5 % of its volume. Sinusoidal hepatic endothelial cells, Kupffer cells and hepatic stellate cells are non-parenchymal cells.

The microscopic structure is conceptualized in several ways, the two most common being the acinus and the lobule. The acinus is a unit that contains a small portal tract at the center and terminal hepatic venules at the periphery. It is the smallest functional unit and is divided into zones 1, 2, and 3, wherein zone 1 surrounds the portal tract and zone 3 surrounds the hepatic venule. Alternatively, the traditional concept of lobule may also be used, in which the central structure is the terminal hepatic venule ("central vein") and the periphery is delineated by portal tracts [1] (Fig. 25.1).

The microscopic anatomy of the liver consists of clusters of cells called lobules. Each lobule, measuring about 1 mm in diameter, consists of numerous cords of hepatocytes, that radiate from central vein toward a thin layer of connective tissue that separates the lobule from other neighboring lobules. The cords of liver cells are one cell thick and are separated from one another on several surfaces by spaces called sinusoids, compartment between endothelial cells and hepatocytes that contains the microenvironment for exchange between blood and hepatocytes is called as 'Space of Disse' [2, 3].

The portal triad at periphery of the lobule contains three important structures, the portal vein, the hepatic artery and the bile duct. These structures lie in a network of connective tissue and are surrounded on all sides by hepatocytes. The ring of hepatocytes abutting the connective tissue of the triad is called the periportal limiting plate [1, 2].

Physiology of Liver

Liver performs many physiologically important functions.

G.K. Dhali, MD, DM • G. Lakamaji, MBBS, MD (⊠) Department of Gastroenterology, IPGMER and SSKM Hospital, 244, A.J.C. Bose Road 20, Kolkata, India

School of Digestive and Liver Diseases, Institute of Post Graduate Medical Education and Research, Kolkata 700020, India e-mail: gkdhali@yahoo.co.in; ravi.therisingsunat6@gmail.com

[©] Springer International Publishing Switzerland 2016 N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development: First and Second Trimesters*, DOI 10.1007/978-3-319-14874-8_25



Bile consist of bile acids, bile pigments, and lipid components. About 500 mL is secreted per day. Some of the components of the bile are reabsorbed in the intestine and then excreted again by the liver (enterohepatic circulation). It has important role in digestion and absorption of fats, and serves as route for lipid-soluble waste products excretion [4].

The various nutrients absorbed in GI tract are transported to liver through portal vein which are processed by hepatocytes before they enter systemic circulation. Liver plays key roles in carbohydrate metabolism such as glycogen storage, gluconeogenesis, and regulate blood glucose levels, it is also a key organ involved in cholesterol and ammonia metabolism [4, 5].

Liver inactivates and detoxifies various endogenous and exogenous substances. It is mediated in their first stages by cytochrome P-450 enzymes expressed in hepatocytes. These convert (xenobiotics) and other toxins to inactive, less lipophilic metabolites [4, 5].

Liver synthesizes albumin which accounts for the majority of plasma oncotic pressure. Other substances that are synthesized by liver are transport proteins for steroids and hormones, clotting factors, and various acute-phase proteins [4].

Introduction to Development of Liver

GI tract is principally derived from endoderm with contribution from other mesoderm and ectoderm. Around 16th day of human development. Endoderm-lined yolk sac cavity is formed due to horizontal and median folding of embryo into primitive gut. In the cephalic and caudal parts of the embryo, the primitive gut forms a blindending tube, the foregut and hindgut, respectively. The middle part, the midgut, remains temporally connected to the yolk sac by means of the vitelline duct.

The liver and biliary system arise from cells of the ventral foregut endoderm; the parenchyma of the liver (cords of hepatocytes and branched tubules of bile ducts) intercalates within the tissue of the septum transversum and the plexus of vitelline vessels, accounting for the overall architecture observed in the adult (plates of hepatocytes, which are endoderm derived, surrounded by vascular sinusoids, which are mesoderm derived)., the ventral bud of endoderm that gives rise to liver grows into the ventral mesentery, thus becoming surrounded by mesodermal tissue. The portion of the mesentery into which the liver grows is involved in the formation of the septum transversum. Thus the developing liver becomes enclosed in the septum. The mesodermal tissue gives rise to the fibrous capsule of Glisson and to the small amount of connective tissue within the gland. This developmental story is a complex process involving interaction between cells and various growth factors most important of this is BMP, HNF etc.

The first molecular evidence of liver development of liver is the expression of albumin, transthyretin, and alpha-fetoprotein in a region of the ventral endoderm. In vivo DNA/protein analyses

anatomy of liver

have shown that the albumin gene promoter is bound by Forkhead box (Fox) A and GATA-4 transcription factors at a stage that precedes transcription of the gene. FoxA and GATA factors decompact chromatin, binding of such factors opens chromatin to provide access to additional transcription factors resulting in the transcriptional activation of albumin [6, 7].

Hepatocyte nuclear factor (HNF)-1 stimulates expression of FoxA1 and FoxA2 in the prehepatic endoderm; in its absence, the endoderm cells lack competence and hepatic specification is impaired. Thus, HNF-1, FoxA1, FoxA2, and GATA-4 are critical initiators of liver development [8].

Extracellular signals act in specific endodemal cell lines and induce their specification to a hepatic fate. Fibroblast growth factors (FGFs) secreted by the cardiogenic mesoderm is one of such important factor. Hepatic gene induction by FGF is mediated by a mitogen-activated protein kinase signaling cascade [9] and is dependent on low concentrations of FGF [9].

Septum transversum also involved in regulation of liver development it produces bone morphogenetic protein (BMP)-2 and BMP-4, which along with FGF involved in hepatic gene expression. BMP saction is probably mediated by involvement of GATA-4 transcription factor [10].

Wnt signaling has got important role in liver development. When the primitive gut becomes patterned along its anteroposterior axis, Wnt signaling must be repressed anteriorly to maintain foregut identity and to allow subsequent liver development which is achieved by secretion of Wnt inhibitors by endoderm [11].

Liver Bud Formation

When endoderm cells have been specified, a diverticulum forms from the primitive gut. This occurs day 22 in humans. The liver diverticulum is lined by endoderm cells, which at this stage are called hepatoblasts, that become columnar and undergo a transition to a pseudostratified epithelium. The hepatoblasts proliferate and form a tissue bud delineated by a basement membrane that contains laminin, collagen IV, nidogen, fibronectin, and heparan sulfate proteoglycan. The hepatoblasts then leave the endoderm epithelium, migrate through the basement membrane, and invade the septum transversum [44].

A network of transcription factors controls the onset of liver development. Studies of knockout mice have shown that the hematopoietically expressed homeobox factor (Hex) promotes interkinetic nuclear migration and hepatoblast proliferation GATA-6 is required to maintain the differentiation state of the hepatoblasts. absence of either factor, liver development is arrested soon after budding. In prospero-related homeobox 1 (Prox-1)-deficient mice, hepatoblasts remain abnormally clustered in the liver bud. It has been proposed that the lack of Prox-1 leads to a failure of the hepatoblasts to delaminate from the liver bud and to migrate in the septum transversum, resulting from overexpression of E-cadherin, which holds the cells together by cell-cell contacts. Other factors involved in development are, HNF-6, Onecut (OC)-2, Tbx3 [44, 45].

There is necessity of continuous interactions between hepatoblasts and adjacent mesodermal tissues. When the activity of metalloproteases produced by the hepatoblasts and mesenchymal cells is blocked, migration of the hepatoblasts through the basement membrane is inhibited.

Also, the endothelial cells that surround the liver bud promote hepatoblast proliferation, as illustrated by the lack of liver expansion in mice with deficiencies in endothelial cell development.

Onset of Liver Development

The parenchymal cells of the liver i.e. hepatocytes are derived from the anterior portion of the endoderm. Mesothelial cells of septum transversum which are formed out of lateral plate mesoderm also have got important role to play in liver development. Coculture studies using chick embryos demonstrated that the developing cardiac mesoderm plays a crucial instructive role during the induction of hepatic cell fate. These inductive signals were found to be members of the fibroblast growth factor family, since FGF1 and FGF2 could substitute for cardiac tissue in inducing the onset of Albumin expression, a characteristic marker of hepatic cell fate, in explants of mouse endoderm. FGF-mediated specification of hepatic cell fate is concentration dependent and this appears to be controlled by the position of the endoderm relative to the heart [7, 12].

The induction of hepatic gene expression by FGF is controlled through activation of the MAPK pathway. FGFs expressed on cardiac mesoderm during embryogenesis include Fgf1, Fgf2, Fgf8, and Fgf10. Interestingly FGF signaling in liver development is evolutionarily conserved, with FGFs displaying hepatogenic properties in Xenopus, chick [13, 14].

GATA4, a zinc finger transcription factor, is also expressed in septum transversum mesenchymal cells that surround the liver bud and is required for the liver bud to expand. It regulates expression of the secreted bone morphogenetic protein, BMP4, which, like GATA4, is highly expressed in the septum transversum mesenchymal cells at an early stage of development. Analyses of Bmp4–/– mouse embryos also revealed a delay in expansion of the liver bud and addition of BMP inhibitors were found to block hepatic specification in vitro [15, 16].

Zaret and colleagues worked on a model whereby TGFb signaling acts to restrict endoderm specification as cell movements position progenitors within the appropriate inductive environments. In This proposed model, TGFb acts as a developmental timer to ensure that the endoderm retains hepatic competency and is prevented from inappropriately differentiating.

The WNT signaling pathway has also been implicated during the onset of hepatic development, the contribution of WNT signaling appear complex. WNT signaling makes different contributions depending on the developmental stage. At early somite stages WNT signaling acts in the posterior endoderm to repress expression of Hhex, an essential transcriptional regulator of hepatic development. Repression of WNT signaling by expression of WNT antagonists in the anterior endoderm is, therefore, required to relieve repression of Hhex in the anterior endoderm and so facilitate commitment of the endoderm to a hepatic fate. In contrast to the repressive effects of WNTs at early somite stages, following specification, WNT signaling appears to promote hepatogenesis. This factor is expressed in the lateral plate mesoderm that is positioned adjacent to the endoderm [11].

Transcriptional Regulation of Hepatogenesis

Multiple transcriptional factors act in association with cell signaling molecules to regulate development of liver. Albumin expression is seen in early stage of development by nascent hepatic cells. DNA footprinting analyses revealed that FoxA and GATA4 both of which are expressed in the anterior endoderm, bind to the Albumin enhancer and interact with their respective binding sites in the context of compacted chromatin and results in displacement of linker histone H1 and repositioning of nucleosomes and subsequently albumin synthesis [9]. But, hepatocyte differentiation and control of gene expression, following specification of the liver progenitors, seems to be independent of FoxA1/A2.

Hepatocyte growth factor (HGF) is expressed by the septum transversum, the endothelial cells, and the hepatoblasts. Its receptor, c-met, is found at the surface of hepatoblasts, where it initiates a signaling cascade mediated by SEK1/MKK4 and possibly by c-jun. This cascade promotes hepatoblast proliferation, as indicated by the reduced proliferation of cells in mice with disruptions in genes that encode HGF, c-Met, SEK1/MMK4 [46].

The transforming growth factor (TGF) B Smad2/Smad3 pathway also stimulates proliferation, because mice with heterozygous mutations in the Smad2 and Smad3 genes showed liver hypoplasia. HGF and TG pathways functionally interact. Cultured explants from livers deficient in Smad2 and Smad3 had reduced expression of A 1-integrin, but this reduction was rescued by the addition of HGF to the medium. It is likely that the HGF and TGF b pathways converge on a1-integrin expression, which is necessary for proliferation [47]. Hepatoma-derived growth factor (HDGF) is produced by fetal hepatoblasts and stimulates their proliferation when they are cultured in vitro. The expression of HDGF remains high at the hepatoblast stage and is extinguished when the hepatoblasts mature to hepatocytes [48].

Studies in mouse embryos have found that the homeodomain transcription factor HNF1b is also essential for hepatic specification. In embryos lacking HNF1b, the liver bud fails to express any markers of hepatic parenchymal cell progenitors and the level of mRNAs encoding the FoxA factors was severely reduced [8].

The homeobox transcription factor Hhex regulates proliferation and positioning of the ventral endoderm within the cardiogenic field. In the absence of Hhex, mutant mouse embryos initiate hepatic specification but fail to complete liver bud morphogenesis, resulting in hepatic structures that lack a parenchymal cell component. It is suggested that GATA4 and/or GATA6 may be involved in transactivating the Hhex promoter [17].

The prospero-related homeobox transcription factor Prox1 also promotes hepatoblast proliferation and migration from the primary liver bud. Although the mechanism through which Prox1 controls hepatoblast migration is unclear, the mutant hepatoblasts were found to maintain high levels of E-cadherin and failed to degrade the basal matrix surrounding the liver bud. Recent work has shown that the T box transcriptional repressor Tbx3 may act upstream of Prox1. In control embryos the expression of key regulators of hepatocyte differentiation including Hnf4a and c/EBPa are strongly expressed in the migrating hepatoblasts, whereas expression of transcription factors that primarily control cholangiocyte fate, such as HNF6 and HNF1b, are found to be at very low levels. In Tbx3 mutant mice expression of Hnf4a and c/EBPa is lost while levels of HNF6 and HNF1b are increased suggesting that Tbx3 normally promotes a hepatocyte fate and represses a cholangiocyte fate [18].

TNF stimulates a signaling cascade that activates the transcription factor nuclear factor kB

(NF-kB). NF-kB is composed of the subunits p50 and p65/RelA and is activated by IkB kinases. Mice deficient in IkB kinase k or IkB kinase gamma, showed massive apoptosis in the liver. Interestingly, the dramatic phenotypes of IkB kinas -null mice were rescued by the additional inactivation of the TNF receptor, indicating that NF-kB protects hepatoblasts against TNFinduced apoptosis.

Thus from above discussion it is obvious that formation of ventral bud and subsequent morphogenesis is regulated by multiple factors which are intricately connected and complex.

Seggregation of Hepatobiliary Lineage

Concept of the cell fate decision applies, in theory, to hepatoblasts that have reached a developmental stage at which they become committed either to the hepatocyte lineage or the cholangiocyte lineage. Hepatocytes have a number of specific markers that are already present at the endodermal stage, so the onset of hepatocyte differentiation cannot be accurately defined. The analysis of mice with a disruption in the gene encoding the transcription factor HNF-4 showed deficient expression at E12 of a number of hepatocytespecific proteins such as apolipoproteins or albumin, suggesting that HNF-4 is critical for hepatocyte fate determination [49].

Another way to determine the timing of lineage segregation is by analysis of expression of biliary markers. For example, the expression of the intermediate filament proteins cytokeratin 19 and 7 is considered a sign of biliary differentiation. However, because cytokeratin 19 is expressed at low levels in hepatoblasts and at increasing levels in differentiating cholangiocytes SRY-related HMG box transcription factor 9 (SOX9) is present in endodermal cells that line the hepatic diverticulum, but its expression disappears when the cells start to invade the septum transversum. SOX9 becomes reexpressed at later developmental stages, it is restricted to biliary cells. The expression of SOX9 at can therefore be considered the earliest indication biliary lineage differentiation.

Hepatocyte Maturation

Cells that have differentiated to the hepatocyte lineage undergo a process of maturation that consists of the progressive acquisition of morphology and physiologic functions. Several microarray analyses of gene expression at various stages of liver development support the view that maturation is a process that extends throughout development, even after birth. The transcription factors that control hepatocyte maturation nicely illustrate the concept of a "dynamic transcriptional network" [50].

Metabolic and cellular functions of hepatocytes is regulated by various transcriptional activators and related cofactors. In addition to the expression of hepatic genes throughout the parenchyma, expression of some genes is restricted to zonal regions that are often related to the position of the portal triad (periportal) or central veins (pericentral/perivenous) which lead to diversity of metabolic functions of hepatocytes depending on their location [18].

WNT/b-catenin signaling pathway are found to be involved in controlling the positional identity of hepatocytes within the liver lobule. b-catenin is important for zonal gene expression in perivenous hepatocytes and this activity is antagonized by adenomatous polyposis coli (APC) in the periportal regions. In studies involving mice that lack HNF4a in hepatocytes found an increase in periportal expression of a subset of perivenous expressed genes. HNF4a was found to directly interact with the glutamine synthetase enhancer suggesting that HNF4a inhibits expression of pericentral mRNAs in periportal hepatocytes possibly by recruiting the histone deacetylase protein HDAC1. Activation of the Wnt pathway converted hepatocytes that exhibited a periportal character to those that expressed perivenous markers. In response to activation of Wnt signaling, a transcription factor activated by b-catenin called LEF1 was found to physically interact with HNF4a [19–21].

Interestingly expression of several genes, including alphafetoprotein, H19, and Glypican-3, is seen in fetal hepatocytes but reduced in fully differentiated cells. This is often reversed as hepatic cells become cancerous. Recent studies of the alphafetoprotein promoter have revealed an important role of transcriptional repressors in controlling the transition of the gene expression profile within hepatocytes from fetal to an adult. Alphafetoprotein regulator 1 (Afr) was genetically defined as a locus that conferred a high level of AFP expression. Recently, genetic mapping studies revealed that the increase in AFP expression in adult livers was a consequence of a retrovirus insertion into the Zhx2 gene. Zhx2 in mediating repression of Afp expression, a direct interaction with the Afp promoter has not been identified, raising the possibility that Zhx2 regulates Afp indirectly. Zhx2 also represses expression of H19 and glypican-3 [22].

Hepatic Vasculature Development

The fetal liver is in contact with two major venous systems, the umbilical veins and the vitelline veins. The vitelline veins participate in the formation of the efferent venous system of the liver. The umbilical vein is the major afferent vessel in the fetal liver, but its presence is transient and it disappears after birth. When the umbilical vein collapses, the portal vein replaces it as the major afferent vein. Hepatic artery development occurs later than venous development. It starts to form along the intrahepatic portal vein within the parenchyma and gradually extends toward the periphery. The current model suggests that intrahepatic arterial development in humans is driven by the ductal plate that forms at the same stage and is a source of VEGF [23, 24].

Blood that is carried into liver enters into specialized vascular channels known as sinosoids. They are important route of exchange of various substances. Sinusoids are the first blood vessels to form during hepatogenesis, they develop by angiogenesis from existing vessels in the septum transversum mesenchyme. As development progresses, the sinusoidal endothelial cells gradually adopt the functional and structural characteristics of mature sinusoids. Although the molecular mechanisms that control growth and maturation of sinusoidal endothelial cells are not well defined, several studies support a role for Wnt signaling in their proliferation and differentiation. In particular, Wnt2 was shown to be expressed in rat hepatic sinusoidal endothelial cells and could increase their proliferation through activation of canonical b-catenin signaling. When Wnt2 levels were depleted it resulted in a decrease in expression of VEGF receptor 2 in rat sinusoidal endothelial cells. This implies that the autocrine activity of Wnt2 cooperates with VEGF signaling to control sinusoidal endothelial cell growth in the liver [24, 25].

Hematopoiesis

Soon after the liver progenitors invade the surrounding mesenchyme, the fetal liver is colonized by hematopoietic progenitors and transiently becomes the principal hematopoietic organ. Immature hepatic progenitor cells can generate an environment that supports hematopoiesis; however, when hepatic progenitor cells are induced to differentiate to a mature form, this is abolished leading to movement of hematopoietic stem cells from the fetal liver to the adult bone marrow, during this general timeframe. It involves activation of Wnt signaling pathways by progenitor cells. Hematopoietic cells within the fetal liver express the cytokine Oncostatin M (OSM) which enhance differentiation into hepatocyte, thus there is a dynamic interplay between the blood and parenchymal compartments within the fetal liver that controls the timing of both hepatogenesis and hematopoiesis [26].

Stellate Cells

Stellate cells to perform many important function such as they store vitamin A and to modulate hepatic microcirculation in response to endothelin signaling. After injury to liver cells, stellate cells can also become activated to adopt a myofibroblast character and chronic activation of stellate cells leads to liver fibrosis.

Initial research suggested that these cells could be of endodermal, neural crest, or mesenchymal origin. Recent studies have suggested that mesothelial cells derived from the proepicardium and septum transversum mesenchyme could give rise to both endothelial and stellate cells [28].

Hepatic stellate cells and derivatives of the septum transversum mesenchyme may contribute toward the fate of other hepatic cell lineages. The homeobox protein Hlx is expressed in the septum transversum and visceral mesenchyme. In Hlx-/- mouse embryos livers were severely hypoplastic containing only 3 % of the cells found in control liver, this could be due to regulation of expression of mitogens or growth factors by Hlx [27].

Kupffer Cells

Kupffer cells are resident macrophages on the surface of hepatic sinusoidal endothelial cells. They represent 15 % of the liver cell population and 50 % of resident macrophages in the body. There is no clear report on the role of Kupffer cells in liver organogenesis. However, some data suggest that Kupffer cells or their progenitors may be involved in maturation of erythrocytes during fetal liver hematopoiesis. They could be possibly of mesenchymal origin [29].

Development of Extrahepatic Bile Ducts

In the embryo approximately 5 mm in length, the hepatic diverticulum shows a protruding bud in its distal part. Some investigators accordingly distinguish in the hepatic diverticulum a cranial part (pars hepatica) and a caudal part (pars cystic).

The caudal bud or pars cystica grows in length and represents the anlage of the gallbladder, the cystic duct, and common bile duct (ductus choledochus). For up to 8 weeks of gestation, the extrahepatic biliary tree further develops through lengthening of the caudal part of the hepatic diverticulum. At 29 days after fertilization, the gallbladder anlage is visible as a right anterolateral dilatation along the distal half of the hepatic diverticulum, with a cystic duct present at 34 days. At that stage, the gallbladder and cystic duct are provided with a lumen. Outpockings appear in the gallbladder wall in the 42-mm embryo; folds develop on the interior surface of the bladder at the 78-mm stage [30].

Myoblasts develop around the 30-mm stage, resulting in the establishment of all three layers of the wall of the future gallbladder: the mucosa, the muscular layer, and the serosa [31].

The pars cystica of the hepatic diverticulum begins initially from the anterior side of the future duodenum. Approximately the fifth week, the duodenum rotates to the right, so that the attachment of the developing common bile duct becomes displaced to its definitive position on the dorsal side of the duodenum [32].

In the 34-day embryo, the common hepatic duct is a broad, funnel-like structure in direct contact with the developing liver, without a recognizable left or right hepatic duct. During the fifth week, a rapid entodermal proliferation takes place in the dilated funnel-shaped structure above the junction of common bile duct and cystic duct; this proliferation gives rise to several folds, resulting in several channels at the porta hepatis. The distal portions of the right and left hepatic ducts develop from the extrahepatic ducts and are clearly defined tubular structures by 12 weeks of gestation. The proximal portions of the main hilar ducts derive from the first intrahepatic ductal plates. The extrahepatic bile ducts and the developing intrahepatic biliary tree maintain luminal continuity from the very start of organogenesis throughout further development [33].

Hes 1, the protein product of the Hes 1 gene is expressed in the epithelium of the extrahepatic bile ducts throughout their development. In vivo loss of Hes 1 (a transcription factor directly regulated by the Notch signalling pathway) results in agenesis of the gallbladder and hypoplasia of the extrahepatic bile ducts. The mRNA of the onecut transcription factor HNF6 is expressed in mouse liver from the onset of its development, and is detected not only in hepatoblasts but also in the extrahepatic bile ducts and the gall bladder primordium. In the mouse, development of the gallbladder further requires normal functioning of the forkhead box f1 gene (foxf1 gene). Haploinsufficiency of the Foxf1 transcription factor results in gallbladder malformation, with inadequate external smooth muscle layer, insufficient mesenchymal cell number [34, 35].

Intra Hepatic Bile Ducts

Seventh week as the time point for the first appearance of the intrahepatic bile duct system and describes a junction between the extrahepatic bile ducts and the earliest intrahepatic bile duct structure at the time of its first appearance in the liver hilum [36].

There are multiple theories of intra hepatic bile radical development. One theory maintains that the intrahepatic biliary tree is derived from ingrowth of the epithelium of the extrahepatic ducts. Another postulates that the entire intrahepatic biledraining system develops from hepatocyte precursor cells. A third theory combines elements of both of the first two. Most investigators favor the second hypothesis [37].

The hepatoblastic origin of the intrahepatic bile duct system received additional support from several studies that reinvestigated the embryologic development of the intrahepatic bile ducts with immunohistochemical techniques. These studies made use of immunohistochemical stains for cytokeratins, tissue polypeptide antigen, carcinoembryonic antigen, epithelial membrane antigen, and other markers for parenchymal and bile duct cell phenotypes [37].

Cytokeratins are the intermediate filaments of the cytoskeleton characteristic for epithelial cells. Different cytokeratins have been identified and catalogued. Normal adult human liver parenchymal cells express only the cytokeratins 8 and 18; intrahepatic bile duct cells express in addition the cytokeratins 7 and 19 and 20 [38].

Around the eighth week of gestation, the primitive hepatoblasts adjacent to the mesenchyme around the largest hilar portal vein branches become more strongly immunoreactive for their cytokeratins 8, 18, and 19. This layer of cells, surrounding the portal vein branches like a cylindrical sleeve, is termed the ductal plate. Recent studies unraveled that biliary cell differentiation is induced in the fetal liver by a periportal gradient of activin/transforming growth factor-beta (TGFb) signaling, the extent of which is controlled by the inhibitory influence of HNF6 and the Onecut factor OC-2 [39].

Forkhead Box m1 transcription factor (Foxm1b) is also essential for normal intrahepatic bile duct cell differentiation, besides development of vessels and sinusoids. The ductal plates adjacent to the mesenchyme (the portal ductal plate layer) become duplicated by a second layer of more keratin-rich cells over variably long segments of their perimeter (the second or lobular ductal plate layer; During the following weeks, ductal plates also appear around the smaller portal vein branches at a distance from the hilum. In the meantime, the hepatoblasts not involved in ductal plate formation gradually lose cytokeratin 19, and by 14 weeks of gestation the future parenchymal cells are immunoreactive only for cytokeratins 8 and 18, the cytokeratin pair normally expressed in adult liver parenchymal cells [40].

From approximately 12 weeks of gestation onward, a progressive "remodeling" of the ductal plates takes place, starting again in the earliest ductal plates around the larger portal vein branches near the hilum. Over short segments of the perimeter of the double-layered ductal plates, a "tubular" dilatation occurs of the slitlike lumen, lined now by taller keratin-rich cells. These cells acquire epithelial membrane antigen and lose their biliary glycoprotein I. The "tubular" parts of the ductal plate become incorporated into the mesenchyme surrounding the portal vein (the future portal tract) by ingrowth of mesenchyme between the lobular ductal plate layer and the hepatoblasts. The tubules incorporated into the portal mesenchyme are the future portal ducts; they remain connected, however, to the ductal plate and its adjacent parenchyma by thin epithelial channels, assuring continuity between the portal ducts and the already developed canalicular network located between the primitive hepatocyte precursor cells [41].

By 20 weeks of gestation, weak immunoreactivity for cytokeratin 7 appears in the cells of the developing ducts, again appearing first in the older ducts near the hilum. The immunoreactivity for cytokeratin 7 gradually increases and extends into more peripheral ducts, to reach the level of immunoreactivity observed in ducts of the adult liver at approximately 1 month after birth. At the time of birth, the most peripheral branches of the intrahepatic biliary tree are still immature: The finest portal vein radicles still are surrounded by ductal plates, which require an additional 4 weeks after birth to develop into small portal ducts. This finding indicates some degree of immaturity and incompleteness of the intrahepatic bile duct system in the neonate and may explain the lower ratio of the total number of bile ducts to the total number of portal tracts in premature infants [42].

The factors determining the developmental fate of the bipotential hepatoblasts remain incompletely known. There is evidence that components of the portal mesenchyme are crucial for inducing the phenotypic shift into bile duct-type cells in the layers of the ductal plate Several components, including laminin, fibonectin, collagen types I and IV, expression of corresponding cellular integrins, catenin, and N-CAM have all been shown to be involved. The process of intrahepatic bile duct development comprises schematically the following components:

- (a) A gradual phenotypic change of the hepatoblasts toward bile duct type cells;
- (b) A remarkable remodeling of the tridimensional structure of the ductal plate; and
- (c) An ongoing further maturation of the remodeled, tubular ducts [43].

The remodeling of the ductal plate involves epithelial changes—construction of new epithelial structures (by proliferation) and the simultaneous deletion of other parts (by apoptosis), mesenchymal influence, and determination by the portal vein. Autocrine stimulation of ductal plate cell proliferation is suggested by the immunohistochemical positivity for transforming growth factor-a, hepatocyte growth factor, and parathyroid hormone-related peptide. Apoptosis in the remodeling ductal plate is indicated by positive histochemical terminal deoxynucleotidyl transferase mediated bio-deoxy UTP nick end labeling (TUNEL) staining and expression of Fas, C-myc, and Lewisy.

References

- Krishna M. Microscopic anatomy of the liver. Clin Liver Dis. 2013;2(S1):S4–7.
- Suriawinata AA, Thung SN. Liver. In: Mills SE, editor. Histology for pathologists. 3rd ed. Philadelphia: Lippincott Williams & Wilkins; 2007. p. 685–703.
- Crawford AR, Lin X-Z, Crawford JM. The normal adult human liver biopsy: a quantitative reference standard. Hepatology. 1998;28:323–31.
- Sendensky A, Dufour J-F. Liver physiology. In: Ginès P, et al, editors. Liver physiology clinical gastroenterology: chronic liver failure. New York: Springer Science; 2011.
- Barrett KE, Barman SM, Boitano S, Brooks HL. Ganong's review of medical physiology. 23rd ed. Mc Graw Hill – Lange Publication, USA.
- Gualdi R, Bossard P, Zheng M, et al. Hepatic specification of the gut endoderm in vitro: cell signaling and transcriptional control. Genes Dev. 1996;10:1670–82.
- Jung J, Zheng M, Goldfarb M, et al. Initiation of mammalian liver development from endoderm by fibroblast growth factors. Science. 1999;284:1998–2003.
- Lokmane L, Haumaitre C, Garcia-Villalba P, et al. Crucial role of vHNF1 in vertebrate hepatic specification. Development. 2008;135:2777–86.
- Zaret KS. Genetic programming of liver and pancreas progenitors: lessons for stem-cell differentiation. Nat Rev Genet. 2008;9:329–40.
- Huang H, Ruan H, Aw MY, et al. Mypt1-mediated spatial positioning of Bmp2-producing cells is essential for liver organogenesis. Development. 2008;135: 3209–18.
- McLin VA, Rankin SA, Zorn AM. Repression of Wnt/ beta-catenin signaling in the anterior endoderm is essential for liver and pancreas development. Development. 2007;134:2207–17.
- Serls AE, Doherty S, Parvatiyar P, Wells JM, Deutsch GH. Different thresholds of fibroblast growth factors pattern the ventral foregut into liver and lung. Development. 2005;132:35–47.
- Calmont A, Wandzioch E, Tremblay KD, Minowada G, Kaestner KH, Martin GR, Zaret KS. An FGF response pathway that mediates hepatic gene induction in embryonic endoderm cells. Dev Cell. 2006;11:339–48.
- Shin D, Shin CH, Tucker J, Ober EA, Rentzsch F, Poss KD, Hammerschmidt M, Mullins MC, Stainier DY. Bmp and Fgf signaling are essential for liver specification in zebrafish. Development. 2007;134:2041–50.
- Watt AJ, Zhao R, Li J, Duncan SA. Development of the mammalian liver and ventral pancreas is dependent on GATA4. BMC Dev Biol. 2007;7:37.
- Rossi JM, Dunn NR, Hogan BL, Zaret KS. Distinct mesodermal signals, including BMPs from the septum transversum mesenchyme, are required in combination for hepatogenesis from the endoderm. Genes Dev. 2001;15:1998–2009.
- Keng VW, Yagi H, Ikawa M, Nagano T, Myint Z, Yamada K, Tanaka T, Sato A, Muramatsu I, Okabe M,

et al. Homeobox gene Hex is essential for onset of mouse embryonic liver development and differentiation of the monocyte lineage. Biochem Biophys Res Commun. 2000;276:1155–61.

- Jungermann K, Katz N. Functional specialization of different hepatocyte populations. Physiol Rev. 1989; 69:708–64.
- Burke ZD, Reed KR, Phesse TJ, Sansom OJ, Clarke AR, Tosh D. Liver zonation occurs through a betacatenin-dependent, c-Myc-independent mechanism. Gastroenterology. 2009;136:2316–24.
- Stanulovic' VS, Kyrmizi I, Kruithof-de Julio M, Hoogenkamp M, Vermeulen JL, Ruijter JM, Talianidis I, Hakvoort TB, Lamers WH. Hepatic HNF4alpha deficiency induces periportal expression of glutamine synthetase and other pericentral enzymes. Hepatology. 2007;45:433–44.
- Kaestner KH. In the zone: how a hepatocyte knows where it is. Gastroenterology. 2009;137:425–7.
- 22. Perincheri S, Dingle RW, Peterson ML, Spear BT. Hereditary persistence of alpha-fetoprotein and H19 expression in liver of BALB/cJ mice is due to a retrovirus insertion in the Zhx2 gene. Proc Natl Acad Sci U S A. 2005;102:396–401.
- 23. Gouysse G, Couvelard A, Frachon S, Bouvier R, Nejjari M, Dauge MC, Feldmann G, He'nin D, Scoazec JY. Relationship between vascular development and vascular differentiation during liver organogenesis in humans. J Hepatol. 2002;37:730–40.
- Collardeau-Frachon S, Scoazec JY. Vascular development and differentiation during human liver organogenesis. Anat Rec (Hoboken). 2008;291:614–27.
- Couvelard A, Scoazec JY, Dauge MC, Bringuier AF, Potet F, Feldmann G. Structural and functional differentiation of sinusoidal endothelial cells during liver organogenesis in humans. Blood. 1996;87:4568–80.
- Martin MA, Bhatia M. Analysis of the human fetal liver hematopoietic microenvironment. Stem Cells Dev. 2005;14:493–504.
- Hentsch B, Lyons I, Li R, Hartley L, Lints TJ, Adams JM, Harvey RP. Hlx homeo box gene is essential for an inductive tissue interaction that drives expansion of embryonic liver and gut. Genes Dev. 1996;10:70–9.
- Asahina K, Tsai SY, Li P, Ishii M, Maxson Jr REJ, Sucov HM, Tsukamoto H. Mesenchymal origin of hepatic stellate cells, submesothelial cells, and perivascular mesenchymal cells during mouse liver development. Hepatology. 2009;49:998–1011.
- Naito M, Hasegawa G, Ebe Y, Yamamoto T. Differentiation and function of Kupffer cells. Med Electron Microsc. 2004;37:16–28.
- Tan CEL, Moscoso GJ. The developing human biliary system at the porta hepatis level between 29 days and 8 weeks of gestation: a way to understanding biliary atresia. Part 1. Pathol Int. 1994;44:587–99.
- Das KM, Squillante L, Chitayet D, Kalousek DK. Simultaneous appearance of a unique common epitope in fetal colon, skin and biliary epithelial cells. A possible link for extracolonic manifestations in ulcerative colitis. J Clin Gastroenterol. 1992;15:311–6.

- Dubois AM. The embryonic liver. In: Rouiller C, editor. The liver. New York: Academic; 1963. p. 1–39.
- Tan J, Hytiroglou P, Wieczorek R, Park YN, Thung SN, Arias B, Theise ND. Immunohistochemical evidence for hepatic progenitor cells in liver diseases. Liver. 2002;22:365–73.
- 34. Kalinichenko VV, Zhou Y, Bhattacharyya D, Kim W, Shin B, Bambal K, Costa RH. Haploinsufficiency of the mouse Forkhead Box f1 gene causes defects in gall bladder development. J Biol Chem. 2002; 277:12369–74.
- Mahlapuu M, Enerback S, Carlsson P. Haploinsufficiency of the forkhead gene Foxf1, a target for sonic hedgehog signaling, causes lung and foregut malformations. Development. 2001;128:2397–406.
- Ruebner BH, Blankenberg TA, Burrows DA, Soohoo W, Lund JK. Development and transformation of the ductal plate in the developing human liver. Pediatr Pathol. 1990;10:55–68.
- Desmet VJ, Van Eyken P, Sciot R. Cytokeratins for probing cell lineage relationships in developing liver. Hepatology. 1990;12:1249–51.
- 38. Faa G, Van Eyken P, Roskams T, Miyazaki H, Serreli S, Ambu R, Desmet V. Expression of cytokeratin 20 in developing rat liver and in experimental models of ductular and oval cell proliferation. J Hepatol. 1998;29:628–33.
- 39. Clotman F, Jacquemin P, Plumb-Rudewiez N, Pierreux CE, Van der Smissen P, Dietz HC, Courtoy PJ, Rousseau GG, Lemaigre FP. Control of liver cell fate decision by a gradient of TGF beta signaling modulated by Onecut transcription factors. Genes Dev. 2005;19:1849–54.
- 40. Krupczak-Hollis K, Wang X, Kalinichenko VV, Gusarova GA, Wang IC, Dennewitz MB, Yoder HM, Kiyokawa H, Kaestner KH, Costa RH. The mouse Forkhead Box m1 transcription factor is essential for hepatoblast mitosis and development of intrahepatic

bile ducts and vessels during liver morphogenesis. Dev Biol. 2004;276:74–88.

- Blankenberg TA, Lund JK, Reubner BH. Normal and abnormal development of human intrahepatic bile ducts. An immunohistochemical perspective. Basel: Karger; 1991.
- Kahn E, Markowitz J, Aiges H, Daum F. Human ontogeny of the bile duct to portal space ratio. Hepatology. 1989;10:21–3.
- Balistreri W. Concluding remarks. 5th International Sendai Symposium on Biliary Atresia. In: Ohi R, editor. Biliary atresia. Tokyo: ICOM Associates; 1991. p. 293–7.
- Bort R, Signore M, Tremblay K, et al. Hex homeobox gene controls the transition of the endoderm to a pseudostratified, cell emergent epithelium for liver bud development. Dev Biol. 2006;290:44–56.
- Shiojiri N, Sugiyama Y. Immunolocalization of extracellular matrix components and integrins during mouse liver development. Hepatology. 2004;40: 346–55.
- Nishina H, Vaz C, Billia P, et al. Defective liver formation and liver cell apoptosis in mice lacking the stress signaling kinase SEK1/MKK4. Development. 1999;126:505–16.
- 47. Fässler R, Meyer M. Consequences of lack of beta 1 integrin gene expression in mice. Genes Dev. 1995;9: 1896–908.
- Enomoto H, Yoshida K, Kishima Y, et al. Hepatomaderived growth factor is highly expressed in developing liver and promotes fetal hepatocyte proliferation. Hepatology. 2002;36:1519–27.
- Li J, Ning G, Duncan SA. Mammalian hepatocyte differentiation requires the transcription factor HNF-4alpha. Genes Dev. 2000;14:464–74.
- Petkov PM, Zavadil J, Goetz D, et al. Gene expression pattern in hepatic stem/progenitor cells during rat fetal development using complementary DNA microarrays. Hepatology. 2004;39:617–27.

Embryonic Development of Liver and Its Future Implications

26

Abhijeet Chaudhuri

Introduction

The growth and maturation of human liver, as an organ, keeps pace with the structural and metabolic needs of the developing fetus and is responsive to the highly dynamic milieu that fetal growth entail. Understanding aspects of this development, including its' regulation, is an intriguingly important matter in molecular and cell biology. Aberrations in development can give rise to human diseases that have health impact. More importantly, liver is the abode for a variety of stem cells - both for the bone marrow derived that are pluripotent and home to the liver as well as the hepatic progenitor cells (HPC) that are more differentiated, oval cells playing roles in hepatocyte regeneration and response to injury. Development of the liver entail organization of the micro and macro elements of this organ in a simplistic view. More consequential of understanding this mechanism would be to serve as a reserve source for replenishment and refurbishment of the organ in times of repair and replacement during liver

Department of Hepatology and Transplantation, School of Digestive and Liver Diseases, Institute of Post Graduate Medical Education & Research and SSKM Hospital, Kolkata 700020, India e-mail: achowdhury2002@yahoo.co.in injury and cell loss. Everyday, we are learning this with greater precision and with the advent of molecular tools as well as refinements in cell biology, understanding of normal and abnormal development of the liver is fairly robust now. This has implications that extend far beyond just enrichment of the knowledge. Understanding the process of development might enable us to create new cell therapies- growing tissues ex vivo for use in transplantation or for coaxing cells in vivo to acquire characteristics that can restore function in disease states.

Adult Liver Architecture/ Morphology

Grossly adult liver measures 12–15 cm coronally and 15–20 cm transversely. The median liver weight is 1,800 g in men and 1,400 g in women The adult liver weight is between 1.8 and 3.1 % of body weight in 80 % of individuals [1, 2]. While liver weights in fetuses and children are relatively greater, being 5.6 % at 5 months' gestational age, 4–5 % at birth, and 3 % at 1 year of age [3, 4].

The falciform ligament anteriorly and the lesser omentum and umbilical fissure posteriorly divide the liver into the conventional right and left lobes.

Histological sections through the liver reveal a rather homogeneous landscape of

A. Chaudhuri, MD, DM

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_26

hepatocytes periodically infiltrated with vascular tissue and bile ducts. The basic architectural unit of the liver is the liver lobule. The lobule consists of plates of hepatocytes lined by sinusoidal capillaries that radiate toward a central efferent vein. Liver lobules are roughly hexagonal with each of six corners demarcated by the presence of a portal triad of vessels consisting of a portal vein, bile duct, and hepatic artery (Figs. 26.1 and 26.2). Both the portal vein and hepatic artery supply blood to the lobule, which flows through a network of sinusoidal capillaries before leaving the lobule through the central vein. Although hepatocytes are the major parenchymal cell type of the liver and account for 78 % of liver volume [5], they function in concert with cholangiocytes (biliary epithelial cells), endothelial cells, sinusoidal endothelial cells, Kupffer cells (resident liver macrophages), pit cells (natural killer cells), and hepatic stellate cells. Predominant cell types and their functions within adult liver are shown in Table 26.1 [6].

The hepatocytes, which are polarized epithelial cells, are arranged as cords that are one cell thick in mammals. The basolateral surfaces of the hepatocyte face fenestrated sinusoidal endothelial cells, which facilitates the transfer of endocrine secretions from the hepatocytes into the blood stream. Tight junctions formed between neighboring hepatocytes generate a canaliculus that surrounds each hepatocyte and is responsible for collection of bile acids and bile salts that are transported across the hepatocyte's apical surface. Bile collected by the canaliculi is carried to the bile ducts within the portal triad and subsequently transported for storage in the gall bladder. The complex arrangement between the polarized hepatocytes with the capillaries and cholangiocytes underlies both endocrine and exocrine functions of the liver.

Normal Liver Histology



Fig. 26.1 Showing arrangement of hepatocytes in cord around central vein



Fig. 26.2 Showing portal traid

Cell type	Position in liver	Function
Hepatocyte	Parenchyma	70 % of liver cell population
		Protein secretion
		Bile secretion
		Cholesterol metabolism
		Detoxification
		Urea metabolism
		Glucose/glycogen metabolism
		Acute phase response
		Blood clotting
Cholangiocyte/bile duct cell	Duct epithelium	3 % of liver cell population
		Form bile ducts to transport bile
		Control rate of bile flow
		Secrete water and bicarbonate
		Control pH of bile
Endothelial cell	Vasculature	Form veins, arteries, venuoles, and arterioles
		Control blood flow
		_
Liver sinusoidal endothelial cells		2.5 % of lobular parenchyma
		Form sinusoidal plexus to facilitate blood circulation
		Highly specialized
		Allow transfer of molecules and proteins between serum and hepatocytes
		Scavenger of macromolecular waste
		Cytokine secretion
		Antigen presentation
		Blood clotting
Pit cells	Liver natural killer cells	Rare
		Cytotoxic activity
Kupfer cells	Sinusoids	2 % of liver cells
		Scavengers of foreign material
		Secrete cytokines and proteases
Hepatic stellate cells	Perisinusoidal	1.4 % of liver cells
		Maintainence of extracellular matrix,
		vitamin A, retinoid storage
		Control of microvascular tone
		Activated to become myofibroblast
		Contributes towards regenerative response to injury
		Secretion of cytokine

 Table 26.1
 Various cell types and their functions in adult liver [6]

With permission from Elsevier

Embryologic Liver Morphogenesis

Liver primordium appears in the middle of third week as an outgrowth of endodermal epithelium of foregut (mouse e9). This outgrowth consists of rapidly proliferating cells that penetrate the septum transversum mesenchyme forming, what is called liver bud. As this grows connection between this liver bud and foregut narrows which forms the bile duct. A small ventral out growth from bile duct gives rise to gall bladder and common bile duct.

During further development these form epithelial liver cords which intermingle with vitteline and umbilical veins which forms hepatic sinusoids. These liver cords differentiate in to hepatocytes and biliary epithelial cells, while kupffer cells, connective tissue cells, hematopoietic cells are derived from septum transversum mesenchyme. In the tenth week weight of liver is approximately 10 % of body weight that is due to large no. of sinusoids with hematopoietic progenitors colonizing these sinusoids making it as principal hematopoietic organ (Fig. 26.5). But in last 2 months these hematopoietic progenitor cells decrease in number and liver constitute only 5 % of total body weight [7].

Carnegie stages based on the external and/or internal morphological development of the embryo, developed in Carnegie institute of Washington after studying embryos up to 8 weeks described the liver developmental events according to stages. Summary of key events described are given in Table 26.2

Development of Sinusoids and Portal Vein [7]

In the fifth week three major can be distinguished (1) Vitteline veins or omphalomesenteric veins carrying blood from yolk sac to sinus venosus (2) Umbilical veins originating in chorionic villi and carrying oxygenated blood to embryo (3) Cardinal veins draining the embryo proper.

These beds converge on the sinal horns before entering the heart. The left and right vitelline veins are joined by three anastomoses to form a
 Table 26.2
 Summary of key events in liver development according to Carnegie staging (source-UNSW Embryology)

Carnegie stage	Feature	
Stage 11 (29 days)	Hepatic diverticulum development	
Stage 12 (30 days)	Cell differentiation	
	Septum transversum forming liver stroma	
	Hepatic diverticulum forming hepatic trabeculae	
Stage 13 (32 days)	Epithelial cord proliferation enmeshing stromal capillaries	
Stage 14 (33 days)	Hepatic gland and its vascular channels enlarge	
	Hematopoietic function appeared	
Stage 18 (44 days)	Obturation due to epithelial proliferation	
	Bile ducts became reorganized (continuity between liver cells and gut)	
Stage 18–23 (44–56 days)	Biliary ductules developed in periportal connective tissue	
	Produces ductal plates that receive biliary capillaries	

ladder-like structure with the intestinal tract intertwined. The liver cords growing in to septum interrupt the course of veins and extensive vascular network, the hepatic sinusoids. Extrahepatic portal vein develops from these vessels after selective obliteration of portions of the ladder. The left vitelline vein receives a tap from the left umbilical vein. The intrahepatic segment of this tap becomes the umbilical portion of the left portal vein. Flow in this segment reverses after birth and supplies segments of the left hemiliver. As the liver develops, the venous drainage of the parenchyma becomes focused into two vessels, the future right and left hepatic veins, and later the middle vein (not shown), which usually drains into the left hepatic vein. The ductus venosus develops as a through-channel from the left portal vein to the common hepatic vein. The remainder of the portal vein blood perfuses sinusoids before reaching the hepatic veins. The vasculature simplifies with the removal of several segments including the most caudal anastomosis between the vitelline veins, the rostral portions of the left vitelline and left umbilical veins, and the right umbilical vein. The right lobe has grown

faster than the left as the left lobe has lost the supply from the left vitelline vein and left umbilical vein blood is shunted through the ductus venosus. The left umbilical vein actually lies in the midline and later shifts to the right of midline.

Development of the Biliary Tree [8]

For up to 8 weeks of gestation, the extrahepatic biliary tree develops through lengthening of the caudal part of the hepatic diverticulum. The hepatic duct (ductus hepaticus) develops from the cranial part (pars hepatica) of the hepatic diverticulum. The distal portions of the right and left hepatic ducts develop from the extrahepatic ducts and are clearly defined tubular structures by 12 weeks of gestation. The proximal portions of the main hilar ducts derive from the first intrahepatic ductal plates. The extrahepatic bile ducts and the developing intrahepatic biliary tree maintain luminal continuity from the very start of organogenesis throughout further development, contradicting a previous study in the mouse suggesting that the extrahepatic bile duct system develops independently from the intrahepatic biliary tree and that the systems are initially discontinuous but join up later. The normal development of intrahepatic bile ducts requires finely timed and precisely tuned epithelial - mesenchymal interactions, which proceed from the hilum of the liver toward its periphery along the branches of the developing portal vein (Figs. 26.3, 26.4, and 26.5).



Fig. 26.3 Histological section of liver at 8 weeks of development showing radiating pattern of sinusoids around central vein



Fig. 26.4 Histological section of liver at 8 weeks showing presence of erythrocytes within central vein and development of endothelial lining



Fig. 26.5 Histological sections of liver at 10 weeks showing hematopoietic cells within sinusoids

Liver Development as Seen in Mouse Embryo

The embryonic liver originates from the ventral foregut endoderm [9], which becomes the hepatic diverticulum, the first morphological sign of the embryonic liver. The hepatic diverticulum is located adjacent to the developing heart and can be identified at E9.0. The anterior portion of the hepatic diverticulum gives rise to the liver and intrahepatic biliary tree, whereas the posterior portion forms the gall bladder and the extrahepatic bile ducts (EHBD). At E9.5, hepatoblasts delaminate from the anterior portion of the

hepatic diverticulum and invade the adjacent septum transversum mesenchyme (STM) to form the liver bud [10]. The STM contributes the fibroblasts and stellate cells of the liver. Between E9.5 and E15, the liver bud undergoes significant growth and, following invasion of hematopoietic cells, becomes the primary site of fetal haematopoiesis.

Hepatoblasts are bi-potential precursor cells which give rise to hepatocytes and biliary epithelial cells (BECs; also known as cholangyocytes), the two major liver cell types. Hepatocytes are parenchymal cells that comprise about 70 % of hepatic cells [11, 12]. BECs line the lumen of the intrahepatic bile ducts (IHBD). The remaining 30 % of the adult liver is comprised of nonparenchymal cells, including Kupffer cells, stromal cells and stellate cells of mesodermal origin.

Triggers for Development

A large no. of transcription factors have been identified which act as signals/triggers for fate specification of cells, differentiation and patterning (regulated assignment of fates over a large area in a manner that ensures proper positional relationship between tissues. This signaling can be cell-cell signaling between neighboring cells or between adjacent cell layers. Such signals can be divided into two classes: permissive signals, which allow a tissue to progress to a fate that has already been assigned, and instructive or induc*tive* signals, which divert a tissue to a new fate that would not otherwise have been followed. Instructive signals play an important role in regulating patterning by committing multipotent cells to specific lineages [13].

Developmental signals have traditionally been identified through transplantation studies, in which different embryonic structures (e.g., epithelium and mesenchyme) are cocultured. The resulting fate (or absence thereof) indicates whether signals are present or absent, and if present, whether the signals are permissive or instructive.

But these signals cannot be considered simply as cause and effect. There occurs reciprocal interaction for example, the epithelium may respond to signal "A" from the mesenchyme by supplying signal "B", which in turn prompts the mesenchyme to secrete signal "C", and so forth. And if one chain of signals is broken, abnormal development ensues, or in other way epithelium may require sequential signals e.g. signal A followed by B and then C.

As development is a highly dynamic process, cells and cell layers are in constant motion relative to each other. Cell or tissue interactions may exist only transiently – long enough for a signal to be received, but not long enough to be easily characterized experimentally. The number of secreted factors encoded in the genome is vast, leading to the potential for a level of signaling complexity that may preclude straight forward analysis.

Sequence of events occurring during liver development and triggers involved can be summarized as

1. Endodermal fate assignment

Definitive endoderm emerges as a sheet of cells from the anterior end of the primitive streak during gastrulation. Signaling by the TGF β growth factor *Nodal* initiates both endoderm and mesoderm formation in a concentration-dependant manner, with low Nodal doses inducing mesoderm and higher doses inducing endoderm [14, 15]

2. Gut tube patterning and assignment of hepatic fate to endodermal cells

Throughout gastrulation and the early somite stages of development morphogenetic movements turn the endoderm into an epithelial gut tube surrounded by mesoderm. During this time the gut tube epithelium is further patterned along the anterior-posterior (A-P) axis into foregut, midgut and hindgut domains by secreted factors from the adjacent mesoderm. These domains can be identified by the expression of transcription factors such as Hhex in the foregut, Pdx1 in the midgut and Cdx in the posterior endoderm [16, 17]. Establishment of the foregut progenitors is important step in hepatogenesis because in vivo only the foregut endoderm (but not the hindgut) is competent to develop into the liver [18-21]. This intrinsic hepatic potential is probably due to the expression of transcription factors such as Foxa2, Gata4–6 and Hhex, which have important roles in early foregut organogenesis. so these transcription factors give foregut endoderm, anterior endoderm fate.

At early somite stages of development (e8.5; four to seven somites stage in mice), FGF signals from the developing heart and BMPs from the STM induce hepatic fate in the ventral foregut endoderm [18, 22-25]. This has been extensively studied using mouse embryo foregut explants, which when isolated at the two to four-somite stage will express Albumin after 1-2 days in culture if the cardiac mesoderm is present. However, if the cardiac mesoderm is removed, or if either FGF or BMP signaling is blocked, liver induction does not occur [22, 23, 26]. Furthermore, exogenous FGF1 or FGF2 can replace the cardiac mesoderm and induce Albumin expression in foregut endoderm explants [24]. BMP signaling however is required, but not sufficient, for hepatic induction in explants and may act in part by maintaining Gata4/6 expression [22]. Mouse explant studies also suggest that different concentrations of FGF are critical for segregating different foregut organ lineages are from a common progenitor cell population, with high, intermediate and low levels of FGF signaling promoting lung, liver and ventral pancreas respectively [27, 28]. It is still unclear whether it is the proximity, or the length of time that the endoderm is in contact with the cardiac mesoderm that controls the FGF dose.

3. Liver Bud development

Liver bud development Once hepatic fate is acquired by endodermal cells by signalling from cardiogenic mesoderm, role of homeobox transcription factor Hhex comes in to play which regulates proliferation and positioning of the ventral endoderm within the cardiogenic field that controls induction of hepatic cell fate and is required to ensure pseudostratification. In the absence of Hhex, mutant mouse embryos initiate hepatic specification [29] but fail to complete liver bud morphogenesis, resulting in hepatic

structures that lack a parenchymal cell component [30, 31]. Conditional ablation of the Hhex gene in the early hepatoblasts also disrupts their differentiation into hepatocytes [32] suggesting that Hhex has multiple roles in controlling the onset of hepatogenesis. Between the 7 and 11 somite stages of development, in response to the inductive cues from the heart and mesenchyme, the cells forming the hepatic endoderm that lie proximal to the sinus venosus transition to a columnar morphology [33] and express several hepatic genes including Albumin, Afp, Ttr (transthyretin), Rbp (retinol binding protein), and the transcription factor Hnf4a, all of which are reliable indicators of early hepatic cell fate

4. Liver bud growth and migration of cells in to septum transversum mesenchyme

At around 21 somites in the mouse, nuclear migration within the epithelial cells results in a pseudostratified epithelial morphology. The transition in cellular morphology results in a thickening of the epithelium, which bulges into the surrounding stroma. The basal face of the diverticulum is surrounded by a matrix which contains laminin, nidogen, type IV collagen, fibronectin, and heparin sulfate proteoglycans [34]. The matrix surrounding the basal surface of the epithelium is then degraded and E-cadherin expression is downregulated in the hepatic cells as they delaminate and invade the surrounding stroma as migrating cords of hepatoblasts [35]. The migration of the hepatic progenitor cells into the stroma requires the action of matrix metalloproteinases (MMPs) [36]. Several MMPs have been identified in the vicinity of the liver bud including MMP-14 in the hepatic progenitors and MMP-2 in the surrounding mesenchyme. A number of other transcriptional regulators have been characterized as playing a role in later events For example, the homeodomain factors HNF6 (also called Onecut-1) and Onecut-2 are required for hepatoblast migration [37]. Homeobox transcription factor Prox1 also promotes hepatoblast proliferation and migration from the primary liver bud

CK7 CK19x10

Fig. 26.6 (a, b) Histological sections of liver at 10 week with immune staining for cytokeratin 19 and cytokeratin 7 marking the differentiation of biliary epithelial cells

[38]. Although the mechanism through which Prox1 controls hepatoblast migration is unclear, the mutant hepatoblasts were found to maintain high levels of E-cadherin and failed to degrade the basal matrix surrounding the liver bud.

In *Prox1-/-* embryos hepatoblasts are specified and begin to proliferate, but the basal lamina fails to degrade and the cells remain trapped in the hepatic diverticulum. OC-1 and OC-2 are expressed in the foregut endoderm and hepatoblasts, and are redundantly required for the timely degradation of the basal lamina [37]. At e9.5, *OC-1;OC-2* double mutants resemble the *Prox1-/-* phenotype, but later hepatoblast invasion recovers resulting in a hypoplastic fetal liver. These mutant livers also have bile duct and gall bladder defects due to later roles of OC-1 and OC-2

The importance of cell-ECM interactions is also illustrated by the fact that hepatoblasts deficient for the laminin receptor β 1-integrin are unable to colonize the liver bud At e9.0, prior to vascularization of the liver bud, endothelial precursor cells lay between the hepatic epithelium and the STM. This close contact with blood vessels persists as hepatoblasts migrate into the stroma. Null mutations in the vascular endothelial growth factor receptor gene *Vegfr-2* results in embryos that lack endothelial cells and the hepatoblasts fail to delaminate in these embryos [39]. In addition angiogenesis inhibitors repress liver bud growth in culture, suggesting that endothelial cells provide unknown paracrine factors promoting hepatoblast migration and/or proliferation

5. Differentiation of hepatocytes and biliary epithelial cells

The hepatoblasts that migrate into the septum transversum appear to have the potential to differentiate into either cholangiocytes or hepatocytes. As evidenced by expression of genes associated with adult hepatocytes (hnf4alpha, albumin) and biliary epithelial cells (cytokeratin 19) as well as fetal liver genes such as AFP. Hepatoblasts in contact with the portal veinform a monolayer, and then a bi-layer, of cuboidal biliary precursors that increase cytokeratin-19 (CK-19) expression and down-regulate hepatic genes (Fig. 26.6). Between e17 and into the perinatal period focal dilations appear in the bilayer and these become surrounded by portal mesenchyme to form IHBD, while the remaining bi-layer cells regress. This process, which involves tubulogenesis and apoptosis, is known as ductal plate remodeling. Hepatoblasts in the liver parenchyma that are not in contact with portal veins gradually differentiate into mature hepatocytes. At e17 hepatocytes acquire their characteristic epithelial morphology arranged in hepatic

chords with bile canaliculi on the apical surfaces

6. Maturation of the hepatocytes.

As stated earlier The hepatoblasts that migrate into the septum transversum have the potential to differentiate into either cholangiocytes or hepatocytes. Cells that follow a hepatocyte cell fate progressively mature and progressively accumulate the gene expression and physiological profile of mature hepatic parenchyma cells [40, 41]. The maturation of hepatocytes is facilitated through network of transcription factors that regulate hepatocyte gene expression. By comparing developmental time points, the complexity of cross-regulation among factors is increasing gradually as development is progressing and the more number of transcription factors binding a given promoter are being identified. various transcriptional factors interact to form a circuit which ensures terminal differentiation of hepatocytes as development progresses. Six transcription factors, (HNF1a, HNF1b, FoxA2, HNF4a1, HNF6, and LRH-1 [Nr5a2]), have been found to form the core of this regulatory circuitry by occupying each other's promoters as well as the promoters of peripheral hepatic transcription factors., gene deletion studies in mouse embryos have found that HNF1b, FoxA2, and HNF6 all have roles in controlling the onset of hepatic gene expression during specification and liver bud formation, which is consistent with these factors having important roles in establishing the transcription factor network within the liver progenitor cells. HNF4a is required for development of normal architecture development. When HNF4a is specifically removed from fetal hepatoblasts, hepatic architecture is severely affected, with livers exhibiting loss of endothelial cells and disrupted hepatocellular polarity. The loss of hepatocyte polarity in Hnf4a-/- livers appears to reflect a requirement for HNF4a in controlling expression of several proteins involved in cell junction assembly [42]. Moreover, in the absence of HNF4a the core regulatory network is severely disrupted in fetal hepatic progenitors [43]; however, in adult hepatocytes, maintenance of the transcription factor network appears to be less dependent on HNF4a, although HNF4a does continue to have an important role in maintaining adult hepatocyte function [44].

Another important aspect in hepatocyte maturation is generation of membrane polarity Which directs molecules to sinusoidal or basolateral membrane. In non polarized hepatocytes progenitor cells, would be sinusoidal and canalicular membrane proteins are sorted at transgolgi network in to distinct transport vesicles by basolateral sorting signals and raft mechanisms.

Polarity is established as the liver bud invades the periheaptic mesoderm and likely mechanism as suggested by Yeaman et al. is as a result of from cell- cell, cell mesenchymal interaction which establishes membrane asymmetry triggering the events of polarization which includes establishment of apical junctional complexes (tight junction and cadherin adherens junction) and restriction of signaling events and polarized attachment of actin and microtubules, which then lead to the reorganization of the cytoskeleton in the cell [45].

7. Developmental interaction of hepatocytes and hematopoietic progenitor cells

Liver is primary hematopoietic organ during fetal development because in mammals soon after endodermal cells with hepatocytic fate enters the surrounding mesenchyme hematopoietic progenitors colonize the developing liver and this is because liver progenitor cells generate an environment that support hematopoiesis as suggested by coculture studies [46] and as the hepatocytes mature and undergo further differentiation they loose the property to support hematopoiesis resulting in change of hematopoiesis site from fetal liver to bone marrow.

This interaction between hematopoietic cells and hepatocyte progenitors is not one way because these hematopoietic cells secrete cytokine oncostatin M which in conjunction with other factors e.g. HGF and wnt promote hepatocyte maturation [47–51]. OSM induces metabolic maturation by activating the gp130

receptor and a JAK/Stat3 signaling pathway [52, 53], while promoting morphological maturation into polarized epithelium via K-ras and E-cadherin [48, 54]. Some evidence suggests that HGF and OSM activity is balanced by TNF α , which inhibits maturation and maintains the proliferative capacity of fetal hepatocytes, allowing the liver to grow to the appropriate size before differentiating [55]

8. Origin of hepatic stellate cells and kupffer cells

The origin of hepatic stellate cells has been debated, with various lines of evidence suggesting that the cells are of endodermal, neural crest, or mesenchymal origin. Most of the conclusions drawn from such studies were based on expression of shared sets of marker genes; however, shared gene expression does not necessarily correlate with cell lineage. Direct lineage tracing experiments have been performed in avian embryos [56], which led to the conclusion that mesothelial cells derived from the proepicardium and septum transversum mesenchyme could give rise to both endothelial and stellate cells within the hepatic sinusoids. Recent studies in human [57] and in mouse [58] support a mesothelial origin of hepatic stellate cells.

9. Development of biliary tree

The extrahepatic biliary tract originates from a portion of the ventral endoderm that is positioned immediately rostral to the ventral pancreatic bud. Reports demonstrate that the extrahepatic biliary tract derives from pancreatobiliary precursors coexpressing PDX1 and SOX17 [59]. This precursor population gives rise to SOX17+/PDX1-extrahepatic biliary cells and SOX17-/PDX1+ pancreatic cells. The segregation of the pancreatobiliary precursor population depends on SOX17. This factor is required for extrahepatic biliary tract development and overexpression inhibits pancreas development. The expression of SOX17 is controlled by homolog of hairy/enhancerof-split (Hes-1): in the absence of Hes-1 the mice not only display accelerated differentiation of pancreatic endocrine cells from pancreatic progenitors [60], the bile duct cells also differentiate to a pancreatic phenotype [61, 62]. Other transcription factors involved in extrahepatic biliary development include Hhex: in Hhex null embryos the common bile duct is replaced by duodenal-like tissue suggesting that the decision between a dudodenal or biliary fate appears to depend, at least in part, on the function of this transcription factor. Mice deficient in HNF6, Hes-1, HNF1b, or FoxF1 show lack or abnormal shape of the gallbladder [63–65].

Intrahepatic biliary tree develops as ductal plate which is continuous ring of cells arranges as monolayer that surrounds periportal mesenchyme. TGF β , Wnt and Notch are candidate signals from periportal mesenchyme that promotes BEC development. There is evidence that a TGF β signaling gradient emanating from the portal region promotes biliary differentiation in the adjacent hepatoblasts [66–68]. Wnt/ β -catenin signaling also promotes BEC development [69–71], and may act in part by stimulating the expression of EGF, which along with HGF can induce the formation of biliary structures in cultured hepatocytes [49, 72, 73].

The Notch signaling pathway is also considered contributing to biliary development based on the finding that patients affected with Alagille syndrome, a disease with bile duct paucity, had mutations in the JAGGED1 and NOTCH2 genes [74–76]. The analysis of this pathway in the liver has been challenging due to the presence of multiple ligands and receptors with overlapping functions. However, recent data favor a model in which Notch signaling controls multiple steps in biliary development, including the initial differentiation of cholangiocytes. Analysis of mice that have a liver-specific inactivation of RBP-Jk, a common transcriptional mediator of Notch signaling, revealed a reduced number of biliary cells differentiating from hepatoblasts [77]. Since expression of Jagged1 occurs in the periportal mesenchyme and biliary cells while Notch2 is present in the biliary cells [78], it appears that Notch signaling contributes not only to differentiation of biliary cells but also restricts differentiation to a periportal location.

Liver Stem Cells

The liver has remarkable regenerative capacity and can regrow when up to 70 % of its mass is removed. Liver regenerates primarily through proliferation of mature hepatocytes [79]. However adult liver also contains hepatic progenitor cells. Adult hepatic progenitor cells are believed to provide a backup system for replenishing hepatocytes and biliary epithelial cells when the regenerative capabilities of these cells are impaired such as in chronic injury states. These progenitor cells emerge and expand in periportal areas of the injured mouse, rat and human liver. Hepatic progenitor cells reside within or derive from the epithelial lining of bile ducts has been confirmed in mice by lineage tracing of cells expressing the transcription factor Sox9 [80]. This reserve compartment of hepatic progenitor cells is activated when mature hepatocytes and cholangiocytes are damaged or inhibited in their replication, The activation of the stem cell compartment, referred to as a "ductular reaction" in humans and "oval cell reaction" in rodents, is observed in circumstances of prolonged necrosis, cirrhosis, and chronic inflammatory liver diseases. This process involves expansion of bipotential transit amplifying progenitor cells, which can differentiate into hepatocytes and biliary cells. Intermediate hepatocytes, with an intermediate phenotype between progenitor cells and mature hepatocytes, are seen in moderate to severe inflammatory hepatitis [81].

Liver Stem Cells and Disease States/ Hepatocellular Carcinoma

The existence of a hepatic stem cell compartment gave rise to expectations regarding the practical applications of such research. Understanding and identification of the bipotential progenitor cells may hold promise for new therapeutic treatments to a wide range of liver pathological conditions ranging from congenital metabolic diseases, endstage liver cirrhosis, and hepatocarcinogenesis. Existence of cancer stem cells (CSCs) was first proposed over 40 years ago, but have these cells been identified in past decade, in hematological malignancies, and more recently in solid tumors that include liver, breast, prostate, brain, and colon. Constant proliferation of stem cells is a vital component in liver tissues.

Exploration of the difference between CSCs from normal stem cells is crucial not only for the understanding of tumor biology but also for the development of specific therapies that effectively target these cells in patients. These ideas have drawn attention to control of stem cell proliferation by the transforming growth factor beta (TGF- β), Notch, Wnt, and Hedgehog pathways. Recent evidence also suggests a key role for the TGF-β signaling pathway in both hepatocellular cancer suppression and endoderm formation, suggesting a dual role for this pathway in tumor suppression as well as progression of differentiation from a stem or progenitor stage. So understanding the mechanism of development and differentiation of stem cells will help us target CSCs via inducing differentiation of CSCs into nontumorigenic cells or completely eliminating the cells via inhibition of the self-renewing stem-cell state [81].

Developmental Anomalies of Liver

Variation in lobulation and vascularity are the most common developmental anomalies (or variations) during development but these are not clinically significant. But their knowledge becomes important in pathological conditions as in surgeries for GB or other surrounding organs.

One classical example of abnormal development is Alagille syndrome (ALGS) which is a complex autosomal dominant disorder due to defects in the Notch signaling pathway. Main clinical features and malformations are chronic cholestasis due to paucity of intrahepatic bile ducts, congenital heart disease primarily affecting the pulmonary outflow tract and vasculature, butterfly vertebrae, characteristic facies with a broad forehead, posterior embryotoxon and/or anterior segment abnormalities of the eyes, and pigmentary retinopathy. The importance of Notch signaling and why it results in abnormal biliary development has been discussed earlier in development of biliary tree.

Implications

Understanding the mechanism of development and differentiation of hepatocytes and applying these developmental protocols on mouse and human embryonic stem cells, researchers have successfully generated cultures where up to 70 % of the cells exhibit a "hepatocyte-like" phenotype. These cells exhibit many features of hepatocytes including; (1) expression of hepatic enzymes, (2) hepatocyte morphology, (3) robust glycogen storage, (4) uptake and metabolizism of drugs and (5) secretion of albumin [82].

These hepatocytes can be used as

- Tool to rescue liver function in disease states Several groups have found that transfusion of embryonic stem cells derived hepatocytes into mice with various liver injury models exhibited a modest rescue of liver function, although engraftment of the cells into the host liver was very low [82]. So one of the concerns about transplanting cells into a liver undergoing failure is that the environment of the liver may not allow the graft to 'take'.
- Artificial liver devices

There is possibility of creating an artificial liver, similar to a dialysis machine for kidney failure, that would perform all the functions of a normal liver which might replace the need for transplants or provide a bridge to therapy for patients waiting for transplants. Despite providing some respite for patients, these artificial liver bioreactors have not provided any overall survival benefit or reduced the need for transplants. In addition, obtaining huge numbers (10^{10}) of fresh hepatocytes to load into the bioreactors is an ongoing issue.

• Neo Livers [83]

Researchers at the McGowan Institute for Regenerative Medicine in Pittsburg are trying alternate approach where they co-opt part of the body, in this case the lymph node, to act as a reservoir for transplanted hepatocytes. The hepatocytes grow well in the lymph nodes, perhaps because there is a lot of blood supply to the nodes, and the resultant 'hepatasized lymph nodes' are even able to rescue mice with lethal liver failure. One issue with translating this approach to humans is the risk of hepatocyte rejection given the donor population of hepatocytes. Researchers theorize that someday perhaps induced pluripotent cells created from the patients themselves could generate the hepatocytes needed for the transplant and thus circumvent the issue of graft rejection.

In an international collaboration, researchers from the USA, Italy and Japan are attempting to create '**neolivers**' by reseeding livers stripped down to a three dimensional scaffold with fresh hepatocytes. They have had some success testing this approach in rats and are now optimizing their seeding strategies and making sure that the stripping technique does not compromise the ability of the neoliver to be infiltrated with blood vessels.

Drug Testing

Pharmaceutical companies require large numbers of hepatocytes in order to identify which chemicals could be potential new drugs. This procedure is invaluable because 50 % of drugs are taken off the market because of toxicity to the liver. Currently, hepatocytes left over from transplants or hepatocyte cell lines derived from liver cancers are used as sources for drug testing but variation and functionality are issues that can confound the results. This is where stem cells could be a tremendous boon. Stem cells could theoretically generate limitless numbers of hepatocytes, and induced pluripotent stem cells could even provide patient-specific hepatocytes to verify that a drug therapy would not harm the patient's liver [83].

Conclusion

Through rapid evolution of molecular genetics and advancements in field of experimental developmental biology with development of new animal models our understanding of molecular mechanisms controlling liver development is improving. These findings of molecular mechanisms are applicable to all tissues and organs. In last decade researchers have uncovered many genes regulating hepatogenesis and transcription factors involved but as the knowledge is improving more questions are also coming up after observations such as why hepatocytes when placed in culture system dedifferentiate, what regulates organ size. As our understanding is improving it is likely that with time we will get answers to many questions and many answers to the questions known till date are likely to change.

References

- Furbank RA. Conversion data, normal values, nomograms and other standards. In: Simpson K, editor. Modern trends in forensic medicine. New York: Appleton-Century-Crofts; 1967. p. 344–64.
- Ludwig J. Current methods of autopsy practice. Philadelphia: WB Saunders; 1972.
- Sunderman FW, Boerner F. Normal values in clinical medicine. Philadelphia: WB Saunders; 1950.
- Schulz DM, Giordano DA, Schulz DH. Weights of organs of fetuses and infants. Arch Pathol. 1962; 74:244–50.
- Blouin A, Bolender RP, Weibel ER. Distribution of organelles and membranes between hepatocytes and nonhepatocytes in the rat liver parenchyma. A stereological study. J Cell Biol. 1977;72:441–55.
- Si-Tayeb K, et al. Organogenesis and development of liver. Dev Cell. 2010;18:177.
- Sadler TW. Langman's embryology. 12th ed, Lippincott Williams & Wilkins; 2012. p. 218–9
- Roskams T et al. Embryology of extra- and intrahepatic bile ducts, the ductal plate. Anat Rec. 2008; 291:628–35
- Tremblay KD, Zaret KS. Distinct populations of endoderm cells converge to generate the embryonic liver bud and ventral foregut tissues. Dev Biol. 2005;280:87–99.
- Houssaint E. Differentiation of the mouse hepatic primordium. An analysis of tissue interactions in hepatocyte differentiation. Cell Differ. 1980;9:269–79.
- Lemaigre FP. Development of the biliary tract. Mech Dev. 2003;120:81–7.
- Shiojiri N. The origin of intrahepatic bile duct cells in the mouse. J Embryol Exp Morphol. 1984;79:25–39.
- Yamada T. (ed.) Text book of Gastroenterology, 5th edition, Blackwell Publishing; 2009, p 572–73.
- Shen MM. Nodal signaling: developmental roles and regulation. Development. 2007;134:1023–34.
- Zorn AM, Wells JM. Molecular basis of vertebrate endoderm development. Int Rev Cytol. 2007;259:49–111.
- Moore-Scott BA, Opoka R, Lin SC, Kordich JJ, Wells JM. Identification of molecular markers that are expressed in discrete anterior-posterior domains of the endoderm from the gastrula stage to mid-gestation. Dev Dyn. 2007;236:1997–2003.
- Grapin-Botton A. Antero-posterior patterning of the vertebrate digestive tract: 40 years after Nicole Le Douarin's PhD thesis. Int J Dev Biol. 2005;49:335–47.

- Fukuda-Taira S. Hepatic induction in the avian embryo: specificity of reactive endoderm and inductive mesoderm. J Embryol Exp Morphol. 1981;63: 111–25.
- Le Douarin. An experimental analysis of liver development. Med Biol. 1975;53:427–55.
- Okada TS. Epithelio-mesenchymal relationships in the regional differentiation of the digestive tract in the amphibian embryo. Rouxs Arch Dev Biol. 1960;152: 1–21.
- Takata N. The differentiation in vivo of the isolated endoderm under the influence of the mesoderm in Triturus pyrrhogaster. Embryologica. 1960;5:38–70.
- 22. Rossi JM, Dunn NR, Hogan BL, Zaret KS. Distinct mesodermal signals, including BMPs from the septum transversum mesenchyme, are required in combination for hepatogenesis from the endoderm. Genes Dev. 2001;15:1998–2009.
- Gualdi R, Bossard P, Zheng M, Hamada Y, Coleman JR, Zaret KS. Hepatic specification of the gut endoderm in vitro: cell signaling and transcriptional control. Genes Dev. 1996;10:1670–82.
- Jung J, Zheng M, Goldfarb M, Zaret KS. Initiation of mammalian liver development from endoderm by fibroblast growth factors. Science. 1999;284:1998–2003.
- Le Douarin NM. An experimental analysis of liver devleopment. Med Biol. 1975;53:427–55.
- Calmont A, Wandzioch E, Tremblay KD, Minowada G, Kaestner KH, Martin GR, Zaret KS. An FGF response pathway that mediates hepatic gene induction in embryonic endoderm cells. Dev Cell. 2006; 11:339–48.
- Deutsch G. et al. A bipotential precursor population for pancreas and liver within the embryonic endoderm. Development 2001:128;871–881
- Serls AE, Doherty S, Parvatiyar P, Wells JM, Deutsch GH. Different thresholds of fibroblast growth factors pattern the ventral foregut into liver and lung. Development. 2005;132:35–47.
- Bort R, Martinez-Barbera JP, Beddington RS, Zaret KS. Hex homeobox gene-dependent tissue positioning is required for organogenesis of the ventral pancreas. Development. 2004;131:797–806.
- Keng, et al. Homeobox gene Hex is essential for onset of mouse embryonic liver development and differentiation of the monocyte lineage. Biochem Biophys Res Commun. 2000;276:1155–61.
- Martinez Barbera, et al. The homeobox gene Hex is required in definitive endodermal tissues for normal forebrain, liver and thyroid formation. Development. 2000;127:2433–45.
- 32. Hunter, et al. The homeobox gene Hhex is essential for proper hepatoblast differentiation and bile duct morphogenesis. Dev Biol. 2007;308:355–67.
- Bort, et al. Hex homeobox gene controls the transition of the endoderm to a pseudostratified, cell emergent epithelium for liver bud development. Dev Biol. 2006; 290:44–56.
- Shiojiri N, Sugiyama Y. Immunolocalization of extracellular matrix components and integrins during mouse liver development. Hepatology. 2004;40:346–55.

- Medlock ES, Haar JL. The liver hemopoietic environment: I. Developing hepatocytes and their role in fetal hemopoiesis. Anat Rec. 1983;207:31–41.
- Margagliotti, et al. Role of metalloproteinases at the onset of liver development. Dev Growth Differ. 2008;50:331–8.
- Margagliotti, et al. The Onecut transcription factors HNF-6/OC-1 and OC-2 regulate early liver expansion by controlling hepatoblast migration. Dev Biol. 2007;311:579–89.
- Sosa-Pineda, et al. Hepatocyte migration during liver development requires Prox1. Nat Genet. 2000;25: 254–5.
- Matsumoto, et al. Liver organogenesis promoted by endothelial cells prior to vascular function. Science. 2001;294:559–63.
- 40. Ge, et al. Interpreting expression profiles of cancers by genomewide survey of breadth of expression in normal tissues. Genomics. 2005;86:127–41.
- Jochheim, et al. Multi-stage analysis of differential gene expression in BALB/C mouse liver development by high-density microarrays. Differentiation. 2003;71:62–72.
- 42. Battle, et al. Hepatocyte nuclear factor 4alpha orchestrates expression of cell adhesion proteins during the epithelial transformation of the developing liver. Proc Natl Acad Sci U S A. 2006;103:8419–24.
- Kyrmizi, et al. Plasticity and expanding complexity of the hepatic transcription factor network during liver development. Genes Dev. 2006;20:2293–305.
- 44. Hayhurst, et al. Hepatocyte nuclear factor 4alpha (nuclear receptor 2A1) is essential for maintenance of hepatic gene expression and lipid homeostasis. Mol Cell Biol. 2001;21:1393–403.
- Wang L, Boyer JL. The maintenance and generation of membrane polarity in hepatocytes. Hepatology. 2004;39(4):896.
- 46. Hata, et al. Establishment of a hepatocytic epithelial cell line from the murine fetal liver capable of promoting hemopoietic cell proliferation. J Cell Physiol. 1993;154:381–92.
- Kamiya, et al. Oncostatin M and hepatocyte growth factor induce hepatic maturation via distinct signaling pathways. FEBS Lett. 2001;492:90–4.
- Matsui T, Kinoshita T, Morikawa Y, Tohya K, Katsuki M, Ito Y, Kamiya A, Miyajima A. K-Ras mediates cytokine-induced formation of E-cadherin-based adherens junctions during liver development. EMBO J. 2002;21:1021–30.
- Michalopoulos, et al. HGF-, EGF-, and dexamethasoneinduced gene expression patterns during formation of tissue in hepatic organoid cultures. Gene Expr. 2003;11:55–75.
- Suzuki, et al. Role for growth factors and extracellular matrix in controlling differentiation of prospectively isolated hepatic stem cells. Development. 2003; 130:2513–24.

- Tan, et al. Beta-catenin deletion in hepatoblasts disrupts hepatic morphogenesis and survival during mouse development. Hepatology. 2008;47:1667–79.
- 52. Ito, et al. Retroviral gene transfer of signaling molecules into murine fetal hepatocytes defines distinct roles for the STAT3 and ras pathways during hepatic development. Hepatology. 2000;32:1370–6.
- Kamiya, et al. Fetal liver development requires a paracrine action of oncostatin M through the gp130 signal transducer. EMBO J. 1999;18:2127–36.
- Imamura, et al. Oncostatin M induces upregulation of claudin-2 in rodent hepatocytes coinciding with changes in morphology and function of tight junctions. Exp Cell Res. 2007;313:1951–62.
- Kamiya, Gonzalez. TNF-alpha regulates mouse fetal hepatic maturation induced by oncostatin M and extracellular matrices. Hepatology. 2004;40:527–36.
- Pe' rez-Pomares, et al. Contribution of mesotheliumderived cells to liver sinusoids in avian embryos. Dev Dyn. 2004;229:465–74.
- Loo, Wu. Origin of stellate cells from submesothelial cells in a developing human liver. Liver Int. 2008;28:1437–45.
- Asahina, et al. Mesenchymal origin of hepatic stellate cells, submesothelial cells, and perivascular mesenchymal cells during mouse liver development. Hepatology. 2009;49:998–1011.
- Spence, et al. Sox17 regulates organ lineage segregation of ventral foregut progenitor cells. Dev Cell. 2009;17:62–74.
- Jensen, et al. Control of endodermal endocrine development by Hes-1. Nat Genet. 2000;24:36–44.
- Fukuda, et al. Ectopic pancreas formation in Hes1 -knockout mice reveals plasticity of endodermal progenitors of the gut, bile duct, and pancreas. J Clin Invest. 2006;116:1484–93.
- Sumazaki, et al. Conversion of biliary system to pancreatic tissue in Hes1-deficient mice. Nat Genet. 2004;36:83–7.
- Clotman, et al. The onecut transcription factor HNF6 is required for normal development of the biliary tract. Development. 2002;129:1819–28.
- 64. Coffinier, et al. Bile system morphogenesis defects and liver dysfunction upon targeted deletion of HNF1beta. Development. 2002;129:1829–38.
- Kalinichenko, et al. Haploinsufficiency of the mouse Forkhead Box f1 gene causes defects in gall bladder development. J Biol Chem. 2002;277:12369–74.
- 66. Clotman, et al. Control of liver cell fate decision by a gradient of TGF beta signaling modulated by Onecut transcription factors. Genes Dev. 2005;19:1849–54.
- 67. Clotman, Lemaigre. Control of hepatic differentiation by activin/TGFbeta signaling. Cell Cycle. 2006;5:168–71.
- Weinstein, et al. Smad proteins and hepatocyte growth factor control parallel regulatory pathways that converge on beta1-integrin to promote normal liver development. Mol Cell Biol. 2001;21:5122–31.

- Decaens, et al. Stabilization of beta-catenin affects mouse embryonic liver growth and hepatoblast fate. Hepatology. 2008;47:247–58.
- Hussain, et al. Wnt impacts growth and differentiation in ex vivo liver development. Exp Cell Res. 2004;292:157–69.
- Monga, et al. Beta-catenin antisense studies in embryonic liver cultures: role in proliferation, apoptosis, and lineage specification. Gastroenterology. 2003; 124:202–16.
- 72. Block, et al. Population expansion, clonal growth, and specific differentiation patterns in primary cultures of hepatocytes induced by HGF/SF, EGF and TGF alpha in a chemically defined (HGM) medium. J Cell Biol. 1996;132:1133–49.
- Tan, et al. Epidermal growth factor receptor: a novel target of the Wnt/beta-catenin pathway in liver. Gastroenterology. 2005;129:285–302.
- 74. Li, et al. Alagille syndrome is caused by mutations in human Jagged1, which encodes a ligand for Notch1. Nat Genet. 1997;16:243–51.
- Oda, et al. Mutations in the human Jagged1 gene are responsible for Alagille syndrome. Nat Genet. 1997;16:235–42.

- McDaniell, et al. NOTCH2 mutations cause Alagille syndrome, a heterogeneous disorder of the notch signaling pathway. Am J Hum Genet. 2006;79:169–73.
- Zong, et al. Notch signaling controls liver development by regulating biliary differentiation. Development. 2009;136:1727–39.
- Geisler, et al. Liver-specific inactivation of Notch2, but not Notch1, compromises intrahepatic bile duct development in mice. Hepatology. 2008;48:607–16.
- Michalopoulos GK. Liver regeneration. J Cell Physiol. 2007;213:286–300.
- Furuyama K, et al. Continuous cell supply from a Sox9- expressing progenitor zone in adult liver, exocrine pancreas and intestine. Nat Genet. 2011;43: 34–41 [PubMed: 21113154].
- Mishra, et al. Liver stem cells and hepatocellular carcinoma. Hepatology. 2009;49(1):318–29. doi:10.1002/ hep.22704.
- Agarwal S, et al. Efficient differentiation of functional hepatocytes from human embryonic stem cells. Stem Cells. 2008;26(5):1117–27.
- http://oirm.ca/sites/default/files/disease-liver_failure. pdf. Accessed on 18/3/16.
Part VIII

Fetal Cardiovascular Development: Up to Second Trimester

Embryological Basis for Vascular Anomalies

27

Trenton R. Foster, Jason A. Chin, Kirstyn E. Brownson, Hualong Bai, and Alan Dardik

Formation of the mature vascular system occurs through a well organized process beginning early in the embryo. Deviation from the normal sequence of events leads to well described vascular anomalies. The cause for deviations from normal development often remains unclear, although research has identified certain signaling pathways that are essential to normal development. This chapter describes an embryologic basis for a variety of vascular anomalies.

Vasculogenesis

Vasculogenesis is the process by which a new vascular system is created de novo in the early embryo, e.g. the formation of endothelial cells and tubes from previously undifferentiated cells. This process is directed by molecular signaling pathways beginning during the third week in utero. At this early stage of development, the yolk sac's mesoderm forms hemangioblastic aggregates that

Vascular Biology and Therapeutics Program, Department of Surgery, Yale University

School of Medicine, 10 Amistad Street, Room 437,

Veterans Affairs Connecticut Healthcare Systems, West Haven, CT, USA e-mail: alan.dardik@yale.edu contain a core of hematopoietic stem cells and an outer layer of endothelial cells. This aggregation is guided by the influence of signaling molecules including Indian hedgehog, bone morphogenic protein, and transforming growth factor β (TGF- β). The hemangioblastic aggregates coalesce to form the connecting stalk and the chorion. In a similar fashion, the embryonic endoderm secretes bone morphogenic protein and TGF- β to induce vessel formation from the embryonic splanchnic mesoderm. The embryonic splanchnic mesoderm develops a cardiogenic area by creating a horseshoe shaped pair of endocardial tubes that will ultimately give rise to the heart. The splanchnic mesoderm also forms paired dorsal aortas that connect to the endocardial tubes.

After vasculogenesis forms the early vessels, the primitive vascular elements develop buds and sprout new branches. The process of developing new branches from existing vascular structures is termed angiogenesis, and the dilation of these vessels into yet larger ones is termed arteriogenesis. The branches develop into capillary networks that remodel into arterial, venous, or lymphatics, under the influence of local signaling molecules. A diagram of the developing embryonic vascular system is shown in Fig. 27.1. Vasculogenesis is typically restricted to embryogenesis and not normally thought to occur in adults. However, angiogenesis and arteriogenesis occur in adults; an example in a patient with peripheral arterial disease is shown in Fig. 27.2.

T.R. Foster, MD • J.A. Chin, MD • K.E. Brownson, MD H. Bai, MD • A. Dardik, MD, PhD (\boxtimes)

PO Box 208089, New Haven, CT 06520-8089, USA

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_27



Fig. 27.2 Arterial angiography of lower extremity with arterial occlusive disease. *Left panel* shows the distal thigh including the area of popliteal artery occlusion, as well as collateral circulation around the occlusion (*yellow arrows*)

show arteriogenesis). *Right panel* shows the leg of the same patient, with capillary growth in the distal muscle (*yellow arrows* show angiogenesis)

VEGF is an important growth factor that promotes the development of fenestrated endothelium found in capillaries, endocrine organs, and the kidneys. Angiopoietin induces the formation of vascular structures containing tight junctions which are essential in the central nervous system to maintain a blood brain barrier. Angiopoeitin-2 and Nitric Oxide synthase are critical for the development of vascular structures. Hypoxia is also known to induce angiogenesis through upregulation of angiogenic genes [1].

Vascular Identity

There are distinct differences between arterial and venous cells. The identity of cells is genetically determined and conferred through signaling pathways. VEGF is common to the development of both arterial and venous cells although differences in the receptor complexes have been identified. Arterial cells are generated from the binding of VEGF to VEGF receptor 2 (VEGFR2) associated with the co-receptor neuropilin-1 (NP1). The VEGFR2-NP-1 complex initiates a signaling cascade that stimulates the delta-notch pathway resulting in the expression of Ephrin-B2, an arterial specific cell surface marker. Venous identity develops when the transcription factor COUP-TFII suppresses the delta-notch pathway, allowing constitutive expression of Eph-B4, a venous specific cell surface marker. Eph-B4 and Ephrin-B2 are transmembrane tyrosine kinase molecules that are critical in vascular development and remodeling [2]. Figure 27.3 shows VEGF signaling with differentiation toward arterial or venous fated cells. Figure 27.4 shows staining of a full-term human umbilical cord comparing arterial and venous endothelial markers.

Lymphatics are created as buds off the developing venous system. The differentiation of lymphatic endothelium from veins occurs under the stimulation of Prospero-related homeobox- 1 (PROX1) gene [3].

Aortic Arch Development

The developing embryo contains two endocardial tubes that fold and join along the midline to create the primitive cardiac area. The cardiac area contains an inflow end and an outflow end. Approximately at days 22-24, the heart moves into a more ventral position causing a bend to develop along the outflow tract forming the first aortic arch. There are a total of six paired aortic arches that will develop from corresponding pharyngeal arches in a rostral to caudal direction. Aortic arches 1 and 2 quickly regress in normal human development, contributing little to adult structures. The third aortic arch becomes the common carotid arteries and proximal internal carotid arteries. The dorsal aortas regress between arches 3 and 4 causing blood to flow selectively through the third aortic arch destined for the head and neck.

The left fourth aortic arch forms part of the arch between the left common carotid and left subclavian arteries. The right fourth aortic arch forms the proximal right subclavian artery.



Fig. 27.3 Arterial fated cell expresses Ephrin-B2 via signaling of the delta-notch pathway. Venous fated cell expresses Eph-B4 due to suppression of this pathway by COUP-TFII



Additional portions of the right subclavian artery are derived from the right dorsal aorta. The fifth aortic arch does not develop in humans. The sixth aortic arch connects the left pulmonary artery to the left dorsal aorta forming the ductus arteriorsus.

Patent Ductus Arteriosus

The most common vascular anomaly is a patent ductus arteriosus. This connection is essential in fetal life as a mechanism to shunt blood away from the immature lungs to the aorta. Premature closure of the ductus arteriosus results in fetal hydrops. Patency of the ductus arteriosus is maintained by molecular signals including prostaglandin, prostacyclin, and low oxygen concentration. After birth, smooth muscle cells surrounding the ductus are exposed to increased oxygen and a fall in prostaglandin and prostacylcin concentration. These changes induce constriction of smooth muscle cells to close the ductus arteriosus, leaving a remnant structure termed the ligamentum arteriosus. Failure of the ductus to close after birth results in a patent ductus arteriosus (PDA). PDA is known to occur in at least 1/2000 term births with some estimates indicating incidence as high as 1/500. There is a genetic predisposition to PDA although the exact mechanism remains unclear. PDA has also been associated with in utero infections including rubella and particular environmental exposures. The long term consequence of PDA is pulmonary hypertension and eventual heart failure. Management of PDA is guided by the severity of left to right shunting. When indicated, closure is performed by transcatheter or surgical approaches [4].

Coarctation of the Aorta

A similar signaling process that induces closure of the ductus arteriosus is thought to contribute to coarctation of the aorta. Aortic coarctation occurs in the 1/2500 births. The exact mechanism of coarctation is unclear and is likely to have a multifactorial etiology. However, it is thought that changes in flow dynamics and signaling molecules induce constriction of muscle cells in the aortic wall in the region near or immediately distal to the ductus. Over time, the aorta remodels due to this constriction resulting in the formation of collateral vessels with increased flow around the coarctation through the intercostal arteries. The increased flow through intercostals results in notched ribs, a hallmark of this condition. Long term consequences of coarctation include hypertension and eventual heart failure [5].

Anomalous Aortic Arch

The transition from paired aortic arches to only a left sided aortic arch requires a particular sequence of development and regression. The arch originates as paired aortic arches arising from the cardiac outflow that bend caudally and fuse distal to the seventh intersegmental artery. The fused dorsal aorta continues distally through the developing embryo. In normal development, the right dorsal aorta forms the proximal right subclavian artery. Distal to the right subclavian artery, the right dorsal aortic arch normally regresses and the connection from the right arch to the aorta is thus lost. If regression of the distal right dorsal aorta fails to occur or if there is abnormal regression of the left side, then an anomalous aortic arch is created. Anomalies of the aortic arch may create vascular rings around the trachea or esophagus which can cause compression of these structures.

Specific examples of aortic arch anomalies include a double aortic arch resulting from failed regression the right aortic arch distal to the right subclavian artery. This failed regression forms a double aortic arch as diagramed in Fig. 27.5.

The segment of abnormally retained right dorsal aorta generally traverses posterior to the esophagus before joining the left aortic arch which passes anterior to the trachea. This configuration results in a vascular ring. A vascular ring can also occur from development of a right aortic arch instead of a normal left sided arch. In this case, the right aortic arch is the conduit to the distal aorta while the left aortic arch regresses.



Fig. 27.5 The right aortic arch undergoes selective regression of the segment distal to the right subclavian artery, leaving only a normal left sided aortic arch

The right aortic arch gives rise to the ligamentum arteriosum which connects to the aorta by crossing anterior to the esophagus. If the right aortic arch crosses over to the left side behind the esophagus, then a vascular ring around the esophagus is created with the ligamentum arteriosum. If the aortic arch crosses over anterior to the esophagus and trachea, no vascular ring is formed.

Anomalous Subclavian Artery

The right subclavian artery normally arises from the right aortic arch. Inappropriate involution of the right fourth aortic arch causes aberrant development of the right subclavian artery originating from the left side. The aberrant right subclavian artery crosses from the left sided aorta to the right upper extremity. This path generally occurs behind the esophagus but it can occur between the esophagus and the trachea or in front of the trachea. The aberrant course of the subclavian artery can cause compressive symptoms as it passes around the esophagus and trachea, termed dysphagia lusoria. Aberrant subclavian artery is also associated with development of an aneurysm at its origin, known as Kommerell's diverticulum, which can also cause compressive symptoms or rupture. However, the majority of patients with aberrant subclavian arteries are asymptomatic [6].

Persistent Median Artery and High Radial Artery

As development continues, segments are created along the length of the embryo, in a body plan common from fruit flies to humans. Each segment is supplied by intersegmental vascular branches. These intersegmental branches are found throughout the cervical, thoracic, and lumbar regions. Development of the arterial supply to the upper extremities originates from the seventh cervical intersegmental artery that forms the axillary artery. The axillary artery continues to the brachial artery that connects to the hand through the median artery. A series of branching and regression then occurs to form the final configuration of the ulnar and radial arteries. During this process, the median artery regresses. Failure of regression results in persistence of the median artery, occurring in 12 % of the population.

During normal development, the ulnar artery originates from the brachial artery and supplies the superficial palmar arch. The radial artery originates from the proximal brachial artery and supplies the deep palmar arch. A final division occurs in which a new, more distal connection develops from the brachial artery to the radial artery. The original connection from the radial artery to the proximal brachial artery then regresses. Failure of this final revision results in a preserved connection of the radial artery to the proximal brachial artery, the basis for a high radial artery [7].

Renal Artery Variations

In the thoracic region, intersegmental branches form intercostal arteries. In the lumbar region, dorsal branches become lumbar arteries and the common iliac arteries. The lumbar ventral branches form allantoic vessels that become umbilical arteries while other paired branches will form the vitelline arteries. The vitelline arteries fuse to form the three dominant vessels to the gut, the celiac artery, the superior mesenteric artery, and the inferior mesenteric artery. Lateral segmental branches also supply the developing kidneys that are migrating rostrally. During the migration, a series of arterial connections are created and then regress. The process of forming arterial connections and regression during renal migration is prone to develop variations in final renal artery configuration. These common variations are important to recognize when performing abdominal operations involving the renal vasculature.

Persistent Sciatic Artery

The lower limb arterial supply is normally via the external iliac artery through the iliofemoral vessels. There is a contribution from the internal iliac artery through an axial artery although this usually regresses with only remnants persisting as the sciatic artery, part of the popliteal artery, and part of the peroneal artery. However, if the iliofemoral artery fails to develop properly, the sciatic artery may persist as the dominant vessel. In cases of complete sciatic preservation, blood travels from the internal iliac through the sciatic and into the popliteal artery to bypass an undeveloped or underdeveloped iliofemoral vessel. There are also 11 cases of persistent sciatic artery in combination with a normal ileofemoral vessel. The etiology of these anomalies is unknown. The location of the abnormal persistent sciatic artery is typically superficial to the buttocks leaving it susceptible to trauma. The majority of patients with a persistent sciatic artery develop symptoms of lower extremity claudication, ischemia, or neurologic sequelae. The neurologic symptoms are often due to compression of the sciatic nerve. Aneurysm of the sciatic artery is also described [8].

Popliteal Entrapment

Popliteal entrapment can occur due to abnormal development of the popliteal fossa. During embryologic development, there is medial migration of the medial head of the gastrocnemius muscle attachment. If this migration is incomplete or delayed, the abnormal lateral attachment of the gastrocnemius muscle displaces the popliteal artery. As the muscle contracts, it compresses the popliteal artery resulting in an anatomic popliteal entrapment syndrome. Anatomic popliteal entrapment is diagramed in Fig. 27.6.

Functional popliteal entrapment is also described. In cases of functional entrapment, the popliteal fossa is normally developed yet patients demonstrate compression with particular movements of the foot. Hypertrophied gastrocnemius muscle from repetitive use was previously believed to be the cause of functional entrapment; however, this assertion remains controversial as there are young untrained people also demonstrating popliteal artery compression [9].

Venous Anomalies

Superior Vena Cava

The superior vena cava normally develops from the right anterior cardinal and right common cardinal veins. The left common cardinal vein becomes the coronary sinus. Cranially, the internal jugular

veins develop from the paired anterior cardinal veins. The internal jugular veins join with the external jugular veins draining the face. The subclavian veins arise from the upper limbs and also connect to drain into the anterior cardinal veins. The left and right anterior cardinal veins connect to form the left brachiocephalic vein with regression of the left anterior cardinal vein caudal to this point. Regression of the left side results in a normal right sided superior vena cava. In the case of the opposite situation, where the right anterior cardinal vein regresses instead, an anomalous left sided superior vena cava is created. A double superior vena cava develops when there is no regression of the anterior cardinal vein on either side leaving both anterior cardinal veins patent.

Inferior Vena Cava

The inferior vena cava develops from a series of connections and regression originating from the posterior cardinal veins. Subcardinal veins develop with connections to the posterior cardinal veins. The subcardinal veins become dominant and both of the posterior cardinal veins regress. The subcardinal veins fuse to form the suprarenal inferior vena cava. The infrarenal



Fig. 27.6 Anatomic popliteal entrapment results from displacement of the popliteal artery by the medial head of the gastrocnemius muscle resulting in popliteal arterial compression. *Left panel*, normal popliteal artery; *right panel*, abnormally displaced popliteal artery

inferior vena cava is formed by the supracardinal veins during which the left supracardinal vein regresses leaving a right sided infrarenal vena cava. In the opposite case, the right supracardinal vein regresses leaving an anomalous left sided infrarenal vena cava. If both supracardinal veins persist, a double inferior vena cava is created [10].

Renal Vein Anomalies

Renal veins develop from the subcardinal veins. The right subcardinal vein gives rise to the right renal vein. The left subcardinal vein forms two branches to the left kidney. One branch runs anterior to the aorta and the other posterior to the aorta. Normally, the posterior vein regresses leaving only the anterior branch to form the left renal vein. If the opposite occurs, with regression of the anterior component, the left renal vein will run posterior to the aorta. If neither component regresses, a left circum-aortic renal vein occurs. In this case, the aorta is surrounded by an anterior and posterior renal vein which may have one or more connections between them. The presence of a circum-aortic renal vein is important to consider when dissecting around the aorta to avoid inadvertent renal vein injury during surgery [11].

Duplicated Femoral Vein

Variations in venous patterns of the extremities are common, the majority of which are without clinical significance. However, variations in femoral vein anatomy can have clinical consequence due to increased risk of deep vein thrombosis (DVT). There are three venous outflow tracts in the mature lower extremity venous network including the femoral vein, the deep femoral vein, and the venous arcades of the sciatic nerve (developed from the primitive axillary vein). Of these three vessels, the femoral vein is normally the dominant outflow tract. In approximately 12 % of people, there is duplication of the femoral vein outflow through either a bifid femoral vein, or through a hypoplastic femoral vein with collateral outflow through an enlarged deep femoral vein or axillary vein. Bilateral cases of duplicated femoral veins are also reported. Duplicated femoral veins have an increased risk of DVT and its associated complications. Detection of DVT in the setting of variable femoral vein anatomy can pose a diagnostic challenge in ultrasound evaluation [12].

Arteriovenous Malformations

Abnormal development of arteries, veins, capillaries or a combination of each results in vascular malformations. Port wine stains are capillary malformations of the dermis causing red or pink lesions to appear on the skin. The lesions occur in 0.1-2 % of newborns. The lesions grow as the child ages. Proposed mechanisms causing capillary malformations include vascular ectasia, lack of neuronal control of blood flow, or over expression of VEGF or VEGF receptors [13]. Genomic studies have identified an association with an activating mutation in the gene encoding guanine nucleotide binding protein G-alpha-q (GNAQ). It is hypothesized this mutation alters cell proliferation through the extracellular signal-regulated kinase (ERK) pathway. These same mutations are also implicated in Sturge-Weber syndrome that includes vascular malformation as a component of a larger syndrome. Sturge-Weber syndrome is a sporadic congenital neurocutaneous disorder that includes a port-wine stain affecting skin in the distribution of the ophthalmic branch of the trigeminal nerve, abnormal capillary venous vessels in the leptomeninges of the brain and choroid, along with glaucoma, seizures, stroke, and intellectual disability. It is thought that the gene mutations causing port wine stains occur 15 later in embryonic development while earlier mutations result in a wider variety of cell types affected causing the syndromic presentation of Sturge-Weber [14].

Parkes-Weber syndrome (PWS) is a disorder that includes a large capillary malformation on an extremity, in addition to soft tissue and bone hypertrophy of the affected limb. The vascular malformations create multiple fast-flow arteriovenous shunts. Patients with PWS have overgrowth of the affected limb and their clinical course is complicated by the effects of the arteriovenous shunting which cause ulcerations and high output cardiac failure. Similar to PWS, Klippel-Trenaunay syndrome (KTS) is a rare disorder that also includes a vascular malformation. In KTS, patients exhibit a combination of capillary malformations, bone or soft tissue hypertrophy, and lateralizing atypical varicosity. In contrast to PWS, KTS does not have significant arteriovenous shunting. The vascular abnormalities found in KTS generally involve both capillary and venous malformations. There is often a persistence of embryonic veins, most commonly the lateral marginal vein running superficially up the lateral border of the leg. When present, this vein is thickened and incompetent due to the absence of valves. Abnormal veins can also be found involving abdominal or pelvic organs. These malformations are at risk of causing bleeding and thromboembolic events. KTS is important for the clinician, as these patients occasionally seek treatment for varicose veins; due to the underlying hypoplasia of the deep leg veins, treatment of their dilated superficial leg veins is generally contraindicated. The exact developmental cause of KTS is unknown but it is thought to result from abnormal embryonic vascular remodeling, possibly through alterations in VEGF mediated signaling pathways. While KTS is primarily sporadic, familial cases have occurred and genetic studies have also identified an association with mutated angiogenic factor VG5Q [15, 16].

Lymphatic System

The lymphatic system develops from buds off the venous system. Researchers have shown regions of endothelial cells located along the walls of the anterior cardinal veins to express the PROX1 gene. It is from these regions that the lymphatic network arises. Animal models lacking the PROX1 gene fail to develop a lymphatic system. The first step in lymphatic development is the creation of six lymph sacs. These sacs bud off from the cardinal and mesonephric venous systems. These lymph sacs generate additional lymph sacs that spread along the course of developing veins. On both the left and right side, thoracic lymph vessels form from jugular sacs that empty into the venous system at the internal jugular and subclavian junction. In normal development, the left and right side fuse at the thoracic level with regression of the right thoracic vessel to form the left sided thoracic duct draining into the left internal jugular vein. The right sided jugular sac will give rise to the right lymphatic duct. Additional lymphatics develop from the original lymph vessels to form a network of lymphatic channels that all drain into the venous system via either the right lymphatic duct or the thoracic duct. Maturation of the lymphatic vessels is orchestrated through additional cell signaling pathways. The forkhead box protein C2 (FOXC2) plays a critical role in normal lymphatic differentiation into a mature collecting vessel. Ephrin interactions are also important in lymphatic development. Eph-B4 is found on both initial and collecting lymphatics while Eph-B2 is found only on the mature collecting lymphatic vessels [17].

References

- Patel-Hett S, D'Amore PA. Signal transduction in vasculogenesis and developmental angiogenesis. Int J Dev Biol. 2011;55(4–5):353–63. doi:10.1387/ ijdb.103213sp.
- Fancher TT, Muto A, Fitzgerald TN, Magri D, Gortler D, Nishibe T, Dardik A. Control of blood vessel identity: from embryo to adult. Ann Vasc Dis. 2008;1(1):28–34. doi:10.3400/avd.AVDrev07011. Epub 2008 Feb 15.
- Hong YK, Detmar M. Prox1, master regulator of the lymphatic vasculature phenotype. Cell Tissue Res. 2003;314(1):85–92. Oct. Epub 2003 Jul 22.
- 4. Schneider DJ, Moore JW. Patent ductus arteriosus. Circulation. 2006;114(17):1873–82.
- Kenny D, Hijazi ZM. Coarctation of the aorta: from fetal life to adulthood. Cardiol J. 2011;18(5):487–95.
- Cinà CS, Althani H, Pasenau J, Abouzahr L. Kommerell's diverticulum and right-sided aortic arch: a cohort study and review of the literature. J Vasc Surg. 2004;39(1):131–9.
- 7. Ciervo A, Kahn M, Pangilinan AJ, Dardik H. Absence of the brachial artery: report of a rare human variation

and review of upper extremity arterial anomalies. J Vasc Surg. 2001;33(1):191–4.

- van Hooft IM, Zeebregts CJ, van Sterkenburg SM, de Vries WR, Reijnen MM. The persistent sciatic artery. Eur J Vasc Endovasc Surg. 2009;37(5):585–91.
- Pillai J. A current interpretation of popliteal vascular entrapment. J Vasc Surg. 2008;48(6 Suppl):61S–5; discussion 65S.
- Spentzouris G, Zandian A, Cesmebasi A, Kinsella CR, Muhleman M, Mirzayan N, Shirak M, Tubbs RS, Shaffer K, Loukas M. The clinical anatomy of the inferior vena cava: a review of common congenital anomalies and considerations for clinicians. Clin Anat. 2014;27(8):1234–43.
- Natsis K, Tsitouridis I, Totlis T, Levva S, Tsikaras P, Skandalakis P. Proposal for classification of the circumaortic renal collar's morphology. Am Surg. 2008;74(12):1190–4.

- Uhl JF, Gillot C, Chahim M. Anatomical variations of the femoral vein. J Vasc Surg. 2010;52(3):714–9.
- Vural E, Ramakrishnan J, Cetin N, et al. The expression of vascular endothelial growth factor and its receptors in port-wine stains. Otolaryngol Head Neck Surg. 2008;139:560.
- Shirley MD, Tang H, Gallione CJ, et al. Sturge-Weber syndrome and port-wine stains caused by somatic mutation in GNAQ. N Engl J Med. 2013;368:1971.
- Gloviczki P, Driscoll DJ. Klippel-Trenaunay syndrome: current management. Phlebology. 2007;22(6):291–8.
- Jacob AG, Driscoll DJ, Shaughnessy WJ, et al. Klippel-Trénaunay syndrome: spectrum and management. Mayo Clin Proc. 1998;73:28.
- Koltowska K, Betterman KL, Harvey NL, Hogan BM. Getting out and about: the emergence and morphogenesis of the vertebrate lymphatic vasculature. Development. 2013;140(9):1857–70.

Cardio Vascular Developmental Abnormality of the Human Fetus Appearing Within Second Trimester: Detection and Treatment

28

Bhabotosh Biswas

Introduction

Fetal Cardio Vascular defects (CVD) have been studied extensively over last two decades. Advancement in imaging technology has helped substantially the Paediatric Cardiologists to gather detailed information of the fetal heart and enabled them to counsel families on "What to expect after delivery".

Earlier the responsibility of treating the babies born with major Cardio Vascular defects were with Obstetrician, Neonatologists and Paediatric Cardiologists. Babies with major CVD used to reach tertiary cardiac care centres in critical condition. As a result mortality used to be unacceptably high.

With the introduction of advanced imaging modalities, the fetal medicine became multi disciplinary. Once the structural cardiac defects are identified by obstetric sonologist, the precise details of CV defects are studied by fetal echocardiographer. In selected subsets of fetuses further details may be gathered with other imaging modalities. Finally a detailed treatment policy can be formulated well before the baby is born. Thus fetal medicine has emerged as a high end speciality for collaborative multi disciplinary care for the fetuses with complex structural CV defects and arrhythmias.

Major congenital heart disease (CHD) is not uncommon with an incidence of 6–12 per 1000 live births [1–4]. Unfortunately sufficient information about major CHD in fetuses is not available in current literature. A study from Belgium [5] reported an incidence of 8.3 % in live and stillborn infants of ≥ 26 weeks of gestation without chromosome abnormalities. There is likely an even higher incidence in early gestation given spontaneous and elective pregnancy termination [5].

Major CHD is responsible for 40 % of perinatal deaths [3, 4] and more than 20 % of deaths within the first month of life [6]. Unfortunately detection rate of CHD still remains slow. Prenatal detection rate of CHD varies widely ranging from 16 % to 65 % [7, 8]. CHD is wrongly diagnosed even in tertiary care centres by expert teams consisting of Cardiologists and Fetal Sonologists. Poor detection of CHD may be attributed to several factors including gestational age, imaging technique and experience of the Sonologists in addition to maternal and amniotic fluid factors.

Most of the CHDs develop during the first 8 weeks of pregnancy. Many of them continue to develop with advancing gestation [9-12]. Arch obstruction and obstructive lesions of Cardiac chambers may progress from mild to a critical state with advancing gestation. Though Cardiac

B. Biswas, MS, MCh (Cardiothoracic Surgery) Vice-Chancellor, West Bengal University of Health Sciences, Kolkata, West Bengal, India

Formerly, Professor, Cardio-Cascular Surgery, RG Kar Hospital, Kolkata, West Bengal, India e-mail: bhabatos@yahoo.co.in

[©] Springer International Publishing Switzerland 2016 N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development:*

First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_28

dysrhythmias develop and progress through gestation, Cardiac neoplasm develops during later half of pregnancy only.

Fetal echocardiography is the most valuable tool for detecting major congenital Cardiac defects and has proved itself invaluable for early detection of such lesions. Though majority of the defects can be diagnosed in early pregnancy, few of them are detected only during the later part of gestation.

Fetal echocardiography is usually performed from 18 weeks onwards. Presently first trimester echocardiography is increasingly used for earlier detection of major congenital Cardiac defects. Several studies on first trimester echocardiography detected major CHD at 11–14 weeks in high risk fetuses [13, 14].

A Scientific Statement from the American Heart Association on Diagnosis and Treatment of Fetal Cardiac Disease was published by in Circulation, 2014 [5]. The statement reviewed available literature and commented on the current practice of fetal Cardiac medicine including diagnosis and management [5]. The authors concluded that fetal Cardiac medicine has progressed considerably over the past two decades. Advances in imaging technology and innovations in therapies have been remarkable. Fetal echocardiography has proved to be invaluable in the diagnosis of major congenital Cardiac defects. New technologies including 3D and 4D echo, MRI, fetal Electrocardiography as well as Magnetocardiography are now available. Medical therapy as well as interventions for complex disorders have made substantial changes in outcome for high risk fetuses.

Extensive studies have identified the causes of referral for fetal Cardiac evaluation. Suspicion of a structural abnormality of heart on Obstetric Ultrasonography was the leading cause of referral in 40–50 % of fetuses. A family history of CHD as well as maternal metabolic diseases were other important referring factors.

Fetal echocardiography may be considered when risk is estimated at 1–2 %, although the relative benefit of this additional testing in this population is less clear. When risk approaches that of the general population (≤ 1 %), fetal echocardiography is not indicated. It should be noted, however, that all fetuses with an abnormal screening ultrasound of the heart should have a detailed fetal echocardiogram by a trained examiner [5].

Common Indications for Referral for Fetal Echocardiography

Several factors, both maternal as well as fetal may be responsible for complicating pregnancies and increasing the risk for major fetal CHD. Following few factors may be mentioned in this context.

- Maternal factors
 - Diabetes Mellitus (DM) affects 3–10 % of pregnancies and causes fivefold increase in CHD compared to the general population of women with DM
 - Medicines taken during pregnancy may be responsible for several congenital defects including Cardiac deformities. A number of teratogens are in clinical use during child bearing age. Use of these medicines during early pregnancy (Cardiogenesis period) increase the risk of CHD. Most important medicines known to cause Cardiac defects are anticonvulsants, vitamin K antagonist, NSAID, Lithium, ACE inhibitors etc.
 - Maternal infections are also responsible for teratogenesis including defective formation of heart.
 - Infertility treatment adopting assisted reproductive technology have also been held responsible for increase in fetal CHD.
 - Family history of CHD has also been evidenced to increase fetal CHD.
- Fetal factors
 - Suspected anatomical defect of heart in obstetric ultrasound warrants further evaluations by fetal echocardiography for early detection and planning of therapy for fetal CHD.
 - Fetal arrhythmias may be associated with CHD. Fetal bradycardia resulting from abnormal AV conduction is more likely to be associated with CHD compared to fetal tachycardia.

- CHD may be associated with non Cardiac malformations as well.
- Increased nuchal translucency on first trimester screening is an important indicator to suspect fetal CHD.
- Abnormalities of umbilical cord and venous system is associated with higher incidence of CHD.

Fetal Echocardiography

A paragraph regarding Fetal echocardiography at this juncture may help further understanding of fetal CHD and enable the physician as well as family to take a right decision in time

Fetal echocardiography remains the primary tool for the evaluation of major CHD, though its scope may be controversial in some cases. Current fetal echocardiography has included a wide variety of modalities which enable to study the fetal Cardiac structure and functions more precisely and elaborately. Various components of fetal echocardiography have been recommended by different authors.

Fetal echocardiography for screening of CHD of high risk pregnancies are usually performed between 18th and 22nd weeks of gestation. However this timing may fail to identify Cardiac disorders that progress from subtle pathology in midterm to major CHD in full term. Detection of abnormal finding on routine Obstetric Ultrasound usually indicates fetal echocardiograph for detailed Cardio-Vascular evaluation. After detailed initial evaluation, follow up fetal echocardiogram has a specific role for detecting progressive fetal CHD. Addition of color Doppler imaging provide invaluable information regarding valve function and vessel patency.

Prenatal Counseling has a specific role in the work up of fetal CHD.

Once an accurate diagnosis of fetal CHD is made after proper evaluation, it must be communicated to the family members providing proper management plan and a truthful picture of prognosis. This helps the parents to take appropriate decision about termination of pregnancy or about palliative care to be given to the baby after birth.

Therapeutic Options for Fetal Cardio: Vascular Defects

Institution of appropriate therapy for congenital heart defects before birth is now a reality.

Improved accuracy in diagnosis of Cardiac defects in utero helps to make therapeutic plan well in time. Varieties of therapeutic options are now available for treating various fetal Cardiac defects.

Followingtherapeuticoptionsarecurrentlyavailable-

- (a) Maternal administration of medication with transplacental transfer to the fetus
- (b) Minimally invasive fetoscopic guided techniques/
- (c) Open uterine fetal surgery

However fetal therapy is still in infancy. Risk to the mother as well as fetus are real concerns for current day fetal therapy.

Certain Cardiac lesions are now considered suitable for fetal surgery. Aortic Stenosis (AS) with evolving hypoplastic left heart syndrome (HLHS), HLHS with restrictive or intact atrial septum, pulmonary atresia with intact ventricular septum, mitral valve dysplasia with MR and AS etc. are few conditions in which fetal Cardiac surgery has proved useful.

Therapy for Fetal Cardiovascular Defects before Birth

(a) Maternal administration of medication with transplacental transfer to the fetus –

This modality has been established as a therapeutic option specially for the following clinical conditions—

Fetal Arrhythmias

Fetal Bradycardia

The cause and mechanism of fetal bradycardia dictates treatment strategy. It is usually close observation for signs of fetal distress. Decisions on early delivery and resultant prematurity must be weighed against therapies available. Risk of both mother and fetus needs consideration. If bradycardia persists, it should be evaluated thoroughly after delivery.

AV Block

Treatment of CHB depends on several factors including the origin, the ventricular rate, and the degree of heart failure. Regardless of the origin, the use of β -sympathomimetics (terbutaline, salbutamol, isoprenaline) to augment fetal ventricular rates when <55 bpm has been reported. Terbutaline is well tolerated, although maternal resting heart rates of 100–120 bpm and benign ectopy may be encountered [15].

Fetal pacing does not improve survival [16] and still experimental and therefor not recommended.

Fetal Tachycardia

Fetal tachycardia is a rare but an important cause of intrauterine fetal nonimmune hydrops, premature delivery, and perinatal morbidity and mortality. Treatment of fetal tachycardia depends on its cause. Conversion to 100 % sinus rhythm is not the target but to achieve sufficient sinus rhythm to allow reasonable ventricular function.

Irregular Rhythm

Is found in 1-3 % of all pregnancies and is usually harmless of pregnancy.

Anti Arrhythmic Drugs

Relatively high doses are required during the second and third trimesters of pregnancy due to increased maternal circulating blood volume and renal clearance. Maternal transplacental treatment is initiated in the hospital and in most cases, oral antiarrhythmic agents are continued.

Congestive Heart Failure of Fetus

Transplacental digoxin may be considered, although its usefulness is doubtable

(b) Minimally invasive fetoscopic guided techniques/interventions

Cardiac Catheter Intervention in Fetus

Cardiac lesions that are amenable to fetal intervention are distinctive in that they can lesion progressing rapidly to severely during gestation myocardial damage need fetal catheterisation.

Fetal cardiac intervention usually leading to alter the natural history of the disease and results in an improved state at birth reducing short- or long-term morbidity or mortality.

Following Cardiac Lesions May Be Amenable to Fetal Intervention

AS with Evolving HLHS

Severe AS in early and mid gestation evolving into HLHS at birth, are suitable candidates for such therapy [17–26].

HLHS with Intact Atrial Septum

HLHS with intact atrial septum have very high mortality and morbidity even after neonatal survival [27]. The fetus with this condition is stable in utero, [27, 28] the newborn becomes critically ill immediately after birth and succumbs to hypoxia, acidosis, and pulmonary oedema. If diagnosed prenatally, a planned delivery with urgent catheterization and balloon dilation or stent dilation of the atrial septum may be done. However favourable, outcome is not expected [29, 30].

Mitral Valve Dysplasia with Mitral Regurgitation and Aortic Stenosis

Fetal cardiac intervention to open the aortic valve may be useful but may end in gross aortic regurgitation [31] Opening of the atrial septum to decompress the left atrium may also help [32].

Pulmonary Atresia with Intact Ventricular Septum

Fetuses with PA and IVS may be candidates for fetal cardiac intervention. It may prevent the need for single-ventricle palliation after birth [33–35].

(c) Open uterine fetal surgery

Surgical Techniques

Fetal surgery may be performed with hysterotomy and exposure of the fetus or through laparoscopy.

Fetal surgery may be suitable in the following conditions ---

Giant sacrococcygeal teratomas large congenital cystic adenomatoid malformations with signs of hydrops, severe congenital diaphragmatic hernia, and meningomyeloceles.

Fetal echocardiography may be useful for assessment of cardiac function and fetal circulation before, during, and after surgery.

Left ventricular hypoplasia may be associated with congenital diaphragmatic hernias resulting from ventricular compression or diminished filling secondary to pulmonary hypoplasia and decreased pulmonary venous return [36].

Open Fetal Surgery

Surgical repair of CHD in a fetus may offer greater benefit than over postnatal repair in selected patients.

However optimal techniques and method of selection of proper candidates, are yet to be established.

In animal models, Cardio Pulmonary Bypass in the fetus results in significant placental dysfunction [37].

Impacts of Fetal Surgery

Need for ventricular shunt procedures was found to be reduced after Open fetal surgery for meningomyelocele repair done before 26 weeks of gestation. Improvement in motor outcome at 30 months of age were also comparable in this subset of cases.

Resection and acute relief of cardiac tamponade in congenital cystic adenomatoid malformation was found to result in acute mismatch in volume with filling impairment along with ventricular dysfunction.

Potential Benefits of Perinatal Management of CHD in Fetuses

Provides sufficient time for extensive prenatal counselling and allows better prediction of the clinical course in utero as well as during the circulatory transition occurring with delivery.

Benefits of Prenatal Diagnosis and Perinatal Management

Critical neonatal CHD diagnosed prenatally, affects neonatal morbidity and mortality. Fetuses

with ductal-dependent pulmonary or systemic blood flow require institution of prostaglandin infusion soon after birth to prevent ductal closure. Because the ductus arteriosus does not close at delivery, these newborns are not expected to be compromised in the delivery room [38–40] and can be stabilized by collaborative team efforts of neonatologists and pediatric cardiologist before transfer for surgical intervention.

Ductus dependent CHD diagnosed prenatally, have been shown to be less compromised preoperatively than infants diagnosed after birth. Improved arterial pH, improved oxygenation, less myocardial dysfunction, and less end-organ disease such as necrotizing enterocolitis and renal injury have been observed in cases diagnosed prenatally [41, 42].

Prenatal diagnosis of CHD may improve fetal and perinatal outcome by reducing intrauterine heart failure, as it helps to initiate intrauterine medical therapy and optimization of perinatal management strategies, including early delivery, if necessary.

Delivery Planning

For assuring optimal benefit for fetal CHD diagnosed prenatally, perfect coordination of intrapartum care is essential between Obstetric, Neonatal, Cardiology including Interventional and Electrophysiology, Intensive care, as well as Cardiac Surgery. Overall neonatal condition and surgical outcomes are improved by delivery in close proximity to a Cardiac center with the resources needed to provide medical and surgical interventions for infants with specific major cardiac defects [42–46].

Delay of elective delivery until 39 completed weeks of gestation has been shown to improve neonatal outcomes [47]; however, waiting beyond 42 weeks has been shown to be detrimental [48].

Recommendations for Transitional Care

Fetal diagnosis of high risk complex CHD prevents the postnatal hemodynamic instability during transition at delivery [42, 44, 49–53]. Usually two major systems-Cardio Vascular and Respiratory, play a significant role in successful fetal-neonatal transition. In case it is apprehended that one or both of these systems cannot transition normally, a specialized plan of care needs to be organised for a favourable outcome.

Left-to-right shunt lesions (Ventricular septal defects and Atrial Septal Defects) without severe Pulmonary Arterial Hypertension (PAH) are generally stable. Mild valve disease with normal cardiac function are usually asymptomatic in the neonatal period, although progression of valve dysfunction may occur rapidly in some cases [40, 54–57]. For minimal-risk neonates, specialized care is not recommended in the delivery room.

Fetal Arrhythmias

Fetal arrhythmias may require intervention- electric conversion to sinus rhythm or administration of antiarrhythmic drugs, in the delivery room itself, specially if the delivery is due to impending heart failure, hydrops, or fetal distress [58–60].

Fetal medicine has already emerged as a high end speciality. Role of fetal medicine in the field of major Cardio Vascular disorder occurring in first and second trimester of gestation has been well established. Advanced imaging modalities specially fetal echocardiography has a major role for in-utero diagnosis and planning multispecialty well co-ordinated collaborative therapy for fetal complex cardio vascular disorders. Though extensive studies during last two decades have contributed tremendously to enrich fetal medicine, the super speciality has to go a long way to offer safe remedies for most of the complex cardio vascular disorders diagnosed during first and second trimester of gestation.

References

- Ferencz C, Rubin JD, McCarter RJ, Brenner JI, Neill CA, Perry LW, Hepner SI, Downing JW. Congenital heart disease: prevalence at livebirth: the Baltimore-Washington Infant Study. Am J Epidemiol. 1985;121:31–6.
- Hoffman JI. Congenital heart disease: incidence and inheritance. Pediatr Clin North Am. 1990;37:25–43.
- Tegnander E, Williams W, Johansen OJ, Blaas HG, Eik-Nes SH. Prenatal detection of heart defects in a non-selected population of 30,149 fetuses: detection rates and outcome. Ultrasound Obstet Gynecol. 2006;27:252–65.

- Wren C, Richmond S, Donaldson L. Temporal variability in birth prevalence of cardiovascular malformations. Heart. 2000;83:414–9.
- 5. Donofrio MT, Moon-Grady AJ., Hornberger LK., Copel JA, Sklansky MS, Abuhamad A, Cuneo BF, Huhta JC, Jonas RA, Krishnan A, Lacey S, Lee W, Michelfelder EC Sr, Rempel GR, Silverman NH, Spray TL, Strasburger JF, Tworetzky W, Rychik J; on behalf of the American Heart Association Adults with Congenital Heart Disease Joint Committee of the Council on Cardiovascular Disease in the Young and Council on Clinical Cardiology, Council on Cardiovascular Surgery and Anesthesia, and Council on Cardiovascular and Stroke Nursing. AHA scientific statement diagnosis and treatment of fetal cardiac disease a scientific statement from the American Heart Association. Circulation. 2014;129:2183–242.
- Moons P, Sluysmans T, De Wolf D, Massin M, Suys B, Benatar A, Gewillig M. Congenital heart disease in 111 225 births in Belgium: birth prevalence, treatment and survival in the 21st century. Acta Paediatr. 2009;98:472–7.
- Rowland TW, Hubbell Jr JP, Nadas AS. Congenital heart disease in infants of diabetic mothers. J Pediatr. 1973;83:815–20.
- Ray JG, O'Brien TE, Chan WS. Preconception care and the risk of congenital anomalies in the offspring of women with diabetes mellitus: a meta-analysis. QJM. 2001;94:435–44.
- Axt-Fliedner R, Kreiselmaier P, Schwarze A, Krapp M, Gembruch U. Development of hypoplastic left heart syndrome after diagnosis of aortic stenosis in the first trimester by early echocardiography. Ultrasound Obstet Gynecol. 2006;28:106–9.
- Gardiner HM. Progression of fetal heart disease and rationale for fetal intracardiac interventions. Semin Fetal Neonatal Med. 2005;10:578–85.
- Trines J, Hornberger LK. Evolution of heart disease in utero. Pediatr Cardiol. 2004;25:287–98.
- Yagel S, Weissman A, Rotstein Z, et al. Congenital heart defects: natural course and in utero development. Circulation. 1997;96:550–5.
- Carvalho JS. Fetal heart scanning in the first trimester. Prenat Diagn. 2004;24:1060–7.
- Haak MC, Twisk JW, Van Vugt JM. How successful is fetal echocardiographic examination in the first trimester of pregnancy? Ultrasound Obstet Gynecol. 2002;20:9–13.
- Cuneo BF, Zhao H, Strasburger JF, Ovadia M, Huhta JC, Wakai RT. Atrial and ventricular rate response and patterns of heart rate acceleration during maternalfetal terbutaline treatment of fetal complete heart block. Am J Cardiol. 2007;100:661–5.
- Cuneo BF, Strasburger JF, Yu S, Horigome H, Hosono T, Kandori A, Wakai RT. In utero diagnosis of long QT syndrome by magnetocardiography. Circulation. 2013;128:2183–91.
- Makikallio K, McElhinney DB, Levine JC, Marx GR, Colan SD, Marshall AC, Lock JE, Marcus EN, Tworetzky W. Fetal aortic valve stenosis and the evo-

lution of hypoplastic left heart syndrome: patient selection for fetal intervention. Circulation. 2006;113:1401–5.

- McElhinney DB, Marshall AC, Wilkins-Haug LE, Brown DW, Benson CB, Silva V, Marx GR, Mizrahi-Arnaud A, Lock JE, Tworetzky W. Predictors of technical success and postnatal biventricular outcome after in utero aortic valvuloplasty for aortic stenosis with evolving hypoplastic left heart syndrome. Circulation. 2009;120:1482–90.
- Maxwell D, Allan L, Tynan MJ. Balloon dilatation of the aortic valve in the fetus: a report of two cases. Br Heart J. 1991;65:256–8.
- Kohl T, Sharland G, Allan LD, Gembruch U, Chaoui R, Lopes LM, Zielinsky P, Huhta J, Silverman NH. World experience of percutaneous ultrasoundguided balloon valvuloplasty in human fetuses with severe aortic valve obstruction. Am J Cardiol. 2000;85:1230–3.
- Simpson JM, Sharland GK. Natural history and outcome of aortic stenosis diagnosed prenatally. Heart. 1997;77:205–10.
- 22. Tworetzky W, Wilkins-Haug L, Jennings RW, van der Velde ME, Marshall AC, Marx GR, Colan SD, Benson CB, Lock JE, Perry SB. Balloon dilation of severe aortic stenosis in the fetus: potential for prevention of hypoplastic left heart syndrome: candidate selection, technique, and results of successful intervention. Circulation. 2004;110:2125–31.
- Wilkins-Haug LE, Tworetzky W, Benson CB, Marshall AC, Jennings RW, Lock JE. Factors affecting technical success of fetal aortic valve dilation. Ultrasound Obstet Gynecol. 2006;28:47–52.
- Marshall AC, Tworetzky W, Bergersen L, McElhinney DB, Benson CB, Jennings RW, Wilkins-Haug LE, Marx GR, Lock JE. Aortic valvuloplasty in the fetus: technical characteristics of successful balloon dilation. J Pediatr. 2005;147:535–9.
- Arzt W, Wertaschnigg D, Veit I, Klement F, Gitter R, Tulzer G. Intrauterine aortic valvuloplasty in fetuses with critical aortic stenosis: experience and results of 24 procedures. Ultrasound Obstet Gynecol. 2011;37:689–95.
- Mizrahi-Arnaud A, Tworetzky W, Bulich LA, Wilkins-Haug LE, Marshall AC, Benson CB, Lock JE, McElhinney DB. Pathophysiology, management, and outcomes of fetal hemodynamic instability during prenatal cardiac intervention. Pediatr Res. 2007;62:325–30.
- Rychik J, Rome JJ, Collins MH, DeCampli WM, Spray TL. The hypoplastic left heart syndrome with intact atrial septum: atrial morphology, pulmonary vascular histopathology and outcome. J Am Coll Cardiol. 1999;34:554–60.
- Graziano JN, Heidelberger KP, Ensing GJ, Gomez CA, Ludomirsky A. The influence of a restrictive atrial septal defect on pulmonary vascular morphology in patients with hypoplastic left heart syndrome. Pediatr Cardiol. 2002;23:146–51.
- Vlahos AP, Lock JE, McElhinney DB, van der Velde ME. Hypoplastic left heart syndrome with intact or

highly restrictive atrial septum: outcome after neonatal transcatheter atrial septostomy. Circulation. 2004;109:2326–30.

- Glatz JA, Tabbutt S, Gaynor JW, Rome JJ, Montenegro L, Spray TL, Rychik J. Hypoplastic left heart syndrome with atrial level restriction in the era of prenatal diagnosis. Ann Thorac Surg. 2007;84:1633–8.
- 31. Vogel M, McElhinney DB, Wilkins-Haug LE, Marshall AC, Benson CB, Juraszek AL, Silva V, Lock JE, Marx GR, Tworetzky W. Aortic stenosis and severe mitral regurgitation in the fetus resulting in giant left atrium and hydrops: pathophysiology, outcomes, and preliminary experience with pre-natal cardiac intervention. J Am Coll Cardiol. 2011;57:348–55.
- Rogers LS, Peterson AL, Gaynor JW, Rome JJ, Weinberg PM, Rychik J. Mitral valve dysplasia syndrome: a unique form of left-sided heart disease. J Thorac Cardiovasc Surg. 2011;142:1381–7.
- 33. Roman KS, Fouron JC, Nii M, Smallhorn JF, Chaturvedi R, Jaeggi ET. Determinants of outcome in fetal pulmonary valve stenosis or atresia with intact ventricular septum. Am J Cardiol. 2007;99:699–703.
- 34. Salvin JW, McElhinney DB, Colan SD, Gauvreau K, del Nido PJ, Jenkins KJ, Lock JE, Tworetzky W. Fetal tricuspid valve size and growth as predictors of outcome in pulmonary atresia with intact ventricular septum. Pediatrics. 2006;118:e415–20.
- 35. Gardiner HM, Belmar C, Tulzer G, Barlow A, Pasquini L, Carvalho JS, Daubeney PE, Rigby ML, Gordon F, Kulinskaya E, Franklin RC. Morphologic and functional predictors of eventual circulation in the fetus with pulmonary atresia or critical pulmonary stenosis with intact septum. J Am Coll Cardiol. 2008;51:1299–308.
- Vogel M, McElhinney DB, Marcus E, Morash D, Jennings RW, Tworetzky W. Significance and outcome of left heart hypoplasia in fetal congenital diaphragmatic hernia. Ultrasound Obstet Gynecol. 2010;35:310–7.
- Hanley FL. Fetal cardiac surgery. Adv Card Surg. 1994;5:47–74.
- Penny DJ, Shekerdemian LS. Management of the neonate with symptomatic congenital heart disease. Arch Dis Child Fetal Neonatal Ed. 2001;84:F141–5.
- Johnson BA, Ades A. Delivery room and early postnatal management of neonates who have prenatally diagnosed congenital heart disease. Clin Perinatol. 2005;32:921–46. ix.
- Rowland DG, Hammill WW, Allen HD, Gutgesell HP. Natural course of isolated pulmonary valve stenosis in infants and children utilizing Doppler echocardiography. Am J Cardiol. 1997;79:344–9.
- 41. Divanovic A, Hor K, Cnota J, Hirsch R, Kinsel-Ziter M, Michelfelder E. Prediction and perinatal management of severely restrictive atrial septum in fetuses with critical left heart obstruction: clinical experience using pulmonary venous Doppler analysis. J Thorac Cardiovasc Surg. 2011;141:988–94.
- Tworetzky W, McElhinney DB, Reddy VM, Brook MM, Hanley FL, Silverman NH. Improved surgical

outcome after fetal diagnosis of hypoplastic left heart syndrome. Circulation. 2001;103:1269–73.

- Friedberg MK, Silverman NH, Moon-Grady AJ, Tong E, Nourse J, Sorenson B, Lee J, Hornberger LK. Prenatal detection of congenital heart disease. J Pediatr. 2009;155:26–31. 31.e1.
- 44. Bonnet D, Coltri A, Butera G, Fermont L, Le Bidois J, Kachaner J, Sidi D. Detection of transposition of the great arteries in fetuses reduces neonatal morbidity and mortality. Circulation. 1999;99:916–8.
- 45. Copel JA, Tan AS, Kleinman CS. Does a prenatal diagnosis of congenital heart disease alter short-term outcome? Ultrasound Obstet Gynecol. 1997;10: 237–41.
- 46. Verheijen PM, Lisowski LA, Stoutenbeek P, Hitchcock JF, Brenner JI, Copel JA, Kleinman CS, Meijboom EJ, Bennink GB. Prenatal diagnosis of congenital heart disease affects preoperative acidosis in the newborn patient. J Thorac Cardiovasc Surg. 2001;121:798–803.
- 47. Tita AT, Landon MB, Spong CY, Lai Y, Leveno KJ, Varner MW, Moawad AH, Caritis SN, Meis PJ, Wapner RJ, Sorokin Y, Miodovnik M, Carpenter M, Peaceman AM, O'Sullivan MJ, Sibai BM, Langer O, Thorp JM, Ramin SM, Mercer BM, Eunice Kennedy Shriver NICHD Maternal-Fetal Medicine Units Network. Timing of elective repeat cesarean delivery at term and neonatal outcomes. N Engl J Med. 2009;360:111–20.
- Hilder L, Costeloe K, Thilaganathan B. Prolonged pregnancy: evaluating gestation-specific risks of fetal and infant mortality. Br J Obstet Gynaecol. 1998;105:169–73.
- Eapen RS, Rowland DG, Franklin WH. Effect of prenatal diagnosis of critical left heart obstruction on perinatal morbidity and mortality. Am J Perinatol. 1998;15:237–42.
- 50. Kumar RK, Newburger JW, Gauvreau K, Kamenir SA, Hornberger LK. Comparison of outcome when hypoplastic left heart syndrome and transposition of the great arteries are diagnosed prenatally versus when diagnosis of these two conditions is made only postnatally. Am J Cardiol. 1999;83:1649–53.
- 51. Verheijen PM, Lisowski LA, Stoutenbeek P, Hitchcock JF, Bennink GB, Meijboom EJ. Lactacidosis

in the neonate is minimized by prenatal detection of congenital heart disease. Ultrasound Obstet Gynecol. 2002;19:552–5.

- 52. Fuchs IB, Muller H, Abdul-Khaliq H, Harder T, Dudenhausen JW, Henrich W. Immediate and longterm outcomes in children with prenatal diagnosis of selected isolated congenital heart defects. Ultrasound Obstet Gynecol. 2007;29:38–43.
- 53. Jaeggi ET, Sholler GF, Jones OD, Cooper SG. Comparative analysis of pattern, management and outcome of pre- versus postnatally diagnosed major congenital heart disease: a population-based study. Ultrasound Obstet Gynecol. 2001;17:380–5.
- 54. Xu Z, Owens G, Gordon D, Cain C, Ludomirsky A. Noninvasive creation of an atrial septal defect by histotripsy in a canine model. Circulation. 2010;121: 742–9.
- Anand R, Mehta AV. Progressive congenital valvar aortic stenosis during infancy: five cases. Pediatr Cardiol. 1997;18:35–7.
- Gielen H, Daniels O, van Lier H. Natural history of congenital pulmonary valvar stenosis: an echo and Doppler cardiographic study. Cardiol Young. 1999;9:129–35.
- 57. Anagnostou K, Messenger L, Yates R, Kelsall W. Outcome of infants with prenatally diagnosed congenital heart disease delivered outside specialist paediatric cardiac centres. Arch Dis Child Fetal Neonatal Ed. 2013;98:F218–21.
- 58. Jaeggi ET, Fouron JC, Silverman ED, Ryan G, Smallhorn J, Hornberger LK. Transplacental fetal treatment improves the outcome of prenatally diagnosed complete atrioventricular block without structural heart disease. Circulation. 2004;110:1542–8.
- 59. Glatz AC, Gaynor JW, Rhodes LA, Rychik J, Tanel RE, Vetter VL, Kaltman JR, Nicolson SC, Montenegro L, Shah MJ. Outcome of high-risk neonates with congenital complete heart block paced in the first 24 hours after birth. J Thorac Cardiovasc Surg. 2008;136:767–73.
- 60. Mirlesse V, Cruz A, LeBidois J, Diallo P, Fermont L, Kieffer F, Magny JF, Jacquemard F, Levy R, Voyer M, Daffos F. Perinatal management of fetal cardiac anomalies in a specialized obstetric-pediatrics center. Am J Perinatol. 2001;18:363–71.

Part IX

Fetal Neurological Development: Up to Second Trimester and Its Implication in Adult Neurodegenerative Diseases

Implications of Foetal Development and Neurodegeneration in Adult Life

29

Biman Kanti Ray

Introduction

The role of early life in health and development of an individual is very important. Developmental period is not controlled by a strict, hard-wired genetic programme. Now this period is considered as a plastic phase which allows the organism to adopt to changes in the environment, specially during early development when cells are differentiating and tissues are developing.

Environmental Factors in Neurodegeneration

Interactions between genes and environmental factors during developmental period play an important role not only in human behavior but also in susceptibility to disease [1]. Environmental factors such as foods, metals, pollutants, microorganisms in early life interact with the specific loci thereby modifying their expression and play a direct or indirect role in brain health [2–8]. Epidemiological and animal based studies have suggested that adverse prenatal and postnatal environmental conditions increase the risk of neurodegenerative diseases [9–13].

Department of Neurology, RG Kar Medical College, Kolkata, India e-mail: biman.kanti@rediffmail.com

Human Studies

Studies have revealed that chronic placental insufficiency may result in foetal hypoxemia which leads to synaptic dysfunction that triggers neurodegeneration in neonates [14]. Maternal hormonal disturbances also affects foetus adversely. Welldesigned human studies showing the effect of inutero exposure on neurodegeneration are limited. Most studies are either retrospective or short duration prospective.

Animal Studies

Animals provide an excellent model for prospective study of early life exposure due to shorter life span and better follow-up. Perinatal hypoxia results in retinal ganglion cell degeneration in rat models and brain damage in the rabbits and the developing brain is more vulnerable than the adult brain to the same insult [15–17]. In an interesting study, pups of female exposed to bacterial endotoxin, Lipopolysaccharide (LPS) during pregnancy showed loss of dopaminergic neurons enhancing susceptibility to Parkinsons disease [27].

The effect of following environmental factors during in-utero and developmental stages of life have been studied and based on these studies, various models have been proposed.

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_29

B.K. Ray

[©] Springer International Publishing Switzerland 2016

Dietary Exposure

Foetal development is dependent on nourishment provided by the maternal system and thus deficiency or excess of any nutritive supplement to the mother adversely affects the foetus [18].

Human Studies

Vitamin B12 deficient diet to the mother during pregnancy adversely affects the myenilation in nervous system of offspring [19]. Inappropriate micronutrient supplementation in mother affect the level of anti-oxidant enzymes in the offspring, thus increasing the risk of neurodegenerative diseases [31].

Animal Studies

Correlation between maternal diet and foetal neurodegeneration has been reported in several animal studies. Maternal folate depletion results in epigenetic changes in the off-springs leading to neurodegeneration [20]. Elevated Homocysteine level in mother was shown to increase oxidative stress in pups brain leading to apoptosis [21]. High dose of Iron in neonatal stage has been shown to result in neurodegeneration of midbrain at later age [32].

Metal Exposure

Blood-Brain-Barrier and formation of metalprotein complex protects the mature brain tissue from foetal toxicity. In foetal brain this sequestering mechanism is absent.

Human Studies

Metals such as aluminium, zinc, iron, copper and mercury have been shown to be associated with neurodegenerative diseases [33]. Metals like zinc and copper are believed to result in A-beta aggregation [22]. Lead exposure in first trimester of pregnancy results in increased amplitude of a and b waves in 7–10 year old children. High level of Mercury and Lead in umbilical cord blood due to prenatal exposure impaired the visual processing in children after 11 years [23].

Animal Studies

Long – term potentiation (LTP) is considered the key mechanism underlying information storage and memory formation. LTP is dependent on presynaptic increase of neurotransmitter release and enhanced function of glutamate receptor at the post synaptic end. NMDA receptor function has been found crucial for the LTP induction in hippocampus. Exposure to metals such as zinc, copper, aluminium and lead is associated with a negative effect on LTP during developmental stage. Various studies show that the lead exposure in developmental stages results in the increased level of beta amyloid in brain causing Alzheimer in later age [24, 25].

Pesticides

Pesticides including insecticides (organophosphates, organochlorines, carbamates), herbicides and fungicides enter the body through respiratory tract, gastrointestinal tract, dermal contact and rarely via ocular exposure.

Human Studies

Pesticides get accumulated in the body and change the gene expression profile in exposed tissues through epigenetic changes. Pesticide exposure has been shown to result in neuronal loss, cognitive impairment and motor dysfunction [26].

Animal Studies

Exposure to dieldrin and paraquet during gestation and lactation has been reported to affect the dopaminergic responses in off-springs and leads to Parkinsons disease in adult life [28].

Lifestyle, Smoking and Drug Abuse

Healthy lifestyle prevents disease whereas bad habits increase susceptibility to disease. Exercise particularly aerobic exercise has a positive impact on brain functioning.

Human Studies

Childhood aerobics increases the resilience of the brain in later life. Similarly, the association of caffeine, smoking and alcohol consumption has been well documented in neurodegenerative diseases [29, 30].

Animal Studies

One study showed that pups born to mothers undergoing low intensity treadmill exercise during pregnancy have more hippocampal cell survival [33]. Similarly, pups performing treadmill exercises at postnatal day 21–60 showed enhanced spatial memory as compared to controls. Female rats exposed to methamphetamine at gestational stage have shown smaller optic nerve diameter than controls [33].

Mechanism, Hypothesis and Models

Epigenetics

Epigenetics is an emerging field that focuses on the mechanisms that alter the function of genes independent of the primary DNA sequence [33, 34]. Thus, independently of the genotype, different epigenetic profiles may result in different phenotypes. Thus, each individual has one genome, but multiple epigenomes. Mechanisms such as DNA methylation, histone tail modification and regulation by non-coding RNAs are responsible for these alterations. These epigenetic marks, particularly if they are near the promoter region of functional genes, can influence transcription by altering access of transcriptional machinery to the DNA. DNA Methylation may occur through recruitment of Methyl-CpG binding domain proteins while histone modifications can change chromatin conformation between open and closed states, resulting in altered availability of DNA for transcription. The non-coding RNA referred to as microRNA is believed to act at post transcriptional stage thereby exerting epigenetic regulation of such changes. Recent evidence suggests micro RNAs play role in DNA Methylation. Alterations in gene expression patterns induced by these mechanisms may be more frequent than genetic mutations. It generally takes into account the gene and environment interaction such that these changes are inherited. Such epigenetic modifications can be passed from one cell generation to next and in some case, can be transgenerationally transmitted. Recent evidence supports that adult neurogenesis is under intensive regulation by epigenetic mechanisms [34]. These changes can be cell, tissue and sex specific and time dependent. The epigenome is altered by environmental signals, not only during the period of exposure but even later in life. The most sensitive window for epigenetic effects varies among different tissues and may extend into early childhood or beyond for some tissues such as brain.

Baker Hypothesis or Foetal Basis of Adult Disease (FeBAD)

According to this hypothesis, adult diseases are consequences of foetal adverse conditions [18]. Although Baker's work was mainly confined to cardiovascular diseases, this hypothesis fits well to other diseases too. Early life origins of later life disease is often referred to as programming [35]. Hales and Baker defined programming as permanent or long term change in the structure or function of an organism resulting from a stimulus or insult acting at a critical period of early life. The adult outcome of early life programming are dependent on the nature of insult, the timing in which insult occurs and the duration of insult. The insult adversely affects the organs that undergo rapid growth in the offspring at the time of the insult occurring.

Developmental Origin of Health and Disease (DOHaD)

This model is a modified version of FeBAD which postulated that postnatal period of development also plays an equal role as foetal life in health and changing environmental factors affect the patterns of diseases.

Predictive Adaptive Response (PAR)

Gluckman and Hanson have suggested that foetus undergoes adaptations during gestation or early postnatal period based on the predicted postnatal environment [35]. The foetus predicts the extrauterine environment from intra-uterine conditions and makes changes for its better survival. However if the predictive environment does not match the actual environment, then the programmed adaptations will become detrimental.

LEARn Model (Latent Early Life Associated Regulation)

Lahiri et al. suggested the role of early environmental factors in disease etology especially with respect to Alzheimer's disease [36]. LEARn model describes the environmental exposures such as nutrition, head trauma, metal exposure and life style as hits. They postulated that all neurodegenerative disorders come under the category of n hit latent model and early life environmental factors lead to epigenetic perturbations in the genes but do not result in any disease symptom. When there is a second trigger, the disease develops and the time interval between first hit and disease onset is called latency period. Genes are divided in two groups (1) which respond late in relation to early life responses (LEARned) (2) Which do not respond (unLEARned).

Epigenetic Regulation of Adult Neural Stem Cells

Since the early 1900s, regenerative mechanisms in adult CNS have been thought to be very limited. The predominant repair mechanisms considered in the CNS were post-mitotic such as sprouting of axon terminals, changes in receptor expression and synaptic reorganization. However, recent findings have clearly demonstrated that new neurons are indeed born from neural stem cells in restricted regions of adult mammalian CNS throughout life [34].

What Is Neural Stem Cells (NSC)?

NSC are defined as cells present in some areas of adult brain that have the capacity to produce new, functional neurons that are added to existing brain circuits.

During development of the mammalian CNS, neurons and glia arise from multipotent neural stem cells in which neurons are generated first, primarily during the embryonic period followed by glia, the majority of which differentiate after most neurons are born.

Although NSC may exist along the entire adult neuroaxis, adult neurogenesis has consistently been found only in subventricular zone (SVZ) of the lateral ventricle and in subgranular zone (SGZ) of the hippocampal dentate gyrus. Some authors have suggested that neurogenesis occurs also in other regions of CNS such as the cortex and substantia nigra. Neurogenic niches are unique tissue microenvironments that are permissive to the presence of NSC in the adult brain [34]. Steps involved in neurogenesis are: proliferation of neural stem cells, generation of a rapidly amplifying progenitor cell, differentiation in immature neuron, migration to the final location, growth of axon and dendrites, formation of synapses with other neurons in the circuit and ultimately maturation into a fully functional neuron. The fundamental difference between developmental and adult neurogenesis is that new adult neurons undergo these processes in an already mature environment and therefore have to integrate into preexisting circuits. Multiple intrinsic and extrinsic factors have been identified e.g. growth factors, morphogens, transcription factors and cell cycle regulators which control neurogenesis. A combination of niche signals and cell intrinsic programmes control the transition from an undifferentiated NSC state to a progenitor cell committed to the neuronal fate. Epigenitic mechanisms are likely key players within these signaling networks, as DNA methylation, chromatin remodeling and small non-coding RNAs from the microRNAs superfamily are required for fine-tuning and coordination of gene expression during adult neurogenesis.

Is Neural Stem Cells in the Adult CNS an Endogenous Source to Repair?

During the past decade, the progress in the field of stem cells has fuelled our hopes that endogenous stem cells can be mobilized to replace dying neurons in neurodegenerative diseases and cure the currently incurable diseases by cell replacement [37]. The use of endogenous sources for cell replacement offers several advantages. Many ethical concerns and political restrictions regarding the use of foetal tissue and embryonic stem cells do not apply for endogenous sources and immunological reactions are avoided. However, for successful use of endogenous neural stem cells for cell replacement, there are multiple specific challenges and problems to be overcome.

Repair from Adult Stem Cells from Neurogenic CNS Regions

Stem cells in neurogenic areas respond to injuries. There is proliferation of precursor cells in SVZ and SGZ shortly after seizure and ischaemia. But it is premature to conclude that increased neurogenesis equals the regeneration of the compromised circuit. First, there is no direct evidence that the new neurons are replacing neurons that degenerate owing to injury. Second, the generation of new neurons in neurogenic areas has not been proven to be causally linked to functional recovery. Third, lesion induced neurogenesis might also contribute to pathological alteration of function due to aberrant migration, network reorganization and altered physiological properties of the newly generated ectopic neurons. For injury induced neurogenesis to be advantageous, the new neurons have to integrate appropriately to the injured neuronal circuit, display functional properties similar to the lost neurons and be generated in number that are comparable to the number of lost neurons.

Repair from Adult Stem Cells in Nonneurogenic Regions

Several studies reported increased cell proliferation and generation of immature neurons in normally non-neurogenic region of CNS following focal stroke by occlusion of middle cerebral artery.

Implications for Treatment of Human Diseases

There are many diseases with different pathophysiologies affecting diverse regions of human CNS. We are far from understanding the processes regulating physiological neurogenesis and lesion-induced neurogenesis and it is too early to comment on the suitability of cell replacement from endogenous stem cells in the context of specific diseases. Some specific aspects of CNS diseases that may have a major impact on the potential and strategy for cell replacement therapy include:

- **Injury/Disease timing**—Whether the CNS disease is acute or death of neuron occurred over extended time-period e.g. Neurodegenerative diseases.
- **Disease location**—Neuronal cell replacement has been primarily observed in regions close to Lateral Ventricle and has been ascribed to neurons that have origin from SVZ.
- **Extent of cell loss**—Greatly influence whether functional regeneration from endogenous stem cells can be achieved.
- **Genetic basis of disease**—Genetic disease will also be present in the new neurons derived from endogenous stem cell population. It is not known whether this will lead to degeneration of the newly generated neuronal population.
- Effect of the disease on the environment— Injury or disease may create a milieu that is hostile to the generation of new neurons e.g. acute stroke increase the presence of scar inducing signals that have an inhibitory effect on endogenous stem cells. Disease like stroke affect not only neurons but also glia. Glial cells create the necessary milieu for neuronal

function. It is therefore very important to focus not only on the generation of new neurons but also on the cellular environment that supports the maturation, survival and function of new neurons.

- Therapeutic recruitment of endogenous Neural Stem cells: some unanswered questions— Are neural stem cells truly present throughout the neuroaxis? What degree of plasticity do the adult stem cells have and do the adult neural stem cells in different regions display the same degree of plasticity?
- Till now, it has been demonstrated that adult neural stem cells can give rise to the functional neuronal phenotypes of the olfactory bulb and the hippocampal dentate gyrus. But it remains unanswered whether endogenous neural stem cells can also generate other neuronal phenotypes such as functional dopaminergic neurons of the substantia nigra pars compacta or motor neurons of ventral horn. Finally, we cannot assume that our findings in animal models are readily translatable into the human system. We still have to go a long way before we can realistically propose cell replacement using endogenous stem cells as a potential source.

References

- Lahiri DK, Maloney B. The "LEARN" (latent earlylife associated regulation) model integrates environmental risk factors and the developmental basis of Alzheimer's disease and proposes remedial steps. Exp Gerontol. 2010;45:91–296.
- Charieta L, Chapronb Y, Faller P, Kirscha R, Stoned AT, Baveyee PC. Neurodegenerative diseases and exposure to the environmental metals Mn, Pb, and Hg. Coord Chem Rev. 2012;256:2147–63.
- Oteiza PL, Mackenzie GG, Verstraeten SV. Metals in neurodegeneration: involvement of oxidants and oxidant-sensitive transcription factors. Mol Aspects Med. 2004;25:103–15.
- Parron T, Requena M, Hernandez AF, Alarcon R. Association between environmental exposure to pesticides and neurodegenerative diseases. Toxicol Appl Pharmacol. 2011;256:379–85.
- Caldwell KA, Tucci ML, Amagost J, Hodges TW, Chen J, et al. Investigating bacterial sources of toxicity as an environmental contributor to dopaminergic neurodegeneration. PLoS One. 2009;4:e7227. doi:10.1371/journal.pone.0007227.

- Ali SF, Binienda ZK, Imam SZ. Molecular aspects of dopaminergic neurodegeneration: gene-environment interaction in parkin dysfunction. Int J Environ Res Public Health. 2011;8:4702–13.
- Gordon PH. Amoytrophic lateral sclerosis: an update for 2013 clinical features, pathophysiology, management and therapeutic trials. Aging Dis. 2013;4:295–310.
- Baldi I, Lebailly P, Mohammed-Brahim B, Letenneur L, Dartigues JF, Brochard P. Neurodegenerative diseases and exposure to pesticides in the elderly. Am J Epidemiol. 2002;157:409–14.
- Li N, Yu ZL, Wang L, Zheng YT, Jia JX, et al. Increased tau phosphorylation and beta amyloid in the hippocampus of mouse pups by early life lead exposure. Acta Biol Hung. 2010;61:123–34.
- Anderson OS, Sant KE, Dolinoy DC. Nutrition and epigenetics: an interplay of dietary methyl donors, one-carbon metabolism and DNA methylation. J Nutr Biochem. 2012;23:853–9.
- Rauh VA, Perera FP, Horton MK, Whyatt RM, Bansal R, Hao X, Liu J, et al. Brain anomalies in children exposed prenatally to a common organophosphate pesticide. Proc Natl Acad Sci U S A. 2012;109:7871–6.
- Xiong N, Long X, Xiong J, Jia M, et al. Mitochondrial complex I inhibitor rotenone-induced toxicity and its potential mechanism in Parkinson's disease models. Crit Rev Toxicol. 2012;42:613–32.
- Gluckman PD, Hanson MA, Cooper C, Thomberg KL. Effect of in utero and early life conditions on adult health and disease. N Engl J Med. 2008;359: 61–73.
- Kiss P, Szogyi D, Reglodi D, Horvath G, Farkas J, et al. Effects of perinatal asphyxia on the neurobehavioral and retinal development of newborn rats. Brain Res. 2009;1255:42–50.
- 15. Piscopo P, Bernardo A, Calamandrei G, Venerosi A, et al. Altered expression of cyclooxygenase-2, presenilins and oxygen radical scavenging enzymes in a rat model of global perinatal asphyxia. Exp Neurol. 2008;209:192–8.
- Van Viet E, Eixarch E, Illa M, Arbat-Plana A, Gonzalez-Tendero A, et al. Metabolomics reveals metabolic alterations by intrauterine growth restriction in the fetal rabbit brain. PLoS One. 2013;8:e64545. doi:10.1371/journal.pone.0064545.
- Johnston MV, Nakajima W, Agberg H. Mechanisms of hypoxic neurodegeneration in the developing brain. Neuroscientist. 2002;8:212–20.
- Barker DJ, Gluckman PD, Godfrey KM, Harding JE, Owens JA, Robinson JS. Foetal nutrition and cardiovascular disease in adult life. Lancet. 1993;341:938–41.
- Lovblad K, Ramelli G, Remonda L, Nirkko AC, Ozdoba C, Schroth G. Retardation of myelination due to dietary vitamin B12 deficiency: cranial MRI findings. Pediatr Radiol. 1997;27:155–8.
- 20. Langie SA, Achterfeldt S, Gomiak JP, Halley-Hogg KJA, Oxley D, et al. Maternal folate depletion and high fat feeding from weaning affects DNA methylation and DNA repair in brain of adult offspring. FASEB J. 2013;27:3323–34.

- Koza ST, Gouwyb NT, Demirc N, et al. Effects of maternal hyperhomocysteinemia induced by methionine intake on oxidative stress and apoptosis in pup rat brain. Int J Dev Neurosci. 2010;28:325–9.
- Kkozlowski H, Luczkowski M, Remelli M, Valensin D. Copper, zinc and iron in the neurodegenerative diseases(Alzheimers, Parkinsons and prion diseases). Coord Chem Rev. 2012;256:2129–41.
- 23. Ethier A, Muckle G, Bastien C, Dewilly E, Ayotte P, Arfken C, Jacobson SW, Jacobson JL, Sant AD. Effects of environmental contaminant exposure on visual brain development: a prospective electrophysiological study in school-aged children. Neurotoxicology. 2012;33:1075–85.
- 24. Basha MR, Wei W, Bakheet SA, Benitez N, Siddiqi HK, et al. The fetal basis of amyloidogenesis: exposure to lead and latent overexpression of amyloid precursor protein and beta amyloid in the aging brain. J Neurosci. 2005;25:823–9.
- 25. Wu J, Basha MR, Brock B, Cox DP, et al. Alzheimer disease-like pathology in aged monkeys after infantile exposure to environmental metal lead: evidence for a developmental origin and environmental link for AD. J Neurosci. 2008;28:3–9.
- Relton CL, Davey Smith G. Epigenetic epidemiology of common complex disease: prospect for prediction, prevention and treatment. PLoS Med. 2010;7:e1000356. doi:10.1371/journal.pmed1000356.
- Ling J, Gayle DA, Ma SY, Lipton JW, Tong CW, et al. In utero bacterial endotoxin exposure causes loss of tyrosine hydroxylase neurons in the postnatal rat midbrain. Mov Disord. 2002;17:116–24.
- Richardson JR, Caudie WM, Wang M, Dean ED, et al. Developmental exposure to the pesticide dieldrin alters the dopamine system and increases neurotoxicity in an animal model of Parkinsons disease. FASEB J. 2006;20:1695–7.

- Costa J, Lunet N, Santos C, Santos J, Vaz-Caneiro A. Caffeine exposure and the risk of Parkinsons disease: a systematic review and meta-analysis of observational studies. J Alzheimers Dis. 2010;20:5221–38.
- 30. De Zeeuw P, Zwart F, Schrama R, et al. Prenatal exposure to cigarette smoke or alcohol and cerebellar volume in attention deficit/hyperactivity disorder and typical development. Transl Psychiatry. 2012;2:e84. doi:10.1038/tp.2012.12.
- Roy S, Sable P, Khaire A, Randhir K, Kale A, Joshi S. Effect of maternal micronutrients and omega 3 fatty acids on indices of brain oxidative stress in the offspring. Brain Dev. 2014;36:219–27.
- 32. Kaur D, Peng J, Chinta SJ, Rajagopalan S, et al. Increased murine neonatal iron intake results in Parkinson-like neurodegeneration with age. Neurobiol Aging. 2007;28:907–13.
- Modgil S, Lahiri DK, Sharma VL, Anand A. Role of early life exposure and environment on neurodegeneration: implications on brain disorders. Transl Neurodegener. 2014;3:9.
- Fitzsimons CP, Van Bodegraven E, Schouten M, et al. Epigenitic regulation of adult neural stem cells: implications for Alzheimers disease. Mol Neurodegener. 2014;9:25.
- Laker RC, Wlodek ME, Connelly JJ, Yan Z. Epigenitic origins iof metabolic disease: the impact of the maternal condition to the offspring epigenome and later health consequences. Food Sci Hum Wellness. 2013; 2:1–11.
- Lahiri DK, Maloney B, Zawia NH. The LEARn model: an epigenetic explanation for idiopathic neurobiological diseases. Mol Psychiatry. 2009;14:992–1003.
- Lie DC, Song H, Colamarino SA, et al. Neurogenesis in the adult brain: new strategies for central nervous system diseases. Annu Rev Pharmacol Toxicol. 2004;44:399–421.

Part X

Fetal Neuropsychiatric Development: Up to Second Trimester

Very Low Birth Weight Babies and Their Mental Health Outcome

30

Pradeep K. Saha

Introduction

Preterm birth and low birth weight babies are an emerging public health concern in India due to several factors as improvement of early neonatal services and booking of pregnancies in India, where the survival rates of the babies of these categories are increasing. Improvements in antenatal care and advances in neonatal medicine have resulted in increased survival of infants, in particular those of very low birth weight (VLBW) (<1,500 g) and extremely low birth weight (ELBW) (<1,000 g). If simple gain in life years is taken as a standard, then neonatal care is the most successful discipline in medicine today [1]. The psychological development and quality of life of VLBW and ELBW children has become an increasing focus of recent research [2]. The focus of this chapter will be the outcomes of the survivor babies of the birth-weight range of 1,000–1,500 g.

The Institute of Medicine Committee on Understanding Premature Birth and Assuring Healthy Outcomes Board on Health Sciences

Department of Psychiatry, IPGME&R, Kolkata, 700020, India

Institute of Psychiatry – A Centre of Excellence, Kolkata, 700025, India e-mail: pradeepnimhan@gmail.com Outcomes, Washington, DC, USA (IOM) report describes prematurity not as a single disease, but a common complex condition that results from multiple gene-environmental interactions that lead to several final pathophysiological pathways to birth before 37 weeks gestation [3]. The many biological and environmental risk factors for preterm birth, including biological factors, maternal medical conditions, genetics, environmental exposures, assisted reproductive technology, behavioural and psychosocial factors, and neighborhood social characteristics, vary among populations. A prior preterm birth is the strongest risk factor. Pathways to preterm birth include inflammation (resulting from environmental exposures to antigens and infections, as well as foetal and maternal immune responses), uterine distension (e.g. multiple gestations) and deteriorating foetal or maternal health (e.g. preeclampsia). For over 50 years, preterm outcome studies have reported higher rates of cerebral palsy, intellectual disability and sensory impairment in preterm survivors as compared with peers born at term [3, 4]. The focus of more recent studies has shifted to describing school and behaviour problems (e.g. minor neuromotor dysfunction, specific learning disability and language, visual-perceptual and attention deficits), and the contribution of preterm complications and their treatments to neurodevelopmental outcomes.

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_30

P.K. Saha, MD

[©] Springer International Publishing Switzerland 2016

An Overview of the Different Outcomes of the 1,000–1,500 g Birth Weight Survivors

Neurodevelopmental Outcomes

Much of the variation in reported rates of neurodevelopmental disabilities may be due to differing methodologies, but the challenge is in discerning how much is due to neonatal intensive care unit (NICU) treatment strategies [3–9]. Commonly used terms that describe study samples (e.g. very low birth weight) have become so confusing that exact descriptions (e.g. infants with birth weight below 1,500 g) will be used here. Variations in length of follow-up, success in tracking subjects, evaluation measures and diagnostic criteria influence disability rates, as demonstrated in a German study that annually evaluated children with birth weight below 1,000 g with neurological examinations and a variety of psychometric tests (based on age) [10]. At 6-10 years, 17 % had major impairment (cerebral palsy, intelligence or developmental quotient below 70, blindness or intractable epilepsy), 42 % had minor impairment (neuromotor abnormalities, intellectual or developmental quotients of 70-84, socio emotional problems, or language, gross motor, fine motor, visual, auditory, or attention deficits), and 41 % were normal. Only 1 of 12 infants with cerebral palsy had not been diagnosed at 18 months, but cognitive impairments were more difficult to identify. The proportion of assessments that concurred with 6-10-year outcomes increased with length of follow-up: only 49 % at term (i.e. 40 weeks postmenstrual age), but 59 %, 68 %, 70 % and 70 % and at 1, 2, 3 and 4 years, respectively.

Motor Outcomes

Although many preterm infants demonstrate neuromotor abnormalities on examination, most do not develop cerebral palsy [3, 7, 10–18]. Rates of cerebral palsy in infants born after 1990 range from 4 % to 12 % for survivors with birth weight below 1,000 g, 6–20 % for infants born before

27 weeks gestation, and 21-23 % for infants born before 25 weeks gestation [3]. In children born 1994–1997 in Northern Ireland, cerebral palsy prevalence was 1.2 per 1,000 live births for birth weight over 2,499 g, 11.3 for birth weight 1,500-2,499 g and 44.5 for birth weight below 1,500 g [16]. Mortality is high in infants with birth weights below 1,000 g, so cerebral palsy developed in 47 per 1,000 live births and 99.5 per 1,000 (9.9 %) survivors. Although they constitute only 1 % of births, 22 % of children with cerebral palsy were born with birth weight below 1,500 g. An increase in cerebral palsy with decreasing birth weight and gestational age category is a consistent finding in preterm outcome studies. This finding is not limited to extremely preterm infants. Marret et al. [14] report an increase in cerebral palsy with each preterm week: 0.7 %, 3.7 %, 4.1 %, 8.7 % and 6.3 % in children with gestational age 34, 33, 32, 31 and 30 weeks, respectively (P<0.01). In a Swedish study, children born at 32-36 weeks gestation had a six times higher prevalence of cerebral palsy than children born at term (7 per 1,000 vs. 1.1 per 1,000 live births) [18]. Being born just a few weeks preterm confers enough vulnerability and complications of prematurity to increase both mortality and morbidity rates over those of fullterm neonates. A recent report from the American Academy of Paediatrics Committee on Foetus and New-born proposes guidelines for evaluating and managing 'late preterm' infants born at 34–36 6/7 weeks gestation [19]. Many children born preterm demonstrate mild fine or gross motor delay, mild but persistent neuro-motor abnormalities (e.g. asymmetries, tight heel cords), motor planning problems and/or sensorimotor integration problems that lead to funcimpairments (e.g. tying shoelaces), tional academic difficulties (e.g. writing), and socialemotional problems (e.g. poor self-esteem and peer relationships) [3, 7, 17, 20, 21]. In an excellent Australian study of 8-9-year-olds, children (especially males) with birth weight below 1,000 g or gestational age below 28 weeks with developmental coordination disorder (9.5 % vs. 2 % in normal birth weight controls, P < 0.001) had more problems with cognitive function,

academic achievement, and adaptive behaviour [20]. In a Swiss study of 6-year-olds, children with birth weight below 1,500 g had low scores on multiple motor tasks (especially quality and speed of movements) [21]. Scores correlated with brain injury on neonatal ultrasounds and neurological abnormalities on examination at age 6, but even preterm children with no neurological abnormalities had lower scores than the referent population. A British study of 6-year-olds noted poorer performance on a variety of motor tasks in children with gestational age below 26 weeks compared with full-term controls, as well as a higher prevalence of left-handed preference (28 % vs. 10 %), overflow movements during motor tasks, sensorimotor difficulties and visuospatial problems [17]. These problems adversely influenced the children's classroom performance as rated by their teachers.

Cognition and Academic Achievement

Recent studies have not only confirmed that children born preterm have more cognitive impairments and academic difficulties than full-term controls, but they also suggest that these are more common than motor, visual or hearing impairments [3, 11, 15, 17, 22]. In a study of 2-yearolds, 54 % of children born before 27 weeks gestation had a Griffith Mental Developmental Quotient greater than two standard deviations below the mean; only 40 % had normal cognitive abilities [12]. A study comparing children born with birth weight below 1,500 g during three epochs found no significant changes in cognitive tests from 1982 to 2002, but 20-24 % had Bayley Mental Developmental Index scores below 70 [15]. The French study of children born at 30-34 weeks gestation found cognitive impairment to be the most common disability, and with increasing gestational age, mean cognitive scores and proportion of children with scores above 85 improved (from 94 to 98 and from 65 % to 76 %, respectively) [14]. Difficulties with reading and spelling increased with decreasing gestational age (and birth weights) in a Danish study of 11-13 year olds [22]. They found significant differences between children born at 33-36 weeks (and at 37-38 weeks) compared with children born at 39–40 weeks gestation. In addition to birth weight and gestational age, factors associated with cognitive outcomes include neuroimaging evidence of brain injury, neuromotor abnormalities on exam, male gender and some factors related to severity of neonatal illness or chronic lung disease [3, 12, 14, 22, 23]. Vlastos et al. [24] found that foetuses with absent foetal heart rate reactivity prior to delivery had higher risks of brain injury and low Bayley scores. In a ventilation trial of children born before 28 weeks gestation, factors associated with lower vocabularies at age 2 included severe neurosensory impairment, male gender, length of hospital stay, and weight at 12 months [23]. Sensory impairments Visual and auditory impairments are associated with preterm birth, and the most immature and sick preterm infants have the highest risks [3, 12, 14, 25-27]. In a study of caregiver-reported outcomes of preschool children, 6 % of children born before 28 weeks gestation had moderate to severe visual impairment and 4 % had moderate to severe hearing impairment, compared with 0.5 % for vision and hearing in children born at 28–32 weeks gestation and 0.3 % for each in full-term controls [28]. Severe visual impairment occurs in 1-3 % of children with birth weight below 1,000 g and 2–6 % in children with gestational age below 27 weeks [3, 12, 14]. Although as many as 6 % of children with gestational age below 26-27 weeks have severe to profound hearing impairment, many function well after early placement of cochlear implants [3, 12].

Adult Outcomes

Many young adults with birth weight below 1,000–1,500 g finish high school, find employment, marry and live independently [3, 29]. Nonetheless, some studies have found that more young adults with birth weight below 1,500 g leave school early and fewer go on to college and graduate from 4-year courses. Although more young adults born preterm demonstrate shy and withdrawn behaviours, they have fewer risktaking behaviours (e.g. use of alcohol or illegal drugs, delinquent behaviours, teenage pregnancies) than full-term controls. In an impressive Australian 31-year follow-up study of offspring of mothers in an antenatal steroid trial, there were no differences between 126 adults born preterm (mean 34 weeks gestational age) and 66 full-term controls in marital status, educational attainment, socioeconomic status, cognitive functioning, working memory, attention or psychiatric symptoms [30]. They found that preterm birth was associated with fewer symptoms of depression and greater satisfaction with respect to general health, pain, and social functioning. A Canadian study of young adults with birth weight below 1,000 g reports that despite more functional cognitive, mobility, and self-care limitations, they considered themselves to have a similar quality of life as their full-term peers [31].

Depression and Infants Born with Birth Weight 1,000–1,500 g

Over the past two decades a growing body of research has linked low birth weight to a variety of adult health outcomes, first and most compellingly for cardiovascular disease [32–34], but also for hypertension and type II diabetes [35, 36] and less clearly some mental disorders [37, 38]. These findings support what is variously termed the 'Barker hypothesis', 'foetal origins hypothesis' or 'developmental origins of adult health and disease hypothesis' [39]. It describes developmental plasticity in the foetus and young infant as a mechanism for permanently adjusting or 'programming' aspects of its physiology in response to the intrauterine and post-natal environment. This may provide adaptive fitness: exposure to poor nutrition in early life may programme glucose-insulin metabolism to maximise fitness by increased insulin resistance. The survival advantage gained in better handling early malnutrition may come at the cost of increased susceptibility to chronic disease in later life [40, 41]. Population-based cohort studies have presented mixed results and there is currently no consensus whether low birth weight is associated with depression in later life [42–45]. High-risk populations, including those exposed to famine, provide particularly strong evidence of an association. For example, Brown et al. [46] found that men and women exposed to the Dutch Hunger Winter of 1944–1945 during their second or third trimesters were at significantly increased risk of affective disorder [odds ratio (OR) 1.54, 95 % confidence interval (CI) 1.12-2.13] for exposure in the second trimester. However, the interpretation of this finding is complex and difficult to generalise to less extreme conditions. Famine no doubt represents a complex cluster of exposures, which in the perinatal period could include extreme material deprivation, maternal ill health and severe stress. There is a need to consolidate the epidemiological data from population studies under more normal circumstances. In an exhaustive meta-analysis of 18 studies by Wojcik et al. [47], demonstrated a weak association between low birth weight and later depression, including psychological distress, which became non-significant after correction for probable publication bias. The same was true if the analysis was restricted to studies examining depression only. Of the 13 studies that reported null findings in relation to their predetermined definitions of low birth weight and depression outcomes, the majority (n=11) emphasised 'positive' findings in their abstracts, such as subgroup analyses or secondary exposures. Only two of these null studies did not prioritise other positive findings [45, 48]. This suggests a preference for authors and journals to report positive findings while null findings are often downplayed, a phenomenon widely recognised [49]. They concluded that there was evidence of reporting and publication bias and that their meta-analysis should be interpreted as indicating no compelling association between low birth weight and adult depression or psychological distress in the populations studied. There was, however, heterogeneity in the studies reviewed, with a wide age range of participants, outcomes used and populations. There may therefore be subgroups where the association between low birth weight and depression is stronger than that seen overall. Some variables that might plausibly contribute to heterogeneity (e.g. family history of depression, smoking, maternal

physical health) were not routinely reported in most of the studies linking depression to VLBW. To conclude from this meta-analysis, data is still misleading as to link babies born between 1,000 and 1,500 g and depression.

Neurobiology behind the Insults in Utero/Being 1,000–1,500 g in Weight at Birth; the Influencers of the Adverse Outcomes

Serial head ultrasounds are a valuable bedside tool for following brain development in even the sickest preterm infants. Two studies have demonstrated that many preterm infants have a reduction in the size of the corpus callosum at term (compared with term controls); this is associated with lower gestational age at birth and with cerebral palsy and lower cognitive scores [50, 51]. Another study found that more infants with birth weight below 1,500 g who had sub-ependymal cysts noted on neonatal ultrasounds had delayed motor development at 2 years than controls with normal head ultrasounds [52]. In a thoughtful and convincing paper, Leviton et al. [53] suggest abandoning the current widely used but outmoded grading system for intraventricular haemorrhage and propose a simpler method that indicates presence or absence of subependymal haemorrhage, intraventricular haemorrhage, ventriculomegaly, and parenchymal abnormality. Although ultrasound evidence of intraparenchymal haemorrhage or white matter injury is a strong predictor, it is not diagnostic: 28-30 % of infants with these findings and birth weight below 1,000 g have no neurodevelopmental impairment at 18-22 months [54]. As many as 39 % of children with normal head ultrasounds had neurodevelopmental impairment. A number of MRI studies describe normal and abnormal brain development in preterm infants. A Swedish study of infants born before 27 weeks gestation in 2004–2005 found that 18 % had moderate to severe white matter abnormalities on head MRI at term [55].

White matter abnormalities were associated with severity of illness, use of inotropes, severe intraventricular haemorrhage and post-haemorrhagic hydrocephalus. Postnatal dexamethasone decreased total and regional cerebral tissue volumes on MRI in infants with birth weight below 1,000 g [56]. Although previous studies have shown that preterm infants with brain injury have decreased volume of both white and cortical gray matter, preterm infants with no brain injury have lower volume of white, but not gray, matter at 42 weeks postmenstrual age than full term controls [57]. A study of MRI at term in 167 infants born before 31 weeks gestation found that 21 % had moderate to severe white matter injury, and this was predictive of severe motor delay and cerebral palsy, even when controlling for ultrasound, perinatal, and neonatal risk factors [23].

A number of studies using diffusion tensor magnetic resonance imaging (DTI) to visualize white matter tracts in neonates with white matter injuries and in older children with cerebral palsy have published promising results [58–62]. One study of 24 infants with birth weight below 1,500 g who had DTI at 37 weeks postmenstrual age found a strong correlation between low fractional anisotropy values in the posterior limbs of the internal capsule and both diagnosis of cerebral palsy and severity of gait problems on outcome evaluations at 4 years [61].

Despite these promising reports, definitive predictive data that include sensitivity, specificity, and especially positive and negative predictive values for neurodevelopmental disabilities are needed before MRI or DTI become standard clinical practice, as discussed by Dammann and Leviton [63]. Neonatal illnesses, especially chronic lung disease/broncho-pulmonary dysplasia, are risk factors for neurodevelopmental impairment in preterm infants [3, 4]. A study of infants with birth weight below 1,000 g found that their logistic regression model with clinical variables better predicted neurodevelopmental impairment at 18-22 months than one with only neonatal ultrasound data; adding the ultrasound data to their model did not improve prediction [64].

Anderson and Doyle [64] demonstrated that lung disease is not only a risk factor for cerebral palsy and cognitive impairment, but also for language delay, visual-motor impairments, low average intelligence, academic difficulties, attention and behaviour problems, memory deficits and executive dysfunction. Necrotizing enterocolitis, especially with surgical intervention, increases a preterm infant's risk for cerebral palsy, and cognitive and visual impairment [65, 66]. Trials that evaluate neurodevelopmental outcome are providing important data regarding safety and efficacy of NICU treatment strategies. Although hypotension is a risk factor, we do not know what constitutes hypotension in extremely preterm infants and whether treatment with inotropes or hydrocortisone influences neurodevelopmental outcomes [67–70]. A number of studies reporting follow-up evaluations of preterm infants enrolled in NICU randomized controlled trials provide important information as to whether rates of neurodevelopmental disability (or death or major neurodevelopmental disability) are lower (e.g. caffeine [71]), unchanged (e.g. methods of ventilation, positive airway pressure or lucinactant surfactant [72–75]) or higher (e.g. some adverse effects with inhaled nitric oxide [76, 77]). The controversy over beneficial vs. adverse effects of inhaled nitric oxide in preterm infants with lung disease can only be answered by more follow-up data from trials, especially as they vary as to selection criteria and dose administered [76–82]. An intriguing paper looked at the effect of COX2 genotype in 5.5-year old preterm children, and found lower cognitive scores with the C allele (92.7 vs. 97.6, P<0.04) [83]. As the effect was independent of confounding variables, the authors speculate that polymorphisms of the COX2 gene may influence the preterm brain's response to inflammation and limit the efficacy of prophylactic indomethacin on neurocognitive outcomes. This paper reminds us that preterm birth and neurodevelopmental outcome of children born preterm are the result of multiple gene-environment interactions.

Putative Developmental Mechanisms Involved

Long term prenatal and perinatal cohort studies before the introduction of neonatal intensive care concluded that social factors and the quality of the home environment can compensate for perinatal and neonatal disadvantage [84]. Recent evidence

shows that favourable social and environmental factors are predictive of catch up in cognitive and behavioural development in larger LBW and preterm infants [85]. After approximately 2 years of age, IQ scores in low to moderate risk premature children are better explained by caretaking environment than their initial neonatal morbidity [86]. These findings indicate that in the vast majority of larger LBW children no persistent central nervous system insult is present. Larger preterm infants benefit from educational stimulation and home interventions [87, 88]. In contrast, although social factors are important for predicting psychological outcome in VLBW infants [89], biological factors have been found to be by far the best predictors of cognitive and behavioural outcome into school age in VPIs [17, 90–92]. Intensive intervention programmes that implemented improvements of educational stimulation and home environment have been disappointing; resulting in no long term benefits for VLBW children [93]. Taken together, this evidence suggests that VLBW children have been subject to various degrees of central nervous system insult that reduce the ability to take advantage of environmental offers. The pathogenic pathways are not fully understood but injuries to the white matter (subcortical ischaemic/infarctive brain lesions) with subsequent implications for late migration, brain organisation, and myelination are a likely cause [90, 91].

Conclusion

Although the majority of children born preterm do not develop major impairments, more preterm children than full term children develop cerebral palsy and/or cognitive impairments, and the risk increases with decreasing gestational age. The additional risk of the more subtle impairments of attention, executive function, language, visual-perceptual abilities, and fine motor function that influence the ability to function at school and at home has become apparent. While most studies focus on the most immature infants, there is growing recognition that infants born at 34-36 weeks gestation have higher mortality, morbidity, and cerebral palsy rates than fullterm infants. Mortality and neurodevelopmental outcomes of preterm infants are due to the causes of preterm birth, immature organ systems not being up to the task of fully sustaining extrauterine life, adverse effects of obstetric and neonatal treatments, and genetic factors that we know little about. Multicentric studies using uniform methodology often note significant differences in mortality and morbidity among centres. Just as large studies of mothers who deliver preterm provide insight into the pathophysiology of preterm birth, preterm neurodevelopmental outcome studies provide insight into how the preterm brain develops and recovers from injury. Knowledge of neurodevelopmental outcome is the key to developing better treatment strategies.

References

- Tyson J. Evidence based ethics and the care of premature infants. Future Child. 1995;5:197–213.
- McCormick M. The outcomes of very low birth weight infants: are we asking the right questions? Pediatrics. 1997;99:869–76.
- Behrman RE, Stith Butler A, editors. Institute of Medicine Committee on Understanding Premature Birth and Assuring Healthy Outcomes Board on Health Sciences Outcomes: preterm birth: causes, consequences, and prevention. Washington, DC: The National Academies Press; 2007.
- Allen MC. Preterm outcomes research: a critical component of neonatal intensive care. Ment Retard Dev Disabil Res Rev. 2002;8:221–33.
- Aylward GP. Cognitive and neuropsychological outcomes: more than IQ scores. Ment Retard Dev Disabil Res Rev. 2002;8:234–40.
- Bracewell M, Marlow N. Patterns of motor disability in very preterm children. Ment Retard Dev Disabil Res Rev. 2002;8:241–8.
- Fawke J. Neurological outcomes following preterm birth. Semin Fetal Neonatal Med. 2007;12:374–82.
- Gibson AT. Outcome following preterm birth. Best Pract Res Clin Obstet Gynaecol. 2007;21:869–82.
- Johnson S. Cognitive and behavioural outcomes following very preterm birth. Semin Fetal Neonatal Med. 2007;12:363–73.
- Voss W, Neubauer AP, Wachtendorf M, et al. Neurodevelopmental outcome in extremely low birth weight infants: what is the minimum age for reliable developmental prognosis? Acta Paediatr. 2007;96: 342–7.
- Tommiska V, Heinonen K, Lehtonen L, et al. No improvement in outcome of nationwide extremely low birth weight infant populations between 1996– 1997 and 1999–2000. Pediatrics. 2007;119:29–36.
- 12. Sommer C, Urlesberger B, Maurer-Fellbaum U, et al. Neurodevelopmental outcome at 2 years in 23

to 26 weeks old gestation infants. Klin Padiatr. 2007;219:23–9.

- Robertson CM, Watt MJ, Yasui Y. Changes in the prevalence of cerebral palsy for children born very prematurely within a population-based program over 30 years. J Am Med Assoc. 2007;297:2733–40.
- Marret S, Ancel PY, Marpeau L, et al. Neonatal and 5-year outcomes after birth at 30–34 weeks of gestation. Obstet Gynecol. 2007;110:72–80.
- Wilson-Costello D, Friedman H, Minich N, et al. Improved neurodevelopmental outcomes for extremely low birth weight infants in 2000–2002. Pediatrics. 2007;119:37–45.
- Dolk H, Parkes J, Hill N. Trends in the prevalence of cerebral palsy in Northern Ireland, 1981–1997. Dev Med Child Neurol. 2006;48:406–12.
- Marlow N, Hennessy EM, Bracewell MA, Wolke D. Motor and executive function at 6 years of age after extremely preterm birth. Pediatrics. 2007;120:793–804.
- Himmelmann K, Hagberg G, Beckung E, et al. The changing panorama of cerebral palsy in Sweden. IX. Prevalence and origin in the birth-year period 1995–1998. Acta Paediatr. 2005;94:287–94.
- Engle WA, Tomashek KM, Wallman C, Committee on Foetus and New-born. 'Late-preterm' infants: a population at risk. Pediatrics. 2007;120:1390–401.
- 20. Davis NM, Ford GW, Anderson PJ, Doyle LW. Developmental coordination disorder at 8 years of age in a regional cohort of extremely-low-birth weight or very preterm infants. Dev Med Child Neurol. 2007;49:325–30.
- Schmidhauser J, Caflisch J, Rousson V, et al. Impaired motor performance and movement quality in verylow-birth weight children at 6 years of age. Dev Med Child Neurol. 2006;48:718–22.
- 22. Kirkegaard I, Obel C, Hedegaard M, Henriksen TB. Gestational age and birth weight in relation to school performance of 10-year-old children: a follow-up study of children born after 32 completed weeks. Pediatrics. 2006;118:1600–6.
- Woodward LJ, Anderson PJ, Austin NC, et al. Neonatal MRI to predict neurodevelopmental outcomes in preterm infants. N Engl J Med. 2006;355:685–94.
- 24. Vlastos EJ, Tomlinson TM, Bidirici I, et al. Fetal heart rate accelerations and the risk of cerebral lesions and poor neurodevelopmental outcome in very low birth weight neonates. Am J Perinatol. 2007;24:83–8.
- 25. Marston L, Peacock JL, Calvert SA, et al. Factors affecting vocabulary acquisition at age 2 in children born between 23 and 28 weeks' gestation. Dev Med Child Neurol. 2007;49:591–6.
- 26. Section on Ophthalmology American Academy of Pediatrics, American Academy of Ophthalmology, American Association for Pediatric Ophthalmology and Strabismus. Screening examination of premature infants for retinopathy of prematurity. Pediatrics. 2006;117:572–6.
- O'Connor AR, Wilson CM, Fielder AR. Ophthalmological problems associated with preterm birth. Eye. 2007;21:1254–60.

- Schiariti V, Hoube JS, Lisonkova S, et al. Caregiverreported health outcomes of preschool children born at 28 to 32 weeks' gestation. J Dev Behav Pediatr. 2007;28:9–15.
- Saigal S, Stoskopf B, Streiner D, et al. Transition of extremely low-birth-weight infants from adolescence to young adulthood: comparison with normal birth weight controls. J Am Med Assoc. 2006;295:667–75.
- Dalziel SR, Lim VK, Lambert A, et al. Psychological functioning and health related quality of life in adulthood after preterm birth. Dev Med Child Neurol. 2007;49:597–602.
- Saigal S, Stoskopf B, Pinelli J, et al. Self-perceived health-related quality of life of former extremely low birth weight infants at young adulthood. Pediatrics. 2006;118:1140–8.
- Barker DJ, Winter PD, Osmond C, Margetts B, Simmonds SJ. Weight in infancy and death from ischaemic heart disease. Lancet. 1989;2:577–80.
- Barker DJ. The developmental origins of adult disease. J Am Coll Nutr. 2004;23:5885–95.
- 34. Huxley R, Owen CG, Whincup PH, Cook DG, Rich-Edwards J, Smith GD, Collins R. Is birth weight a risk factor for ischemic heart disease in later life? Am J Clin Nutr. 2007;85:1244–50.
- Barker DJ, Bagby SP, Hanson MA. Mechanisms of disease: in utero programming in the pathogenesis of hypertension. Nat Clin Pract Nephrol. 2006;2:700–7.
- Whincup PH, Kaye SJ, Owen CG, Huxley R, Cook DG, Anazawa S. Birth weight and risk of type 2 diabetes: a systematic review. JAMA. 2008;300:2886–97.
- 37. Jones PB, Rantakallio P, Hartikainen AL, Isohanni M, Sipila P. Schizophrenia as a long-term outcome of pregnancy, delivery, and perinatal complications: a 28-year follow-up of the 1966 north Finland general population birth cohort. Am J Psychiatry. 1998;155:355–64.
- Breslau N, Chilcoat HD. Psychiatric sequelae of low birth weight at 11 years of age. Biol Psychiatry. 2000; 47:1005–11.
- Barker DJ. The origins of the developmental origins theory. J Intern Med. 2007;261:412–7.
- Barker DJ. Fatal origins of coronary heart disease. Br Med J. 1995;311:171–4.
- 41. Heijmans BT, Tobi EW, Stein AD, Putter H, Blauw GJ, Susser ES, Slagboom PE, Lumey LH. Persistent epigenetic differences associated with prenatal exposure to famine in humans. Proc Natl Acad Sci U S A. 2008;105:17046–9.
- Thompson C, Syddall H, Rodin I, Osmond C, Barker DJ. Birth weight and the risk of depressive disorder in late life. Br J Psychiatry. 2001;179:450–5.
- Gale CR, Martyn CN. Birth weight and later risk of depression in a national birth cohort. Br J Psychiatry. 2004;184:28–33.
- 44. Osler M, Nordentoft M, Andersen AM. Birth dimensions and risk of depression in adulthood: cohort study of Danish men born in 1953. Br J Psychiatry. 2005;186:400–3.
- 45. Inskip HM, Dunn N, Godfrey KM, Cooper C, Kendrick T. Is birth weight associated with risk of depressive symptoms in young women? Evidence

from the Southampton Women's Survey. Am J Epidemiol. 2008;167:164–8.

- 46. Brown AS, van Os J, Driessens C, Hoek HW, Susser ES. Further evidence of relation between prenatal famine and major affective disorder. Am J Psychiatry. 2000;157:190–5.
- 47. Wojcik W, Lee W, Colman I, Hardy R, Hotopf M. Foetal origins of depression? A systematic review and meta-analysis of low birth weight and later depression. Psychol Med. 2013;43:1–12.
- Vasiliadis HM, Gilman SE, Buka SL. Fetal growth restriction and the development of major depression. Acta Psychiatr Scand. 2008;117:306–12.
- Fanelli D. Do pressures to publish increase scientists' bias? An empirical support from US States Data. PLoS One. 2010;5:e10271.
- Anderson NG, Laurent I, Woodward LJ, Inder TE. Detection of impaired growth of the corpus callosum in premature infants. Pediatrics. 2006;118:951–60.
- Narberhaus A, Segarra D, Caldu X, et al. Gestational age at preterm birth in relation to corpus callosum and general cognitive outcome in adolescents. J Child Neurol. 2007;22:761–5.
- Chuang YC, Lee C, Chiu NC, et al. Neurodevelopment in very low birth weight premature infants with postnatal subependymal cysts. J Child Neurol. 2007;22:402–5.
- Leviton A, Kuban K, Paneth N. Intraventricular haemorrhage grading scheme: time to abandon? Acta Paediatr. 2007;96:1254–6.
- 54. Broitman E, Ambalavanan N, Higgins RD, et al. Clinical data predict neurodevelopmental outcome better than head ultrasound in extremely low birth weight infants. J Pediatr. 2007;151:500–5.
- 55. Horsch S, Hallberg B, Leifsdottir K, et al. Brain abnormalities in extremely low gestational age infants: a Swedish population based MRI study. Acta Paediatr. 2007;96:979–84.
- Parikh NA, Lasky RE, Kennedy KA, et al. Postnatal dexamethasone therapy and cerebral tissue volumes in extremely low birth weight infants. Pediatrics. 2007;119:265–72.
- Mewes AU, Huppi PS, Als H, et al. Regional brain development in serial magnetic resonance imaging of low-risk preterm infants. Pediatrics. 2006;118:23–33.
- Anjari M, Srinivasan L, Allsop JM, et al. Diffusion tensor imaging with tract basedspatial statistics reveals local white matter abnormalities in preterm infants. Neuroimage. 2007;35:1021–7.
- Counsell SJ, Dyet LE, Larkman DJ, et al. Thalamocortical connectivity in children born preterm mapped using probabilistic magnetic resonance tractography. Neuroimage. 2007;34:896–904.
- 60. Nagae LM, Hoon Jr AH, Stashinko E, et al. Diffusion tensor imaging in children with periventricular leukomalacia: variability of injuries to white matter tracts. Am J Neuroradiol. 2007;28:1213–22.
- Rose J, Mirmiran M, Butler EE, et al. Neonatal microstructural development of the internal capsule on diffusion tensor imaging correlates with severity of gait and motor deficits. Dev Med Child Neurol. 2007;49:745–50.
- 62. Yoo SS, Park HJ, Soul JS, et al. In vivo visualization of white matter fiber tracts of pretermand term-infant brains with diffusion tensor magnetic resonance imaging. Invest Radiol. 2005; 40:110–5.
- Dammann O, Leviton A. Neuroimaging and the prediction of outcomes in preterm infants. N Engl J Med. 2006;355:727–9.
- Anderson PJ, Doyle LW. Neurodevelopmental outcome of bronchopulmonary dysplasia. Semin Perinatol. 2006;30:227–32.
- 65. Rees CM, Pierro A, Eaton S. Neurodevelopmental outcomes of neonates with medically and surgically treated necrotizing enterocolitis. Arch Dis Child Fetal Neonatal Ed. 2007;92:F193–8.
- 66. Schulzke SM, Deshpande GC, Patole SK. Neurodevelopmental outcomes of very low-birthweight infants with necrotizing enterocolitis: a systematic review of observational studies. Arch Pediatr Adolesc Med. 2007;161:583–90.
- 67. Barrington K. Time for pressure tactics. Pediatrics. 2007;119:396–7.
- Fanaroff JM, Wilson-Costello DE, Newman NS, et al. Treated hypotension is associated with neonatal morbidity and hearing loss in extremely low birth weight infants. Pediatrics. 2006;117:1131–5.
- Rademaker KJ, Uiterwaal CS, Groenendaal F, et al. Neonatal hydrocortisone treatment: neurodevelopmental outcome and MRI at school age in preterm born children. J Pediatr. 2007;150:351–7.
- Subhedar NV, Duffy K, Ibrahim H. Corticosteroids for treating hypotension in preterm infants. Cochrane Database Syst Rev. 2007;24(1):CD003662.
- Schmidt B, Roberts RS, Davis P, et al. Long-term effects of caffeine therapy for apnea of prematurity. N Engl J Med. 2007;357:1893–902.
- Moya F, Sinha S, Gadzinowski J, et al. One-year follow-up of very preterm infants who received lucinactant for prevention of respiratory distress syndrome: results from 2 multicenter randomized, controlled trials. Pediatrics. 2007;119:e1361–70.
- 73. Stack JA, Jalaludin B. Developmental outcomes at the age of two years for very premature babies managed with nasal prong continuous positive airway pressure. J Paediatr Child Health. 2007;43:480–5.
- 74. Truffert P, Paris-Llado J, Escande B, et al. Neuromotor outcome at 2 years of very preterm infants who were treated with high-frequency oscillatory ventilation or conventional ventilation for neonatal respiratory distress syndrome. Pediatrics. 2007;119:e860–5.
- Wintermark P, Tolsa JF, Van MG, et al. Long-term outcome of preterm infants treated with nasal continuous positive airway pressure. Eur J Pediatr. 2007;166:473–83.
- 76. Hintz SR, Van Meurs KP, Perritt R, et al. Neurodevelopmental outcomes of premature infants with severe respiratory failure enrolled in a random-

ized controlled trial of inhaled nitric oxide. J Pediatr. 2007;151:16–22.

- Barrington KJ, Finer NN. Inhaled nitric oxide for respiratory failure in preterm infants. Cochrane Database Syst Rev. 2007;18(3):CD000509.
- Ballard RA, Truog WE, Cnaan A, et al. Inhaled nitric oxide in preterm infants undergoing mechanical ventilation. N Engl J Med. 2006;355:343–53.
- Barrington KJ, Finer NN. Inhaled nitric oxide for preterm infants: a systematic review. Pediatrics. 2007;120:1088–99.
- Kinsella JP, Cutter GR, Walsh WF, et al. Early inhaled nitric oxide therapy in premature newborns with respiratory failure. N Engl J Med. 2006;355:354–64.
- Steinhorn RH, Porta NF. Use of inhaled nitric oxide in the preterm infant. Curr Opin Pediatr. 2007;19: 137–41.
- Tanaka Y, Hayashi T, Kitajima H, et al. Inhaled nitric oxide therapy decreases the risk of cerebral palsy in preterm infants with persistent pulmonary hypertension of the newborn. Pediatrics. 2007;119:1159–64.
- 83. Harding DR, Humphries SE, Whitelaw A, et al. Cognitive outcome and cyclooxygenase-2 gene (765 G/C) variation in the preterm infant. Arch Dis Child Fetal Neonatal Ed. 2007;92:F108–12.
- Werner E, Smith R. Vulnerable but invincible: Kauai's children come of age. New York: McGraw Hill; 1982.
- Aylward G. The relationship between environmental risk and developmental outcome. Dev Behav Pediatr. 1992;13:222–9.
- Wolke D. The preterm responses to the environment—long term effects? In: Cockburn F, editor. Advances in perinatal medicine. Carnforth: Parthenon Publishing; 1997. p. 305–14.
- Wolke D. Supporting the development of low birth weight infants [annotation]. J Child Psychol Psychiatry. 1991;32:723–41.
- Brooks-Gunn J, McCarton C, Casey P, et al. Early intervention in low birth weight premature infants. JAMA. 1994;272:1257–62.
- Breslau N. Psychiatric sequelae of low birth weight. Epidemiol Rev. 1995;17:96–106.
- 90. Whitaker AH, Van Rossem R, Feldman JF, et al. Psychiatric outcomes in low-birth-weight children at age 6 years: relation to neonatal cranial ultrasound abnormalities. Arch Gen Psychiatry. 1997;54: 847–56.
- Whitaker A, Feldman J, Van Rossem R, et al. Neonatal cranial ultrasound abnormalities in low birth weight infants: relation to cognitive outcomes at six years of age. Pediatrics. 1996;98:719–29.
- Roth SC, Baudin J, Pezzani-Goldsmith M, Towndsend J, Reynolds EOR, Stewart AL. Relation between neurodevelopmental status of very preterm infants at one and eight years. Dev Med Child Neurol. 1994;36:1049–62.
- Baumeister A, Bacharach V. A critical analysis of the infant health and development programme. Intelligence. 1996;23:79–104.

Part XI

Fetal Nephrological Development: Up to Second Trimester

Development of Renal Tissues in First Twenty Weeks

31

Rajendra Pandey

We have on average one million nephrons and they're fully formed before we're born. Premature birth and low birth weight coincides with fewer nephrons. No new nephrons are formed after birth and we lose up to 6,000 of them each year, making old age and low birth weight a risk factor for kidney disease.

The kidneys are the most intelligent organ ever evaluated in nature. They not only filter out all that is unwanted, but reclaim almost every wanted thing that escapes filtration barrier. They play a very important role in keeping skeleton stronger by participating in vitamin D synthesis. The Kidney plays a major role as an endocrine organ by secreting Erythropoietin responsible for effective erythropoiesis. The kidney by regulating RAS (Renin Angiotensin System) maintains blood pressure to make blood reach every tissue in human body.

The Urinary system serves the purpose of maintaining the electrolyte and the water balance of the body fluid that bathes the tissue. The three nephric system develop from the intermediate mesoderm located on either side of the dorsal body wall. The first of this called **Pronephros is a**

Department of Nephrology, Institute of Post Graduate Medical Education & Research, 244, Acharya Jagadish Chandra Bose Road, Kolkata 700 020, India e-mail: rajensankrityan@gmail.com small group of nephrotomes developing in the cervical region which are short lived and remain nonfunctional. They regress by fourth week and are succeeded by the **Mesonephros** developing in thoracic and lumbar region. This is comprised of simple nephrons complete in all respect, functional and draining into mesonephric (Wolffian) duct which grows caudally to open in to the primitive urogenital sinus. A pair of ureteric buds appears by fifth week, sprouting from the distal mesonephric duct and induces the overlying sacral intermediate mesoderm to develop in to **Metanephros** (definitive kidney) (*Human Embryology*, 3rd Ed, Pg. 265).

During gastrulation the mesoderm is deposited on either side of the midline and thereafter differentiates into three subdivisions: the paraxial, intermediate and lateral plate mesoderm. The intermediate mesoderm gives rise to the nephric structure of the embryo, portions of the gonads, and to the male genital duct system. Three sets of nephric structures develop from the intermediate mesoderm in cradiocaudal sequence. These are known as **cervical nephrotomes**, the **mesonephros**, and the **metanephros** (*Human Embryology*, 3rd Ed., Pg. 268).

The Cervical Nephrotomes

By early fourth week, five to seven paired cervical segments of intermediate mesoderm give rise to small but hollow ball of epithelium known as

© Springer International Publishing Switzerland 2016 N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development: First and Second Trimesters*, DOI 10.1007/978-3-319-14874-8_31

R. Pandey, MD, DM

nephric vesicle or **nephrotome** or **pronephros**. Unlike some lower vertebrate these do not develop into the primitive functional excretory structure of a true pronephros, rather they cease developing and disappear by day 24 or 25 (*Human Embryology*, 3rd Ed., Pg. 268).

The Mesonephros

The next series of development takes place in thoracic and lumbar region (up to third lumbar level). Here by fourth week **nephric tubules** begin to develop within a pair of elongated swelling of intermediate mesoderm on either side of vertebral column. These swellings are known as **mesoneproi** or **mesonephric ridge**. About 40 mesonephric tubule develop in craniocaudal succession. While new ones are formed the older one regresses so that at one time only 30 pairs are there. By fifth week the cranial regions of mesonephros goes in to massive regression leaving only about 20 pairs of tubules occupying the first three lumbar levels (*Human Embryology*, 3rd Ed., Pg. 269).

These mesenephric tubules differentiate into excretory units that resemble an abbreviated version of the adult nephron. The **Bowman's capsule** a cup shaped sac) formed out of medial end of these tubules wraps around a bunch of capillaries called **Glomerulus** to form **renal corpuscle**.

The appearance of **mesonephric ducts** at about 24 days, a pair of solid longitudinal rods that condense in the intermediate mesoderm of the thoracic region dorsolateral to the developing mesonephric tubules. These grow caudally through the proliferation and migration of the cells at their caudal tip. With their growth into the lumbar region they diverge from the intermediate mesoderm and grow towards ventrolateral wall of cloaca to fuse with them eventually. During fusion with cloaca they begin to cavitate at their distal end to form a lumen which progresses cranially, transforming the rods into the **mesonephric duct** (*Human Embryology*, 3rd Ed, Pg. 269).

Each mesonephric tubules at its lateral tip fuses with the mesonephric duct to form a continuous passage from the excretory unit to the cloaca. These units function between 6 and 10 weeks and gradually stop functioning and then eventually regress.

The Metanephros

This is the final stage of development of definitive kidney. These start as early as fifth week. A pair of new structure known as ureteric bud arises out of intermediate mesoderm at the sacral region from the distal portion of the mesonephric duct (day 28). After about 4 days these ureteric bud penetrates a portion of sacral intermediate mesoderm known as metanephric blastema and starts bifurcating. As the ureteric bud branches, each new growing tip (ampulla) acquires a cap like aggregate of tissue from metanephric blastema, giving the metanephros a lobulated appearance. By halfway sixth week the developing metanephros consists of two lobes with a sulcus in between, but acquires 14-16 lobes by the end of 16th week. The lobular structure eventually is obscured as the sulci get filled (Human Embryology, 3rd Ed, Pg. 271).

The metanephric blastema and the ureteric bud exert reciprocal inductive effects. Interaction between these tissues is a classic model of induction where several hours of direct contact between ureteric bud ampulla and metanephric blastema is needed to induce differentiation in blastema tissue. Conversely, metanephric blastema through inductive signals regulates orderly branching and growth of the bifurcating tips of the ureteric buds. The collecting duct system is result of sequential bifurcation of the ureteric bud. The ureteric bud undergoes an exact sequence of bifurcations, and the expanded major and minor calyces arise through phases of intussusceptions in which occurs coalescence of earlier formed branches. Urine produced by the mature nephron flows through collecting tubules, minor calyces, major calyces, the renal pelvis and finally the ureter. In the sixth week uretric bud by four time bifurcation forms 16 branches which then coalesce to form major calyces extending from the renal pelvis (formed by splaying of ampulla of the ureteric bud. In the seventh week next four generation of branches also coalesce to form **minor calyces**. By end of 32 weeks 11 additional branching takes place to form one to three million branches which will become **collecting tubules (collecting ducts)** in future (*Human Embryology*, 3rd Ed, Pg. 272–3).

The ampulla of a collecting duct is surrounded by a cap of metanephric blastema which gives rise to origin of a Nephron as a vesicle. As the vesicle elongates to form a tubular structure, a capillary glomerulus forms near one end of it. The tubule epithelium abutting the glomeruli thins and invaginates to form a Bowman's capsule. Bowman's capsule in union with glomerulus is named Renal corpuscle. While renal corpuscle is shaping the lengthening nephric tubules differentiates to form proximal convulated tubule, the descending and ascending limbs of loop of Henle, and the distal convulated tubules. This definitive nephron along with renal corpuscle is also called a metanephric excretory unit. By tenth week the distal convulated tubules unite with the collecting duct to make metanephroi functional. Now the plasma flowing through the glomeruli forms a dilute filtrate that travels through the different segments of the tubule to form concentrated urine (Human Embryology, 3rd Ed, Pg. 273).

The Kidney architecture gets its definite shape between 5th and 15th week. The outer cortex comprises of nephron whereas the inner medulla contains collecting ducts and loop of Henle. Each minor calyx drains a tree of collecting ducts within a **renal pyramid**. These renal pyramids are interspersed with nephron containing cortical tissue named **renal columns** of **Bertin. Renal Pyramids** converge to form the **renal papilla**. The neurons of the kidney, responsible for blood flow regulation and secretion, arise from the neural crest cells that invade the metanephron early in their development (*Human Embryology*, 3rd Ed, Pg. 274).

Kidney Diseases

Kidney diseases usually involve damage to the nephrons and can be acute or chronic. In acute kidney disease there is a sudden drop in kidney function. It is usually caused by loss of large amounts of blood or an accident and is often short lived, though it can occasionally lead to lasting kidney damage. Chronic kidney disease (CKD) is defined as loss of a third or more of kidney function for at least 3 months. In CKD kidney function worsens over a number of years and the problem often goes undetected for many years because its effects are relatively mild. Some of the symptoms associated with CKD are: headache, fatigue, high blood pressure, itching, fluid retention, shortness of breath.

However, kidney disease can lead to kidney failure (less than 10 % kidney function). Once this happens, patients need dialysis or a kidney transplant to stay alive. The risk of developing CKD is increased by old age, diabetes, high blood pressure, obesity and smoking. At least 8 % of the European population (40 million individuals) currently has a degree of CKD, putting them at risk of developing kidney failure. This figure is increasing every year and there are not enough organ donors to provide transplants for so many patients. This makes the development of new therapeutic options for treating CKD increasingly important.

Kidney Stem Cells

Scientists are still debating whether kidney stem cells exist in the adult body and if so, where they are found and how they can be identified. Cells found in a number of places within the nephrons have been proposed as candidates for kidney stem cells. The most convincing evidence for the existence of such stem cells is the discovery of a group of cells at the urinary pole of the Bowman's capsule of the nephron (marked in blue in the diagram above). These cells have some of the key features of stem cells and researchers have shown them to be responsible for production of podocytes - specialised cells involved in the filtration work of the nephron and that need to be replaced continuously throughout our lifetime. Studies also suggest that these same proposed stem cells might be able to generate a second type of specialised cell found in the nephron lining, called proximal tubular epithelial cells. Other suggested locations for kidney

stem cells include certain places in the tubules (marked green in the diagram). As well as kidney stem cells, cells with some of the characteristics of mesenchymal stem cells have very recently been isolated from the kidney.

Kidney Disease and Mesenchymal Stem Cells

A number of different types of cells from the bone marrow have been tested in animals and in clinical studies for potential use in kidney disease. Amongst all the cells under investigation, mesenchymal stem cells (MSCs) have shown the most promising results to date. Studies suggest that MSCs may be able to enhance the intrinsic ability of the kidney to repair itself.

MSCs of the bone marrow can differentiate to produce specialised bone, fat and cartilage cells. Researchers investigating the therapeutic effects of these MSCs within the kidney have suggested these cells may release proteins that can help kidney cells to grow, inhibit cell death and that could encourage the kidney's own stem cells to repair kidney damage. Further research is needed to establish whether these ideas are correct and if so, how this could lead to a treatment for patients.

Cells with some of the features of MSCs appear to exist in many other organs as well as the bone marrow, though there is much controversy amongst scientists about the exact nature of such cells and their roles in the body. Recently, cells with MSClike features have been isolated from the kidney. These so-called kidney MSCs are distinctly different from bone marrow MSCs and heart MSCs. More research is needed to identify their precise role in normal kidney maintenance and to investigate their potential to enhance the kidney's ability to regenerate or repair itself after damage.

How Else Could Stem Cells Help Tackle Kidney Disease?

Another type of stem cell that scientists are using in kidney research is the induced pluripotent stem cell (iPSC). Induced pluripotent stem cells are made by reprogramming adult, specialised cells of the body to act like embryonic stem cells. They have the ability to develop into any cell or tissue in the body. Recently researchers have been able to use iPSCs to produce kidney cells in a very early stage of development. These very early kidney cells resemble cells found in the embryo that will turn into the cells that eventually make up the kidney in foetal development. These cells could have the potential to make the glomerulus and tubules, the building blocks of the nephron. However, a lot of research needs to be done before such cells can be used in patients to treat CKD.

An alternative approach to organ replacement is also under investigation and may help kidney disease patients in the future: The use of organ scaffolds to produce whole, transplantable organs. Organ scaffolds are organs from which all the cells have been removed. What remains is the extracellular matrix – the part of the organ that supports its shape. This matrix can be seeded with a patient's own cells, which can be carefully nurtured to grow and multiply to re-cover the scaffold. By using the patient's own cells, the complications of immune rejection that can occur with organ transplantations are drastically reduced. The challenge with this approach is identifying and obtaining the right types of cells to seed the scaffold, especially in organs with complex structures made up of many different cells. IPSCs or the recently identified kidney MSCs could be useful candidate cells for seeding kidney organ scaffolds. Very recently, experiments in rats have shown the feasibility of this approach.

Can Stem Cells Be Used to Treat Kidney Disease Today?

Stem cell treatments for kidney disease have not yet been developed. The kidney is a very complex organ consisting of a large number of different types of cells. To make a new kidney in the lab, all these different cells would need to be produced in a different way and mixed together in the hope that they would eventually recreate a functional kidney. What's more, kidney disease comes in many flavours with different cells affected and so treatments aiming to replace damaged cells within a patient's kidney would need to supply different types of cells for different patients. Research on organ or cell replacement therapies is ongoing, but this is likely to be a long-term goal.

In the meantime, stem cells may benefit patients in other ways. For example, stem cells can be used to help progress our understanding of the disease through studies on the development and behaviour of kidney cells grown in large numbers in the laboratory. Stem cell research may also enable us to utilise the body's own repair mechanisms to find treatments for kidney disease. In acute kidney disease, the body can often repair kidney damage itself, but it is unable to do this well enough to tackle the progressive damage that occurs in chronic kidney disease. The recent identification of mesenchymal-stem-cell-like cells in the kidney may open up new possibilities for enhancing the body's own capacity for regeneration and repair of damaged kidneys. Investigating these possibilities by studying how these newly discovered cells work is currently an important area of research. Researchers also continue to explore new ideas using emerging technologies in stem cell research, such as reprogramming cells to change their behaviour.

 Congenital anomalies of the kidney and urinary tract (CAKUT) constitute approximately 20–30 % of all anomalies identified in the prenatal period [1].

Defects can be bilateral or unilateral, and different defects often coexist in an individual child. The antenatal screening and postnatal evaluation of infants with CAKUT are discussed in greater detail separately. (See "Evaluation of congenital anomalies of the kidney and urinary tract (CAKUT)" and "Congenital ureteropelvic junction obstruction" and "Primary megaureter in infants and children" and "Ectopic ureter" and "Renal ectopic and fusion anomalies" and "Autosomal recessive polycystic kidney disease in children". Congenital abnormalities of the kidney and urinary tract are frequently observed in children and represent a significant cause of morbidity and mortality. These conditions are phenotypically variable, often affecting several segments of the urinary tract simultaneously, making clinical classification and diagnosis difficult. Renal agenesis/hypoplasia and dysplasia account for a significant portion of these anomalies, and a genetic contribution to its cause is being increasingly recognized. Nevertheless, overlap between diseases and challenges in clinical diagnosis complicate studies attempting to discover new genes underlying this anomaly. Most of the insights in kidney development derive from studies in mouse models or from rare, syndromic forms of human developmental disorders of the kidney and urinary tract. The genes implicated have been shown to regulate the reciprocal induction between the ureteric bud and the metanephric mesenchyme. Strategies to find genes causing renal agenesis/ hypoplasia and dysplasia vary depending on the characteristics of the study population available. Congenital abnormalities of the kidney and urinary tract are frequently observed in children and represent a significant cause of morbidity and mortality. These conditions are phenotypically variable, often affecting several segments of the urinary tract simultaneously, making clinical classification and diagnosis difficult. Renal agenesis/ hypoplasia and dysplasia account for a significant portion of these anomalies, and a genetic contribution to its cause is being increasingly recognized. Nevertheless, overlap between diseases and challenges in clinical diagnosis complicate studies attempting to discover new genes underlying this anomaly. Most of the insights in kidney development derive from studies in mouse models or from rare, syndromic forms of human developmental disorders of the kidney and urinary tract. The genes implicated have been shown to regulate the reciprocal induction between the ureteric bud and the metanephric mesenchyme. Strategies to find genes causing renal agenesis/ hypoplasia and dysplasia vary depending on the characteristics of the study population available [2]. Abnormal genes cause polycystic kidney disease, and the genetic defects mean the disease runs in families.

There are two types of polycystic kidney disease, caused by different genetic flaws:

Autosomal dominant polycystic kidney disease (ADPKD). Signs and symptoms of ADPKD often develop between the ages of 30 and 40. In the past, this type was called adult polycystic kidney disease, but children can develop the disorder.

Only one parent needs to have the disease in order for it to pass along to the children. If one parent has ADPKD, each child has a 50 % chance of getting the disease. This form accounts for about 90 % of cases of polycystic kidney disease.

Autosomal recessive polycystic kidney disease (ARPKD). This type is far less common than is ADPKD. The signs and symptoms often appear shortly after birth. Sometimes, symptoms don't appear until later in childhood or during adolescence.

Both parents must have abnormal genes to pass on this form of the disease. If both parents carry a gene for this disorder, each child has a 25 % chance of getting the disease. Researchers have identified two genes associated with ADPKD and one associated with ARPKD.

In some cases, a person with ADPKD has no known family history of the disease. However, it's possible that someone in the affected person's family actually did have the disease, but didn't show signs or symptoms before dying of other causes.

In a smaller percentage of cases where no family history is present, ADPKD results from a spontaneous gene mutation.

Acknowledgement

- 1. Gray's Anatomy 35th edition
- 2. Human Embryology William J. Larsen, 3rd Edition
- 3. Medical Embryology T.W. Sadler, 11th Edition

References

- Queisser-Luft A, Stolz G, Wiesel A, et al. Malformations in newborn: results based on 30,940 infants and fetuses from the Mainz congenital birth defect monitoring system (1990–1998). Arch Gynecol Obstet. 2002;266:163.
- Sanna-Cherchi S, Caridi G, Weng PL, Scolari F, Perfumo F, Gharavi AG, Ghiggeri GM. Genetic approaches to human renal agenesis/hypoplasia and dysplasia. Pediatr Nephrol. 2007;22(10):1675.

Part XII

Fetal Haematological Development

Development of the Haemopoiesis System before the Second Trimester of Pregnancy

Mainuddin Naskar and Niranjan Bhattacharya

Introduction

During embryogenesis the extra-embryonic yolk sac, fetal liver, preterm bone marrow and developing placenta are spatially and temporarily distinct sites where haemopoiesis [1] occurs. Soon after the implantation of blastocyst, erythropoiesis is established and primitive erythroid cells appear in the Yolk Sac Blood Islands by day 18th of gestation [2].

The origin of haemopoietic cells is closely related with gastrulation and formation of mesoderm. Inducers of mesoderm are Transforming Growth Factor b (TGF-b), Fibroblast Growth Factor (FGF), Bone Morphogenic Protein-4 (BMP-4). These factors play an important role in haemopoiesis. Yolk sac erythroblast develops in close association with first embryonic blood vessel-which suggests that blood cells and endo-

M. Naskar, MBBS

e-mail: sanjuktaniranjan@gmail.com

thelial cells are of common haemangioblast precursor's source.

In Yolk sac haemopoiesis the yolk sac erythroblasts have several characteristic features thereby distinguishing them from their later counterparts. Primitive nucleated erythroblast differentiates within vascular network rather than extravascular space and in circulation.

Primitive erythroblasts are characterized by:

- 1. More rapid maturation.
- 2. Increased sensitivity to erythropoietin.
- 3. Shortened life span compared to fetal and adult erythroblast.

Yolk sac erythroblasts are extremely large red cells with mean cell volume (MCV) of 450 fl/cell. The erythroid progenitors burst forming-erythroid (BRU-E), and Colony forming unit-erythroid (CFU-E) appear in yolk sac at fourth week of gestation. Primitive erythroblast and erythroid progenitors then enter the embryo proper through the circulation. After 7 weeks, haemopoietic progenitors are no longer detected in the yolk sac. The primitive erythroblasts derived from yolk sac continue to circulate up to 12 weeks of gestation [3].

From 9th to 24th weeks of gestation the liver serves as a primary source of red cells. Between 7th and 15th week of gestation, 60 % of liver cells are haemopoietic. The fetal-liver derived definitive macrocytes are smaller than yolk sac megaloblasts and contain one third the amount of

Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical medicine, Kolkata, India

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (\boxtimes) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_32

haemoglobin. Differentiation of erythroid cells in fetal liver is dependent on erythropoietin signaling through its receptor and the JAK2 kinase. In the liver, erythropoietin transcripts are present during the first trimester of pregnancy. The liver remains the primary site of erythropoietin transcription throughout the fetal life. In 17 weeks of gestation erythropoietin transcripts also remain present in fetal kidney and it increases after 30 weeks. Erythropoietin is expressed both in fetal liver and postnatal kidney. Like primitive erythropoiesis in the yolk sac, definitive erythropoiesis in the fetal liver is essential for continued survival of the embryo [4].

In yolk sac haemopoiesis is restricted to erythroid and macrophage cells but haemopoiesis in fetal liver includes other myeloid and lymphoid lineages. Megakaryocytes are present in the liver at 6 weeks of gestation. Platelets are first evident in the circulation at 8-9 weeks of gestation. Small numbers of circulating leukocytes are present in the 11th week of gestation. Granulopoiesis is present in liver parenchyma and in some areas of connective tissue as early as 7 weeks of gestation. Despite the low number and immature appearance of hepatic neutrophil, the fetal liver contains haemopoietic progenitor cells such as colonyforming unit-granulocyte-monocyte-macrophage (CFU-GEMM) and colony-forming unitgranulocyte-monocyte (CFU-GM). CFU-GM growth depends upon several cytokines, including granulocyte colony stimulating factor (G-CSF), granulocyte-monocyte colony stimulating factor (GM-CSF), and interleukins [4, 5].

In marrow haemopoiesis, haemopoietic cells are first seen in the marrow of 10–11 week's embryo, [1, 2] and they remain confined to the diaphyseal region of long bones till 15 weeks of gestation.

Initially there is equal number of myeloid and erythroid cells in the fetal marrow. However, by 12 weeks of gestation myeloid cells predominate and the myeloid to erythroid ratio approaches to adult level of 3–1 by 21 weeks of pregnancy. Macrophage cells in the fetal marrow express the lipopolysaccharide receptor CD14. The marrow becomes the major site of haemopoiesis after the 24th week of gestation [6, 7].

Lymphopoiesis is present in lymph plexus and the thymus beginning at 9 weeks of gestation. B cells with surface IgM are present in liver and circulating lymphocytes at 9 weeks of gestation. Lymphocyte subpopulation are detected in fetal liver at 13 weeks of gestation [8]. Absolute numbers of major lymphoid subsets in 20–26 weeks old fetuses defined by antigens CD2, CD3, CD4, CD8 CD19, CD20 and CD16, are similar to those of the newborn [9, 10].

The reconstitution of haemopoiesis by cord blood transplantation indicates that hematopoietic stem cells are present at birth. The immunological reconstitution of an immunodeficient human fetus with fetal liver derived cells indicates that haemopoietic stem cells are present in the fetal liver. Yolk sac stem cells were first thought to seed the liver and eventually the bone marrow. The transient appearance of CD34-positive blood cells associated with ventral wall of the aorta in a 5 week human embryo suggests that the AGMregion-derived stem cell seed the liver and the marrow to provide lifelong haemopoiesis [11].

Human hemoglobin is a tetramer composed of two A-type and two B-type globin chains. The A-globin gene cluster is located on chromosome no. 16 [12] and the B-globin gene cluster is located on chromosome no. 11 [12]. During embryogenesis the genes of both chromosomes are activated sequentially [13].

Hb Gower 1 is the major hemoglobin in an embryo less than 5 weeks of gestation. Hb Gower 2 has been found in embryos of gestational age as low as 4 weeks and absent in embryos older than 13 weeks [14, 15].

A newborn represents the culmination of developmental evidence from conception and implantation through organogenesis. The embryo requires red cells for transport of maternal oxygen to permit the growth and development. Embryonic haemopoiesis begins with the generation of primitive RBC which supplies embryos with immediate oxygen demand. In the placental villi of first trimester, maturation and enucleation of human primitive RBC occurs. Frequency of enucleated RBC is found to be higher in placental villous stroma than circulating erythrocytes during 5-7 weeks of embryonic development. Macrophage progenitors of fetal origin present in the chorionic plate of the placenta before fetoplacental circulation, and migration of macrophage to villous of placenta after the fetoplacental circulation, proves that the macrophage has its role in fetal haemopoiesis and enucleation of RBC. Birth brings dramatic changes in circulation and oxygenation which affects haemopoiesis as the newborn makes a separate biological existence [16, 17].

Hb F is the major hemoglobin of fetal life [18]. Synthesis of HbA can be demonstrated in fetuses as young as 9 weeks of gestation [19, 20]. In fetuses of 9-21 weeks of gestation the amount of Hb A rises from 4 to 13 % of the total hemoglobin. This level of Hb A enables the antenatal diagnosis of b Thalassemia using globin chain synthesis.

The fetal hemoglobin or HbF concentration in blood decreases after birth by about 3 % per week and comprises less than 2–3 % of total hemoglobin at 6 month of age. This rate of decrease is related to gestational age and not to the environment or oxygen tension that occurs at the time of birth.

The fetal blood composition changes markedly during the second and third trimesters. The mean hemoglobin in fetuses progressively increases from 9.0 ± 2.8 g/dl at 10 weeks to 16.5 ± 4.0 g/dl at 39 weeks.

There is decrease in the MCV fetal red cells from a mean of 134 fl/cell at 18 weeks to 118 fl/ cell at 30 weeks of gestation.

The total white blood cell count during the middle trimester is between 4 and $4.59 \times 109/1$ with an 80–85 % preponderance of lymphocytes and 5–10 % neutrophils. The circulating nucleated red cell decreases from a mean of 12 % at 18 weeks to 4 % at 30 weeks. The platelet count remains greater than 15,000/µ ml from 15 weeks gestation to term [21, 22].

Large numbers of haemopoietic progenitors circulate in fetal blood. Blood samples obtained by fetoscopy at 12–19 weeks of gestation reveals a mean of 20,450 BFU-E/ml and 12,490 CF-GM/ ml [23].

Placenta is the site for feto-maternal exchange where nutrients, oxygen, metabolic by-products, hormones and many other molecules pass from fetus to maternal circulation. As per concentration of many molecules in maternal circulation placenta transfers them in fetomaternal pool. Maternal blood flow, fetoplacental blood flow and placental trophoblastic [23] membrane permeability are the main modifiers of placental transfer function. Fetal oxygenation, nutrition and metabolism are modified by

- 1. Altered maternal perfusion
- 2. Altered placental perfusion
- 3. Reduction of placental permeability
- 4. Increase in placental metabolic requirement.

Fetomaternal transfer through placenta may be through diffusion, or be a carrier mediated transfer or active transfer. All are dependent upon placental properties such as fetomaternal blood flow, pattern of perfusion through placenta, surface thickness and property of placental membrane and the metabolic activity of the placenta.

Optimal fetal oxygenation is assured by:

- 1. High uterine blood flow.
- Production of fetal hemoglobin in sufficient amount which has greater oxygen affinity than maternal hemoglobin.
- 3. Uniform fetal perfusion of placenta.
- 4. Increased tissue perfusion.
- 5. High fetal cardiac activity.

For embryonal growth, fetal growth and metabolic requirement, normal placental development and maximizing placental efficacy are essential [24].

For flourishing intrauterine pregnancy, the following are essential

- (a) Trophoblast to anchor and invade the deciduas.
- (b) Ability of uterine vasculature for progressive dramatic increase of blood flow.

After implantation a remodeling of uterine epithelium and stroma occurs and a line of nutrient develops from the maternal tissue to the growing fetus. Endometrial invasion occurs by:

- (a) Interstitial invasion of trophoblast in decidua and stroma.
- (b) Endovascular invasion by which vascular remodeling is initiated.

Both infiltrative and phagocytic components are involved in interstitial invasion of the endometrium by the conceptus. The erosive procedure is indicated by far more molecular signals [25] and there may be a linkage with primitive immune mechanisms; there is a presence of adequate killer cells in the late luteal phase and early pregnant human epithelium. Retrograde trophoblast extension in spiral vessels begins in venous circuit and its flow is against the arterial tree [26]. Remodeling of trophoblast is a continuous process and occurs in two sequences. The first sequence is completed in late first trimester and the second sequence of remodeling is completed by early second trimester. After the second sequence, the trophoblast penetrates the superficial layer of the myometrium.

Uterine artery flow in non-pregnant state is 1-2 % of cardiac output and this increases to 30-50 % of cardiac output at 20-24 weeks of gestation. Non-pregnant spiral artery flow is high resistance and low volume whereas spiral vessels of remodeling trophoblast are low resistance and high capacity.

There is comparative and temporary denervation of uterine artery vasculature in pregnancy [26, 27].

Uteroplacental and fetoplacental circulation should depend on anatomic and humoral mediator. Trophoblastic invasion is less in the periphery of the placenta than the center [28] and thus less involved in fetomaternal exchange.

Anatomically there are three types of trophoblast:

- 1. Villous trophoblast lining the intervillous space with cytrotrophoblast "stem cell" and thus producing syncytiotrophoblast.
- 2. Anchoring mononuclear trophoblast attached to basal plate.
- 3. Invasive, Interstitial and endovascular trophoblast.

All types of trophoblast differentiate and express antigen in highly coordinated and distinct pattern that are influenced by soluble factors and extracellular matrix components [29, 30]. Invasive trophoblast expresses human leucocyte antigen G (HLA-G), a major histocompatibility antigen, expression of which is restricted to a few cell types, including trophoblast [29]. Implantation and early placental development occurs under great regulation and modulation of trophoblastic function.

Placental functional maturation occurs by villous morphological changes in pregnancy. The primary villi are columns of cytotrophoblast. "Secondary villi" develops after mesodermal invasion in the primary villi. The appearance of blood vessels in secondary villi transforms them into tertiary villi. The mature villous functional unit is a barrel, with the staves formed by the large fetal stem vessels and the villous tree arborizing towards the center where recently formed villi are found.

By the end of the fifth week of gestation the embryonic heart begins to pump nucleated erythrocytes from the yolk sac throughout villous circulation. There is a good co-relation between the proportion of nucleated and anucleated RBC and crown- rump length of the fetus [31]. Villous macrophages (Hofbauer cells) are numerous throughout gestation (about 40 % of stromal cells); near term its number becomes less.

Villi develop over the entire conceptus but atrophy over extra placental membranes occurs by the end of the first trimester. Normal villous atrophy may be confused with placental infarct. Remnants of villi in placental membrane are chorionic lave and the villi developed in the placental disc are chorion frondosum.

The delivered placenta is deflated compared to the uterus. The average diameter of the placenta is 18.5 cm (range 10.5-24.5 cm) with mean thickness of 2.3 cm (range 1.1-4.1 cm).

Early stage growth of placenta is due to growth of chorionic disc whereas late stage of gestation placental weight gain is due to increase in placental thickness.

The mean diameter of the placenta at the third to sixth month is 5.8 cm, 8.2 cm, 10.8 cm and 13.0 cm respectively. The number of major villous trunks remain the same. Ultrasonic data review of the mean weekly increase of placental volume between 16 and 24 weeks is 31 ± 8 cm.

This is in a nutshell the development of the hemopoiesis system before the second trimester of pregnancy.

References

- Zon L. Developmental biology of hematopoiesis. Blood. 1995;86:2876.
- McGrath KE, Palis J. Hematopoiesis in the yolk sac: more than meets the eye. Exp Hematol. 2005;33: 1021–8.
- 3. Bloom W, Bartelmez GW. Hematopoiesis in young human embryos. Am J Anat. 1940;67:21.
- 4. Peschle C, Migliaccio AR, Migliaccio G, Ptrini M, Calandrini M, Russo G, et al. Embryonic to fetal Hb switch in humans: studies on erythroid bursts generated by embryonic progenitors from yolk sac and liver. Proc Natl Acad Sci USA. 1984;81:2416.
- Migliaccio G, Migliaccio AR, Petti S, et al. Human embryonic hemopoiesis. Kinetics of progenitors and precursors underlying the yolk sac—liver transition. J Clin Invest. 1986;78:51.
- Emerson SG, Shanti T, Ferrara JL, Greenstein JL. Developmental regulation of erythropoiesis by hematopoietic growth factors: analysis on populations of BFU-E from bone marrow, peripheral blood, and fetal liver. Blood. 1989;74:49.
- Slayton WB, Juul SE, Calhoun DA, Li Y, Braylan RC, Christensen RD. Hematopoiesis in he liver and marrow of human fetuses at 5 to 16 weeks postconception: quantitative assessment of macrophage and neutrophil populations. Pediatr Res. 1998;43:774.
- Gupta S, Pahwa R, O'Reilly R, et al. Ontogeny of lymphocyte subpopulation in human fetal liver. Proc Natl Acad Sci USA. 1976;73:919.
- Rainaut M, Pagniez M, Hercend T, Daffos F, Forestier F. Characterization of mononuclear cell subpopulations in normal fetal peripheral blood. Hum Immunol. 1987;18:331.
- Hann IM, Gibson BES, Letsky EA. Fetal and neonatal haematology. Philadelphia: Bailliere Tindall; 1991.
- Cairo MS, Wagner JE. Placental and/or umbilical cord blood: an alternative source of hematopoietic stem transplantation. J Am Soc Hematol. 1997;90:4665.
- Dame C, Fahnenstich H, Feitag P, Hofmann D, Abdul-Nour T, Bartmann P, Fandrey J. Erythropoietin mRNA expression in human fetal and neonatal tissue. Blood. 1998;92:3218.
- Grosveld F, van Assendelft GB, Greaves DR, Kolias B. Position independent, high-level expression of the human b globin gene in transgenic mice. Cell. 1987;51:975.
- Hecht F, Motulsky AG, Lemire RJ, et al. Predominance of hemoglobin Gower 1 in early human embryonic development. Science. 1966;152:91.

- Huehns ER, Dance N, Beaven GH, et al. Human embryonic hemoglobins. Cold Spring Harb Symp Quant Biol. 1964;29:327.
- Bard H. The effect of placental insufficiency on fetal and adult hemoglobin synthesis. Am J Obstet Gynecol. 1974;120:67.
- 17. McCue CM, Garner FB, Hurt WG, et al. Placental transfusion. J Pediatr. 1968;72:15.
- Pataryas HA, Stomatoyannopoulos G. Hemoglobins in human fetuses: evidence of adult hemoglobin production after the 11th gestational week. Blood. 1972;39:688.
- Thomas ED, Lochte Jr HL, Greenough III WB, et al. In vitro synthesis of foetal and adult haemoglobin by foetal haematopoietic tissues. Nature. 1960;185:396.
- Kazazian HH, Woodhead AP. Hemoglobin A synthesis in the developing fetus. N Engl J Med. 1973;289:58.
- Forestier F, Daffos F, Galacteros F, et al. Haematological values of 163 normal fetuses between 18 and 30 weeks of gestation. Paediatr Res. 1986;20:342.
- 22. Millar DS, Davis LR, et al. Normal blood cell values in the early mid- trimester fetus. Prenat Diagn. 1985;5:367.
- Linch DC, Knott LJ, et al. Studies of circulating hemopoietic progenitor cells in human fetal blood. Blood. 1982;59:976.
- 24. Yao AC, Hirvensalo M, Lind J. Placental transfusion rate and uterine contraction. Lancet. 1968;1:380.
- Fujiwara Y, Browne CP, Cuniff K, Goff SC, Orkin SH. Arrested development of embryonic red cell precursors in mouse embryos lacking transcription factor GATA-1. Proc Natl Acad Sci USA. 1996;93:12355.
- Porcellini A, Manna A, Manna M, et al. Ontogeny of granulocyte- macrophage progenitor cells in the human fetus. Int J Cell Cloning. 1983;1:92.
- Nicola NA, Metcalf D. Specificity of action of colonystimulating factors in the differentiation of granulocytes and macrophages. CIBA Found Symp. 1986;118:7.
- Cudennec CA, Johnson GR. Presence of multipotentialhemopoietic cells in teratocarcinoma cultures. J Embryol Exp Morphol. 1981;61:51.
- Carpenter KL, Turpen JB. Experimental studies on hemopoiesis in the pronephros of Rana pipiens. Differentiation. 1979;14:167.
- Dieterlen-Lievre F. On the origin of hematopoietic stem cells in the avian embryo: an experimental approach. J Embryol Exp Morphol. 1975;33:607.
- Gale RE, Clegg JB, Huehns ER. Human embryonic haemoglobins Gower 1 and Gower 2. Nature. 1979;280:162.

Part XIII

Pharmacological Implication of Fetal Development: Up to Second Trimester

Liabilities of the Fetus as a Patient

33

Mursheed Ali and Subhas Chakraborty

Issues Related to Fetal Development

Twin-Twin Transfusion Syndrome

Twin-twin transfusion syndrome affects roughly 15 % of monochorionic twin pregnancies and regardless of contemporary obstetric and neonatal treatment procedures, is connected with 30-50 % perinatal mortality [1-3]. The haemodynamic changes in twin-twin transfusion syndrome are because of uneven incessant intrafetal transfusion. It is important to note that often neurodevelopmental result is poor in surviving newborn children which is inferable from complexities of the ailment itself and the high rate of preterm birth that goes with this condition [4]. The old criteria of hemoglobin discrepancy and birth weight have been replaced by a ultrasound stage based classification [5]. Overall rates of perinatal survival have increased as a consequence of a number of treatment modalities, including amnioreduction, septostomy, and laser ablation. Amnioreduction offers promising results if it is done at the right stage of the disease, with at least one fetus surviving in more

Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

e-mail: chakrabortysubhas.2007@gmail.com

than 85 % of cases and two surviving in 66.7 % of cases at stage I or stage II disease [6]. The general survival rate, according to a study, for cases before 28 weeks that were treated by laser was 58 %; similarly, amnioreduction offers good results in early stage disease, with at least one fetus surviving in more than 85 % of cases and two surviving in 66.7 % of cases with stage I or stage II disease [7–9]. Laser treatment enhances the degree of single survivor, by decreasing the number of double survivors and double deaths. The key consideration is to oversee pregnancies affected by twin-twin transfusion syndrome with early referral to a tertiary fetal solution unit with experienced practitioners. These patients need cautious assessment before any mediation, which often has to be individualized.

Selective termination of pregnancy in serious twin-twin transfusion syndrome is one alternative that is done in only a handful of centres.

Multifetal Pregnancy Reduction

Fetal reduction can be either specific or nonspecific. In specific fetal decrease, one of the fetuses may have an anomaly that may be lethal and its proximity may jeopardise the survival/ growth of its cotwin(s). Non-selective fetal reduction is generally done early in gestation in high order multiple pregnancies to lessen the probability of high order births with all their

M. Ali, MSc • S. Chakraborty (🖂)

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_33

complications. Fetal reduction has the double goal of preventing the birth of a baby that may have significant abnormalities and in addition avoiding the risk of preterm delivery that is regularly connected with multiple pregnancy. Contingent upon chorionicity, multifetal pregnancy decrease could be achieved by ultrasound guided intracardiac potassium infusion, bipolar cord coagulation, or interstitial laser. In high order multiple pregnancies (triplets or more) multifetal pregnancy reduction has a certain advantages in terms of perinatal result, which is seen in lessened risks for prematurity, cerebral paralsy, and pregnancy related complications. Unless there is a discordance between the fetuses for an anomaly that caries a serious risk of handicap, most fetal terminations in the United Kingdom are done before 24 weeks of gestation, primarily between 11 and 14 weeks. One reason for choosing this period is because it is harder to do transabdominal procedures before 10 weeks because of the little fetal size and the inaccessibility of the fetuses when the uterus essesntially remains a pelvic organ. Moreover, before this time, the spontaneous loss of a fetus may happen. According to one study, multifetal pregnancy reduction is normally done between 11 and 14 weeks, principally in view of a lower miscarriage rate (5.4 %) compared with the risk of spontaneous miscarriage (12 %).

The transabdominal procedure has supplanted the transvaginal technique [10]. As the perinatal result of fewer twins approaches, yet does not reach, that of spontaneous twins, the decrease of higher order multiple pregnancies to a number of two is currently standard practice, as many groups feel that the perinatal mortality and morbidity of twin pregnancies are within acceptable limits. The Human Fertilisation and Embryological Authority has now decreed that at most two fetuses can be replaced at one time, to reduce the risks connected with high order multiples. According to a recent study of the International Registry, in 3513 patients before 24 weeks' gestation undergoing multifetal pregnancy reduction in 11 centres, the overall loss of pregnancy was 9.6 %, with 3.7 % preterm deliveries between 25 and 28 weeks of gestation [10], both of which appear better than the published results for unreduced numerous pregnancies [11]. There is a strong correlation between the starting number of fetuses and the completing number after multifetal pregnancy reduction, with the probability of poor pregnancy result (loss and prematurity) increasing with higher order multiples [12].

Fetal Red Cell and Platelet Alloimmunisation

Fetal Rh or Kell status can now be determined non-intrusively by utilizing coursing fetal DNA in maternal plasma. This mechanical development technological advancement has rendered early intrusive testing redundant in alloimmunised pregnancies. Although in utero transfusions remain the preferred treatment for iron deficient babies affected by red cell alloimmunisation, techniques for monitoring the at risk fetus have developed. Previously, serial amniocentesis was needed to measure optical density 450 levels for calculating the time of the next transfusion. Every amniocentesis carried a risk of miscarriage or preterm labour, and as a few pregnancies required many of these procedures, the risk was considerable. The approach of Doppler velocity assessment in the fetal middle cerebral artery has revolutionized the management of fetal anemia, whatever the cause. This strategy has 100 % sensitivity with a 12 % false positive rate and permits accurate timing of intrauterine transfusions without any requirement for extra invasive procedures [13]. Feto-maternal alloimmune thrombocytopenia is caused by human platelet antigen incompatibility. Fetal intracranial hemorrhages can happen in 10-20 % of cases. The accessible treatment alternatives incorporate maternal treatment with high dosage intravenous immunoglobulin, corticosteroids, a blend of both, or serial intrauterine platelet transfusions. A recent European study on the antenatal administration of feto-maternal alloimmune thrombocytopenia noted that the start of treatment can now be stratified on the basis of sibling history and supports the utilization of first line maternal immunoglobulin treatment, in this way staying away from various intrusive procedures [14].

Screening of Fetus- Prenatal Stage

As more than 90 % of auxiliary and chromosomal abnormalities emerge in pregnancies without any risk factors, anomaly and aneuploidy screening is offered universally. In a few circumstances screening for particular genetic issues may be restricted to certain ethnic groups. In case of Down's syndrome, despite older women being at higher risk, even women about 35 years are screened with this syndrome [15]. In England and Wales, prenatal diagnosis of Down's syndrome cases increased from 28 % in 1989 to 53 % in 1999, and screening programs for Down syndrome that were based on maternal serum biochemistry or ultrasound were more effective and efficient than the screening programs that used advanced maternal age alone [16]. By April 2007 the NHS is required to provide a test that has a detection rate over 75 % and a false positive rate of less than 3 %. Only the consolidated, integrated, quadruple, and serum integrated tests will meet this more stringent criteria.

Diagnostic Testing for Fetal Abnormalities

Recent advancements in fluorescence in situ hybridization and quantitative fluorescence polymerase chain reaction methods have prompted quick reporting times (1-3 days) for Down's syndrome identification of other trisomies. The quick testing of prenatal samples has brought up the issue of whether full karyotypic examination and reporting should be done for these samples. Most women who experience invasive testing do so due to the fact that they have been distinguished as being at high risk by a specific screening method. Full karyotypic examination may detect abnormalities of unknown significance (small "marker" chromosomes, balanced chromosome rearrangements, or regions of variability), which may be inherited. These findings of genetic abnormalities frequently raise ethical issues for the mothers to choose termination of pregnancy or continue

with it and remain anxious for the pregnancy. Generally, 0.07–0.14 % of pregnancies which are karyotyped will have a clinically significant chromosomal abnormality that would not be detected through rapid testing [17, 18]. Preimplantation genetic diagnosis is now recognized as a reliable for diagnozing chromosomal abnormalities arising from parental balanced translocations or rearrangements [19, 20]. This procedure can be utilized to screen fetuses of in vitro fertilization pregnancies. It is expensive and intrusive, but it is suitable for women at particularly high risk because of chromosome rearrangement or those had in vitro fertilization.

Non-invasive Prenatal Diagnosis Strategies

As all current screening techniques include an invasive diagnosis test (for instance, amniocentesis or chorionic villous sampling), which carries a small but definite risk of miscarriage, endeavors have been made to develop less invasive diagnostic strategies that examine the fetal genome through isolation and characterization of DNA from fetal cells distinguished in the maternal blood circulation or investigation of free fetal DNA in maternal plasma. Circulating fetal nucleated red blood cells, mesenchymal stem cells, and trophoblast have all been utilized for different prenatal diagnostic tests [21-23]. The limitation of this technique is the rarity of such cells in the maternal circulation and further, the availability of a reliable fetal marker. The number of fetal cells in maternal circulation vary according to stages of gestation and methods employed for analysis; circulating fetal cells are rare and is estimated to be around one fetal cell in 10⁴ to one in 10⁹ maternal cells in normal pregnancy [24]. As the duration of pregnancy increases, the free fetal DNA increases to around 3-6 % of total DNA in maternal plasma. The fetal DNA concentrations increases during aneuploidy pregnancies [25]. Another new technique of investigating is the analysis of plasma RNA which is employed for non-invasive profiling of gene expression [26].

Recent Imaging Techniques- Fetal Magnetic Resonance Imaging

High resolution ultrasonography allows for the detection of anomalies, but poor views because of maternal obesity or oligohydramnions are limitations. There are faster imaging techniques that allow a single slice to be obtained in less than 400 ms, which obviates most fetal motion artefacts. No harmful impact on the developing fetus have yet been reported. Moreover, the fetal magnetic resonance imaging is useful for assessing abnormalities of the CNS. Levine et al. in 1997 reported that fetal magnetic resonance imaging provided additional information on abnormalities in the brain in 55 % of fetuses [27]. Another advantage of this technique is it its evaluation of changes in the developing brain due to neuronal migration, gyral formation, and myelination. MRI helps in the evaluation of other fetal anomalies like sacrococcygeal teratoma, diaphragmatic hernia, and spinal anomalies. Image acquisition time is expected to decline further and this will enhance the clear visualization of almost any fetal organs.

Fetus Examination- Three Dimensional Ultrasonography

Three dimensional ultrasonography is a moderately new strategy for examination. For a while it appeared to give only aesthetic images with little clinical value. In some clinical and research settings, three dimensional ultrasonography has been demonstrated to be advantageous in assessing the fetus [28, 29]. The indications for three dimensional ultrasound are not yet clear. Most reports have dealt with the identification of abnormalities of the fetal surface, especially cleft lip and palate and spina bifida. Very few studies have looked at the capacity of three dimensional ultrasonography to survey more profound structures. Also, three dimensional ultrasonography permits precise estimation of the volume of any fetal organ. Very few studies have investigated the clinical utility of fetal volume estimations. Once more, as this technology

advances and real time three dimensional images enhance, it may become the imaging modality of choice.

Current Tests for Fetal Abnormalities

- *Fetal viability scans:* This is an ultrasound examination done at 6–10 weeks of gestation with an aim to focus the number of fetuses present, and whether the pregnancy is advancing typically inside the uterus.
- First trimester screening for fetal aneuploidies: This incorporates maternal blood tests and ultrasound evaluation of the fetus. The most effective screening test in the first trimester uses a combination of biochemical markers. pregnancy-associated plasma protein Α (PAPP-A) and human chorionic gonadotropin (hCG), and the nuchal translucency (NT) measurement to adjust a woman's risk for trisomies 21 and 18. Collection of blood for biochemical analysis is performed between 9 and 13 6/7 weeks' gestation and Ultrasound assessment of the NT measurement is performed between 11 and 13 6/7 weeks [30]. Doppler ultrasound assessment of the uterine arteries is more effective in identifying preeclampsia [30].
- *Fetal anomaly scans:* Done at 18–23 weeks of development of fetus. Definite fetal anatomy and structural abnormalities can be analyzed.
- *Risk reassessment scan:* This scan focuses on markers and heart abnormalities of Down's syndrome (trisomy 21). This scan is attempted when a serum biochemistry test risk is high.
- *Cervical screening:* A transvaginal ultrasound gives an image of the cervix; if it shorter than normal or there is funneling, there is possibility of preeclampsia and preterm birth.
- *Fetal echocardiography:* Fetal echocardiography (echoCG) is considered to be an accurate diagnostic tool, reflecting postnatal outcomes well. Fetal echoCG is now widely used in pediatric cardiology and perinatology and even for fetal cardiac intervention, improving the preoperative condition, morbidity and mortality of patients with congenital heart

disease (CHD) [31]. Definite assessment of fetal heart is carried-out by a fetal cardiologist.

- *Fetal scan:* Done at 24–40 weeks growth of fetus. The scan expects to:
 - 1. Assess fetal head size, abdomen and fetal thigh bone for general estimation of fetal weight.
 - 2. Assessment of placenta position and appearance.
 - 3. Examination of fetal developments.
 - 4. Estimation of the amount of amniotic fluid.
 - Evaluation of blood flow to the fetus and placenta through color Doppler.

Invasive Strategies

- *Chorionic villus sampling (CVS):* Helpful for diagnosing fetal chromosomal and genetic disorders. CVS is a diagnostic procedure which involves removing some chorionic villi cells from the placenta at the point where it attaches to the uterine wall. This is usually done at 11–14 weeks of gestation [32].
- *Amniocentesis:* Similar to CVS, but done utilizing amniotic fluid which contain fetal tissues. Both invasive methods convey risk of 1 % miscarriage.
- Fetal MRI: It is a non-invasive, safe methodology executed as supplementary to ultrasound, and is valuable in fetal visceral and soft-tissue depiction like central nervous system abnormalities. MRI provides multiplanar imaging as well a large field of view, facilitating examination of fetuses with large or complex anomalies, and visualization of the lesion within the context of the entire fetal body. Fetal MRI may detect subtle neural tube defects not shown by US and determine the level of the defect in myelomeningocele for potential fetal surgery. However, it is prudent to wait until 17-18th weeks of gestation before performing fetal MRI because of the potential risk to the developing fetus and the current limitations of fetal MRI created by the small size and excessive motion of younger fetuses [33].
- *Fetal intrauterine transfusion:* Intrauterine fetal blood transfusion (IUT) is thought to

be the only life saving therapy, and very effective in the management of preterm Rh isoimmunized pregnancies. Universal use of prophylactic Rh(D) immune globulin has reduced the need for IUT dramatically; however, the procedure continues to be an essential modality for treatment of severe fetal anemia from a variety of causes, such as non-Rh(D) alloimmunization, parvovirus B19 infection, chronic fetomaternal hemorrhage, and homozygous alpha-thalassemia. The procedure is generally limited to pregnancies between 18 and 35 weeks of gestation because before 18 weeks the small size of the relevant anatomic structures causes technical challenges and after 35 weeks IUT is considered riskier than delivery followed by postnatal transfusion therapy [34].

Recent Advancements in Fetal Medicine

Fetal medicine involves the assessment of fetal health, development and wellbeing and therefore methods of fetal intervention to correct aberrations prior to birth. It requires a multidisciplinary approach involving obstetricians, maternal medicine specialists, neonatologists, paediatric cardiologists, paediatric surgical specialists, geneticists and specialist nursing counselors. Fetal medicine addresses the unborn patient. The advances made in fetal medicine now allows for prenatal screening in early firsttrimester for aneuploidy fetuses and for forecast of preeclampsia. This has enhanced the detection rate to 95 % and 80 % with a false-positive rate of 2 % and 10 %, respectively [35].

Recent improvements in fluorescence insitu hybridization or quantitative fluorescence PCR have prompted quick results within 2–6 h of amniocentesis for trisomy 21 [36]. Preimplantation genetic testing is a technique used to identify genetic defects in embryos created through in vitro fertilization (IVF) before pregnancy. Preimplantation genetic *diagnosis* (PGD) refers specifically to when one or both genetic parents has a known genetic abnormality and testing is performed on an embryo to determine if it also carries a genetic abnormality. In contrast, preimplantation genetic screening (PGS) refers to techniques where embryos from presumed chromosomally normal genetic parents are screened for an euploidy [37]. Advance in imaging with introduction of fetal MRI assesses CNS and other anomalies in the fetus [48]. Three- and four dimensional ultrasonography gives added data to investigations of skeletal deformities, facial abnormalities and evaluation of neural tube defects [38]. This technology now offers new opportunities to study embryonic and fetal growth which would make information available on specific movement patterns and quality of movement in the high-risk fetus for detection of fetal neuro-developmental impairment [39].

Until recently, the main inquiry raised by the prenatal diagnosis of a fetal abnormality was whether to abort the fetus or to wait for delivery. Presently, treatment of a few fetal disorders has turned out to be achievable in-utero. With the recent advancement in negligibly invasive fetal surgery, frequency of open hysterotomy has diminished. The latter is performed for conditions, for example, myelomeningocele, large congenital cystic adenomatiod malformation of lungs or a large sacrococcygeal teratoma [40]. Prenatal repair of fetal myelomeningocele is presently at present under scrutiny through a an appropriate randomized trial supported by National Institute of Child Health and Human Development [49]. Introduction of minimally intrusive technique for laser photocoagulation for the treatment of twin-twin transfusion syndrome [41], ablation of posterior urethral valves causing bladder outlet obstruction [42] and tracheal occlusion using a balloon catheter for congenital diaphragmatic hernia [43] are the latest fetal mediation being done in-utero. Non-intrusive prenatal diagnosis of fetal RhD status [44] and fetal anaemia has turned into a reality [45]. Then again, appraisal of fetal growth and subsequent monitoring of fetal oxygenation status can be effectively performed at the level of the fetal and umbilical vessels and haemodynamic changes can be watched and utilized for improving the characteristics of management [46]. With these advances, incorporation of the information of genomics, proteomics, and stem cell exploration and gene therapy into fetal medicine is now required so as to make a wider range of therapeutic interventions possible [47]. Further, the field needs to move into the next phase, that is, evidence-based practice of fetal medicine and surgery where it is essential not to introduce exciting techniques without solid evidence, which eventually would require establishment of and collaborations between multicentre networks of excellence globally; this would help organize and set the agenda for good quality research and clinical progress in this field.

Conclusions

Despite the fact that the diagnosis of numerous fetal anomalies can be done in-utero, the dilemma regarding what is the most suitable treatment remains. Open fetal surgery is technically possible for some of these conditions. Minimally invasive techniques and enhanced imaging coupled with novel methods to anticipate preterm labor will presumably provide more scope for therapeutic mediations. Rapid developments in genomics, proteomics and stem cell research will likewise make in-utero treatment of some genetic conditions possible. The future of fetal medicine lies in the application of basic science to advances in technology and medical science like stem cell therapy and gene therapy to further ensure normal fetal development.

References

- Mohammed NB, Bui T. Foetus as a patient: art and science of foetal medicine. JPMA J Pak Med Assoc. 2010;60(6):417–8.
- O'Brien AL, Kumar S. Advances in fetal therapy. Obstet Gynecol. 2005;7(3):183–9.
- Hecher K, Plath H, Bregenzer T, Hansmann M, Hackelöer BJ. Endoscopic laser surgery versus serial amniocenteses in the treatment of severe twin-twin transfusion syndrome. Am J Obstet Gynecol. 1999;180(3):717–24.

- Haverkamp F, Lex C, Hanisch C, Fahnenstich H, Zerres K. Neurodevelopmental risks in twin-to-twin transfusion syndrome: preliminary findings. Eur J Paediatr Neurol. 2001;5(1):21–7.
- Quintero RA, Morales WJ, Allen MH, Bornick PW, Johnson PK, Kruger M. Staging of twin-twin transfusion syndrome. J Perinatol: Off J Calif Perinat Assoc. 1999;19(8 Pt 1):550–5.
- Quintero RA, Dickinson JE, Morales WJ, Bornick PW, Bermúdez C, Cincotta R, et al. Stage-based treatment of twin-twin transfusion syndrome. Am J Obstet Gynecol. 2003;188(5):1333–40.
- Taylor M, Farquharson D, Cox P, Fisk N. Identification of arterio–venous anastomoses in vivo in monochorionic twin pregnancies: preliminary report. Ultrasound Obstet Gynecol. 2000;16(3):218–22.
- Welsh A, Taylor M, Cosgrove D, Fisk N. Freehand threedimensional Doppler demonstration of monochorionic vascular anastomoses in vivo: a preliminary report. Ultrasound Obstet Gynecol. 2001;18(4):317–24.
- Evans M, Adzick N, Holzgreve W, Harrison M. The unborn patient: the art and science of fetal therapy. Philadelphia: Saunders; 2001.
- Evans MI, Berkowitz RL, Wapner RJ, Carpenter RJ, Goldberg JD, Ayoub MA, et al. Improvement in outcomes of multifetal pregnancy reduction with increased experience. Am J Obstet Gynecol. 2001;184(2):97–103.
- 11. Yaron Y, Bryant-Greenwood PK, Dave N, Moldenhauer JS, Kramer RL, Johnson MP, et al. Multifetal pregnancy reductions of triplets to twins: comparison with nonreduced triplets and twins. Am J Obstet Gynecol. 1999;180(5):1268–71.
- Harrison MR, Golbus MS, Filly RA, Nakayama DK, Callen PW, de Lorimier AA, et al. Management of the fetus with congenital hydronephrosis. J Pediatr Surg. 1982;17(6):728–42.
- Mari G, Deter RL, Carpenter RL, Rahman F, Zimmerman R, Moise Jr KJ, et al. Noninvasive diagnosis by Doppler ultrasonography of fetal anemia due to maternal red-cell alloimmunization. N Engl J Med. 2000;342(1):9–14.
- Birchall JE, Murphy Kaplan MF, Kroll H. European collaborative study of the antenatal management of feto-maternal alloimmune thrombocytopenia. Br J Haematol. 2003;122(2):275–88.
- Egan JF, Benn P, Borgida AF, Rodis JF, Campbell WA, Vintzileos AM. Efficacy of screening for fetal down syndrome in the United States from 1974 to 1997. Obstet Gynecol. 2000;96(6):979–85.
- Smith-Bindman R, Chu P, Bacchetti P, Waters JJ, Mutton D, Alberman E. Prenatal screening for Down syndrome in England and Wales and population-based birth outcomes. Am J Obstet Gynecol. 2003;189(4):980–5.
- 17. Thein AT, Abdel-Fattah SA, Kyle PM, Soothill PW. An assessment of the use of interphase FISH with chromosome specific probes as an alternative to cytogenetics in prenatal diagnosis. Prenat Diagn. 2000;20(4):275–80.

- Ryall RG, Callen D, Cocciolone R, Duvnjak A, Esca R, Frantzis N, et al. Karyotypes found in the population declared at increased risk of Down syndrome following maternal serum screening. Prenat Diagn. 2001;21(7):553–7.
- Scriven P, Flinter F, Braude P, Ogilvie CM. Robertsonian translocations—reproductive risks and indications for preimplantation genetic diagnosis. Hum Reprod. 2001;16(11):2267–73.
- Pettenati MJ, Kap-Herr V, Jackle B, Bobby P, Mowrey P, Schwartz S, et al. Rapid interphase analysis for prenatal diagnosis of translocation carriers using subtelomeric probes. Prenat Diagn. 2002;22(3):193–7.
- 21. Choolani M, O'Donnell H, Campagnoli C, Kumar S, Roberts I, Bennett PR, et al. Simultaneous fetal cell identification and diagnosis by epsilon-globin chain immunophenotyping and chromosomal fluorescence in situ hybridization. Blood. 2001;98(3): 554–7.
- Campagnoli C, Roberts IA, Kumar S, Bennett PR, Bellantuono I, Fisk NM. Identification of mesenchymal stem/progenitor cells in human first-trimester fetal blood, liver, and bone marrow. Blood. 2001;98(8):2396–402.
- Oudejans C, Tjoa ML, Westerman BA, Mulders MA, Van Wijk IJ, Van Vugt JM. Circulating trophoblast in maternal blood. Prenat Diagn. 2003;23(2):111–6.
- 24. Krabchi K, Gros-Louis F, Yan J, Bronsard M, Masse J, Forest JC, et al. Quantification of all fetal nucleated cells in maternal blood between the 18th and 22nd weeks of pregnancy using molecular cytogenetic techniques. Clin Genet. 2001;60(2):145–50.
- 25. Yan Zhong X, Bürk MR, Troeger C, Jackson LR, Holzgreve W, Hahn S. Fetal DNA in maternal plasma is elevated in pregnancies with aneuploid fetuses. Prenat Diagn. 2000;20(10):795–8.
- Costa J-M, Benachi A, Olivi M, Dumez Y, Vidaud M, Gautier E. Fetal expressed gene analysis in maternal blood: a new tool for noninvasive study of the fetus. Clin Chem. 2003;49(6):981–3.
- Levine D, Barnes PD, Madsen JR, Li W, Edelman RR. Fetal central nervous system anomalies: MR imaging augments sonographic diagnosis. Radiology. 1997;204(3):635–42.
- Timor-Tritsch IE, Platt LD. Three-dimensional ultrasound experience in obstetrics. Curr Opin Obstet Gynecol. 2002;14(6):569–75.
- Dyson R, Pretorius D, Budorick N, Johnson D, Sklansky M, Cantrell C, et al. Three-dimensional ultrasound in the evaluation of fetal anomalies. Ultrasound Obstet Gynecol. 2000;16(4):321–8.
- Cristina Z, Amar B. Fetal medicine the way forward. J Obstet Gynecol. 2009;59(4):301–307.
- 31. Yu C, Khouri O, Onwudiwe N, Spiliopoulos Y, Nicolaides K. Prediction of pre-eclampsia by uterine artery Doppler imaging: relationship to gestational age at delivery and small-for-gestational age. Ultrasound Obstet Gynecol. 2008;31(3):310–3.
- Bahado-Singh RO, Wapner R, Thom E, Zachary J, Platt L, Mahoney MJ, et al. Elevated first-trimester nuchal

translucency increases the risk of congenital heart defects. Am J Obstet Gynecol. 2005;192(5):1357–61.

- 33. Rosatelli MC, Saba L. Prenatal diagnosis of β-thalassemias and hemoglobinopathies. Mediterr J Hematol Infect Dis. 2009;1(1):e2009011. http:// www.ncbi.nlm.nih.gov/pmc/articles/PMC3033155/.
- Brahmaiah U, Ratha C. Foetus as a patient. Am J Obstet Gynecol. 2005;192:1357–61.
- Deka D, Sharma N, Dadhwal V, Suneeta M. Successful application of middle cerebral artery peak systolic velocity to time intrauterine transfusions in Rh isoimmunised fetus. J Obstet Gynecol India. 2006;56(6):534–6.
- Nicolaides KH. Some thoughts on the true value of ultrasound. Ultrasound Obstet Gynecol. 2007;30(5): 671–4.
- 37. Choolani M, Ho S, Razvi K, Ponnusamy S, Baig S, Fisk N, et al. FastFISH: technique for ultrarapid fluorescence in situ hybridization on uncultured amniocytes yielding results within 2 h of amniocentesis. Mol Hum Reprod. 2007;13(6):355–9.
- Rutherford MA. Magnetic resonance imaging of the fetal brain. Curr Opin Obstet Gynecol. 2009;21(2):180–6.
- Kurjak A, Miskovic B, Andonotopo W, Stanojevic M, Azumendi G, Vrcic H. How useful is 3D and 4D ultrasound in perinatal medicine? J Perinat Med. 2007;35(1):10–27.
- 40. Kurjak A, Tikvica A, Stanojevic M, Miskovic B, Ahmed B, Azumendi G, et al. The assessment of fetal neurobehavior by three-dimensional and fourdimensional ultrasound. J Matern Fetal Neonatal Med. 2008;21(10):675–84.

- Moise Jr KJ, Johnson A, Carpenter RJ, Baschat AA, Platt LD. Fetal intervention: providing reasonable access to quality care. Obstet Gynecol. 2009;113(2, Part 1):408–10.
- 42. Senat M-V, Deprest J, Boulvain M, Paupe A, Winer N, Ville Y. Endoscopic laser surgery versus serial amnioreduction for severe twin-to-twin transfusion syndrome. N Engl J Med. 2004;351(2):136–44.
- 43. Welsh A, Agarwal S, Kumar S, Smith RP, Fisk NM. Fetal cystoscopy in the management of fetal obstructive uropathy: experience in a single European centre. Prenat Diagn. 2003;23(13):1033–41.
- Grethel EJ, Nobuhara KK. Fetal surgery for congenital diaphragmatic hernia. J Paediatr Child Health. 2006;42(3):79–85.
- Hung E, Chiu R, Lo Y. Detection of circulating fetal nucleic acids: a review of methods and applications. J Clin Pathol. 2009;62(4):308–13.
- 46. Pretlove S, Fox C, Khan K, Kilby M. Noninvasive methods of detecting fetal anaemia: a systematic review and meta analysis. BJOG Int J Obstet Gynecol. 2009;116(12):1558–67.
- Mari G. Doppler ultrasonography in obstetrics: from the diagnosis of fetal anemia to the treatment of intrauterine growth-restricted fetuses. Am J Obstet Gynecol. 2009;200(6):613.e1–e9.
- Kumar S, O'Brien A. Recent developments in foetal medicine. BMJ. 2004;328:1002–6.
- 49. Hopkins LM, Feldstein VA. The use of ultrasound in foetal surgery. Clin Perinatol. 2009;36:255–72.

Fetus as a Patient During the First and Second Trimesters of Growth and Development

34

Priyodarshi Sengupta, Mainuddin Naskar, Raj Gupta, Nandita Bose, Sushanta Banerjee, and Niranjan Bhattacharya

Introduction

Fetus as a patient is a new concept which is starting to get more prominence in the field of obstetrics and gynecology along with other fields of medicine like pediatric cardiology, pediatric neurology and surgery and even among basic biomedical scientists [1]. Fetal medicine depends on the well being of the in utero fetus during the time of pregnancy and it mainly focuses on the different aspects of repairing and healing various fetal abnormalities which might appear during the

M. Naskar, MBBS Department of Regenerative Medicine and Translational Science, Kolkata, India

- R. Gupta, MSc Department of Regenerative Medicine, Calcutta School of Tropical Medicine, Kolkata, India
- N. Bose, MD Director, School of Tropical Medicine, Kolkata, India

Formerly, Professor, Department of Pathology, IPGMER, SSKM Hospital, Kolkata, India

S. Banerjee, MD Director, Medical Education, Govt. of West Bengal, Kolkata, India

Formerly, Professor and Head, Department of Pharmacology, RG Kar Medical College, Kolkata, India whole duration of pregnancy. Previously treatment was restricted to the neonatal periods but with recent advancements in the basic science of medicine and surgery there is a high possibility to detect and treat any anomaly of the fetus prenatally [2].

Diagnosis of Fetal Abnormalities by Invasive and Non-invasive Techniques

Currents advancements in diagnostic tools and techniques like fetal viability scan an ultrasound examination which can determine the number of fetuses present between 6 and 10 weeks time along with detection of fetal aneuploidies by maternal blood test and a special ultrasound technique to assess the fetus for any chromosomal abnormalities particularly for Down's syndrome can be conducted [2]. In the second trimester fetal anomaly scans can be performed by assessing the detailed anatomical and physiological structures of the fetus. Also a type of risk assessment scan can be used for examining soft markers and cardiac

© Springer International Publishing Switzerland 2016 N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development: First and Second Trimesters*, DOI 10.1007/978-3-319-14874-8_34

P. Sengupta, MSc

Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India e-mail: sanjuktaniranjan@gmail.com

anomalies and can be replaced with a serum biochemistry test in case if there is too much risk for the fetus. Fetal echocardiography tests can be also carried out for detailed evaluations of fetal cardiac anomalies [2]. Fetal well being scan is another important and latest technology by which the fetal abdominal size, head size, fetal weight, placental position, fetal movements and the volume of amniotic fluid can be assessed. Coloured Doppler Scan can be utilized for assessing the materno fetal blood flow through the placenta [2].

Fetal intrauterine transfusion (IUT) is considered to be the best choice for treating any anemia stemming from various blood disorders in the fetus. In case of Rh isoimmunised fetuses this intervention is accepted as the standard method by which intravenous blood transfusion can be given to an anemic fetus [3].

With the usage of ultrasound imaging in the first trimester there has been encouraging effects and influences in the improvement of fetal diagnosis and treatment. Two dimensional and three dimensional scans for embryonic improvement [4, 5], early embryonic circulation [6, 7], uterine receptivity, formation of yolk sac [8], embryo implantation [9] and detecting ultrasound biochemical disorders have become possible in the first trimester itself. Ultrasound imaging has immense application for invasive procedures like amniocentesis (which is best performed in the first trimester due to the abundance of amniotic fluid) chorionic biopsy, cardiocentesis and funiculocentesis [10].

Basic understanding of the fetal physiology and anatomy like renal function [10], fetal cardiac pathology, fetal hearing [11] and even behavioral changes of the fetus [12] by the help of ultrasound imaging techniques are also now possible. Certain invasive and cervical procedures include the transvaginal scan to detect preeclampsia of the fetus and karyotyping for fetal chromosomal problems. This process is also known as Chorionic villus sampling (CVS) and is usually done at 11-14 weeks of the first trimester period using placental tissues [13]. However the above invasive procedures are associated with a 1 % chance of miscarriage. Transvaginal USG can be also employed in the mother during the first trimester to assess fetal malformations occurring due to chromosomal aberrations. In the first trimester the sensitivity of this diagnostic tool is around 35 % to a maximum of 65 % or more and in the 20th week of the second trimester it is more than 80 % [14, 15].

Physiological midgut herniation, ascites, fetal edema or hydrops fetalis and some degree of hydrocephaly can be also detected in the first trimester. However not all anomalies can be detected in the first 3 months of development by imaging techniques as the fetus is still immature.

Fetal Down's syndrome, another major genetic disorder that is frequent among fetuses of mothers who possess a great risk of genetic aberrations can be detected in the first trimester by using three dimensional ultrasound imaging of the nuchal structures although there are many practical problems associated with this method. Total nuchal structure determination in the first trimester by three dimensional imaging was first introduced in 1998. In the first trimester there are often mistakes in identifying the exact anatomy which might be due to the limitations of the operator technique. Mistaking the amnion for the nuchal skin or measuring the nuchal translucency with fetal head flexed or extended can lead to wrong diagnosis of the disease. Certain set of parameters are required for diagnosing fetal Down's syndrome like appropriate image magnification, precise caliper placements and the optimal positioning of the fetus. However this diagnosis sometimes becomes extremely difficult in the first trimester as the anatomical status of the fetus still remains in a developmental phase. Also the age of the mother is a vital criteria for detecting Down's syndrome in the fetus [16]. Incidence of brain herniation mass due to encephalocoele in fetus is around 1-4 in 10,000 live births. Diagnosis of this disease in the first trimester might be impossible since the tumor size of the fetus will be small and can be often confused with some other anatomical anomalies thereby leading to a wrong diagnosis. Also at the end of the first trimester and the beginning of the second trimester meninges maybe the only structures protruding out to the exterior part of the brain thereby having the appearance of a cyst and can be easily confused with that of a perinatal brain tumor [17].

Megacystis, urethral obstruction and anomalies of the urethral valve, hypertrophic bladder, dysplastic kidneys, ureteral dilations and absence of abdominal muscles can be easily detected by 3D USG in the first trimester [18]. Other neurological problems associated with failure in closure of the neural tube can be determined in the sixth week of the first trimester development. Also different pulmonary herniation and cysts can be diagnosed in the first trimester as the diaphragm completes development by the 9–12th week of fetal development. Defects of the fetus like vertebral ossification by guided USG scan can be determined in the second trimester as the vertebral development completes its formation between the 10th and 12th weeks of the first trimester [19].

Diagnosis of holoprosencephaly is also possible during the tenth week of the first trimester [20].

Anomalies of the umbilical cord like a single artery formation or hematomasis are, however, tough to detect in the first trimester as the full maturation of this cord happens during 12–14th week. This anomaly can be detected by Doppler angiography in the mid second trimester phase.

Fetal hands, fingers, feet and toes can be detected in the late first trimester with the help of 2D USG with a diagnostic efficiency of 12–41 % and around 41–65 % in case of a 3D USG system [21]. Facial malformations can be detected between week no.12 and 20 with 3D imaging along with the whole fetal face and body. Other anomalies like changes in the ductus venosus blood flow velocities can be also assessed in the first and early second trimesters [22–26].

MRI of the fetus can be used mainly to detect any fetal central nervous system abnormality along with any associated shortcomings as it is a safe, non-invasive process. MRI scan of the fetus can be extremely helpful in detecting cases of non conclusive complex CNS malformations also [27]. However applications of MRI in the first trimester are extremely limited except in cases where the assessment of the pregnant mother's health or voluntary interruption of pregnancy by the mother herself is essentially required. Use of MRI in the second trimester can be performed to detect any anomaly of the fetus along with ultrasound examination. Since the last two decades, the role of ultrasound screening in the first and second trimesters has not been emphasized by medical the community due to undue risks that might be associated with fetal growth and development. The American College of Obstetrics and Gynecology (ACOG) does not endorse its use in the initial stages of pregnancy due to ethical values and exploitations that might be associated with peri natal screening in many countries. However it is still widely practiced because ultrasound imaging is the most effective way by which early detection of fetal anomalies can be diagnosed. Also a study revealed that mothers who underwent a second trimester USG imaging favor the use of this tool also in the first trimester, as it will help them to opt for an early abortion in case if they want to opt for such processes [28].

Effect of Drugs, a Serious Contender for Making the Fetus a Patient in the 1st and 2nd Trimester

Both excessive drinking and heavy drug use have a massive effect on the fetus especially during the first and second trimesters. In the first trimester along with organogenesis, morphogenesis is also initiated. In the first 8 days before implantation, the embryo is known to be in a stage also termed as "all or none phase" where any damage during this phase will result in embryo abortion. However if the blastocyst can manage to escape and negate the harmful effects of toxins or drugs during this stage then it can undergo the process of implantation and further proceed for the growth and development without any anomalies associated. However it is always advised by physicians that abuse of drugs in a non-scientific manner can have a profound effect on the neurobehavior of the fetus, delay in synapse formation of the CNS and other teratogenic effects. ACE inhibitors used for treating hypertension in mothers were previously thought to be teratogenic during the second trimester, however now it has also shown to increase the chances of fetal malformations in the first trimester having a direct effect on the fetal CNS system [29].

The Australian health authorities have published a guideline detailing the effect of different drugs during the various stages of trimester development [30]. Cardiovascular drug Angiotensin II Receptor Antagonists (ARAS) has no reports for fetal adverse effect during the first trimester which is probably because of lack of clinical data. However pregnant mothers taking such drugs should be changed to anti hypertensive drugs in order to maintain a normal blood pressure. Drugs like Phenindone D and Warfarin clot dissolving agents can lead to embryopathy of the fetus in the first trimester itself [30].

Lithium toxicity from Lithium salts D in the first and second trimesters can prove fatal to the fetus also. One to 2 % of neural tube defects like Spina Bifida have been seen to be associated with sodium valproate in the first trimester of pregnancy. Other allergic and immune system drugs like Levocarbastine B3 have shown to cause teratogenic results in animal models during the first trimester [30]. Nicotine as a chewing gum and cigarette smoking can cause harm to the fetus in the first trimester if it is exposed to its effect for long period [30].

Azole anti-fungal drugs can also lead to potential teratogenesis of the fetus due to aberrant sterol metabolism [53, 54]. Exposure to such drugs in the second trimester can often result in the fetus undergoing dysmorphic anomalies like hypoplasia of the nasal bones, craniosynostosis, bowed tibia and femur, cleft palate humeral radial fusion [31].

FDA has established five categories of drugs namely the A, B, C, D and X based on their level of toxicity and teratogenicity to the fetus. 'A' category shows no fetal risks, 'B' also shows no toxicities to humans but has shown adverse effects in animal studies whereas 'C' refers to adverse fetal effects in case of animal models but not in case of humans. 'D' shows fetal risks but can be prescribed in cases where it is necessary to outweigh the risks in order to impart beneficial effects. 'X' is where drugs can cause fetal toxicities and outweighs the patient benefits and induces unnecessary risks to the fetus and the mother [32].

Despite widespread concern about drug toxicity to the fetus it has been reported that only 2-3 % of all fetal congenital malformations and anomalies are directly related to the adverse effect of drugs. In fact genetic, environmental and other unknown causes have been reported to be some of the leading causes behind fetal anomalies. The passage of drugs through the placenta to the fetal compartment is actually dependant on a number of important factors like the molecular weight and size of the drug (less than 500 Da drugs can easily pass but not drugs of high molecular weight), its physico-chemical and biological properties. Not all drugs can pass through the placenta in the first and second trimester of fetal development as the placenta remains intact and allows only selective exchange of molecules across the placenta.

Also the fetal age is an extremely important criteria for drug toxicity or drug induced toxicity to the fetus. As discussed elsewhere in this chapter, drugs in the first trimester especially before the implantation of the blastocyst till the 20th day of fertilization which is also known as the "all or none stage", can either abort the development of the embryo or it can undergo successful implantation due to protective mechanisms in place without any adverse events. Teratogenesis is seen at a maximum between the third and eighth weeks of post fertilization. During the second trimester when organogenesis is nearing its end no observable level of acute toxicity of drugs to the fetus can be observed due to enhanced functions of the fetal liver and kidney, two main centers for drug metabolism and excretion in humans. However some potent drugs still might be able to alter the growth and development of the fetus in the second trimester although teratogenesis is unlikely. With increase in placental metabolism doses can be higher which might have a direct effect on the fetus also.

A study was conducted regarding the effects of anti retroviral therapies like acyclovir, valacyclovir or faciclovir during the first 3 months of 1,804 early pregnant mothers in the first trimester who received ARV therapies against a sample size of 830,000 [33]. It was found that the percentage of major birth defects in case of the 830,000 infants were 2.4 % or 19,000, and in the case of the 1,804 pregnant mothers who received anti retroviral therapies in their first trimester, it was found that 2.2 % or 40 infants had major birth defects [33]. Therefore the study confirmed that there was not a significant difference in the number of major birth defects between the ARV group and the normal group thereby concluding that it is not always the effect of drugs that induces fetal toxicities and the usage of certain drugs in certain periods of gestational time with safe doses can be done without imparting any harm to the developing fetus [33].

Adverse events like teratomas are often associated with drugs taken in the first trimester, that is, when organogenesis and morphogenesis occur side by side [34], whereas in the case of second trimester adverse events of drugs are often associated with intra uterine growth reduction, bleeding and infections [35]. However some of the important criterias on which the fetal effect depends are the type of drugs used by the mother during pregnancy and at what stages, duration of exposure to the drugs and the gestation phase of the fetus [36].

Anti-tubercular Drug Mediated Toxicity to the Fetus in Early Pregnancy

Treatment of tuberculosis of the mother with Paraamino salicylic acid in the first trimester which is classified as a C class drug by the USFDA during pregnancy can result in congenital abnormalities to the fetus. Therefore it should be used in pregnant women only when there is no alternative treatment option available with any other suitable drug that can affect the fetus to a minimal level [37, 38]. Other C category drugs like fluroquinoles can induce fetal toxicities like damage to the articular cartilage, arthritis, lesions in animal models, [37, 38] ethionamide which can cross the placental barrier and has shown to induce fetal malformations [39, 40]. Aminoglycosides on the other hand are D category drugs which can rapidly cross the placental barrier and induce ototoxicity and nephrotoxicity [37, 38, 41].

Teratogenic Effect of Drugs on the Fetus during 1st and 2nd Trimesters

Evidences for the teratogenic effects of most anti neoplastic drugs in the first trimester are little as very few women are detected with cancer in this stage. However with limited data available, it has been noted that chemotherapy during the first trimester is often associated with teratogenic effects on the fetus and causing malformations. The most vulnerable and susceptible period of teratogenic effects of cancer drugs in the first trimester is generally between 2 and 8 weeks with a high risk of 20–30 % fetal teratogenesis [42].

Chemotherapy has a detrimental effect on the development of the fetus during the first and second trimester. During the first trimester, i.e., at about 17 days of post conception in the preembryonic period rapid division of the cells are observed and any chemotherapeutic toxicity during this stage can lead to abortion. On the other hand there are many protective and preventive mechanisms in place, which may help the developing embryo during the 2-8 weeks phase to escape the chemotherapeutic drug toxicities and may lead to organogenesis and the successful development of the fetus without anomalies [43]. However any end organ damage to the developing and immature heart, lung, renal system, limbs and neural tube can be irreversible in nature during the first and second trimester period.

During this period of development the renal and gastrointestinal tracts of the fetus along with the cerebral cortex remain highly susceptible to chemotherapy induced drug toxicity which can often prove fatal for the fetus [43]. It has been noted that there is an increased risk of around 10-20 % of fetal abortion, inhibition of blastocyst implantation and trophoblast migration in the first trimester if the chemotherapeutic drugs are exposed to the fetus [44, 45].

Continuous exposure of the fetus to cancer drugs in the second trimester can further complicate things leading to Intra uterine growth retardation (IUGR), fetal pre term delivery but no increase in the case of congenital anomalies [46]. However genetic predisposition factors also play an important role in fetal toxicity in respect to cancer drugs as it has been observed that all chemotherapeutic drugs do not express their adverse effects and toxicity levels on the developing embryo at the same level as they do in some other embryos [47].

In case of slow going cancers it is better to advice the patient to wait till the second trimester to start chemotherapy during which a multi drug treatment can be started. However in cases where the chemotherapy should be initiated from the first trimester it is advisable to start with a single dose of safer chemotherapeutic drug like a vinca alkaloid or anthracycline. Unfortunately in a real life scenario in order to assess the risks of the mother's cancer it is often not an easy choice to delay the chemotherapy regimen till the second trimester as it might bring in more harm to the mother's as well as the fetus' health [42]. However it has been noted that most babies where the maternal chemotherapy was started during the second trimester and beyond were born healthy without any structural malformations [42].

One of the most common non steroid antiinflammatory drug (NSAID) used by pregnant mothers is Indomethacin [48]. It is an analgesic used commonly to treat symptoms of rheumatoid arthritis, ankylosing spondylitis and osteoarthritis. Indomethacin functions by inhibiting the activities of cyclo-oxygenase enzyme. In animal models it has been observed that Indomethacin can cross the placenta freely during the second trimester onwards whereas in the first trimester the passage is minimal [49–51].

Regarding its use in humans, Aselton et al. (1985) reported that 1 congenital defect was observed in 50 women who had taken this drug during their pregnancy [52]. Also another report revealed that nine babies of eight mothers who had consumed Indomethacin in the first trimester had a mild degree of hypospadia and no adverse reports were reported in the remaining group [53]. In a large study conducted at Michigan with a sample size of 229,101 pregnancies over 5 years where the mother had consumed Indomethacin prescribed during the first trimester, only 6.1 % of birth defects were observed [54].

Studies by Moise et al. [50] revealed the noticeable passage of Indomethacin in the second half of pregnancy from the second trimester onwards. Indomethacin has profound negative effects on the fetus like constriction of the ductus arteriosus leading to tricuspid regurgitation in utero and resulting in hydrops fetalis and fetal death [55, 56], decreased urine output [57], and oligohydraminos [58].

Radiotherapy Induced Fetal Anomalies

Incidence of cervical cancer, Hodgkin's lymphoma, breast cancer and melanoma are quite frequent at the time of pregnancy. Radiotherapy often used to treat such cases of malignancies can result in fetal abortion if the abdomens of pregnant women are not properly shielded. Abdominal shielding can reduce the risks of fetal death by 50 %. However, in most cases, radiotherapy in the pregnant mother is contraindicated. Chromosomal abnormalities can be one of the leading causes behind miscarriages and fetal abortions in the first trimester. Progesterone deficiencies also result in a high incidence of fetal deaths in the second trimester. Also 15 % of pregnancy losses in the second trimester can be attributed to other factors like uterus malformations, uterine fibroid formations and other cervical problems which might also contribute to premature birth [59].

Are Anti Retroviral Drugs Safe for Use in the 1st and 2nd Trimesters?

Use of antiretroviral drugs during the first trimester of pregnancy is extremely harmful for the fetus. This is the time of organogenesis when the ectoderm, endoderm and mesoderm form internal organs. Drug toxicity during this time results in the most severe birth defects. The Antiretroviral Pregnancy Registry, an international database compiling data on HIV+ pregnant women and antiretroviral treatment, reports that the rate of birth defects in children exposed to ARVs at any time in utero is 2.7 % [18]. Children exposed to ARVs during the first trimester show a slightly higher incidence of birth defects at 2.9 % [60].

Therefore the need to follow certain standard guidelines for application of ART in the first

trimester is always recommended. Based on ethical considerations a clinician has to assess the risk benefit ratio of the mother and the fetus and is encouraged to start ART's from the first trimester in cases of severely infected mothers who have a high risk of transmitting the disease to the fetus [61].

Risk of late acidosis [62, 63] with certain drugs like Didanosine (ddl) in second line RT regimens should be avoided in pregnancy as it can lead to fetal and maternal deaths. A mother may be given didanosine (ddl) in second-line ART regimens. Exposure to AZC and 3TC drugs in the first trimester has shown around twofold increase of risk to the fetus. 6.3 % birth defects were observed in live born infants who were exposed to ddl drug compared to just 1.1 % in cases of late exposure to this drug [64]. Also a French study reported that the combinatorial treatment of AZC and 3TC given to women during the second trimester period of fetal development resulted in a high rate of fetal anemia, neutropenia when compared to infants who were only exposed to AZT [65]. In cases of oral administration of calcium channel antagonists like verapamil, nifedipine in pregnant mothers during the first trimester for treating hypertension did not show any teratogenic risk to the fetus [66].

Fluconazole should be avoided during the first trimester of pregnancy as there are four reports of craniofocal and skeletal congenital malformations at high doses during the first 3 months of pregnancy [67, 68]. However in low doses no malformations have been observed in more than 800 cases of pregnancies [69-72]. Rifampacin a very potent anti tuberculosis drug should be also used with caution especially in the first trimester as it may have teratogenic effect on the fetus. Use of Clarithromycin in animal models has been associated with major malformations but in human studies comprising a study group of 265 women showed no evidence of teratogenicity [73, 74]. Also, in Toronto, Canada, 123 patients when treated with azithromycin in the first trimester, no increase in fetal malformations were observed when compared to a non-teratogenic antibiotic drug [74]. Also there is a huge concern regarding the safety of trimethoprim sulphamethoxazole which is the preferred prophylactic drug for Pneumocystis jirovecii or PCP regarding its use in the first trimester [75–77].

Physiological and Anatomical Disorders of the Fetus in the Early Stages of Pregnancy

Malformations of the brain are usually associated with the first and second trimester of pregnancy. Lesions in the white matter of the fetus has been observed between the mid second trimester to the middle of the third trimester as this is the time when the functional and structural maturation of the brain takes place [78, 79].

Insulin like growth factors-1 and 2 are extremely important in the initial phases of growth and development of the fetus as they are directly related to the fetal size and brain development [80, 81]. Excess secretion or rather passage of insulin can result in conditions like macrosmia [82] whereas underproduction has found to be associated with congenital agenesis of the pancreas [83]. Diabetes mellitus in neonates has shown to induce growth retardation in both animal and human systems due to depletion of the beta pancreatic cells further leading to cell cycle arrest and reduction in angiogenesis thus limiting the overall capacity of the fetus to produce insulin [84]. It should be noted that maternal IGF-1 and 2 do not cross the placenta directly. They instead have shown to interact with the placenta and help in establishing a feto placental unit and by the help of the IGF binding proteins and proteases, insulin like growth factors are delivered into the placenta [85]. Thyroid and sex hormones have little effect on fetal growth and development. Hormone leptin has shown to help in the growth of the fetus in the initial phases of pregnancy. Also in the absence of growth hormones, it has been seen that the fetus tends to be small in size [86].

Intrauterine growth development another common complication observed during pregnancy is often found to be associated with the second trimester. Any deformity or harm to the placenta can result in its integrity getting destroyed and reduction in its size leading to an extreme adverse environment for the fetus. Preclampsia, a very common condition in the mother, can result in the decrease of placental growth, leading to development of less number of terminal villi. This further results in to a severe reduction in the metabolic rate by cutting down the level of oxygen and glucose transport to the fetus. Due to this there will be a direct effect on the metabolic activity of the fetus in the organogenesis period as there will be already a huge demand for high energy to perform various physiological activities and functions. This failure of the placental layer will have a severe effect on the development of the fetus in the first and second trimesters [87]. Other factors like multiple gestation, chromosomal anomaly, fetal infection or toxicity and delivery of toxic and harmful substrates can also induce intrauterine growth reduction (IUGR) among neonates in the first 6 months of pregnancy [88].

IUGR has direct consequences on infants who are born with a high hematocrit. This is an erythropoietic response to hypoxic events due to insufficiency of the placenta. Hypertension in mothers is also related to thrombocytopenia and neutropenia in the fetus [89]. To prevent IUGR or to detect the fetal viability and its gross malformations an ultrasound examination is performed normally during the first 8–10 weeks of gestational period [90]. Physical measurements like symphysis-fundus height (SFH) are also recommended when USG is not available. A study in The Netherlands revealed that starvation of the mother during the first phases of pregnancy can also lead to IUGR [91].

Fetuses of Insulin dependent diabetic mothers normally have two to eightfold risks of teratogenesis [92, 93] in the CNS like Spina bifida, heart, kidney and skeleton like caudal regression [92–94]. Grave's disease, Systemic Lupus Erythromatosus, rhesus allo-immunization, anemia due to hemolytic disease of the newborn are also commonly associated with second trimester pregnancy due to transplacental passage of maternal antibodies. Fetal hypothyroidism is linked to growth retardation and advanced bone ageing and craniosynostosis [95].

Ischemic lesion of the cerebrum can cause schizencephaly of the fetus which is accompanied by the presence of abnormal grey matter particularly in the gyra [96]. Injuries due to trauma or accidents during the 24th week or end of second trimester can lead to cystic lesions of the brain with smooth walls devoid of gray matter line and clefts in the brain [96].

Introduction to Hematological Disorders of the Fetus and in Utero Transfusion

Forty years ago the fetus would have been conceived as a mere passenger in a closed and restricted environment inside the mother's womb. With the advent of two and three dimensional and real time USG scans it has been possible to consider the fetus now as a patient in terms of medical definition although ethically it is still debatable. In the late 1970s the fetus's hematologic stature could be assessed only through the bilirubin level biomarkers obtained from the amniotic fluid. In 1973 Valenti used a 27 gauge needle and punctured the chorionic blood vessels under direct visualization [97]. This procedure came to be known as fetoscopy [98] which is now restricted in its use due to a 5 % fetal mortality rate. In 1983 Daffos and his group further reported a new minimally invasive diagnostic technique with the help of real time USG guided imaging whereby a 20 gauge needle was directly inserted into the umbilical vessels of the fetus; a low mortality rate of 1 % was reported [99]. Also as time went by different parameters for assessing fetal blood disorders also changed, thereby giving rise to a new field of medicine also known as fetal hematology where the fetal hematocrit and the leukocyte count became some of the important parameters to assess the well being of the fetus. The fetal hematocrit and the leukocytic count increases with gestational time whereas erythroblasts and reticulocytes count decreases [100–102]. However the fetal platelet count remains more or less stable during the whole pregnancy period without much observable changes [100].

Liley in 1963 was the pioneer of the first successful treatment of hemolytic disease of the fetus when he introduced the concept of intraperitoneal transfusion [103]. In 1981 Rodeck et al. modified the technique a bit and the first intravascular fetal blood transfusion (IUT) using a needle into the chorionic plate vessels directly under visualization or fetos-copy was performed [104].

Another group from Denmark, the following year, reported an intravascular transfusion of a fetus by puncturing the umbilical cord under the guidance of USG [105]. Since then many researchers have shared their different thoughts and opinions in the process of intra uterine blood transfusion and serious efforts are still constantly being undertaken in order to upgrade the current standard of treatment. Over the years with the evolution of fetal therapies, the intraperitoneal transfusion technique was abandoned. The intravascular technique became the standardized intervention process and therefore it became a major technical advancement in the field of fetal hematology and surgery. Survival of hydropic fetuses was increased by the intravascular mode of blood transfusion although it is still not clear whether this process would be an efficient one in case of non hydropic fetuses [106]. Through this IUT it has been possible to treat different disorders of the red blood cells such as red cell alloimmunization, fetomaternal haemorrhage and parovirus infections. Fetal thrombocytopenia occurring due to platelet alloimmunization in the fetus has also been undertaken through this IUT process.

Due to the presence of the blood placental barrier, there is no mixing of the maternal and fetal blood but different FACS analysis has shown the presence of small fetomaternal hemorrhages in most cases of pregnancies. When the fetal erythrocytes with their fetal red cell antigen enter the maternal circulation, formation of antibodies

takes place mediated through a process also known as red cell alloimmunization. The placenta plays an active role in transferring the IgG antibodies via itself into the fetal circulation thereby causing antigen and antibody reactions in fetal erythrocytes resulting in cell medicated destruction of red blood cells especially in the spleen, which decreases the rate of RBC production and ultimately causes fetal anemia. Apart from Rh and D antigens, there are 43 other antigens that are implicated in hemolytic disease of the new born like Duffy, Kidd and Kell. However the intravascular transfusion of erythrocytes too has certain shortcomings like the formation of cord hematoma, fetal bradycardia, wide variable swings in the parameters between transfusions and formation of porencephalic cysts. A combined technique of intraperitoneal and intravascular transfusion also revealed this to be a much better process than either of the two when applied individually. It resulted in a more stable hematocrit and allowed longer intervals between each transfusion to the fetus. The objective of one study was to reach a hematocrit value of 0.35-0.40. The authors did an intraperitoneal transfusion in fetuses of gestational age 20 weeks along with a dose of vecuronium (0.1 mg per kg of USG estimated fetal body weight) before the start of the next intravascular transfusion in order to stabilize the fetus movement for 2 h thereby reducing the chance of fetal injury during the process. Every 2 weeks interval, transfusion was undertaken for the first two sittings and then the interval period was further increased to 3-4 weeks time. Hydropic or severely anemic fetuses of the early second trimester do not tolerate intravascular transfusions compared to fetuses of older gestational time. The mean end point of the hematocrit which is often regarded as an important biomarker for hemolytic anemias in case of early second trimester fetuses is normally between 0.20 and 0.25 in case of first transfusion and 0.35 in cases of second transfusions to achieve a normal level of hematocrit function [107, 108]. Some clinicians look for a fetal hematocrit value of 0.50-0.65 via intravascular transfusions whereas in case of the combined

intraperitoneal and intravascular transfusions, a fetal hematocrit value of 0.35–0.40 is often regarded as a standard.

In 1 out of 1,000 births, fetal mortality occurs due to substantial fetomaternal hemorrhage and around 1 in 800 results in fetal morbidity. In a study of five intrauterine transfusions, three such cases reported complaints regarding decreased fetal movement due to fetal bradycardia, sinusoidal heart pattern, falling hematocrit levels due to continuous fetomaternal haemorrhage after IUT and in one case hydrops fetalis was observed via USG. However pregnancy was prolonged in one of these five cases where three received a single intrauterine transfusion and two received double intrauterine transfusion. This one case where pregnancy prolonging was reported after an intraperitoneal transfusion during the 21st week helped in reversing the condition of hydrops fetalis with a normal baby delivery at 34 weeks of gestation. However due to the low success rate where only one fetus responded to the IUT as observed in the case of five fetuses the safety and efficacy of this intervention is still questioned.

Ten percent of in utero intracranial hemorrhage has been documented as early as second trimester [109, 110]. The clinician becomes aware of this hemorrhage or fetal thrombocytopenia only after the infant is born thereby resulting in a high mortality rate at the time of birth [111]. Regarding initial evaluation of the disease it has been suggested that cordiocentesis as early as 20 weeks during the second trimester can be conducted in order to assess the degree of fetal platelet count and antigen status. However this approach of cordiocentesis is also associated with high mortality as 14 cases of fetal death caused due to fetal hemorrhage during blood sampling of this disorder have been reported [99]. Maternal platelets if transfused during this time can help in coagulation of the blood thereby stopping the loss of excess blood from the site of puncture or else previous analysis of the fetal platelet count should be undertaken before embarking on this method.

Infection of the Fetus during the First and Second Trimesters

Infections of the fetus can be vertically transmitted from the mother to the fetus which includes mycobacterium tuberculosis [112, 113], cryptococcal infections [114, 115], Pneumocystis jirovecii or PCP [116, 117], Cytomegalovirus (CMV) and toxoplasmosis gondii [118]. Other factors like the intactness of the blood placental barrier, the time of gestational period and the type of pathogen that is infecting the mother and its potency also plays an important role in transmitting the infections from the mother to the fetus thus making the fetus a patient.

Toxoplasmosis gondii is mainly acquired from uncooked raw meat and cat feces. The risk of fetus infection in the first trimester is 15 % when compared to 45 % in the second trimester. However it is seen that in mothers who get infected in the first trimester, there is 80 % transmission of the infections to the fetus in the later stages, whereas it is only 50 % if acquired by the mother during the second trimester. However infections in the mother in the later stage can cause fetal damage or still birth with damage to the CNS in a live infant. Spiramycin treatment of the mother has shown to reduce the incidence of vertical transmission by 60 % [119]. The above phenomenon might be due to the fact that in the initial stages of development the placental barrier is extremely intact thereby not allowing many pathogens to cross the blood placental barrier and enter the fetal compartment. This mechanism of placenta is in place to support the immature fetal immune system. However if the maternal infection is not treated in the initial stages then there are high chances that the pathogen might be able to cross the placental barrier when it loses its intactness due to cellular devascularization in the later trimester phases. Although the fetal immune system also gradually develops by the time it reaches its 14th week still it remains too immature to combat the virulency and pathogenicity of many infections.

Cytomegalovirus (CMV) is another common cause of in utero infections and thereby screening for this virus is extremely essential. Around 50 % of women of child bearing age remain susceptible to this virus and only 1 % can have the risk of getting affected by this virus. These 1 % women getting affected with CMV can transmit 45 % of this virus to the fetus in the initial stages of pregnancy thereby posing an immense risk to the fetus. Infections like Herpes Simplex Virus (HSV) are extremely rare in terms of in utero infections but if it is transmitted to the fetus it can cause severe damage to it at any stage. Genital herpes which usually causes cold sores has little risk of transmission to the fetus if it infects the mother in the first and second trimester [119].

Human Parovirus B 19 can affect around 2.5-38% of fetuses and inhibits fetal erythropoiesis resulting in the development of aplastic anemia, non immune hydrops and even fetal abortions [120, 121]. Maternal infection is assessed by the presence of a specific IgM antibody which appears 3-4 days after the onset of clinical symptoms and continues to persist for another 3–4 months [122]. Within this window period of 4-6 weeks time after the maternal infection, the fetus might acquire this infection which can be detected by means of USG imaging weekly. Diagnosis of the Human Parovirus infection in the fetus can be assessed by the presence of fetalis hydrops followed by severe anemia in some cases, a negative direct Coombs test, an inappropriately low reticulocyte count, elevated levels of total IgM and hepatic enzyme levels [123], presence of viral particles in the fetal ascetic fluid [124] or the presence of viral DNA collected from fetal fluids. Such fetuses undergo a serial intrauterine transfusion process at short intervals ranging between 1 and 7 days. Increase in the fetal reticulocyte count signals a recovery of the fetal bone marrow and further reduces the chances for intra vascular uterine transfusions. Enhanced intrauterine transfusions among anemic Rh positive fetuses can result in the suppression of the fetal erythropoiesis and lead to the dependence of the fetus on serial procedures in

order to maintain the hematocrit level. Non immune parovirus infections may require weeks of transfusions to correct the fetal hematocrit level to a normal one. Hepatits B virus can be acquired in utero by the fetus through maternal infection or during the course of fetal development. Klein et al. demonstrated evidence that there is a causal relationship of rubella, cytomegalovirus and toxoplasmosis infection with IUGR [125].

Primary Varicella Zoster in the first and second trimester of pregnancy may result in the birth of a fetus born with congenital malformations. This is also referred to as the 'Congenital Varicella Syndrome' which is rare affecting only 2.8% of women with chicken pox in the first 20 weeks post conception. The maximum risk of getting infected with this pathogen is however between the 12 and 20 weeks of gestational time. The fetus might develop symptoms like limb hypoplasia, ocular abnormalities, mental retardation and pre mature death. Varicella Pneumonia another sub group of this infection can be highly infectious during the third trimester of fetal development. Normally systemic antiviral medication is prescribed [126].

In case of HIV, viral transmission from mother to fetus is around 25 %. The fetus is more likely to acquire the disease if the mother has a high viremia or a low CD34 positive cell count. The risk of congenital infection in case of Rubella is also very high to the fetus in case of the first trimester. However between the 23rd and 26th gestation week the risk of the transmission of this disease decreases to around 25 %. The first 11 weeks of the trimester can produce teratogenic effects to the fetus; however, no such risks are associated with the fetus once it crosses the 16 weeks of the second trimester phase [127].

Bacterial infections like Group B streptococcal (GBS) can be one of the main life threatening infections among the new born as it has the potency to induce pneumonia infection; if acquired by the mother in the first 2 weeks of pregnancy in the first trimester it can result in the fetal hearing and vision loss too. GBS is also found in 20 % pregnant women in their vaginas during the last trimester and this has the potentiality to affect the child during birth. Most of the bacterial infections however infect the fetus in the later stages of pregnancy or post natal [127].

Experimental Treatment Options for the Fetus in Utero during the First and Second Trimester of Growth and Development

Stem cell therapy in utero of the fetus is an extremely new and a very exciting concept in treating different disorders of the fetus. It holds great promise in treating many fetal complications. The reason why stem cell transplantation in the late first and second trimester and beyond can be a very exciting opportunity is because the immunological system of the fetus is still immatured during this stage and it fails to recognize foreign antigen or cells thereby reducing the chances of abortion and other immune mediated rejections. The size of the fetus is another important criteria as on it is dependent the dosage regime [128].

In sickle cell anemia stem cell therapy is an extremely good option. Actiology of sickle cell anemia is because of the change in the shape of the erythrocytes which become crescent shaped and a decrease in their life span compared to normal blood cells. They also tend to clot small blood vessels [128]. Normally the onset of anemia is seen in the fetus during the first 4–8 weeks of gestation in the first trimester [129].

Scientists like Flake are currently investigating the potentials of blood forming stem cells to treat sickle cell anemia as the delivery of stem cells through intrauterine transfer involves minimally invasive surgery and is thereby less risky [128]. These blood stem cells can be co-cultured in the lab under in vitro conditions by taking half from the mother and half from the father. The fetus's weakened immune system especially in the first and second trimester might take up the cells thereby making it a chimeric and a tolerant fetus to future donors thus increasing the chances of accepting more stem cells from the same donors in future especially during post natal development.

Use of Umbilical Cord Blood as a Viable Substitute and Option for Intrauterine Transfusions

Umbilical cord blood has shown immense promise in treating thalassemia, anemia and other blood related disorders and malignancies. Bhattacharya et al. have shown encouraging reports of cord blood transfusion in more than 1,000 patients suffering from different ailments [130]. New concepts in this area stem from this very thought of using whole umbilical cord blood transfusion in treating fetal anemias and blood disorders. Volume of the cord blood transfused, it is conceptualized, is an important criteria for treating the fetus. Normally for neonates of extremely low birth weight, approximately 10-20 ml/kg of body weight, one or two bottles of autologus blood transfusions can be sufficient, but for the fetuses in the first and second trimester, dose is a big factor to make sure that excess of dose will not result in abortion as the fetus in the first two trimesters, is extremely prone to abortion in the presence of foreign molecules and cells [131, 132]. Umbilical cord blood transfusion is also an alternate but feasible option for treating premature deliveries where the fetuses are of the age of 27 or even 24 weeks of gestational phase. Several clinical trials have also shown the importance of umbilical cord blood transfusions obtained from both autologus and allogeneic sources [133, 134].

The procedure in doing so will involve the aid of USG guided image for proper transfusion of the fetus in utero. The inherent properties of cord blood like a higher fetal hemoglobin which is almost 50 % more than that of adult hemoglobin and a larger capacity to transport and release oxygen after transfusion due to higher oxygen dissociation curve makes cord blood an attractive option for intrauterine transfusion [135, 136]. Platelet content in cord blood is not so high. Also presence of small amounts of pro and anti-coagulant factors in the plasma of the fetal cord blood makes it an important alternative to adult blood transfusion as it helps in reducing the chances of coagulation post intra uterine fetal transfusion. This is also one of the reasons why neonate plasmas are not efficient in correcting bleeding due to the absence of coagulating factors. In case of allogeneic cord blood transfusion in adults normally no HLA matching is required apart from the standard blood group matching. It has been found in most patients that after transfusion there has been a rise of CD34 positive cells in the peripheral blood. If this is true in case of the adult immune system where it is highly developed and specialized to recognize between self and non self antigens then this same principle would be true in case of the fetus also where the immune system is naive. Umbilical cord blood can be stored for 35 days; however after 2-3 weeks of storage the physiological potassium level of the cord blood increases leading to hyperkalemia and if transfused after 3 weeks time it can cause fetal cardiac arythmias [137].

Important disadvantage of cord blood is that the volume of cord blood collected is low and therefore it requires multiple units for a minimal effective level of transfusion dose. Chances of viral transmissions are also high especially during the term time but it has been shown through various studies that the risk of HIV transmission is between 1 in 100,000 to a million [138]. Risk of transmission of Hepatitis B is 1 in 50,000. This can be attributed to the fact that Nature's most dynamic screening system, the placenta, can selectively screen at the nano-molecular level thereby allowing the selective transmission of molecules and inhibiting the passage of pathogens through its fetomaternal barrier. However, there are still chances of viruses and pathogens to move through the placental barrier especially during the near term time as the fetomaternal barrier can gradually undergo cellular de-vascularization. This risk in adult blood is not perceived in the west due to stringent screening protocols that are in place but in developing countries like India, Bangladesh, Sri Lanka, Pakistan although there are standard operating procedures in place for proper screening of adult blood many of them

are almost nonexistent. Also in the west molecular screening of the adult blood is conducted by means of PCR which is unlikely in case of developing countries due to high cost. Therefore cord blood plays an important role here as it is dynamically screened by the placenta itself. Another important factor in obtaining adult blood for transfusion purposes during emergencies especially in developing countries is that it is an extremely arduous and time consuming process. However in case of cord blood after a proper informed consent from the mother it can be easily available. Also cord blood is less thrombogenic. The erythrocytes of newborns do not express Kelly like antigens and other antigens like A and B, therefore cord blood can be less immunogenic causing no adverse event in the fetus after transfusion [129].

In case of stem cell transplantations it has been seen in mouse and lamb models that the fetus during its early stages of development has an excellent potentiality of accepting donor or foreign cells and has a unique tolerance to foreign antigens and also the capacity to transport large cell doses within a short time due to the small fetus size and also an efficient engraftment potential of the stem cells in the fetus [139, 140]. Successful haemotopietic stem cell transplantations have been observed and achieved in immunodeficient fetuses thereby suggesting the role of an immunological barrier that often exists in the early stages of pregnancy in the human fetal compartment [141]. However it was observed that allogenic responses from the 9th to 12th week started, which increased with gestational time ultimately leading to the theory of GvHD suggesting that although it might be considered that the fetus is immune resistant it can still elicit responses specific to stem cells, the mechanisms of which are still known [142]. In these cases the role of mesenchymal stem cells can be important as these groups of stem cells have excellent immune modulatory functions. Mesenchymal stem cells contain an potentials for treating adult osteogenic diseases, skeletal disease, cardiac diseases, pancreatic disorders leading to diabetes mellitus, liver regeneration, and even in some cases it has shown some potentialities to treat
neurological disorders but with inconsistent reports. However it is believed that in case of blood diseases even allogenic haemotopoietic stem cells (HSC) will be more potent and viable than mesenchymal stem cells (MSC) because of the HSC ontogeny and presence of many blood forming precursor cells [141].

Intrauterine stem cell transplantation has been performed in cases of haemoglobinopathies, osteogenesis imperfecta, immunodeficiencies but more clinical data is needed to substantiate the procedure's curative value. Tiblad and Westgren it was found that in the case of intervention of stem cells, source and type of the donor stem cells and the specific disease in question varies greatly between the first and second trimesters [143].

The same can be said in case of gene therapies where the introduction of a foreign viral vector can give rise to a cascade of immune reactions, but there can be an exception in the fetus during the first and second trimester as the fetal immunity is still in its nascent stage of development. Gene therapy can be applied by intra uterine delivery of the viral vector in the fetus to treat various genetic disorders like hemophilia A and B. It has already been shown that the previously deficient blood clotting factors in case of hemophilia tends to function normally after the introduction of the correct gene. Fetal lamb model has shown low mortality rate of 3–15 % after a minimally invasive USG imaging was conducted targeting the muscles, diaphragm, cerebral ventricles, peritoneum and liver via the intra uterine system. Therefore gene therapy can give a unique solution to many genetic diseases like congenital blindness, fetal cystic fibrosis prevailing in the early fetus. This mode of treatment can also help in stopping the irreversible damage to organs before or shortly after delivery among neonates [141].

Surgical Interventions of the Fetus during 1st and 2nd Trimester

Interventions in the fetus are extremely risky procedures as it can cause harm to both the mother and the fetus. A simple process like amniocentesis itself is associated with 0.5–1.4 % fetal mortality [144]. Open fetal surgical procedures often pose a greater risk of uterine dehiscence and rupture having a direct negative implication on the current and future pregnancy prospective.

Any rectification of the fetus in terms of anatomy and physiology can be conducted by means of two types of interventions: open fetal surgery and intra uterine therapy. Open fetal surgery is normally preferred during the second trimester whereas intra uterine therapies which include blood transfusion and stem cell therapies can be conducted between the late first trimester and early second trimester. Open fetal surgery itself is also a very risky procedure [145]. In 1990 under the guidance of Dr. Michael Harrison and his team at UCSF the first successful intervention for congestive diaphragmatic hernia of the fetus was performed [146]. This was soon followed by a prospective trial comparing open fetal surgical procedures to post natal repair in a sample size of seven patients in the case of an isolated left sided chronic diaphragmatic hernia with a significant displacement in the ipsilateral lung and intrathoracic stomach region. However the trial failed to show any prospective difference in survival between the two groups. Also the intervention in the fetal group had a much longer duration of hospitalization compared to the second group where this surgical intervention was conducted during the post natal stage of development [147]. Therefore clinicians target another alternative repair form known as the fetal tracheal occlusion intervention (FeTO) since open fetal repair has been rejected by many surgeons in favor of tracheal occlusion. However FeTO also had a mixed success as it has lost its momentum among North American clinicians following the UCSF trial. However in Europe the trend is quite different as more and more tracheal occlusions of the fetus are being treated by the FeTO way which has shown high survival rates among CDH fetuses reported to be at less than 30 % for left sided CDH and less than 45 % for right sided CDH [148, 149].

It is believed that tumors that gets manifested during the first 3 months of development normally get termed as congenital tumors which progress with increase in gestation time [150, 151]. The number of fetal neuroblastic nodules increase during the 17–20th week of gestation and are indistinguishable from the neuroblastoma in situ [152, 153]. Therefore the presence of neuroblastoma of the fetus in autopsy series is significantly higher than the actual clinical manifestation of the disease and the precise way to actually identify this condition is by the help of USG guided imaging. Surgery in such cases remains the standard operating procedure with a conservative detection and diagnosis of the neuroblastomas [154].

Hemangioblastomas, another class of solid tumors can be detected in a 16 weeks fetus and might have different sonographic appearances and can be either hypoechoic or hyperechoic or maybe even of mixed echogenicity. Fetal MRI or a color Doppler evaluation can be performed in such cases in the second trimester; these will show the increase of flow along with a flow void in the mass. Also in some cases fetuses may develop consumptive coagulopathy known as the Kasabach-Merritt sequence which if not detected in the early trimester can result in congestive heart failure of the fetus [155].

Sacrococcygeal tumors (SCT) are a rare form of neoplasms which are being diagnosed by USG in utero frequently and are often related to intra uterine fetal demise (IUFD). SCT's if not treated from the first to second trimester can result in abnormal size of the fetus resulting in cardiac failure and non-immune hydrops with vascular shunting. Tumors can rarely cause internal and external hemorrhages [156].

Spina bifida or myelomeningocele that occurs in almost 5–10 pregnancies in 10,000 women in the United States is a non-lethal neural tube defect. It occurs mostly during the third or fourth week of the first trimester however with improved peri natal screening the incidence rate has been reduced to 3.4 per 10,000 live births currently [157, 158].

Spina bifida involves the failure of the development of the neural plate and thereby fails to close along its length often leading to an open spinal canal condition with exposure of meninges and other natural elements of the brain. The aetiology of spina bifida can be due to the teratogenic effects of drugs like carbamazepine, pregestational diabetes or valproic acid exposure. Usually maternal serum alpha feto protein measurement and USG screening is conducted during the second trimester to assess the condition [159].

In order to treat spina bifida in the fetus surgical resection is required which can be performed through maternal hysterotomy and laparotomy. Normally open repair is required in cases of large lesions. However, risks are involved to both the mother and the fetus. Open fetal surgery in most of the cases results in pre term labour [160, 161].

Common cardiac lesions like intrapericardial teratomas and rhabdomyomas are not easy to detect in the first trimester mainly because of the immature anatomy of the fetus. Other cardiac lesions include teratomas, fibromas, myxoma and vascular malformations [162, 163]. The incidence of cardiac neoplasms are extremely less with an approximately 0.1 % of the fetus getting affected [164, 165]. Cardiac lesions of the fetus are usually detectable after the first 22 weeks of gestation by means of prenatal USG imaging techniques which involves the imaging of the pericardium, myocardium, cardiac valves and/or the major pericardial blood vessels [166–168]. The survival rates of cardiac neoplasms are quite low with 57 % chances of IUFD. The most common type of cardiac neoplasm observed in the fetal heart is the rhabdomyoma [166].

There can be a chance of sound and correct diagnosis of such cardiac lesions during the second trimester as that is the time when the heart of the fetus matures [169]. Also nuchal translucency, a diagnosis meant for Down's syndrome, can be prescribed in case of fetal heart diseases even if the fetus has normal chromosomes [170, 171].

In case of CHD, early first trimester fetal echocardiography conducted trans vaginally or trans abdominally cannot have the same degree of detection efficiency as in the case of second trimester for CHD as many forms of fetal CHD develop only during the late first trimester or during the course of the second trimester [172–174]. However it is always logical to start monitoring the fetus for any anomaly in the first trimester itself.

Some surgeons have opined that due to the presence of non-immune hydrops open fetal cardiac surgical resection through maternal hysterotomy and fetal median hysterotomy can be conducted. However till date there have been no reported cases of open fetal resection of a cardiac neoplasm.

Another congenital anomaly of the fetal cardiac system is Valvar aortic stenosis which represents the development of a thick valve during the second trimester. Severe cases of valvar aortic stenosis if present during the mid second trimester can potentially affect the peri natal development of the fetus's left ventricle resulting in a form of dilated cardiomyopathy [175, 176].

Other conditions like obstruction of the fetus and trachea can result in cardiac failure of the fetus along with craniofacial defects and cranial nerve injuries. Highly vascularized lesions can lead to fetal cardiac arrest too. Majority of the fetal lesions are comprised of cervical teratoma, cystic hygroma and other vascular malformations [177]. Rare defects of the neck like thymic cysts or congenital neuroblastomas can be also observed in the fetus during its initial stages of development.

Incomplete closure of neura tube defects can give rise to myelomeningocele resulting in an open spinal canal with exposed neural tissues and leakage of the cerebrospinal fluid in the fetal compartment. The occurrence of such congenital malformation is 1 in every 2,900 live births in the US thus making it one of the most common congenital birth defects. The failure to close the neural tube often results in the exposure of the fetal neural tissue to the external environment like the amniotic fluid resulting in intra uterine traumas during the different phases of development. Other reasons can be also attributed to this neural tube defect like the abnormality of the neural placode, exposure to drugs, toxins and genetic abnormalities [166].

Twin to twin transfusion syndrome also denoted as TTTS is observed in around 10–15 % of all monochromic pregnancies and is often related to morbidity and multiple anastomoses in the fetal circulation. This syndrome is characterized by the presence of polyhydramnios in the donor twin and oligohydramnios in the sac of the recipient twin during a monochromic pregnancy. Patho physiological conditions like early myocardiac dysfunction in the recipient twin, placental insufficiency in the donor to recipient twin are noticed [178]. TTTS if not treated can result in a high mortality rate of around 80 % and risk of disability in 20 % [179]. USG imaging is used in detecting this condition in the early trimesters. Fetoscopy laser ablation techniques have been shown to be effective over the standard procedure of amnio-reduction [180] with a median survival rate of 60-70 % before 26 weeks of gestation. The risk of severe handicap is however 5 %. This rate of overall long term survival is similar in case of amnio reduction method as well; but the major drawback of amnio reduction is that the rate of handicap is extremely high at 25 %. However another study revealed less satisfactory results for laser ablation therapy followed by an amnion reduction [178].

Discussion

In the first 8–10 days post conception, the zygote undergoes transformation into a pro-embryo and with subsequent cleavage forms the morula and the blastula stage. This early stage represents the 'All or no phase" which itself means if the embryo survives the initial toxic effect of drugs it will undergo further development and if not then it will simply lose its embryonic viability. Therefore the question apparently is whether teratogenic drugs can induce congenital anomalies in the first week of gestation itself or do they have a more profound effect at the later stages of fetal development? [181]. One paper suggests that as organogenesis is initiated from the 29th day and continues till the 70th day, this is when the DNA activation of the stem cells takes place resulting proliferation [181]. in differentiation and Therefore even if a teratogenic drug is consumed during this nascent and early period of pre embryonal development it might not have severe teratogenic implications on the embryo. Therefore considering the first trimester as a whole is a seriously challenged concept and it is only in the second and third month when the fate of the embryo is already decided, that the fetus is highly susceptible to congenital abnormalities [181].

The use of stem cell therapies in utero is also a very exciting emerging field. Attempts have already been undertaken to treat diseases like sickle cell anemia with in utero transplantation of hematopoietic stem cells [182]. It has proved to be successful in case of allogeneic donors also. Animal studies have revealed the involvement of the immune barrier in the fetus to be stronger than it was originally supposed [183].

An attractive source of stem cells is the amniotic fluid [184]. Amniotic fluid consists of fetal origin stem cells, mesenchymal stem cells and can be easily differentiated into a variety of different lineages like myogenic, adipogenic, hepatic, neuronal and endothelial [185, 186]. Also amniotic fluid obtained from a diseased fetus through a routine clinical amniocenteses technique during the 15th week of gestation can be a viable autologus transplantation source for the ailing fetus [187, 188]. Hematopoietic stem cells harvested from umbilical cord blood collected from a first trimester fetus can be transplanted back to the fetus itself by USG guided blood sampling [189]; thereby this can act as autologus cord blood derived hematopoietic stem cell transplantation. This can also be an alternative approach as it will reduce the risk of allogeneic reaction of the fetus in its later stages of growth and development. It has been seen in mice and sheep models that amniotic fluid derived stem cells can be easily transplanted without losing any of their general characteristics [190–192].

The human immune system starts to develop between the 12th and 14th weeks of gestation and during this time there is an increase in the number of circulating T Lymphocytes in the fetus [196].

As amniotic fluid derived stem cells are fetal in origin they can show resistance against Natural Killer (NK) cells thereby playing an important role in immune suppression of the host [193]. This immune modulatory property of amniotic fluid derived stem cells is quite similar to that of mesenchymal stem cells which are regarded as one of the best immune compatible cells for transplantation. Amniotic fluid stem cells can also release high levels of cytokine factors [194]. After collection of amniotic fluid from the baby in the second trimester these cells can be easily quantified and harvested in the lab as they have a superior average doubling time of 25–38 h compared to bone marrow derived MSC's which have a doubling time of 30–90 h respectively [195]. Further these stem cells derived from the amniotic fluid have a higher clonogenic potential and can express both pluripotent and mesenchymal stem cell markers SSEA-4, Oct-4 [186].

Therefore the advantages of using amniotic fluid derived stem cells are quite paramount. After collecting amniotic fluid during the 15th week of gestation and then harvesting these stem cells and re-transplanting them back to the fetus would not result in immune rejection in the later stages of fetal development as these cells, like mesenchymal stem cells, have the ability to evade immune rejection thereby increasing the chances of enhanced engraftment in the fetus. The property of autologus amniotic membrane derived stem cell to be transplanted back to a diseased fetus in utero, can be a viable approach, limiting the number of invasive and open fetal surgeries often associated with high risk to both the mother and the fetus.

Gene therapies, like stem cell therapy, also have limitations in their approach. Gene therapy is not possible in case of thalassemia and sickle cell anemia due to the presence of the large globin gene and the globin promoter whose insertion in vectors can be technically very tough [184]. A study conducted on sheep models where autologusly derived fetal liver stem cells were isolated from a first trimester pre-immune fetal sheep with the help of USG guidance and were labeled with a PLH26 contrast and transplanted intra peritoneally into both autologus and allogeneic fetal recipients. Observations revealed equal engraftment in both the allogeneic and the autologus cases but the autologus fetal cell loss was observed to be 73 %, much higher than that of the allogeneic which was only 29 % [197].

Treatment and diagnosis of Severe combined Immunodeficiency (SCID) is feasible only in cases of neonates. However with rapid advances in the field of in utero 2D and 3D imaging and the availability of different non invasive techniques, assessment of SCID can be confirmed even in the first trimester. Allogeneic bone marrow stem cell is the only transplantation procedure available till date through IUSCT technique [184].

In utero bone marrow derived mesenchymal stem cells have also shown long term engraftment potentialities in sheep models [198]. In case of spina bifida the therapeutic potentialities for combining surgical repair and transplantation have been shown to be effective in a rat model [199].

Apart from the scientific concepts of the fetus as a patient it is also imperative to consider the ethical view in treating the fetus as a patient. Questions arise like when does an individual human life start, rather, when does biological life begin? [200, 201]. What are its biological, ethical and religious aspects based on which we can consider the fetus as a viable or a non viable living organism? These are topics that attract a tangential amount of ideas and views. In the pre embryonic phase the zygote divides to form an embryo. During this stage of development the cells are not yet differentiated in terms of their fate for specific cells or organs [200]. By the time they reach the 14th day post conception via the formation of the morula and blastula, placentation starts and that marks the beginning of the actual embryonic phase which is accompanied along with organogenesis and morphogenesis. Therefore, it may be possible that individual human life begins with the emergence of the embryonal stage and distinct human life or a biological life starts during the initial pre-embryonic phase [200].

Any abnormalities observed during the first two trimesters of growth and development of the fetus can be rectified by means of several therapies and techniques which have already been discussed. However moral and ethical dilemmas especially in the case of open fetal surgeries still persist. The concept of open fetal surgery is a very risky option as it increases the chance of fetal and maternal mortality. Open fetus interventions become extremely necessary in cases of emergencies or for diseases that require an early intervention or might carry the risks of abortion or congenital abnormality in the later stages of gestational period. In a normal adult surgery, administration of anesthetics is not a major concern both medically and ethically, however it becomes a major issue in case of an open fetal surgery. The sole reason of anesthetics is to numb and alleviate the patients from experiencing pain while undergoing the procedure. But this concept of pain is extremely debatable in the fetus. There seems to exist different opinions among surgeons and clinicians regarding when the fetus starts to feel pain and sensation. It is suggested by many that from the 20th week onwards the fetus can start to sense some amount of pain and others claim that the receptors of pain come into prominence between the 8th and 12th weeks and the fetus can feel pain between the 26th and 28th week of its development. It is a well known fact that application of anesthetics in cases of pregnant mothers undergoing surgeries in the first and second trimester can be fatal to the fetus as there are high chances that these molecules might be able to cross the placental barrier and enter the fetal compartment. Therefore the question of moral responsibility and ethics comes into the picture when a fetus in its first or second trimester is given an anesthetic during a surgical procedure. In case of abortion the main aim is fetal destruction but in cases of surgical repair the application of anesthetics becomes highly debatable as questions of direct toxicity to the fetus arise. On the other hand if anesthetics are not administered then there are high chances that the fetus might feel pain during surgery as the feeling of fetal pain is still a debatable topic.

Conclusion

Therefore the concept of the fetus as a patient involves an extremely dynamic and diversified field of medicine and surgery which has opened doors for many new avenues and exciting ventures in the field of experimental medicine, particularly regenerative medicine, stem cell and gene therapy. The ultimate aim of treating the fetus as a patient is to maximize the chances of better fetal health and to minimize the risks associated with fetal disease and abnormalities. The advent of pioneer biomedical imaging and instrumentation techniques will further enhance the field of pre natal diagnosis and in utero surgery.

References

- Mohammed NB, Bui TH. Fetus as a patient: art and science of fetal medicine. Pak Med Assoc. 2010;60(6):417–8.
- Brahmaiah U, Ratha C. Fetus as a patient. Fetal medicine. JIACM. 2012;13(3):223.
- Deka D, Sharma N, Dadhwal V, Mittal S. Successful application of middle cerebral artery peak systolic velocity to time intrauterine transfusions in Rh isoimmunised fetus. J Obstet Gynecol India. 2006;56:534–6.
- Montenegro N, Beires J, Pereira L. Reverse enddiastolic umbilical artery flow at 11 weeks gestation. Ultrasound Obstet Gynecol. 1995;5:141–2.
- Bonilla-Musoles F. Ecografía Vaginal; Doppler y Tridimensión. Madrid: Editorial Panamericana; 2000.
- Kurjak A, Kupesic S, Banovic I. The study of morphology and circulation of early embryo by 3D ultrasound and power Doppler. J Perinat Med. 1999;27:145–57.
- Kupesic S, Bekavac I, Bjelos D, et al. Assessment of endometrial receptivity by transvaginal color Doppler and three-dimensional power Doppler ultrasonography in patients undergoing in vitro fertilization procedures. Ultrasound Med. 2001;20:125–34.
- Jauniaux E, Burton GJ, Moscoso GJ, et al. Development of the early placenta—morphometric study. Placenta. 1992;12:269–76.
- Sterzik K, Grab D, Sasse V, Hutter W, Rosenbusch B, Terinde R. Doppler sonographic findings and their correlation with implantation and in an in-vitro fertilization program. Fertil Steril. 1989;52:825–8.
- Campbell S, Pearce JMF. Ultrasound in obstetrics and gynaecology. In: McDonald RR, editor. Scientific basis of obstetrics and gynaecology. Edinburgh: Churchill Livingstone; 1985. p. 304–9.
- Arabin B, van Eyck J, Wisser J, Versmold H, Weizel HK. Fetal behavior in multiple pregnancy: methodologic, clinical and scientific aspects. Gegurtshilfe Frauenheilkd. 1991;51:86–75.
- Nijhuis JG, editor. Fetal behavior, developmental and perinatal aspects. Oxford: Oxford University Press; 1992. p. 3–17. de Vries JIP. The first trimester.
- Rosatelli MC, Saba L. Prenatal diagnosis of b-thalassaemias and haemoglobinopathies. Mediterr J Hematol Infect Dis. 2009;1(1). http://www.ncbi. nlm.nih.gov/pmc/articles/PMC3033155/
- 14. De Vore GR. The use of color Doppler imaging to examine the fetal heart. Normal and pathologic anatomy. In: Jaffe R, editor. Color Doppler imaging

in obstetrics and gynecology. New York: McGraw-Hill; 1992. p. 121–54.

- Bonilla-Musoles F, Raga F, Ballester MJ, et al. Early detection of embryonic malformations by transvaginal and color Doppler sonography. J Ultrasound Med. 1994;13:347–55.
- 16. Chasen ST, Skupski DW, McCullough LB, Chervenak FA. Ethical dimensions of nuchal translucency screening. In: Carrera JM, Chervenak FA, Kurjak A, editors. Controversies in perinatal medicine, studies on the fetus as a patient. New York: Parthenon Publishing Group; 2003. p. 397.
- Carrera JM, Chervenak FA, Kurjak A, editors. Controversies in perinatal medicine, studies on the fetus as a patient. New York: Parthenon Publishing Group; 2003.
- 18. Bonilla-Musoles F, Machado L, Osborne N, Raga F, Lima-Couy I, Bonilla Jr F, Torres F. Early diagnosis of congenital anomalies 3. Alterations of fetal arterial and venous flow. In: Carrera JM, Chervenak FA, Kurjak A, editors. Controversies in perinatal medicine, studies on the fetus as a patient. New York: Parthenon Publishing Group; 2003. p. 114–38.
- Bonilla-Musoles F, Machado L, Osborne N, Raga F, Lima-Couy I, Bonilla Jr F, Torres F. Early diagnosis of congenital anomalies 2. Thoracic and abdominal malformations. In: Carrera JM, Chervenak FA, Kurjak A, editors. Controversies in perinatal medicine, studies on the fetus as a patient. New York: Parthenon Publishing Group; 2003. p. 104.
- Blaas HG, Eik-Nes S, Vainio T, Vogt IC. Alobar holoprosencephaly at 9 weeks' gestational age visualized by two- and three-dimensional ultrasound. Ultrasound Obstet Gynecol. New York: Parthenon Publishing Group, London; 2000;16:62–5.
- 21. Bonilla-Musoles F, Machado L, Osborne N, Raga F, Lima-Couy I, Bonilla Jr F, Machado F. Early diagnosis of congenital anomalies 4. Adnexal markers of aneuploidy. In: Carrera JM, Chervenak FA, Kurjak A, editors. Controversies in perinatal medicine, studies on the fetus as a patient. New York: Parthenon Publishing Group; 2003. p. 137.
- Matias A, Gomes C, Flack N, Montenegro N, Nicolaides KH. Screening for chromosomal abnormalities at 10–14 weeks: the role of ductus venosus blood flow. Ultrasound Obstet Gynecol. 1998;12:380–4.
- Borrell A, Antolín E, Costa D, Farré MT, Martínez JM, Fortuny A. Abnormal ductus venosus blood flow in trisomy 21 fetuses during early pregnancy. Am J Obstet Gynecol. 1998;179:1612–7.
- 24. Antolín E, Comas C, Torrents M, et al. The role of ductus venosus blood flow assessment in screening for chromosomal abnormalities at 10–16 weeks of gestation. Ultrasound Obstet Gynecol. 2001;17:295–300.
- Comas C, Antolín E, Torrents M, et al. Early screening for chromosomal abnormalities: new strategies combining biochemical, sonographic and Doppler parameters. Prenat Neonat Med. 2001;6:95–102.

- Comas C, Torrents M, Antolín E, et al. Firsttrimester sonographic markers for chromosomal abnormalities. Ultrasound Rev Obstet Gynecol. 2002;2:213–20.
- Agrawal SK, Singh P, Kumar P. MRI in antenatal and perinatal congenital anomalies. J Neonat 2008;22(4):223–8.
- Yasuyuki Y, Tomohico N, Yasuko A, et al. MR imaging of the fetus by a HASTE sequence. Am J Roentgenol. 1997;168:513–9.
- Cooper WO, Hernandez-Diaz S, Arbogast PG, et al. Major congenital malformations after firsttrimester exposure to ACE inhibitors. N Engl J Med. 2006;354:2443–51.
- Kennedy D. Classifying drugs in pregnancy. Aust Prescr. 2014;37:38–40.
- Milojkovic D, Apperley JF. How I treat leukemia during pregnancy. American Society of Hematology, Submitted 17 Aug 2013; Accepted 4 Nov 2013. Prepublished online as Blood First Edition paper, 22 Nov 2013.
- Douglas Wilson R. Principles of human teratology: drug, chemical, and infectious exposure. J Obstet Gynaecol Can. 2007;29(11):911–7. Philadelphia.
- 33. Gunatilake R, Patil AS. Drugs in pregnancy. Merck Manual Online for Health Care Professionals. 2013. www.merckmanuals.com/professional/gynecology_and_obstetrics/drugs_in_pregnancy/drugs_in_ pregnancy.html?qt=drugspregnancy&alt=sh. See more at: http://www.uspharmacist.com/content/ c/43164/#sthash.xQjRtkWc.dpuf.
- 34. Holmes LB, Harvey EA, Coull BA, Huntington KB, Khoshbin S, Hayes AM, et al. The teratogenicity of anticonvulsant drugs. N Engl J Med. 2001;344:1132–8.
- 35. Muaed Jamal Alomar. Factors affecting the development of adverse drug reactions (Review article) Faculty of Pharmacy and Health Sciences, Clinical Pharmacy Department, United Arab Emirates. Available online 24 Feb 2013.
- Meloni P, D'Angeli I, Piazze J, Cerekya A, Simari T, Pala A. First trimester PAPP-A levels associated with early prediction of pregnancy induced hypertension. Hypertens Pregnancy. 2009;28(4):361–8.
- World Health Organization. Guidelines for the programmatic management of drug-resistant tuberculosis: emergency update 2008. Geneva: World Health Organization, Stop TB Department; 2008.
- 38. Blumberg HM, Burman WJ, Chaisson RE, Daley CL, Etkind SC, Friedman LN, et al. American Thoracic Society, Centers for Disease Control and Prevention/Infectious Diseases Society of America: treatment of tuberculosis. Am J Respir Crit Care Med. 2003;167(4):603–62.
- Ormerod P. Tuberculosis in pregnancy and the puerperium. Thorax. 2001;56(6):494–9.
- Shin S, Guerra D, Rich M, Seung KJ, Mukherjee J, Joseph K, et al. Treatment of multidrug-resistant tuberculosis during pregnancy: a report of 7 cases. Clin Infect Dis. 2003;36(8):996–1003.

- World Health Organization. The treatment of tuberculosis guidelines. Geneva: World Health Organization; 2010.
- 42. Cancer during pregnancy, publications & resources, Annual Report. Scientific Review Books SOGC News Reports. Cancer during Pregnancy. http:// sogc.org/publications/cancerduringpregnancy/2/3.
- Cardonick E, Iacobucci A. Use of chemotherapy during human pregnancy. Lancet Oncol. 2004;5(5):283–91.
- 44. Amant F. Safety of chemotherapy in pregnancy. Clin Adv Hematol Oncol. 2012;10(4):258–9.
- 45. Matalon ST, Ornoy A, Fishman A, Drucker L, Lishner M. The effect of 6-mercaptopurine on early human placental explants. Hum Reprod. 2005;20(5):1390–7.
- 46. Abdel-Hady e-S, Hemida RA, Gamal A, El-Zafarany M, Toson E, El-Bayoumi MA. Cancer during pregnancy: perinatal outcome after in utero exposure to chemotherapy. Arch Gynecol Obstet. 2012;286(2):283–6.
- Cassina M, Salviati L, Di Gianantonio E, Clementi M. Genetic susceptibility to teratogens: state of the art. Reprod Toxicol. 2012;34(2):186–91.
- Norton ME. Teratogen update: fetal effects of indomethacin. Administration during pregnancy. Teratology. 1997;56:282–92.
- Traeger A, Noschel H, Zaumseil J. Pharmacokinetics of indomethacin in pregnancy and parturient women and in their newborn infants. Zentralbl Gynakol. 1973;95:635–41.
- Moise Jr KJ, Ou C-N, Kirshon B, Cano LE, Rognerud C, Carpenter Jr RJ. Placental transfer of indomethacin in the human pregnancy. Am J Obstet Gynecol. 1990;162:549–54.
- Klein KL, Scott WJ, Clark KE, Wilson JG. Indomethacin- placental transfer, cytotoxicity, and teratology in the rat. Am J Obstet Gynecol. 1981;141:448–52.
- Aselton PA, Jick H, Milunsky A, Hunter JR, Stergachis A. First-trimester drug use and congenital disorders. Obstet Gynecol. 1985;65:451–5.
- Katz Z, Lancet M, Borenstein R, Chemke J. Absence of teratogenicity of indomethacin in ovarian hyperstimulation syndrome. Int J Fertil. 1984;29:186–8.
- Briggs GG, Freeman RK, Yaffe SJ, editors. Drugs in pregnancy and lactation. Baltimore: Williams & Wilkins; 1994. p. 443–52.
- Hallak M, Reiter AA, Ayres NA, Moise KJ. Indomethacin for preterm labor: fetal toxicity in a dizygotic twin gestation. Obstet Gynecol. 1991;78:911–3.
- Moise Jr KJ, Huhta JC, Sharif DS, Ou C-N, Kirshon B, Wasserstrum N, Cano L. Indomethacin in the treatment of premature labor: effects on the fetal ductus arteriosus. N Engl J Med. 1988;319:327–31.
- 57. Hickok DE, Hollenbach KA, Reilley SF, Nyberg DA. The association between decreased amniotic fluid volume and treatment with nonsteroidal antiinflammatory agents for preterm labor. Am J Obstet Gynecol. 1989;160:1525–31.

- Bivins HA, Newman RB, Fyfe DA, Campbell BA, Stramm SL. Randomized trial of oral indomethacin and terbulatine sulfate for the long-term suppression of preterm labor. Am J Obstet Gynecol. 1993;169:1065–70.
- 59. Recommendations and guidelines for peri natal medicine, an initiative of World Association of peri natal Medicine, World Association of Perinatal Medicine (Wapm) & Matres Mundi International. Barcelona; Editor-in-Chief: José M. Carrera, 2007. http://www. wapm.info/portals/0/recommendations_perinatal. pdf.
- 60. The Antiretroviral Pregnancy Registry. Kendle Media. http://apregistry.com/index.htm. HIV positive pregnant women and antiretroviral treatment. From MicrobeWiki, the student edited microbiology resource. 2012.
- 61. HIV Programme, WHO. Antiretroviral Drugs for treating pregnant women and preventing HIV infection in infants: Towards Universal access. Recommendations for a public health approach, 2006 version. Geneva, Switzerland. http://www. who.int/hiv/pub/guidelines/pmtctguidelines3.pdf.
- Mandelbrot L, et al. Case report: nucleoside analogue-induced lactic acidosis in the third trimester of pregnancy. AIDS. 2003;17(2):272–3.
- 63. Sarner L, Fakoya A. Acute onset lactic acidosis and pancreatitis in the third trimester of pregnancy in HIV-1 positive women taking antiretroviral medication. Sex Transm Infect. 2002;78(1):58–9.
- 64. Antiretroviral Pregnancy Registry Steering Committee. Antiretroviral Pregnancy Registry international interim report for 1 January 1989 – 31 July 2005. Wilmington, Registry Coordinating Center, 2005 (http://www. APRegistry.com. Accessed 13 Jul 2006.
- Mandelbrot L, et al. Lamivudine-zidovudine combination for prevention of maternal-infant transmission of HIV-1. JAMA. 2001;285(16):2083–93.
- 66. Dwyer B, Lyell DJ. Hypertensive disorders of pregnancy. Chapter 11. In: Stevenson DK, Benitz WE, Sunshine P, Hintz SR, Druzin ML, editors. Fetal and neonatal brain injury. 4th ed. New York, USA: Cambridge University Press; 2009. p. 146.
- Riera C. Observations on 'seroepidemiology study of Leishmania infantum infection in Castilla-Leon, Spain'. Am J Trop Med Hyg. 2005;73:231.
- Russo R, Nigro LC, Minniti S, et al. Visceral leishmaniasis in HIV infected patients: treatment with high dose liposomal amphotericin B (AmBisome). J Infect. 1996;32:133–7.
- 69. Laguna F, Videla S, Jimenez-Mejias ME, et al. Amphotericin B lipid complex versus meglumine antimoniate in the treatment of visceral leishmaniasis in patients infected with HIV: a randomized pilot study. J Antimicrob Chemother. 2003;52:464–8. 37. Bern C, Adler-Moore J, Berenguer J, et al. Liposomal amphotericin B for the treatment of visceral leishmaniasis. Clin Infect Dis. 2006;43:917–24.
- Pintado V, Martin-Rabadan P, Rivera ML, Moreno S, Bonza E. Visceral leishmaniasis in human immu-

nodeficiency virus (HIV)-infected and non-HIVinfected patients. A comparative study. Medicine (Baltimore). 2001;80:54–73.

- Laguna F. Treatment of leishmaniasis in HIVpositive patients. Ann Trop Med Parasitol. 2003;97 Suppl 1:135–42.
- Perez-Molina JA, Lopez-Velez R, Montilla P, Guerrero A. Pentamidine isethionate as secondary prophylaxis against visceral leishmaniasis in HIVpositive patients. AIDS. 1996;10:237–8.
- Pappas PG, Pottage JC, Powderly WG, et al. Blastomycosis in patients with the acquired immunodeficiencysyndrome. Ann Intern Med. 1992;116:847–53.
- Ampel NM. Coccidioidomycosis in persons infected with HIV type 1. Clin Infect Dis. 2005;41:1174–8.
- Potasman I, Beny A, Seligmann H. Neuropsychiatric problems in 2,500 long-term young travelers to the tropics. J Travel Med. 2000;7:5–9.
- 76. Van Riemsdijk MM, Sturkenboom MC, Ditters JM, et al. Low body mass index is associated with an increased risk of neuropsychiatric adverse events and concentration impairment in women on mefloquine. Br J Clin Pharmacol. 2004;57:506–12.
- Van Luin M, Van der Ende ME, Richter C, et al. Lower atovaquone/proguanil concentrations in patients taking efavirenz, lopinavir/ritonavir or atazanavir/ritonavir. AIDS. 2010;24:1223–6.
- Bax M, Tydeman C, Flodmark O. Clinical and MRI correlates of cerebral palsy: the European Cerebral Palsy Study. JAMA. 2006;296:1602–8.
- Wu YW, Croen LA, Shah SJ, et al. Cerebral palsy in a term population: risk factors and neuroimaging findings. Pediatrics. 2006;118:690–7.
- Evain-Brion D. Hormonal regulation of fetal growth. Horm Res. 1994;42:207–14.
- Woods KA, Camacho-Hubner C, Savagex MO, et al. Intrauterine growth retardation and postnatal growth failure associated with deletion of the insulinlike growth factor I gene. N Engl J Med. 1996; 335:1363–7.
- Cornblath M, Schwartz R. Disorders of carbohydrate metabolism in infancy. 3rd ed. Boston: Blackwell; 1991.
- Lemons JA, Ridenour R, Orsini EN. Congenital absence of the pancreas and intrauterine growth retardation. Pediatrics. 1979;64:255–7.
- Limesand SW, Jensen J, Hutton JC, et al. Diminished beta-cell replication contributes to reduced beta-cell mass in fetal sheep with intrauterine growth restriction. Am J Physiol Regul Integr Comp Physiol. 2005;288:R1297–305.
- 85. Philip AGS, Stevenson DK, Hay Jr WW. Intrauterine growth restriction. Chapter 7. In: Stevenson DK, Benitz WE, Philip S, Hintz SR, Druzin ML, editors. Fetal and neonatal brain injury. 4th ed. Cambridge: Cambridge University Press; 2009. p. 92.
- 86. Gluckman PD, Gunn AJ, Wray A, et al. Congenital idiopathic growth hormone deficiency associated with prenatal and early postnatal growth failure. The International Board of the Kabi

Pharmacia International Growth Study. J Pediatr. 1992;121:920–3.

- DiGiacomo JE, Hay WW. Fetal glucose metabolism and oxygen consumption during sustained hypoglycemia. Metabolism. 1990;39:193–202. 40.
- O'Callaghan MJ, Harvey JM, Tudehope DI, et al. Aetiology and classification of small for gestational age infants. J Paediatr Child Health. 1997;33:213–8.
- Koenig JM, Christensen RD. Incidence, neutrophil kinetics, and natural history of neonatal neutropenia associated with maternal hypertension. N Engl J Med. 1989;321:557–62.
- Kennedy A. Fetal ultrasound. Curr Probl Diagn Radiol. 2000;29:109–40.
- Lumey LH. Decreased birthweights in infants after maternal in utero exposure to the Dutch famine of 1944–1945. Paediatr Perinat Epidemiol. 1992;6:240–53.
- 92. Sorem KA, Druzin ML. Maternal diseases that affect fetal development. In: Stevenson DK, Benitz WE, Sunshine P, editors. Fetal and neonatal brain injury: mechanisms, management, and risks of practice. 3rd ed. Cambridge: Cambridge University Press; 2003. p. 191–211.
- Moore TR. Diabetes in pregnancy. In: Creasy RK, Resnik R, Iams JD, editors. Maternal–fetal medicine: principles and practice. 5th ed. Philadelphia: Saunders; 2004. p. 1023–61.
- 94. Gabbay-Benziv R, Reece EA, Wang F, Yang P. Birth defects in pregestational diabetes: Defect range, glycemic threshold and pathogenesis. World J Diabetes. 2015;6(3):481–8. doi: 10.4239/wjd.v6.i3.481.
- 95. Dwyer B, Druzin ML. Maternal diseases that affect fetal development. Chapter 8. In: Stevenson DK, Benitz WE, Philip S, Hintz SR, Druzin ML, editors. Fetal and neonatal brain injury. 4th ed. New York, USA: Cambridge University Press; 2009. p. 113.
- 96. Hahn JS, Lemire RJ. Specific conditions associated with fetal and neonatal brain injury congenital malformations of the brain. Chapter22. In: Stevenson DK, Benitz WE, Philip S, Hintz SR, Druzin ML, editors. Fetal and neonatal brain injury. 4th ed. New York, USA: Cambridge University Press USA; 2009. p. 288.
- Valenti C. Antenatal detection of hemoglobinopathies-a preliminary report. Am J Obstet Gynecol. 1973;115:851–3.
- Rodeck CH, Nicolaides KH. Fetoscopy and fetal tissue sampling. Br Med Bull. 1983;39:332–7.
- Moise Jr KJ. Intrauterine transfusion with red cells and platelets. West J Med. 1993;159:318–24. In Fetal Medicine [Special Issue].
- Forestier F, Daffos F, Catherine N, Renard M, Andreus JP. Development of hematopoiesis in normal fetal blood. Blood. 1991;77:2360–3.
- Leduc L, Moise KJ, Carpenter RJ, Cano LE. Fetoplacental blood volume estimation in pregnancies with Rh alloimmunization. Fetal Diagn Ther. 1990;5:138–46.

- Nicolaides KH, Thilaganathan B, Mibashan RS. Cordocentesis in the investigation of fetal erythropoiesis. Am J Obstet Gynecol. 1989;161:1197–200.
- Liley AW. Intrauterine transfusion of foetus in haemolytic disease. Br Med J. 1963;2:1107–9.
- 104. Rodeck CH, Kemp JR, Holman CA, Whitmore DN, Kamicki J, Austin MA. Direct intravascular fetal blood transfusion by fetoscopy in severe Rhesus isoimmunisation. Lancet. 1981;1:625–7.
- 105. Bang J, Bock JE, Trolle D. Ultrasound-guided fetal intravenous transfusion for severe rhesus haemolytic disease. Br Med J. 1982;284:373–4.
- 106. Harman CR, Bowman JM, Manning FA, Menticoglou SM. Intrauterine transfusion-intraperitoneal versus intravascular approach: a case-control comparison. Am J Obstet Gynecol. 1990;162:1053–9.
- 107. Radunovic N, Lockwood C, Alvarez M, Plecas D, Chitkara U, Berkowitz R. The severely anemic and hydropic isoimmune fetus: changes in fetal hematocrit associated with intrauterine death. Obstet Gynecol. 1992;79:390–3.
- 108. Hallak M, Moise KJ, Hesketh DE, Cano LE, Carpenter RJ. Intravascular transfusion of fetuses with rhesus incompatibility: prediction of fetal outcome by changes in umbilical venous pressure. Obstet Gynecol. 1992;8:1–5.
- Deaver JE, Leppert PC, Zaroulis CG. Neonatal alloimmune thrombocytopenia purpura. Am J Perinatol. 1986;3:127–31.
- 110. Giovangrandi Y, Daffos F, Kaplan C, Forestier F, MacAleese J, Moirot M. Very early intracranial haemorrhage in alloimmune fetal thrombocytopenia (Letter). Lancet. 1990;336:310.
- 111. Muller JY, Reznikoff-Etievant MF, Patereau C, Dangu C, Chesnel N. Thrombopenies neo-natales allo-immunes: Etude clinique et biologique de 84 cas. Presse Med. 1985;14:83–6.
- 112. Soyinka JO, Onyeji CO, Omoruyi SI, Owolabi AR, Sarma PV, Cook JM. Pharmacokinetic interactions between ritonavir and quinine in healthy volunteers following concurrent administration. Br J Clin Pharmacol. 2010;69:262–70.
- 113. Soyinka JO, Onyeji CO, Omoruyi SI, Owolabi AR, Sarma PV, Cook JM. Effects of concurrent administration of nevirapine on the disposition of quinine in healthy volunteers. J Pharm Pharmacol. 2009;61:439–43.
- 114. Newton PN, Ward S, Angus BJ, et al. Early treatment failure in severe malaria resulting from abnormally low plasma quinine concentrations. Trans R Soc Trop Med Hyg. 2006;100:184–6.
- 115. Dorsey G, Staedke S, Clark TD, Njama-Meya D, Nzarubara B, Maiteki-Sebuguzi C, Dokomajilar C, Kamya MR, Rosenthal PJ. Combination therapy for uncomplicated falciparum malaria in Ugandan children: a randomized trial. JAMA. 2007;297:2210–9.
- 116. German P, Greenhouse B, Coates C, et al. Hepatotoxicity due to a drug interaction between amodiaquine plus artesunate and efavirenz. Clin Infect Dis. 2007;44:889–91.

- 117. Kredo T, van der Walt JS, Mauff K, Weisner L, Maartens G, Cohen K, Smith P, Barnes KI. Nevirapine increases lumefantrine exposure in HIV-infected patients. 17th conference on oportunistic infections and retroviruses. Feb 2010. San Francisco. Abstract N-140.
- 118. Gazzard B, Lundgren J. editors. British HIV Association and British Infection Association guidelines for the treatment of opportunistic infection in HIV-seropositive individuals 2011. HIV Medicine. Editorial Board: Jose' Gatell, Margaret Johnson, Jurgen Rockstroh, Schlomo Staszewski, Ian Williams. British HIV, 2011 British HIV Association HIV Medicine. 2011;12 (Suppl 2):1–5.
- 119. Chamberlain NR. Infections of the fetus and newborn infant: Congenital and perinatal infections. Lecture. 2004. http://www.atsu.edu/faculty/chamberlain/website/lectures/lecture/congen.htm.
- 120. Daffos F, Capella-Pavlovsky M, Forestier F. A new procedure for fetal blood sampling in utero: preliminary results of fifty-three cases. Am J Obstet Gynecol. 1983;146:985–7.
- 121. Rodis JF, Hovick TJ, Quinn DL, Rosengren SS, Tattersall P. Human parvovirus infection in pregnancy. Obstet Gynecol. 1988;72:733–8.
- 122. Anderson LJ, Tsou C, Parker RA, et al. Detection of antibodies and antigens of human parvovirus B 19 by enzyme-linked immunosorbent assay. J Clin Microbiol. 1986;24:522–6.
- 123. Peters MT, Nicolaides KH. Cordocentesis for the diagnosis and treatment of human fetal parvovirus infection. Obstet Gynecol. 1990;75:501–4.
- 124. Naides SJ, Weiner CP. Antenatal diagnosis and palliative treatment of non-immune hydrops fetalis secondary to fetal parvovirus B 19 infection. Prenat Diagn. 1989;9:105–14.
- 125. Klein JO, Baker CJ, Remington JS, et al. Current concepts of infection of the fetus and newborn infants. In: Remington JS, Klein JO, Wilson CB, Baker CJ, editors. Infectious diseases of the fetus and newborn infant. 6th ed. Philadelphia: Elsevier Saunders; 2006. p. 1–25.
- 126. Clinical Guidelines. Section B: Obstetrics and Midwifery Care. Women and Newborn Health Service, King Edward Memorial Hospital, Perth, Western Australia. 2001. Infection Control Policy Manual 4.2: Varicella-Zoster: Varicella (Chicken Pox) in Pregnancy. http://citeseerx.ist.psu.edu/viewdoc/download;jsessionid=7495AA62EB86E879268 B3E2C300F06CF?doi=10.1.1.445.3483&rep=rep1 &type=pdf.
- 127. Maternal to Fetal Infections. Definition of maternal to fetal. Infections by Medical dictionary. http:// medical-dictionary.thefreedictionary.com/Maternal +to+Fetal+Infections.
- Willyard C. Tinkering in the womb: the future of fetal surgery Washington, DC. Nat Med. 2008;14:1176–7.
- 129. Pranke P, Onsten T. Umbilical cord blood transfusion and its therapeutic potentialities. Chapter 5. In: Bhattacharya N, Stubblefield P, editors. Pregnancy

specific substances in regenerative medicine. 1st ed. London: Springer Publications; 2011. p. 83–93.

- Gluckman E. Umbilical cord blood transfusions in low-income countries. Lancet Hematol. 2015;2:e85–6.
- 131. Surbek DV, Glanzmann R, Senn H-P, et al. Can cord blood be used for autologous transfusion in preterm neonates? Eur J Pediatr. 2000;159:790–91. 12. Luban NLC. Neonatal red blood cell transfusions. Vox Sang. 2004;87(Suppl 2):S184–8.
- Luban NLC. Neonatal red blood cell transfusions. Vox Sang. 2004;87 suppl 2:S184–8.
- 133. Bhattacharya N, Mukherjee K, Chettri MK, et al. A study report of 174 units of placental umbilical cord whole blood transfusion in 62 patients as a rich source of fetal hemoglobin supply in different indications of blood transfusion. Clin Exp Obstet Gynecol. 2001;28(1):47–52.
- 134. Garritsen HSP, Brune T, Louwen F, et al. Autologus red cells derived from cord blood: collection, preparation, storage and quality controls with optimal additive storage medium (Sag-mannitol). Transfus Med. 2003;13:303–10.
- 135. Bhattacharya N. Placental umbilical cord whole blood transfusion: a safe and genuine blood substitute for patients of the under-resourced world at emergency. J Am Coll Surg. 2005;200(4):557–63.
- Walsh TS, Salch E-E-D. Anaemia during critical illness. Br J Anaesth. 2006;97:278–91.
- 137. Widing L, Bechensteen AG, Mirlashari MR, et al. Evaluation of nonleukoreduced red blood cell transfusion units collected at delivery from the placenta. Transfusion. 2007;47:1481–7.
- Strauss RG. Autologous transfusions for neonates using placental blood; a cautionary note. Am J Dis Child. 1992;146:21–2.
- Flake A, Zanjani E. In utero hematopoietic stem cell transplantation: ontogenic opportunities and biologic barriers. Blood. 1999;94:2179–91.
- 140. Fleischmam R, Mintz B. Prevention of genetic anaemias in mice by microinjection of normal hematopoietic stem cells into the fetal placenta. Proc Natl Acad Sci. 1979;76:5736–40. doi:10.1073/ pnas.76.11.5736.
- 141. Morris RK, Chan BC, Kilby MD. Review: advances in fetal therapy. Obstet Gynecol. 2010;12:94–102.
- 142. Lindton B, Markling L, Ringden O, Westgren M. Mixed lymphocyte culture of human fetal liver cells. Fetal Diagn Ther. 2000;15:71–8. doi:10.1159/000020979.
- Tiblad E, Westgren M. Fetal stem-cell transplantation. Best Pract Res Clin Obstet Gynaecol. 2008;22:189– 201. doi:10.1016/j.bpobgyn.2007.07.007.
- 144. Tabor A, Vestergaard CH, Lidegaard O. Fetal loss rate after chorionic villus sampling and amniocentesis: an 11-year national registry study. Ultrasound Obstet Gynecol. 2009;34:19–24.
- 145. Adzick NS. Open fetal surgery for life-threatening fetal anomalies. Semin Fetal Neonatal Med. 2010;15:1–8.

- 146. Harrison MR, Adzick NS, Longaker MT, et al. Successful repair in utero of a fetal diaphragmatic hernia after removal of herniated viscera from the left thorax. New Eng J Med. 1990;322:1582–4.
- 147. Harrison MR, Adzick NS, Bullard KM, et al. Correction of congenital diaphragmatic hernia in utero VII: a prospective trial. J Pediatr Surg. 1997;32:1637–42.
- 148. Jani JC, Nicolaides KH, Gratacos E, et al. Severe diaphragmatic hernia treated by fetal endoscopic tracheal occlusion. Ultrasound Obstet Gynecol. 2009;34:304–10.
- 149. Iqbal CW. Chapter 3. Congenital diaphragmatic hernia. Page no: 14. Brad Feltis and Christopher Muratore: Chief editors. Fetal diagnosis and therapy: a reference handbook for pediatric surgeons. American Pediatric Surgical Association from the Fetal Diagnosis and Therapy Committee, Apr 2013. https://www.eapsa.org/about-apsa/ committees/?cid=FETAL.
- 150. Parkes SE, Muir KR, Southern L, Cameron AH, Darbyshire PJ, Stevens MC. Neonatal tumours: a thirty-year population-based study. Med Pediatr Oncol. 1994;22:309–17.
- 151. Isaacs H Jr. Perinatal (congenital and neonatal) neoplasms: a report of 110 cases. Pediatr Pathol. 1985;3:165–216. 3. Werb P, Scurry J, Ostor A, Fortune D, Attwood H. Survey of congenital tumors in perinatal necropsies. Pathology. 1992;24:247–53.
- 152. Lonergan GJ, Schwab CM, Suarez ES, Carlson CL. Neuroblastoma, ganglioneuroblastoma, and ganglioneuroma: radiologic-pathologic correlation. Radio- Graph. 2002;22:911–34.
- Sklansky M, Greenberg M, Lucas V, Gruslin-Giroux A. Intrapericardial teratoma in a twin fetus: diagnosis and management. Obstet Gynecol. 1997;89:807–9.
- 154. Woodward PJ, Sohaey R, Kennedy A, Koeller KK. From the archives of the AFIP: a comprehensive review of fetal tumors with pathologic correlation. Capt Radio Graph. 2005;25:215–42. doi:10.1148/ rg.251045156. Published online.
- 155. Keslar PJ, Buck JL, Selby DM. Infantile hemangioendothelioma of the liver revisited. Radio Graph. 1993;13:657–70.
- 156. Corey W. Iqbal. Saccrococcygeal tumors. Chapter 4, Page no: 21. Brad Feltis and Christopher Muratore: Chief editors. Fetal diagnosis and therapy: a reference handbook for pediatric surgeons. American Pediatric Surgical Association from the Fetal Diagnosis and Therapy Committee, Apr 2013. https://www.eapsa. org/about-apsa/committees/?cid=FETAL.
- 157. Shaer CM, Chescheir N, Schulkin J. Myelomeningocele: a review of the epidemiology, genetics, risk factors for conception, prenatal diagnosis, and prognosis for affected individuals. Obstet Gynecol Surv. 2007;62:471–9. PubMed: 17572919.
- 158. Boulet SL, Yang Q, Mai C, Kirby RS, Collins JS, Robbins JM, Meyer R, Canfield MA, Mulinare J. Trends in the postfortification prevalence of spina

bifida and anencephaly in the United States. Birth Defects Res A Clin Mol Teratol. 2008;82:527–32. PubMed: 18481813.

- 159. Ferschl M, Ball R, Lee H, Rollins MD. Anesthesia for in utero repair of myelomeningocele. Anesthesiology. 2013;118(5):1211–23. doi:10.1097/ ALN.0b013e31828ea597).
- 160. Tulipan N, Bruner JP, Hernanz-Schulman M, Lowe LH, Walsh WF, Nickolaus D, Oakes WJ. Effect of intrauterine myelomeningocele repair on central nervous system structure and function. Pediatr Neurosurg. 1999;31:183–8. PubMed: 10705927.
- 161. Sutton LN, Adzick NS, Bilaniuk LT, Johnson MP, Crombleholme TM, Flake AW. Improvement in hindbrain herniation demonstrated by serial fetal magnetic resonance imaging following fetal surgery for myelomeningocele. JAMA. 1999;282:1826–31. PubMed: 10573273.
- 162. Corey W. Iqbal and Abdalla E. Zarroug. Fetal Neoplasms, Chapter 5, Page nos:19–23. Brad Feltis and Christopher Muratore: Chief editors. Fetal diagnosis and therapy: a reference handbook for pediatric surgeons. American Pediatric Surgical Association from the Fetal Diagnosis and Therapy Committee, Apr 2013. https://www.eapsa.org/about-apsa/ committees/?cid=FETAL.
- 163. Chaoui R. The four-chamber view: four reasons why it seems to fail in screening for cardiac abnormalities and suggestions to improve detection rates. Ultrasound Obstet Gynecol. 2003;22:3.
- 164. Holley DG, Martin GR, Brenner JL, et al. Diagnosis and management of fetal cardiac tumors: a multicenter experience and review of the published reports. J Am Call Cardiol. 1995;25:516–20.
- 165. Groves AM, Fagg NL, Cook AC, Allan LD. Cardiac tumors in intrauterine life. Arch Dis Child. 1992;67:1189–92.
- 166. Sklansky M. Fetal cardiac malformations and arrhythmias. Detection, diagnosis, management, and prognosis. In: Creasy RK, Resnik R, Iams JD, Lockwood CJ, Moore TR, editors. Creasy and Resnik's Maternal and Fetal Medicine: Principles and Practice. 6th ed. Philadelphia: Saunders; 2009. p. 305–46.
- 167. Gava G, Buoso G, Beltrame GL, et al. Cardiac rhabdomyoma as a marker for the prenatal detection of tuberous sclerosis: case report. Br J Obstet Gynaecol. 1990;97:1154–7.
- 168. Tegnander E, Eik-Nes S. The examiner's ultrasound experience has a signify cant impact on the detection rate of congenital heart defects at the secondtrimester fetal examination. Ultrasound Obstet Gynecol. 2006;28:8.
- 169. Weber H, Kleinman C, Hellenbrand W, et al. Development of a benign intrapericardial tumor between 20 and 40 weeks of gestation. Pediatr Cardiol. 1988;9:153.
- 170. Carvalho J. The fetal heart or the lymphatic system or ...? The quest for the etiology of increased nuchal translucency. Ultrasound Obstet Gynecol. 2005;25:215.

- Jones K. Smith's recognizable patterns of human malformation. 6th ed. Philadelphia: Elsevier Saunders; 2006.
- 172. Trines J, Hornberger LK. Evolution of heart disease in utero. Pediatr Cardiol. 2004;25:287.
- 173. Hornberger LK, Sanders SP, Sahn DJ, et al. In utero pulmonary artery and aortic growth and potential for progression of pulmonary outflow tract obstruction in tetralogy of Fallot. J Am Coll Cardiol. 1995;25:739.
- 174. Hornberger LK, Sanders SP, Rein AJ, et al. Left heart obstructive lesions and left ventricular growth in the midtrimester fetus: A longitudinal study. Circulation. 1995;92:1531.
- 175. Tworetzky W, Wilkins-Haug L, Jennings RW, et al. Balloon dilation ofsevere aortic stenosis in the fetus. Circulation. 2004;11:2125.
- 176. Makikallio K, McElhinney D, Levine J, et al. Fetal aortic valve stenosis and the evolution of hypoplastic left heart syndrome: patient selection for fetal intervention. Circulation. 2006;113:1401.
- 177. Cooper M, Enderlein M, Dyson D, et al. Fetal echocardiography: retrospective review of clinical experience and an evaluation of indications. Obstet Gynecol. 1995;86:577.
- 178. Ville Y. Twin-to-twin transfusion syndrome: time to forget the Quintero staging system? Ultrasound Obstet Gynecol. 2007;30:924–7. doi:10.1002/ uog.5221.
- 179. Yamamoto M, El Murr L, Robyr R, Leleu F, Takahashi Y, Ville Y. Incidence and impact of perioperative complications in 175 fetoscopy guided laser coagulations of chorionic plate anastomoses in fetofetal transfusion syndrome before 26 weeks of gestation. Am J Obstet Gynecol. 2005;193:1110–6. doi:10.1016/j.ajog.2005.07.003.
- 180. Senat M, Deprest J, Boulvain M, Paupe A, Winer N, Ville Y. Endoscopic laser surgery versus serial amnioreduction for severe twin-to-twin transfusion syndrome. N Engl J Med. 2004;351:182–4. doi:10.1056/NEJMoa032597.
- 181. Bánhidy F, Lowry RB, Czeizel AE. Risk and Benefit of Drug Use During Pregnancy. Int J Med Sci 2005;2(3):100–6. doi: 10.7150/ijms.2.100.
- 182. Westgren M, Ringden O, Eik-Nes S, Ek S, Anvret M, Brubakk A-M, et al. Lack of evidence of permanent engraftment after in utero fetal stem cell transplantation in congenital hemoglobinopathies. Transplantation. 1996;61:1176–9. doi:10.1097/00007890-199604270-00010.
- Peranteau WH, Endo M, Adibe OO, Flake AW.Evidence for an immune barrier after in utero hematopoietic-cell transplantation. Blood. 2007;109:1331– 3. doi:10.1182/blood-2006-04-018606.
- 184. Ramachandra DL, Shaw SSW, Shangaris P, Loukogeorgakis S, Guillot PV, De Copp P, David AL. In utero therapy for congenital disorders using amniotic fluid stem cells. Review Article published: 19 Dec 2014. doi:10.3389/fphar.2014.00270.

- 185. Tsai MS, Hwang S-M, Tsai Y-L, Cheng F-C, Lee J-L, Chang Y-J. Clonal amnioticfluid-deri vedstemcellsexpresscharacteristicsofboth. Am J Obstet Gynecol. 2004;191:309–14. doi:10.1095/ biolreprod.105.046029.
- 186. DeCoppi P, Bartsch Jr G, Siddiqui MM, Xu T, Santos CC, Perin L, et al. Isolation of amniotic stem cell lines with potential for therapy. Nat Biotechnol. 2007;25(100):106. doi:10.1038/nbt1274.
- 187. Gosden CM. Amniotic fluid cell types and culture. Br Med Bull. 1983;39:348–54.
- Delo DM, De Coppi P, Bartsch G. Amniotic fluid and placental stem cells. Methods Enzymol. 2006;419:426–38. doi:10.1016/S0076-6879(06) 19017-5.
- 189. Chan J, Kumar S, Fisk NM. Firsttrimesterembryofetoscopic and ultrasound-guided fetal blood sampling for ex vivo viral transduction of cultured human fetal mesenchymal stem cells. Hum Reprod (Oxford, Engl). 2008;23:2427–37. doi:10.1093/ humrep/den302.
- 190. Ditadi A, deCoppi P, Picone O, Gautreau L, Smati R, Six E, et al. Human and murine amniotic fluidc-kitCL in cells display hematopoietic activity. Blood. 2009;113:3953–60. doi:10.1182/ blood-2008-10-182105.
- 191. Mehta V, Nader KA, Waddington S, David AL. Organtargeted prenatal gene therapy—how far are we? Prenat Diagn. 2011;31:720–34. doi:10.1002/ pd.2787.
- 192. Shaw SWS, Bollini S. Autologoustransplantationof amnioticfluid- derived mesenchymal stem cells into sheep fetuses. Cell Transplant. 2011;20:1015–31. doi:10.3727/096368910X543402.
- 193. Takahama Y. Journey through the thymus: stromal guides for T-cell development and selection. Nat Rev Immunol. 2006;6:127–35. doi:10.1038/nri1781.
- 194. DiTrapani M, Bassi G, Fontana E, Giacomello L, Pozzobon M, Guillot PV, et al. Immuneregulatoryp ropertiesofcd117(pos)amniotic fluid stem cells vary according to gestational age. Stem Cells Dev. 2014. doi:10.1089/scd.2014.0234.
- 195. Moorefield EC, McKee EE, Solchaga L, Orlando G, Yoo JJ, Walker S, et al. Cloned, CD117selectedhumanamnioticfluidstemcells are capable of modulating the immune response. PLoS ONE. 2011;6, e26535. doi:10.1371/journal. pone.0026535.
- 196. Kaviani A, Perry TE, Dzakovic A, Jennings RW, Ziegler MM, Fauza DO. Theamnioticfluidasasour ceofcellsforfetaltissueengineering. J Pediatr Surg. 2001;36:1662–5. doi:10.1053/jpsu.2001.27945.
- 197. Schoeberlein A, Holzgreve W, Dudler L, Hahn S, Surbek DV. In utero transplantation of autologous and allogeneic fetal liver stem cell sin ovinefetuses. Am J Obstet Gynecol. 2004;191:1030–6. doi:10.1016/j.ajog.2004. 06.042.
- Mackenzie TC, Flake AW. Human mesenchymal stem cells persist, demonstrate site-specific multi-

- 199. Li H, Gao F, Ma L, Jiang J, Miao J, Jiang M, et al. Therapeutic potential of in utero mesenchymal stemcell (MSCs)transplantation in rat foetuses with spina bifida aperta. J Cell Mol Med. 2012;16:1606–17. doi:10.1111/j.1582-4934.2011.01470.
- 200. Kurjak A, Carrera JM, Mc Cullough LB, Chervenak FA. The ethical concept of the fetus as a patient and the beginning of human life. PERIOD BIOL J 2009;111(3):341–8.
- 201. Beller FK, Zlatnik GP. The beginning of human life. J Assist Reprod Genet. 1995;12(8):477–83.

35

Structural and Functional Developmental Perspectives of the Placental Barrier and Its Role in the Fetal Development During the First and Second Trimesters

Priyodarshi Sengupta, Biplabendu Talukdar, Indranil Roy, Santanu Tripathi, Nandita Bose, Sushanta Banerjee, and Niranjan Bhattacharya

Introduction

The term placenta has been derived from the Latin word meaning 'Cake' and the Greek word for 'flat slab like structure'. The primary function of the placenta is to act as a fetomaternal organ and a barrier with two of its components, the fetal placenta also termed as the "Chorion frondosum" and the maternal placenta termed as "Decidua basalis" [1, 2]. The fetal placenta develops from the blastocyst

Formerly, Professor, Department of Pathology, IPGMER, SSKM Hospital, Kolkata, India

S. Banerjee, MD Director, Medical Education, Govt. of West Bengal, Kolkata, India

Formerly, Professor and Head, Department of Pharmacology, RG Kar Medical College, Kolkata, India and the maternal placenta from the maternal uterine tissues. The first concept regarding the presence of a fetomaternal or a placental barrier was coined around 46 years ago by using transport physiology and ultrastructural analysis [3]. However in terms of evolutionary trend it is observed that the human fetomaternal barrier which is 10 nm in size is less selective than the epitheliochorial barriers in animals. The metabolic capacity of the trophoblast and macrophage along with the fetal blood cells plays an important role in the fetomaternal barrier in the first trimester and the early stages of second trimester by metabolizing and converting harmful toxins into less harmful substrates and by also forming a sequential barrier to the maternal activated immune cells [4]. Also the placenta plays a very important role in providing a micro niche which supports the fetal growth through transfer of nutrients and immunological

P. Sengupta, MSc, MPhil • B. Talukdar, MD I. Roy, MSc

Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

S. Tripathi, MD Professor, Department of Clinical and Experimental Pharmacology, School of Tropical Medicine, Kolkata, India

N. Bose, MD

Director, School of Tropical Medicine, Kolkata, India

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India e-mail: sanjuktaniranjan@gmail.com

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_35

properties which helps in imparting protective immunological properties to the fetus. It also acts as a major fetomaternal barrier against pathogens and the maternal immune system. The placenta also undertakes the secretion of different hormones, cytokines, growth factors and other bioactive products essential for the fetus.

Development of the Embryo

Pre Lacunar Stage

After the formation of a two cell stage, there are a series of divisions resulting in a process known as cleavage. This cleavage produces a three cell stage from a two cell stage, a four cell stage and a five cell stage. This subsequent cleavage of the ovum results in the formation of 16 cells where it looks like a mulberry thereby attaining its name as the morula. The morula remains surrounded by the zona pellucida and contains an inner cell mass surrounded completely by an outer layer of cells which gives rise to the trophoblast. This trophoblast provides nutrition to the developing embryo and forms the trophoectoderm and gives rise to the placenta in later stages of development. The inner cell mass gives rise to the embryo proper. The amnion is derived from the trophoblast layer and the chorion from the epiblast after 8 days of fertilization. Both the amnion and chorion are derivatives of the embryo [5]. With subsequent increase in the quantity of the fluid the morula takes the appearance of a cyst like structure. The trophoblastic cells attain a more flat shaped like structure and the inner cell mass gets attached to the inner side of the trophoblastic layer on one side only. This is known as the embryonic or the Animal pole and the opposite side is known as the aembryonic pole. This stage of development is often regarded as the blastocyst stage [6].

Lacunar Stage

On the eighth day of conception, the implantation of this blastocyst in the endometrium walls of the uterus commences the development of the placenta. The trophoblast layer of the blastocyst stage gives rise to the chorion and the inner cell mass later on develops to form the umbilical cord and the amnion. After adhering to the inner walls of the uterine endometrium the trophoblast layer of the blastocyst invades the endometrial epithelium, thereby eroding the deciduas and allowing the embedment of the blastocyst, a biological process often termed as the implantation stage [7]. After the implantation process the trophoblastic cells undergo rapid cell division to form a bi layered trophoblast which is composed of multinucleated outer bi layered syncytiotrophoblast and an inner mononucleated cytotrophoblast layer [7]. Appearance of small intra syncytial vacuoles in the syncytiotrophoblast starts to develop and by day 8 it becomes confluent and gets separated by the presence of lamellae and trabecule [7]. Primary villi are formed after the invasion of the cytotrophoblast into the trabeculae whereas the lacunaes form the intervillous spaces where there is a rich flow of maternal blood. This also marks the beginning of the villous stages of placentation [7].

In the first trimester of early pregnancy the anti-implantation pole gives rise to the smooth chorion due to the degradation of the villous and fibroid deposition whereas at the implantation pole appearance of leafy chorion frondosum occurs [7]. After 8-9 days of fertilization morphological changes occur in the inner cell mass forming the epiblast and the hypoblast. The epiblast constitutes the amniotic epithelium from where the three germ layers namely the endoderm, mesoderm and the ectoderm occurs whereas from the hypoblast, cells migrate to form the exocoelomic membrane which modifies to form the yolk sac [7]. This exocoelomic membrane along with the blastocoel also forms the extra embryonic reticulum. Further, some more cells from the hypoblast migrate to form the extra embryonic mesoderm which surrounds the yolk sac and the amniotic cavity and later on forms the amniotic and chorionic mesoderm. This whole process occurs before gastrulation initiates [7].

Development of the Fetomaternal Barrier

The syncytiotrophoblast layer is a continuous layer which extends all over the surface of the villi. Physiological evidence suggests that there is an existence of two different routes through the fetomaternal barrier, namely the transcellular and the paracellular routes [8].

In the presence of tertiary villi there is an establishment of the intraplacental fetal circulation where the maternal and the fetal blood in the intervillous space comes into close contact with one another but remains separated due to the presence of the fetomaternal barrier also termed as the placental barrier which is further composed of five layers. The first layer is a continuous syncytiotrophoblast layer covering the villi and thus lining the intervillous spaces. The second layer is the cytotrophoblast layer which is consistent in its appearance in the first trimester but later discontinuous in the second trimester [8]. A trophoblastic laminal layer and the connective tissue layer which is derived from the extraembryonic mesoderm is the next layer. The last layer is referred to as the fetal endothelium layer which remains uncovered during the first and the second trimester. After the second month of pregnancy and with increase in gestational time and trimesters this fetomaternal barrier undergoes quantitative changes rather than qualitative like the thickness of the syncytiotrophoblast layer decreases from 15 μ m to a mere 4.1 μ m in the third trimester. Also rapid expansion of the villous surface occurs as the cytotrophoblast becomes rarified [9].

Comparison of the Fetomaternal Barrier during the First and Second Trimesters of Development

The placental barrier is composed of structures which separate the maternal and the fetal blood from one another. In the first trimester the placental barrier consists of the cytotrophoblast, syncytiotrophoblast and the villus mesenchymal layer which inhibits the entry of macrophages. During the second trimester the cytotrophoblast layer disappears from the villus wall as the thickness of the barrier subsequently decreases with increase in the vascularization and the surface area. By the beginning to mid of fifth month, i.e., in the second trimester there is a considerable increase in the number of fetal vessels along their numerous branching and sub branching which gets closer to the villous surface [10]. By the end of the second trimester the nuclei of the syncytiotrophoblast forms proliferative notes which is due to the aggregation and grouping of the cells together. The other regions of the syncytiotrophoblast lack nuclei and comes into peripheral and close proximity to the exchange zones of the capillaries thereby enhancing the rate of exchange between the maternal and the fetal compartments [10].

Also the placenta in the first trimester consists of the chorionic plate, intervillous space surrounding the chorionic villi, cell islands, basalplate from which the septa protrudes into the anchoring villi [10]. These anchoring villi further remain connected to the basal plate via the cell columns. In the beginning of the second trimester these intervillous spaces remain filled with maternal blood. The chorionic plate in the first trimester is usually devoid of the amnion and it remains covered by a conspicuous incomplete layer of mesothelium and a thick layer of chorionic mesoderm where the chorionic vessels generally remain embedded [11]. In the intervillous space with the progression of pregnancy and gestation time the syncytiotrophoblast is replaced by the acellular eosinophilic fibrinoid [12].

At the end of the fourth week all of the placental villi become the tertiary villi and they adapt themselves according to the needs of the growing embryo. At the beginning of the eighth week there is development of the mesenchymal villi which show a homogenous, rather, a dense cellular stroma. These villi remain immature even during the second trimester and are known as the intermediate immature villi [13]. The cell islands in the first trimester are generally accumulations of the extravillous trophoblastic cells embedded in fibrinoid generally, whereas the placental septa resembles the rough pillar shape like extensions of the basal plate which protrudes into the intervillous spaces [13]. The basal plate which is also the bottom part of the intervillous space represents the part of the fetomaternal junctional zone which remains adhered to the placenta after delivery [13].

The placenta in the second trimester generally has a similar structure to that of the placental development as observed in the first trimester with the exception that in this stage the layers are much better defined [10]. In the second trimester the cytotrophoblast slowly disappears from the tertiary villi walls and thereby diminishes the distance between the maternal and the fetal circulation. These cytotrophoblasts also disappear from the chorionic plate whereas in the basal plate the cytrophoblast layer still persists [10]. Along with the deciduous formation and the deposition of fibrin, protrusions are formed which project into the intervillous spaces and thereby form the cotyledons which are normally primitive in appearance during this time. In the meantime septas are formed due to the spread of the placenta [14].

Villous Development in the First and Second Trimesters

On the 12th day after conception the transition phase starts where the implanted embryo moves from the lacunar phase to the primary villous, accompanied by secondary and tertiary villous formation stages, and this continues till the 18th day to term [8]. In the tertiary villi as soon as the intra placental fetal circulation develops, the fetal blood in the villi and the maternal blood in the intervillous come into close contact with each other but remain separated by the placental barrier [8]. This placental barrier is made up of a continuous syncytiotrophoblast layer, an initially complete cytotrophoblast layer in the first trimester. However this becomes inconsistent in the second and third trimester period. With increase in the trimester period there are extensive structural and functional changes in the blood placental barrier like the thickness of the syncytiotrophoblast layer decreases; the cytotrophoblast becomes relatively rarified due to the constant increase in the villous surface areas [8]. The intervillous position of the fetal capillaries comes closer to the villous surface with advancements in maturation resulting in the reduction of mean maternofetal diffusion distance [8]. This distance decreases from 50 to 10 μ m and further between 4 and 5 μ m at term gestation [15]. The exchange between the fetus and the mother is focused on the presence of terminal and matured villi which remain supplied by fetal capillaries lined by continuous epithelium without any pores [15]. Its main function is to act as a passive filter and limit the transfer of molecules with larger weight, just like in the blood brain barrier, the presence of pericytes and smooth muscle cells also helps in the control of permeability and vaso regulation [16].

The synctiotrophoblast separates the mixing of maternal and fetal blood. Generally it is continuous and uninterrupted and extends all over the villous trees. The transplacental exchange mainly takes place by the paracellular and the transcellular route. The paracellular route of exchange is mediated via the extracellular water filled pathways and the transcellular involves the bi lipid plasma membrane layer and the cytoplasm of the synctiotrophoblast. Occurrence of blood clots and exchange of toxic substances prematurely can be attributed to the fact that the syncytiotrophoblast does not always maintain an intact integrity [8].

Blood Circulation across the Placenta during the First and Second Trimesters

Maternal Vascularization and Placental Circulation

Maternal vascularization is often defined as an important step in the pre implantation stage where the structural changes of the spiral arteries occur resulting in the disorganization of the smooth muscle cell layers and the matrix by means of proteinases such as MMPs [17] along with vasodilation and a reduction in blood pressure. This further assists in increased flow of maternal blood towards the placental surface due to the decrease of maternal arterial resistance [18-20]. The maternal blood placental circulation is matured and completed by the end of the first trimester. During the pre implantation of the blastocyst stage, the walls of the uterine endometrium epithelial cells undergoes morphological changes also defined as "decidualisation." The spiral arteries are remodeled and there is an increase in the diameter size of these arteries [21]. They also become less convoluted. The increase in the diameter helps in the increased pressure of blood flow from the maternal end to the intervillous space of the placenta thereby bathing the fetal villi. In the case of human placentas the maternal blood comes into direct contact with the fetal chorion although there is no exchange of fluid with the fetal chorion. Further due to a decrease

in the pressure, the deoxygenated blood flows back through the endometrial veins.

Fetoplacental Circulation

On the other hand deoxygenated blood from the fetus is passed though the umbilical arteries to the placenta. These umbilical arteries radiate to form chorionic arteries at the junction of the umbilical cord and the placenta and further divides to form an extensive network of arterio capillary venous branches in the villi brining the fetal blood in close proximity to the maternal blood but with no mixing due to the presence of the placental barrier [8]. Vasoconstriction is caused by endothelin and prostanoids in placental arteries whereas nitric oxides cause vasodilation. Free radicals may be generated as the feto placental circulation is extremely sensitive to hypoxia resulting in preeclampsia [22]. Melanin can play a major role in such cases by acting as an anti-oxidant [22]. Also certain other molecules like oxygen, electrolytes, water, carbohydrates, lipids, proteins, amino acids, vitamins, hormone, drugs including pathogens and viruses can enter the fetal compartment during the first trimester from the rich maternal spiral arteries. On the other hand in the fetomaternal exchange system, carbon dioxide, water and urea along with any waste products or hormones can enter the uterine circulatory system from the fetus. During the course of development the maternal and the fetal circulatory exchanges comes into close contact without any mixing due to the presence of a thin layer of syncytiotrophoblast. Any disruption in the flow of blood can eventually hamper the fetomaternal circulatory system thus causing hypoxia to the fetus during the first and the second trimester [8, 23, 24].

Physiological Properties of the Placenta during the First and Second Trimesters

The placenta consists of Umbilical vessels, chorionic plate arteries and multiple dividing villi. It lacks any form of autonomic regulation and instead, the vascular function is under the direct influence of the hormones [25] thereby stimulating the roles of agonists in the placenta function. There is a delicate balance between the role of vaso constrictors and vaso dilators in controlling the placental circulation. Presence of mechanosensors in the placenta can be a possibility as it has been shown to adapt to various hemodynamic changes [26]. Another important property of the placental circulation is that it depends on the adequate matching of the placental and uterine blood flow and has shown to have affinity towards Angiotensin II receptors and ACE [27–29]. Other potent vasoconstrictors include Et-1 released from the endothelial cells and are found abundantly across the placenta and umbilical vessels [30–33], thrombaxanes and 5-HT [33, 34].

Role of Vasculogenesis and Angiogenesis in the First and Second Trimester Development of the Placenta

Angiogenesis starts from day 32 and proceeds till week 25 after conception. During week 15 when a regression of peripheral capillary nets takes place, the remaining central capillaries develop into primitive veins and arteries. Due to vascularization of the maternal spiral arteries the maternal adaptions to pregnancy are almost complete by the end of the first trimester during which the maternal blood via the developing fetomaternal passage enters and floods the placental intervillous spaces but does not mix with the fetal blood. This begins at the eighth to ninth week and it increases more dynamically by the time it reaches the end of the first trimester, i.e., the 12th week [35–38].

Presence of Vascular endothelial growth factor (VEGF) and Placental growth (PGF) factors in the maternal plasma normally remains high at around 25 days post conception and this trend continues till the second trimester [39, 40]. In the presence of angiogenesis and a relatively hypoxic environment, the trophoblast cells increase their proliferation to form the highly vascularized villi and eventually a complex network of blood vessels which begin to transport blood [19, 41, 42, 44]. Due to further enhancement of the level of angiogenic factors, there is an increase in the formation of blood vessels from pre existing ones and eventually develops into well defined arteries, veins and arterioles [43]. Therefore the exchange of nutrients, gases, oxygenated and deoxygenated blood initiates at this stage [37]. The placenta helps in a stable passage of blood continuously through its vessels due to its low resistance system [27] and a pressure difference between the vascular connections which further drives the blood from the basal part towards the chorionic plate [19, 35, 42, 44]. The function of the early or primary villous trees are incomplete at this stage as they fail to develop into a fully functional system lacking autonomic control to help keep the fetus constantly supplied with blood. In the second trimester no new vessels are formed rather sprouting of multiple villi branches takes place which are constantly replaced by immediate villi for the rest of the trimesters [8, 25].

Mechanisms of Fetomaternal Exchange

There are different types of mechanisms by which feto maternal exchange takes place. They are either simple diffusion by which mainly gaseous substances like oxygen, carbon dioxide and fatty acids gets exchanged. Active transport includes the passage of Iron, Calcium and Iodine whereas facilitated diffusion includes the passage for glucose. Amino acids can pass through secondary active transport mechanisms whereas electrolytes and water are exchanged via bulk transport system. However taking advantage of this nature's unique system of selective exchange between the mother and the fetus, often harmful substances like nicotine, cocaine, alcohol, sedatives, anesthetics and analgesics can indulge in maternal fetal exchange causing inhibiting effects on the fetus.

The law by which fetomaternal exchange is governed is as follows: $Q/T = K \times A(Cm-Ct)/D$ where Q/T is defined as the rate of diffusion, K is the diffusion co-efficient, A is the area of the membrane, D is the thickness of the membrane and (Cm-Ct) is the concentration gradient. This is also known as Fick's Law [8].

Pharmacokinetics of Maternal and Fetal Exchange of Drugs



The above graph depicts when the rate of placental transfer is high. The drug crosses the placenta and reaches the fetus rapidly. The curve overlaps and reaches an equilibrium state because the maternal and fetal plasma protein binding level rapidly rises and reaches concentration equilibrium with each other over time.



This type of graph is obtained when there is no feto maternal exchange. This not possible practically as in reality it does shows some varying degree of exchange. Several reasons like the fetal metabolic status, plasma protein binding capacity, and its excretory function may be cited.



The graph above depicts a case where the maternal fetal exchange is rapid across the placenta. With increase in time fetal concentration of drug and protein may increase suggesting that the fetal concentration of drugs might be higher than that of the mother. This can be attributed to the presence of drugs in the maternal system and lack of maternal renal clearance or in the fetus because of the lack of a developed renal clearance function or hepatic metabolism due to which the fetal drug concentration increases over time.

The placenta favors the entry of lipophilic molecules. Processes like endocytosis, exocytosis, pinocytosis and the free movement of vesicular vehicles also contribute towards the fetomaternal exchange. Low density lipoprotein, transferrin-iron complex, IgG all enters the placenta by mechanisms of endocytosis and exocytosis [45]. Uptake of IgG protein is by a receptor mediated endocytosis and it is transported in vesicles so as to negate the effect of phagolysosomal degradation before it enters the fetus.

Hydrophilic permeability of the lipid insoluble molecules in the absence of transporters is extremely difficult to cross the placental barriers. However presence of extracellular pores can facilitate this transfer of such molecules due to the fact that the surface area is quite small. This phenomenon is termed as Membrane Limited [46–48]. Permeability of hydrophilic substances is also dependant on the diffusion co-efficient of water. Presence of para cellular channels and pores are also sufficient enough to allow the free diffusion of low molecular weight inorganic substances like chloride, phosphate and sulphate ions. These inorganic ionic concentrations are seen to be in a higher concentration in the maternal plasma thereby providing an evidence for transcellular active transport mechanism.

Also it has been seen that stereo-specificity plays a major role in the materno-fetal transfer not only in the case of transfer of amino acids but also in cases of some drugs which might explain the unpredictability that why some drugs in spite of being lipophilic and non ionized find it difficult to cross the placenta. In case of an anthelmintic drug Albendazole Sulfoxide which was administered as a single dose to a sheep during the last trimester, it was found that there was a higher impairment of Albendazole Sulfoxide (+) transfer from the mother to the fetus unlike the Albendazole Sulfoxide (-) [49, 50]. In another a chiral drug, NSAID Ketotifen, which can cross the placenta readily has shown that the S(+) Ketotifen due to its stereoselective protein binding can increasingly cross the placenta unlike R(-) Ketotifen thereby indicating the great variability of transplacental transfer even among the same drugs based on their streochemical properties [51].

Role of Efflux Transporters and Their Level of Expression in the Fetomaternal Barrier during the First and Second Trimesters

In the course of pregnancy the expression of the different active transporters in the placental cellular structure evolves and some of them are found to dominate in the first trimester and others in the second trimester. Still others are expressed continuously throughout the gestational period [52, 53]. High levels of P-gp are expressed and detected on the human and mouse syncytiotrophoblast layer forming the essential part of the hemochorial placental barrier [54-59]. This presence of efflux transporters in the placental barrier plays an important role in the placental resistance in the passage and exchange of drugs and proteins during the first and second trimesters of development. P-glycoprotein (encoded by the ABCB1 gene) and a member of the active binding cassette (ABC) group are found to be expressed on both the maternal and the fetal sides although expression on the maternal villi is more [57].

However as mentioned above, the expression of the various placental transporters varies at different gestational periods and during each trimester of pregnancy [53, 60, 61]. The expression of P-gp efflux transporter protein is observed to be the highest in the first trimester, up to 45-folds higher when compared to the second and third trimesters and it decreases subsequently with increase in gestational time [56], thus suggesting the protective role of P-gp during the early trimester phases and its direct importance in providing a protective mechanism in the early pregnancy stages when the fetus is at its most vulnerable stage due to the occurrence of organogenesis [52]. However expression of an organic anion transporter like OATP2B1 and the breast cancer resistance protein BCRP are more or less consistent throughout the first and second trimesters without any noticeable changes in their expression levels [52]. To compensate for the loss of P-gp expression in the subsequent second trimester it is thought that other efflux transporters such as the Multi drug resistance associated protein 2 (MRP 2) plays an important role as their level of expression increases more towards the end of second trimester [52, 62, 63].

During the first trimester period, the epithelium remains stratified and the thickness of the fetomaternal barrier often exceeds more than 20-30 µm [64]. With increase in gestational time and as the weight of the fetus increases the placental barrier undergoes thinning with most noticeable changes observed in the late second trimester phase and continuing till term pregnancy. The size of the fetomaternal barrier gets reduced to 2-5 µm from 20 to 30 μ m in the first trimester [64]. The reduction in barrier thickness can be attributed to a number of factors in terms of change in morphology and structure like the extension of the chorionic villi capillaries, atrophy and reduction of the continuous arrangement of the cytotrophoblast cells and transformations in the poly nuclear structure of the syncytiotrophoblast. These changes in the fetomaternal barrier have a direct consequence on the transport ability of the drugs and the role of the placenta as a drug metabolizing organ in the later stages of development [65].

The above changes in the placental structure play an important role in affecting the expression of the various ABC and other organic and anionic transporter families. The barrier role of the placenta undergoes subsequent evolutionary and adaptive changes, i.e., from the importance of its protective function to the embryo to the free exchange of selective substances between the mother and the fetus by the end of second trimester [64, 66]. However in the first 10 weeks of the first trimester due to extensive and intensive organogenesis there is a lack of extensive fetomaternal circulation as observed in the case of the second trimester. This does not helps in complete elimination of embryo toxicity and its effects with respect to xenobiotics as some of them depending on their small size can still enter the fetal compartment by ways of bodily fluid tissue via the intercellular spaces [65].

Multi Drug resistance protein 3, whose transportation is still not well documented, has shown to express on the baso-lateral segments of the syncytiotrophoblast plasmalemma and in the villi endothelial cells of the capillaries. It has shown to increase with gestational time with the first trimester showing the lowest expression compared to the second trimester.

Other important transporter members of multi drug resistance are the MDR 1 and 2 efflux proteins. These have also shown to increase their expression with increase in the gestational phase [67]. MRP 5 another sub class of the MRP family has shown its activity to be high during the first trimester as it helps in the regulation of organogenesis and differentiation of the fetus. MRP 7 another member is found in low concentration at the placenta as its cellular localization is yet to be established. Their main role is to help in resisting the entry of anti viral drugs and effluxing cyclic nucleotides, bile acids, conjugated steroids and eicosanoids [65].

Therefore it can be concluded that the functions of the above discussed transporters in the placenta is still not established as also their role in inferring resistance to the placental barrier for the exchange and passage of substances between the maternal and the fetal compartment.

Role of Placenta as an Immunological Barrier during the First and Second Trimesters

Medawar and Billingham during 1953 proposed a hypothesis in a *Nature* article to explain the idea why the fetus is not rejected as a graft by the mother [68]. According to them either the con-

ceptus lacks immunogenicity or there is a significant lowering of the immune response during the onset of pregnancy or the uterus is an immune privileged site like the brain. They also postulated that the presence of a placental barrier might regulate the immune components selectively. After studies were conducted in mouse models by Hoskin and Murgita it was revealed that there were immune reactions against the fetal mouse thereby refuting the theory of lack of immunogenicity in the conceptus [69]. Also the occurrence of ectopic pregnancy further disapproved the claims that the uterus might be an immune privileged organ site. Therefore the remaining two hypotheses, i.e., whether the maternal immune system gets lowered during pregnancy and the presence of the placental barrier controls the immune response were tested and remain under intense scrutiny [70]. Cytokines have also been shown to play an important role in regulating embryo survival apart from maintaining the angiogenesis and vasculogenesis in the initial stages of pregnancy. Due to the production of huge cytokines from the uterine decidual and placental cells an immune response deviation occurs from Th1 to Th2. Although this leaves the mother at of high risk of infection but on the other hand it imparts stability and maintenance of pregnancy by controlling the endocrine system thus promoting the function of trophoblast at the implantation site [71]. The fetal trophoblast plays the major role in evading recognition by the maternal immune system. Formation of the placenta, the organ that feeds the fetus, involves cooperation between maternal NK cells and fetal trophoblast cells that remodels the blood supply. This process and, consequently, human reproductive success, are influenced by polymorphic human leukocyte antigen (HLA)-C ligands and their killer cell immunoglobulin-like receptors [72]. HLA plays a crucial role in the process of implantation [73]. During the time of embryonal development in the first trimester half of the fetus receives fetal antigen from the mother and the rest is from the paternal antigen. Due to this characteristic over expression of HLA-G is found on the trophoblast whereas expression of the HLA-A and B are absent and HLA-C is weakly expressed during the

first trimester [74]. HLA-G has shown to bind to KIRs (killer cell like Ig-like receptors of NK cells) and block the cells' toxicity and infer tolerance to the fetus by reducing the NK cells activity [70]. LIF also synthesized in the first trimester by the deciduas on the maternal part of the placenta plays an important role by binding to LIF receptors; this further helps in trophoblastic growth and differentiation thereby providing fetomaternal tolerance and acting as an immune barrier [75].

In fetuses of the second trimester of pregnancy, the Igs, SC and J chain are located in the syncytio- and cytotrophoblast of the chorion [76]. The villous stroma contains a small amount of different subsets of lymphocytes. Macrophages account for up to 45 % of the stromal cells of the villi and contain IgG and J-chain. In the maternal part of the placenta, the Secretory Immune System proteins (SIS) are in the decidual cells. Relatively few lymphocytes and macrophages are observed in the decidual stroma. A different origin and composition of immunocompetent cells and a different course of immune reactions in fetal and maternal parts of the placenta have been shown [77]. These systems are already in place at the beginning of the embryonic period starting from weeks 3.5-4 to 5 and function during the entire first trimester of pregnancy.

The trophoblastic villi has several protective mechanisms for the fetus in place against the maternal system, for instance, it fails to express MHC class I and II molecules and down regulates maternal NK cells by the expression of non-classical MHC gene [78–81]. Also expression of the complement proteins like CD46, CD55 and CD59 by the surface of the trophoblast helps in protecting the embryo in the first trimester [82, 83].

The Secretory Immune System in the Barrier Structures during the First and Second Trimesters

In the mucosal and systemic compartments of an organism after an antigen expression, the secretory immune system (SIS) participates and helps in imparting immune protection of the organs. Two types of the SIS have been suggested to be present at the border between maternal and embryonic tissues [76, 84]. Therefore it acts like a barrier between the external and the internal environment for the organ [85, 86]. The SIS expressed on the maternal placenta [76] consists of an integrated extensive pathway of lymphoid tissues which are made up of inductive and effector sites for the protection of the host organ against the foreign pathogens/antigens [87]. The fetal part of the placenta also comprises of IgG, IgA, IgM along with trans membrane secretory components (SC), macrophages, J chain, antigen presenting cells and plasma cells [82, 85, 86, 88]. Secretory components are classified as glycoprotein components and are an important receptor site for SIS [78, 89]. In addition to the mucous membranes and glands of the human embryos and fetuses, secretory cells (SC) are also detected in the trophoblast and amnion. During the first trimester at around the fourth week, presence of SC and J chains in an up regulated state are detected in the absence of lymphoid organs in the ectodermal and endodermal structures of the embryo thus helping in functioning as a pathological barrier against the entry of foreign pathogens to the fetal compartment [82]. During this 3-4 weeks time and also during the onset of the second trimester the fetomaternal part of the placenta starts to express all types of components for the secretory immune system SIS [76]. Although SIS is fully expressed during the third trimester of gestation, in the second trimester itself the placenta expresses all the typical components of SIS [90].

Therefore the fetomaternal barrier function can be very much related to that of the mucosal SIS although some exceptions like secretion of IgG at a higher level occurs in the placenta and IgA and IgM to a much lesser extent partly due to the fact that IgG helps in providing the fetal immunity primarily. Unlike the mucosal SIS the placental barrier in the first and second trimesters, does not possess a large surface area [91].

Role of Epigenetic Factors in the Formation of Placenta from the Early Gestation Phase till the First Trimester

The placenta through its blood placental barrier protects the fetus from the maternal immune system. It attacks and helps in the detoxification of harmful drugs and in the secretion of growth factors and pregnancy associated hormones. A major factor in the placental growth and development is its genetic imprintation and how it is controlled [92]. After fertilization, there is a genome wide DNA methylation and remethylation observed in the embryos during the process of blastocyst formation in the first trimester. There is a lack of DNA methylation in the Tropho Ectoderm however [93]. It is also observed that the level of DNA methylation is extremely active in the embryonal formation period rather than the placentation process [93]. In many ways the development of the placenta is related to the properties of tumorogenesis like trophoblast proliferation and migration, invasion of the uterine endometrial walls and escape immune regulation, thereby conferring the fetus an immunological privilege [94]. These all are controlled by DNA methylation and histone modifications of tumor suppressor genes such as Maspin, RASSFIA and APC in human placentas [95]. Novakovic et al. in 2008 also demonstrated through their investigation that during the first trimester there is a methylation dependant reduction in the expression of a small subset of genes during the normal placenta formation process [96]. Also pluripotency of the inner cell mass is maintained by the Histone 3 arginine methylation of the blastomeres [97]. Mutants of the polycomb group family fail to form the chorion and the amnion of the placenta [98, 99].

Is the placenta an important site for epigenetic action? It has been shown that the placenta fails to develop in the first trimester in the absence of Ascl2, Phlda2 and Peg10 genes. Igf2 gene has been shown to play an important role in the development of the selective diffusional placental

capacity thereby raising the inevitable question regarding whether the placenta is a hub for epigenetic actions or not [100–105].

Therefore it can be said that epigenetics which is a relatively new field is gradually getting recognition as an important diagnostic and developmental study tool around the globe for understanding the different transcription factors which are responsible for placental development especially during the first two trimesters as the maximum development of the placenta is associated with those two stages. The epigenetic regulation is a highly dynamic process that starts from the first trimester and continues till fetal development and includes susceptibility in later life [94].

Discussion

The placental barrier which is a highly specialized organ containing the maternal and fetal parts, acts as the main structure providing normal function and development to the fetus [78]. The growth and development of the placenta in the first two trimesters are highly coordinated and regulated by the different transcription factors at the molecular level. The formation of the placental barrier and its intactness in the first two trimesters is meant for selective exchange of substances like removal of carbon dioxide from the fetus and introduction of oxygen through the maternal blood circulation. Glucose, another important molecule, can pass freely via the pores of the syncytiotrophoblast layer via GLUT 1 [106] and during moderate glucose deprivation through the GLUT 3 transport protein [107]. However severe glucose deprivation can have an adverse effect on fetal growth and development.

Another role of the placenta in the first two trimesters is to help in metabolizing harmful substrates that might cross the feto maternal barrier thereby making entry into the fetal compartment [64]. The placenta helps in providing protective mechanisms against entry of xenobiotics, different pathogens and proteins that might cause harm to the fetus during the first and second trimester

as layout of organogenesis is initiated during the initial phases of the first trimester. The placenta also primarily eliminates fetal acid equivalents of respiratory and metabolic origin from the fetus. Passage of IgG protein molecules is initiated even in the presence of strict placental barrier from the first trimester in order to compensate the immature fetal immunity. This transport of IgG is initiated between week numbers 4 and 5 of post conception in the first trimester phase [84]. It has also been claimed that only the IgG molecule can pass across the placenta into the fetus and not other Ig's [108, 109]. However in case of Toxoplasma Gondii, an infectious disease, it has been seen that IgA can also transfer from the mother to the fetus [109, 110]. This once again highlights the very true nature of the fetomaternal barrier which allows the exchange and passage of selective molecules just like the blood brain barrier in the adult system. However any inflammation in the placenta can result in the hyper presence of Ig's in the fetal compartment as the transport of all Ig's will increase respectively. By the time the ninth week is reached, fetuses are able to synthesize their own IgG sub class 1 and 2 [111]. Of late, new findings have emerged on the fetomaternal exchange of fetal stem cells between the mother and the fetus. This phenomenon is known as microchimerism where fetal cells persist in the maternal circulatory system and tissues, playing a major role in healing with response to injury. One such case was reported in a mother suffering from Hepatitis C where the fetal cells aided the maternal system to control her disease. On the other hand maternal chimerism can result in the transmission of cancer, dermatological and systemic auto-immune diseases to children [5].

Conclusion

The placenta functions as a fetomaternal barrier which remains highly selective during the first trimester. With subsequent decrease in the fetomaternal barrier the selectivity also decreases by the end of the second trimester. Also the physical and anatomical appearances of the placental barrier throughout the first and second trimesters are synchronized with fetal development, thereby proving the important point that fetal growth and development during these first two trimesters essentially rely on the presence of the placental barrier, since abnormalities in the placental structure can disrupt the integrity of the fetomaternal barrier leading to abortions or other fetal adverse events in the first and second trimesters.

References

- 1. Placenta, World Publishing Library, World Heritage Encyclopedia.
- 2. Maternal fetal barrier. Pedia View.com. Open source encyclopedia.
- 3. Challier JC. The placental barrier: structure, resistance, asymmetry. Reprod Nutr Dev. 1989;29:703.
- Zorzi W, Thellin O, Coumans B, Melot F, Hennen G, Lakaye B, Igout A, Heinen E. Demonstration of the expression of CD95 ligand transcript and protein in human placenta. Placenta. 1998;19:269.
- Bhattacharya N, Stubblefield P, editors. Frontiers of cord blood science. ISBN: 978-1-84800-166-4 e-ISBN: 978-1-84800-167-1. doi:10.1007/978-1-84800-167-1.
- Singh I. Human embryology. 5th ed. Chapter 4. Formation of germ layers. Madras: McMillan India Press. 1993. p. 42.
- Gilbert SF. Developmental biology. 6th edition. Sunderland (MA): Sinauer Associates; 2000. Early Mammalian Development. Available from: http:// www.ncbi.nlm.nih.gov/books/NBK10052/.
- Polin RA, Fox WW, Abman SH. Fetal and neonatal physiology. vol 2 4th ed. Saunders, Elsevier. Chapter 2 section 11, 2004.
- 9. Fox P, Benirschke K, Kaufmann P, Baergen R. Pathology of the human placenta. 5th ed. New York: Springer; 2006.
- Baergen RN. Manual of the human placenta. Chapter
 2nd ed. Springer 2011.
- 11. Boyd JD, Hamilton WJ. The human placenta. Cambridge: W Heffer & Sons; 1970.
- Kaufmann P, Huppertz B, Frank HG. The fibrinoids of the human placenta: origin, composition and functional relevance. Anat Anz. 1996;178(6):485– 501. Review.
- http://www.embryology.ch/anglais/fplacenta/villosite07.html.
- Gruenwald P. The development of the placental lobular pattern in the human. Review and reinterpretation of the material. Obstet Gynecol. 1977;49(6):728–32. Review.
- Benirschke K, Kaufmann P, Baergen R. Pathology of the human placenta. 5th ed. New York: Springer; 2006.

- Macara LM, Kingdom JCP, Kaufmann P, et al. Control of the fetoplacental circulation. Fetal Matern Med Rev. 1993;5:167.
- Baker PN, Kingdom J. Pre-eclampsia current perspectives on management. London: The Parthenon Publishing Group; 2004.
- Robertson WB, Brosens IA, Dixon HG. Placental bed vessels. Am J Obstet Gynecol. 1973;117: 294–5.
- Roberts JM, Taylor RN, Musci TJ, Rogers GM, Hubel CA, Mc Laughlin MK. Preeclampsia: an endothelial cell disorder. Am J Obstet Gynecol. 1989;161:1200–4.
- Babawale MO, Mobberley MA, Ryder TA, Elder MG, Sullivan MH. Ultrastructure of the early human feto-maternal interface co-cultured in vitro. Hum Reprod. 2002;17:1351–7.
- Lyall F. Priming and remodelling of human placental bed spiral arteries during pregnancy – a review. Placenta. 2005;26(Suppl A):S31–6.
- Reiter RJ, Tan DX, Korkmaz A, RosalesCorral SA. Melatonin and stable circadian rhythms optimize maternal, placental and fetal physiology. Hum Reprod Update. 2013;20(2):293–307. doi:10.1093/ humupd/dmt054. ISSN 13554786.
- 23. Schneider H, Danko J, Huch R, Huch A. Homeostasis of fetal lactate metabolism in late pregnancy and the changes during labor and delivery. Eur J Obstet Gynecol Reprod Biol. 1984;17:183–92.
- Blechner JN. Maternal-fetal acid-base physiology. Clin Obstet Gynecol. 1993;36:3–12.
- Kaufmann P, Mayhew TM, Charnock-Jones DS. Aspects of human fetoplacental vasculogenesis and angiogenesis. II. Changes during normal pregnancy. Placenta. 2004;25:114–26.
- Mulvany MJ, Aalkjaer C. Structure and function of small arteries. Physiol Rev. 1990;70:921–61.
- Myatt L, Brewer AS, Langdon G, Brockman DE. Attenuation of the vasoconstrictor effects of thromboxane and endothelin by nitric oxide in the human fetal placental circulation. Am J Obstet Gynecol. 1992;166:224–30.
- Myatt L, Webster RP. Vascular biology of preeclampsia. J Thromb Haemost. 2009;7:375–84.
- Wang Y, Zhao S. Vascular biology of the placenta. San Rafael: Morgan & Clypool Life Sciences; 2010.
- Chaudhuri G, Furuya K. Endothelium-derived vasoactive substances in fetal placental vessels. Semin Perinatol. 1991;15:63–7.
- Mc Carthy AL, Woolfson RG, Evans BJ, Davies DR, Raju SK, Poston L. Functional characteristics of small placental arteries. Am J Obstet Gynecol. 1994;170:945–51.
- Khalil RA, Granger JP. Vascular mechanisms of increased arterial pressure in preeclampsia: lessons from animal models. Am J Physiol Regul Integr Comp Physiol. 2002;283:R29–45.
- 33. Maigaard S, Forman A, Andersson KE. Relaxant and contractile effects of some amines and prostanoids in myometrial and vascular smooth muscle

within the human uteroplacental unit. Acta Physiol Scand. 1986;128:33–40.

- Allen J, Skajaa K, Maigaard S, Forman A. Effects of vasodilators on isolated human uteroplacental arteries. Obstet Gynecol. 1991;77:765–71.
- Brosens IA, Robertson WB, Dixon HG. The role of the spiral arteries in the pathogenesis of preeclampsia. Obstet Gynecol Annu. 1972;1:177–91.
- 36. Pijnenborg R, Anthony J, Davey DA, Rees A, Tiltman A, Vercruysse L, Van Assche A. Placental bed spiral arteries in the hypertensive disorders of pregnancy. Br J Obstet Gynaecol. 1991;98:648–55.
- Lyall F. Mechanisms regulating cytotrophoblast invasion in normal pregnancy and pre-eclampsia. Aust N Z J Obstet Gynaecol. 2006;46:266–73.
- Huppertz B. The feto-maternal interface: setting the stage for potential immune interactions. Semin Immunopathol. 2007;29:83–94.
- 39. Maynard SE, Min JY, Merchan J, Lim KH, Li J, Mondal S, Libermann TA, Morgan JP, Sellke FW, Stillman IE, Epstein FH, Sukhatme VP, Karumanchi SA. Excess placental soluble fms-like tyrosine kinase 1 (sFlt1) may contribute to endothelial dysfunction, hypertension, and proteinuria in preeclampsia. J Clin Invest. 2003;111:649–58.
- 40. Demir R, Kayisli UA, Seval Y, Celik-Ozenci C, Korgun ET, Demir-Wuesten AY, Huppertz B. Sequential expression of VEGF and its receptors in human placental villi during very early pregnancy: differences between placental vasculogenesis and angiogenesis. Placenta. 2004;25:560–72.
- Kaufmann P, Sen DK, Schweikhart G. Classification of human placental villi. I. Histology. Cell Tissue Res. 1979;200:409–23.
- 42. Pijnenborg R, Robertson WB, Brosens I, Dixon G. Review article: trophoblast invasion and the establishment of haemochorial placentation in man and laboratory animals. Placenta. 1981;2:71–91.
- Huppertz B, Abe E, Murthi P, Nagamatsu T, Szukiewicz D, Salafia C. Placental angiogenesis, maternal and fetal vessels—a workshop report. Placenta. 2007;28(Suppl A):S94–6.
- 44. Goldenberg RL, Iams JD, Miodovnik M, Van Dorsten JP, Thurnau G, Bottoms S, Mercer BM, Meis PJ, Moawad AH, Das A, Caritis SN, Mc Nellis D. The preterm prediction study: risk factors in twin gestations. National Institute of Child Health and Human Development Maternal-Fetal Medicine Units Network. Am J Obstet Gynecol. 1996;175: 1047–53.
- King BF. Absorption of peroxidase-conjugated immunoglobulin G by human placenta: an in vitro study. Placenta. 1982;3:395–406.
- 46. Faber JJ, Thornburg KL. The forces that drive inert solutes from water across the epitheliochorial placenta of the sheep and the goat and the hemochorial placentae of the rabbit and the guinea pig. 1981.
- 47. Sibley CP, Boyd RDH. Control of transfer across the mature placenta. In: Clarke JR, editor. Oxford

reviews of reproductive biology. Oxford: Oxford University Press; 1988. p. 382–435.

- 48. Faber JJ. Diffusional exchange between foetus and mother as a function of the physical properties of diffusing materials. In: Comline KS, Cross KW, Dawes GS, Nathanielsz PW, editors. Fetal and neonatal physiology. Cambridge: Cambridge University Press; 1973. p. 306–27.
- 49. Tran A, O'Mahoney T, Rey E, et al. Vigabatrin: placental transfer in vivo and excretion into breast milk of the enantiomers. Br J Clin Pharmacol. 1998;45:409.
- Capece BP, Pérez R, Andaluz A, et al. Placental transfer of albendazole sulphoxide enantiomers in sheep. Vet J. 2002;163:155.
- Lagrange F, Pehourcq F, Bannwarth B, et al. Passage of S- (+) – and R- (–) – ketotifen across the human isolated perfused placenta. Fundam Clin Pharmacol. 1998;12:286.
- Mathias A, Hitti J, Unadkat J. P-glycoprotein expression in human placenta of various gestational ages. Am J Physiol Regul Integr Comp Physiol. 2005;289:R963–9.
- 53. Patel P, Weeraskekera N, Hitchins M, et al. Semiquantitative expression analysis of MDR3, FIC1, BSEP, OATP-A, OATP-C, OATP-D, OATP-E and NTCP gen transcript in 1st and 3rd trimester human placenta. Placenta. 2003;24:39–44.
- 54. Cordon-Cardo C, O'Brien JP, Casals D, Rittman-Grauer L, Biedler JL, Melamed MR, Bertino JR. Multidrug-resistance gene (P-glycoprotein) is expressed by endothelial cells at blood-brain barrier sites. Proc Natl Acad Sci U S A. 1989;86:695–8.
- 55. Smit JW, Huisman MT, van Tellingen O, Wiltshire HR, Schinkel AH. Absence of pharmacological blocking of placental P-glycoprotein profoundly increases fetal drug exposure. J Clin Invest. 1999;104:1441–7.
- 56. Nakamura Y, Ikeda S-i, Furukawa T, Sumizawa T, Tani A, Akiyama S-i, Nagata Y. Function of P-glycoprotein expressed in placenta and mole. Biochem Biophys Res Commun. 1997;235: 849–53.
- 57. Ushigome F, Takanaga H, Matsuo H, Yanai S, Tsukimori K, Nakano H, Uchiumi T, Nakamura T, Kuwano M, Ohtani H, Sawada Y. Human placental transport of vinblastine, vincristine, digoxin and progesterone: contribution of P-glycoprotein. Eur J Pharmacol. 2000;408:1–10.
- Lankas GR, Wise LD, Cartwright ME, Pippert T, Umbenhauer DR. Placental P-glycoprotein deficiency enhances susceptibility to chemically induced birth defects in mice. Reprod Toxicol. 1998;12:457–63.
- 59. Tanabe M, Ieiri I, Nagata N, Inoue K, Ito S, Kanamori Y, Takahashi M, Kurata Y, Kigawa J, Higuchi S, Terakawa N, Otsubo K. Expression of P-glycoprotein in human placenta: relation to genetic polymorphism of the multidrug resistance (MDR)-1 gene. J Pharmacol Exp Ther. 2001;297:1137–43.

- Unadkat JD, Dahlin A, Vijay S. Placental drug transporters. Curr Drug Metab. 2004;5:125–31.
- 61. Sun M, Kingdom J, Baczyk D, Lye SJ, Matthews SG, Gibb W. Expression of the multidrug resistance P-glycoprotein, (ABCB1 glycoprotein) in the human placenta decreases with advancing gestation. Placenta. 2006;27:602–9.
- Kalabis GM, Kostaki A, Andrews MH, Petropoulos S, Gibb W, Matthews SG. Multidrug resistance phosphoglycoprotein (ABCB1) in the mouse placenta: fetal protection. Biol Reprod. 2005;73:591–7.
- 63. May K, Minarikova V, Linnemann K, Zygmunt M, Kroemer HK, Fusch C, Siegmund W. Role of the multidrug transporter proteins ABCB1 and ABCC2 in the diaplacental transport of talinolol in the term human placenta. Drug Metab Dispos. 2008;36:740–4.
- 64. Syme M, Paxton J, Keelan J. Drug transfer and metabolism by the human placenta. Clin Pharmacokinet. 2004;43:487–514.
- Włoch S, Pałasz A, Kamiński M. Active and passive transport of drugs in the human placenta. Ginekol Pol. 2009;80:772–7.
- Evseenko D, Paxton J, Keelan J. Active transport across the human placenta: impact on drug efficacy and toxicity. Expert Opin Drug Metab Toxicol. 2006;2:51–69.
- 67. Pascolo L, Fernetti C, Pirulli D, et al. Effects of maturation on RNA transcription and protein expression of four MRP genes in human placenta and in behio cells. Biochem Biophys Res Commun. 2003;303:259–65.
- Billingham RE, Medawar PB. 'Actively acquired tolerance' of foreign cells. Nature. 1953;172:603–6.
- Hoskin DW, Murgita RA. Specific maternal antifetal lymphoproliferative responses and their regulation by natural immunosuppressive factors. Clin Exp Immunol. 1989;76:262–7.
- Thellin O, Coumans B, Zorzi W, Igout† A, Heinen E. Tolerance to the foeto-placental 'graft': ten ways to support a child for nine months. Curr Opin Immunol. 2000;12:731–7.
- Saito S. Cytokine network at the feto-maternal interface. J Reprod Immunol. 2000;47:87.
- Parham P. NK cells and trophoblasts: partners in pregnancy. JEM. 2004;200:951.
- Loke YW, King A. Immunology of implantation. Baillieres Best Pract Res Clin Obstet Gynaecol. 2000;14:827.
- 74. King A, Boocock C, Sharley AM, Gardner L, Beretta A, Siccardi AG, Loke YW. Evidence for the expression of HLA-C class I mRNA and protein by human first trimester trophoblast. J Immunol. 1996;156:2068–76.
- Moreau P, Paul P, Rouas-Freiss N, Kirszenbaum M, Dausset J, Carosella ED. Molecular and immunologic aspects of the nonclassical HLA class I antigen HLA-G: evidence for an important role in the maternal tolerance of the fetal allograft. Am J Reprod Immunol. 1998;40:136–44.

- 76. Ben-Hur H, Gurevich P, Berman V, Tchanyshev R, Gurevich E, Zusman I. The secretory immune system as part of the placental barrier in the second trimester of pregnancy in humans. In Vivo. 2001;15:429–39.
- 77. Fox H. Normal and abnormal placentation. In: Reece EA, Hobbins JC, editors. Medicine of the fetus and mother. Philadelphia: Lippincott-Raven Publ; 1999. p. 47.
- Weetman AP. The immunology of pregnancy. Thyroid. 1999;9:643.
- 79. Christiansen OB, Mohapeloa HP, Pedersen B, Rosgaard A. Is the expression of classical HLA class I antigens on trophoblast of importance for human pregnancy? Am J Reprod Immunol. 1998;40:158.
- Hutter H, Hammer A, Dohr G, Hunt JS. HLA expression at the maternal-fetal interface. Dev Immunol. 1998;6:197.
- Bulla R, Bossi F, Radillo O, de Seta F, Tedesco F. Placental trophoblast and endothelial cells as target of maternal immune response. Autoimmunity. 2003;36:11.
- Zusman I, Gurevich P, Ben-Hur H. Immune systems and human intrauterine development. The fetus book. Chapter 1. Transworld Research Network, 2008.
- Brandtzaeg P, Berstad AE, Farstad IN, Haraldsen G, Helgeland L, Jahnsen FL, Johansen FE, Natvig IB, Nilsen EM, Rugtveit J. Mucosal immunity – a major adaptive defense mechanism. Behring Inst Mitt. 1997;98:1.
- 84. Gurevich P, Elhayany A, Ben-Hur H, Moldavsky M, Szvalb S, Zandbank J, Schneider I, Zusman I. An immunohistochemical study of the secretory immune system in human fetal membranes and decidua of the first trimester of pregnancy. Am J Reprod Immunol. 2003;50:13.
- 85. Goldblum RM, Hansen LÅ, Brandtzaeg P. The mucosal defense system. In: Stiehm ER, editor. Immunologic disorders in infants and children. Philadelphia: Saunders Publ. Co.; 1996. p. 159.
- McGhee JR, Kiyono H. The mucosal immune system. In: Paul WE, editor. Fundamental immunology. Philadelphia: Lippincott-Raven Publ; 1999. p. 909.
- Iijima H, Takahashi I, Kiyono H. Mucosal immune network in the gut for the control of infectious diseases. Rev Med Virol. 2001;11:117.
- Brandtzaeg P. Molecular and cellular aspects of secretory immunoglobulin system. Acta Pathol Microbiol Immunol Scand. 1995;103:1.
- Brandtzaeg P. The human intestinal immune system: basic cellular and humoral mechanisms. Baillieres Clin Rheumatol. 1996;10:1.
- Ackerman J, Gonzalez EF, Gilbert-Barness E. Immunological studies of the placenta in maternal connective tissue disease. Pediatr Dev Pathol. 1999;2:19.
- Aherne W, Dunnill MS. Quantitative aspects of placental structure. J Pathol Bacteriol. 1966;91:123.
- 92. Reik W, Walter J. Genomic imprinting: parental influence on the genome. Nat Rev Genet. 2001;2: 21–32.

- Santos F, Hendrich B, Reik W, Dean W. Dynamic reprogramming of DNA methylation in the early mouse embryo. Dev Biol. 2002;241:172–82.
- Nelissen ECM, van Montfoort APA, Dumoulin JCM, Evers JLH. Epigenetics and the placenta. Hum Reprod Update. 2011;17(3):397–417.
- 95. Wong NC, Novakovic B, Weinrich B, Dewi C, Andronikos R, Sibson M, Macrae F, Morley R, Pertile MD, Craig JM, et al. Methylation of the adenomatous polyposis coli (APC) gene in human placenta and hypermethylation in choriocarcinomacells. Cancer Lett. 2008;268:56–62.
- 96. Novakovic B, Rakyan V, Ng HK, Manuelpillai U, Dewi C, Wong NC, Morley R, Down T, Beck S, Craig JM, et al. Specific tumour-associated methylation in normal human term placenta and first-trimester cytotrophoblasts. Mol Hum Reprod. 2008;14: 547–54.
- Torres-Padilla ME, Parfitt DE, Kouzarides T, Zernicka-Goetz M. Histone arginine methylation regulates pluripotency in the early mouse embryo. Nature. 2007;445:214–8.
- O'Carroll D, Erhardt S, Pagani M, Barton SC, Surani MA, Jenuwein T. The polycomb-group gene Ezh2 is required for early mouse development. Mol Cell Biol. 2001;21:4330–6.
- 99. Pasini D, Bracken AP, Jensen MR, Lazzerini Denchi E, Helin K. Suz12 is essential for mouse development and for EZH2 histone methyltransferase activity. Embo J. 2004;23:4061–71.
- Guillemot F, Nagy A, Auerbach A, Rossant J, Joyner AL. Essential role of Mash-2 in extraembryonic development. Nature. 1994;371:333–6.
- 101. Salas M, John R, Saxena A, Barton S, Frank D, Fitzpatrick G, Higgins MJ, Tycko B. Placental growth retardation due to loss of imprinting of Phlda2. Mech Dev. 2004;121:1199–210.
- 102. Angiolini E, Fowden A, Coan P, Sandovici I, Smith P, Dean W, Burton G, Tycko B, Reik W, Sibley C, et al. Regulation of placental efficiency

for nutrient transport by imprinted genes. Placenta. 2006;27:S98–102.

- 103. Ono R, Nakamura K, Inoue K, Naruse M, Usami T, Wakisaka-Saito N, Hino T, Suzuki-Migishima R, Ogonuki N, Miki H, et al. Deletion of Peg10, an imprinted gene acquired from a retrotransposon, causes early embryonic lethality. Nat Genet. 2006;38:101–6.
- 104. Constancia M, Hemberger M, Hughes J, Dean W, Ferguson-Smith A, Fundele R, Stewart F, Kelsey G, Fowden A, Sibley C, et al. Placental-specific IGF-II is a major modulator of placental and fetal growth. Nature. 2002;417:945–8.
- 105. Sibley CP, Coan PM, Ferguson-Smith AC, Dean W, Hughes J, Smith P, Reik W, Burton GJ, Fowden AL, Constancia M. Placental-specific insulin-like growth factor 2 (Igf2) regulates the diffusional exchange characteristics of the mouse placenta. Proc Natl Acad Sci U S A. 2004;101:8204–8.
- Takata K, Hirano H. Mechanism of glucose transport across the human and rat placental barrier: a review. Microsc Res Tech. 1997;38:145.
- 107. Bell AW, Hay Jr WW, Ehrhardt RA. Placental transport of nutrients and its implications for fetal growth. J Reprod Fertil Suppl. 1999;54:401.
- Saji F, Samejima Y, Kamiura S, Koyama M. Dynamics of immunoglobulins at the fetomaternal interface. Rev Reprod. 1999;4:81.
- 109. Jauniaux E, Gulbis B. In vivo investigation of placental transfer early in human pregnancy. Eur J Obstet Gynecol Reprod Biol. 2000;92:45.
- 110. Jauniaux E, Jurkovic D, Gulbis B, Liesnard C, Lees C, Campbell S. Materno-fetal immunoglobulin transfer and passive immunity during the first trimester of human pregnancy. Hum Reprod. 1995;10:3297.
- 111. Okoko BJ, Wesumperuma HL, Fern J, Yamuah LK, Hart CA. The transplacental transfer of IgG subclasses: influence of prematurity and low birthweight in the Gambian population. Ann Trop Paediatr. 2002;22:325.

Part XIV

Surgical Implications of Fetal Development: Up to Second Trimester

Surgery of the Unborn/Fetal Surgery

Mursheed Ali, Priyodorshi Sengupta, and Niranjan Bhattacharya

Introduction

Fetal surgery is part of the expansive scope of surgical systems that are utilized to treat birth defects in the fetal stage. Fetal surgery has improved from a health curiosity to a multidisciplinary specialty for treating various diseases and to improve patient outcomes. There are two predominant types of fetal surgery techniques: one is the open fetal surgery method in which the uterus is opened for operating the fetus; the other technique is fetoscopic surgery, in which incision is minimal. The techniques adopted for this surgery are sonography and fetoscopy. In initial stages of fetal surgery during 1990s only folic acid supplements were prescribed to women considering high risk, which was borne previously with a child by NTD. But, the recent advancements in the fetal surgery techniques allow the physicians to diagnose abnormality of the fetus accurately

e-mail: sanjuktaniranjan@gmail.com

and to treat fetal anomalies with high level maternal safety. Therefore, neonatologist needs to be more familiar with recent techniques which are shaping the field. The purpose of the present review is to examine emerging operative treatments for fetal abnormalities including percutaneous, fetoscopy and open hysterotomy methods that are used for fetal surgery. The effectiveness of these experimental therapies is also analyzed.

Emergence of the Field of Fetal Surgery

In 1963, Liley, a New Zealand perinatologist, successfully initiated the new area of fetal surgery by transfusing blood into the peritoneum of a hydropic fetus affected with severe Rh disease [1]. The complete transfusion, yet, involves straight access to the fetal circulation by hysterectomy [2] or rod lens fetoscopic surgery [3]. A group of surgeons in Nashville during 1999 operated the spina bifida. They performed the surgery for a 147 days old fetus in the womb. The operation was performed by a surgical team who developed a fetal correcting technique in mid-pregnancy by removing the uterus temporarily, then draining it of amniotic fluid, performing a tiny operation, and finally restoring the fetus to the uterus [4].

The first clinical fetoscopic surgeries were done through obstetrical endoscopy, where the outcome of a clinical trial demonstrating fetoscopic laser

M. Ali, MSc • P. Sengupta, MSc, Mphil Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_36

coagulation for chorionic plate vessels proved effective for Twin-to-Twin Transfusion Syndrome (TTTS). At present for severe congenital diaphragmatic hernia, fetal tracheal occlusion surgery has been performed and been proved safe. Minimal access fetal surgery is important in preventing preterm labor which can occur with invasive surgery; it is risky for the mother because there is a risk of preterm prelabour rupture of the membrane (PPROM). This proved to be a bottleneck for further developments. Risto Juhana Rintala et al. in an article in 2014 highlighted the research methods by which valid evidence-based research data is obtained in observational studies, randomized controlled trials, and meta-analyses. Consequently; very few designed trials in neonatal surgical treatment have been performed [5]. Hence valid evidence-based research data are few. However, the authors suggest that the problem of small numbers of patients may be overcome by meta-analyses, multi-center trials, and networking.

Cystic Lung Lesions

Congenital Cystic Adenomatoid Malformation (CCAM) is a condition in which abnormal tissue grows as cysts in a fetus's lungs. This may require open fetal surgery if hydrops appear. Open fetal surgery is a modified type of cesarean where physicians make an incision in the uterus, exposing the fetus's chest and perform surgery to remove the cysts. The fetus is then returned to the womb and the uterus closed and the pregnancy carried out until term. CCAM fetuses who have open fetal surgery have an increased survival rate of 50 - 80 percent. However, it places the mother at a high risk for preterm labor and complications. Cases have been documented at the University of California, San Francisco, Medical Center (UCSF) and Children's Hospital of Philadelphia (CHOP). A group of researchers determined CVR (CAM volume ratio) by measuring 3 dimensions of the CAM using the formula for the volume of an ellipse and dividing by the head circumference to correct for differences in gestational age [6]. The study showed that when the CVR is greater than 1.6, the danger for hydrops is 80 %, and in such cases, fetal intervention may be

necessary (thoracoamniotic shunting, fetal thoracotomy, EXIT delivery). In the appropriate situation, this maternal fetal surgery approach for CCAM is life-saving for the affected fetus with acceptable maternal morbidity risks in the present and future pregnancies [7, 8]. Hydrops can be resected in utero if they are predominantly solid or multicystic. Fetal sacrococcygeal teratoma complicated with progressive high output cardiac failure may benefit from in-utero resection of the tumor [9, 10]. At the point when introducing late in pregnancy they can be resected while on placental flow [11]. Macrocystic masses can be punctured or shunted. The Center for Fetal Diagnosis and Treatment, The Children's Hospital of Philadelphia (CHOP) has done the largest number of highly complex surgical interventions to repair birth defects in the womb, approximately 1000 out of 4,000 fetal surgeries done worldwide, by 2013. As far as they can tell, fetal intercession lessens the CCAM volume by 70 %, turns around hydrops and results in a survival rate of 74 % [12]. This rate has been affirmed by others [13]. Recently there have been studies on the utilization of steroids on the mother. Peranteau et al showed that maternal prenatal administration of Betamethasone (12.5 mg/ day, im, two consecutive days) may have beneficial effects on clinical fetal evolution in cases with massive CCAM (microcystic) with or without hydrops. However, more studies are needed to prove its adequacy [14, 15].

Sacrococcygeal Teratoma

Some sacrococcygeal teratomas (SCT) do not result in pre-birth issues; larger and quickly growing tumors increase the metabolic requirements of the fetus, thereby causing fetal anemia and acting as a vast arteriovenous shunt, in the long run bringing on high-yield cardiac arrest. This prompts polyhydramnios, hydrops and intra-uterine fetal death (intra-uterine fetal demise IUFD). Moreover, the mother may develop mirror syndrome. The groups from UCSF and CHOP showed that fetal hydrops and placentomegaly are indicators of poor outcome [16, 17]. Others have proposed extra prognostic criteria, for example, rate of tumor development taking into account which fetal treatment may be considered [17]. A study conducted by a group at the Hôpital Necker-Enfants Malades AP-HP-Université Paris V on 44 fetuses showed that in prenatally diagnosed sacrococcygeal teratoma (SCT), there is good maternal and perinatal outcome when the tumor diameter less than 10 cm, absent or mild vascularity and slow growth or when it is diameter 10 cm or greater, predominantly cystic lesion with absent or mild vascularity and slow growth although shunting or drainage of the SCT could be necessary. Large fast-growing SCT with rich vascularity is associated with a higher perinatal mortality and morbidity than smaller lesions with mild vascularity [18].

Symptomatic treatments, for example, amniodrainage (polyhydramnios), intra-uterine transfusion (anaemia) and bladder shunting (urinary obstacle) have been reported as well. Open fetal resection of Type 1 or prevalently additional pelvic tumors has been reported in five cases. Mean age at childbirth was 30 weeks and four survived [19]. One survivor had to undergo postnatal treatment of pulmonary metastases of a germ cell tumor and at age of 11 years had no confirmation of illness; however, another patient had significant morbidity, which could most likely be identified with emboli at the time of tumor resection. The other two survivors stayed healthy. There are also reports of less invasive procedures, arresting flow in vessels either by fetoscopic laser [20], interstitial thermocoagulation [21] and also radiofrequency removal [22]. The latter can result in collateral damage of the tissue [23].

A group of researchers have also suggested that in utero embolization with Histoacryl may be a useful tool to reduce cardiac output in large vascular sacrococcygeal teratoma [18, 21, 24].

Myelomeningocoele and Other Open Fetal Surgical Methodologies

Myelomeningocoele (MMC), one of the most common congenital malformations, can result in severe lifelong disabilities, including paraplegia, hydrocephalus, Arnold-Chiari II malformation, incontinence, sexual dysfunction, skeletal deformations, and mental impairment. MMC was the first nonlethal anomaly to be treated by fetal surgery. The first cases of in utero spina bifida repair were performed in 1994 with an endoscopic technique that proved unsatisfactory and was abandoned. In 1998, in utero repair of MMC was performed by hysteromy at the Vanderbilt University and at the Children's Hospital of Philadelphia (CHOP). Adzick reviewed the natural history, pathophysiology, and in-utero intervention for fMMC. The author conceded that little progress has been made in the postnatal surgical management of the child with spina bifida. Postnatal surgery is usually aimed at covering the exposed spinal cord, preventing infection, and treating hydrocephalus with a ventricular shunt. In-utero repair of open spina bifida is now performed in selected patients and presents an additional therapeutic alternative for expectant mothers carrying a fetus with fMMC. Tests and early clinical experience demonstrated that prebirth treatment could enhance result [25]. Observational studies demonstrated that pre-birth microsurgical layered repair turns around hindbrain herniation, diminishes the requirement for shunting, enhances leg and bladder capacity, and in addition later cognitive function [26-28]. The National Institutes of Health has supported the Management of Myelomeningocoele Study (MOMS) randomized trial. The objective of the trial was to evaluate if intrauterine repair of MMC between 19 to 25 weeks gestation improves outcome compared with standard neurosurgical repair. The MOMS trial has shown a significant reduction of VP shunt placement at one year of age following fetal MMC surgery (prenatal group: 40 % vs. postnatal group: 82 %, P<0.001). The trial also demonstrated a substantial improvement in the overall neuromotor function at 30 months of age by a variety of measures including the finding that 42 % in the fetal surgery group were walking independently compared to only 21 % in the postnatal surgery group (P < 0.01). Finally, hindbrain herniation was significantly reversed in the fetal surgery group compared to the postnatal surgery group (no hindbrain herniation in 36 % and 4 % of the infants, respectively, and severe herniation in 6 % and 22 %, respectively, P < 0.001). But it also revealed that fetal MMC surgery increases the risks for spontaneous rupture of membranes (prenatal surgery: 46 % vs. postnatal surgery: 8 %, P<0.001), oligohydramnios (21 % vs. 4 %, P = 0.001), and preterm delivery (79 % vs. 15 %, P<0.001) including 13 % of fetal surgery group that were born before 30 weeks of gestation. This trial is intriguing for some reasons. Initially, it is being conducted only in the US. Second, the trials are helping to bring together contending opinions regarding MMS. The main downside is that it is not open to non-US subjects. Additionally such a trial may stall the advancement of negligibly intrusive methods. However, other trials have provided promising results. For instance, an experiment using in utero repair of congenital myelomeningocele through a hysterotomy appeared to be technically superior to procedures performed endoscopically [29, 30].

Open fetal surgery is substantially less popular in Europe. Most centers limit themselves to procedures performed on placental support. The EXIT (ex utero intrapartum treatment) procedure was at first designed as a delivery technique for safely establishing upper airways following tracheal occlusion [31, 32]. The specialized subtle elements of this treatment modality have been precisely detailed [11, 33], and most fetal surgical centers should be required to have the capacity to offer this treatment modality safely. The time between induction of anesthesia and cord clamping is kept as long as is clinically needed. Utilizing inhalational anesthesia for maximal uterine relaxation, uteroplacental blood stream and gas exchange are maintained and amnioinfusion and incomplete conveyance of the fetus keep uterine volume within normal limits. The list of indications for utilizing this method has developed through the years yet basically incorporates airway obstruction because of laryngeal atresia, vast tumors or iatrogenic tracheal impediment. The procedure may also be used to create vascular access for extracorporeal circulation, such as for managing cardiac defects, severe CDH, lung lesions or conjoined twins [34].

Michael Tchirikov et al. in 2012 investigated the impact of undertaking long-distance air travel to a specialized medical center while pregnant in order to undergo fetoscopic laser coagulation (FLC) for twin-to-twin transfusion syndrome (TTTS) [35]. A retrospective cohort study was conducted on a women having TITS who travelled by land (n=61) or air (n=16) to a surgical center in Germany between 2006 and 2010. All women underwent FLC on arrival at the surgical center. The postoperative survival rate for a single twin was 100.0 % (n=16) in the flight group and 98.3 % in the land transportation group (n=60). The postoperative survival rate for both twins was 81.3 % in the flight group (n=13) and 75.4 % (n=46) in the land transportation group. No differences in neonatal outcome or the rate of adverse effects were observed between the 2 groups. No flight-related pregnancy complications were recorded. It was concluded that the long-distance air travel to a specialized medical center is sufficiently safe to warrant recommendation to pregnant women with TTTS who require FLC.

Recent Advances in the Treatment of Different Fetal Diseases

Yan Chen in 2013 noted that diaphragmatic hernia complicating pregnancy rarely occurs while it is frequently misdiagnosed [36]. A pregnant woman with right upper-quadrant abdominal pain for 4 months was hospitalized because of severe unrelieved abdominal pain. Here a part of omentum and a part of transverse colon were trapped in the thorax through a patient's diaphragm. Emergency caesarean section was done and the trapped intestine, about 40 cm long, released, and a diaphragmatic hernia was repaired at the same time. The author concluded that careful examination and a timely operation are needed to treat diaphragmatic hernia complicating pregnancy.

Salim Al-Gailani in 2014 noted that health organizations and governments all over had adopted policies in order to increase intake of folic acid during the first weeks of pregnancy, which is rich in vitamin B [37]. This folic acid supplementation has been given to reduce the Neural Tube Defects (NTDs). This article focused principally on the mandatory fortification of vitamin grain products and noting how folic acid has been transformed from a routine prenatal supplement to reduce anaemic risk to a pre-conceptional supplement to reduce birth defects. Also folic acid, which was earlier given to 'high-risk' women, is now recommended for all women of childbearing age.

Anke C. Heitkamp, 2015, reported the Splenic Artery Aneurysm (SAA) prevalence which varies from 0.01 to 10.4 %; because of asymptomatic behavior the true prevalence is uncertain [38]. These SAAs occur in younger patients with 58 % prevalence in childbearing women; 95 % diagnosed were pregnant women. The authors reported the case of a 26-year old woman thirtyone weeks pregnant, who suffered a severe abdominal pain while boarding a plane, and an urgent laparotomy had to be performed; a ruptured SAA was found and a splenectomy had also to be performed. But both mother and baby survived. The authors concluded that pregnant women with hypovolemia and acute abdomen, may be suffering from ruptured SAA. Timely multidisciplinary treatment could save the life of the mother and the child.

Discussion

Although it is now accepted that the fetus is also a patient, the specific role of surgery on the unborn as treatment of maladies is yet unclear. For fetendo surgery, it is expected that cases will increase quietly over the next few years and may decline as other approaches develop. Myelomeningocele and TTTS show promise but will be tied up in clinical trials in future. Open fetal surgery cases could double or triple if treatment is proved to be appropriate. Less invasive surgeries can become more common, because these can be mastered by practitioners who will not need specialized open fetal surgery centers. Twin-twin transfusion has the highest volume procedure and requires tertiary/quaternary expertise but if this procedure is proven effective through randomized trials, then more centers can be justified. Several models have been suggested regarding what kind of centers should be set up. The Perinatologists Surgeon Model, best exemplified at Vanderbilt, is based on treating myelomeningocele with a team including a few dedicated perinatologists, neonatologists, nurses and administrators and supportive obstetricians, neurologists and other specialists, each with specific focused duties. This model is effectively easier to implement as development of further new skills can be avoided. In another model, the Virtuoso Model, a talented practitioner uses new minimally invasive (often fetoscopic) techniques. This model requires only a well-known capable individual and can be done in a less intensive environment. What model will be cost effective etc. remains to be seen. Ultimately, trials will determine the future of fetal surgery.

Prenatal conditions will need medical management and rare conditions may need prenatal intervention to save the life of the fetus, or stop permanent organ damage. However, these are uncommon conditions, and extremely few patients might choose for fetal medical care. This implies that there is a case-load limitation.

Suggestions

The main problem is not the background of the specialist, but the to get talented individuals from disparate cultures and various medical disciplines to work together where real problems of the patients can be resolved through consultation and discussion from various angles. It is suggested that collaborative, multi-disciplinary clinical and research papers can be presented to present problems and solutions at the national level.

The controversy surrounding fetal surgery today revolves around what is ideal quality and how this can be ensured, versus sufficient quantities of patients. Here the issue of advancement and option approaches again becomes pertinent. The suggestion here is to harmonize the two approaches. This is a new exciting area where researchers and clinicians can work together for the advancement of medical practice.

Conclusion

There are a series of health issues that come under the scanner when a fetus is recognized as a patient, for instance, early recognition of a disease or defect, reducing the risk of anaemia, guidelines for prenatal nutrition, avoiding risk-aversive behaviours, overcoming genetic disabilities, encouraging newborn screening for phenylketonuria, finding out babies with CDH, diagnosing diaphragmatic hernia before caesarean section, reducing the risk of certain cancers, immediate ambulance transfer with patient informaemergency tion to the department, multidisciplinary treatment by high skilled surgeons, to name a few.

The difficulties involved in infant and medical specialty surgery, its infrequency and the small numbers involved makes analysis of the process and consequently solely a also, very few well-designed have been performed so far. Surgery is usually an invasive procedure, and therefore many prefer different alternative approaches. But surgery on the unborn holds promise and the need of the day is to focus on how this surgical technique can be improved. It was not so long back that such surgery was unheard of; only the future will unfold its full potential.

References

- Liley AW. Intrauterine transfusion of foetus in haemolytic disease. Br Med J. 1963;2(5365):1107–9.
- 2. Adamsons K. Fetal surgery. N Engl J Med. 1966;275:204–6.
- Rodeck C, Nicolaides K, Warsof S, Fysh W, Gamsu H, Kemp J. The management of severe rhesus isoimmunization by fetoscopic intravascular transfusions. Am J Obstet Gynecol. 1984;150(6):769–74.
- Bartoo C. Vanderbilt-pioneered fetal surgery procedure yields positive results. 2011 [cited]. Available from: http://news.vanderbilt.edu/2011/02/vanderbiltpioneered-fetal-surgery-procedure-yields-positiveresults/.

- Rintala RJ, Pakarinen MP, Koivusalo AI. Neonatal surgery: towards evidence-based practice and management. Semin Pediatr Surg. 2014;23(5):303–8. Elsevier; 2014.
- Crombleholme TM, Coleman B, Hedrick H, Liechty K, Howell L, Flake AW, et al. Cystic adenomatoid malformation volume ratio predicts outcome in prenatally diagnosed cystic adenomatoid malformation of the lung. J Pediatr Surg. 2002;37(3):331–8.
- Davenport M, Warne S, Cacciaguerra S, Patel S, Greenough A, Nicolaides K. Current outcome of antenally diagnosed cystic lung disease. J Pediatr Surg. 2004;39(4):549–56.
- Wilson RD, Hedrick HL, Liechty KW, Flake AW, Johnson MP, Bebbington M, et al. Cystic adenomatoid malformation of the lung: review of genetics, prenatal diagnosis, and in utero treatment. Am J Med Genet A. 2006;140A(2):151–5.
- 9. Adzick NS. Management of fetal lung lesions. Clin Perinatol. 2003;30(3):481–92.
- Adzick NS. Open fetal surgery for life-threatening fetal anomalies. Semin Fetal Neonatal Med. 2010;15(1):1–8. Elsevier; 2010a.
- Liechty KW. Ex-utero intrapartum therapy. Semin Fetal Neonatal Med. 2010;15(1):34–9. Elsevier; 2010.
- Wilson R, Baxter J, Johnson M, King M, Kasperski S, Crombleholme T, et al. Thoracoamniotic shunts: fetal treatment of pleural effusions and congenital cystic adenomatoid malformations. Fetal Diagn Ther. 2004;19(5):413.
- Knox E, Kilby M, Martin W, Khan K. In-utero pulmonary drainage in the management of primary hydrothorax and congenital cystic lung lesion: a systematic review. Ultrasound Obstet Gynecol. 2006;28(5): 726–34.
- Tsao K, Hawgood S, Vu L, Hirose S, Sydorak R, Albanese CT, et al. Resolution of hydrops fetalis in congenital cystic adenomatoid malformation after prenatal steroid therapy. J Pediatr Surg. 2003;38(3):508–10.
- 15. Peranteau W, Wilson R, Liechty K, Johnson M, Bebbington M, Hedrick H, et al. Effect of maternal betamethasone administration on prenatal congenital cystic adenomatoid malformation growth and fetal survival. Fetal Diagn Ther. 2007;22(5):365–71.
- Bond SJ, Harrison MR, Schmidt KG, Silverman NH, Flake AW, Slotnick RN, et al. Death due to highoutput cardiac failure in fetal sacrococcygeal teratoma. J Pediatr Surg. 1990;25(12):1287–91.
- Westerburg B, Feldstein VA, Sandberg PL, Lopoo JB, Harrison MR, Albanese CT. Sonographic prognostic factors in fetuses with sacrococcygeal teratoma. J Pediatr Surg. 2000;35(2):322–6.
- Benachi A, Durin L, Maurer SV, Aubry M-C, Parat S, Herlicoviez M, et al. Prenatally diagnosed sacrococcygeal teratoma: a prognostic classification. J Pediatr Surg. 2006;41(9):1517–21.
- Hedrick HL, Flake AW, Crombleholme TM, Howell LJ, Johnson MP, Wilson RD, et al. Sacrococcygeal teratoma: prenatal assessment, fetal intervention, and outcome. J Pediatr Surg. 2004;39(3):430–8.

- Hecher K, Reinhold U, Gbur K, Hackelöer BJ. Unterbrechung des umbilikalen Blutflusses bei einem akardischen Zwilling durch endoskopische Laserkoagulation. Geburtshilfe Frauenheilkd. 1996;56(02):97–100. 18.03.2008.
- Makin EC, Hyett J, Ade-Ajayi N, Patel S, Nicolaides K, Davenport M. Outcome of antenatally diagnosed sacrococcygeal teratomas: single-center experience (1993-2004). J Pediatr Surg. 2006;41(2):388–93.
- Hang Lam Y, Hoi Yin Tang M, Wai Hung Shek T. Thermocoagulation of fetal sacrococcygeal teratoma. Prenat Diagn. 2002;22(2):99–101.
- Paek BW, Jennings RW, Harrison MR, Filly RA, Tacy TA, Farmer DL, et al. Radiofrequency ablation of human fetal sacrococcygeal teratoma. Am J Obstet Gynecol. 2001;184(3):503–7.
- 24. Perrotin F, Herbreteau D, Machet M, Potin J, Lardy H, Arbeille P. OP06. 20: in utero Doppler ultrasound-guided embolization for the treatment of a large, vascular sacrococcygeal teratoma causing fetal hydrops. Ultrasound Obstet Gynecol. 2006;28(4):458–9.
- Adzick NS. Fetal myelomeningocele: natural history, pathophysiology, and in-utero intervention. Semin Fetal Neonatal Med. 2010;15(1):9–14. Elsevier; 2010b. doi:10.1016/j.siny.2009.05.002.
- Bruner JP, Tulipan N, Paschall RL, et al. FEtal surgery for myelomeningocele and the incidence of shunt-dependent hydrocephalus. JAMA. 1999;282(19):1819–25.
- 27. Stirnemann JJ, Nasr B, Quarello E, Ortqvist L, Nassar M, Bernard J-P, et al. A definition of selectivity in laser coagulation of chorionic plate anastomoses in twin-to-twin transfusion syndrome and its relationship to perinatal outcome. Am J Obstet Gynecol. 1999;198(1):62.e1–e6.
- Danzer E, Gerdes M, Zarnow DM, Bebbington M, Adzick NS, Johnson M. Preschool neurodevelopmental outcome of children following fetal myelomeningocele closure. Am J Obstet Gynecol. 2008;199(6):S15. http://www.ncbi.nlm.nih.gov/ pubmed/20347433.
- Bruner JP, Tulipan NB, Richards WO, Walsh WF, Boehm FH, Vrabcak EK. In utero repair of

myelomeningocele: a comparison of endoscopy and hysterotomy. Fetal Diagn Ther. 2000;15(2):83–8.

- 30. Kohl T, Hering R, Heep A, Schaller C, Meyer B, Greive C, et al. Percutaneous fetoscopic patch coverage of spina bifida aperta in the human early clinical experience and potential. Fetal Diagn Ther. 2006;21(2): 185–93.
- 31. Liechty KW, Crombleholme TM, Flake AW, Morgan MA, Kurth CD, Hubbard AM, et al. Intrapartum airway management for giant fetal neck masses: the EXIT (ex utero intrapartum treatment) procedure. Am J Obstet Gynecol. 1997;177(4):870–4.
- Mychaliska GB, Bealer JF, Graf JL, Rosen MA, Adzick NS, Harrison MR. Operating on placental support: the ex utero intrapartum treatment procedure. J Pediatr Surg. 1997;32(2):227–31.
- Bouchard S, Johnson MP, Flake AW, Howell LJ, Myers LB, Adzick NS, et al. The EXIT procedure: experience and outcome in 31 cases. J Pediatr Surg. 2002;37(3):418–26.
- 34. Kunisaki SM, Barnewolt CE, Estroff JA, Myers LB, Fauza DO, Wilkins-Haug LE, et al. Ex utero intrapartum treatment with extracorporeal membrane oxygenation for severe congenital diaphragmatic hernia. J Pediatr Surg. 2007;42(1):98–106.
- 35. Tchirikov M, Oshovskyy V, Steetskamp J, Thäle V. Neonatal outcome following long-distance air travel for fetoscopic laser coagulation treatment of twin-to-twin transfusion syndrome. Int J Gynecol Obstet. 2012;117(3):260–3.
- 36. Chen Y, Bai J, Guo Y, Zhang G. The simultaneous repair of an irreducible diaphragmatic hernia while carrying out a cesarean section. Int J Surg Case Rep. 2013;4(9):771–2.
- Al-Gailani S. Making birth defects 'preventable': preconceptional vitamin supplements and the politics of risk reduction. Stud Hist Philos Sci C Stud Hist Philos Biol Biomed Sci. 2014;47:278–89.
- Heitkamp AC, Dickhoff C, Nederhoed JH, Franschman G, de Vries JI. Saved from a fatal flight: a ruptured splenic artery aneurysm in a pregnant woman. Int J Surg Case Rep. 2015;8C:32–4.
Reflections on Reality: Towards Scarless Healing: Déjà Vu

37

Andrew Burd

For this Chapter in this Book the story of the journey towards scarless healing should begin with the birth of Ian Donald on the 27th of December, 1910. He was born in Liskeard, an old market town located on the southern edge of Bodmin Moor in South West England. The First World War was just around the corner and within the span of one generation the world was going to be again involved in a further struggle for dominance and control. Such conflicts bring to the fore the need for science to address urgent problems of survival and there are tremendous ethical and moral issues at stake. The science of war is the science of death and destruction but also of defence and protection and it was the science of war that brought salvation to that beleaguered island in the North East of Europe; it was the use of ultrasound, 'sonar' that allowed the navel protection vessels for the vital North Atlantic convoys to detect the most formidable underwater fighters, the German U-boats and radar, a defence against the most powerful aerial fighting force the world had ever assembled at that time, the German Luftwaffe. Radar was the pioneered by a brilliant and perhaps somewhat eccentric scientist Robert Watson-Watt who was convinced that

Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Calcutta, India e-mail: darburd@gmail.com radio waves could be used to detect thunderstorms. Time and circumstances forced him to change his focus from thunderstorms to airplanes.

There were many others who were looking at these new technologies of sonar and radar and wondered about their potential medical applications and one of them was a doctor who was specialising in a field diametrically opposite from war, he was an obstetrician concerned about bringing healthy and live babies into the world. In 1954 Ian Donald, now 44 years of age, became the Regius Professor of Obstetrics and Gynaecology at the University of Glasgow. After much trial and experimentation he reported on the first diagnostic ultrasound machine in the Lancet in 1958 [94]. This new imaging device became rapidly and widely popular and a rapidly expanding field of research and development produced a stream of new machines for clinical application. Dr Joseph Woo has written the three part short history of the development of ultrasound in Obstetrics and Gynaecology and it is clear that by the 1970s the technology and its application was having a profound effect on the practice of obstetrics [104].

In 1975 another classic paper appeared in the Lancet describing the diagnosis of spina bifida at 17 weeks of gestation [92]. With the development of high definition ultrasound it was becoming increasingly possible to identify anatomic anomalies that were compatible with life in utero but were

A. Burd, MD, FRCS

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_37

fatal upon delivery. A classic example would be the congenital diaphragmatic hernia, which occupies the thoracic cavity in utero [36]. This does not affect the fetus which receives oxygenated blood via the placenta. However at birth the new-born baby would not be able to inflate the lungs which would have been aplastic and have no space to expand.

Other conditions could be detected such as a urethral stricture, which would obstruct the passage of the fetal urine. Backpressure in the urinary system would prevent the renal parenchyma from developing which again would not be compatible with neonatal existence. Early identification of closed space pressure problems led to a whole new field of interventional decompressing procedures that could be performed percutaneously with ultrasound guidance. This did not address the more complex anatomical anomalies that required direct surgical intervention. This was not long in coming. The real challenge was to be able to expose the fetus through a maternal laparotomy and then hysterotomy; perform the corrective surgery and return the fetus to the uterus without initiating a spontaneous abortion.

This required extensive animal experimentation and it was in on April 26, 1981, that the first human open fetal surgery in the world was performed at University of California, San Francisco under the direction of Dr. Michael Harrison. This was a case of congenital hydronephrosis. As experience accumulated more and more complex procedures were performed but the inherent risks of interrupting a healthy and viable pregnancy, restricted the indications to life-threatening fetal anomalies. There was however a clinical observation associated with those who had undergone fetal surgery and successfully survived to term. The relevance and impact of this clinical observation cannot be underestimated in terms of the way it continues to drive a mighty sector of the medical research field. And what was the observation? The clinical observation? The wounds created surgically in the 21 week old fetus could not be found when the full term baby was delivered.

This was not an animal model. This was real and very relevant. The outcome of post-natal skin wound healing following surgery could always be identified, no matter how many years later, by the outcome of post-natal wound healing. A scar. Now scars are having a bit of a bad press in Science these days and yes indeed they are involved in a whole host of pathological processes. The problem is that scar tissue is a vital biological response to a defect in tissue integrity. You do not have the safety and the comfort, and the time that you have in the womb. The 'real' world is far more threatening and so wound closure is a priority. Now, a defect in the skin? How does that heal? Well the first point is to acknowledge that the skin is a highly complex, multifunctional organ that has a number of functional layers [41, 42]. The first and foremost of these layers is the stratum corneum. This is a layer of dead keratinocytes that provide a formidable barrier between the inner biological world and the external environment that presents far more than just biological risks. But the stratum corneum does not just happen. It is the end stage of keratinocyte differentiation, а terminal differentiation. Keratinocytes are highly organized cells and they need a specific architectural contiguity to perform their primary role, producing keratins and ultimately the stratum corneum.

So if a gap appears in the epidermis, a gap in terms of a discontinuity of functional integrity of the outer layer of the skin, the response will be to rapidly spread a new contact layer of specialised basal keratinocytes. Invading transit amplifying cells or in situ proliferation of basal cells will cover these. The horizontal cellular proliferation is halted by horizontal contact inhibition at which point the outward terminal differentiation occurs which is part of the fundamental programming of the keratinocyte which works on a regular cycle throughout life [5]. Now the end result of this rapid healing of the epidermis with restoration of the vital barrier function is indeed a structural and functional regeneration of the original epidermal architecture. Apart from slight and usually temporary disturbance in the melanocyte- keratinocyte unit organization, which may cause some pigmentary anomalies, there is no other indication that a wound has ever been present.

Postnatal wound healing of the dermis is a completely different biological process. It is absolutely reasonable to expect something very different as the dermis has very different functions from the epidermis. It has a major biomechanical role reflected in a complex dermal architecture that provides the property of plasticity, to the skin. In terms of survival though, the barrier function of the epidermis is both key and critical so that has to be the priority. The epidermis needs a good connective tissue support and this is provided by rapid reconstitution of a supporting "dermal-like' layer which is primarily composed of the fibrillar collagens Type III and Type I [19]. These are assembled into connecting three-dimensional configurations that in the uninjured dermis support the particular properties of the dermis where the matrix, is the key and the cells are there to provide materials. When the dermis has been injured, restoring the functional integrity of the dermis is paramount and speed trumps organization. And nature did as nature does and made a compromise. I can give you rapid healing by a process we shall call repair. Repair is distinct from regeneration in that it is a healing that results in a permanent focal area of connective tissue that does not have the characteristic features of the uninjured tissue. Of course in the skin this would also include the lack of epidermal appendages, sweat glands, hair follicles, sebaceous glands that are found in Normal skin. It is this 'abnormal' region that is called scar tissue [22].

Scars come in all different shapes and sizes and have a considerable range of severity with some pathological scar types recognized, principally hypertrophic and keloid. It is important as a doctor and a scientist to acknowledge the very positive survival benefits of scar tissue formation but if you could have the same benefits without the scar? It is that goal that drives so many to try to unlock the secrets of scarless healing. It is not all humanity and compassion and if, there were a simple secret that could be elucidated, patented and commercialised the financial return would be beyond imagination. And both fame and fortune are powerful motivators for individuals working in the field of fundamental biological research.

Scarless healing has been called many things. In a review published this year it was referred to as the Holy Grail [100]. I really do not think that is a very appropriate term and at this level of science, terms, are important. The "Holy Grail" is the original chalice, or cup, that Jesus and his disciples drank from for the very last time. The night he was betrayed. The last supper. So there is a mythical, mystical link with that chalice and generations have searched for, but have failed to find it. Does it even exist?

It did. There was some receptacle used for the wine that was shared at the Last Supper. But the important point for our younger scientists who still have some neural plasticity is to appreciate that it did indeed exist. That is a matter of logic, not fact. We can't find it. Does that really matter? But there is more to it than just the semantics. It is the very limitation set upon the scientific community in the 1970s by the emerging multi-billion dollar biomedical research industry. The motivation was, too often, a romantic fairy tale of how science works. Long nights in the laboratory, repeating experiments in true reductive style, again and again, and then one day: a breakthrough. A molecule. The key to unimaginable fortune. The Corporate world has bought up so many patents based on this naively simple concept of how nature works. This was a dark time in modern science when ethics became very edgy. Retraction of papers from the world's leading scientific journals, back peddling in predictions and failures of business plan after business plan. Unit pricing of drugs, dressings devices were set at levels that assumed the funding available for purchase would reflect the need for the product. Life is not like that; no economy can support unlimited health funding.

The seduction of venture capitalists on the basis of a complete and utter fantasy really shows how divorced science had become in terms of its relevance to real life. What is this 'scarless healing' that had caught the imagination of so many scientists? In reality it is a clinical observation that has remained a constant throughout from 1985 to 2015. What about the Science? What indeed? When a science is corrupted what reality does it present? The age of the reductionism is over. Consider for the moment the logistics of preparing a baby for birth; the controls, the checks, the lists, the preparations that need to be assessed, reassessed at local, regional and systemic levels. I am of course referring to the biological processes of foetal growth and maturation which are a reflection of the complicated

470

nature and the biology of life. We are now in the era of 'Big Data' but we also need a more pragmatic attitude towards "truth". We still, we must, relay on facts. But the absence of a fact does not deny the presence of a possibility. Hence the Holy Grail once existed, but does it exist now?

Scarless Healing is no Holy Grail. It is a biological phenomenon which does not have a historical context out with its own intrinsic maturation. And logic and scientific evidence support the hypothesis that the scarless healing of the human fetus in the second trimester of pregnancy is an autonomous response of a more socially cohesive tissue environment where local factors, matrix factors and the cell-matrix interactions are primarily involved with the local processes of tissue generation which are perceived to be regeneration. A regeneration that is so perfect that no scar is evident within a few months.

What follows are sections of a thesis titled, "Towards Scarless Healing". This was presented for an MD degree (a higher research degree) at the University of Aberdeen in 1995. It covers work undertaken in the late 1980s and reflects the state of research and understanding about the clinical phenomenon of scarless healing in the human fetus following surgery at that time. It is a salutary lesson to look at this in the context of where we are now. In essence we are no further forward but we are beginning to realise that something as ubiquitous as scarring is going to be linked to some very basic but biologically, hugely complex, survival mechanisms. It is in this context that we are now viewing a new era of medicine which is focussed on regeneration through a biologically programmed approach using stem cell and stem cell based therapies. This is going to be a great challenge for the Health Industry as it grapples with the definitions both of pathology and of therapy based on outcomes.

Introduction and Overview

A wound is a discontinuity in tissue integrity. Healing is the process of restoring that integrity.

The specific focus of this thesis is directed at the phenomenon of scarring which, in the post-natal wound, is a constant sequel of the healing process when the integrity of the dermis is breached and dermal continuity has to be restored. The inevitability of scarring, in the post-natal wound, is now under intense investigation primarily due to the observation that foetal wounds, surgically repaired and allowed to heal in utero, can heal without scarring. In this introductory chapter the structure of skin is briefly outlined, followed by an overview of clinical skin wound healing. The major extracellular components of skin wound healing relevant to this thesis are described and finally foetal wound healing is introduced.

The experimental sections of this thesis describe work which covered a period of intensive investigation into the major differences between foetal and post natal wound healing and falls into three stages; the first two stages address specific hypotheses and the third part presents a new hypothesis for further investigation. The two hypotheses which are directly examined are:

- The scarless healing observed in foetal animals is due to the lack of new collagen deposition in foetal wound healing [47]
- 2. The scarless healing observed in foetal animals is due to the hyaluronic acid rich environment which allows for more effective remodelling of the repair tissue [26].

Skin

Skin is a complex, bilaminar structure, comprising a predominantly cellular superficial layer, the epidermis, overlying a supporting layer of dense fibroelastic connective tissue, the dermis (reviewed in Odland [68]) (Fig.37.1).

The dermis is composed of collagen fibres, elastic fibres and an interfibrillar gel of glycosaminoglycans, salts and water. The principal cell of the dermis is the **Fibroblast** which synthesises and organises the major structural elements [30]. **Collagen**, which makes up nearly 80 % of the dry, fat free, weight of skin, accounts for the tensile strength of the dermal fabric [87]. **Type I** col-



lagen is the major collagen in the dermis with **Type III** constituting approximately 15 %. Interwoven among the bundles of collagen is a network of elastic fibres which restore the normal fibrous array following its deformation by external mechanical forces (Fig. 37.2).

The dermis is arbitrarily divided into two anatomical regions, the papillary and the reticular dermis. The thinner of these is the papillary dermis which is the outermost portion of the dermal connective tissue. It is moulded against the overlying epidermis and accordingly is thrown into surface contours of papillae and folds conforming to the basal epithelial ridges and grooves of the epidermis [6–8]. The papillary dermis contains smaller and more loosely distributed elastic and collagen fibrils than does the reticular dermis, and has a substantially greater proportion of interfibrillar gel and connective tissue cells in addition to enclosing the vast microcirculatory and lymphatic plexuses just beneath the papillary projections of the dermis. The collagen fibres of the papillary dermis are 0.3–3 mm in diameter [9]. The second region of the dermis, the reticular dermis, constitutes the greater bulk and lies beneath the papillary dermis. It is relatively acellular and avascular containing bundles of elastin and collagen fibres. The collagen fibres are considerably larger than in



the papillary dermis (10–40 mm diameter) and a proportionately smaller amount of interfibrillar gel exists.

The major macromolecular component of the non-collagenous extracellular matrix is the broad class of anionic polysaccharides [24]. These compounds, glycosaminoglycans (GAG) and proteoglycans (PG), form a diverse group of heterogeneous glycoconjugates which are formed by a protein core to which different GAG chains and N- and/or 0-linked oligosaccharides are covalently attached. GAG chains present in the ground substance of skin consists mainly of hyaluronic acid, dermatan sulphate (and smaller amounts of chondroitin-6-sulphate), heparin and heparan sulphate.

These substance play an important role in maintaining salt and water distribution; PGs control a "domain," so that a volume of water as much as 1,000 times the volume of the PG can be contained within the molecule. GAGs consist of hexosamine (either glucosamine or galactosamine), alternating with acidic sugars (glucuronic or iduronic acid). These extracellular matrix molecules are metabolised and degraded by fibroblasts and mast cells. In the dermis they exist as a viscoelastic sol-gel of tangled polymers.

Wound Healing

In the skin, there are **two basic elements to healing**;

The restoration of

- (i) Cover and
- (ii) Support.

New cover is achieved by re-epithelialization, new support is achieved by the deposition of a collagen rich connective tissue matrix which undergoes subsequent remodelling and results in what is recognised histomorphologically as scar tissue.

Dermal repair involves a fibroproliferative process with the re-establishment of structural integrity by the deposition of connective tissue. Classically, repair, in this context, involves three overlapping phases: (i) the inflammatory phase (ii) the phase of granulation tissue formation (iii) the phase of matrix deposition and remodelling (Fig. 37.3).

The Inflammatory Phase

This phase is characterised by a sequence of overlapping cellular events in which the breach in tissue integrity is recognised, and attempts are made to contain and limit damaged tissue before the process of repair and reconstruction commences (reviewed in Clarke [23]). Blood vessel disruption causes:

- (i) Extravasation of blood constituents and platelet aggregation
- (ii) Blood coagulation
- (iii) Bradykinin production
- (iv) Complement-derived anaphylotoxin production (these cause the release of histamine and other granule constituents from mast cells resulting in vasodilatation) as well as having a chemotactic effect on neutrophils and monocytes.

The individual cellular elements will be described sequentially beginning with the platelets. These appear in the wound after the disruption of blood vessels results in the extravasation of blood constituents and platelet aggregation.

- (i) Platelets are anucleate, discoid fragments approximately 2 mm in diameter derived from marrow megakaryocytes. They contain several distinct types of storage organelles containing substances which can initiate and influence the process of repair. The main function of the platelet is to initiate haemostasis by forming and maintaining a cellular plug at the site of vascular injury. Platelets adhere to the exposed sub-endothelial collagen in the vessel walls and in doing so become activated, releasing mediators that promote further aggregation and vasoconstriction. Among these mediators is plateletderived growth factor (PDGF) which is stored in the alpha granules.
- (ii) Neutrophils are the first leukocytes seen in the area of inflammation and injury. Despite their early and abundant presence it appears that neutrophils do not play a central or essential role in the wound healing process. This was demonstrated by the effective healing of wounds in guinea pigs rendered aneutropenic by the administration of antineutrophil serum [78].

(iii) Macrophages. Tissue macrophages are largely derived from circulating monocytes of bone marrow origin. The reason for the sequential migration of monocytes into the tissues after the neutrophil infiltration is still a matter of conjecture. Several possible mechanisms have been proposed to account for the apparent sequence: it may be that both cell types respond to the same chemotactic factors but that monocytes are slower overall, either in accumulating in the general area of the wound or specifically in migrating through the walls of the blood vessels and extracellular matrix, alternatively monocyte migration occurs later and is due to different stimuli, possibly initiated, facilitated or amplified as a result of earlier neutrophil migration.

Whatever the case it is apparent that macrophages from inflammatory sites display features of activation, tending to be larger than resident macrophages and having more secondary lysosomes, vacuoles and mitochondria. For many years it was thought that the primary function of macrophages was the removal and degradation of injured tissues prior to the onset of repair. Studies of wound healing in guinea pigs, however, have shown that the depletion of circulating monocytes and local tissue macrophages leads to a severe retardation in tissue debridement and a marked delay in fibroblast proliferation and connective tissue deposition [46, 52]. It is now apparent that wound macrophages secrete growth factors that stimulate the proliferation of fibroblasts, smooth muscle cells and endothelial cells [53]. They are therefore critical in the transition from the phase of inflammation to the phase of granulation tissue production.

Wound macrophages phagocytose and digest organisms, they scavenge tissue debris and effete PNM's in addition they release a large number of active substances:

- (a) To recruit more inflammatory cells and help in tissue debridement and
- (b) To initiate the formation of granulation tissue.

(iv) **Lymphocytes**. The role of lymphocytes in wound healing is unclear.

Lymphocytes are present within wounds and they do produce a wide range of lymphokines which have particular effects on macrophage and fibroblast function. Recent interest has focused on the possibility that lymphocytes and their products may be involved in abnormal wound healing seen in keloid and hypertrophic scars [61].

(v) **Eosinophils**. These cells are also seen in wounds and again their role is unclear.

Granulation Tissue Formation

Granulation tissue consists of abundant macrophages, fibroblasts, new vessels and a loose matrix of collagen, fibronectin and hyaluronic acid. Fibroblasts proliferate and migrate into the wound space. The cells alter in shape, deposit extracellular matrix materials, proteoglycans and collagen and become increasingly mobile. They form cell-cell and cell-matrix links and generate tension which results in reorganisation and contraction of the wound matrix.

Angiogenesis occurs at the same time as fibroplasia. New vessels bud from pre existing ones in a process involving a number of steps:

- (a) There is enzymatic degradation of basement membrane of the parent vessel.
- (b) Endothelial cells migrate towards the angiogenic stimulus (fibronectin and tumour necrosis factor (TNF).
- (c) There is proliferation of endothelial cells behind the migrating cells.
- (d) The endothelial cells mature and organise into capillary tubes.

The new vessels have leaky inter endothelial junctions and proteins and erythrocytes pass into the extravascular space.

Matrix Remodelling

The third and final stage of wound healing begins at the same time as granulation tissue formation. The granulation tissue is removed and the matrix altered to remove fairly rapidly most of the fibronectin present. Collagenase from fibroblasts, macrophages and neutrophils helps with the remodelling. There is a slow accumulation of large bundles of **Type I** collagen and an increase in the tensile strength in the residual scar. The numbers of active fibroblasts and new vessels decrease; many blood vessels thrombose and degenerate, the cells are resorbed and digested by macrophages.

The end result of healing is a scar composed of inactive-looking, spindle-shaped fibroblasts, dense collagen, and fragments of elastic tissue, extracellular matrix and relatively few vessels.

Extracellular Components of Repair

The foregoing account has briefly reviewed the phenomenon of wound healing. It is obviously a complex process and the mechanisms of repair in the post-natal wound are far from clear.

In this third section of this chapter I shall give an overview of the extracellular components of particular relevance to this thesis:

- Growth factors
- Collagen
- Hyaluronic acid

Growth Factors

In populations of normal cells, growth is a balance between growth stimulators and inhibitors. These can lead to a shortened cell cycle or decreased cell loss. There are an increasing number of peptide regulatory factors which have been identified in the serum and produced by cells which affect cell proliferation, metabolism and movement [67, 81]. The main growth factors are:-

1. Epidermal growth factor (EGF)

EGF is a polypeptide of 53 amino acids (MW=6 kd) which contains three intramolecular disulphide bonds required for biological activity. The effects of EGF are mediated through cell surface receptors. EGF is mitogenic for epithelial cells and fibroblasts in culture. In vivo EGF stimulates epidermal regeneration and

collagen synthesis in a dose dependent effect [48]. The sustained release of EGF from subcutaneously implanted polyvinyl alcohol sponges was reported to result in a significant increase in granulation tissue deposition and protein accumulation [10].

2. Platelet-derived growth factor (PDGF)

PDGF is a dimer of molecular weight approximately 30,000. It consists of two peptide chains, denoted A and B, of approximately 16,000 and 14,000 molecular weight, respectively. PDGF is stored in the platelet a-granules and is released on platelet activation. It is also produced by activated macrophages, endothelium and smooth muscle cells. PDGF has both mitogenic and chemotactic activity. It is the most potent mitogen in serum for cells of mesenchymal origin, stimulating cell growth through interaction with specific high affinity receptors on the cell surface of target cells. PDGF is chemotactic for a variety of cells. The response of different cells is concentration dependent, thus maximum neutrophil chemotaxis occurs at concentration of 1-5 ng/ml whereas maximum monocyte chemotaxis occurs at concentrations of 20 ng/ml. Neutrophils are activated at higher concentrations, 20-40 ng/ml, resulting in superoxide synthesis, release of granule content and neutrophil aggregation.

It has been postulated that PDGF may play an important role in wound healing by the expression of both its mitogenic and chemotactic properties mediating the recruitment of inflammatory and connective tissue cells into the site of injury. Subsequent activation of neutrophils causes the release of lysosomal enzymes and neutral proteases which remove damaged tissues. Superoxide anions are produced which kill micro-organisms. Cellular proliferation is an essential step in repair following the inflammatory response. Fibroblasts are responsible for the synthesis and secretion of new matrix elements but also PDGF stimulates fibroblasts to secrete collagenase and this may be involved in the removal of damaged collagen and also in the remodelling process. PDGF also stimulates the production of insulin-like growth factor by fibroblasts.

3. Fibroblast growth factor (FGF)

FGF was first defined as a single factor that was mitogenic for fibroblasts. Now, there are at least five structurally related proteins having molecular weights of approximately 16,500. The most abundant is basic FGF which was initially isolated form the pituitary. Basic FGF is produced by activated macrophages; it is a fibroblast mitogen but even more mitogenic for capillary endothelial cells and can induce all the steps in angiogenesis.

4. Transforming growth factor alpha (TGF- α) TGF- α is a 50 amino acid polypeptide that has a substantial sequence homology with EGF and binds to the same receptor protein. It has similar functions to EGF but in vivo studies suggest it is a more potent angiogenic mediator [71, 74].

5. Transforming growth factor beta (TGF- β)

TGF- β has a molecular weight of 25,000 and is a dimer of two apparently identical subunits (MW=12,500) held together by disulphide bonds. It has a broad array of biological effects, including the ability to stimulate and inhibit the proliferation of cells in culture [69, 80]. In vivo, subcutaneous ad ministration of TGF- β induces marked collagen deposition and a single topical application of TGF- β to incisional wounds appears to promote healing. A slow release vehicle seems to be necessary for prolonged action on the wounds [65]. The application of a neutralizing antibody to TGF- β in adult rat wounds appears to eliminate scaring [76]. TGF- β is produced by platelets, endothelium, T-cells and macrophages.

6. Interleukin one (IL-1) and tumour necrosis factor (TNF).

These cytokines, both molecular weight of 17,000, produce the same spectrum of biological effects but are structurally different and do not compete for a common receptor. IL-1 is produced by nearly all cell types including lymphocytes, activated macrophages and vascular endothelium. TNF is produced by stimulated macrophages and causes production of surface adhesion molecules which result in the adhesion of neutrophils, monocytes and lymphocytes to vascular





endothelium. It causes production of PGI2 a potent vasodilator and inhibitor of platelet aggregation. It increases synthesis of platelet activating factor. TNF produces the haemodynamic effects of septic shock and is angiogenic. It is chemotactic for fibroblasts and stimulates them to produce more collagen and collagenase. It thus has a probable role in connective tissue remodelling.

Collagen

The word collagen is derived from the Greek word Kolla, which means glue. The term collagen or glue former was originally used in the nineteenth century for the part of skin, bone, tendon and cartilage, which, when the tissues were boiled and the extracts evaporated produced glue. Collagen is a ubiquitous protein which has been the subject of extensive investigation over many years. What was once considered a relatively inert structural protein now appears to be a group of substances which have considerable in vivo metabolic potential. In order to clarify the constituent molecules regarded as collagens an inclusive definition is necessary. In general a protein can be classified as a collagen by the presence of a typical helical, collagenous domain containing peptide chains with repeating Gly-X-Y triplets, and by the presence of hydroxyproline and hydroxylysine which are relatively specific to collagens.

Collagen is the most abundant protein in the human body accounting for about 30 % of all proteins and is the major constituent of most connective tissues. In skin and tendons, collagen accounts for over 80 % of the dry weight. Since collagen constitutes the bulk of most connective tissues it makes a major contribution to their properties. The critical function of collagen is to give strength to and to maintain the structural integrity of various tissues and organs.

The Collagen Molecule

The basic unit of collagen formerly known as tropocollogen is now known as the collagen molecule. It is shaped like a pencil with a diameter of 15 Å and a length of 3,000 Å. The collagen molecule consists of three polypeptide chains that are coiled into a unique type of helical structure (Fig. 37.4). The most abundant body collagen, Type I, has a collagen molecule with two chains of one type, designated alpha 1, and one of another type, designated alpha 2. Both chains are almost identical, but their amino acid compositions differ enough for the chains to be separable by ion exchange chromatography.

The alpha chains each contain about 1,000 amino acids and therefore have a molecular weight of about 95,000 and each chain is characterized by the presence of the typical GLY-X-Y repeating sequence throughout 90 % or more of its length. Proline and hydroxyproline follow each other relatively frequently and the Gly-Pro-Hyp sequence makes up about 10 % of the molecule.

Glycine constitutes 33 % of the amino acid residues in collagen. Twenty-two percent is made up of the imino acids, Pro and Hyp. Other unique features of the amino acid composition of collagen are the presence of hydroxylysine (Hyl) a low content of tyrosine and the absence of tryptophan. Cysteine is also absent which eliminates



the possibility of disulphide cross-links in the molecule. Both Hyp and Hyl have been considered specific to collagen, but they do occur in very small quantities in other animal proteins (elastin, the Clq component of complement, acetyl cholinesterase, glycoproteins obtained from lung lavage and a collagen-like protein absorbed to human blood cells).

The biosynthesis of collagen is a complex process. It is apparent that each collagen is coded for by a different and specific gene thus chromosome number 17 contains the coding information for the alpha-1 and alpha-2 chains of Type I collagen. After the gene is transcribed, it is spliced to yield a functional mRNA which contains about 3,000 base pairs. This is the template for the formation of the amino-acid sequence. Post translational changes occur to the amino-acids in this sequence which is going to be unique for an individual collagen alpha chain. Hydroxylation of proline and lysine is particularly important. The responsible enzymes are prolyl and lysyl hydroxylases. Their substrate is oxygen and co factors are ascorbic acid, ferric iron and alpha-ketoglutarate. Without ascorbate, the collagen molecule can be synthesised up to but not beyond the point of praline and lysine hydroxylation. The degree of hydroxylation varies from tissue to tissue depending on substrate availability, rate of synthesis and turnover time.

Glycosylation of hydroxylysine then occurs and is catalysed by galactosyl transferase and the addition of glucose by glucosyltransferase. Once synthesis of the individual pro-a chains has occurred, formation of the triple helical structure ensues. The procollagen is then passed through the cisternae of the rough endoplasmic reticulum and through a transitional endoplasm to the Golgi where it is packaged into secretory vesicles prior to extrusion by exocytosis.

Once the molecule has left the cell the terminal, or registration peptides are cleaved and it becomes involved in a process of organisation with the formation of intra and inter molecular cross-linkage (Fig. 37.5). **Fig. 37.6** The repeating disaccharides of hyaluronic acid



Crosslinking

Crosslinking stabilises the collagen fibres giving them an adequate degree of tensile strength and visco-elasticity to perform their structural role. The process involves an aldehyde-mediated mechanism with the conversion of specific lysine and hydroxylysine residues in collagen to peptide-bound aldehydes. Enzymes responsible in this step are lysyl hydroxylase and lysyl oxidase which converts the hydroxyl group to an aldol and initiates the lysine-lysine linkage.

Hyaluronic Acid

Hyaluronic acid (HA) was first isolated by Meyer and Palmer in 1934 [63], and has been shown to be a ubiquitous constituent of connective tissues. The chemical structure was established by enzymatic [64] and chemical [43] studies carried out on preparations obtained from human umbilical cord and consist of alternating 4-0-linked b-D glucopyranosyluronic and 3-0-linked 2-acetamido-2-deoxy-b-D-glucopyranosyl residues (Fig.37.6).

Because of its composition of hexosamine and uronic acid, and also of its biosynthesis, it has been classified among the glycosaminoglycans such as dermatan sulphate, chondroitin sulphate, heparin sulphate, heparin, etc., although it is the only polymer of this class of compounds in which a linkage to protein backbone (to form a proteoglycan) has not been demonstrated [24]. The history of the structure determination of glycosaminoglycans, where new sugar components or new anomeric configurations have been found repeatedly, suggests that variations in the structure of HA are possible.

Recently N-deacetylated HA has been reported in old human skin [60]. Although a linear polysaccharide, its macromolecular structure is that of an expanded coil. This is susceptible to enzymatic degradation, resulting in the release of lowmolecular weight oligosaccharides. HA and its degradation products have been associated with a number of cell regulatory actions. Considerable attention has been directed towards the role of HA in growth and development [84], and it would appear that the structural configuration of hyaluronic acid facilitates the movement of cells within the extracellular matrix [28].

Now increasing attention is being directed towards the role of HA in wound healing [4]. In the early stages of repair, when HA levels are high, cell migration plays a prominent role in restoring cellular continuity [1]. Later, when HA concentrations fall and sulphated GAG concentrations are high, cell differentiation, collagen production, and tissue organisation occur [82, 83]. The high molecular weight HA seems to inhibit the formation of capillaries [29] while angiogenesis is induced by degradation products of HA [88]. Weigel et al. [86] have proposed a model for the role of HA in wound healing in which an HA-fibrin matrix forms which attracts inflammatory cells into the wound. This matrix is in turn modified by the cells entering the wound as they secrete hyaluronidase and plasminogen activator into the extracellular space to degrade hyaluronan and fibrin. The degradation products from both HA and fibrin are then seen as important regulatory molecules for controlling cellular functions involved in the inflammatory response and inducing new blood vessel formation in the healing wounds.

There is now evidence to suggest that postnatal wound healing can be modulated, resulting in a decrease in scarring. Hellstrom [40] reported enhanced wound healing in tympanic membrane perforations in a rat model following the topical application of tissue-extracted hyaluronic acid (HA). Membranes treated with HA appeared otomicroscopically normal (translucent) 3 months later in contrast to untreated membranes which showed extensive opacification. In a follow up study it was reported that this effect was independent of molecular size or rheological properties of the HA but related to the concentration of the preparation [50]. Abatangelo et al. [1] reported that HA facilitated wound healing in diabetic rats. Its mode of action appeared to be the promotion of epithelial migration and differentiation. Radaelli et al. [70] using a rat model reported that HA treated wounds developed a greater early wound breaking strength compared to untreated controls. The reason for enhanced wound strength was due to an early accumulation of oriented collagen fibres.

Foetal Wound Healing

Human foetal surgery has been successfully performed in a small number of highly selected cases for foetal urinary tract obstruction, diaphragmatic hernia and sacrococcygeal teratoma, which threaten foetal viability [37, 38, 49]. This experimental therapy involves maternal laparotomy and hysterotomy, delivering the foetus and per forming the necessary foetal surgery, after which the foetus is returned to the uterus and the pregnancy allowed to continue. At subsequent delivery, it was observed that there was a lack of scarring and contracture associated with the surgery (M. R. Harrison., personal communication). The absence of scarring makes the prospect of reconstructive human foetal surgery very attractive [20], but the inherent risks of interrupting a healthy and viable pregnancy, at present, restricts human foetal surgery to life-threatening foetal anomalies [21, 25, 32-34, 39]. Perhaps more relevant, at this stage, is investigating the biology of foetal wound healing to discover the mechanisms of scarless healing. The aim of such studies is to discover insights into the modulation of postnatal wound healing and, thus, decrease scarring, a matter of relevance to all surgeons [55].

It must be said, at the outset, that the fundamental mechanisms of foetal wound repair remain a mystery. To date, they have been described in terms which are appropriate to postnatal repair and, yet, there do seem to be a number of major differences between the foetal and post-natal repair. From the point of view of the immunohistochemical and biochemical analysis, there are well described differences between foetal and post-natal repair both at cellular and extra cellular levels [3, 44, 45, 47, 56, 57, 59, 72].

In the foetal wound healing model there is a minimal inflammatory response with little deposition of granulation tissue or fibrosis seen [18]. The role of collagen was a subject of major controversy in the mid-1980s and is examined in some detail in this thesis. The glycosaminoglycans content of the healing wound in the foetus has been the subject of intensive investigation and it has been noted that hyaluronic acid is deposited during early stages of foetal repair and that fibronectin and tenascin deposition are early features of foetal wounds [89–91].

The investigation of foetal wound healing is moving on from phenomenological studies to more mechanistic investigations and the focus of interest now is directed more towards what makes the foetal wound heal. Studies that look at the regenerative as well as the developmental and oncogenic processes have uncovered some intriguing aspects of cellular control that might have significance in foetal wound repair. Homeobox genes are a class of genes which code for peptide regulatory factors that bind with varying degrees of specificity to DNA and promote the transcription of structural and/or metabolic genes. The idea, basically, is that in foetal wound repair there is reformation of injured tissue to create the architecture of the pre-injury tissue. The central role of DNA in the process of coordinated production of organised tissue is a matter of some debate. Some investigators regard the DNA as the ultimate controlling factor in tissue organisation. The predominance of the cell, however, is not universally accepted and other investigators feel that the extra-cellular matrix has a profound influence upon the ultimate tissue architecture which results from the co-ordinated repair of a tissue defect.

In essence, then, the process of foetal wound repair is more usefully recognised as a process of regeneration rather than the healing and the regulatory mechanisms for regeneration are, in all probability, very different from those for repair.

As mentioned in the introductory paragraphs to this chapter, this thesis is a historical record that describes the experimental work which had a major influence in directing the focus of research in foetal wound healing.

The Essential Difference Between Foetal and Post-Natal Wound Repair Appears to Be Not What Is Deposited but What Happens to It

In the foetus reorganisation of repair tissue results in the restoration of the original tissue architecture.

In the post-natal wound, reorganisation of the repair tissue does not result in the restoration of the tissue architecture.

In the former **regeneration** has occurred.

In the latter there is the formation of **scar tissue**.

Foetal Rabbit Studies

Introduction

It is perhaps surprising that the presence of collagen in the foetal wound healing process has been questioned. Krummel et al. [47] reported no detectable hydroxyproline (a highly specific marker for collagen) using a highly sensitive liquid chromatography analytic technique [54], in silastic/poly vinyl alcohol (PVA) tubes implanted in foetal rabbits. Adzick et al. [3] had looked for collagen deposition in another wound healing model in foetal rabbits and had reported elevated levels of hydroxyproline. Adzick had used implanted polytetrafluoroethylene tubing [31]. The histological events in both wound healing models are reported to be similar to those found in incisional wounds. Adzick had not been able to characterise the collagen although it was apparent that the reported yields of collagen in the Gore-Tex tube model were well within the limits of detection of a micro analytical typing technique which I had developed under the guidance of H. Paul Ehrlich, Director of the Wound Healing Laboratory at the Shriners Burns Institute, Boston. This technique involved the typing of collagen by the chromatographic separation of peptides derived from cyanogen bromide digests of tissue samples. Adzick's work was repeated in collaboration with Dr John Siebert MD, a resident in Plastic Surgery at the Institute of Plastic and Reconstructive Surgery at New York University, New York. The foetal animal surgery was performed in New York and the harvested material was delivered to Boston for histological and biochemical analyses.

Amino acid analysis (residues per 1000) (40 pmole sensitivity)

	Foetal		Newborn	
	Day 4	Day 7	Day 4	Day 7
HPro	1	4	4	33
Asp	89	55	94	77
Thr	52	55	51	41
Ser	54	55	51	46
Glu	116	126	125	109
Pro	57	60	62	78
Gly	67	78	72	160
Ala	99	90	92	92
Val	74	65	69	55
Cys	27	0	32	13
Met	3	12	2	3
Ile	34	37	35	29
Leu	100	90	71	69
Tyr	29	27	30	20
Phe	37	35	34	26
His	28	47	37	28
HLys	0	1	4	4
Lys	96	95	91	71
Arg	39	35	45	46
Collagen (ng/tube)	41.5	43.5	198	291
	<i>n</i> = 22	<i>n</i> = 8	<i>n</i> = 24	n = 24

This table reflected the amino acid analysis of the Gore-Tex tube model.

Discussion

The findings presented here, again using a rabbit model, do not support Adzick's work which reported abundant quantities of collagen in Gore-Tex implants. At the same time, this study does not support Krummel's concept that collagen is not involved in the healing process in the foetus. The Gore-Tex tube implant model has been criticized because of a tendency to be encapsulated leading to poor ingrowth of new tissue [27]; however, this was not our experience. One observation is that when using the Gore-Tex tubing, care must be taken to ensure that the tissue surrounding the tubing is not analysed with the content of the tubing. Incomplete removal of the connective tissue capsule from the outside of the tube before analysis, would contribute to elevated hydroxyproline levels attributed to the tube contents. The silastic/PVA model overcomes this problem by enclosing a PVA sponge within a perforated silastic tube. The tube forms an inert zone between the sponge and the surrounding tissue. Only new tissue will grow into the sponge which can be isolated intact from the surrounding tubing. One possible explanation for the reported absence of hydroxyproline, despite a very sensitive analytical technique, relates to sample size. Preparative steps, in the technique used to process the PVA sponges, result in dilution of the sample. Under such circumstances the detection limits may have been reached due to dilution. In our study, amino acid analysis failed to detect hydroxylysine in the foetal Day 4 samples although hydroxyproline was detected. In the Foetal Day 7 sample both were detected. Hydroxyproline is not unique to collagen. It occurs in other proteins [2]. The combination of hydroxyproline and hydroxylysine, however, supports the presence of collagen. In addition there is the morphologic appearance of the striated fibre on electron microscopy. Native collagen does have a banded appearance with a periodicity of 67 nm. The finding of aperiodicity of 54 nm does not exclude the possibility that this is fibrillar collagen. Fixation artefacts can affect the observed periodicity. The presence of collagen deposition and the absence of scarring following foetal repair would suggest that the

process of reorganization of the tissue at the repair site is far more efficient than in the postnatal healing process. Another possibility is that healing in the foetus occurs by a process of regeneration as suggested by Rowsel [73]. In terms of reconstitution of a dermal matrix, repair with a high degree of reorganization or regeneration, would produce the same result. However, skin is far more than a layer of epithelium over a dermal matrix. Our histological findings support the concept of a high degree of matrix reorganization, but the decreased numbers of appendageal elements suggest that true regeneration has not occurred.

While the evidence suggesting the presence of collagen within the wound healing model in foetal rabbits remains, it was obvious that very small amounts were present which precluded further characterization. It was thus decided not to pursue further biochemical studies using this animal model but explore the possibility of gaining further information from a larger animal model, the foetal sheep, which is described in the next section.

Foetal Sheep Studies

Introduction

Adzick and Krummel had come to diametrically opposite conclusions from their studies of collagen deposition in similar foetal rabbit models, they had however used two different wound implant models. We had been unable to reproduce Adzick's findings using the similar animal and wound healing models and yet had not been able to support Krummel's conclusions. At this time there were more reports coming from other investigators which tended to increase rather than resolve the controversy regarding the presence and role of collagen in scarless wound healing in the foetus. These studies all involved small animal models [3, 35, 47, 62, 66, 73, 77].

One possibility was that the conflicting results were associated with the small size of the animal models used and in order to finally resolve the controversy it was decided to use a



Fig. 37.7 The wound healing model showing the sandwiched sponges, from the side (*left*), and above (*centre*) and the bisected sponges (*right*)



Fig. 37.8 Breaking strength of the wound tissue in the sponge implants (*error bars* \pm SEM). Six sponges from each time point were used and the results are the mean of six values

well-characterized large animal, the sheep [36], to investigate the deposition of collagen in a model of wound healing (Figs. 37.7, 37.8, 37.9, and 37.10). This study was again a collaborative venture with the animal surgery being performed in San Francisco by Michael Longaker, MD. at the Fetal Treatment Program, University of California, and the histological and biochemical analysis of the harvested tissues being performed in Boston.

Wound Healing Model

Two configurations of polyvinyl alcohol sponge (PVA) discs ($12 \text{ mm} \times 3 \text{ rnrn}$) were implanted

Hydroxyproline content of sponges



Fig. 37.9 Hydroxyproline levels in implants from a 75-day foetus, a 100-day foetus and an adult. For this assay, the sponge halves from the breaking strength assay were pooled for each time point and stage of gestation

subcutaneously, sponges were either cut in half, and the two halves re-joined by a staple, or an intact sponge was sandwiched between two equal-sized silicone rubber discs. The "sandwiched" sponge allows new tissue deposition to occur only from the periphery of the sponge. Histological studies of PVA sponges implanted in adults have demonstrated that the cellular and matrix sequence of events in the sponges reflects the events of incisional wound healing in postnatal skin [75].



Fig. 37.10 A comparison between the content and composition of the collagen in adult and foetal sponges using conventional extraction techniques. The x-axis on the left shows the % of Type l collagen. This is a more rigid form of collagen and is associated with more mature or stable

forms of extracellular matrices. The % is demonstrated in the histogram. The *curves* demonstrate the amount of collagen with the value expressed in mg of extracted collagen/sponge in the right side x-axis

Discussion

Evidence for the deposition of collagen in this wound healing model in foetal sheep is presented both histologically and biochemically [14]. Other investigators have looked for collagen deposition in foetal wounds in small animal models. Rowsell could find no evidence of new collagen deposition in foetal rat wounds using trichrome staining [73]. This histochemical stain lacks the sensitivity of currently available immunohistochemical stains, which have now demonstrated the presence of Types I, III, and VI collagen in incisional wounds in foetal mouse and sheep. Trichrome staining in our study was useful because of the abundant deposition of collagen in the sponges.

Merkel and colleagues reported the presence of collagen from pooled tissues from excisional wounds in foetal rats [62]. Their study was not supported with histological evidence that the biopsied material was definitely wound repair tissue rather than fascia underlying the wound. Previous studies with foetal rabbits show that excisional wounds do not close and that granulation tissue is absent [79]. These small animal studies were all performed late in gestation. When collagen deposition was reported, the yield was too low in individual animals to permit qualitative analysis.

The use of a large animal model, such as the sheep, facilitates surgical manipulations and allows more extensive analysis of wound tissue. The sheep has a 145-day period of gestation compared to 31–32 days for the rabbit. Foetal surgery can be performed, with a low abortion rate, as early as 55 days in the sheep (early second trimester), compared to 22 days (early third trimester) in the rabbit, thus giving a much longer time to follow the wound healing process.

It should be noted that these studies have been carried out on animals which represent a range of embryonic development. This is reflected in the varying rates of maturation of different body systems in utero. This makes inter species comparisons difficult as well as making extrapolations from animal studies to the potential behaviour of human tissues uncertain. That being stated it should be noted, as mentioned in the introduction, that scarless healing has been observed in human foetuses.

Since collagen is the major component of scar tissue in post-natal animals, how does scarless healing proceed in foetal animals, which also lay down a collagen-rich repair matrix? Perhaps the first point to emphasise is that in healing incisional wounds in foetal sheep the initial repair tissue deposited, is disorganized [58]. For up to a week, histologically, scar tissue is present but by 14 days this is no longer apparent. So there are similarities in the fundamental processes of healing in adult and foetal animals; a defect is quickly filled in with connective tissue components which are subsequently rearranged to effect a more permanent repair. The nature of the repair in the foetus will be determined by the composition of the materials used and the manner in which they are rearranged. The findings in this section demonstrate some qualitative and quantitative differences in the collagens deposited in foetal and adult wounds.

This work disproves the **first hypothesis** that the scarless healing observed in foetal animals is due to the lack of new collagen deposition [47]. This thesis, however, continues with the exploration of the phenomenon of scarless healing and the next chapter considers the **second hypothesis** that the scarless healing observed in the foetal animals is due to the hyaluronic acid rich environment which allows for more effective remodelling of the repair tissue [26].

The Role of Hyaluronic Acid in Wound Healing

Introduction

The second hypothesis to be explored was proposed by DePalma et al. [26] from the Medical college of Virginia. This group, using a foetal rabbit model, had attempted a characterisation and quantification of the wound matrix in the foetal rabbit and had reported high levels of hyaluronic acid. They proposed that the scarless repair in the foetus was due to a regenerative process facilitated by a wound matrix rich in hyaluronic acid (HA). We had confirmed the elevated levels of HA in the Gore-Tex tube implants but were also intrigued by reports of the modulating effect on post-natal healing achieved by the topical application of tissue extracted HA resulting in reduced scarring. Hellstrom reported (1987) reduced scarring in tympanic membrane perforations in an adult rat model. Topical HA treated membranes appeared otomicroscopically normal (translucent) 3 months after treatment in contrast to untreated membranes, which showed extensive opacification. Laurent et al. [50] reported that closure time for the perforations was related to the concentration of HA but not to its molecular weight or viscosity. Also using a rat model, Radaelli et al. [70] reported that HA-treated wounds developed a greater early wound breaking strength compared to untreated controls due to an accelerated accumulation of oriented collagen fibres. These experimental studies were performed using HA extracted from rooster comb.

HA is present in all tissue in varying concentrations, with the highest concentration occurring in soft connective tissue [51]. Although a linear polysaccharide, its macromolecular structure is that of an expanded coil, which makes it susceptible to enzymatic degradation, resulting in the release of low molecular weight oligosaccharides. HA and its degradation products have been associated with a number of cell regulatory actions. In the past considerable attention has been directed towards determining the role HA plays in growth and development [84]. Currently attention is beginning to focus on the function HA serves in wound healing. In the early stages of repair, when HA levels are high, cell migration plays a prominent role in restoring cellular continuity. Later, when HA concentrations fall and sulphated GAG concentrations increase, cell differentiation, collagen production, and tissue organization occur [82, 83]. A high molecular weight HA seems to inhibit the formation of capillaries [29], while angiogenesis is induced by degradation products of HA [88]. A theoretical model has been proposed by Weigel et al. [86] for the role of HA in wound healing; an HA-fibrin matrix forms that attracts inflammatory cells into the wound, and this matrix is then broken down by cells entering the wound. These cells secrete hyaluronidase (Hyase) and plasminogen activator into the extracellular space to degrade HA and fibrin. The degradation products from

both HA and fibrin are viewed as regulatory molecules for modulating the cellular functions involved in the inflammatory response and inducing new blood vessel formation in the healing wound.

The question remains, however, concerning the mechanism for the proposed effect of HA in post-natal wound healing and scarring? The possibilities are:

- Macrophage-effected postnatal repair process is suppressed,
- An intrinsic tissue repair process is augmented, or,
- 3. A combination of the two effects occurs.

Whatever the case, a mechanism must be proposed where HA has a definite role in harnessing and manipulating the natural reparative capacity of tissue fibroblasts. This raises the question whether HA alone, has the capacity to control events of such complexity as wound healing and scar maturation. Preliminary studies have identified a heterogeneous group of HA-protein complexes in normal skin and post-burn scar [11]. The association of highly purified HA and protein is well established [85]. The protein composition of these complexes suggest that it is not random contamination; collagen comprises <5 % of the protein in the complexes, yet is the most abundant tissue protein. A functional association is suggested by the resistance of the collagen in the complexes to collagenase digestion.

Discussion

This study demonstrates that HA produced in the clinical laboratory from human tissue extracts or commercially from human and animal extracts varies both in its biochemical composition and also in its biological activity. A major contributing factor dictating this variation would appear to be the proteins associated with the HA. When an HA-protein complex extracted from human scar tissue was added to an in vitro, foetal sheep, wound healing model, there was gross evidence of a resulting dermal repair [16]. The relevance

of this experiment concerns the putative nature of the foetal tissue wound healing process. As it demonstrates more autonomy, what directs the fibroblasts in the repair process? It would appear that it is not a self-directed process in that the fibroblasts are present and viable in the control wounds but healing of the dermal matrix does not occur. Similarly the possibility of epidermal control of the dermal healing process can be excluded as the epidermis is viable, proliferates and restores continuity but again no dermal matrix repair occurs. DePalma and colleagues have suggested that the HA enriched environment of the foetal wounds creates a permissive environment for dermal repair [26], however, the application of HA alone does not result in dermal repair. It is only when tissue extracted HA containing associated proteins, is added that healing occurs. This suggests that even in the foetal wounds, extrinsic factors are important in the initiating repair and it is possible that macrophages play a role in this process in vivo.

General Discussion

The thesis began by exploring the hypothesis that foetal and postnatal wound healing were fundamentally different processes due to a lack of new collagen deposition in foetal wounds. When we consider the precept that a wound is a 'discontinuity in tissue integrity' it does appear strange that such a hypothesis should be advanced as the major structural contribution to tissue integrity is indeed collagen. The hypothesis, however, was proposed by several investigators who looked for histological and biochemical evidence of collagen deposition in several wound healing models and found none. The observation that stimulated this investigation was that incisional wounds in foetal skin appeared to heal in utero without any evidence of scarring.

The experimental aspect of the thesis began by exploring this hypothesis using a small animal model, the rabbit. The conclusion of the work described in Chap. 2 is that collagen deposition does occur in this animal model but the amounts were very small. It was only when the results of the large animal study described in Chap. 3 were presented that the controversy was finally put to rest [11].

A new hypothesis was thus needed to explain the difference in the wound healing processes in adult and foetal animals which led to the scarring in the post-natal situation and the scarless healing in the foetal model. At this stage in the evolution of the thinking about the differences in foetal and post-natal wound healing another interesting observation was being made and verified in a number of animal studies, namely that the healing of post-natal wounds could be modulated by the application of tissue extracted HA, resulting in a decreased amount of scar tissue. The question thus arose, what is the relationship between HA and the organization of the repair tissue. Was it purely physical, promoting an environment in which the new collagen could be more easily manipulated and organized? Was it chemical? Or was it not the HA at all or at least not entirely the HA? The second hypothesis that is examined in this thesis is that it is the hyaluronic acid rich wound matrix in the foetus which promotes the scarless repair.

The experimental work presented in Chap. 4 does not support this hypothesis. A number of experiments are described looking at the effect of tissue extracted HA on epidermal cells, fibroblasts and in vitro wounds. The conclusion is that it is the proteins associated with HA which appear to have biological effects which persist even when the HA is enzymatically degraded. These effects are, however, lost when the complex is heated, presumably due to denaturing the protein.

This then is the extent of the experimental work of the thesis and several points can be made regarding the possible manipulation of the wound healing process:

- 1. Foetal and post-natal wound healing process are fundamentally similar.
- 2. The **MAJOR difference** is in the organization of the collagen which is deposited.
- 3. Animal studies suggest HA stimulates reduced scarring in post-natal animals.
- 4. The in vitro studies described in this thesis suggest that the reduced scarring may be a function of the activity of the associated proteins.

5. If scarless healing is going to be achieved in the post-natal wound then two stages must be considered: the work presented in this thesis strongly supports the idea that in the foreseeable future it will be possible to suppress the adult type response to injury by targeting a specific factor(s) and in a similar fashion it will be possible to augment/enhance the foetal response which results in neogeneration rather than repair.

This leads to a **new hypothesis**:

Scarless healing in skin wounds in post-natal animals will be achieved by the application of appropriate peptide regulatory factors which will suppress the response to repair and allow a foetal -like ability to generate new tissue.

How can this hypothesis be explored?

Firstly a working hypothesis has to be proposed: and this is that scar maturation, as a function of matrix remodelling, is determined by cell matrix interactions which are, in turn, determined by peptide regulatory factors. The questions that need to be addressed are:-

- 1. What peptide regulatory factors are involved in matrix remodelling?
- 2. How do these factors produce their effects?
- 3. How can these factors be extrinsically modulated to reduce scarring.

To address the first question it would be possible to continue to look at the proteins associated with highly purified, tissue extracted, hyaluronan (HA) from post-burn scar tissue of varying stages of maturity. I have proposed that this HA-protein association represents a select 'window' into the biological activity of the remodelling matrix [11, 15, 17].

The second question could be addressed in two ways:

Firstly, the tissue extracted proteins could be assessed individually and in combination using a variety of in vitro and in vivo models including monolayer cell proliferation studies, the fibroblast populated collagen lattice (splinted and unsplinted), and incisional wound repair using tissue explants [13]. Small and large animals could be used in in vivo studies, incorporating incisional wound and PVA sponge implant models [12, 75]. Wound healing could be assessed biomechanically and histologically using light and electron microscopy and immunohistochemistry.

Secondly, individual and combinations of known, and well characterised, peptide regulatory factors could be assessed using the same models. The third question would require in vitro and in vivo animal models but eventually clinical studies would be required.

This chapter has described part of the journey towards the ultimate clinical goal of scarless healing. The prospect of modulating the healing of surgical incisional wounds is very encouraging. The goal extends far beyond this however, to the healing of those with more extensive injury particularly the scarring which results from burns. When this affects the face, the resulting deformity can lead to the isolation of the patient from society. When the scarring affects the hands, the disability from functional loss can lead to the loss of work and economic deprivation. The aim then, of this work is to gain further insights into the fundamental biology of wound healing and scar maturation.

To reduce or eliminate scarring is to give new life, new hope, to those countless victims of trauma whose future, at present, is characterised by deformity and disability.

Summary and Conclusions to "Towards Scarless Healing: Déjà vu"

Science has undergone some major changes in the last 30 years and now must come to terms with the major limitations in the classical reductionist view. Biological events are highly sophisticated and complex processes where neither cell nor matrix reign supreme. The molecular biologists must recognize the power and the influence of the biology of molecules which have a profound effect on the outcome of such broad concepts as tissue regeneration and tissue repair. It is also essential to understand the key role of clinical observation as an ultimate arbiter in determining life and death (literally). Whilst clinical observations seem to stand the test of time, scientific observations are more transitory in nature relating to both the prevailing understanding of biological processes and the ability to measure them.

The extrapolation of research from small animal models to further the understanding of potential therapeutic manipulations in human post-natal wound healing continues to cause concern. A point to note is that when the fetal skin is injured in the third trimester of pregnancy then scarring does result. That scarring begins to become apparent when polymorphonuclear leucocyte cells appear in the fetal circulation might suggest that the fetus is getting ready for the changing priorities of tissue repair once the ex-utero status has been achieved. To attribute the presence or absence of microbiota to acceleration or delay in wound healing and the outcome in terms of scarring or no scarring might be supported by studies of genetically manipulated mice [93], however does it fit in with the reported clinical observations? The lack of clinical understanding becomes of greater concern when reports of research from globally prestigious centres claim to hold a "contrarian view" that inflammation might even be a positive benefit to regenerative healing (also known as scarless healing). To give apparent clinical relevance to very sophisticated molecular biological studies again on mutant mice by suggesting the observations should promote some future clinical studies on the effects of NSAIDs on wound healing is bordering on the surreal [97]. This also illustrates the gulf of missunderstanding between the "Ivory Towers" and the wards and clinics where hundreds of thousands of patients present each day seeking treatment for the care of wounds both acute and chronic. It is perhaps also somewhat disappointing to find that science and scientists have fixations on "magic molecules" that singularly and uniquely will unravel the mysteries of the universe. Do such molecules exist? TGF- β and its various isoforms continue to excite research scientists particularly

when they can be put in the context of a fibrosisassociated canonical pathway [101]. But how specific, reliable and relevant are such pathways when the reality of cellular context, the multiplicity of signalling events and potential multiple outcomes have rendered the concept of linear pathways to the garbage bin. The new era of science is going to have to come to terms with the reality of complex programs of response that are going to need very different methods of analysis than are currently practised. It is important to think "out of the box" but in so doing it is important to maintain the clinical perspective. Looking at the healing of the oral mucosa is a good example. Clinical observation confirms that post-natal scarless healing occurs both with a regularity and predictability following injury to the oral mucosa. But the human oral mucosa is critically different from the skin in not having to provide such a vital and complex barrier function. Again it is disappointing to read published research which reveals a lack of basic knowledge of the clinically observed phenomena; to state that "foetal wounds heal fast without scar formation" is just not true [95, 96].

Recent reviews on scarless healing make little or no mention of the collagen story revealed in this thesis nor are the needs for different forms of wound healing given appropriate consideration. The conclusions are little different from 20 years ago, "The precise mechanism of fetal regulation remains unclear with a number of differences identified between the fetal and adult wound healing" [99]. These authors did acknowledge that the introduction of stem cells is going to change the game. Stem cells can provide a multiplicity of factors which involve post-translational modifications and matrix organization. This is a very important point to keep in focus as ultimately the scar is a matrix and not a cellular phenomenon [102]. Another review which again overlooks the original collagen question does correctly identify the control of the extra cellular collagen matrix as a key challenge. They propose educating adult fibroblasts to behave like fetal fibroblasts to reduce scarring through cell therapy [105]. This seems to partially reflect the hypothesis proposed over 20 years ago that the assembly of the dermal architecture is more likely to be achieved when the biological imperative of rapid and strong repair is overruled. The interaction of fibroblasts and keratinocytes remain a target for exploration although the interpretation of the significance of the results is difficult [103]. Also the dynamics of events in utero and ex utero particularly as indicated by gene expression has received considerable attention [98] but we have to come back to the clinical observation that scarring is the outcome, not the process. In the review referred to before 'Scarless Wound Healing: Chasing the Holy Grail' [100] the authors conclude, "Achieving scarless wound healing in the adult will require not only an understanding of signalling molecules and growth factors but also a thorough understanding of lineage-specific cellular origin and function during both fetal and adult stages." With great respect for the senior author of this paper, Michael T Longaker, I would like to think that the emerging discipline of regenerative medicine is going to provide us with clinical observations that are real and stand the test of time, but, like the scarless healing achievable in the human fetus may remain a biological mystery.

We now have a story that began over a hundred years ago with the birth of a doctor who was driven to cross-fertilize his ideas with cutting edge technology in other disciplines. The result, high-definition ultrasound, opened up a whole new world of interventional obstetric care creating the reality of human fetal surgery. The ability of the human fetus to heal surgical wounds created in the second trimester without scarring was a clinical observation, not a scientific breakthrough. The failure to understand the fundamental biological mechanisms involved despite the massive allocation of time and resource from a multi-disciplinary scientific community should not diminish the energy for the quest. Rather it supports the case that the focus should return to clinical observation in the context of human regenerative medicine, the 'new reality'.

Acknowledgement The Boston Days: Paul Ehrlich introduced me to Collagen, Toni Rittenberg introduced me to the fibroblast populated collagen lattice, Rita Greco introduced me to gels and Hari Garg introduced me to Hyaluronan. I am indebted to each and every one.

Hong Kong: I am indebted to my friend and colleague Dr Lin Huang who has kept alive for me the magic and responsibility of science. Thank you.

References

- Abetangelo G, Martelli M, Vecchia P. Healing of hyaluronic acidenriched wounds: histological observations. J Surg Res. 1983;35:410.
- Adams E, Frank L. Metabolism of proline and the hydroxyprolines. Ann Rev Biochem. 1980;49:1005–61.
- Adzick NS, Harrison MR, Glick PL, Beckstead JH, Villa RL, Scheunstuhl H, Goodson WH. Comparison of fetal, new-born, and adult healing by histologic, enzyme-histochemical and hydroxyproline determination. J Pediatr Surg. 1985;20:315.
- Balazs EA, Darzynkiewicz Z. The effect of hyaluronic acid on fibroblasts, mononuclear phagocytes and lymphocytes. In: Kulonen E, Pikkarainen J, editors. Biology of fibroblasts. London: Academic; 1973. p. 237–52.
- Bergstresser PR, Taylor JR. Epidermal 'turnover time' – a new examination. Br J Dermatol. 1977;96:503–9.
- Briggaman RA, Dalldorf FG, Wheeler Jr CE. Formation and origin of basal lamina and anchoring fibrils in adult human skin. J Cell Biol. 1971;51:384–95.
- Briggaman RA, Wheeler Jr CE. The epidermal-dermal junction. J Invest Dermatol. 1975;65:71–84.
- Briggaman RA, Wheeler Jr CE. Epidermolysis bullosa dystrophica recessive: a possible role of anchoring fibrils in the pathogenesis. J Invest Dermatol. 1975;65:203–11.
- Brown IA. Scanning electron microscopy of human dermal fibrous tissue. J Anat. 1972;113:159–68.
- Buckley A, Davidson JM, Kamerath CD, Wolt TB, Woodward SC. Sustained release of epidermal growth factor accelerates wound repair. Proc Natl Acad Sci U S A. 1985;82:7340–4.
- Burd DAR, Siebert JW, Ehrlich HP, Garg HG. Human skin and post-scar hyaluronan; demonstration of association with collagen and other proteins. Matrix. 1989;9:322.
- Burd DAR, Longaker MT, Adzick NS, Harrison MR, Ehrlich HP. Fetal wound healing in a large animal model: the deposition of collagen is confirmed. Br J Plast Surg. 1990;43:571.
- Burd DAR, Longaker MT, Adzick NS, Compton CC, Harrison MR, Siebert JW, Ehrlich HP. Fetal wound healing: an in vitro explant model. J Pediatr Surg. 1990;25:898.
- Burd DAR, Siebert J, Longaker MT. Collagen deposition in fetal wounds. In: Scott Adzick N, Longaker MT, editors. Fetal wound healing: a paradigm for tissue repair. New York: Elsevier Science Publishing Company; 1992.

- Burd DAR, Siebert J, Garg HG. Hyaluronan-protein interactions. In: Scott Adzick N, Longaker MT, editors. Fetal wound healing: a paradigm for tissue repair. New York: Elsevier Science Publishing Company; 1992.
- 16. Burd DAR, Longaker MT, Adzick NS. In vitro models of fetal wound healing. In: Scott Adzick N, Longaker MT, editors. Fetal wound healing: a paradigm for tissue repair. New York: Elsevier Science Publishing Company; 1992.
- Burd DAR, Greco RM, Regauer S, Longaker MT, Siebert JW, Garg HG. Hyaluronan and wound healing: a new perspective. Br J Plast Surg. 1992;44:579–84.
- Burrington JD. Wound healing in the fetal lamb. J Paediatr Surg. 1971;6:523.
- Chandrarajan J. Separation of type III collagen from type I collagen and pepsin by differential denaturation and renaturation. Biochem Biophys Res Commun. 1978;83:180.
- 20. Christ JE. Fetal surgery: a frontier for plastic surgery. Plast Reconstr Surg. 1986;77:645.
- Christ JE. Plastic surgery for the fetus. Plast Reconstr Surg. 1990;86:1238.
- Chvapil M, Koopman CF. Scar formation: physiology and pathological states. Otolaryngol Clin N Am. 1984;17:265.
- Clarke RAF, Henson DM, editors. The molecular and cellular biology of wound repair. New York: Plenum Press; 1988.
- Comper WW, Laurent TC. Physiological function of connective tissue polysaccharides. Physiol Rev. 1978;58:255.
- Dado DV, Kernahan DA, Gianopoulos JG. Intrauterine repair of cleft lip: what's involved? Plast Reconstr Surg. 1990;85:461–5.
- 26. DePalma RL, Krummel TM, Durham LA, Michna BA, Thomas BL, Nelson JM, Diegelmann RF. Characterization and quantitation of wound matrix in the fetal rabbit. Matrix. 1989;9:224.
- Diegelmann RF, Lindblad WJ, Cohen IK. A subcutaneous implant for wound healing studies in humans. J Surg Res. 1986;40:229–37.
- Docherty R, Forrester JV, Lackie JM, Gregory DW. Glycosaminoglycans facilitate the movement of fibroblasts through three-dimensional collagen matrices. J Cell Sci. 1989;92:263.
- Feinberg RN, Beebe DC. Hyaluronate in vasculogenesis. Science. 1983;220:1177.
- 30. Gibson T, Kenedi RM. The structural components of the dermis and their mechanical characteristics. In: Montagna W, Bentley JP, Dobson RL, editors. Advances in biology of the skin, vol. 10. New York: Appleton-Century Crofts; 1970. p. 19–38.
- Goodson III WH, Hunt TK. Development of a new miniature method for the study of wound healing in human subjects. J Surg Res. 1982;33:394.
- Goss AN. Intrauterine healing of fetal rat oral mucosal, skin and cartilage wounds. J Oral Pathol. 1977;6:35.

- Hallock GG. In utero cleft lip repair in A/J mice. Plast Reconstr Surg. 1985;75:785.
- Hallock GG, Rice DC, McClure HM. In utero lip repair in the rhesus monkey: an update. Plast Reconstr Surg. 1987;80:855.
- Hallock GG, Rice DC, Merkel JR, DiPaolo BR. Analysis of collagen content in the fetal wound. Ann Plast Surg. 1988;21:310.
- Harrison MR, Jester JA, Ross NA. Correction of congenital diaphragmatic hernia in utero. I. The model: intrathoracic balloon produces fatal pulmonary hypoplasia. Surgery. 1980;88:174.
- Harrison MR, Golbus MS, Filly RA, Callen PW, Katz M, De Lorimier AA, Rosen M, Jonsen AR. Fetal surgery for congenital hydronephrosis. N Engl J Med. 1982;306:591.
- Harrison MR, Golbus MS, Filly RA, Anderson RL, Flake AW, Rosen M, Huf RW. Fetal hydronephrosis: selection and surgical repair. J Pediatr Surg. 1987;22:556.
- Harrison MR, Golbus MS, Filly RA. Experimental models for fetal treatment. In: Harrison MR, Golbus MS, Filly RA, editors. The unborn patient. Orlando: Grune and Stratton; 1984.
- Hellstrom S, Laurent C. Hyaluronan and healing of tympanic membrane perforations. An experimental study. Acta Otolaryngol Suppl. 1987;442:54.
- Holbrook KA, Odland GF. Regional differences in the thickness (cell layers) of the human stratum corneum: an ultrastructural analysis. J Invest Dermatol. 1974;62:415–22.
- Holbrook KA, Odland GF. The fine structure of developing human epidermis: light, scanning, and transmission electron microscopy of the periderm. J Invest Dermatol. 1975;65:16–38.
- Jeanloz RW. Chemistry of mucopolysaccharide. Proceedings 3rd international congress of biochemistry. 1985. p. 65.
- Kistler A, Utsugi R, Ibara S. Wound healing in fetal limb organ culture. Ann Plast Surg. 1988;21:303.
- Kistler A, Ibara S. Requirements of serum factors for wound closure of embryonic skin *in vitro*. Ann Plast Surg. 1989;23:479.
- Knighton DR, Fiegel UD. The macrophage: effector cell in wound repair. Prog Clin Biol Res. 1989;299:217.
- 47. Krummel TM, Nelson JM, Dieglemann RF, Lindblad WJ, Salzberg AM, Greenfield LJ, Cohen IK. Fetal response to injury in the rabbit. J Pediatr Surg. 1987;22:640.
- Laato M, Niinikoski J, Lundberg C, Arfors KE. Effect of epidermal growth factor (EGF) on experimental granulation tissue. J Surg Res. 1986;41:252–5.
- 49. Langer JC, Harrison MR, Schmidt KG, Silverman NH, Anderson RL, Goldberg JD, Filly RA, Crombleholme TM, Longaker MT, Golbus MS. Fetal hydrops and death from sacrococcygeal teratoma: rationale for fetal surgery. Am J Obstet Gynecol. 1989;160:1145.
- Laurent C, Hellstrom S, Fellenius E. Hyaluronan improves the healing of experimental tympanic membrane perforations. Arch Otolaryngol Head Neck Surg. 1988;114:435.

- 51. Laurent TC. Biochemistry of hyaluronan. Acta Otolaryngol Suppl. 1987;442:7–24.
- Leibovich SJ, Ross R. The role of the macrophage in wound repair. A study with hydrocortisone and antimacrophage serum. Am J Pathol. 1975;78: 71–100.
- Leibovich SJ, Ross R. A macrophage-dependent factor that stimulates the proliferation of fibroblasts in vitro. Am J Pathol. 1976;84:501.
- Lindblad WJ, Diegelmann RF. Quantitation of hydroxyproline isomers in acid hydrolysates by highperformance liquid chromatography. Anal Biochem. 1984;138:390–5.
- Lindgren VV, Hollister DW, Marshall WR. Wound healing and unsatisfactory scars. Plastic Reconstr Surg. 1987;80:321.
- 56. Longaker MT, Harrison MR, Crombleholme TM, Langer JC, Decker M, Verrier ED, Spendlove R, Stern R. Studies in fetal wound healing: I. A factor in fetal serum that stimulates deposition of hyaluronic acid. J Pediatr Surg. 1989;24:789–92.
- Longaker MT, Harrison, Langer, Cromblehome TM, Verrier ED, Spendlove R, Stern R. Studies in fetal wound healing: II. A fetal environment accelerates fibroblast migration in vitro. J Pediatr Surg. 1989;24:793–8.
- 58. Longaker MT, Whitby DJ, Adzick NS, Crombleholme TM, Langer JC, Duncan BW, Bradley SM, Stern R, Ferguson MWJ, Harrison MR. Studies in fetal wound healing: VI. Second and early third trimester fetal wounds demonstrate rapid collagen deposition without scar formation. J Pediatr Surg. 1990;25:63.
- Longaker MT, Chiu ES, Adzick NS, Stern M, Harrison MR, Stern R. Studies in fetal wound healing: V. A prolonged presence of hyaluronic acid characterizes fetal wound fluid. Ann Surg. 1991;213:292.
- Longas MO. Evidence for structural changes in dermatan sulphate and hyaluronic acid with aging. Carbohydr Res. 1987;159:127.
- 61. Martin CW, Muir IFK. The role of lymphocytes in wound healing. Br J Plast Surg. 1990;43:655–62.
- 62. Merkel JR, DiPaolo BR, Hallock GG, Rice DC. Type I and type III collagen content of healing wounds in fetal and adult rats. Proc Soc Exp Biol Med. 1988;187:493.
- 63. Meyer K, Palmer JW. The polysaccharide of vitreous humor. J Biol Chem. 1934;107:629.
- Meyer K. Chemical structure of hyaluronic acid. Fed Proc. 1958;17:1075.
- 65. Mustoe TA, Pierce GF, Thomason A, Gramates P, Sporn MB, Deuel TF. Accelerated healing of incisional wounds in rats induced by transforming growth factor – b. Science. 1987;237:1333–6.
- Nathan CF. Secretory products of macrophages. J Clin Invest. 1987;79:319.
- 67. Nemeth GG, Bolander ME, Martin GR. Growth factors and their role in wound and fracture healing. In: Barbul A, Pines E, Caldwell M, Hunt TK, editors. Growth factors and other aspects of wound healing. New York: Alan R Liss; 1988. p. 1–17.

- Odland GF. Structure of the skin. In: Goldsmith LA, editor. Biochemistry and physiology of the skin. Oxford: Oxford University Press; 1983.
- 69. Pierce GF, Mustoe TA, Deuel TF. Transforming growth factor β induces increased directed cellular migration and tissue repair in rats. In: Barbu! A, Pines E, Caldwell M, Hunt TK, editors. Growth factors and other aspects of wound healing. New York: Alan R. Liss; 1988. p. 93–102.
- Radaelli E, Piazzini A, Marazzi M, Rizzitelli E, Rosina M, Baratti C, Trabucchi E. Role of hyaluronic acid in the wound healing. Ultrastructural observations. International symposium on cutaneous development, aging and repair, University of Padova; 1982. p. 42.
- Rappole DA, Mark D, Banda MJ, Werb Z. Wound macrophages express TGF-α and other growth factors in vivo: analysis by mRNA phenotyping. Science. 1988;241:708–12.
- Robinson BW, Goss AN. Intrauterine healing of fetal rat cheek wounds. Cleft Palate J. 1981;18:251.
- Rowsell AR. The intra-uterine healing of foetal muscle wounds: experimental study in the rat. Br J Plast Surg. 1984;37:635.
- 74. Schreiber AB, Winkler ME, Derynck R. Transforming growth factor-α: a more potent angiogenic mediator than epidermal growth factor. Science. 1986;232:1250–3.
- 75. Schilling JA, Joel W, Shurley HM. Wound healing: a comparative study of the histochemical changes in granulation tissue contained in stainless steel wire mesh and polyvinyl sponge cylinders. Surgery. 1959;46:702.
- 76. Shah M, Foreman DM, Ferguson MW. Control of scarring in adult wounds by neutralising antibody to transforming growth factor β. Lancet. 1992;339:213–4.
- Siebert JW, Burd DAR, McCarthy J, Ehrlich HP. Fetal wound healing: a biochemical study of scarless healing. Plast Reconstr Surg. 1990;85:495–502.
- Simpson DM, Ross R. The neutrophilic leukocyte in wound repair a study with antineutrophil serum. J Clin Invest. 1972;51:2009–23.
- Somasundaram K, Prathap K. Intra-uterine healing of skin wounds in rabbit foetuses. J Pathol. 1970; 100:81.
- Sporn MB, Roberts AB, Wakefield LM, Assoian RK. Transforming growth factor – β: biological function and chemical structure. Science. 1986;233:532–4.
- Sprugel KH, McPherson JM, Clowes AW, Ross R. The effects of different growth factors in subcutaneous wound chambers. In: Barbul A, Pines E, Caldwell M, Hunt TK, editors. Growth factors and other aspects of wound healing. New York: Alan R. Liss; 1988. p. 77–91.
- Toole BP. Hyaluronate turnover during chondrogenesis in the developing chick limb and axial skeleton. Dev Biol. 1972;29:321.
- Toole BP, Trelstad RL. Hyaluronate production and removal during corneal development in the chick. Dev Biol. 1971;26:28.

- 84. Toole BP, Goldberg RL, Chi-Rosso G, Underhill CB, Orkin RW. Hyaluronate cell reactions. In: Trelstad RL, editor. The role of the extracellular matrix in development. New York: Alan R. Lin; 1984. p. 43–66.
- 85. Underhill CB, Green SJ, Comoglio PM, Tarone G. The hyaluronate receptor is identical to a glycoprotein of Mr 85,000 (gp85) as shown by a monoclonal antibody that interferes with binding activity. J Biol Chem. 1987;262:13142–6.
- 86. Weigel PH, Fuller GM, LeBoeuf RD. A model for the role of hyaluronic acid and fibrin in the early events during the inflammatory response and wound healing. J Theor Biol. 1986;119:219.
- Weinstein GB, Boucek RJ. Collagen and elastin of human dermis. J Invest Dermatol. 1960;35:227–9.
- West DC, Hampson IN, Arnold F, Kumar S. Angiogenesis induced by degradation products of hyaluronic acid. Science. 1985;228:1324.
- Whitby DJ, Longaker MT, Harrison MR, Adzick NS, Ferguson MWJ. Rapid epithelialization of fetal wounds is associated with early deposition of tenascin. J Cell Sci. 1991;99:583–6.
- Whitby DJ, Ferguson MWJ. The extracellular matrix in fetal and adult wound healing. Development. 1991;112:651–68.
- Whitby DJ, Ferguson MWJ. Immunohistochemical localisation of growth factors in lip wounds of fetal, neonatal and adult mice. Dev Biol. 1991;147:207–15.

Additional References

- Campbell S, Pryse-Davies J, Coltart TM, Seller MJ, Singer JD. Ultrasound in the diagnosis of spina bifida. Lancet. 1975;1(7915):1065–8.
- 93. Canesso MC, Vieira AT, Castro TB, Schirmer BG, Cisalpino D, Martins FS, Rachid MA, Nicoli JR, Teixeira MM, Barcelos LS. Skin wound healing is accelerated and scarless in the absence of commensal microbiota. J Immunol. 2014;193:5171–80.
- Donald I, MacVicar J, Brown TG. Investigation of abdominal masses by pulsed ultrasound. Lancet. 1958;1(7032):1188–95.
- Glim JE, Everts V, Niessen FB, Ulrich MM, Beelen RH. Extracellular matrix components of oral mucosa differ from skin and resemble that of foetal skin. Arch Oral Biol. 2014;59:1048–55.
- 96. Glim JE, Beelen RH, Niessen FB, Everts V, Ulrich MM. The number of immune cells is lower in healthy oral mucosa compared to skin and does not increase after scarring. Arch Oral Biol. 2015;60:272–81.
- 97. Gourevitch D, Kossenkov AV, Zhang Y, Clark L, Chang C, Showe LC, Heber-Katz E. Inflammation and its correlates in regenerative wound healing: an alternate perspective. Adv Wound Care (New Rochelle). 2014;3:592–603.
- Larson BJ, Longaker MT, Lorenz HP. Scarless fetal wound healing: a basic science review. Plast Reconstr Surg. 2010;126:1172–80.

- 99. Rolfe KJ, Grobbelar AO. A review of fetal scarless healing. ISRN Dermatol. 2012;2012:698034.
- 100. Walmsley GG, Maan ZN, Wong VW, Duscher D, Hu MS, Zielins ER, Wearda T, Muhonen E, McArdle A, Tevlin R, Atashroo DA, Senarath-Yapa K, Lorenz HP, Gurtner GC, Longaker MT. Scarless wound healing: chasing the holy grail. Plast Reconstr Surg. 2015;135:907–17.
- 101. Walraven M, Beelen RH, Ulrich MM. Transforming growth factor-β (TGF-β) signalling in healthy human fetal skin: a descriptive study. J Dermatol Sci. 2015;78:117–24.
- Wang YC, Peterson SE, Loring JF. Protein post-translational modifications and regulation of pluripotency in human stem cells. Cell Res. 2014;24:143–60.
- 103. Wang Z, Liu X, Zhang D, Wang X, Zhao F, Zhang T, Wang R, Lin X, Shi P, Pang X. Phenotypic and functional modulation of 20–30 year old dermal fibroblasts by mid- and late-gestational keratinocytes in vitro. Burns. 2015. doi:10.1016/j. burns.2014.12.013. pii: S0305-4179(14)00447-1, [Epub ahead of print].
- 104. Woo J. A short history of the development of ultrasound in obstetrics and gynecology (http://www.obultrasound.net/history1.html. Accessed 15th Apr 2015).
- Yates CC, Hebda P, Wells A. Skin wound healing and scarring: fetal wounds and regenerative restitution. Birth Defects Res C Embryo Today. 2012;96: 325–33.

Part XV

Fetal Growth and Adult Life Implications

Implications of Gross IUGR in Adult Life with Respect to Some Major Diseases

Priyodarshi Sengupta and Niranjan Bhattacharya

Introduction

Intrauterine growth retardation is often defined as a medical phenomenon where the weight of the newborn during the gestational stage according to its sex and age are considered to be poor or less than expected according to medical and population parameters. The causes of IUGR, as it is often abbreviated its causes are still unknown and not definite although a multitude of factors can be attributed, like poor nutrition of the mother, lack of placental insufficiency resulting in diminished intrauterine fetal oxygenation, exposure to teratogens and pathogens and in some cases, molecular aberrations also termed fetal reprogramming. The long term complications of a baby born through IUGR are immense and grave like hypertension, increased risk of metabolic diseases like dyslipidemia, diabetes mellitus and cardiovascular disorders, renal impairments in

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India

e-mail: sanjuktaniranjan@gmail.com

some cases, improper brain and neuro-functional development and in some rare cases, onset of auto-immune diseases like rheumatoid arthritis and systemic lupus erythematosus.

IUGR and Increased Risk of Cardiovascular Diseases, Hypertension and Atherosclerosis Conditions

There is a high rate of cardiovascular diseases, metabolic diseases and hypertension in the early stages of development related to IUGR according to different epidemiological studies. A study conducted by Cosmi and Zanardo [1] suggested that blood pressure measurements are significantly higher in IUGR children than those born under normal growth conditions. It is postulated that due to changes in the Angiotensin activity, diseases can be triggered in IUGR patients. Also hypoxia, increased sympathetic nerve activity and catecholamine production may augment the pathogenetic trigger [2].

A strong positive correlation was found between the reduced number of nephrons [3–7] and hypertension in case of sheep [8], rat [9] and humans [10]. Many animal models have shown, due to lack of proper growth of the renal system, onset of hypertension is extremely high. Studies conducted by Michael G. Ross and Marie H. Beall in rat models has proved altered gene

P. Sengupta, MSc, MPhil

Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development: First and Second Trimesters*, DOI 10.1007/978-3-319-14874-8_38

expression in the rat fetal kidneys which has a direct relation with an increased propensity to adult hypertension in later life [11]. It is further suggested that the end of the second trimester till 34 weeks is the most vulnerable period during which altered renal development can result in hypertension of the neonate [4].

The aetiology of hypertension among IUGR patients are multi factorial. Another important factor leading to hypertension in IUGR affected children are the changes in the composition and structure of the extra cellular matrix which often helps in maintaining the arterial pressure [12]. The status of the ECM often changes as seen in adult rats with 8 weeks of high salt intake; Changes like increase in the aortal wall thickness, decrease in the content of collagen, elastin, smooth muscles, which are important extracellular components of the ECM [13]. This overall change of the ECM also affects the vascular changes, as observed in IUGR adults in animal models. Also the incidence of hypertension due to IUGR is observed more in male offspring [11].

The vascular endothelium is another area where several studies have shown that endothelial mediated or non mediated vasodilation is normally impaired in IUGR patients, resulting in a decrease of flow mediated dilation in the birth weight of neonates, later childhood and early adulthood [14–16]. Undernourished rats in the first 18 days of gestation also showed reduced vasoconstriction response to phenylephrine and norepipnephrine in isolateral femoral arteries [11].

Impaired growth in utero often results in endothelial dysfunction which is an early stage of atherosclerosis. This event silently develops in the neonates and often culminates in increased CVD's or an early stroke [17]. This can be detected by means of ultrasound imaging. Skilton and co-workers in the year 2005 compared intima-mediated thickness or aIMT of the aortic wall in newborn low weight infants with respect to normal weight babies and thereby developed a biomarker of atherosclerosis in infants via USG guided assistance [18]. Similarly carotid thickness intima was shown to be greater in IUGR children which was further confirmed by Crispi et al. [19]. These changes further persisted in childhood stages of IUGR born infants with increased risk for myocardial infarction and stroke. The reason behind the increase in the arterial wall thickness can be attributed to the fact that in IUGR babies there is an increased chance of metabolic reprogramming which further has a direct effect on the changes in cardiac morphology, clinical cardiac dysfunction and hypertension. These factors were also found to be independent of gestational age at delivery, lipid profile and body mass index of the infant [2].

It is postulated that cardiovascular diseases in adults are often due to quality of life as well as due to genetic factors. However a new concept has emerged regarding the onset of cardiovascular diseases which can be attributed to low birth weight especially among IUGR infants. As David Barker in 1989 [20] in Southampton confirmed a direct correlation between increased cardiovascular mortality and IUGR, it is now a well known fact also observed in rat and other animal models [21, 22] that IUGR born infants have a high susceptibility to disorders like dyslipidemia, coagulation, diabetes and hypertension [23]. This can be attributed to the high level of genetic complexity and their different epigenetic interactions and is known as the "Fetal reprogramming" where any molecular or genetic changes to the fetus in utero can result in key organ changes [22-25]. During the time of fetal development the cells undergo extensive mitotic divisions and thereby remain sensitive even to the smallest changes that can impair the structural growth and development of the fetus [23].

The proposed mechanism of IUGR induced CVD and hypertensive disorders can be related to four important factors: (a) lack of proper nutritional diet by the mother, (b) uteroplacental dysfunction, (c) exposure to harmful toxic drugs/ chemicals and (d) environmental, external factors and genetic changes.

Many rat studies have confirmed that maternal low-protein diet during the gestational phase gave rise to a consistent rise in blood pressure, endothelial dysfunction, vascular dysfunction, increased ACE activity, decreased nephron numbers and increased oxidative stress in adulthood [26–32]. Also it was seen that a permanent ligation of both the uterine arteries led to hypoxia and increased obstruction of blood flow which later had a direct implication on the onset of diabetes type 2, altered brain development and proteinuria in both the early and adult stages [33, 34]. It has been further observed in rats that with low protein diet, pulmonary and cardiac genes can get affected too.

Treatment with glucorticoid dexamethasone of pregnant rats has shown low birth weights and persistent elevation of the arterial blood pressure in both adults and offsprings [35, 36]. Also low protein diet has been found to increase the maternal glucocorticoid level due to a decrease in activity of 11 Beta-HSD2 in placenta as observed in animal models leading to low birth weights and persistent hypertension in adult life too [37]. Environmental factors like high altitude, maternal ventilation during pregnancy, can give rise to increased arterial oxygen saturation [37]. When pregnant rats were exposed to chronic hypoxia between 15 and 21 days of gestation, there was an increase in the percentage of fetal cardiac apoptosis and an increased susceptibility to high CVD, cardiac hypertrophy and ischemic reperfusion injuries [37, 38]. Maternal smoking can be also one of the most important aetiology to risks of hypertension and development of IUGR in neonates leading to fatal consequences in late adulthood [39, 40].

The fetal environment can also result in epigenetic modifications like changes in gene expression, mutation or alteration in the DNA sequences, activation or silencing of the gene expression through DNA methylation, histone modifications and micro RNAs. These alterations can have a serious effect on the onset of CVD among IUGR induced babies and in adults.

Lack of follow ups in epidemiological studies is one of the commonest obstacles faced by researchers and clinicians while assessing the relation of IUGR and low birth weights. A study which included 15,000 Swedish men and women over 50 years with a 97 % follow up, revealed a strong correlation ship between CVD and low birth weight [37, 41] whereas another study by Rich-Edwards inferred that that was a strong negative relationship between incidence of birth weights and non fatal coronary heart disease and stroke [37, 41, 42].

IUGR Effects on the Brain and Cognitive Developments

Multiple follow up studies, clinical datas and observations have shown that intra-uterine growth factor is consistently related to many disabilities related to cognitive function, cognitive skills, language, abstract reasoning and thinking, gross motor skills, memory, concentration and mood related disorders [43]. Animal studies have recently related the negative effects of chronic intra-uterine hypoxia and malnutrition to IUGR which in turn has a direct effect on the neuronal cell number restriction, cell size, lighter weight of the brain, lower neuronal DNA content as well as on the (reduced) number of functional synaptic connections [43, 44].

Further progressive deterioration of the hippocampal pyramidal neurons, loss of oligodendritic branches and reduction of the cellularity by 30 % [45] has been observed in animal models with IUGR. These losses resulted in a decrease of the overall hippocampal volume accompanied with loss of density in the granular neurons [46]. With age there have been reports of increasing neuro endocrine anomalies resulting in the modification of the hypothalamo-pituitary adrenal axis (HPA) [43]. Also recent studies pertaining to IUGR infants have observed that repeated antenatal glucocorticoid therapies and post natal corticosteroid therapies have resulted in significant reductions of the overall brain size and in the cortical gray matter volume in case of IUGR infants [47, 48].

It has been well documented that early fetal growth retardation has a direct implication on the neuronal and neurobehavioral development of the individual [49–51]. The different behavioural domains like academic skills, cognitive abilities, problem solving skills are significantly lower in cases with children born with IUGR defects. These limitations were related to the frontal lobe function which fails to develop completely in case of IUGR born babies. This can be also attributed to maternal malnutrition especially during the third trimester that is when the brain growth maintains maximum maturity and becomes more vulnerable. Fetal retardation can often lead to the

birth of babies with smaller head size. Cerebral arterial flow via Doppler findings have shown that changes in the cerebral flow can induce specific developmental abnormalities of the fetal brain which can have a profound effect on the child during the onset of adulthood [52]. Abnormal neural development in children and adults are among one of the most important consequences of intrauterine growth restriction at the time of birth.

IUGR can be normally classified into symmetric and asymmetric. In case of symmetric IUGR, the head, length and weight of the body becomes proportionally small for gestational age (SGA); it is termed as asymmetric when the head grows bigger compared to the length and weight of the body. This phenomenon is known as the "Brain Sparing effect" which is a haemodynamic effect with increased flow of blood to the brain thereby directing most of its activity towards the main vital organs [52, 53]. Children born through IUGR also show signs of restricted brain development with major reduction in the frontal lobe region and volume thereby affecting various daily skills like speech, language and creativity. According to the statistics released by the Institute of Medicine, one-third American women who have given births to babies with encephalopathies and neural tube defects, had been subject to excess nutrition where BMI >25 kg/m²; further, more than 30 % were obese [54]. Therefore maternal nutrition has a profound effect on the outcome of the neural and brain development of the child which if not properly developed can have a debilitating effect in the later stages of adolescence and mainly impairment in cognitive skills as it has been observed among IUGR born 7 year old kids with regressive IQ [55].

Autoimmunity and IUGR

Among auto immune diseases, Systemic lupus erythromatosis and rheumatoid arthritis deserves special mention as these are the most widely studied auto immune diseases in infants, especially children who are born via IUGR method. Early life events like IUGR have long lasting effects on the immune function thereby resulting in RA and SLE [56]. IUGR is primarily linked to malnutrition of the mother and this malnutrition has a severe effect on the thymus maturation and development. One of the main functions of the pre natal and neonatal immune systems is to continuously learn the process of recognizing and destroying infections without any induction of auto immune disease. T cells of the thymus play a major role in this function and failure to form the thymus in the developmental stages or rather the formation of a small immature thymus results in the variation of the self-tolerance phenomenon [57] as seen in animal models. Normally in human IUGR conditions there is a long lasting effect on immunity including a reduced response to recall antigens from childhood vaccination stages [56].

Implications of IUGR on Metabolic Disorders and Diabetes and in Adult Life

Many of the metabolic disorders like insulin resistance, dyslipidemia or fat deposition can be attributed to IUGR, since there is an endocrine origin to changes in the hormone bioavailability during early growth and fetal development [58-61]. Abnormal levels of catecholamines, growth hormones, insulin have been observed in children born with IUGR [62, 63]. Seventy percent of the whole body glucose uptake is initiated by the skeletal muscles which is also the primary centre for insulin mediated glucose uptake apart from other organs like liver, pancreas and adipose tissues. It has been observed that hypoxia induced in the in utero environment can result in the diversion of blood flow maximum to the brain and adrenal glands thereby resulting in a baby with a higher head mass and low skeletal and body mass [64, 65] resulting in severe altered insulin resistance and other metabolic disorders. These observations tend towards a hypothesis that changes in key endocrine factors like leptin, adiponectin can often be associated with intrauterine conditions which might give rise to adult metabolic diseases in the fetus in its early childhood stage [61].

IUGR is also found to be associated with abnormality in the fetal adipose tissue and growth apart from the fetal skeletal muscles. IUGR born babies show a remarkable reduction in the body fat mass mainly affecting the amount of adipocytes. Although total body fat reduces, increase in the visceral fat has shown to occur [66]. This results in rapid increase of distribution of centralized fat tissues even if they are not overweight [67, 68] with abdominal tissue showing hyper reactivity to catecholamines and early insulin resistance [69]. Also polymorphism of the gene peroxisome proliferative activated receptor g2 (PPARG) has been shown to impair metabolic function resulting in a case of insulin resistance leading to a high risk of Diabetes Mellitus type 2 [69–71].

Adipocytokines of the adipocyte derived hormones resembling an endocrine type of activity helps in regulating the body metabolism and homeostasis [72–74] and plays an important role in the metabolic activities in early life and might be abrogated in cases of babies with IUGR [67].

According to recent data, malnutrition of the fetus can result in obesity or obesity related disorders that might alter the secretion of important metabolic hormones like leptin and adiponectin [75]. Leptin hormone plays a central role in maintaining the functional characteristics of mass before birth and in the control of substrate utilization. IUGR is directly related to lower concentration of leptin as several studies have shown decreased placental leptin production [76] resulting in reduced fat mass and adipose tissue accumulation [77]. However other researchers have concluded higher concentration of leptin in IUGR infants, children and adults when compared with normal born babies, children or adults. This might be due to the adipocyte dysfunction associated with IUGR [78].

Further, higher concentration of leptin in IUGR, according to an older report, might be attributed to the differences in the fetal oxygenation status as leptin is highly sensitive to oxygen abundance [79] and IUGR fetuses with severe distress have shown to include significantly higher leptin concentration per kilogram of weight [77] which may lead to the pre disposition of excess fat in later life [80]. It has been observed that maternal administration of leptin results in an increase in fetal pancreatic insulin content which further gives the fetus a long term protection from type 2 Diabetes mellitus and obesity.

Adiponectin, another important and abundant cytokine found in the adipose tissue, plays an important role in modulation of glucose and lipid metabolism in insulin sensitive tissues [81]. Circulating adiponectin concentrations help in the decrease of insulin resistance states and its content is inversely proportional to the body mass and weight, unlike leptin [82]. Adiponectin is also produced in the intra uterine environment by both the placenta and the fetus [67, 83] thereby suggesting its important role in fetal growth and promotion of insulin via its insulin sensitizing mode of action [83]. Through different clinical observations among IUGR children it is believed that low adiponectin levels in IUGR infants may actually predict the subsequent development of visceral fat and insulin resistance [67]. Contradictory to this certain researches observed normal levels of adiponectin in animals and humans inspite of the fact that many are insulin resistant [67].

An interesting observation is also that the shift in insulin sensitivity appears to be sex specific as seen in case of rodents. It has been noticed that that in cases of males there is a higher rate of Insulin impairment when compared to that of females [84].

Adult IUGR human subjects having insulin resistance have demonstrated a failure to up regulate muscle GLUT 4 after insulin stimulation [85–87]. In low weight human infants a reduced expression of the transcription factors p85, p110 beta subunits of PI3k has been observed with a reduced skeletal muscle expression of the GLUT4 expression and further a blunted phosphorylation of Akt with respect to insulin infusion [88].

OXPHOS or oxidative phosphorylation is also impaired in low birth animals due to decrease in mitochondrial NOS [89]. This results in low birth weights and decreased ATP and insulin production [90, 91]. It has been reported that in humans with low birth weights, an increase in glycolytic fibers also precedes insulin resistance [92].

Additionally high amounts of saturated fatty acids have also been related to insulin resistance [62]. Any alteration in fatty acid metabolism and oxidation impairs the ability of the skeletal muscle metabolic activities and is shown to have a direct effect on IUGR [89, 93]. This results in the accumulation of lipid metabolites in the skeletal structure which is linked with increase in stress induced protein kinases C (pkC), c-JNK, inhibition of NFkappa Beta kinase, further leading to insulin impairment.

Another important reason for the later onset of diabetes disorders as found in IUGR born individuals can be related to the fact that due to increased hepatic glucogenesis observed in rats [11], it can often precede the development of hyperglycaemia resulting in the resistance to insulin effects. Up regulation of PPAR (peroxisome proliferator-activated) gamma co-activator 1 alpha, mRNA regulator of glucose-6-phosphate and other gluconeogenesis enzymes may also alter the hepatic glucose production resulting in intracellular cell signalling changes and high glucose production which might overcome the effect of insulin [11].

Conclusion

From the above review it can be said that malnutrition is one of the major factors behind the cause of IUGR in rodent model though the problem is not so simple in the human system. There was an observation made during the second world war that the weight of the fetus upto 24 weeks showed no statistical difference in mothers belonging to the social upper or elite class starvation induced mothers with progressive malnutrition, who were deprived by the invading army.

IUGR has severe implications from the post natal period to adult life of an individual and can be fatal sometimes. With the development and advent of modern diagnostic techniques like three-dimensional USG, MRI and other invasive techniques, IUGR can be detected at a much earlier stage in the pregnant woman and thereby necessary steps can be taken to prevent further deterioration. Normally fetuses have an alternative mechanism to combat nutritional restrictions and severe fetal hypoxia but often an increase of such high risk factors can severely impair the growth and function of the fetus in utero and can have major implications in future growth and development.

References

- Zanardo V, Fanelli T, Weiner G, Fanos V, Zaninotto M, Visentin S, Canallin F, Trevisanuto D, Cosmi E. Intra uterine growth retardation is associated with persistent aortic wall thickening and glomerular proteinuria during infancy. Kidney Int. 2011;80(1):119–23.
- Cosmi E, Fanelli T, Visentin S, Trevisanuto D, Zanardo V. Consequences in infants that were intrauterine growth restricted. J Pregnancy. 2011;2011, 364381.6 pages.
- Manalich R, Reyes L, Herrera M, et al. Relationship between weight at birth and the number and size of renal glomeruli in humans: a histomorphometric study. Kidney Int. 2000;58:770–3. PubMed: 10916101.
- Konje JC, Bell SC, Morton JJ, et al. Human fetal kidney morphometry during gestation and the relationship between weight, kidney morphometry and plasma active renin concentration at birth. Clin Sci (Lond). 1996;91:169–75. PubMed: 8795440.
- Schreuder MF, Nyengaard JR, Remmers F, et al. Postnatal food restriction in the rat as a model for a low nephron endowment. Am J Physiol Renal Physiol. 2006;291:F1104–7.
- Brenner BM, Chertow GM. Congenital oligonephropathy: an inborn cause of adult hypertension and progressive renal injury? Curr Opin Nephrol Hypertens. 1993;2:691–5.
- Lisle SJ, Lewis RM, Petry CJ, et al. Effect of maternal iron restriction during pregnancy on renal morphology in the adult rat offspring. Br J Nutr. 2003;90:33–9. PubMed: 12844373.
- Gilbert JS, Lang AL, Grant AR, et al. Maternal nutrient restriction in sheep: hypertension and decreased nephron number in offspring at 9 months of age. J Physiol. 2005;565:137–47. PubMed: 15790663.
- Langley-Evans SC, Welham SJ, Jackson AA. Fetal exposure to a maternal low protein diet impairs nephrogenesis and promotes hypertension in the rat. Life Sci. 1999;64:965–74. PubMed: 10201645.
- Brenner BM, Mackenzie HS. Nephron mass as a risk factor for progression of renal disease. Kidney Int Suppl. 1997;63:S124–7.
- 11. Ross MG M.D., M.P.H., Beall MH M.D. Adult sequelae of intrauterine growth restriction. Semin

Perinatol. 2008;32(3):213–8. doi:10.1053/j.semperi.2007.11.005. NIH main paper.

- Shadwick RE. Mechanical design in arteries. J Exp Biol. 1999;202:3305–13. PubMed: 10562513.
- De Rooij SR, Painter RC, Phillips DI, et al. Impaired insulin secretion after prenatal exposure to the Dutch famine. Diabetes Care. 2006;29:1897–901 [Pub Med: 16873799].
- Goh KL, Shore AC, Quinn M, et al. Impaired microvascular vasodilatory function in 3-month-old infants of low birth weight. Diabetes Care. 2001;24:1102–7. PubMed: 11375378.
- Goodfellow J, Bellamy MF, Gorman ST, et al. Endothelial function is impaired in fit young adults of low birth weight. Cardiovasc Res. 1998;40:600–6. PubMed: 10070502.
- Martin H, Hu J, Gennser G, et al. Impaired endothelial function and increased carotid stiffness in 9- year-old children with low birth weight. Circulation. 2000; 102:2739–44.
- Järvisalo MJ, Jartti L, Näntö-Salonen K, et al. Increased aortic intima-media thickness: a marker of preclinical atherosclerosis in high-risk children. Circulation. 2001;104(24):2943–7.
- Skilton MR, Evans N, Griffiths KA, Harmer JA, Celermajer DS. Aortic wall thickness in newborns with intrauterine growth restriction. Lancet. 2005;365(9469):1484–6.
- Crispi F, Bijnens B, Figueras F, et al. Fetal growth restriction results in remodeled and less efficient hearts in children. Circulation. 2010;121(22):2427–36.
- Barker DJ. In utero programming of cardiovascular disease. Theriogenology. 2000;53(2):555–74.
- Barker DJ. Fetal origins of coronary heart disease. BMJ. 1995;311:171–4.
- 22. Tintu A, Rouwet E, Verlohren S, Brinkmann J, Ahmad S, Crispi F, van Bilsen M, Carmeliet P, Staff AC, Tjwa M, Cetin I, Gratacos E, Hernandez-Andrade E, Hofstra L, Jacobs M, Lamers WH, Morano I, Safak E, Ahmed A, le Noble F. Hypoxia induces dilated cardiomyopathy in the chick embryo: mechanism, intervention, and long-term consequences. PLoS One. 2009;4:e5155.
- Gluckman PD. Developmental origins of disease paradigm: a mechanistic and evolutionary perspective. Pediatr Res. 2004;56:311–7.
- Palinski W, Napoli C. Impaired fetal growth, cardiovascular disease, and the need to move on. Circulation. 2008;117:341–3.
- Phillips D. Insulin resistance as a programmed response to fetal under nutrition. Diabetologia. 1996; 39:1119–22.
- Gluckman PD, Hanson MA, Cooper C, Thornburg KL. Effect of in utero and early life conditions on adult health. N Engl J Med. 2008;359:61–73.
- Tarry-Adkins JL, Ozanne SE. Mechanisms of early life programming: current knowledge and future directions. J Clin Nutr. 2011;96(Suppl):1765S–71.
- McMillen IC, Robinson JS. Developmental origins of the metabolic syndrome: prediction, plasticity, and programming. Physiol Rev. 2005;85:571–633.

- Langley SC, Jackson AA. Increased systolic blood pressure in adult rats induced by fetal exposure to maternal low-protein diets. Clin Sci (Lond). 1994;86:217–22.
- Snoeck A, Remacle C, Reusens B, Hoet JJ. Effect of a low-protein diet during pregnancy on the fetal rat endocrine pancreas. Biol Neonate. 1990;57:2107–18.
- Nuyt AM, Alexander BT. Developmental programming and hypertension. Curr Opin Nephrol Hypertens. 2009;18:144–52.
- 32. Watkins AJ, Lucas ES, Torrens C, Cleal JK, Green L, Osmond C, Eckert JJ, Gray WP, Hanson MA, Fleming TP. Maternal low-protein diet during mouse preimplantation development induces vascular dysfunction and altered renin-angiotensin-system homeostasis in the offspring. Br J Nutr. 2010;103:1762–70.
- Barnes SK, Ozanne SE. Pathways linking the early environment to long-term health and lifespan. Prog Biophys Mol Biol. 2011;106:323–36.
- Rehn AE, Van Den Buuse M, Copolov D, Briscoe T, Lambert G, Rees S. An animal model of chronic placental insufficiency: relevance to neurodevelopmental disorders including schizophrenia. Neuroscience. 2004;129:381–91.
- Benediktsson R, Lindsay RS, Noble J, Seckl JR, Edwards CR. Glucocorticoid exposure in utero: new model for adult hypertension. Lancet. 1993;341: 339–41.
- 36. Levitt NS, Lindsay RS, Holmes MC, Seckl JR. Dexamethasone in the last week of pregnancy attenuates hippocampal glucocorticoid receptor gene expression and elevates blood pressure in the adult offspring. Neuroendocrinology. 1996;64:412–8.
- Demicheva E, Crispi F. Long-term follow-up of intrauterine growth restriction: cardiovascular disorders. Fetal Diagn Ther. 2014;36:143–53. doi: 10.1159/000353633.
- Li G, Xiao Y, Estrella JL, Ducsay CA, Gilbert RD, Zhang L. Effect of fetal hypoxia on heart susceptibility to ischemia and reperfusion injury in the adult rat. J Soc Gynecol Investig. 2003;10:265–74.
- Blake KV, Gurrin LC, Evans SF, Beilin LJ, Landau LI, Stanley FJ, Newnham JP. Maternal cigarette smoking during pregnancy, low birth weight and subsequent blood pressure in early childhood. Early Hum Dev. 2000;57:137–47.
- 40. Brion MJ, Leary SD, Lawlor DA, Smith GD, Ness AR. Modifiable maternal exposures and offspring blood pressure: a review of epidemiological studies of maternal age, diet, and smoking. Pediatr Res. 2008; 63:593–8.
- 41. Leon DA, Lithell HO, Vâgerö D, Koupilová I, Mohsen R, Berglund L, Lithell UB, McKeigue PM. Reduced fetal growth rate and increased risk of death from ischemic heart disease: cohort study of 15,000 Swedish men and women born 1915–29. BMJ. 1998;317:241–5.
- Rich-Edwards JW, Stampfer MJ, Manson JE, Rosner B, Hankinson SE, Colditz GA, Willett WC, Hennekens CH. Birth weight and risk of cardiovascular disease in

a cohort of women followed up since 1976. BMJ. 1997;315:369-400.

- 43. Tolsa CB, Zimine S, Warfield SK, Freschi M, Rossignol AS, Lazeyras F, Hanquinet S, Pfizenmaier M, Hüpp PS. Early alteration of structural and functional brain development in premature infants born with intrauterine growth restriction. Pediatr Res. 2004;56(1):132–8. doi:10.1203/01.PDR.0000128983.54614.7E.
- 44. Cragg BG. The development of cortical synapses during starvation in the rat. Brain. 1972;95:143–50.
- 45. Uno H, Lohmiller L, Thieme C, Kemnitz JW, Engle MJ, Roecker EB, Farrell PM. Brain damage induced by prenatal exposure to dexamethasone in fetal rhesus macaques. I. Hippocampus. Brain Res Dev Brain Res. 1990;53:157–67.
- 46. Uno H, Eisele S, Sakai A, Shelton S, Baker E, DeJesus O, Holden J. Neurotoxicity of glucocorticoids in the primate brain. Horm Behav. 1994;28:336–48.
- 47. Modi N, Lewis H, Al Naqeeb N, Ajayi-Obe M, Dore CJ, Rutherford M. The effects of repeated antenatal glucocorticoid therapy on the developing brain. Pediatr Res. 2001;50:581–5.
- 48. Murphy BP, Inder TE, Hüppi PS, Zientara GP, Warfield S, Zientara GP, Kikinis R, Jolesz FA, Volpe JJ. Impaired cerebral cortical gray matter growth following treatment with dexamethasone for neonatal chronic lung disease. Pediatrics. 2001;107:217–21.
- Arduini D, Rizzo G, Caforio L, Boccolini MR, Romanini C, Mancuso S. Behavioural state transitions in healthy and growth retarded fetuses. Early Hum Dev. 1989;19:155–65.
- Arduini D, Rizzo G, Romanini C, Mancuso S. Computerized analysis of behavioural states in asymmetrical growth retarded fetuses. J Perinat Med. 1988;16:357–63.
- Nijhuis IJ, ten Hof J, Nijhuis JG, Mulder EJ, Narayan H, Taylor DJ, Visser GH. Temporal organisation of fetal behaviour from 24-weeks gestation onwards in normal and complicated pregnancies. Dev Psychobiol. 1999;34:257–68.
- Baschat AA. Neurodevelopment after fetal growth restriction. Fetal Diagn Ther. 2014;36:136–42. doi:10.1159/000353631.
- Turan OM, Turan S, Gungor S, Berg C, Moyano D, Gembruch U, Nicolaides KH, CR H, Baschat AA. Progression of Doppler abnormalities in intrauterine growth restriction. Ultrasound Obstet Gynecol. 2008;32:160–7.
- 54. Institute of Medicine. Influence of pregnancy weight on maternal child health: a workshop report. Washington DC: National Academy Press; 2007.
- Georgrieff MK. Nutrition and the developing brain: nutrient priorities and measurement 1,2,3. Am J Clin Nutr. 2007;85(2):614S–20S.
- Edwards CJ, Cooper C. Early environmental factors and rheumatoid arthritis. Clin Exp Immunol. 2006;143(1): 1–5. doi:10.1111/j.1365-2249.2005.02940.
- 57. Muaku SM, Thissen JP, Gerard G, Ketelslegers JM, Maiter D. Postnatal catch-up growth induced by growth hormone and insulin-like growth factor-I in

rats with intrauterine growth retardation caused by maternal protein malnutrition. Pediatr Res. 1997;42:370–7.

- Osmond C, Barker DJ. Fetal, infant and childhood growth are predictors of coronary heart disease, diabetes, and hypertension in adult men and women. Environ Health Perspect. 2000;108:545–53. doi:10.1289/ehp.00108s3545 [PubMed].
- 59. Kind KL, Clifton PM, Grant PA, Owens PC, Sohlstrom A, Roberts CT, Robinson JS, Owens JA. Effect of maternal feed restriction during pregnancy on glucose tolerance in the adult guinea pig. Am J Physiol Regul Integr Comp Physiol. 2003;284:R140–52 [PubMed].
- Leger J, Levy-Marchal C, Bloch J, Pinet A, Chevenne D, Porquet D, Collin D, Czernichow P. Reduced final height and indications for insulin resistance in 20 year olds small for gestational age: regional cohort study. BMJ.1997;315:341–7. doi:10.1136/bmj.315.7104.341 [PubMed].
- Fowden AL, Giussani DA, Forhead AJ. Endocrine and metabolic programming during intrauterine development. Early Hum Dev. 2005;81:723–34.
- Fowden AL, Forhead AJ. Endocrine mechanisms of intrauterine programming. Reproduction. 2004;127: 515–26.
- Phillips DI. Fetal growth and programming of the hypothalamic– pituitary–adrenal axis. Clin Exp Pharmacol Physiol. 2001;28:967–70.
- 64. Froguel P, Zouali H, Vionnet N, Velho G, Vaxillaire M, Sun F, Lesage S, Stoffel M, Takeda J, Passa P, Permutt MA, Beckmann JS, Bell GI, Cohen D. Familial hyperglycemia due to mutations in glucokinase. Definition of a subtype of diabetes mellitus. N Engl J Med. 1993;328:679–702.
- 65. Hattersley AT, Beards F, Ballantyne E, Appleton M, Harvey R, Ellard S. Mutations in the glucokinase gene of the fetus result in reduced birth weight. Nat Genet. 1998;19:268–70.
- 66. Yajnik CS, Lubree HG, Rege SS, Naik SS, Deshpande JA, Deshpande SS, Joglekar CV, Yudkin JS. Adiposity and hyperinsulinemia in Indians are present at birth. J Clin Endocrinol Metab. 2002;87:5575–80.
- Briana DD, Malamitsi-Puchner A. Intrauterine growth restriction and adult disease: the role of adipocytokines. Eur J Endocrinol. 2009;160:337–47.
- Ibanez L, Lopez-Bermejo A, Suarez L, Marcos MV, Diaz M, de Zegher F. Visceral adiposity without overweight in children born small for gestational age. J Clin Endocrinol Metab. 2008;93:2079–83.
- Jaquet D, Gaboriau A, Czernichow P, Levy-Marchal C. Insulin resistance early in adulthood in subjects born with intrauterine growth retardation. J Clin Endocrinol Metab. 2000;85:1401–6.
- 70. Eriksson JG, Lindi V, Uusitupa M, Forsen TJ, Laakso M, Osmond C, Barker DJ. The effects of the Pro12Ala polymorphism of the peroxisome proliferatorsactivated receptor-gamma2 gene on insulin sensitivity and insulin metabolism interact with size at birth. Diabetes. 2002;51:2321–4.

- Boiko J, Jaquet D, Chevenne D, Rigal O, Czernichow P, LevyMarchal C. In situ lipolytic regulation in subjects born small for gestational age. Int J Obes. 2005;29:565–70.
- Hoggard N, Hoggarty P, Thomas L, Lea RG. Leptin expression in placental and fetal tissues: does leptin have a functional role? Biochem Soc Trans. 2001;29:57–62.
- Christou H, Serdy S, Mantzoros CS. Leptin in relation to growth and developmental processes in the fetus. Semin Reprod Med. 2002;20:123–30.
- 74. Cortelazzi D, Corbetta S, Ronzoni S, Pelle F, Marconi A, Cozzi V, Cetin I, Cortelazzi R, Beck-Peccoz P, Spada A. Maternal and foetal resistin and adiponectin concentrations in normal and complicated pregnancies. Clin Endocrinol. 2007;66:447–53.
- Krechowec SO, Vickers M, Gertler A, Breier BH. Prenatal influences on leptin sensitivity and susceptibility to diet-induced obesity. J Endocrinol. 2006;189:355–63.
- Lepercq J, Guerre-Millo M, Andre J, Cauzac M, Hauguel-de Mouzon S. Leptin: a potential marker of placental insufficiency. Gynecol Obstet Investig. 2003;55:151–5.
- 77. Cetin I, Morpurgo PS, Radaelli T, Taricco E, Cortellazzi D, Bellotti M, Pardi G, Beck-Peccoz P. Fetal plasma leptin concentrations: relationship with different intrauterine growth patterns from 19 weeks to term. Pediatr Res. 2000;48:646–51.
- Jaquet D, Leger J, Tabone MD, Czernichow P, Levy-Marchal C. High serum leptin concentrations during catch-up growth of children born with intrauterine growth retardation. J Clin Endocrinol Metab. 1999;84:1949–53.
- 79. Grosfeld A, Andre J, Hauguel-De Mouzon SH, Berrat E, Pouyssegur J, Guerre-Millo M. Hypoxia-induced factor 1 transactivates the human leptin gene promoter. J Biol Chem. 2002;277:42953–7. 83 Stocker C, O'Dowd J, Morton NM, Wargent E.
- Shekawat PS, Garland JS, Shivpuri C, Mick GJ, Sasidharan P, Pelz CJ, McCormick KL. Neonatal cord blood leptin: its relationship to birth weight, body mass index, maternal diabetes, and steroids. Pediatr Res. 1998;43:338–43.
- Schondorf T, Maiworm A, Emmison N, Forst T, Pfutzner A. Biological background and role of adiponectin as marker for insulin resistance and cardiovascular risk. Clin Lab. 2005;51:489–94.
- 82. Arita Y, Kihara S, Ouchi N, Takahashi M, Maeda K, Miyagawa J, Hotta K, Shimomura I, Nakamura T, Miyaoka K, Kuriyama H, Nishida M, Yamashita S, Okudo K, Matsubara K, Muraguchi M, Ohmoto Y, Funahashi T, Matsuzawa Y. Paradoxical decrease of an adipose-specific protein, adiponectin, in obesity. Biochem Biophys Res Commun. 1999;257:79–83.
- Tsai PJ, Yu CH, Hsu SP, Lee YH, Chiou CH, Hsu YW, Ho SC, Chu CH. Cord plasma concentrations of adiponectin and leptin in healthy term neonates: posi-

tive correlation with birthweight and neonatal adiposity. Clin Endocrinol. 2004;61:88–93.

- 84. Owens JA, Thavaneswaran P, De Blasio MJ, McMillen IC, Robinson JS, Gatford KL. Sex-specific effects of placental restriction on components of the metabolic syndrome in young adult sheep. Am J Physiol Endocrinol Metab. 2007;292:E1879–89. doi:10.1152/ ajpendo.00706.2006 [PubMed].
- 85. Brons C, Jacobsen S, Nilsson E, Ronn T, Jensen CB, Storgaard H, Poulsen P, Groop L, Ling C, Astrup A, et al. Deoxyribonucleic acid methylation and gene expression of PPARGC1A in human muscle is influenced by high-fat overfeeding in a birth-weightdependent manner. J Clin Endocrinol Metab. 2010;95:3048–56. doi:10.1210/jc.2009-2413 [PubMed].
- Blesson CS, Sathishkumar K, Chinnathambi V, Yallampalli C. Gestational protein restriction impairs insulin regulated glucose transport mechanisms in gastrocnemius muscles of adult male offspring. Endocrinology. 2014. doi:10.1210/en.2014-1094 [PMC free article] [PubMed].
- Shulman GI. Cellular mechanisms of insulin resistance. J Clin Investig. 2000;106:171–6. doi:10.1172/ JCI10583 [PMC free article] [PubMed].
- Dunlop K, et al. Altered fetal skeletal muscle nutrient metabolism following an adverse in utero environment and the modulation of later life insulin sensitivity. Nutrients. 2015;7.2:1202–16. *PMC*. Web. 2 May 2015.15. Vuguin P, Raab E, Liu B, et al. Hepatic insulin resistance precedes the development of diabetes in a model of intrauterine growth retardation. Diabetes. 2004;53:2617–22. [PubMed: 15448092]
- Beauchamp B, Ghosh S, Dysart MW, Kanaan GN, Chu A, Blais A, Rajamanickam K, Tsai EC, Patti ME, Harper ME. Low birth weight is associated with adiposity, impaired skeletal muscle energetics and weight loss resistance in mice. Int J Obes. 2014. doi:10.1038/ ijo.2014.120.
- Albrecht E, Lembcke C, Wegner J, Maak S. Prenatal muscle fiber development and bundle structure in beef and dairy cattle. J Anim Sci. 2013;91:3666–73. doi:10.2527/jas.2013-6258 [PubMed].
- 91. St-Pierre J, Lin J, Krauss S, Tarr PT, Yang R, Newgard CB, Spiegelman BM. Bioenergetic analysis of peroxisome proliferator-activated receptor gamma coactivators lalpha and lbeta (PGC-lalpha and PGC-lbeta) in muscle cells. J Biol Chem. 2003;278:26597–603. doi:10.1074/jbc.M301850200 [PubMed].
- 92. Jensen CB, Storgaard H, Madsbad S, Richter EA, Vaag AA. Altered skeletal muscle fiber composition and size precede whole-body insulin resistance in young men with low birth weight. J Clin Endocrinol Metab. 2007;92:1530–4. doi:10.1210/jc.2006-2360.
- Selak MA, Storey BT, Peterside I, Simmons RA. Impaired oxidative phosphorylation in skeletal muscle of intrauterine growth-retarded rats. Am J Physiol Endocrinol Metab. 2003;285:E130–7.
Part XVI

Understanding Fetal Growth from the Perspective of Alternative Medicine

Growth and Development of the Human Fetus Up to the Second Trimester: A Perspective of Traditional Chinese Medicine

39

Mursheed Ali and Niranjan Bhattacharya

Introduction

Traditional ways of practicing medicines during pregnancy are still prevalent in many nations, for example, Turkey [1], African countries [2], Thailand [3] and China. There is a likelihood that following traditional practices during pregnancy has both medicinal and harmful [1, 4] outcomes. With regard to pregnancy, there are no significant differences among distinctive socioeconomic groups. Some women may have doubts about traditional customs, but still take after the advice of the older generation [3], while other women may modify their social behavior [4, 5] to defend their unborn baby from danger [5].

In Chinese culture, there are traditional pregnancy confinements to protect the fetus from "malign influences" and to prevent pregnancyrelated issues, for example, miscarriage of the

M. Ali, MSc

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India

e-mail: sanjuktaniranjan@gmail.com

fetus, stillbirth, death of the mother, and defects in the baby [5]. The fundamental ideas of these limitations originate from the notion of yin and yang in Traditional Chinese Medicine (TCM), since 2000 BC [6, 7]. TCM originates from the Taoist theory of an individual's harmony with the universe. As per this hypothesis, the human body is similar to the universe, and can be divided into a positive force (yang) and a negative force (yin) which are integral to one another. Health is considered to be a balance between the yin and the yang [8]. TCM has particular key ideas, including the qi (vital energy), which is accepted to circle along 14 channels or meridians, 12 of which impact are affected by the major internal organs. The qi is kept in balance by the double polarities of yin and yang [8], and the disturbance of the qi in a pregnant woman is thought to bring miscarriage or fetal malformation [9].

Chinese Medicines for Pregnancy and Their Impact on the Developing Fetus

Chinese medicine incorporates a few separate medications which are connected in an unexpected way; however they all take into account the same assumptions and insights of the nature of the human body. Fundamental therapeutic strategies [10] incorporate acupuncture, Chinese

First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_39

Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

[©] Springer International Publishing Switzerland 2016 N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development:*

508

medicines (applying traditional medications, for the most part derived from herbs, animals, and minerals, to cure ailment and maintain good health), nourishment treatment (diet to meet specific nourishment requirements and herbs to adjust internal bodily needs), Qi Gong (advancing health through breathing and meditation exercise), Tai Chi (the development of muscles and exercise of related joints), Tui Na (back rubs to stimulate the meridians and enhance the blood flow), Cupping (relieving blood stasis and pain by creating vacuums on the body surface), Die Da (bone setting) and Gua Sha (scaling the skin to stimulate specific acu-points until mild to moderate subcutaneous haemorrhage occurs) [11, 12].

For centuries, Chinese medicines have been generally used to mitigate numerous side effects and to treat complications during pregnancy [13]. Some Chinese medicines are prescribed for utilization during pregnancy and particular herbs are guaranteed to be safe and can be very viable. It has been utilized to prevent miscarriage and preterm labor, and to manage common colds, low back pain, low fetal weight, placenta previa, uterine fibroid and other obstetric issues [14]. More than 9 % of pregnant women devour herbal medicines or supplements around the world. The commonness of utilization in pregnant women is high in the Pacific nations; for instance, 78.7 % in Japan [15], 36 % in Australia [16], 32 % in Mainland China [17] and 24 % in Taiwan [18]. Chinese medicines are frequently utilized in threatened miscarriage to prevent pregnancy loss [19–22].

Lu Li reported on threatened miscarriages, that if there was an unevenness in the mother's system (however, this was not applicable in case of genetic issues with the fetus), it could frequently be overcome with the use of herbs and treatment with moxa or acupuncture [23]. The method to be utilized and the strategies to follow should be discussed early in the pregnancy so that appropriate steps can be taken should bleeding, fetal agitation, or early contractions occur. It is critical to note that most instances of early miscarriage (sometimes called spontaneous abortion) are not identified with an irregularity in the mother's systems. Later in the pregnancy, shortcomings in the mother's system or fetal development not considered to be normal. There is a specific herb formula, called Tang-kuei and Peony Formula (Dang Gui Shao Yao San), which aims at avoiding miscarriage; however, the formula is expected to be utilized primarily as an everyday preventive treatment instead of an emergency treatment. Extensive testing in the oriental countries has shown this formula to be safe and successful; modified versions of this formula, such as Tang-kuei Formula (Dang Gui San), are utilized to address particular concerns and are likely to be equally safe and efficacious. Other formulas, for example, "A Tai Yin" (which implies Peaceful Fetus Formula) and "Shou Tai Wan" (which implies Fetus Longevity Pill) are prominent cures in China for utilization during the latter part of pregnancy.

Jiang Hong assessed the impact of Jinye Baidu Granule (JYBDG), a traditional Chinese medicine compound prescription, on fetal development and maternal active human cytomegalovirus infection [24]. As a result of active human cytomegalovirus infection (HCMV), mother-fetus vertical transmission may cause abortion, malformation, fetal intrauterine death, development retardation, and in addition long haul sequelae, such as, damage of hearing or vision and intelligent disturbance of neonates. It has been noted that active HCMV infection rate in pregnant women is around 11.2 % [25], which causes obstruction to the growth and development of the fetus and neonates; so far, few effective treatments have been found. Therefore, A study was conducted in 3,187 pregnant women with active HCMV infection using Jinye Baidu Granule, JYBDG, a traditional Chinese medicine compound. mRNA was screened by enzyme linked immunoabsorbent examine (ELISA) and Reverse Transcription Polymerase Chain Reaction (RT-PCR). The growth and development of the fetus were also investigated to examine whether the Chinese medicine had a remedial role.

Traditional Chinese Pregnancy Restrictions and Health-Related Quality of Life

Health related quality of life is an undeniably vital issue. It is an expansive, multifactoral development that surveys the level of prosperity felt by people and can fluctuate as a result of diverse social impacts [26]. Health related quality of life is one part of this development [26], and is related to the impact of an individual's health status on his or her subjective physical, mental, and social wellbeing. It incorporates both physical and mental areas, which has concentrated on particular issues, for example, an individual's encounters, convictions, desires and observations [27]. During pregnancy, hormonal and organ-particular changes adjust affect as well as mental wellbeing. As a result, the impression of quality of life may change [28]. Hypothetically, pregnancy confinements may be seen as having both positive and negative effects on health related quality of life. These limitations may be connected to antenatal health practices that persuade women to have a positive disposition toward pregnancy, to get ready for child rearing, and to recognize this life move [29]. One study found that a more inspired state of mind was connected with a better quality of life [30]. Then again, pregnancy may limit women's physical movement and eating regimen; accordingly, women lose some of their flexibility [4]. One study found that health related quality of life was lower in pregnant than in non-pregnant women [31]. Thus, pregnancy can compromise physical and mental function [4]. One study found that taking traditional Chinese medicines during the perinatal period may have a constructive outcome on women's health related quality of life; however. The persistent use of these medicines may have a contrary impact [32].

Pregnancy is a vital stage in life for some women. The physiological changes caused by fetal development (including postnatal adjustment) and birth of a baby influences maternal health. A person's constitution, as per the hypotheses of Traditional Chinese Medicine (TCM), is characterized by intrinsic hereditary and obtained elements, which incorporate appearance, physical function and mental status [36]. There are complex morphological, auxiliary, physical and psychological dimensions. These qualities can affect an individual's susceptibility to disease and pathological tendencies [4]. Chinese medicine has four diagnostic strategies (counting examination, auscultation/olfaction, history taking and palpation) to diagnose, for instance, the 'Qi vacuity pattern' [37]. Humans are complex organisms who display continuous changes in their constitutions following changes in age, diet and living environment [38]. TCM claim that giving birth causes an insufficiency in qi and blood causing women to become physically weak. Consequently, postnatal women are inclined to blood stasis and illness when exposed to either cold or dampness. Furthermore, insufficiency in qi/blood renders postnatal women susceptible to common cold and other illnesses [39]. Therefore, constitutional changes during the perinatal period are especially imperative. During pregnancy, minor constitutional changes can lead to the replenishment of qi and blood, which guarantee proper fetal development [38]. An ancient Chinese colloquialism states that women are inclined to fear cold during early pregnancy (which is sometimes uncomfortably manifestated), but have adverse responses to heat in the later pregnant stages. This suggests that the constitution experiences considerable changes during the various phases of pregnancy. Wang [40] reported significant differences in constitution before and after pregnancy, observing that the average scores from a constitutional scale increased during pregnancy.

Traditional Chinese Medication is presently acknowledged as a standard form of medical care all over East Asia and is viewed as a corresponding or optional solution in the Western world. Chinese herbal medicines have been utilized to prevent abortion and preterm labor and to deal with common cold, low back pain, placenta previa, fetal development confinement and other obstetric issues [14]. Some Chinese herbal medicines are commonly utilized during pregnancy and they are asserted to be "safe" and "successful" [33]. Unlike those pharmaceutical medications not suggested for utilization during pregnancy due to known adverse or teratogenic impacts in animal studies and/or clinical studies, Chinese herbal medicines are endorsed based on individual involvement in clinical practices; but, there are no sufficient statistics or scientific information with respect to the adverse maternal and perinatal effects of Chinese medicine. Therefore, the actual impact of TCM on pregnancy, embryo fetal development, prenatal and post-natal growth is yet unknown. The procedure for testing herbal medicines in humans has not yet been established. Before clinical studies can be attempted in humans, the initial phase is to develop guidelines on the assessment of quality, efficacy and safety of herbal medicines through testing in animals. There are traditional convictions that certain sorts of Traditional Chinese Medicines (TCMs) are valuable in pregnancy [34]. However, the adequacy of most TCMs is largely unproven and its safety has yet to be established.

Maternal and Fetal Safety Concerns of Chinese Medicines

Chinese pharmaceuticals are normally respected by people as safe [35]. Albeit there is no scientific basis for the safety claim, around 30-50 % of pregnant women use it to maintain good health and decrease the requirement for Western medicine [7]. Chinese medicines have been historically used in the cultural context, and there is no sufficient record with respect to any adverse impacts in utilizing these herbs. There is no data on how safe the medicines are during pregnancy and if there are any antagonistic impact of the medicines on embryo-fetal development. More than 2,000 Chinese medicines and prepared slices of Chinese crude drugs, vegetable oil, fats and extracts, patented Chinese traditional medicines, and single ingredients of Chinese crude drug and preparations are recorded in Chinese Pharmacopoeia [41]. At present, there are more than 300 Chinese medications which are regularly utilized during pregnancy as a part of daily clinical practice; approximately 200 are applied for promoting maternal and fetal wellbeing.

Conclusion

Chinese medicines should conform to the same modern pharmacological standards as western medicine. The active elements of the Chinese medicines are likewise chemicals that are similar to prescription pharmaceuticals. Chinese medicines are not free of danger; they are similar to western pharmaceutical medicines and have potential to cause antagonistic impacts. Chinese medicines may be beneficial, yet might likewise antagonistically influence both mothers and fetuses in utero, and may harm the fetus specifically. Until more dependable and scientific information is available, clinicians ought to consider both the dangers and advantages before prescribing Chinese solutions to pregnant ladies. More deliberate examination of the safety implications of the use of Chinese medicines in animals is suggested, and more studies with clinical trials in humans are also needed to ensure the clinical safety and usefulness of TCM.

References

- Ayaz S, Yaman Efe S. Potentially harmful traditional practices during pregnancy and postpartum. Eur J Contracept Reprod Healthcare. 2008;13(3):282–8.
- Kaaya S, Mbwambo J, Fawzi MS, Van Den Borne H, Schaalma H, Leshabari M. Understanding women's experiences of distress during pregnancy in Dar es Salaam, Tanzania. Tanzan J Health Res. 2010;12(1): 36–46.
- Liamputtong P, Yimyam S, Parisunyakul S, Baosoung C, Sansiriphun N. Traditional beliefs about pregnancy and child birth among women from Chiang Mai, Northern Thailand. Midwifery. 2005;21(2): 139–53.
- Lee DT, Ngai IS, Ng MM, Lok IH, Yip AS, Chung TK. Antenatal taboos among Chinese women in Hong Kong. Midwifery. 2009;25(2):104–13.
- Helaine S. Childbirth across cultures: ideas and practices of pregnancy, childbirth and the postpartum (science across cultures: the history of non-western science) Dordrecht: Springer; 2009.
- Kong Y, Liang S. The cultural fabric of Chinese medicine: how to know your body through Chinese medicine. Hong Kong: The Commercial Press; 2005.
- West Z. Acupuncture in pregnancy and childbirth. John F. Kennedy Boulevard, Philadelphia: Elsevier Health Sciences; 2008.
- Kong YC. Huangdi Neijing. A Synopsis with Commentaries (English and Mandarin Chinese Edition). Hong Kong, China: The Chinese University Press; 2010.
- Yeh H-Y, Chen Y-C, Chen F-P, Chou L-F, Chen T-J, Hwang S-J. Use of traditional Chinese medicine among pregnant women in Taiwan. Int J Gynecol Obstet. 2009;107(2):147–50.

M. Ali and N. Bhattacharya

- Wang HH. The guidelines of Chinese medicine therapy. Hong Kong: Reader's Digest Yuan Dong Company; 2003.
- Luo SMWWQ. Image of Chinese medicine: the study on Chinese medicine history. Beijing: People's Health Publishing House; 2007.
- Zhong Guo Yun Dong Yi Xue Za Zhi. Comparison of effects of eliminating muscular acute fatigue by acupoint electric stimulation at limbs and torso. 2003:29(1):100–2.
- Flaws B. Chinese medical obstetrics. Colorado, USA: Blue Poppy Enterprises; 2005.
- Fu qing. mTranslated by Yang-Shou-zhong & Liu Da-Wei. Fu Qing-Zhu's Gynecology. USA: Blue Poppy Press; 1996.
- Mantani N, Kasahara Y, Kamata T, Sekiya N, Shimada Y, Usuda K, et al. Effect of Seihai-to, a Kampo medicine, in relapsing aspiration pneumonia an open-label pilot study. Phytomedicine. 2002;9(3):195–201.
- Forster DA, Denning A, Wills G, Bolger M, McCarthy E. Herbal medicine use during pregnancy in a group of Australian women. BMC Pregnancy Childbirth. 2006;6(1):21.
- Wang X, Shi B, Lin G, Wang Q. A survey on applications of drugs among 416 pregnant women in ShangHai. Chin J Pharmacoepidemiol. 1995;4:167–9.
- Chuang CH, Hsieh WS, Guo YL, Tsai YJ, Chang PJ, Lin SJ, et al. Chinese herbal medicines used in pregnancy: a population-based survey in Taiwan. Pharmacoepidemiol Drug Saf. 2007;16(4):464–8.
- Liu R. Clinical observation of integrated traditional and western medicine. Chin J Integr Trad West Med. 2002;22:68–9.
- Li L, Dou L, Leung PC, Wang CC. Chinese herbal medicines for threatened miscarriage. The Cochrane Library. New York ,USA: John Wiley and Sons; 2012.
- Li L, Dou LX, Neilson JP, Leung PC, Wang CC. Adverse outcomes of Chinese medicines used for threatened miscarriage: a systematic review and metaanalysis. Hum Reprod Update. 2012;18(5):504–24.
- 22. Tang L, Li L, Borchert A, Lau CB, Leung P, Wang C. Molecular studies of the congenital malformation induced by Largehead Atractylodes Rhizome, the most commonly used Chinese medicine for threatened miscarriage. Mol Hum Reprod. 2012;18(12):585–92.
- Li L, Leung PC, Chung TKH, Wang CC. Systematic review of Chinese medicine for miscarriage during early pregnancy. Evid-Based Complement Alternat Med. 2014;2014:753856.
- 24. Jiang H, Chen S-h, Wen L-z. Effects of Jinye Baidu Granule (金叶败毒颗粒) on fetal growth and development with maternal active human cytomegalovirus infection. Chin J Integr Med. 2006;12:250–4.
- 25. Wen L, Wu S, Lu S. The epidemiological study on human cytomegalovirus infection of pregnant women and the maternal-fetal transmission in three Chinese metropolis. Zhonghua Fu Chan Ke Za Zhi. 1996; 31(12):714–7.
- Salter KL, Moses MB, Foley NC, Teasell RW. Healthrelated quality of life after stroke: what are we measuring? Int J Rehabil Res. 2008;31(2):111–7.

- Lena NW. Health-related quality of life and physical ability among pregnant women with and without back pain in late pregnancy. Acta Obstet Gynecol Scand. 2004;83(4):351–7.
- 28. Förger F, Oestensen M, Schumacher A, Villiger PM. Impact of pregnancy on health related quality of life evaluated prospectively in pregnant women with rheumatic diseases by the SF-36 health survey. Ann Rheum Dis. 2005;64(10):1494–9.
- Haslam C, Lawrence W, Haefeli K. Intention to breastfeed and other important health-related behaviour and beliefs during pregnancy. Fam Pract. 2003;20(5):528–30.
- Mancuso CA, Sayles W, Allegrante JP. Knowledge, attitude, and self-efficacy in asthma self-management and quality of life. J Asthma. 2010;47(8): 883–8.
- Mota N, Cox BJ, Enns MW, Calhoun L, Sareen J. The relationship between mental disorders, quality of life, and pregnancy: findings from a nationally representative sample. J Affect Disord. 2008;109(3): 300–4.
- 32. Chang P-J, Tseng Y-C, Chuang C-H, Chen YC, Hsieh W-S, Hurng B-S, et al. Use of Sheng-Hua-Tang and health-related quality of life in postpartum women: a population-based cohort study in Taiwan. Int J Nurs Stud. 2010;47(1):13–9.
- Lei ZQ CS, Gao XM. The Chinese medicines. Shanghai: Science and Technology Publishing House; 1995.
- Wilkowski R. Chinese medicine for pregnancy and childbirth. Midwifery Today Int Midwife. 2000; 58:39–41.
- Marcus DM, Snodgrass WR. Do no harm: avoidance of herbal medicines during pregnancy. Obstet Gynecol. 2005;105(5, Part 1):1119–22.
- Tang W, Wang J, Hu JP, Wang YH, Han XX. An investigation of "determination and treatment" for clinical practice. Chin Arch Tradit Chin Med. 2005;23:2181–3.
- 37. Su YC. The study of the classic literature in Chinese medicine for constitutional enhancement. In: The committee on Chinese Medicine Pharmacy, Department of Health, Executive Yuan; 2006.
- Zeng SH. A little experience in syndrome differentiation of Jin Kui Yao Lue. Guid J Tradit Chin Med Pharmacol. 2005;11:10–5.
- 39. Pi KW. Chinese Medicine diet in doing the month. Basic training program in traditional Chinese medicine nursing. In: Symposium at the meeting of the Taiwan Traditional Chinese Medicine Nurses Association; Taiwan, 2007.
- Wang HL. Relationships between constitution and uncomfortable symptoms of first trimester for pregnant women. ([Unpublished Master's Thesis]) Kaohsiung Medical University, Taiwan, ROC; 2008.
- Chinese Pharmacopoeia Commission. Pharmacopoeia of the Peoples Republic of China 2010 (English edition), vol. 1. Beijing: Medical Science Press; 2011. p. 2001.

Perspectives of Tribal Medicine on the First 6 Months of Pregnancy

40

Priyodarshi Sengupta, Dibyendu Bandhopadhyay, and Niranjan Bhattacharya

Introduction

Tribal medicine is essentially a very age old medicine without recognition among the modern scientific community due to its lack of presence in the scientific fraternity which can be attributed to the fact that most tribal medicines lack advertisement and are secluded as they are only used by a fraction of the people across the globe in various tribal areas of Africa. India. North America. Australia, New Zealand, Pakistan and Central Asia. Traditional remedies form an essential and important part of the cultural and religious life of tribals. Tribal medicines are mainly herbs in nature just like in the case of Ayurveda, homeopathy, traditional Chinese medicine and Unani system of medicine. Nearly half the population globally relies on herbal medicine as do many tribals [1]. A wide range of herbs are normally included in many forms of medicine during various stages of pregnancy and to regulate the menstrual cycle [2].

e-mail: sanjuktaniranjan@gmail.com

Tribal Medicine and Methodologies in 1st and 2nd Trimester of Pregnancy

Tribal communities are often forest dwellers and their medicinal practices have often been termed as 'traditional health care system' over the years. Most of the healing systems of traditional medicine are based on faith healing which is often equated to as confidence building according to the modern system of medicine. Magical spells play an important role in tribal medicine [3].

Most tribal people in India believe that at the time of pregnancy consumption of alcohol can cause deleterious effects in the body. It has been claimed that a certain group of tribals in the Indian state of Chhattisgarh can guess the sex of the baby by just observing the features of a pregnant mother which includes measuring the circumference of the abdomen of the mother mainly during the late 2nd trimester of development. According to them if the circumference of the stomach is large then the fetus is a girl and if she looks thin then it s a boy. Also during the second trimester if the pregnant woman feels the movement of the child inside her abdomen then it can be said that the fetus is a female child.

A certain sect of a tribal community in Assam known as the Karbis, use herbal products during the various stages of pregnancy and to treat various gynecological conditions. To carry out abortion during the 1st or 2nd trimester Ceylon Leadwort, a herbaceous plant with glabrous and erect stem which according to the folklore

P. Sengupta, MSc, MPhil • D. Bandhopadhyay, MSc Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_40

resembles the size of the penis is tied around the thigh region with a long thread and then inserted inside the vagina to initiate the process of abortion; normally it is kept tied for a day or two. Also it has been reported from tribal groups in the Indian state of Gujarat that root paste of the plant is used in the vagina for increasing the chance of abortion [4]. It is also reported that ground root mixed with a little amount of sugar if taken orally can be effective in inducing permanent sterility [5].

Concept of Pregnancy among Different Tribes

The Cholanaickan tribes in Kerala believe that the initiation of pregnancy is induced by the sexual union of female and the male where the male is regarded as the progenitor. According to the folklore of these tribes it is suspected that pregnancy which is also referred to as oddalayarathu happens as the young married woman misses her periods. Further according to these tribes, pregnancy lasts for ten lunar months and the sexual union up to the seventh month of the last trimester. During the 1st and 2nd trimester food acts as the principal medicine to these tribes where roots and tubers are regarded as roughage foods and need to be avoided. Other holistic approaches during the 1st and 2nd trimester are followed in order to keep the fetus and the mother healthy; these are massage of the abdomen with coconut oil, magic spells and spiritualism. This also continues to the last trimester and before the delivery of the baby the intensity of such rituals increases. The pregnant women during the early stages of pregnancy can be allowed to accompany men in the forest for gathering and hunting foods. Husbands, on the other hand, after the end of second trimester, are not allowed to come near to the woman [6].

Like the Ayurvedic, Unani, Chinese and Homeopathy system of medicine diet also plays a very integral and major role in maintaining a safe pregnancy period.

According to North American tribes, including the Apaches, pregnancy is a very laborious and time consuming process where the blood (semen) from a man enters a woman's system during coitus.

This male blood (semen) is opposed by the woman time and again so that repeated love is made in order to ensure the sufficient seeding of the baby and to overcome the resistance from the mother. Other tribes like the Hopi tribes believed that continuous coitus was necessary for pregnancy. Other tribes, on the other hand, abstained from sex thinking that sex during pregnancy might give birth to a baby filled with filth [7].

Among a certain tribe in South Africa, herbs are used during pregnancy orally on a regular and normal basis as it is thought that doing such will help in cleaning the womb of the mother as the herb functions as a cleansing tonic [8, 9]. It is also thought that practicing such methods it will facilitate easy delivery and protect the new born child from evil [10].

Herbs play an important role in tribal medicine. For example, plumbaginaceae herb, used by the tribal people of Boudh district in Orissa, functions as an abortifaciant by inducing abortion in the tribal mother within 3 months of pregnancy [11], whereas in the case of the various tribes of West Bengal, the abortifacient herbs used are amaranthaceae, bromeliaceae. These herbs help in inducing abortion during the first 3–4 months of gestation also [12]. Also herbs like Acantheceae can induce early abortion in the mother during the first trimester itself. Ebenaceae, another herb, is used in the second trimester to cure anaemia in tribal mothers [11].

Although tribal medicine lays a special emphasis on food as an important medicine because it provides nutritional status to the fetus in the early pregnancy months, it has been found that 90 % female Kannikar tribes in India showed anaemia, 30 % suffering from Vitamin A deficiency and 10 % from niacin deficiency [13, 14]. This trend was also common in case of female tribes from the Bastar district in India. In general tribal medicine can be defined as that branch of medicine which is still hugely unexplored and is a diversified form of medicine based on the ancient traditions of that country, like in India where most tribal medicines are based on the herbs and principles of Ayurved.

Moreover the course of treatment and standard of life maintained by different tribes in India are also based on the principles of Ayurved and the Unani form of medicine where spiritualism, holistic measures and approaches and diet as the main course of medicine during the first 6 months of pregnancy, play an important role. Serious shortcomings of scientific explanations behind the many forms of tribal medicine and the lack of understanding of the different phases of pregnancy from the scientific point among the tribal people really makes it very difficult for scientists, public health workers and researchers to extract data regarding the types of medicines used in the first and second trimesters of fetal development. Also diversification of the tribes and their knowledge is another important criteria in making it impossible to come to a single basic approach of tribal medicine during the first 6 months of pregnancy. For example, a Maori tribal or an Apache from North America and the Kannikar tribes from Maharashtra might share a common philosophy in pregnancy based tribal medicine but their approach in applying this thought might be different and can be extremely diverse in the form of medicines they use; for instance, the Kannikar tribes would prefer the use of herbs more than the Maori or the Apaches.

Conclusion

With the emergence of more efficient healthcare access and providers, tribal medicine is slowly becoming a thing of the past with more number of tribal communities embracing and accepting the modern standard care of health and medicine. Also governments across the globe through different awareness and initiatives are starting to provide easy and free healthcare access to the different tribal communities. In India it is mandatory for government health care practitioners to spend some time in rural and tribal areas. One can see such initiatives taken in the Andaman and Nicobar islands for Jarwas, an indigenous ethnic tribal community residing in the Nicobar islands for more than 2000 years thereby resulting in a slow extinction of many of the indigenous health care practices that were followed by them over the years.

References

- Sandhya B, Thomas S, Isabel W, Shenbagarathai R. Ethnomedicinal plants used by the valaiyan community of pairanmalai hills (reserved forest), Tamilnadu, India- a pilot study. Afr J Tradit Complement Alternat Med. 2006;3(1):101–14.
- Shukla R, Chakravarty M, Gautam MP. Indigenous medicine used for treatment of gynecological disorders by tribal of Chhattisgarh, India. J Med Plants Res. 2008;2(12):356–60.
- Balgir, RS (2011) Genetic Disease Burden, Nutrition and Determinants of Tribal Health Care in Chhattisgarh State of Central-East India: A Status Paper. [Journal (On-line/Unpaginated)] Online Journal of Health and Allied Sciences.
- Patek PK, Patel MK. Ethnogynaecological uses of plants from Gujarat. India Bangladesh J Plant Taxon. 2012;19(1):93–4.
- Terangpi R, Basumatary TK, Teron R: Ethnomedicinal plants of the Karbi ethnic group in Assam state (India) for management of gynaecological disorders. International Journal of Pharm Life Science, 2014; 5(10):3910–3916.
- Viswanathan N. Tribal health and medicine in Kerala. DC Books, Kottayam 686001, Kerala State, India. http://ebooks.dcbooks.com/assets/preview/tribalhealth-and-medicine-in-kerala.pdf 2008;30.
- John C. Avise, Evolutionary Perspectives on Pregnancy, 2013, Columbia University Press, 2013. Printed in the USA.
- van der Kooi R, Theobald S. Traditional medicine in late pregnancy and labour: perceptions of Kgaba remedies amongst the Tswana in South Africa. Afr J Tradit Complement Alternat Med. 2006;3(1): 11–22.
- Varga CA, Veale DJH. Isihlambezo: utilization patterns and potential health effects of pregnancy related traditional herbal medicine. Soc Sci Med. 1997; 44(7):911–24.
- Gonçalves R. Assessment of maternal and fetal outcomes following Ingestion of Herbal Remedies. AMFIHR Study Protocol 2001. Pretoria. 2001.
- Sahu CR, Nayak RK, Dhal NK. Herbal remedies for various diseases used by tribals of Boudh District, Odisha, India for sustainable development. Int J Herb Med. 2013;1(3):12.
- Mitra S, Mukherjee SK. Some abortifacient plants used by the tribal people of West Bengal. Nat Prod Radiance. 2009;8(2):167–71.
- Prema L, Thomas F. Nutrition and health problems faced by Kanikkar women. In: Tiwari PD, Tripathi RS, editors. Dimensions of scheduled tribes development in India. New Delhi: Uppa Publishing House; 1992.
- Gopaldas T. Nutritional status of selected tribes of western and central India. Proc Nutr Soc India. 1987;33:76–89.

41

General Perspectives and Applications of Unani **Medicine in Pregnancy Up to** the 2nd Trimester

Priyodarshi Sengupta, Anamika Ishani Upadhyay, Abdus Salam Ansari, Akash Bhattacharya, Nandita Bose, Sushanta Banerjee, and Niranjan Bhattacharya

Introduction

Unani Medicine or Graeco Arabic Medicine or the well versant term of Unani Tibb is an ancient traditional system of Medicine that links itself to the Middle East and South Asian Countries especially Yunan (Greece), Persia, Syria and India. Though Martin Levey in 1952 produced a report that justified Babylon to be the earliest centre of medicine [1], the concept of health and healing is much more primitive in origin. The Tibbi system of medicine is a rich storehouse of principles and philosophies of medicine which if understood in

P. Sengupta, MPhil • A.S. Ansari, BUMS

A. Bhattacharya, MTech

Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

A.I. Upadhyay, BUMS, MPhil Calcutta School of Tropical Medicine, Kolkata, India

N. Bose, MD Director, School of Tropical Medicine, Kolkata, India

Formerly, Professor, Department of Pathology, IPGMER, SSKM Hospital, Kolkata, India

S. Banerjee, MD Director, Medical Education,, Govt. of West Bengal, Kolkata, India

Formerly, Professor and Head, Department of Pharmacology, RG Kar Medical College, Kolkata, India its proper perspective can prove of immense value. However, since all Tibbi literature is in the language and terminology of the contemporary sciences of the Middle ages (Persian, Arabic, Greek, Urdu etc.) it is not understood, or is rather inaccessible to modern scientists. Thus this system of medicine has been labelled as traditional and unscientific, though it has rendered service to millions of people for thousands of years and is the mother of the modern system of medicine. Hippocrates (460-337 BC), also known as the Father of Medicine was a Unani Tabib (physician). Other eminent Unani physicians include:

- Asclepius (also known as the God of Health in the Unani System) [2].
- Aristotle.
- Joannes Grammaticus.
- Herophilus (also known as the Father of Anatomy in the Unani System) [3].
- Galen.

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (🖂) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India

e-mail: sanjuktaniranjan@gmail.com

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_41

[©] Springer International Publishing Switzerland 2016

- Geber (also known as the Father of Chemistry in the Unani System) [4].
- Rhazes.
- Haly Abbas.
- Juhannitus.
- RabbanTabri.
- Thabit Ibn Qurrah.
- Yaqub Ibn Ishaque.
- Abu SahlMasihi.
- Avicenna.
- Abulcasis.
- JesuHaly.
- Hazen, and many more.

Methods of Unani System of Medicine

The concept of Unani system of medicine is primarily based on prevention of disease and the well being of the individual by means of improving their standard of living and quality of life by maintaining a healthy diet and healthy lifestyle especially during pregnancy as it believed that the well being of the mother is related to the well being of the baby inside her womb. However like other principles of medicine Unani also lays emphasis on curative or pharmacotherapeutic approaches. The treatment guidelines of the Unani system of medicine is based on two important principles, namely, the preventive and curative principles.

The preventive principles are further based on the different elements of nature like the atmosphere, air, water, temperature, the mental and emotional condition of the individual, dietary intake like food and drinks, amount of stress in life and how well is the sleep and wakefulness of the mother including the movement and repose of the body.

The curative principles are based on curative regimen therapies like cupping or Majamat where the application of a glass vessel exhausted by heat is done to the skin, Diaphoresis or Tareeq which helps in lowering the body temperature and cleaning of morbid materials through the skin pores. Other methods include leeching by which toxic blood is released from the body, therapeutic bath in a hot water bath is quite common in Turkey for cleansing the body and provides beneficial support to the skin and prevents humoral diseases. Blood letting or Fasd which involves a small cut at the affected area and allowing the blood to drain out; Diuresis, also termed as Adrar-e-Bowl to cure heart, cardiac, lung and liver ailments; massage or Dalak to tone up the muscles and the nerves along with cauterization performed by a hot metal, or corrosive medicines to remove pathogens have been successfully practiced in Unani over the years. Diet therapies and Pharmacotherapy including dosage timings, dosage route, forms and shape of the drugs also form an important constituent of the curative mechanisms under Unani mode of therapy.

Unani system is very similar to the ayurvedic system of medicine in terms of its principles. Both systems of medicine, Ayurved and Unani, consider natural elements like air, water, earth and fire as key elements and any imbalance in these key elements can result in the imbalance of the nature of the human body leading to a disease state. Therefore in order to treat such imbalances herbal remedies are often prescribed in order to rebalance these elements within the human body and fight disease conditions [5].

Like Ayurveda, in the Unani system of medicine it is postulated that if treatment is possible by diet therapy then there is no use of drugs in such cases as diet plays an important and basic role in maintaining a healthy body. Certain herbs which are used as essential ingredients of cooking in countries like India, Pakistan, Bangladesh, Sri Lanka including most western nations and the Middle East are used extensively in ayurvedic and Unani medicine too. Cloves for example are used as carminative to increase the hydrochloric acid in the stomach, thereby helping to increase the peristalsis effect. The essential oil extracted from clove is also used to alleviate pain and healing in Unani medicine. It is also used for analgesic, anesthetic, antidotal and antiperspirant purposes. Coriander, another essentiality in Asian cooking, is used for treating disorders of the digestive, respiratory and urinary system, diabetes and loss of appetite. Bay leaves have important properties in controlling flatulence and relieving the stomach of indigestion. It is also used for sprains, bruises and skin rashes. Powdered seeds can be given to young girls after menarche [6].

However if the shift of the body from normal health condition is consider to be great and can

carry the risk of affecting the body to a greater extent, then diet therapy alone is not sufficient. Pharmacotherapy (Ilaj bid Dawa), like the principles of modern medicine, is also important. These drugs can be either herbal type, mineral type or maybe even derived from animals. In Unani, single drugs or their combinations in raw forms are preferred over compound formulations. Most of the drugs applied in the raw form are natural as it is believed that naturally occurring drugs are a symbol of life and have minimal side effects.

However the safety of the drug composition is also given importance in Unani medicine where drugs which are toxic in its crude forms are processed and purified. In Unani medicine, compound formulations are also used for treating various complex and chronic disorders since emphasis is laid on a particular temperament of the individual [7].

Principles of Unani System

The Unani system states that 'Tabiyat' (physisnature) is the supreme planner of our body. It is a natural power- a prime mover which when exists in a body, becomes cause direct or cause proximate for its active motion or rest. Thus 'Al Umur Al Tabaiyah' [8] deals with the nature (Tabiyat) of a thing (Human Body) which is comprised of seven principles-

- Al-Arkan or Al-Anasir (Elements)
- Al-Mizaj (Temperament)
- Al-Akhlat (Humours or Body fluids)
- Al-Aza (Organs or Members)
- Al-Arwah (Pneuma or Vital spirit)
- Al-Quwa (Faculties or Powers)
- Al-Afal (Functions)

In addition to the above seven UmurTabaiyah, there are AsbabSittahZaruriyah [9] (Six Essential causes) that too influence the human body with respect to preservation of health or causation of disease. They are

- Al Hawawa'lMuhit (Atmospheric Air)
- Al Makulwa'lMashrub (Food and Drinks)
- Al Harkatwa'lSukun al Badaniyan (Physical Body Movement and Repose)

- AlHarkatwa'lSukun al Nafsaniyan (Mental or Psychic Movement and Repose)
- Al Naumwa'lYaqzah (Sleep and Wakefulness)
- Al Istifraghwa'llhtibas (Evacuation and Retention)

The above six factors essentially influence each and every human being and likewise there are Asbabghayrzaruriyah (non- essential causes) which do not influence each and every human being, but may influence some, like habitat, profession, sex, social factors etc.

The term physiology also originated from the Greek root 'Physiologikos' meaning discourse or knowledge of nature or Tabiyat [10]. Thus Umurtabaiyah and Physiology are synonymous and both deal with the body and its natural functions.

The study of modern physiology begins with the 'Cell' that is regarded as the structural and functional unit of the body. Then the tissues (AlAza al-Mufradah) are studied. Thereafter comes organs (AlAzaal Murakabbah) and the systems. But in Tibb, contrary to the modern concept of cells, the lowest level of organization of the human body and other creations is the Unsur (element) or the atom (JuzLayataza). About 2500 years back, Greek Scholar Democritus and his pupils thought that all matter is made up of small indivisible units called atoma (atoms) [11]. They associate in different ways to constitute Akhlat (humour).

Now the concept is these Arkan (elements) come into Imtizaj (intermixture or chemical combinations) and produce a uniform state or state of equilibrium for each Anasiror Arkan (element) and thus this new state of matter is called Mizaj (temperament).

The humoral theory postulated by Hippocrates (460BC) in his famous book *Tabiyat al Insan* (Human nature) states that "The body contains four (major kinds of) humours, Dam (blood), Balgham (phlegm), Safra (yellow bile) and Sauda (black bile); a right proportion, according to quality and quantity, and mixing of which (homeostasis) constitutes health and unright proportion and irregular distribution, according to their quantity and quality constitutes disease" [12].

Ali Ibn al-Abbas [13] says Akhlat (humours) are the proximate principles for the human body. But the more proximate are the AzaBasitah (cells and tissues) which are composed of Akhlat, and Aza Aliyah (organs), which are composed of AzaBasitah. Ibn Sina also has the same view [14].

The concept of Quwa (faculties, power) is unique in Tibb. The Quwa is that property of the body through which life is manifested. It provides the basis of many bodily functions. These Quwa (faculties) are specific for a tissue or organ. For example, the Brain can only perform functions (Afal) pertaining to Quwat e Nafsaniyan (nervous system) and cannot perform the functions (Afal) of the heart, liver or kidney.

Thus the organs (Aza), faculties (Quwa) and functions (Afal) are interrelated and interdependent. For each function there is a Quwat (faculty or power) and where there is Quwat, there are Afaal. (Nafis) [15].

Concepts of Unani Medicine in Terms of Modern Medicine

The concept of modern allopathic medicine is based largely on the disease and suffering of the patient. It is based largely on curing and treating the individual whereas Unani and Ayurveda are based on the concept of prevention rather than cure through different holistic and natural methods. However with increasing evidence that many diseases actually arise due to the ingestion of materials that cause abnormal reactions to the body, more and more emphasis is focusing on diet control and therapies, including at the time of pregnancy. Modern therapeutic approaches come at a cost with adverse drug reactions; so to minimize such reactions often traditional complementary and alternative medicines, also known as (TCAM), play an important role. Various Unani formulations include rationality of using various medicinal plants, animal products, methods of preservation and administration, indications and contra indications, avoidance and abstinence of certain diets (Parhez), complete drug profile, drug effect and kinetics, adverse drug interactions, drug to drug and drug food interactions, thus making it acceptable and on lines with modern systems of medicine [16].

In Unani formulations of drugs, proper and logical reasoning in the preparations of drugs are followed thus making it a very scientific field of medicine. The above methods and considerations are based on lines of the principles of modern medicine where correctives (Muslehat) to drugs have been in use for a long time so to as negate the undesirable effects of different drugs to the body, which otherwise in the normal combined prescription of both single and combined drugs can produce unwarranted effects. In spite of the fact that crude toxic drugs are purified in many ways (Tadbir) these drugs may still have side effects (muzarrat) and the sole responsibility of prescribing these drugs depends on the well informed and experienced physician so as to avoid any unnecessary adverse events [16].

Another characteristic of Unani medicine is the use of wide varieties of herbal drugs which are different in different countries. These herbal drugs have a profound effect on the physiology of the body and mechanism of different herbal drugs varies from country to country and within its genera, thereby increasing the potential chances of toxic diversities. In order to safely monitor such use of herbal drugs, an international program has already been undertaken by the WHO Collaborating Centre for International Drug Monitoring of Herbal medicines [17].

Fetal Development in Unani Medicine

After copulation when the sperm enters the uterus, mani (germ cells) rests for a while there, thus causing pregnancy (Halat-e-Hamal) [18].

According to Hippocrates [19] Quwat-e-Muwallidah (generative faculty) extracts semen material from every organ of the body, thus it is capable of forming every organ from which it is derived. Conception occurs when both mani (sperm and ovum) combine together resulting in the formation of Nutfa (zygote). Since both matter (element) and quality (temperament) are present in mani, this mixing produces a new temperament (Mizaj). If this temperament lies within human limits, conception occurs and the zygote gets implanted in the uterus for further development. If the uterine environment and zygote interact and exceed the limit of human temperament in total, abortion occurs.

At first instance the Nutfa gets frothy due to Quwat-e-Musawirah (formative faculty) which is an inherent property of the uterus. The sperm already has Ruh-e-Nafsaniyah (psychic pneuma) and Ruh-e-Haiwaniyah (vital pneuma). These inherent properties of sperm get stimulated by the Quwat-e-Musawirah of the uterus and as a consequence each Ruh leads to genesis of their specific organ (Ruh-e-Nafsaniyah leads to genesis of cardiovascular system and Ruh-e-Tabaiyah leads to the formation of the gastrointestinal system) [20].

Ruh is a gaseous substance obtained from the inspired air (oxygen); it helps in all metabolic activities of the body. It forms the AkhlatLatifa (fine humours) to produce all kinds of Quwa (powers) and HararatGhariziyah (innate heat); it is the source of vitality for all organs [21].

According to Ibn Nafis [22] Quwat-e-Musawirah is that quwat which gives shape to each part of mani (sperm and ovum) which is required by that particular species. Giving the shape means it produces lines in the organs, forms cavities and depressions and performs other functions.

Al-Abbas [23] is also of the same opinion. Thus, Quwat-e-Musawirah controls [24]:

- Male and female sexual powers and functions of the heart copulation (potency)
- Fertilization of ovum
- Transplantation of ovum
- Cleavage and differentiation of zygote
- Formation of membranes and fetal parts
- Development of fetus
- Parturition

To elaborate, the frothy product formed at the time of conception reaches a place in the uterus where there is a favourable condition for the formation of the heart of the foetus. From the point where the heart generates, two prominences, one on its right and the other on its upper side appear. Thereafter they separate out forming a zone for the heart form the right prominence and the liver form the left prominence [25].

Subsequently the frothy product is filled with whitish blood (SafediMaelkhoon) and then an orifice is formed by a Pneuma (NafakhRehi) so the conceptus can get its nutrition, that is, blood (dam) and oxygen (ruh) form the uterus. Thus the umbilicus (naaf) is formed. So the generation of heart, liver and brain occurs prior to the umbilicus and the umbilical cord.

According to Hippocrates, the first organ formed in the developing embryo is the brain and two eyes. He justifies it by citing the example of an egg where these two are the first organs to be formed, whereas Aristotle claims that the heart is the first organ to be formed because it the centre of both life and Hararat-e-Ghariziyah (innate heat) [26].

At the time of conception the amniotic membrane (Gisha-e Janini), formed due to the movement of ovum and sperm, provides protection and nourishment to the conceptus till the requirement of nourishment is less and the membrane is still thin.

Ruh-e-Haiwani (vital pneuma) and Hararat-e-Ghariziyah from the heart goes to the other viscera and the Quwat-e-Musawirah (formative faculty) already present in them, get activated. As a consequence, the chief organs of the system get ready to function separately with the help of their supportive structures. For example, the brain starts functioning separately along with the peripheral nerves.

During the first week of conception, the mani gets frothy and Quwat-e-Musawirah starts performing its function in the zygote. After 3 days of this period (on ninth day of conception) with the help of the uterus, cleavage appears in the zygote. On the 15th day blood enters the zygote and a mass of flesh forms. From this fleshy mass, the heart, brain and liver are consequently formed which separate out and the spaces between them are filled with mucus. On the 34th day, the head gets differentiated from the shoulder and becomes prominent. At the same time, the ribs and the abdominal cavity are formed [27].

The temperaments (Mizaj) of maternal and paternal germ cells are the determining factors of the sex of the embryo. As in Tibb, sex is a temperamental character, which if hot, the embryo would definitely grow into a male. The reason for more expansion of the organs in males than females is this higher level of heat, thus giving a strong physique to the male compared to a female of the same age. After the embryonic temperament has been established, Quwat-e-Mughayyirahula (primary transformative power) works in the embryo and prepares its different parts to acquire the properties of different organs. It differentiates parts of the embryo into organ specific temperaments and consistencies, whereas Quwat-e-Musawirah (formative faculty) has the specific ability to prepare the specific dimension and shape of the organ. The failure of Quwat-e-Mughayyirahula results in absence of organs, whereas failure of Quwat-e-Musawirah results in structural anomalies though the organ exists. These two Quwat work with ample heat and its deficiency in early developmental life can result in death. One such condition is a mole (Hydatidiform mole) [19].

The functions of Quwat-e-Mughayyirahula stops when every part of the embryo is ready to form a specific organ, accepting the action of Quwat-e-Musawirah. In early stages of embryonal development, maternal Quwat-e-Haiwaniyah (vital faculties) give the embryo the ability to accept the functions of Quwat-e-Tabaiyah (natural faculties), but after sometime the embryo acquires it from the development of its own heart. Now the fetus is ready to accept the functions of Quwat-e-Tabaiyah and Quwat-e-Nafsaniyah (psychic or mental faculties).

The uterus along with providing protection and nourishment to the fetus, also provides the site of action to Quwat-e-Mughayyirahula and Quwat-e-Musawirah as neither of these function outside the uterine cavity or in post natal life. As congenital abnormalities are a result of malfunctioning of Quwat-e-Musawirah, Quwat-e-Mughayyirah and temperament, all these need to be modified in order to bring back the healthy state. Temperamental change though can be applied in that case, but as stated earlier, these two Quwats exist only in utero posing difficulty in permanent cure.

Quwat-e-Masika (retaining faculty) helps the fetus to remain in the uterus while its weakness strengthens Quwat-e-Dafiyah (expelling faculty), thus causing its expulsion.

Pharmacovigilance and Safety of Unani Drugs

During the first 2 months of pregnancy Unani medicine stresses on the well being of the mother and the fetus by stressing on diet therapy and very much less on pharmacotherapy. Apart from diet therapy other important considerations like the emotional state of the mother, sanitation and hygiene of the mother and her surroundings, standard and quality of life are all important factors which play a dominant role in the well being of the child and the mother during the first and second trimesters of pregnancy. Unani drugs are basically based on temperament, potency and efficiency. Higher the degree, higher is its adverse effect. Temperament or Mizaj is an important constituent of the Unani drugs and is based on 3° like cold, hot and dry.

During pregnancy if the mother does not respond well to the above concepts of Unani therapy, substitutes' or Abad al Adina can be used only. If a drug or therapy is substituted by some other means of treatment or medicine then that new medicine or therapy should be unique and equally or more effective than the previous one, thus rationalizing the same concept and principle used in case of modern principles of medicine. The main concept of Unani medicine is to assist in the natural recuperative power of the body and to eradicate any disease or abnormal condition from the body through the process of sweating, urination and defection. This is again in accordance to the modern ADME principles of allopathic medicine where the each of the letters stand for Absorption of the drug, Distribution of the drug, Metabolism of the drug followed by Excretion from the body. Therefore in Unani medicine before prescribing a drug to the pregnant mother the Unani practitioner needs to emphasize on the diet regime, as correct diet and digestion along with a good quality of life can help in a safe pregnancy period.

Pharmaco environmentology, another branch of Unani medicine, is important as it deals with the entry of chemicals, drugs and therapeutic molecules into the environment from the living systems post therapy [17, 28–30]. Reasons for adverse drug reactions in herbal medicines related to the Unani system of medicine can be associated with individualizing of treatment by the patients themselves, parenteral use and most importantly, adulteration [1]. The regulatory environment including the legal status of herbal drugs, their rationale and mechanistic approval for actions are also unknown as many of these approaches are based on holistic and traditional beliefs. However, generally, most of the herbal drugs used in Unani medicine are deemed to be safe as many of them are used in daily life ranging from cooking to the early stages of pregnancy till the late phases in life [17].

In Unani like in Ayurveda, therapeutics are based on a constitutional approach and the medicinal herbs are selected or excluded according to their compatibility or incompatibility to the constitutional makeup of a given individual. Unani like Ayurveda and some tribal medicines, is also dependent on the patient's constitutional health plan [31].

References

- 1. QadeerAshhar, Tareekh e- TibwaIkhlaqiyat (urdu), p. 25.
- QadeerAshhar, Tareekh e- TibwaIkhlaqiyat (urdu), p. 67.
- 3. QadeerAshhar, Tareekh e- TibwaIkhlaqiyat (urdu), p. 86.
- 4. QadeerAshhar, Tareekh e- TibwaIkhlaqiyat (urdu), p. 125.
- Find a Vitamin or Supplement. Unani Medicine http:// www.webmd.com/vitamins-supplements/ ingredientmono-1212).
- Spices used in Unani Medicine. Spice J. http:// thespicejournal.com/natural-medicines/ spices-used-in-unani-medicine/).
- Ilaj Bin Dawa. Pharmaco therapy, The Unani system of medicine by Dr Roohi Zaman BUMS, BAMS (Integrated), MD Lecturer, National Institute of Unani Medicine Bangalore, p. 30, 31 and 32.

- 8. Ishtiaq Hakim Sayed, Al-Umur Al-Tabaiyah, p. 2
- 9. Ishtiaq Hakim Sayed, Al-Umur Al-Tabaiyah, p. 3.
- 10. Ishtiaq Hakim Sayed, Al-Umur Al-Tabaiyah, p. 1
- 11. Ishtiaq Hakim Sayed, Al-Umur Al-Tabaiyah, p. 5.
- 12. Ishtiaq Hakim Sayed, Al-Umur Al-Tabaiyah, p. 76.
- Al Abbas Ali Ibn, Kamil Al- Sin'at Al-Tibbiyah, vol. 1, Cairo, 1294, p. 48.
- 14. Ishtiaq HakimSayed, Al-Umur Al-Tabaiyah, p. 143.
- Nafis Burhan Al-Din, Kulliyat Nafisi, Daftar Al-Masih, Delhi. 1935. p. 142–4.
- 16. Rahman HSZ. Historical perspective of traditional medicine with special reference to ADR's. National Symposium on Relevance of Pharmacovigilance for Indian system of Medicine. Department of AYUSH, Ministry of Health and Welfare, Government of India and Society of Pharmacovigilance, Iindia, 4th Nov 2006. p. 53–61.
- Syed Ziaur Rahman, Rahat Ali Khan and Abdul Latif. Importance of Pharmacovigilance in Unani system of medicine.
- Sina Ibn (Avicenna), Al Qanoon Fi'l -Tib (Tibb e-Islamika Encyclopedia). p. 1059.
- Haseeb AA, ZulkifleMohd, Zaidi IH. Unimed. 2005;1(1):17.
- Sina Ibn (Avicenna), Al Qanoon Fi'l -Tib (Tibb e-Islamika Encyclopedia), 3(part 2), p. 1059.
- 21. Ishtiaq Hakim Sayed, Al-Umur Al-Tabaiyah, p. 153.
- Qarshi Ala-al-Din (Ibn Nafis), Ifada-e-Kabir(Mufassal), Hyderabad, 1947. p. 157.
- Al Abbas Ali Ibn, Kamil Al-Sin'at Al-Tibbiyah, Vol. 1, Cairo, 1294, p.131, 132.
- 24. Ishtiaq Hakim Sayed, Al-Umur Al-Tabaiyah, p. 187.
- Sina Ibn (Avicenna), Al Qanoon Fi'l -Tib(Tibb e-IslamikaEncyclopedia), 3(part 2). p. 1060.
- 26. Tabri Abu Al-Hasan Ali Bin SahalRabban, Firdaus Al-Hikmat, p. 43.
- 27. Sina Ibn (Avicenna), Al Qanoon Fi'l -Tib (Tibb e-Islamika Encyclopedia). 3(part 2). 1061.
- Rahman SZ, Khan RA, Gupta V, Uddin M. Pharmacoenvironmentology: a component of pharmacovigilance. Environ Health. 2007;6:20.
- Rahman SZ. Impact of human medicines on environment-a new emerging problem. Popul ENVIS. 2006;3:3–4.
- Rahman SZ, Khan RA. Environmental Pharmacology-a new discipline. Indian J Pharmacol. 2006;34:1–2.
- Zollman C, Vickers A. ABC of complementary medicine – what is complementary medicine? Br Med J. 1999;319:693–6.

42

Applications of Traditional Chinese Medicine Concepts in the Safety and Development of the Fetus During the 1st and 2nd Trimesters of Pregnancy

Priyodarshi Sengupta, Akash Bhattacharya, and Niranjan Bhattacharya

Introduction

Chinese medicine is a relatively new concept or rather an alternative medicine to the west [1]. Chinese medicine is considered to be an assimilation of philosophical and scientific thoughts and ideas that influence us in our daily lives. Chinese medicine believes in Yin Yang which metaphors the idea to a dualist reality that prevails in our society today. Like Ayurveda, Chinese medicine also represents the different elements of nature like wood, fire, metal, water and earth. Qi another concept in traditional Chinese medicine (TCM) is regarded as the main force of life filled with energy [2]. It is compared to the hot steam that comes out from cooking rice and consists of the wind, cold, heat and damp. In modern civilization Chinese medicine is slowly becoming the

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India

e-mail: sanjuktaniranjan@gmail.com

main stream of medicine in many east and Far East nations and is often regarded as an alternative source of treatment [3]. Since 5000 years and beyond Chinese medicine has offered many therapeutic potentials in pregnancy related indications ranging from infertility issues to threatened miscarriages [4, 5]. Chinese herbal medicine, diet and acupressure/acupuncture are some of the widely evolved and accepted Chinese method of practicing medicine. Although Chinese medicine is considered safe and gentle by some health care practitioners, public health workers and and therapists, major concerns pertaining to its safety and therapeutic potency remains a big question till date.

Pharmacodynamic and Pharmacokinetic Principles of Chinese Medicine

The basic principles of most drugs are its physicochemical properties which govern the binding of the drug with the plasma protein. There are certain other important factors influencing the pharmacokinetics and pharmacodynamics of a drug like dosage and time and route of administration, age and any pre-existing conditions. Whether a drug should be taken as a single dose or a multiple dose and the issue of how long it should be taken

P. Sengupta, MSc, MPhil • A. Bhattacharya, MTech Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_42

to elicit a minimal therapeutic response, is a major factor in governing the properties of drugs.

Age is another important consideration; which drugs with what mode of action, in what type of population (whether geriatrics, paediatrics or pregnant mothers), plays a major role in deciding the drug activity and its efficiency. Any pre existing conditions like liver cirrhosis or hepatitis or renal disease like proteinuria where the proteins escape from the renal tubules, and even dermatological constraints like eczema or dermatitis, can seriously malign and impair the proper metabolic and excretory functions by which the drug can be safely metabolized and eliminated from the body.

In case of Chinese medicine, especially in case of herbal medicine, like any other principles of medicine, there is a debate whether a whole plant extract or a single plant extract is more effective and superior for treating a certain ailment. In western medicine a certain purified drug can have a certain beneficial effect on a certain disease but this comes at a cost of certain degree of side effects. Chinese herbalists believe that a whole plant contains a cocktail of different factors and compounds that not only facilitate to negate a certain disease but also help to reduce the other side effects by entering into those organs where there is a maximal risk of metabolite interference produced from the main compound. Also in a mixture or cocktail of compounds a potent chemical can act both as an inducer and inhibitor at the same time as observed in the case of the Chinese herb Dang Gui Radix Angelicae sinesis which can contract and relax the walls of the uterus at the same time and Ren Shen Radix Ginseng which can both stimulate and inhibit the CNS. Therefore the above ideas conclude that an isolated and purified content of a Chinese herbal drug may not be potent enough to elevate the same level of therapeutic potential as a whole herb would do, thereby highlighting the point that a mixture of several drugs has a much higher therapeutic activity than a single extract. As observed and stated by Borchers et al., this can be attributed due to the synergistic or antagonistic mode of the extract interaction as observed and stated by Borchers et al. [2, 6].

Also Chinese herbalists have laid down a fourfold structure to make it easier for formulating prescriptions of Chinese drugs. According to them the first type of classification is known as the Emperor herb which forms the main prescription, followed by the assisting herb known as the Minister herb. The third type is defined as the Assistant herb and its main role is to moderate the influence of the first two herbs and thereby counterbalance and negate their side effects. The fourth classification is termed as the Messenger herbs and their main function is to help in directing the drugs to a specific organ of the body [2].

Certain medications like Huang Lian, An-Tai-Yin have been found to be associated with increased risks of congenital malformations in the first trimester of fetal development such as congenital nervous system malformations including the eyes, musculoskeletal and connective tissues. However a direct causal relationship of the interactions of these drugs with the fetal malformations cannot be established yet as more scientific explanations and observations are found wanting. In the case of An-Tai-Yin it is advised to use these drugs in the third trimester in order to help in the smooth delivery process which is also similar for another drug, Huan Lian, which is used for improving skin conditions. It should be allowed from the second trimester to evade complications to the fetus [7]. Further observations from Chuang CH et al. [8] suggested that there was a non significant slight increase in the birth weight as well as an increased risk of low birth weight when Huan Lian was administered to 9,895 pregnant mothers for more than 8 weeks.

TCM and Fetal Development

Several pre eleventh century Chinese texts describe the conception and development of the foetus. Though some essential elements of these theories remain unchanged, the later works, fabricated in principle from the earlier ones reflect a progressive sophistication of concepts professed by the former, often maintaining several simultaneously. Conceptions which were initially philosophical, by the medieval period, shifted to a great extent in focus towards the medical aspects of the health of the mother and the foetus. Having been conceived in a patriarchal regime of thought, early discourses on the subject focus on the pain of pregnancy as an intense experience rather than joy. Therein lay a fundamental social dilemma – it contrasted strongly with the (at the time) stigma of obligatorily fathering a male child [9].

The following text provides perspectives on fetal development.

Fetal development is described in the Huainanzi as follows [9, 10]

In the first month the lard occurs In the second month a lump develops In the third month an embryo forms In the fourth month the flesh is produced In the fifth month the muscles form In the sixth month the bones develop In the seventh month the foetus forms In the eighth month the foetus starts to move In the ninth month its movements become more pronounced In the tenth month, the birth occurs In this way, the physical body is complete and the five orbs (Zang) are formed

According to Chinese medicine, the general view of pregnancy is philosophically related with respect to Qi and Yi Yang. Chinese medicine also has a way of interpreting the development of the fetus with respect to the natural elements which is denoted below.

The wood resembles the first to fifth week of fetal liver development and is related to Hun as the ethereal soul that carries messages from the forefathers.

Fire resembles the heart between the 8th and 16th weeks and is resembled in Chinese medicine as Shen, the spirit that animates us and with which we live for the rest of our lives.

The Earth resembles the fetal spleen between 16th and 24th weeks and is signified as Yi, the ability which is based on learning remembering and reflecting.

Metal represents the 24–32 weeks lungs and in Chinese medicine it is denoted by Po an earthly soul which gives humans their identities in the form of body.

Water resembles the fetal kidney during the 32th and 40th weeks of development and is classified as Zhi which gives us the will and desires to live in this world [11].

Conception is generally regarded as the first essential step towards a fruitful reproduction and according to Chinese medicine both pre natal and post natal essences combine together to form the embryo which in other words means that the maternal and paternal health at the time of conception has a profound effect on the future health and well being of the child. Hence this indicates that both the mother and the father need to have a sound health at the time of conception. After conception it is regarded that the mother produces kidney essence increasingly to provide greater nourishment to the fetus via the maternal blood circulation, as well as an increase of the functional efficiency of the different organs of the body. Qi in the meantime, depending on the regulation of the emotions, ascends for the maturation and development of the breasts in the mother and dark lines along the lower part of the abdomen. The side effects of Qi are however noted as morning sickness, mouth sores and changes in body temperature, hemorrhoids and an increase in appetite. Strong emotions can have inhibitory effects on the fetus. To help the smooth flowing process of Qi, yoga, exercises and acupressure techniques are encouraged and the mothers are constantly asked to undergo a normal stress free life with adequate and sufficient rest and maintain a healthy risk free diet [11].

Role of Diet, Exercise and Acupressure in the Initial Stages of Pregnancy

Like traditional medicines practiced in India, Ayurveda and Unani, Chinese medicine also believes that during the initial phases of pregnancy, especially in the first trimester, the body's Yin and Yang, stomach's Qi and circulation change. Which is quite similar to concepts currently prevailing in western medical science. Therefore in order to negate any inhibitory effect to the fetus, maintaining a balanced and a healthy diet is the utmost need of the day. Some traditional medicine books including the famous herbal book Ben Cao Gang Mu shiyi have stated that lemons are neutral in characters and primarily work on the stomach by keeping it cool and calm [12]. Lemons can be also used for treating varicose veins in pregnant mothers during lymph massage along with lavender essential oils. Morning sickness accompanied with nausea and constipation, common side effects of pregnancy, can be treated by acupressure methods and by avoiding rich foods with ample exercise and yoga involving the stretching and massaging of the lower abdominal walls on a regular basis. Also, nausea can be treated by simply taping a magnet or a wrist band to provide constant pressure. Heart burns in the second trimester a common symptom in pregnant mothers can be reduced by taking sufficient apples and cabbages, nettle lemons, balm tea and peppermint three or four times each day. With growing gestation period as the mass and volume of the uterus increases, a constant pressure on the backbone also increases resulting in a chronic back pain. This can be treated with lavender and lemon essential oils massage. In cases of dehydration and edema along with muscle cramps, higher intake of barley, squashes accompanied with garlic acid is encouraged. Other complications like anemia, hypertension, and hemorrhoids are treated with herbal diets like passion flower, special herbal teas as an infusion for stabilizing the fetus [11].

It is advised that internal use of oils in pregnant mothers should be avoided along with different nervous system muscle stimulants like ephedra, kola nut and gurana as well as highly concentrated herbal extracts.

Diet, discussed later, in the context, is an essential and compulsory content of Chinese medicine. Foods like carrots, dark leafy and root vegetables, sprouted seeds, Brazil nuts and flax seeds are encouraged to be consumed by the mother for proper intake of protein, calcium and vitamins.

Chinese Medicine in the First and Subsequent Trimesters of Pregnancy and Dilemmas Pertaining to Its Usage

Due to the lack of standardization and characterization of the different Chinese drugs and their implications in pregnancy, utmost care by the Chinese therapists should be taken in prescribing remedies especially during the first trimester period as this is often the most sensitive period. Organogenesis initiates from this stage and it is during this period that the nervous system starts its initial development. Its primitive function is noticed between the 15th and 25th day post conception followed by the development of the eyes and cardiac tissues between the 20th and 40th day. The lower limbs also start its shaping between the 24th and 36th day of gestation period. Therefore being a very sensitive period for the fetus it is highly recommended to abstain from prescribing any Chinese medicine in the first 3 months of pregnancy; this is similar to western medicine, as any adverse events stemming from Chinese medicine can result in having serious implications on the fetal health often causing teratogenesis or mutagenesis [2].

To assess the effects of teratogenesis in pregnant mothers a comparative study was conducted at a Hong Kong University Hospital where it was found that 2.7 % of women who undertook western medicine underwent termination of pregnancy because of anxiety compared to none in case of pregnant mothers who took Chinese medicines [12]. There have been reported cases of teratogenicity related to the use of Chinese medicine during pregnancy. Takei et al. (1997) reported that a mother who took Lei Gong Teng during the initial trimester's period gave birth to a baby with occipital meningoencephalocele and cerebellar agenesis. A second example of such teratogenicity as reported by Koren et al. (1990) resulted in a baby with excessive hair, a health condition also known as androgenisation [13]. Therefore like western medicine, Chinese medicine has also been known to cause serious side effects many of which might get unreported due to lack of resources or awareness among the Chinese medical health care practitioners thereby suggesting that Chinese medicinal herbs are not always safe for pregnancy and should be handled with caution before prescribing them to pregnant mothers [14].

Also, after 3 months of pregnancy that is when the second trimester starts and organogenesis is already in a matured stage, careful considerations should be made in selecting the dosage of the drugs starting off from a low sub lethal toxicity dose and increasing it accordingly without any observable side effects. In cases of patients with poor prognosis they can easily opt out of the treatment regimen without causing any further harm to the developing fetus [2].

Below are given the list of certain Chinese herbal medicines which are identified as having very strong Qi and blood movers and if not used appropriately can result in serious side effects during the time of pregnancy.

- 1. Semen persicae Tao ren
- 2. Flos carthami Hong hua
- 3. Flos carthami Hong hua
- 4. Rhizoma and Radix Rhei Da huang
- 5. Fructus aurantii Zi shi
- 6. Fructus aurantii Zi shi
- 7. Radix aconite Fu zhi
- 8. Rhizoma zingiberis Gan jiang
- 9. Cortex cinnamomi Rou gui [15].

Is Chinese Medicine a Viable Solution and an Alternative to Modern Medicine?

Different studies in both animal models and human clinical trials have failed to report serious side effects, which can be attributed to the fact either due to the lack of safety awareness among the research and clinical groups or the level of understanding regarding how these Chinese medicines act on the fetus. Here, knowledge is severely limited due to lack of definitive well controlled in vivo and clinical studies where the use of placebo and double blind trials can generally reduce the level of doubt regarding the effectiveness of a medicine bias for such findings. In a case study of WS model of Chinese herbal medicine for pregnancy as reviewed by a German research group it was found that the scientific integrity of the animal model studies were severely lacking consistency. The group as reported had a total of 1751 animals of which only 46 animals were under control trial thereby indicating that there can be large fluctuations in reporting adverse events as the group was 38 times bigger than the control models. Safety aspects of Chinese herbal medicine in pregnancy—Re-evaluation of experimental data of two animal studies and the clinical experience [1]

Also, another important issue can be the combinatorial treatment of Chinese medicinal drugs with allopathic medicines. Dissemination of safety datas from such reports may not be possible given the current stage of clinical research [16].

How Safe Are Chinese Medicinal Herbs?

Chinese medicine is not immune to adverse events after its application. However lack of animal datas, proper controlled clinical trials without scientific explanations, proper training, lack of awareness, poor follow up and personalizing Chinese medicine has resulted in under-reporting of many cases of adverse events properly [1].

According to the author of this manuscript such adverse reactions to Chinese herbal medicine could be due to either wrong identification of herbs or interactions with heavy metals thereby contaminating the drug; including another factor could be its combined use with western medicine.

Wrong diagnosis, like in the case of any medicine along with bad practice and self-medication are some of the major problems and the root cause behind the occurrence of adverse effects of Chinese drugs [2]. One such example of bad practice resulting in adverse events is reported in the use of Chinese herbs in Hong Kong causing aconite poisoning [17].

Combined treatment of Chinese medicine along with western medicine has been found to malign the many therapeutic effects of the Chinese drugs as opined by many schools of thoughts among the Chinese therapists. However, two randomized control trials involving 130 participants were analyzed for toxicity and adverse events among a group of 130 participants for combined Chinese and western medicine and no adverse reports were observed in the group. under combined medicine but On the other hand, around 2–8 % minor adverse drug effects were reported in case of Chinese medicine when applied alone [18, 19] thereby raising the question of whether Chinese drugs should be administered alone or in combination doses.

Other Alternative Treatment Strategies of Chinese Medicine during Pregnancy

Acupuncture has been used successfully in treating infertility. Pregnancy rates in Assisted Reproduction Therapy were analyzed in a prospective randomized study in 80 patients with acupuncture. Results showed a higher rate of fertility in women receiving acupuncture than the control [20]. It has been shown to demonstrate beneficial hormonal responses in threatened miscarriage, with reduced miscarriage rates [21]. It has been used to treat back pain in pregnancy. Studies show that the mechanisms of regression of symptoms of pain in acupuncture are related to the production of endorphins [22]. Acupuncture has been used for amelioration of nausea and dry retching in pregnancy. A RCT conducted in Adelaide showed a significant improvement (p < 0.01) in both parameters, although a placebo effect was found among some individuals [23]. Apart from these, acupuncture has also been used to treat tension type headaches in pregnancy, dyspepsia, emotional complaints and insomnia [23–27].

Other general benefits of acupuncture include improved uterine lining, increased blood flow to the uterus, hormone regulation and in alleviating stress.

Acupuncture practices are generally regarded as safe. Adverse events of acupuncture in pregnancy have been reviewed. Total AE incidence was 1.9 % and AEs evaluated as certainly, probably or possibly causally related to acupuncture was 1.3 %. All AEs were mild or moderate [27]. Most of the studies regarding the use of Chinese medicine in the 1st trimesters are based on weak observations and less follow up time period. Also it has been noted that most of the end point of the studies were related to the normal delivery of the birth without a scientific insight into fetal wellbeing during the early gestation period that might where there is a possibility that acupuncture treatment may result in give rise to silent mutations and diseases even after a healthy delivery. Therefore most of the trials designed are based on weak clinical outcomes and parameters.

On the other hand, because of fewer adverse events reported from the use of Chinese medicine in pregnant mothers, pregnant mothers may wish to use Chinese medicines during pregnancy. The traditional Chinese medicine approach to health and disease dictates that all aspects of a system must be considered when treating a problem. Only with a comprehensive analysis on the patients' physical condition, age, lifestyle, medical history, career and mental state can a correct diagnosis be made and then treated properly. Though some herbal preparations might be toxic, they are widely used because of the prevalent ideas of their safety. Some are known to be effective in various complications of pregnancy. Their effects can be potent, yet their use is not controlled. Their guided use can curb unwanted side effects. But there is no pressing need to address the efficacy of these treatments and they may be used in moderation as a supplementary to conventional treatment. Advances in the field of quality control are important for the standardisation of the practices of TCM and will allow for a better control of the bioactivity and interaction of the ingredients and in the identification of its active principles. Currently practice is not standard and opinions of practitioners vary widely. Its clinical appropriateness should be analysed by randomized controlled trials and specified for particular conditions.

Discussion

TCM and Ayurveda are the two most common types of TCAM that have been practised and prescribed not only in Asia but throughout the globe for a long time. TCM and its treatment is more common in the eastern parts of Asia. TCM is primarily focused on and is concerned about the symptoms or symptomatic manifestations whereas Ayurveda, Unani and some tribal medicines are dependent on the patient's constitutional health [28]. Also traditional Chinese medicine or TCM is focussed more on the evaluation and differentiation of syndrome [29]. In terms of pathology Chinese medicine believes more in physical examinations including the organ structure. TCM medicinal herbs are classified according to the therapeutic effects of the herb itself, namely, dispersive quality, Yin tonifying quality and so forth. Also the Chinese Suwen and Indian Caraka of Ayurvedic system both represent the steps taken in daily life to promote one's health. It also describes different causes of diseases such as environmental, dietary and emotional factors. The main focus of Suwen is based on acupuncture. In terms of diet the Chinese system believes that normal digested food produces normal qi and blood which promotes health and well being. Chinese medicine depends on tastes like sweet, sour, salty, pungent (acrid), bitter, and bland. However during the time of pregnancy all Complementary and Traditional Alternative Medicine (TCAM) advise the mother to have a stress free life, abstain from smoking and drinking, maintain proper hygiene, consume foods that would keep the body devoid of toxins and would help in the nourishment of the fetus. Therefore all of the TCAM's lay emphasis on the standard quality of life of the pregnant mother and her happiness.

However comparing other branches of TCAM's like Unani, Ayurveda, Homeopathy and tribal medicine it can be noted that Chinese medicine has a more prominent stake in the global alternative pharmaceutical industry rather than Ayurveda or Unani. Some of the reasons can be attributed to the fact that during the past 25 years, numerous schools of Chinese medicine have opened in the west, and currently in the U.S, 36 states has licenses to formally allow the practice of acupuncturists as health professionals [30]. Acupuncture, an alternative to physiotherapy, is an exciting concept that was embraced by the west some 30 years back although now Yoga or Asanas from the Ayurvedic system of therapy are also gaining more prominence in western culture. Lastly the presence of a large Chinese diaspora has lead to huge exports of Chinese medicine in the west and has eventually helped Chinese herbs and traditional medicine to catch global attention.

References

 Wiebrechta A, Gausb W, Beckera S, Hummelsbergera J, Kuhlmannaa K. Safety aspects of Chinese herbal medicine in pregnancy—Re-evaluation of experimental data of two animal studies and the clinical experience working group. Safety of Chinese herbal medicine during pregnancy. Berlin: 2014. Elsevier Ltd. http://www.redkank.com/file/downloads/64f11df 23723a07cd90b0631772e8e8a.pdf.

- Maciocia G. Safety of Chinese herbal medicine 1999. Su Wen Press. Buckinghamshire, UK. http://www. three-treasures.com/pdf/safetyofchinesemedicine. pdf.
- Li L, Dou LX, Neilson JP, Leung PC, Wang CC. Adverse outcomes of Chinese medicines used for threatened miscarriage: a systematic review and metaanalysis. Hum Reprod Update. 2012;18(5):504–24. Advanced Access publication on June 2, 2012 doi: 10.1093.
- Hemminki E, Mantyranta T, Malin M, Koponen P. A survey on the use of alternative drugs during pregnancy. Scand J Soc Med. 1991;19:199–204.
- Nordeng H, Havnen G. Use of herbal drugs in pregnancy: a survey among 400 Norwegian women. Pharmacoepidemiol Drug Saf. 2004;13:371–80.
- Borchers AT, Hackman RM, Keen CL, Stern JS, Gershwin ME. Complementary medicine: a review of immunomodulatory effects of Chinese herbal medicine. Am J Clin Nutr. 1997;66:1303–12.
- Chuang CH, et al. Herbal medicines used during the first trimester and major congenital malformations: an analysis of data from a pregnancy cohort study. Drug Saf. 2006;29:537–48.
- Chuang CH, et al. Use of Coptidis Rhizoma and fetal growth: a follow-up study of 9,895 pregnancies. Pharmacoepidemiol Drug Saf. 2006;15: 185–92.
- Choo J-JV. That fatty lump: discourses on the fetus, fetal development and filial piety in China before the 11th Century CE, Nan Nü. 2012;14:177–221. https://www.academia. edu/2402164/That_Fatty_Lump_Discourses_on_ the_Fetus_Fetal_Development_and_Filial_Piety_ in_China_Before_the_Eleventh_Century_CE.
- Blanc CL. Early Chinese texts: a bibliographical guide. In: Lowe M (ed). Early Chinese Texts: A Bibliographical Guide. Berkeley: Society for the Study of Early China; Institute of East Asian Studies, University of California, Berkeley; 1993.
- Chinese medicine workshop natural remedies for a healthy pregnancy, delivery, and postpartum recovery. Acupuncture Medical Group – 862 Folsom St., San Francisco 94107 415-810-7434.
- Ping W. Lemon and maintaining early stage of pregnancy. Digestive health, fertility & pregnancy, food & diet therapy. Australia; 2010. http://www.pingminghealth.com/article/574/lemon-and-diet-therapy/.
- Takei A, et al. Meningoencephalocele associated with Tripterygium wilfordii treatment. Pediatr Neurosurg. 1997;27:45–8.
- Leung KY, et al. Are herbal medicinal products less teratogenic than Western pharmaceutical products? Acta Pharm Sin. 2002;23:1169–72.
- Koren G, et al. Maternal ginseng use associated with neonatal androgenization. JAMA. 1990;264:2866.

- Zhu YP. Chinese materia medica: chemistry, pharmacology and applications. Amsterdam: Harwood Academic Publishers; 1998. p. 30.
- Thai HC. Chinese and Western herbal medicine: a guide to potential risks and drug interactions. Date Authored: 01 Sept 2004.
- Williams MA, Mittendorf R, Lieberman E, Monson RR. Adverse infant outcomes associated with firsttrimester vaginal bleeding. Obstet Gynecol. 1991;78:14–8.
- But P. Attitudes and approaches of traditional Chinese medicine to herbal toxicity. J Nat Toxins. 1995;4(2):212.
- Zhang LL. Combined Chinese medicines for threatened miscarriage. Chin Fore Med Treat. 2009;34:89 [Chinese].
- Teng J, Wu XQ. Bu Shen Yi Qi formula and Progesterone for threatened miscarriage. Master thesis. Hubei Chin Med College; 2008. p. 1–10 [Chinese].
- 22. Paulus WE, Zhang M, Strehler E, El-Danasouri I, Sterzik K. Influence of acupuncture on the pregnancy rate in patients who undergo assisted reproduction therapy. Fertility and Sterility. Elsevier Science Inc. 2002;77(4):721–24. http://www.comoxvalleyacupuncture.com/pdf/paulus%20study.pdf.

- Betts D, Smith CA, Hannah DG. Acupuncture as a therapeutic treatment option for threatened miscarriage. BMC Complement Alternat Med. 2012;12:20.
- 24. Han JS. Acupuncture and endorphins. Neurosci Lett. 2004;361(1–3):258–61.
- Smith C, Crowther C, Beilby J. Acupuncture to treat nausea and vomiting in early pregnancy: a randomized controlled trial. Birth. 2002;29(1):1–9.
- Guerreiro da Silva JB, Nakamura MU, Cordeiro JA, et al. Acupuncture for tension-type headache in pregnancy: a prospective, randomised, controlled study. Eur J Integr Med. 2012;4:366–70.
- Guerreiro da Silva JB, Nakamura MU, Cordeiro JA, et al. Acupuncture for dyspepsia in pregnancy: a prospective, randomised, controlled study. Acupunct Med. 2009;27:50–3.
- Zollman C, Vickers A. ABC of complementary medicine—what is complementary medicine? Br Med J. 1999;319:693–6.
- Tang JL, Liu BY, Ma KW. Traditional Chinese medicine. Lancet. 2008;372:1938–40.
- Dharmananda S. Ayurvedic herbal medicine and its relation to chinese herbal medicine. Institute for traditional medicine. Portland, Oregon. http://www.itmonline.org/arts/ayurherb.htm.

43

General Perspectives of Homeopathic Medicine During the First and Second Trimesters of Pregnancy

Priyodarshi Sengupta, Nutan Gavhane, Nandita Bose, Sushanta Banerjee, and Niranjan Bhattacharya

Introduction

Homoeopathy is a system of medicine developed by Dr. Samuel Hahnemann, a German Physician who lived between 1755 and 1843 [1]. Homoeopathy is a scientific method of treatment that is based on the application of the law of similars and potentization. The law of similars states that "a substance that causes, in a healthy person. Symptoms similar to those of a disease state can

P. Sengupta, MSc, MPhil • N. Gavhane, BHMS Department of Regenerative Medicine and Translational Science, School of Tropical Medicine, Kolkata, India

N. Bose, MD

Director, School of Tropical Medicine, Kolkata, India

Formerly, Professor, Department of Pathology, IPGMER, SSKM Hospital, Kolkata, India

S. Banerjee, MD Director, Medical Education, Govt. of West Bengal, Kolkata. India

Formerly, Professor and Head, Department of Pharmacology, RG Kar Medical College, Kolkata, India

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India

e-mail: sanjuktaniranjan@gmail.com

cure a sick person in that similar disease state". The Latin phrase Similia Similibus Curentur that means "likes cured by likes". This law of similars is based on years of observation and a number of discoveries and reflections found throughout the history of medicine, long before Hahnemann's time. Even Hippocrates, the "father of Medicine", had stated that cure could be achieved through the action of "Similars". Hahnemann first experienced the law of similars while translating a textbook of medicine in which it was reported that Cinchona bark was used to cure malaria. He decided to test Hippocrates' theory by taking some Cinchona bark himself, and found that as a healthy person, he actually developed symptoms very similar to those of malaria. This led Hahnemann to develop a hypothesis that Cinchona appears to cure malaria because it produces the symptoms of malaria in healthy people [1, 2]. Potentization on the other hand is a process by which serial dilutions along with succession or vigorous shaking is applied to prepare homeopathic remedies. This process of alternative dilution and shaking is thought to remove the toxic side effects of the main component of the drug to have a deeper curative property of the drug.

The relative safety of Homoeopathic medicines makes them invaluable in pregnancy. "There's nothing safer", says Ananda Zaren, a nurse, midwife and Homoeopath in Santa Barbara, California, who has used Homoeopathic medicines in hundreds of births. Zaren adds, "The medicines help strengthen the woman physically and psychologically".

Homoeopaths differentiate the symptoms between acute symptoms and chronic symptoms. Acute symptoms represent self protective efforts of the organisms dealing with some type of recent stress or infection whereas chronic symptoms refer to recurrent, unsuccessful efforts of the organism to re-establish health. Instead of prescribing a medicine primarily for the most prominent symptoms, the homoeopath may prescribe 'Constitutional Medicine' which is individualized to the totality of a woman's symptoms [3, 4].

Jaceques Imberechts, M.D., a respected Belgian homoeopath, notes, "The Homoeopathic medicines are very effective in healing so many symptoms and syndromes of pregnancy that I have found that my patients rarely request or need anything other than Homoeopathic treatment" [5]. Richard Moskowitz, M.D., a Boston homoeopath, has found the best results in treating varicose veins with Pulsatilla and Hamamelis [6]. Marcel Simons, M.D., a Belgium obstetrician and homoeopath has also observed good results with these medicines [5, 6].

Why Homeopathy and Traditional Complementary and Alternative Medicines (TCAM) Are an Increasingly Popular Field of Medicine: Their Current Status in Different Countries

In 1991, USA instructed the National Institute of Health (NIH) to create a new federal office that will follow FDA guidelines; this came to be known as the Office of Alternative Medicine (OAM). In the UK, although chiropractice and osteopathy are officially regulated, incidence of TCAM usage is around 40 %. In other developed countries usage of TCAM is comparably low, for instance, it is 46 % in Australia, 70 % in Canada, 31 % in Belgium and 49 % France However, in the developing countries, the usage of TCAM is quite staggeringly high with Ethiopia reporting the highest usage of these forms of drugs at 90 %, followed by Benin at 80 %, India and Rwanda at 70 % while Tanzania and Uganda are at 60 % [7].

This might be attributed to the fact that in developing countries, cost is a major factor as

many people cannot afford allopathic drugs due to socio-economic and financial constraints: therefore TCAM ha become a better option for treatment. Another important area which is the emergence of more and more chronic diseases in the developing countries [8]. Harmful side effects of allopathy can be one of the main reasons why the use of TCAM is gaining grounds among many patients in the west and developed nations [8]. A survey carried out in Chhattisgarh, a state in India, reported that 41 % of people were satisfied with ayurvedic drugs [3]. However a negative trend was observed when around 94 % of 492 respondents in another study, preferred allopathic drugs, with 3.25 % favouring ayurvedic and 2.85 % homeopathic drugs, in spite of another opinion expressed by the same respondents that the biggest side effect was associated with allopathic drug at 93.9 %, followed by 3.45 % in ayurveda, and 2.65 % in homeopathy [9]. Also it was found that ayurveda was the preferred choice of medication in case of long term treatment followed by allopathy and homeopathy whereas people found allopathic medicine to be the most common medicine that the group would opt for in the first instance [9]. Therefore the above data reveals that ayurvedic and homeopathic medicines perhaps play a prominent, maybe a dominant role, among patients when it comes to opting for the best suitable treatment.

Rationale behind the Usage of Homeopathic Medicine: a Perspective into Some Clinical Trials

Before moving into the perspectives and usage of homeopathic drugs during the first and second trimester of pregnancy and their role in growth and development of the fetus it is important to consider certain facets of homeopathy in terms of scientific principles. One of the major challenges of homeopathy like other traditional complementary and alternative medicine is its lack of definitive clinical data in terms of clinical trials which can be attributed to the fact that most of the trials lack a sound clinical design and approach [10]. A study of osteoarthritis in a group of 36 patients concluded that one third of the patients who received a dose of 30× Rhus Tox showed no better symptoms than the other placebo group, whereas a third group showed some response to standard care of NSAID's. This team, when they conducted a similar study on hay fever symptoms with a $30 \times$ dose of pollen grains processed, found that the group which received the homeopathic treatment showed six times fewer symptoms than the placebo group. Also against radiation in albino rat, homeopathy has shown less chromosomal aberration and cell death with respect to placebo albino rat. Also another study in an animal model reported that homeopathy can reduce the reddish appearance and infer a protective role against causing inflammation of the skin when exposed to X-rays for long periods [11].

Homeopathy has now become a major contribution of alternative and complementary medicine to many countries in Europe. In the UK 17 % of people use homeopathy according to a House of Lords report in 2000 [10]. Also another study found that 55 % of English people related homeopathy to be a useful medicine against a mere 14 %. In Ireland a 4 month study conducted in children found that 57 % of parents reported using CAM [12] for their children among which usage of homeopathy were 16 % which was also incidentally the highest and the most common of these CAM medicines used among the respondent groups [12]. According to a 2004 study, statistics showed that in France, 62 % of French mothers used homeopathy 12 months ahead of their pregnancy [13]. Also another study found that around 95 % of French pharmacists recommended prescribing homeopathic medicine to pregnant mothers [14]. In Germany during 1993, 1993 trained homeopathic physicians were present which rose in 2006 to around 6,073 trained physicians in this branch of medicine [15].

Fetal Development

Events during the First Week of Human Development

- 1. Oocyte immediately after ovulation.
- 2. Fertilization, approximately 12–24 h after ovulation
- 3. Stage of the male and female pro nuclei.

- 4. Spindle of the first mitotic division.
- 5. Two cell stage approximately 30 h of age.
- Morula containing 12–16 blastomeres approximately 3 days of age.
- 7. Advanced Morula stage, reaching the uterine lumen at approximately 4 days of age.
- 8. Early Blastosyst stage, approximately 4.5 days of age, the zona pelucida has disappeared.

After 9 or 10 days of conception the blastocyst is fully attached to the endometrium. Primitive placental blood circulation begins in 12 days or so after conception. In the meantime the blastocyst starts to produce hormones like Human Chorionic Gonadotrophin Hormone (hCG) which is essentially one of the most important hormones responsible for inferring selective safe passage across the transplacental membrane (Miasm-Psora). This hormone is also detected in women's urine defining the start of pregnancy. It is in this stage that the fetus starts its development.

Stages of Development of Fetus after Fertilization

In the first stage of development, the embryo is called a zygote and with subsequent divisions termed as cleavage the different stages of development take place like the formation of morula, blastomere compaction resulting in the formation of blastocyst with the blastocoel cavity. The blastocyst contains the inner cell mass, a rich and potent source of embryonic and pluripotent stem cells. Appearance of epiblast, hypoblast, extra embryonic mesoderm and visceral mesoderm along with the trophoblast is initiated.

First Trimester

Following 13 or 14 days after conception, a "primitive streak" appears which later develops into the fetal central nervous system. The preembryo is now referred to as an embryo. Structurally this embryo appears to be a very small blob of undifferentiated tissue at this stage of development. In 3 weeks time the embryo matures. The cardiac muscle slowly develops and in about 3 weeks time light heart beat can be observed. In the third week after conception the in utero size of the fetus matures to 1/5" long and resembles a tadpole. The head, which starts developing in this organogenesis phase starts developing and to compensate fetal breathing activities, structures resembling like those of the fish gills develop in the future throat area. Tiny arms and leg buds start forming from 5 weeks time. Hands with webs between the fingers are formed at the end of the arm buds. The face has a distinctly reptilian aspect [16–18].

In 6 weeks time the embryo is about 1/2''long. The face has one eye on each side of its head. The front of the face has connected slits where the month and nose eventually develop. By the time of the seventh week the embryo almost loses its tail. The face is mammalian but is somewhat resembles pig-like. Pain sensors appear. However the higher functions of the brain are yet to develop and the pathways to transfer pain signals from the pain sensors to the brain are still immature. In 2 months time the embryo's face resembles that of a primate although not fully human in appearance. Some of the brain regions begins to evolve and mature; this is the primitive "reptilian brain" that functions throughout life.

At 10 weeks time the embryo is now called a fetus. Its face looks human and its gender may be detectable via ultrasound. The fetus is about 3 in. long by the time it reaches 13 weeks time and weighs about an ounce. Fingernails and bones can be seen [19].

Second Trimester

During second trimester, i.e., after 17 weeks or 3.9 months of conception the fetus is about 8 in and weighs half a pound. The movements of the fetus may begin to be felt. Its heartbeat can usually be detected in the mother's abdomen. In about 22 weeks or 5 months the size of the fetus further increases to anout 12 in. in length and it weighs about a pound and fetal hair starts to develop on the dermal side. Its movements can be felt [20, 21].

In about 26 weeks or 6 months the fetus is 14 in. long and almost 2 lb. In the meantime the lung bronchioles develop. Functional interlinking of the neurons initiates the higher functional development of the brain for the first time. This is the time that the fetus is able to feel pain for the first time [6].

Management of Homeopathic Medicines in First and Second Trimesters of Pregnancy

First Trimester

Morning sickness is one of the first signs of pregnancy. It is often characterized by a feeling of nausea or sickness and in many cases it can induce vomiting too. This symptom usually lasts until the patient is 14–16 weeks pregnant but can occasionally persist through the whole pregnancy. If very severe, it is referred to as "Hyperemesis Gravidarum" and may require a hospital admission and/or conventional medication [17, 18, 22]. This can be treated by Nux Vomica if the vomiting is worse in the morning. Coculus Indica derived from the cockle flower can be very helpful if nausea starts immediately after lifting the head from the pillow [23].

Due to the changes in hormonal regulation breast tenderness is another common ailment that is often observed among pregnant mothers in the first trimester. Pulsatilla can be recommended as a remedy usually taken 30 c once or twice daily as it normally helps to soothe the tenderness [23]. Also, Natrummur, Belladona and Phytolacca may be suggested [3, 24, 25].

Treatment

Sepia is one of the most important remedies during this pregnancy period where the smell of food triggers sickness and eating does not help as it aggravates the problem of vomiting. Different drugs used for treating this symptom are Arsenicum, Ipecac, Kreosote, Symphoricarpus Racemosa, Cuprum ars, Gossypium, Iris V and Carbo Veg [5, 16, 26, 27]. Urinary frequency in the mother is increased which can be due to the pressure of the gravid uterus during first trimester. Treatment includes Causticum, Coccus Cacti and Merc. Sol which can be used to treat sour smelling urine.

Second Trimester

Backache is a common ailment during 3–6 months of pregnancy as the bump will be getting bigger creating pressure on the spinal cord region including slight and tender inflammation of the pelvis enlarging as the ligaments stretch due to the effect of pregnancy hormones.

Treatments in homeopathy include drugs like Kali. Carb, Natrummur and Arnica.

Constipation, another common syndrome in pregnant mothers, is due to added pressure on the bowels from the growing baby and also due to the effects of the hormone progesterone.

Treatment includes Lycopodium, Alumina, Nux vomica, Opium and Platina.

Haemorrhoids like constipation can be treated by Collinsonia, Lachesis and Sulphur. Varicose veins are treated by apismel, Arnica, Carbo Veg and Hamamelis [28–30].

Due to the maturation of the fetus in the second trimester the womb of the mother considerably increases in size thereby putting pressure on the pelvic main arteries and the veins. This results in the feeling of faintness while lying back. This phenomenon is known as varicose veins. Avoiding long periods of standing during this period is normally recommended. However the latter is not possible in case of working women. Pulsatilla of dose 6 c twice daily can be prescribed. Carbo Vegetabilis can be recommended in cases where Pulsatilla is not proving to be effective.

Anemia in Pregnancy

Anemia is common in pregnancy and it has been described as the second leading cause of maternal death. According to WHO guidelines, anemia during pregnancy is considered when the hemoglobin level <11.0 g in pregnancy, but as per the Indian standard it is considered when hemoglobin level is less than 10.5 g. As the pregnancy progresses, the blood becomes diluted and the woman may become anemic. The dilution of blood in pregnancy is a natural process and starts at approximately the 8th week of pregnancy and progresses until the 32nd–34th week of pregnancy.

Requirement of Iron: 0.8 mg daily in the first trimester and 4–5 mg daily in the second trimester [29].

Treatment

Homeopathy offers a number of remedies that may be helpful in treating anemia. Homeopathy considers anemia as the symptomatic results of some other underlined conditions and attempts to treat these conditions. Before prescribing a remedy, homeopaths should take into account a person's constitutional types and then accordingly determine the most appropriate treatment for each individual. Indications of some of the commonly used Homeopathic medicines are as follows [20, 21].

- Ferrum Met.
- Ferrum Phos.
- Pulsatilla
- Cinchona
- Nat. Mur
- Acetic acid
- Calc carb
- Calc Phos
- Alumina
- Nux Vomica
- Arsenicum
- Kali Carb

Concepts of Homeopathy with Respect to Modern Medicine

With respect to modern medicine many of the theories, ideas and logic behind the use of TCAM's are scientific although no scientific answers exist. For example, homeopathy believes that there are diseases that are already manifest in the human body, which is quite similar to the modern medical concepts, for instance, aberrations of cancer controlling genes or activation of the gut flora that can cause serious implications to our health [31]. On the other hand most homeopathic drugs are manufactured on the basis of dilution principle which raises the question as to how can diluted drugs elicit potent activity inside the body if they do not have a minimal therapeutic threshold, or rather, what are the criteria for designing a homeopathic drug with a minimal response level to negate the side effects of the drug constituents. According to homeopathic physicians, to maintain the safety of such drugs the potency of the main drug composition is reduced into serial dilutions [31]. This concept of reducing the harmful components of a drug by diluting it is sometimes very confusing, showing that due to lack in scientific design of clinical trials and experiments, most homeopathic manufacturers or clinicians remain confused [31]. Such occasional anomalies question the very scientific basis of treatment of diseases by Traditional Complementary and Alternative Medicines (TCAM) and raise serious questions regarding their validation and importance [31].

Therefore homeopathy focuses more on the curative properties of diseases which are already manifest in the human body by means of diluted drugs which again is a concept under the scanner by many theorists and scientists [31].

Conclusion

Homoeopathy is a very specialized therapy. Remedies are prescribed on the basis of a person's individual symptoms. The medicines are very mild and given in minute doses, hence they are perfectly safe to take during pregnancy. In fact homoeopathic medicines boost up immunity of the patient, thereby allowing the patient to acquire resistance against diseases. However like every other branch of medicine more scientific and clinical data are required before the safety and efficacy of homeopathic medicines during pregnancy are clinically proven.

References

- Hahnemann S. The chronic diseases, their peculiar nature and their homoeopathic cure. Vol. 1 and 2. New Delhi: Jain Publishing Co.; Reprint 1983.
- Hahnemann S. Organon of medicine (trans. W.Boericke). Leipzig; 1920. Paperback 15 August 2003. http://www.amazon.in/Organon-Medicine-Samuel-Hahnemann/dp/8170210852. http://organonofmedicine.com/sixth-edition/
- Kaplan B. Prof Care Mother Child. 1994; 4(6):185–7.
 Katz T'. Complement Ther Nurse Midwifery. 1995; 1(6):159–64. 9).
- Wertz RW, Wertz DC. Lying-in: a history of childbirth in America. New York: Schoken; 1979. p. 137.
- Ullman D. Homeopathy: medicine for the 21st century. Berkeley: North Atlantic Books; 1987. http:// www.amazon.in/Homeopathy-Medicine-Century-Dana-Ullman/dp/1556430159.
- 6. Ullman D. Homeopathic perspective on pregnancy and labor: getting off to a good start by (excerpted from discovering homeopathy: medicine for the 21st century). Berkeley: North Atlantic Books; 1991. http://www.brauer.com.au/how-homeopathy-works.
- Syed SB, Haering, SA. Traditional medicine, Johns Hopkins Bloomberg School of Public Health December 2007. Prepared as part of an educational project of the Global Health Education Consortium and collaborating partners. http://cugh.org/sites/ default/files/content/resources/modules/To%20 Post%20Both%20Faculty%20and%20Trainees/44_ Traditional_Medicine_FINAL.pdf.
- Pal SK. Complementary and alternative medicine: an overview. Curr Sci. 2002;82(5):518–24 by D.P. Agarwal.
- Nagori K, Sharma M, Agarwal, A, Agarwal AK, Sharma A, Verma H, Tripathi DK. General Awareness on Allopathic, Ayurvedic and Homeopathic system of Medicine in Chhattisgarh, India. Int J Pharm Pharm Sci. 2011;3(Suppl 4). http://www.ijppsjournal.com/ Vol3Suppl4/2379.pdf.
- Ullman D. Homeopathy scientific evidence for homeopathic medicine. Homeopathic educational services. https://www.homeopathic.com/Articles/ Homeopathic_research/Scientific_Evidence_for_ Homeopathic_Medicine.html.
- Bildet J, Guyot M, Bonini F, et al. Demonstrating the effects of Apis mellifica and Apium virus dilutions on erythema induced by UV radiation on guinea pigs. Berl J Res Homeopathy. 1990;1:28.
- Ullman D. Homeopathic medicine: Europe's no. 1 alternative for doctors. http://www.huffingtonpost. com/dana-ullman/homeopathic-medicineeuro_ b_402490.html?ir=India&adsSiteOverride=in.
- Weiser M, Strosser W, Klein P. Homeopathic vs. conventional treatment of vertigo: a randomized doubleblind controlled clinical trial. Arch Otolaryngol Head Neck Surg. 1998;124:879–85.

- Damase-Michel C, Vie C, Lacroix I, Lapeyre-Mestre M, Montastruc JL. Drug counselling in pregnancy: an opinion survey of French Community Pharmacists. Pharmacoepidemiol Drug Saf. 2004;13(10):711.
- Joos S, Musselmann B, Miksch A, Rosemann T, Szecsenyi J. The role of complementary and alternative medicine (CAM) in Germany – a focus group study of GPs. BMC Health Serv Res. 2008;8:127. doi:10.1186/1472-6963-8-127. http://www.ncbi.nlm. nih.gov/pmc/articles/PMC2442431/pdf/1472-6963-8-127.pdf.
- Castro M. Homoeopathy for pregnancy, birth and the first year. New York: St. Martins; 1992.
- Sharma RK. Pregnancy and thyroid dysfunction. Homeopathic J. 2012;5(4) (General Theme). http:// www.homeorizon.com/homeopathic-articles/endocrinology/pregnancy-and-thyroid-dysfunction.
- Dr. Ellen kramer MCPH, ARH; The College of Homoeopathic Article, 23 Sep 2011.
- Hamlin FW. A manual of practical obstetrics. New York: Boericke and Runyon; 1908.
- Edmonds K. Dewhurst's textbook of obstetrics and gynaecology. 7th, 8th ed. Wiley Blackwell; 2009. http://as.wiley.com/WileyCDA/WileyTitle/productCd-0470753331.html#.
- Duncan TC. Diseases of infants and childrens and their homeopathic treatments, vol. 4. Chicago: Duncan Brothers; 1880. p. 492.
- 22. Borland DM. Homeopathy for mother and infant. New Delhi: Indian books and periodical syndicate, B. Jain; 1929 (reprint). http://etheses.whiterose.ac. uk/4089/1/thesis_all_include_Final_Submission_ 13th_June_2013.pdf.
- Its baby time by Jenifer Worde. British Homeopathic Association. http://www.britishhomeopathic.org/bha-

charity/how-we-can-help/articles/pregnancyand-labour.

- Kleijenen J, Knipschild P, ter Riet G. Trials of homoeopathy. BMJ. 1991;302(6782):960.
- Moskowitz R. Homoeopathic medicines for pregnancy and childbirth. Berkeley: North Atlantic; 1992.
- Fisher C. A handbook on the diseases of children and their homoeopathic treatment. Chicago: Medical Century; 1895.
- Castro M. Homoeopathy. A theoretical framework and clinical application. J Nurse Midwifery. 1999;44(3):280–90.
- Banerjea SK. Miasmatic diagnosis. Practical tips with clinical comparisons. New Delhi: B. Jain Publishers; 1999.
- 29. Stoltzfus RJ, Dreyfuss ML. Guidelines for the use of iron supplements to prevent and treat iron deficiency anemia. Washington DC: International Nutritional Anemia Consultative Group (INACG). http://www. who.int/nutrition/publications/micronutrients/guidelines_for_Iron_supplementation.pdf.
- 30. Brandon J, Md. Bankowski, Amy E., MD Hearne, Nicholas C., MD Lambrou, Harold E., MD Fox, Edward E., MD Wallach, (Editors), The Johns Hopkins University Department (Producer). The Johns Hopkins manual of gynecology and obstetrics. 2nd ed. Philadelphia: Lippincott Williams & Wilkins Publishers; 2002.
- 31. Difference between Ayurveda and Homeopathyl difference between Ayurveda vs Homeopathy. http:// www.differencebetween.net/science/health/ differencebetweenayurvedaandhomeopathy/3/6.

44

Ayurvedic Perspective of Pregnancy and Fetal Development During the First and Second Trimester

Priyodarshi Sengupta, Madhav Rayate, Abhishek Kumar, Anamika Kumari Prasad, Nandita Bose, Sushanta Banerjee, and Niranjan Bhattacharya

Introduction

Healing is defined as the concept or rather an art of the flow of energy from one part of the body to the other where it is most needed in order to re-establish a balance in the individual. This is the most basic concept on which the major traditional, complementary and alternative system of medicine has been relying upon since age old time [1]. Traditional complementary and alternative medicine also termed as TCAM encompasses a wide spectrum of healing resources such as health systems, holistic approaches, theories, beliefs and their age old practices since time immemorial. TCAM deals with that branch of medicine that has not yet been readily integrated into the standardized general health care models due to its challenges in scientific findings as perceived by the medical fraternity [1]. The word complementary or alternative medicine evolved gradually over the years when people started to use these type of medicines when they found limitations in the current allopathic system [1]. TCAM's are based on the common concept of spirituality and controlling the quality and standard of life. According to Dr. Pal, TCAM can be further classified into seven categories [1]. Mind and body medicine which deals with behavioural, psychological, social and spiritual healing is the first category. An alternative medical approach which is the next one deals with yoga, puncture, hypnosis and meditation and Qi Gong [1]. The third type is related to the application of ayurvedic principles like panchkarma, health and lifestyle promotion and electrodermal diagnosis and is termed as

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India e-mail: sanjuktaniranjan@gmail.com

P. Sengupta, MSc, MPhil • M.Rayate, BAMS A. Kumar, BAMS • A.K. Prasad, BAMS, MPhil Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

N. Bose, MD Director, School of Tropical Medicine, Kolkata, India

Formerly, Professor, Department of Pathology, IPGMER, SSKM Hospital, Kolkata, India

S. Banerjee, MD Director, Medical Education, Govt. of West Bengal, Kolkata, India

Formerly, Professor and Head, Department of Pharmacology, RG Kar Medical College, Kolkata, India

lifestyle based disease prevention [1]. The different formulations of herbal drugs and medicines used extensively and widely in case of Unani, Ayurveda, tribal medicine, Homeopathy and Ayurveda is referred to as biologically based therapies which is the 4th type. Chiropractice, massage, kinesiology, Chinese tuina massage are all related to the fifth concept of manipulative and body based systemic therapies. Biofield and bioelectromagnetic therapies which comprises of the Reiki, external Qi Gong, use of electronic magnetic fields for medical diagnosis forms the sixth and seventh basis of TCAM based medical therapies [1].

Growth and Development of Human Fetus as per Ayurveda

Ayurveda, literally meaning the science of life (ayur-life,veda = science) is an ancient medical science developed in India. It is believed to have been passed on to humans by the Gods themselves [2]. Modern conceptions of Ayurveda have developed and evolved over thousands of years, originating from the confluence of ideas of several ancient treatises. Widely regarded as the oldest form of healthcare in the world, the fundamentals of Ayurveda can be found in Hindu scriptures called the Vedas – the ancient Indian books of wisdom. The Rig Veda, written over 6000 years ago, contains a series of prescriptions that aid humans in overcoming various ailments [3]. The aims of this system of medicine are clearly delineated.

They can be summed up as follows: Swasthyas swasthya rakshanam: To protect health and prolong life; Aturasya vikar prashamanamcha: To eliminate diseases and dysfunctions of the body [4]. Ayurvedic practices mainly depend on the treatment of the three dosas (evils) of an individual, i.e., vata, pitta, kapha [5] by the use of herbal medicines, minerals or metal supplementation (rasa sastra), surgical techniques, opium and application of oils and massages in panchkarma. The practice of Ayurveda is not directed at symptomatic treatment, but aimed at the root of disease by addressing the state of imbalance of the three dosas. Methodologically, this involves their expulsion from the system by Panchkarma therapies [6]: Vaman (emesis), Virechan (Puragation) and Vasti (enema), Shirovasti(retention of medicated oils over the scalp) Nashya. (Insufflation of medicinal herbs) In Ayurveda, fetal development is described comprehensively. Health problems pertaining to the mother and the growing fetus as well as dietary recommendations finds a prominent place in a number of Ayurvedic texts.

Embryology

The union of sukra (sperm) artava (ovum) and atma or jiva inside the uterus, i.e., kuksi is known as garbha (embryo or fetus). Besides atma the association of prakriti and vikaras are also essential for the development of arms, legs, tongue, nose, ears, hips and other body parts and it is termed as sarira. As per ayurvedic classics the body is composed of five tatwas (elements), five indiriyas (sense organs) and seven dhatus (liquids, blood, muscle, fat, bone, marrow, semen). The garbha (fetus) formation is mainly due to vayu function as per Carak and Susruta; this indicates that vayu which moves agni for the formation of the five srotas (channels) therefore is the most important [7]. The garbha has six parts: (a) JIVA/ATAMA (conciousness) - This is the root element of garbha. Nothing can happen in life without atma. (b) Five mahabhutas (Elements) - Prithvi (Earth), Akash (space), Vayu (Air), Agni (Fire), Jal (Water) (C) The essential part of the garbha is ojas or sometimes called first dhatu [7] (Table 44.1).

Although Ayurvedic medicine is one of the oldest branches of medicine, many of its concepts in development of the fetus are extremely scientific, some of which may have laid the basis of modern medicine. In Ayurveda, space, air, fire, water and earth are described as panchmahabhutas and each one of these elements give qualities of life to the fetus. Looking at the development of the embryo with respect to Table 44.1, it can be observed that any deficiency in the qualities is directly linked to the five elements also known as the Panch Bhutas [9]. They are listed as below:

(a) Vayu or air grants the dosha body type to the fetus. It is this element that decides whether the child will have Kapha dosha type, Pitta dosha type or Vata dosha type body. Amount of 'vayu' in one's body or the capability of

Panch Bhūtas	Ākāśa (Space)	Vāyu (Air)	Agni (Fire)	Jala (Water)	Prithvī(Earth)
Qualities	Sound	Touch	Sight	Taste	Smell
	Hearing	Feeling	Seeing	Tasting	Evacuation
	Lightness	Roughness	Warmth	Coldness	Heaviness
	Smallness	Impulsion	Metabolism	Unctuousness	Stability
	Division	Activity	Enzymes	Moisture	Hair
	Srotas(channels)	Formation of dhatus Transportation of Dhatus	Luster	Blood	Bones
	Orifices	Expiration	Body temperature	Semen	
				Liquid	
				Fat	

 Table 44.1
 Qualities of panchamahabhutas in the development of the fetus

retaining or releasing it from the body decides the dosha body type of any human being [8].

- (b) Agni or fire decides what will be the metabolism of the baby's body. Metabolism is defined by digestive capability of the body if agni is high, the baby will have good digestion power and vice versa. It also decides about body temperature, light and lusture, sharpness, intellect, complexion [8]
- (c) Jal or water is called the lifeline by Ayurveda. It is responsible for cleansing the entire bodily organs. It decides about the creation of rasa or taste, coldness, softness, moistening ability of the body, blood and semen characteristics, as also of fat, muscle mass and saliva [8].
- (d) Prithvi or earth is solid by nature. It decides on how strong the baby will be- physically as well as emotionally. It decides such structural attributes of the body as heaviness, stability, bones, nails, courage, skin, muscles, body hair etc. It also decides upon the stability and smelling ability of the baby [8].
- (e) Akash or sky is eternal and it represents growth. It is responsible for such attributes of the baby as lightness, observing power, psych, intelligence, pride as also for lethargy, hearing ability, sleep etc. [8]

So, when an element is in excess or in lesser quantity in the fetus, they affect them in different ways and grant them the body as well as mental and emotional characteristics they have.

According to Ayurvedic classics it is believed that all soft parts of the body come from the mother and the hard parts of the body from the father [9].

All the classic authors have detailed accounts of the embryology and monthly development of the fetus. However a detailed description of such is outside the scope of this review. Aspects which are relevant are the desires, longing and cravings of the pregnant mother which must be fulfilled as they are indirectly linked to the desires of the fetus inside the mother's womb irrespective of whether the fetus is healthy or unhealthy [10]. It is also suggested that in order to keep the woman satisfied: both her wholesome and unwholesome desires should be fulfilled. Unfulfilled desires of the mother lead to aggravated Vata, which can either destroy or deform the fetus. On the other hand a pregnant woman, whose desires are fulfilled, will "beget a son endowed with valour and long life". The key events of human development including the fetal period following fertilization and organogenesis are quite well established clinically in Ayurvedic system of medicine. It is described in the following section.

Monthly Development of the Human Fetus According to Concept of Ayurvedic Medicine, Garbha (Embryo) Up to Second Trimester

According to *Charak Samhita* and *Sushrut Samhita*, although there are some conceptual differences between Ayurvedic and Western medicine still both these branches of medical sciences complement each other in terms of basic principles and overall similarities.

First month:

Sushrut and Charak Samhita: The kalal is formed in the first month which is composed of mucous similar to that of the nose and is semi solid, sticky and slimy. The kalal possesses all organs, systems and building blocks of development. It is described as the ultramicro version of a life that gets manifested later with increase in gestation period. Like each and every entity in the universe, it is made up of the five great elements namely Space, Air, Fire, Water and the Earth all combined together in different proportions [11–14].

Second month:

The embryo becomes solid in the presence of the heat produced by the kapha, pitta and vata which are regarded as the three basic energy sources of the body. The Garbha acquires either circular (pinda) or elongated (peshi) or semispherical (arbuda) form. If it is pinda, the Garbha develops into male and if peshi it develops into female. If the garbha is semispherical then it will possess both male and female characteristics [11, 12, 15, 16].

Third month:

Five pedestals/min. form (pindaka) for hands, feet, head and miniature forms of all body parts develop thus laying the very concept of organogenesis as per western medicine is concerned [17, 18].

Fourth month:

The minute forms of all body parts take specific form and shape. As the fetal heart becomes perceivable, the chetna dhatu (soul) becomes evident. This organ becomes the home of the soul. Garbha responds to sensory stimuli and that is why pregnant women have cravings (Douhrida). The woman is called Douhridini (with two hearts) [11, 12, 16].

Fifth month:

During this period according to Ayurveda the mind wakes up or becomes alert and the blood and the muscle tissue of the fetus enhances and matures. Most of the nutrition is provided to the muscle tissues and the fetal blood for its enrichment and nourishment. As the exchange of nutrients from mother to fetus enhances the mother feels weak and may start losing weight also [15, 16].

Sixth month:

The Buddhi (intellect with the power of grasping and retaining) matures or progresses mainly in this stage. The fetus continues to become strong and healthy [16].

Trends in Ayurvedic Medicine

The Ayurvedic approach towards motherhood including early stage pregnancy till childbirth is indeed a holistic one. Ayurveda through its different spiritual and philosophical ways prepare the woman for pregnancy. Due to acute shortage of trained and qualified Ayurvedic practitioners there is always a high chance of drug abuse. Also the reliability, by which Ayurvedic medicines are controlled, commercialized, marketed and regulated remains questionable. Currently there are no definitive protocols or laws in place as specified by the Drugs and Cosmetics Acts of India in order to ensure the safe marketing and commercialization of Ayurvedic medicines especially related to vulnerable groups like pregnant mothers. There are medications to control hormonal imbalances but most of the Ayurvedic medicines have heavy regulations on them. Also due to lack of proper knowledge among practitioners and general awareness among the masses one can see hundreds of quack and self proclaimed ayurved clinicians, second-hand forcible prints and advertisements which often malign the very basic principles and fundamentals of Ayurvedic treatment which is to heal with the laws of science and nature implemented together. The concept of Ayurveda is originally of herbarium medication even though Gynecology in modern medicine has many applications with natural ingredients and formulations. Some of the best Ayurvedic products of interest are Ovarin and Vatvidhwansan Ras [19–21] containing more than 24,000 ppm lead, 70,000 ppm Mercury and 1000 ppm Arsenic.

Role of Ayurveda for the Pregnant Mother and the Fetus in the Early Stages of Pregnancy

In Ayurveda the main concept during pregnancy is the science of well being of both the mother and the fetus. The mother should always try to be in a happy mood and should follow proper sanitation and hygiene like keeping neat and clean, well dressed, wear simple clothes and sleep under a roof always in a serene environment. The food she eats should be moist, tasty, enriched with nutrients mainly in liquid form. It should be also ensured that the food is enriched with all the six rasas (tastes) and treated by deepan drugs which are known to increase appetite and digestive power [20, 21]. An important and essential Ayurvedic recommendation for a pregnant mother is to control her dietary intake by avoiding rich foods and consuming more natural foods [22, 23]. Also the mother's quality and standard of life has a profound effect on her well being and behavior. Apart from ayurvedic herbal remedies and applications it is also postulated that yoga and meditation just like acupuncture in Chinese medicine can help a woman to obtain serenity and a sense of tranquility during her pregnancy period which in turn can aid in the nourishment and development of the mental health of the baby [13, 24].

Quality of life is another important aspect and concept of Ayurvedic medicine for pregnant women. Mothers during this stage should always abstain from excessive sex particularly during early and late pregnancy including overeating or fasting, sleeping during the daytime and staying up late at night. She should also avoid wearing tight clothes and tight belts, witnessing or listening to things which might give rise to feelings of sorrow, anger, horror pain or agony. She needs to avoid bumpy road rides and should restrict herself from excessive travelling and squatting for a long time or sitting in an uncomfortable position or on a hard surface. Mothers should also be warned against lifting heavy things or remaining in a bending position for a long time and oleation massage. Unless positively indicated, beholding

natural urges are essential except in emergency cases [25]. Dry, fermented, heavy, hot or strong food, alcohol and meat (although fish is allowed), visiting abandoned and remote places, leaning into a deep well or change in posture that may increase the chance of harm to the abdominal position during pregnancy should be avoided [20, 21]. The general rule is to take greater care during the first 3 months of pregnancy up to the completion of the seventh month. During the first trimester, stress is laid on stabilizing the pregnancy and nurturing the uterine bed through rasa and rakta dhatus. The embryo gets nourishment directly by percolation (upsnehan) hence more jaleeya (liquid) substances such as juicy fruits, coconut water, milk and so on are advised. In the first month, sipping cold milk and maintaining light diet and during the next 2 months, the intake of milk medicated with herbs like vidhari, shatavari, yasthimadhu, brahmin which are jeevaniya (life building) and garbhasthapak (helping nidation) are encouraged [21]. Honey and ghee are also recommended during this period. By the end of the third month, the body parts of the fetus becomes differentiated, sensory perceptions and motor reactions start developing, the heart starts beating and is said to express its desires through the maternal circulation. This is the period when the woman craves for certain foods/flavors required for both the mother and fetus. Hence Ayurveda recommends that her cravings be fulfilled as per as the requirements otherwise it can lead to contraindications. Brahmi helps in calming the nerves and is also a good prajasthapan (sustainer of pregnancy) [11–14, 21].

During the fourth month and extending up to the seventh month, drugs which give strength to the uterine muscles and nourishment to the embryo are advised, e.g., ashwagandha, kraunch beej and guduchi helps to prevent intrauterine growth retardation (IUGR). Nourishment starts through the umbilical cord by the kedar kulya method. During this time the diet should be consisting of rice, milk, butter and ghee. Fruits which are orange or yellow in color such as mangoes, apples, carrots, amalaki etc. along with leafy vegetables are also advised [26]. During the seventh month, the abdominal
skin gets stretched giving rise to itching and striations which are called kikkis. It should be either treated by taking sips of the infusion of berries or butter medicated with manjistha or by the application of the pulp of sandalwood and lotus or by a paste made of neem, basil and manjistha or oil medicated with karveer leaves or jasmine.

As soon as a pregnant woman enters the ninth month, she is supposed to move to the delivery stage. After an asthapan basti (simple enema), she should undergo anuvasan basti (retention enema of oil boiled with some herbs) which may be repeated. Tampons soaked in the same oil are kept in the vagina to make the pelvis soft and elastic. It enhances excretory function of apan vayu (urination and defecation) and expulsion of the fetus. The maternal skin and nails become soft, her strength and complexion are rejuvenated. Spotting of blood during any month of pregnancy should be taken seriously by the ayurvedic clinician [11, 19–21].

Ayurveda and Modern Science in Lieu of Fetal Growth and Development

In the first month, sperm and ovum unite and they turn into seed which further divides to take a more spherical seed like structure. These further divisions are referred to as cleavage in western medicine, help in forming the inner and outer seed layers and in between them a cavity is formed which remains filled with fluids known as blastocoele in modern medicine [11, 12]. According to modern science once the zygote has reached a two cell stage, it undergoes a series of divisions, forming blastula, morula stage then blastocoele or blastocyst stage [27]. In the third month all germinating parts appear. The head is quite large and fingers become visible. Male and female sex organs are formed although the eyelids and lip are not separated still [11, 12]. Modern science states that the head and upper limbs are still disproportionately large in the 3rd month and by the 12th week, external genitalia develop which can be diagnosed by ultrasonography. During the third month eyelids meet and fuse [27, 28]. According to *Charak Samhita*, after birth, a hole in the middle flap of the heart chamber closes referred to as the foramen ovale in modern medicine concept which closes after birth [29, 30]. By the third month, the heart appears and starts pumping. Designation of the heart chambers occurs and the fetal heart sounds can be heard in the maternal abdomen around the fourth month. This also has been elaborated in a similar manner in modern science. Mucous membrane of the vagina gives a bluish tinge and this sign appears by the beginning of the second month as per *Charak Samhita* [11, 12]. In modern science, it is referred to as Jacquemiers or chadwicks sign [31, 32].

In the fourth month fetal movements are felt and fine hair like structures can be seen on head and other sites of the body. Such hair like appearances are also known as lanugo [6]. Prior to 6 months timeframe, fetal skin remains wrinkled according to Ayurveda which is again consistent with respect to modern science [18, 33]. According to Ayurveda, testes descend into the scrotum at 9 months post conception. The fetus measures 20 in. in length and 3.5 kg in weight (normal and healthy) which is in accordance to both Ayurvedic and modern medical science concept. The size of the finger nails also increase with increased gestational time which is about 50 cm in case of modern medicine [34, 35]. Therefore from the above reviews and findings it can be noted that both the concepts of Ayurveda and modern medical science are based on the same principles and concepts of fetal growth and development as stated previously in this chapter [13, 14, 19, 33, 35].

Role of Yoga in Ayurveda Therapy of the Mother during the First and Second Trimester of Pregnancy

Yogas in the First Trimester of Pregnancy

Like acupuncture, acupressure and in traditional Chinese medicine, yoga in ayurvedic medicine during pregnancy in the first and second trimester plays an important role in helping the pregnant mother to stay active and energetic during these periods of fetal development [36].

Some of the most practised first trimester asanas are Ardha Titali asana or half butterfly as it is known, which is an excellent yoga that can help in loosening of the hip and knee joints thus enabling the mother to have a safer and a faster natural delivery. Poorna Titali asana or full butterfly plays an important role in relieving pain and tiredness from the legs. Supta Udarakarshan asana or sleeping abdominal stretch pose relieves stiffness from the spine and overcomes the problems of constipation. Tadasana and Utthanasana are also good for spinal loosening, clearing any nerve congestion, strengthening the muscles of the uterus, thighs and ankles. Kati Chakrasana is good for toning the waist, hip and back muscles whereas Kashta Takshana asana loosens the pelvic muscles. Chakki Chalan asana is good for regulating the menstrual cycle and toning of the nerves, pelvis and abdominal region. Marjariasana is known to induce flexibility of the neck, shoulders and spine area apart from helping in toning the female reproductive system [36].

Yogas in the Second Trimester

Some of the most practiced yogas or the asanas as it is known in Ayurveda system of therapy during the second trimester are listed below. Matsya Kridasana which helps in re-distributing the excess weight around the waistline helps in digestion, easing constipation and relaxing the nerves of the legs. It also helps in relaxing and regulating the sleeping patterns of the pregnant mother, who remains stress free during the second trimester. Vajrasana and Bhadrasana are good for digestion and relief from acidity. Meru Akarshanasan is good for the thigh and abdominal muscles and their strengthening whereas Marjarisana improves the neck, spine and shoulder muscle flexibility. Hasta Utthanasan or hand raising pose helps in reducing muscle spasms of the neck, shoulders and upper back and is extremely good for cardiac and blood circulation and helps in more oxygen supply to the brain. However certain asanas or yogas like forward/backward bending, Sarvangasana, Matsyasana, Sasankasana, Vakrasana and Ardhamatsyendrasana should be avoided by the pregnant mother during the time of pregnancy [36].

It is also documented that different massages with ayurvedic oils like head, breast, foot and body massages can help in relieving back pressure, pains, reducing stretch mark and can improve the overall quality of life during pregnancy [37].

Safety of Ayurvedic Medicine: Do We Have Enough Conclusive Evidence?

Six cases of lead poisoning was reported by the New York City Department during the year 2011–2012 among six cases of foreign born pregnant women with the use of ten oral ayurvedic drugs made in India. The amount of lead in the blood levels of these six pregnant mothers ranged from 16 to 64 μ g/dL and also after investigation it was found that most of the medications had a lead concentration of 2.4 % as well as toxic levels of mercury and arsenic which could have caused an adverse health event [38]. Anesthesiologists (ASA) have taken a conservative stance and recommended that it is essential to stop using herbal or Ayurvedic medicine at 2-3 weeks prior to anesthesia and surgery [39, 40]. Also, an American study published that in 2004, 70 Ayurvedic medicines purchased over the counter which were all manufactured in Asia contained lead, mercury and arsenic at an alarming level. In 2004, CDC reported 12 reports of lead poisoning due to herbal medicines.

List of Ayurvedic Herbs That Can Be Used during the First Two Trimesters of Fetal Growth and Development

To reduce vomiting and morning sickness among pregnant women in the first trimester Vilvaadhi laham can be prescribed at a dose of 1–2 g for

five to six times till it dissolves in the mouth. Also drinking a lot of natural fluids like coconut water is extremely beneficial especially when the mother is experiencing stress and tiredness resulting from continuous vomiting and nausea. Also Garbha raksheena guliga or maha-dhanwanthira can be prescribed to the pregnant mother with jeera water (cumin seed) in the morning and evening once. In the second trimester 15 ml of Garbharaksha kashayam boiled in 60 ml of water and then cooled can be prescribed to the pregnant mother in the second trimester of pregnancy. Also one garbharaksheena-guliga can be added and this whole mixture can be taken 30 min before breakfast and if needed then in the evening. Also two spoons of 10 ml of dhadhi maathi hritham before lunch is recommended. After bathing everyday application of dhanwanthiram kuzhambu or thailam is beneficial to the pregnant mother during this trimester [41].

Garlic, an important herb used for cooking purposes across the world helps in boosting the immune system and reducing the chances of ischemic induced neuronal injuries [42]. It is regarded safe by USFDA and has less toxicities as observed in post operative cases of bleeding especially in case of urethral resection of the prostate gland.

Aswaganda is a sympathomimetic drug used by athletes to boost their energy levels. Adverse events like hypertension, tachycardia, hypoglycemia, insomnia, nervousness and skin rashes are observed. Turmeric or haldi works as an anti septic, anti-inflammatory and antibacterial substance, but can cause very rare iron deficiency. Tulsi is used mainly for treating respiratory syndromes, cough and cold, flu, diarrhea, malaria and cardiac diseases. Currently no known side effects with tulsi are reported as of yet. Amla is a rich source of Vitamin C which is 30 times the amount found in oranges. It helps in building up the immunity of the body, acts as an anti oxidant and free radical scavenger and preventing age related disorders [43]. Also no side effects are reported with Amla yet. Guggul helps in the treatment of heat diseases as well as a good anti oxidant, anti-obesity and hypolipidemic drug [44]. It can also help in reducing C-reactive proteins in the body thus reducing fat. Giloe acts as a cardiotonic, expectorant, analgesics, anti pyretic, anti diabetic and anti inflammatory drug [44, 45]. According to WHO 75 % of the world's population use herbal remedies for basic health care.

Pharmacovigilance of Ayurvedic Drugs, a General Perspective

The word drug has come from the ancient word "Root" and by definition herbs are drugs [39]. Herbal tea is not a problem but some traditional medicinal herbs are the problem.

Classical Ayurveda classifies metals and minerals as bhasmas which can be given in combination with plants as herbomineral formulations. Manufacturing protocols for these type of medicines are extremely stringent and can only cause adverse event if in case they are not properly manufactured [47, 48].

Ayurved in India is regarded as a formal medicine along with allopathy. It is an old system of medicine that was incorporated into the Indian medical education system over a century ago. Now the country has 196 undergraduate colleges and 55 postgraduate centres where Ayurveda is taught. Currently 438,721 [46] licensed and registered clinicians in Ayurvedic medicine are present in the country. Manufacture and marketing of Ayurvedic drugs in the country is controlled by the Drugs and Cosmetics Act of 1940 [47].

In general two types of Ayurvedic drugs are available in the market/globally. The first type of drugs is part of the classical Ayurvedic formulations which follow the descriptions of Ayurved Samhita during its preparation and manufacture. The other is derived through patents and proprietary formulations made from herbal extracts [49]. There are currently 8403 pharmacies in India dispensing Ayurved drugs thus making India the largest consumer and exporter of Ayurved drugs in the world [47]. The Ayurvedic formulations and productions form one third of the total pharmaceutical business in the country and helps the nation in earning revenue of 4000 crores Indian rupees every year. However despite having such a large use of Ayurveda drug in India the total number of adverse events reported is negligible when compared to that of alternative medicines and allopathy [47]. This is the same trend observed in other parts of the world resulting in the lack of side effect monitoring and reporting. Certain reasons due to this lack of Pharmacovigilance awareness in Ayurved system of medicine can be attributed to different factors like the safety methods that are used in this field of medicine have not evolved adequately and scientifically over time [47]. This can be further related to the fact that in none of the text books of Ayurveda, how to monitor and evaluate ayurvedic drugs has been mentioned. One example can be given in support of the above fact. Urmila et al. [46] interviewed a group of 80 Vaidyas or Ayurvedic physicians in India and found that 14 of the physicians refused to accept that there is any side effect associated with Ayurved and out of the remaining 66 clinicians, 48 responded by saying that they experienced some kind of adverse events in patients after Ayurved drugs were prescribed to patients. Out of these 48 clinicians only 14 of them reported the adverse events to the National Pharmacovigilance Centre in India thereby showcasing the fact that many of the trained Vaidyas actually lack the knowledge on safety aspects and perspectives related to these kinds of medicines. Also the increase of counterfeit drugs in the market and lack of quality manufacturers and manufacturing are questioning the safety of Ayurvedic drugs like never before. Understanding the pharmacokinetics and toxicokinetics of Ayurved drugs is also an area which has not been explored properly till date as most of the drugs are multi-ingredient in composition. Individualization of Ayurvedic medicine like Homeopathic medicine has also aggravated the lack of reporting of adverse events among patients [47].

Conclusion

Ayurvedic samhita and modern science = share almost the same principles of fetal growth and development. Many similarities between the two are observed like at the second month, the shape of the fetus decides its sex. However scientific principles behind this

mechanism are still confounding. More than 5000 years back, Ayurvedic physicians wrote books and made conclusions according to their observations and knowledge without any scientific tools to support their ideas. Modern science today has all diagnostic laboratory equipments which were not available at that time, so the contribution of the Ayurvedic samhita is noticeable and paramount to any field of medicine as it can be opined that many of the basic and conceptual contexts of western or modern medicine that evolved at a much later stage are based on different traditional medicines in which Ayurvedic medicine has a very prominent position.

References

- Pal SK. Complementary and alternative medicine: an overview. Curr Sci. 2002;82(5):518–24. by D.P. Agarwal.
- 2. Ayurveda en.wikipedia.org/wiki/Ayurveda.
- Sharma RK, Bhagwan Dash. vol 1 sutrastha. Charak samhita editor – 30:21 pg 598 ed. 2002.
- Sharma RK, Bhagwan Dash. Charak samhita editor vol 1 sutrasthan 30:26 pg 600 ed. 2002.
- Sharma RK, Bhagwan Dash. Charak samhita editor vol 1 sutrasthan 20:9, 10 pg 362 ed. 2002.
- Sharma RK, Bhagwan Dash. Charak samhita editor vol 1 sutrasthan 2nd chapter pg 64 ed. 2002.
- Sharma RK, Bhagwan Dash. Charak samhita editor vol 2 sarirasthan 4:5, 6 pg 388 ed. 2009.
- Deepa Roy Choudhary. Ayurveda and holistic health – Ayurveda on how a baby gets bodily features during pregnancy. 4 Mar 2010. ayurvedahealthblog. blogspot.in.
- Embryology (version 2). http://connectpro91461020adobeconnect.com.
- 10. Srikanta Murty. Astanga sanghra editor Sarirasthan 2:11, 12 pg no 21 ed. 2005.
- Joshi YGV. Fetal growth and development and management of pregnancy. Charak Samhita n.d.; 1 (2003 1st ed.):1–20.
- Charak Sharirsthana YGJ. Charak Samhita n.d.; 4 (2003 1st ed.):5–11.
- Journal R. Is the use of Ayurvedic products safe in Pregnancy? 1970, viewed 11 Apr 2015. http://www. researchgate.net/post/is_the_use of_Ayurvedic_ Products_safe_in_Pregnancy1970; volume (11 Apr 2015):1–1.
- Vedic M. Maharishi Vedic approach to health 1970, viewed 11 Apr 2015. http://en.wikipedia.org/wiki/ Maharishi_vedic_Approach_to_Health 1970; volume (11 Apr 2015).

- Article J, Article P. An Ayurvedic Perspective of low birth weight – a conceptual study 1970 viewed 11 Apr 2015. www.jahm.in/index.php/JAHM/article/view/146. Shad Garbhakara Bhavas vis 4368, viewed 11 Apr 2015. www.ncbi.nlm.nih.gov/pmc/articles/PMC3215361, 1970; volume (Apr 2015).
- Sushrut A. Fetal growth and development and pregnancy management month by month. Sushrut Sharirsthana 3–18, 33, n.d; 18–33.
- Cunningham FG, Leveno KJ, Bloom SL. Fetal growth and development. In: Cunningham F, editor. Williams obstetrics. 23rd ed. New York: McGraw – Hill; 2010. p. 4–4.
- 18. Science in Hinduism: embryology in Garbhopanishad and Charaka Samhita.
- Articles N.work 8494, work 8494, viewed 11 Apr 2015. www.ncbi.nlm.nih.gov/pmc/articles/PMC2699761, n.d; volume (11 Apr 2015).
- Ayurveda. Pregnancy care/stress free pregnancy and delivary-1970, viewed 11 Apr 2015. www.prokerala. com/health/ayurveda/ayurveda-pregnancy-care.htm 1970; volume (11 Apr 2015).
- Clinic A. Ayurveda and care 5292, www.ayurclinic. com.au/treatment/ayurveda. and pregnancy-care n.d.; volume (11 Apr 2015).
- Brewer TH. Human pregnancy nutrition: an examination of traditional assumptions. Aust N Z J Obstet Gynaecol. 1970;10(2):87–92.
- Rodwell SW. Nutrition and diet therapy. Nutrition and diet therapy, 2nd ed. St. Louis: Mosby; 1973, Chapter 17. 1973;17(1973):17–17.
- Journal Y. Prenatal Yoga Poses 1970, www.yogajournal.com/article/practice section/yoga – for – moms – to – be 1970; 1(11 Apr 2015).
- Lagercrantz H TAS. The stress of being born. The 'Stress' of being born scientific American (Apr 1986) 100–107, 1986; volume (Apr 1986):100–107.
- Roy P, Harold K, Michael N. Maternal nutrition: a selective review of clinical topics. Obstet Gynecol. 1972;40:773–85.
- 27. Primrose T, Higgins A. A study in human antepartum nutrition. J Reprod Med. 1971;7:257–64.
- MGR, MGE, D.N. Placental and fetal physiology. Placental and fetal physiology. In: Richard Harding & Alan D. Bocking, fetal growth and development, Cambridge: Cambridge University Press; 2001 volume.
- 29. Rucker MP. The behavior of the uterus in eclampsia. Am J Obstet Gynecol. 1921;2:179–83.
- By R. Fetal growth and development .Web MD medical reference reviewed by Dr. Rob Hicks on 03 Sept 2013. Volume (03 Sept 2013).
- 31. James C. Trans Am Gynecol Soc. 1887;1887(11): 399–418.
- Gleichert JE. Etienne Joseph Jacquemin, discoverer of Chadwick's sign. J Hist Med Allied Sci. 1971;26(1):75–80.
- 33. Gabbe S, Niebyl J, Simpson J. Normal and problem pregnancies. 6th ed. Philadelphia: Saunders Elseviers; 2012. chap 2. eds.Obstetrics:normal & problem pregnancies.6 th ed.Gabbe, Niebyl and Simpson 1.

- Hytten F. The physiology of human pregnancy. 2nd ed. Oxford: Blackwell Scientific Publications; 1970. p. 2–2.
- Gina K. The baby doctors: probing the limits of fetal medicine. New York: Delcorte press; 1990.
- 36. Yogic Interventions, Asanas, Pranayama and Meditation for safe delivery and pregnancy. http:// w w w . g y a n u n l i m i t e d . c o m / h e a l t h / yogic-interventions-asanas-pranayama-and-meditation-for-safe-delivery-and-pregnancy/901.
- 37. Precautions and benefits of massage in pregnancy. http://easyayurveda.com/2011/01/18/ massage-in-pregnancy-15-benefits-and-precautions/.
- 38. Paromita hore, Munerah Ahmed, Jacqueline Ehrlich, Celia Ng, Lourdes Steffen, Slavenka Sedlar, Phyllis Curry-Johnson, Nathan Graber, Deborah Nagin, Nancy Clark, New York City Department of Health and Hygiene, Robert Saper, Marissa Scalia Sucosky, Corresponding author: Paromita Hore PhD. Lead Poisoning in Pregnant Women Who Used Ayurvedic Medications from India – New York City, 2011–2012. Weekly. 24 Aug 2012/61(33); 641–646.
- Pradhan SL, Pradhan PS. Ayurvedic medicine and anaesthesia. Indian J Anaesth. 2011;55:334–9.
- Chang B, Hung CT, Chiu W. Herbal medicine and anesthesia. Hong Kong Med J HKMJ. 2002;8:123– 30. Pubmed.
- Ayurvedic medicines used in pregnancy. http://www. uni5.co/index.php/en/herbs-in-pregnancy/264-herbalayurvedic-medicines-in-pregnancy.html.
- 42. Aguilera P, Chanez-Cardena ME, Ortiz-Plata A, Leon-Apaticio D, Burrera D, Espinoza-Rojo M, et al. Aged garlic delay the appearance of infarct area in a cerebral ischemia model, an effect likely conditioned by the cellular antioxidant system. Phytomedicine. 2010;17(3–4):241–7. Pub Med.
- 43. Yokozawa T, Kim HY, Kim HJ, Tanaka J, Sugino H, Okubo T, et al. AMLA (Emblica officinali Gaertn) attenuates age related renal dysfunction by oxidative stress. J Agric Food Chem. 2007;55:7744–52. Pub Med.
- Nityananda S, Srivastava JS, Asthana OP. Clinical trials with gugulipid: a new hypolipidaemic agent. J Assoc Phys India. 1989;37:323–8. Pub Med.
- Prince PS, Menon VP. Antioxidant activity of Tinospora corlifolia roots in experimental diabetes. J Ethmopharmacol. 1999;65:277081. Pub Med.
- Panchabhai TS, Kulkarni UP, Rege NN. Validation of therapeutic claims of Tinospora Cordifolia: a review. Phyto Ther Res. 2008;22:424–41.
- Thattee U, Bhalerao S. Pharmacovigilance of Ayurvedic medicines in India. Indian J Pharmacol. 2008;40(Suppl no.1):S10–2. Pubmed.
- Shastri PK, editor. 2nd Adhyaya, Rasantarangini. 11th ed. New Delhi: Sri Jainendra Press; 1994. p. 22–4.
- Dahanukar SA, Thattle UM. Can we prescribe Ayurvedic drugs rationally? Indian Pract. 1998;51:882–6.

Part XVII

Miscellaneous

Problems of Prematurity

45

Annie Abraham and C.S. Rejiya

Definition of Preterm or Premature

The average length of a normal pregnancy is 40 weeks (280 days) from the date of conception. Infants born before 37 weeks gestation are considered premature and may be at risk for complications.

Although, the rate of premature birth appears to vary by geographic region, the reported incidence varies between 6 and 10 %. Despite significant improvements in perinatal care, there has not been a concomitant reduction in the rate of premature births in developed countries. More than one out of every ten infants born in the United States is born prematurely. Advances in medical technology have made it possible for infants born as young as 23 weeks gestational age (17 weeks premature) to survive. These premature infants, however, are at higher risk for death or serious complications, which include heart defects, respiratory problems, blindness, and brain damage.

C.S. Rejiya, PhD

Causes and Symptoms

The birth of a premature baby can be brought on by several different factors, including the following:

- Premature labor
- Placental abruption, in which the placenta detaches from the uterus
- Placenta previa, in which the placenta grows too low in the uterus
- Premature rupture of membranes, in which the amniotic sac is torn, causing the amniotic fluid to leak out
- Incompetent cervix, in which the cervix opens too soon
- Maternal toxemia or preeclampsia

Prematurity is much more common in pregnancy of multiples and for mothers who have a history of miscarriages or prior premature birth. Another identifiable cause of prematurity is drug abuse (e.g. cocaine) by the mother. Infants born prematurely may experience major complications due to their low birth weight and the immaturity of their organ systems. Some of the common problems among premature infants are **jaundice** (yellow discoloration of the skin and whites of the eyes), apnea (a long pause in breathing), and inability to breast or bottle feed. Body temperature, blood pressure, and heart rate may be difficult to regulate in premature infants. The

A. Abraham, PhD (🖂)

Department of Biochemistry, University of Kerala, Kariavattom PO, Thiruvananthapuram, Kerala, India e-mail: annieab2001@gmail.com

Department of Biochemistry, Sree Ayyappa College (TDB), Eramallikkara, Kerala, India

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_45

lungs, digestive system, and nervous system (including the brain) are underdeveloped in premature babies and are particularly vulnerable to complications.

Short-Term Complications

In the first weeks, the complications of premature birth may include:

Breathing Problems

The primary function of the lung is gas exchange (i.e., they inhale oxygen and exhale carbon dioxide). Fetal breathing movements begin as early as 10 weeks of gestation, and the breathing of amniotic fluid in and out is essential for the stimulation of lung development. Fetal breathing movements tend to be erratic and occur only 30–40 % of the time up to 30 weeks of gestation. The failure of fetal breathing movements or a lack of amniotic fluid that can be breathed in and out results in underdeveloped lungs (i.e., pulmonary hypoplasia), which can be incompatible with extrauterine life. By approximately 30-32 weeks of gestation, the lungs make surfactant, a soap like substance that helps keep the air sacs (alveoli) open. Infants born before 28-30 weeks gestation lack alveoli and breath with their terminal bronchioles and primitive air sacs. After delivery, the breathing pattern generally becomes more regular and continuous, but immature regulatory systems can lead to brief episodes of not breathing.

Respiratory Distress Syndrome (RDS) is the most common problem in premature infants. Babies born too soon have immature lungs that have not developed surfactant, a protective film that helps air sacs in the lungs to stay open. With RDS, breathing is rapid and the center of the chest and rib cage pull inward with each breath. Extra oxygen can be supplied to the infant through tubes that fit into the nostrils of the nose or by placing the baby under an oxygen hood. In more serious cases, the baby may have to have a breathing tube inserted and receive air from a respirator or ventilator. A surfactant drug can be given in some cases. Extra oxygen may be needed for a few days or weeks. This condition primarily affects infants born before 35 weeks.

The chronic lung disease (CLD) that sometimes follows RDS in preterm infants born between 23 and 32 weeks, is also called bronchopulmonary dysplasia (BPD). BPD/CLD is a chronic disorder that results from inflammation, injury, and scarring of the airways and the alveoli. It is associated with growth, health, and neurodevelopmental problems during childhood. Positive-pressure ventilation, high oxygen concentrations, infection, and other inflammatory triggers all contribute to lung injury; but the primary cause of BPD/CLD is lung immaturity. Especially for infants born at less than 28–30 weeks of gestation, the lung tissue is very fragile and the injured lung tissue tends to trap air, collapse, or fill with mucus and other fluids, which further compromise lung growth and development.

Another complication of preterm birth is apnea, in which infants may stop breathing for 20 s or more, sometimes accompanied by a slow heart rate (bradycardia). Immaturity of the control of breathing is the major cause of apnea and bradycardia, although sometimes preterm infants have obstructive apnea (an obstruction to the movement of air in their airways). They require constant monitoring but generally respond quickly to stimulation (or in the case of obstructive apnea, repositioning). They may occasionally need to be given some positive-pressure breaths to get them breathing again. There is no agreement as to what constitutes pathologic apnea or the threshold of apnea that requires treatment.

Heart Problems

Preterm infants can experience a variety of cardiovascular disorders, ranging from major morphological defects to dysfunctional autoregulation of blood vessels (hypotension). By embryonic day 20, the cells that will form the heart begin to differentiate. The primitive heart beats by 4 weeks of gestation and is fully formed at the end of the sixth week. Because gas exchange occurs in the placenta, most of the fetal blood flow bypasses the lungs through the ductus arteriosus.

The most common heart problems premature babies experience are **Patent Ductus Arteriosus** (**PDA**) and low blood pressure (hypotension). PDA, which tends to affect babies born before 30 weeks, is a persistent opening between two major blood vessels leading from the heart. While this heart defect often closes on its own, left untreated it can cause too much blood to flow through the heart and cause heart failure as well as other complications. Low blood pressure may require adjustments in intravenous fluids, medicines, and sometimes blood transfusions.

Brain Problems

Babies born before 28 weeks are at risk of bleeding in the brain, known as an Intra Ventricular **Hemorrhage** (IVH). IVH is another serious complication of prematurity. It is a condition in which immature and fragile blood vessels within the brain burst and bleed into the hollow chambers (ventricles) normally reserved for cerebrospinal fluid and into the tissue surrounding them. Physicians grade the severity of IVH according to a scale of I through IV, with I being bleeding confined to a small area around the burst vessels and IV being an extensive collection of blood in the ventricles and in the brain tissue itself. Grades I and II are not uncommon, and the baby's body usually reabsorbs the blood with no ill effects. However, more severe IVH can result in hydrocephalus, a potentially fatal condition in which too much fluid collects in the ventricles, exerting increased pressure on the brain and causing the baby's head to expand abnormally. To drain fluid and relieve pressure on the brain, doctors either perform lumbar punctures, a procedure in which a needle is inserted into the spinal canal to drain fluid; install a reservoir, a tube that drains fluid from a ventricle and into an artificial chamber under or on top of the scalp; or install a ventricular shunt, a tube that drains fluid from the

ventricles and into the abdomen, where it is reabsorbed by the body. Infants who are at high risk for IVH usually have an ultrasound taken of their brain in the first week after birth, followed by others if bleeding is detected. IVH cannot be prevented; however, close monitoring can ensure that procedures to reduce fluid in the brain are implemented quickly to minimize possible damage. Most hemorrhages are mild and resolve with little short-term impact. But some babies may have larger brain bleeding which causes permanent brain injury. Larger brain bleeds may lead to fluid accumulation in the brain (hydrocephalus) over a number of weeks. Some babies who develop hydrocephalus will require an operation to relieve the fluid accumulation.

Temperature Control Problems

Premature infants are not able to regulate their body temperatures as well as term infants. Factors that contribute to this problem are immaturity of the hypothalamic regulatory center, lack of subcutaneous fat (insulation shield), lack of brown fat (allows for thermogenesis in adverse climatic environmental conditions) and a relatively large body surface area to body mass ratio.

Typically, temperature control is maintained by providing an external heat source (radiant warmers or incubators/isolettes). These sources are set to provide heat to maintain a neutral thermal environment. In this environment, the infant has minimal energy expenditure to maintain core body temperature.

The majority of premature infants are able to regulate their body temperatures by a post conceptional age of 34 weeks. However, they remain at continued risk for poor thermal regulation at the extremes of environmental temperature. Parents should be appropriately counseled and encouraged to avoid subjecting their infant to temperature extremes.

Premature babies can lose body heat rapidly; they don't have the stored body fat of a full-term infant and they can't generate enough heat to counteract what's lost through the surface of their bodies. If body temperature dips too low, hypothermia can result. Hypothermia in a preemie can lead to breathing problems and low blood sugar levels. In addition, a preemie may use up all of the energy gained from feedings just to stay warm, not to grow bigger. That's why smaller preemies require additional heat from a warmer or an incubator until they're larger and able to maintain their temperature without assistance. After you bring your baby home, you won't need to keep your house or apartment any warmer than you typically would.

Gastrointestinal Problems

The gastrointestinal (GI) tract digests and absorbs food, but it also has immune and endocrine functions and receives a good deal of input from the nervous system. It begins to form as early as the fourth week of gestation, and the stomach and the intestines are fully formed by 20 weeks of gestation. The intestines double in length in the last 15 weeks of gestation (to 275 cm at term). The intestinal absorptive cells form as early as 9 weeks of gestation, and endocrine and immune functions also begin early. Taste buds form at between 7 and 12 weeks of gestation. However, preterm infants have difficulty with digesting nutrients because many specialized cells are not fully functional.

The earliest coordinated reflexes are related to stimulation around the mouth, with mouth opening in response to perioral stimulation occurring at 9.5 weeks of gestation and head turning occurring by 11.5 weeks of gestation. The fetus swallows by 10–12 weeks of gestation and can suck by 20 weeks of gestation. After birth, the newborn's GI tract becomes colonized with bacteria, which aids with food digestion. Antibiotics alter this process.

Feeding intolerance is a common complication of preterm birth. The immature GI tract has difficulty digesting food necessary for ongoing growth and development. Very immature and sick infants receive parenteral (intravenous) nutrition with amino acids, glucose, electrolytes, and lipids. Preterm infants below 34–35 weeks of postmenstrual age require tube feeding because they cannot coordinate sucking, swallowing, or breathing. Providing the preterm infant with sufficient nutritional requirements for growth and development can complicate the treatment of other conditions.

Necrotizing Entero Colitis (NEC) is another complication of prematurity. In this condition, part of the baby's intestine is destroyed as a result of bacterial infection. In cases where only the innermost lining of the bowel dies, the infant's body can regenerate it over time; however, if the full thickness of a portion dies, it must be removed surgically and an opening (ostomy) must be made for the passage of wastes until the infant is healthy enough for the remaining ends to be sewn together. Because NEC is potentially fatal, doctors are quick to respond to its symptoms, which include lethargy, vomitting, a swollen and/or red abdomen, fever, and blood in the stool. Measures include taking the infant off mouth feedings and feeding him or her intravenously, administering antibiotics, and removing air and fluids from the digestive tract via a nasal tube. Approximately 70 % of NEC cases can be successfully treated without surgery. Premature babies who receive only breast milk have a much lower risk of developing NEC.

Gastro Esophageal Reflux (GER) is common in preterm and full-term infants, often presents as regurgitation, and may adversely affect growth and health. It may also be manifested by aspiration pneumonia, wheezing, or worsening of BPD/ CLD because of an inability to protect the airway when refluxing. The presence of a nasogastric feeding tube increases the likelihood of reflux. Severe GER with aspiration of the stomach contents into the lungs is life threatening. GER is often treated with medications, including H2 blockers or protein pump inhibitors, which neutralize gastric acidity (and which may increase vulnerability to infection via the GI tract), and prokinetic compounds, which increase GI motility.

Blood Problems

Hematopoiesis is the generation of blood cells from stem cell progenitors. It begins in the embryo 7 days after conception. Stem cells are active in the aortogonadomesonephron at 10 days and then shift to the liver and, finally, the bone marrow. There are developmental changes in the numbers and functions of hematopoietic stem cells and in the various differentiated blood cells (e.g., red blood cells, white blood cells, and platelets). Red blood cells in the fetus contain fetal hemoglobin, which is necessary for intrauterine gas exchange because it has a higher affinity for oxygen. Fetal hemoglobin levels decrease after birth.

Preemies are at risk of blood problems such as anemia and infant jaundice. Fetal blood loss, fetomaternal hemorrhage, and hemolysis can all result in congenital anemia, but the most common hematologic complication in preterm infants is anemia of prematurity. Anemia of prematurity is an exaggeration of the physiological anemia of infancy because of suppressed hematopoiesis for 6-12 weeks after birth and is earlier in onset and symptomatic. Its causes are multifactorial and include blood loss from frequent blood sampling, the shorter survival of red blood cells in preterm infants, a suboptimal response to anemia, and a greater need for red blood cells with growth. Preterm infants often need red blood cell transfusions, and many of the sickest and most immature infants need multiple transfusions.

Infant jaundice is a yellow discoloration in a newborn baby's skin and eyes that occurs because the baby's blood contains an excess of a yellowcolored pigment of red blood cells (bilirubin). Infant jaundice is common in babies born before 38 weeks.

Metabolic Problems

Premature babies often have problems with their metabolism. Some preemies may develop an abnormally low level of blood sugar (hypoglycemia). This can happen because preemies typically have smaller stores of glycogen (stored glucose) than do full-term babies and because preemies' immature livers have trouble producing glucose.

Problems of Immune System

Premature infants are at higher risk for infections. This risk is multifactorial. The primary source of immunity for the neonate is passively derived antibodies from the mother and this tends to occur primarily in the third trimester. Thus, the relative amount of antibody transferred is affected by the duration of gestation. Additionally, a significant proportion of premature infants who are hospitalized in intensive care units, require interventions such as IV therapy, and placement of central vascular catheters for providing nutrition, and invasive monitoring. All of these factors contribute to the increased risk of infections in this population. Premature infants present with nonspecific signs and symptoms of infection. This mandates close monitoring for infectious complications, both during hospitalization, in the immediate neonatal period, and in subsequent months during the first year of life.

Given their propensity for infections, the American Academy of Pediatrics recommends that all childhood immunizations be administered to premature infants at the appropriate chronological age. The only exception to this rule is the hepatitis B immunization, which should be initiated only after the infant's weight exceeds 2 kg. Despite lower titers of antibody response in these infants, there is no recommendation for additional doses of specific immunizations.

Passive prophylaxis for Respiratory Syncytial Virus (RSV) infection is currently recommended during the cooler winter months for certain premature infants at highest risk for serious complications from RSV. These guidelines are evolving. These infants will also benefit from receiving influenza immunization at 6 months chronological age during the cooler winter months (3).

An underdeveloped immune system, common in premature babies, can lead to infection. Infection in a premature baby can quickly spread to the bloodstream causing sepsis, a life-threatening complication. As a result, when a preemie's condition is getting worse, baby's doctor might check for an infection—even if there's no fever. Often, in such situations, baby may be given antibiotics until it's apparent that there's no infection.

Long-Term Complications

In the long term, premature birth may lead to these complications:

- **Cerebral palsy**. Cerebral palsy is a disorder of movement, muscle tone or posture that is caused by injury to a preemie's developing brain either during pregnancy or while the baby is still young and immature. Brain injury from poor circulation, insufficient oxygen supply, undernourishment or infection can all lead to cerebral palsy or other neurological problems.
- **Impaired cognitive skills**. Premature babies are more likely to lag behind their full-term counterparts on various developmental milestones. Upon school age, a child who was born prematurely might be more likely to have learning disabilities.
- Vision problems. Preemies born before 30 weeks may develop retinopathy of prematurity (ROP), a disease that occurs when blood vessels swell and overgrow in the lightsensitive layer of nerves at the back of the eye (retina). Sometimes the abnormal retinal vessels gradually scar the retina, pulling it out of position. When the retina is pulled away from the back of the eye it's called retinal detachment, a condition that, if undetected, can impair vision and cause blindness.

Preterm infants are more likely than term infants to have significant abnormalities of all parts of the visual system, leading to reduced vision. The optic vesicles that will become the eyes form during the fifth and sixth weeks after conception. The eyeball is well formed by the lower limit of viability (22-25 weeks gestation). However, a pupillary membrane covers the anterior vascular capsule of the lens and gradually disappears between 27 and 34 weeks of gestation. The retina is a vascular layer in the back of the eye that translates light into electrical messages to the brain. The retina is the one of the last organs to be vascularized in the fetus. Blood vessel-forming cells originate near the optic disc (where the optic nerve enters the retina) from spindle cell precursors at 16 weeks of gestation and gradually spread across the surface of the retina, from the center to the periphery. Vessels cover only 70 % of the retina by 27 weeks of gestation, but in most cases the retina is completely vascularized to the nasal side by 36 weeks of gestation and to the temporal side by 40 weeks of gestation.

The visual system functions very early, with the preterm infant blinking in response to bright light by 23–25 weeks of gestation and with papillary constriction in response to light by 29–30 of weeks gestation. By 30–32 weeks of postmenstrual age, the preterm infant begins to differentiate visual patterns. Visual acuity progressively improves with increasing postmenstrual age. The full-term neonate sees shapes (approximate visual acuity of 20/150) and colors and has a fixed focal length of 8 in. (anything closer or farther away becomes more blurry).

Hearing problems. Premature babies are at increased risk of some degree of hearing loss. All babies will have their hearing checked before going home. The ear begins to develop at the end of 6 weeks of gestation and is fully developed by 20 weeks of gestation. A response to sound can be demonstrated in fetuses and infants born at 23 and 24 weeks of gestation, and auditory brainstem-evoked responses can be recorded this early in preterm infants. The shape of the waveform changes and the conduction time decreases with increasing gestational or postmenstrual age.

One to two of 1,000 newborns suffer from congenital or perinatally acquired hearing disorders. The prevalence of neonatal hearing disorders has been reported to be increased 10- to 50-fold in infants at risk, which includes preterm infants. In addition to hearing impairment as a result of heredity, which is the cause of the largest percentage of hearing disorders, a number of in utero and neonatal complications (e.g., infections, immaturity, asphyxia, ototoxic medications, and hyperbilirubinemia) have been described to be risk factors for neonatal hearing disorders. Ventilated infants are at increased risk for otitis media.

Significant hearing impairment, often requiring hearing aids, occurs in 1-5 % of infants born at gestational ages of less than 25 or 26 weeks. Moderate to severe bilateral hearing impairment can distort the developing child's perception of speech and may interfere with his or her attempt at speech production. If the impairment remains undetected hearing through the critical period of language acquisition, that is, within the first 2 years of life, a profound impairment of receptive and expressive speech and language development can result. Early detection of hearing impairment facilitates early remediation (e.g., hearing aids or cochlear implants) and early intervention for speech and language acquisition. The prognosis for functional speech and language skills improves with the early detection and treatment of hearing impairment.

- **Dental problems**. Preemies who have been critically ill are at increased risk of developing dental problems, such as delayed tooth eruption, tooth discoloration and improperly aligned teeth.
- Behavioral and psychological problems. Children who experienced premature birth are more likely than full-term infants to have certain behavioral and psychological problems, such as attention deficit hyperactivity disorder, depression or generalized anxiety, and difficulties interacting with kids their own age.
- Chronic health issues. Premature babies are more likely to have chronic health issues some of which may require hospital care than are full-term infants. Infections, asthma

and feeding problems are more likely to develop or persist. Premature infants are also at increased risk of Sudden Infant Death Syndrome (SIDS).

For some premature babies, difficulties may not appear until later in childhood or even adulthood. Not performing well in school is often a prime concern. Some studies suggest that premature babies may face an increased risk of type 2 diabetes and cardiovascular disease in adulthood.

The premature infant is ready for discharge when he/she is able to fulfill the following criteria: (1) ability to appropriately regulate their temperature without the need for technological support, (2) ability to ingest adequate calories to achieve consistent growth, and (3) to have demonstrated other parameters of global physiologic stability (the absence of clinically significant apnea, bradycardia, or hypoxemia). In addition, and most importantly, it is critical that the parents/caregivers feel comfortable with the care of the infant in the home environment. One of the issues that may alleviate some of the parental anxiety is training in infant CPR. Thus, the process of discharge of the infant is a continuum that begins several days to weeks prior to the actual discharge of the infant. Many of these infants will have additional needs and it is important that all of these needs and appropriate community resources are identified prior to discharge. At the time of discharge, the routine mandated screening for hearing and metabolic diseases should be completed with the results forwarded to the primary care physician.

Problems of Extreme Prematurity

Prasanta Choudhuri, Dhritidipa Chaudhuri, and Niranjan Bhattacharya

What Is Extreme Prematurity: Definition Incidence and Trend Consequences

- Embryology which organ is at which stage of development and maturation at that moment both structurally and functionally?
- What is likely to be the problem with that state of specific development of that organ and that organ system?
- How will it affect the structural and functional development of growth and mayuration? Would there be any support required?
- What is the effect of the non mature or delay in maturation on the ultimate organ structure and function, and the impact on the individual immediately after birth, and morbidity and

mortality due to this both in the immediate and in the long term?

- Is there any disease which might be predisposed by this defective and/or incomplete and/or late organ maturation both structurally and functionally?
- Can it be prevented? If yes, how?
- Are there any advantages of extreme prematurity in the immediate or later adaptation to extrauterine life and final adaptation to the environment? If yes, what?

On 21st February 2007, Susan Donaldson James of ABC News presented an interesting perspective on premature babies and their issues of survival: Life at 21 weeks: Immature Lungs and a Handful of Fragile Skin and Pain – Last year, one of every eight babies born in the United States was premature, a number that is growing because of an increase the use of in vitro fertilization and other reproductive advances.

But the medical triumphs that allow a tiny 10-ounce baby to survive are often fraught with trauma and tragedy.

Consider this: At 25 weeks gestation or younger, a tiny baby can fit in the palm of a hand. Its skin is gelatinous – red and shiny and prone to infection like a burn victim – and sloughs off in the doctor's hand. The baby's windpipe is so small it can be crushed by a breathing tube.

P. Choudhuri, MBBS, MSc (Thalassaemia) D. Chaudhuri, MBBS, Dip National Board Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India e-mail: sanjuktaniranjan@gmail.com

And there is the pain, treated with narcotics that can cause dependency and withdrawal symptoms. The long-term prognosis for babies this young includes mental retardation, blindness, deafness, cerebral palsy and – in the best scenario – learning disabilities, wrote Susan Donaldson James of ABC News on 21st Feb 2007

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development: First and Second Trimesters*, DOI 10.1007/978-3-319-14874-8_46

What Is Extreme Prematurity? Definition

Preterm is defined as babies born alive before 37 completed weeks of pregnancy. The fetus is said to have attained viability if it has crossed 20th completed weeks of gestation and with the current level of neonatal care almost 50 % of such babies to be born after 24 completed weeks of gestation are likely to survive extra-uterine onslaught. Preterm is further subdivided depending upon the time of gestation in completed weeks post conception:

- Extremely preterm born before 28 completed weeks of gestation
- Very preterm born between 28 to 32 completed weeks of gestation
- Moderate to Late preterm more than 32 but less than 37 completed weeks of gestation [1].

Incidence

An estimated 15 million babies are born too early each year, which is more than one in ten babies, and about one million children die each year due to complications arising out of preterm birth. Many survivors face life time disabilities, including learning disabilities, visual and hearing problems. Globally, prematurity is the leading cause of death in children less than 5 years of age, and in almost all countries, incidence of prematurity is on the rise. In low socio-economic groups, 50 % of babies who are born at less than 32 completed weeks of gestation die due to lack of feasible and basic life care support, but in high income countries almost all the babies survive.

Why Does Preterm Birth Happen?

Most occur spontaneously, but in some cases, multiple births, chorio-amnionitis, other infections and chronic conditions, such as diabetes, high blood pressure, may result in preterm birth, but often the cause is not identified. Obesity or the low weight of the mother, vaginal infections, tobacco smoking, psychological stress and preeclampsia are important predisposing causes; few have suggested genetic predisposition and influence.

There are 3,519,100 preterm births in India in a year [2].

Preterm birth is one of the most common causes of death among infants worldwide [3]. About 15 million babies are preterm each year (5–18 % of all deliveries). In many countries rates of premature births have increased between the 1990s and 2010s [4]. Complications from preterm births resulted in 0.74 million deaths in 2013 down from 1.57 million in 1990 [5]. The chance of survival at less than 23 weeks is close to zero, while at 23 weeks it is 15 %, 24 weeks 55 % and 25 weeks about 80 % [6]. The chances of survival without long term difficulties is less [7].

Introduction

In recent years, excellent collaborations between the obstetrician and neonatologists and advances in neonatal care have made survival of pre-term neonates, especially extremely premature neonates possible [8]. Though preterm birth rates have increased all over the world, so has their survival especially in developed countries. Although most of the organs have been formed but their tertiary and terminal differentiation, for it to be structurally mature, is still pending in preterm babies. Moreover most organs are functionally, hence physiologically immature, which may or may not be entirely due to its anatomical and micro-structural immaturity. It is to be noted that the brain and lungs are the most susceptible among other organs, to the consequences of prematurity [9, 10]. Due to increasing specialization and cost of neonatal intensive care and the social and economic burden of disabilities on society, it is very important to justify intensive care and clinicians to be aware of the changing outcomes and the effect of long term disabilities and health problems on the survivors, their families and society as a whole. The focus should be on mortality and both short-term and long-term sequel of preterm birth, covering a broad range of outcomes such as neuro-developmental, educational, behavioral, psycho-social, growth and health outcomes. The principal areas of concern are higher rates of temperature instability, respiratory distress, apnoea, hypoglycaemia, seizures, anaemia, jaundice, which can lead to kernicterus, feeding difficulties, visual and auditory impairment and peri-ventricular leukomalacia [11–14].

Respiratory System

Development Between months 3 and 6, the lung changes from pseudo-glandular to glandular; conducting epithelial tubes surrounded by thick mesenchyme are formed and there is extensive airway branching; after that the bronchioles are produced, the number of capillaries in close contact with the cuboidal epithelium increase and the alveolar epithelium begin to develop from about 16th week and completed by the 25th week, which is known as the canalicular stage. In this stage there is differentiation of pulmonary epithelium resulting in the formation of the future air-blood tissue barrier, and surfactant synthesis starts. Next is the saccular stage, which stretches from 24 to 40 weeks, where the alveolar ducts and air sacks are developed [15, 16]. Most peripheral airways form the widened airspaces, termed sacchules, which widen and lengthen the airspace, which is to become the future gas exchange region; this expands significantly [17, 18]. The fibroblastic cells undergo differentiation, producing extracellular matrix, collagen and elastin, which play a role in epithelial differentiation and control of surfactant secretion. This prenatal secretary lung epithelium has to drastically change to post natal absorptive epithelium, which is influenced by epinephrine, oxygen, glucocorticoids and thyroid hormone [19]. The role of Clara cells, along with ciliated and pulmonary neuro-endocrine cells that make up the epithelium of the bronchioles along the conducting airways, has also been indicated in regeneration after injury [20].

The Problem As the formation of terminal bronchioles and its maturation into sacchules is incomplete, so is the transition from secretary to absorptive epithelium; the lung fails to expel fluid, such that the transformation from a cystic space to pneumatic organ is hampered. The secretion of surfactant is insufficient, as only from the sixth month the alveolar type 2 cells starts secreting it. The amount is highly insufficient to inflate and keep inflated the fluid filled vesicles, the surface tension of which is huge compared to the meagre amount of surfactant being produced. So just after birth, respiratory distress starts, which has been termed as 'Infant Respiratory Distress Syndrome (IRDS/RDS)'. This can only be tackled by administration of artificial surfactant and not with glucocorticoid injection to the mother, as there are least number of alveolar type 2 cells mature enough to secrete it in response to the stimulation. Moreover, though it has been used for decades, meta-analysis shows that its use may be harmful, though it might be having some added advantages too [21-25]. Chronic lung disease is another problem which can develop if the baby survives the initial insult with artificial lung surfactant, specialized ventilation and controlled oxygenation via extracorporeal membrane oxygenation (ECMO), which was previously known as broncho-pulmonary dysplasia (BPD). Due to this respiratory insufficiency other organs also suffer as they too become victims of hypoxia, specially the brain and the hearing apparatus.

Respiratory Distress Syndrome and Bronchopulmonary Dysplasia

RDS is one of the most common lung disorders in premature infants, affecting about 10 %. Most infants who develop RDS show signs of breathing difficulty at birth or within the following few hours. The cause is largely the result of surfactant deficiency and lung immaturity in both structure and function. Many infants born with serious RDS go on to develop BPD. BPD usually develops within 1–2 weeks after birth; injury to small airways and microvascular development is implicated, the combined result of lung immaturity prolonged oxygen and mechanical plus ventilation. Infants with BPD are at risk for repeated pulmonary infections [26, 27].

Long Term Outcome

Respiratory Premature infants with a history of severe neonatal respiratory distress such as bronchopulmonary dysplasia may need more frequent follow-up to be vigilant for wheezing or pulmonary infections. Infants are likely to have persistent respiratory issues up to about 1 year of age. For example, hospitalization for a cold that causes wheezing and difficulty in breathing is not uncommon for infants who have had early respiratory difficulties. Respiratory difficulties tend to become less frequent after the first year; group daycare therefore may not be advisable in the first year. Infants with bronchopulmonary dysplasia can experience poor lung function until adolescence, when lung function typically becomes normal [28, 29].

Nervous System

Neurulation happens between 0 and 4 weeks, thereafter till the 12th week neuronal proliferation occurs, which is followed by neural migration, which continues till birth. Selective apoptosis and synaptogenesis starts within 17–18th week, continues till adolescence. Myelination starts after the 28th week and continues till adulthood [30, 31].

Results from multiple studies indicate that the effects of early life stress on the developing brain are significant and include, but are not limited to the following: increased amygdala volume, decreased activity in frontal cortical and limbic brain structures, and altered white matter structures.

Early life stress is believed to produce changes in brain development by interfering with neurogensis, synaptic production, and pruning of synapses and receptors. Interference with these processes could result in increased or decreased brain region volumes, potentially explaining the findings that early life stress is associated with increased amygdala volume and decreased anterior cingulate cortex volume [32].

Children born preterm are more likely to have white matter brain abnormalities early on causing higher risks of cognitive dysfunction [33]. White matter connectivity between the frontal and posterior brain regions are critical in learning to identify patterns in language [35]. Preterm children are at a greater risk for having poor connectivity between these areas leading to learning disabilities [36]. Neurological problems include apnea of prematurity, hypoxic-ischemic encephalopathy (HIE), retinopathy of prematurity (ROP), developmental disability, transient hyperammonemia of the newborn, cerebral palsy and intraventricular hemorrhage, the latter affecting 25 % of babies born preterm, usually before 32 weeks of pregnancy [34]. Mild brain bleeds usually leave no or few lasting complications, but severe bleeds often result in brain damage or even death [22]. Neurodevelopmental problems have been linked to lack of maternal thyroid hormones, at a time when their own thyroid is unable to meet postnatal needs [36].

Apnea of Prematurity (AOP) Apnea is the most common problem of ventilatory control in the premature infant, defined by a cessation of breathing for 20 s, or for a shorter period of time if accompanied by bradycardia, cyanosis, or pallor. The incidence and severity of apnea in premature infants are inversely related to gestational age. Approximately 50 % of VLBW infants require either pharmacologic intervention or ventilatory support. The peak incidence occurs between 5 and 7 days postnatal age. AOP usually resolves at 34–36 weeks postconceptual age [26, 27].

Apnea of prematurity tends to improve as the infant matures, usually resolving by about 42–44 weeks adjusted age [28, 29].

Retinopathy of prematurity (ROP) ROP occurs mainly in ELBW infants. Premature birth interrupts development of the retinal vasculature; prolonged exposure to supplemental oxygen required for the premature infant's lung function further exacerbates the condition. ROP often regresses or heals, but it can lead to severe visual impairment or blindness; myopia, amblyopia, or strabismus may occur [35].

Cerebral Palsy (CP) CP is an umbrella term describing non-progressive brain lesions that occur during early development, with resulting disorders varying by the week of gestation when the injury occurred. Preterm birth is only one of many risk factors for CP. The motor disorders of cerebral palsy may be accompanied by other disturbances including cognition, communication, perception, behavior and a seizure disorder [35, 37].

Developmental Disabilities As increasing numbers of VLBW and ELBW infants survive the early physical challenges of prematurity, attention is focusing on understanding longer-term implications for the child's developmental progress. Developmental disabilities resulting from premature birth range from cognitive and academic abilities, fine and gross motor skills, vision and hearing, and attention and behavioral deficits.

Long Term Outcome

Vision Visual problems, such as those associated with retinopathy of prematurity, should be followed regularly after discharge, with regular vision exams. Premature infants are also at increased risk for issues with eye muscle strength (e.g., strabismus) and may require glasses for correction [35].

Hearing All preterm infants should have their hearing tested at least once during their first year to make sure they do not have hearing problems. Early identification of hearing impairment is critical for the child's language development [35, 37].

Developmental Issues Ongoing follow-up physical examinations and monitoring of developmental milestones help identify cerebral palsy or other developmental problems as early as possible to enable an early start to appropriate interventions [35, 37].

Compared with full-term infants, premature infants have been found to demonstrate poorer scores relating to behavioral organization, attention to sensory stimuli, and self-regulation. It is thought that these early deficits are related to later deficits in children's attention regulation and executive functioning.

Numerous follow-up studies have been conducted on premature infants to assess later evidence of developmental delays:

- The EPICure Study followed a group of infants born at 25 or fewer weeks' gestation in 1995. More than 60 % received steroid treatment and 84 % received surfactant. At 30 months of age, 24 % of the survivors had severe disabilities in neuromotor, cognitive, hearing and vision domains, with cognitive impairment most common. These children were evaluated again at 6 years of age. Of the children observed to have severe disability at 30 months of age, 86 % still had moderate-to-severe disability at age 6. However, children with less severe disabilities at 30 months were developmentally comparable at age 6 to those without disability at 30 months [38].
- A study comparing rates of survival and neurodevelopmental impairment at corrected age 20 months in two groups of infants, one group born before the introduction of surfactant therapy vs. one group born after, found that overall survival rates increased from 49 % pre-surfactant to 67 % with surfactant therapy, but the rates of survival both without impairment and with impairment increased [39].
- A 12-year study completed in 1997 evaluated a group of premature infants born in presurfactant, transitional, and surfactant eras, and found that the rate of neurologic impairment remained constant at approximately 11 % [40].
- A 15-year follow-up study of children born at less than 29 weeks' gestation who received surfactant therapy found that a significant minority required intensive special education services through secondary school age [41].
- Presence of disabilities in more than one developmental domain predicts a more serious outcome for preterm infants. A follow-up study of children born <30 weeks' gestation evaluated neurological, motor, cognitive and behavioral effects. Thirty nine percent were

observed to have a normal developmental outcome, 17 % had a single disability, and 44 % had multiple disabilities. Multiple disabilities were associated with lower birth weight, BPD and continuing neuro-developmental difficulties as observed at age 2 [42].

- VLBW infants have a higher likelihood of having behavioral problems, delays in motor development, and difficulties in school, all issues that lie within the domain of executive functioning. The fact that preterm children show more generalized learning problems rather than specific learning disabilities (e.g., only a reading problem) supports the theory that preterm children's learning difficulties are caused by a global processing deficit rather than difficulties with particular skills [43].
- A study of 10-year-old VLBW preterm children showed that 20 % suffered from attention deficits compared with 8 % of full-term children [43].
- A study of 241 on children born between 22 and 25 weeks who were currently at school age found that 46 % had severe or moderate disabilities such as cerebral palsy, vision or hearing loss and learning problems. Thirty four percent were mildly disabled and 20 % had no disabilities, while 12 % had disabling cerebral palsy [44].

Gastrointestinal System

Gastrointestinal Immaturities Gastrointestinal and metabolic issues can arise from neonatal hypoglycemia, feeding difficulties, rickets of prematurity, hypocalcemia, inguinal hernia, and necrotizing enterocolitis (NEC). The "mechanical" immaturities of the premature infant's GI tract affect feeding tolerance, digestion, upper and lower GI motility and absorption. GERD and necrotizing enterocolitis are serious results of GI immaturity.

GERD – Both premature and full-term infants experience gastroesophageal reflux (GER), which is distinguished from GERD by the number and severity of reflux episodes and complications including strictures, malnutrition, respiratory disorders, and bleeding. For premature infants, an incompetent lower esophageal sphincter combined with delayed gastric emptying can lead to GERD. An indirect sign, especially in premature babies, is apnea associated with bradycardia.

Necrotizing enterocolitis (NEC) – occurs in about 10 % of premature infants weighing less than 1500 g. The pathogenesis of NEC is not completely established, but evidence suggests that it is a function of multiple causes including the presence of abnormal bacterial flora, intestinal ischemia, and intestinal mucosal immaturity/dysfunction. NEC is rare in infants who have not received feedings. When feedings are started and food moves into a weakened area of the intestinal tract, bacteria from the food can damage the intestinal tissues. Severe tissue damage can cause a perforation to develop, leading to infection in the abdomen. Initial signs of NEC include feeding intolerance, delayed gastric emptying and abdominal distension and tenderness.

Gastroesophageal Reflux Gastroesophageal reflux is common in premature infants and usually presents as spitting up after feeding. Severe GER can cause feeding difficulties, irritability, poor weight gain and respiratory problems. Keeping the infant in a semi-upright position after feeding may improve symptoms. GER tends to resolve as the infant's GI system matures.

Other Complications

- Hematologic complications include prematurity, thrombocytopenia, and hyperbilirubinemia (jaundice) that can lead to kernicterus.
- Infection, including sepsis, pneumonia, and urinary tract infection

Integumentary System

The basal layer of the epidermis becomes the stratum germinativum, which produces new cells that are displaced into the more superficial layers. By 11 weeks, cells from the stratum germinativum have formed an intermediate layer. Replacement of peridermal cells continues until

approximately the 21st week; thereafter, the periderm disappears and the stratum corneum forms. Proliferation of cells in the stratum germinativum also forms epidermal ridges, which extend into the developing dermis. These ridges begin to appear in embryos at 10 weeks and are permanently established by 17 weeks [45].

Late in the embryonic period, neural crest cells migrate into the mesenchyme of the developing dermis and differentiate into melanoblasts. Later these cells migrate to the dermoepidermal junction and differentiate into melanocytes. The differentiation of melanoblasts into melanocytes involves the formation of pigment granules. Wnt signaling regulates this process. Melanocytes appear in the developing skin at 40-50 days, immediately after the migration of neural crest cells. In white races, the cell bodies of melanocytes are usually confined to basal layers of the epidermis; however, their dendritic processes extend between the epidermal cells. Only a few melanin-containing cells are normally present in the dermis. The melanocytes begin producing melanin before birth and distribute it to the epidermal cells. Pigment formation can be observed prenatally in the epidermis of dark-skinned races; however, there is little evidence of such activity in light-skinned fetuses. Increased amounts of melanin are produced in response to ultraviolet light. The relative content of melanin inside the melanocytes accounts for the different colors of skin.

Anemia Anemia can be managed with iron supplementation. Supplemental iron should be given to preterm infants for 12–15 months.

Growth and Nutrition During the first 2 years, growth of the premature infant follows their age corrected for prematurity using a special premature growth chart. After the age of two, the child's progress may be followed using the standard growth chart.

Nutrient reserves in premature infants tend to be low at discharge, therefore nutritional status should be assessed at each follow-up visit. Ongoing supplements are recommended for preterm infants who are breastfed to meet additional nutritional needs. Formula-fed infants who gain weight well can transition from a preterm formula to a regular routine formula, while those who require additional calories to meet weight goals may be fed a discharge formula until they meet weight goals and can be transitioned to a routine formula.

Conclusion

Medical problems originating in the initial weeks of life may require care for months or years. Other conditions may be identified later in infancy or in childhood, therefore continual attentiveness to early signs is valuable.

Each follow-up appointment should include review of developmental milestones so that early intervention can be initiated if a developmental delay is identified or suspected. As with any developmental problem, early intervention and therapy is a key strategy.

References

- Blencove H, Consens S, Ostergaard M, et al. National, regional and worldwide estimates of preterm birth. Lancet. 2012;379(9832):2162–72.
- World Health Organization. Fact Sheet No. 363. Preterm Birth. Updated 2015. http://www.who.int/ mediacentre/factsheets/fs363/en/
- Preterm labor and birth: condition information. http:// www.nichd.nih.gov. 03/11/2014. Retrieved 7 Mar 2015.
- World Health Organization. Preterm birth Fact sheet No 363. who.int. 2014. Retrieved 6 Mar 2015.
- GBD 2013 Mortality and Causes of Death, Collaborators. Global, regional, and national age-sex specific all-cause and cause-specific mortality for 240 causes of death, 1990-2013: a systematic analysis for the Global Burden of Disease Study 2013. Lancet. 2014;385:117–71. doi:10.1016/S0140-6736(14)61682-2. PMC 4340604. PMID 25530442.
- Cloherty JP. Care of the extremely low birth weight infant. In: Manual of neonatal care. 7th ed. Philadelphia: Wolters Kluwer Health/Lippincott Williams & Wilkins; 2012. p. 146. ISBN 9781608317776.
- Jarjour IT. Neurodevelopmental outcome after extreme prematurity: a review of the literature. Pediatr Neurol. 2015;52(2):143–52. doi:10.1016/j. pediatrneurol.2014.10.027.
- Goldberg R, Culhane JF, Iams J, Romero R. Epidemiology and causes of preterm birth. Lancet. 2007;371:73–82.

- Rees S, Inder T. Fetal and neonatal origins of altered brain development. Early Hum Dev. 2005;81:753–61.
- Jobe AH, Bancalari E. Bronchopulmonary dysplasia. Am J Respir Crit Care Med. 2001;163:1723–9.
- Raju TN. The problem of pre-term births: a workshop summary. Pediatr Res. 2006;60:775–6.
- Wang ML, Dorer DJ, Fleming MP, Catlin EA. Clinical outcomes of near-term infants. Pediatrics. 2004;114:372–6.
- Kinney HC. Near-term human brain and the risk of periventricular leukomalacia: a review. Semin Perinatol. 2006;30:81–8.
- 14. Escobar GJ, McCornick MC, Zupancic JA, et al. Unstudies infants: outcomes of moderately premature infants in the neonatal intensive care unit. Arch Dis Child Fetal Neonatal Ed. 2006;91:F238–44.
- Warburton D, El-Hashash A, Carraro G, Tiozzo C, Sala F, Rogers O, De Langhe S, Kemp PJ, Riccardi D, Torday J, Bellusci S, Shi W, Lubkin SR, Jesudason E. Lung organogenesis. Curr Top Dev Biol. 2010;90:73–158. PMID: 20691848.
- Morrisey EE, Hogan BLM. Preparing for the first breath: genetic and cellular mechanisms in lung development. Dev Cell. 2010;18(1):8–23. PMID: 20152174.
- Chinoy MR. Lung growth and development. Front Biosci. 2003;8:d392–415. PMID: 12456356.
- Burri PH. Fetal and postnatal development of the lung. Annu Rev Physiol. 1984;46:617–28. PMID: 6370120.
- Barker PM, Olver RE. Invited review: clearance of lung liquid during the perinatal period. J Appl Physiol. 2002;93(4):1542–8. PMID: 12235057.
- Xing Y, Li C, Li A, Sridurongrit S, Tiozzo C, Bellusci S, Borok Z, Kaartinen V, Minoo P. Signaling via Alk5 controls the ontogeny of lung Clara cells. Development. 2010;137(5):825–33. PMID: 20147383.
- Roberts D, Dalziel S. Antenatal corticosteroids for accelerating fetal lung maturation for women at risk of preterm birth. Cochrane Database Syst Rev. 2006;3(3), CD004454. doi:10.1002/14651858. CD004454.pub2.
- Yeh TF, Lin YJ, Lin HC, Huang CC, Hsieh WS, Lin CH, Tsai CH. Outcomes at school age after postnatal dexamethasone therapy for lung disease of prematurity. N Engl J Med. 2004;350(13):1304–13. doi:10.1056/NEJMoa032089.
- Murphy KE, Hannah ME, Willan AR, Hewson SA, Ohlsson A, Kelly EN, Matthews SG, Saigal S, Asztalos E, Ross S, Delisle MF, Amankwah K, Guselle P, Gafni A, Lee SK, Armson BA. Multiple courses of antenatal corticosteroids for preterm birth (MACS): a randomised controlled trial. Lancet. 2008;372(9656):2143–51. doi:10.1016/ S0140-6736(08)61929-7. PMID 19101390.
- Noguchi KK, Walls KC, Wozniak DF, Olney JW, Roth KA, Farber NB. Acute neonatal glucocorticoid exposure produces selective and rapid cer-

ebellar neural progenitor cell apoptotic death. Cell Death Differ. 2008;15(10):1582–92. doi:10.1038/ cdd.2008.97.PMC 2636573.

- Steroids used in preemies may kill brain cells. USA Today. 2008-11-17. Retrieved 22 May 2010.
- Kessenich M. Developmental outcomes of premature, low birth weight and medically fragile infants. http:// www.medscape.com/viewarticle/461571.
- 27. Neonatology on the web. http://www.neonatology. org.
- LaHood A, Bryant CA. Outpatient care of the premature infant. Am Fam Physician. 2007;76(8):1159–64.
- Trachtenbarg DE, Golemon TB. Office care of the premature infant: part II. Common medical and surgical problems. Am Fam Physician. 1998; 57(10):2383–90.
- Brown TT, Kuperman JM, Chung Y, et al. Neuroanatomical assessment of biological maturity. Curr Biol. 2012;22(18):1693–8. doi:10.1016/j. cub.2012.07.002. PMC 3461087.PMID 22902750.
- Tau GZ, Peterson BS. Normal development of brain circuits. Neuropsychopharmacology. 2010;35(1):147– 68. doi:10.1038/npp.2009.115. PMC 3055433.PMID 19794405.
- 32. Baker LM, Williams LM, Korgaonkar MS, Cohen RA, Heaps JM, Paul RH. Impact of early vs. late childhood early life stress on brain morphometrics. Brain Imaging Behav. 2013;7(2):196–203. doi:10.1007/s11682-012-9215-y.
- 33. March of Dimes, PMNCH, Save the Children, WHO. Born Too Soon: The Global Action Report on Preterm Birth. Eds CP Howson, MV Kinney, JE Lawn. World Health Organization. Geneva, 2012. http://www.bettercaretogether.org/content/born-too-soon-globalaction-report-preterm-birth; http://www.bettercareto gether.org/sites/default/files/resources/Born%20 Too%20Soon.pdf
- 34. Frye RE, Landry SH, Swank PR, Smith KE. Executive dysfunction in poor readers born prematurely at high risk. Dev Neuropsychol. 2009;34(3):254–71. doi:10.1080/87565640902805727. PMC 2692028. PMID 19437202.
- 35. Marlow N, Wolke D, Bracewell MA, Samara M; for the EPICure Study Group. Neurologic and developmental disability at six years of age after extremely preterm birth. N Engl J Med. 2005;352(1):9–19. doi:10.1056/NEJMoa041367. Retrieved 6/7/2013.
- 36. Berbel P, Navarro D, Ausó E, Varea E, Rodríguez AE, Ballesta JJ, Salinas M, Flores E, Faura CC, de Escobar GM. Role of late maternal thyroid hormones in cerebral cortex development: an experimental model for human prematurity. Cereb Cortex. 2010;20(6):1462– 75. doi:10.1093/cercor/bhp212. PMC 2871377.
- Wilson-Costello D, et al. Improved survival rates with increased neurodevelopmental disability for extremely low birth weight infants in the 1990s. Pediatrics 2005;115:997–1003.
- Marlow N et al. for the EPICure Study Group. Neurologic and Developmental Disability at Six

Years of Age after Extremely Preterm Birth. NEJM 2005;352(1):9–19.

- Wilson-Costello D et al. Improved survival rates with increased neurodevelopmental disability for extremely low birth weight infants in the 1990s. Pediatrics 2005;115:997–1003.
- 40. D'Angio CT et al. Longitudinal, 15-Year Follow-up of Children Born at Less than 29 Weeks' Gestation After Introduction of surfactant Therapy Into a Region: Neurologic, Cognitive, and Educational Outcomes. Pediatrics 2002;110:1094–102.
- van Baar AL et al. Very Preterm Birth is Associated with Disabilities in Multiple Developmental Domains. J Pediatr Psychol. 2005;30(3):247–55.

- Trachtenbarg DE, Golemon TB. Care of the Premature Infant: Part I. Monitoring Growth and Development. Am Fam Physician. 1998;57(9):2123–30.
- 43. Kessenich M. Developmental outcomes of premature, low birth weight and medically fragile infants. http:// www.medscape.com/viewarticle/461571
- 44. Marlow N, Wolke D, Bracewell MA, Samara M (5/1/2005). "Neurologic and Developmental Disability at Six Years of Age after Extremely Preterm Birth". NEJM. 2005;352(1):9–19. doi:10.1056/ NEJMoa041367.
- Lund CH, Kuller JM. Integumentary System. In Kenner C, Lott JW, eds. Comprehensive Neonatal Nursing Care. New York: Springer. 2014. 5th Edition: 301.

Sequencing of Naturally Aborted Human Foetuses: A Resource for New Knowledge

47

Samir K. Brahmachari

Human Genome Project since its inception in 1990 has raised many ethical issues. However, after the successful completion of Human Genome Sequencing in 2003, it was very clear that there exist significant variations in the genome of two individuals. The HapMap project [1] and the Indian Genome Variation Database [2, 3] led to many discoveries using which one could predict how individuals are prone to various diseases including drug response.

The advancement of technology has allowed us to determine the genome sequence of fetal DNA in the circulating blood of the pregnant mother. This has great medical implications especially with old age pregnancy where there is a higher risk of Down's Syndrome and other genetic disorders in the family. Sequencing the Unborn [4] has met with great triumph as well as protest. In this short commentary, I would like to focus on the idea of sequencing the naturally aborted fetuses and its potential biological implications. And it is believed that such a resource can easily be generated in India and such a database can provide answers to many crucial yet unsolved problems related to premature delivery

Former, Director-General, Council of Scientific and Industrial Research, Govt of India, New Delhi, India e-mail: skb@igib.res.in to rejection of the fetus. This is also likely to answer several developmental issues that decide the progress from successful fertilization and fetal development to a fully mature baby.

Most powerful tool in genetics is to observe phenotypes by knocking out a gene. This has been very successful in smaller organisms and mammals like mouse. However, establishing a correlation between gene knock out and a phenotype has been successful for simple Mendelian genetic disorders but for complex disorders these are yet to be solved. Although large number of developmental genes and epigenetic modifications have been discovered through successful gene knockouts but application of this technique is not allowed for human embryos due to ethical reasons. Although recently discovered CRISPR-CAS technology [5] allows genome editing, the international scientific community has put major restrictions on the application of such technology for human embryo due to the uncertainty of their side effects resulting from wrong editing, which are yet to be deciphered. Recent human embryo editing by Chinese researchers [6] has raised many ethical concerns amongst the world's genomics community.

While we have been sequencing centenarians [7, 8] to understand the longevity and developmental genes that allows slow metabolism and slow decay of the cells and tissues, sequencing of naturally aborted fetus are like natural knockouts of the human genome. Sequencing of naturally

S.K. Brahmachari, PhD

[©] Springer International Publishing Switzerland 2016

N. Bhattacharya, P.G. Stubblefield (eds.), Human Fetal Growth and Development: First and Second Trimesters, DOI 10.1007/978-3-319-14874-8_47

aborted fetus will aid the discovery of deleterious mutations and rearrangement of genomes which will help discover the underlying principles of developments leading to fetal abortion.

The naturally aborted fetal genome sequencing will open up new possibilities of discovering the developmental errors that has finite likelihood to occur. While various chromosomal trisomy are not lethal at the fetal level it is likely that one will discover novel chromosomal rearrangements, duplications and deletions involved in development.

This database along with the genome sequence of premature born babies as in contrast to centenarian genomes will allow a comparative genomic analysis of fundamentally essential genes which are highly conserved and are highly sensitive to mutation, insertion/deletion, duplication etc.

The sequencing of naturally aborted human fetuses will aid in furthering our understanding of the underlying genetic factors behind miscarriages and reduce infant and maternal mortality.

The most crucial underpinning of this initiative would be that this will not result in any ethical issues. With its high frequency of population growth, India is likely to have higher number of naturally aborted fetuses. Thus India has a strategic advantage. However, it is important to remember that this project will not be successful for artificially aborted fetuses.

References

- International HapMap Consortium. The international HapMap project. Nature. 2003;426(6968):789–96.
- Indian Genome Variation Consortium. Genetic landscape of the people of India: a canvas for disease gene exploration. J Genet. 2008;87(1):3–20.
- Indian Genome Variation Consortium. The Indian Genome Variation database (IGVdb): a project overview. Hum Genet. 2005;118(1):1–11. PubMed PMID: 16133172. http://www.igvdb.res.in/.
- Kai Kupferschmidt. Sequencing the Unborn. Science Magazine, 2012. http://news.sciencemag.org/ health/2012/06/sequencing-unborn.
- Sander JD, Joung JK. CRISPR-Cas systems for editing, regulating and targeting genomes. Nat Biotechnol. 2014; 32(4):347–55. doi: 10.1038/ nbt.2842. Epub 2014. PubMed PMID: 24584096; PubMed Central PMCID: PMC4022601. http://www. lifetechnologies.com/in/en/home/life-science/ genome-editing.
- Liang P, Xu Y, Zhang X, Ding C, Huang R, Zhang Z, Lv J, Xie X, Chen Y, Li Y, Sun Y, Bai Y, Songyang Z, Ma W, Zhou C, Huang J. CRISPR/Cas9-mediated gene editing in human tripronuclear zygotes. Protein Cell. 2015;6(5):363–72.
- 7. The project is ongoing and the link to the online portal is: https://sites.google.com/a/igib.in/100g/.
- Gierman HJ, Fortney K, Roach JC, Coles NS, Li H, Glusman G, Markov GJ, Smith JD, Hood L, Coles LS, Kim SK. Whole-genome sequencing of the world's oldest people. PLoS One. 2014;9(11):e112430.

Part XVIII

Ethics and Human Fetal Tissue Research

Ethics Pertaining to the Use of Aborted Human Tissues for Research and Therapeutic Purposes

48

Priyadarshi Sengupta, Niranjan Bhattacharya, Sanjukta Bhattacharya, and Phillip G. Stubblefield

Introduction

With recent advancements in the field of medical science, tissue engineering, organ transplantation, human embryonic stem cell research and aborted fetal tissue transplants are becoming an attractive source for new scientific exploration and innovation in the field of medicine and healthcare. Human tissues can be used for transplantation studies and therapeutics as well as for in vitro purposes in understanding the simple and basic biology of different ailments as well as in the search for a better cure. However, like any other drug therapy or drug discovery process, tissue experimentation generates intensive debate in terms of ethics and law. Whether it is moral or immoral to use different aborted materials like embryonic stem cells from the inner cell mass or fetal tissues are discussed in this chapter, as these are the two primary materials that are obtained from aborted fetuses and are currently intense topics of discussion regarding their ethical use for experimental and therapeutic purposes.

Religious Views regarding the Life of a Fetus and Its Abortion

Different religions have different views on when life begins in a fetus and therefore whether abortion should be allowed or not, and if it is allowed, under what circumstances or conditions. Most religions make an exception if the mother's health is in danger, but, beyond that, are uncompromising. Research on the growth and development of the fetus or on surgery of the unborn or fetal stem cells, may require aborted fetuses, and therefore it is important to understand religious beliefs. While one country may regularize abortion under certain circumstances including the right of the mother to have a choice in the matter, another country may forbid abortions on religious grounds. In Islam, abortion is normally regarded as wrong or haram (forbidden); however, under special circumstances abortion is permitted in

P. Sengupta, MSc, MPhil

Department of Regenerative Medicine and Translational Science, Calcutta School of Tropical Medicine, Kolkata, India

N. Bhattacharya, Dsc(Medical Science), MD(Ob/Gyn), MS(Gen Surg), FSOG(Canada), FICS, FACS(USA) (⊠) Dr. Subhas Mukherjee, Chair Professor, Department of Regenerative Medicine and Translational Science, Director General of the Public Cord Blood Bank and Convener of Bidhan Chandra Roy Biorepository, Calcutta School of Tropical Medicine, Kolkata, West Bengal, India e-mail: sanjuktaniranjan@gmail.com

S. Bhattacharya, PhD Professor, Department of International Relations, Jadavpur University, Kolkata, West Bengal, India

P.G. Stubblefield, MD Emeritus Professor of Obstetrics and Gynecology at Boston University School of Medicine, Boston University, Jamaica Plain, MA, USA e-mail: phillip.stubblefield@gmail.com

[©] Springer International Publishing Switzerland 2016 N. Bhattacharya, P.G. Stubblefield (eds.), *Human Fetal Growth and Development: First and Second Trimesters*, DOI 10.1007/978-3-319-14874-8_48

order to save a mother's life. According to in Islam, abortion is permitted after 120 days of pregnancy under specific circumstances and some Islamic states even permit abortion after 16 weeks of the second trimester if it is thought that the pregnancy may cause major concern with regard to the mother's health and wellbeing or if the pregnancy outcome may result in a child with life long health issues [1].

According to the Jewish custom or Halacha a fetus becomes a fully fledged human when the head emerges from the womb at the time of birth. Before then the fetus is normally considered as a partial life [2]. The fetus to the Jewish community has a supreme value as it helps in forming a fully fledged life. Abortions are not permitted on the grounds of genetic anomalies; however, like in Islam, abortion is permitted only when it poses a great risk to the mother's life and health [2]. In Christianity, both the Catholic and the Orthodox church are against abortion. According to the Catholic Church, life is sacred and inviolable and it should not be hurt [4].

However, in ancient times, before the advent of the Jewish religion or Christianity or Islam, the understanding of when exactly life begins, was different. During the fourth century BCE to first century CE, Aristotle's theory of ensoulment was widely accepted in pagan Greece and Rome [3]. Aristotle taught that the fetus is originally a vegetable which evolves as an animal soul with further increase in gestation time. Ensoulment or the life of a human starts at 40 days after conception in case of males and 90 days for female fetuses according to this theory [3]. Therefore the idea of abortion was not condemned in the early stages of life as the fetus was believed to have a vegetable soul or an animal soul except in the later stages after ensoulment has occurred [3]. St. Augustine (354–430 CE) opined that a human soul can reside in an organized entity during early pregnancy; an abortion cannot be called a murder as no human soul is destroyed (only an animal or vegetable soul is destroyed) [3]. This was very much in accordance with Aristotle's philosophy regarding the abortion of a fetus. Much later, at the turn of the seventeenth century, the Greek concept was totally discarded (the religions mentioned earlier, in any case, viewed abortion as morally wrong, and the new theories

provided a kind of justification); the concept of simultaneous animation gained momentum: it was surmized that the embryo acquires its soul right from the conception stage and not in the later weeks of gestation [3].

In Hinduism, abortion is generally viewed as a wrongdoing since it destroys an atma (soul). Like the Jewish and Muslim communities, Hindus permit the abortion process only when pregnancy poses a great risk to the mother. Normally, according to Hinduism, killing a fetus is like killing a priest, or it is even seen as worse than killing one's parents; in ancient times many would have lost their caste for committing this act. Many Hindus consider pregnancy as a duty since procreation is considered a duty to continue the family name, the caste and the community. Hindus also believe in re-incarnation, that is, when a person dies, his or her soul is recycled to this world in the form of a fetus in a different mother's womb; thus, in the eternal cycle of life, the fetus is held in high respect [5].

In contemporary times, however, like some other developing nations, India, which has a majority Hindu population with Muslims forming a large minority, has legalized abortion till a fixed number of weeks; this is to contain the population and for other reasons like the mother's health, the status of the fetus' health etc. At the same time, illegal abortions of female fetuses, known as 'female foeticide' also occur because of cultural preferences for male children [5].

Buddhism believes that harming the fetus is a murder or killing; but according to the views of the Dalai Lama, the Buddhist supremo who is remarkably modern in his thinking, while killing a fetus is wrong, depending on circumstances like whether the unborn child will be regressed and retarded from birth or create a serious problem for the parents, abortion can be implemented [4].

From the above discussion, it may be gathered, that an intense debate revolves around the issue as to whether the fetus can be considered as a life form, and this debate is largely fed by conceptual differences in various religions on when life begins, the sanctity of life and whether man has any right to kill or destroy God's creation. These arguments should also extend to man's so-called claimed right to kill in wars or develop nuclear bombs that threaten mass killing. It almost appears that self-proclaimed defenders of the 'fundamentals' of many religions place less sanctity on the lives of innocent adults, who are at times slaughtered in the name of religion, at other times, in the pursuit of power and national interest, than on the innocent unborn, who has yet to emerge into life. Interestingly, debates on issues of power struggles, regime change, nuclearization and the threat of a nuclear war/disaster. which may involve mass killing of innocents, never seem to take a religious form, while the abortion debate and ethical issues surrounding fetal research are largely religious. Those who are against abortion believe that life starts at the time of conception, and those who favour abortion believe that abortion can be implemented only before the fetus is considered as a sufficient human and has the right to live [6].

According to the Catholic Church as discussed above, the embryo should be treated as a person right from its conception. It is believed that the fertilized egg during the conception period contains the full genetic code of a human being. The fertilization stage can be regarded as the beginning of a process that involves development and maturation and initiates the beginning of biological life. Some people also consider brain activity of the fetus as the first sign of introduction of life in the fetus, i.e., it is the time when the fetus starts to develop into a new individual human being. However, medical science has proved that the stage of brain activity in the fetus is more like a pre-conditioning phase where the fetal consciousness is actually not very well demonstrated [6].

Guidelines and Ethical Principles Pertaining to Aborted Material Research for Therapeutic Purposes

The advances in medical research and the scope for new ideas to develop into new life saving therapies cannot be denied. In fact, since the beginnings of medical science, particularly Western medicine, research has proved invaluable in developing new cures, and some of this research involved cadaveric as well as real life research. While emphasizing the need for experimentation, without which theory will remain theory, it should also be emphasized that certain principles, guidelines and ethics should be followed. Medical Councils of many countries have tried to frame ethical guidelines which suggest broad principles for empirical research. One such guideline is that of the Medical Research Council, UK, which suggests that human samples, be it cell lines or an aborted fetus, should be treated like donations [7]. Research pertaining to the use of human aborted samples and biological specimens should be treated with dignity, respect and transparency [7]. Cultural and religious differences play an important part while considering the use of aborted materials for research and medical purposes which researchers should be aware of. Therefore a connection of trust, belief and honesty should be developed between the researcher and the donor [8]. The Guideline also says that an important objective of research is to develop new products that will have less risk and optimum efficiency when compared to standard modes of therapy currently available in the market. Therefore the parameter of assessing the potential benefits and potential risks should be taken into serious consideration before aborted materials can be used for research or therapeutic purposes. Research, using aborted materials, should continue only when it is observed that the potential benefits outweigh the potential risks [9]. Researchers should also emphasize on the full use of existing collections rather than trying to accord new aborted material [8].

Another objective of research, according to the aforementioned Guideline is to know the unknown through trial and error methods. However, it suggests that special considerations should be followed while using human aborted materials by improving the awareness of the researcher with respect to the experiment or research or the therapeutic efficacy that certain trials or studies demand through different enriched and high quality journals and research materials for clinical datas and efficacy results; this can be easily accessed online so that there can be minimal waste of human and biological aborted samples [8].

Coming to the point of financial gains and transactions through selling or acquiring of aborted human parts or the fetus, the Guideline notes that no such vested interests should be encouraging the donors, physicians, researchers and the recipients of such aborted therapies, nor should there be any conflict of interest [9].

The informed consent form is one of the best ethical practices to maintain transparency in any medical or biomedical research pertaining to the use of human aborted tissues and samples. It gives the donor a chance to verify information pertaining to the purpose and use of their samples, thereby giving them an honest chance to decide whether to donate or not [8]. Through a lucid and easily understandable 'Informed Consent Form' (ICF), donors get a first hand and detailed information whether their samples would be used for commercial use like therapeutic purposes or just for pure research. Literacy is another important aspect of taking informed consent from human donors. This is a very serious issue in many developing countries where illiterate or minimally educated donors can be coerced or foxed into donations of aborted tissues or fetus in lieu of personal financial and commercial gains of the researcher, clinician or the recipient of the tissue. The donor should be adequately accompanied by someone who can maintain transparency and not influence the donor's decision and in case if the donor is not able to apprehend the language or the content of the ICF, the person accompanying should explain the ICF to the donor [10]. Details of the process and the purpose of collecting the aborted material, where it will be stored, whether it will be given or sent to any other place for therapeutic purposes, and if used for transplantation studies, the methods by which it will be transplanted and for what medical condition, should be mentioned.

Ownership is another important concept while donating aborted materials for research and medical purposes. In case of fetal transplantations, aborted fetuses are normally used as therapeutics. While the fetus is inside the womb, the mother controls the ownership; however, once it has been donated after a proper informed consent process, the question remains whether the mother will have the same say as before or whether the ownership role of the mother will be diminished as

the power of attorney of the aborted fetus will be transferred to the hands of the clinicians and the surgeons who will be conducting a fetal tissue transplantation or experimental procedures. It is a serious matter of debate and no straight forward answers are possible and the best way to ascertain such ownership or proprietary rights is to ascertain the situation on a case by case basis. There are other Guidelines like the world Health Organization's (WHO) International Ethical Guidelines for Biomedical Research involving Human Subjects and the Human Tissue Authority's (HTA) Codes of Practice, which are working on how best to conduct research in a fair manner without harming any community's sentiments or violating human or other rights or allowing unscrupulous persons to make financial gains instead of using aborted tissues for research that will ultimately benefit mankind [11]. It is useful to look into these well thought out but evolving guidelines as an aid to ethical research instead of dumping the baby with the bathwater in the rather intense debae on whether abortion or abortion research or fetal research should be permitted on ethical grounds. Contemporary state-of the-art research requires cell lines etc that may need the destruction of the embryo. If proper guidelines are followed and if a mother's right to choice is recognized, maybe some of the issues raised by religious and other bodies, may be circumnavigated.

Use of Human Embryonic Stem Cells from Pre Implanted Blastocyst Stage

One of the issues surround human embryonic stem cells which are an extremely attractive tool for cell therapeutic purposes and were first isolated by Thomson et al in 1998 at the University of Wisconsin, Madison [12]. These human embryonic stem cells (hES) as they are termed, are derived from the inner cell mass of the blastocyst stage of the embryos within 5 days of fertilization of the oocyte. hES cells are regarded as pluripotent stem cells as they can differentiate into any lineage of the ectoderm, mesoderm and the endoderm except the extraembryonic tissues which forms the placenta [13]. Hence these cells are pluripotent but not totipotent stem cells. These hES cells are also immortal due to the presence of the telomerase gene, the protein product which ensures that the telomeric ends of the chromosomes are maintained after each cell division and does not undergo senescence [13]. The other cells which are pluripotent like hESC's are the human embryonal germ cells (hEG) derived from the primordial germ cell lines which would have formed the gametes ultimately if the fetus was not aborted. Most of the work pertaining to hESC and hEG are still in the pre clinical and trial phase and this has led to an intense debate over the use of such cells for therapeutic research, as isolation of these cells before the implantation stage requires an intervention and the termination of the pregnancy [13].

Therefore the status of the pre-implanted embryos is one of the most debated and sensitive issues, disputing on the isolation of the inner cell mass for the derivation of the viable human embryonic stem cells. However, it may be pointed out here that normally, from a medical point of view, pre-implanted embryos are mostly non viable due to 50-60 % of them being aneuploidies in nature [13]. Therefore selecting a non viable embryo for clinical and therapeutic purposes to treat different diseases does not collide with the ethical views of destroying a potential or viable life [13]. The debate surrounding the pre implanted embryo therefore is more of a moral and symbolic value; but in terms of research it has potential value for cure and the development of new therapies [13].

From the scientific point of view, an isolated inner cell mass (ICM) alone cannot give rise to a fetus; this is the misconception among many people especially in the non medical community who condemn the use of embryonic stem cell research. Trophoblast and extra-embryonic endoderm cells along with the ICM are required for proper implantation and nutrition of the embryo to help in the formation of the fetus. Thereby the ICM alone is a non viable material which cannot contribute to later stages of development [13]. Some others also opine that the cells of the inner cell mass are an important part of the embryo; however, when the ICM is cultured in vitro, the non viable cells perish, giving way to the viable cells of the ICM that can further divide into three lineages. This further strengthens the argument that the ICM may contain viable human embryonic stem cells that along with other constructs like the tropohoblastic layer and the extra embryonic endoderm could give rise to a fetus, but not alone by itself, and therefore the cells of the ICM only, can never be considered as an embryo [13].

Inspite of the above mentioned scientific observations and explanations, many opinions suggest that any part of the ICM can be considered as a viable tool for fetal development if the proper constructs were present, thereby depriving the chance of fetal development to the isolated ICM. These very people also share the opinion that any cell that can give rise to a fetus should be considered as an embryo, thereby contradicting scientific ideas, where an adult stem cell after a somatic cell nuclear transfer into an enuclated oocyte can give rise to an adult. In such cases, then, adult stem cells will be also regarded as embryos which very much refute the scientific observation and concepts pertaining to embryology [13]. Therefore the moral value of the non viable pre implanted embryos is much lesser than the viable pre implanted embryos; this concept needs further introspection [13]. The ban on embryonic stem cell research on flimsy unscientific grounds based on religious morality rather than on scientific perceptions and observations, appear a little confounding and not very justifiable [13].

There are two types of opinions regarding the developmental source and moral status of the preimplantation embryo [13, 14]. Some view the embryo as a person and others see it as a non person devoid of any moral/rights status. Another view sandwiched between these two is the moderate view that the embryo needs protection as it can become a potential individual, and that it needs respect and is not to be played around with as a research material [13]. However, before the appearance of a primitive streak and the differentiation of the three germ layers, embryos have a diversified and unpredictable function from where it can either give rise to twins or a normal individual, or abort; thus the embryo, at this stage, can be considered more like the units of a cell present in an adult system which together forms an organization called the tissue and performs a specialized function.

Also, ethically, human embryonic research is far more superior in terms of moral acceptance when compared to the use of aborted fetal tissue transplants because a fetus has a higher moral and symbolic value than a pre-implanted embryo, and secondly, because abortion of the fetus can often lead to pregnancy specific disorders to the mother [15].

Use of Fetal Tissue as a Research and Transplantation Material: Moral and Ethical Dilemmas

Modern research has indicated the use of fetal tissue as an attractive option for treating intractable diseases like diabetes, Parkinson's and various other neurological disorders. It is also an attractive option for preparing vaccines; the 1954 Nobel Prize was awarded for a polio vaccine that was developed from fetal kidney cells. Also, fetal cells were widely used for making vaccines for measles. On November 10, 1988 fetal tissue was used for the transplantation of fetal cells into a 52 year old suffering from Parkinson's disease at the University of Colorado Medical Centre [16]. More recently in India, Bhattacharya et al used fetal thymus tissue in a patient with arthritis.

Yet many bioethicists like Arthur Caplan believe that the very concept of using aborted fetal tissue for research and therapies is like a ticking time bomb of medical ethics [17].

But the possibilities of unveiling the mysteries of nature to provide cure for diseases that have plagued mankind for centuries through fetal tissue, outweigh outright rejection of the idea of fetal tissue research. Fetal tissue has some unique properties for transplantation and cell therapy purposes as it is hypo immune and can result in less rejection of the graft. It has a high proliferation and differentiation rate in vivo as seen from various pre clinical trials and medical therapies. However, the paucity of research has led to limited success, though the possibilities are immense. The best way to characterize fetal tissue research and transplantation till date is "Experimental" [18]. In the US, currently, 8 states prohibit the use of fetal tissue from dead fetuses and 17 states further prohibit the sale of tissue and fetal organs [19].

Current prominent and identifiable ethical concerns pertaining to the use of fetal tissues as a substance for biomedical and medical research are mainly based on the possibilities that a pregnant woman might be influenced upon to undergo abortion, or some women would be encouraged to get pregnant to induce abortion in exchange for financial gains. Thirdly, a woman's decision to donate an aborted fetus as a gift for human research and development may also be influenced by persuasion tactics or pressure [19]. The very concept of gift is flawed, because abortion is induced due to the non viability of a fetus or in other words, fetuses that suffer normal abortion, are not welcomed to join the human community. Therefore questions arise regarding how a non viable fetus can be considered as a gift [18]. The mother also cannot consider herself a donor of a gift because a gift has certain values whereas an aborted fetus does not have any such properties and is destined for closure. Therefore the very concept of donation or gift may itself be debatable.

However, there is evident truth in all these opinions. The role of a physician in influencing a mother's decision for fetus donation due to vested interests or financial gain, is not unknown, especially in developing countries where povertystricken pregnant mothers with many children are extremely vulnerable to exploitation.

Therefore, it is important to reiterate some of the points made earlier regarding the strict implementation of guidelines which are made after considerable consideration. These guidelines should also be not too rigid, allowing for flexibility; what is taboo today may be acceptable tomorrow. It must be remembered that in early Christendom, cadavers were clandestinely dug up by scientists and artists who wished to study muscle structure, the veinous system etc; today, cadavers are used in medical schools all over the world for the study of human organs and systems. However, in the current context, certain measures may help maintain ethical standards: [19]

- Proper informed consent form, which is easily understood, and which contains all relevant information, is a must. Fostering and developing a relationship other than personal interests between the physician and the mother should be also encouraged. If the fetal tissue will be stored, it should also be mentioned in the ICF [19, 20].
- Health care practitioners who will be participating in the research should be in no way involved with the ICF process so as to avoid any conflict of interest. Also confidentiality of the donor should be maintained strictly [20].
- Ownership of the dead fetus or the fetal tissue, i.e., who will assume the proprietary or ownership rights, the mother, the physician or the recipients, should be clear [20].

Discussions

According to the World Medical Association (WMA) fetal tissues are not to be exploited, and should be provided according to the current guidelines and laws of the country, or international guidelines like the Declaration of Helsinki and the Declaration on Human Organ Transplantation should be followed [19]. The question that always rises up in case of fetal research and transplantation is that "Is it necessary to destroy a life for the sake of another life"? It is a highly ambiguous question that has no direct or correct answers because of the fact that a number of ethical, moral and scientific opinions are involved before coming to a general consensus. The same is the case for using human embryonic stem cells for therapeutic and clinical research. Under the circumstances, what is needed are strong independent Institutional Ethical Committees consisting of experts from diverse disciplines including law and the social sciences apart from subject experts, which can judge the merits of research on a case by case basis, and who are knowledgeable about their country's and international guidelines on medical research on human subjects, tissues or cells. The Ethics Committee must review the IFCs to safeguard the interests and health of the mother and the aborted fetus.

Conclusion

It is immensely important to understand what are the ethical dilemmas involved in aborted tissue research and whether this research is superior to the current approach towards cadaveric organ transplantation or xenotransplantation. The pros and cons of this kind of research have to be weighed. Modern medical theory speculates on the spectacular benefits of fetal tissue research and the potential cures of intractable diseases through the transplantation of cells derived from fetal tissues. On the other hand, such research can easily be exploited by doctors, donors, family members of donors, touts etc for personal financial gain. Further, one must also take into account the raging controversy on when life begins and whether abortion is ethical on religious grounds because it could mean the murder of an unborn human life, which also has a right to become a full human being at birth. Various national and international medical councils and other bodies have speculated on these issues and have come up with guidelines to prevent misuse of research and also take into account the mother's health and willingness to donate, thus also addressing religious and cultural issues. Perhaps, if strict surveillance is kept by ethical committees, which should bear in mind the long term benefits of research as well as the interests of all concerned, the new and novel ideas of medical scientific theorization can bear fruit in cures and therapies for humans in the future without harming the interests of the humans and unborn humans of the present.

References

- Sanctity of life, Islamic teachings on abortion. http:// www.bbc.co.uk/religion/religions/islam/islamethics/ abortion_1.shtml.
- Robinson BA. When does human personhood begin? Belief system 4: Jewish beliefs, Ontario Consultants on Religious Tolerance. Latest update: 09 Nov 2005.
- Robinson BA. Roman Catholicism and abortion access. Pagan & Christian beliefs 400 BCE–1983 CE. Copyright © 1997 to 2013 by Ontario Consultants on Religious Tolerance. Latest update: 22 Oct 2013.
- Religious views on abortion. https://www.spuc.org. uk/youth/student_info_on_abortion/religion.
- Hinduism and abortion. http://www.bbc.co.uk/religion/ religions/hinduism/hinduethics/abortion_1.shtml.
- 6. When does a fetus get the right to life? http://www. bbc.co.uk/ethics/abortion/child/alive_1.shtml.
- Health Research Authority/MRC: consent and participant information sheet preparation guidance. http:// www.hra-decisiontools.org.uk/consent/index.html.
- Human Tissue and Biological Samples for Use in Research: Operational and Ethical Guidelines. Medical Research Council November 2014. http:// www.mrc.ac.uk/research/facilities/regulatory-support-centre/human-tissue.
- International Ethical Guidelines for Biomedical Research Involving Human Subjects. Geneva 2002. Copyright by the Council for International Organizations of Medical Sciences (CIOMS). http:// www.cioms.ch/publications/layout_guide2002.pdf.
- General Medical Council (GMC) 'Consent to research' (updated 2013). http://www.gmc-uk.org/ guidance/ethical_guidance/research.asp.
- Human Tissue Authority (HTA) Codes of practice: code 1 – consent. http://www.hta.gov.uk/legislation-

policies and codes of practice/codes of practice/codelconsent.cfm.

- Thomson JA, Itskovitz-Eldor J, Shapiro SS, Waknitz MA, Swiergiel JJ, Marshall VS, Jones JM. Embryonic stem cell lines derived from human blastocysts. Science. 1998;282:1145–7.
- de Wert G, Mummery C. Human embryonic stem cells: research, ethics and policy. Hum Reprod. 2003;18(4):672–82.
- Hursthouse R. Beginning lives. Oxford: Blackwell; 1987.
- 15. De Wert G, Berghmans R, Boer GJ, Andersen S, Brambati B, Carvalho AS, Dierickx KM, Elliston S, Nunez P, Osswald W, et al. Ethical guidance on human embryonic and fetal tissue transplantation: a European overview. Med Health Care Philos. 2002;5:79–90.
- Denver Hospital defies ban on fetal tissue transplant. Med Ethics Advis. 1988;4:161.
- Human fetal tissue transplantation research, Report to the advisory committee to the Director, National Institute of Health, December 14, 1988,Behesda, Maryland. https://repository.library.georgetown.edu/ bitstream/handle/10822/559348/fetal_tissue_report. pdf?sequence=1.
- Rae SB, DeGiorgio CM. Ethical issues in fetal tissue transplants. Lincare Q. 1991;58(3):12–32.
- World Medical Association Statement on Fetal Tissue Transplantation, Adopted by the 41st World Medical Assembly Hong Kong in September 1989 and rescinded at the WMA General Assembly, Pilanesberg, South Africa, 2006. http://www.wma.net/en/30publica tions/10policies/20archives/f7/index.html.pdf?printmedia-type&footer-right=[page]/[toPage].
- 20. Nuffield Council on Bioethics. Human tissue ethical and legal issues. 1995.

Index

A

Abortion, 12 guidelines and ethical principles, 577-578 human embryonic stem cells, 578-580 intra amniotic antigens (see Intra amniotic antigens) moral and ethical dilemmas, 580-581 religious views, 575-577 Accessory olfactory bulb (AOB), 211 Acid mucopolysaccharides (AMPS), 103-104 brain, 111-113 extraction of, 104 fractionation of, 104-105 functions of, 102, 103 liver, 105-107, 109 lung, 109-111 matrix, 102 negative charges of, 103 Acupuncture, 530 Adrenal cortex, 136 DHEA-S, 215, 216 fetal hair and sebaceous glands adrenal androgen secretion, regulation of, 216-217 medullary cells, paracrine effect of, 217 Merkel cells and adrenal medulla, 217-218 physiological role of, 216 histological study 2 days of age, 142-143 7 days of age, 143 8 weeks of gestation, 137-138 9 weeks of gestation, 138 12 weeks of gestation, 138-139 13 weeks of gestation, 139-140 17 weeks of gestation, 140-141 19 weeks of gestation, 141 24 weeks of gestation, 141-142 28 weeks of gestation, 141 32 weeks of gestation, 142 postnatal life, autoantibodies in

Ouchterlony system, 152 passive cutaneous anaphyl axis test, 153 passive hemagglutination test, 153 quantitative immunoprecipitation test, 152-153 sensitized lymphocytes and macrophages, migration inhibition factor of, 155-156 rabbit antibodies, human fetal and adult, 156 fetal 1M antigens, characterization of, 151 1M NaCl extracts, 148-152, 159 PBS extracts (see Phosphate buffered saline (PBS))soluble proteins, fractionation of electrophoretic patterns, 144-146 gestation period, 144-146 Adult adrenal glands, antibodies 1M NaCl extracts, 148-151 PBS extracts, 148-149 Adult hypertension, 496 Advisory Committee on Immunization Practices (ACIP), 11 Aerobic exercise, 371 Alpha-fetoprotein (AFP), 59 Amniocentesis, 20, 30, 59, 411 AMP-activated protein kinase (AMPK), 241 AMPS. See Acid mucopolysaccharides (AMPS) Amylopectin, 94 Aneuploidy, 409 in oocytes, 232-233 preimplantation embryos, abnormalities in, 234 screening, 402 in spermatozoa, 233 Anthropometric measurement, 175 brain, 163, 175 fetal body weight, 68 crown-rump length, 68, 70 head circumference, 68, 71 and volume, 71 and grouping, 67-69 growth

Anthropometric measurement (cont.) of brain, 73-74 of human fetal ovaries, 76-77 of human fetal testes, 75-76 of kidney and adrenal gland, 74-75 of liver, 72 of lung, 72-73 of thymus, 76-77 weight length relationship, 68, 69 Aorta-gonad-mesonephros (AGM), 270 Apnea, 554 Apnea of prematurity (AOP), 564 Arginase, 123 Argininosuccinic acid synthetase, 123, 125, 126 Autosomal dominant polycystic kidney disease (ADPKD), 396 Autosomal recessive polycystic kidney disease (ARPKD), 396 Avurveda herbs Aswaganda, 548 Garbharaksha kashayam, 548 Garbha raksheena guliga/maha-dhanwanthira, 548 garlic, 548 Vilvaadhi laham, 547 human fetal growth and development monthly development, 543-544 qualities of panchamahabhutas, 542-543 and modern science, 546 pharmacovigilance, 548-549 pregnant mother and fetus, 545-546 safety concern, 547 yogas first trimester, 546-547 second trimester, 547

B

Baker hypothesis, 371 Blast colony forming cells (BL-CFCs), 270 Bovine serum albumin (BSA), 28 Brain anthropometric measurement, 163, 175 biochemical changes central nervous system, histological studies of (see Central nervous system (CNS)) DNA and RNA content, 164, 175, 178, 179 glutamine synthetase, 166, 167, 175 P³² incorporation, 165–166, 175–176 protein content, 164-165, 176, 178 brain cells, number of, 163 brain cortex and personhood, 208-209 differentiation, 162-163 glucosamine-6-phosphate synthetase, 113-114, 118 migration, 162 mucopolysaccharides, 111-113, 117 proliferation, 161-162 sodium and potassium concentrations in, 113 13th week of development, adult type hippocampus (see Hippocampus)

Branched-chain amino acids (BCAAs), 241–246, 248 Broncho-pulmonary dysplasia (BPD), 554, 563

С

Carbohydrate metabolism blood sugar concentration, fetal and maternal blood, 87-88, 93 human fetal liver fructose 1,6 diphosphatase, 91-92, 96, 97 glucose-6-phosphatase activity, 90-91, 95-97 glycogen content in, 88-89, 94-96 inorganic phosphorus content in, 89 phosphorylase activity in, 89-90, 96, 97 Cardio vascular defects (CVD) arch obstruction and obstructive lesions, 359 cardiac catheter intervention, 362 delivery planning, 363 detection, 359 fetal arrhythmias anti arrhythmic drugs, 361 fetal bradycardia, 361 fetal tachycardia, 361 irregular rhythm, 361 fetal echocardiography, 360, 361 fetal factors, 360 fetal surgery, impacts of, 363 incidence, 359 maternal factors, 360 obstetric ultrasonography, 360 open fetal surgery, 363 perinatal management, 363 prenatal diagnosis, 363 surgical techniques, 362-363 therapy, 361 transitional care, 363-364 vascular defects, 361 Cardio-vascular system, 53 Central nervous system (CNS), 54, 162, 272 cell replacement therapy, 373-374 cerebrum-subgerminal layer, 172, 173 cortical histogenesis, 177 development of cerebellum, 178-179 of eye, 167, 170, 171, 177 of midbrain, 177 of somites, 177 of sulcus, 172, 173 extensive capillary network, 172 nerve fiber, layer of, 167, 168 neural tube, disposition of, 167 neuronal cells, peripheral layer of, 174 neuronal migration, 172 nucleated cells, 171 sub-cortical region, 172, 173 undifferentiated cells, 161, 170 ventricular zone, intermediate zone and marginal layer, 167, 168 Cerebral palsy (CP), 558, 565 Childhood aerobics, 371

Chinese medicine. See Traditional Chinese medicine (TCM) Cholesterol (CHO) cholesterol biosynthetic pathway, defects in, 192 intake and steroid hormonal synthesis regulation, 184 and lipid exchange, 185 maternal contribution, 186 production of, 186 transport, 186-187 Chorionic villus sampling (CVS), 60, 411 Chronic lung disease (CLD), 554 Citrulline, 122, 125, 127, 130 ¹⁴C-leucine incorporation, 164, 165, 176, 180 Clostridium tetani, 22 CNS. See Central nervous system (CNS) Congenital anomalies of the kidney and urinary tract (CAKUT), 395 Congenital diaphragmatic hernia, 468 Connective tissue (CT), 299, 319 electrolytes control in, 103-104 structure of, 102 C reactive protein (CRP), 16 CVD. See Cardio vascular defects (CVD) Cytogenetics, 231, 232

D

Dehydroepiandrosterone sulfate (DHEA-S), 215, 216 Developmental origin of health and disease (DOHaD), 372 Diabetes, 57, 93, 190, 191, 246, 498–500 Diabetes mellitus (DM), 360 Diaphragmatic hernia, 428, 462 Dietary exposure, 370 Down's syndrome, 409, 410, 415, 416, 429, 571 Dyptheria, 27, 31

Е

Early embryonic development chromatin modifications and cell differentiation, 201-202 gene regulatory networks, 200-202 ICM, epiblast and primitive endoderm, 203-204 morphological and lineage specification steps, 199, 200 transcription factors, zygote, 203 Elongation factor 2 (EF2), 27 Embryology embryonic stem cells, 225-227 gamets, transperitoneal migration of, 224 left ovary and right abdominal testis, 224 male and female gonads, 223-224 pregnancies in chimera, 224 self-fertilization, 224-225 virgin birth, 225 Embryonic stem cells (ESCs), 207, 225-227, 267-269

Endocrine maturation adrenal system, 294-295 anterior pituitary adult gland, 292 bone morphogenetic protein 4 (BMP4), 292 corticotroph cells, 292 definitive Rathke's pouch, 292 genetic mutations and alterations, 293 hypothalamic cell condensations, 293 pituitary placode, 291 Ptx1 expression, 292 rudimentary Rathke's pouch, 291-292 signal-dependent coactivating factors, 292 TSH and gonadotropin, 292 autonomic nervous system, 295 gonadal system, 298-299 intermediate lobe pituitary, 293 pancreas, 299-300 posterior pituitary, 294 thyroid clinical importance, 296 congenital hypothyroidism, 297 maternal hypothyroxinemia, 297 parathyroid gland, 297 pharyngeal mesoderm, 296 thyroglobulin (Tg), 296 thyroglossal duct, 296 thyroxine-binding globulin (TBG), 296 Eosinophil granulopoiesis, 311 Eosinophils, 474 Epidermal growth factor (EGF), 474-475 Epigenetics, 371 Extravillous trophoblast (EVT) cells, 57 Extremely low birth weight (ELBW), 379

F

Familial hypercholesterolemia (FH), 190-191 Fatty acid (FA), 187-188 Fetal abnormalities cervical screening, 410 Doppler scan, 416 fetal anomaly scans, 410 fetal echocardiography, 410-411, 416 fetal medicines, 411-412 fetal scan, 411 fetal viability scans, 410 first trimester screening, 410 high resolution ultrasonography, 410 intrauterine transfusion (IUT), 416 invasive and cervical procedures, 416 invasive strategies, 411 magnetic resonance imaging, 410 non-invasive prenatal diagnosis strategies, 409 risk reassessment scan, 410 three dimensional ultrasonography, 410 Fetal alcohol syndrome (FAS), 56, 272 Fetal arrhythmias, 360, 361, 364 Fetal bradycardia, 360 Fetal echocardiography, 360, 361

Fetal growth, 49-50 antibody production, 17, 18 B lymphocytes, 17 brain cortex and personhood, 208-209 CHO (see Cholesterol (CHO)) first two trimesters alcohol and illicit drugs, 56 alpha feto protein assay, 59 cardio-vascular system, 53 cigarette smoking, 56 CNS. 54 CVS, 60 cytokine response, prenatal infections, 56-57 diagnostic amniocentesis, 59 fetal allograft, 57-58 gastrointestinal system, 53-54 genetic and epigenetic perspectives, 54-55 genetic factors, 55 impaired utero-placental and feto-placental blood flow, 56 maternal nutrition, 55-56 microchimerism, 56 nine to twelve weeks, 50-52 non-invasive prenatal diagnosis, 59 nutritional factors, 55 parity, maternal age and multiple pregnancy, 55 respiratory system, 51, 53 seventeen to twenty four weeks, 51 spectrophotometric studies, 59-60 thirteen to sixteen weeks, 51 tribal medicine (see Tribal medicine) ultrasonography, 58-59 urinary system, 54 head circumference, 5 intra amniotic antigen and disruption (see Intra amniotic antigens) length, 5 maternal milk consumption, mTORC1 (see Mechanistic (mammalian) target of rapamycin complex 1 (mTORC1)) natural immunity CRP. 16 fibronectins, 16 lactoferrin. 16 opsonin activity, 16 polymorphonuclear leucocytes, 16 nucleic acids, 5-6 physical growth and measurements, 3-4 placental transfer, 18 protein non-specific protein synthesis, 6 physiological functions and biochemical processes, 7 specific protein synthesis, 6, 8 T lymphocytes, 16-17 vaccination (see Vaccination) weight, 5 Fetal inflammatory response syndrome (FIRS), 311 Fetal reduction, 407-408 Fetal surgery cystic lung lesions, 460

development of, 459-460 medication, 463 myelomeningocele, 461-462 sacrococcygeal teratoma, 460-461 TTTS, 462 Fibroblast growth factor 21 (FGF21), 247 First trimesters ACE inhibitors, 417 anti retroviral drugs, 420-421 anti-tubercular drug, 419 Ayurveda (see Ayurveda) blastocyst, 417 chromosomal abnormalities, 420 embryo abortion, 417 fetal abnormalities, diagnosis abdominal muscles, 417 chorionic villus sampling (CVS), 416 facial malformations, 417 holoprosencephaly, 417 MRI, 417 neuro-logical problems, 417 pulmonary herniation and cyst, 417 transvaginal USG, 416 two dimensional and three dimensional scans, 416 ultrasound imaging, 416 umbilical cord, anomalies, 417 fetal age, 418 fetal infection, 424-426 hematological disorders, 422-424 hematopoietic stem cells, 431 human immune system, 431 lithium toxicity, 418 natural killer (NK) cells, 431 nicotine, 418 Phenindone D, 418 physiological and anatomical disorders, 421-422 sickle cell anemia, 431 stem cell therapy, 431 surgical interventions congenital anomaly, 430 fetal tracheal occlusion intervention (FeTO), 428 hemangioblastomas, 429 intrapericardial teratomas and rhabdomyomas, 429 Kasabach-Merritt sequence, 429 open fetal surgery, 428 sacrococcygeal tumors (SCT), 429 spina bifida/myelomeningocel, 429 twin to twin transfusion syndrome, 430 valvar aortic stenosis, 430 teratogenic drug, 430 teratogenic effect, 419-420 teratomas, 419 treatment options, 426 umbilical cord blood, intrauterine transfusions allogeneic cord blood transfusion, 427 clinical trials, 426 coagulation post intra uterine fetal transfusion, 427 gene therapies, 428 HIV transmission, 427 HLA matching, 427 HSC, 428

intrauterine stem cell transplantation, 428 mesenchymal stem cells, 427, 428 platelet content, 426 stem cell transplantations, 427 utero bone marrow, 432 Warfarin, 418 Fluorescence in situ hybridization (FISH), 231–234 Foetal basis of adult disease (FeBAD), 371 Free fatty acids (FFAs), 187–188

G

Garbha Upanishada, 50 Gastrointestinal system, 53-54, 566 Gene regulatory networks, 200-202 Gestational diabetes mellitus (GDM), 191 Glucose-dependent insulinotropic polypeptide (GIP), 242 Glucose metabolism amino-acids and lactate, 286 anabolic processes, 286 chronic glucose deprivation, 288 endocrine pancreatic cells, 286 enzymatic systems, 286 fat oxidation, 286 foetal insulin secretion, 287-288 hormonal affinity, 286 hyperglycemia, 285 intrauterine growth restriction (IUGR) and amino acid metabolism, 288 energy substrates, 285 maternal glucose intolerance and insulin resistance, 285 Glucose transporter 1 (GLUT1), 247 Glutamate dehydrogenase (GDH), 121, 122, 243 Glutamine synthetase, 166, 167, 175, 179-180

H

Haematoxylineosin (HE) staining, 138 Haemopoiesis system characteristic features, 399 cord blood transplantation, 400 erythropoietin, 400 fetal blood composition, 401 fetal macrophage progenitors, 400 fetomaternal transfer, 401 Hb Gower, 400 human hemoglobin, 400 lymphopoiesis, 400 megakaryocytes, 400 optimal fetal oxygenation, 401-402 placenta, 401 primitive erythroblasts, 399 white blood cell count, 401 Hematopoietic stem cells (HSCs), 267, 307-308, 431 allo-transplantation of, 273 blood formation, 270-272 MDS, 272, 273 Hepatocyte growth factor (HGF), 322 Hepatoma-derived growth factor (HDGF), 323 hESC. See Human embryonic stem cell (hESC)

High density lipoprotein (HDL), 184, 187, 189-192 Hippocampus behavioral strategies, 209 fetal MRI, 209 hippocampal birth theory, 209 memory and emotion, 209 mental functions, 210 vomeronasal system, 210-211 Homoeopathy acute symptoms and chronic symptoms, 534 anemia in pregnancy, 537 clinical trial, 534-535 fetal development after fertilization, 535 first trimester, 535-536 first week event, 535 second trimester, 536 medicinal management first trimester, 535-536 second trimester, 537 treatment, 536-537 modern medicine, respect to, 537-538 Human embryonic stem cell (hESC), 262-263 chromosomal disease, 260-261 genetic disease autosimal recessive disease, 261 dominant disease, 262 X-linked disease, 261, 262 Human fetus adrenal cortex (see Adrenal cortex) anthropometric measurement (see Anthropometric measurement) brain development (see Brain) carbohydrate metabolism (see Carbohydrate metabolism) lipid metabolism (see Lipid metabolism) liver (see Liver) stem cells (see Stem cells) survival of alloimmunisation, 279 antibody, 278 anti-phospholipid antibody, 278 cardiolipin, 278 cytostatic and cytotoxic lymphokines, 280 genetic information, 279 immunity, 277 immunoglobins (IgG), 279 immunological testing and HLA tissue typing, 278 non-specific killer cells and inflammatory, 280 phospholipids, 278 recurrent spontaneous abortion, 278 recurrent spontaneous miscarriages, 277 reproductive immunophenotype, 280-281 Rh negative mother, 277 Rh positive cells, 277 Rh sensitization, 277 secondary abortions, 278 TJ6 protein, pregnancy, 280 uterine local and systemic immune responses, 277 white blood cell type, 278

Human immune system B cell development, 310 CD95 receptor, 306 cytokine profile, 308 eosinophil granulopoiesis, 311 FasL expression, 306 gastrointestinal and respiratory tracts, 306 hematopoietic stem cells (HSCs), 307-308 immunoglobulin production, 310 macrophages and dendritic cells, 308-309 MHC antigens, 306 mucosal immunity, 311 neonates, placental infections, 311 progesterone, 306 pro-inflammatory cytokines, 305 regulatory T cells, 305 T cell development, 309 T-cell immunity, 306 Th1 immunity, 306 TOLL receptors, 306 Treg development, 309 trophoblasts, 306 5-Hydroxytryptamine (5-HT), 218

I

Immunoprotection, 58 Induced pluripotent stem cells (iPSCs), 267, 394 Inner cell mass (ICM), 199, 203-204 Insulinemic index, 241 Insulin-like growth factor-1 (IGF-1), 240, 242 Insulin receptor substrate-1 (IRS-1), 246 Integumentary system, 566-567 Interleukin-1β (IL-1β), 56, 57 Interleukin one (IL-1), 475-476 Intermediate progenitor cell (IPC), 269 Intra amniotic antigens allogeneic amniotic fluid, 5 cc of, 29 auto immune profile, 36 bacterial polysaccharide injection, 2 cc of, 29 10 cc maternal whole blood, leucocytes of, 29 clinical events, 31, 32 hematopoietic stem cell transplantation, 41-46 immune response gene, 20 immune suppressor gene, 20 immuno-endocrino-hematological impact, 35, 36 maternal blood injection, 28 paternal and maternal HLA DR/DQ study, 36, 37 post-abortion maternal effect, 32-35 pre-immune/hypo-immune fetus, 39-41 tetanus toxoid injection 2 cc 20 % BSA antigen, 28 1 cc glutamate BCG, 26 1 cc single injection, 22-24 1.5 cc single injection, 22-24 2 cc single injection, 19, 20 3 cc single injection, 22-24 4 cc single injection, 22–24 consecutive second pregnancy termination attempt, 30, 31

direct 2 cc intra-fetal injection, 24, 25 double antigen challenge, 27 immunomodulation and fetotoxic substance, vitamin A, 24, 25 maternal non-immunization, 25, 26 multiple 2 cc injection, 20, 21 oral cimetidine, 26 oral levamisole, 25 single ¹/₂ cc age parity injection, 22, 23 single fixed dosage schedule, 36, 38, 39 thiomersol and aluminium phosphate, 30 Intrauterine fetal blood transfusion (IUT), 405, 416, 423, 424 Intrauterine growth restriction (IUGR), 56 adult hypertension, 496 angiotensin activity, 495 animal models, 495 autoimmunity, 498 blood pressure measurements, 495 brain and cognitive developments, 497-498 environmental factor, 497 glucorticoid dexamethasone, 497 low birth weight, 496 maternal smoking, 497 metabolic disorders and diabetes, 498-500 myocardial infarction and stroke, 496 rat studies, 496 vascular endothelium, 496 Intra ventricular hemorrhage (IVH), 555 IUGR. See Intrauterine growth restriction (IUGR)

K

Kidney cervical nephrotomes, 391–392 kidney diseases CKD, 393 and mesenchymal stem cells, 394 mesonephros, 391, 392 metanephros, 391–393 pronephros, 391 stem cells (*see* Stem cells)

L

Lactic acid, 93 Lanugo, 51, 216 Large for gestational age (LGA), 191 Late endosomes and lysosomes (LEL), 239, 240 Latent Early Life Associated Regulation (LEARn) Model, 372 LDL receptor-related protein 1 (LRP-1), 187 LDL receptor-related protein 2 (LRP-2), 187 Leucyl-tRNA synthetase (LeuRS), 240 Lipid metabolism CHO (*see* Cholesterol (CHO)) fetal disorders IUGR, 191–192 SLOS, 192 fetal fatty acids and triglycerides, 187–188

fetal lipoproteins and molecular mechanisms, 188 - 190lipids synthesis and transport, 184-186 maternal disorders diabetes, 191 familial hypercholesterolemia, 190-191 pre-eclampsia, 191 Lipoprotein (LP), 184, 186-192 Liver, 115 albumin expression, 321 AMPS concentration in, 105-107 anatomy, 319 and biliary system, 320 coculture studies, 321 developmental anomalies, 341 drug testing, 342 embryonic development biliary tree, 335, 340 cell types and function, 333 endodermal fate assignment, 336 gut tube pattern, 336-337 hematopoietic progenitor cells, 339-340 hepatic stellate cells and kupffer cells, 340 hepatocyte maturation, 339 hepatocytes, 332 hepatocytes and biliary epithelial cells, 338-339 implications, 342 liver bud development, 337 lobule, 332 morphogenesis, 334 in mouse, 335-336 septum transversum mesenchyme, 337-338 sinusoids and portal vein, 334–335 endoderm-lined yolk sac cavity, 320 extrahepatic bile ducts, 325-326 fibroblast growth factors (FGFs), 321 fructose 1,6 diphosphatase, 91-92 GATA4, 322 GATA-4 transcription factors, 321 glucosamine-6-phosphate synthetase, 113-114, 118 glucose-6-phosphatase activity at gestation period, 90, 91 non specific activity of, 90 subcellar fraction, 91 glycogen content in at gestation period, 88, 89 glycogen stability, 88 hematopoiesis, 325 hepatic gene expression, 322 hepatocyte maturation, 324 hepatocyte nuclear factor (HNF)-1, 321 inorganic phosphorus content in, 89 intra hepatic bile ducts, 326-327 in vivo DNA/protein analyses, 320 Kupffer cells, 325 lineage segregation, 323-324 liver bud formation, 321 mesodermal tissue, 320 mesothelial cells, 321

parenchymal cells, 321 phosphorylase activity in, 89-90 physiology, 319-320 septum transversum, 321 stellate cells, 325 stem cells, 341 TGFb signaling, 322 transcriptional regulation, 322-323 urea biosynthesis (see Urea biosynthesis) vasculature development, 324-325 water and electrolyte content of, 107-108, 115-116 Wnt signaling, 321, 322 Low density lipoprotein (LDL), 186, 187, 189-192 Lung, 116 glucosamine-6-phosphate synthetase, 113-114, 118 total AMPS, fractions of, 109-111 water and electrolyte contents of, 111, 116-117

М

Major histocompatibility (MHC) antigens, 57, 58 Maternal hyperglycemia, 93 Maternal milk consumption, mTORC1. See Mechanistic (mammalian) target of rapamycin complex 1 (mTORC1) Maternal nutrition, 55-56 Mechanistic (mammalian) target of rapamycin complex 1 (mTORC1) amino acid-mediated activation, 239-240 glucose activation, 241 glutaminolysis pathway activation, 240-241 insulin/IGF-1 signaling activation, 238-239 nutrient-and growth factor-sensitive kinase, 238 palmitate activation, 241 Merkel cells, 217-218 Mesenchymal stem cells (MSCs), 394, 428 Metabolic disorders, 498-500 Metal exposure, 370 Metaphase comparative genomic hybridisation (mCGH), 231, 232 Microchimerism (Mc), 56 Milk consumption, mTORC1 amino acids activation, 242-243 cow's milk consumption, 240, 241 glucose activation, 243 GLUT1, FGF21-mediated over-expression of, 247 hyperinsulinemia, 241-242 maternal weight, 249 microRNA-21 (miR-21), 247-248 palmitate activation, 243 PGH-induced insulin resistance, 245–246 placental, fetal and birth weight, 243-245 in placental nutrient transfer, 248-249 placental weight gain and growth hormone signaling, 245 serum IGF-1 levels, 240, 242 MPS. See Mucopolysaccharide (MPS) mTORC1. See Mechanistic (mammalian) target of rapamycin complex 1 (mTORC1)

Mucopolysaccharide (MPS) AMPS (*see* Acid mucopolysaccharides (AMPS)) function of, 102–104 structure of, 99, 100 classification of, 102 proteoglycal aggregate, 99, 101 proteoglycan monomer structure, tentative model for, 99, 101 water and electrolytes (*see* Water and electrolytes) Myelodysplastic syndrome (MDS), 272–273 Myocardial infarction, 31, 496

Ν

Natural killer (NK) cells, 15–16, 431 Naturally aborted fetus, 571–572 N-cadherin, 53 Necrotizing entero colitis (NEC), 556 Neural stem cells (NSCs), 269–272 cell replacement, 373 definition, 372 neurogenic areas, 373 Neural tube defects (NTDs), 463 Neurodegeneration animal studies, 369 environmental factors, 369 human studies, 369 Niemann-Pick C1-Like 1 (NPC1L1), 186 Nucleated red blood cells (NRBC), 59

0

Ornithine transcarbamylase, 125, 131–133 Ossification, 51 Ouchterlony system, 147, 150 Ovulation, 223

Р

Patent ductus arteriosus (PDA), 555 Pesticides, 370 Phosphate buffered saline (PBS) adult adrenal glands, 148-151 fetal adrenal glands, 148, 151 P32 incorporation, 165-166, 175-176, 180 Placenta blastocyst, 441 DNA methylation, 450 embryo development, 442 fetomaternal barrier intraplacental fetal circulation, 443 syncytiotrophoblast layer, 442 fetomaternal barrier, first and second trimesters basal plate, 443 chorionic plate, 443 cytotrophoblast, 443 efflux transporters, 447-448 fetoplacental circulation, 445 immunological barrier, 448-449 maternal vascularization and placental circulation, 444-445

physiological properties, 445 secretory immune system, 449 tertiary villi, 443 vasculogenesis and angiogenesis, 445-446 villous development, 444 fetomaternal exchange, 446-447 hormone secretion, 442 micro niche, 441 xenobiotics, 451 Placental growth hormone (PGH), 245-246 Platelet-derived growth factor (PDGF), 475 Predictive adaptive response (PAR), 372 Pre-eclampsia (PE), 191 Preimplantation genetic diagnosis (PGD), 259 Prematurity causes and symptoms, 553-554 definition, 553 extreme prematurity anatomical and micro-structural immaturity, 562 complications, 562 definition, 562 gastrointestinal system, 566 incidence, 562 integumentary system, 566-567 nervous system, 564-566 respiratory distress syndrome and bronchopulmonary dysplasia, 563 respiratory system, 563 long-term complications, 558-559 short-term complications blood problems, 556-557 brain problems, 555 breathing problems, 554 gastrointestinal problems, 556 heart problems, 554-555 immune system, 557 metabolic problems, 557 temperature control problems, 555-556 Preterm birth and low birth weight babies adult outcomes, 381-382 biological and environmental risk factors, 379 cognition and academic achievement, 381 depression and infants, 382-383 extremely low birth weight (ELBW), 379 motor outcomes, 380-381 neurobiology, 383-384 neurodevelopmental outcomes, 380 putative developmental mechanisms, 384 very low birth weight (VLBW), 379 Pyrexia, 31, 32 Pyruvic acid, 86

Q

Quantitative immunoprecipitation test, 152-153

R

RAS-homolog enriched in brain (RHEB), 238, 239 Red cell alloimmunisation, 408 Respiratory distress syndrome (RDS), 554, 563 Respiratory system canalicular period, 51, 53 pseudoglandular period, 51 Retinopathy of prematurity (ROP), 564

S

Scarless wound healing biological events, 487 biological mechanisms, 488 biological phenomenon, 470 biomechanical role, 469 clinical observation, 488 collagen collagen molecule, 476-477 crosslinking, 477 collagen deposition, 485 epidermis, 469 extracellular components, 470 fibroblasts, 488 in foetal animals, 470 foetal rabbits studies, 480-481 foetal sheep studies, 481-484 foetal wound healing, 479-480 granulation tissue formation, 474 growth factors epidermal growth factor (EGF), 474-475 interleukin one (IL-1), 475-476 platelet-derived growth factor (PDGF), 475 transforming growth factor alpha (TGF-a), 475 transforming growth factor beta (TGF- β), 475 tumour necrosis factor (TNF), 475-476 high-definition ultrasound, 488 human oral mucosa, 488 hyaluronic acid, 478-479, 484-485 inflammatory phase eosinophils, 474 lymphocytes, 474 macrophages, 473 platelets, 473 in vitro and in vivo models, 486 keratinocytes, 468, 488 light and electron microscopy and immunohistochemistry, 487 monolayer cell proliferation studies, 486 NSAID, effects of, 487 pathological scare types, 469 peptide regulatory factors, 487 polymorphonuclear leucocyte cells, 487 postnatal wound healing, 468 post-translational modifications and matrix organization, 488 skin fibroblast, 470 GAG chains, 472 non-collagenous extracellular matrix, 472 papillary dermis, 471 type I collagen, 470-471 type III collagen, 471 wound repair, phase of, 472

stem cell and stem cell based therapies, 470 stem cells, 488 stratum corneum, 468 tissue integrity, 485 Scavenger receptor class B (SR-B1), 187 Second trimester abnormalities, 432 amniotic fluid, 431 anti retroviral drugs, 420-421 Ayurveda (see Ayurveda) azole anti-fungal drugs, 418 endocrine maturation (see Endocrine maturation) fetal abnormalities, diagnosis brain herniation mass, 416 Down's syndrome, 415 fetal age, 418 fetal infection, 424-426 heart burns, 528 hematological disorders, 422-424 lithium toxicity, 418 natural killer (NK) cells, 431 organogenesis, 418 physiological and anatomical disorders, 421-422 sickle cell anemia, 431 stem cell therapy, 431 surgical interventions congenital anomaly, 430 fetal tracheal occlusion intervention (FeTO), 428 hemangioblastomas, 429 intrapericardial teratomas and rhabdomyomas, 429 Kasabach-Merritt sequence, 429 open fetal surgery, 428 sacrococcygeal tumors (SCT), 429 spina bifida/myelomeningocel, 429 twin to twin transfusion syndrome, 430 valvar aortic stenosis, 430 teratogenic drug, 430 teratogenic effect, 419-420 treatment options, 426 umbilical cord blood, intrauterine transfusions allogeneic cord blood transfusion, 427 clinical trials, 426 coagulation post intra uterine fetal transfusion, 427 gene therapies, 428 HIV transmission, 427 HLA matching, 427 HSC. 428 intrauterine stem cell transplantation, 428 MSC, 427, 428 platelet content, 426 stem cell transplantations, 427 utero bone marrow, 432 Self-fertilization, 224-225 Severe combined immunodeficiency (SCID), 431 Single nucleotide polymorphisms (SNP), 260 Small for gestational age (SGA), 191 Smith-Lemli-Opitz syndrome (SLOS), 192 Spermatogenesis, 223 Spermatozoa, 233

Stem cells **ADPKD**, 396 ARPKD, 396 auto-immune diseases, 273 breast model, 269, 270 CAKUT, 395 early embryo development and origin of, 268-269 ECSs, 267 emerging technologies, 395 FAS, 272 HSCs, 267, 270-272, 307-308, 431 iPSCs, 267 MDS, 272-273 and mesenchymal stem cells, 394 MSCs, 394, 428 neural stem cells, 269-272 Streptococcus agalactiae, 308 Stroke, 496 Suppressor of cytokine signaling proteins (SOCS), 246

Т

TCM. See Traditional Chinese medicine (TCM) Tissue macrophages, 473 Traditional Chinese medicine (TCM) acupuncture, 530 and allopathic medicines, 529 cupping, 508 dehydration and edema, 528 Die Da, 508 diet, role of, 527-528 and fetal development, 526-527 Gua Sha, 508 health-related quality of life, 508-510 heart burns, 528 in vivo and clinical studies, 529 Jinye Baidu Granule (JYBDG), 508 maternal and fetal safety concerns, 510 morning sickness, 527 moxa/acupuncture, 508 organogenesis, 528 Peony Formula, 508 pharmacodynamic and pharmacokinetic principles, 525-526 Qi Gong, 508 Tai Chi, 508 Tang-kuei, 508 teratogenesis effects, 528 therapeutic effects, 530 Tui Na. 508 Traditional complementary and alternative medicine (TCAM), 529–530 Transforming growth factor alpha (TGF- α), 475 Transforming growth factor beta (TGF-β), 475 Tribal medicine Acantheceae, 514 Ayurved and Unani, 515 Cholanaickan tribes, in Kerala, 514 Hopi tribes, 514 Kannikar tribes, 515

Maori tribal, 515 methodologies, 513–514 North American tribes, 514 Triglycerides (TGs), 187–188 Tumor necrosis factor-α (TNF-α), 56, 57, 308, 311, 340 Tumour necrosis factor (TNF), 323, 475–476 Twin to twin transfusion syndrome (TTTS), 407, 430

U

U 14C-glucose incorporation, 176-177 Unani medicine bay leaves, 518 cloves, 518 coriander, 518 diet therapy, 518 drug composition, 519 fetal development, 520-522 modern therapeutic approaches, 520 pharmacovigilance and safety, 522-523 principles, 519-520 Tibbi system, 517 Urea biosynthesis, 129-133 bladder fluid and plasma, ammonia nitrogen and urea nitrogen concentrations of, 124-125 citrulline, 122 concentration of ornithine, 127, 128 polyamines, 128-129 enzyme activity arginase, 126, 127 argininosuccinase, 126 argininosuccinic acid synthetase, 125, 126 ornithine amino transferse, 129 ornithine transcarbamylase, 125 glutamate dehydrogenase, 121, 122 in mammals, 123-124 urea production, rate of, 126, 127 Urethral stricture, 468 Urinary system, 54 U.S. vaccination program, 11

v

Vaccination antigenic stimulation, role of, 12 graft vs. host reaction and autoimmunity, 12-13 hypoimmune fetal system, biomedical research, 13 maternal immunization advantages, 11 disadvantages, 11-12 neonatal immunization advantages, 12 disadvantages, 12 theoretical prospect/potentialities, 11 Vaccination Assistance Act. 11 Vascular anomalies aortic arches, 351-353 aortic coarctation, 352-353 arteriovenous malformations, 356-357

duplicated femoral vein, 356 inferior vena cava, 355-356 lymphatic system, 357 median artery and high radial artery, 354 patent ductus arteriosus, 352 popliteal entrapment, 354-355 renal artery variations, 354 renal vein anomalies, 356 sciatic artery, 354 subclavian artery, 353-354 superior vena cava, 355 vascular identity, 351 vasculogenesis, 349-350 Vernix caseosa, 51, 216 Very low birth weight (VLBW) behavioral problems, 566 psychological outcome, 384

Very low density lipoprotein (VLDL), 184–187, 189 Vomeronasal organ (VNO), 211 Vomeronasal system (VNS), 210–211

W

Water and electrolytes brain, 113, 117–118 and extracellular water, 104 liver sodium and potassium contents of, 107–109 thiocyate concentrations, 107, 108 lung, 111

Y

Yolk sac erythroblasts, 399